State of Misconsin



2017 Assembly Bill 64

Date of enactment: September 21, 2017 Date of publication*: September 22, 2017

2017 WISCONSIN ACT 59

(Vetoed in Part)

AN ACT relating to: state finances and appropriations, constituting the executive budget act of the 2017 legislature.

The people of the state of Wisconsin, represented in senate and assembly, do enact as follows:

SECTION 1c. 7.52 (8) of the statutes is amended to read:

7.52(8) The board of absentee ballot canvassers shall make full and accurate return of the votes cast for each candidate and proposition on the tally sheet forms. Each tally sheet shall record the returns for each office or referendum by ward, unless combined returns are authorized in accordance with s. 5.15 (6) (b), in which case the tally sheet shall record the returns for each group of combined wards. After recording the votes, the board of absentee ballot canvassers shall seal in a carrier envelope outside the ballot bag or container one inspector's statement under sub. (4) (d), one tally sheet, and one poll list for delivery to the county clerk, unless the election relates only to municipal or school district offices or referenda or municipal offices or referenda. The board of absentee ballot canvassers shall also similarly seal one statement, one tally sheet, and one poll list for delivery to the municipal clerk.

SECTION 1d. 8.06 of the statutes is amended to read: **8.06 Special elections may be called.** Towns, cities, villages, and, subject to ss. 67.05 (6a) (a) 2. and 121.91 (3) (a), school districts, may call special elections for any purpose authorized by law. If an election is called for a special referendum, the election shall be noticed under s. 8.55.

SECTION 1e. 13.093 (2) (a) of the statutes is amended **Vetoed** to read:

In Part

13.093 (2) (a) Any bill making an appropriation, any bill increasing or decreasing existing appropriations or state or general local government fiscal liability or revenues, and any bill that modifies an existing surcharge or creates a new surcharge that is imposed under ch. 814, shall, before any vote is taken thereon by either house of the legislature if the bill is not referred to a standing committee, or before any public hearing is held before any standing committee or, if no public hearing is held, before any vote is taken by the committee, incorporate a reliable estimate of the anticipated change in appropriation authority or state or general local government fiscal liability or revenues under the bill, including to the extent possible a projection of such changes in future biennia. The estimate shall also indicate whether any increased costs incurred by the state under the bill can be mitigated through the use of contractual service contracts let in accordance with competitive procedures. For purposes of this paragraph, a bill increasing or decreasing the liability or revenues of the unemployment reserve fund is considered to increase or decrease state fiscal liability or revenues. Except as otherwise provided by joint rules of the legislature or this paragraph, such estimates shall be made by the department or agency administering the appropriation or fund or collecting the revenue. The legislative council staff shall prepare the fiscal estimate with

^{*} Section 991.11, WISCONSIN STATUTES: Effective date of acts. "Every act and every portion of an act enacted by the legislature over the governor's partial veto which does not expressly prescribe the time when it takes effect shall take effect on the day after its date of publication.'

respect to the provisions of any bill referred to the joint survey committee on retirement systems which create or modify any system for, or make any provision for, the retirement of or payment of pensions to public officers or employees. The director of state courts shall prepare the fiscal estimate with respect to the provisions of any bill that modifies an existing surcharge or creates a new surcharge that is imposed under ch. 814. The executive director of the state prosecutors office shall prepare the fiscal estimate with respect to the provisions of any bill that affects prosecutors or the state prosecutors office, including bills modifying or creating crimes or sentencing practices. When a fiscal estimate is prepared after the bill has been introduced, it shall be printed and distributed as are amendments.

Vetoed In Part **SECTION 1L.** 13.0967 of the statutes is created to read:

13.0967 Review of bills affecting state prosecutors office. Any bill that is introduced in either house of the legislature that directly affects the state prosecutors office shall have a notation to that effect on its jacket when the jacket is prepared. When a bill that has that notation on the jacket is introduced, the legislative reference bureau shall submit a copy of the bill to the state prosecutors office.

SECTION 2. 13.101 (6) (a) of the statutes is amended to read:

13.101 (6) (a) As an emergency measure necessitated by decreased state revenues and to prevent the necessity for a state tax on general property, the committee may reduce any appropriation made to any board, commission, department, or the University of Wisconsin System, or to any other state agency or activity, by such amount as it deems feasible, not exceeding 25 percent of the appropriations, except appropriations made by ss. 20.255 (2) (ac), (bc), (bh), (cg), and (cr), 20.395 (1), (2) (cq), (eq) to (ex) and (gq) to (gx), (3), (4) (aq) to (ax), and (6) (af), (aq), (ar), and (au), 20.435 (4) (a) and (5) (da), and 20.437 (2) (a) and (dz) or for forestry purposes under s. 20.370 (1) (2), or any other moneys distributed to any county, city, village, town, or school district. Appropriations of receipts and of a sum sufficient shall for the purposes of this section be regarded as equivalent to the amounts expended under such appropriations in the prior fiscal year which ended June 30. All functions of said state agencies shall be continued in an efficient manner, but because of the uncertainties of the existing situation no public funds should be expended or obligations incurred unless there shall be adequate revenues to meet the expenditures therefor. For such reason the committee may make reductions of such appropriations as in its judgment will secure sound financial operations of the administration for said state agencies and at the same time interfere least with their services and activities.

SECTION 8bd. 13.48 (33e) of the statutes is created to read:

13.48 (33e) LA CROSSE CENTER. (a) The legislature finds and determines that the meetings and conventions and the sports and entertainment industries are of vital importance in creating jobs and contributing to economic development and tourism in this state and are statewide responsibilities of statewide dimension. It is therefore in the public interest, and it is the public policy of this state, to assist the city of La Crosse in the remodeling and expansion of the La Crosse Center.

- (b) The building commission may authorize up to \$5,000,000 in general fund supported borrowing to assist the city of La Crosse in the remodeling and expansion of the La Crosse Center. The state funding commitment shall be in the form of a grant to the city of La Crosse. Before approving any state funding commitment for the remodeling and expansion of the La Crosse Center, the building commission shall determine that the city of La Crosse has secured additional funding for the project of at least \$42,000,000 from nonstate revenue sources.
- (c) If the building commission authorizes a grant to the city of La Crosse under par. (b), and if, for any reason, the space that is remodeled and expanded with funds from the grant is not used for meetings and conventions or sports and entertainment, the state shall retain an ownership interest in the remodeled and expanded space equal to the amount of the state's grant.

SECTION 8be. 13.48 (33m) of the statutes is created to read:

13.48 (33m) St. Ann Center for Intergenera-TIONAL CARE, INC.; BUCYRUS CAMPUS. (a) The legislature finds and determines that the improvement of the health and well-being of residents of all ages of this state and the promotion of community and economic development are statewide responsibilities of statewide dimension. The legislature further finds and determines that St. Ann Center for Intergenerational Care, Inc., provides vital intergenerational care at its Bucyrus Campus in the city of Milwaukee that includes services for individuals suffering from Alzheimer's disease and dementia, services for individuals with mental and physical disabilities, services for minority and economically disadvantaged children and their families, and services that encourage entrepreneurship and business development. It is therefore in the public interest, and it is the public policy of this state, to assist St. Ann Center for Intergenerational Care, Inc., in the completion of its Bucyrus Campus.

(b) The building commission may authorize up to \$5,000,000 in general fund supported borrowing to assist St. Ann Center for Intergenerational Care, Inc., in the completion of its Bucyrus Campus. The state funding commitment shall be in the form of a grant to St. Ann Center for Intergenerational Care, Inc. Before approving any state funding commitment for completion of the Bucyrus Campus, the building commission shall determine that St. Ann Center for Intergenerational Care, Inc.,

has secured additional funding for the project of at least \$20,268,200 from nonstate revenue sources.

(c) If the building commission authorizes a grant to St. Ann Center for Intergenerational Care, Inc., under par. (b), and if, for any reason, the Bucyrus Campus is not used for the provision of intergenerational care, the state shall retain an ownership interest in the Bucyrus Campus equal to the amount of the state's grant.

SECTION 8bm. 13.48 (33s) of the statutes is created to read:

13.48 (33s) Brown County innovation center. (a) The legislature finds and determines that meeting manufacturing workforce needs and supporting innovation and entrepreneurship in the manufacturing industry in this state are of vital importance in expanding the manufacturing industry in this state, creating jobs, and improving the municipal, regional, and state economies and are statewide responsibilities of statewide dimension. It is therefore in the public interest, and it is the public policy of this state, to assist Brown County in the construction of a science, technology, engineering, and mathematics innovation center located on or adjacent to the University of Wisconsin - Green Bay campus.

- (b) The building commission may authorize up to \$5,000,000 in general fund supported borrowing to assist Brown County in the construction of a science, technology, engineering, and mathematics innovation center located on or adjacent to the University of Wisconsin -Green Bay campus. The state funding commitment shall be in the form of a grant to Brown County. Before approving any state funding commitment for the construction of the innovation center, the building commission shall determine that Brown County has secured additional funding for the project of at least \$10,000,000 from nonstate revenue sources.
- (c) If the building commission authorizes a grant to Brown County under par. (b), and if, for any reason, the center that is constructed with funds from the grant is not used as a science, technology, engineering, and mathematics innovation center, the state shall retain an ownership interest in the center equal to the amount of the state's grant.

Vetoed In Part

SECTION 8bt. 13.489 (1g) of the statutes is renumbered 13.489 (1g) (a) and amended to read:

13.489 (1g) (a) There is created a transportation projects commission consisting.

(b) The commission consists of the governor, 3 2 citizen members appointed by the governor to serve at his or her pleasure, and 5 3 senators and 5 3 representatives to the assembly appointed as are the members of standing committees in their respective houses, and 4 citizen members, one appointed by each the senate majority leader, the senate minority leader, the speaker of the assembly, and the assembly minority leader. Of the members from each house, 3 shall be chosen from the majority party and 2 shall be chosen from the minority party the

senate and the assembly, 2 shall be appointed by each the Vetoed speaker of the assembly and the senate majority leader and one shall be appointed by each the assembly minority leader and senate minority leader. The governor shall appoint the secretary of transportation shall serve or the secretary of administration as a nonvoting member.

- (c) The governor shall serve as chairperson of the commission.
- (d) Citizen members of the commission shall be reimbursed for their actual and necessary expenses incurred as members of the commission from the appropriation under s. 20.395 (4) (aq).

SECTION 8c. 13.489 (1i) of the statutes is created to

- 13.489 (1i) STAFF. (a) The commission shall appoint a director and submit the appointment to the senate for confirmation. The director may serve prior to senate confirmation. The commission shall make the initial appointment of a director under this paragraph no later than January 12, 2018.
- (b) The director shall appoint staff necessary for performing the duties of the commission. Staff appointed under this paragraph shall include an engineer, legal counsel, and a financial auditor. Staff appointed under this paragraph report to and serve at the pleasure of the director.

SECTION 8d. 13.489 (2) of the statutes is renumbered 13.489 (2) (a).

SECTION 8e. 13.489 (2) (b) of the statutes is created to read:

- 13.489 (2) (b) 1. Annually, the department of transportation shall provide the commission with a list of potential major highway projects and southeast Wisconsin freeway megaprojects that are not yet being considered for an environmental impact statement or an environmental assessment or enumeration under s. 84.013 (3) or approval under s. 84.013 (6) and the estimated cost and scope of each project.
- 2. In each even-numbered year, the department of transportation shall provide the commission with a list of proposed or planned state highway rehabilitation projects and southeast Wisconsin freeway megaprojects, the estimated cost and scope of each project, and the location of each project.

SECTION 8f. 13.489 (3) of the statutes is renumbered 13.489 (3) (a).

SECTION 8g. 13.489 (3) (b) of the statutes is created to read:

13.489 (3) (b) When the department of transportation submits its biennial budget request under s. 16.42, the department shall provide a copy of the request to the commission.

SECTION 8h. 13.489 (7) of the statutes is created to read:

13.489 (7) REVIEW OF DEPARTMENT ACCOUNTS AND RECORDS. (a) The commission shall periodically review In Part

the records and accounts of the department of transportation.

- (b) Annually, the commission shall evaluate the department of transportation based on goals and performance measures established by the commission. Not later than December 31 of each year, the commission shall submit the evaluation to the governor, the joint committee on finance, the standing committees of the legislature with jurisdiction over transportation matters, and the department of transportation.
- (c) The director of the commission may periodically enter into a contract for an independent audit of the department of transportation.

SECTION 8i. 13.489 (8) of the statutes is created to read:

- 13.489 (8) MEETINGS. (a) The commission shall meet at least twice each year.
 - (b) The commission may hold public meetings.

SECTION 8j. 13.489 (9) of the statutes is created to read:

13.489(9) DEENUMERATION. In each even-numbered year the commission shall consider recommending the removal of projects that are at least 10 years old from the schedule of enumerated projects.

SECTION 8k. 13.489 (10) of the statutes is created to read:

13.489 (10) COMMISSION REPORTS. The commission shall prepare all of the following reports:

- (a) A report describing the short-term and long-term impacts of each department of transportation biennial budget request on state and local roads. The commission shall submit the report under this paragraph to the governor and the standing committees of the legislature with jurisdiction over transportation matters no later than 30 days after the department of transportation submits its biennial budget request under s. 16.42.
- (b) A report describing the short-term and long-term impacts of the executive budget bill on state and local roads. The commission shall submit the report under this paragraph to the governor and the standing committees of the legislature with jurisdiction over transportation matters no later than 30 days after the executive budget bill is introduced under s. 16.47.

SECTION 8L. 13.489 (11) of the statutes is created to read:

13.489 (11) LONG-RANGE PLANNING. If the commission issues long-range planning recommendations, the department of transportation, to the extent permitted by state and federal law, shall adopt the recommendations.

SECTION 8m. 13.489 (12) of the statutes is created to read:

13.489 (12) BUDGET REQUEST. The commission shall submit a biennial budget request under s. 16.42 for commission operations.

SECTION 8n. 13.489 (13) of the statutes is created to read:

13.489 (13) SUNSET. Subsections (1g) (b), (1m), (3), Vetoed (4), (4m), (7), (8), (9), (10), and (11) do not apply after **In Part** June 30, 2021.

SECTION 8p. 13.90 (10) of the statutes is created to **Vetoed** read:

13.90 (10) The cochairpersons of the joint committee on legislative organization shall authorize all expenditures from the appropriation under s. 20.765 (4) (b).

SECTION 8s. 13.94 (1) (u) of the statutes is created to **Vetoed** read:

13.94(1) (u) Audit the financial records of a commission created under s. 66.0304 and any entity created under s. 66.0304 (4e) at the direction of the joint legislative audit committee.

SECTION 9. 15.01 (2) of the statutes is amended to read:

15.01 (2) "Commission" means a 3-member governing body in charge of a department or independent agency or of a division or other subunit within a department, except for the employment relations commission which shall consist of one chairperson, the Wisconsin waterways commission which shall consist of 5 members, the elections commission which shall consist of at least 6 members, the ethics commission which shall consist of at least 6 members, and the parole commission which shall consist of <u>8</u> <u>4</u> members. A Wisconsin group created for participation in a continuing interstate body, or the interstate body itself, shall be known as a "commission", but is not a commission for purposes of s. 15.06. The parole commission created under s. 15.145 (1) shall be known as a "commission", but is not a commission for purposes of s. 15.06.

SECTION 12. 15.06 (1) (bm) of the statutes is created to read:

15.06 (1) (bm) The employment relations commission shall consist of a chairperson, nominated by the governor, and with the advice and consent of the senate appointed, for a 6-year term, except that the term of the first chairperson appointed after the effective date of this paragraph [LRB inserts date], expires on March 1, 2023.

SECTION 14. 15.06 (3) (a) 4. of the statutes is repealed.

SECTION 15. 15.06 (3) (c) of the statutes is repealed. **SECTION 17.** 15.06 (10) of the statutes is amended to read:

15.06 (10) COMPENSATION. Members A member of the elections commission and members a member of the ethics commission shall receive a per diem of \$227 for each day they were actually and necessarily engaged in In Part performing their duties a per diem equal to the amount prescribed under s. 753.075 (3) (a) for reserve judges sitting in circuit court on which the member attends or participates by audio or video conference call in a meeting of the member's commission.

In Part

In Part

Vetoed

SECTION 17m. 15.07 (1) (a) 6. of the statutes is created to read:

15.07 (1) (a) 6. Members of the public leadership board appointed under s. 15.915 (7) (d) and (e) shall be appointed by the governor without senate confirmation.

SECTION 17n. 15.07 (1) (b) 24. of the statutes is created to read:

15.07 (1) (b) 24. The 6 members of the group insurance board appointed under s. 15.165 (2) (j).

SECTION 22. 15.07 (5) (i) of the statutes is repealed. **SECTION 31.** 15.105 (3) of the statutes is repealed.

Vetoed In Part

Vetoed

In Part

SECTION 31n. 15.105 (7) of the statutes is created to read:

15.105 (7) STATE PROSECUTORS OFFICE. There is created a prosecutors office that is attached to the department of administration under s. 15.03. The executive director shall be appointed by the prosecutor board.

SECTION 32. 15.105 (10) of the statutes is amended to read:

15.105 (10) BOARD ON AGING AND LONG-TERM CARE. There is created a board on aging and long-term care, attached to the department of administration under s. 15.03. The board shall consist of 7 members who are members of the public and who are appointed for staggered 5-year terms. Members shall have demonstrated a continuing interest in the problems of providing delivering and financing long-term care for the aged or disabled. At least 4 members shall be public members with no interest in or affiliation with any nursing home persons who are 60 years of age or older or who are beneficiaries of the Medicare program. No person who currently owns or who, within the previous 5 years, owned or who had any operational or substantial financial or employment interest in or any other affiliation with any long-term care provider or health care insurance company may be appointed to or retained as a member of the board. No person who is or has been an employee or volunteer of the board may be appointed to or retained as a member.

SECTION 34. 15.105 (25m) of the statutes is renumbered 15.185 (5), and 15.185 (5) (intro.) and (a), as renumbered, are amended to read:

15.185 (5) COLLEGE SAVINGS PROGRAM BOARD. (intro.) There is created a college savings program board that is attached to the department of administration financial institutions under s. 15.03 and that consists of all of the following members:

(a) The secretary of administration financial institutions or his or her designee.

SECTION 34m. 15.107 (18) of the statutes is renumbered 15.315 (1), and 15.315 (1) (a), as renumbered, is amended to read:

15.315 (1) (a) There is created an interoperability council, attached to the department of administration military affairs under s. 15.03.

SECTION 35. 15.137 (2) of the statutes is renumbered 15.227 (15), and 15.227 (15) (a) (intro.), as renumbered, is amended to read:

15.227 (15) (a) (intro.) There is created in the department of agriculture, trade and consumer protection workforce development an agricultural education and workforce development council consisting of the following members:

SECTION 38j. 15.145 (1) of the statutes is amended to read:

15.145 (1) PAROLE COMMISSION. There is created in the department of corrections a parole commission consisting of -8-4 members. Members shall have knowledge of or experience in corrections or criminal justice. The members shall include a chairperson who is nominated by the governor, and with the advice and consent of the senate appointed, for a 2-year term expiring March 1 of the odd-numbered years, subject to removal under s. 17.07 (3m), and the remaining members in the classified service appointed by the chairperson.

SECTION 39d. 15.165 (2) of the statutes is renum- **Vetoed** bered 15.165 (2) (intro.) and amended to read:

In Part

15.165 (2) GROUP INSURANCE BOARD. (intro.) There is created in the department of employee trust funds a group insurance board. The board shall consist of the following members:

(a) The governor, the or his or her designee.

(b) The attorney general, the or his or her designee.

(c) The secretary of administration, the director of the office of state employment relations, and the or his or her designee.

(e) The commissioner of insurance or their designees, and 6 persons his or her designee.

(i) Six individuals appointed for 2-year terms, of whom one shall be an insured participant in the Wisconsin Retirement System who is not a teacher, one shall be an insured participant in the Wisconsin Retirement System who is a teacher, one shall be an insured participant in the Wisconsin Retirement System who is a retired employee, one shall be an insured employee of a local unit of government, and one shall be the chief executive or a member of the governing body of a local unit of government that is a participating employer in the Wisconsin Retirement System.

SECTION 39f. 15.165 (2) (d) of the statutes is created

15.165 (2) (d) The administrator of the division of personnel management in the department of administration or his or her designee.

SECTION 39g. 15.165 (2) (f) of the statutes is created to read:

15.165 (2) (f) One member appointed by the speaker of the assembly.

SECTION 39h. 15.165 (2) (g) of the statutes is created to read:

15.165 (2) (g) One member appointed by the minority leader of the assembly.

SECTION 39j. 15.165 (2) (h) of the statutes is created to read:

15.165 (2) (h) One member appointed by the majority leader of the senate.

SECTION 39k. 15.165 (2) (i) of the statutes is created to read:

15.165 (2) (i) One member appointed by the minority leader of the senate.

SECTION 39m. 15.315 of the statutes is created to read:

15.315 Same; councils. (2) 911 SUBCOMMITTEE. (a) There is created a 911 subcommittee of the interoperability council, attached to the department of military affairs under s. 15.03. The 911 subcommittee consists of one member serving a 3–year term who is appointed by the adjutant general and the following members serving 3–year terms who are appointed by the governor:

- 1. An individual recommended by an association of Wisconsin cities, villages, or towns.
- 2. An individual recommended by an association of Wisconsin counties.
- 3. An individual recommended by a Wisconsin association, or a Wisconsin chapter of an association, that promotes a universal emergency telephone number system.
- 4. An individual recommended by an association of Wisconsin county sheriffs.
- 5. Two individuals, each of whom represents a different commercial mobile radio service provider, as defined in s. 196.01 (2g), operating in Wisconsin: one serving a primarily regional market and one serving a national market.
- 6. Two individuals recommended by a Wisconsin association, or a Wisconsin chapter of an association, of public safety communications professionals.
- 7. Two individuals recommended by an association of Wisconsin telecommunications providers, as defined in s. 196.01 (8p), each of whom represents an incumbent local exchange carrier.
- 8. An individual who represents a competitive local exchange carrier.
- 9. An individual who represents a voice over Internet protocol provider.
- 10. A police chief recommended by an association of Wisconsin police chiefs.
- 11. A fire chief recommended by an association of Wisconsin fire chiefs.
- 12. An individual recommended by a Wisconsin association that promotes emergency management.
- 13. An individual who represents a video service provider, as defined in s. 196.01 (12r).

- 14. An individual recommended by a Wisconsin association of emergency medical service providers.
- 15. An individual recommended by an association of land information professionals.
- (b) In making appointments under par. (a), the governor shall consider the geographical diversity of, and the representation of urban and rural interests by, the membership of the 911 subcommittee.

SECTION 48m. 15.405 (18) of the statutes, as affected by 2017 Wisconsin Act (this act), is repealed.

SECTION 52m. 15.406 (6) (a) 1. of the statutes is amended to read:

15.406 (6) (a) 1. Six massage therapists or bodywork therapists licensed under ch. 460 who have engaged in the practice of massage therapy or bodywork therapy for at least 2 years preceding appointment. One member appointed under this subdivision shall be a representative of a massage therapy or bodywork therapy school approved by the educational approval board under s. 38.50 440.52. One member appointed under this subdivision shall be a representative of a massage therapy or bodywork therapy program offered by a technical college in this state. No other members appointed under this subdivision shall be directly or indirectly affiliated with a massage therapy or bodywork therapy school or program.

SECTION 520. 15.406 (6) (a) 1. of the statutes, as affected by 2017 Wisconsin Act (this act), is amended to read:

15.406 (6) (a) 1. Six massage therapists or bodywork therapists licensed under ch. 460 who have engaged in the practice of massage therapy or bodywork therapy for at least 2 years preceding appointment. One member appointed under this subdivision shall be a representative of a massage therapy or bodywork therapy school approved by the educational approval board department of safety and professional services under s. 440.52. One member appointed under this subdivision shall be a representative of a massage therapy or bodywork therapy program offered by a technical college in this state. No other members appointed under this subdivision shall be directly or indirectly affiliated with a massage therapy or bodywork therapy school or program.

SECTION 68. 15.675 (1) (d) of the statutes is repealed and recreated to read:

15.675 (1) (d) The secretary of the department of safety and professional services or his or her designee.

SECTION 68g. 15.77 of the statutes is created to read: **15.77 Prosecutor board.** There is created a prosecutor board consisting of 11 members, appointed for staggered 3–year terms, as follows:

(1) From each district under s. 752.11 (1) (b), (c), and (d), 2 district attorneys appointed by a majority of district attorneys from the district.

Vetoed In Part

- (2) From the district under s. 752.11 (1) (a), the district attorney and a deputy district attorney appointed by the district attorney.
- (3) Two nonelected prosecutors, each from a different county, appointed by a majority of nonelected prosecutors. Under this subsection, "prosecutor" does not include a special prosecutor appointed under s. 978.045 or 978.05 (8) (b).
 - (4) The attorney general or his or her designee.

SECTION 68m. 15.915 (7) of the statutes is created to read:

- 15.915 (7) PUBLIC LEADERSHIP BOARD. There is created a public leadership board attached to the University of Wisconsin System under s. 15.03. The board consists of the following members:
- (a) The director of the Tommy G. Thompson Center on Public Leadership appointed under s. 36.68 (3).
- (b) The president of the Thompson Family Charitable Foundation, Inc., or his or her designee.
- (c) One member, appointed for a 3-year term, who worked under the personal direction of former governor Tommy G. Thompson during the former governor's career in state or federal government.
- (d) Two members appointed for 3-year terms from a list of at least 3 names submitted by the speaker of the assembly.
- (e) Two members appointed for 3-year terms from a list of at least 3 names submitted by the senate majority leader.

SECTION 69d. 15.945 (title) of the statutes is repealed.

SECTION 69e. 15.945 (1) of the statutes is renumbered 15.405 (18) and amended to read:

15.405 (18) EDUCATIONAL APPROVAL BOARD. There is created an educational approval board which is attached to the technical college system board department of safety and professional services under s. 15.03. The board shall consist of not more than 7 members, who shall be representatives of state agencies and other persons with a demonstrated interest in educational programs, appointed to serve at the pleasure of the governor.

SECTION 73. 16.004 (20) of the statutes is created to read:

- 16.004 (20) SHARED SERVICES AGENCIES. (a) In this subsection, "shared services agency" means any of the following:
- 1. A department created in subch. II of ch. 15 except the department of employee trust funds, the department of justice, the department of public instruction, and the department of military affairs.
- 2. An independent agency created under subch. III of ch. 15 except the investment board, the public defender board, the Board of Regents of the University of Wisconsin System, and the technical college system board.
- (b) Subject to par. (c), the division of personnel management in the department shall administer for each

- shared service agency its responsibilities to provide human resources services and payroll and benefit services. Subject to par. (c), the department may charge shared services agencies for services provided under this paragraph.
- (c) 1. The department shall prepare an annual report that includes the following:
- a. Information identifying the assessments that the department intends to charge each shared services agency under par. (b) in the upcoming fiscal year.
- b. The number of positions that the department is using to administer human resources services and payroll and benefits services under par. (b).
- c. The number of vacant and filled positions that the department no longer needs to administer human resources services and payroll and benefits services under par. (b).
- d. The cost savings to the state due to the administration of human resources services and payroll and benefits services by the department.
- e. Metrics evaluating the effectiveness of human resources services and payroll and benefits services provided to shared services agencies by the department in the previous fiscal year, as well as a comparison of the metrics for the previous fiscal year to similar metrics in previous reports.
- 2. On April 15 of each year, the department shall submit the report under subd. 1. to the joint committee on finance. If the cochairpersons of the joint committee on finance do not notify the department that the committee has scheduled a meeting for the purpose of reviewing the report within 14 working days after the date of the submission, the department may provide the human resources services and payroll and benefits services as proposed in the report and may charge the assessments as proposed in the report. If within 14 working days after the date of the notification, the cochairpersons of the committee notify the department that the committee has scheduled a meeting for the purpose of reviewing the report, the department may provide the human resources services and payroll and benefits services proposed in the report and charge the assessments proposed in the report only upon approval of the committee.
- (d) The department shall provide human resources **Vetoed** services and payroll and benefits services on site for the In Part State Fair Park Board, the department of corrections, the department of health services, and the department of veterans affairs.

SECTION 74m. 16.004 (24) of the statutes is created

16.004 (24) School district employee health CARE. Annually, the secretary shall report to the joint committee on finance and to the appropriate standing committees of the legislature under s. 13.172 (3) the information it receives from school districts under s. 120.12 (24) (b).

SECTION 75. 16.009 (1) (a) of the statutes is repealed.
SECTION 76. 16.009 (1) (ac) of the statutes is created to read:

16.009 (1) (ac) "Access" means the ability to have contact with a person or to obtain, examine, or retrieve information or data pertinent to the activities of the board with respect to a person.

SECTION 77. 16.009 (1) (ar) of the statutes is amended to read:

16.009 (1) (ar) "Client" means an individual who requests <u>or is receiving</u> services of the office, <u>or a resident on whose behalf a request is made</u>.

SECTION 78. 16.009 (1) (b) of the statutes is repealed. SECTION 79. 16.009 (1) (br) of the statutes is created to read:

16.009 (1) (br) "Disclosure" means the release, the transfer, the provision of access to, or divulging in any manner of information outside the entity holding the information.

SECTION 80. 16.009 (1) (c) of the statutes is repealed. SECTION 81. 16.009 (1) (cg) of the statutes is created to read:

16.009 (1) (cg) "Enrollee" means an enrollee, as defined in s. 46.2805 (3), an individual receiving services under the Family Care Partnership Program or the program of all–inclusive care for the elderly, or an individual receiving long–term care benefits as a veteran.

SECTION 82. 16.009 (1) (cm) of the statutes is created to read:

16.009 (1) (cm) "Family Care Partnership Program" means an integrated health and long-term care program operated under an amendment to the state Medical Assistance plan under 42 USC 1396u-2 and a waiver under 42 USC 1396n (c).

SECTION 83. 16.009 (1) (cr) of the statutes is created to read:

16.009 (1) (cr) "Family care program" means the program under ss. 46.2805 to 46.2895 that provides the family care benefit, as defined in s. 46.2805 (4).

SECTION 84. 16.009 (1) (d) of the statutes is repealed. SECTION 85. 16.009 (1) (e) of the statutes is repealed. SECTION 86. 16.009 (1) (ef) of the statutes is created to read:

16.009 (1) (ef) "Immediate family member" means a member of a client's household or a relative of a client with whom the client has a close personal or significant financial relationship.

SECTION 87. 16.009 (1) (gr) of the statutes is amended to read:

16.009 (1) (gr) "Ombudsman" means the <u>state</u> longterm care ombudsman, as specified in sub. (4) (a), <u>or any employee or volunteer who is a representative of the office and who is designated by the state long-term care ombudsman to fulfill the duties under this section, 42 <u>USC 3058g</u>, and 45 CFR 1324.</u>

SECTION 88. 16.009 (1) (gv) of the statutes is created to read:

16.009 (1) (gv) "Program of all–inclusive care for the elderly" means in integrated health and long–term care program operated under 42 USC 1395eee or 1396u–4.

SECTION 89. 16.009 (1) (h) of the statutes is repealed.
SECTION 90. 16.009 (1) (k) of the statutes is created to read:

16.009 (1) (k) "Self-directed services option" has the meaning given in s. 46.2899 (1).

SECTION 91. 16.009 (2) (a) of the statutes is amended to read:

16.009 (2) (a) Appoint an executive director within the classified service who shall serve as the state longterm care ombudsman as specified under sub. (4) (a) and who shall employ staff within the classified service.

SECTION 92. 16.009 (2) (b) 1. of the statutes is amended to read:

16.009 (2) (b) 1. Investigate complaints from any person concerning improper conditions or treatment of aged or disabled persons who are 60 years of age or older and who receive long-term care in certified or licensed long-term care facilities or under programs administered by state or federal governmental agencies or concerning noncompliance with or improper administration of federal statutes or regulations or state statutes or rules related to long-term care for the aged or disabled persons who are 60 years of age or older.

SECTION 93. 16.009 (2) (b) 2. of the statutes is amended to read:

16.009 (2) (b) 2. Serve as mediator or advocate to resolve any problem or dispute relating to long–term care for the aged or disabled persons who are 60 years of age or older.

SECTION 94. 16.009 (2) (b) 3. of the statutes is created to read:

16.009 (2) (b) 3. Comply with the requirements of 42 USC 3058f to 3058h and 45 CFR 1321 and 1324.

SECTION 95. 16.009 (2) (d) of the statutes is amended to read:

16.009 (2) (d) Promote public education, planning, and voluntary acts to resolve problems and improve conditions involving long—term care for the aged or disabled persons who are 60 years of age or older.

SECTION 96. 16.009 (2) (e) of the statutes is amended to read:

16.009 (2) (e) Monitor, evaluate, and make recommendations concerning the development and implementation of federal, state, and local laws, regulations, rules, ordinances, and policies that relate to long-term care facilities and programs for the aged or disabled persons who are 60 years of age or older.

SECTION 97. 16.009 (2) (em) of the statutes is amended to read:

16.009 (2) (em) Monitor, evaluate, and make recommendations concerning long—term community support services received by clients of the long—term support community options program under s. 46.27, the family care program, the Family Care Partnership Program, and the program of all—inclusive care for the elderly.

SECTION 98. 16.009 (2) (f) of the statutes is amended to read:

16.009 (2) (f) As a result of information received while investigating complaints and resolving problems or disputes, publish material that assesses collect and publish materials that assess existing inadequacies in federal and state laws, regulations, and rules concerning long–term care for the aged or disabled persons who are 60 years of age or older. The board shall initiate legislation as a means of correcting collaborate with appropriate state agencies on efforts to resolve systemic concerns and shall recommend to the governor and the legislature legislation to remedy these inadequacies.

SECTION 99. 16.009 (2) (h) of the statutes is amended to read:

16.009 (2) (h) Conduct statewide hearings on issues of concern to aged or disabled persons who are 60 years of age or older and who are receiving or who may receive long—term care.

SECTION 100. 16.009 (2) (i) of the statutes is amended to read:

16.009 (2) (i) Report annually to the governor and the chief clerk of each house of the legislature for distribution to the appropriate standing committees under s. 13.172 (3). The report shall set forth the scope of the programs for providing long—term care for the aged or disabled persons who are 60 years of age or older developed in the state, the findings regarding the state's activities in the field of long—term care for the aged and disabled persons who are 60 years of age or older, any recommendations for a more effective and efficient total program, and the actions taken by the agencies of the state to carry out the board's recommendations.

SECTION 101. 16.009 (2) (p) (intro.) of the statutes is amended to read:

16.009 (2) (p) (intro.) Employ staff within the classified service or contract with one or more organizations to provide advocacy services to potential or actual recipients enrollees of the family care benefit, as defined in s. 46.2805 (4), or their families or guardians program, the Family Care Partnership Program, or the program of all–inclusive care for the elderly or potential or actual recipients of the self–directed services option. The board and contract organizations under this paragraph shall assist these persons in protecting their rights under all applicable federal statutes and regulations and state statutes and rules. An organization with which the board contracts for these services may not be a provider, nor an affiliate of a provider, of long–term care services, a resource center under s. 46.283 or a care management organization under

s. 46.284. For potential or actual recipients of the self—directed services option who are 60 years of age or older and for enrollees of the family care benefit program who are 60 years of age or older, advocacy services required under this paragraph shall include all of the following:

SECTION 102. 16.009 (2) (p) 5. of the statutes is amended to read:

16.009 (2) (p) 5. Providing individual case advocacy services in administrative hearings and legal representation for judicial proceedings regarding self-directed services option or family care services or benefits.

SECTION 103. 16.009 (3) (intro.) of the statutes is amended to read:

16.009 (3) (intro.) The board may shall:

SECTION 104. 16.009 (3) (a) of the statutes is amended to read:

16.009 (3) (a) Contract with any state agency to carry Carry out the board's long-term care ombudsman activities, as described in 42 USC 3058g (a) (2) and 45 CFR 1321 and 1324, and the activities of the Medigap Helpline program as specified in sub. (2) (j).

SECTION 105. 16.009 (3) (bm) of the statutes is amended to read:

16.009 (3) (bm) Employ an attorney for provision of legal services in accordance with requirements of the long-term care ombudsman program under 42 USC 3027 (a) (12) and 42 USC 3058g (g), as specified in 45 CFR 1324.15 (j).

SECTION 106. 16.009 (4) (a) of the statutes is amended to read:

16.009 (4) (a) The board shall operate the office in order to carry out the requirements of the long–term care ombudsman program, as defined in 42 USC 3058g (a) (2), under 42 USC 3027 (a) (12) (A) and 42 USC 3058f to 3058h and in compliance with 42 CFR 1321 and 1324. The executive director of appointed by the board shall serve as the state long–term care ombudsman under the office. The executive director of the board may delegate operation of the office to the staff employed under sub. (2) (a), as designated representatives of the ombudsman.

SECTION 107. 16.009 (4) (b) (intro.) of the statutes is amended to read:

16.009 (4) (b) (intro.) The ombudsman or his or her designated representative may have the following access to clients, residents, enrollees, and long-term care facilities:

SECTION 108. 16.009 (4) (d) of the statutes is amended to read:

16.009 (4) (d) An ombudsman acting as specified under 45 CFR 1324.11 (e) (2) (vii) is not subject to the provisions of the federal privacy rule under 45 CFR 160.101 to 164.534. A long–term care facility or personnel of a long–term care facility that disclose information make a disclosure as authorized under this subsection are not liable for that disclosure.

SECTION 109. 16.009 (4) (e) (intro.) of the statutes is amended to read:

16.009 (4) (e) (intro.) Information A disclosure of information of the office relating to a client, complaints, or investigations under the program may be disclosed made only at the discretion of the ombudsman or his or her designated representative. The identity of A disclosure of information relating to a client or named witness or of a resident who is not a client may be revealed made under this paragraph only if one of the following conditions is met:

SECTION 110. 16.009 (5) (a) 3. of the statutes is created to read:

16.009 (5) (a) 3. Willfully interfere with the actions of an ombudsman by acting or attempting to act to intentionally prevent, interfere with, or impede the ombudsman from performing any of the functions or responsibilities under this section.

SECTION 111. 16.047 of the statutes is created to read: 16.047 Volkswagen settlement funds. (1) DEFINITIONS. In this section:

- (a) "Settlement funds" means moneys allocated to this state from the environmental mitigation trust specified in par. (d) and received by the state from the trustee.
- (b) "Settlement guidelines" means the eligible mitigation actions established under the partial consent decree specified in par. (d) and all other partial consent decrees entered in the federal court case specified in par. (d) under which this state receives settlement funds.
- (c) "State agency" has the meaning given in s. 20.001 (1).
- (d) "Trustee" means the trustee of the environmental mitigation trust required to be established under the partial consent decree entered on October 25, 2016, by the United States District Court for the Northern District of California, San Francisco Division, Case No: MDL No. 2672 CRB (JSC).
- (2) REPLACEMENT OF STATE VEHICLES. (a) From the appropriation under s. 20.855 (4) (h), the department may use settlement funds for the payment of all costs incurred in accordance with the settlement guidelines to replace vehicles in the state fleet.
- (b) Any use of settlement funds under par. (a) shall take precedence over any distribution under sub. (4m).
- (c) The department may expend no more than \$10,000,000 under par. (a) during the 2017–19 fiscal biennium.
- (3) STATE AGENCY LAPSES. If the department replaces a state agency's vehicle under sub. (2) (a), the secretary may calculate the general purpose revenue or program revenue savings for the state agency resulting from expenditures under s. 20.855 (4) (h) and may lapse to the general fund from the state agency's general purpose revenue or program revenue appropriations the amount calculated.

- **(4m)** Transit capital assistance grants. (a) In this subsection:
- 1. "Eligible applicant" has the meaning given in s. 85.20 (1) (b).
- 2. "Public transit vehicle" means any vehicle used for providing transportation service to the general public that is eligible for replacement under the settlement guidelines
- (b) The department shall establish a program to award grants of settlement funds from the appropriation under s. 20.855 (4) (h) to eligible applicants for the replacement of public transit vehicles. Any eligible applicant may apply for a grant under the program.
- (c) The department shall award grants under this subsection on a competitive basis and shall give preference to the replacement of public transit vehicles in communities or on routes that the department determines are critical for the purpose of connecting employees with employers.
- (d) An eligible applicant may use settlement funds awarded under this subsection only for the payment of costs incurred by the eligible applicant to replace public transit vehicles in accordance with the settlement guidelines
- (e) The department may not award more than a total of \$32,000,000 in grants under this subsection.
- (5) SUNSET. This section does not apply after June 30, 2027.

SECTION 112. 16.25 (3) (d) 1. of the statutes, as affected by 2017 Wisconsin Act 12, is amended to read: 16.25 (3) (d) 1. Subject to subd. 2., the department shall provide a match equal to twice the amount of all annual municipal contributions paid for volunteer fire fighters, emergency medical responders, and emergency medical services practitioners up to \$250 a state match of \$390 per fiscal year, other than contributions paid for the purchase of additional years of service under par. (e), to be paid from the appropriation account under s. 20.505 (4) (er). This amount shall be adjusted annually on July 1 to reflect any changes in the U.S. consumer price index for all urban consumers, U.S. city average, as determined by the U.S. department of labor, for the 12-month period ending on the preceding December 31. The department shall pay all amounts that are matched under this paragraph to the individuals and organizations offering the plans selected by the municipalities.

SECTION 113. 16.25 (3) (g) of the statutes, as affected by 2017 Wisconsin Act 12, is amended to read:

16.25 (3) (g) A volunteer fire fighter, emergency medical responder, or emergency medical services practitioner shall be paid a length of service award either in a lump sum or in a manner specified by rule, consisting of all municipal and state contributions made on behalf of the volunteer fire fighter, emergency medical responder, or emergency medical services practitioner and all

Vetoed In Part

earnings on the contributions, less any expenses incurred in the investment of the contributions and earnings, after the volunteer fire fighter, emergency medical responder, or emergency medical services practitioner attains 20 15 years of service for a municipality and reaches the age of 60. If a volunteer fire fighter, emergency medical responder, or emergency medical services practitioner has satisfied all vesting requirements under the program but has at least 10 but less than 20 15 years of service for a municipality or has reached the age of 53 but has not reached the age of 60, the program shall provide for the payment of a length of service award either in a lump sum or in a manner specified by rule in an amount to be determined by the department, but less than the amount paid to a volunteer fire fighter, emergency medical responder, or emergency medical services practitioner who has attained 20 15 years of service for a municipality and has reached the age of 60. The department shall promulgate rules implementing this paragraph.

SECTION 113c. 16.25 (3) (h) of the statutes is repealed.

SECTION 113d. 16.25 (3) (j) of the statutes, as affected by 2017 Wisconsin Act 12, is amended to read:

16.25 (3) (j) The account of any volunteer fire fighter, emergency medical responder, or emergency medical services practitioner who has not met all of the vesting requirements under the program, who has not provided volunteer fire fighting, emergency medical responder, or emergency medical technical services for a municipality for a period of 6-12 months or more, who does not meet any other program requirement established by the municipality, and who has not been granted a leave of absence by his or her supervisor shall be closed.

SECTION 114. 16.255 (title) of the statutes is renumbered 224.51 (title).

SECTION 115. 16.255 (1) of the statutes is renumbered 224.51 (1m), and 224.51 (1m) (intro.), as renumbered, is amended to read:

224.51 (**1m**) (intro.) The department shall determine the factors to be considered in selecting a vendor of the program under s. 16.641 224.50, which shall include:

SECTION 116. 16.255 (2) of the statutes is renumbered 224.51 (2) and amended to read:

224.51 (2) The department shall solicit competitive sealed proposals under s. 16.75 (2m) from nongovernmental persons to serve as vendor of the college savings program. The department shall select the vendor based upon factors determined by the department under sub. (1m).

SECTION 117. 16.255 (3) of the statutes is renumbered 224.51 (3), and 224.51 (3) (d), as renumbered, is amended to read:

224.51 (3) (d) That the vendor communicate to the beneficiary and account owner the requirements of s. 16.641 224.50 (8).

SECTION 118. 16.303 (2) (c) of the statutes is repealed.

SECTION 119. 16.306 (title) of the statutes is amended to read:

 $\begin{array}{ccc} \textbf{16.306} & (\text{title}) & \textbf{Transitional housing} & \underline{\textbf{Housing}} \\ \textbf{grants.} \end{array}$

SECTION 120. 16.306 (1) (intro.) and (a) (intro.) of the statutes are consolidated, renumbered 16.306 (1) (intro.) and amended to read:

16.306 (1) DEFINITIONS <u>DEFINITION</u>. (intro.) In this section: (a) (intro.) "Eligible, "eligible applicant" means any of the following:

SECTION 121. 16.306 (1) (a) 1., 2., 3., 4. and 5. of the statutes are renumbered 16.306 (1) (am), (bm), (c), (d) and (e).

SECTION 122. 16.306 (1) (b) of the statutes is repealed.

SECTION 123. 16.306 (2) (a) of the statutes is amended to read:

16.306 (2) (a) From the appropriation under s. 20.505 (7) (fm), the department may award a grant to an eligible applicant for the purpose of providing transitional housing and associated supportive services to homeless individuals and families to facilitate their movement to independent living if the conditions under par. (b) are satisfied. The department shall ensure that the funds for the grants are reasonably balanced among geographic areas of the state that correspond to the geographic areas served by each continuum of care organization designated by the federal department of housing and urban development, consistent with the quality of applications submitted.

SECTION 124. 16.306 (2) (b) (intro.) of the statutes is amended to read:

16.306 (2) (b) (intro.) A recipient of a grant under par. (a) shall agree to use the grant to support a transitional housing program that does all of the following:

SECTION 125. 16.306 (2) (b) 6. of the statutes is repealed.

SECTION 126. 16.306 (3) (b) of the statutes is amended to read:

16.306 (3) (b) The length of stay in transitional housing of each person served.

SECTION 127. 16.306 (3) (c) of the statutes is amended to read:

16.306 (3) (c) The housing and employment status of each person served, at the time that the person leaves the transitional housing program.

SECTION 128. 16.306 (3) (d) of the statutes is amended to read:

16.306 (3) (d) Any other information that the department determines to be necessary to evaluate the effectiveness of the transitional housing program operated by the recipient.

SECTION 129. 16.3085 of the statutes is created to read:

- **16.3085** Homeless case management services grants. (1) DEFINITION. In this section, "shelter facility" has the meaning given in s. 16.308 (1) (d).
- (2) GRANTS. (a) From the appropriation under s. 20.505 (7) (kg), the department may award up to 10 grants, of up to \$50,000 each, annually to any of the following:

Vetoed In Part

1. A shelter facility.

Vetoed In Part 2. A nonprofit organization that partners with local governments, religious organizations, local businesses, or charitable organizations to provide individuals and families with rent assistance and intensive case management.

Vetoed In Part

- (b) A shelter facility shall use all grant moneys awarded to it under par. (a) 1. for the purpose of providing intensive case management services to homeless families, including any of the following:
 - 1. Services related to financial management.
- 2. Employment–related services, including connecting parents who are job training graduates or who have a recent work history with their local workforce development board established under 29 USC 2832 and assisting them with using the job center Web site maintained by the department of workforce development.
- 3. Services intended to ensure continuation of school enrollment for children.
- 4. Services related to the enrollment of unemployed or underemployed parents in a food stamp employment and training program under s. 49.79 (9) or in the Wisconsin Works program under ss. 49.141 to 49.161.

Vetoed In Part (c) A nonprofit organization shall use all grant moneys awarded to it under par. (a) 2. for the purpose of providing immediate housing relocation services to individuals and families, including paying rent on behalf of participants in private housing.

SECTION 130. 16.311 (title) of the statutes is repealed.

SECTION 131. 16.311 (1) of the statutes is repealed. **SECTION 132.** 16.311 (2) of the statutes is renumbered 51.047 and amended to read:

51.047 Mental health services. From the appropriation under s. 20.505 (7) 20.435 (5) (fr), the department may not award more than \$45,000 in each fiscal year to applying public or nonprofit private entities for the costs of providing certain mental health services to homeless individuals with serious and persistent mental illness. Entities that receive funds awarded by the department under this subsection section shall provide the mental health services required under 42 USC 290cc–24. The amount that the department awards to an applying entity may not exceed 50 percent of the amount of matching funds required under 42 USC 290cc–23.

SECTION 133. 16.313 of the statutes is created to read:

- **16.313** Employment grants. (1) In this section, "municipality" means a county, city, village, or town.
- (2) (a) Any municipality may apply for a grant under this section.
- (b) The department may award a grant of up to \$75,000 to a municipality that submits an application under par. (a). The grant and all moneys contributed by the municipality under sub. (3) shall be used for the purpose of connecting homeless individuals with permanent employment.
- (3) A municipality receiving a grant under sub. (2) shall itself contribute at least \$50,000 for the purpose specified in sub. (2) (b).
- (4) In considering grant applications submitted under sub. (2) (a), the department shall give preference to a municipality that obtains an agreement from a nonprofit organization to provide additional employment and support services to homeless individuals participating in the grant program.
- (5) In considering grant applications submitted under sub. (2) (a), the department shall give preference to a municipality that places a priority on using the grant moneys and the moneys contributed by the municipality under sub. (3) for the purpose of paying the wages of homeless individuals participating in the grant program under this section.

SECTION 136. 16.401 (14) of the statutes is amended to read:

16.401 (14) APPORTION INTEREST. Apportion at least quarterly the interest earned on state moneys in all depositories among the several funds as provided in s. 25.14 (3), except that earnings attributable to the investment of temporary excess balances under sub. (4) (b) shall be distributed according to a formula prescribed by the depository selection board secretary or his or her designee. To the maximum extent deemed administratively feasible by the depository selection board secretary or his or her designee, the formula shall approximate the distribution of earnings among funds which would occur if earnings were allocated in proportion to each fund's actual contribution to the earnings. Interest so apportioned shall be added to and become a part of such funds.

SECTION 137. 16.417 (1) (e) of the statutes is created to read:

16.417 (1) (e) "Health care professional" means any of the following:

- 1. A registered nurse who is licensed under s. 441.06 or in a party state, as defined in s. 441.50 (2) (j), or permitted under s. 441.08.
- 2. A licensed practical nurse who is licensed or has a temporary permit under s. 441.10 or who is licensed as a licensed practical/vocational nurse in a party state, as defined in s. 441.50 (2) (j).
- 3. A physician who is licensed to practice medicine and surgery under s. 448.02.

3m. A physician assistant who is licensed under s. 448.04 (1) (f).

4. A psychologist who is licensed to practice psychology under ch. 455.

SECTION 138. 16.417 (2) (a) of the statutes is amended to read:

16.417 (2) (a) No individual other than an elective state official who is employed or retained in a full-time position or capacity with an agency or authority may hold any other position or be retained in any other capacity with an agency or authority from which the individual receives, directly or indirectly, more than \$12,000 from the agency or authority as compensation for the individual's services during the same year any 12-month period.

SECTION 139. 16.417 (2) (f) 3. of the statutes is created to read:

16.417 (2) (f) 3. A health care professional who is employed or retained in a full-time position or capacity with an agency or authority and who holds another position or is retained in any other capacity with an agency or authority for less than 1,040 hours during any 12-month period.

Vetoed In Part **SECTION 139m.** 16.42 (5) of the statutes is created to

16.42 (5) (a) In this subsection, "fee" means any amount of money other than a tax that an agency charges a person other than a governmental entity.

- (b) Each agency required to submit a budget request under sub. (1) shall include with its request a report that lists each fee the agency is required or otherwise authorized to charge and that, for each fee, includes all of the
- 1. The amount of the fee, or, if the fee does not have a fixed amount, the method of calculating the fee.
- 2. An identification of the agency's statutory authority to charge the fee.
- 3. A statement of whether the agency currently charges the fee.
- 4. A description of whether and how the fee has increased or decreased since the agency was first authorized to charge the fee.
- 5. Any recommendation the agency has concerning the fee.

Vetoed In Part

SECTION 140k. 16.47 (1d) of the statutes is created to read:

16.47 (1d) The executive budget bill or bills shall satisfy the requirement applicable to bills adopted by the legislature under s. 20.003 (4m).

SECTION 141. 16.505 (2) (am) of the statutes is created to read:

16.505 (2) (am) The state public defender board may request the governor to create or abolish a full-time equivalent position or portion thereof funded from revenues specified in s. 20.001 (2) (a) in the office of the state public defender. Upon receiving such a request, the governor may change the authorized level of full-time equivalent positions funded from such revenues in the office of the state public defender in accordance with this subsection. The governor may approve a different authorized level of positions than is requested by the state public defender board. If the governor proposes to change the number of full-time equivalent positions in the office of the state public defender funded from revenues specified in s. 20.001 (2) (a), the governor shall notify the joint committee on finance in writing of his or her proposed action. If the cochairpersons of the committee do not notify the governor that the committee has scheduled a meeting for the purpose of reviewing the proposed action within 14 working days after the date of the governor's notification, the position changes may be made as proposed by the governor. If, within 14 working days after the date of the governor's notification, the cochairpersons of the committee notify the governor that the committee has scheduled a meeting for the purpose of reviewing the proposed action, the position changes may be made under this subsection only upon approval of the committee.

SECTION 142. 16.505 (2) (b) of the statutes is amended to read:

16.505 (2) (b) This subsection does not apply to fulltime equivalent positions funded from the appropriation under s. 20.370 (2) (bg) (4) (co) or (8) (mg).

SECTION 143. 16.515 (3) of the statutes is amended

16.515 (3) This section does not apply to supplementation of the appropriation under s. 20.370 (2) (bg) (4) (co) or (8) (mg).

SECTION 144. 16.5185 (intro.) of the statutes is renumbered 16.5185 (1) (intro.).

SECTION 145. 16.5185 (1) of the statutes is renumbered 16.5185 (1) (a).

SECTION 146. 16.5185 (2) of the statutes is renumbered 16.5185 (1) (b).

SECTION 147. 16.5185 (2m) of the statutes is created

16.5185 (**2m**) Beginning on June 30, 2020, in each fiscal year, the secretary shall transfer the unencumbered balance of the petroleum inspection fund on June 30, less an amount sufficient to meet the reserve requirement under this subsection, from the petroleum inspection fund to the transportation fund. The petroleum inspection fund balance after a transfer under this subsection may not be less than 5 percent of gross revenues received during the fiscal year in which the transfer is made.

SECTION 147d. 16.5185 (2n) of the statutes is created **Vetoed** to read:

In Part

16.5185 (**2n**) In fiscal year 2017–18, the secretary shall transfer \$50,000 from the general fund to the transportation fund. Beginning on June 30, 2019, in each fiscal year, the secretary shall transfer \$200,000 from the general fund to the transportation fund.

SECTION 148. 16.64 of the statutes is renumbered 224.48.

SECTION 149. 16.641 of the statutes is renumbered 224.50, and 224.50 (2) (a), as renumbered, is amended to read:

224.50 (2) (a) Except as provided in s. 16.255 224.51, establish and administer a college savings program that allows an individual, trust, legal guardian, or entity described under 26 USC 529 (e) (1) (C) to establish a college savings account to cover tuition, fees, and the costs of room and board, books, supplies, and equipment required for the enrollment or attendance of a beneficiary at an eligible educational institution, as defined under 26 USC 529.

SECTION 150. 16.642 of the statutes is renumbered 224.52 and amended to read:

- **224.52 Repayment to the general fund.** (1) The secretary of administration shall transfer from the tuition trust fund, the college savings program trust fund, the college savings program bank deposit trust fund, or the college savings program credit union deposit trust fund to the general fund an amount equal to the amount expended from the appropriations under s. 20.505 (9) (a), 1995 stats., s. 20.585 (2) (a), 2001 stats., and s. 20.585 (2) (am), 2001 stats., when the secretary of administration determines, after consultation with the secretary of financial institutions, that funds in those trust funds are sufficient to make the transfer. The secretary of administration may make the transfer in installments.
- (2) Annually, by June 1, the secretary of financial institutions, after consultation with the secretary of administration, shall submit a report to the joint committee on finance on the amount available for repayment under sub. (1), the amount repaid under sub. (1), and the outstanding balance under sub. (1).

SECTION 151. 16.705 (1b) (d) of the statutes is created to read:

16.705 **(1b)** (d) The department of financial institutions under s. 224.51.

SECTION 153. 16.71 (5r) of the statutes is created to read:

16.71 (**5r**) The department shall delegate authority to the department of financial institutions to enter into vendor contracts under s. 224.51.

SECTION 161. 16.84 (2) of the statutes is amended to read:

16.84 (2) Appoint such number of police officers as is necessary to safeguard all public property placed by law in the department's charge, and provide, by agreement with any other state agency, police and security services at buildings and facilities owned, controlled, or occupied by the other state agency. The department may charge the other state agency for the cost of providing security services at multitenant buildings or multitenant state facilities. The governor or the department may, to the extent it is necessary, authorize police officers

employed by the department to safeguard state officers, state employees, or other persons. A police officer who is employed by the department and who is performing duties that are within the scope of his or her employment as a police officer has the powers of a peace officer under s. 59.28, except that the officer has the arrest powers of a law enforcement officer under s. 968.07 regardless of whether the violation is punishable by forfeiture or criminal penalty. The officer may exercise the powers of a peace officer and the arrest powers of a law enforcement officer while located anywhere within this state. Nothing in this subsection limits or impairs the duty of the chief and each police officer of the police force of the municipality in which the property is located to arrest and take before the proper court or magistrate persons found in a state of intoxication or engaged in any disturbance of the peace or violating any state law in the municipality in which the property is located, as required by s. 62.09 (13).

SECTION 161d. 16.84 (5) of the statutes is renumbered 16.84 (5) (a) and amended to read:

Vetoed In Part

16.84 (5) (a) Have responsibility, subject to approval of the governor, for all functions relating to the leasing, acquisition, allocation, and utilization of all real property by the state, except where such responsibility is otherwise provided by the statutes. In exercising this connection responsibility, the department shall may not enter into, extend, or renew a lease for an executive branch agency, as defined in s. 16.70 (4), involving an annual rent of more than \$500,000 unless the secretary signs the lease, a copy of the proposed lease is submitted electronically to the chief clerk of each house for distribution, and the department notifies the joint committee on finance of the proposed lease and provides the committee with the information under par. (b) as well as a summary report of that information, including the terms of the lease and the lease rate per square foot of the proposed property and the comparable options. If the cochairpersons of the joint committee on finance do not notify the secretary that the committee has scheduled a meeting for the purpose of reviewing the proposed lease within 14 working days after the date of the notification, the lease may be entered into, extended, or renewed. If, within 14 working days after the date of the notification, the cochairpersons of the committee notify the secretary that the committee has scheduled a meeting for the purpose of reviewing the proposed lease, the lease may be entered into, extended, or renewed only upon approval of the committee.

- (c) When exercising the responsibility under par. (a), with the governor's approval, require physical consolidation of office space utilized by any executive branch agency, as defined in s. 16.70 (4), having fewer than 50 authorized full–time equivalent positions with office space utilized by another executive branch agency, whenever feasible. The department shall lease
- (d) Lease or acquire office space for legislative offices or legislative service agencies at the direction of

the joint committee on legislative organization. In this subsection, "executive branch agency" has the meaning given in s. 16.70 (4).

SECTION 161e. 16.84 (5) (b) of the statutes is created to read:

16.84 (5) (b) Before entering into, extending, or renewing a lease, do all of the following:

- 1. Conduct a cost-benefit analysis comparing the lease with purchasing the space or another suitable space.
- 2. Evaluate comparable lease options within a 10-mile radius of the property proposed in the lease, or if there are not sufficient comparable properties within a 10-mile radius to perform a meaningful comparison, a wider radius as needed, to ensure the lease rate per square foot does not exceed the lease rate per square foot on comparable properties or the market rate by more than 5 percent.

SECTION 162. 16.848 (2) (f) of the statutes is amended to read:

16.848 (2) (f) Subsection (1) does not apply to lands acquired with revenues collected paid into the conservation fund under s. 70.58.

SECTION 164. 16.856 of the statutes is repealed.

SECTION 169e. 16.9645 (title) of the statutes is renumbered 323.29 (2) (title).

SECTION 169m. 16.9645 (1) of the statutes is renumbered 323.29 (1), and 323.29 (1) (a), as renumbered, is amended to read:

323.29 (1) (a) "Council" means the interoperability council created under s. 15.107 (18) 15.315 (1) (a).

SECTION 169s. 16.9645 (2) of the statutes is renumbered 323.29 (2), and 323.29 (2) (d), (e) and (f) (intro.), as renumbered, are amended to read:

- 323.29 (2) (d) Assist the department of justice in identifying and obtaining funding to implement a statewide public safety interoperable communication system.
- (e) Advise the department of justice and the department of military affairs on allocating funds, including those available for homeland security, for the purpose of achieving the goals under par. (b).
- (f) (intro.) Make recommendations to the department of justice on all of the following:

Vetoed In Part

SECTION 169t. 16.971 (2) (cg) of the statutes is created to read:

16.971 (2) (cg) In October 2017, and every 6 months thereafter, submit a report to the joint committee on finance and the joint committee on information policy and technology relating to the management of the enterprise resource planning system maintained under par. (cf). Each report shall include all of the following:

- 1. An accounting of all expenditures in the current fiscal year from the appropriations under ss. 20.505 (1) (iv) and (kd) and 20.865 (2) (i) and (r).
- 2. An identification of all master leases originated since the date of the immediately preceding report under this paragraph.

3. An accounting of all state agency assessments Vetoed charged in the immediately preceding fiscal year, an accounting of all assessments charged in the current fiscal year, and an estimate of the charges anticipated for future fiscal years.

- 4. An accounting of the status of any deficit in the appropriation accounts under s. 20.505 (1) (iv) and (kd).
- 5. Current information concerning the department's efforts with respect to benefits realization, including all actual or anticipated savings and efficiencies associated with the enterprise resource planning system.

SECTION 171. 16.971 (9) of the statutes is amended to read:

16.971 (9) In conjunction with the public defender board, the director of state courts, the departments of corrections and justice and district attorneys, the department may maintain, promote and coordinate automated justice information systems that are compatible among counties and the officers and agencies specified in this subsection, using the moneys appropriated under s. 20.505 (1) (kh), (kp), and (kq). The department shall annually report to the legislature under s. 13.172 (2) concerning the department's efforts to improve and increase the efficiency of integration of justice information systems.

SECTION 171b. 16.971 (9) of the statutes, as affected **Vetoed** by 2017 Wisconsin Act (this act), is amended to read: In Part

16.971 (9) In conjunction with the public defender board, the prosecutor board, the director of state courts, and the departments of corrections and justice and district attorneys, the department may maintain, promote and coordinate automated justice information systems that are compatible among counties and the officers and agencies specified in this subsection, using the moneys appropriated under s. 20.505 (1) (kh) and (kq). The department shall annually report to the legislature under s. 13.172 (2) concerning the department's efforts to improve and increase the efficiency of integration of justice information systems.

SECTION 171c. 16.971 (10) of the statutes is amended to read:

16.971 (10) The department shall maintain, and provide the department of justice and the state prosecutors office with general access to, a case management system that allows the state prosecutors office and district attorneys to manage all case-related information and share the information among prosecutors.

SECTION 172. 16.973 (15) of the statutes is created to **Vetoed** read:

16.973 (15) By October 1 of each year, submit to the joint committee on finance and the legislature under s. 13.172 (2) a report on the administration of the information technology and communication services selffunded portal. The report shall include the following information regarding the portal for the immediately preceding fiscal year:

In Part

In Part

- (a) A financial statement of state revenues and expenditures.
- (b) A list of services available through the portal, identifying services added since the previous reporting
- (c) Fees charged for each service available through the portal.
- (d) The activity level of each service available through the portal.
- (e) Any other information the department determines to be appropriate to include.

SECTION 174. 16.993 (10) of the statutes is repealed. **SECTION 175.** 16.9945 of the statutes is created to read:

16.9945 Information technology block grants. (1) COMPETITIVE GRANTS. In fiscal years 2017-18 and 2018-19, the department may annually award grants on a competitive basis to eligible school districts for the purpose of improving information technology infrastructure. For purposes of awarding grants under this section, "improving information technology infrastructure" includes purchasing and installing on a bus a portable device that creates an area of wireless Internet coverage and purchasing for individuals to temporarily borrow from a school a portable device that creates an area of wireless Internet coverage. In awarding grants under this section, the department shall give priority to applications for school districts in which the percentage of pupils who satisfy the income eligibility criteria under 42 USC 1758 (b) (1) for a free or reduced–price lunch is greater than in other applicant school districts. The department shall require an applicant for a grant under this section to provide all of the following:

- (a) A description of the specific information technology infrastructure, including any equipment, that the applicant intends to purchase with grant proceeds.
- (b) The applicant's plan to purchase, install, and use the information technology infrastructure described in
- (c) A description of the applicant's readiness to use information technology infrastructure purchased with grant proceeds.
- (2) ELIGIBLE SCHOOL DISTRICTS. (a) A school district is eligible for a grant under this section in fiscal year 2017–18 if the school district's membership in the previous school year divided by the school district's area in square miles is 16 or less.
- (b) A school district is eligible for a grant under this section in fiscal year 2018-19 if the school district's membership in the previous school year divided by the school district's area in square miles is 16 or less.
- (3) MAXIMUM AWARDS. The total amount the department may award to an eligible school district under sub. (1) during a fiscal biennium may not exceed the following:

- (a) If the membership of the eligible school district is fewer than 750 pupils, \$30,000.
- (b) If the membership of the eligible school district is 750 pupils to 1,500 pupils, \$40 multiplied by the school district's membership.
- (c) If the membership of the eligible school district is more than 1,500 pupils, \$60,000.
- (4) FUNDING LIMITATION. (a) The department may not award grants under this section that total more than \$15,000,000 in the 2017–18 fiscal year.
- (b) The department may not award grants under this section that total more than \$7,500,000 in the 2018–19 fiscal year.
- (5) SUNSET. The department may not award grants under this section after July 1, 2019.

SECTION 177. 19.11 (1) to (3) of the statutes are amended to read:

- 19.11 (1) The secretary of state, and treasurer and attorney general shall each furnish a bond to the state, at the time each takes and subscribes the oath of office required of that officer, conditioned for the faithful discharge of the duties of the office, and the officer's duties as a member of the board of commissioners of public lands, and in the investment of the funds arising therefrom. The bond of each of said officers shall be further conditioned for the faithful performance by all persons appointed or employed by the officer in his or her office of their duties and trusts therein, and for the delivery over to the officer's successor in office, or to any person authorized by law to receive the same, of all moneys, books, records, deeds, bonds, securities and other property and effects of whatsoever nature belonging to the officer's
- (2) Each of said bonds shall be subject to the approval of the governor and shall be guaranteed by resident freeholders of this state, or by a surety company as provided in s. 632.17 (2). The amount of each such bond, and the number of sureties thereon if guaranteed by resident freeholders, shall be as follows: secretary of state, \$25,000, with sufficient sureties; and treasurer, \$100,000, with not less than 6 sureties; and the attorney general, \$10,000, with not less than 3 sureties.
- (3) The attorney general shall renew the bond required under this section in a larger amount and with additional security, and the The treasurer shall give an additional bond, when required by the governor.

SECTION 177s. 19.32 (1) of the statutes is amended **Vetoed** to read:

In Part

19.32 (1) "Authority" means any of the following having custody of a record: a state or local office, elective official, agency, board, commission, committee, council, department or public body corporate and politic created by the constitution or by any law, ordinance, rule or order; a governmental or quasi-governmental corporation except for the Bradley center sports and entertainment

In Part

In Part

corporation; a special purpose district; any court of law; the assembly or senate; a nonprofit corporation which receives more than 50 percent of its funds from a county or a municipality, as defined in s. 59.001 (3), and which provides services related to public health or safety to the county or municipality; a university police department under s. 175.42; a commission, as defined in s. 66.0304 (1) (c); or a formally constituted subunit of any of the

SECTION 178. 19.36 (3) of the statutes is amended to read:

19.36 (3) CONTRACTORS' RECORDS. Subject to sub. (12), each Each authority shall make available for inspection and copying under s. 19.35 (1) any record produced or collected under a contract entered into by the authority with a person other than an authority to the same extent as if the record were maintained by the authority. This subsection does not apply to the inspection or copying of a record under s. 19.35 (1) (am).

Vetoed In Part

SECTION 179. 19.36 (12) of the statutes is repealed. **SECTION 179e.** 19.42 (7w) (f) of the statutes is created to read:

19.42 (7w) (f) The position of member of the board of a commission created under s. 66.0304.

SECTION 179f. 19.45 (11) (e) of the statutes is created to read:

19.45 (11) (e) A commission established under s. 66.0304 shall establish a code of ethics for members of the board, and employees, contract staff, and agents of a commission established under s. 66.0304 who are not state public officials and shall file the code of ethics with the department of administration. A commission shall provide the department of administration with any amendments to the code of ethics within 30 days of adoption of the amendment.

SECTION 179s. 19.82 (1) of the statutes is amended to read:

19.82 (1) "Governmental body" means a state or Vetoed local agency, board, commission, committee, council, department or public body corporate and politic created by constitution, statute, ordinance, rule or order; a governmental or quasi-governmental corporation except for the Bradley center sports and entertainment corporation; a local exposition district under subch. II of ch. 229; a long-term care district under s. 46.2895; the board of a commission, as defined in s. 66.0304 (1) (c); or a formally constituted subunit of any of the foregoing, but excludes any such body or committee or subunit of such body which is formed for or meeting for the purpose of collective bargaining under subch. I, IV, or V of ch. 111.

SECTION 180. 20.002 (2) (a) of the statutes is amended to read:

20.002 (2) (a) Solely for purposes of relating annual taxes to estimated expenses, amounts withheld under s. 71.64 prior to July 1 and taxes imposed by subch. III of ch. 77 for periods ending prior to July 1 shall be deemed accrued tax receipts as of the close of the fiscal year but no revenue shall be deemed accrued tax receipts unless deposited by the state on or before the August 15 following the end of the fiscal year. Solely for purposes of relating annual taxes to estimated expenses, fees imposed under subch. II of ch. 77, taxes imposed under ss. 139.02, 139.03 (2m) and (2n), 139.31 and 139.76 and assessments imposed under s. 50.14 (2) shall be deemed accrued tax receipts as of the close of the fiscal year, but no revenue shall be deemed accrued tax receipts unless deposited by this state on or before July 31. Solely for purposes of relating annual taxes to estimated expenses, taxes imposed under s. 70.58 shall be deemed accrued tax receipts as of the close of the fiscal year, but no revenue shall be deemed accrued tax receipts unless it is deposited by this state on or before August 31.

SECTION 181. 20.005 (1) of the statutes is repealed and recreated to read:

20.005 (1) SUMMARY OF ALL FUNDS. The budget governing fiscal operations for the state of Wisconsin for all funds beginning on July 1, 2017, and ending on June 30, 2019, is summarized as follows: [See Figure 20.005 (1) following]

Figure: 20.005 (1)

GENERAL FUND SUMMARY					
2017–18 2018–19					
Opening Balance, July 1	\$ 467,148,800	\$ 442,294,800			
Revenues					
Taxes	\$ 16,077,884,900	\$ 16,638,840,900			
Departmental Revenues					
Tribal Gaming Revenues	25,909,500	25,838,400			
Other	493,207,400	443,289,900			

017 Wisconsin Act 59	– 18 –	20	2017 Assembly Bill 64		
		2017–18	2018–19		
Total Available		\$ 17,064,150,600	\$17,550,264,000		
Appropriations, Transfers, and Reserves					
Gross Appropriations		\$ 16,877,365,200	\$ 17,693,422,100		
Other Acts/Bills*		19,712,500	10,055,000		
Transfers to:					
Transportation Fund		40,244,700	41,797,100		
Compensation Reserves		3,080,500	52,081,600		
Less Lapses		(318,547,100)	(441,515,200)		
Net Appropriations		\$ 16,621,855,800	\$ 17,355,840,600		
Balances					
Gross Balance		\$ 442,294,800	\$ 194,423,400		
Less Required Statutory Balance		(70,000,000)	(75,000,000)		

^{*2017} Acts 1 to 58.

Net Balance, June 30

SUMMARY OF APPROPRIATIONS — ALL FUNDS

\$ 372,294,800 \$ 119,423,400

	2017–18	2018-19
General Purpose Revenue	\$ 16,877,365,200	\$ 17,693,422,100
F. I. I.D.	ф 10 <i>(</i> 24 001 400	Ф 11 001 1 <i>5</i> 0 100
Federal Revenue	\$ 10,624,981,400	\$ 11,001,150,100
Program	(9,688,025,800)	(10,048,467,400)
Segregated	(936,955,600)	(952,682,700)
Program Revenue	\$ 5,950,607,900	\$ 6,079,142,300
State	(5,061,833,100)	(5,149,503,800)
Service	(888,774,800)	(929,638,500)
Segregated Revenue	\$ 3,687,236,200	\$ 3,676,657,900
State	(3,472,393,300)	(3,457,460,400)
Local	(115,325,600)	(115,325,600)
Service	(99,517,300)	(103,871,900)
GRAND TOTAL	\$ 37,140,190,700	\$ 38,450,372,400

SUMMARY OF COMPENSATION RESERVES — ALL FUNDS

)17–18	2018–19		
General Purpose Revenue	\$	3,080,500	\$	52,081,600	
Federal Revenue		796,900		13,329,800	
Program Revenue		2,421,000		44,181,400	
Segregated Revenue		520,300		7,574,700	
TOTAL	\$	6,818,700	\$	117,167,500	
LOTTERY FUND SUM	1MA	RY			
	2	017–18	2	2018–19	
Gross Revenue					
Ticket Sales	\$	609,675,900	\$	619,157,100	
Miscellaneous Revenue		83,200	·	154,900	
	\$	609,759,100	\$	619,312,000	
Expenses					
Prizes	\$	369,630,800	\$	375,372,700	
Administrative Expenses		78,803,400		79,893,600	
Less GPR Retailer Compensation		-8,000,000		<u>-40,000,000</u>	
	\$	440,434,200	\$	415,266,300	
Net Proceeds	\$	169,324,900	\$	204,045,700	
Total Available for Tax Relief					
Opening Balance	\$	6,677,500	\$	12,195,100	
Net Proceeds		169,324,900		204,045,700	
Interest Earnings		1,091,100		1,763,400	
	\$	177,093,500	\$	218,004,200	
Property Tax Relief	\$	164,898,400	\$	205,617,900	
Gross Closing Balance	\$	12,195,100	\$	12,386,300	
Reserve	\$	12,195,100	\$	12,386,300	
Net Balance		\$0		\$0	

SECTION 182. 20.005 (2) of the statutes is repealed and recreated to read:

14,100,000

20.005 (2) STATE BORROWING PROGRAM SUMMARY. The following schedule sets forth the state borrowing program summary: [See Figures 20.005 (2) (a) and (b) following]

Figure: 20.005 (2) (a)

Harbor assistance

SUMMARY OF BONDING AUTHORITY MODIFICATIONS 2017–19 FISCAL BIENNIUM

2017-19 FISCAL BIENNIUM						
Source and Purpose		Amount				
GENERAL OBLIGATIONS						
Administration						
Energy conservation projects	\$	20,000,000				
Agriculture, Trade and Consumer Protection						
Soil and water		7,000,000				
Building Commission						
Other public purposes (all agency projects)		186,168,000				
Housing state agencies		97,000,000				
La Crosse Center		5,000,000				
St. Ann Center for Intergenerational Care, Inc.; Bucyrus Campus		5,000,000				
Brown County Innovation Center		5,000,000				
Corrections						
Correctional facilities		44,333,000				
Environmental Improvement Fund						
Clean Water Fund		-40,460,000				
Safe Drinking Water Loan Program		5,800,000				
Health Services						
Mental health facilities		22,695,000				
Military Affairs						
Armories and military facilities		10,218,100				
•		, ,				
Natural Resources						
Dam safety projects		4,000,000				
Nonpoint source		6,150,000				
Urban nonpoint source cost-sharing		3,700,000				
SEG revenue supported facilities		5,805,800				
Transportation						
Freight rail		12,000,000				

2017 Assembly Bill 64	- 21 -	2017 Wisconsin Act 59
Source and Purpose		Amount
Southeast Wisconsin megaproj	ects (Foxconn legislation)	252,400,000
University of Wisconsin System		
Academic facilities		210,912,000
Self-amortizing facilities		31,502,300
Veterans Affairs		
Veterans facilities		4,332,600
Self-amoritizing facilities		8,046,400
Self-amortizing mortgage loan	S	

TOTAL General Obligation Bonds

\$ 647,403,200*

*Excludes \$1,500,000,000 of economic refunding bonds that would be authorized in the bill.

REVENUE OBLIGATIONS

Transportation

Transportation facilities, major highway projects and southeast Wisconsin megaprojects

\$ 123,900,000

GRAND TOTAL General and Revenue Obligation Bonds

\$ 771,303,200

Figure: 20.005 (2) (b)

GENERAL OBLIGATION DEBT SERVICE FISCAL YEARS 2017-18 AND 2018-19

STAT	UTE, A	GENCY AND PURPOSE	Source	201	7–18	201	8–19
20.11	15 Agri	culture, trade and consumer protectio	n, department d	of .			
(2)	(d)	Principal repayment and interest	GPR	\$	4,200	\$	4,500
(7)	(b)	Principal repayment and interest, conservation reserve enhancement	GPR		934,800		957,000
20.19	90 State	e fair park board					
(1)	(c)	Housing facilities principal repayment, interest and rebates	GPR		683,000		247,100
(1)	(d)	Principal repayment and interest	GPR	,	2,373,400	,	2,290,700
20.22	25 Edu	cational communications board					
(1)	(c)	Principal repayment and interest	GPR	,	2,692,200	,	2,540,400
20.24	45 Hist	orical society					
(1)	(e)	Principal repayment, interest, and rebates	GPR	4	4,601,200	4	4,892,300
20.23	50 Med	lical College of Wisconsin					
(1)	(c)	Principal repayment, interest, and rebates; biomedical research and technology incubator	GPR	;	3,209,000	,	3,043,200

STAT	TUTE, A	GENCY AND PURPOSE	Source	2017–18	2018-19
(1)	(e)	Principal repayment and interest	GPR	189,800	186,600
20.2	55 Publ	ic instruction, department of			
(1)	(d)	Principal repayment and interest	GPR	1,077,900	1,051,300
20.28	85 Univ	ersity of Wisconsin System			
(1)	(d)	Principal repayment and interest	GPR	223,168,600	218,577,800
20.32	20 Envi	ronmental improvement program			
(1)	(c)	Principal repayment and interest – clean water fund program	GPR	10,805,500	10,516,000
(2)	(c)	Principal repayment and interest – safe drinking water loan program	GPR	5,569,000	5,595,100
20.3	70 N	atural resources, department of			
(7)	(aa)	Resource acquisition and development – principal repayment and interest	GPR	70,715,700	74,329,900
(7)	(cb)	Principal repayment and interest – pollution abatement bonds	GPR	0	0
(7)	(cc)	Principal repayment and interest – combined sewer overflow; pollution abatement bonds	GPR	1,511,600	2,013,000
(7)	(cd)	Principal repayment and interest – municipal clean drinking water grants	GPR	177,500	176,200
(7)	(ea)	Administrative facilities – principal repayment and interest	GPR	696,100	693,800
20.3	95 Tran	sportation, department of			
(6)	(ae)	Principal repayment and interest, contingent funding of major highway rehabilitation projects, state funds	GPR	14,829,800	14,532,500
(6)	(af)	Principal repayment and interest, local roads for job preservation program, major highway and rehabilitation projects, southeast			
		megaprojects, state funds	GPR	111,754,000	102,091,500
		ections, department of			
(1)	(e)	Principal repayment and interest	GPR	72,056,200	66,480,100
(1)	(ec)	Prison industries principal, interest and rebates	GPR	0	0
(3)	(e)	Principal repayment and interest	GPR	3,856,500	3,519,900
	35 Heal	th services, department of			
(2)	(ee)	Principal repayment and interest	GPR	21,228,100	18,922,900
		ary affairs, department of			
(1)	(d)	Principal repayment and interest	GPR	6,856,300	6,448,200
		rans affairs, department of			
(1)	(f)	Principal repayment and interest	GPR	1,546,200	1,462,400

		GENCY AND PURPOSE	Source	2017–18	2018–19
		inistration, department of			
(4)	(es)	Principal, interest, and rebates; general purpose revenue – schools	GPR	911,400	1,000,600
(4)	(et)	Principal, interest, and rebates; general purpose revenue – public library boards	GPR	8,000	6,200
(5)	(c)	Principal repayment and interest; Black Point Estate	GPR	201,900	193,400
20.83	55 Misc	ellaneous appropriations			
(8)	(a)	Dental clinic and education facility; principal repayment, interest and rebates	GPR	2,272,600	2,281,200
20.8	67 Build	ling commission			
(1)	(a)	Principal repayment and interest; housing of state agencies	GPR	0	0
(1)	(b)	Principal repayment and interest; capitol and executive residence	GPR	10,717,900	9,479,600
(3)	(a)	Principal repayment and interest	GPR	9,365,300	20,187,700
(3)	(b)	Principal repayment and interest	GPR	1,643,300	1,434,000
(3)	(bb)	Principal repayment, interest and rebates; AIDS Network, Inc.	GPR	23,900	23,900
(3)	(bc)	Principal repayment, interest and rebates; Grand Opera House in Oshkosh	GPR	41,400	42,100
(3)	(bd)	Principal repayment, interest and rebates; Aldo Leopold climate change classroom and interactive laboratory	GPR	37,900	37,900
(3)	(be)	Principal repayment, interest and rebates; Bradley Center Sports and Entertainment Corporation	GPR	3,245,300	3,193,800
(3)	(bf)	Principal repayment, interest and rebates; AIDS Resource Center of			
(3)	(bg)	Wisconsin, Inc. Principal repayment, interest, and	GPR	63,700	63,700
		rebates; Madison Children's Museum	GPR	20,000	19,900
(3)	(bh)	Principal repayment, interest, and rebates; Myrick Hixon EcoPark, Inc.	GPR	40,000	40,000
(3)	(bj)	Principal repayment, interest and rebates; Lac du Flambeau Indian Tribal Cultural Center	GPR	19,000	18,800
(3)	(bL)	Principal repayment, interest and rebates; family justice center	GPR	698,100	814,000
(3)	(bm)	Principal repayment, interest, and rebates; HR Academy, Inc.	GPR	136,400	136,900
(3)	(bn)	Principal repayment, interest and rebates; Hmong cultural centers	GPR	21,400	16,800
		-		•	•

	UTE, AG	ENCY AND PURPOSE	Source	2017–18	2018–19
(3)	(bq)	Principal repayment, interest and rebates; children's research institute	GPR	969,500	930,100
(3)	(br)	Principal repayment, interest and rebates	GPR	90,300	88,900
(3)	(bt)	Principal repayment, interest and rebates; Wisconsin Agriculture Education Center, Inc.	GPR	62,500	263,200
(3)	(bu)	Principal repayment, interest and rebates; Civil War exhibit at the Kenosha Public Museums	GPR	42,400	33,000
(3)	(bv)	Principal repayment, interest, and rebates; Bond Health Center	GPR	73,100	74,200
(3)	(bw)	Principal repayment, interest, and rebates; Eau Claire Confluence Arts, Inc.	GPR	125,000	927,500
(3)	(bx)	Principal repayment, interest, and rebates; Carroll University	GPR	161,200	179,200
(3)	(cb)	Principal repayment, interest and rebates; Domestic Abuse Intervention Services, Inc.	GPR	41,000	40,700
(3)	(cd)	Principal repayment, interest, and rebates; K I Convention Center	GPR	138,700	137,600
(3)	(cf)	Principal repayment, interest, and rebates; Dane County; livestock facilities	GPR	681,400	675,300
(3)	(ch)	Principal repayment, interest, and rebates; Wisconsin Maritime Center of Excellence	GPR	185,600	557,600
(3)	(cj)	Principal repayment, interest, and rebates; Norskedalen Nature and Heritage Center	GPR	0	84,200
(3)	(cq)	Principal repayment, interest, and rebates; La Crosse Center	GPR	0	0
(3)	(cr)	Principal repayment, interest, and rebates; St. Ann Center for Intergenerational Care Bucyrus Campus	GPR	0	0
(3)	(cs)	Principal repayment, interest, and rebates; Brown County innovation center	GPR	0	0
(3)	(e)	Principal repayment, interest and rebates; parking ramp	GPR	0	0
тот	'AL Ger	neral Purpose Revenue Debt Service	OTK	\$ 592,574,800	\$ 583,554,400
20.19		ate fair park board			
(1)	(j)	State fair principal repayment, interest and rebates	PR	\$ 3,718,000	\$ 3,716,000

		SENCY AND PURPOSE	Source	2017–18	2018–19
20.22	25 E	ducational communications board			
(1)	(i)	Program revenue facilities; principal repayment, interest, and rebates	PR	13,700	13,600
20.24	45 Histo	rical society			
(1)	(j)	Self-amortizing facilities; principal repayment, interest, and rebates	PR	767,600	1,116,800
20.28	85 Unive	ersity of Wisconsin System			
(1)	(gj)	Self-amortizing facilities principal and interest	PR	141,717,700	146,584,800
20.37	70 Natu	ral resources, department of			
(7)	(ag)	Land acquisition – principal repayment and interest	PR	0	0
(7)	(cg)	Principal repayment and interest – nonpoint repayments	PR	0	0
20.41	10 Corre	ections, department of			
(1)	(ko)	Prison industries principal repayment, interest and rebates	PR	96,000	93,200
20.48	85 Veter	ans Affairs, department of			
(1)	(go)	Self-amortizing facilities; principal repayment and interest	PR	2,038,500	2,345,000
20.50	05 Admi	inistration, department of			
(4)	(ha)	Principal, interest, and rebates; program revenue – schools	PR	0	0
(4)	(hb)	Principal, interest, and rebates; program revenue – public library boards	PR	0	0
(5)	(g)	Principal repayment, interest and rebates; parking	PR	1,178,100	1,111,100
(5)	(kc)	Principal repayment, interest and rebates	PR	23,702,800	31,365,000
20.86	67 Build	ling commission			
(3)	(g)	Principal repayment, interest and rebates; program revenues	PR	0	0
(3)	(h)	Principal repayment, interest, and rebates	PR	0	0
(3)	(i)	Principal repayment, interest and rebates; capital equipment	PR	0	0
(3)	(kd)	Energy conservation construction projects; principal repayment, interest and rebates	PR	1,810,300	2,660,000
(3)	(km)	Aquaculture demonstration facility; principal repayment and interest	PR	<u>271,900</u>	265,800
TOT	'AL Pro	gram Revenue Debt Service		\$ 175,314,600	\$ 189,271,300

STAT	UTE, A	GENCY AND PURPOSE	Source	2017-18	2018–19
20.11	5 Agric	culture, trade and consumer protection	n, department of		
(7)	(s)	Principal repayment and interest; soil and water, environmental fund	SEG	\$ 4,234,900	\$ 4,569,000
20.32	20 Envi	ronmental improvement program			
(1)	(t)	Principal repayment and interest – clean water fund program bonds	SEG	8,000,000	8,000,000
20.37	70 Natu	ral resources, department of			
(7)	(aq)	Resource acquisition and development – principal repayment and interest	SEG	100	100
(7)	(ar)	Dam repair and removal – principal repayment and interest	SEG	522,900	475,700
(7)	(at)	Recreation development – principal repayment and interest	SEG	0	0
(7)	(au)	State forest acquisition and development – principal repayment and interest	SEG	13,500,000	13,500,000
(7)	(bq)	Principal repayment and interest – remedial action	SEG	2,688,200	2,642,400
(7)	(br)	Principal repayment and interest – contaminated sediment	SEG	1,940,800	2,012,100
(7)	(cq)	Principal repayment and interest – nonpoint source grants	SEG	6,211,500	5,796,800
(7)	(cr)	Principal repayment and interest – nonpoint source	SEG	2,329,400	2,492,700
(7)	(cs)	Principal repayment and interest – urban nonpoint source cost–sharing	SEG	3,280,000	3,464,800
(7)	(ct)	Principal repayment and interest – pollution abatement, environmental fund	SEG	6,146,900	7,047,000
(7)	(eq)	Administrative facilities – principal repayment and interest	SEG	5,815,600	6,062,900
(7)	(er)	Administrative facilities – principal repayment and interest; environmental fund	SEG	799,300	835,100
20.39	95 Tran	sportation, department of			
(6)	(aq)	Principal repayment and interest, transportation facilities, state highway rehabilitation, major highway projects, state funds	SEG	58,096,100	60,870,300
(6)	(ar)	Principal repayment and interest, buildings, state funds	SEG	28,700	29,000
(6)	(au)	Principal repayment and interest, southeast rehabilitation projects, southeast megaprojects, and high-cost bridge projects, state			
		funds	SEG	91,079,100	95,821,300

STATUTE, AGENCY AND PURPOSE		Source	2017–18	2018-19	
(6)	(av)	Principal repayment and interest, contingent funding of major highway and rehabilitation projects, state funds	SEG	6,085,700	17,957,100
20.48	85 Veter	ans affairs, department of			
(3)	(t)	Debt service	SEG	6,614,500	5,249,300
(4)	(qm)	Repayment of principal and interest	SEG	78,300	22,700
20.86	67 Build	ling commission			
(3)	(q)	Principal repayment and interest; segregated revenues	SEG	0	0
TOTAL Segregated Revenue Debt Service				\$ 217,452,000	\$ 236,848,300
GRA	ND TO	OTAL All Debt Service		\$ 985,341,400	\$1,009,674,000

SECTION 183. 20.005 (3) of the statutes is repealed and recreated to read:

20.005 (3) APPROPRIATIONS. The following schedule sets forth all annual, biennial, and sum certain continuing appropriations and anticipated expenditures from other appropriations for the programs and other purposes indicated. All appropriations are made from the general fund unless otherwise indicated. The letter abbreviations shown designating the type of appropriation apply to both fiscal years in the schedule unless otherwise indicated. [See Figure 20.005 (3) following]

Fi	gure: 20.005 (3)				
STATUT	E, AGENCY AND PURPOSE	Source	Түре	2017-2018	2018-2019
		Comme	rce		
20.115	Agriculture, Trade and Consumer F	Protection, D	epartment o	of	
(1)	FOOD SAFETY AND CONSUMER PROTECT	ΓΙΟΝ			
(a)	General program operations	GPR	A	-0-	-0-
	Food inspection	GPR	A	3,633,800	3,636,900
	Meat and poultry inspection	GPR	A	4,076,900	4,077,300
	Trade and consumer protection	GPR	A	1,688,100	1,691,400
	NET APPROPRIATION			9,398,800	9,405,600
(c)	Petroleum products; storage tank				
	inventory	GPR	A	-0-	-0-
(g)	Related services	PR	A	57,700	57,700
(gb)	Food, lodging, and recreation	PR	A	9,261,900	9,262,000
(gc)	Testing of petroleum products	PR	C	-0-	-0-
(gf)	Fruit and vegetable inspection	PR	C	894,200	894,400
(gh)	Public warehouse regulation	PR	A	87,900	87,900
(gm)	Dairy trade regulation	PR	A	127,100	127,200
(h)	Grain inspection and certification	PR	C	1,004,300	1,004,300
(hm)	Ozone-depleting refrigerants and				
	products regulation	PR	A	-0-	-0-
(i)	Sale of supplies	PR	A	10,400	10,400
(im)	Consumer protection; telephone				
	solicitor fees	PR	A	302,200	302,600
(ip)	Bisphenol A enforcement	PR	C	-0-	-0-
(j)	Weights and measures inspection	PR	A	1,657,700	1,657,700

17 Wi	isconsin Act 59	– 28 –		2017 Assembly Bill 64			
Statut	E, AGENCY AND PURPOSE	Source	ТүрЕ	2017-2018	2018-2019		
(jb)	Consumer protection, information,						
	and education	PR	A	147,800	147,800		
(jm)	Telecommunications utility trade						
	practices	PR	A	378,600	379,600		
(m)	Federal funds	PR-F	C	5,463,600	5,464,500		
(q)	Dairy, grain, and vegetable security	SEG	A	1,167,600	1,169,300		
(r)	Unfair sales act enforcement	SEG	A	252,200	253,000		
(s)	Weights and measures; petroleum						
	inspection fund	SEG	A	881,100	885,700		
(t)	Petroleum products; petroleum						
	inspection fund	SEG	A	4,501,200	4,504,700		
(u)	Recyclable and nonrecyclable						
	products regulation	SEG	A	-0-	-0-		
(v)	Agricultural producer security;		_	_	_		
	contingent financial backing	SEG	S	-0-	-0-		
(w)	Agricultural producer security;	ana	a	200,000	200.000		
(1)	payments	SEG	S	200,000	200,000		
(wb)	Agricultural producer security;						
	proceeds of contingent financial	CEC	С	0	0		
(1110)	backing	SEG	C	-0-	-0-		
(wc)	Agricultural producer security;						
	repayment of contingent financial backing	SEG	S	-0-	-0-		
	Dacking		GRAM TO		_0_		
	GENERAL PURPOSE REVENUE	(1) TKOC	JKAWI 10	9,398,800	9,405,600		
	PROGRAM REVENUE			19,393,400	19,396,100		
	FEDERAL			(5,463,600)	(5,464,500)		
	OTHER			(13,929,800)	(13,931,600)		
	SEGREGATED REVENUE			7,002,100	7,012,700		
	OTHER			(7,002,100)	(7,012,700)		
	TOTAL–ALL SOURCES			35,794,300	35,814,400		
				33,794,300	33,614,400		
(2)	ANIMAL HEALTH SERVICES	GPR	A	2,627,700	2,629,700		
(a)	General program operations						
(b)	Animal disease indemnities	GPR	S	108,600	108,600		
(c)	Financial assistance for paratuberculosis testing	GPR	A	-0-	-0-		
(d)	Principal repayment and interest	GPR	S	4,200	4,500		
	Livestock premises registration	GPR	A	250,000	250,000		
(e)	Related services	PR	A C	230,000 -0-	230,000 -0-		
(g)		PR PR		28,400	28,400		
(h)	Sale of supplies Inspection, testing and enforcement	PR PR	A C	28,400 664,800	28,400 576,800		
(ha)		ſΚ	C	004,800	370,800		
(j)	Dog licenses, rabies control, and related services	PR	C	427,000	427,300		
(im)		PR	C	323,900	280,000		
(jm)	Veterinary examining board Federal funds	PR-F	C	323,900	320,700		
(m)	Animal health inspection, testing and	LV-L	C	320,300	320,700		
(q)	enforcement	SEG	A	331,200	331,200		
(r)	Livestock premises registration —	SEC	Λ	331,200	331,400	Veto	
(r)	agrichemical management fund	SEG	A	100,000	100,000	In Pa	
	agriciiciincai management tunu				100,000	111 1 (
		(2) DD (0)	CDAM TO	TAIC			
	GENERAL PURPOSE REVENUE	(2) PROC	GRAM TO	ГALS 2,990,500	2,992,800		

2017 A	2017 Assembly Bill 64		-	2017	2017 Wisconsin Act 59		
Statu'	STATUTE, AGENCY AND PURPOSE		Түре	2017-2018	2018-2019		
	FEDERAL			(320,500)	(320,700)		
	OTHER			(1,444,100)	(1,312,500)		
	SEGREGATED REVENUE			431,200	431,200		
	OTHER			(431,200)	(431,200)		
	TOTAL-ALL SOURCES			5,186,300	5,057,200		
(3)	AGRICULTURAL DEVELOPMENT SERVICES						
(a)	General program operations	GPR	A	2,076,400	2,079,800		
(at)	Farm to school program						
	administration	GPR	A	66,400	66,400		
(g)	Related services	PR	A	-0-	-0-		
(h)	Loans for rural development	PR	C	58,700	58,700		
(i)	Marketing orders and agreements	PR	C	103,300	103,400		
(j)	Stray voltage program	PR	A	237,700	238,600		
(ja)	Agricultural development services						
	and materials	PR	C	103,400	103,500		
(jm)	Stray voltage program; rural electric						
	cooperatives	PR	A	19,700	19,700		
(L)	Something special from Wisconsin	D.D.		55 600	55 5 00		
	promotion	PR	A	57,600	57,700		
(m)	Federal funds	PR-F	С	1,290,700	1,291,500		
		(3) PROC	ìRAM T				
	GENERAL PURPOSE REVENUE			2,142,800	2,146,200		
	PROGRAM REVENUE			1,871,100	1,873,100		
	FEDERAL			(1,290,700)	(1,291,500)		
	OTHER			(580,400)	(581,600)		
	TOTAL-ALL SOURCES			4,013,900	4,019,300		
(4)	AGRICULTURAL ASSISTANCE						
(a)	Aid to Wisconsin livestock breeders	CDD		0	0		
()	association	GPR	A	-0-	-0-		
(am)	•	GPR	В	200,000	200,000		
(as)	Farm to school grants	GPR	A	-0-	-0-		
(b)	Aids to county and district fairs	GPR	A	431,400	431,400		
(c)	Agricultural investment aids	GPR	В	-0-	-0-		
(d)	Dairy industry promotion	GPR	A	200,000	200,000		
(dm)		GPR	A	200,000	200,000		
(e)	Aids to World Dairy Expo, Inc.	GPR	A	20,100	20,100		
(q)	Grants for agriculture in the	ara		02.000	02.000		
()	classroom program	SEG	A	93,900	93,900		
(r)	Agricultural investment aids,	CEC	D	0	0		
	agrichemical management fund	SEG	В	-0-	-0-		
	CENED AL DIDDOCE DEVENITE	(4) PROC	JKAWI I		1.051.500		
	GENERAL PURPOSE REVENUE SEGREGATED REVENUE			1,051,500 93,900	1,051,500		
	OTHER				93,900		
				(93,900)	(93,900)		
(7)	TOTAL-ALL SOURCES			1,145,400	1,145,400		
(7)	AGRICULTURAL RESOURCE MANAGEMENT		Δ.	756 500	759 200		
(a)	General program operations	GPR	A	756,500	758,300		
(b)	Principal repayment and interest, conservation reserve enhancement	CDD	S	024 900	057 000		
(-)		GPR	S	934,800	957,000		
(c)	Soil and water resource management program	GPR	C	3,027,200	3,027,200		
	L. Agram	J1 IX	_	3,021,200	3,027,200		

STATUTI	E, AGENCY AND PURPOSE	Source	Түре	2017-2018	2018–2019
(dm)	Farmland preservation planning				
	grants	GPR	A	210,000	210,000
(g)	Agricultural impact statements	PR	C	321,400	321,900
(ga)	Related services	PR	C	328,000	328,600
(gm)	Seed testing and labeling	PR	C	82,800	82,900
(h)	Fertilizer research assessments	PR	C	255,600	255,600
(ha)	Liming material research funds	PR	C	21,100	21,100
(i)	Agricultural conservation easements;				
	gifts and grants	PR	C	-0-	-0-
(ja)	Plant protection	PR	C	165,300	165,400
(k)	Agricultural resource management				
	services	PR-S	C	290,800	290,800
(m)	Federal funds	PR-F	C	1,233,300	1,233,500
(qc)	Plant protection; conservation fund	SEG	A	1,585,900	1,587,600
(qd)	Soil and water administration;				
	environmental fund	SEG	A	2,214,800	2,217,600
(qe)	Soil and water management; local				
	assistance	SEG	A	5,936,900	5,936,900
(qf)	Soil and water management; aids	SEG	A	3,325,000	3,325,000
(r)	General program operations;				
	agrichemical management	SEG	A	6,721,700	6,728,600
(s)	Principal repayment and interest; soil				
	and water, environmental fund	SEG	S	4,234,900	4,569,000
(tg)	Agricultural conservation easements	SEG	A	-0-	-0-
(tm)	Farmland preservation planning				
	grants, working lands fund	SEG	A	-0-	-0-
(ts)	Working lands programs	SEG	A	12,000	12,000
(va)	Clean sweep grants	SEG	A	750,000	750,000
(wm)	Agricultural chemical cleanup				
	reimbursement	SEG	C	900,000	900,000
		(7) PROC	GRAM 7	ΓOTALS	
(GENERAL PURPOSE REVENUE			4,928,500	4,952,500
F	PROGRAM REVENUE			2,698,300	2,699,800
	FEDERAL			(1,233,300)	(1,233,500)
	OTHER			(1,174,200)	(1,175,500)
	SERVICE			(290,800)	(290,800)
5	SEGREGATED REVENUE			25,681,200	26,026,700
	OTHER			(25,681,200)	(26,026,700)
7	TOTAL-ALL SOURCES			33,308,000	33,679,000
(8)	CENTRAL ADMINISTRATIVE SERVICES				
(a)	General program operations	GPR	A	5,940,600	5,962,700
(g)	Gifts and grants	PR	C	801,800	802,000
(gm)	Enforcement cost recovery	PR	A	4,600	4,600
(h)	Sale of material and supplies	PR	C	9,600	9,600
(ha)	General laboratory related services	PR	C	44,200	44,200
(hm)	Restitution	PR	C	-0-	-0-
(i)	Related services	PR	A	15,600	15,600
(j)	Electronic processing	PR	C	-0-	-0-
(jm)	Telephone solicitation regulation	PR	C	697,700	699,000
(k)	Computer system equipment, staff		-	5,7,700	0,7,000
(-)	and services	PR-S	A	2,784,400	2,835,700
				* *	

2017 A	ssembly Bill 64	- 31 -	_	2017 W	isconsin Act 59
STATU'	ГЕ, AGENCY AND PURPOSE	Source	Түре	2017-2018	2018-2019
(kL)	Central services	PR-S	C	653,300	654,000
(km)	General laboratory services	PR-S	В	3,183,900	3,291,600
(ks)	State services	PR-S	C	188,300	188,400
(m)	Federal funds	PR-F	C	232,700	234,000
(pz)	Indirect cost reimbursements	PR-F	C	2,011,100	2,012,700
		(8) PRO	GRAM TO	TALS	
	GENERAL PURPOSE REVENUE			5,940,600	5,962,700
	PROGRAM REVENUE			10,627,200	10,791,400
	FEDERAL			(2,243,800)	(2,246,700)
	OTHER			(1,573,500)	(1,575,000)
	SERVICE			(6,809,900)	(6,969,700)
	TOTAL-ALL SOURCES			16,567,800	16,754,100
	20).115 DEPA	RTMENT	TOTALS	
	GENERAL PURPOSE REVENUE			26,452,700	26,511,300
	PROGRAM REVENUE			36,354,600	36,393,600
	FEDERAL			(10,551,900)	(10,556,900)
	OTHER			(18,702,000)	(18,576,200)
	SERVICE			(7,100,700)	(7,260,500)
	SEGREGATED REVENUE			33,208,400	33,564,500
	OTHER			(33,208,400)	(33,564,500)
	TOTAL-ALL SOURCES			96,015,700	96,469,400
20.144	Financial Institutions, Department of				
(1)	SUPERVISION OF FINANCIAL INSTITUTIONS		S REGULATIO	ON AND OTHER FUNCTION	NS
(a)	Losses on public deposits	GPR	S	-0-	-0-
(g)	General program operations	PR	A	16,608,100	16,823,000
(h)	Gifts, grants, settlements, and				
. ,	publications	PR	C	58,500	58,500
(i)	Investor education and training fund	PR	A	84,500	84,500
(j)	Payday loan database and financial literacy	PR	С	900,000	900,000
(m)	Credit union examinations, federal	110	C	700,000	, , , , , , , , , , , , , , , , , , ,
(111)	funds	PR-F	C	-0-	-0-
(u)	State deposit fund	SEG	S	-0-	-0-
. ,	•	(1) PRO0	GRAM TO	TALS	
	GENERAL PURPOSE REVENUE	. ,		-0-	-0-
	PROGRAM REVENUE			17,651,100	17,866,000
	FEDERAL			(-0-)	(-0-)
	OTHER			(17,651,100)	(17,866,000)
	SEGREGATED REVENUE			-0-	-0-
	OTHER			(-0-)	(-0-)
	TOTAL-ALL SOURCES			17,651,100	17,866,000
(3)	College Tuition and Expenses and Co	LLEGE SAVI	NGS PROGRA		,,
(tb)	Payment of qualified higher education expenses and refunds;				
	college tuition and expenses program	SEG	S	-0-	-0-
(td)	Administrative expenses; college				
	tuition and expenses program	SEG	A	118,300	118,300
(tf)	Payment of qualified higher				
	education expenses and refunds;	SEC	C	0	0
	college savings program trust fund	SEG	S	-0-	-0-

2017 W	2017 Wisconsin Act 59		_	2017 A	Assembly Bill 64
STATUT	TE, AGENCY AND PURPOSE	Source	Түре	2017-2018	2018-2019
(th)	Administrative expenses; college savings program trust fund	SEG	A	640,700	640,900
(tj)	Payment of qualified higher education expenses and refunds; college savings program bank				
(tL)	deposit trust fund Administrative expenses; college savings program bank deposit trust	SEG	S	-0-	-0-
(tn)	fund Payment of qualified higher	SEG	A	-0-	-0-
,	education expenses and refunds; college savings program credit union deposit trust fund	SEG	S	-0-	-0-
(tp)	Administrative expenses; college savings program credit union deposit			·	Ü
	trust fund	SEG	A	-0-	-0-
		(3) PROC	GRAM TO		
	SEGREGATED REVENUE			759,000	759,200
	OTHER			(759,000)	(759,200)
	TOTAL-ALL SOURCES			759,000	759,200
).144 DEPA	RTMENT		
	GENERAL PURPOSE REVENUE			-0-	-0-
	PROGRAM REVENUE			17,651,100	17,866,000
	FEDERAL			(-0-)	(-0-)
	OTHER			(17,651,100)	(17,866,000)
	SEGREGATED REVENUE			759,000	759,200
	OTHER			(759,000)	(759,200)
	TOTAL-ALL SOURCES			18,410,100	18,625,200
	Insurance, Office of the Commissione	er of			
(1)	SUPERVISION OF THE INSURANCE INDUSTR				
(g)	General program operations	PR	A	18,499,700	18,679,500
(gm)	•	PR	C	-0-	-0-
(h)	Holding company restructuring expenses	PR	С	-0-	-0-
(m)	Federal funds	PR-F	C	601,000	601,000
(111)	reactar rands		GRAM TO		001,000
	PROGRAM REVENUE	(1) 1100		19,100,700	19,280,500
	FEDERAL			(601,000)	(601,000)
	OTHER			(18,499,700)	(18,679,500)
	TOTAL-ALL SOURCES			19,100,700	19,280,500
(2)	INJURED PATIENTS AND FAMILIES COMPEN	SATION FUN	D	17,100,700	17,200,500
(a)	Supplement for claims payable	GPR	S	-0-	-0-
(q)	Interest earned on future medical				_
	expenses	SEG	S	-0- 1 220 000	<u>-()-</u>
(u)	Administration	SEG	A	1,228,000	1,228,600
(um)	Peer review council	SEG	A	143,100	143,200
(v)	Specified responsibilities, investment board payments, and future medical				
	expenses	SEG	C	54,150,400	54,150,400
		(2) PROC	GRAM TO	DTALS	
	GENERAL PURPOSE REVENUE			-0-	-0-
	SEGREGATED REVENUE			55,521,500	55,522,200

2017 A	2017 Assembly Bill 64		-	2017 Wisconsin Act 59	
STATU	TE, AGENCY AND PURPOSE	SOURCE	Түре	2017-2018	2018-2019
	OTHER			(55,521,500)	(55,522,200)
	TOTAL-ALL SOURCES			55,521,500	55,522,200
(3)	LOCAL GOVERNMENT PROPERTY INSURAN	NCE FUND			
(u) (v)	Administration Specified payments, fire dues and	SEG	A	1,426,700	1,426,800
(v)	reinsurance	SEG	С	29,660,200	29,660,200
	Temodranee		GRAM TO		27,000,200
	SEGREGATED REVENUE	(3) 11(0)		31,086,900	31,087,000
	OTHER			(31,086,900)	(31,087,000)
	TOTAL-ALL SOURCES			31,086,900	31,087,000
(4)	STATE LIFE INSURANCE FUND			,,	,,
(u)	Administration	SEG	A	640,400	640,800
(v)	Specified payments and losses	SEG	C	4,493,000	4,493,000
(.)	specifica payments and resses		GRAM TO	· · · · · · · · · · · · · · · · · · ·	.,.,,,,,,,
	SEGREGATED REVENUE	(1)		5,133,400	5,133,800
	OTHER			(5,133,400)	(5,133,800)
	TOTAL-ALL SOURCES			5,133,400	5,133,800
		0.145 DEPA	ARTMENT '		2,122,000
	GENERAL PURPOSE REVENUE	011 10 2 211		-0-	-0-
	PROGRAM REVENUE			19,100,700	19,280,500
	FEDERAL			(601,000)	(601,000)
	OTHER			(18,499,700)	(18,679,500)
	SEGREGATED REVENUE			91,741,800	91,743,000
	OTHER			(91,741,800)	(91,743,000)
	TOTAL-ALL SOURCES			110,842,500	111,023,500
20.155	Public Service Commission				
(1)	REGULATION OF PUBLIC UTILITIES				
(g)	Utility regulation	PR	A	14,446,600	14,130,100
(h)	Holding company and nonutility				
	affiliate regulation	PR	C	703,100	703,100
(i)	Relay service	PR	A	2,879,000	2,879,100
(j)	Intervenor financing and grants	PR	В	742,500	742,500
(L)	Stray voltage program	PR	A	305,100	305,100
(Lb)	Gifts for stray voltage program	PR	C	-0-	-0-
(Lm)		PR	C	-0-	-0-
(m)	Federal funds	PR-F	C	2,633,800	2,634,000
(n)	Indirect costs reimbursement	PR-F	C	50,000	50,000
(q)	Universal telecommunications				
	service; broadband service	SEG	A	5,940,000	5,940,000
(r)	Nuclear waste escrow fund	SEG	S	-0-	-0-
		(1) PRO0	GRAM TO		
	PROGRAM REVENUE			21,760,100	21,443,900
	FEDERAL			(2,683,800)	(2,684,000)
	OTHER			(19,076,300)	(18,759,900)
	SEGREGATED REVENUE			5,940,000	5,940,000
	OTHER TOTAL ALL SOURCES			(5,940,000)	(5,940,000)
(2)	TOTAL-ALL SOURCES			27,700,100	27,383,900
(2)	OFFICE OF THE COMMISSIONER OF RAILR	OADS			
(g)	Railroad and water carrier regulation	DD	٨	558,700	550,000
	and general program operations	PR	A	330,700	559,000

2017 Wisconsin Act 59		- 34 -		2017 A	2017 Assembly Bill 64		
STATU'	TE, AGENCY AND PURPOSE	Source	Түре	2017-2018	2018-2019		
(m)	Railroad and water carrier regulation; federal funds	PR-F	С	-0-	-0-		
	DDOCD AM DEVENUE	(2) PROC	GRAM T		770 000		
	PROGRAM REVENUE			558,700	559,000		
	FEDERAL			(-0-)	(-0-)		
	OTHER TOTAL-ALL SOURCES			(558,700) 558,700	(559,000) 559,000		
(3)	AFFILIATED GRANT PROGRAMS			338,700	339,000		
(r)	Broadband expansion grants; transfer funding	SEG-S	С	14,000,000	-0-		
(rm)	_	SEG	C	-0-	-0-		
(s)	Energy efficiency and renewable	DEG	C	V	O .		
(5)	resource programs	SEG	A	408,300	408,300		
(t)	Police and fire protection fee			,	,		
()	administration	SEG	A	166,600	166,600		
		(3) PROC	GRAM T	OTALS			
	SEGREGATED REVENUE			14,574,900	574,900		
	OTHER			(574,900)	(574,900)		
	SERVICE			(14,000,000)	(-0-)		
	TOTAL-ALL SOURCES			14,574,900	574,900		
	20).155 DEPA	RTMEN'	T TOTALS			
	PROGRAM REVENUE			22,318,800	22,002,900		
	FEDERAL			(2,683,800)	(2,684,000)		
	OTHER			(19,635,000)	(19,318,900)		
	SEGREGATED REVENUE			20,514,900	6,514,900		
	OTHER			(6,514,900)	(6,514,900)		
	SERVICE			(14,000,000)	(-0-)		
	TOTAL-ALL SOURCES			42,833,700	28,517,800		
	Safety and Professional Services, Dep	artment of	•				
(1)	Professional regulation and adminis	STRATIVE SE	RVICES				
(a)	General program operations –	~~~					
	executive and administrative services	GPR	A	-0-	-0-		
(g)	General program operations	PR	A	10,291,700	10,344,700		
(gc)	Chiropractic examination	PR	C	-0-	-0-		
(gm)		DD	0	112 000	112 000		
(1-)	reimbursement	PR	C	113,000	113,000		
(h)	Technical assistance; nonstate agencies and organizations	PR	C	-0-	-0-		
(ha)	General program operations; medical	rĸ	C	-0-	-0-		
(hg)	examining board; interstate medical						
	licensure compact; prescription drug						
	monitoring program	PR	В	2,354,800	2,446,800		
(i)	Examinations; general program			, ,	, ,		
· · · · · · · · · · · · · · · · · · ·	operations	PR	C	1,382,700	1,382,700		
(im)	Boxing and unarmed combat sports;						
	enforcement	PR	C	-0-	-0-		
(jm)	Nursing workforce survey						
	administration	PR	В	9,000	9,000		
(jr)	Proprietary school programs	PR	A	318,400	645,400		
(jt)	Student protection	PR	C	28,300	56,600		
(jv)	Closed schools; preservation of						
	student records	PR	C	6,000	12,100		

2017 Assembly Bill 64		- 35 -		2017	2017 Wisconsin Act 59	
STATUT	TE, AGENCY AND PURPOSE	Source	Түре	2017-2018	2018-2019	
(k)	Technical assistance; state agencies	PR-S	C	-0-	-0-	
(ka)	Sale of materials and services – local assistance	PR-S	C	-0-	-0-	
(kb)	Sale of materials and services –	111 5	Ü	v	v	
()	individuals and organizations	PR-S	C	-0-	-0-	
(kc)	Sale of materials or services	PR-S	C	35,600	35,600	
(ke)	Transfer of unappropriated balances	PR-S	C	-0-	-0-	
(m)	Federal funds	PR-F	C	-0-	-0-	
(n)	Federal aid, local assistance	PR-F	C	-0-	-0-	
(o)	Federal aid, individuals and					
	organizations	PR-F	C	-0-	-0-	
(pz)	Indirect cost reimbursements	PR-F	C	2,700	2,700	
(s)	Wholesale drug distributor bonding	SEG	C	-0-	-0-	
		(1) PRO0	GRAM 7	ΓOTALS		
	GENERAL PURPOSE REVENUE			-0-	-0-	
	PROGRAM REVENUE			14,542,200	15,048,600	
	FEDERAL			(2,700)	(2,700)	
	OTHER			(14,503,900)	(15,010,300)	
	SERVICE			(35,600)	(35,600)	
	SEGREGATED REVENUE			-0-	-0-	
	OTHER			(-0-)	(-0-)	
	TOTAL-ALL SOURCES			14,542,200	15,048,600	
(2)	REGULATION OF INDUSTRY, SAFETY AND E	BUILDINGS				
(a)	General program operations	GPR	Α	-0-	-0-	
(g)	Gifts and grants	PR	C	18,000	18,000	
(ga)	Publications and seminars	PR	C	21,000	21,000	
(gb)	Local agreements	PR	C	-0-	-0-	
(h)	Local energy resource system fees	PR	Α	-0-	-0-	
(j)	Safety and building operations	PR	Α	14,001,600	14,059,400	
(ka)	Interagency agreements	PR-S	C	135,000	135,000	
(kd)	Administrative services	PR-S	A	2,282,800	2,285,600	
(ke)	Private on–site wastewater treatment system replacement and					
	rehabilitation	PR	C	840,000	840,000	
(ks)	Data processing	PR-S	C	-0-	-0-	
(L)	Fire dues distribution	PR	C	20,310,000	20,530,000	
(La)	Fire prevention and fire dues	D.D.		72 0 (00	53 0 (00	
()	administration	PR	A	728,600	728,600	
(m)	Federal funds	PR-F	C	471,900	471,900	
(ma)	Federal aid – program administration	PR–F	С	-0-	-0-	
(q)	Groundwater – standards;	SEG	A	-0-	-0-	
	implementation	(2) PROC			_0_	
	GENERAL PURPOSE REVENUE	(2) TROC	JIXANI	-0-	-0-	
	PROGRAM REVENUE			38,808,900	39,089,500	
	FEDERAL			(471,900)	(471,900)	
	OTHER			(35,919,200)	(36,197,000)	
	SERVICE			(2,417,800)	(2,420,600)	
	SERVICE SEGREGATED REVENUE			(2,417,800) -0-	(2,420,000)	
	OTHER			(-0-)	(-0-)	
	TOTAL-ALL SOURCES			38,808,900	39,089,500	
	TO ITAL SOURCES			30,000,300	37,009,300	

STATU	TE, AGENCY AND PURPOSE	Source 20.165 DEI		2017–2018 T TOTALS	2018–2019
	GENERAL PURPOSE REVENUE			-0-	-0-
	PROGRAM REVENUE			53,351,100	54,138,100
	FEDERAL			(474,600)	(474,600)
	OTHER			(50,423,100)	(51,207,300)
	SERVICE			(2,453,400)	(2,456,200)
	SEGREGATED REVENUE			-0-	-0-
	OTHER			(-0-)	(-0-)
	TOTAL-ALL SOURCES			53,351,100	54,138,100
20 190	State Fair Park Board			33,331,100	31,130,100
(1)	STATE FAIR PARK				
(c)	Housing facilities principal				
(C)	repayment, interest and rebates	GPR	S	683,000	247,100
(d)	Principal repayment and interest	GPR	S	2,373,400	2,290,700
(h)	State fair operations	PR	C	16,875,200	16,875,200
(i)	State fair capital expenses	PR	C	180,000	180,000
(i) (j)	State fair principal repayment,	1 IX	C	100,000	100,000
())	interest and rebates	PR	S	3,718,000	3,716,000
(jm)	Gifts and grants	PR	C	-0-	-0-
(m)	Federal funds	PR-F	C	-0-	_0_ _0_
(111)	rederal fullds		OGRAM T		-0-
	GENERAL PURPOSE REVENUE	(1) PKC	JUKAM I		2 527 900
				3,056,400	2,537,800
	PROGRAM REVENUE			20,773,200	20,771,200
	FEDERAL			(-0-)	(-0-)
	OTHER			(20,773,200)	(20,771,200)
	TOTAL-ALL SOURCES	20 100 DEI		23,829,600	23,309,000
	CELVED AT DANDS OF DELICITIES	20.190 DEI	PARTMEN		2 525 000
	GENERAL PURPOSE REVENUE			3,056,400	2,537,800
	PROGRAM REVENUE			20,773,200	20,771,200
	FEDERAL			(-0-)	(-0-)
	OTHER			(20,773,200)	(20,771,200)
	TOTAL-ALL SOURCES			23,829,600	23,309,000
	Wisconsin Economic Development	_	n		
(1)	PROMOTION OF ECONOMIC DEVELOPM				
(a)	Operations and programs	GPR	S	1,519,500	16,512,500
(m)	Federal aids; programs	PR–F	C	-0-	-0-
(r)	Economic development fund;				
	operations and programs	SEG	C	32,731,200	24,038,200
(s)	Brownfield site assessment grants	SEG	В	1,000,000	1,000,000
		(1) PRO	OGRAM T		
	GENERAL PURPOSE REVENUE			1,519,500	16,512,500
	PROGRAM REVENUE			-0-	-0-
	FEDERAL			(-0-)	(-0-)
	SEGREGATED REVENUE			33,731,200	25,038,200
	OTHER			(33,731,200)	(25,038,200)
	TOTAL-ALL SOURCES			35,250,700	41,550,700
		20.192 DEF	PARTMEN	T TOTALS	
	GENERAL PURPOSE REVENUE			1,519,500	16,512,500
	PROGRAM REVENUE			-0-	-0-
	FEDERAL			(-0-)	(-0-)
	SEGREGATED REVENUE			33,731,200	25,038,200
				•	•

017 Assembly Bill 64		- 37 -	_	2017 Wisconsin Act 59		
STATU	ΓΕ, AGENCY AND PURPOSE	Source	Түре	2017-2018	2018-2019	
	OTHER			(33,731,200)	(25,038,200)	
	TOTAL-ALL SOURCES			35,250,700	41,550,700	
			ommerce			
		UNCTION	AL ARE	A TOTALS		
	GENERAL PURPOSE REVENUE			31,028,600	45,561,600	
	PROGRAM REVENUE			169,549,500	170,452,300	
	FEDERAL			(14,311,300)	(14,316,500)	
	OTHER			(145,684,100)	(146,419,100)	
	SERVICE			(9,554,100)	(9,716,700)	
	SEGREGATED REVENUE			179,955,300	157,619,800	
	FEDERAL			(-0-)	(-0-)	
	OTHER			(165,955,300)	(157,619,800)	
	SERVICE			(14,000,000)	(-0-)	
	LOCAL			(-0-)	(-0-)	
	TOTAL-ALL SOURCES			380,533,400	373,633,700	
		Educat	ion			
20.220	Wisconsin Artistic Endowment Found	dation				
(1)	SUPPORT OF THE ARTS					
(a)	Education and marketing	GPR	C	-0-	-0-	
(q)	General program operations	SEG	A	-0-	-0-	
(r)	Support of the arts	SEG	C	-0-	-0-	
		(1) PRO	GRAM 7	ΓOTALS		
	GENERAL PURPOSE REVENUE			-0-	-0-	
	SEGREGATED REVENUE			-0-	-0-	
	OTHER			(-0-)	(-0-)	
	TOTAL-ALL SOURCES			-0-	-0-	
).220 DEPA	ARTMEN	IT TOTALS		
	GENERAL PURPOSE REVENUE			-0-	-0-	
	SEGREGATED REVENUE			-0-	-0-	
	OTHER			(-0-)	(-0-)	
	TOTAL-ALL SOURCES			-0-	-0-	
	Educational Communications Board					
(1)	INSTRUCTIONAL TECHNOLOGY					
(a)	General program operations	GPR	A	2,874,500	2,877,700	
(b)	Energy costs; energy-related	CDD		022 200	0.52.200	
()	assessments	GPR	A	932,200	953,300	
(c)	Principal repayment and interest	GPR	S	2,692,200	2,540,400	
(eg)	Transmitter construction	GPR	C	-0-	-0-	
(er)	Transmitter operation	GPR	A	16,000	16,000	
(g)	Gifts, grants, contracts, leases, instructional material, and copyrights	PR	C	12,342,800	12,970,800	
(i)	Program revenue facilities; principal repayment, interest, and rebates	PR	S	13,700	13,600	
(k)	Funds received from other state agencies	PR-S	C	-0-	-0-	
(kb)	Emergency weather warning system					
	operation	PR-S	A	134,700	134,700	
/ \	Federal grants	PR-F	C	-0-	-0-	
(m)	1 Courtai Sianto					
(m)	GENERAL PURPOSE REVENUE	(1) PRO	GRAM 7	ГОТАLS 6,514,900	6,387,400	

	E, AGENCY AND PURPOSE PROGRAM REVENUE	Source	ТүрЕ	2017–2018 12,491,200	2018–2019 13,119,100
1	FEDERAL			(-0-)	(-0-)
	OTHER			(12,356,500)	(12,984,400)
	SERVICE				
-				(134,700)	(134,700)
-	TOTAL-ALL SOURCES	0 225 DED	DTMEN	19,006,100	19,506,500
,		0.225 DEPA	AKIMEN.		(207 400
	GENERAL PURPOSE REVENUE			6,514,900	6,387,400
i	PROGRAM REVENUE			12,491,200	13,119,100
	FEDERAL			(-0-)	(-0-)
	OTHER			(12,356,500)	(12,984,400)
_	SERVICE			(134,700)	(134,700)
	ГОТAL-ALL SOURCES			19,006,100	19,506,500
	Higher Educational Aids Board				
(1)	STUDENT SUPPORT ACTIVITIES				
(b)	Wisconsin grants; private, nonprofit		_		
	college students	GPR	В	27,856,800	28,504,600
(cg)	Nursing student loans	GPR	A	-0-	-0-
(cm)	Nursing student loan program	GPR	A	445,500	445,500
(cr)	Minority teacher loans	GPR	A	134,500	134,500
(ct)	Teacher loan program	GPR	A	147,200	147,200
(cu)	School leadership loan program	GPR	C	500,000	-0-
(cx)	Loan program for teachers and				
	orientation and mobility instructors	~~~		00.000	00.000
	of visually impaired pupils	GPR	A	99,000	99,000
(d)	Dental education contract	GPR	A	1,733,000	1,733,000
(e)	Minnesota-Wisconsin student	CDD		7.12 0.000	7.12 0.000
(0.)	reciprocity agreement	GPR	S	7,130,000	7,130,000
(fc)	Independent student grants program	GPR	В	-0-	-0-
(fd)	Talent incentive grants	GPR	В	4,458,800	4,458,800
(fe)	Wisconsin grants; University of	CDD	D	60 407 500	(1.004.100
(00	Wisconsin System students	GPR	В	60,487,500	61,894,100
(ff)	Wisconsin grants; technical college	CDD	D	22.506.400	22 071 700
(C.)	students	GPR	В	22,506,400	22,971,700
(fg)	Minority undergraduate retention	CDD	D	910 000	910 000
(£;)	grants program	GPR	В	819,000	819,000
(fj)	Handicapped student grants	GPR	В	122,600	122,600 2,930,000
(fm)	Wisconsin covenant scholars grants	GPR	S	5,040,000	2,930,000
(fp)	Primary care and psychiatry shortage	GPR	C	-0-	-0-
(f)	grant program Technical excellence higher	UFK	C	-0-	-0-
(fw)	education scholarships	GPR	S	832,000	890,000
(fy)	Academic excellence higher	OLK	S	652,000	890,000
(fy)	education scholarships	GPR	S	2,964,000	2,964,000
(fz)	Remission of fees and	OIK	5	2,704,000	2,704,000
(IL)	reimbursement for veterans and				
	dependents	GPR	В	6,496,700	6,496,700
(g)	Student loans	PR	A	-0-	-0-
(gg)	Nursing student loan repayments	PR	C	-0-	-0-
(gm)	Indian student assistance;		Č	v	9
(5111)	contributions	PR	C	-0-	-0-
(i)	Gifts and grants	PR	C	-0-	-0-
\ - /			-	~	3

)17 A	ssembly Bill 64	- 39 -	_	2017 W	isconsin Act 5
Statu	TE, AGENCY AND PURPOSE	Source	Түре	2017-2018	2018-2019
(k)	Indian student assistance	PR-S	В	779,700	779,700
(kc)	Tribal college payments	PR-S	A	405,000	405,000
(km)					
	students	PR-S	В	470,900	481,800
(no)	Federal aid; aids to individuals and				
	organizations	PR-F	C	150,000	150,000
		(1) PROC	GRAM TO		
	GENERAL PURPOSE REVENUE			141,773,000	141,740,700
	PROGRAM REVENUE			1,805,600	1,816,500
	FEDERAL			(150,000)	(150,000)
	OTHER			(-0-)	(-0-)
	SERVICE			(1,655,600)	(1,666,500)
	TOTAL-ALL SOURCES			143,578,600	143,557,200
(2)	Administration				
(aa)	General program operations	GPR	A	951,000	954,500
(bb)	Student loan interest, loans sold or		_		
	conveyed	GPR	S	-0-	-0-
(bc)	Write-off of uncollectible student	ar-		_	
	loans	GPR	A	-0-	-0-
(bd)	Purchase of defective student loans	GPR	S	-0-	-0-
(ga)	Student interest payments	PR	C	900	900
(gb)	Student interest payments, loans sold	DD	C	0	0
<i>(</i> ')	or conveyed	PR	C	-0-	-0-
(ia)	Student loans; collection and administration	PR	С	-0-	-0-
(io)	Write-off of defaulted student loans	PR	A	-0- -0-	_0- _0-
(ja) (n)	Federal aid; state operations	PR–F	C	-0-	_0_ _0_
	Student loan revenue obligation	I IX-I	C	-0-	_0_
(qa)	repayment	SEG	C	-0-	-0-
	repayment		GRAM TOT		Ü
	GENERAL PURPOSE REVENUE	(2) 1100		951,000	954,500
	PROGRAM REVENUE			900	900
	FEDERAL			(-0-)	(-0-)
	OTHER			(900)	(900)
	SEGREGATED REVENUE			-0-	-0-
	OTHER			(-0-)	(-0-)
	TOTAL-ALL SOURCES			951,900	955,400
(3)	DISTANCE LEARNING AUTHORIZATION BO.	ARD		,	,
(g)	General operations	GPR	C	-0-	-0-
ν.Ον	1	(3) PROC	GRAM TO		
	GENERAL PURPOSE REVENUE			-0-	-0-
	TOTAL-ALL SOURCES			-0-	-0-
	20).235 DEPA	RTMENT 7	ΓOTALS	
	GENERAL PURPOSE REVENUE			142,724,000	142,695,200
	PROGRAM REVENUE			1,806,500	1,817,400
	FEDERAL			(150,000)	(150,000)
	OTHER			(900)	(900)
	SERVICE			(1,655,600)	(1,666,500)
	SEGREGATED REVENUE			-0-	-0-
	OTHER			(-0-)	(-0-)
	OTTIER			(-0-)	(0)

17 Wi	isconsin Act 59	- 40 -	_	2017 Assembly Bill 64		
	TE, AGENCY AND PURPOSE Historical Society	Source	Түре	2017–2018	2018–2019	
(1)	HISTORY SERVICES					T 7 . T
(a)	Concret program operations	GPR	A	13,021,600	14,486,100 14,442,100	Vetoed In Part
(a) (b)	General program operations Wisconsin Black Historical Society	GPK	А	13,021,000	14,442,100	111 1 41 0
(0)	and Museum	GPR	A	84,500	84,500	
(c)	Energy costs; energy-related					
	assessments	GPR	A	728,700	740,600	
(e)	Principal repayment, interest, and rebates	GPR	S	4,601,200	4,892,300	
(h)	Gifts, grants, and membership sales	PR	C	809,000	809,000	
(ii) (j)	Self–amortizing facilities; principal	110	C	007,000	002,000	
07	repayment, interest and rebates	PR	S	767,600	1,116,800	
(k)	Storage facility	PR-S	В	201,100	201,100	
(km)	Northern Great Lakes Center	PR-S	A	217,100	217,100	
(ks)	General program operations – service		_			
<i>a</i> \	funds	PR-S	C	1,575,200	1,576,100	
(kw)	Records management — service funds	PR-S	С	260,700	260,700	
(m)	General program operations; federal	1 K-3	C	200,700	200,700	
(111)	funds	PR-F	C	1,285,300	1,237,500	
(n)	Federal aids	PR-F	C	-0-	-0-	
(pz)	Indirect cost reimbursements	PR-F	C	101,200	101,200	
(q)	Endowment	SEG	C	570,200	570,200	
(r)	History preservation partnership trust					
	fund	SEG	C	4,055,900	4,055,900	
(y)	Northern great lakes center;	SEG	٨	51,000	51,000	
	interpretive programming		A GRAM TO	51,900	51,900	
	GENERAL PURPOSE REVENUE	(1) TKO	JKAWI 10.	18,436,000	20,203,500	
	PROGRAM REVENUE			5,217,200	5,519,500	
	FEDERAL			(1,386,500)	(1,338,700)	
	OTHER			(1,576,600)	(1,925,800)	
	SERVICE			(2,254,100)	(2,255,000)	
	SEGREGATED REVENUE			4,678,000	4,678,000	
	OTHER			(4,678,000)	(4,678,000)	
	TOTAL-ALL SOURCES			28,331,200	30,401,000	
		0.245 DEPA	ARTMENT T			
	GENERAL PURPOSE REVENUE			18,436,000	20,203,500	
	PROGRAM REVENUE			5,217,200	5,519,500	
	FEDERAL			(1,386,500)	(1,338,700)	
	OTHER			(1,576,600)	(1,925,800)	
	SERVICE SECREGATED DEVENIUE			(2,254,100)	(2,255,000)	
	SEGREGATED REVENUE OTHER			4,678,000 (4,678,000)	4,678,000 (4,678,000)	
	TOTAL-ALL SOURCES			28,331,200	30,401,000	
	Medical College of Wisconsin			20,331,200	50,701,000	
	_					
(1)	TRAINING OF HEALTH PERSONNEL					
(1) (a)	Training of Health Personnel Medical student tuition assistance	GPR	A	1,926,600	1,926,600	

17 A	ssembly Bill 64	- 41 -	_	2017 Wisconsin Act 59		
TATU	TE, AGENCY AND PURPOSE	Source	Түре	2017-2018	2018-2019	
(c)	Principal repayment, interest, and rebates; biomedical research and					
	technology incubator	GPR	S	3,209,000	3,043,200	
(e)	Principal repayment and interest	GPR	S	189,800	186,600	
(k)	Tobacco-related illnesses	PR-S	C	-0-	-0-	
		(1) PRO0	GRAM TOT			
	GENERAL PURPOSE REVENUE			9,936,800	9,767,800	
	PROGRAM REVENUE			-0-	-0-	
	SERVICE			(-0-)	(-0-)	
a \	TOTAL-ALL SOURCES			9,936,800	9,767,800	
2)	RESEARCH	D.D.		2.47. 500	245 500	
(g)	Cancer research	PR	C	247,500	247,500	
(h)	Prostate cancer research	PR	C	-0 -	-0-	
	DDOCD AM DEVENIUS	(2) PROC	GRAM TOT		247.500	
	PROGRAM REVENUE			247,500	247,500	
	OTHER TOTAL ALL SOURCES			(247,500)	(247,500)	
	TOTAL-ALL SOURCES	050 DEDA	DTMENT T	247,500	247,500	
	GENERAL PURPOSE REVENUE).230 DEPA	ARTMENT T		0.767.900	
	PROGRAM REVENUE			9,936,800 247,500	9,767,800 247,500	
	OTHER			(247,500)	(247,500)	
	SERVICE			(-0-)	(247,300)	
	TOTAL-ALL SOURCES			10,184,300	10,015,300	
0 255	Fublic Instruction, Department of			10,104,500	10,015,500	
1)	EDUCATIONAL LEADERSHIP					
(a)	General program operations	GPR	A	11,816,200	11,905,800	
(b)	General program operations; Wisconsin Educational Services Program for the Deaf and Hard of Hearing and Wisconsin Center for					
(c)	the Blind and Visually Impaired Energy costs; Wisconsin Educational Services Program for the Deaf and Hard of Hearing and Wisconsin Center for the Blind and Visually Impaired; energy-related	GPR	A	10,918,900	10,918,900	
(cm)	assessments Electric energy derived from	GPR	A	505,900	512,200	
	renewable resources	GPR	A	14,500	14,500	
(d)	Principal repayment and interest	GPR	S	1,077,900	1,051,300	
(dw)	1	GPR	A	18,558,400	18,558,400	
(e)	Student information system	GPR	C	3,400,000	3,400,000	
(ee)	Educator effectiveness evaluation system	GPR	A	973,300	973,300	
(eg)	Rural school teacher talent pilot	CDD		5 00.000	5 00.000	
7.15	program	GPR	A	500,000	500,000	
(ek)	Longitudinal data system	GPR	A	3,488,100	3,488,100	
(eL)	WISElearn	GPR	A	1,359,000	1,359,000	
	Academic and career planning	GPR	C	1,100,000	1,100,000	
(em) (ep)	Mental health training program	GPR	A	220,000	220,000	

2017 Wi	sconsin Act 59	- 42 -	_	2017 A	2017 Assembly Bill 64		
STATUT	E, AGENCY AND PURPOSE	Source	Түре	2017-2018	2018-2019		
(fm)	Value-Added Research Center	GPR	A	-0-	-0-		
(fp)	Study on school district						
	reorganization; certain school	ann		77 000	0		
	districts	GPR	В	75,000	-0-		
(g)	Student activity therapy	PR	A	100	100		
(gb)	Wisconsin Educational Services Program for the Deaf and Hard of						
	Hearing and Wisconsin Center for						
	the Blind and Visually Impaired;						
	nonresident fees	PR	C	-0-	-0-		
(ge)	Educator effectiveness evaluation						
,	system; fees	PR	C	4,309,500	4,309,500		
(gL)	Wisconsin Educational Services						
	Program for the Deaf and Hard of						
	Hearing and Wisconsin Center for						
	the Blind and Visually Impaired; leasing of space	DD	C	2,000	2 000		
(22)	Wisconsin Educational Services	PR	С	2,000	2,000		
(gs)	Program for the Deaf and Hard of						
	Hearing and Wisconsin Center for						
	the Blind and Visually Impaired;						
	services	PR	C	7,000	7,000		
(gt)	Wisconsin Educational Services						
	Program for the Deaf and Hard of						
	Hearing and Wisconsin Center for						
	the Blind and Visually Impaired; pupil transportation	PR	A	1,210,000	1,210,000		
(he)	Student information system; fees	PR	C	-0-	-0-		
(he)	Personnel licensure, teacher supply,	I IX	C	O	O		
(1-8)	information and analysis and teacher						
	improvement	PR	A	3,425,300	3,481,500		
(hj)	General educational development						
	and high school graduation						
	equivalency	PR	C	146,000	146,000		
(hm)	Services for drivers	PR-S	A	141,600	141,600		
(i)	Publications	PR	C	152,800	152,800		
(im)	Library products and services	PR	C	141,100	141,100		
(j)	Milwaukee Parental Choice Program and the parental choice program for						
	eligible school districts and other						
	school districts; financial audits	PR	C	135,700	135,700		
(jg)	School lunch handling charges	PR	A	10,062,200	10,062,200		
(jm)	Professional services center charges	PR	A	106,300	106,300		
(jr)	Gifts, grants and trust funds	PR	C	1,500,000	1,500,000		
(jz)	School district boundary appeal						
	proceedings	PR	C	10,000	10,000		
(kd)	Alcohol and other drug abuse	DD . G		600.000	600.200		
<i>A</i>	program	PR-S	A	600,200	600,200		
(ke)	Funds transferred from other state	DD C	C	2 666 700	2 666 700		
(lem)	agencies; program operations	PR-S	С	2,666,700	2,666,700		
(km)	State agency library processing center	PR-S	A	8,100	8,100		
				0,100	0,100		

2017 As	sembly Bill 64	- 43 -	_	2017	Wisconsin Act 59	
STATUT	E, AGENCY AND PURPOSE	Source	Түре	2017-2018	2018-2019	
(ks)	Data processing	PR-S	C	8,957,500	8,957,500	
(me)	Federal aids; program operations	PR-F	C	50,459,000	50,391,900	
(pz)	Indirect cost reimbursements	PR-F	C	4,117,000	4,117,000	
(q)	Digital learning collaborative	SEG	A	1,000,000	1,000,000	
		(1) PROC	GRAM TO	OTALS		
(GENERAL PURPOSE REVENUE			56,158,200	56,152,500	
]	PROGRAM REVENUE			88,158,100	88,147,200	
	FEDERAL			(54,576,000)	(54,508,900)	
	OTHER			(21,208,000)	(21,264,200)	
	SERVICE			(12,374,100)	(12,374,100)	
,	SEGREGATED REVENUE			1,000,000	1,000,000	
	OTHER			(1,000,000)	(1,000,000)	
ŗ	TOTAL-ALL SOURCES			145,316,300	145,299,700	
(2)	AIDS FOR LOCAL EDUCATIONAL PROGRAM	IMING				
(ac)	General equalization aids	GPR	A	4,584,098,000	4,656,848,000	
(ad)	Supplemental aid	GPR	A	100,000	100,000	
(ae)	Sparsity aid	GPR	A	18,496,200	18,759,300	
(af)	Belmont school library aid	GPR	A	-0-	-0-	
(ag)	General equalization aids; hold					
	harmless	GPR	S	-0-	-0-	
(aq)	Per pupil aid	GPR	S	378,180,000	549,098,400	
(ar)	Low revenue adjustment aid	GPR	A	-0-	-0-	
(aw)	Personal electronic computing					
	devices; grant program	GPR	A	-0-	9,187,500	
(az)	Special needs scholarship program	GPR	S	4,272,500	9,320,300	
(b)	Aids for special education and school					
	age parents programs	GPR	A	368,939,100	368,939,100	
(bb)	Aid for high poverty school districts	GPR	A	16,830,000	16,830,000	
(bc)	Aid for children-at-risk programs	GPR	A	-0-	-0-	
(bd)	Additional special education aid	GPR	A	9,239,000	9,353,800	
(be)	Supplemental special education aid	GPR	A	1,750,000	1,750,000	
(bf)	Aid for special education transition					
	grants	GPR	A	2,700,000	3,600,000	
(bg)	Special education transition					
	readiness grants	GPR	A	-0-	1,500,000	
(bh)	Aid to county children with	~~~				
<i>a</i> .	disabilities education boards	GPR	A	4,067,300	4,067,300	T 7 4 1
(bp)	Aid for whole grade sharing	CDD	A	0	750,000	Vetoed In Part
<i>a</i>)	agreements	GPR	A	-0-	750,000	III Part
(br)	School district consolidation aid	GPR	S	-0-	-0-	
(bs)	School district consolidation grants	GPR	A	-0-	-0-	1 7-43
(bt)	Shared services pilot program	GPR	C	_0_	2,000,000	Vetoed In Part
(cc)	Bilingual-bicultural education aids	GPR	A	8,589,800	8,589,800	III I al t
(cf)	Alternative education grants	GPR	A	-0-	-0-	
(cg)	Tuition payments; full–time open	CDD		9 242 000	9 242 000	
()	enrollment transfer payments	GPR	A	8,242,900	8,242,900	
(cm)	Reimbursement for school breakfast	GPR	С	2,510,500	2,510,500	
(on)	programs Aids for school lunches and	OI K	C	2,310,300	2,310,300	
(cn)	nutritional improvement	GPR	A	4,218,100	4,218,100	
(cp)	Wisconsin school day milk program	GPR	A	617,100	617,100	
(cp)	Wisconsin school day milk program	OLK	А	017,100	017,100	

2017 Wi	isconsin Act 59	- 44 -	_	2017 Assembly Bill	
STATUT	E, AGENCY AND PURPOSE	Source	Түре	2017-2018	2018-2019
(cq)	High cost transportation aid	GPR	A	12,700,000	12,700,000
(cr)	Aid for pupil transportation	GPR	A	24,000,000	24,000,000
(cs)	Aid for debt service	GPR	A	133,700	133,700
(cu)	Achievement gap reduction contracts	GPR	A	109,184,500	109,184,500
(cy)	Aid for transportation; open				
	enrollment and course options	GPR	A	454,200	454,200
(da)	Aid for school mental health				
	programs	GPR	A	-0-	3,000,000
(dg)	School performance improvement				
	grants	GPR	A	-0-	3,690,600
(dj)	Summer school programs; grants	GPR	Α	-0-	1,400,000
(dp)	Four-year-old kindergarten grants	GPR	A	1,350,000	1,350,000
(dr)	Robotics league participation grants	GPR	A	250,000	250,000
(ds)	STEM grants	GPR	В	-0-	-0-
(dt)	School-based mental health service				
	grants	GPR	C	-0-	3,250,000
(eb)	Grant for information technology	CDD		077.000	077.000
(1)	education	GPR	A	875,000	875,000
(eh)	Head start supplement	GPR	A	6,264,100	6,264,100
(ek)	Educator effectiveness evaluation	CDD	A	5.746.000	<i>5.746.</i> 000
(f ₀)	system; grants to school districts	GPR	A	5,746,000	5,746,000
(fg)	Aid for cooperative educational service agencies	GPR	A	-0-	-0-
(fk)	Grant program for peer review and	OFK	A	_0_	_0_
(IK)	mentoring	GPR	A	1,606,700	1,606,700
(fm)	Charter schools	GPR	S	63,802,000	68,920,000
(fp)	Charter schools; office of educational	OTT	S	03,002,000	00,520,000
(-P)	opportunity	GPR	S	-0-	-0-
(fr)	Parental choice program for eligible				
	school districts and other school				
	districts	GPR	S	56,612,100	82,428,100
(fs)	Opportunity schools and partnership				
	programs	GPR	S	-0-	-0-
(fu)	Milwaukee parental choice program	GPR	S	212,365,800	222,227,100
(fv)	Milwaukee Parental Choice Program				
	and the parental choice program for				
	eligible school districts and other	CDD	S	0	0
(f)	school districts; transfer pupils	GPR	3	-0-	-0-
(fy)	Grants to support gifted and talented pupils	GPR	A	237,200	237,200
(k)	Funds transferred from other state	OIK	Λ	257,200	231,200
(K)	agencies; local aids	PR-S	C	11,500,000	11,500,000
(kd)	Aid for alcohol and other drug abuse	111 5	C	11,000,000	11,000,000
(RG)	programs	PR-S	A	1,284,700	1,284,700
(km)	Tribal language revitalization grants	PR-S	A	222,800	222,800
(m)	Federal aids; local aid	PR-F	C	760,633,500	760,633,500
(q)	Grants for literacy and early				. ,
• •	childhood development programs	SEG	S	-0-	-0-
(s)	School library aids	SEG	C	35,000,000	37,000,000
		(2) PROC	GRAM 7	TOTALS	
	GENERAL PURPOSE REVENUE			5,908,431,800	6,223,999,300
	PROGRAM REVENUE			773,641,000	773,641,000

STATUTE, AGENCY AND PURPOSE SOURCE TYPE 2017-2018 FEDERAL (760,633,500) (7					-04-7		
FEDERAL	2017 As	sembly Bill 64	- 45 -	_	2017 Wisconsin Act 59		
SERVICE (13,007,500) (13,007,500) SEGREGATED REVENUE 35,000,000 37,000,000 TOTAL—ALL SOURCES 6,717,072,800 7,034,640,300 Adult literacy grants GPR A 83,200 83,200 Common data backer certification or master educator licensure GPR S 2,910,000 2,910,000 Common data backer certification or master educator licensure GPR A 73,900 73,900 Common data backer certification or master educator licensure GPR A 73,900 73,900 Common data backer certification or master educator licensure GPR A 73,900 73,900 Common data backer certification or master educator licensure GPR A 73,900 73,900 Common data backer certification or master educator licensure GPR A 40,000 70,000 Common data backer certification or master education GPR A 40,000 150,000 Common data backer compact on educational opportunity for military children GPR A 42,200 42,200 Common data backer compact on educational opportunity for military children GPR A 75,000 75,000 Common data backer certification GPR A 75,000 75,000 Common data backer certification GPR A 75,000 75,000 Common data backer certification GPR A 300,000 700,000 Common data backer certification GPR A 1,931,500 1,931,500 Common data backer certification GPR A 1,931,500 1,931,500 Common data backer certification GPR A 1,170,400 1,174,300 Common data backer certification GPR A 1,170,400 1,174,300 Common data backer certification GPR A 1,170,400 1,174,300 Common data backer GPR A 1,170,400 1,174,300 Common data backer GPR C GPR C GPR C GPR GPR C GPR GPR C GPR G	STATUTI	E, AGENCY AND PURPOSE	Source	Түре		2018-2019	
SEGREGATED REVENUE		FEDERAL			(760,633,500)	(760,633,500)	
OTHER					(13,007,500)	(13,007,500)	
TOTAL-ALL SOURCES	5	SEGREGATED REVENUE			35,000,000	37,000,000	
(3) AIDS TO LIBRARIES. INDIVIDUALS AND ORGANIZATIONS (b) Adult literacy grants GPR A 83,200 83,200 (c) Grants for national teacher certification or master educator licensure GPR S 2,910,000 2,910,000 (d) Eliks and Easter Seals Center for Respite and Recreation GPR A 73,900 73,900 (dn) Project Lead the Way grants GPR A -0- 0-0- (eb) Grants for bullying prevention GPR A 150,000 150,000 (eg) Milwaukee Public Museum GPR A 150,000 150,000 (eg) Milwaukee Public Museum GPR A 42,200 42,200 (f) Interstate compact on educational opportunity for military children GPR S 900 900 (fa) Very special arts GPR A 75,000 75,000 (fc) College Possible, Inc. GPR A 500,000 500,000 (fg) Special Olympics GPR A 500,000 500,000 (fr) Wisconsin Reading Corps GPR A 300,000 700,000 (fr) Precollege scholarships GPR A 1,931,500 1,931,500 (ge) Special Olympics Wisconsin PR C -0- 0- (mm) Federal funds; local assistance PR-F C 1,300,000 1,300,000 (ms) Federal funds; individuals and organizations PR-F C 62,868,500 62,868,500 (q) Periodical and reference information databases; Newsline for the Blind (price of the Blind Organizations) SEG A 1,170,400 1,174,300 (f) Library service contracts SEG A 1,170,400 1,174,300 (f) Cibrary service contracts SEG A 1,170,400 1,174,300 (f) Cibrary service contracts SEG A 1,170,400 1,174,300 (f) Cibrary SEC		OTHER			(35,000,000)	(37,000,000)	
(b) Adult literacy grants	7	TOTAL-ALL SOURCES			6,717,072,800	7,034,640,300	
(c) Grants for national teacher certification or master educator licensure GPR S 2,910,000 2,910,000 (d) Elks and Easter Seals Center for Respite and Recreation GPR A 73,900 73,900 (d) Project Lead the Way grants GPR A -0- -0- (eb) Grants for bullying prevention GPR A 150,000 150,000 (eg) Milwaukee Public Museum GPR A 42,200 42,200 (f) Interstate compact on educational opportunity for military children GPR S 900 900 (fa) Very special arts GPR A 75,000 75,000 (fc) College Possible, Inc. GPR A 75,000 75,000 (fc) Special Olympics GPR A 75,000 75,000 (fc) Very special arts GPR A 300,000 700,000 (fc) Precollege scholarships GPR A 1,931,500 1,931,500 (ge) Special Olympics	(3)	AIDS TO LIBRARIES, INDIVIDUALS AND	ORGANIZATION	NS			
Certification or master educator licensure GPR S 2,910,000 2,910,000	(b)	Adult literacy grants	GPR	A	83,200	83,200	
licensure GPR S 2,910,000 2,910,000	(c)						
(d) Elks and Easter Seals Center for Respite and Recreation GPR A 73,900 73,900 (dn) Project Lead the Way grants GPR A -0- -0- (eb) Grants for bullying prevention GPR A 150,000 150,000 (eg) Milwaukee Public Museum GPR A 42,200 42,200 (f) Interstate compact on educational opportunity for military children GPR A 75,000 75,000 (fa) Very special arts GPR A 75,000 75,000 (fc) College Possible, Inc. GPR A 500,000 500,000 (fc) Special Olympics GPR A 75,000 75,000 (fc) Special Olympics GPR A 1,931,500 1,931,500 (ge) Special Olympics Wisconsin PR C -0- -0- (mm) Federal funds; iocal assistance PR-F C 1,300,000 1,300,000 (ms) Federal funds; iocal assistance PR-							
Respite and Recreation			GPR	S	2,910,000	2,910,000	
(dn) Project Lead the Way grants GPR A -0- -0- (eb) Grants for bullying prevention GPR A 150,000 150,000 (eg) Milwaukee Public Museum GPR A 42,200 42,200 (f) Interstate compact on educational opportunity for military children GPR S 900 900 (fa) Very special arts GPR A 75,000 75,000 (fc) College Possible, Inc. GPR A 500,000 75,000 (fg) Special Olympics GPR A 300,000 75,000 (fg) Special Olympics GPR A 300,000 75,000 (fg) Special Olympics Wisconsin PR C -0- -0- (mm) Federal funds; iccal assistance PR-F C 1,300,000 1,300,000 (ms) Federal funds; individuals and organizations PR-F C 62,868,500 62,868,500 (qm) Aid to public library systems SEG	(d)						
(eb) Grants for bullying prevention GPR A 150,000 (eg) Milwaukee Public Museum GPR A 42,200 42,200 (f) Interstate compact on educational opportunity for military children GPR S 900 900 (fa) Very special arts GPR A 75,000 75,000 (fc) College Possible, Inc. GPR A 500,000 500,000 (fg) Special Olympics GPR A 75,000 75,000 (fr) Wisconsin Reading Corps GPR A 75,000 75,000 (fr) Wisconsin Reading Corps GPR A 1,931,500 75,000 (ge) Special Olympics Wisconsin PR C -0- -0- -0- (mm) Federal funds; individuals and organizations PR-F C 1,300,000 1,300,000 (ms) Federal funds; individuals and organizations PR-F C 62,868,500 62,868,500 (qm) Aid to public library systems <		_					
(eg) Milwaukee Public Museum GPR A 42,200 42,200 (f) Interstate compact on educational opportunity for military children GPR S 900 900 (fa) Very special arts GPR A 75,000 75,000 (fc) College Possible, Inc. GPR A 500,000 500,000 (fg) Special Olympics GPR A 75,000 75,000 (fr) Wisconsin Reading Corps GPR A 300,000 700,000 (fz) Precollege scholarships GPR A 1,931,500 1,931,500 (ge) Special Olympics Wisconsin PR C -0- -0- (mm) Federal funds; local assistance PR-F C 1,300,000 1,300,000 (ms) Federal funds; individuals and organizations PR-F C 6,868,500 62,868,500 (q) Periodical and reference information databases; Newsline for the Blind SEG A 2,919,100 2,937,500 (am) Aid to							
(f) Interstate compact on educational opportunity for military children GPR S 900 9900 (fa) Very special arts GPR A 75,000 75,000 (fc) College Possible, Inc. GPR A 500,000 500,000 (fg) Special Olympics GPR A 75,000 75,000 (ff) Wisconsin Reading Corps GPR A 300,000 700,000 (fz) Precollege scholarships GPR A 1,931,500 1,931,500 (ge) Special Olympics Wisconsin PR C -0- 0- 0- (mm) Federal funds; local assistance PR-F C 1,300,000 1,300,000 (ms) Federal funds; individuals and organizations PR-F C 62,868,500 62,868,500 (q) Periodical and reference information databases; Newsline for the Blind SEG A 2,919,100 2,937,500 (qm) Aid to public library systems SEG A 15,513,100 16,013,100 (f) Library service contracts SEG A 15,713,100 16,013,100 (f) Library service contracts SEG A 15,713,100 16,013,100 (f) PROGRAM REVENUE 64,168,500 64,168,500 (64,168,500) (7) HER (19,602,600) 20,124,900 OTHER (19,602,600) 20,124,900 OTHER (19,602,600) 20,124,900 PROGRAM REVENUE 925,967,600 90,835,100 PROGRAM REVENUE 925,967,600 925,956,700 FEDERAL (879,378,000) (879,310,900) OTHER (21,208,000) (21,264,200) SEGNEGATED REVENUE 925,967,600 925,956,700 FEDERAL (879,378,000) (879,310,900) OTHER (21,208,000) (21,264,200) SEGNEGATED REVENUE 55,602,600 58,124,900 OTHER (21,208,000) (25,381,600) SEGNEGATED REVENUE 55,602,600 58,124,900 OTHER (21,208,000) (25,381,600)							
opportunity for military children GPR S 900 900 (fa) Very special arts GPR A 75,000 75,000 (fc) College Possible, Inc. GPR A 500,000 500,000 (fg) Special Olympics GPR A 500,000 75,000 (ff) Wisconsin Reading Corps GPR A 300,000 700,000 (fz) Precollege scholarships GPR A 1,931,500 1,931,500 (ge) Special Olympics Wisconsin PR C -0- -0- (mm) Federal funds; local assistance PR-F C 1,300,000 1,300,000 (ms) Federal funds; individuals and organizations PR-F C 62,868,500 62,868,500 (q) Periodical and reference information databases; Newsline for the Blind SEG A 2,919,100 2,937,500 (qm) Aid to public library systems SEG A 1,513,100 16,013,100 (r) Library service contracts SEG<			GPR	A	42,200	42,200	
(fa) Very special arts GPR A 75,000 75,000 (fc) College Possible, Inc. GPR A 500,000 500,000 (fg) Special Olympics GPR A 75,000 75,000 (fr) Wisconsin Reading Corps GPR A 300,000 700,000 (fr) Precollege scholarships GPR A 1,931,500 1,931,500 (ge) Special Olympics Wisconsin PR C -0- -0- (mm) Federal funds; local assistance PR-F C 1,300,000 1,300,000 (mm) Federal funds; individuals and organizations PR-F C 62,868,500 62,868,500 (q) Periodical and reference information databases; Newsline for the Blind stable in the	(f)			~			
(fc) College Possible, Inc. GPR A 500,000 500,000 (fg) Special Olympics GPR A 75,000 75,000 (fr) Wisconsin Reading Corps GPR A 300,000 700,000 (fz) Precollege scholarships GPR A 1,931,500 1,931,500 (ge) Special Olympics Wisconsin PR C -0- -0- (mm) Federal funds; local assistance PR-F C 1,300,000 1,300,000 (ms) Federal funds; individuals and organizations PR-F C 62,868,500 62,868,500 (g) Periodical and reference information databases; Newsline for the Blind SEG A 2,919,100 2,937,500 (qm) Aid to public library systems SEG A 15,513,100 16,013,100 (r) Library service contracts SEG A 1,170,400 1,174,300 (ge) PROGRAM REVENUE 64,168,500 64,168,500 64,168,500 FEDERAL (64,168,500	(0.)						
(fg) Special Olympics GPR A 75,000 75,000 (fr) Wisconsin Reading Corps GPR A 300,000 700,000 (fz) Precollege scholarships GPR A 1,931,500 1,931,500 (ge) Special Olympics Wisconsin PR C -0- -0- (mm) Federal funds; local assistance PR-F C 1,300,000 1,300,000 (ms) Federal funds; individuals and organizations PR-F C 62,868,500 62,868,500 (q) Periodical and reference information databases; Newsline for the Blind SEG A 2,919,100 2,937,500 (qm) Aid to public library systems SEG A 1,170,400 1,174,300 (r) Library service contracts SEG A 1,170,400 1,174,300 (r) Library service contracts SEG A 1,170,400 1,174,300 (r) Library service contracts SEG A 1,170,400 1,174,300 GENERAL PURPOSE REVENUE <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>							
(fr) Wisconsin Reading Corps GPR A 300,000 700,000 (fz) Precollege scholarships GPR A 1,931,500 1,931,500 (ge) Special Olympics Wisconsin PR C -0- -0- (mm) Federal funds; local assistance PR-F C 1,300,000 1,300,000 (ms) Federal funds; individuals and organizations PR-F C 62,868,500 62,868,500 (q) Periodical and reference information databases; Newsline for the Blind SEG A 2,919,100 2,937,500 (qm) Aid to public library systems SEG A 15,513,100 16,013,100 (r) Library service contracts SEG A 1,170,400 1,174,300 (r) Library service contracts SEG A 1,170,400 1,174,300 (r) Library service contracts SEG A 1,170,400 1,174,300 (r) Library service contracts 6,141,700 6,541,700 6,541,600 PROGRAM REVENUE							
(fz) Precollege scholarships GPR A 1,931,500 1,931,500 (ge) Special Olympics Wisconsin PR C -0- -0- (mm) Federal funds; local assistance PR-F C 1,300,000 1,300,000 (ms) Federal funds; individuals and organizations PR-F C 62,868,500 62,868,500 (q) Periodical and reference information databases; Newsline for the Blind SEG A 2,919,100 2,937,500 (qm) Aid to public library systems SEG A 15,513,100 16,013,100 (r) Library service contracts SEG A 1,170,400 1,174,300 (r) Library service contracts SEG A 1,170,400 1,174,300 (r) Library service contracts SEG A 1,170,400 1,174,300 (r) Library service contracts SEG A 1,170,400 6,541,700 FEDERAL (64,168,500)		* * *					
(ge) Special Olympics Wisconsin PR C -0- -0- (mm) Federal funds; local assistance PR-F C 1,300,000 1,300,000 (ms) Federal funds; individuals and organizations PR-F C 62,868,500 62,868,500 (q) Periodical and reference information databases; Newsline for the Blind SEG A 2,919,100 2,937,500 (qm) Aid to public library systems SEG A 15,513,100 16,013,100 (r) Library service contracts SEG A 1,170,400 1,174,300 (r) Library service contracts SEG A 1,170,400 6,541,700 PROGRAM REVENUE 6,141,700 6,541,700 6,541,68,500 64,168,500 OTHER							
(mm) Federal funds; local assistance PR-F C 1,300,000 1,300,000 (ms) Federal funds; individuals and organizations PR-F C 62,868,500 62,868,500 (q) Periodical and reference information databases; Newsline for the Blind databases; Newsline for the Blind SEG A 2,919,100 2,937,500 (qm) Aid to public library systems SEG A 15,513,100 16,013,100 (r) Library service contracts SEG A 1,170,400 1,174,300 (r) DROGRAM REVENUE 6,141,700 6,541,700 PROGRAM REVENUE 920,255 DEPARTMENT TOTALS GENERAL PURPOSE REVENUE 925,967,600 <		-					
(ms) Federal funds; individuals and organizations PR-F C 62,868,500 62,868,500 (q) Periodical and reference information databases; Newsline for the Blind SEG A 2,919,100 2,937,500 (qm) Aid to public library systems SEG A 15,513,100 16,013,100 (r) Library service contracts SEG A 1,170,400 1,174,300 (r) Library service contracts SEG A 1,170,400 1,174,300 (3) PROGRAM TOTALS GENERAL PURPOSE REVENUE 6,141,700 6,541,700 PROGRAM REVENUE 64,168,500 64,168,500 64,168,500 OTHER (-0-) (-0-) (-0-) SEGREGATED REVENUE 19,602,600 20,124,900 OTHER (19,602,600) (20,124,900) TOTAL-ALL SOURCES 89,912,800 90,835,100 PROGRAM REVENUE 5,970,731,700 6,286,693,500 PROGRAM REVENUE 925,967,600 925,956,700 FEDERAL (879,378,000) (879,310,900)	_	* * *					
organizations PR-F C 62,868,500 62,868,500 (q) Periodical and reference information databases; Newsline for the Blind SEG A 2,919,100 2,937,500 (qm) Aid to public library systems SEG A 15,513,100 16,013,100 (r) Library service contracts SEG A 1,170,400 1,174,300 (r) Library service contracts SEG A 1,170,400 1,174,300 (3) PROGRAM TOTALS GENERAL PURPOSE REVENUE 66,141,700 6,541,700 PROGRAM REVENUE 64,168,500 64,168,500 64,168,500 64,168,500 64,168,500 64,168,500 OTHER (-0-) <td>` ,</td> <td></td> <td>PR–F</td> <td>С</td> <td>1,300,000</td> <td>1,300,000</td>	` ,		PR–F	С	1,300,000	1,300,000	
(q) Periodical and reference information databases; Newsline for the Blind SEG A 2,919,100 2,937,500 (qm) Aid to public library systems SEG A 15,513,100 16,013,100 (r) Library service contracts SEG A 1,170,400 1,174,300 (r) Library service contracts SEG A 1,170,400 1,174,300 (GENERAL PURPOSE REVENUE 6,141,700 6,541,700 PROGRAM REVENUE 64,168,500 64,168,500 FEDERAL (64,168,500) (64,168,500) OTHER (19,602,600) 20,124,900 OTHER (19,602,600) (20,124,900) TOTAL-ALL SOURCES 89,912,800 90,835,100 PROGRAM REVENUE 5,970,731,700 6,286,693,500 PROGRAM REVENUE 5,970,731,700 6,286,693,500 PROGRAM REVENUE 925,967,600 925,956,700 FEDERAL (879,378,000) (879,310,900) OTHER (21,208,000) (21,264,200) SERVICE (25,381,600) (25,381,600	(ms)		DD E	~	(2.0(0.500	62 0 60 7 00	
databases; Newsline for the Blind (qm) SEG A 2,919,100 2,937,500 (qm) Aid to public library systems SEG A 15,513,100 16,013,100 (r) Library service contracts SEG A 1,170,400 1,174,300 (r) Library service contracts SEG A 1,170,400 1,174,300 GENERAL PURPOSE REVENUE 64,141,700 6,541,700 PROGRAM REVENUE 64,168,500 64,168,500 OTHER (-0-) (-0-) OTHER (19,602,600 20,124,900 TOTAL-ALL SOURCES 89,912,800 90,835,100 PROGRAM REVENUE 5,970,731,700 6,286,693,500 PROGRAM REVENUE 925,967,600 925,956,700 FEDERAL (879,378,000) (879,310,900) OTHER (21,208,000) (21,264,200) SERVICE (25,381,600) (25,381,600) SEGREGATED REVENUE 55,602,600 58,124,900 OTHER (55,602,600) (58,124,900)		=		C	62,868,500	62,868,500	
(qm) Aid to public library systems SEG A 15,513,100 16,013,100 (r) Library service contracts SEG A 1,170,400 1,174,300 (3) PROGRAM TOTALS GENERAL PURPOSE REVENUE 6,141,700 6,541,700 PROGRAM REVENUE 64,168,500 64,168,500 FEDERAL (64,168,500) (64,168,500) OTHER (-0-) (-0-) OTHER (19,602,600) 20,124,900 TOTAL-ALL SOURCES 89,912,800 90,835,100 PROGRAM REVENUE 5,970,731,700 6,286,693,500 PROGRAM REVENUE 925,967,600 925,956,700 FEDERAL (879,378,000) (879,310,900) OTHER (21,208,000) (21,264,200) SERVICE (25,381,600) (25,381,600) SEGREGATED REVENUE 55,602,600 58,124,900 OTHER (55,602,600) (58,124,900)	(q)			٨	2.010.100	2 027 500	
(r) Library service contracts SEG A (3) PROGRAM TOTALS 1,170,400 1,174,300 GENERAL PURPOSE REVENUE 6,141,700 6,541,700 PROGRAM REVENUE 64,168,500 64,168,500 FEDERAL (64,168,500) (64,168,500) OTHER (-0-) (-0-) SEGREGATED REVENUE 19,602,600 20,124,900 OTHER (19,602,600) (20,124,900) TOTAL-ALL SOURCES 89,912,800 90,835,100 20.255 DEPARTMENT TOTALS GENERAL PURPOSE REVENUE 5,970,731,700 6,286,693,500 PROGRAM REVENUE 925,967,600 925,956,700 FEDERAL (879,378,000) (879,310,900) OTHER (21,208,000) (21,264,200) SERVICE (25,381,600) (25,381,600) SEGREGATED REVENUE 55,602,600 58,124,900 OTHER (55,602,600) (58,124,900)	()						
GENERAL PURPOSE REVENUE 6,141,700 6,541,700 PROGRAM REVENUE 64,168,500 64,168,500 FEDERAL (64,168,500) (64,168,500) OTHER (-0-) (-0-) SEGREGATED REVENUE 19,602,600 20,124,900 OTHER (19,602,600) (20,124,900) TOTAL-ALL SOURCES 89,912,800 90,835,100 20.255 DEPARTMENT TOTALS GENERAL PURPOSE REVENUE 5,970,731,700 6,286,693,500 PROGRAM REVENUE 925,967,600 925,956,700 FEDERAL (879,378,000) (879,310,900) OTHER (21,208,000) (21,264,200) SERVICE (25,381,600) (25,381,600) SEGREGATED REVENUE 55,602,600 58,124,900 OTHER (55,602,600) (58,124,900)	_						
GENERAL PURPOSE REVENUE 6,141,700 6,541,700 PROGRAM REVENUE 64,168,500 64,168,500 FEDERAL (64,168,500) (64,168,500) OTHER (-0-) (-0-) SEGREGATED REVENUE 19,602,600 20,124,900 OTHER (19,602,600) (20,124,900) TOTAL-ALL SOURCES 89,912,800 90,835,100 20.255 DEPARTMENT TOTALS GENERAL PURPOSE REVENUE 5,970,731,700 6,286,693,500 PROGRAM REVENUE 925,967,600 925,956,700 FEDERAL (879,378,000) (879,310,900) OTHER (21,208,000) (21,264,200) SERVICE (25,381,600) (25,381,600) SEGREGATED REVENUE 55,602,600 58,124,900 OTHER (55,602,600) (58,124,900)	(r)	Library service contracts				1,1/4,300	
PROGRAM REVENUE 64,168,500 64,168,500 FEDERAL (64,168,500) (64,168,500) OTHER (-0-) (-0-) SEGREGATED REVENUE 19,602,600 20,124,900 OTHER (19,602,600) (20,124,900) TOTAL-ALL SOURCES 89,912,800 90,835,100 20.255 DEPARTMENT TOTALS GENERAL PURPOSE REVENUE 5,970,731,700 6,286,693,500 PROGRAM REVENUE 925,967,600 925,956,700 FEDERAL (879,378,000) (879,310,900) OTHER (21,208,000) (21,264,200) SERVICE (25,381,600) (25,381,600) SEGREGATED REVENUE 55,602,600 58,124,900 OTHER (55,602,600) (58,124,900)		SENIED AT DUDDOCE DEVIENTE	(3) PROC	JKAM I		(541 700	
FEDERAL (64,168,500) (64,168,500) OTHER (-0-) (-0-) SEGREGATED REVENUE 19,602,600 20,124,900 OTHER (19,602,600) (20,124,900) TOTAL-ALL SOURCES 89,912,800 90,835,100 20.255 DEPARTMENT TOTALS GENERAL PURPOSE REVENUE 5,970,731,700 6,286,693,500 PROGRAM REVENUE 925,967,600 925,956,700 FEDERAL (879,378,000) (879,310,900) OTHER (21,208,000) (21,264,200) SERVICE (25,381,600) (25,381,600) SEGREGATED REVENUE 55,602,600 58,124,900 OTHER (55,602,600) (58,124,900)							
OTHER (-0-) (-0-) SEGREGATED REVENUE 19,602,600 20,124,900 OTHER (19,602,600) (20,124,900) TOTAL-ALL SOURCES 89,912,800 90,835,100 20.255 DEPARTMENT TOTALS GENERAL PURPOSE REVENUE 5,970,731,700 6,286,693,500 PROGRAM REVENUE 925,967,600 925,956,700 FEDERAL (879,378,000) (879,310,900) OTHER (21,208,000) (21,264,200) SERVICE (25,381,600) (25,381,600) SEGREGATED REVENUE 55,602,600 58,124,900 OTHER (55,602,600) (58,124,900)	I						
SEGREGATED REVENUE 19,602,600 20,124,900 OTHER (19,602,600) (20,124,900) TOTAL-ALL SOURCES 89,912,800 90,835,100 20.255 DEPARTMENT TOTALS GENERAL PURPOSE REVENUE 5,970,731,700 6,286,693,500 PROGRAM REVENUE 925,967,600 925,956,700 FEDERAL (879,378,000) (879,310,900) OTHER (21,208,000) (21,264,200) SERVICE (25,381,600) (25,381,600) SEGREGATED REVENUE 55,602,600 58,124,900 OTHER (55,602,600) (58,124,900)					, , , , , , , , , , , , , , , , , , , ,		
OTHER (19,602,600) (20,124,900) TOTAL-ALL SOURCES 89,912,800 90,835,100 20.255 DEPARTMENT TOTALS GENERAL PURPOSE REVENUE 5,970,731,700 6,286,693,500 PROGRAM REVENUE 925,967,600 925,956,700 FEDERAL (879,378,000) (879,310,900) OTHER (21,208,000) (21,264,200) SERVICE (25,381,600) (25,381,600) SEGREGATED REVENUE 55,602,600 58,124,900 OTHER (55,602,600) (58,124,900)	c						
TOTAL-ALL SOURCES 20.255 DEPARTMENT TOTALS GENERAL PURPOSE REVENUE 5,970,731,700 6,286,693,500 PROGRAM REVENUE 925,967,600 925,956,700 FEDERAL (879,378,000) (879,310,900) OTHER (21,208,000) SERVICE (25,381,600) SEGREGATED REVENUE 55,602,600 OTHER (55,602,600) (58,124,900)	2						
20.255 DEPARTMENT TOTALS GENERAL PURPOSE REVENUE 5,970,731,700 6,286,693,500 PROGRAM REVENUE 925,967,600 925,956,700 FEDERAL (879,378,000) (879,310,900) OTHER (21,208,000) (21,264,200) SERVICE (25,381,600) (25,381,600) SEGREGATED REVENUE 55,602,600 58,124,900 OTHER (55,602,600) (58,124,900)	7				, , , , , , , , , , , , , , , , , , , ,		
GENERAL PURPOSE REVENUE 5,970,731,700 6,286,693,500 PROGRAM REVENUE 925,967,600 925,956,700 FEDERAL (879,378,000) (879,310,900) OTHER (21,208,000) (21,264,200) SERVICE (25,381,600) (25,381,600) SEGREGATED REVENUE 55,602,600 58,124,900 OTHER (55,602,600) (58,124,900)	J	IOTAL-ALL SOURCES	20 255 DEDA			90,835,100	
PROGRAM REVENUE 925,967,600 925,956,700 FEDERAL (879,378,000) (879,310,900) OTHER (21,208,000) (21,264,200) SERVICE (25,381,600) (25,381,600) SEGREGATED REVENUE 55,602,600 58,124,900 OTHER (55,602,600) (58,124,900)		SENIED AT DUDDOCE DEVIENTE	20.255 DEPA	AKIMEN		(20((02 500	
FEDERAL (879,378,000) (879,310,900) OTHER (21,208,000) (21,264,200) SERVICE (25,381,600) (25,381,600) SEGREGATED REVENUE 55,602,600 58,124,900 OTHER (55,602,600) (58,124,900)							
OTHER (21,208,000) (21,264,200) SERVICE (25,381,600) (25,381,600) SEGREGATED REVENUE 55,602,600 58,124,900 OTHER (55,602,600) (58,124,900)	ŀ						
SERVICE (25,381,600) (25,381,600) SEGREGATED REVENUE 55,602,600 58,124,900 OTHER (55,602,600) (58,124,900)					` ' ' '		
SEGREGATED REVENUE 55,602,600 58,124,900 OTHER (55,602,600) (58,124,900)							
OTHER (55,602,600) (58,124,900)							
	2						
	7				(55,602,600) 6,952,301,900	(58,124,900) 7,270,775,100	

	SEGREGATED REVENUE			55,602,600	58,124,900
	OTHER			(55,602,600)	(58,124,900)
	TOTAL-ALL SOURCES			6,952,301,900	7,270,775,100
20.285	University of Wisconsin System				
(1)	University education, research	AND PUBLIC SE	RVICE		
(a)	General program operations	GPR	В	818,372,600	839,718,400

7 Wi	sconsin Act 59	- 46 -	_	2017	2017 Assembly Bill 64	
TATUT	E, AGENCY AND PURPOSE	Source	Түре	2017-2018	2018-2019	
(am)	Electric energy derived from renewable resources	GPR	A	4,367,000	4,367,000	
(b)	Tommy G. Thompson Center on Public Leadership	GPR	A	1,500,000	1,500,000	
(d)	Principal repayment and interest	GPR	S	223,168,600	218,577,800	
(e)	Grants to meet emergency financial					
(fd)	need State laboratory of hygiene; general	GPR	С	130,000	130,000	
	program operations	GPR	A	11,037,300	11,037,300	
(fj)	Veterinary diagnostic laboratory	GPR	A	4,974,800	4,974,800	
(gb) (ge)	General program operations Gifts and nonfederal grants and	PR	С	2,511,574,600	2,511,574,600	
(gj)	contracts Self–amortizing facilities principal	PR	C	575,445,500	575,445,500	
(8))	and interest	PR	S	141,717,700	146,584,800	
(i)	State laboratory of hygiene	PR	C	20,888,100	20,888,100	
(ia)	State laboratory of hygiene, drivers	PR-S	C	1,619,200	1,619,200	
(je)	Veterinary diagnostic laboratory; fees	PR	C	4,445,100	4,445,100	
(k)	Funds transferred from other state	1 K	C	7,773,100	4,445,100	
` /	agencies	PR-S	C	66,858,500	66,858,500	
(kg)	Veterinary diagnostic laboratory;	DD C	C	0	0	
<i>a</i>	state agencies	PR-S	C	-0-	-0-	
(Li)	General fund interest	PR	C	-0-	-0-	
(m)	Federal aid	PR–F	C	1,668,452,700	1,668,452,700	
(mc)	Veterinary diagnostic laboratory; federal funds	PR-F	C	193,300	193,300	
(q)	Telecommunications services	SEG	A	1,054,800	1,054,800	
(q) (qe)	Rural physician residency assistance					
(qj)	Physician and dentist and health care provider loan assistance programs; critical access hospital assessment	SEG	В	855,300	855,300	
, ,	fund	SEG	В	250,000	250,000	
_	Grants for forestry programs	SEG	A	134,500	134,500	
(qr) (rm)	Discovery farm grants Environmental program grants and	SEG	A	249,800	249,800	
(sp)	scholarships Wisconsin Institute for Sustainable	SEG	С	301,600	301,600	Veto
	Technology	SEG	A	440,000	440,000	In P
(u)	Trust fund income	SEG	C	22,794,700	22,794,700	
(w)	Trust fund operations	SEG	C	-0-	-0-	
	GENERAL RUBBOGE REVENUE	(1) PROC	GRAM TO		1 000 207 200	
	GENERAL PURPOSE REVENUE			1,063,550,300	1,080,305,300	
J	PROGRAM REVENUE			4,991,194,700	4,996,061,800	
	FEDERAL			(1,668,646,000)	(1,668,646,000)	
	OTHER			(3,254,071,000)	(3,258,938,100)	
	SERVICE			(68,477,700)	(68,477,700)	
,	SEGREGATED REVENUE			26,080,700	26,080,700	
	OTHER			(26,080,700)	(26,080,700)	
-	FOTAL-ALL SOURCES).285 DEPA	RTMFNT	6,080,825,700 TOTALS	6,102,447,800	
	GENERAL PURPOSE REVENUE	203 DEFA	TATELLA I	1,063,550,300	1,080,305,300	

	E, AGENCY AND PURPOSE	Source	Түре	2017-2018	2018-2019
	PROGRAM REVENUE			4,991,194,700	4,996,061,800
	FEDERAL			(1,668,646,000)	(1,668,646,000)
	OTHER			(3,254,071,000)	(3,258,938,100)
	SERVICE			(68,477,700)	(68,477,700)
	SEGREGATED REVENUE			26,080,700	26,080,700
	OTHER			(26,080,700)	(26,080,700)
	TOTAL-ALL SOURCES			6,080,825,700	6,102,447,800
	Technical College System Board			, , ,	, , ,
(1)	TECHNICAL COLLEGE SYSTEM				
(a)	General program operations	GPR	A	2,703,100	2,708,700
(am)	Fee remissions	GPR	A	14,200	14,200
(d)	State aid for technical colleges;	OTIC		11,200	11,200
(u)	statewide guide	GPR	A	88,534,900	88,534,900
(dp)	Property tax relief aid	GPR	S	406,000,000	406,000,000
(e)	Grants to meet emergency financial	OTK	S	400,000,000	400,000,000
(C)	need	GPR	C	320,000	320,000
(f)	Grants to district boards	GPR	C	21,874,200	21,874,200
	Text materials	PR		115,500	115,500
(g)		PR	A C	15,200	15,200
(ga)	Auxiliary services				
(gm)	Fire schools; state operations	PR	A	400,100	400,900
(gr)	Fire schools; local assistance	PR	A	600,000	600,000
(h)	Gifts and grants	PR	C	20,600	20,600
(hm)	Truck driver training	PR-S	C	150,000	150,000
(i)	Conferences	PR	C	72,600	72,600
(j)	Personnel certification	PR	A	236,400	236,900
(k)	Gifts and grants	PR	C	30,200	30,200
(ka)	Interagency projects; local assistance	PR-S	A	2,000,000	2,000,000
(kb)	Interagency projects; state operations	PR-S	A	239,900	239,900
(kd)	Transfer of Indian gaming receipts;				
	work-based learning programs	PR-S	A	594,000	594,000
(km)	Master logger apprenticeship grants	PR-S	C	-0-	-0-
(kx)	Interagency and intra-agency				
	programs	PR-S	C	57,900	57,900
(L)	Services for district boards	PR	A	46,000	46,000
(m)	Federal aid, state operations	PR-F	C	3,386,100	3,391,800
(n)	Federal aid, local assistance	PR-F	C	28,424,300	28,424,300
(o)	Federal aid, aids to individuals and				
	organizations	PR-F	C	800,000	800,000
(pz)	Indirect cost reimbursements	PR-F	C	196,000	196,000
(q)	Agricultural education consultant	GPR	A	67,100	67,100
(r)	Veteran grant jobs pilot program	SEG	A	-0-	-0-
		(1) PROC	GRAM T	ΓOTALS	
	GENERAL PURPOSE REVENUE			519,513,500	519,519,100
	PROGRAM REVENUE			37,384,800	37,391,800
	FEDERAL			(32,806,400)	(32,812,100)
	OTHER			(1,536,600)	(1,537,900)
	SERVICE			(3,041,800)	(3,041,800)
	SEGREGATED REVENUE			-0-	-0-
	OTHER			(-0-)	(-0-)
	TOTAL-ALL SOURCES			556,898,300	556,910,900
(2)	EDUCATIONAL APPROVAL BOARD			, , -	, , ,
` '					

2017 Wi	sconsin Act 59	- 48 -	_	2017	Assembly Bill 64
STATUT	E, AGENCY AND PURPOSE	Source	Түре	2017-2018	2018-2019
(g)	Proprietary school programs	PR	A	308,100	-0-
(gm)	Student protection	PR	C	28,300	-0-
(i)	Closed schools; preservation of			,	
()	student records	PR	C	6,100	-0-
		(2) PROC	GRAM TO	OTALS	
	PROGRAM REVENUE			342,500	-0-
	OTHER			(342,500)	(-0-)
,	TOTAL–ALL SOURCES			342,500	-0-
	20	0.292 DEPA	RTMENT	TOTALS	
	GENERAL PURPOSE REVENUE			519,513,500	519,519,100
	PROGRAM REVENUE			37,727,300	37,391,800
	FEDERAL			(32,806,400)	(32,812,100)
	OTHER			(1,879,100)	(1,537,900)
	SERVICE			(3,041,800)	(3,041,800)
	SEGREGATED REVENUE			-0-	-0-
	OTHER			(-0-)	(-0-)
i	TOTAL–ALL SOURCES			557,240,800	556,910,900
		Е	ducation		
	F	UNCTION	AL AREA	TOTALS	
	GENERAL PURPOSE REVENUE			7,731,407,200	8,065,571,800
	PROGRAM REVENUE			5,974,652,000	5,980,113,800
	FEDERAL			(2,582,366,900)	(2,582,257,700)
	OTHER			(3,291,339,600)	(3,296,898,800)
	SERVICE			(100,945,500)	(100,957,300)
	SEGREGATED REVENUE			86,361,300	88,883,600
	FEDERAL			(-0-)	(-0-)
	OTHER			(86,361,300)	(88,883,600)
	SERVICE			(-0-)	(-0-)
	LOCAL			(-0-)	(-0-)
	TOTAL–ALL SOURCES			13,792,420,500	14,134,569,200
	Enviro	nmental	Resou	rces	
20.320	Environmental Improvement Progra	m			
(1)	CLEAN WATER FUND PROGRAM OPERATIO				
(a)	Environmental aids — clean water				
	fund program	GPR	A	-0-	-0-
(c)	Principal repayment and interest —				
	clean water fund program	GPR	S	10,805,500	10,516,000
(r)	Clean water fund program repayment				
	of revenue obligations	SEG	S	-0-	-0-
(s)	Clean water fund program financial				
	assistance	SEG	S	-0-	-0-
(sm)	Land recycling loan program	~=~	~		
	financial assistance	SEG	S	-0-	-0-
(t)	Principal repayment and interest —	and.		0.000.000	0.000.000
	clean water fund program bonds	SEG	A	8,000,000	8,000,000
(u)	Principal repayment and interest —				
	clean water fund program revenue	SEG	С	-0-	Λ
()	obligation repayment	SEC	C	-0-	-0-
(x)	Clean water fund program financial assistance; federal	SEG-F	С	-0-	_^_
	assistance, icuciai	(1) PROC		· ·	- 0-
		(1) 1100	J. 1. 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	J 1. 11.	

017 A	Assembly Bill 64	- 49	_	2017 W	isconsin Act 59	
STATU	TTE, AGENCY AND PURPOSE GENERAL PURPOSE REVENUE	Source	Түре	2017–2018 10,805,500	2018–2019 10,516,000	
	SEGREGATED REVENUE			8,000,000	8,000,000	
	FEDERAL			(-0-)	(-0-)	
	OTHER			(8,000,000)	(8,000,000)	
	TOTAL-ALL SOURCES			18,805,500	18,516,000	
(2)	SAFE DRINKING WATER LOAN PROGRAM	OPERATIONS	1			
(c)	Principal repayment and interest —					
	safe drinking water loan program	GPR	S	5,569,000	5,595,100	
(s)	Safe drinking water loan programs	~=~	~			
	financial assistance	SEG	S	-0-	-0-	
(x)	Safe drinking water loan programs		~			
	financial assistance; federal	SEG-F	C	-0-	-0-	
		(2) PRO	GRAM TOT			
	GENERAL PURPOSE REVENUE			5,569,000	5,595,100	
	SEGREGATED REVENUE			-0-	-0-	
	FEDERAL			(-0-)	(-0-)	
	OTHER			(-0-)	(-0-)	
	TOTAL-ALL SOURCES			5,569,000	5,595,100	
		20.320 DEP	ARTMENT T	TOTALS		
	GENERAL PURPOSE REVENUE			16,374,500	16,111,100	
	SEGREGATED REVENUE			8,000,000	8,000,000	
	FEDERAL			(-0-)	(-0-)	
	OTHER			(8,000,000)	(8,000,000)	
	TOTAL-ALL SOURCES			24,374,500	24,111,100	
20.36	0 Lower Wisconsin State Riverway B	oard				
(1)	CONTROL OF LAND DEVELOPMENT AND	USE IN THE	LOWER WISCO	NSIN STATE RIVERWAY		
(g)	Gifts and grants	PR	C	-0-	-0-	
(q)	General program operations —			238,900	238,900	Veto
	conservation fund	SEG	A	224,300	224,300	In P
		(1) PRO	GRAM TOT	TALS		
	PROGRAM REVENUE			-0-	-0-	
	OTHER			(-0-)	(-0-)	
	SEGREGATED REVENUE			238,900	238,900	
	OTHER			(238,900)	(238,900)	
	TOTAL-ALL SOURCES			238,900	238,900	
		20.360 DEP	ARTMENT T			
	PROGRAM REVENUE	5.2 00 DEA		- 0-	-0-	
	OTHER			(-0-)	(-0-)	
	SEGREGATED REVENUE			238,900	238,900	
	OTHER			(238,900)	(238,900)	
	TOTAL-ALL SOURCES			238,900)	238,900)	
20 27	0 Natural Resources, Department of			250,700	230,300	
	FISH, WILDLIFE & PARKS					
(1)		CDD	٨	0	0	
(ea)		GPR	A	-0-	-0-	
(eq)		CEC	C	^	0	
()	maintenance	SEG	S	-0-	-0-	
(er)	Parks — campground reservation	SEC	C	1 250 000	1 250 000	
	fees	SEG	C	1,250,000	1,250,000	
(.)	Daulas intermedia o			^		
(es)	Parks – interpretive programs	SEG	С	-0-	-0-	
(es) (fb)	Endangered resources – general					
		SEG GPR	C A	-0- -0-	-0- -0-	

2017 As	2017 Assembly Bill 64		_	2017 V	2017 Wisconsin Act 59		
STATUT	E, AGENCY AND PURPOSE	Source	Түре	2017-2018	2018-2019		
(Lt)	Wildlife management	SEG	A	-0-	-0-		
(Lu)	Fish and wildlife habitat	SEG	S	-0-	-0-		
(Lv)	Deer management assistance						
	program	SEG	C	-0-	-0-		
(ma)	General program operations — state						
	funds	GPR	A	1,315,100	1,315,100		
(mi)	General program operations —	DD		(20, 200	(20, 200		
(1)	private and public sources	PR	C	638,200	638,200		
(mk)	General program operations — service funds	PR-S	C	277,600	277,600		
(mm)	General program operations —	1 K-5	C	277,000	277,000		
(111111)	federal funds	PR-F	C	227,800	227,800		
(mq)	General program operations – state	110 1	C	227,000	227,000		
(1114)	snowmobile trails and areas	SEG	A	211,100	211,100		
(mr)	General program operations – state			,	,		
, ,	utility terrain vehicle projects	SEG	A	16,900	16,900		
(ms)	General program operations – state						
	all-terrain vehicle projects	SEG	A	310,500	310,500		
(mt)	Land preservation and management –						
	endowment fund	SEG	S	-0-	-0-		
(mu)	General program operations — state	ana		0	0		
	funds	SEG	A	-0-	-0- 751 500		
	Land program management	SEG	A	819,900	751,500		
	Wildlife management Southern forests	SEG	A	11,391,400	11,255,200		
	Parks and recreation	SEG SEG	A A	5,151,500 15,474,000	4,943,400 14,937,000		
		SEG	A A	881,300	881,300		
	Natural Heritage Conservation Fisheries Management	SEG	A A	15,226,400	15,140,900		
	Facilities and lands	SEG	A	3,024,200	5,037,700		
	NET APPROPRIATION	SEC	Λ	51,968,700	52,947,000		
(my)	General program operations —			31,700,700	32,747,000		
(IIIy)	federal funds	SEG-F	C	-0-	-0-		
	Wildlife management	SEG-F	C	8,826,800	8,677,800		
	Southern forests	SEG-F	C	113,600	113,600		
	Parks and recreation	SEG-F	C	625,400	625,400		
	Natural Heritage Conservation	SEG-F	C	2,051,100	2,051,100		
	Fisheries Management	SEG-F	C	6,727,100	6,672,800		
	Facilities and lands	SEG-F	C	765,900	1,531,000		
	NET APPROPRIATION			19,109,900	19,671,700		
		(1) PRO0	GRAM T	OTALS			
(GENERAL PURPOSE REVENUE			2,606,700	2,606,700		
I	PROGRAM REVENUE			1,612,400	1,612,400		
	FEDERAL			(227,800)	(227,800)		
	OTHER			(776,700)	(776,700)		
	SERVICE			(607,900)	(607,900)		
5	SEGREGATED REVENUE			79,747,500	81,287,600		
	FEDERAL			(19,109,900)	(19,671,700)		
_	OTHER			(60,637,600)	(61,615,900)		
	FOTAL-ALL SOURCES			83,966,600	85,506,700		
` /	FORESTRY	CEC	C	100 500	100 500		
(cq)	Forestry — reforestation	SEG	С	100,500	100,500		

2017 Wisconsin Act 59		- 52 -	-	2017	2017 Assembly Bill 64		
STATUT	TE, AGENCY AND PURPOSE	Source	Түре	2017-2018	2018-2019		
(cr)	Forestry — recording fees	SEG	C	89,100	89,100		
(cs)	Forestry — forest fire emergencies	SEG	C	-0-	-0-		
(ct)	Timber sales contracts — repair and reimbursement costs	SEG	C	-0-	-0-		
(cu)	Forestry — forestry education						
(cv)	curriculum Forestry — forestry emergency	SEG	A	350,000	350,000		
(cw)	reserve Forestry — Pattison communications	SEG	C	5,000,000	-0-		
(CW)	tower	SEG	C	1,241,700	-0-		
(cx)	Forestry – management plans	SEG	C	316,800	316,800		
(cy)	Forestry – cooperating foresters and	DEG	C	310,000	310,000		
(-3)	private contractors	SEG	C	-0-	-0-		
(cz)	Forestry — management of national						
. ,	forest land	SEG	C	-0-	-0-		
(gt)	Habitat conservation plan fees	SEG	C	9,900	9,900		
(jr)	Rental property and equipment —						
(mi)	maintenance and replacement General program operations —	SEG	C	245,500	245,500		
(1111)	private and public sources	PR	C	183,000	183,000		
(mk)	-	PR-S	С	402,800	402,800		
(mv)				,,,,,,	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,		
(mz)	funds; forestry Forest fire emergencies — federal	SEG	A	49,790,900	49,126,600		
	funds	SEG-F	C	-0-	-0-		
(nz)	General program operations — federal funds	SEG-F	С	1,461,900	1,461,900		
	reactar rands	(2) PROC			1,401,700		
	PROGRAM REVENUE	(2) 1 KOC	JIXAWI I	585,800	585,800		
	OTHER			(183,000)	(183,000)		
	SERVICE			(402,800)	(402,800)		
	SERVICE SEGREGATED REVENUE			58,606,300	51,700,300		
	FEDERAL			(1,461,900)	(1,461,900)		
	OTHER TOTAL ALL SOURCES			(57,144,400)	(50,238,400)		
	TOTAL-ALL SOURCES			59,192,100	52,286,100		
(3)	Public Safety						
(ak)	Law enforcement – snowmobile enforcement and safety training; service funds	PR-S	٨	1,122,100	1 122 100		
(aq)	Law enforcement – snowmobile		A	, ,	1,122,100		
(ar)	enforcement and safety training Law enforcement – boat enforcement	SEG	A	123,400	123,400		
(as)	and safety training Law enforcement – all–terrain	SEG	A	2,759,100	2,759,100		
()	vehicle and utility terrain vehicle						
	enforcement	SEG	A	1,262,600	1,262,600		
(at)	Education and safety programs	SEG	C	337,600	337,600		
(ax)	Law enforcement – water resources enforcement	SEG	A	227,600	227,600		
(ay)	Off-highway motorcycle safety			,	,		
. • /	certification program	SEG	C	-0-	-0-		

2017 As	sembly Bill 64	- 53 -		2017	Wisconsin Act 59
STATUT	E, AGENCY AND PURPOSE	Source	Түре	2017-2018	2018-2019
(bg)	Enforcement – stationary sources	PR	A	85,200	85,200
(ma)	General program operations – state funds	GPR	A	1,469,300	1,469,300
(mi)	General program operations – private and public sources	PR	C	4,200	4,200
(mk)	General program operations – service funds	PR-S	C	-0-	-0-
(mm)		PR-F	C	537,800	537,800
(mq)	General program operations – environmental fund	SEG	A	1,724,900	1,724,900
(mu)	General program operations – state funds	SEG	A	19,067,400	19,812,500
(my)	General program operations – federal funds	SEG-F	C	3,283,700	3,283,700
	141145	(3) PROG			2,202,700
(GENERAL PURPOSE REVENUE	(3) 1100	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	1,469,300	1,469,300
	PROGRAM REVENUE			1,749,300	1,749,300
1	FEDERAL			(537,800)	(537,800)
	OTHER			(89,400)	(89,400)
				* ' '	, , ,
	SERVICE			(1,122,100)	(1,122,100)
	SEGREGATED REVENUE			28,786,300	29,531,400
	FEDERAL			(3,283,700)	(3,283,700)
_	OTHER			(25,502,600)	(26,247,700)
	TOTAL-ALL SOURCES			32,004,900	32,750,000
` '	Environmental Management				
(ac)	Wisconsin River monitoring and	CDD		0	0
(- f)	study	GPR	A	-0- 50 000	-0-
(af)	Water resources – remedial action	GPR	C	50,000	50,000
(ag) (ah)	Water resources – pollution credits Water resources – Great Lakes	PR	C	-0-	-0-
	protection fund	PR	C	214,900	214,900
(ai)	Water resources – water use fees	PR	C	802,000	802,000
(aj)	Water resources – ballast water	PR	С	255,200	255 200
(ng)	discharge permits Water resources management – lake,	PK	C	233,200	255,200
(aq)	river, and invasive species management	SEG	A	2,285,700	2,285,700
(ar)	Water resources – groundwater management	SEG	В	91,900	91,900
(au)	Cooperative remedial action; contributions	SEG	C	-0-	-0-
(av)	Cooperative remedial action; interest on contributions	SEG	S	-0-	-0-
(aw)	Water resources — public health	SEG	A	24,700	24,700
(bL)	Wastewater management – fees	PR	C	133,200	133,200
(bn)	Air management — emission analysis	PR	C	-0-	-0-
(bo)	Air management — permit review and enforcement	PR	A	2,098,500	2,098,500
(bp)	Air waste management — incinerator operator certification	PR	C	-0-	-0-
	-				

PR

PR

SEG

SEG

Α

C

C

Α

76,300

38,100

1,471,800

-0-

76,300

38,100

1,471,800

-0-

Nonferrous metallic mining

Recycling; administration

regulation and administration

Ferrous metallic mining operations

Solid waste management — mining

(gh)

(gi)

(gr)

(hq)

programs

017 Assembly Bill 64		- 55 -	_	2017 W	2017 Wisconsin Act 59	
TATUTI	E, AGENCY AND PURPOSE	Source	Түре	2017-2018	2018-2019	
(hr)	Electronic waste recycling	SEG	C	144,300	144,300	
(ma)	General program operations – state	CDD		0	0	
	funds	GPR	A	-0- 2 424 200	-0- 2 424 200	
	Drinking water and groundwater	GPR	A	2,434,300	2,434,300	
	Water Quality	GPR	A	5,313,300	5,313,300	
	Waste and Materials Management	GPR	A	407,700	407,700	
	Remediation and redevelopment	GPR	A	937,000	937,000	
	Water program management NET APPROPRIATION	GPR	A	748,100 9,840,400	748,100 9,840,400	
(mi)	General program operations – private and public sources	PR	С	186,700	186,700	
(mk)	General program operations – service	TK	C	180,700	180,700	
	funds	PR-S	C	-0-	-0-	
(mm)	General program operations – federal funds	PR-F	С	-0-	-0-	
	Drinking water and groundwater	PR-F	C	4,965,700	4,965,700	
	Water Quality	PR-F	C	8,541,800	8,166,400	
	Air management	PR-F	C	3,178,700	3,178,700	
	Waste and Materials Management	PR-F	C	1,834,100	1,834,100	
	Remediation and redevelopment	PR-F	C	2,457,300	2,457,300	
	NET APPROPRIATION			20,977,600	20,602,200	
(mq)	General program operations –	CEC		0	0	
	environmental fund	SEG	A	-0- 2 270 500	-0- 2 270 500	
	Drinking water and groundwater	SEG	A	2,279,500	2,279,500	
	Water Quality	SEG	A	1,069,200	1,069,200	
	Air management	SEG	A	143,300	143,300	
	Waste and Materials Management	SEG	A	1,304,900	1,304,900	
	Remediation and redevelopment	SEG	A	1,885,700	1,885,700	
	Water program management NET APPROPRIATION	SEG	A	201,700 6,884,300	201,700 6,884,300	
(mr)	General program operations —	CEC		204 200	220 200	
(ms)	nonpoint source General program operations —	SEG	A	294,200	229,200	
	environmental fund; federal funds	SEG-F	C	1,047,000	1,047,000	
(mt)	General program operations –					
	environmental improvement					
	programs; state funds	SEG	A	513,700	513,700	
(mv)	General program operations —					
(mx)	brownfields General program operations – clean	SEG	A	291,300	291,300	
(11111)	water fund program; federal funds	SEG-F	C	757,800	757,800	
(my)	General program operations –	CEC E	C	0	0	
(nz)	environmental fund; federal funds General program operations – safe	SEG-F	С	-0-	-0-	
()	drinking water loan programs;					
	federal funds	SEG-F	C	2,292,400	2,292,400	
		(4) PRO	GRAM TO	TALS		
(GENERAL PURPOSE REVENUE			9,890,400	9,890,400	
F	PROGRAM REVENUE			36,761,500	36,386,100	
	FEDERAL			(20,977,600)	(20,602,200)	
	OTHER			(15,783,900)	(15,783,900)	

-56-

2017 Assembly Bill 64

2017 Wisconsin Act 59

17 As	ssembly Bill 64	<i>–</i> 57 <i>–</i>		2017 Wisconsin Act 59		
STATUT	E, AGENCY AND PURPOSE	Source	Түре	2017-2018	2018-2019	
(ct)	Recreation aids – all–terrain vehicle	BOCKEL			_010 _013	
` ′	project aids; gas tax payment Recreation aids – all–terrain vehicle	SEG	C	1,892,100	1,931,000	
(cu)	project aids	SEG	C	1,670,000	1,670,000	
(cw)	Recreation aids – supplemental snowmobile trail aids	SEG	C	800,000	800,000	
(cx)	Recreation aids – all–terrain vehicle and utility terrain vehicle safety	SEC	S	207.000	207.000	
(cy)	enhancement program Recreation and resource aids, federal	SEG		297,000	297,000	
	funds	SEG-F	C	3,162,100	3,162,100	
(cz)	Resource aids – interpretive center	SEG	A	27,000	27,000	
(da) (dq)	Aids in lieu of taxes – general fund Aids in lieu of taxes – lands acquired	GPR	S	6,840,000	6,930,000	
_	before a specified date	SEG	S	780,000	780,000	
(dr)	Aids in lieu of taxes – lands acquired after a specified date	SEG	S	6,480,000	6,570,000	
(dx)	Resource aids – payment in lieu of taxes; federal	SEG-F	C	440,000	440,000	
(dy)	Resource aids – distribution of closed acreage fees	SEG	A	114,000	-0-	
(ea)	Enforcement aids – spearfishing enforcement	GPR	C	-0-	-0-	
(eq)	Enforcement aids – boating enforcement	SEG	A	1,386,000	1,386,000	
(er)	Enforcement aids – all–terrain vehicle and utility terrain vehicle enforcement	SEG	A	495,000	495,000	
(es)	Enforcement aids – snowmobiling enforcement	SEG	A	396,000	396,000	
(eu)	Recreation aids- utility terrain					
	vehicle project aids	SEG	C	95,600	95,600	
(ex)	Enforcement aids – federal funds	SEG-F	C	-0-	-0-	
(fc) (fq)	Summer tribal youth program Wildlife damage claims and	GPR	A	250,000	250,000	
(fr)	abatement Wildlife abatement and control	SEG	C	2,950,000	2,950,000	
(11)	grants	SEG	В	24,700	24,700	
(fs)	Venison and wild turkey processing	SEG	В	300,000	300,000	
(ft)	Venison and wild turkey processing;			200,000	200,000	
(10)	voluntary contributions	SEG	C	14,800	14,800	
(fv)	Wolf depredation program	SEG	C	-0-	-0-	
(fw)	Resource Aids – Natural Resources Foundation of Wisconsin payments	SEG	С	20,000	20,000	
(gr)	Recreation aids- utility terrain					
	vehicle project aids; gas tax payment	SEG	C	336,700	351,500	
	CENEDAL DUBBOGE BEVENUE	(5) PRO(GRAM TO		7 100 000	
	GENERAL PURPOSE REVENUE			7,090,000	7,180,000	
	SEGREGATED REVENUE			38,111,000	37,952,000	
	FEDERAL			(4,384,300)	(4,384,300)	
	OTHER			(33,726,700)	(33,567,700)	
,	TOTAL–ALL SOURCES			45,201,000	45,132,000	

2017 Wi	2017 Wisconsin Act 59		-	2017	Assembly Bill 64
STATUT	E, AGENCY AND PURPOSE	Source	Түре	2017-2018	2018-2019
(6)	Environmental aids				
(aq)	Environmental aids; nonpoint source	SEG	В	100,000	100,000
(ar)	Environmental aids – lake protection	SEG	C	2,252,600	2,252,600
(as)	Environmental aids – invasive				
	aquatic species and lake monitoring	CEC	D	4.520.100	4.020.100
(-)	and protection	SEG	В	4,529,100	4,029,100
(au)	Environmental aids — river protection; environmental fund	SEG	A	-0-	-0-
(av)	Environmental aids – river	SEC	A	_0_	_0_
(av)	protection; lake monitoring and				
	protection contracts; conservation				
	fund	SEG	В	289,500	289,500
(aw)	Environmental aids – river				
	protection, nonprofit organization				
	contracts	SEG	C	-0-	-0-
(bu)	Financial assistance for responsible	~~~		40.000.000	40.000.000
	units	SEG	A	19,000,000	19,000,000
(bw)	Recycling consolidation grants	SEG	A	1,000,000	1,000,000
(cm)	Environmental aids – federal funds	PR–F	C	800,000	800,000
(cr)	Environmental aids – compensation for well contamination and				
	abandonment	SEG	С	200,000	200,000
(da)	Environmental planning aids – local	SLO	C	200,000	200,000
(ua)	water quality planning	GPR	A	196,400	196,400
(dm)	Environmental planning aids –			,	
` '	federal funds	PR-F	C	150,000	150,000
(dq)	Environmental aids – urban nonpoint				
	source	SEG	В	550,000	500,000
(ef)	Brownfields revolving loan		_	_	_
	repayments	PR	C	-0-	-0-
(eg)	Groundwater mitigation and local	DD	C	0	0
(ah)	assistance Brownfields revolving loan funds	PR	C	-0-	-0-
(eh)	administered for other entity	PR	C	-0-	-0-
(em)	Federal brownfields revolving loan	110	C	-0-	_0_
(CIII)	funds	PR-F	C	1,000,000	1,000,000
(eq)	Environmental aids – dry cleaner			,,	,,
` 1/	environmental response	SEG	В	763,600	763,600
(er)	Vapor control system removal grants	SEG	В	-0-	-0-
(ev)	Reimbursement for disposal of				
	contaminated sediment	SEG	A	-0-	-0-
(fr)	Petroleum storage environmental	ar-~			- - - - - - - - -
/c :	remedial action; awards	SEG	В	7,500,000	7,500,000
(fv)	Removal of underground petroleum	SEC	٨	100 000	100 000
(22)	storage tanks	SEG SEG	A B	100,000 100,000	100,000 -0-
(gs)	Village of Plover grant	(6) PROC			-0-
	GENERAL PURPOSE REVENUE	(U) FROC	JIXAJVI I	196,400	196,400
	PROGRAM REVENUE			1,950,000	1,950,000
	FEDERAL			(1,950,000)	(1,950,000)
	OTHER			(-0-)	(-0-)
	SEGREGATED REVENUE			36,384,800	35,734,800
	· · · · · · ·			, ,	., ,~~

17 A	ssembly Bill 64	- 59 -		2017 Wisconsin Act 59	
STATU	TE, AGENCY AND PURPOSE OTHER	Source	ТүрЕ	2017–2018 (36,384,800)	2018–2019 (35,734,800)
	TOTAL-ALL SOURCES			38,531,200	37,881,200
(7)	DEBT SERVICE AND DEVELOPMENT				
(aa)	Resource acquisition and				
	development – principal repayment	CDD	C	70 715 700	74 220 000
(ad)	and interest	GPR	S	70,715,700	74,329,900
(ad)	Land sales – principal repayment Land acquisition – principal	PR	С	-0-	-0-
(ag)	repayment and interest	PR	C	-0-	-0-
(aq)	Resource acquisition and			v	· ·
(1)	development – principal repayment				
	and interest	SEG	S	100	100
(ar)	Dam repair and removal - principal				
	repayment and interest	SEG	S	522,900	475,700
(at)	Recreation development – principal	CEC	S	0	0
(011)	repayment and interest State forest acquisition and	SEG	3	-0-	-0-
(au)	development – principal repayment				
	and interest	SEG	A	13,500,000	13,500,000
(bq)	Principal repayment and interest –				
	remedial action	SEG	S	2,688,200	2,642,400
(br)	Principal repayment and interest –				
	contaminated sediment	SEG	S	1,940,800	2,012,100
(cb)	Principal repayment and interest –	GPR	S	-0-	-0-
(cc)	pollution abatement bonds Principal repayment and interest –	UFK	S	-0-	_0_
(cc)	combined sewer overflow; pollution				
	abatement bonds	GPR	S	1,511,600	2,013,000
(cd)	Principal repayment and interest -				
	municipal clean drinking water	ann		455.500	456.000
()	grants	GPR	S	177,500	176,200
(cg)	Principal repayment and interest – nonpoint repayments	PR	С	-0-	-0-
(cq)	Principal repayment and interest –	110	C	-0-	-0-
(04)	nonpoint source grants	SEG	S	6,211,500	5,796,800
(cr)	Principal repayment and interest –				
	nonpoint source	SEG	S	2,329,400	2,492,700
(cs)	Principal repayment and interest –		_		
	urban nonpoint source cost–sharing	SEG	S	3,280,000	3,464,800
(ct)	Principal and interest – pollution abatement, environmental fund	SEG	S	6,146,900	7,047,000
(dr)	Petroleum inspection fund – revenue	SEG	S	0,140,900	7,047,000
(ur)	obligation repayment	SEG	S	-0-	-0-
(ea)	Administrative facilities – principal				
	repayment and interest	GPR	S	696,100	693,800
(eq)	Administrative facilities – principal		_		
()	repayment and interest	SEG	S	5,815,600	6,062,900
(er)	Administrative facilities – principal repayment and interest;				
	environmental fund	SEG	S	799,300	835,100
(fa)	Resource maintenance and	- · ·		,	222,230

2017 Wisconsin Act 59		- 60 -		2017	2017 Assembly Bill 64		
STATUT	TE, AGENCY AND PURPOSE	Source	Түре	2017-2018	2018-2019		
(fk)	Resource acquisition and development – service funds; transportation moneys	PR-S	C	-0-	-0-		
(fr)	Resource acquisition and development – boating access to southeastern lakes	SEG	C	92,400	92,400		
(fs)	Resource acquisition and development – state funds	SEG	C	889,100	889,100		
(ft)	Resource acquisition and development – boating access	SEG	C	184,800	184,800		
(fu)	Resource acquisition and development – nonmotorized boating improvements	SEG	C	-0-	-0-		
(fw)	Resource acquisition and development – Mississippi and St. Croix rivers management	SEG	C	57,700	57,700		
(fy)	Resource acquisition and development – federal funds	SEG-F	C	9,112,800	9,112,800		
(gg)	Ice age trail – gifts and grants	PR	C	-0-	-0-		
(gq)	State trails – gifts and grants	SEG	C	-0-	-0-		
(ha)	Facilities acquisition, development and maintenance	GPR	C	144,400	144,400		
(hq)	Facilities acquisition, development and maintenance – conservation fund	SEG	C	372,400	372,400		
(ht)	Property development – conservation fund	SEG	C	-0-	-0-		
(hu)	Parks development – conservation fund	SEG	C	1,000,000	1,000,000		
(jr)	Rental property and equipment – maintenance and replacement	SEG	C	180,000	180,000		
(mc)	Resource maintenance and development – state park, forest, and riverway roads; general fund	GPR	C	-0-	-0-		
(mi)	General program operations – private and public sources	PR	C	-0-	-0-		
(mk)	General program operations – service funds	PR-S	C	-0-	-0-		
(mr)	Resource maintenance and development – state park, forest, and riverway roads; conservation fund	SEG (7) PROC	C	2,000,000 TOTALS	2,000,000		
	GENERAL PURPOSE REVENUE	(/) FROC	JKAWI	74,000,900	78,112,900		
	PROGRAM REVENUE			74,000,900 -0-	78,112,900 -0-		
	OTHER			(-0-)	(-0-)		
	SERVICE			(-0-)	(-0-)		
	SEGREGATED REVENUE			57,123,900	58,218,800		
	FEDERAL			(9,112,800)	(9,112,800)		
	OTHER			(48,011,100)	(49,106,000)		
	TOTAL–ALL SOURCES			131,124,800	136,331,700		
(8)	INTERNAL SERVICES			131,124,000	150,551,700		
(ir)	Promotional activities and publications	SEG	C	82,200	82,200		

)17 A	ssembly Bill 64	- 61 -	_	2017 W	isconsin Act
Statu	TE, AGENCY AND PURPOSE	Source	Түре	2017-2018	2018-2019
(iw)	Statewide recycling administration	SEG	A	447,100	456,600
(ma)	General program operations – state funds	GPR	A	3,397,000	3,427,300
(mg)		PR	A	-0-	-0-
(mi)	General program operations – private and public sources	PR	C	-0-	-0-
(mk)	1	PR-S	C	4,014,300	4,014,300
(mq)		SEG	A	978,700	1,001,700
(mr)	General program operations – environmental improvement fund	SEG	A	382,500	391,500
(mt)	Equipment and services	SEG-S	C	-0-	-0-
(mu)		SEG	A	24,884,900	23,859,800
(mv)	General program operations – environmental fund	SEG	A	2,347,700	2,373,800
(my)		SEG-F	C	1,883,800	1,322,000
(mz)		SEG-F	C	7,460,700	7,501,100
(ni)	Geographic information systems, general program operations – other funds	PR	С	32,700	32,700
(nk)	Geographic information systems, general program operations – service		C	22,700	52,700
	funds	PR-S	C	1,139,100	1,139,100
(zq)	Gifts and donations	SEG	C	-0-	-0-
		(8) PRO	GRAM TO		
	GENERAL PURPOSE REVENUE			3,397,000	3,427,300
	PROGRAM REVENUE			5,186,100	5,186,100
	OTHER			(32,700)	(32,700)
	SERVICE			(5,153,400)	(5,153,400)
	SEGREGATED REVENUE			38,467,600	36,988,700
	FEDERAL			(9,344,500)	(8,823,100)
	OTHER			(29,123,100)	(28,165,600)
	SERVICE			(-0-)	(-0-)
	TOTAL-ALL SOURCES			47,050,700	45,602,100
(9)	EXTERNAL SERVICES				•
(aq)	Water resources management — lake, river, and invasive species				
(as)	management Water resources — trading water	SEG	A	822,700	822,700
(at)	pollution credits Watershed — nonpoint source	SEG	С	-0-	-0-
(bg)	contracts Water regulation and zoning —	SEG	В	767,600	767,600
	computer access fees	PR	C	-0-	-0-
(bi)	Water regulation and zoning — fees	PR	C	1,302,500	1,302,500
(bj) (bm)	Storm water management — fees Wetland restoration — fees;	PR	A	1,646,700	1,646,700
	payments	PR	C	-0-	-0-

SEG

SEG

SEG-F

Α

A

 \mathbf{C}

940,800

194,500

1,316,700

940,800

194,500

1,316,700

environmental fund

recreation

(mx)

(mw) Aids administration - snowmobile

fund program; federal funds

Aids administration – clean water

2017 Assembly Bill 64		- 63 -	-	2017 W	2017 Wisconsin Act 59		
STATU'	TE, AGENCY AND PURPOSE	Source	Түре	2017-2018	2018-2019		
(my)	General program operations - federal						
	funds	SEG-F	C	439,800	439,800		
(mz)		SEG-F	C	946,400	946,400		
(nq)	Aids administration – dry cleaner	ana		00.000	00.000		
()	environmental response	SEG	A	89,800	89,800		
(ny)	Aids administration – safe drinking water loan programs; federal funds	SEG-F	C	171,400	171,400		
	water toan programs, rederar funds		GRAM TO		171,400		
	GENERAL PURPOSE REVENUE	(<i>)</i>) TROC	JIM IVI TO	8,707,100	8,707,100		
	PROGRAM REVENUE			11,485,500	11,485,500		
	FEDERAL			(4,123,200)	(4,123,200)		
	OTHER			(4,547,600)	(4,547,600)		
	SERVICE			(2,814,700)	(2,814,700)		
	SEGREGATED REVENUE			21,136,000	21,136,000		
	FEDERAL			(2,874,300)	(2,874,300)		
	OTHER			(18,261,700)	(18,261,700)		
	TOTAL-ALL SOURCES			41,328,600	41,328,600		
	20	0.370 DEPA	RTMENT	TOTALS			
	GENERAL PURPOSE REVENUE			107,357,800	111,590,100		
	PROGRAM REVENUE			59,330,600	58,955,200		
	FEDERAL			(27,816,400)	(27,441,000)		
	OTHER			(21,413,300)	(21,413,300)		
	SERVICE			(10,100,900)	(10,100,900)		
	SEGREGATED REVENUE			382,554,800	376,276,000		
	FEDERAL			(53,668,600)	(53,709,000)		
	OTHER			(328,886,200)	(322,567,000)		
	SERVICE TOTAL ALL SOURCES			(-0-)	(-0-)		
20 373	TOTAL-ALL SOURCES Fox River Navigational System Author	ority		549,243,200	546,821,300		
(1)	INITIAL COSTS	ority					
(g)	Administration, operation, repair,						
(5)	and rehabilitation	PR	C	-0-	-0-		
(r)	Establishment and operation	SEG	C	125,400	125,400		
	1	(1) PROC	GRAM TO	TALS	,		
	PROGRAM REVENUE			-0-	-0-		
	OTHER			(-0-)	(-0-)		
	SEGREGATED REVENUE			125,400	125,400		
	OTHER			(125,400)	(125,400)		
	TOTAL-ALL SOURCES			125,400	125,400		
		0.373 DEPA	RTMENT				
	PROGRAM REVENUE			-0-	-0-		
	OTHER			(-0-)	(-0-)		
	SEGREGATED REVENUE			125,400	125,400		
	OTHER TOTAL ALL SOURCES			(125,400)	(125,400)		
20.255	TOTAL-ALL SOURCES			125,400	125,400		
	Lower Fox River Remediation Author	ority					
(1)	INITIAL COSTS Initial costs	GPR	В	-0-	-0-		
(a)	minai cosis		в GRAM TO		-0-		
	GENERAL PURPOSE REVENUE	(1) FROC	DIVINI IO	-0-	-0-		
	TOTAL-ALL SOURCES			-0-	-0- -0-		
				v	v		

STATUT	E, AGENCY AND PURPOSE	Source 0.375 DEF		2017–2018 T TOTALS	2018–2019
	GENERAL PURPOSE REVENUE			-0-	-0-
	TOTAL-ALL SOURCES			-0-	-0-
20.380	Tourism, Department of				
(1)	TOURISM DEVELOPMENT AND PROMOTION	N			
(a)	General program operations	GPR	A	2,464,600	2,506,500
(b)	Tourism marketing; general purpose				
	revenue	GPR	В	1,909,600	1,827,100
(g)	Gifts, grants and proceeds	PR	C	100	100
(h)	Tourism promotion; sale of surplus				
	property receipts	PR	C	-0-	-0-
(ig)	Golf promotion	PR	C	-0-	-0-
(ir)	Payments to the WPGA Junior				
	Foundation	PR	C	-0-	-0-
(j)	Tourism promotion – private and				
	public sources	PR	C	99,000	99,000
(k)	Sale of materials or services	PR-S	C	-0-	-0-
(ka)	Sale of materials and services–local		_	_	_
	assistance	PR-S	C	-0-	-0-
(kb)	Sale of materials and				
	services-individuals and	DD C	C	0	0
(1)	organizations	PR-S	C	-0- 210 200	-0- 210 200
(kc)	Marketing clearinghouse charges	PR-S	A	319,300	319,300
(kg)	Tourism marketing; gaming revenue	PR-S	В	8,967,100	8,967,100
(km)	Grants for regional tourist	DD C	A	160,000	160,000
()	information centers	PR-S	A	160,000	160,000
(m)	Federal aid, state operations	PR–F PR–F	C C	-0- -0-	-0- -0-
(n)	Federal aid, local assistance	РК-Г	C	-0-	-0-
(o)	Federal aid, individuals and organizations	PR-F	С	-0-	-0-
(q)	Administrative	I IX-I	C		_0_
(4)	services—conservation fund	SEG	A	12,100	12,100
(w)	Tourism marketing; transportation	SEC	11	12,100	12,100
(w)	fund	SEG	В	1,591,400	1,591,400
	Tunu		OGRAM T		1,001,100
	GENERAL PURPOSE REVENUE	(1) 1110	, 0111111	4,374,200	4,333,600
	PROGRAM REVENUE			9,545,500	9,545,500
	FEDERAL			(-0-)	(-0-)
	OTHER			(99,100)	(99,100)
	SERVICE			(9,446,400)	(9,446,400)
	SEGREGATED REVENUE			1,603,500	1,603,500
	OTHER			(1,603,500)	(1,603,500)
	TOTAL-ALL SOURCES			15,523,200	15,482,600
(3)	SUPPORT OF ARTS PROJECTS			15,525,200	13,102,000
(a)	General program operations	GPR	A	256,400	262,100
(b)	State aid for the arts	GPR	A	518,800	359,300
(c)	Portraits of governors	GPR	A	-0-	-0-
(d)	Challenge grant program	GPR	A	-0-	-0-
(e)	High Point fund	GPR	A	-0-	-0-
(f)	Wisconsin regranting program	GPR	A	116,700	116,700
(g)	Gifts and grants; state operations	PR	C	20,000	20,000
(5)	one grand, suite operations		_	20,000	20,000

17 Assembly Bill 64		- 65 -	_	2017 Wisconsin A	
TATUT	TE, AGENCY AND PURPOSE	Source	Түре	2017-2018	2018-2019
(h)	Gifts and grants; aids to individuals				
	and organizations	PR	C	-0-	-0-
(j)	Support of arts programs	PR	C	-0-	-0-
(km)	State aid for the arts; Indian gaming				
	receipts	PR-S	A	24,900	24,900
(m)	Federal grants; state operations	PR-F	C	239,200	239,200
(o)	Federal grants; aids to individuals				
	and organizations	PR-F	C	524,500	524,500
		(3) PROC	GRAM TOT	ALS	
	GENERAL PURPOSE REVENUE			891,900	738,100
	PROGRAM REVENUE			808,600	808,600
	FEDERAL			(763,700)	(763,700)
	OTHER			(20,000)	(20,000)
	SERVICE			(24,900)	(24,900)
	TOTAL-ALL SOURCES			1,700,500	1,546,700
	2	20.380 DEPA	ARTMENT T	OTALS	
	GENERAL PURPOSE REVENUE			5,266,100	5,071,700
	PROGRAM REVENUE			10,354,100	10,354,100
	FEDERAL			(763,700)	(763,700)
	OTHER			(119,100)	(119,100
	SERVICE			(9,471,300)	(9,471,300
	SEGREGATED REVENUE			1,603,500	1,603,500
	OTHER			(1,603,500)	(1,603,500)
	TOTAL-ALL SOURCES			17,223,700	17,029,300
	Kickapoo Reserve Management Boa	rd		, ,	, ,
1)	KICKAPOO VALLEY RESERVE				
(g)	Kickapoo reserve management				
(0)	board; program services	PR	C	166,000	166,000
(h)	Kickapoo reserve management				
	board; gifts and grants	PR	C	-0-	-0-
(k)	Kickapoo valley reserve; law				
	enforcement services	PR-S	A	69,400	69,400
(m)	Kickapoo reserve management				
•	board; federal aid	PR-F	C	-0-	-0-
(q)	Kickapoo reserve management				
(1)	=	SEG	Α.	450 000	453,000
(4)	board; general program operations	SEG	A	453,000	,
	Kickapoo valley reserve; aids in lieu				,
		SEG	S S	453,000 280,000	,
	Kickapoo valley reserve; aids in lieu	SEG		280,000	,
(r)	Kickapoo valley reserve; aids in lieu	SEG	S	280,000	280,000
(r)	Kickapoo valley reserve; aids in lieu of taxes	SEG	S	280,000 ALS	280,000 235,400
(r)	Kickapoo valley reserve; aids in lieu of taxes PROGRAM REVENUE	SEG	S	280,000 ALS 235,400	280,000 235,400 (-0-
(r)	Kickapoo valley reserve; aids in lieu of taxes PROGRAM REVENUE FEDERAL	SEG	S	280,000 ALS 235,400 (-0-)	280,000 235,400 (-0- (166,000
(r)	Kickapoo valley reserve; aids in lieu of taxes PROGRAM REVENUE FEDERAL OTHER	SEG	S	280,000 PALS 235,400 (-0-) (166,000)	280,000 235,400 (-0- (166,000) (69,400)
(r)	Kickapoo valley reserve; aids in lieu of taxes PROGRAM REVENUE FEDERAL OTHER SERVICE	SEG	S	280,000 ALS 235,400 (-0-) (166,000) (69,400)	280,000 235,400 (-0- (166,000 (69,400 733,000
(r)	Kickapoo valley reserve; aids in lieu of taxes PROGRAM REVENUE FEDERAL OTHER SERVICE SEGREGATED REVENUE	SEG	S	280,000 PALS 235,400 (-0-) (166,000) (69,400) 733,000	280,000 235,400 (-0- (166,000 (69,400 733,000 (733,000
(r)	Kickapoo valley reserve; aids in lieu of taxes PROGRAM REVENUE FEDERAL OTHER SERVICE SEGREGATED REVENUE OTHER TOTAL-ALL SOURCES	SEG (1) PROO	S	280,000 PALS 235,400 (-0-) (166,000) (69,400) 733,000 (733,000) 968,400	280,000 235,400 (-0- (166,000 (69,400 733,000 (733,000
(r)	Kickapoo valley reserve; aids in lieu of taxes PROGRAM REVENUE FEDERAL OTHER SERVICE SEGREGATED REVENUE OTHER TOTAL-ALL SOURCES	SEG (1) PROO	S GRAM TOT	280,000 PALS 235,400 (-0-) (166,000) (69,400) 733,000 (733,000) 968,400	280,000 235,400 (-0-) (166,000) (69,400) 733,000 (733,000) 968,400
(r)	Kickapoo valley reserve; aids in lieu of taxes PROGRAM REVENUE FEDERAL OTHER SERVICE SEGREGATED REVENUE OTHER TOTAL-ALL SOURCES	SEG (1) PROO	S GRAM TOT	280,000 PALS 235,400 (-0-) (166,000) (69,400) 733,000 (733,000) 968,400 POTALS 235,400	280,000 235,400 (-0-) (166,000) (69,400) 733,000 (733,000) 968,400 235,400
(r)	Kickapoo valley reserve; aids in lieu of taxes PROGRAM REVENUE FEDERAL OTHER SERVICE SEGREGATED REVENUE OTHER TOTAL-ALL SOURCES PROGRAM REVENUE	SEG (1) PROO	S GRAM TOT	280,000 ALS 235,400 (-0-) (166,000) (69,400) 733,000 (733,000) 968,400 OTALS	280,000 235,400 (-0-) (166,000) (69,400) 733,000 (733,000) 968,400

017 Wisconsin Act 59		- 66 -		2017 Assembly Bill 64		
;	TE, A GENCY AND P URPOSE SEGREGATED REVENUE OTHER	Source	ТүрЕ	2017–2018 733,000 (733,000)	2018–2019 733,000 (733,000)	
	TOTAL-ALL SOURCES			968,400	968,400	
20.395	Transportation, Department of					
(1)	AIDS					
(ar)	Corrections of transportation aid payments	SEG	S	-0-	-0-	
(as)	Transportation aids to counties, state funds	SEG	A	101,573,600	111,093,800	
(at)	Transportation aids to municipalities, state funds	SEG	A	334,949,900	348,639,300	
(bq)	Intercity bus assistance, state funds	SEG	C	-0-	-0-	
(bs)	Transportation employment and mobility, state funds	SEG	C	332,600	332,600	
(bv)	Transit and other					
	transportation-related aids, local funds	SEG-L	C	110,000	110,000	
(bx)	Transit and other transportation–related aids, federal					
	funds	SEG-F	C	20,136,100	20,538,800	
(ck)	Tribal elderly transportation grants	PR-S	A	396,000	396,000	
(cq)	Seniors and individuals with disabilities specialized transportation	SEC	C	012.700	012.700	
(cr)	aids, state funds Seniors and individuals with disabilities specialized transportation	SEG	C	912,700	912,700	
(cv)	county aids, state funds Seniors and individuals with disabilities specialized transportation	SEG	A	14,193,900	14,477,800	
(cx)	aids, local funds Seniors and individuals with	SEG-L	С	605,500	605,500	
` /	disabilities specialized transportation aids, federal funds	SEG-F	C	2,938,100	2,996,900	
(ex)	Highway safety, local assistance, federal funds	SEG-F	C	6,734,700	6,869,400	
(fq)	Connecting highways aids, state funds	SEG	A	12,063,500	12,063,500	
(fs)	Disaster damage aids, state funds	SEG	S	1,064,000	1,000,000	
(ft)	Lift bridge aids, state funds	SEG	В	2,659,200	2,659,200	
(fu)	County forest road aids, state funds	SEG	A	284,700	284,700	
(gq)	Expressway policing aids, state funds	SEG	A	1,023,900	1,023,900	
(gt)	Soo Locks improvements, state funds	SEG	A	-0-	-0-	
(hq)	Paratransit aids	SEG	A	2,750,000	2,750,000	
(hr)	Tier B transit operating aids, state funds	SEG	A	24,486,700	24,486,700	
(hs)	Tier C transit operating aids, state funds	SEG	A	5,188,900	5,188,900	
(ht)	Tier A–1 transit operating aids, state funds	SEG	A	64,193,900	64,193,900	

017 Assembly Bill 64 Statute, Agency and Purpose		- 67 -		2017 Wisconsin Act 5	
		Source	Түре	2017-2018	2018-2019
(hw)	Tier A-3 transit operating aids, state				
` /	funds	SEG	A	-0-	-0-
(ig)	Professional football stadium				
	maintenance and operating costs,				
	state funds	PR	C	450,000	450,000
(ih)	Child abuse and neglect prevention,				
	state funds	PR	C	125,000	125,000
		(1) PROC	GRAM TO		
	PROGRAM REVENUE			971,000	971,000
	OTHER			(575,000)	(575,000
	SERVICE			(396,000)	(396,000
	SEGREGATED REVENUE			613,069,900	637,095,600
	FEDERAL			(29,808,900)	(30,405,100
	OTHER			(582,545,500)	(605,975,000
	LOCAL TOTAL ALL SOURCES			(715,500)	(715,500
	TOTAL-ALL SOURCES			614,040,900	638,066,600
(2)	LOCAL TRANSPORTATION ASSISTANCE				
(aq)	Accelerated local bridge	SEC	C	Λ	0
(071)	improvement assistance, state funds	SEG	С	-0-	-0-
(av)	Accelerated local bridge improvement assistance, local funds	SEG-L	С	-0-	-0-
(ax)	Accelerated local bridge	OLU-L	C	_0_	-0-
(ax)	improvement assistance, federal				
	funds	SEG-F	C	-0-	-0-
(bq)	Rail service assistance, state funds	SEG	C	1,262,600	1,262,600
(bt)	Freight rail preservation	SEG	C	-0-	-0-
(bu)	Freight rail infrastructure	·- • -	-	v	Ü
\- ·-/	improvements, state funds	SEG	C	-0-	-0-
(bv)	Rail service assistance, local funds	SEG-L	C	500,000	500,000
(bw)	Freight rail assistance loan				
	repayments, local funds	SEG-L	C	4,000,000	4,000,000
(bx)	Rail service assistance, federal funds	SEG-F	C	-0-	-0-
(cq)	Harbor assistance, state funds	SEG	C	3,851,000	651,000
(cr)	Rail passenger service, state funds	SEG	C	7,000,000	6,800,000
(cs)	Harbor assistance, federal funds	SEG-F	C	-0-	-0-
(ct)	Passenger railroad station				
	improvement and commuter rail	a= ~		_	
	transit system grants, state funds	SEG	В	-0-	-0-
(cu)	Passenger railroad station				
	improvement and commuter rail	CEC I	C	^	0
(011)	transit system grants, local funds	SEG-L	C	-0- -0-	-0- 0
(cv)	Rail passenger service, local funds Harbor assistance, local funds	SEG-L SEG-L	C	-0- -0-	-0- -0-
(cw)	Rail passenger service, federal funds	SEG-L SEG-F	C C	-0- -0-	-0- -0-
(cx)		SEG-F SEG	C	18,942,500	
(dq)	Aeronautics assistance, state funds	SEG SEG	A		13,242,500
(ds)	Aviation career education, state funds	SEG-L	A C	178,800 42,000,000	178,800 42,000,000
$(d\mathbf{v})$	Aeronautics assistance, local funds Aeronautics assistance, federal funds	SEG-L SEG-F	C	56,125,100	
(dx)		SEU-L	C	30,123,100	56,125,100
(eq)	Highway and local bridge	SEG	C	19,496,300	18,470,600
(01/)	improvement assistance, state funds Local bridge improvement	SEC	C	17,470,300	10,470,000
(ev)	assistance, local funds	SEG-L	C	11,157,600	11,157,600
	assistance, local fullus	SLU-L	C	11,137,000	11,137,000

2017 Wi	2017 Wisconsin Act 59		_	2017	2017 Assembly Bill 64		
STATUT	E, AGENCY AND PURPOSE	Source	Түре	2017-2018	2018-2019		
(ex)	Local bridge improvement assistance, federal funds	SEG-F	C	24,363,700	24,363,700		
(fb)	Local roads for job preservation, state funds	GPR	C	-0-	-0-		
(fq)	Local transportation facility improvement assistance, state funds	SEG	C	-0-	-0-		
(fr)	Local roads improvement program, state funds	SEG	C	17,865,600	17,865,600		
(ft)	Local roads improvement program; discretionary grants, state funds	SEG	C	15,167,400	15,167,400		
(fv)	Local transportation facility improvement assistance, local funds	SEG-L	C	43,898,600	43,898,600		
(fx)	Local transportation facility improvement assistance, federal funds	SEG-F	C	72,211,300	72,211,300		
(fz)	Local roads for job preservation, federal funds	SEG-F	C	-0-	-0-		
(gj)	Railroad crossing protection installation and maintenance, state funds	SEG	С	-0-	-0-		
(gq)	Railroad crossing improvement and protection maintenance, state funds	SEG	A	2,112,000	2,112,000		
(gr)	Railroad crossing improvement and protection installation, state funds	SEG	C	1,595,700	1,595,700		
(gs)	Railroad crossing repair assistance, state funds	SEG	C	234,700	234,700		
(gv)	Railroad crossing improvement, local funds	SEG-L	C	-0-	-0-		
(gx)	Railroad crossing improvement, federal funds	SEG-F	C	3,291,800	3,291,800		
(hq)	Multimodal transportation studies, state funds	SEG	C	-0-	-0-		
(hx)	Multimodal transportation studies, federal funds	SEG-F	C	-0-	-0-		
(iq)	Transportation facilities economic assistance and development, state funds	SEG	C	3,402,600	3,402,600		
(iv)	Transportation facilities economic assistance and development, local funds	SEG-L	С	3,588,700	3,588,700		
(iw)	Transportation facility improvement loans, local funds	SEG-L	C	-0-	-0-		
(ix)	Transportation facilities economic assistance and development, federal funds	SEG-F	С	-0-	-0-		
(js)	Transportation alternatives program, state funds	SEG	C	20,000	-0-		
(jv)	Transportation alternatives program, local funds	SEG-L	C	2,012,300	2,012,300		
(jx)	Transportation alternatives program, federal funds	SEG-F	C	7,049,300	7,049,300		
(kv)	Congestion mitigation and air quality improvement, local funds	SEG-L	C	3,124,700	3,124,700		

2017 Assembly Bill 64		- 69 -	_	2017 W	Visconsin Act 5	
STATUT	TE, AGENCY AND PURPOSE	Source Type		2017-2018	2018-2019	
(kx)	Congestion mitigation and air quality					
	improvement, federal funds	SEG-F	C	10,719,000	10,719,000	
(mq)	Astronautics assistance, state funds	SEG	C	-0-	-0-	
(mv)	Astronautics assistance, local funds	SEG-L	C	-0-	-0-	
(mx)	Astronautics assistance, federal funds	SEG-F	C	-0-	-0-	
(ph)	Transportation infrastructure loans, gifts and grants	SEG	C	-0-	-0-	
(pq)	Transportation infrastructure loans, state funds	SEG	C	4,600	4,600	
(pu)	Transportation infrastructure loans, service funds	SEG-S	C	-0-	-0-	
(pv)	Transportation infrastructure loans, local funds	SEG-L	C	-0-	-0-	
(px)	Transportation infrastructure loans, federal funds	SEG-F	С	-0-	-0-	
	reactat tutius	(2) PRO		•	-0-	
	GENERAL PURPOSE REVENUE	(2) TKO	JKAWI I	-0-	-0-	
	SEGREGATED REVENUE			375,175,900	365,030,200	
	FEDERAL			(173,760,200)	(173,760,200)	
	OTHER			(91,133,800)	(80,988,100)	
	SERVICE			(-0-)	(-0-)	
	LOCAL			(110,281,900)	(110,281,900)	
	TOTAL-ALL SOURCES			375,175,900	365,030,200	
(3)	STATE HIGHWAY FACILITIES			373,173,900	303,030,200	
(aq)	Southeast Wisconsin freeway megaprojects, state funds	SEG	C	6,488,300	11,660,100	
(av)	Southeast Wisconsin freeway	SEC	C	0,400,300	11,000,100	
(av)	megaprojects, local funds	SEG-L	C	-0-	-0-	
(ax)	Southeast Wisconsin freeway	DLG L	C	· ·	Ü	
(un)	megaprojects, federal funds	SEG-F	C	53,895,000	29,138,100	
(bq)	Major highway development, state	SEC 1	C	22,022,000	25,130,100	
(04)	funds	SEG	C	50,676,800	22,867,100	
(br)	Major highway development, service			, ,	, ,	
` /	funds	SEG-S	C	48,232,700	66,587,300	
(bv)	Major highway development, local					
	funds	SEG-L	C	-0-	-0-	
(bx)	Major highway development, federal funds	SEG-F	C	166,159,900	209,176,200	
(cq)	State highway rehabilitation, state funds	SEG	C	387,002,400	383,602,300	
(cr)	Southeast Wisconsin freeway rehabilitation, state funds	SEG	C	-0-	-0-	
(ct)	Owner controlled insurance program, service funds	SEG-S	C	-0-	-0-	
(cv)	State highway rehabilitation, local funds	SEG-L	C	2,059,200	2,059,200	
(cw)	Southeast Wisconsin freeway rehabilitation, local funds	SEG-L	C	-0-	-0-	
(cx)	State highway rehabilitation, federal funds	SEG-F	C	424,783,000	424,044,700	
(cy)	Southeast Wisconsin freeway rehabilitation, federal funds	SEG-F	C	-0-	-0-	

2017 Wi	2017 Wisconsin Act 59		_	2017 Assemb	
STATUT	E, AGENCY AND PURPOSE	Source	Түре	2017-2018	2018-2019
(dq)	Major interstate bridge construction, state funds	SEG	C	8,000,000	-0-
(dr)	High-cost state highway bridge projects, state funds	SEG	C	-0-	-0-
(dv)	Major interstate bridge construction, local funds	SEG-L	C	-0-	-0-
(dw)	High-cost state highway bridge projects, local funds	SEG-L	С	-0-	-0-
(dx)	Major interstate bridge construction, federal funds	SEG-F	С	-0-	-0-
(dy)	High-cost state highway bridge projects, federal funds	SEG-F	С	-0-	-0-
(eg)	Supplement from sponsorship agreements, state funds	PR	C	10,500	10,500
(eq)	Highway system management and operations, state funds	SEG	C	100,617,400	98,594,400
(er)	State—owned lift bridge operations and maintenance, state funds	SEG	A	2,380,100	2,380,100
(es)	Routine maintenance activities, state				
(et)	funds Intelligent transportation systems and	SEG	С	185,366,500	188,366,500
(eu)	traffic control signals, state funds Intelligent transportation systems and	SEG	С	10,000,000	10,000,000
(ev)	traffic control signals, local funds Highway system management and	SEG-L	С	-0-	-0-
(ew)	operations, local funds Routine maintenance activities, local	SEG-L	C	1,900,000	1,900,000
(ex)	funds Highway system management and	SEG-L	C	-0-	-0-
	operations, federal funds Routine maintenance activities,	SEG-F	C	1,102,500	1,102,500
(ey)	federal funds	SEG-F	C	-0-	-0-
(ez)	Intelligent transportation systems and traffic control signals, federal funds	SEG-F	C	-0-	-0-
(iq)	Administration and planning, state funds	SEG	A	14,767,600	14,767,600
(ir)	Disadvantaged business mobilization assistance, state funds	SEG	C	-0-	-0-
(iv)	Administration and planning, local funds	SEG-L	C	-0-	-0-
(ix)	Administration and planning, federal funds	SEG-F	С	3,521,500	3,521,500
(jg)	Surveying reference station system	PR	C	470,000	590,000
(jh)	Utility facilities within highway	DD	C	270 700	270 700
(jj)	rights-of-way, state funds Damage claims	PR PR	C C	279,700 4,128,100	279,700 4,087,200
(js)	Telecommunications services,	1 IX	C	4,120,100	4,007,200
(10)	service funds	SEG-S	C	-0-	-0-
		(3) PROC	GRAM T		
]	PROGRAM REVENUE			4,888,300	4,967,400
	OTHER			(4,888,300)	(4,967,400)
;	SEGREGATED REVENUE			1,466,952,900	1,469,767,600

2017 Assembly Bill 64		- 71 -	-	2017 Wisconsin Act 59		
Statut	TE, AGENCY AND PURPOSE	Source	Түре	2017-2018	2018-2019	
	FEDERAL			(649,461,900)	(666,983,000)	
	OTHER			(765,299,100)	(732,238,100)	
	SERVICE			(48,232,700)	(66,587,300)	
	LOCAL			(3,959,200)	(3,959,200)	
	TOTAL-ALL SOURCES			1,471,841,200	1,474,735,000	
(4)	GENERAL TRANSPORTATION OPERATIONS			1,471,041,200	1,474,733,000	V 04
		GPR	В	150,000	-0-	Vet In I
(ab)	Transportation projects commission	GPK	D	69.154.800	-0-	Vet
(aq)	Departmental management and operations, state funds	SEG	A	66,654,800	65,528,900	In I
(ar)	Minor construction projects, state					
	funds	SEG	C	-0-	-0-	
(as)	Transit safety oversight, state funds	SEG	C	72,700	72,700	
(at)	Capital building projects, service					
	funds	SEG-S	C	4,540,000	4,540,000	
(av)	Departmental management and					
	operations, local funds	SEG-L	C	369,000	369,000	
(ax)	Departmental management and					
	operations, federal funds	SEG-F	C	15,053,000	15,081,700	
(ay)	Transit safety oversight, federal					
	funds	SEG-F	C	299,000	305,000	
(ch)	Gifts and grants	SEG	C	-0-	-0-	
(dq)	Demand management	SEG	A	393,600	393,600	
(eq)	Data processing services, service					
	funds	SEG-S	C	15,034,900	15,034,900	
(er)	Fleet operations, service funds	SEG-S	C	12,570,700	12,570,700	
(es)	Other department services,					
	operations, service funds	SEG-S	C	5,139,000	5,139,000	
(et)	Equipment acquisition	SEG	A	-0-	-0-	
(ew)	Operating budget supplements, state					
	funds	SEG	C	-0-	-0-	
		(4) PROC	GRAM TO	TALS		
	GENERAL PURPOSE REVENUE			150,000	-0-	
	SEGREGATED REVENUE			122,626,700	119,035,500	
	FEDERAL			(15,352,000)	(15,386,700)	
	OTHER			(69,621,100)	(65,995,200)	
	SERVICE			(37,284,600)	(37,284,600)	
	LOCAL			(369,000)	(369,000)	
	TOTAL-ALL SOURCES			122,776,700	119,035,500	
(5)	MOTOR VEHICLE SERVICES AND ENFORCE	MENT		122,770,700	115,000,000	
(cg)	Convenience fees, state funds	PR	C	218,400	118,400	
(ch)	Repaired salvage vehicle	110	C	210,100	110,100	
(CII)	examinations, state funds	PR	C	145,900	145,900	
(ci)	Breath screening instruments, state	110	C	113,500	113,500	
(C1)	funds	PR-S	C	299,200	299,200	
(cj)	Vehicle registration, special group	110		277,200	277,200	
(0)	plates, state funds	PR	C	-0-	-0-	
(cL)	Football plate licensing fees, state	110		· ·	Ü	
(CL)	funds	PR	C	-0-	-0-	
			\sim	U	v	
(ca)						
(cq)	Vehicle registration, inspection and maintenance, driver licensing and					

2017 Wi	2017 Wisconsin Act 59		_	2017 A	2017 Assembly Bill 64		
	E, AGENCY AND PURPOSE	Source	Түре	2017-2018	2018-2019		
(cx)	Vehicle registration and driver licensing, federal funds	SEG-F	C	1,200,800	1,224,700		
(da)	State traffic patrol equipment, general fund	GPR	A	3,550,000	-0-		
(dg)	Escort, security and traffic enforcement services, state funds	PR	C	350,200	328,300		
(dh)	Traffic academy tuition payments, state funds	PR	С	655,400	655,400		
(di)	Chemical testing training and services, state funds	PR-S	A	1,578,300	1,578,300		
(dk)	Public safety radio management, service funds	PR-S	С	1,001,000	1,001,000		
(dL)	Public safety radio management, state funds	PR	C	182,900	160,900		
(dq)	Vehicle inspection, traffic enforcement and radio management,			,	,		
	state funds	SEG	A	62,902,200	64,319,500		
(dr)	Transportation safety, state funds	SEG	A	1,840,200	1,840,200		
(dx)	Vehicle inspection and traffic enforcement, federal funds	SEG-F	C	4,662,000	4,747,000		
(dy)	Transportation safety, federal funds	SEG-F	C	5,022,800	5,123,400		
(eg)	Payments to the Wisconsin Lions Foundation	PR	C	7,000	7,000		
(eh)	Motorcycle safety program supplement, state funds	PR	C	38,300	38,300		
(ei)	Payments to Wisconsin Trout Unlimited	PR	C	-0-	-0-		
(ej)	Baseball plate licensing fees, state	PR	С	5,000	5,000		
(ek)	funds Safe–ride grant program; state funds	PR-S	C	401,400	161,400		
(ek)	Payments resulting from the issuance of certain special plates	PR	C	5,000	5,000		
(fg)	Payments to the Boy Scouts of	ΓK	C	3,000	3,000		
(1g)	America National Foundation	PR	C	5,000	5,000		
(fh)	Payments to Whitetails Unlimited	PR	C	5,000	5,000		
(fi)	Payments to the Wisconsin Rocky Mountain Elk Foundation	PR	C	5,000	5,000		
(fj)	Payments to Wisconsin Organization of Nurse Executives	PR	C	5,000	5,000		
(gg)	Basketball plate payments to the Milwaukee Bucks Foundation	PR	С	5,000	5,000		
(gh)	Payment to Midwest Athletes Against Childhood Cancer	PR	С	5,000	5,000		
(gi)	Payments to the Wisconsin Women's Health Foundation	PR	С	-0-	-0-		
(gj)	Payments to Donate Life Wisconsin	PR	C	-0-	-0-		
(hq)	Motor vehicle emission inspection and maintenance program; contractor costs and equipment grants; state	-	-	, and the second	Ü		
	funds	SEG	A	3,193,300	3,193,300		

2017 Assembly Bill 64		− 73 −		2017 V	2017 Wisconsin Act 59	
STATU: (hx)	TE, AGENCY AND PURPOSE Motor vehicle emission inspection	Source	Түре	2017–2018	2018-2019	
,	and maintenance programs, federal funds	SEG-F	C	-0-	-0-	
(iv)	Municipal and county registration fee, local funds	SEG-L	C	-0-	-0-	
		(5) PROC	GRAM TO	OTALS		
	GENERAL PURPOSE REVENUE			3,550,000	-0-	
	PROGRAM REVENUE			4,918,000	4,534,100	
	OTHER			(1,638,100)	(1,494,200)	
	SERVICE			(3,279,900)	(3,039,900)	
	SEGREGATED REVENUE			151,727,700	153,462,000	
	FEDERAL			(10,885,600)	(11,095,100)	
	OTHER			(140,842,100)	(142,366,900)	
	LOCAL			(-0-)	(-0-)	
	TOTAL-ALL SOURCES			160,195,700	157,996,100	
(6)	DEBT SERVICES					
(ae)	Principal repayment and interest, contingent funding of major highway and rehabilitation projects, state	CDD	C	14 820 800	14.522.500	
(- ()	funds	GPR	S	14,829,800	14,532,500	
(af)	Principal repayment and interest, local roads for job preservation program, major highway and rehabilitation projects, southeast megaprojects, state funds	GPR	S	111,754,000	102,091,500	
(aq)	Principal repayment and interest, transportation facilities, state highway rehabilitation, major	OI K	5	111,754,000	102,001,300	
	highway projects, state funds	SEG	S	58,096,100	60,870,300	
(ar)	Principal repayment and interest, buildings, state funds	SEG	S	28,700	29,000	
(au)	Principal repayment and interest, southeast rehabilitation projects, southeast megaprojects, and high —					
	cost bridge projects, state funds	SEG	S	91,079,100	95,821,300	
(av)	Principal repayment and interest, contingent funding of major highway and rehabilitation projects, state					
	funds	SEG	S	6,085,700	17,957,100	
		(6) PROC	GRAM TO			
	GENERAL PURPOSE REVENUE	· /		126,583,800	116,624,000	
	SEGREGATED REVENUE			155,289,600	174,677,700	
	OTHER			(155,289,600)	(174,677,700)	
	TOTAL-ALL SOURCES			281,873,400	291,301,700	
(9)	GENERAL PROVISIONS			, ,	, ,	
(qd)	Freeway land disposal reimbursement clearing account	SEG	С	-0-	_^_	
(qh)	Highways, bridges and local	SLU	C	- v-	- 0-	
	transportation assistance clearing account	SEG	C	-0-	-0-	

2017 Wi	2017 Wisconsin Act 59		_	2017	Assembly Bill 64
STATUT	E, AGENCY AND PURPOSE	Source	Түре	2017-2018	2018-2019
(qj)	Highways, bridges and local				
	transportation assistance clearing				
	account, federally funded positions	SEG-F	C	-0-	-0-
(qn)	Motor vehicle financial	CEC	C	0	0
(th)	responsibility Temporary funding of projects	SEG	С	-0-	-0-
(ui)	financed by revenue bonds	SEG	S	-0-	-0-
	imaneed by revenue bonds		GRAM TO		v
	SEGREGATED REVENUE	(-)		-0-	-0-
	FEDERAL			(-0-)	(-0-)
	OTHER			(-0-)	(-0-)
	TOTAL–ALL SOURCES			-0-	-0-
	20).395 DEPA	ARTMENT	ΓTOTALS	
	GENERAL PURPOSE REVENUE			130,283,800	116,624,000
	PROGRAM REVENUE			10,777,300	10,472,500
	OTHER			(7,101,400)	(7,036,600)
	SERVICE			(3,675,900)	(3,435,900)
	SEGREGATED REVENUE			2,884,842,700	2,919,068,600
	FEDERAL			(879,268,600)	(897,630,100)
	OTHER SERVICE			(1,804,731,200) (85,517,300)	(1,802,241,000) (103,871,900)
	LOCAL			(115,325,600)	(115,325,600)
	TOTAL-ALL SOURCES			3,025,903,800	3,046,165,100
	TOTAL TILL SOURCES	Environr	nental Res		3,010,103,100
	F	UNCTION			
	GENERAL PURPOSE REVENUE			259,282,200	249,396,900
	PROGRAM REVENUE			80,697,400	80,017,200
	FEDERAL			(28,580,100)	(28,204,700)
	OTHER			(28,799,800)	(28,735,000)
	SERVICE			(23,317,500)	(23,077,500)
	SEGREGATED REVENUE			3,278,098,300	3,306,045,400
	FEDERAL			(932,937,200)	(951,339,100)
	OTHER			(2,144,318,200)	(2,135,508,800)
	SERVICE			(85,517,300)	(103,871,900)
	LOCAL TOTAL–ALL SOURCES			(115,325,600) 3,618,077,900	(115,325,600) 3,635,459,500
		non Do	~~~~		3,033,439,300
• • • • • •		nan Res	sources		
	Corrections, Department of				
(1)	ADULT CORRECTIONAL SERVICES	CDD		760 216 200	777 020 200
(a)	General program operations Institutional repair and maintenance	GPR GPR	A A	769,216,300 4,291,200	777,020,300 4,333,800
(aa) (ab)	Corrections contracts and agreements	GPR	A	24,707,800	21,498,100
(ab) (b)	Services for community corrections	GPR	A	150,998,400	152,621,400
(bd)	Services for drunken driving	OLK	Α	130,770,400	132,021,400
(64)	offenders	GPR	A	5,175,900	4,887,300
(bm)	Pharmacological treatment for			, ,	, ,
. /	certain child sex offenders	GPR	A	58,900	58,900
(bn)	Reimbursing counties for probation,				
	extended supervision and parole	CDD		4.005.500	4.005.500
	holds	GPR	A	4,885,700	4,885,700

2017 As	2017 Assembly Bill 64		_	2017 W	isconsin Act 59
STATUT	E, AGENCY AND PURPOSE	Source	Түре	2017-2018	2018-2019
(c)	Reimbursement claims of counties				
	containing state prisons	GPR	S	45,000	45,000
(cw)	Mother-young child care program	GPR	A	198,000	198,000
(d)	Purchased services for offenders	GPR	A	31,190,000	31,190,000
(ds)	Becky Young community				
	corrections; recidivism reduction				
	community services	GPR	A	11,486,700	11,486,700
(e)	Principal repayment and interest	GPR	S	72,056,200	66,480,100
(ec)	Prison industries principal, interest	~~~	~		
	and rebates	GPR	S	-0-	-0-
(ed)	Correctional facilities rental	GPR	A	-0-	-0-
(ef)	Lease rental payments	GPR	S	-0-	-0-
(f)	Energy costs; energy-related	CDD		24.040.700	25 420 700
(6.)	assessments	GPR	A	24,849,700	25,438,700
(fm)	Electric energy derived from	CDD	٨	560 900	560 900
(ah)	renewable resources	GPR PR	A C	560,800 -0-	560,800
(gb)	Drug testing	PR PR	C	340,800	-0- 240 800
(gc)	Sex offender honesty testing	PR PR	C A	909,100	340,800 909,100
(gd)	Sex offender management Probation, parole, and extended	PK	Α	909,100	909,100
(gf)	supervision	PR	A	5,282,400	5,282,400
(gh)	Supervision of persons on lifetime	1 IX	А	3,202,400	3,202,400
(gii)	supervision of persons on meanic	PR	A	-0-	-0-
(gi)	General operations	PR	A	5,439,600	5,439,600
(gk)	Global positioning system tracking	110	11	3,133,000	3,133,000
(811)	devices for certain sex offenders	PR	C	264,800	281,500
(gL)	Global positioning system tracking			,	,
ν,	devices for certain violators of				
	restraining orders	PR	C	139,400	139,400
(gm)	Sale of fuel and utility service	PR	A	-0-	-0-
(gn)	Interstate compact for adult offender				
	supervision	PR	A	375,900	375,900
(gr)	Home detention services; supervision	PR	A	252,500	151,100
(gt)	Telephone company commissions	PR	A	1,404,600	1,404,600
(h)	Administration of restitution	PR	A	849,000	850,000
(hm)	Private business employment of				
	inmates and residents	PR	A	-0-	-0-
(i)	Gifts and grants	PR	C	33,400	33,400
(jz)	Operations and maintenance	PR	C	521,500	534,600
(kc)	Correctional institution enterprises;		_		
	inmate activities and employment	PR-S	C	2,776,000	2,776,000
(kd)	Victim notification	PR-S	A	682,300	682,300
(ke)	American Indian reintegration	DD C		7 0.000	7 0.000
4.6	program	PR-S	A	50,000	50,000
(kf)	Correctional farms	PR-S	A	9,333,300	8,633,800
(kh)	Victim services and programs	PR-S	A	218,500	218,500
(kk)	Institutional operations and charges	PR-S	A	12,765,900	12,766,000
(km)	Prison industries	PR-S	A	20,300,600	22,456,200
(ko)	Prison industries principal	DD C	C	06.000	02.200
(1)	repayment, interest and rebates	PR-S	S A	96,000	93,200
(kp)	Correctional officer training	PR-S	A	2,468,100	2,468,100

2017 Wi	2017 Wisconsin Act 59		_	2017	2017 Assembly Bill 64		
STATUT	TE, AGENCY AND PURPOSE	Source	Түре	2017-2018	2018-2019		
(kx)	Interagency and intra-agency						
	programs	PR-S	C	2,228,900	2,228,900		
(ky)	Interagency and intra-agency aids	PR-S	C	1,427,700	1,427,700		
(kz)	Interagency and intra–agency local assistance	PR-S	С	-0-	-0-		
(m)	Federal project operations	PR-F	C	2,473,100	2,473,100		
(n)	Federal program operations	PR-F	C	86,800	86,800		
(qm)	Computer recycling	SEG	A	-0-	-0-		
(4111)	computer recycling	(1) PROC			· ·		
	GENERAL PURPOSE REVENUE	(1) 1110 (1,099,720,600	1,100,704,800		
	PROGRAM REVENUE			70,720,200	72,103,000		
	FEDERAL			(2,559,900)	(2,559,900)		
	OTHER			(15,813,000)	(15,742,400)		
	SERVICE			(52,347,300)	(53,800,700)		
	SEGREGATED REVENUE			-0-	-0-		
	OTHER			(-0-)	(-0-)		
	TOTAL-ALL SOURCES			1,170,440,800	1,172,807,800		
(2)	PAROLE COMMISSION						
(a)	General program operations	GPR	A	753,800	755,300		
	1 0 1	(2) PROC	GRAM T				
	GENERAL PURPOSE REVENUE			753,800	755,300		
	TOTAL-ALL SOURCES			753,800	755,300		
(3)	JUVENILE CORRECTIONAL SERVICES						
(a)	General program operations	GPR	A	3,192,300	3,385,500		
(ba)	Mendota juvenile treatment center	GPR	A	1,365,500	1,365,500		
(c)	Reimbursement claims of counties containing juvenile correctional						
	facilities	GPR	S	36,000	36,000		
(cg)	Serious juvenile offenders	GPR	В	17,755,800	19,077,000		
(dm)	Interstate compact for juveniles			, ,	, ,		
` /	assessments	GPR	A	-0-	-0-		
(e)	Principal repayment and interest	GPR	S	3,856,500	3,519,900		
(g)	Legal services collections	PR	C	-0-	-0-		
(gg)	Collection remittances to local units						
	of government	PR	C	-0-	-0-		
(hm)	Juvenile correctional services	PR	A	30,221,800	30,133,400		
(ho)	Juvenile alternate care services	PR	A	7,278,900	7,763,400		
(hr)	Juvenile corrective sanctions services	D.D.		7.266.200	7.071.400		
	program	PR	A	5,266,300	5,251,400		
(i)	Gifts and grants	PR	C	7,700	7,700		
(jr)	Institutional operations and charges	PR	A	180,100	180,100		
(jv)	Secure detention services	PR	C	200,000	200,000		
(kx)	Interagency and intra–agency	PR-S	C	810,800	810,800		
(ky)	programs Interagency and intra–agency aids	PR-S	C	-0-	-0-		
(kz)	Interagency and intra–agency and Interagency and intra–agency local	rk-3	C	_0_	_0_		
(KZ)	assistance	PR-S	C	-0-	-0-		
(m)	Federal project operations	PR-F	C	-0-	-0-		
(n)	Federal program operations	PR-F	C	30,000	30,000		
(q)	Girls school benevolent trust fund	SEG	C	-0-	-0-		
(1)		(3) PROC			-		

017 A	ssembly Bill 64	- 77	_	2017 V	Visconsin Act
STATUT	TE, AGENCY AND PURPOSE	Source	Түре	2017-2018	2018-2019
	GENERAL PURPOSE REVENUE			26,206,100	27,383,900
	PROGRAM REVENUE			43,995,600	44,376,800
	FEDERAL			(30,000)	(30,000)
	OTHER			(43,154,800)	(43,536,000)
	SERVICE			(810,800)	(810,800)
	SEGREGATED REVENUE			-0-	-0-
	OTHER			(-0-)	(-0-
	TOTAL-ALL SOURCES			70,201,700	71,760,700
		20.410 DEF	PARTMENT	TOTALS	
	GENERAL PURPOSE REVENUE			1,126,680,500	1,128,844,000
	PROGRAM REVENUE			114,715,800	116,479,800
	FEDERAL			(2,589,900)	(2,589,900)
	OTHER			(58,967,800)	(59,278,400)
	SERVICE			(53,158,100)	(54,611,500)
	SEGREGATED REVENUE			-0-	-0-
	OTHER			(-0-)	(-0-
	TOTAL-ALL SOURCES			1,241,396,300	1,245,323,800
	Employment Relations Commission	n		, ,	, -,,
(1)	LABOR RELATIONS				
(a)	General program operations	GPR	Α	991,300	992,200
(i)	Fees, collective bargaining training.			,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	
(-)	publications, and appeals	, PR	A	145,600	145,600
	, 11	(1) PRO	OGRAM TO		
	GENERAL PURPOSE REVENUE	. ,		991,300	992,200
	PROGRAM REVENUE			145,600	145,600
	OTHER			(145,600)	(145,600
	TOTAL-ALL SOURCES			1,136,900	1,137,800
		20.425 DEF	PARTMENT	TOTALS	
	GENERAL PURPOSE REVENUE			991,300	992,200
	PROGRAM REVENUE			145,600	145,600
	OTHER			(145,600)	(145,600
	TOTAL-ALL SOURCES			1,136,900	1,137,800
20.427	Labor and Industry Review Comm	nission			
(1)	REVIEW COMMISSION				
(a)	General program operations, review	V			
	commission	GPR	A	242,600	243,100
(g)	Agency collections	PR	C	-0-	-0-
(k)	Unemployment administration	PR-S	C	1,856,000	1,859,400
(km)	Equal rights; other moneys	PR-S	C	224,900	225,100
(m)	Federal moneys	PR-F	C	-0-	-0-
(ra)	Worker's compensation operations				
	fund; worker's compensation				
	activities	SEG	A	764,000	764,000
		(1) PRC	OGRAM TO		
	GENERAL PURPOSE REVENUE			242,600	243,100
	PROGRAM REVENUE			2,080,900	2,084,500
	FEDERAL			(-0-)	(-0-
	OTHER			(-0-)	(-0-
	SERVICE			(2,080,900)	(2,084,500)
	SEGREGATED REVENUE			764,000	764,000
	OTHER			(764,000)	(764,000)

STATUTE, AGENCY AND PURPOSE TOTAL-ALL SOURCES	Source	ТүрЕ	2017–2018 3,087,500	2018–2019 3,091,600
TOTAL-ALL SOURCES	20.427 DEPA	ARTMEN		3,091,000
GENERAL PURPOSE REVENUE			242,600	243,100
PROGRAM REVENUE			2,080,900	2,084,500
FEDERAL			(-0-)	(-0-)
OTHER			(-0-)	(-0-)
SERVICE			(2,080,900)	(2,084,500)
SEGREGATED REVENUE			764,000	764,000
OTHER			(764,000)	(764,000)
TOTAL-ALL SOURCES			3,087,500	3,091,600
20.432 Board on Aging and Long-Term (
(1) IDENTIFICATION OF THE NEEDS OF THE		SABLED		
(a) General program operations	GPR	A	1,414,600	1,432,900
(i) Gifts and grants	PR	C	-0-	-0-
(k) Contracts with other state agencies	PR-S	C	1,393,200	1,412,900
(kb) Insurance and other information,	<i>DD</i> . 0		700.00	5 00 400
counseling and assistance	PR-S	A	508,600	509,400
(m) Federal aid	PR-F	C CDAM T	-0-	-0-
CENEDAL DUDDOGE DEVENUE	(I) PROC	GRAM T		1 422 000
GENERAL PURPOSE REVENUE PROGRAM REVENUE			1,414,600	1,432,900
FEDERAL			1,901,800	1,922,300
OTHER			(-0-) (-0-)	(-0-) (-0-)
SERVICE			(1,901,800)	(1,922,300)
TOTAL-ALL SOURCES			3,316,400	3,355,200
TOTAL-ALL SOURCES	20.432 DEPA	ATMEN		3,333,200
GENERAL PURPOSE REVENUE	20.432 DEI	XIXI IVILIV	1,414,600	1,432,900
PROGRAM REVENUE			1,901,800	1,922,300
FEDERAL			(-0-)	(-0-)
OTHER			(-0-)	(-0-)
SERVICE			(1,901,800)	(1,922,300)
TOTAL-ALL SOURCES			3,316,400	3,355,200
20.433 Child Abuse and Neglect Preventi	on Board		-,,	-,,
(1) Prevention of Child abuse and Ne				
(b) Grants to organizations	GPR	A	995,000	995,000
(g) General program operations	PR	A	635,000	635,800
(h) Grants to organizations	PR	C	850,600	850,600
(i) Gifts and grants	PR	C	-0-	-0-
(jb) Fees for administrative services	PR	C	15,000	15,000
(k) Interagency programs	PR-S	C	-0-	-0-
(m) Federal project operations	PR-F	C	197,700	197,700
(ma) Federal project aids	PR-F	C	450,000	450,000
(q) Children's trust fund; gifts and gra	nts SEG	C	15,000	15,000
	(1) PRO0	GRAM T		
GENERAL PURPOSE REVENUE			995,000	995,000
PROGRAM REVENUE			2,148,300	2,149,100
FEDERAL			(647,700)	(647,700)
OTHER			(1,500,600)	(1,501,400)
SERVICE			(-0-)	(-0-)
SEGREGATED REVENUE			15,000	15,000
OTHER			(15,000)	(15,000)

2017 Assembly Bill 64

	E, AGENCY AND PURPOSE	Source	Түре	2017-2018	2018-2019
-	ГОТAL–ALL SOURCES			3,158,300	3,159,100
	2	0.433 DEPA	RTMENT	TOTALS	
	GENERAL PURPOSE REVENUE			995,000	995,000
]	PROGRAM REVENUE			2,148,300	2,149,100
	FEDERAL			(647,700)	(647,700)
	OTHER			(1,500,600)	(1,501,400)
	SERVICE			(-0-)	(-0-)
\$	SEGREGATED REVENUE			15,000	15,000
	OTHER			(15,000)	(15,000)
	ГОТAL–ALL SOURCES			3,158,300	3,159,100
	Health Services, Department of				
` '	PUBLIC HEALTH SERVICES PLANNING, RE		D DELIVERY		
(a)	General program operations	GPR	A	7,688,000	7,726,600
(am)	Services, reimbursement, and				
	payment related to human	CDD		(220 000	6.220.000
4.5	immunodeficiency virus	GPR	A	6,220,900	6,220,900
(b)	General aids and local assistance	GPR	A	543,600	543,600
(bg)	Alzheimer's disease; training and	CDD		121 400	121 400
(1)	information grants Purchased services for clients	GPR	A	131,400	131,400
(bm)		GPR	A	93,900	93,900
(bn)	Workplace wellness program grants	GPR	S	3,000,000	3,000,000
(br)	Respite care	GPR	A	225,000	225,000
(c)	Public health emergency quarantine costs	GPR	S	-0-	-0-
(cb)	Well-woman program	GPR	A	2,328,200	2,328,200
(cc)	Cancer control and prevention	GPR	A	333,900	333,900
(ce)	Primary health for homeless	OI K	Λ	333,900	333,900
(CC)	individuals	GPR	C	-0-	-0-
(cf)	Communicable disease control and	OI IX	C	· ·	O .
(61)	prevention	GPR	C	500,000	500,000
(cg)	Guardianship grant program	GPR	A	100,000	100,000
(ch)	Emergency medical services; aids	GPR	A	1,960,200	1,960,200
(cm)	Immunization	GPR	S	-0-	-0-
(cx)	Independent living centers	GPR	A	1,017,700	1,017,700
(da)	Interpreter services and				
, ,	telecommunication aid for the				
	hearing impaired	GPR	A	178,200	178,200
(de)	Dental services	GPR	A	2,974,300	2,974,300
(dg)	Clinic aids	GPR	В	66,800	66,800
(dh)	Programs for senior citizens; elder				
	abuse services; benefit specialist			4.7.7.000	4.7.7.7.000
	program	GPR	A	15,707,800	15,707,800
(dk)	Low-income dental clinics	GPR	A	850,000	850,000
(dm)	Rural health dental clinics	GPR	A	895,500	895,500
(dn)	Food distribution grants	GPR	A	288,000	288,000
(ds)	Statewide poison control program	GPR	A	382,500	382,500
(e)	Public health dispensaries and drugs	GPR	В	661,000	661,000
(ed)	Radon aids	GPR	A	26,700	26,700
(ef)	Lead-poisoning or lead-exposure	CDD	A	004.700	004.700
()	services	GPR	A	894,700	894,700
(eg)	Pregnancy counseling	GPR	A	69,100	69,100

 \mathbf{C}

92,735,300

92,735,300

PR-F

(na)

Federal program aids

STATUTE, AGENCY AND PURPOSE SOURCE Type 2017-2018 302,200 302,200 302,200 302,200 302,200 302,200 302,200 302,200 302,200 302,200 302,200 302,200 302,200 303,200 303,300 303,200 302,2	2017 Assembly Bill 64		- 81 -	_	2017 V	2017 Wisconsin Act 59		
SEG A 302,200 302,200 302,200 Company Segment Segm			Source	Түре	2017-2018	2018-2019		
Canaly program; inpatient psychiatric beds Cape	(4)		SEG	A	302,200	302,200		
PROGRAM REVENUE				GRAM TO		,		
FEDERAL	(GENERAL PURPOSE REVENUE	, ,		62,756,700	63,795,300		
OTHER (33,433,000) (33,398,500) SERVICE (6,139,700) (6,139,700) (6,139,700) (6,139,700) (6,139,700) (6,139,700) (6,139,700) (6,139,700) (7,705,800) (7,103,8]	PROGRAM REVENUE			225,944,300	225,948,400		
SERVICE (6,139,700) (6,139,700) SEGREGATED REVENUE 302,200 302,2		FEDERAL			(186,371,600)	(186,410,200)		
SEGREGATED REVENUE		OTHER			(33,433,000)	(33,398,500)		
OTHER		SERVICE			(6,139,700)	(6,139,700)		
TOTAL—ALL SOURCES 289,003,200 290,045,900		SEGREGATED REVENUE			302,200	302,200		
(2) MENTAL HEALTH AND DEVELOPMENTAL DISABILITIES SERVICES; FACILITIES (a) General program operations GPR A 79,214,900 82,899,800 (aa) Institutional repair and maintenance (bp) Competency examinations and treatment, and conditional release, supervised release, and community supervision services GPR B 14,836,600 16,213,900 (bm) Secure mental health units or facilities GPR A 103,969,700 105,840,300 (cm) Grant program; inpatient psychiatric beds GPR A 30,000 30,000 (ee) Principal repayment and interest GPR S 21,228,100 18,922,900 (ef) Lease rental payments GPR S 21,228,100 18,922,900 (f) Energy costs; energy—related assessments GPR A 4,528,600 4,583,900 (g) Alternative services of institutes and centers PR C 10,018,100 10,030,500 (g) Institutional operations and charges (g) Extended intensive treatment surcharge PR C 100,000 100,000 (gs) Sex offender honesty testing PR C 100,000 100,000 (gs) Sex offender honesty testing PR C 93,800 93,800 (km) Indian mental health placement PR—S A 250,000 250,000 (kx) Interagency and intra-agency programs PR—S C 7,455,800 7,455,800 (ky) Interagency and intra-agency plocal assistance PR—F C ——————————————————————————————————		OTHER			(302,200)	(302,200)		
(a) General program operations (GPR A 79,214,900 82,899,800 (aa) Institutional repair and maintenance (GPR A 715,200 7	,	TOTAL–ALL SOURCES			289,003,200	290,045,900		
(aa) Institutional repair and maintenance (b) Competency examinations and treatment, and conditional release, supervised release, and community supervision services (GPR B 14,836,600 16,213,900) (bm) Secure mental health units or facilities (GPR A 103,969,700 105,840,300) (cm) Grant program; inpatient psychiatric beds (GPR A 30,000 30,000) (ce) Principal repayment and interest (GPR S 21,228,100 18,922,900) (cf) Lease rental payments (GPR A 4,528,600 4,583,900) (fm) Electric energy derived from renewable resources (GPR A 241,400 241,400) (g) Alternative services of institutes and centers (GPR A 170,269,400 168,780,300) (gL) Extended intensive treatment surcharge (GR) (GR) (GR) (GR) (GR) (GR) (GR) (GR)	(2)	MENTAL HEALTH AND DEVELOPMENTAL I	DISABILITIES	SERVICES; I	FACILITIES			
(bj) Competency examinations and treatment, and conditional release, supervised release, and community supervision services (cm) Secure mental health units or facilities	(a)	General program operations	GPR	A	79,214,900	82,899,800		
treatment, and conditional release, supervised release, and community supervision services GPR B 14,836,600 16,213,900 (bm) Secure mental health units or facilities Grant program; inpatient psychiatric beds GPR A 103,969,700 105,840,300 (cm) Grant program; inpatient psychiatric beds GPR A 30,000 30,000 (ce) Principal repayment and interest GPR S 21,228,100 18,922,900 (cf) Lease rental payments GPR A 4,528,600 4,583,900 (fm) Energy costs; energy-related assessments GPR A 4,528,600 4,583,900 (fm) Electric energy derived from renewable resources GPR A 241,400 241,400 (g) Alternative services of institutes and centers PR C 10,018,100 10,030,500 (gk) Institutional operations and charges PR A 170,269,400 168,780,300 (gs) Sex offender honesty testing PR C 100,000 100,000 (gs) Sex offender honesty testing PR C 93,800 93,800 (km) Indian mental health placement PR-S A 250,000 250,000 (kx) Interagency and intra-agency programs PR-S C 7,455,800 7,455,800 (ky) Interagency and intra-agency local assistance PR-S C -00- (m) Federal project operations PR-F C -00- (m) Federal Propose REVENUE PROGRAM REVENUE PROGRAM REVENUE 188,187,100 186,710,400 FEDERAL (-0-) (-0-) OTHER	(aa)	Institutional repair and maintenance	GPR	A	715,200	715,200		
Supervised release, and community supervision services GPR B 14,836,600 16,213,900	(bj)							
Supervision services								
(bm) Secure mental health units or facilities			CDD	ъ	14.026.600	16.212.000		
facilities	(1,)		GPR	В	14,836,600	16,213,900		
(cm) Grant program; inpatient psychiatric beds GPR A 30,000 30,000 (ee) Principal repayment and interest GPR S 21,228,100 18,922,900 (ef) Lease rental payments GPR S -0- -0- (f) Energy costs; energy-related assessments GPR A 4,528,600 4,583,900 (fm) Electric energy derived from renewable resources GPR A 241,400 241,400 (g) Alternative services of institutes and centers GPR A 241,400 241,400 (g) Alternative services of institutes and centers PR C 10,018,100 10,030,500 (gk) Institutional operations and charges PR A 170,269,400 168,780,300 (gL) Extended intensive treatment surcharge PR C 100,000 100,000 (gs) Sex offender honesty testing PR C 100,000 100,000 (gs) Sex offender honesty testing PR C 93,800 93,800 <td>(bm)</td> <td></td> <td>CDD</td> <td>٨</td> <td>102 060 700</td> <td>105 940 200</td>	(bm)		CDD	٨	102 060 700	105 940 200		
beds	(cm)		OFK	A	103,909,700	103,640,300		
(ee) Principal repayments GPR S 21,228,100 18,922,900 (ef) Lease rental payments GPR S -0- -0- (f) Energy costs; energy—related assessments GPR A 4,528,600 4,583,900 (fm) Electric energy derived from renewable resources GPR A 241,400 241,400 (g) Alternative services of institutes and centers PR C 10,018,100 10,030,500 (gk) Institutional operations and charges PR C 10,018,100 10,030,500 (gt) Extended intensive treatment surcharge PR A 170,269,400 168,780,300 (gt) Extended intensive treatment surcharge PR C 100,000 100,000 (gs) Sex offender honesty testing PR C 0-0- -0- (gz) Costs of housing persons on supervised release PR C 0-0- -0- (km) Indian mental health placement PR-S A 250,000 250,000	(CIII)		GPR	Α	30,000	30,000		
(ef) Lease rental payments GPR S -0- -0- (f) Energy costs; energy-related assessments GPR A 4,528,600 4,583,900 (fm) Electric energy derived from renewable resources GPR A 241,400 241,400 (g) Alternative services of institutes and centers PR C 10,018,100 10,030,500 (gk) Institutional operations and charges PR A 170,269,400 168,780,300 (gL) Extended intensive treatment surcharge PR C 100,000 100,000 (gs) Sex offender honesty testing PR C -0- -0- (gz) Costs of housing persons on supervised release PR C -0- -0- (gz) Costs of housing persons on supervised release PR C 93,800 93,800 (km) Indian mental health placement PR-S A 250,000 250,000 (kx) Interagency and intra-agency aids PR-S C 7,455,800 7,455,800	(ee)							
(f) Energy costs; energy-related assessments GPR A 4,528,600 4,583,900 (fm) Electric energy derived from renewable resources GPR A 241,400 241,400 (g) Alternative services of institutes and centers PR C 10,018,100 10,030,500 (gk) Institutional operations and charges PR A 170,269,400 168,780,300 (gL) Extended intensive treatment surcharge PR C 100,000 100,000 (gS) Sex offender honesty testing PR C -0- -0- (gz) Costs of housing persons on supervised release PR C -0- -0- (gz) Gifts and grants PR C 93,800 93,800 (km) Indian mental health placement PR-S A 250,000 250,000 (kx) Interagency and intra-agency aids PR-S C -0- -0- (ky) Interagency and intra-agency local assistance PR-S C -0- -0- <td< td=""><td>` /</td><td></td><td></td><td></td><td></td><td></td></td<>	` /							
Assessments			0110	J	v	· ·		
(fm) Electric energy derived from renewable resources GPR A 241,400 241,400 (g) Alternative services of institutes and centers PR C 10,018,100 10,030,500 (gk) Institutional operations and charges PR A 170,269,400 168,780,300 (gL) Extended intensive treatment surcharge PR C 100,000 100,000 (gs) Sex offender honesty testing PR C -0- -0- (gz) Costs of housing persons on supervised release PR C -0- -0- (i) Gifts and grants PR C 93,800 93,800 (km) Indian mental health placement PR-S A 250,000 250,000 (kx) Interagency and intra-agency programs PR-S C 7,455,800 7,455,800 (ky) Interagency and intra-agency local assistance PR-S C -0- -0- (kz) Interagency and intra-agency local assistance PR-F C -0- -0-	(-)		GPR	A	4,528,600	4,583,900		
Rememble resources	(fm)	Electric energy derived from						
Centers			GPR	A	241,400	241,400		
(gk) Institutional operations and charges PR A 170,269,400 168,780,300 (gL) Extended intensive treatment surcharge PR C 100,000 100,000 (gs) Sex offender honesty testing PR C -0- -0- (gz) Costs of housing persons on supervised release PR C -0- -0- (i) Gifts and grants PR C 93,800 93,800 (km) Indian mental health placement PR-S A 250,000 250,000 (kx) Interagency and intra-agency programs PR-S C 7,455,800 7,455,800 (ky) Interagency and intra-agency local assistance PR-S C -0- -0- (kz) Interagency and intra-agency local assistance PR-F C -0- -0- (m) Federal project operations PR-F C -0- -0- (2) PROGRAM REVENUE 188,187,100 186,710,400 FEDERAL (-0-) (-0-) (-0-) <	(g)	Alternative services of institutes and						
(gL) Extended intensive treatment surcharge PR C 100,000 100,000 (gs) Sex offender honesty testing PR C -0- -0- (gz) Costs of housing persons on supervised release PR C -0- -0- (i) Gifts and grants PR C 93,800 93,800 (km) Indian mental health placement PR-S A 250,000 250,000 (kx) Interagency and intra-agency programs PR-S C 7,455,800 7,455,800 (ky) Interagency and intra-agency agency assistance PR-S C -0- -0- (kz) Interagency and intra-agency local assistance PR-S C -0- -0- (m) Federal project operations PR-F C -0- -0- (m) Federal project operations PR-F C -0- -0- (2) PROGRAM REVENUE 188,187,100 186,710,400 FEDERAL (-0-) (-0-) (-0-) <t< td=""><td></td><td></td><td></td><td>C</td><td></td><td></td></t<>				C				
surcharge PR C 100,000 100,000 (gs) Sex offender honesty testing PR C -0- -0- (gz) Costs of housing persons on supervised release PR C -0- -0- (i) Gifts and grants PR C 93,800 93,800 (km) Indian mental health placement PR-S A 250,000 250,000 (kx) Interagency and intra-agency programs PR-S C 7,455,800 7,455,800 (ky) Interagency and intra-agency aids assistance PR-S C -0- -0- (kz) Interagency and intra-agency local assistance PR-S C -0- -0- (m) Federal project operations PR-F C -0- -0- (2) PROGRAM TOTALS 224,764,500 229,447,400 PROGRAM REVENUE 188,187,100 186,710,400 FEDERAL (-0-) (-0-) (-0-) OTHER (180,481,300) (179,004,600) (7,705,800)	_	=	PR	A	170,269,400	168,780,300		
(gs) Sex offender honesty testing PR C -0- -0- (gz) Costs of housing persons on supervised release PR C -0- -0- (i) Gifts and grants PR C 93,800 93,800 (km) Indian mental health placement PR-S A 250,000 250,000 (kx) Interagency and intra-agency programs PR-S C 7,455,800 7,455,800 (ky) Interagency and intra-agency aids assistance PR-S C -0- -0- (kz) Interagency and intra-agency local assistance PR-S C -0- -0- (m) Federal project operations PR-F C -0- -0- (2) PROGRAM TOTALS 224,764,500 229,447,400 PROGRAM REVENUE 188,187,100 186,710,400 FEDERAL (-0-) (-0-) (-0-) OTHER (180,481,300) (179,004,600) SERVICE (7,705,800) (7,705,800)	(gL)		DD	a	100.000	100.000		
(gz) Costs of housing persons on supervised release PR C -0- -0- (i) Gifts and grants PR C 93,800 93,800 (km) Indian mental health placement PR-S A 250,000 250,000 (kx) Interagency and intra-agency aids programs PR-S C 7,455,800 7,455,800 (ky) Interagency and intra-agency aids assistance PR-S C -0- -0- (kz) Interagency and intra-agency local assistance PR-S C -0- -0- (m) Federal project operations PR-F C -0- -0- (2) PROGRAM TOTALS 224,764,500 229,447,400 229,447,400 PROGRAM REVENUE 188,187,100 186,710,400 FEDERAL (-0-) (-0-) OTHER (180,481,300) (179,004,600) SERVICE (7,705,800) (7,705,800)		_						
Supervised release			PR	C	-0-	-0-		
(i) Gifts and grants PR C 93,800 93,800 (km) Indian mental health placement PR-S A 250,000 250,000 (kx) Interagency and intra-agency programs PR-S C 7,455,800 7,455,800 (ky) Interagency and intra-agency aids PR-S C -0- -0- (kz) Interagency and intra-agency local assistance PR-S C -0- -0- (m) Federal project operations PR-F C -0- -0- (2) PROGRAM TOTALS GENERAL PURPOSE REVENUE 224,764,500 229,447,400 PROGRAM REVENUE 188,187,100 186,710,400 FEDERAL (-0-) (-0-) OTHER (180,481,300) (179,004,600) SERVICE (7,705,800) (7,705,800)	(gz)		DD	C	0	0		
(km) Indian mental health placement PR-S A 250,000 250,000 (kx) Interagency and intra-agency programs PR-S C 7,455,800 7,455,800 (ky) Interagency and intra-agency aids assistance PR-S C -0- -0- (m) Federal project operations PR-S C -0- -0- (2) PROGRAM TOTALS GENERAL PURPOSE REVENUE 224,764,500 229,447,400 PROGRAM REVENUE 188,187,100 186,710,400 FEDERAL (-0-) (-0-) OTHER (180,481,300) (179,004,600) SERVICE (7,705,800) (7,705,800)	(i)	=						
(kx) Interagency and intra–agency programs PR-S C 7,455,800 7,455,800 (ky) Interagency and intra–agency aids assistance PR-S C -0- -0- (kz) Interagency and intra–agency local assistance PR-S C -0- -0- (m) Federal project operations PR-F C -0- -0- (2) PROGRAM TOTALS GENERAL PURPOSE REVENUE 224,764,500 229,447,400 PROGRAM REVENUE 188,187,100 186,710,400 FEDERAL (-0-) (-0-) OTHER (180,481,300) (179,004,600) SERVICE (7,705,800) (7,705,800)								
PR-S C 7,455,800 7,455,800			1 K-5	Λ	230,000	230,000		
(ky) Interagency and intra-agency aids PR-S C -0- -0- (kz) Interagency and intra-agency local assistance PR-S C -0- -0- (m) Federal project operations PR-F C -0- -0- (2) PROGRAM TOTALS 224,764,500 229,447,400 PROGRAM REVENUE 188,187,100 186,710,400 FEDERAL (-0-) (-0-) OTHER (180,481,300) (179,004,600) SERVICE (7,705,800) (7,705,800)	(KA)		PR-S	C	7.455.800	7.455.800		
(kz) Interagency and intra–agency local assistance PR–S C -0- -0- (m) Federal project operations PR–F C -0- -0- (2) PROGRAM TOTALS 224,764,500 229,447,400 PROGRAM REVENUE 188,187,100 186,710,400 FEDERAL (-0-) (-0-) OTHER (180,481,300) (179,004,600) SERVICE (7,705,800) (7,705,800)	(kv)							
assistance PR-S C -00- (m) Federal project operations PR-F C -00- (2) PROGRAM TOTALS GENERAL PURPOSE REVENUE 224,764,500 229,447,400 PROGRAM REVENUE 188,187,100 186,710,400 FEDERAL (-0-) (-0-) OTHER (180,481,300) (179,004,600) SERVICE (7,705,800)	-		111 5	· ·		· ·		
(2) PROGRAM TOTALS GENERAL PURPOSE REVENUE 224,764,500 229,447,400 PROGRAM REVENUE 188,187,100 186,710,400 FEDERAL (-0-) (-0-) OTHER (180,481,300) (179,004,600) SERVICE (7,705,800) (7,705,800)	()		PR-S	C	-0-	-0-		
(2) PROGRAM TOTALS GENERAL PURPOSE REVENUE 224,764,500 229,447,400 PROGRAM REVENUE 188,187,100 186,710,400 FEDERAL (-0-) (-0-) OTHER (180,481,300) (179,004,600) SERVICE (7,705,800) (7,705,800)	(m)	Federal project operations	PR-F	C	-0-	-0-		
GENERAL PURPOSE REVENUE 224,764,500 229,447,400 PROGRAM REVENUE 188,187,100 186,710,400 FEDERAL (-0-) (-0-) OTHER (180,481,300) (179,004,600) SERVICE (7,705,800) (7,705,800)	• •			GRAM TO	OTALS			
PROGRAM REVENUE 188,187,100 186,710,400 FEDERAL (-0-) (-0-) OTHER (180,481,300) (179,004,600) SERVICE (7,705,800) (7,705,800)	(GENERAL PURPOSE REVENUE				229,447,400		
OTHER (180,481,300) (179,004,600) SERVICE (7,705,800) (7,705,800)]	PROGRAM REVENUE						
SERVICE (7,705,800) (7,705,800)		FEDERAL			(-0-)	(-0-)		
		OTHER			(180,481,300)	(179,004,600)		
TOTAL-ALL SOURCES 412,951,600 416,157,800		SERVICE			(7,705,800)	(7,705,800)		
	,	TOTAL-ALL SOURCES			412,951,600	416,157,800		

	E, AGENCY AND PURPOSE	SOURCE	TYPE	2017-2018	2018-2019
(4)	MEDICAID SERVICES				
(a)	General program operations	GPR	A	38,726,700	39,323,000
(b)	Medical Assistance program benefits	GPR	В	2,882,581,300	3,035,063,000
(bd)	Long-term care programs	GPR	A	80,321,200	80,321,200
(bf)	Graduate medical training support				
. ,	grants	GPR	C	750,000	750,000
(bk)	Mental health pilot projects	GPR	C	266,700	266,700
(bm)	Medical Assistance, food stamps,			,	
()	and Badger Care administration;				
	contract costs, insurer reports, and				
	resource centers	GPR	В	65,371,700	68,828,300
(bn)	Income maintenance	GPR	В	14,327,100	14,327,100
(bp)	Food stamp employment and training	0110	2	11,027,100	1.,027,100
(op)	program administration	GPR	C	10,637,500	17,625,000
(bq)	Substance abuse treatment costs	GPR	В	-0-	-0-
(bq)	Cemetery, funeral, and burial	OTK	Ь	O	O
(01)	expenses program	GPR	В	9,095,300	9,410,600
(bv)	Prescription drug assistance for	OIK	D	7,073,300	7,410,000
(01)	elderly; aids	GPR	В	18,013,800	20,927,400
(a)	Disease aids	GPR	В		
(e)		GFK	Ь	3,971,600	4,128,300
(ed)	State supplement to federal				
	supplemental security income	GPR	S	160 721 600	163,289,200
(-)	program			160,721,600	
(g)	Family care benefit; cost sharing	PR	C	-0-	-0-
(gm)	Medical assistance; provider refunds	DD	C	776 994 300	950 250 900
()	and collections	PR	C	776,884,200	850,259,800
(gr)	Income maintenance; county	PR	C	-0-	0
(1.)	payments				-0-
(h)	County contributions	PR	C	60,880,400	60,436,800
(hp)	Disabled children's long-term	DD	C	1.567.200	1.567.200
(1)	support waivers	PR	C	1,567,300	1,567,300
(i)	Gifts, grants, and payments; health	DD		2 20 5 000	2 20 5 000
	care financing	PR	C	3,385,900	3,385,900
(iL)	Medical assistance provider				
	assessments; health services	D.D.		455.000	455.000
	regulation	PR	C	175,000	175,000
(im)	Medical assistance; correct payment				
	recovery; collections; community	DD	C	77.565.100	77.015.100
<i>(</i> ,)	services; other recoveries	PR	C	77,565,100	77,815,100
(in)	Community options program; family	D.D.		244.000	244.000
	care; recovery of costs administration	PK	A	244,900	244,900
(j)	Prescription drug assistance for		~	<0.4 0 0.400	
	elderly; manufacturer rebates	PR	C	69,428,100	75,312,300
(jb)	Prescription drug assistance for		~		
	elderly; enrollment fees	PR	C	4,154,900	4,154,900
(jc)	Fees for administrative services	PR	C	30,000	30,000
(jd)	Electronic benefit transfer card				
	replacement costs	PR	C	455,000	455,000
(je)	Disease aids; drug manufacturer				
	rebates	PR	C	991,900	1,036,700
(jt)	Care management organization,				
	insolvency assistance	PR	C	-0-	-0-

17 As	sembly Bill 64	- 83 -	_	2017	Wisconsin Act 5
TATUT	E, AGENCY AND PURPOSE	Source	Түре	2017-2018	2018-2019
(jw)	BadgerCare Plus and hospital				
9 /	assessment	PR	C	2,030,200	2,030,200
(jz)	Medical Assistance and Badger Care				
	cost sharing, and employer penalty				
	assessments	PR	C	10,200,000	14,971,600
(kb)	Relief block grants to tribal	DD C		712 000	712 000
(1-4)	governing bodies	PR-S	A	712,800	712,800
(kt)	Medical assistance outreach and reimbursements for tribes	PR-S	В	961,700	961,700
(kv)	Care management organization;	1 K-5	Ъ	701,700	701,700
(K*)	oversight	PR-S	C	-0-	-0-
(kx)	Interagency and intra-agency				
` /	programs	PR-S	C	6,656,700	6,656,700
(ky)	Interagency and intra–agency aids	PR-S	C	40,420,100	40,362,800
(kz)	Interagency and intra-agency local				
	assistance	PR-S	C	845,300	845,300
(L)	Fraud and error reduction	PR	C	788,000	788,000
(m)	Federal project operations	PR-F	C	5,835,500	5,835,500
(ma)	Federal project aids	PR-F	C	2,700,000	2,700,000
(md)	Federal block grant aids	PR-F	C	-0-	-0-
(n)	Federal program operations	PR-F	C	69,400,000	70,073,500
(na)	Federal program aids	PR-F	C	12,485,000	12,485,000
(nn)	Federal aid; income maintenance	PR-F	C	58,455,200	58,571,200
(np)	Federal aid; food stamp employment	DD E	a	20.227.400	22 446 406
(-)	and training program	PR-F	C	20,337,400	23,446,400
(o)	Federal aid; medical assistance	PR-F	C	5,232,926,200	5,549,940,600
(p)	Federal aid; Badger Care health care program	PR-F	С	-0-	-0-
(pa)	Federal aid; Medical Assistance and	110 1	C	O .	O .
(pu)	food stamps contracts administration	PR-F	C	181,131,500	190,416,400
(pg)	Federal aid; prescription drug			, ,	, ,
10,	assistance for elderly	PR-F	C	18,133,000	21,067,700
(w)	Medical Assistance trust fund	SEG	В	326,027,700	324,998,700
(wm)	Medical assistance trust fund;				
	nursing homes	SEG	S	-0-	-0-
(wp)	Medical Assistance trust fund;	~~~	~		
	county reimbursement	SEG	S	-0-	-0-
(x)	Medical Assistance trust fund;	CEC	C	0	0
(va)	Badger Care health care program Hospital assessment fund; hospital	SEG	С	-0-	-0-
(xc)	payments	SEG	C	252,588,000	250,892,700
(xe)	Critical access hospital assessment	SEG	C	232,300,000	250,072,700
(AC)	fund; hospital payments	SEG	C	4,023,100	3,739,500
	, 1 1 2	(4) PROC	GRAM TO		, ,
(GENERAL PURPOSE REVENUE			3,284,784,500	3,454,259,800
]	PROGRAM REVENUE			6,659,781,300	7,076,739,100
	FEDERAL			(5,601,403,800)	(5,934,536,300)
	OTHER			(1,008,780,900)	(1,092,663,500)
	SERVICE			(49,596,600)	(49,539,300)
\$	SEGREGATED REVENUE			582,638,800	579,630,900
	OTHER			(582,638,800)	(579,630,900)
,	TOTAL-ALL SOURCES			10,527,204,600	11,110,629,800

MT / AAT	isconsin Act 59	- 84 -	_	2017 A	Assembly Bill 04	
STATUT	E, AGENCY AND PURPOSE	Source	Түре	2017-2018	2018-2019	
(5)	CARE AND TREATMENT SERVICES					
(a)	General program operations	GPR	A	2,393,400	2,393,400	
(bc)	Grants for community programs	GPR	A	8,931,100	8,681,100	
(be)	Mental health treatment services	GPR	A	1,551,500	1,551,500	
(bf)	Brighter futures initiative	GPR	A	865,000	865,000	
(bw)	Child psychiatry consultation					
	program	GPR	В	1,000,000	1,000,000	
(cd)	Crisis intervention training grants	GPR	В	125,000	125,000	
(cf)	Mobile crisis team grants	GPR	В	125,000	125,000	
(co)	Initiatives for coordinated services	GPR	A	2,599,100	2,599,100	
(da)	Reimbursements to local units of					
	government	GPR	S	507,800	507,800	
(fr)	Mental health for homeless			44.000	44.000	
	individuals	GPR	A	41,900	41,900	
(gb)	Alcohol and drug abuse initiatives	PR	C	468,000	468,000	
(gg)	Collection remittances to local units	DD	C	4.400	4 400	
<i>a</i> >	of government	PR	C	4,400	4,400	
(hx)	Services related to drivers, receipts	PR	A	-0-	-0-	
(hy)	Services for drivers, local assistance	PR	A	1,000,000	1,000,000	
(i)	Gifts and grants	PR	C	73,900	73,900	
(jb)	Fees for administrative services	PR	C	23,900	23,900	
(kc)	Severely emotionally disturbed children	DD C	C	724 500	724 500	
(kd)	Youth crisis stabilization facilities	PR-S PR-S	A	724,500 	724,500 - 0-	Vetoed
	Compulsive gambling awareness	rk-3	A	- 0-	-0-	In Part
(kg)	campaigns	PR-S	A	396,000	396,000	111 1 41 0
(kL)	Indian aids	PR-S	A	242,000	242,000	
(kE)	Indian drug abuse prevention and	TK 5	11	242,000	242,000	
(KIII)	education	PR-S	A	445,500	445,500	
(kp)	Veterans peer–run respite center	PR-S	C	-0-	450,000	Vetoed
(kx)	Interagency and intra–agency				,	In Part
()	programs	PR-S	C	2,891,500	2,891,500	
(ky)	Interagency and intra–agency aids	PR-S	C	-0-	-0-	
(kz)	Interagency and intra–agency local					
	assistance	PR-S	C	-0-	-0-	
(m)	Federal project operations	PR-F	C	695,000	495,500	
(ma)	Federal project aids	PR-F	C	4,639,400	4,639,400	
(mb)	Federal project local assistance	PR-F	C	-0-	-0-	
(mc)	Federal block grant operations	PR-F	C	3,065,400	3,065,400	
(md)	Federal block grant aids	PR-F	C	10,927,000	10,927,000	
(me)	Federal block grant local assistance	PR-F	C	9,359,500	9,359,500	
(n)	Federal program operations	PR-F	C	876,900	876,900	
(na)	Federal program aids	PR-F	C	835,100	835,100	
(nL)	Federal program local assistance	PR-F	C	-0-	-0-	
(o)	Federal aid; community aids	PR-F	C	12,249,100	12,249,100	
		(5) PRO	GRAM TO	OTALS		
	GENERAL PURPOSE REVENUE			18,139,800	17,889,800	
	PROGRAM REVENUE			48,917,100	49,167,600	
	FEDERAL			(42,647,400)	(42,447,900)	
	OTHER			(1,570,200)	(1,570,200)	
	SERVICE			(4,699,500)	(5,149,500)	

	E, AGENCY AND PURPOSE TOTAL–ALL SOURCES	Source	Түре	2017–2018 67,056,900	2018–2019 67,057,400
(6)	QUALITY ASSURANCE SERVICES PLANNING	G, REGULATI	ON AND I		
(a)	General program operations	GPR	A	5,451,900	5,451,900
(dm)	Nursing home monitoring and				
	receivership supplement	GPR	S	-0-	-0-
(g)	Nursing facility resident protection	PR	C	220,300	220,300
(ga)	Community-based residential facility				
	monitoring and receivership	DD	C	0	0
(*)	operations	PR	C	-0-	-0-
(i)	Gifts and grants	PR	C	_0_ 106 100	-0- 106 100
(jb)	Fees for administrative services	PR	C	196,100	196,100
(jm)	Licensing and support services Nursing home monitoring and	PR	A	5,406,800	5,406,800
(k)	receivership operations	PR	С	-0-	-0-
(kx)	Interagency and intra–agency	1 IX	C	-0-	-0-
(KA)	programs	PR-S	C	-0-	-0-
(ky)	Interagency and intra–agency aids	PR-S	C	-0-	-0-
(kz)	Interagency and intra-agency local				
` /	assistance	PR-S	C	-0-	-0-
(m)	Federal project operations	PR-F	C	-0-	-0-
(mc)	Federal block grant operations	PR-F	C	-0-	-0-
(n)	Federal program operations	PR-F	C	16,223,300	16,223,300
(na)	Federal program aids	PR-F	C	-0-	-0-
(nL)	Federal program local assistance	PR-F	C	-0-	-0-
		(6) PROC	GRAM 7	ΓOTALS	
	GENERAL PURPOSE REVENUE			5,451,900	5,451,900
]	PROGRAM REVENUE			22,046,500	22,046,500
	FEDERAL			(16,223,300)	(16,223,300)
	OTHER			(5,823,200)	(5,823,200)
	SERVICE			(-0-)	(-0-)
	TOTAL-ALL SOURCES			27,498,400	27,498,400
(7)	DISABILITY AND ELDER SERVICES				
(b)	Community aids and medical	GPR	Δ.	202 120 200	205 072 100
(ha)	assistance payments	GPR	A A	203,139,300 131,200	205,072,100
(bt)	Grants for community programs Early intervention services for	UFK	А	131,200	131,200
(01)	infants and toddlers with disabilities	GPR	C	5,789,000	5,789,000
(ky)	Interagency and intra–agency aids	PR-S	C	-0-	-0-
(kz)	Interagency and intra–agency local	111 5	C	v	v
(112)	assistance	PR-S	C	1,257,800	1,257,800
(ma)	Federal project aids	PR-F	C	-0-	-0-
(mb)	Federal project local assistance	PR-F	C	-0-	-0-
(md)	Federal block grant aids	PR-F	C	-0-	-0-
(me)	Federal block grant local assistance	PR-F	C	-0-	-0-
(na)	Federal program aids	PR-F	C	25,577,000	25,577,000
(nL)	Federal program local assistance	PR-F	C	7,560,000	7,560,000
(o)	Federal aid; community aids	PR-F	C	36,502,500	36,421,900
		(7) PROC	GRAM 7		
	GENERAL PURPOSE REVENUE			209,059,500	210,992,300
]	PROGRAM REVENUE			70,897,300	70,816,700
	FEDERAL			(69,639,500)	(69,558,900)

STATUTE, AGENCY AND PURPOSE SERVICE SOURCE Type 2017-2018 (1.257,800) (1.207,800) (1.257,800) (1.2
(8) GENERAL ADMINISTRATION (a) General program operations GPR A 14,973,100 14,973,100 (b) Inspector general; general operations GPR A 4,537,200 4,537,200 (c) Inspector general; local assistance GPR A 750,000 750,000 (i) Gifts and grants PR C 10,000 10,000 (k) Administrative and support services PR-S A 29,245,300 29,245,300 (kw) Inspector general; interagency and intra-agency programs PR-S C 778,100 778,100 (kx) Interagency and intra-agency programs PR-S C 41,800 41,800 (ky) Interagency and intra-agency place PR-S C 2,000,000 2,000,000 (kz) Interagency and intra-agency local assistance PR-S C -000- (m) Federal project operations PR-F C -00- (m) Income augmentation services receipts PR-F C 1,418,100 1,418,100 (mc) Federal block grant operations PR-F C 925,900 922,400 (mm) Reimbursements from federal government PR-F C 3,029,600 3,029,600 (o) Inspector general; federal program operations PR-F C 1,033,200 1,033,200 (p) Inspector general; federal program operations PR-F C 2,491,900 2,491,900 (p) Inspector general; federal program operations PR-F C 2,491,900 2,491,900 (ERFORMAN EVENUE PROGRAM REVENUE PROGRAM REVENUE PROGRAM REVENUE PROGRAM REVENUE PROGRAM REVENUE 48,201,200 48,197.700 (ERFORMAN EVENUE PROGRAM REVENUE PROGRAM REVENUE 48,201,200 48,197.700 (ERFORMAN EVENUE 48,201,200
(a) General program operations (BPR A 14,973,100 14,973,100 (b) Inspector general; general operations (GPR A 4,537,200 4,537,200 (c) Inspector general; local assistance (GPR A 750,000 750,000 (i) Gifts and grants (Rifts and grants (Rifts and grants) (Rifts and
(b) Inspector general; general operations (c) Inspector general; local assistance (c) Inspector general; local assistance (d) GPR A 750,000 750,000 (i) Gifts and grants PR C 10,000 10,000 (k) Administrative and support services PR-S A 29,245,300 29,245,300 (kw) Inspector general; interagency and intra-agency programs PR-S C 778,100 778,100 (kx) Interagency and intra-agency programs PR-S C 41,800 41,800 (ky) Interagency and intra-agency local assistance PR-S C 2,000,000 2,000,000 (kz) Interagency and intra-agency local assistance PR-F C -00- (ma) Federal project operations PR-F C -00- (mb) Income augmentation services receipts PR-F C 925,900 922,400 (mm) Reimbursements from federal government PR-F C 3,029,600 3,029,600 (o) Inspector general; federal program operations PR-F C 1,033,200 1,033,200 (p) Inspector general; federal program operations PR-F C 1,033,200 1,033,200 (p) Inspector general; federal program operations PR-F C 2,491,900 2,491,900 (8) PROGRAM TOTALS GENERAL PURPOSE REVENUE PROGRAM TOTALS (32,065,200) (32,065,200)
(c) Inspector general; local assistance GPR A 750,000 750,000 (i) Gifts and grants PR C 10,000 10,000 (k) Administrative and support services PR-S A 29,245,300 29,245,300 (kw) Inspector general; interagency and intra-agency programs PR-S C 778,100 778,100 (kx) Interagency and intra-agency aids programs PR-S C 41,800 41,800 (ky) Interagency and intra-agency local assistance PR-S C -0- -0- (m) Federal project operations PR-F C -0- -0- (ma) Federal project operations PR-F C -0- -0- (mb) Income augmentation services receipts PR-F C 1,418,100 1,418,100 (mc) Federal block grant operations PR-F C 925,900 922,400 (mm) Reimbursements from federal government PR-F C 3,029,600 3,029,600 (o)
(i) Gifts and grants PR C 10,000 10,000 (k) Administrative and support services (kw) Inspector general; interagency and intra-agency programs PR-S C 778,100 778,100 (kx) Interagency and intra-agency programs PR-S C 41,800 41,800 (ky) Interagency and intra-agency aids (kz) Interagency and intra-agency aids (kz) Interagency and intra-agency local assistance PR-S C 2,000,000 2,000,000 (kz) Interagency and intra-agency local assistance PR-F C -00- (m) Federal project operations PR-F C -00- (m) Federal project aids PR-F C 1,418,100 1,418,100 (mc) Federal block grant operations PR-F C 925,900 922,400 (mm) Reimbursements from federal government PR-F C 3,029,600 3,029,600 (o) Inspector general; federal program local assistance PR-F C 1,033,200 1,033,200 (pz) Indirect cost reimbursements PR-F C 2,491,900 2,491,900 (REPAGRAM REVENUE PROGRAM REVENUE PROGRAM REVENUE PROGRAM REVENUE 48,201,200 48,197,700 FEDERAL (16,126,000) (16,122,500) OTHER (10,000) (32,065,200)
(k) Administrative and support services (kw) Inspector general; interagency and intra-agency programs (kw) Inspector general; interagency and intra-agency programs (kx) Interagency and intra-agency programs (ky) Interagency and intra-agency aids (ky) Interagency and intra-agency aids (ky) Interagency and intra-agency aids (kz) Interagency and intra-agency local assistance (kz) Interagency and intra-agency local assistance (kz) Interagency and intra-agency local assistance (kz) (kz) (kz) (kz) (kz) (kz) (kz) (kz)
(kw) Inspector general; interagency and intra-agency programs PR-S C 778,100 778,100 (kx) Interagency and intra-agency programs PR-S C 41,800 41,800 (ky) Interagency and intra-agency aids PR-S C 2,000,000 2,000,000 (kz) Interagency and intra-agency local assistance PR-S C -0- -0- (m) Federal project operations PR-F C -0- -0- (ma) Federal project aids PR-F C -0- -0- (ma) Federal project aids PR-F C -0- -0- (mb) Income augmentation services receipts PR-F C 1,418,100 1,418,100 (mc) Federal block grant operations PR-F C 925,900 922,400 (mm) Reimbursements from federal government PR-F C 3,029,600 3,029,600 (n) Federal program operations PR-F C 3,029,600 3,029,600 (p)
(kx) Interagency and intra-agency programs
PR-S C 41,800 41,800
(ky) Interagency and intra-agency aids PR-S C 2,000,000 2,000,000 (kz) Interagency and intra-agency local assistance PR-S C -0- -0- (m) Federal project operations PR-F C -0- -0- (ma) Federal project aids PR-F C -0- -0- (mb) Income augmentation services receipts PR-F C 1,418,100 1,418,100 (mc) Federal block grant operations PR-F C 925,900 922,400 (mm) Reimbursements from federal government PR-F C -0- -0- (m) Federal program operations PR-F C 3,029,600 3,029,600 (o) Inspector general; federal program operations PR-F C 1,033,200 1,033,200 (pz) Indirect cost reimbursements PR-F C 7,227,300 7,227,300 (pz) Indirect cost reimbursements PR-F C 2,491,900 2,491,900 (a) PROGRAM
(kz) Interagency and intra-agency local assistance PR-S C -0- -0- (m) Federal project operations PR-F C -0- -0- (ma) Federal project aids PR-F C -0- -0- (mb) Income augmentation services receipts PR-F C 1,418,100 1,418,100 (mc) Federal block grant operations PR-F C 925,900 922,400 (mm) Reimbursements from federal government PR-F C 925,900 922,400 (mm) Reimbursements from federal government PR-F C 3,029,600 3,029,600 (o) Inspector general; federal program local assistance PR-F C 3,029,600 3,029,600 (p) Inspector general; federal program operations PR-F C 1,033,200 1,033,200 (pz) Indirect cost reimbursements PR-F C 7,227,300 7,227,300 (pz) Indirect cost reimbursements PR-F C 2,491,900 2,491,900
assistance PR-S C -00- (0) (m) Federal project operations PR-F C -0- (0) (ma) Federal project aids PR-F C -0- (0) (ma) Federal project aids PR-F C -0- (0) (mb) Income augmentation services receipts PR-F C 1,418,100 1,418,100 (mc) Federal block grant operations PR-F C 925,900 922,400 (mm) Reimbursements from federal government PR-F C -0- (0) (ma) Federal program operations PR-F C 3,029,600 3,029,600 (no) Inspector general; federal program local assistance PR-F C 1,033,200 1,033,200 (p) Inspector general; federal program operations PR-F C 7,227,300 7,227,300 (pz) Indirect cost reimbursements PR-F C 2,491,900 2,491,900 (REMERAL PURPOSE REVENUE PROGRAM TOTALS GENERAL PURPOSE REVENUE PROGRAM REVENUE 48,201,200 48,197,700 FEDERAL (16,126,000) (16,122,500) OTHER (10,000) (10,000) SERVICE (32,065,200) (32,065,200)
(m) Federal project operations PR-F C -0- -0- (ma) Federal project aids PR-F C -0- -0- (mb) Income augmentation services receipts PR-F C 1,418,100 1,418,100 (mc) Federal block grant operations PR-F C 925,900 922,400 (mm) Reimbursements from federal government PR-F C -0- -0- (n) Federal program operations PR-F C 3,029,600 3,029,600 (o) Inspector general; federal program local assistance PR-F C 1,033,200 1,033,200 (p) Inspector general; federal program operations PR-F C 7,227,300 7,227,300 (pz) Indirect cost reimbursements PR-F C 2,491,900 2,491,900 (g) PROGRAM REVENUE PR-F C 2,491,900 20,260,300 PROGRAM REVENUE 48,201,200 48,197,700 48,197,700 FEDERAL (10,000) (16,122,500) </td
(ma) Federal project aids PR-F C -0- -0- (mb) Income augmentation services receipts PR-F C 1,418,100 1,418,100 (mc) Federal block grant operations PR-F C 925,900 922,400 (mm) Reimbursements from federal government PR-F C -0- -0- (n) Federal program operations PR-F C 3,029,600 3,029,600 (o) Inspector general; federal program local assistance PR-F C 1,033,200 1,033,200 (p) Inspector general; federal program operations PR-F C 7,227,300 7,227,300 (pz) Indirect cost reimbursements PR-F C 2,491,900 2,491,900 (general program operations PR-F C 2,260,300 20,260,300 (pz) Indirect cost reimbursements PR-F C 2,491,900 2491,900 (general program operations PR-F C 2,491,900 20,260,300 20,260,300 (general program o
(mb) Income augmentation services receipts PR-F C 1,418,100 1,418,100 (mc) Federal block grant operations PR-F C 925,900 922,400 (mm) Reimbursements from federal government PR-F C -0- -0- (n) Federal program operations PR-F C 3,029,600 3,029,600 (o) Inspector general; federal program local assistance PR-F C 1,033,200 1,033,200 (p) Inspector general; federal program operations PR-F C 7,227,300 7,227,300 (pz) Indirect cost reimbursements PR-F C 2,491,900 2,491,900 (g) PROGRAM REVENUE 20,260,300 20,260,300 20,260,300 PROGRAM REVENUE 48,201,200 48,197,700 FEDERAL (16,126,000) (16,122,500) OTHER (10,000) (10,000) SERVICE (32,065,200) (32,065,200)
receipts PR-F C 1,418,100 1,418,100 (mc) Federal block grant operations PR-F C 925,900 922,400 (mm) Reimbursements from federal government PR-F C -00- (n) Federal program operations PR-F C 3,029,600 3,029,600 (o) Inspector general; federal program local assistance PR-F C 1,033,200 1,033,200 (p) Inspector general; federal program operations PR-F C 7,227,300 7,227,300 (pz) Indirect cost reimbursements PR-F C 2,491,900 2,491,900 (8) PROGRAM TOTALS GENERAL PURPOSE REVENUE PROGRAM REVENUE FEDERAL (16,126,000) (16,122,500) OTHER (10,000) SERVICE (32,065,200) (32,065,200)
(mc) Federal block grant operations PR-F C 925,900 922,400 (mm) Reimbursements from federal government PR-F C -0- -0- (n) Federal program operations PR-F C 3,029,600 3,029,600 (o) Inspector general; federal program local assistance PR-F C 1,033,200 1,033,200 (p) Inspector general; federal program operations PR-F C 7,227,300 7,227,300 (pz) Indirect cost reimbursements PR-F C 2,491,900 2,491,900 (general program operations PR-F C 2,260,300 20,260,300 (pz) Indirect cost reimbursements PR-F C 2,491,900 2,491,900 (general program operations PR-F C 2,491,900 20,260,300 20,260,300 (general program operations PR-F C 2,491,900 20,260,300 20,260,300 (general program operations PR-F C 2,491,900 20,260,300 20,260,300
(mm) Reimbursements from federal government PR-F C -0- -10- -0- -0- -10- -10- -10- -0- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10-
government PR-F C -0- (n) Federal program operations PR-F C 3,029,600 3,029,600 (o) Inspector general; federal program local assistance PR-F C 1,033,200 1,033,200 (p) Inspector general; federal program operations PR-F C 7,227,300 7,227,300 (pz) Indirect cost reimbursements PR-F C 2,491,900 2,491,900 (8) PROGRAM TOTALS GENERAL PURPOSE REVENUE 20,260,300 20,260,300 PROGRAM REVENUE 48,201,200 48,197,700 FEDERAL (16,126,000) (16,122,500) OTHER (10,000) (10,000) SERVICE (32,065,200) (32,065,200)
(n) Federal program operations PR-F C 3,029,600 3,029,600 (o) Inspector general; federal program local assistance PR-F C 1,033,200 1,033,200 (p) Inspector general; federal program operations PR-F C 7,227,300 7,227,300 (pz) Indirect cost reimbursements PR-F C 2,491,900 2,491,900 (8) PROGRAM TOTALS GENERAL PURPOSE REVENUE PROGRAM REVENUE 48,201,200 48,197,700 FEDERAL (16,126,000) (16,122,500) OTHER (10,000) (32,065,200)
(o) Inspector general; federal program local assistance PR-F C 1,033,200 1,033,200 (p) Inspector general; federal program operations PR-F C 7,227,300 7,227,300 (pz) Indirect cost reimbursements PR-F C 2,491,900 2,491,900 (8) PROGRAM TOTALS GENERAL PURPOSE REVENUE PROGRAM REVENUE 48,201,200 48,197,700 FEDERAL (16,126,000) (16,122,500) OTHER (10,000) (32,065,200)
local assistance
operations PR-F C 7,227,300 7,227,300 (pz) Indirect cost reimbursements PR-F C 2,491,900 2,491,900 (8) PROGRAM TOTALS GENERAL PURPOSE REVENUE PROGRAM REVENUE 48,201,200 48,197,700 FEDERAL OTHER (10,000) (16,122,500) SERVICE (32,065,200) (32,065,200)
(pz) Indirect cost reimbursements PR-F C 2,491,900 2,491,900 (8) PROGRAM TOTALS 20,260,300 20,260,300 PROGRAM REVENUE 48,201,200 48,197,700 FEDERAL (16,126,000) (16,122,500) OTHER (10,000) (10,000) SERVICE (32,065,200) (32,065,200)
(8) PROGRAM TOTALS GENERAL PURPOSE REVENUE PROGRAM REVENUE FEDERAL OTHER (10,000) SERVICE (8) PROGRAM TOTALS 20,260,300 20,260,300 48,197,700 (16,122,500) (16,122,500) (10,000) (10,000) (32,065,200)
GENERAL PURPOSE REVENUE 20,260,300 20,260,300 PROGRAM REVENUE 48,201,200 48,197,700 FEDERAL (16,126,000) (16,122,500) OTHER (10,000) (10,000) SERVICE (32,065,200) (32,065,200)
PROGRAM REVENUE 48,201,200 48,197,700 FEDERAL (16,126,000) (16,122,500) OTHER (10,000) (10,000) SERVICE (32,065,200) (32,065,200)
FEDERAL (16,126,000) (16,122,500) OTHER (10,000) (10,000) SERVICE (32,065,200) (32,065,200)
OTHER (10,000) (10,000) SERVICE (32,065,200) (32,065,200)
SERVICE (32,065,200) (32,065,200)
TOTAL-ALL SOURCES 68,461,500 68,458,000
20.435 DEPARTMENT TOTALS
GENERAL PURPOSE REVENUE 3,825,217,200 4,002,096,800
PROGRAM REVENUE 7,263,974,800 7,679,626,400
FEDERAL (5,932,411,600) (6,265,299,100)
OTHER (1,230,098,600) (1,312,470,000)
SERVICE (101,464,600) (101,857,300)
SEGREGATED REVENUE 582,941,000 579,933,100
OTHER (582,941,000) (579,933,100)
TOTAL-ALL SOURCES 11,672,133,000 12,261,656,300
20.437 Children and Families, Department of
(1) CHILDREN AND FAMILY SERVICES
(a) General program operations GPR A 11,215,000 11,215,200
(ab) Child abuse and neglect prevention
(ac) Child abuse and neglect prevention technical assistance GPR A -00-

2017 As	2017 Assembly Bill 64		_	2017 Wisconsin Act 59		
STATUT	E, AGENCY AND PURPOSE	Source	ТүрЕ	2017-2018	2018-2019	
(b) (bc)	Children and family aids payments Grants for children's community	GPR	A	26,135,200	26,396,400	
	programs	GPR	A	625,200	625,200	
(bd)	Tribal family services grants	GPR	A	1,271,900	1,271,900	
(cd)	Domestic abuse grants	GPR	A	12,434,600	12,434,600	
(cf)	Foster parent insurance and liability	GPR	A	59,400	59,400	
(cj)	Community youth and family aids	GPR	A	88,591,400	88,591,400	
(cm)	Community intervention program	GPR	A	3,712,500	3,712,500	
(cw)	Milwaukee child welfare services; general program operations	GPR	A	18,808,000	18,853,200	
(av)	Child welfare services; aids	GPR	A	62,962,000		
(cx)		GPK	Α	02,902,000	62,883,600	
(dd)	State out-of-home care, guardianship, and adoption services	GPR	A	56,463,500	56,473,400	
(dg)	State adoption information exchange	CDD		160.600	160,600	
	and state adoption center	GPR	A	169,600	169,600	
(e)	Services for sex-trafficking victims	GPR	В	3,000,000	3,000,000	
(eg) (er)	Brighter futures initiative Grants for services for homeless and	GPR	A	864,900	864,900	
	runaway youth	GPR	A	100,000	100,000	
(f)	Second-chance homes	GPR	A	-0-	-0-	
(fm)	Literacy improvement aids	GPR	A	23,600	23,600	
(gg)	Collection remittances to local units					
	of government	PR	C	-0-	-0-	
(gx)	Milwaukee child welfare services;					
	collections	PR	C	4,487,200	4,487,200	
(hh)	Domestic abuse surcharge grants	PR	C	540,000	540,000	
(i)	Gifts and grants	PR	C	5,000	5,000	
(j)	Statewide automated child welfare		~	201 200	7 04 7 00	
	information system receipts	PR	C	581,300	581,300	
(jb)	Fees for administrative services	PR	C	78,000	78,000	
(jj)	Searches for birth parents and adoption record information; foreign					
	adoptions	PR	A	90,400	90,400	
(jm)	Licensing activities	PR	C	97,800	97,800	
(kb)	Interagency aids; brighter futures initiative	PR-S	C	865,000	865,000	
(km)	Interagency and intra-agency aids; children and family aids; local					
(kw)	assistance Interagency and intra–agency aids;	PR-S	С	7,288,700	7,273,400	
(kx)	Milwaukee child welfare services Interagency and intra–agency	PR-S	A	20,101,300	20,101,300	
` /	programs	PR-S	C	3,454,300	3,454,300	
(ky)	Interagency and intra-agency aids	PR-S	C	3,290,100	3,290,100	
(kz)	Interagency and intra-agency aids; tribal placements and guardianships	PR-S	A	717,500	717,500	
(m)	Federal project operations	PR-F	C	1,025,800	1,025,800	
(ma)	Federal project aids	PR-F	C	3,234,600	3,234,600	
(mb)	Federal project local assistance	PR-F	C	3,495,000	3,495,000	
(mc)	Federal block grant operations	PR-F	C	-0-	-0-	
(md)	Federal block grant aids	PR-F	C	-0-	-0-	
	-					

/OI / VVI	sconsin rice 37	00		2017	1155cmbiy bii 0-
	E, AGENCY AND PURPOSE	Source	Түре	2017-2018	2018-2019
(mw)	Federal aid; Milwaukee child welfare				
	services general program operations	PR-F	C	3,726,400	3,756,800
(mx)	Federal aid; Milwaukee child welfare				
	services aids	PR-F	C	15,798,700	16,088,700
(n)	Federal program operations	PR-F	C	10,894,100	10,780,400
(na)	Federal program aids	PR-F	C	12,872,800	12,606,200
(nL)	Federal program local assistance	PR-F	C	11,894,900	11,819,100
(o)	Federal aid; children, youth, and				
	family aids	PR-F	C	38,963,000	42,814,000
(pd)	Federal aid; state out-of-home care,				
	guardianship, and adoption services	PR-F	C	45,250,700	45,254,000
(pm)	Federal aid; adoption incentive				
	payments	PR-F	C	136,000	136,000
(q)	Grants for literacy and early				
	childhood development programs	SEG	S	-0-	-0-
		(1) PROG	RAM	TOTALS	
(GENERAL PURPOSE REVENUE			287,422,500	287,660,600
]	PROGRAM REVENUE			188,888,600	192,591,900
	FEDERAL			(147,292,000)	(151,010,600)
	OTHER			(5,879,700)	(5,879,700)
	SERVICE			(35,716,900)	(35,701,600)
9	SEGREGATED REVENUE			-0-	-0-
	OTHER			(-0-)	(-0-)
,	TOTAL-ALL SOURCES			476,311,100	480,252,500
(2)	ECONOMIC SUPPORT			170,511,100	100,232,300
(a)	General program operations	GPR	A	4,054,000	4,242,700
(bc)	Child support local assistance	GPR	C	8,500,000	8,500,000
, ,	Wisconsin works child care	GPR			
(cm)		GPK	A	28,849,400	28,849,400
(dz)	Temporary Assistance for Needy				
	Families programs; maintenance of effort	GPR	A	131,077,000	131,077,000
(2)	Incentive payments for identifying	OFK	A	131,077,000	131,077,000
(e)	children with health insurance	GPR	A	300,000	300,000
(am)		GPR	A	250,000	250,000
(em)	Drug testing and treatment costs				,
(f)	Emergency Shelter of the Fox Valley	GPR	A	50,000	50,000
(fr)	Skills enhancement grants	GPR	A	250,000	250,000
(i)	Gifts and grants	PR	C	2,500	2,500
(ja)	Child support state operations – fees,	DD	0	10 202 000	10 202 000
(1)	reimbursements, and collections	PR	C	19,393,900	19,393,900
(jb)	Fees for administrative services	PR	C	725,000	725,000
(jL)	Job access loan repayments	PR	C	610,200	610,200
(jn)	Child care licensing and certification		~	4 6 7 0 0 0 0	4 5 7 0 0 0 0
	activities	PR	C	1,650,000	1,650,000
(k)	Child support transfers	PR-S	C	7,027,800	7,027,800
(kx)	Interagency and intra-agency		_		
	programs	PR-S	C	3,182,300	2,754,400
(L)	Public assistance overpayment				
	recovery, fraud investigation, and	D D		4 4 0 4 0 -	
, .	error reduction	PR	C	160,600	160,600
(ma)	Federal project activities and	- nn			
, .	administration	PR-F	C	4,139,700	161,000
(mc)	Federal block grant operations	PR-F	A	50,079,000	49,835,500

J17 As	sembly Bill 64	- 89 -	_	2017 V	Visconsin Act
Statuti	E, AGENCY AND PURPOSE	Source	Түре	2017-2018	2018-2019
(md)	Federal block grant aids	PR-F	A	375,107,900	404,317,300
(me)	Child care and temporary assistance				
	overpayment recovery	PR–F	C	4,287,600	4,287,600
(mg)	Community services block grant; federal funds	PR-F	C	8,659,900	8,659,900
(mm)	Reimbursements from federal government	PR-F	C	-0-	-0-
(n)	Child support state operations; federal funds	PR-F	С	18,078,300	18,039,40
(nL)	Child support local assistance; federal funds	PR–F	С	70,340,600	70,340,600
(om)	Refugee assistance; federal funds	PR-F	C	5,616,800	5,616,800
(q)	Centralized support receipt and	IKI	C	3,010,000	3,010,000
	disbursement; interest	SEG	S	35,000	35,000
(qm)	Child support state operations and reimbursement for claims and				
	expenses; unclaimed payments	SEG	S	100,000	100,000
(s)	Economic support – public benefits	SEG	A	9,139,700	9,139,70
(5)	Zeonomie support public cenemis		GRAM TO		3,133,70
(GENERAL PURPOSE REVENUE	(2) TRO	310 111 10	173,330,400	173,519,100
	PROGRAM REVENUE			569,062,100	593,582,500
1	FEDERAL			(536,309,800)	(561,258,100
	OTHER				
				(22,542,200)	(22,542,200
	SERVICE			(10,210,100)	(9,782,200
2	SEGREGATED REVENUE			9,274,700	9,274,700
-	OTHER			(9,274,700)	(9,274,700
	TOTAL-ALL SOURCES			751,667,200	776,376,300
` ′	GENERAL ADMINISTRATION	CDD		1 041 200	1 0 4 1 2 0
(a)	General program operations	GPR	A	1,841,300	1,841,30
(i)	Gifts and grants	PR	C	5,000	5,000
(jb)	Fees for administrative services	PR	C	-0-	-0-
(k)	Administrative and support services	PR-S	A	22,085,500	22,120,500
(kp)	Interagency and intra–agency aids; income augmentation services	DD 0	~	0.50.000	7.50 000
<i>a</i> >	receipts	PR-S	C	950,000	550,000
(kx)	Interagency and intra–agency	PR-S	С	14,786,400	14,786,40
(1)	programs				
(ky) (kz)	Interagency and intra–agency aids Interagency and intra–agency local	PR-S	С	-0-	-0
()	assistance	PR-S	C	-0-	-0-
(mc)	Federal block grant operations	PR-F	C	-0-	-0-
(md)	Federal block grant aids	PR-F	C	-0-	-0-
(mf)	Federal economic stimulus funds	PR-F	C	-0-	-0-
(mm)	Reimbursements from federal government	PR-F	С	-0-	-0-
(n)	Federal project activities	PR-F	C	-0-	-0-
(pz)	Indirect cost reimbursements	PR-F	C	-0-	-0-
(PL)	The state of the s		GRAM TO		-0
,	GENERAL PURPOSE REVENUE	(2) I KO	5141111 10	1,841,300	1,841,300
				1,071,000	1,071,500
	PROGRAM REVENUE			37,826,900	37,461,900

2017 W	isconsin Act 59	- 90 -	_	2017	Assembly Bill 64
STATU	TE, AGENCY AND PURPOSE	Source	Түре	2017-2018	2018-2019
	OTHER			(5,000)	(5,000)
	SERVICE			(37,821,900)	(37,456,900)
	TOTAL-ALL SOURCES			39,668,200	39,303,200
		20.437 DEPA	ARTMENT	TOTALS	
	GENERAL PURPOSE REVENUE			462,594,200	463,021,000
	PROGRAM REVENUE			795,777,600	823,636,300
	FEDERAL			(683,601,800)	(712,268,700)
	OTHER			(28,426,900)	(28,426,900)
	SERVICE			(83,748,900)	(82,940,700)
	SEGREGATED REVENUE			9,274,700	9,274,700
	OTHER			(9,274,700)	(9,274,700)
	TOTAL-ALL SOURCES			1,267,646,500	1,295,932,000
20.438	Board for People with Developmen	ntal Disabiliti	es		
(1)	DEVELOPMENTAL DISABILITIES				
(a)	General program operations	GPR	A	42,600	43,400
(h)	Program services	PR	C	-0-	-0-
(i)	Gifts and grants	PR	C	-0-	-0-
(mc)	Federal project operations	PR-F	C	881,300	882,500
(md)	Federal project aids	PR-F	C	543,600	543,600
		(1) PRO	GRAM TO	OTALS	
	GENERAL PURPOSE REVENUE			42,600	43,400
	PROGRAM REVENUE			1,424,900	1,426,100
	FEDERAL			(1,424,900)	(1,426,100)
	OTHER			(-0-)	(-0-)
	TOTAL-ALL SOURCES			1,467,500	1,469,500
		20.438 DEPA	ARTMENT	TOTALS	
	GENERAL PURPOSE REVENUE			42,600	43,400
	PROGRAM REVENUE			1,424,900	1,426,100
	FEDERAL			(1,424,900)	(1,426,100)
	OTHER			(-0-)	(-0-)
	TOTAL-ALL SOURCES			1,467,500	1,469,500
20.440	Health and Educational Facilities	Authority			
(1)	CONSTRUCTION OF HEALTH AND EDUC	CATIONAL FACIL	LITIES		
(a)	General program operations	GPR	C	-0-	-0-
		(1) PRO	GRAM TO	DTALS	
	GENERAL PURPOSE REVENUE			-0-	-0-
	TOTAL-ALL SOURCES			-0-	-0-
(2)	RURAL HOSPITAL LOAN GUARANTEE				
(a)	Rural assistance loan fund	GPR	C	-0-	-0-
		(2) PRO	GRAM TO	OTALS	
	GENERAL PURPOSE REVENUE			-0-	-0-
	TOTAL-ALL SOURCES			-0-	-0-
		20.440 DEPA	ARTMENT	TOTALS	
	GENERAL PURPOSE REVENUE			-0-	-0-
	TOTAL-ALL SOURCES			-0-	-0-
20.445	Workforce Development, Departm	nent of			
(1)	WORKFORCE DEVELOPMENT				
(a)	General program operations	GPR	A	7,927,700	7,946,000
(aa)		GPR	S	525,000	525,000
	Special death benefit				

017 Assembly Bill 64		- 91 -		2017 Wisconsin Act 59	
STATUT	E, AGENCY AND PURPOSE	Source	Түре	2017-2018	2018-2019
(aL)	Unemployment insurance administration; controlled substances		_	•••	
(b)	testing and treatment Workforce training; programs, grants	GPR	В	250,000	250,000
(0)	and services	GPR	C	26,095,900	13,595,900
(bm)	Workforce training; administration	GPR	В	3,582,000	3,582,000
(bt)	Workforce development; grants for				
	teacher training and recruitment	GPR	В	500,000	500,000
(cr)	State supplement to employment	CDD		200 (00	200 (0)
<i>(</i> 1)	opportunity demonstration projects	GPR	A	200,600	200,600
(d)	Reimbursement for tuition payments	GPR	A	-0-	1,753,500
(f)	Death and disability benefit payments; public insurrections	GPR	S	-0-	-0-
(g)	Gifts and grants	PR	C	-0-	-0- -0-
(ga)	Auxiliary services	PR	C	379,800	379,800
(ga)	Local agreements	PR	C	261,500	261,500
(gc)	Unemployment administration	PR	C	-0 -	-0-
(gd)	Unemployment interest and penalty	TIC	C	V	Ü
(gu)	payments	PR	C	1,859,100	1,864,700
(gg)	Unemployment information technology systems; interest and			, ,	, ,
	penalties	PR	C	-0-	-0-
(gh)	Unemployment information		~		
	technology systems; assessments	PR	C	-0-	-0-
(gk)	Child labor permit system; fees	PR	A	367,500	367,500
(gm) (gr)	Unemployment insurance handbook Agricultural education and workforce development council, gifts	PR	С	-0-	-0-
	and grants	PR	C	-0-	-0-
(ka)	Interagency and intra-agency				
	agreements	PR-S	C	36,755,200	36,755,200
(kc)	Administrative services	PR-S	A	35,102,300	35,145,500
(km)	Nursing workforce survey and grants	PR-S	C	155,600	155,600
(m)	Workforce investment and assistance; federal moneys	PR-F	C	74,650,900	74,650,900
(n)	Employment assistance and				
	unemployment insurance administration; federal moneys	PR-F	С	52,752,900	52,843,900
(na)	Employment security buildings and	1 IX-1	C	32,732,700	32,043,700
()	equipment	PR-F	C	-0-	-0-
(nb)	Unemployment administration; information technology systems	PR-F	C	-0-	-0-
(nd)	Unemployment administration; apprenticeship and other	DD 5		500.000	500.000
()	employment services	PR–F	A	523,000	523,000
(ne)	Unemployment insurance administration and bank service costs	PR-F	C	-0-	-0-
(o)	Equal rights; federal moneys	PR-F	C	850,200	850,200
(p)	Worker's compensation; federal	DD E	C	^	-
()	moneys	PR-F	C	-0- 25 200	-0- 25 200
(pz)	Indirect cost reimbursements	PR-F	С	25,300	25,30

2017 Wisconsin Act 59		- 92 -		2017	2017 Assembly Bill 64		
STATUT	TE, AGENCY AND PURPOSE	Source	Түре	2017-2018	2018-2019		
(ra)	Worker's compensation operations fund; administration	SEG	A	12,701,900	12,711,000		
(rb)	Worker's compensation operations fund; contracts	SEG	C	93,900	93,900		
(rp)	Worker's compensation operations fund; uninsured employers program;						
	administration	SEG	A	1,130,200	1,130,200		
(s)	Self-insured employers liability fund	SEG	C	-0-	-0-		
(sm) (t)	Uninsured employers fund; payments Work injury supplemental benefit fund	SEG SEG	S C	5,500,000 5,360,000	5,500,000 5,360,000		
(u)	Unemployment interest payments			3,300,000	3,300,000		
	and transfers	SEG	S	-0-	-0-		
(v)	Unemployment program integrity	SEG	S	-0-	-0-		
		(1) PROC	GRAM '				
	GENERAL PURPOSE REVENUE			39,081,200	28,353,000		
	PROGRAM REVENUE			203,683,300	203,823,100		
	FEDERAL			(128,802,300)	(128,893,300)		
	OTHER			(2,867,900)	(2,873,500)		
	SERVICE			(72,013,100)	(72,056,300)		
	SEGREGATED REVENUE			24,786,000	24,795,100		
	OTHER			(24,786,000)	(24,795,100)		
	TOTAL-ALL SOURCES			267,550,500	256,971,200		
(5)	VOCATIONAL REHABILITATION SERVICES						
(a)	General program operations;						
	purchased services for clients	GPR	C	17,446,800	17,446,800		
(gg)	Contractual services	PR	C	-0-	-0-		
(gp)	Contractual aids	PR	C	-0-	-0-		
(h)	Enterprises and services for blind	DD	C	140 100	140 100		
(1, .)	and visually impaired	PR	C	149,100	149,100		
(he)	Supervised business enterprise	PR PR	C C	125,000	125,000		
(i)	Gifts and grants	PK	C	1,000	1,000		
(kg)	Vocational rehabilitation services for tribes	PR-S	A	314,900	314,900		
(kx)	Interagency and intra–agency programs	PR-S	C	-0-	-0-		
(ky)	Interagency and intra-agency aids	PR-S	C	-0-	-0-		
(kz)	Interagency and intra-agency local						
	assistance	PR-S	C	-0-	-0-		
(m)	Federal project operations	PR-F	C	50,000	50,000		
(ma)	Federal project aids	PR-F	C	6,388,400	5,970,000		
(n)	Federal program aids and operations	PR-F	C	68,296,800	68,296,800		
(nL)	Federal program local assistance	PR-F (5) PROC	C GRAM '	–0– TOTALS	-0-		
	GENERAL PURPOSE REVENUE	(S) TROC		17,446,800	17,446,800		
	PROGRAM REVENUE			75,325,200	74,906,800		
	FEDERAL			(74,735,200)	(74,316,800)		
	OTHER			(275,100)	(275,100)		
	SERVICE			(314,900)	(314,900)		
	TOTAL-ALL SOURCES			92,772,000	92,353,600		
).445 DEPA	RTMEN	NT TOTALS	2,555,000		

2017 As	sembly Bill 64	- 93 -		2017 Wisconsin Act 59		
(E, AGENCY AND PURPOSE GENERAL PURPOSE REVENUE PROGRAM REVENUE	Source	ТүрЕ	2017–2018 56,528,000 279,008,500	2018–2019 45,799,800 278,729,900	
•	FEDERAL			(203,537,500)	(203,210,100)	
	OTHER			(3,143,000)	(3,148,600)	
	SERVICE			(72,328,000)	(72,371,200)	
S	SEGREGATED REVENUE			24,786,000	24,795,100	
	OTHER			(24,786,000)	(24,795,100)	
	TOTAL-ALL SOURCES			360,322,500	349,324,800	
	Justice, Department of			200,222,200	0.5,02.1,000	
	Legal services					
(a)	General program operations	GPR	A	13,969,100	13,970,200	
(d)	Legal expenses	GPR	В	738,700	738,800	
(gh)	Investigation and prosecution	PR	C	650,700	650,700	
(gs)	Delinquent obligation collection	PR	A	7,000	7,000	
(hm)	Restitution	PR	C	-0-	-0-	
(k)	Environment litigation project	PR-S	C	482,000	482,000	
(km)	Interagency and intra–agency					
, ,	assistance	PR-S	C	1,675,800	1,679,700	
(m)	Federal aid	PR-F	C	1,094,800	1,095,900	
		(1) PROC	GRAM T	OTALS		
(GENERAL PURPOSE REVENUE			14,707,800	14,709,000	
I	PROGRAM REVENUE			3,910,300	3,915,300	
	FEDERAL			(1,094,800)	(1,095,900)	
	OTHER			(657,700)	(657,700)	
	SERVICE			(2,157,800)	(2,161,700)	
7	TOTAL-ALL SOURCES			18,618,100	18,624,300	
(2)	LAW ENFORCEMENT SERVICES					
(a)	General program operations	GPR	A	22,575,900	22,635,700	
(am)	Officer training reimbursement	GPR	S	150,000	150,000	
(b)	Investigations and operations	GPR	A	-0-	-0-	
(c)	Crime laboratory equipment	GPR	В	-0-	-0-	
(cv)	Shot Spotter Program	GPR	A	175,000	175,000	
(dg)	Weed and seed and law enforcement					
	technology	GPR	A	-0-	-0-	
(eg)	Drug courts	GPR	A	500,000	500,000	
(em)	Alternatives to prosecution and incarceration for persons who use alcohol or other drugs; presentencing					
(g)	assessments Gaming law enforcement; racing	GPR	A	2,500,000	2,500,000	
	revenues	PR	A	-0-	-0-	
(gb)	Gifts and grants	PR	C	-0-	-0-	
(gc)	Gaming law enforcement; Indian gaming	PR	A	165,600	165,800	
(gm)	Criminal history searches; fingerprint identification	DD	C	4 041 700	4.055.000	
(~~)	Crime information alerts	PR PR	C C	4,941,700	4,955,900	
(gp) (gr)	Handgun purchaser record check; checks for licenses or certifications			50,200	50,200	
	to carry concealed weapons	PR	C	2,588,000	2,305,000	
(gu)	Sobriety programs	PR	A	-0-	-0-	

2017 Wi	017 Wisconsin Act 59		_	2017 Assembly Bill 64	
STATUT	E, AGENCY AND PURPOSE	Source	Түре	2017-2018	2018-2019
(h)	Terminal charges	PR	A	2,055,800	2,056,000
(hd)	Internet crimes against children	PR	C	750,000	750,000
(i)	Penalty surcharge, receipts	PR	A	-0-	-0-
(j)	Law enforcement training fund, local assistance	PR-S	A	4,364,800	4,364,800
(ja)	Law enforcement training fund, state operations	PR-S	A	3,255,800	3,260,600
(jb)	Crime laboratory equipment and supplies	PR-S	A	683,100	824,100
(jc)	Law enforcement overtime grants	PR	A	1,000,000	1,000,000
(jd)	Alternatives to incarceration grant	1 K	Λ	1,000,000	1,000,000
	program	PR	A	250,000	250,000
(k)	Interagency and intra–agency assistance	PR-S	С	1,467,000	1,467,300
(kb)	Law enforcement officer supplement grants	PR-S	A	1,224,900	1,224,900
(kc)	Transaction information management of enforcement system	PR-S	A	715,500	715,700
(kd)	Drug law enforcement, crime laboratories, and genetic evidence				
(ke)	activities Drug enforcement intelligence	PR-S	A	9,126,200	9,150,900
(kg)	operations Interagency and intra–agency	PR-S	A	1,710,900	1,714,600
(6)	assistance; fingerprint identification	PR-S	A	-0-	-0-
(kj)	Youth diversion program	PR-S	A	672,400	672,400
(km)	Lottery background investigations	PR-S	A	-0-	-0-
(kn)	Alternatives to prosecution and incarceration for persons who use alcohol or other drugs; justice				
(ko)	information fee Wisconsin justice information	PR-S	A	1,078,400	1,078,400
, ,	sharing program Drug crimes enforcement; local	PR-S	A	657,800	657,900
(kp)	grants	PR-S	A	717,900	717,900
(kq)	County law enforcement services	PR-S	A	490,000	490,000
(kr)	Alternatives to prosecution and incarceration for persons who use				
(kt)	alcohol and other drugs; grants County–tribal programs, local	PR	С	-0-	-0-
(ku)	assistance County-tribal programs, state	PR-S	A	631,200	631,200
	operations Grants for substance abuse treatment	PR-S	A	95,500	95,600
(kv)	programs for criminal offenders	PR	C	10,600	10,800
(kw)	Tribal law enforcement assistance	PR-S	A	695,000	695,000
(ky)	Law enforcement programs and youth diversion – administration	PR-S	A	143,600	143,700
(Lm)	Crime laboratories; deoxyribonucleic acid analysis	PR-S	C	5,813,100	5,826,300
(Lp)	Crime laboratories; deoxyribonucleic acid analysis surcharges	PR	С	-0-	-0-
	-				

017 A	ssembly Bill 64	- 95 -		2017 Wisconsin Act 59	
STATU	TE, AGENCY AND PURPOSE	Source	Түре	2017-2018	2018-2019
(m)	Federal aid, state operations	PR-F	C	3,647,200	3,483,700
(n)	Federal aid, local assistance	PR-F	C	5,755,000	5,755,000
(r)	Gaming law enforcement; lottery				
	revenues	SEG	A	351,900	352,300
		(2) PRO	GRAM TO	TALS	
	GENERAL PURPOSE REVENUE			25,900,900	25,960,700
	PROGRAM REVENUE			54,757,200	54,513,700
	FEDERAL			(9,402,200)	(9,238,700)
	OTHER			(11,811,900)	(11,543,700)
	SERVICE			(33,543,100)	(33,731,300)
	SEGREGATED REVENUE			351,900	352,300
	OTHER			(351,900)	(352,300)
	TOTAL-ALL SOURCES			81,010,000	80,826,700
(3)	ADMINISTRATIVE SERVICES				
(a)	General program operations	GPR	A	6,381,400	6,392,000
(g)	Gifts, grants and proceeds	PR	C	-0-	-0-
(m)	Federal aid, state operations	PR-F	C	-0-	-0-
(pz)	Indirect cost reimbursements	PR-F	C	609,300	609,300
_		(3) PRO0	GRAM TO	TALS	
	GENERAL PURPOSE REVENUE			6,381,400	6,392,000
	PROGRAM REVENUE			609,300	609,300
	FEDERAL			(609,300)	(609,300
	OTHER			(-0-)	(-0-
	TOTAL-ALL SOURCES			6,990,700	7,001,300
(5)	VICTIMS AND WITNESSES				
(a)	General program operations	GPR	A	1,296,500	1,297,400
(b)	Awards for victims of crimes	GPR	A	2,388,100	2,388,100
(br)	Global positioning system tracking	GPR	A	-0-	-0-
(d)	Reimbursement for forensic				
` '	examinations	GPR	S	600,000	600,000
(e)	Sexual assault victim services	GPR	A	2,138,700	2,138,700
(es)	Court appointed special advocates	GPR	A	-0-	-0-
(g)	Crime victim and witness assistance				
_	surcharge, general services	PR	A	6,752,500	6,752,800
(gj)	General operations; child				
	pornography surcharge	PR	C	75,000	75,000
(h)	Crime victim compensation services	PR	A	72,000	72,100
(hh)	Crime victim restitution	PR	C	267,300	267,300
(i)	Victim compensation, inmate				
	payments	PR	C	10,600	10,700
(k)	Interagency and intra-agency				
	assistance; reimbursement to				
	counties	PR-S	A	566,300	566,400
(ke)	Child advocacy centers	PR-S	A	238,000	238,000
(kp)	Reimbursement to counties for	DD C		7.40.000	5 40.000
<i>a</i> :	victim-witness services	PR-S	A	748,900	748,900
(kr)	Court appointed special advocates	PR	A	80,000	80,000
(m)	Federal aid; victim compensation	PR-F	C	1,823,900	1,823,900
(ma)		DD E	C	000.000	000.00
, .	to crime victim services	PR-F	C	988,900	989,000
(mh)	Federal aid; victim assistance	PR-F	C	9,407,200	9,411,000

STATUTE, AGENCY AND PURPOSE	Source	түрг	2017–2018	2018–2019
STATULE, AGENCI AND I UNIOSE			TOTALS	2010-2017
GENERAL PURPOSE REVENUE	(-)		6,423,300	6,424,200
PROGRAM REVENUE			21,030,600	21,035,100
FEDERAL			(12,220,000)	(12,223,900)
OTHER			(7,257,400)	(7,257,900)
SERVICE			(1,553,200)	(1,553,300)
TOTAL-ALL SOURCES			27,453,900	27,459,300
	20.455 DEF	PARTME	ENT TOTALS	
GENERAL PURPOSE REVENUE			53,413,400	53,485,900
PROGRAM REVENUE			80,307,400	80,073,400
FEDERAL			(23,326,300)	(23,167,800)
OTHER			(19,727,000)	(19,459,300)
SERVICE SECRECATED DEVENUE			(37,254,100)	(37,446,300)
SEGREGATED REVENUE OTHER			351,900 (351,900)	352,300
TOTAL-ALL SOURCES			134,072,700	(352,300) 133,911,600
20.465 Military Affairs, Department of			134,072,700	155,911,000
(1) National guard operations				
(a) General program operations	GPR	A	6,650,100	6,650,100
(b) Repair and maintenance	GPR	A	839,900	839,900
(c) Public emergencies	GPR	S	40,000	40,000
(d) Principal repayment and interest	GPR	S	6,856,300	6,448,200
(e) State flags	GPR	A	400	400
(f) Energy costs; energy–related				
assessments	GPR	A	2,006,000	2,031,200
(g) Military property	PR	A	1,025,300	1,025,300
(h) Intergovernmental services	PR	A	-0-	-0-
(i) Distance learning centers	PR	C	-0-	-0-
(km) Agency services	PR-S	A	60,800	60,800
(Li) Gifts and grants	PR	C	135,000	135,000
(m) Federal aid	PR–F	C	32,165,000	32,168,100
(pz) Indirect cost reimbursements	PR-F	C	912,700	912,700
GENERAL BURDOGE BEVENUE	(I) PRC	JGRAM	TOTALS	16,000,000
GENERAL PURPOSE REVENUE			16,392,700	16,009,800
PROGRAM REVENUE			34,298,800	34,301,900
FEDERAL OTHER			(33,077,700) (1,160,300)	(33,080,800) (1,160,300)
SERVICE			(60,800)	(60,800)
TOTAL-ALL SOURCES			50,691,500	50,311,700
(2) GUARD MEMBERS' BENEFITS			30,071,300	30,311,700
(a) Tuition grants	GPR	S	5,500,000	5,500,000
(r) Military family relief	SEG	C	-0-	-0-
(1) 1.111111111 101101			TOTALS	Ů
GENERAL PURPOSE REVENUE	() -		5,500,000	5,500,000
SEGREGATED REVENUE			-0-	-0-
OTHER			(-0-)	(-0-)
TOTAL-ALL SOURCES			5,500,000	5,500,000
(3) EMERGENCY MANAGEMENT SERVICES				
(a) General program operations	GPR	A	1,049,200	1,049,200
(am) Worker's compensation for local un				
of government volunteers	GPR	S	27,600	27,600

2017 As	sembly Bill 64	- 97 -	-	2017 V	Visconsin Act 59
STATUT	E, AGENCY AND PURPOSE	Source	Түре	2017-2018	2018-2019
(b)	State disaster assistance	GPR	A	-0-	-0-
(dd)	Regional emergency response teams	GPR	A	1,247,400	1,247,400
(df)	Regional emergency response grants	GPR	C	500,000	-0-
(dm)	Mobile field force grants	GPR	C	500,000	-0-
(dp)	Emergency response equipment	GPR	A	417,000	417,000
(dr)	Emergency response supplement	GPR	C	-0-	-0-
(dt)	Emergency response training	GPR	В	57,900	57,900
(e)	Disaster recovery aid; public health				
	emergency quarantine costs	GPR	S	2,500,000	2,500,000
(f)	Civil air patrol aids	GPR	A	16,900	16,900
(g)	Program services	PR	C	3,062,700	2,749,300
(h)	Interstate emergency assistance	PR	A	-0-	-0-
(i)	Emergency planning and reporting; administration	PR	A	1,208,500	1,209,400
(j)	Division of emergency management;				
(jm)	gifts and grants Division of emergency management;	PR	C	-0-	-0-
	emergency planning grants Regional emergency response	PR	C	1,043,800	1,043,800
(jt)	reimbursement	PR	C	-0-	-0-
(ke)	Interagency and intra–agency assistance	PR-S	C	-0-	-0-
(km)	Interoperable communications system	PR-S	A	1,166,300	1,166,400
(ks)	Public safety interoperable communication system; state fees	PR-S	A	-0-	-0-
(L)	Public safety interoperable communication system; general				
	usage fees	PR	A	-0-	-0-
(m)	Federal aid, state operations	PR-F	C	4,502,700	4,504,200
(mb)	Federal aid, homeland security	PR-F	C	16,973,900	16,973,900
(n)	Federal aid, local assistance	PR-F	C	12,800,000	12,800,000
(o)	Federal aid, individuals and			,,	,,
. ,	organizations	PR-F	C	1,926,400	1,926,400
(q)	Interoperability council	SEG	A	199,400	212,600
(qm)	Next Generation 911	SEG	A	-0-	6,700,000
(r)	Division of emergency management; petroleum inspection fund	SEG	A	462,100	462,100
(s)	State disaster assistance; petroleum inspection fund	SEG	С	711,200	711,200
(t)	Emergency response training – environmental fund	SEG	В	7,600	7,600
		(3) PROC		TOTALS	
	GENERAL PURPOSE REVENUE			6,316,000	5,316,000
]	PROGRAM REVENUE			42,684,300	42,373,400
	FEDERAL			(36,203,000)	(36,204,500)
	OTHER			(5,315,000)	(5,002,500)
	SERVICE			(1,166,300)	(1,166,400)
\$	SEGREGATED REVENUE			1,380,300	8,093,500
	OTHER			(1,380,300)	(8,093,500)
	TOTAL-ALL SOURCES			50,380,600	55,782,900
(4)	NATIONAL GUARD YOUTH PROGRAMS				

2017 W	isconsin Act 59	- 98	3 –	2017 A	Assembly Bill 64
STATUT	TE, AGENCY AND PURPOSE	Sourc	е Түре	2017-2018	2018-2019
(h)	Gifts and grants	PR	C	-0-	-0-
(ka)	Challenge academy program; public	e			
	instruction funds	PR-S	C	1,141,300	1,141,300
(m)	Federal aid	PR-F	C	3,423,800	3,423,800
		(4) PR	OGRAM TO	ΓALS	
	PROGRAM REVENUE			4,565,100	4,565,100
	FEDERAL			(3,423,800)	(3,423,800)
	OTHER			(-0-)	(-0-)
	SERVICE			(1,141,300)	(1,141,300)
	TOTAL-ALL SOURCES			4,565,100	4,565,100
		20.465 DE	PARTMENT T	TOTALS	
	GENERAL PURPOSE REVENUE			28,208,700	26,825,800
	PROGRAM REVENUE			81,548,200	81,240,400
	FEDERAL			(72,704,500)	(72,709,100)
	OTHER			(6,475,300)	(6,162,800)
	SERVICE			(2,368,400)	(2,368,500)
	SEGREGATED REVENUE			1,380,300	8,093,500
	OTHER			(1,380,300)	(8,093,500)
	TOTAL-ALL SOURCES			111,137,200	116,159,700
20.475	District Attorneys				
(1)	DISTRICT ATTORNEYS				
(d)	Salaries and fringe benefits	GPR	A	43,113,800	43,154,900
(em)	Salary adjustments	GPR	A	1,562,600	3,162,900
(h)	Gifts and grants	PR	C	2,880,400	2,739,100
(i)	Other employees	PR	A	305,000	305,000
(k)	Interagency and intra-agency				
` '	assistance	PR-S	C	-0-	-0-
(km)	Deoxyribonucleic acid evidence				
	activities	PR-S	A	90,400	90,400
(m)	Federal aid	PR-F	C	-0-	-0-
		(1) PR	OGRAM TO	ΓALS	
	GENERAL PURPOSE REVENUE			44,676,400	46,317,800
	PROGRAM REVENUE			3,275,800	3,134,500
	FEDERAL			(-0-)	(-0-)
	OTHER			(3,185,400)	(3,044,100)
	SERVICE			(90,400)	(90,400)
	TOTAL-ALL SOURCES			47,952,200	49,452,300
		20.475 DE	PARTMENT T	TOTALS	
	GENERAL PURPOSE REVENUE			44,676,400	46,317,800
	PROGRAM REVENUE			3,275,800	3,134,500
	FEDERAL			(-0-)	(-0-)
	OTHER			(3,185,400)	(3,044,100)
	SERVICE			(90,400)	(90,400)
	TOTAL-ALL SOURCES			47,952,200	49,452,300
	Veterans Affairs, Department of				
(1)	VETERANS HOMES				
(a)	Aids to indigent veterans	GPR	A	178,200	178,200
(d)	Cemetery maintenance and				
	beautification	GPR	A	23,200	23,200
(e)	Lease rental payments	GPR	S	-0-	-0-
(f)	Principal repayment and interest	GPR	S	1,546,200	1,462,400

17 As	ssembly Bill 64	- 99 -	-	2017 V	Visconsin Act
STATUT	E, AGENCY AND PURPOSE	Source	Түре	2017-2018	2018-2019
(g)	Home exchange	PR	A	266,100	266,100
(gd)	Veterans home cemetery operations	PR	C	5,000	5,000
(gf)	Veterans home member care	PR	C	-0-	-0-
(gk)	Institutional operations	PR	A	107,637,900	107,642,000
(go)	Self–amortizing facilities; principal				
(0)	repayment and interest	PR	S	2,038,500	2,345,000
(h)	Gifts and bequests	PR	C	239,600	239,600
(i)	State-owned housing maintenance	PR	C	59,700	59,700
(kc)	Electric energy derived from				
	renewable resources	PR-S	A	54,000	54,000
(kg)	Grants to counties	PR-S	A	76,200	76,200
(kj)	Grants to local governments	PR-S	В	150,000	150,000
(m)	Federal aid; care at veterans homes	PR-F	C	-0-	-0-
(mn)	Federal projects	PR-F	C	21,700	21,700
(t)	Veterans homes member accounts	SEG	C	-0-	-0-
		(1) PROC	GRAM TO	TALS	
(GENERAL PURPOSE REVENUE			1,747,600	1,663,800
]	PROGRAM REVENUE			110,548,700	110,859,300
	FEDERAL			(21,700)	(21,700
	OTHER			(110,246,800)	(110,557,400
	SERVICE			(280,200)	(280,200
	SEGREGATED REVENUE			-0-	-0-
	OTHER			(-0-)	(-0-
,	TOTAL-ALL SOURCES			112,296,300	112,523,100
2)	Loans and aids to veterans				
(db)	General fund supplement to veterans				
	trust fund	GPR	A	-0-	-0-
(g)	Consumer reporting agency fees	PR	C	-0-	-0-
(h)	Public and private receipts	PR	C	18,200	18,200
(kg)	American Indian services				
	coordinator	PR-S	A	87,800	87,800
(km)	American Indian grants	PR-S	A	61,200	61,200
(m)	Federal payments; veterans				
	assistance	PR-F	C	373,300	373,300
(qm)	Veterans employment and				
	entrepreneurship grants	SEG	A	500,000	500,000
(rm)	Veterans assistance programs	SEG	В	830,700	830,70
(rn)	Fish and game vouchers	SEG	В	15,000	15,000
(rp)	Veterans assistance program receipts	SEG	C	115,500	115,500
(s)	Transportation payment	SEG	A	300,000	300,000
(sm)	Military funeral honors	SEG	S	304,500	304,500
(tf)	Veterans tuition reimbursement				
	program	SEG	В	1,153,100	1,153,100
(th)	Grants to nonprofit organizations	SEG	В	250,000	250,000
(tj)	Retraining assistance program	SEG	A	210,000	210,000
(tm)	Facilities	SEG	C	-0-	-0-
(u)	Administration of loans and aids to				
	veterans	SEG	A	8,775,100	8,782,400
(vm)	Assistance to needy veterans	SEG	A	970,000	970,000
(1111)	Grants to Camp American Legion			75,000	75,000

2017 Wi	sconsin Act 59	- 100 -	_	201	7 Assembly Bill 64
STATUT	E, AGENCY AND PURPOSE	Source	Түре	2017-2018	2018-2019
(vu)	Grants to American Indian tribes and bands	SEG	A	48,800	48,800
(vw)	Payments to veterans organizations for claims service	SEG	A	348,000	348,000
(vx)	County grants	SEG	A	342,400	342,400
(\mathbf{x})	Federal per diem payments	SEG-F	C	1,343,600	1,343,600
(yn)	Veterans trust fund loans and	SLO-I	C	1,545,000	1,545,000
(y11)	expenses	SEG	В	50,000	50,000
(yo)	Debt payment	SEG	S	-0-	-0-
(z)	Gifts	SEG	C	-0-	-0-
(2)		(2) PROC			v
	GENERAL PURPOSE REVENUE			-0-	-0-
	PROGRAM REVENUE			540,500	540,500
	FEDERAL			(373,300)	(373,300)
	OTHER			(18,200)	(18,200)
	SERVICE			(149,000)	(149,000)
,	SEGREGATED REVENUE			15,631,700	15,639,000
	FEDERAL			(1,343,600)	(1,343,600)
	OTHER			(14,288,100)	(14,295,400)
,	TOTAL-ALL SOURCES			16,172,200	16,179,500
(3)	SELF-AMORTIZING MORTGAGE LOANS FOR	NETERANS		-, -,	-,,
(b)	Self insurance	GPR	S	-0-	-0-
(e)	General program deficiency	GPR	S	-0-	-0-
(q)	Foreclosure loss payments	SEG	C	801,000	801,000
(r)	Funded reserves	SEG	C	50,000	50,000
(rm)	Other reserves	SEG	C	-0-	-0-
(s)	General program operations	SEG	A	253,000	253,000
(sm)	County grants	SEG	A	342,400	342,400
(t)	Debt service	SEG	C	6,614,500	5,249,300
(wd)	Loan-servicing administration	SEG	A	-0-	-0-
(wg)	Escrow payments, recoveries, and				
(6)	refunds	SEG	C	-0-	-0-
(wp)	Loan-servicing rights	SEG	В	-0-	-0-
· 1/		(3) PROC	GRAM	TOTALS	
(GENERAL PURPOSE REVENUE			-0-	-0-
,	SEGREGATED REVENUE			8,060,900	6,695,700
	OTHER			(8,060,900)	(6,695,700)
,	TOTAL-ALL SOURCES			8,060,900	6,695,700
(4)	VETERANS MEMORIAL CEMETERIES				
(g)	Cemetery operations	PR	A	271,400	271,400
(h)	Gifts, grants and bequests	PR	C	-0-	-0-
(m)	Federal aid; cemetery operations and burials	PR–F	C	1,057,600	1,057,600
(q)	Cemetery administration and		-	-,,	-,,
\1/	maintenance	SEG	A	542,600	542,600
(qm)	Repayment of principal and interest	SEG	S	78,300	22,700
(r)	Cemetery energy costs;				
. /	energy-related assessments	SEG	A	106,300	106,300
		(4) PROC	GRAM	TOTALS	
]	PROGRAM REVENUE			1,329,000	1,329,000
	FEDERAL			(1,057,600)	(1,057,600)

2017 A	ssembly Bill 64	- 101 ·	_	2017 W	isconsin Act 59
STATU'	TE, AGENCY AND PURPOSE	Source	Түре	2017–2018	2018–2019
	OTHER			(271,400)	(271,400)
	SEGREGATED REVENUE			727,200	671,600
	OTHER			(727,200)	(671,600)
(5)	TOTAL-ALL SOURCES			2,056,200	2,000,600
(5)	WISCONSIN VETERANS MUSEUM				
(c)	Operation of Wisconsin Veterans Museum	GPR	A	248,500	248,500
(mn)		UFK	Α	246,300	240,300
(mn)	acquisitions and operations	PR-F	C	-0-	-0-
(tm)	Museum facilities	SEG	C	52,800	52,800
(v)	Museum sales receipts	SEG	C	170,700	170,700
(v) (vo)	Veterans of World War I	SEG	A	2,500	2,500
(wd)		SEC	Λ	2,300	2,500
(wu)	Museum	SEG	A	2,976,300	3,321,700
(zm)		SEG	C	-0-	-0-
(ZIII)	Museum girts and bequests		GRAM TO		· ·
	GENERAL PURPOSE REVENUE	(3) 1100	SIGNIT TO	248,500	248,500
	PROGRAM REVENUE			-0-	-0-
	FEDERAL			(-0-)	(-0-)
	SEGREGATED REVENUE			3,202,300	3,547,700
	OTHER			(3,202,300)	(3,547,700)
	TOTAL-ALL SOURCES			3,450,800	3,796,200
(6)	ADMINISTRATION			3,430,000	3,770,200
(b) (k)	Funds received from other state				
(K)	agencies	PR-S	C	-0-	-0-
	ageneres		GRAM TO		v
	PROGRAM REVENUE	(0)		-0-	-0-
	SERVICE			(-0-)	(-0-)
	TOTAL-ALL SOURCES			-0-	-0-
		20.485 DEPA	ARTMENT	TOTALS	
	GENERAL PURPOSE REVENUE			1,996,100	1,912,300
	PROGRAM REVENUE			112,418,200	112,728,800
	FEDERAL			(1,452,600)	(1,452,600)
	OTHER			(110,536,400)	(110,847,000)
	SERVICE			(429,200)	(429,200)
	SEGREGATED REVENUE			27,622,100	26,554,000
	FEDERAL			(1,343,600)	(1,343,600)
	OTHER			(26,278,500)	(25,210,400)
	TOTAL-ALL SOURCES			142,036,400	141,195,100
20.490	Wisconsin Housing and Economic	Development	Authority		
(1)	FACILITATION OF CONSTRUCTION	-	·		
(a)	Capital reserve fund deficiency	GPR	C	-0-	-0-
	-	(1) PRO0	GRAM TO	TALS	
	GENERAL PURPOSE REVENUE			-0-	-0-
	TOTAL-ALL SOURCES			-0-	-0-
(2)	HOUSING REHABILITATION LOAN PROC	GRAM			
(a)	General program operations	GPR	C	-0-	-0-
(q)	Loan loss reserve fund	SEG	C	-0-	-0-
_		(2) PRO0	GRAM TO	TALS	
	GENERAL PURPOSE REVENUE			-0-	-0-
	SEGREGATED REVENUE			-0-	-0-

2017 W	Visconsin Act 59	- 102	_	2017	Assembly Bill 64
STATU	TE, AGENCY AND PURPOSE	SOURCE	Түре	2017-2018	2018-2019
	OTHER			(-0-)	(-0-)
	TOTAL-ALL SOURCES			-0-	-0-
(3)	HOMEOWNERSHIP MORTGAGE ASSISTANCE	Е			
(a)	Homeowner eviction lien protection	CDD	a	0	0
	program	GPR	CDAM TOT	-O-	-0-
	GENERAL PURPOSE REVENUE	(3) PRO	GRAM TO	-0-	-0-
	TOTAL-ALL SOURCES			-0- -0-	-0- -0-
(4)	DISADVANTAGED BUSINESS MOBILIZATION	N ASSISTANO	~E	_0_	_0_
(g)	Disadvantaged business mobilization	V ASSISTAN	CE		
(8)	loan guarantee	PR	C	-0-	-0-
		(4) PRO	GRAM TO	ΓALS	
	PROGRAM REVENUE			-0-	-0-
	OTHER			(-0-)	(-0-)
	TOTAL-ALL SOURCES			-0-	-0-
(5)	WISCONSIN DEVELOPMENT LOAN GUARAI	NTEES			
(a)	Wisconsin development reserve fund	GPR	C	-0-	-0-
(q)	Environmental fund transfer to	ana.	a		0
	Wisconsin development reserve fund	SEG	С	-0-	-0-
(r)	Agrichemical management fund transfer to Wisconsin development				
	reserve fund	SEG	С	-0-	-0-
(s)	Petroleum inspection fund transfer to			·	-
	Wisconsin development reserve fund	SEG	A	-0-	-0-
		(5) PRO	GRAM TO	ΓALS	
	GENERAL PURPOSE REVENUE			-0-	-0-
	SEGREGATED REVENUE			-0-	-0-
	OTHER			(-0-)	(-0-)
	TOTAL-ALL SOURCES			-0-	-0-
).490 DEP.	ARTMENT 1		0
	GENERAL PURPOSE REVENUE			-0-	-0-
	PROGRAM REVENUE			-0-	-0-
	OTHER SEGREGATED REVENUE			(-0-) -0-	(-0-) -0-
	OTHER			(-0-)	(-0-)
	TOTAL-ALL SOURCES			(=0=) -0-	(- 0 -)
	TOTAL TALE SOURCES	Hum	nan Resource		· ·
	F		IAL AREA T		
	GENERAL PURPOSE REVENUE			5,603,000,600	5,772,010,000
	PROGRAM REVENUE			8,738,727,800	9,183,377,100
	FEDERAL			(6,921,696,800)	(7,282,771,100)
	OTHER			(1,462,206,600)	(1,544,484,100)
	SERVICE			(354,824,400)	(356,121,900)
	SEGREGATED REVENUE			647,135,000	649,781,700
	FEDERAL			(1,343,600)	(1,343,600)
	OTHER			(645,791,400)	(648,438,100)
	SERVICE			(-0-)	(-0-)
	LOCAL			(-0-)	(-0-)
	TOTAL-ALL SOURCES			14,988,863,400	15,605,168,800

2017 Assembly Bill 64	– 103 –	2017 Wisconsin Act 59
STATUTE, AGENCY AND PURPOSE	SOURCE TYPE	2017-2018 2018-2019

STATUT	E, AGENCY AND PURPOSE	Source	TYPE	2017-2018	2018-2019
	General I	Executiv	e Func	ctions	
20.505	Administration, Department of				
	SUPERVISION AND MANAGEMENT				
(a)	General program operations	GPR	A	6,476,100	6,382,200
(b)	Midwest interstate low-level				
	radioactive waste compact; loan from				
	general fund	GPR	C	-0-	-0-
(bq)	Appropriation obligations				
	repayment; tobacco settlement revenues	GPR	A	76,783,800	105,433,400
(br)	Appropriation obligations	OFK	А	70,785,800	103,433,400
(01)	repayment; unfunded liabilities under				
	the Wisconsin Retirement System	GPR	A	279,969,800	307,789,700
(cm)	Comprehensive planning grants;			, ,	, ,
. ,	general purpose revenue	GPR	A	-0-	-0-
(cn)	Comprehensive planning;				
	administrative support	GPR	A	-0-	-0-
(d)	Special counsel	GPR	S	611,900	611,900
(fm)	Fund of funds investment program	GPR	A	-0-	-0-
(fo)	Federal resource acquisition support	CDD		0	0
(-)	grants	GPR	A	-0-	-0-
(g)	Midwest interstate low–level radioactive waste compact;				
	membership and costs	PR	A	-0-	-0-
(gc)	Processing services	PR	A	156,300	156,400
(ge)	High-voltage transmission line				,
(8.7)	annual impact fee distributions	PR	C	-0-	-0-
(gm)	Federal resource acquisition	PR	A	248,900	248,900
(gr)	Disabled veteran-owned,				
	woman-owned, and minority				
	business certification fees	PR	C	31,500	31,500
(gs)	High-voltage transmission line				
	environmental impact fee distributions	PR	С	-0-	-0-
(ic)	Services to nonstate governmental	1 IX	C	_0_	_0_
(10)	units	PR	A	215,300	215,600
(id)	Justice information fee receipts	PR	C	-0-	-0-
(im)	Services to nonstate governmental				
` /	units; entity contract	PR	A	1,569,000	1,571,000
(ip)	Information technology and				
	communication services; self-funded				
	portal	PR	A	7,859,100	8,034,000
(is)	Information technology and				
	communications services; nonstate entities	PR	A	12,707,000	12,707,200
(it)	Appropriation obligations;	1 K	Λ	12,707,000	12,707,200
(11)	agreements and ancillary				
	arrangements	PR	C	-0-	-0-
(iu)	Plat and proposed incorporation and				
•	annexation review	PR	C	324,900	325,400
(iv)	Enterprise resource planning system;		_		
	nonstate entities	PR	C	-0-	-0-

2017 Wi	sconsin Act 59	- 104 ·	_	2017 A	ssembly Bill 64	
STATUT	E, AGENCY AND PURPOSE	Source	Түре	2017-2018	2018-2019	
(j)	Gifts, grants, and bequests	PR	C	300	500	
(jc)	Employee development and training services	PR	A	233,900	234,200	
(ka)	Materials and services to state agencies and certain districts	PR-S	A	6,009,000	6,247,300	
(kb)	Transportation, records, and document services	PR-S	A		19,172,200	
(kc)	Capital planning and building			19,099,800		
4.1	construction services	PR-S	A	12,496,100	12,520,200	
(kd)	Enterprise resource planning system	PR-S	C	13,608,900	11,013,700	
(kf)	Procurement services	PR-S	C	4,530,300	4,540,600	
(kg)	Federal resource acquisition	PR-S	C	-0-	-0-	
(kh)	Justice information systems	PR-S	A	4,209,900	4,223,300	
(ki)	Postage costs	PR-S	C	15,710,100	15,710,100	
(kj)	Financial services	PR-S	A	9,083,900	9,096,100	
(kL)	Printing, mail, communication, and information technology services;					
(km)	state agencies; veterans services University of Wisconsin–Green Bay	PR-S	A	97,690,900	97,802,200	Vetoed
	programming	PR-S	A	247,500	247,500	In Part
(kn)	Publications	PR	A	102,100	102,100	
(kq)	Justice information systems					
	development, operation and					
	maintenance	PR-S	A	-0-	-0-	
(kr)	Legal services; relocation assistance	PR-S	A	1,095,100	1,098,800	
(ks)	Collective bargaining grievance					
	arbitrations	PR-S	A	30,000	30,000	
(ku)	Management assistance grants to					
	counties	PR-S	A	563,200	563,200	
(kx)	American Indian economic					
	development; technical assistance	PR-S	A	79,500	79,500	
(kz)	General program operations	PR-S	A	6,147,600	38,675,000	
(mb)	Federal aid	PR-F	C	6,650,800	6,653,100	
(n)	Federal aid; local assistance	PR-F	C	90,000,000	90,000,000	
(ng)	Sale of forest products; funds for			, ,	, ,	
ν υ,	public schools and public roads	PR	C	-0-	-0-	
(pz)	Indirect cost reimbursements	PR-F	C	4,300	4,800	
(s)	Diesel truck idling reduction grant					
· /	administration	SEG	A	77,000	77,100	
(sa)	Diesel truck idling reduction grants	SEG	A	1,000,000	1,000,000	
(ub)	Land information program, state operations; reviews of municipal incorporations and annexations;					
	planning grants	SEG	A	761,600	761,900	
(uc) (ud)	Land information program; local aids Comprehensive planning grants; land	SEG	С	6,945,300	6,945,300	
(v)	information fund General program operations —	SEG	A	-0-	-0-	
()	environmental improvement					
	programs; state funds	SEG	A	807,900	808,500	
(x)	General program operations — clean water fund program; federal funds	SEG-F	С	-0-	-0-	
	1 0 /					

017 A	ssembly Bill 64	- 105	_	2017 V	Visconsin Act
STATU	TE, AGENCY AND PURPOSE	Source	Түре	2017-2018	2018-2019
(y)	General program operations — safe drinking water loan program; federal funds	SEG-F	С	-0-	-0-
(z)	Transportation planning grants to	SLO 1	C	O	U
(-)	local governmental units	SEG-S (1) PROO	B GRAM TO	-0- TALS	-0-
	GENERAL PURPOSE REVENUE	· /		363,841,600	420,217,200
	PROGRAM REVENUE			310,705,200	341,304,400
	FEDERAL			(96,655,100)	(96,657,900)
	OTHER			(23,448,300)	(23,626,800)
	SERVICE			(190,601,800)	(221,019,700)
	SEGREGATED REVENUE			9,591,800	9,592,800
	FEDERAL			(-0-)	(-0-)
	OTHER			(9,591,800)	(9,592,800)
	SERVICE			(-0-)	(-0-)
	TOTAL-ALL SOURCES			684,138,600	771,114,400
(2)	RISK MANAGEMENT				
(a)	General fund supplement — risk				
	management claims	GPR	S	-0-	-0-
(am)	<i>y</i>	GPR	S	-0-	-0-
(k)	Risk management costs	PR-S	C	35,406,200	35,406,200
(ki)	Risk management administration	PR-S	A	10,671,600	10,677,400
		(2) PRO	GRAM TO		
	GENERAL PURPOSE REVENUE			-0-	-0-
	PROGRAM REVENUE			46,077,800	46,083,600
	SERVICE			(46,077,800)	(46,083,600
	TOTAL-ALL SOURCES			46,077,800	46,083,600
(3)	UTILITY PUBLIC BENEFITS AND AIR QUAL	ITY IMPROV	EMENT		
(q)	General program operations; utility	CEC		11 412 000	11 414 404
()	public benefits	SEG	A	11,413,900	11,414,400
(r)	Low-income assistance grants	SEG	S	19,447,300	19,447,300
(rr)	Air quality improvement grants	SEG	S	-0-	-0-
(s)	Transfer to air quality improvement fund	SEG	S	-0-	-0-
	Tuna		GRAM TO	_	-0-
	SEGREGATED REVENUE	(3) TRO	JIVANI 10	30,861,200	30,861,700
	OTHER			(30,861,200)	(30,861,700
	TOTAL-ALL SOURCES			30,861,200	30,861,700
(4)	ATTACHED DIVISIONS AND OTHER BODIES			30,001,200	30,001,700
(a)	Adjudication of tax appeals	GPR	A	569,800	571,400
(b)	Adjudication of equalization appeals	GPR	S	-0-	-0-
(d)	Claims awards	GPR	S	25,000	25,000
(ea)	Women's council operations	GPR	A	145,200	145,400
(ec)	Service award program; general	0111		1.0,200	1.0,100
()	program operations	GPR	A	17,200	17,200
(er)	Service award program; state awards	GPR	S	2,400,000	2,500,000
(es)	Principal, interest, and rebates;				
	general purpose revenue — schools	GPR	S	911,400	1,000,600
(et)	Principal, interest, and rebates;				
	general purpose revenue — public				
	library boards	GPR	S	8,000	6,200

2017 Wi	isconsin Act 59	- 106	_	2017 A	Assembly Bill 64
STATUT	E, AGENCY AND PURPOSE	Source	Түре	2017-2018	2018-2019
(h)	Program services	PR	A	27,200	27,200
(ha)	Principal, interest, and rebates; program revenue — schools	PR	С	-0-	-0-
(hb)	Principal, interest, and rebates; program revenue — public library boards	PR	C	-0-	-0-
(j)	National and community service				_
(js)	board; gifts and grants Educational technology block grants; Wisconsin Advanced Telecommunications Foundation assessments	PR PR	C C	-0- -0-	-0-
(k)	Waste facility siting board; general program operations	PR-S	A	45,500	45,500
(ka)	State use board — general program				
(kb)	operations National and community service	PR-S	A	134,800	141,700
	board; administrative support	PR-S	A	317,700	319,900
(kp)	Hearings and appeals fees	PR-S	A	10,847,200	10,660,500
(L)	Equipment purchases and leases	PR	C	-0-	-0-
(Lm)	Educational telecommunications; additional services	PR	C	-0-	-0-
(mp)	Federal e-rate aid	PR-F	C	5,576,700	5,576,900
(o)	National and community service board; federal aid for administration	PR-F	C	623,600	623,800
(p)	National and community service board; federal aid for grants	PR-F	C	3,354,300	3,354,300
(r)	State capitol and executive residence board; gifts and grants	SEG	C	-0-	-0-
(s)	Telecommunications access for educational agencies, infrastructure				
	grants, and teacher training grants	SEG	В	21,984,200	15,984,200
		(4) PROC	GRAM T		
	GENERAL PURPOSE REVENUE			4,076,600	4,265,800
	PROGRAM REVENUE			20,927,000	20,749,800
	FEDERAL			(9,554,600)	(9,555,000)
	OTHER			(27,200)	(27,200)
	SERVICE			(11,345,200)	(11,167,600)
	SEGREGATED REVENUE			21,984,200	15,984,200
	OTHER			(21,984,200)	(15,984,200)
	TOTAL-ALL SOURCES			46,987,800	40,999,800
(5)	FACILITIES MANAGEMENT				
(c)	Principal repayment and interest; Black Point Estate	GPR	S	201,900	193,400
(g)	Principal repayment, interest and rebates; parking	PR-S	S	1,178,100	1,111,100
(ka)	Facility operations and maintenance;				
(1.1.)	police and protection functions	PR-S	A	41,890,800	42,817,100
(kb)	Parking	PR	A	1,775,500	1,777,500
(kc)	Principal repayment, interest and rebates	PR-S	C	23,702,800	31,365,000

017 A	ssembly Bill 64	- 107 ·	_	2017 W	isconsin Act
STATUT	TE, AGENCY AND PURPOSE	Source	Түре	2017-2018	2018-2019
(ke)	Additional energy conservation				
` ′	construction projects	PR-S	C	-0-	-0-
(kg)	Electric energy derived from				
	renewable resources	PR-S	A	325,400	325,400
(ks)	Security services	PR-S	A	175,000	175,000
		(5) PRO0	GRAM TO		
	GENERAL PURPOSE REVENUE			201,900	193,400
	PROGRAM REVENUE			69,047,600	77,571,100
	OTHER			(1,775,500)	(1,777,500)
	SERVICE			(67,272,100)	(75,793,600)
	TOTAL-ALL SOURCES			69,249,500	77,764,500
(7)	HOUSING AND COMMUNITY DEVELOPMEN	NT			
(a)	General program operations	GPR	A	884,200	886,200
(b)	Housing grants and loans; general				
	purpose revenue	GPR	В	3,097,800	3,097,800
(c)	Payments to designated agents	GPR	A	-0-	-0-
(fm)	Shelter for homeless and housing				
	grants	GPR	В	1,413,600	1,413,600
(ft)	Employment grants	GPR	A	75,000	75,000
(gg)	Housing program services; other		~	4.50.000	
	entities	PR	C	168,900	168,900
(h)	Funding for the homeless	PR	C	422,400	422,400
(k)	Sale of materials or services	PR-S	C	-0-	-0-
(kg)	Housing program services	PR-S	C	922,400	922,400
(m)	Federal aid; state operations	PR-F	C	1,499,500	1,500,600
(n)	Federal aid; local assistance	PR-F	C	10,000,000	10,000,000
(o)	Federal aid; individuals and	DD E	a	22 164 000	22 164 006
	organizations	PR-F	C	22,164,000	22,164,000
	CENEDAL DUDDOGE DEVENIUE	(/) PROC	GRAM TO		5 4 72 600
	GENERAL PURPOSE REVENUE			5,470,600	5,472,600
	PROGRAM REVENUE			35,177,200	35,178,300
	FEDERAL			(33,663,500)	(33,664,600)
	OTHER			(591,300)	(591,300)
	SERVICE TOTAL ALL SOURCES			(922,400)	(922,400)
(0)	TOTAL-ALL SOURCES			40,647,800	40,650,900
(8)	DIVISION OF GAMING	CDD	C	100	100
(am)	Interest on racing and bingo moneys	GPR	S	100	100
(g)	General program operations; racing General program operations; Indian	PR	A	-0-	-0-
(h)	gaming	PR	A	1,974,800	1,996,300
(hm)		PR	C	-0-	1,990,300 -0-
(iiiii) (j)	General program operations; raffles	PR	A	283,600	284,000
(jm)	General program operations; rames General program operations; bingo	PR	A A	340,700	341,300
(JIII)	General program operations, omgo		GRAM TO		341,300
	GENERAL PURPOSE REVENUE	(6) FROC	JKAWI 10.	100	100
	PROGRAM REVENUE				
	OTHER			2,599,100	2,621,600 (2,621,600)
	TOTAL-ALL SOURCES			(2,599,100)	
		0 505 DEDA	DTMENT	2,599,200	2,621,700
		U.JUJ DEPA	ARTMENT T		420 140 100
	GENERAL PURPOSE REVENUE			373,590,800	430,149,100
	PROGRAM REVENUE			484,533,900	523,508,800

17 Wi	sconsin Act 59	- 108	_	2017	Assembly Bill 64	
STATUT	E, AGENCY AND PURPOSE	Source	Түре	2017-2018	2018-2019	
	FEDERAL			(139,873,200)	(139,877,500)	
	OTHER			(28,441,400)	(28,644,400)	
	SERVICE			(316,219,300)	(354,986,900)	
9	SEGREGATED REVENUE			62,437,200	56,438,700	
	FEDERAL			(-0-)	(-0-)	
	OTHER			(62,437,200)	(56,438,700)	
	SERVICE			(-0-)	(-0-)	
-	TOTAL-ALL SOURCES			920,561,900	1,010,096,600	
	Board of Commissioners of Public La	anda		920,301,900	1,010,090,000	
		anus				
(1)	TRUST LANDS AND INVESTMENTS					
(h)	Trust lands and investments —	PR-S	A	1 625 200	1 627 500	
(i)	general program operations Payments to American Indian tribes	rk-3	A	1,625,300	1,627,500	
(j)	or bands for raised sunken logs	PR	С	-0-	-0-	
(1:)	Trust lands and investments —	rĸ	C	-0-	-0-	
(k)	interagency and intra–agency					
	assistance	PR-S	A	-0-	-0-	
(mg)	Federal aid — flood control	PR-F	C	52,700	52,700	
(mg)	reactar and — flood control		GRAM TO		32,700	
1	PROGRAM REVENUE	(1) FKO	JKAWI 10	1,678,000	1,680,200	
	FEDERAL			(52,700)	(52,700)	
	OTHER			* ' '	* * *	
				(-0-)	(-0-)	
-	SERVICE			(1,625,300)	(1,627,500)	
-	ГОТAL–ALL SOURCES	0 507 DED		1,678,000	1,680,200	
-		0.507 DEPA	ARTMENT		1 (00 200	
j	PROGRAM REVENUE			1,678,000	1,680,200	
	FEDERAL			(52,700)	(52,700)	
	OTHER			(-0-)	(-0-)	
_	SERVICE			(1,625,300)	(1,627,500)	
	ГОТAL-ALL SOURCES			1,678,000	1,680,200	
	Elections Commission					
(1)	ADMINISTRATION OF ELECTIONS					
(a)	General program operations; general			1,817,300	4,733,600	Veto
	purpose revenue	GPR	В	1,807,700	4,419,900	In F
(be)	Investigations	GPR	A	25,000	25,000	
(bm)	Training of chief inspectors	GPR	В	-0-	-0-	
(br)	Special counsel	GPR	A	-0-	-0-	
(c)	Voter identification training	GPR	A	82,600	82,600	
(d)	Election administration transfer	GPR	A	-0-	-0-	
(e)	Elections administration	GPR	A	-0-	-0-	
(g)	Recount fees	PR	A	-0-	-0-	
(h)	Materials and services	PR	A	1,700	1,700	
(jm)	Gifts and grants	PR	A	-0-	-0-	
(m)	Federal aid	PR-F	A	157,700	-0-	
(t)	Election administration	SEG	A	100	100	
(x)	Federal aid; election administration	~- ~		2.674.800	100	Vet
(A)	fund	SEG-F	С	2,370,700	-0-	In F
	Tuna		GRAM TO	· · · · · · · · · · · · · · · · · · ·	_0_	_
,	GENERAL PURPOSE REVENUE	(1) 1 KO	GIVAIVI IU	1,924,900	4,841,200	
,	PROGRAM REVENUE			1,924,900	4,841,200 1,700	
1				179 400	1 /00	
]	FEDERAL			(157,700)	(-0-)	

)17 As	ssembly Bill 64	- 109 ·	_	2017 W	isconsin Act 59	
Statut	E, AGENCY AND PURPOSE	Source	Түре	2017-2018	2018-2019	
	OTHER			(1,700)	(1,700)	
;	SEGREGATED REVENUE			2,674,900	100	
	FEDERAL			(2,674,800)	(-0-)	
	OTHER			(100)	(100)	
,	TOTAL-ALL SOURCES			4,759,200	4,843,000	
		0.510 DEPA	RTMENT T		1,010,000	
	GENERAL PURPOSE REVENUE	0.010 2211		1,924,900	4,841,200	
	PROGRAM REVENUE			159,400	1,700	
•	FEDERAL			(157,700)	(-0-)	
	OTHER			(1,700)	(1,700)	
9	SEGREGATED REVENUE			2,674,900	100	
•	FEDERAL			(2,674,800)	(-0-)	
	OTHER			(100)	(100)	
,	TOTAL-ALL SOURCES			4,759,200	4,843,000	
	Employee Trust Funds, Department	of		7,732,200	7,073,000	
20.515 (1)	Employee Trust Funds, Department Employee benefit plans	O1				
` '		GPR	S	96,500	68,800	
(a)	Annuity supplements and payments Contingencies	GPR GPR	S S	90,300 -0-	-0-	
(c)		PR		_0_ _0_	-0- -0-	
(gm)	Gifts and grants		C C			
(m)	Federal aid	PR-F	C	-0-	-0-	
(sr)	Gifts and grants; public employee trust fund	SEG	C	0	-0-	
(4)			C	-0- 9 202 600		
(t)	Automated operating system	SEG	C	8,393,600	8,393,600	
(tm)	Health savings account plan	SEG	C	-0-	-0-	
(u)	Employee-funded reimbursement	CEC	C	0	0	
()	account plan	SEG	C	-0- 4 000	-0-	
(um)	Benefit administration	SEG	В	4,900	4,900	
(ut)	Health insurance data collection and	CEC		060 100	060 100	
	analysis contracts	SEG	A	968,100	968,100	
(w)	Administration	SEG	A	36,552,800	36,646,700	
		(1) PROC	GRAM TOT		60.000	
	GENERAL PURPOSE REVENUE			96,500	68,800	
]	PROGRAM REVENUE			-0-	-0-	
	FEDERAL			(-0-)	(-0-)	
	OTHER			(-0-)	(-0-)	
;	SEGREGATED REVENUE			45,919,400	46,013,300	
	OTHER			(45,919,400)	(46,013,300)	
,	TOTAL–ALL SOURCES			46,015,900	46,082,100	
	2	0.515 DEPA	RTMENT 7	ΓOTALS		
	GENERAL PURPOSE REVENUE			96,500	68,800	
]	PROGRAM REVENUE			-0-	-0-	
	FEDERAL			(-0-)	(-0-)	
	OTHER			(-0-)	(-0-)	
;	SEGREGATED REVENUE			45,919,400	46,013,300	
	OTHER			(45,919,400)	(46,013,300)	
,	TOTAL–ALL SOURCES			46,015,900	46,082,100	
20.521	Ethics Commission					
(1)	ETHICS, CAMPAIGN FINANCE AND LOBBY	NG REGULAT	TION			
(a)	General program operations; general			614,700	617,100	Vet
()	purpose revenue	GPR	A	605,100	607,500	In
	Investigations	GPR	A	225,000	225,000	

2017 W	2017 Wisconsin Act 59		_	2017 A	2017 Assembly Bill 64		
STATU	TE, AGENCY AND PURPOSE	Source	Түре	2017-2018	2018-2019		
(br)	Special counsel	GPR	A	-0-	-0-		
(g)	General program operations;						
	program revenue	PR	A	31,700	31,700		
(h)	Gifts and grants	PR	A	-0-	-0-		
(i)	Materials and services	PR	A	4,500	4,500		
(im)	Lobbying administration; program						
	revenue	PR	A	462,900	455,800		
(j)	Electronic filing software	PR	A	-0-	-0-		
		(1) PRO0	GRAM TO				
	GENERAL PURPOSE REVENUE			839,700	842,100		
	PROGRAM REVENUE			499,100	492,000		
	OTHER			(499,100)	(492,000)		
	TOTAL-ALL SOURCES			1,338,800	1,334,100		
		0.521 DEPA	ARTMENT				
	GENERAL PURPOSE REVENUE			839,700	842,100		
	PROGRAM REVENUE			499,100	492,000		
	OTHER			(499,100)	(492,000)		
	TOTAL-ALL SOURCES			1,338,800	1,334,100		
	5 Governor, Office of the						
(1)	EXECUTIVE ADMINISTRATION		_				
(a)	General program operations	GPR	S	3,301,000	3,301,000		
(b)	Contingent fund	GPR	S	20,400	20,400		
(c)	Membership in national associations	GPR	S	118,300	118,300		
(d)	Disability board	GPR	S	-0-	-0-		
(i)	Gifts and grants	PR	C	-0-	-0-		
(m)	Federal aid	PR-F	C	-0-	-0-		
	CENED AL DUDDOCE DELEDINE	(1) PROC	GRAM TO		2 420 700		
	GENERAL PURPOSE REVENUE			3,439,700	3,439,700		
	PROGRAM REVENUE			-0-	-0-		
	FEDERAL			(-0-)	(-0-)		
	OTHER TOTAL ALL SOURCES			(-0-)	(-0-)		
(2)	TOTAL-ALL SOURCES			3,439,700	3,439,700		
(2)	EXECUTIVE RESIDENCE	CDD	C	271 000	271 000		
(a)	General program operations	GPR	S SDAM TO	271,000	271,000		
	GENERAL PURPOSE REVENUE	(2) PROC	GRAM TO		271,000		
				271,000	*		
	TOTAL-ALL SOURCES	0 525 DEDA	ARTMENT	271,000	271,000		
	GENERAL PURPOSE REVENUE	0.323 DEFF	AKTIVIENT	3,710,700	3,710,700		
	PROGRAM REVENUE			-0-	-0-		
	FEDERAL			(-0-)	(-0-)		
	OTHER			(-0-)	(-0-)		
	TOTAL-ALL SOURCES			3,710,700	3,710,700		
20 536	6 Investment Board			3,710,700	3,710,700		
(1)	Investment board Investment of funds						
(k)	General program operations	PR	С	53,499,600	53,499,600		
(ka)	General program operations;			22,177,000	55,177,000		
(Hu)	environmental improvement fund	PR-S	C	-0-	-0-		
	r		GRAM TO		-		
	PROGRAM REVENUE	* *		53,499,600	53,499,600		
	OTHER			(53,499,600)	(53,499,600)		
				, , , ,	• • • • •		

2017 A	Assembly Bill 64	- 111 -	_	2017 W	Visconsin Act 59	
STATU	TE, AGENCY AND PURPOSE	Source	Түре	2017–2018	2018–2019	
	SERVICE TOTAL ALL SOURCES			(-0-)	(-0-)	
	TOTAL-ALL SOURCES	20 526 DEDA	DTMENT	53,499,600	53,499,600	
	DDOCD AM DEVENUE	20.536 DEPA	KIMEN		52 400 C00	
	PROGRAM REVENUE			53,499,600	53,499,600	
	OTHER			(53,499,600)	(53,499,600)	
	SERVICE			(-0-)	(-0-)	
20.54	TOTAL-ALL SOURCES			53,499,600	53,499,600	
	0 Lieutenant Governor, Office of the	2				
(1)	EXECUTIVE COORDINATION	CDD		202 100	202 100	
(a)	General program operations	GPR	A	382,100	382,100	
(g)	Gifts, grants and proceeds	PR	C	-0-	-0-	
(k)	Grants from state agencies	PR-S	C	-0-	-0-	
(m)	Federal aid	PR-F	С	-0-	-0-	
		(1) PRO0	GRAM T			
	GENERAL PURPOSE REVENUE			382,100	382,100	
	PROGRAM REVENUE			-0-	-0-	
	FEDERAL			(-0-)	(-0-)	
	OTHER			(-0-)	(-0-)	
	SERVICE			(-0-)	(-0-)	
	TOTAL-ALL SOURCES			382,100	382,100	
		20.540 DEPA	RTMEN	ΓTOTALS		
	GENERAL PURPOSE REVENUE			382,100	382,100	
	PROGRAM REVENUE			-0-	-0-	
	FEDERAL			(-0-)	(-0-)	
	OTHER			(-0-)	(-0-)	
	SERVICE			(-0-)	(-0-)	
	TOTAL-ALL SOURCES			382,100	382,100	
20.548	8 Prosecutor Board					Vetoed
(1)	COORDINATION AND ADMINISTRATION	OF PROSECUTO	R FUNCTIO	ONS		In Part
(a)	Program administration	GPR	A	93,800	225,000	
(g)	Gifts, grants, and proceeds	PR	C	-0-	-0-	
		(1) PROC	GRAM T	OTALS		
	GENERAL PURPOSE REVENUE			93,800	225,000	
	PROGRAM REVENUE			-0-	-0-	
	OTHER			(-0-)	(-0-)	
	TOTAL-ALL SOURCES			93,800	225,000	
		20.548 DEPA	RTMEN	ΓTOTALS		
	GENERAL PURPOSE REVENUE			93,800	225,000	
	PROGRAM REVENUE			-0-	-0-	
	OTHER			(-0-)	(-0-)	
	TOTAL-ALL SOURCES			93,800	225,000	
20.550	O Public Defender Board					
(1)	LEGAL ASSISTANCE					
(a)	Program operation	GPR	В	85,931,300	87,323,100	
(fb)	Payments from clients;			/ /	,,	
()	administrative costs	PR	A	295,300	295,600	
(g)	Gifts, grants and proceeds	PR	C	-0-	-0-	
(h)	Contractual agreements	PR-S	A	-0-	-0-	
(i)	Tuition payments	PR	C	-0-	-0-	
(kj)	Conferences and training	PR-S	A	169,900	170,300	
\ J/	C			,	,	

Company Comp	2017 W	2017 Wisconsin Act 59		_	2017 A	2017 Assembly Bill 64		
(L) Private bar and investigator reimbursement; payments for legal representation representation (m) Federal and PR-C ————————————————————————————————————	STATU	TE. AGENCY AND PURPOSE	Source	Туре	2017-2018	2018-2019		
reimbursement; payments for legal representation		· ·	DOUNCE		_01010	2010 2015		
representation PR C 913,000 913,000 (m) Federal aid PR-F C -0-0-0-0 (1) PROGRAM TOTALS (1) PROGRAM TOTALS (1) PROGRAM TOTALS (1,208,300) 1,378,900 1,378,900 (1,208,600) (1,20	(—)							
Comparison Com			PR	C	913,000	913,000		
GENERAL PURPOSE REVENUE	(m)	Federal aid		-		-0-		
PROGRAM REVENUE			(1) PRO0	GRAM TO	OTALS			
FEDERAL		GENERAL PURPOSE REVENUE			85,931,300			
OTHER		PROGRAM REVENUE			1,378,200	1,378,900		
SERVICE					, ,			
TOTAL-ALL SOURCES								
Q20.550 DEPARTMENT TOTALS								
GENERAL PURPOSE REVENUE						88,702,000		
PROGRAM REVENUE			0.550 DEPA	RTMENT				
FEDERAL								
OTHER								
SERVICE								
TOTAL-ALL SOURCES 87,309,500 88,702,000								
20.566 Revenue, Department of (1) COLLECTION OF TAXES								
(1) COLLECTION OF TAXES (a) General program operations GPR A 63,737,600 63,873,000 (g) Administration of county sales and use taxes PR A 3,071,400 3,079,400 (ga) Cigarette tax stamps PR A 249,300 249,300 (gb) Business tax registration PR A 1,667,400 1,668,800 (gd) Administration of special district taxes PR-S A 443,600 444,100 (ge) Administration of local professional football stadium district taxes PR-S A 128,700 128,700 (gf) Administration of local taxes PR-S A 83,900 83,900 (gg) Administration of local taxes PR A 134,700 134,700 (h) Debt collection PR A 2,702,000 2,851,100 (ha) Administration of liquor tax and alcohol beverages enforcement PR A 1,130,500 1,130,500 (hb) Collections by the department PR A 1,135,400 1,155,400 (hc) Collections from the financial record matching program PR A 506,600 510,100 (hd) Administration of liquor tax and alcohol beverages enforcement; wholesaler fees funding special agent position PR S 357,300 357,300 (hm) Collections under contracts PR S 357,300 357,300 (hm) Collections under the multistate tax commission audit program PR S 58,300 58,300 (ho) Collections under multistate streamlined sales tax project PR S 40,000 40,000 (hp) Administration of income tax checkoff voluntary payments PR C -00- (m) Federal funds; state operations PR-F C -00- (m) Federal funds; state operations	20.566				87,309,500	88,702,000		
(a) General program operations (GPR A 63,737,600 63,873,000 (g) Administration of county sales and use taxes PR A 3,071,400 3,079,400 (ga) Cigarette tax stamps PR A 249,300 249,300 (gb) Business tax registration PR A 1,667,400 1,668,800 (gd) Administration of special district taxes PR-S A 443,600 444,100 (ge) Administration of local professional football stadium district taxes PR-S A 128,700 128,700 (gf) Administration of local taxes PR-S A 83,900 83,900 (gg) Administration of local taxes PR A 134,700 134,700 (h) Debt collection PR A 2,702,000 2,851,100 (ha) Administration of liquor tax and alcohol beverages enforcement PR A 1,130,500 1,130,500 (hb) Collections by the department PR A 1,135,400 1,155,400 (hd) Administration of liquor tax and alcohol beverages enforcement; wholesaler fees funding special agent position PR S 357,300 357,300 (hm) Collections under the multistate tax commission audit program PR S 58,300 58,300 (hm) Collections under the multistate extreamlined sales tax project PR S 40,000 40,000 (hp) Administration of income tax checkoff voluntary payments PR C -00- (q) Economic development surcharge		, <u>*</u>						
(g) Administration of county sales and use taxes PR A 3,071,400 3,079,400 (ga) Cigarette tax stamps PR A 249,300 249,300 (gb) Business tax registration PR A 1,667,400 1,668,800 (gd) Administration of special district taxes PR—S A 443,600 444,100 (ge) Administration of local professional football stadium district taxes PR—S A 128,700 128,700 (gf) Administration of resort tax PR—S A 83,900 83,900 (gg) Administration of local taxes PR A 134,700 134,700 (h) Debt collection PR A 2,702,000 2,851,100 (ha) Administration of liquor tax and alcohol beverages enforcement PR A 1,130,500 1,130,500 (hb) Collections by the department PR A 1,135,400 1,155,400 (hd) Administration of liquor tax and alcohol beverages enforcement; wholesaler fees funding special agent position PR S 357,300 357,300 (hm) Collections under the multistate tax commission audit program PR S 58,300 58,300 (hm) Collections under the multistate tax commission audit program PR S 40,000 40,000 (hp) Administration of income tax checkoff voluntary payments PR C -00- (q) Economic development surcharge	* *		CDD	٨	62 727 600	62 972 000		
use taxes PR A 3,071,400 3,079,400 (ga) Cigarette tax stamps PR A 249,300 249,300 (gb) Business tax registration PR A 1,667,400 1,668,800 (gd) Administration of special district taxes PR A 443,600 444,100 (ge) Administration of local professional football stadium district taxes PR A 128,700 128,700 (gf) Administration of resort tax PR A 83,900 83,900 (gg) Administration of local taxes PR A 134,700 134,700 (gg) Administration of local taxes PR A 2,702,000 2,851,100 (h) Debt collection PR A 1,130,500 1,130,500 (h) Collections by the department PR A 1,135,400 1,155,400 (h) Collections from the financial record matching program PR A 506,600 510,100 (h) Administration of liquor tax and al			GPK	А	05,757,000	03,873,000		
(ga) Cigarette tax stamps PR A 249,300 249,300 (gb) Business tax registration PR A 1,667,400 1,668,800 (gd) Administration of special district taxes PR-S A 443,600 444,100 (ge) Administration of local professional football stadium district taxes PR-S A 128,700 128,700 (gf) Administration of resort tax PR-S A 83,900 83,900 (gg) Administration of local taxes PR A 134,700 134,700 (h) Debt collection PR A 2,702,000 2,851,100 (ha) Administration of liquor tax and alcohol beverages enforcement PR A 1,130,500 1,130,500 (hb) Collections from the financial record matching program PR A 506,600 510,100 (hd) Administration of liquor tax and alcohol beverages enforcement; wholesaler fees funding special agent position PR S 357,300 357,300 (hm) Collections under the multistate tax commission audit p	(g)	•	DD	٨	3 071 400	3 070 400		
(gb) Business tax registration PR A 1,667,400 1,668,800 (gd) Administration of special district taxes PR—S A 443,600 444,100 (ge) Administration of local professional football stadium district taxes PR—S A 128,700 128,700 (gf) Administration of resort tax PR—S A 83,900 83,900 (gg) Administration of local taxes PR A 134,700 134,700 (h) Debt collection PR A 2,702,000 2,851,100 (ha) Administration of liquor tax and alcohol beverages enforcement PR A 1,130,500 1,130,500 (hb) Collections by the department PR A 1,135,400 1,155,400 (hc) Collections from the financial record matching program PR A 506,600 510,100 (hd) Administration of liquor tax and alcohol beverages enforcement; wholesaler fees funding special agent position PR S 357,300 357,300 (hm) Collections under contracts PR S 357,300 357,300 (hn) Collections under the multistate tax commission audit program PR S 58,300 58,300 (ho) Collections under multistate streamlined sales tax project PR S 40,000 40,000 (hp) Administration of income tax checkoff voluntary payments PR C —0— —0— (i) Gifts and grants PR C —0— —0— (ii) Economic development surcharge	(ga)							
(gd) Administration of special district taxes PR-S A 443,600 444,100 (ge) Administration of local professional football stadium district taxes PR-S A 128,700 128,700 (gf) Administration of resort tax PR-S A 83,900 83,900 (gg) Administration of local taxes PR A 134,700 134,700 (h) Debt collection PR A 2,702,000 2,851,100 (ha) Administration of liquor tax and alcohol beverages enforcement PR A 1,130,500 1,130,500 (hb) Collections by the department PR A 1,135,400 1,155,400 (hc) Collections from the financial record matching program PR A 506,600 510,100 (hd) Administration of liquor tax and alcohol beverages enforcement; wholesaler fees funding special agent position PR C 109,800 109,800 (hm) Collections under the multistate tax commission audit program PR S 357,300 357,300 (ho) Collections under multistate streamlined sales tax project PR S 40,000 40,000 (hp) Administration of income tax checkoff voluntary payments PR C -00- (m) Federal funds; state operations PR-F C -00- (q) Economic development surcharge	_							
taxes PR-S A 443,600 444,100 (ge) Administration of local professional football stadium district taxes PR-S A 128,700 128,700 (gf) Administration of resort tax PR-S A 83,900 83,900 (gg) Administration of local taxes PR A 134,700 134,700 (h) Debt collection PR A 2,702,000 2,851,100 (ha) Administration of liquor tax and alcohol beverages enforcement PR A 1,130,500 1,130,500 (hb) Collections by the department PR A 1,135,400 1,155,400 (hc) Collections from the financial record matching program PR A 506,600 510,100 (hd) Administration of liquor tax and alcohol beverages enforcement; wholesaler fees funding special agent position PR C 109,800 109,800 (hm) Collections under contracts PR S 357,300 357,300 (hn) Collections under the multistate tax commission audit program PR S 58,300 58,300 (ho) Collections under multistate streamlined sales tax project PR S 40,000 40,000 (hp) Administration of income tax checkoff voluntary payments PR C -00- (i) Gifts and grants PR C -00- (m) Federal funds; state operations PR-F C -00- (q) Economic development surcharge	_		1 IX	71	1,007,400	1,000,000		
(ge) Administration of local professional football stadium district taxes PR-S A 128,700 128,700 (gf) Administration of resort tax PR-S A 83,900 83,900 (gg) Administration of local taxes PR A 134,700 134,700 (h) Debt collection PR A 2,702,000 2,851,100 (ha) Administration of liquor tax and alcohol beverages enforcement PR A 1,130,500 1,130,500 (hb) Collections by the department PR A 1,135,400 1,155,400 (hc) Collections from the financial record matching program PR A 506,600 510,100 (hd) Administration of liquor tax and alcohol beverages enforcement; wholesaler fees funding special agent position PR S 357,300 357,300 (hm) Collections under the multistate tax commission audit program PR S 58,300 58,300 (ho) Collections under multistate streamlined sales tax project PR S 40,000 40,000 (hp) Administration of income tax checkoff voluntary payments PR C -00- (m) Federal funds; state operations PR-F C -00- (q) Economic development surcharge	(50)	-	PR-S	Α	443,600	444,100		
football stadium district taxes PR-S A 128,700 128,700 (gf) Administration of resort tax PR-S A 83,900 83,900 (gg) Administration of local taxes PR A 134,700 134,700 (h) Debt collection PR A 2,702,000 2,851,100 (ha) Administration of liquor tax and alcohol beverages enforcement PR A 1,130,500 1,130,500 (hb) Collections by the department PR A 1,135,400 1,155,400 (hc) Collections from the financial record matching program PR A 506,600 510,100 (hd) Administration of liquor tax and alcohol beverages enforcement; wholesaler fees funding special agent position PR S 357,300 357,300 (hm) Collections under the multistate tax commission audit program PR S 58,300 58,300 (ho) Collections under multistate streamlined sales tax project PR S 40,000 40,000 (hp) Administration of income tax checkoff voluntary payments PR C -00- (m) Federal funds; state operations PR-F C -00- (q) Economic development surcharge	(ge)				,	,		
(gg) Administration of local taxes PR A 134,700 134,700 (h) Debt collection PR A 2,702,000 2,851,100 (ha) Administration of liquor tax and alcohol beverages enforcement PR A 1,130,500 1,130,500 (hb) Collections by the department PR A 1,135,400 1,155,400 (hc) Collections from the financial record matching program PR A 506,600 510,100 (hd) Administration of liquor tax and alcohol beverages enforcement; wholesaler fees funding special agent position PR S 357,300 109,800 (hm) Collections under contracts PR S 357,300 357,300 (hn) Collections under the multistate tax commission audit program PR S 58,300 58,300 (ho) Collections under multistate streamlined sales tax project PR S 40,000 40,000 (hp) Administration of income tax checkoff voluntary payments PR A 27,300 27,300 (i) Gifts and grants PR C -00- (m) Federal funds; state operations PR-F C -00- (q) Economic development surcharge	(C)		PR-S	A	128,700	128,700		
(h) Debt collection PR A 2,702,000 2,851,100 (ha) Administration of liquor tax and alcohol beverages enforcement PR A 1,130,500 1,130,500 (hb) Collections by the department PR A 1,135,400 1,155,400 (hc) Collections from the financial record matching program PR A 506,600 510,100 (hd) Administration of liquor tax and alcohol beverages enforcement; wholesaler fees funding special agent position PR C 109,800 109,800 (hm) Collections under contracts PR S 357,300 357,300 (hn) Collections under the multistate tax commission audit program PR S 58,300 58,300 (ho) Collections under multistate streamlined sales tax project PR S 40,000 40,000 (hp) Administration of income tax checkoff voluntary payments PR A 27,300 27,300 (i) Gifts and grants PR C -00- (m) Federal funds; state operations PR-F C -00- (q) Economic development surcharge	(gf)	Administration of resort tax	PR-S	A	83,900	83,900		
(h) Debt collection PR A 2,702,000 2,851,100 (ha) Administration of liquor tax and alcohol beverages enforcement PR A 1,130,500 1,130,500 (hb) Collections by the department PR A 1,135,400 1,155,400 (hc) Collections from the financial record matching program PR A 506,600 510,100 (hd) Administration of liquor tax and alcohol beverages enforcement; wholesaler fees funding special agent position PR S 357,300 109,800 (hm) Collections under contracts PR S 357,300 357,300 (hn) Collections under the multistate tax commission audit program PR S 58,300 58,300 (ho) Collections under multistate streamlined sales tax project PR S 40,000 40,000 (hp) Administration of income tax checkoff voluntary payments PR A 27,300 27,300 (i) Gifts and grants PR C -00- (q) Economic development surcharge	(gg)	Administration of local taxes	PR	A	134,700	134,700		
alcohol beverages enforcement PR A 1,130,500 1,130,500 (hb) Collections by the department PR A 1,135,400 1,155,400 (hc) Collections from the financial record matching program PR A 506,600 510,100 (hd) Administration of liquor tax and alcohol beverages enforcement; wholesaler fees funding special agent position PR C 109,800 109,800 (hm) Collections under contracts PR S 357,300 357,300 (hn) Collections under the multistate tax commission audit program PR S 58,300 58,300 (ho) Collections under multistate streamlined sales tax project PR S 40,000 40,000 (hp) Administration of income tax checkoff voluntary payments PR C -00- (m) Federal funds; state operations PR-F C -00- (q) Economic development surcharge		Debt collection	PR	A	2,702,000	2,851,100		
(hb) Collections by the department (hc) Collections from the financial record matching program (hd) Administration of liquor tax and alcohol beverages enforcement; wholesaler fees funding special agent position (hm) Collections under contracts (hm) Collections under the multistate tax commission audit program (ho) Collections under multistate tax streamlined sales tax project (ho) Administration of income tax checkoff voluntary payments (ho) Federal funds; state operations (ho) Federal f	(ha)	Administration of liquor tax and						
(hc) Collections from the financial record matching program PR A 506,600 510,100 (hd) Administration of liquor tax and alcohol beverages enforcement; wholesaler fees funding special agent position PR C 109,800 109,800 (hm) Collections under contracts PR S 357,300 357,300 (hn) Collections under the multistate tax commission audit program PR S 58,300 58,300 (ho) Collections under multistate streamlined sales tax project PR S 40,000 40,000 (hp) Administration of income tax checkoff voluntary payments PR C -00- (m) Federal funds; state operations PR-F C -00- (q) Economic development surcharge		alcohol beverages enforcement	PR	A	1,130,500	1,130,500		
matching program PR A 506,600 510,100 (hd) Administration of liquor tax and alcohol beverages enforcement; wholesaler fees funding special agent position PR C 109,800 109,800 (hm) Collections under contracts PR S 357,300 357,300 (hn) Collections under the multistate tax commission audit program PR S 58,300 58,300 (ho) Collections under multistate streamlined sales tax project PR S 40,000 40,000 (hp) Administration of income tax checkoff voluntary payments PR A 27,300 27,300 (i) Gifts and grants PR C -00- (m) Federal funds; state operations PR-F C -00- (q) Economic development surcharge	. ,		PR	A	1,135,400	1,155,400		
(hd) Administration of liquor tax and alcohol beverages enforcement; wholesaler fees funding special agent position PR C 109,800 109,800 (hm) Collections under contracts PR S 357,300 357,300 (hn) Collections under the multistate tax commission audit program PR S 58,300 58,300 (ho) Collections under multistate streamlined sales tax project PR S 40,000 40,000 (hp) Administration of income tax checkoff voluntary payments PR A 27,300 27,300 (i) Gifts and grants PR C -00- (m) Federal funds; state operations PR-F C -00- (q) Economic development surcharge	(hc)							
alcohol beverages enforcement; wholesaler fees funding special agent position PR C 109,800 109,800 (hm) Collections under contracts PR S 357,300 357,300 (hn) Collections under the multistate tax commission audit program PR S 58,300 58,300 (ho) Collections under multistate streamlined sales tax project PR S 40,000 40,000 (hp) Administration of income tax checkoff voluntary payments PR A 27,300 27,300 (i) Gifts and grants PR C -00- (m) Federal funds; state operations PR-F C -00- (q) Economic development surcharge			PR	A	506,600	510,100		
wholesaler fees funding special agent position PR C 109,800 109,800 (hm) Collections under contracts PR S 357,300 357,300 (hn) Collections under the multistate tax commission audit program PR S 58,300 58,300 (ho) Collections under multistate streamlined sales tax project PR S 40,000 40,000 (hp) Administration of income tax checkoff voluntary payments PR A 27,300 27,300 (i) Gifts and grants PR C -00- (m) Federal funds; state operations PR-F C -00- (q) Economic development surcharge	(hd)							
position PR C 109,800 109,800 (hm) Collections under contracts PR S 357,300 357,300 (hn) Collections under the multistate tax commission audit program PR S 58,300 58,300 (ho) Collections under multistate streamlined sales tax project PR S 40,000 40,000 (hp) Administration of income tax checkoff voluntary payments PR A 27,300 27,300 (i) Gifts and grants PR C -00- (m) Federal funds; state operations PR-F C -00- (q) Economic development surcharge								
(hm) Collections under contracts PR S 357,300 357,300 (hn) Collections under the multistate tax commission audit program PR S 58,300 58,300 (ho) Collections under multistate streamlined sales tax project PR S 40,000 40,000 (hp) Administration of income tax checkoff voluntary payments PR A 27,300 27,300 (i) Gifts and grants PR C -00- (m) Federal funds; state operations PR-F C -00- (q) Economic development surcharge			DD	C	100 800	100 800		
(hn) Collections under the multistate tax commission audit program PR S 58,300 58,300 (ho) Collections under multistate streamlined sales tax project PR S 40,000 40,000 (hp) Administration of income tax checkoff voluntary payments PR A 27,300 27,300 (i) Gifts and grants PR C -00- (m) Federal funds; state operations PR-F C -00- (q) Economic development surcharge	(hm)							
commission audit program PR S 58,300 58,300 (ho) Collections under multistate streamlined sales tax project PR S 40,000 40,000 (hp) Administration of income tax checkoff voluntary payments PR A 27,300 27,300 (i) Gifts and grants PR C -00- (m) Federal funds; state operations PR-F C -00- (q) Economic development surcharge	` ,		1 IX	S	337,300	337,300		
(ho) Collections under multistate streamlined sales tax project PR S 40,000 40,000 (hp) Administration of income tax checkoff voluntary payments PR A 27,300 27,300 (i) Gifts and grants PR C -00- (m) Federal funds; state operations PR-F C -00- (q) Economic development surcharge	(1111)		PR	S	58 300	58 300		
streamlined sales tax project PR S 40,000 40,000 (hp) Administration of income tax checkoff voluntary payments PR A 27,300 27,300 (i) Gifts and grants PR C -00- (m) Federal funds; state operations PR-F C -00- (q) Economic development surcharge	(ho)	ž v	110	5	30,300	20,300		
(hp) Administration of income tax checkoff voluntary payments PR A 27,300 27,300 (i) Gifts and grants PR C -00- (m) Federal funds; state operations PR-F C -00- (q) Economic development surcharge	()		PR	S	40,000	40,000		
checkoff voluntary payments PR A 27,300 27,300 (i) Gifts and grants PR C -00- (m) Federal funds; state operations PR-F C -00- (q) Economic development surcharge	(hp)				,	,		
(i) Gifts and grants PR C -00- (m) Federal funds; state operations PR-F C -0- (q) Economic development surcharge	\ 1 /		PR	A	27,300	27,300		
(m) Federal funds; state operations PR-F C -00- (q) Economic development surcharge	(i)		PR		-0-			
	(m)	=	PR-F	C	-0-	-0-		
administration SEG A 258,300 258,300	(q)							
		administration	SEG	A	258,300	258,300		

2017 As	ssembly Bill 64	- 113 -	_	2017 W	isconsin Act 59
STATUT	E, AGENCY AND PURPOSE	Source	Түре	2017-2018	2018-2019
(qm)	Administration of rental vehicle fee	SEG	A	62,700	62,700
(r)	Administration of dry cleaner fees	SEG	A	18,900	18,900
(s)	Petroleum inspection fee collection	SEG	A	81,700	81,700
(t)	Farmland preservation credit, 2010			,	,
(-)	and beyond	SEG	A	-0-	-0-
(u)	Motor fuel tax administration	SEG	A	1,688,900	1,695,600
		(1) PROC	GRAM TO	OTALS	
(GENERAL PURPOSE REVENUE			63,737,600	63,873,000
	PROGRAM REVENUE			11,846,200	12,028,700
	FEDERAL			(-0-)	(-0-)
	OTHER			(11,190,000)	(11,372,000)
	SERVICE			(656,200)	(656,700)
	SEGREGATED REVENUE			2,110,500	2,117,200
	OTHER			(2,110,500)	(2,117,200)
,	TOTAL-ALL SOURCES			77,694,300	78,018,900
(2)	STATE AND LOCAL FINANCE				
(a)	General program operations	GPR	A	7,757,500	7,778,400
(b)	Valuation error loans	GPR	S	-0-	-0-
(bm)	Integrated property assessment				
	system technology	GPR	A	2,460,900	2,460,900
(g)	County assessment studies	PR	C	-0-	-0-
(ga)	Commercial property assessment	PR	C	-0-	-0-
(gb)	Manufacturing property assessment	PR	A	1,195,200	1,199,000
(gi)	Municipal finance report compliance	PR	A	32,800	32,800
(h)	Reassessments	PR	A	273,500	273,500
(hm)	Administration of tax incremental,				
	and environmental remediation tax		~	4=4.000	4=0.000
40	incremental, financing programs	PR	C	174,800	179,000
(i)	Gifts and grants	PR	C	-0-	-0-
(m)	Federal funds; state operations	PR-F	C	-0-	-0-
(q)	Railroad and air carrier tax	CEC		240.000	240.000
()	administration	SEG	A	249,800	249,800
(r)	Lottery and gaming credit administration	SEG	A	267,900	272,700
	adilillistration		GRAM TO		272,700
	GENERAL PURPOSE REVENUE	(2) TROC	JICANI I C	10,218,400	10,239,300
	PROGRAM REVENUE			1,676,300	1,684,300
-	FEDERAL			(-0-)	(-0-)
	OTHER			(1,676,300)	(1,684,300)
	SEGREGATED REVENUE			517,700	522,500
,	OTHER			(517,700)	(522,500)
,	TOTAL-ALL SOURCES			12,412,400	12,446,100
(3)	ADMINISTRATIVE SERVICES AND SPACE R	ENTAL		,,	, ,
(a)	General program operations	GPR	A	30,884,500	30,957,900
(b)	Integrated tax system technology	GPR	A	4,087,100	4,087,100
(c)	Expert professional services	GPR	В	63,300	63,300
(g)	Services	PR	A	81,300	81,300
(gm)	Reciprocity agreement and	•		2-,000	- , , , , , , , , , , , , , , , , , , ,
(8)	publications	PR	A	36,000	36,000
(go)	Reciprocity agreement, Illinois	PR	A	-0-	-0-
(i)	Gifts and grants	PR	C	-0-	-0-
	-				

2017 W	2017 Wisconsin Act 59		_	2017 A	2017 Assembly Bill 64		
STATU	TE, AGENCY AND PURPOSE	Source	Түре	2017-2018	2018-2019		
(k)	Internal services	PR-S	A	2,909,700	2,911,100		
(m)	Federal funds; state operations	PR-F	C	-0-	-0-		
		(3) PRO0	GRAM TO	ΓALS			
	GENERAL PURPOSE REVENUE			35,034,900	35,108,300		
	PROGRAM REVENUE			3,027,000	3,028,400		
	FEDERAL			(-0-)	(-0-)		
	OTHER			(117,300)	(117,300)		
	SERVICE			(2,909,700)	(2,911,100)		
	TOTAL-ALL SOURCES			38,061,900	38,136,700		
(4)	UNCLAIMED PROPERTY PROGRAM						
(a)	Unclaimed property; contingency	GPR	C	0	0		
(j)	appropriation Unclaimed property; claims	PR	S C	-0- -0-	-0- -0-		
(k)	Unclaimed property; administrative		C	_0_	-0-		
(K)	expenses	PR-S	A	3,828,200	3,838,900		
	expenses		GRAM TO		3,030,700		
	GENERAL PURPOSE REVENUE	(1) 1110		-0-	-0-		
	PROGRAM REVENUE			3,828,200	3,838,900		
	OTHER			(-0-)	(-0-)		
	SERVICE			(3,828,200)	(3,838,900)		
	TOTAL-ALL SOURCES			3,828,200	3,838,900		
(7)	INVESTMENT AND LOCAL IMPACT FUN	D					
(e)	Investment and local impact fund						
	supplement	GPR	A	-0-	-0-		
(g)	Investment and local impact fund			_	_		
	administrative expenses	PR	A	-0-	-0-		
(n)	Federal mining revenue	PR-F	C	-0-	-0-		
(v)	Investment and local impact fund	SEG	C	-0-	-0-		
	CENED AL DUDDOCE DEVENUE	(7) PROC	GRAM TO		0		
	GENERAL PURPOSE REVENUE PROGRAM REVENUE			-0- -0-	-0- -0-		
	FEDERAL			(-0-)	(-0-)		
	OTHER			(-0-)	(-0-)		
	SEGREGATED REVENUE			-0-	- 0-		
	OTHER			(-0-)	(-0-)		
	TOTAL-ALL SOURCES			-0-	-0-		
(8)	LOTTERY			Ů	v		
(b)	Retailer compensation	GPR	A	8,000,000	40,000,000		
(q)	General program operations	SEG	A	19,407,500	19,375,300		
(r)	Retailer compensation	SEG	S	34,765,800	3,427,400		
(s)	Prizes	SEG	S	-0-	-0-		
(v)	Vendor fees	SEG	S	15,708,600	15,952,900		
		(8) PRO	GRAM TO	ΓALS			
	GENERAL PURPOSE REVENUE			8,000,000	40,000,000		
	SEGREGATED REVENUE			69,881,900	38,755,600		
	OTHER			(69,881,900)	(38,755,600)		
	TOTAL-ALL SOURCES			77,881,900	78,755,600		
	GP14PD 14 P24P2 27	20.566 DEPA	ARTMENT T		4.40.220.42		
	GENERAL PURPOSE REVENUE			116,990,900	149,220,600		
	PROGRAM REVENUE			20,377,700	20,580,300		
	FEDERAL			(-0-)	(-0-)		

2017 A	2017 Assembly Bill 64		_	2017 W	2017 Wisconsin Act 59		
STATU	IE, AGENCY AND PURPOSE OTHER SERVICE SEGREGATED REVENUE OTHER TOTAL-ALL SOURCES	Source	Түре	2017–2018 (12,983,600) (7,394,100) 72,510,100 (72,510,100) 209,878,700	2018–2019 (13,173,600) (7,406,700) 41,395,300 (41,395,300) 211,196,200		
20.575	Secretary of State			209,878,700	211,190,200		
(1)	Managing and operating program	I RESPONSIBILI	TIES				
(g)	Program fees	PR	A	262,300	262,300		
(ka)	Agency collections	PR-S	A	3,400	3,400		
		(1) PRO	GRAM TO	TALS			
	PROGRAM REVENUE			265,700	265,700		
	OTHER			(262,300)	(262,300)		
	SERVICE			(3,400)	(3,400)		
	TOTAL-ALL SOURCES			265,700	265,700		
		20.575 DEPA	ARTMENT '				
	PROGRAM REVENUE			265,700	265,700		
	OTHER			(262,300)	(262,300)		
	SERVICE TOTAL-ALL SOURCES			(3,400) 265,700	(3,400) 265,700		
20 585	Treasurer, State			203,700	203,700		
(1)	Custodian of state funds						
(b)	Insurance	GPR	A	-0-	-0-		
(h)	Training conferences	PR	C	-0-	-0-		
(i)	Gifts and grants	PR	C	-0-	-0-		
(k)	Administrative expenses	PR-S	A	113,500	113,500		
(kb)	General program operations	PR-S	A	-0-	-0-		
, ,	1 0 1	(1) PRO	GRAM TO	TALS			
	GENERAL PURPOSE REVENUE			-0-	-0-		
	PROGRAM REVENUE			113,500	113,500		
	OTHER			(-0-)	(-0-)		
	SERVICE			(113,500)	(113,500)		
	TOTAL-ALL SOURCES			113,500	113,500		
		20.585 DEPA	ARTMENT '				
	GENERAL PURPOSE REVENUE			-0- 112.500	-0- 112 500		
	PROGRAM REVENUE			113,500	113,500		
	OTHER SERVICE			(-0-) (113,500)	(-0-) (113,500)		
	TOTAL-ALL SOURCES			113,500	113,500		
	TOTAL-ALL SOURCES	General F	xecutive Fur		113,300		
		FUNCTION					
	GENERAL PURPOSE REVENUE	101(01101)		583,560,700	676,762,700		
	PROGRAM REVENUE			562,505,100	601,520,700		
	FEDERAL			(140,083,600)	(139,930,200)		
	OTHER			(96,896,000)	(97,282,200)		
	SERVICE			(325,525,500)	(364,308,300)		
	SEGREGATED REVENUE			183,541,600	143,847,400		
	FEDERAL			(2,674,800)	(-0-)		
	OTHER			(180,866,800)	(143,847,400)		
	SERVICE			(-0-)	(-0-)		
	LOCAL			(-0-)	(-0-)		

2017 W	2017 Wisconsin Act 59		_	2017	2017 Assembly Bill 64		
STATU	TE, AGENCY AND PURPOSE TOTAL–ALL SOURCES	Source	ТүрЕ	2017–2018 1,329,607,400	2018–2019 1,422,130,800		
		Judici	al	, , ,	, , ,		
20.62	5 Circuit Courts	-					
(1)	COURT OPERATIONS						
(a)	Circuit courts	GPR	S	76,883,700	76,883,700		
(b)	Permanent reserve judges	GPR	A	-0-	-0-		
(cg)	Circuit court costs	GPR	В	24,676,800	24,676,800		
(g)	Sale of materials and services	PR	C	-0-	-0-		
(k)	Court interpreters	PR-S	A	232,700	232,700		
(m)	Federal aid	PR-F	C CDAM TO	-0-	-0-		
	GENERAL PURPOSE REVENUE	(I) PROC	GRAM TO		101 560 500		
	PROGRAM REVENUE			101,560,500 232,700	101,560,500 232,700		
	FEDERAL			(-0-)	(-0-)		
	OTHER			(-0-)	(-0-)		
	SERVICE			(232,700)	(232,700)		
	TOTAL-ALL SOURCES			101,793,200	101,793,200		
		20.625 DEPA	ARTMENT		, ,		
	GENERAL PURPOSE REVENUE			101,560,500	101,560,500		
	PROGRAM REVENUE			232,700	232,700		
	FEDERAL			(-0-)	(-0-)		
	OTHER			(-0-)	(-0-)		
	SERVICE			(232,700)	(232,700)		
•0.66	TOTAL-ALL SOURCES			101,793,200	101,793,200		
	Court of Appeals						
(1)	APPELLATE PROCEEDINGS General program operations	GPR	S	11,149,700	11,171,900		
(a) (m)	General program operations Federal aid	PR-F	S C	11,149,700 -0-	-0-		
(111)	rederar ard		GRAM TO		O		
	GENERAL PURPOSE REVENUE	(1) 1100		11,149,700	11,171,900		
	PROGRAM REVENUE			-0-	-0-		
	FEDERAL			(-0-)	(-0-)		
	TOTAL-ALL SOURCES			11,149,700	11,171,900		
		20.660 DEPA	ARTMENT	TOTALS			
	GENERAL PURPOSE REVENUE			11,149,700	11,171,900		
	PROGRAM REVENUE			-0-	-0-		
	FEDERAL			(-0-)	(-0-)		
20.66	TOTAL-ALL SOURCES			11,149,700	11,171,900		
	5 Judicial Commission						
(1) (a)	JUDICIAL CONDUCT General program operations	GPR	A	287,300	287,900		
	Contractual agreements	GPR	В	16,200	16,200		
	Federal aid	PR-F	C	-0-	-0-		
(11111)	-,		GRAM TO		Ü		
	GENERAL PURPOSE REVENUE	() ===0	0	303,500	304,100		
	PROGRAM REVENUE			-0-	-0-		
	FEDERAL			(-0-)	(-0-)		
	TOTAL-ALL SOURCES			303,500	304,100		
		20.665 DEPA	ARTMENT				
	GENERAL PURPOSE REVENUE			303,500	304,100		

)17 A	Assembly Bill 64	– 117	_	2017 W	isconsin Act 59
Statu	TTE, AGENCY AND PURPOSE	Source	Түре	2017-2018	2018-2019
	PROGRAM REVENUE			-0-	-0-
	FEDERAL			(-0-)	(-0-)
	TOTAL-ALL SOURCES			303,500	304,100
20.67	0 Judicial Council				
(1)	ADVISORY SERVICES TO THE COURTS A	ND THE LEGIS	LATURE		
(a)	General program operations	GPR	A	-0-	-0-
(k)	Director of state courts and law				
	library transfer	PR-S	C	111,400	111,400
(m)	Federal aid	PR-F	C	-0-	-0-
		(1) PRO	GRAM TOT	ALS	
	GENERAL PURPOSE REVENUE			-0-	-0-
	PROGRAM REVENUE			111,400	111,400
	FEDERAL			(-0-)	(-0-)
	SERVICE			(111,400)	(111,400)
	TOTAL-ALL SOURCES			111,400	111,400
		20.670 DEP	ARTMENT T		,
	GENERAL PURPOSE REVENUE			-0-	-0-
	PROGRAM REVENUE			111,400	111,400
	FEDERAL			(-0-)	(-0-)
	SERVICE			(111,400)	(111,400)
	TOTAL-ALL SOURCES			111,400	111,400
20.680	0 Supreme Court			,	,
(1)	SUPREME COURT PROCEEDINGS				
(a)	General program operations	GPR	S	5,529,700	5,529,700
(m)	Federal aid	PR-F	C	-0-	-0-
(111)	1 000101 010		GRAM TOT		v
	GENERAL PURPOSE REVENUE	(-)		5,529,700	5,529,700
	PROGRAM REVENUE			-0-	-0-
	FEDERAL			(-0-)	(-0-)
	TOTAL-ALL SOURCES			5,529,700	5,529,700
(2)	DIRECTOR OF STATE COURTS AND LAW	LIBRARY		-,,,	-,,
(a)	General program operations	GPR	В	11,665,900	11,703,300
	Gifts and grants	PR	C	663,500	663,500
(ga)		PR	C	66,100	66,100
(gc)	9		-	00,100	00,100
(3-)	certification	PR	C	45,100	45,100
(h)	Materials and services	PR	C	103,600	103,600
(i)	Municipal judge training	PR	C	181,200	181,200
(j)	Court information systems	PR	C	7,135,100	7,158,200
(kc)	•	PR-S	A	258,300	258,600
(ke)		-		,	,
\ -/	automation assistance	PR-S	C	-0-	-0-
(kf)	Interagency and intra-agency				
. /	assistance	PR-S	C	-0-	-0-
(L)	Library collections and services	PR	C	145,500	145,500
(m)	Federal aid	PR-F	C	987,100	987,200
(qm)		SEG	C	820,200	821,400
			GRAM TOT		*
	GENERAL PURPOSE REVENUE	. /		11,665,900	11,703,300
	PROGRAM REVENUE			9,585,500	9,609,000
				/ /- 	(987,200)

2017 Wisconsin Act 59		- 118	_	2017 A	2017 Assembly Bill 64		
Statu	TE, AGENCY AND PURPOSE OTHER SERVICE SEGREGATED REVENUE OTHER	Source	Түре	2017–2018 (8,340,100) (258,300) 820,200 (820,200)	2018–2019 (8,363,200) (258,600) 821,400 (821,400)		
(2)	TOTAL-ALL SOURCES			22,071,600	22,133,700		
(3) (b) (c) (g) (h)	BAR EXAMINERS AND RESPONSIBILITY General program operations Contractual agreements Board of bar examiners Office of lawyer regulation	GPR GPR PR PR	A B C C	-0- -0- 839,700 3,271,700	-0- -0- 839,700 3,271,700		
		(3) PRO	GRAM TO				
	GENERAL PURPOSE REVENUE PROGRAM REVENUE OTHER TOTAL-ALL SOURCES			-0- 4,111,400 (4,111,400) 4,111,400	-0- 4,111,400 (4,111,400) 4,111,400		
		20.680 DEP	ARTMENT	, , , , , , , , , , , , , , , , , , ,	, ,		
	GENERAL PURPOSE REVENUE PROGRAM REVENUE FEDERAL OTHER SERVICE SEGREGATED REVENUE OTHER TOTAL-ALL SOURCES GENERAL PURPOSE REVENUE PROGRAM REVENUE FEDERAL OTHER SERVICE SEGREGATED REVENUE FEDERAL OTHER SERVICE SEGREGATED REVENUE FEDERAL OTHER	FUNCTION	Judicial	17,195,600 13,696,900 (987,100) (12,451,500) (258,300) 820,200 (820,200) 31,712,700 TOTALS 130,209,300 14,041,000 (987,100) (12,451,500) (602,400) 820,200 (-0-) (820,200)	17,233,000 13,720,400 (987,200) (12,474,600) (258,600) 821,400 (821,400) 31,774,800 130,269,500 14,064,500 (987,200) (12,474,600) (602,700) 821,400 (-0-) (821,400)		
	SERVICE LOCAL			(-0-) (-0-)	(-0-) (-0-)		
	TOTAL-ALL SOURCES			145,070,500	145,155,400		
		Legisla	tive	, ,	, ,		
20.76 : (1) (a)	5 Legislature ENACTMENT OF STATE LAWS General program operations —	8					
(b)	assembly General program operations —	GPR	S	26,558,800	26,558,800		
(d) (e)	senate Legislative documents Gifts, grants, and bequests	GPR GPR PR	S S C	18,813,300 3,919,100 -0-	18,813,300 3,919,100 -0-		
	GENERAL PURPOSE REVENUE	(1) PRO	GRAM TO	OTALS 49,291,200	49,291,200		
	PROGRAM REVENUE OTHER			-0- (-0-)	-0- (-0-)		

17 A	ssembly Bill 64	– 119 -	_	2017 W	isconsin Act 59	
STATUT	TE, AGENCY AND PURPOSE	Source	Түре	2017-2018	2018-2019	
	TOTAL-ALL SOURCES			49,291,200	49,291,200	
(3)	SERVICE AGENCIES AND NATIONAL ASSOC	IATIONS				
(b)	Legislative reference bureau	GPR	В	5,968,300	5,979,600	
(c)	Legislative audit bureau	GPR	В	6,377,900	6,393,900	
(d)	Legislative fiscal bureau	GPR	В	3,985,700	3,999,100	
(e)	Joint legislative council; execution of functions, conduct of research, development of studies, and the	GPR	В	2.062.700	2.076.200	
(00)	provision of assistance to committees Joint legislative council; contractual	UFK	Б	3,963,700	3,976,800	
(ec)	studies	GPR	В	15,000	-0-	
(em)		OFK	Ь	13,000	O	
(CIII)	bureau	GPR	В	4,370,000	4,382,400	
(f)	Joint committee on legislative			, ,	, ,	
. /	organization	GPR	В	-0-	-0-	
(fa)	Membership in national associations	GPR	S	265,900	269,000	
(g)	Gifts and grants to service agencies	PR	C	10,000	10,000	
(ka)	Audit bureau reimbursable audits	PR-S	A	2,223,200	2,228,300	
(m)	Federal aid	PR-F	C	-0-	-0-	
		(3) PRO0	GRAM TOT	ALS		
	GENERAL PURPOSE REVENUE			24,946,500	25,000,800	
	PROGRAM REVENUE			2,233,200	2,238,300	
	FEDERAL			(-0-)	(-0-)	
	OTHER			(10,000)	(10,000)	
	SERVICE			(2,223,200)	(2,228,300)	
	TOTAL-ALL SOURCES			27,179,700	27,239,100	
4)	CAPITOL OFFICES RELOCATION; 100TH AN		OF STATE CAL	PITOL		Vet
(a)	Capitol offices relocation costs	GPR	В	-0-	-0-	In l
(b)	Celebration of 100th anniversary of					Vet
	state capitol; general purpose	CDD		50,000	0	In I
(1.)	revenue	GPR	A	50,000	-0-	
(h)	Celebration of 100th anniversary of	PR	С	-0-	-0-	
	state capitol; program revenue		GRAM TOT		_0_	
	GENERAL PURPOSE REVENUE	(4) TKO	JKAWI 101	50,000	-0-	
	PROGRAM REVENUE			-0-	_0_ _0_	
	OTHER			(-0-)	(-0-)	
	TOTAL-ALL SOURCES			50,000	-0-	
		765 DFP4	ARTMENT T		O	
	GENERAL PURPOSE REVENUE	103 DEIT	iitiviibivi i	74,287,700	74,292,000	
	PROGRAM REVENUE			2,233,200	2,238,300	
	FEDERAL			(-0-)	(-0-)	
	OTHER			(10,000)	(10,000)	
	SERVICE			(2,223,200)	(2,228,300)	
	TOTAL-ALL SOURCES			76,520,900	76,530,300	
		I.	egislative	, 0,520,700	, 0,550,500	
	FI		AL AREA T	OTALS		
	GENERAL PURPOSE REVENUE			74,287,700	74,292,000	
	PROGRAM REVENUE			2,233,200	2,238,300	
	FEDERAL			(-0-)	(-0-)	

2017 Wi	sconsin Act 59	- 120 -	_	2017 A	Assembly Bill 64
	E, AGENCY AND PURPOSE SERVICE	Source	Түре	2017–2018 (2,223,200)	2018–2019 (2,228,300)
\$	SEGREGATED REVENUE			-0-	-0-
	FEDERAL			(-0-)	(-0-)
	OTHER			(-0-)	(-0-)
	SERVICE			(-0-)	(-0-)
	LOCAL			(-0-)	(-0-)
r	TOTAL-ALL SOURCES			76,520,900	76,530,300
	Genera	ıl Appro	opriati	ons	
20.835	Shared Revenue and Tax Relief				
(1)	SHARED REVENUE PAYMENTS				
(c)	Expenditure restraint program account	GPR	S	58,145,700	59,311,700
(db)	County and municipal aid account	GPR	S	692,375,200	699,088,400
(dm)	Public utility distribution account	GPR	S	74,034,500	74,880,600
(e)	State aid; tax exempt property	GPR	S	94,270,000	94,660,000
(f)	State aid; personal property tax exemption	GPR	S	-0-	74,400,000
(r)	County and municipal aid account;	OTK	S	V	74,400,000
(1)	police and fire protection fund	SEG	C	51,700,600	44,987,400
	ponce and me protection rand	(1) PROC			11,207,100
(GENERAL PURPOSE REVENUE	(1) 1100	SICILIVI I	918,825,400	1,002,340,700
	SEGREGATED REVENUE			51,700,600	44,987,400
,	OTHER			(51,700,600)	(44,987,400)
,	TOTAL-ALL SOURCES			970,526,000	1,047,328,100
	TAX RELIEF			970,320,000	1,047,320,100
(b)	Claim of right credit	GPR	S	191,000	191,000
(b) (bb)	Jobs tax credit	GPR	S	20,000,000	13,000,000
(bc)	Woody biomass harvesting and	OIK	3	20,000,000	13,000,000
(60)	processing credit	GPR	S	-0-	-0-
(bd)	Meat processing facility investment	OIK	5	-0-	-0-
(60)	credit	GPR	S	-0-	-0-
(be)	Food processing plant and food	OTIC	S	V	· ·
(60)	warehouse investment credit	GPR	S	-0-	-0-
(bg)	Business development credit	GPR	S	24,250,000	22,000,000
(bL)	Film production company investment	OTT	S	21,230,000	22,000,000
(CL)	credit	GPR	S	-0-	-0-
(bm)	Film production services credit	GPR	S	-0-	-0-
(bn)	Dairy manufacturing facility investment credit	GPR	S	-0-	-0-
(bp)	Dairy manufacturing facility	GPR	S	-0-	-0-
(br)	investment credit; dairy cooperatives Interest payments on overassessments of manufacturing	OFK	3	_0_	_0_
	property	GPR	S	10,000	10,000
(c)	Homestead tax credit	GPR	S	93,400,000	85,900,000
(co)	Enterprise zone jobs credit	GPR	S	37,600,000	43,000,000
(d)	Research credit	GPR	S	-0-	2,100,000
(dm)	Farmland preservation credit	GPR	S	625,000	475,000
(dn)	Farmland tax relief credit	GPR	S	-0-	-0-
(dn) (do)	Farmland preservation credit, 2010	J. 10	~	V	J
(40)	and beyond	GPR	S	18,700,000	18,950,000

generate and refunds (ep) Cigarette and refunds (f) Earned incord (ka) Farmland tax gaming received (kf) Earned incord assistance for assistance for assistance for assistance for a service (q) Farmland tax GENERAL PUPROGRAM RUSERGATE OTHER TOTAL—ALL (3) STATE PROPER (b) School levy to credit (ef) Transfer to conforestry (q) Lottery and generate as a service (q) Lottery and generate applications GENERAL PUSEGREGATE OTHER TOTAL—ALL (4) COUNTY AND Generate as a service (generate as a service (gene	y Bill 64	- 121 ·	_	2017 V	Visconsin Act :
generate and refunds (f) Earned incorrefunds (f) Earned incorrefunds (f) Earned incorrefunds (f) Earned incorrefunds (gaming received for the following form of the following follo	CY AND PURPOSE	Source	Түре	2017-2018	2018-2019
(en) Beginning far owner tax crees (ep) Cigarette and refunds (f) Earned incores (ka) Farmland tax gaming received (kf) Earned incores assistance for assistance for assistance for assistance for a service (q) Farmland tax GENERAL PUPROGRAM RESERVICE TOTAL—ALL (3) STATE PROPER TOTAL—ALL (3) STATE PROPER (ef) Transfer to conforestry (ef) Lottery and gapplications GENERAL PUSEGREGATE OTHER TOTAL—ALL (4) COUNTY AND SEGREGATE (ef) Special district taxes (gg) Local profess district taxes (gg) Local taxes PROGRAM RESERVICE (ef) TOTAL—ALL (5) PAYMENTS IN INTERIOR (ef) Payments for GENERAL PURPLEMENTS IN INTERIOR (ef) PAYMENTS IN INTERIOR (ef) PAYM	ns and surviving spouses				
owner tax cre (ep) Cigarette and refunds (f) Earned incore (ka) Farmland tax gaming receive (kf) Earned incores assistance for assistance	ty tax credit	GPR	S	29,830,000	30,430,000
(ep) Cigarette and refunds (f) Earned incor (ka) Farmland tax gaming receit (kf) Earned incord assistance for assistance for assistance for GENERAL PUPROGRAM RUSEGREGATE TOTAL-ALL (3) STATE PROPER (a) School levy to credit (b) School levy to credit (cf) Transfer to conforestry (d) Lottery and government of the second of	ning farmer and farm asset				
refunds (f) Earned incor (ka) Farmland tax gaming received: (kf) Earned incor assistance for (q) Farmland tax GENERAL PU PROGRAM R SERVIOUS SEGREGATE: OTHER TOTAL—ALL (3) STATE PROPER (b) School levy of creditity (ef) Transfer to conforestry (q) Lottery and good applications GENERAL PU SEGREGATE: OTHER TOTAL—ALL (4) COUNTY AND C	tax credit	GPR	S	-0-	-0-
(f) Earned incor (ka) Farmland tax gaming received: (kf) Earned incor assistance fo (q) Farmland tax GENERAL PU PROGRAM R SERVIOUS SEGREGATE: OTHER TOTAL-ALL (3) STATE PROPER (b) School levy to credit (ef) Transfer to conforestry (q) Lottery and governed	tte and tobacco product tax				
(ka) Farmland tax gaming recei (kf) Earned incor assistance for the second seco		GPR	S	34,888,300	33,996,000
gaming receives (kf) Earned income assistance for assistance for GENERAL PUPROGRAM RESERVIOUS SEGREGATE: OTHER TOTAL—ALL (3) STATE PROPER (b) School levy of credit (ef) Transfer to composit for forestry (q) Lottery and generated applications (g) Lottery and generated applications (g) County taxes (gb) Special district (gd) Premier reso (ge) Local profession district taxes (gg) Local taxes (gg) PROGRAM RESERVATE (5) PAYMENTS IN ITOTAL—ALL (5) PAYME	l income tax credit	GPR	S	30,100,000	35,000,000
(kf) Earned incor assistance for assistance for GENERAL PUPROGRAM RESERVICE SEGREGATE OTHER TOTAL-ALL (3) STATE PROPER (b) School levy acredit (ef) Transfer to a forestry (q) Lottery and ge applications GENERAL PUSEGREGATE OTHER TOTAL-ALL (4) COUNTY AND SEGREGATE OTHER TOTAL-ALL (4) County taxes (gb) Special district (gd) Premier reso (ge) Local profession district taxes (gg) Local taxes PROGRAM RESTORY OTHER TOTAL-ALL (5) PAYMENTS IN ITOTAL-ALL (5) PAYMENTS IN ITOTAL (5) P	and tax relief credit; Indian				
assistance fo (q) Farmland tax GENERAL PU PROGRAM R SERVICE SEGREGATE OTHER TOTAL-ALL (3) STATE PROPER (b) School levy of credit (ef) Transfer to composite for the compo	=	PR-S	C	-0-	-0-
GENERAL PUPROGRAM R SERVIOUS SEGREGATE OTHER TOTAL-ALL (3) STATE PROPER (b) School levy of credit (ef) Transfer to composite for forestry (q) Lottery and government of applications GENERAL PUSEGREGATE OTHER TOTAL-ALL (4) COUNTY AND OF CO	l income tax credit; temporary				
GENERAL PU PROGRAM R SERVIO SEGREGATE OTHER TOTAL-ALL (3) STATE PROPER (b) School levy to credit (ef) Transfer to composit for stry (q) Lottery and good applications GENERAL PU SEGREGATE OTHER TOTAL-ALL (4) COUNTY AND to composit for stry (gd) Premier resource (gd) Premier resource (gd) Premier resource (ge) Local profest district taxes (gg) Local taxes PROGRAM R OTHER TOTAL-ALL (5) PAYMENTS IN ITOTAL-ALL (5) PAYMENTS IN ITOTAL-ALL (5) PAYMENTS IN ITOTAL-ALL (5) PAYMENTS IN ITOTAL-ALL (6) GENERAL PU	nce for needy families	PR-S	A	69,700,000	69,700,000
PROGRAM R SERVIO SEGREGATE OTHER TOTAL-ALL (3) STATE PROPER (b) School levy of credit (ef) Transfer to composite for forestry (q) Lottery and government of supplications GENERAL PUSEGREGATE OTHER TOTAL-ALL (4) COUNTY AND SEGREGATE (gd) Premier resource (gd) Special district taxes (gd) Premier resource (ge) Local profest district taxes (gg) Local taxes PROGRAM R OTHER TOTAL-ALL (5) PAYMENTS IN INCIPATE TOTAL ALL (5) PAYMENTS IN INCIPATE TOTAL ALL (5) PAYMENTS IN INCIPATE TOTAL PAYMENTS IN INCIPATE TOTAL ALL (5) PAYMENTS IN INCIPATE TOTAL PAYMENTS IN INCI	and tax relief credit	SEG	S	-0-	-0-
PROGRAM R SERVIO SEGREGATE OTHER TOTAL-ALL (3) STATE PROPER (b) School levy of credit (ef) Transfer to composite for forestry (q) Lottery and government of supplications GENERAL PUSEGREGATE OTHER TOTAL-ALL (4) COUNTY AND SEGREGATE (gd) Premier resource (gd) Special district taxes (gd) Premier resource (ge) Local profest district taxes (gg) Local taxes PROGRAM R OTHER TOTAL-ALL (5) PAYMENTS IN INCIPATE TOTAL ALL (5) PAYMENTS IN INCIPATE TOTAL ALL (5) PAYMENTS IN INCIPATE TOTAL PAYMENTS IN INCIPATE TOTAL ALL (5) PAYMENTS IN INCIPATE TOTAL PAYMENTS IN INCI		(2) PRO0	GRAM TO	TALS	
SERVIGUE SEGREGATE OTHER TOTAL-ALL (3) STATE PROPER Credit (b) School levy of credit (ef) Transfer to composite for stry (q) Lottery and government of supplications GENERAL PUSEGREGATE OTHER TOTAL-ALL (4) COUNTY AND SEGREGATE OF SEGREGATE OTHER TOTAL COUNTY AND SEGREGATE OTHER TOTAL COUNTY AND SEGREGATE OTHER TOTAL SEGREGATE OTHER SEGREGATE OTHER TOTAL SEGREGATE OTHER SEGRE	AL PURPOSE REVENUE			289,594,300	285,052,000
SEGREGATE OTHER TOTAL-ALL (3) STATE PROPER (b) School levy of credit (ef) Transfer to composite for stry (q) Lottery and government of supplications GENERAL PUSEGREGATE OTHER TOTAL-ALL (4) COUNTY AND COU	AM REVENUE			69,700,000	69,700,000
OTHER TOTAL-ALL (3) STATE PROPER (b) School levy of credit (ef) Transfer to composite for forestry (q) Lottery and government of supplications GENERAL PUSEGREGATE OTHER TOTAL-ALL (4) COUNTY AND Second district district district taxes (gd) Premier resource district taxes (gg) Local profession district taxes (gg) Local taxes PROGRAM R OTHER TOTAL-ALL (5) PAYMENTS IN INC. (a) Payments for GENERAL PUSE GENERAL PUSE (5) GENERAL PUSE (6) GENERAL PUSE (6) GENERAL PUSE (7) OTHER (8) OTHER (9) OTHER (9) OTHER (10) OTHER (11) OTHER (12) OTHER (13) OTHER (14) OTHER (15) PAYMENTS IN INC. (16) OTHER (17) OTHER (17) OTHER (18) O	ERVICE			(69,700,000)	(69,700,000)
TOTAL-ALL (3) STATE PROPER (b) School levy oredit (ef) Transfer to composite for forestry (q) Lottery and government of applications GENERAL PUSEGREGATE OTHER TOTAL-ALL (4) COUNTY AND to the composite form of the	GATED REVENUE			-0-	-0-
(3) STATE PROPER (b) School levy oredit (ef) Transfer to composite for stry (q) Lottery and government of supplications GENERAL PUSEGREGATE OTHER TOTAL-ALL (4) COUNTY AND Second district district taxes (gd) Premier resoned district taxes (gg) Local profession district taxes (gg) Local taxes PROGRAM R OTHER TOTAL-ALL (5) PAYMENTS IN INCIPATE OF SECOND CONTROL (a) Payments for GENERAL PUSE (5) GENERAL PUSE (6) School levy of credit of second control of second c	OTHER			(-0-)	(-0-
(b) School levy credit (ef) Transfer to c forestry (q) Lottery and g applications GENERAL PU SEGREGATE OTHER TOTAL-ALL (4) County and g (gb) Special district daxes (gc) Local profest district taxes (gg) Local taxes PROGRAM R OTHER TOTAL-ALL (5) PAYMENTS IN IT (a) Payments for	-ALL SOURCES			359,294,300	354,752,000
credit (ef) Transfer to c forestry (q) Lottery and g applications GENERAL PU SEGREGATE OTHER TOTAL-ALL (4) COUNTY AND E (g) County taxes (gb) Special district gd) Premier reso (ge) Local profess district taxes (gg) Local taxes PROGRAM R OTHER TOTAL-ALL (5) PAYMENTS IN IT (a) Payments for GENERAL PU	ROPERTY TAX RELIEF				
credit (ef) Transfer to c forestry (q) Lottery and g applications GENERAL PU SEGREGATE OTHER TOTAL-ALL (4) COUNTY AND E (g) County taxes (gb) Special district gd) Premier reso (ge) Local profess district taxes (gg) Local taxes PROGRAM R OTHER TOTAL-ALL (5) PAYMENTS IN IT (a) Payments for GENERAL PU	levy tax credit and first dollar				
forestry (q) Lottery and g applications GENERAL PU SEGREGATE OTHER TOTAL-ALL (4) COUNTY AND 1 (g) County taxes (gb) Special district (gd) Premier reso (ge) Local profest district taxes (gg) Local taxes PROGRAM R OTHER TOTAL-ALL (5) PAYMENTS IN INTER (6) GENERAL PU GENERAL PU	•	GPR	S	1,001,863,500	1,090,000,000
(q) Lottery and g applications GENERAL PU SEGREGATE: OTHER TOTAL-ALL (4) COUNTY AND 1 (g) County taxes (gb) Special district (gd) Premier reso (ge) Local profess district taxes (gg) Local taxes PROGRAM R OTHER TOTAL-ALL (5) PAYMENTS IN ITOTAL-ALL (5) PAYMENTS IN ITOTAL-ALL (5) PAYMENTS IN ITOTAL-ALL (5) PAYMENTS IN ITOTAL-ALL (5) CENERAL PU	er to conservation fund;				
(s) Lottery and gapplications GENERAL PU SEGREGATE OTHER TOTAL-ALL (4) COUNTY AND (g) County taxes (gb) Special district (gd) Premier reso (ge) Local profess district taxes (gg) Local taxes PROGRAM R OTHER TOTAL-ALL (5) PAYMENTS IN IT (a) Payments for GENERAL PU	у	GPR	S	89,259,600	91,644,000
applications GENERAL PU SEGREGATE OTHER TOTAL-ALL (4) COUNTY AND (g) County taxes (gb) Special distri (gd) Premier reso (ge) Local profest district taxes (gg) Local taxes PROGRAM R OTHER TOTAL-ALL (5) PAYMENTS IN IT (a) Payments for	y and gaming credit	SEG	S	164,640,800	205,360,300
GENERAL PU SEGREGATE OTHER TOTAL-ALL (4) COUNTY AND 1 (g) County taxes (gb) Special district (gd) Premier reso (ge) Local profest district taxes (gg) Local taxes PROGRAM R OTHER TOTAL-ALL (5) PAYMENTS IN 1 (a) Payments for	y and gaming credit; late				
SEGREGATE OTHER TOTAL-ALL (4) COUNTY AND (9) County taxes (gb) Special district gd) Premier reso (ge) Local profest district taxes (gg) Local taxes PROGRAM R OTHER TOTAL-ALL (5) PAYMENTS IN IT (a) Payments for	ations	SEG	S	257,600	257,600
SEGREGATE OTHER TOTAL-ALL (4) COUNTY AND (2) (g) County taxes (gb) Special district ged (gc) Local profest district taxes (gg) Local taxes PROGRAM R OTHER TOTAL-ALL (5) PAYMENTS IN IT (a) Payments for		(3) PRO0	GRAM TO	TALS	
OTHER TOTAL-ALL (4) COUNTY AND (2) (g) County taxes (gb) Special distriction (2) (gc) Local profession distriction (2) (gg) Local taxes PROGRAM R OTHER TOTAL-ALL (5) PAYMENTS IN IT (a) Payments for	AL PURPOSE REVENUE			1,091,123,100	1,181,644,000
(4) COUNTY AND (2) (2) County taxes (2) Special district (2) Premier reso (2) Local profess district taxes (2) Local taxes PROGRAM R OTHER TOTAL-ALL (5) PAYMENTS IN D (a) Payments for GENERAL PU	GATED REVENUE			164,898,400	205,617,900
(4) COUNTY AND (2) (g) County taxes (gb) Special district (gd) Premier reso (ge) Local profess district taxes (gg) Local taxes PROGRAM R OTHER TOTAL-ALL (5) PAYMENTS IN IT (a) Payments for	THER			(164,898,400)	(205,617,900
(g) County taxes (gb) Special district (gd) Premier reso (ge) Local profess district taxes (gg) Local taxes PROGRAM R OTHER TOTAL-ALL (5) PAYMENTS IN IT (a) Payments for	-ALL SOURCES			1,256,021,500	1,387,261,900
(gb) Special district (gd) Premier reso (ge) Local profess district taxes (gg) Local taxes PROGRAM R OTHER TOTAL-ALL (5) PAYMENTS IN IT (a) Payments for GENERAL PU	Y AND LOCAL TAXES				
(gd) Premier reso (ge) Local profess district taxes (gg) Local taxes PROGRAM R OTHER TOTAL-ALL (5) PAYMENTS IN IT (a) Payments for	y taxes	PR	C	-0-	-0-
(ge) Local profess district taxes (gg) Local taxes PROGRAM R OTHER TOTAL-ALL (5) PAYMENTS IN II (a) Payments for GENERAL PU	l district taxes	PR	C	-0-	-0-
district taxes (gg) Local taxes PROGRAM R OTHER TOTAL-ALL (5) PAYMENTS IN II (a) Payments for GENERAL PU	er resort area tax	PR	C	-0-	-0-
district taxes (gg) Local taxes PROGRAM R OTHER TOTAL-ALL (5) PAYMENTS IN I (a) Payments for GENERAL PU	professional football stadium				
PROGRAM R OTHEI TOTAL-ALL (5) PAYMENTS IN I (a) Payments for	taxes	PR	C	-0-	-0-
PROGRAM R OTHER TOTAL-ALL (5) PAYMENTS IN I (a) Payments for GENERAL PU	taxes	PR	C	-0-	-0-
OTHER TOTAL-ALL (5) PAYMENTS IN I (a) Payments for GENERAL PU		(4) PRO	GRAM TO	TALS	
TOTAL-ALL (5) PAYMENTS IN I (a) Payments for GENERAL PU	AM REVENUE			-0-	-0-
(5) PAYMENTS IN I (a) Payments for GENERAL PU	THER			(-0-)	(-0-
(5) PAYMENTS IN I (a) Payments for GENERAL PU	-ALL SOURCES			-0-	-0-
(a) Payments for GENERAL PU	ITS IN LIEU OF TAXES				
	ents for municipal services	GPR	A GRAM TO	18,584,200	18,584,200
	AL PURPOSE REVENUE	(3) 1100	514 11VI 10	18,584,200	18,584,200
IOIAL-ALL				18,584,200	
		0.025 DEDA	DTMENT		18,584,200
CENTED AT DE		0.835 DEPA	AKT WEN I		2 407 620 000
	AL PURPOSE REVENUE AM REVENUE			2,318,127,000 69,700,000	2,487,620,900 69,700,000

2017 W	2017 Wisconsin Act 59		_	2017	2017 Assembly Bill 64		
STATUT	TE, AGENCY AND PURPOSE OTHER SERVICE SEGREGATED REVENUE	Source	Түре	2017–2018 (–0–) (69,700,000) 216,599,000	2018–2019 (-0-) (69,700,000) 250,605,300		
	OTHER TOTAL ALL GOLDGES			(216,599,000)	(250,605,300)		
	TOTAL-ALL SOURCES			2,604,426,000	2,807,926,200		
	Miscellaneous Appropriations						
(1)	CASH MANAGEMENT EXPENSES; INTEREST				0		
(a)	Obligation on operating notes	GPR	S	-0-	-0-		
(b)	Operating note expenses	GPR	S	_0_ 1 125 000	-0- 1 125 000		
(bm)	•	GPR	S	1,125,000	1,125,000		
(c)	Interest payments to program revenue accounts	GPR	S	-0-	-0-		
(4)		UFK	S	-0-	-0-		
(d)	Interest payments to segregated funds	GPR	S	-0-	-0-		
(dm)		OTK	S	V	U		
(um)	government	GPR	S	-0-	-0-		
(e)	Interest on prorated local government	0111	2	Ů	· ·		
(0)	payments	GPR	S	-0-	-0-		
(f)	Payment of fees to financial						
()	institutions	GPR	S	1,500,000	1,500,000		
(gm)	Payment of canceled drafts; program						
	revenues	PR	S	-0-	-0-		
(q)	Redemption of operating notes	SEG	S	-0-	-0-		
(r)	Interest payments to general fund	SEG	S	-0-	-0-		
(rm)	Payment of canceled drafts;						
	segregated revenues	SEG	S	450,000	450,000		
		(1) PRO0	GRAM TO	OTALS			
	GENERAL PURPOSE REVENUE			2,625,000	2,625,000		
	PROGRAM REVENUE			-0-	-0-		
	OTHER			(-0-)	(-0-)		
	SEGREGATED REVENUE			450,000	450,000		
	OTHER			(450,000)	(450,000)		
	TOTAL-ALL SOURCES			3,075,000	3,075,000		
(3)	CAPITOL RENOVATION EXPENSES						
(b)	Capitol restoration and relocation						
	planning	GPR	В	-0-	-0-		
(c)	Historically significant furnishings	GPR	В	-0-	-0-	3 7.4. 1	
(k)	Capitol restoration and relocation	DD C	D	0	0	Vetoed	
	planning; program revenue	PR-S	В	-0-	-0-	In Part	
	CENEDAL DUDDOCE DEVENUE	(3) PROC	GRAM TO		0		
	GENERAL PURPOSE REVENUE			-0-	-0-		
	PROGRAM REVENUE			-0-	-0-		
	SERVICE TOTAL ALL SOURCES			(-0-)	(-0-)		
	TOTAL-ALL SOURCES	· · · ·		-0-	-0-		
(4)	TAX, ASSISTANCE AND TRANSFER PAYMEN		C	1 000 000	1 000 000		
(a)	Interest on overpayment of taxes	GPR	S	1,000,000	1,000,000		
(am)	Great Lakes protection fund contribution	GPR	C	Λ	-0-		
(ha)	Study of engineering	GPR GPR		-0- -0-	-0- -0-		
(be) (bm)		GPR GPR	A S	4,341,700	4,352,100		
(c)	Minnesota income tax reciprocity	GPR	S S	4,341,700 -0-	4,552,100 -0-		
(6)	winnesota meome tax reciprocity	OFK	S	-0-	-0-		

STATUTE, ACENCY AND PURPOSE (Ca) Minnesota income tax reciprocity bench mark (Circ) Illinois income tax reciprocity (Circ) Illinois income tax reciprocity, 1998 and 1999 (Circ) Transfer to local exposition district (dr) Transfer to local exposition district (dr) Transfer to local exposition district (dr) Transfer to environmental fund; environmental management SEG S 1,892,100 1,904,0	2017 A	2017 Assembly Bill 64		_	2017 Wisconsin Act 59		
beach mark GPR A	STATUTE, AGENCY AND PURPOSE		Source	Түре	2017-2018	2018-2019	
(cm) Illinois income tax reciprocity bench mark GPR S 64,000,000 90,500,000 (co) Illinois income tax reciprocity, 1998 and 1999 GPR A -0- -0- (cr) Transfer to local exposition district (dr) GPR A 4,000,000 4,000,000 (dr) Transfer to local exposition district (dr) GPR A 4,000,000 4,000,000 (dr) Transfer to local exposition district (dr) GPR A 4,000,000 4,000,000 (f) Transfer to local exposition district nonpoint sources GPR A 4,000,000 4,000,000 (f) Transfer to terasportation fund; morpoint sources GPR A 7,991,100 7,991,100 (f) Also for certain local purchases and projects GPR A -0- -0- (f) Transfer to transportation fund; macking active transportation fund; disaster damage aids GPR S -0- -0- -0- (g) GPR S -0- -0- -0- -0- -0- -0- -0- -0-<	(ca)						
Cen Illinois income tax reciprocity bench mark GPR							
mark			GPR	S	64,000,000	90,500,000	
CCO	(cn)	± •	GDD.		0	0	
and 1999 GPR A -000- (cr) Transfer to local exposition district GPR A 4,000,000 4,000,000 (dr) Transfer to local exposition district GPR A 4,000,000 4,000,000 (e) Transfer to conservation fund; land acquisition reimbursement GPR S 200 200 (1) Transfer to environmental fund; nonpoint sources GPR A 7,991,100 7,991,100 (ii) Transfer to environmental fund; nonpoint sources GPR A 7,991,100 7,991,100 (iii) Transfer to transportation fund; hub facility exemptions GPR S -00- (iii) Transfer to transportation fund; disaster damage aids GPR S -0- 3,238,200 (gd) American Red Cross, Badger Chapter PR C -00- (iii) Transfer to transportation fund; disaster damage aids GPR S -0- 3,238,200 (gd) American Red Cross, Badger Chapter PR C -00- (iii) Transfer to transportation fund; disaster damage aids GPR S -0- 3,238,200 (iii) Transfer to conservation fund; SEG S 1,996,000 1,906,000 (iv) Petroleum allowance SEG S 1,996,000 1,906,000 (iv) Petroleum allowance SEG S 1,996,000 1,906,000 (iv) Transfer to conservation fund; SEG S 1,995,100 13,140,000 (iv) Transfer to conservation fund; all-terrain vehicle formula SEG S 1,892,100 1,931,000 (iv) Transfer to conservation fund; petroleum inspection fund SEG S 3,336,700 351,500 (iv) Transfer to transportation fund; environmental fund; environmental management SEG A 1,704,800 1,704,800 (iv) Transfer to transportation fund; environmental management SEG A 1,704,800 1,704,800 (iv) Transfer to transportation fund; environmental management SEG A 1,704,800 1,704,800 (iv) PROGRAM TOTALS (iv) PRO			GPR	A	-0-	-0-	
Cer	(co)		CDD	A	0	0	
(dr) Transfer to local exposition district GPR A 4,000,000 4,000,000 (e) Transfer to conservation fund; land acquisition reimbursement and acquisition reimbursement GPR S 200 200 (f) Transfer to tenvironmental fund; nonpoint sources GPR A 7,991,100 7,991,100 (fi) Transfer to tenvironmental fund; nonpoint sources GPR A 7,991,100 7,991,100 (fi) Transfer to tenvironmental fund; nonpoint sources GPR A 7,991,100 7,991,100 (fi) Transfer to tenvironmental fund; nonpoint sources GPR A 7,991,100 7,991,100 (fi) Transfer to transportation fund; nonpoint sources GPR A 7,991,100 7,991,100 (fi) Transfer to transportation fund; nonpoint sources GPR A 7,991,100 7,991,100 (gr) Transfer to transportation fund; nonpoint sources GPR C 0 0 0 0 (gr) Transfer to transportation fund; nonpoint sources SEG S 1,906,000 1,906,000 </td <td>(or)</td> <td></td> <td></td> <td></td> <td></td> <td></td>	(or)						
(e) Transfer to conservation fund; land acquisition reimbursement (f) Transfer to environmental fund; nonpoint sources (GPR A 7,991,100 7,991,100 (Fc) Aids for certain local purchases and projects (GPR A 7,991,100 Family (GPR A) 7,991,100 (Fc) Aids for certain local purchases and projects (GPR A) 7,991,100 (Fc) (Fc) Transfer to transportation fund; that facility exemptions (GPR A) 7,991,100 (Fc) (Fc) (Fc) (Fc) (Fc) (Fc) (Fc) (Fc)		-					
Column		-	OFK	A	4,000,000	4,000,000	
(f) Transfer to environmental fund; nonpoint sources (fc) Aids for certain local purchases and projects (fm) Transfer to transportation fund; hub facility exemptions (gPR S -00-6) (fm) Transfer to transportation fund; hub facility exemptions (gPR S -00-7-6) (gPR S -00-6) (gPR S -00-6) (gPR S -00-6) (gPR S -00-7-6) (gPR S -000-6) (gPR S -00	(6)		GPR	S	200	200	
nonpoint sources	(f)		OTT	S	200	200	
CFC Aids for certain local purchases and projects GPR A -0- -0-	(1)		GPR	A	7,991,100	7,991,100	
projects	(fc)				, ,		
Facility exemptions			GPR	A	-0-	-0-	
(fr) Transfer to transportation fund; disaster damage aids GPR S -O- 3,238,200 (gd) American Red Cross, Badger Chapter PR C -O- -O- (ge) Feeding America; Second Harvest food banks PR C -O-	(fm)						
disaster damage aids		•	GPR	S	-0-	-0-	
Gd	(fr)						
Chapter		_	GPR	S	-0-	3,238,200	
(ge) Feeding America; Second Harvest food banks PR C -0- (h) Volkswagen Settlement Distributions PR C 21,000,000 21,000,000 (q) Terminal tax distribution SEG S 1,906,000 1,906,000 (r) Petroleum allowance SEG S 300,000 300,000 (s) Transfer to conservation fund; motorboat formula SEG S 12,950,100 13,140,000 (t) Transfer to conservation fund; snowmobile formula SEG S 4,858,700 4,670,000 (u) Transfer to conservation fund; all-terrain vehicle formula SEG S 1,892,100 1,931,000 (v) Transfer to conservation fund; utility terrain vehicle formula SEG S 336,700 351,500 (w) Transfer to transportation fund; petroleum inspection fund SEG A 6,258,500 6,258,500 (wc) Petroleum inspection fund supplement to environmental fund; environmental management SEG A 1,704,800 1,704,800 PROGRAM REVENUE 85,333,000 <td< td=""><td>(gd)</td><td>•</td><td>DD</td><td>a</td><td>0</td><td>0</td></td<>	(gd)	•	DD	a	0	0	
Food banks	()		PK	C	-0-	-0-	
(h) Volkswagen Settlement Distributions PR C 21,000,000 21,000,000 (q) Terminal tax distribution SEG S 1,906,000 1,906,000 (r) Petroleum allowance SEG S 300,000 300,000 (s) Transfer to conservation fund; motorboat formula SEG S 12,950,100 13,140,000 (t) Transfer to conservation fund; snowmobile formula SEG S 4,858,700 4,670,000 (u) Transfer to conservation fund; all-terrain vehicle formula SEG S 1,892,100 1,931,000 (v) Transfer to conservation fund; terrain vehicle formula SEG S 336,700 351,500 (w) Transfer to transportation fund; petroleum inspection fund SEG A 6,258,500 6,258,500 (wc) Petroleum inspection fund SEG A 1,704,800 1,704,800 was puplement to environmental fund; environmental management SEG A 1,704,800 115,081,600 PROGRAM REVENUE SEG A 1,704,800	(ge)		DD	C	0	0	
(q) Terminal tax distribution SEG S 1,906,000 1,906,000 (r) Petroleum allowance SEG S 300,000 300,000 (s) Transfer to conservation fund; motorboat formula SEG S 12,950,100 13,140,000 (t) Transfer to conservation fund; snowmobile formula SEG S 4,858,700 4,670,000 (u) Transfer to conservation fund; all-terrain vehicle formula SEG S 1,892,100 1,931,000 (v) Transfer to conservation fund; utility terrain vehicle formula SEG S 336,700 351,500 (w) Transfer to transportation fund; petroleum inspection fund SEG A 6,258,500 6,258,500 (wc) Petroleum inspection fund supplement to environmental fund; environmental management SEG A 1,704,800 1,704,800 FORGRAM REVENUE 85,333,000 115,081,600 17,04,800 17,04,800 12,000,000 12,000,000 12,000,000 12,000,000 12,000,000 12,000,000 12,000,000 12,000,000 12,000,000 12,000,000 <td>(h)</td> <td></td> <td></td> <td></td> <td></td> <td></td>	(h)						
(r) Petroleum allowance SEG S 300,000 300,000 (s) Transfer to conservation fund; motorboat formula SEG S 12,950,100 13,140,000 (t) Transfer to conservation fund; snowmobile formula SEG S 4,858,700 4,670,000 (u) Transfer to conservation fund; all-terrain vehicle formula SEG S 1,892,100 1,931,000 (v) Transfer to conservation fund; utility terrain vehicle formula SEG S 336,700 351,500 (w) Transfer to transportation fund; petroleum inspection fund SEG A 6,258,500 6,258,500 (wc) Petroleum inspection fund supplement to environmental fund; environmental management SEG A 1,704,800 1,704,800 (wc) PROGRAM REVENUE 85,333,000 115,081,600 17,04,800 PROGRAM REVENUE 21,000,000 21,000,000 21,000,000 21,000,000 OTHER 30,206,900 30,261,800 30,261,800 TOTAL-ALL SOURCES 136,539,900 166,343,400 (5) STATE							
(s) Transfer to conservation fund; motorboat formula SEG S 12,950,100 13,140,000 (t) Transfer to conservation fund; snowmobile formula SEG S 4,858,700 4,670,000 (u) Transfer to conservation fund; all-terrain vehicle formula SEG S 1,892,100 1,931,000 (v) Transfer to conservation fund; utility terrain vehicle formula SEG S 336,700 351,500 (w) Transfer to transportation fund; petroleum inspection fund SEG A 6,258,500 6,258,500 (wc) Petroleum inspection fund SEG A 1,704,800 1,704,800 environmental management SEG A 1,704,800 1,704,800 (4) PROGRAM TOTALS GENERAL PURPOSE REVENUE 85,333,000 115,081,600 PROGRAM REVENUE 21,000,000 21,000,000 OTHER (21,000,000) (21,000,000) SEGREGATED REVENUE 30,206,900 30,261,800 OTHER (30,206,900) (30,261,800) TOTAL-ALL SOURCES 136,539,900 166,343,400 (5) STATE HOUSING AUTHORITY RESERVE FUND (a) Enhancement of credit of authority debt GPR A -00- (5) PROGRAM TOTALS GENERAL PURPOSE REVENUE -00-							
motorboat formula			DEG	S	300,000	300,000	
(t) Transfer to conservation fund; snowmobile formula SEG S 4,858,700 4,670,000 (u) Transfer to conservation fund; all-terrain vehicle formula SEG S 1,892,100 1,931,000 (v) Transfer to conservation fund; utility terrain vehicle formula SEG S 336,700 351,500 (w) Transfer to transportation fund; petroleum inspection fund SEG A 6,258,500 6,258,500 (wc) Petroleum inspection fund supplement to environmental fund; environmental management SEG A 1,704,800 1,704,800 PROGRAM REVENUE 85,333,000 115,081,600 115,081,600 115,081,600 PROGRAM REVENUE 21,000,000 21,000,000 21,000,000 21,000,000 OTHER (21,000,000) (21,000,000) 30,261,800 OTHER (30,206,900) 30,261,800 TOTAL-ALL SOURCES 136,539,900 166,343,400 (5) STATE HOUSING AUTHORITY RESERVE FUND 4 Enhancement of credit of authority debt GPR A -0- -0- (5) PROGRAM T	(5)		SEG	S	12,950,100	13,140,000	
Simple S	(t)				, ,	, ,	
All-terrain vehicle formula SEG S 1,892,100 1,931,000			SEG	S	4,858,700	4,670,000	
(v) Transfer to conservation fund; utility terrain vehicle formula SEG S 336,700 351,500 (w) Transfer to transportation fund; petroleum inspection fund supplement to environmental fund; environmental management SEG A 6,258,500 6,258,500 (wc) Petroleum inspection fund supplement to environmental fund; environmental management SEG A 1,704,800 1,704,800 GENERAL PURPOSE REVENUE 85,333,000 115,081,600 PROGRAM REVENUE 21,000,000 21,000,000 OTHER (21,000,000) (21,000,000) (21,000,000) (21,000,000) SEGREGATED REVENUE 30,206,900 30,261,800 OTHER (30,206,900) (30,261,800) TOTAL-ALL SOURCES 136,539,900 166,343,400 (5) STATE HOUSING AUTHORITY RESERVE FUND (a) Enhancement of credit of authority debt GPR A -0- -0- GENERAL PURPOSE REVENUE -0- -0- -0-	(u)	Transfer to conservation fund;					
terrain vehicle formula SEG S 336,700 351,500 (w) Transfer to transportation fund; petroleum inspection fund supplement to environmental fund; environmental management SEG A 1,704,800 1,704,800 (WC) Petroleum inspection fund supplement to environmental fund; environmental management SEG A 1,704,800 1,704,800 (4) PROGRAM TOTALS GENERAL PURPOSE REVENUE 85,333,000 115,081,600 PROGRAM REVENUE 21,000,000 21,000,000 OTHER (21,000,000) (21,000,000) SEGREGATED REVENUE 30,206,900 30,261,800 OTHER (30,206,900) (30,261,800) TOTAL—ALL SOURCES 136,539,900 166,343,400 (5) STATE HOUSING AUTHORITY RESERVE FUND (a) Enhancement of credit of authority debt GPR A —0— —0— (5) PROGRAM TOTALS GENERAL PURPOSE REVENUE —0— —0—			SEG	S	1,892,100	1,931,000	
(w) Transfer to transportation fund; petroleum inspection fund SEG A 6,258,500 6,258,500 (wc) Petroleum inspection fund supplement to environmental fund; environmental management SEG A 1,704,800 1,704,800 GENERAL PURPOSE REVENUE 85,333,000 115,081,600 PROGRAM REVENUE 21,000,000 21,000,000 OTHER (21,000,000) (21,000,000) SEGREGATED REVENUE 30,206,900 30,261,800 OTHER (30,206,900) (30,261,800) TOTAL-ALL SOURCES 136,539,900 166,343,400 (5) STATE HOUSING AUTHORITY RESERVE FUND GPR A -0- -0- (a) Enhancement of credit of authority debt GPR A -0- -0- (5) PROGRAM TOTALS GENERAL PURPOSE REVENUE -0- -0-	(v)	•	~~~	~	22 (200	27. 700	
petroleum inspection fund SEG A 6,258,500 6,258,500		terrain vernere rormana	SEG	S	336,700	351,500	
(wc) Petroleum inspection fund supplement to environmental fund; environmental management SEG A 1,704,800	(w)		CEC		(250 500	(250 500	
SEG A 1,704,800 1,704,800 1,704,800 (4) PROGRAM TOTALS	()		SEG	Α	0,238,300	0,238,300	
environmental management SEG A 1,704,800 1,704,800	(wc)						
(4) PROGRAM TOTALS GENERAL PURPOSE REVENUE			SEG	Α	1.704.800	1.704.800	
GENERAL PURPOSE REVENUE 85,333,000 115,081,600 PROGRAM REVENUE 21,000,000 21,000,000 OTHER (21,000,000) (21,000,000) SEGREGATED REVENUE 30,206,900 30,261,800 OTHER (30,206,900) (30,261,800) TOTAL-ALL SOURCES 136,539,900 166,343,400 (5) STATE HOUSING AUTHORITY RESERVE FUND (a) Enhancement of credit of authority debt GPR A -00- (5) PROGRAM TOTALS GENERAL PURPOSE REVENUE -00-		en vironmentar management				1,701,000	
PROGRAM REVENUE 21,000,000 21,000,000 OTHER (21,000,000) (21,000,000) SEGREGATED REVENUE 30,206,900 30,261,800 OTHER (30,206,900) (30,261,800) TOTAL-ALL SOURCES 136,539,900 166,343,400 (5) STATE HOUSING AUTHORITY RESERVE FUND Enhancement of credit of authority debt GPR A -00- (5) PROGRAM TOTALS GENERAL PURPOSE REVENUE -00-		GENERAL PURPOSE REVENUE				115,081,600	
OTHER SEGREGATED REVENUE OTHER							
OTHER TOTAL-ALL SOURCES 136,539,900 166,343,400 (5) STATE HOUSING AUTHORITY RESERVE FUND (a) Enhancement of credit of authority debt GPR A -00- (5) PROGRAM TOTALS GENERAL PURPOSE REVENUE -00-		OTHER			(21,000,000)	(21,000,000)	
TOTAL-ALL SOURCES (5) STATE HOUSING AUTHORITY RESERVE FUND (a) Enhancement of credit of authority debt (b) GPR A -00- (5) PROGRAM TOTALS GENERAL PURPOSE REVENUE GENERAL PURPOSE REVENUE 136,539,900 166,343,400 166,343,400 166,343,400 166,343,400		SEGREGATED REVENUE			30,206,900	30,261,800	
(5) STATE HOUSING AUTHORITY RESERVE FUND (a) Enhancement of credit of authority debt GPR A -00- (5) PROGRAM TOTALS GENERAL PURPOSE REVENUE -00-		OTHER			(30,206,900)	(30,261,800)	
(a) Enhancement of credit of authority debt GPR A -00- (5) PROGRAM TOTALS GENERAL PURPOSE REVENUE -00-					136,539,900	166,343,400	
debt GPR A -0- -0- (5) PROGRAM TOTALS -0- -0- GENERAL PURPOSE REVENUE -0- -0-	(5)	STATE HOUSING AUTHORITY RESERVE FUN	ND				
(5) PROGRAM TOTALS GENERAL PURPOSE REVENUE -00-	(a)						
GENERAL PURPOSE REVENUE -00-		debt				-0-	
		GENERAL PURPOSE PROPERTY.	(5) PROC	jRAM TO		_	
101AL-ALL SOURCES -00-						-0-	
		TOTAL-ALL SOURCES			-0-	-0-	

2017 Wisconsin Act 59		- 124	_	2017 A	2017 Assembly Bill 64	
STATU'	TE, AGENCY AND PURPOSE	Source	Түре	2017-2018	2018-2019	
(6)	MISCELLANEOUS RECEIPTS					
(g)	Gifts and grants	PR	C	-0-	-0-	
(h)	Vehicle and aircraft receipts	PR	A	-0-	-0-	
(i)	Miscellaneous program revenue	PR	A	-0-	-0-	
(j)	Custody accounts	PR	C	-0-	-0-	
(k)	Aids to individuals and organizations	PR-S	C	-0-	-0-	
(ka)	Local assistance	PR-S	C	-0-	-0-	
(m)	Federal aid	PR-F	C	-0-	-0-	
(pz)	Indirect cost reimbursements	PR-F	C	-0-	-0-	
		(6) PRO0	GRAM T	OTALS		
	PROGRAM REVENUE			-0-	-0-	
	FEDERAL			(-0-)	(-0-)	
	OTHER			(-0-)	(-0-)	
	SERVICE			(-0-)	(-0-)	
	TOTAL-ALL SOURCES			-0-	-0-	
(8)	Marquette University					
(a)	Dental clinic and education facility; principal repayment, interest and					
	rebates	GPR	S	2,272,600	2,281,200	
		(8) PROC	GRAM T			
	GENERAL PURPOSE REVENUE			2,272,600	2,281,200	
	TOTAL-ALL SOURCES			2,272,600	2,281,200	
(9) (a)	State capitol renovation and restor South wing renovation and	RATION				
	restoration	GPR	C	-0-	-0-	
		(9) PRO	GRAM T			
	GENERAL PURPOSE REVENUE			-0-	-0-	
	TOTAL-ALL SOURCES	000 DEDA		-0-	-0-	
).833 DEPA	KIMEN	T TOTALS	110 007 000	
	GENERAL PURPOSE REVENUE PROGRAM REVENUE			90,230,600 21,000,000	119,987,800 21,000,000	
	FEDERAL			(-0-)	(-0-)	
	OTHER			(21,000,000)	(21,000,000)	
	SERVICE			(21,000,000)	(21,000,000)	
	SEGREGATED REVENUE			30,656,900	30,711,800	
	OTHER			(30,656,900)	(30,711,800)	
	TOTAL-ALL SOURCES			141,887,500	171,699,600	
20.865	Frogram Supplements			141,007,500	171,022,000	
(1)	EMPLOYEE COMPENSATION AND SUPPORT					
(a)	Judgments and legal expenses	GPR	S	-0-	-0-	
(c)	Compensation and related adjustments	GPR	S	-0-	-0-	
(ci)	University pay adjustments	GPR	S	-0-	_0_ _0_	
(cj)	Pay adjustments for certain	OTK	5	V	O	
(cj)	university employees	GPR	A	-0-	-0-	
(d)	Employer fringe benefit costs	GPR	S	-0-	-0-	
(dm)				-	-	
()	program	GPR	A	6,000,000	6,000,000	
(e)	Additional biweekly payroll	GPR	A	-0-	-0-	
(em)		GPR	A	-0-	-0-	
(fm)	Risk management	GPR	A	-0-	-0-	

17 Assembly Bill 64		- 125 -		2017 Wi	2017 Wisconsin Act 59	
TATUT	TE, AGENCY AND PURPOSE	Source	Түре	2017-2018	2018-2019	
(fn)	Physically handicapped supplements	GPR	A	5,800	5,800	
(g)	Judgments and legal expenses;			,	,	
	program revenues	PR	S	-0-	-0-	
(i)	Compensation and related					
	adjustments; program revenues	PR	S	-0-	-0-	
(ic)	University pay adjustments	PR	S	-0-	-0-	
(j)	Employer fringe benefit costs;					
	program revenues	PR	S	-0-	-0-	
(jm)	Additional biweekly payroll;		_	_	_	
	nonfederal program revenues	PR	S	-0-	-0-	
(js)	Financial and procurement services;	D D	~	0	0	
	program revenues	PR	S	-0-	-0-	
(kr)	Risk management; program revenues	PR	S	-0-	-0-	
(Ln)	Physically handicapped supplements;	D D	~	0	0	
	program revenues	PR	S	-0-	-0-	
(m)	Additional biweekly payroll; federal	DD E	C	0	0	
()	program revenues	PR–F	S	-0-	-0-	
(q)	Judgments and legal expenses;	CEC	C	0	0	
(-)	segregated revenues	SEG	S	-0-	-0-	
(s)	Compensation and related adjustments; segregated revenues	SEG	S	-0-	-0-	
(si)	University pay adjustments	SEG	S	_0_ _0_	_0_ _0_	
	Employer fringe benefit costs;	SEC	S	-0-	-0-	
(t)	segregated revenues	SEG	S	-0-	-0-	
(tm)	Additional biweekly payroll;	SEC	3	_0_	-0-	
(1111)	nonfederal segregated revenues	SEG	S	-0-	-0-	
(ts)	Financial and procurement services;	520	J		· ·	
()	segregated revenues	SEG	S	-0-	-0-	
(ur)	Risk management; segregated					
` /	revenues	SEG	S	-0-	-0-	
(vn)	Physically handicapped supplements;					
	segregated revenues	SEG	S	-0-	-0-	
(x)	Additional biweekly payroll; federal					
	segregated revenues	SEG-F	S	-0-	-0-	
		(1) PROC	GRAM TO	OTALS		
	GENERAL PURPOSE REVENUE			6,005,800	6,005,800	
	PROGRAM REVENUE			-0-	-0-	
	FEDERAL			(-0-)	(-0-)	
	OTHER			(-0-)	(-0-)	
	SEGREGATED REVENUE			-0-	-0-	
	FEDERAL			(-0-)	(-0-)	
	OTHER			(-0-)	(-0-)	
	TOTAL-ALL SOURCES			6,005,800	6,005,800	
2)	STATE PROGRAMS AND FACILITIES					
(a)	Private facility rental increases	GPR	A	-0-	-0-	
(ag)	State-owned office rent supplement	GPR	A	-0-	-0-	
(am)	Space management	GPR	A	-0-	-0-	
(d)	State deposit fund	GPR	S	-0-	-0-	
(e)	Maintenance of capitol and executive					
	residence	GPR	A	4,508,900	4,508,900	
	Expansion and damps from inhings					
(eb)	Executive residence furnishings replacement	GPR	C	10,200	10,200	

7 Wi	sconsin Act 59	- 126	_	2017 A	ssembly Bill 64	
TATUT	E, AGENCY AND PURPOSE	Source	Түре	2017-2018	2018-2019	
(em)	Groundwater survey and analysis	GPR	A	182,500	182,500	
(g)	Private facility rental increases;					
	program revenues	PR	S	-0-	-0-	
(gg)	State-owned office rent supplement;					
	program revenues	PR	S	-0-	-0-	
(gm)	Space management; program					
	revenues	PR	S	-0-	-0-	
(i)	Enterprise resource planning system;	D.D.	G.	0	0	
<i>(</i> *)	program revenues	PR	S	-0-	-0-	
(j)	State deposit fund; program revenues	PR	S	-0-	-0-	
(L)	Data processing and					
	telecommunications study; program revenues	PR	S	-0-	-0-	
(q)	Private facility rental increases;	1 K	3	_0_	-0-	
(Ψ)	segregated revenues	SEG	S	-0-	-0-	
(qg)	State—owned office rent supplement;	SEC	S	O	Ü	
(15)	segregated revenues	SEG	S	-0-	-0-	
(qm)	Space management; segregated		·-	-	Ž	
(1)	revenues	SEG	S	-0-	-0-	
(r)	Enterprise resource planning system;					
	segregated revenues	SEG	S	-0-	-0-	
(t)	State deposit fund; segregated					
	revenues	SEG	S	-0-	-0-	
		(2) PRO	GRAM TOTA			
	GENERAL PURPOSE REVENUE			4,701,600	4,701,600	
]	PROGRAM REVENUE			-0-	-0-	
	OTHER			(-0-)	(-0-)	
,	SEGREGATED REVENUE			-0-	-0-	
_	OTHER			(-0-)	(-0-)	
	TOTAL-ALL SOURCES			4,701,600	4,701,600	
3)	TAXES AND SPECIAL CHARGES	CDD	G	0	0	
(a)	Property taxes	GPR	S	-0-	-0-	
(g)	Property taxes; program revenues	PR	S	-0-	-0-	
(i)	Payments for municipal services;	DD	C	0	0	
(a)	program revenues	PR SEG	S	-0-	-0-	
(q)	Property taxes; segregated revenues Payments for municipal services;	SEG	S	-0-	-0-	
(s)	segregated revenues	SEG	S	-0-	-0-	
	segregated revenues		GRAM TOTA		-0-	
	GENERAL PURPOSE REVENUE	(3) 1 KO	OM 1171 1011	-0-	-0-	
	PROGRAM REVENUE			-0-	-0-	
,	OTHER			(-0-)	(-0-)	
	SEGREGATED REVENUE			-0-	-0-	
,	OTHER			(-0-)	(-0-)	
,	TOTAL-ALL SOURCES			-0-	-0-	
(-	JOINT COMMITTEE ON FINANCE SUPPLEME	ENTAL APPR	OPRIATIONS	~	Ž	
(a)	General purpose revenue funds			16,879,600		Vetoe
(4)	general program supplementation	GPR	В	16,329,600	21,740,900	In Par
(g)	Program revenue funds general				·,···•,> · ·	Vetoe
10/	program supplementation	PR	S	3,445,500	2,200,000	In Par
(k)	Public assistance programs				. ,	
	supplementation	PR-S	C	-0-	-0-	

017 A	ssembly Bill 64	- 127	_	2017 W	isconsin Act 59
STATUTE, AGENCY AND PURPOSE		Source Type		2017-2018	2018-2019
(m)	Federal funds general program supplementation	PR-F	C	-0-	-0-
(u)	Segregated funds general program supplementation	SEG	S	-0-	-0-
	supplementation		GRAM TOT	-	O
	GENERAL PURPOSE REVENUE			16,879,600	21,740,900
	PROGRAM REVENUE			3,445,500	2,200,000
	FEDERAL			(-0-)	(-0-)
	OTHER			(3,445,500)	(2,200,000)
	SERVICE			(-0-)	(-0-)
	SEGREGATED REVENUE			-0-	-0-
	OTHER			(-0-)	(-0-)
	TOTAL-ALL SOURCES			20,325,100	23,940,900
(8) (g)	Supplementation of program revenue and program revenue—service	;			
	appropriations	PR	S	-0-	-0-
	DD OCD AM DEVENIE	(8) PRO	GRAM TOT		0
	PROGRAM REVENUE			-0-	-0-
	OTHER TOTAL-ALL SOURCES			(-0-) -0-	(-0-) -0-
		0 865 DEP	ARTMENT T		-0-
	GENERAL PURPOSE REVENUE	.0.003 DLIF	AKTIVILIVI I	27,587,000	32,448,300
	PROGRAM REVENUE			3,445,500	2,200,000
	FEDERAL			(-0-)	(-0-)
	OTHER			(3,445,500)	(2,200,000)
	SERVICE			(-0-)	(-0-)
	SEGREGATED REVENUE			-0-	-0-
	FEDERAL			(-0-)	(-0-)
	OTHER			(-0-)	(-0-)
	TOTAL-ALL SOURCES			31,032,500	34,648,300
20.866	6 Public Debt				
(1)	BOND SECURITY AND REDEMPTION FUND)			
(u)	Principal repayment and interest	SEG	S	-0-	-0-
		(1) PRO	GRAM TOT		
	SEGREGATED REVENUE			-0-	-0-
	OTHER			(-0-)	(-0-)
	TOTAL-ALL SOURCES	0000 DED		-0-	-0-
		20.866 DEPA	ARTMENT T		0
	SEGREGATED REVENUE			-0-	-0-
	OTHER TOTAL ALL SOURCES			(-0-)	(-0-)
20 86	TOTAL-ALL SOURCES			-0-	-0-
(1)	7 Building Commission STATE OFFICE BUILDINGS				
(a)	Principal repayment and interest;				
(a)	housing of state agencies	GPR	S	-0-	-0-
(b)	Principal repayment and interest;	OTK	S	O .	O .
(0)	capitol and executive residence	GPR	S	10,717,900	9,479,600
		(1) PP()	GRAM TOT	ZIAT	
	GENERAL PURPOSE REVENUE	(1) PRO	GRAM TOT	TALS 10,717,900	9,479,600

2017 W	2017 Wisconsin Act 59		– 128 –		2017 Assembly Bill 64	
STATU	TE, AGENCY AND PURPOSE	Source	Түре	2017-2018	2018-2019	
(2)	ALL STATE-OWNED FACILITIES					
(b)	Asbestos removal	GPR	A	-0-	-0-	
(c)	Hazardous materials removal	GPR	A	-0-	-0-	
(f)	Facilities preventive maintenance	GPR	A	-0-	-0-	
(q)	Building trust fund	SEG	C	-0-	-0-	
(r)	Planning and design	SEG	C	-0-	-0-	
(u)	Aids for buildings	SEG	C	-0-	-0-	
(v)	Building program funding contingency	SEG	C	-0-	-0-	
(w)	Building program funding	SEG	C	-0-	-0-	
		(2) PROC	GRAM 7	ΓOTALS		
	GENERAL PURPOSE REVENUE			-0-	-0-	
	SEGREGATED REVENUE			-0-	-0-	
	OTHER			(-0-)	(-0-)	
	TOTAL-ALL SOURCES			-0-	-0-	
(3)	STATE BUILDING PROGRAM					
(a)	Principal repayment and interest	GPR	S	9,365,300	20,187,700	
(b)	Principal repayment and interest	GPR	S	1,643,300	1,434,000	
(bb)	Principal repayment, interest and rebates; AIDS Network, Inc.	GPR	S	23,900	23,900	
(bc)	Principal repayment, interest and rebates; Grand Opera House in Oshkosh	GPR	S	41,400	42,100	
(bd)	Principal repayment, interest and rebates; Aldo Leopold climate change classroom and interactive laboratory	GPR	S	37,900	37,900	
(be)	Principal repayment, interest and rebates; Bradley Center Sports and Entertainment Corporation	GPR	S	3,245,300	3,193,800	
(bf)	Principal repayment, interest and rebates; AIDS Resource Center of Wisconsin, Inc.	GPR	S	63,700	63,700	
(bg)	Principal repayment, interest, and rebates; Madison Children's Museum	GPR	S	20,000	19,900	
(bh)	Principal repayment, interest, and rebates; Myrick Hixon EcoPark, Inc.	GPR	S	40,000	40,000	
(bj)	Principal repayment, interest and rebates: Lac du Flambeau Indian Tribal Cultural Center	GPR	S	19,000	18,800	
(bL)	Principal repayment, interest and rebates; family justice center	GPR	S	698,100	814,000	
(bm)	Principal repayment, interest, and rebates; HR Academy, Inc.	GPR	S	136,400	136,900	
(bn)	Principal repayment, interest and rebates; Hmong cultural center	GPR	S	21,400	16,800	
(bq)	Principal repayment, interest and rebates; children's research institute	GPR	S	969,500	930,100	
(br)	Principal repayment, interest and rebates	GPR	S	90,300	88,900	

17 As	ssembly Bill 64	– 129 –		2017 Wisconsin Act 59	
TATUT	E, AGENCY AND PURPOSE	Source	Түре	2017-2018	2018-2019
(bt)	Principal repayment, interest, and rebates; Wisconsin Agriculture Education Center, Inc.	GPR	S	62,500	263,200
(bu)	Principal repayment, interest and rebates; Civil War exhibit at the Kenosha Public Museums	GPR	S	42,400	33,000
(bv)	Principal repayment, interest, and rebates; Bond Health Center	GPR	S	73,100	74,200
(bw)	Principal repayment, interest, and rebates; Eau Claire Confluence Arts,	CDD	C.	125 000	027.500
(bx)	Inc. Principal repayment, interest, and rebates; Carroll University	GPR GPR	S S	125,000 161,200	927,500 179,200
(cb)	Principal repayment, interest, and rebates; Domestic Abuse	Ork	5	101,200	175,200
(cd)	Intervention Services, Inc. Principal repayment, interest and	GPR	S	41,000	40,700
(cf)	rebates; K I Convention Center Principal repayment, interest and rebates; Dane County; livestock	GPR	S	138,700	137,600
(ch)	facilities Principal repayment, interest, and rebates; Wisconsin Maritime Center	GPR	S	681,400	675,300
(cj)	of Excellence Principal repayment, interest, and rebates; Norskedalen Nature and	GPR	S	185,600	557,600
(cq)	Heritage Center Principal repayment, interest, and	GPR	S	-0-	84,200
(cr)	rebates; La Crosse Center Principal repayment, interest, and rebates; St. Ann Center for	GPR	S	-0-	-0-
	Intergenerational Care, Inc; Bucyrus Campus	GPR	S	-0-	-0-
(cs)	Principal repayment, interest, and rebates; Brown County innovation center	GPR	S	-0-	-0-
(d)	Interest rebates on obligation proceeds; general fund	GPR	S	-0-	-0-
(e)	Principal repayment, interest and rebates; parking ramp	GPR	S	-0-	-0-
(g)	Principal repayment, interest and rebates; program revenues	PR	S	-0-	-0-
(h)	Principal repayment, interest, and rebates	PR	S	-0-	-0-
(i)	Principal repayment, interest and rebates; capital equipment	PR	S	-0-	-0-
(k) (kd)	Interest rebates on obligation proceeds; program revenues Energy conservation construction	PR-S	C	-0-	-0-
(KU)	projects; principal repayment, interest and rebates	PR-S	С	1,810,300	2,660,000
(km)	Aquaculture demonstration facility; principal repayment and interest	PR-S	S	271,900	265,800

2017 Wisconsin Act 59		- 130 ·	_	2017 A	2017 Assembly Bill 64	
STATE	TTE, AGENCY AND PURPOSE	Source	Түре	2017-2018	2018-2019	
(q)	Principal repayment and interest;	2001101				
(1)	segregated revenues	SEG	S	-0-	-0-	
(r)	Interest rebates on obligation					
	proceeds; conservation fund	SEG	S	-0-	-0-	
(s)	Interest rebates on obligation					
	proceeds; transportation fund	SEG	S	-0-	-0-	
(t)	Interest rebates on obligation		_	_	_	
	proceeds; veterans trust fund	SEG	S	-0-	-0-	
(w)	Bonding services	SEG	S	1,024,200	1,024,200	
		(3) PRO0	GRAM TO		20.021.000	
	GENERAL PURPOSE REVENUE			17,926,400	30,021,000	
	PROGRAM REVENUE			2,082,200	2,925,800	
	OTHER			(-0-)	(-0-)	
	SERVICE			(2,082,200)	(2,925,800)	
	SEGREGATED REVENUE			1,024,200	1,024,200	
	OTHER TOTAL-ALL SOURCES			(1,024,200)	(1,024,200)	
(4)	CAPITAL IMPROVEMENT FUND INTERES	T FADNINGS		21,032,800	33,971,000	
	Funding in lieu of borrowing	SEG	С	-0-	0	
(q) (r)	Interest on veterans obligations	SEG	C	_0_ _0_	_0_ _0_	
(1)	interest on veterans congations		GRAM TO		-0-	
	SEGREGATED REVENUE	(4) 1 KO	SICINI TO	- 0-	-0-	
	OTHER			(-0-)	(-0-)	
	TOTAL-ALL SOURCES			-0-	-0-	
(5)	SERVICES TO NONSTATE GOVERNMENTA	L UNITS				
(g)	Financial consulting services	PR	C	-0-	-0-	
	<u> </u>	(5) PRO0	GRAM TO	TALS		
	PROGRAM REVENUE			-0-	-0-	
	OTHER			(-0-)	(-0-)	
	TOTAL-ALL SOURCES			-0-	-0-	
		20.867 DEPA	ARTMENT T	ΓOTALS		
	GENERAL PURPOSE REVENUE			28,644,300	39,500,600	
	PROGRAM REVENUE			2,082,200	2,925,800	
	OTHER			(-0-)	(-0-)	
	SERVICE			(2,082,200)	(2,925,800)	
	SEGREGATED REVENUE			1,024,200	1,024,200	
	OTHER			(1,024,200)	(1,024,200)	
20.05	TOTAL-ALL SOURCES			31,750,700	43,450,600	
20.873 (1)	5 Budget Stabilization Fund Transfers to fund					
(a)	General fund transfer	GPR	S	-0-	-0-	
(a)	General fund transfer		GRAM TO		-0-	
	GENERAL PURPOSE REVENUE	(1) 1 KO	JKAWI 101	-0-	-0-	
	TOTAL-ALL SOURCES			-0-	-0-	
(2)	TRANSFERS FROM FUND			· ·	· ·	
(q)	Budget stabilization fund transfer	SEG	A	-0-	-0-	
(4)	= augus amamamam runa trunster		GRAM TO		O .	
	SEGREGATED REVENUE	(-) 1110	10	-0-	-0-	
	OTHER			(-0-)	(-0-)	
	TOTAL-ALL SOURCES			-0-	-0-	
		20.875 DEPA	ARTMENT 7	ΓOTALS		

Vetoed In Part

SECTION 183m. 20.115 (2) (r) of the statutes is created to read:

20.115 (2) (r) Livestock premises registration — agrichemical management fund. From the agrichemical management fund, the amounts in the schedule for administration of the livestock premises registration program under s. 95.51.

SECTION 187g. 20.115 (4) (f) of the statutes is repealed.

SECTION 187r. 20.115 (4) (qm) of the statutes is repealed.

SECTION 188m. 20.115 (7) (dm) of the statutes is amended to read:

20.115 (7) (dm) Farmland preservation planning grants. The amounts in the schedule for farmland preservation planning grants under s. 91.10 (6). No moneys may be encumbered under this paragraph after June 30, 2016.

SECTION 190. 20.115 (7) (wm) of the statutes is amended to read:

20.115 (7) (wm) Agricultural chemical cleanup reimbursement. From the agricultural chemical cleanup fund, as a continuing appropriation, the amounts in the

schedule for reimbursement of corrective action costs under s. 94.73 and for financial assistance to prevent pollution from agricultural chemicals under s. 94.74.

SECTION 191. 20.115 (8) (g) of the statutes is amended to read:

20.115 (8) (g) *Gifts and grants*. Except as provided in par. (ge) and sub. (7) (i), all moneys received from gifts and grants to carry out the purposes for which made.

SECTION 192. 20.115 (8) (ge) of the statutes is renumbered 20.445 (1) (gr).

SECTION 193. 20.144 (intro.) of the statutes is amended to read:

20.144 Financial institutions, department of. (intro.) There is appropriated to the department of financial institutions for the following program programs:

SECTION 194. 20.144 (1) (g) of the statutes is amended to read:

20.144 (1) (g) General program operations. The amounts in the schedule for the general program operations of the department of financial institutions. Except as provided in pars. (a), (h), (i), (j), and (u) and sub. (3), all moneys received by the department, other than by the office of credit unions and the division of banking, and 88

percent of all moneys received by the office of credit unions and the department's division of banking shall be credited to this appropriation, but any balance at the close of a fiscal year under this appropriation shall lapse to the general fund. Annually, \$150,000 of the amounts received under this appropriation account shall be transferred to the appropriation account under s. 20.575 (1) (g).

SECTION 195. 20.144 (3) (title) of the statutes is created to read:

20.144 (3) (title) College Tuition and Expenses and College Savings programs.

SECTION 198. 20.155 (1) (q) of the statutes is amended to read:

20.155 (1) (q) *Universal telecommunications service*: broadband service. From the universal service fund, the amounts in the schedule for the promotion of broadband service and universal telecommunications service for the purposes specified in s. 196.218 (5) (a) 1., 4., 8. and, 9., and 10. Notwithstanding s. 20.001 (3) (a), the unencumbered balance on June 30 of each year shall be transferred to the appropriation account under sub. (3) (r).

SECTION 198m. 20.155 (3) (q) of the statutes is repealed.

SECTION 199. 20.155 (3) (r) of the statutes is amended to read:

20.155 (3) (r) Broadband expansion grants: transfer funding. From the universal service fund, as a continuing appropriation, the amounts in the schedule all moneys transferred from the appropriation accounts under par. (rm), sub. (1) (q), and ss. 20.255 (1) (q) and (3) (q), (qm), and (r), 20.285 (1) (q) and 20.505 (4) (s), under 2015 Wisconsin Act 55, section 9236 (1v), and under 2017 Wisconsin Act (this act), section 9237 (1) and (2) (a), for broadband expansion grants under s. 196.504. All moneys transferred under 2015 Wisconsin Act 55, section 9236 (1v) shall be credited to this appropriation account.

SECTION 199m. 20.155 (3) (rm) of the statutes is created to read:

20.155 (3) (rm) *Broadband grants; other funding*. From the universal service fund, as a continuing appropriation, all moneys transferred under s. 196.218 (3) (a) 2s., for broadband expansion grants under s. 196.504. Notwithstanding s. 20.001 (3) (c), the unencumbered balance on June 30 of each year shall be transferred to the appropriation account under sub. (3) (r).

SECTION 202e. 20.165 (2) (j) of the statutes is amended to read:

20.165 (2) (j) Safety and building operations. The amounts in the schedule for the purposes of chs. 101 and 145 and ss. 167.35, 236.12 (2) (ap), 236.13 (1) (d) and (2m), and 236.335 and for the purpose of transferring the amounts in the schedule under par. (ke) to the appropriation account under par. (ke). All moneys received under ch. 145, ss. 101.178, 101.19, 101.63 (9), 101.654 (3),

101.73 (12), 101.82 (4), 101.955 (2), 101.973 (7), 167.35 (2) (f), and 236.12 (7) and all moneys transferred under 2005 Wisconsin Act 45, section 76 (6), shall be credited to this appropriation account.

SECTION 202g. 20.165 (2) (ke) of the statutes is repealed.

SECTION 203. 20.192 (1) (a) of the statutes is repealed and recreated to read:

20.192 (1) (a) Operations and programs. A sum sufficient in fiscal year 2017–18 equal to the amount obtained by subtracting from \$35,250,700 an amount equal to the sum of the amounts expended in that fiscal year from the appropriations under pars. (r) and (s); and in fiscal year 2018–19 equal to the amount obtained by subtracting from \$41,550,700 the sum of the amounts expended in that fiscal year from the appropriations under pars. (r) and (s); for the operations of the Wisconsin Economic Development Corporation and for funding economic development programs developed and implemented under s. 238.03. No more than \$16,512,500 may be expended from this appropriation in any fiscal year, and no moneys may be expended from this appropriation unless the balance of the appropriation under par. (r) is \$0.

SECTION 204. 20.192 (1) (r) of the statutes is amended to read:

20.192 (1) (r) Economic development fund; operations and programs. From the economic development fund, as a continuing appropriation, the amounts in the schedule after deducting the amounts appropriated from that fund under s. 20.566 (1) (q), all moneys received from the deposits made under s. 77.97, for the operations of the Wisconsin Economic Development Corporation and for funding the economic development programs it administers.

SECTION 204m. 20.235 (1) (cu) of the statutes is created to read:

20.235 (1) (cu) *School leadership loan program.* As a continuing appropriation, the amounts in the schedule for the loan program under s. 39.397.

SECTION 204p. 20.235 (1) (fe) of the statutes is amended to read:

20.235 (1) (fe) Wisconsin grants; University of Wisconsin System students. A sum sufficient equal to \$0 in the 2013–14 fiscal year, equal to \$58,345,400 in the 2014–15 fiscal year, and equal to the amount calculated under s. 39.435 (7) Biennially, the amounts in the schedule for the Wisconsin grant program under s. 39.435 for University of Wisconsin System students, except for grants awarded under s. 39.435 (2) or (5), thereafter. Notwithstanding s. 20.002 (1), the higher educational aids board may transfer moneys under this paragraph between fiscal years.

SECTION 205. 20.255 (1) (c) of the statutes is amended to read:

20.255 (1) (c) Energy costs; Wisconsin Educational Services Program for the Deaf and Hard of Hearing and Wisconsin Center for the Blind and Visually Impaired; energy-related assessments. The amounts in the schedule to be used at the facilities of the Wisconsin Educational Services Program for the Deaf and Hard of Hearing and the Wisconsin Center for the Blind and Visually Impaired to pay for utilities and for fuel, heat, and air conditioning, to pay assessments levied by the department of administration under s. 16.847 (3) for costs incurred and savings generated at departmental facilities, and to pay costs incurred by or on behalf of the department under ss. 16.858 and 16.895. No moneys may be expended from this appropriation for the purposes specified in par. (cm).

SECTION 206. 20.255 (1) (cm) of the statutes is created to read:

20.255 (1) (cm) Electric energy derived from renewable resources. The amounts in the schedule for the premium cost incurred for the generation or purchase of electric energy derived from renewable resources under s. 16.75 (12).

SECTION 206m. 20.255 (1) (eg) of the statutes is created to read:

20.255 (1) (eg) Rural school teacher talent pilot program. The amounts in the schedule for grants to cooperative educational service agencies under s. 115.423.

SECTION 207f. 20.255 (1) (fp) of the statutes is created to read:

20.255 (1) (fp) Study on school district reorganization; certain school districts. Biennially, the amounts in the schedule for the study under 2017 Wisconsin Act (this act), section 9135 (4w).

SECTION 207g. 20.255 (1) (fp) of the statutes, as created by 2017 Wisconsin Act (this act), is repealed.

SECTION 208. 20.255 (1) (q) of the statutes is amended to read:

20.255 (1) (q) Digital learning collaborative. From the universal service fund, the amounts in the schedule for a digital learning collaborative for the statewide web academy and for the delivery of digital content and collaborative instruction under s. 115.28 (53) and (54). Notwithstanding s. 20.001 (3) (a), the unencumbered balance on June 30 of each year shall be transferred to the appropriation account under s. 20.155 (3) (r).

SECTION 208e. 20.255 (2) (aw) of the statutes is created to read:

20.255 (2) (aw) Personal electronic computing devices; grant program. The amounts in the schedule for aid under s. 115.438. No moneys may be encumbered from this appropriation after June 30, 2023.

SECTION 208m. 20.255 (2) (az) of the statutes is amended to read:

20.255 (2) (az) Special Needs Scholarship Program. A sum sufficient to make the payments under s. 115.7915 (4m) (a), (cm), and (e) and (4p).

SECTION 208n. 20.255 (2) (bg) of the statutes is created to read:

20.255 (2) (bg) Special education transition readiness grants. The amounts in the schedule for grants under s. 115.885 to support special education workforce transition support services.

SECTION 208p. 20.255 (2) (bp) of the statutes is cre- Vetoed

20.255 (2) (bp) Aid for whole grade sharing agree*ments*. The amounts in the schedule for payments under s. 118.50 (5m).

SECTION 208r. 20.255 (2) (br) of the statutes is created to read:

20.255 (2) (br) School district consolidation aid. A sum sufficient to provide aid to school districts under ss. 117.08 (6) and 117.09 (6).

SECTION 208t. 20.255 (2) (bt) of the statutes is cre- Vetoed ated to read:

20.255 (2) (bt) Shared services pilot program. As a continuing appropriation, the amounts in the schedule for aid to school districts participating in a shared services plan under s. 115.434.

SECTION 208v. 20.255 (2) (cg) of the statutes is amended to read:

20.255 (2) (cg) Tuition payments; full-time open enrollment transfer payments. The amounts in the schedule for payment of tuition under subch. V of ch. 121 and full-time open enrollment transfer payments under s. 118.51 (16) (b) 2. and (17) (c) 2. and (cm) 2.

SECTION 209. 20.255 (2) (cw) of the statutes is repealed.

SECTION 210. 20.255 (2) (cy) of the statutes is amended to read:

20.255 (2) (cy) Aid for transportation; open enrollment and course options early college credit program. The amounts in the schedule to reimburse parents for the costs of transportation of open enrollment pupils under ss. 118.51 (14) (b) and 118.52 (11) (b) and for the payment of state aid under s. 118.55 (7g) for the transportation of pupils attending a course at an institution of higher education and receiving credit for the course under s. 118.55 (3) (b).

SECTION 211. 20.255 (2) (da) of the statutes is created to read:

20.255 (2) (da) Aid for school mental health programs. The amounts in the schedule for aid to school districts and independent charter schools under s. 115.364.

SECTION 214. 20.255 (2) (dg) of the statutes is created to read:

20.255 (2) (dg) School performance improvement grants. The amounts in the schedule for grants under s. 115.387.

Section 215. 20.255 (2) (dj) of the statutes is created to read:

In Part

In Part

20.255 (2) (dj) Summer school programs; grants. The amounts in the schedule for grants to school boards for summer school grant programs under s. 115.447.

SECTION 215g. 20.255 (2) (dr) of the statutes is amended to read:

20.255 (2) (dr) *Robotics league participation grants*. The amounts in the schedule for grants to robotics teams under s. 115.45. No moneys may be encumbered from this appropriation after June 30, 2017.

SECTION 215k. 20.255 (2) (dt) of the statutes is created to read:

20.255 (2) (dt) School-based mental health services grants. As a continuing appropriation, the amounts in the schedule for the school-based mental health services grant program under s. 115.367.

SECTION 215m. 20.255 (2) (eb) of the statutes is created to read:

20.255 (2) (eb) *Grant for information technology education*. The amounts in the schedule for the grant under s. 115.455.

SECTION 215n. 20.255 (2) (fm) of the statutes is amended to read:

20.255 (2) (fm) *Charter schools*. A sum sufficient to make the payments to charter schools under s. 118.40 (2r) (e) and, (f), and (fm).

SECTION 215p. 20.255 (2) (fp) of the statutes, as affected by 2017 Wisconsin Act 30, is amended to read:

20.255 (2) (fp) Charter schools; office of educational opportunity. A sum sufficient to make the payments to charter schools under s. 118.40 (2x) (e) 1. and (em).

SECTION 216. 20.255 (2) (km) of the statutes is amended to read:

20.255 (2) (km) *Tribal language revitalization grants*. The amounts in the schedule for grants to school districts and, cooperative educational service agencies, and agencies determined by the state superintendent to be eligible under 42 USC 9836 for designation as head start agencies under s. 115.745. All moneys transferred from the appropriation account under s. 20.505 (8) (hm) 5. shall be credited to this appropriation account. Notwithstanding s. 20.001 (3) (a), the unencumbered balance on June 30 of each year shall revert to the appropriation account under s. 20.505 (8) (hm).

SECTION 216m. 20.255 (3) (cm) of the statutes is renumbered 20.445 (1) (bt) and amended to read:

20.445 (1) (bt) *Teach for America Workforce development: grants for teacher training and recruitment.* Biennially, the amounts in the schedule for payments to Teach for America, Inc., grants under s. 115.28 (60) 106.277.

SECTION 217. 20.255 (3) (eb) of the statutes is created to read:

20.255 (3) (eb) *Grants for bullying prevention.* The amounts in the schedule for grants under s. 115.28 (45).

SECTION 217g. 20.255 (3) (fc) of the statutes is created to read:

20.255 (3) (fc) *College Possible, Inc.* The amounts in the schedule for grants to College Possible, Inc., under s. 115.28 (64).

SECTION 217m. 20.255 (3) (fr) of the statutes is created to read:

20.255 (3) (fr) *Wisconsin Reading Corps*. The amounts in the schedule for payments to Wisconsin Reading Corps under s. 115.28 (65). No moneys may be encumbered under this paragraph after June 30, 2019.

SECTION 218. 20.255 (3) (q) of the statutes is amended to read:

20.255 (3) (q) Periodical and reference information databases; Newsline for the Blind. From the universal service fund, the amounts in the schedule for the Newsline for the Blind, provided by the Regional Library for the Blind and Physically Handicapped, and to contract for periodical and reference information databases under s. 115.28 (26). Notwithstanding s. 20.001 (3) (a), the unencumbered balance on June 30 of each year shall be transferred to the appropriation account under s. 20.155 (3) (r).

SECTION 219. 20.255 (3) (qm) of the statutes is amended to read:

20.255 (3) (qm) *Aid to public library systems*. From the universal service fund, the amounts in the schedule for state aid to public library systems under s. 43.24. <u>Not-withstanding s. 20.001 (3) (a)</u>, the unencumbered balance on June 30 of each year shall be transferred to the appropriation account under s. 20.155 (3) (r).

SECTION 220. 20.255 (3) (r) of the statutes is amended to read:

20.255 (3) (r) *Library service contracts*. From the universal service fund, the amounts in the schedule for library service contracts under s. 43.03 (6) and (7). Not-withstanding s. 20.001 (3) (a), the unencumbered balance on June 30 of each year shall be transferred to the appropriation account under s. 20.155 (3) (r).

SECTION 221. 20.285 (1) (a) of the statutes is amended to read:

20.285 (1) (a) General program operations. Biennially, the amounts in the schedule for the purpose of educational programs and related programs. The board of regents may not encumber amounts appropriated under this paragraph for groundwater research without the approval of the secretary of administration. No moneys may be expended from this appropriation for the purposes specified in par. (am).

SECTION 222. 20.285 (1) (am) of the statutes is created to read:

20.285 (1) (am) Electric energy derived from renewable resources. The amounts in the schedule for the premium cost incurred for the generation or purchase of electric energy derived from renewable resources under s. 16.75 (12).

SECTION 222m. 20.285 (1) (b) of the statutes is created to read:

20.285 (1) (b) *Tommy G. Thompson Center on Public Leadership*. The amounts in the schedule for general program operations of the Tommy G. Thompson Center on Public Leadership.

SECTION 223. 20.285 (1) (q) of the statutes is amended to read:

20.285 (1) (q) *Telecommunications services*. From the universal service fund, the amounts in the schedule to provide telecommunications services as specified in s. 196.218 (5) (a) 6. Notwithstanding s. 20.001 (3) (a), the unencumbered balance on June 30 of each year shall be transferred to the appropriation account under s. 20.155 (3) (r).

SECTION 223m. 20.285 (1) (sp) of the statutes is created to read:

20.285 (1) (sp) Wisconsin Institute for Sustainable Technology. From the environmental fund, the amounts in the schedule to support the Wisconsin Institute for Sustainable Technology at the University of Wisconsin–Stevens Point.

SECTION 225. 20.292 (2) (title) of the statutes is repealed.

SECTION 226. 20.292 (2) (g) of the statutes is renumbered 20.165 (1) (jr) and amended to read:

20.165 (1) (jr) *Proprietary school programs*. The amounts in the schedule for the examination and approval of proprietary school programs <u>under s. 440.52</u>. Ninety percent of all moneys received from the issuance of solicitor's permits under s. 38.50 440.52 (8) and from the fees under s. 38.50 440.52 (10) and all moneys received from the fees under s. 38.50 440.52 (13) (d) shall be credited to this appropriation account.

SECTION 227. 20.292 (2) (gm) of the statutes is renumbered 20.165 (1) (jt) and amended to read:

20.165 (1) (jt) Student protection. All moneys received from fees received under s. 38.50 440.52 (10) (c) 4., for the purpose of indemnifying students, parents, or sponsors under s. 38.50 440.52 (10) (a) and for the purpose of preserving under s. 38.50 440.52 (11) the students records of schools, as defined in s. 38.50 440.52 (11) (a) 2., that have discontinued their operations.

SECTION 228. 20.292 (2) (i) of the statutes is renumbered 20.165 (1) (jv) and amended to read:

20.165 (1) (jv) Closed schools; preservation of student records. All moneys received from fees collected under s. 38.50 440.52 (11) (d) to be used for the administrative costs of taking possession of, preserving, and providing copies of student records of schools, as defined in s. 38.50 440.52 (11) (a) 2., that have discontinued their operations.

SECTION 228p. 20.320 (3) of the statutes is repealed. **SECTION 229.** 20.370 (1) (title) of the statutes is repealed and recreated to read:

20.370 (1) (title) FISH, WILDLIFE, AND PARKS.

SECTION 230. 20.370 (1) (cq) of the statutes is renumbered 20.370 (2) (cq) and amended to read:

20.370 (2) (cq) Forestry — reforestation. As a continuing appropriation, from the conservation fund, the amounts in the schedule for reforestation of state forests and nursery operations as provided under chs. 26 and 28.

SECTION 231. 20.370 (1) (cr) of the statutes is renumbered 20.370 (2) (cr) and amended to read:

20.370 (2) (cr) Forestry — recording fees. All From the conservation fund, all moneys received under ss. 77.82 (2m) (d) and (4) and 77.88 (2) (ac) 1. for the payment of fees to the registers of deeds under s. 77.91 (5).

SECTION 232. 20.370 (1) (cs) of the statutes is renumbered 20.370 (2) (cs) and amended to read:

20.370 (2) (cs) Forestry — forest fire emergencies. Except as provided in s. 26.11 (7), from the conservation fund, all moneys received from other states for forest fire fighting activities provided by the department to be used for forest fire fighting activities.

SECTION 233. 20.370 (1) (ct) of the statutes is renumbered 20.370 (2) (ct) and amended to read:

20.370 (2) (ct) Timber sales contracts — repair and reimbursement costs. All From the conservation fund, all moneys received by the department as sureties under s. 28.05 (1) to be used to repair damage and recover costs incurred by the improper performance of timber sales contracts and to reimburse persons who provide sureties as provided in s. 28.05 (1).

SECTION 234. 20.370 (1) (cu) of the statutes is renumbered 20.370 (2) (cu) and amended to read:

20.370 (2) (cu) Forestry — forestry education curriculum. The From the conservation fund, the amounts in the schedule for the development of a forestry education curriculum under s, 26.39 (2).

SECTION 235. 20.370 (1) (cx) of the statutes is renumbered 20.370 (2) (cx) and amended to read:

20.370 (2) (cx) Forestry — management plans. All From the conservation fund, all moneys received under s. 77.82 (2m) (ag) for payment for management plans prepared or completed by plan writers who are under contract with the department under s. 77.82 (3).

SECTION 236. 20.370 (1) (cy) of the statutes is renumbered 20.370 (2) (cy) and amended to read:

20.370 (2) (cy) Forestry — cooperating foresters and private contractors. All From the conservation fund, all moneys received under s. 28.05 (3) (c) for payment to cooperating foresters and private contractors to be used for those payments.

SECTION 237. 20.370 (1) (cz) of the statutes is renumbered 20.370 (2) (cz) and amended to read:

20.370 (2) (cz) Forestry — management of national forest land. All From the conservation fund, all moneys received from the sale of timber from federal land under a cooperative agreement under s. 28.15 to be used to administer, implement, and pay costs associated with the cooperative agreement and any contracts entered into under s. 28.15 (3) and to lapse the amounts under s. 28.15 (5).

SECTION 238. 20.370 (1) (er) of the statutes is amended to read:

20.370 (1) (er) *Parks and forests* — *campground reservation fees.* All moneys not retained by the department under s. 27.01 (11) (cr) 1. for payments to contracting parties under contracts entered into under s. 27.01 (11) (cm) that are applicable to southern state forests or state parks.

SECTION 239. 20.370 (1) (fe) of the statutes is amended to read:

20.370 (1) (fe) Endangered resources — general fund. From the general fund, a sum sufficient in fiscal year 1993–94 and in each fiscal year thereafter that equals the sum of the amount certified in that fiscal year under s. 71.10 (5) (h) 3. for the previous fiscal year and the amounts received under par. (gr) (fu) in that fiscal year for the purposes of the endangered resources program, as defined in s. 71.10 (5) (a) 2. The amount appropriated under this subdivision may not exceed \$500,000 in a fiscal year, except that the amount appropriated under this subdivision in fiscal year 2005–06 may not exceed \$364,000 and the amount appropriated under this subdivision in fiscal year 2006–07 may not exceed \$364,000.

SECTION 239m. 20.370 (1) (fs) of the statutes is amended to read:

20.370 (1) (fs) Endangered resources — voluntary payments; sales, leases, and fees. As a continuing appropriation, from moneys received as amounts designated under ss. 71.10 (5) (b) and 71.30 (10) (b), the net amounts certified under ss. 71.10 (5) (h) 4. and 71.30 (10) (h) 3., all moneys received from the sale or lease of resources derived from the land in the state natural areas system, and all moneys received from fees collected under ss. 29.319 (2), 29.563 (10) (a), and 341.14 (6r) (b) 5. and 12., for the purposes of the endangered resources program, as defined under ss. 71.10 (5) (a) 2. and 71.30 (10) (a) 2. Three percent of the moneys certified under ss. 71.10 (5) (h) 4. and 71.30 (10) (h) 3. in each fiscal year and 3 percent of the fees received under s. 341.14 (6r) (b) 5. and 12. in each fiscal year shall be allocated for wildlife damage control and payment of claims for damage associated with endangered or threatened species, except that this combined allocation may not exceed \$100,000 per fiscal year.

SECTION 240. 20.370 (1) (gr) of the statutes is renumbered 20.370 (1) (fu).

SECTION 241. 20.370 (1) (gt) of the statutes is renumbered 20.370 (2) (gt) and amended to read:

20.370 (2) (gt) *Habitat conservation plan fees.* All From the conservation fund, all moneys received from gifts, grants, and bequests to, and all fees paid by partners in, the Karner blue butterfly habitat conservation plan to be used for the administration and implementation of the plan.

SECTION 242. 20.370 (1) (hw) of the statutes is amended to read:

20.370 (1) (hw) *Pheasant stocking and propagation.* Sixty percent of the moneys received under s. 29.191 (2) and all moneys received under s. 23.09 (15) for the stocking and propagation of pheasants on lands under the department's ownership, management, supervision, or control.

SECTION 243. 20.370 (1) (jb) of the statutes is renumbered 20.370 (9) (jb).

SECTION 244. 20.370 (1) (jr) of the statutes is amended to read:

20.370 (1) (jr) Rental property and equipment — maintenance and replacement. All moneys received by the department from the rental of real property and equipment that are owned by the department and are utilized for land, fisheries, and wildlife management, excluding forestry purposes, to be used for the maintenance and replacement of this real property and equipment.

SECTION 245. 20.370 (1) (ma) of the statutes is amended to read:

20.370 (1) (ma) General program operations — state funds. From the general fund, the amounts in the schedule for general program operations under ch. 23 and ss. 30.40 to 30.49 and, for the trapper education program under s. 29.597, and for general program operations relating to management of the state's fishery resources.

SECTION 246. 20.370 (1) (mi) of the statutes is amended to read:

20.370 (1) (mi) General program operations — private and public sources. From the general fund, all moneys not otherwise appropriated that are received from private or public sources, other than state agencies and the federal government, for facilities, materials, or services, excluding forestry facilities, materials, or services, provided by the department relating to resource management to pay for expenses associated with those facilities, materials, or services.

SECTION 247. 20.370 (1) (mk) of the statutes is amended to read:

20.370 (1) (mk) General program operations — service funds. From the general fund, all moneys received by the department from the department and from other state agencies for facilities, materials, or services, excluding forestry facilities, materials, or services, provided by the department relating to resource management under an agreement or other arrangement with the department or other state agencies to pay for expenses associated with those facilities, materials, or services.

SECTION 248. 20.370 (1) (mm) of the statutes is created to read:

20.370 (1) (mm) General program operations — federal funds. From the general fund, all moneys received as federal aid for the state's fishery resources, as authorized by the governor under s. 16.54, for the purposes for which received.

SECTION 249. 20.370 (1) (mu) of the statutes is amended to read:

Vetoed In Part 20.370 (1) (mu) General program operations — state funds. The amounts in the schedule for general program operations that do not relate to the management and protection of the state's fishery resources and that are conducted under ss. 23.09 to 23.11, 27.01, 30.203, 30.277, and 90.21, and chs. 29 and 169, for activities conducted under the ecological inventory and monitoring program of the endangered resources program, for the aquatic and terrestrial resources inventory under s. 23.09 (2) (km), and for payments of \$53,700 in each fiscal year, to be credited to the appropriation account under s. 20.285 (1) (k), to the University of Wisconsin System for outdoor skills training under s. 29.598.

SECTION 250. 20.370 (1) (mv) of the statutes is renumbered 20.370 (2) (mv) and amended to read:

20.370 (2) (mv) General program operations — state funds; forestry. The From the conservation fund, the amounts in the schedule for general program operations that relate to the management and protection of the state's forestry resources and that are conducted under ss. 23.09 to 23.11 and 27.01, subch. VI of ch. 77, and chs. 26 and 28, to make the payments under s. 77.89 (1) (b), and to pay the initial costs of administering and implementing a cooperative agreement under s. 28.15 and any contracts entered into under s. 28.15 (3).

SECTION 251. 20.370 (1) (my) of the statutes is amended to read:

20.370 (1) (my) General program operations — federal funds. All moneys received as federal aid for land, forestry, and wildlife, fisheries, and recreation management, as authorized by the governor under s. 16.54 for the purposes for which received.

SECTION 252. 20.370 (1) (mz) of the statutes is renumbered 20.370 (2) (mz) and amended to read:

20.370 (2) (mz) Forest fire emergencies — federal funds. Except as provided in s. 26.11 (7), from the conservation fund, all moneys received as federal aid for forest fire fighting activities provided by the department to be used for forest fire fighting activities.

SECTION 253. 20.370 (2) (title) of the statutes is repealed and recreated to read:

20.370 (2) (title) FORESTRY.

SECTION 254. 20.370 (2) (bg) of the statutes is renumbered 20.370 (4) (co) and amended to read:

20.370 (4) (co) Air management — stationary sources. The From the general fund, the amounts in the schedule for purposes related to stationary sources of air contaminants as specified in s. 285.69 (2) (c). All moneys received from fees under s. 285.69 (2) (a) and (e) and (2e), except moneys appropriated under subs. (3) (bg), (8) (mg) and (9) (mh), and all moneys received from fees imposed under s. 285.69 (7) shall be credited to this appropriation.

SECTION 255. 20.370 (2) (bh) of the statutes is renumbered 20.370 (4) (cm) and amended to read:

20.370 (4) (cm) Air management — state permit sources. The From the general fund, the amounts in the schedule for purposes related to stationary sources of air contaminants for which an operation permit is required under s. 285.60 but not under the federal clean air act as specified in s. 285.69 (2m) (bm). All moneys received from fees imposed under s. 285.69 (2m) shall be credited to this appropriation account.

SECTION 256. 20.370 (2) (bi) of the statutes is renumbered 20.370 (4) (cn) and amended to read:

20.370 (4) (cn) Air management — asbestos management. All From the general fund, all moneys received from fees imposed under s. 285.69 (1) (c) on persons proposing asbestos abatement projects and all moneys received under s. 285.69 (3) for asbestos abatement inspections, for costs related to exempting asbestos abatement projects from air pollution control permits and for inspections of asbestos demolition and renovation projects.

SECTION 257. 20.370 (2) (bq) of the statutes is repealed.

SECTION 258. 20.370 (2) (br) of the statutes is renumbered 20.370 (4) (bt).

SECTION 259. 20.370 (2) (ce) of the statutes is renumbered 20.370 (4) (cv) and amended to read:

20.370 (4) (cv) Air quality monitoring station stations. Biennially, from the petroleum inspection fund, the amounts in the schedule for the air quality monitoring station stations under s. 285.72.

SECTION 260. 20.370 (2) (cf) of the statutes is renumbered 20.370 (4) (cw) and amended to read:

20.370 (4) (cw) Air management — motor vehicle emission inspection and maintenance program, state funds petroleum inspection fund. The From the petroleum inspection fund, the amounts in the schedule for the administration of the motor vehicle emission inspection and maintenance program under s. 285.30.

SECTION 261. 20.370 (2) (cg) of the statutes is renumbered 20.370 (4) (cL) and amended to read:

20.370 (4) (cL) Air management — recovery of ozone-depleting refrigerants. The From the general fund, the amounts in the schedule for administration of the recovery of ozone-depleting refrigerants program. All moneys received from fees under s. 285.59 (5) (a) 2. shall be credited to this appropriation.

SECTION 262. 20.370 (2) (ch) of the statutes is renumbered 20.370 (4) (bn) and amended to read:

20.370 (4) (bn) *Air management* — *emission analysis*. All From the general fund, all moneys received from fees collected under s. 285.53 (1) (c) 3. for the purpose of reviewing and preparing analyses of emissions from certain medical waste incinerators.

SECTION 263. 20.370 (2) (ci) of the statutes is renumbered 20.370 (4) (bo) and amended to read:

20.370 (4) (bo) Air management — permit review and enforcement. The From the general fund, the amounts in the schedule for any purpose specified under s. 285.69 (1) or (5), except for purposes described in par. (bi) (cn), and for other activities to reduce air pollution, as provided in s. 285.69 (6). All moneys received from fees imposed under s. 285.69 (1), (1d), and (5), except moneys appropriated under par. (bi) (cn), shall be credited to this appropriation.

SECTION 264. 20.370 (2) (cL) of the statutes is renumbered 20.370 (4) (bp) and amended to read:

20.370 (4) (bp) Air waste management — incinerator operator certification. —All From the general fund, all moneys received from fees under s. 285.51 for the purpose of administering s. 285.51.

SECTION 264n. 20.370 (2) (cv) of the statutes is created to read:

20.370 (2) (cv) Forestry — forestry emergency reserve. From the conservation fund, as a continuing appropriation, the amounts in the schedule for the costs of responding to significant forest fire, disease, infestation, or other natural disasters affecting forests. No expenditure may be made from this appropriation except upon approval of the joint committee on finance after a determination that the moneys are needed and that no appropriation of federal funds is available for that purpose.

SECTION 264p. 20.370 (2) (cw) of the statutes is created to read:

20.370 (2) (cw) Forestry — Pattison communications tower. From the conservation fund, as a continuing appropriation, the amounts in the schedule for the cost of constructing a communications tower at Pattison Ranger Station in Pattison State Park in the town of Superior, Douglas County.

SECTION 265. 20.370 (2) (dg) of the statutes is renumbered 20.370 (4) (dg) and amended to read:

20.370 (4) (dg) Solid waste management — solid and hazardous waste disposal administration. All From the general fund, all moneys received from fees under ss. 289.42 (1), 289.43 (7) (e) 1. and 2., 289.61, 291.05 (7) and 291.33, for the purpose of administering ss. 289.42 (1), 289.43, 289.47, 289.53, 289.95, 291.23, 291.25, 291.29, 291.31 and 291.87 and subch. III of ch. 289.

SECTION 266. 20.370 (2) (dh) of the statutes is renumbered 20.370 (4) (dh) and amended to read:

20.370 (4) (dh) Solid waste management — remediated property. All From the general fund, all moneys received under ss. 292.11 (7) (d) 2., 292.13 (3), 292.15 (5), 292.21 (1) (c) 1. d., 292.31 (7) (d), 292.35 (13), 292.55 (2), 292.57 (2), and 292.94 for the department's activities related to the issuance of determinations under s. 292.13 (2), remedial action cost recovery under s. 292.35, remediation of property under ss. 292.11 (7) (d), 292.15 (2), 292.55 (1), and 292.57, providing management and technical support for remedial action under 42

USC 9601 to 9675, and conducting reviews described in s. 292.94.

SECTION 267. 20.370 (2) (dq) of the statutes is renumbered 20.370 (4) (dq).

SECTION 268. 20.370 (2) (dt) of the statutes is renumbered 20.370 (4) (dt).

SECTION 269. 20.370 (2) (du) of the statutes is renumbered 20.370 (4) (du).

SECTION 270. 20.370 (2) (dv) of the statutes is renumbered 20.370 (4) (dv).

SECTION 271. 20.370 (2) (dw) of the statutes is renumbered 20.370 (4) (dw).

SECTION 272. 20.370 (2) (dy) of the statutes is renumbered 20.370 (4) (dy).

SECTION 273. 20.370 (2) (dz) of the statutes is renumbered 20.370 (4) (dz).

SECTION 274. 20.370 (2) (eg) of the statutes is renumbered 20.370 (4) (eg) and amended to read:

20.370 (4) (eg) *Solid waste facility siting board fee.* All From the general fund, all moneys received from the fee under s. 289.64 to be transferred to the appropriation under s. 20.505 (4) (k).

SECTION 275. 20.370 (2) (eh) of the statutes is renumbered 20.370 (4) (eh) and amended to read:

20.370 (4) (eh) *Solid waste management — source reduction review.* All From the general fund, all moneys received from fees collected under s. 287.07 (8) (d) for the purpose of reviewing medical waste source reduction policies and assessments.

SECTION 276. 20.370 (2) (eq) of the statutes is renumbered 20.370 (4) (eq).

SECTION 277. 20.370 (2) (fq) of the statutes is renumbered 20.370 (4) (fq).

SECTION 278. 20.370 (2) (gh) of the statutes is renumbered 20.370 (4) (gh) and amended to read:

20.370 (4) (gh) *Nonferrous metallic mining regulation and administration*. The From the general fund, the amounts in the schedule for the administration, regulation and enforcement of nonferrous metallic mining exploration, prospecting, mining and mine reclamation activities under ch. 293. All moneys received under ch. 293 shall be credited to this appropriation.

SECTION 279. 20.370 (2) (gi) of the statutes is renumbered 20.370 (4) (gi) and amended to read:

20.370 (4) (gi) Ferrous metallic mining operations. All From the general fund, all moneys received under subch. III of ch. 295 for department of natural resources operations related to ferrous metallic exploration and mining.

SECTION 280. 20.370 (2) (gr) of the statutes is renumbered 20.370 (4) (gr).

SECTION 281. 20.370 (2) (hq) of the statutes is renumbered 20.370 (4) (hq).

SECTION 282. 20.370 (2) (hr) of the statutes is renumbered 20.370 (4) (hr).

SECTION 283. 20.370 (2) (jr) of the statutes is created to read:

20.370 (2) (jr) Rental property and equipment — maintenance and replacement. From the conservation fund, all moneys received by the department from the rental of real property and equipment that are owned by the department and are utilized for forestry operations, to be used for the maintenance and replacement of this real property and equipment.

SECTION 284. 20.370 (2) (ma) of the statutes is repealed.

SECTION 285. 20.370 (2) (mi) of the statutes is amended to read:

20.370 (2) (mi) General program operations — private and public sources. All moneys not otherwise appropriated that are received from private or public sources, other than state agencies of and the federal government, for forestry facilities, materials, or services provided by the department relating to its environmental quality forestry functions to pay for expenses associated with those facilities, materials, or services.

SECTION 286. 20.370 (2) (mk) of the statutes is amended to read:

20.370 (2) (mk) General program operations — service funds. All moneys received by the department from the department and from other state agencies for purposes relating to its air and waste functions forestry facilities, materials, or services provided by the department relating to resource management under an agreement or other arrangement with the department or other state agencies to pay for expenses associated with those facilities, materials, or services.

SECTION 287. 20.370 (2) (mm) of the statutes is repealed.

SECTION 288. 20.370 (2) (mq) of the statutes is repealed.

SECTION 289. 20.370 (2) (mr) of the statutes is renumbered 20.370 (4) (mv).

SECTION 290. 20.370 (2) (my) of the statutes is renumbered 20.370 (4) (ms).

SECTION 291. 20.370 (2) (nz) of the statutes is created to read:

20.370 (2) (nz) General program operations — federal funds. From the conservation fund, all moneys received as federal aid for forestry management, as authorized by the governor under s. 16.54 for the purposes for which received.

SECTION 292. 20.370 (3) (title) of the statutes is amended to read:

20.370 (3) (title) Public safety and business support

SECTION 292m. 20.370 (3) (aw) of the statutes is repealed.

SECTION 293. 20.370 (3) (bL) of the statutes is renumbered 20.370 (9) (fL).

SECTION 294. 20.370 (3) (dg) of the statutes is repealed.

SECTION 295. 20.370 (3) (dh) of the statutes is renumbered 20.370 (9) (dh).

SECTION 296. 20.370 (3) (di) of the statutes is renumbered 20.370 (9) (di) and amended to read:

20.370 (9) (di) Environmental consulting costs — federal power projects. The From the general fund, the amounts in the schedule for reviewing and evaluating activities under s. 23.42. All moneys received from fees the department charges under s. 23.42 shall be credited to this appropriation.

SECTION 297. 20.370 (3) (fj) of the statutes is renumbered 20.370 (9) (fj).

SECTION 298. 20.370 (3) (is) of the statutes is renumbered 20.370 (9) (ks).

SECTION 299. 20.370 (3) (ma) of the statutes is amended to read:

20.370 (3) (ma) General program operations — state funds. From the general fund, the amounts in the schedule for regulatory and enforcement operations under chs. 30, 31 and 280 to 299 and ss. 44.47, 59.692, 59.693, 61.351, 61.353, 61.354, 62.231, 62.233, 62.234 and 87.30, for reimbursement of the conservation fund for expenses incurred for actions taken under s. 323.12 (2) (c); for review of environmental impact requirements under ss. 1.11 and 23.40; and for enforcement of the treaty—based, off—reservation rights to fish, hunt, and gather held by members of federally recognized American Indian tribes or bands.

SECTION 300. 20.370 (3) (ms) of the statutes is renumbered 20.370 (9) (ms).

SECTION 301. 20.370 (3) (mt) of the statutes is repealed.

SECTION 302. 20.370 (3) (mu) of the statutes is amended to read:

20.370 (3) (mu) General program operations — state funds. The amounts in the schedule for law enforcement operations under ss. 23.09 to 23.11, 90.21, and 323.12 (2) (c) and chs. 29, 30, and 169 and for review of environmental impact requirements under ss. 1.11 and 23.40.

SECTION 303. 20.370 (3) (mw) of the statutes is renumbered 20.370 (4) (aw).

SECTION 304. 20.370 (4) (title) of the statutes is repealed and recreated to read:

20.370 (4) (title) Environmental management.

SECTION 306. 20.370 (4) (as) of the statutes is renumbered 20.370 (9) (as).

SECTION 307. 20.370 (4) (at) of the statutes is renumbered 20.370 (9) (at).

SECTION 308. 20.370 (4) (bg) of the statutes is renumbered 20.370 (9) (bg).

SECTION 309. 20.370 (4) (bh) of the statutes is repealed.

SECTION 310. 20.370 (4) (bi) of the statutes is renumbered 20.370 (9) (bi).

SECTION 311. 20.370 (4) (bj) of the statutes is renumbered 20.370 (9) (bj).

SECTION 312. 20.370 (4) (bL) of the statutes is amended to read:

20.370 (4) (bL) Wastewater management — fees. From the general fund, from the moneys received under. ss. 281.17 (3) and s. 281.48 (4s) (a), all moneys not appropriated under sub. (3) (bL) (9) (fL), for the certification of operators of water systems, wastewater treatment plants, and septage servicing vehicles and for wastewater management activities.

SECTION 313. 20.370 (4) (bm) of the statutes is renumbered 20.370 (9) (bm).

SECTION 314. 20.370 (4) (br) of the statutes is renumbered 20.370 (9) (br).

SECTION 314d. 20.370 (4) (cq) of the statutes is created to read:

20.370 (4) (cq) *River and stream monitoring and study.* The amounts in the schedule for the river and stream monitoring and study under s. 281.145. No funds may be encumbered under this paragraph after June 30, 2021.

SECTION 314m. 20.370(4)(cr) of the statutes is created to read:

20.370 (4) (cr) *Hydrologic evaluation and modeling*. As a continuing appropriation, from the environmental fund, the amounts in the schedule to conduct the hydrologic evaluation and modeling under s. 281.34 (7m).

SECTION 315. 20.370 (4) (kb) of the statutes is renumbered 20.370 (1) (kb).

SECTION 316. 20.370 (4) (kc) of the statutes is renumbered 20.370 (1) (kc).

SECTION 317. 20.370 (4) (kg) of the statutes is renumbered 20.370 (1) (kg).

SECTION 318. 20.370 (4) (kk) of the statutes is renumbered 20.370 (1) (kk).

SECTION 319. 20.370 (4) (kr) of the statutes is renumbered 20.370 (1) (kr).

SECTION 320. 20.370 (4) (kt) of the statutes is renumbered 20.370 (1) (kt).

SECTION 321. 20.370 (4) (ku) of the statutes is renumbered 20.370 (1) (ku).

SECTION 322. 20.370 (4) (kv) of the statutes is renumbered 20.370 (1) (kv).

SECTION 323. 20.370 (4) (kw) of the statutes is renumbered 20.370 (1) (kw).

SECTION 324. 20.370 (4) (ky) of the statutes is renumbered 20.370 (1) (ky).

SECTION 325. 20.370 (4) (ma) of the statutes is amended to read:

20.370 (4) (ma) General program operations — state funds. From the general fund, the amounts in the schedule for the management and protection of the state's water resources and the state's fishery resources for envi-

ronmental quality and environmental management purposes.

SECTION 326. 20.370 (4) (mi) of the statutes is amended to read:

20.370 (4) (mi) General program operations — private and public sources. From the general fund, all moneys not otherwise appropriated that are received from private or public sources, other than state agencies and the federal government, for facilities, materials, or services provided by the department relating to its environmental quality functions and to the management of the state's water resources and the state's fishery resources and all moneys required under s. 283.31 (8) (b) to be credited to this appropriation to pay for expenses associated with those facilities, materials, or services.

SECTION 327. 20.370 (4) (mk) of the statutes is amended to read:

20.370 (4) (mk) General program operations — service funds. All From the general fund, all moneys received by the department from the department and from other state agencies for purposes relating to the department's function relating its environmental management functions and to the state's water resources and the state's fishery resources.

SECTION 328. 20.370 (4) (mm) of the statutes is amended to read:

20.370 (4) (mm) General program operations — federal funds. From the general fund, all moneys received as federal aid for the state's water resources and the state's fishery resources for environmental quality purposes, as authorized by the governor under s. 16.54, for the purposes for which received.

SECTION 329. 20.370 (4) (mq) of the statutes is amended to read:

20.370 (4) (mq) General program operations — environmental fund. From the environmental fund, the amounts in the schedule for administration of environmental activities under chs. 160, 281 and, 283, 285, and 289 to 299.

SECTION 330. 20.370 (4) (mt) of the statutes is amended to read:

20.370 (4) (mt) General program operations — environmental improvement programs; state funds. From the environmental improvement fund, the amounts in the schedule for general program operations under s. 281.58, 281.59, 281.60, 281.61 or, 281.62, or 283.31.

SECTION 331. 20.370 (4) (mu) of the statutes is repealed.

SECTION 333. 20.370 (4) (mz) of the statutes is repealed.

SECTION 334. 20.370 (5) (ac) of the statutes is repealed.

SECTION 334g. 20.370 (5) (cx) of the statutes is amended to read:

20.370 (5) (cx) Recreation aids — all-terrain vehicle and utility terrain vehicle safety enhancement program.

The amounts in the schedule for grants A sum sufficient equal to the amount determined under s. 23.33 (5m) (e) to provide funding to organizations to assist with the all-terrain vehicle and utility terrain vehicle safety enhancement program under s. 23.33 (5m) (d).

SECTION 334m. 20.370 (5) (dy) of the statutes is amended to read:

20.370 (5) (dy) Resource aids — distribution of closed acreage fees. The amounts in the schedule for the payments to municipalities under s. 77.89 (1) (c) and the payments to counties and municipalities under 2015 Wisconsin Act 358, section 100 (2), and under 2017 Wisconsin Act (this act), section 9133 (6p).

SECTION 334q. 20.370 (5) (fv) of the statutes is amended to read:

20.370 (5) (fv) Wolf depredation program. All moneys received from the issuance of wolf harvesting licenses under s. 29.185 and all processing fees received as authorized under s. 29.553 (1) (hr) to be used for the wolf depredation program under s. 29.888 (1m).

SECTION 335. 20.370 (6) (ac) of the statutes is repealed.

SECTION 336. 20.370 (6) (bj) of the statutes is repealed.

SECTION 337. 20.370 (6) (bk) of the statutes is repealed.

SECTION 338. 20.370 (6) (ca) of the statutes is repealed.

SECTION 338d. 20.370 (6) (dq) of the statutes is amended to read:

20.370 (6) (dq) Environmental aids — urban non-point source. Biennially, from the environmental fund, the amounts in the schedule to provide financial assistance for urban nonpoint source water pollution abatement and storm water management under s. 281.66 and for municipal flood control and riparian restoration under s. 281.665 and to make the grants under 2009 Wisconsin Act 28, section 9137 (5q) and (6i) and 2017 Wisconsin Act (this act), section 9133 (8t).

SECTION 338g. 20.370 (6) (gs) of the statutes is created to read:

20.370 (6) (gs) *Village of Plover grant*. Biennially, from the environmental fund, the amounts in the schedule for awarding the grant under 2017 Wisconsin Act (this act), section 9133 (7w).

SECTION 338m. 20.370 (6) (gs) of the statutes, as created by 2017 Wisconsin Act (this act), is repealed.

SECTION 339. 20.370 (7) (aa) of the statutes is amended to read:

20.370 (7) (aa) Resource acquisition and development — principal repayment and interest. A sum sufficient to reimburse s. 20.866 (1) (u) for the payment of principal and interest costs incurred in financing the placement of structures and fill under s. 30.203, in financing the acquisition, construction, development, enlargement, or improvement of state recreation facilities under

s. 20.866 (2) (tp) and (tr), in financing state aids for land acquisition and development of local parks under s. 20.866 (2) (tq), in financing land acquisition activities under s. 20.866 (2) (ts) and (tt), in financing the aid program for dams under s. 20.866 (2) (tx), in financing ice age trail development under s. 20.866 (2) (tw), in financing the Warren Knowles-Gaylord Nelson stewardship program under s. 20.866 (2) (tz) and in financing the Warren Knowles-Gaylord Nelson stewardship 2000 program under s. 20.866 (2) (ta), but not including payments made under par. (ac), and to make payments under an agreement or ancillary arrangement entered into under s. 18.06 (8) (a). Payments may not be made from this appropriation account for principal and interest costs incurred in financing land acquisition and development of state forests under ss. 20.866 (2) (ta) and (tz) until all moneys available under s. 20.370 (7) (au) have been expended.

SECTION 340. 20.370 (7) (ac) of the statutes is repealed.

SECTION 340m. 20.370 (7) (hu) of the statutes is created to read:

20.370 (7) (hu) Parks development – conservation fund. From the conservation fund, from moneys received by the department for state parks activities, as a continuing appropriation, the amounts in the schedule for parks development and maintenance on state parks property.

SECTION 341. 20.370 (7) (jr) of the statutes is amended to read:

20.370 (7) (jr) Rental property and equipment — maintenance and replacement. From the conservation fund, all moneys received by the department from the rental of real property and equipment that are owned by the department, except moneys appropriated under subs. sub. (1) (jr) and (4) (kt), to be used for the maintenance and replacement of this real property and equipment.

SECTION 342. 20.370 (8) (title) of the statutes is repealed and recreated to read:

20.370 (8) (title) Internal services.

SECTION 344. 20.370 (8) (my) of the statutes is created to read:

20.370 (8) (my) Land and property management — federal funds. All moneys received from the federal government for land and property management.

SECTION 345. 20.370 (9) (title) of the statutes is repealed and recreated to read:

20.370 (9) (title) EXTERNAL SERVICES.

SECTION 345m. 20.370 (9) (aq) of the statutes is created to read:

20.370 (9) (aq) Water resources management – lake, river, and invasive species management. The amounts in the schedule for lake and river management and other water resource management activities and for the invasive species program under s. 23.22.

SECTION 346. 20.370 (9) (eg) of the statutes is repealed.

SECTION 347. 20.370 (9) (gb) of the statutes is renumbered 20.370 (1) (gb).

SECTION 348. 20.370 (9) (gh) of the statutes is renumbered 20.370 (1) (gh).

SECTION 350. 20.370 (9) (ma) of the statutes is amended to read:

20.370 (9) (ma) General program operations — state funds. From the general fund, the amounts in the schedule for communications, customer services and, aids administration, watershed management, and environmental analysis and sustainability.

SECTION 351. 20.370 (9) (mh) of the statutes is amended to read:

20.370 (9) (mh) General program operations — stationary sources. From the general fund, from the moneys received from fees under s. 285.69 (2) (a) and (e) and (2e), the amounts in the schedule for customer service, communications and aids administration for the operation permit program under ch. 285 and s. 299.15 external relations.

SECTION 352. 20.370 (9) (mi) of the statutes is amended to read:

20.370 (9) (mi) General program operations — private and public sources. From the general fund, all moneys received from public or private sources, other than state agencies and the federal government, for facilities, materials, or services provided by the department related to customer service and external relations, to pay for costs and expenses associated with those facilities, materials, or services.

SECTION 353. 20.370 (9) (mk) of the statutes is amended to read:

20.370 (9) (mk) General program operations — service funds. From the general fund, all moneys received by the department from the department and from other state agencies for facilities, materials, or services provided by the department relating to communications, customer services, licensing and aids administration external relations.

SECTION 354. 20.370 (9) (mm) of the statutes is amended to read:

20.370 (9) (mm) General program operations — federal funds. From the general fund, all moneys received as federal aid for communications, customer services and aids administration external relations, as authorized by the governor under s. 16.54, for the purposes for which received.

SECTION 355. 20.370 (9) (mq) of the statutes is amended to read:

20.370 (9) (mq) General program operations — mobile sources. From the petroleum inspection fund, the amounts in the schedule for customer services, communications and, aids administration, and environmental analysis and sustainability for the mobile source air pollution program under ch. 285.

SECTION 356. 20.370 (9) (mr) of the statutes is created to read:

20.370 (9) (mr) General program operations — non-point source. From the environmental fund, the amounts in the schedule for performing the duties of the department related to nonpoint source pollution standards and enforcement under ss. 281.16, 281.19, 281.20, and 281.65.

SECTION 357. 20.370 (9) (mt) of the statutes is amended to read:

20.370 (9) (mt) Aids administration — environmental improvement programs; state funds. From the environmental improvement fund, the amounts in the schedule for the administration of ss. 281.58, 281.60, 281.61 and, 281.62, and 283.31.

SECTION 358. 20.370 (9) (mu) of the statutes is amended to read:

20.370 (9) (mu) General program operations — state funds. The amounts in the schedule for communications, customer services, licensing, registration and, aids administration, and environmental analysis and sustainability.

SECTION 359. 20.370 (9) (mv) of the statutes is amended to read:

20.370 (9) (mv) General program operations — environmental fund. From the environmental fund, the amounts in the schedule for communications, customer services and, aids administration, watershed management, and environmental analysis and sustainability.

SECTION 359m. 20.380 (1) (b) of the statutes is amended to read:

20.380 (1) (b) Tourism marketing; general purpose revenue. Biennially, the amounts in the schedule for tourism marketing service expenses and the execution of the functions under ss. 41.11 (4) and 41.17 and the grant under 2017 Wisconsin Act (this act), section 9144 (3t). In each fiscal year, the department shall expend for tourism marketing service expenses and the execution of the functions under ss. 41.11 (4) and 41.17 an amount that bears the same proportion to the amount in the schedule for the fiscal year, minus, for fiscal year 2017-18, the amount expended for the grant under 2017 Wisconsin Act (this act), section 9144 (3t), as the amount expended under par. (kg) in that fiscal year bears to the amount in the schedule for par. (kg) for that fiscal year. Of the amounts under this paragraph, not more than 50 percent shall be used to match funds allocated under s. 41.17 by private or public organizations for the joint effort marketing of tourism with the state.

SECTION 359n. 20.380 (1) (b) of the statutes, as affected by 2017 Wisconsin Act (this act), is amended to read:

20.380 (1) (b) *Tourism marketing; general purpose revenue*. Biennially, the amounts in the schedule for tourism marketing service expenses and the execution of

the functions under ss. 41.11 (4) and 41.17 and the grant under 2017 Wisconsin Act (this act), section 9144 (3t). In each fiscal year, the department shall expend for tourism marketing service expenses and the execution of the functions under ss. 41.11 (4) and 41.17 an amount that bears the same proportion to the amount in the schedule for the fiscal year, minus, for fiscal year 2017-18, the amount expended for the grant under 2017 Wisconsin Act (this act), section 9144 (3t), as the amount expended under par. (kg) in that fiscal year bears to the amount in the schedule for par. (kg) for that fiscal year. Of the amounts under this paragraph, not more than 50 percent shall be used to match funds allocated under s. 41.17 by private or public organizations for the joint effort marketing of tourism with the state.

SECTION 359nm. 20.395 (2) (cq) of the statutes is amended to read:

20.395 (2) (cq) Harbor assistance, state funds. As a continuing appropriation, the amounts in the schedule for harbor assistance under s. 85.095 (2) (a), for administration of the harbor assistance program under s. 85.095 and for grants under 1999 Wisconsin Act 9, section 9150 (4f), 2013 Wisconsin Act 20, section 9145 (4i) and (4u), and 2015 Wisconsin Act 55, section 9145 (1c), and 2017 Wisconsin Act (this act), section 9145 (4d).

SECTION 359nq. 20.395 (2) (dq) of the statutes is amended to read:

20.395 (2) (dq) Aeronautics assistance, state funds. As a continuing appropriation, the amounts in the schedule for the state's share of airport projects under ss. 114.34 and 114.35; for developing air marking and other air navigational facilities; for administration of the powers and duties of the secretary of transportation under s. 114.31; for costs associated with aeronautical activities under s. 114.31, except for the program under s. 114.31 (3) (b); and for the administration of other aeronautical activities, except aircraft registration under s. 114.20, authorized by law; and for the grants under 2017 Wisconsin Act (this act), section 9145 (3i).

SECTION 359p. 20.395 (2) (fq) of the statutes is created to read:

Vetoed In Part

Vetoed In Part Vetoed In Part

20.395 (2) (fq) Local transportation facility improvement assistance, state funds. All moneys transferred under 2017 Wisconsin Act (this act), section 9145 (4w), for providing public access roads to navigable waters, for the purposes of ss. 84.27 and 84.28, and for improving transportation facilities, including facilities funded under applicable federal acts or programs, that are not state trunk or connecting highways.

SECTION 360. 20.395 (3) (et) of the statutes is amended to read:

20.395 (3) (et) Intelligent transportation systems and traffic control signals, state funds. As a continuing appropriation, the amounts in the schedule for the installation, replacement, or rehabilitation of traffic control signals and intelligent transportation systems. No moneys may be encumbered from this appropriation account after June 30, 2019 2021.

SECTION 361. 20.395 (3) (eu) of the statutes is amended to read:

20.395 (3) (eu) Intelligent transportation systems and traffic control signals, local funds. All moneys received from any local unit of government or other sources for the installation, replacement, or rehabilitation of traffic control signals and intelligent transportation systems, for such purposes. No moneys may be encumbered from this appropriation account after June 30, 2019 2021.

SECTION 362. 20.395 (3) (ez) of the statutes is amended to read:

20.395 (3) (ez) Intelligent transportation systems and traffic control signals, federal funds. All moneys received from the federal government for the installation, replacement, or rehabilitation of traffic control signals and intelligent transportation systems, for such purposes. No moneys may be encumbered from this appropriation account after June 30, 2019 2021.

SECTION 362m. 20.395 (4) (ab) of the statutes is cre- Vetoed ated to read:

In Part

20.395 (4) (ab) Transportation projects commission. Biennially, from the general fund, the amounts in the schedule for the general program operations of the transportation projects commission.

SECTION 362n. 20.395 (4) (aq) of the statutes is Vetoed amended to read:

In Part

20.395 (4) (aq) Departmental management and operations, state funds. The amounts in the schedule for departmental planning and administrative activities and the administration and management of departmental programs except those programs under subs. (2) (bq), (cq), and (dq) and (3) (iq), including not less than \$220,000 in each fiscal year to reimburse the department of justice for legal services provided the department under s. 165.25 (4) (a) and including activities related to the transportation employment and mobility program under s. 85.24 that are not funded from the appropriation under sub. (1) (bs), (bv) or (bx) and the scholarship and loan repayment incentive grant program under s. 85.107 and to match federal funds for mass transit planning and to pay for the removal and disposal of deer killed by vehicles on state trunk highways under s. 84.07 (7).

SECTION 362p. 20.395 (5) (da) of the statutes is created to read:

20.395 (5) (da) State traffic patrol equipment, general fund. From the general fund, the amounts in the schedule for in-vehicle video camera equipment and tactical vests and helmets for the state traffic patrol.

SECTION 363. 20.395 (5) (dg) of the statutes is amended to read:

20.395 (5) (dg) Escort, security, and traffic enforcement services, state funds. From the general fund, all moneys received under ss. 348.105 and s. 348.26 (2) for motor carrier escort services and under s. 85.51 for security and traffic enforcement services, for those purposes.

SECTION 365. 20.410 (1) (f) of the statutes is amended to read:

20.410 (1) (f) Energy costs; energy-related assessments. The amounts in the schedule to be used at state correctional institutions to pay for utilities and for fuel, heat and air conditioning, to pay assessments levied by the department of administration under s. 16.847 (3) for costs incurred and savings generated at departmental facilities, and to pay costs incurred by or on behalf of the department under ss. 16.858 and 16.895. No moneys may be expended from this appropriation for the purposes specified in par. (fm).

SECTION 366. 20.410 (1) (fm) of the statutes is created to read:

20.410 (1) (fm) Electric energy derived from renewable resources. The amounts in the schedule for the premium cost incurred for the generation or purchase of electric energy derived from renewable resources under s. 16.75 (12).

SECTION 367m. 20.410 (3) (c) of the statutes is amended to read:

20.410 (3) (c) Reimbursement claims of counties containing juvenile correctional facilities. The amounts in the schedule A sum sufficient to pay all valid claims made by county clerks of counties containing state juvenile correctional facilities as provided in s. 16.51 (7).

SECTION 369. 20.435 (1) (a) of the statutes is amended to read:

20.435 (1) (a) General program operations. The amounts in the schedule for general program operations, including public health services regulation, administration, and field services, and for the operation of the council on physical disabilities under s. 46.29.

SECTION 370. 20.435 (1) (b) of the statutes is amended to read:

20.435 (1) (b) General aids and local assistance. The amounts in the schedule for aids and local assistance relating to public health services and for grants for community programs under s. 46.48. Notwithstanding ss. 20.001 (3) (a) and 20.002 (1), the department may transfer funds between fiscal years under this paragraph. Except as otherwise provided in this paragraph, all funds allocated but not encumbered by December 31 of each year lapse to the general fund on the next January 1 unless carried forward to the next calendar year by the joint committee on finance.

SECTION 370r. 20.435 (1) (cf) of the statutes is created to read:

20.435 (1) (cf) *Communicable disease control and prevention*. As a continuing appropriation, the amounts in the schedule for communicable disease control and prevention under s. 252.185.

SECTION 371r. 20.435 (1) (fi) of the statutes is created to read:

20.435 (1) (fi) Allied health professional education and training grants. Biennially, the amounts in the schedule for grants to hospitals, health systems, and educational entities that form health care education and training consortia for training allied health professionals under s. 146.616.

SECTION 371t. 20.435 (1) (fk) of the statutes is created to read:

20.435 (1) (fk) *Grants to establish advanced practice clinician training programs*. Biennially, the amounts in the schedule for grants to hospitals under s. 146.615.

SECTION 372. 20.435 (1) (g) of the statutes is renumbered 20.395 (5) (gj) and amended to read:

20.395 (5) (gj) *Payments to Donate Life Wisconsin.* All From the general fund, all moneys received under ss. 341.14 (6r) (b) 11. and 343.21 (1) (o), for payments to Donate Life Wisconsin the nonprofit organization that promotes organ and tissue donation under s. 250.17 (1) 341.14 (8w).

SECTION 373. 20.435 (1) (gi) of the statutes is renumbered 20.395 (5) (gi) and amended to read:

20.395 (5) (gi) Payments to the Wisconsin Women's Health Foundation. All From the general fund, all moneys received under s. 341.14 (6r) (b) 10. for payments to the Wisconsin Women's Health Foundation, Inc., under s. 250.16 341.14 (8v).

SECTION 374. 20.435 (1) (n) of the statutes is amended to read:

20.435 (1) (n) Federal program operations. All moneys received from the federal government or any of its agencies for the state administration of continuing programs relating to public health services, for the purposes for which received, and for services of resource centers under s. 46.283.

SECTION 375. 20.435 (2) (f) of the statutes is amended to read:

20.435 (2) (f) Energy costs; energy-related assessments. The amounts in the schedule to be used at mental health institutes and centers for the developmentally disabled to pay for utilities and for fuel, heat and air conditioning, to pay assessments levied by the department of administration under s. 16.847 (3) for costs incurred and savings generated at departmental facilities, and to pay costs incurred by or on behalf of the department under ss. 16.858 and 16.895. No moneys may be expended from this appropriation for the purposes specified in par. (fm).

SECTION 376. 20.435 (2) (fm) of the statutes is created to read:

20.435 (2) (fm) Electric energy derived from renewable resources. The amounts in the schedule for the premium cost incurred for the generation or purchase of electric energy derived from renewable resources under s. 16.75 (12).

SECTION 377. 20.435 (2) (gk) of the statutes, as affected by 2015 Wisconsin Act 55, is amended to read:

20.435 (2) (gk) Institutional operations and charges.

The amounts in the schedule for care, other than under s. 51.06 (1r), provided by the centers for the developmentally disabled, to reimburse the cost of providing the services and to remit any credit balances to county departments that occur on and after July 1, 1978, in accordance with s. 51.437 (4rm) (c); for care, other than under s. 46.043, provided by the mental health institutes, to reimburse the cost of providing the services and to remit any credit balances to county departments that occur on and after January 1, 1979, in accordance with s. 51.42 (3) (as) 2.; for maintenance of state-owned housing at centers for the developmentally disabled and mental health institutes; for repair or replacement of property damaged at the mental health institutes or at centers for the developmentally disabled; and for reimbursing the total cost of using, producing, and providing services, products, and care; and to transfer to the appropriation account under sub. (5) (kp) \$450,000 in fiscal year 2018–19 for funding peer-run respite centers for veterans . All moneys received as payments from medical assistance on and after August 1, 1978; as payments from all other sources including other payments under s. 46.10 and payments under s. 51.437 (4rm) (c) received on and after July 1, 1978; as medical assistance payments, other payments under s. 46.10, and payments under s. 51.42 (3) (as) 2. received on and after January 1, 1979; as payments for the rental of state-owned housing and other institutional facilities at centers for the developmentally disabled and mental health institutes; for the sale of electricity, steam, or chilled water; as payments in restitution of property damaged at the mental health institutes or at centers for the developmentally disabled; for the sale of surplus property, including vehicles, at the mental health institutes or at centers for the developmentally disabled; and for other services, products, and care shall be credited to this appropriation, except that any payment under s. 46.10 received for the care or treatment of patients admitted under s. 51.10, 51.15, or 51.20 for which the state is liable under s. 51.05 (3), of forensic patients committed under ch. 971 or 975, admitted under ch. 975, or transferred under s. 51.35 (3), or of patients transferred from a state prison under s. 51.37 (5), to the Mendota Mental Health Institute or the Winnebago Mental Health Institute shall be treated as general purpose revenue earned, as defined under s. 20.001 (4); and except that

Vetoed In Part

Vetoed In Part

SECTION 377b. 20.435 (2) (gk) of the statutes, as affected by 2017 Wisconsin Act (this act), is amended to read:

moneys received under s. 51.06 (6) may be expended

only as provided in s. 13.101 (17).

20.435 (2) (gk) *Institutional operations and charges*. The amounts in the schedule for care, other than under s. 51.06 (1r), provided by the centers for the developmentally disabled, to reimburse the cost of providing the services and to remit any credit balances to county depart-

ments that occur on and after July 1, 1978, in accordance with s. 51.437 (4rm) (c); for care, other than under s. 46.043, provided by the mental health institutes, to reimburse the cost of providing the services and to remit any credit balances to county departments that occur on and after January 1, 1979, in accordance with s. 51.42 (3) (as) 2.; for maintenance of state-owned housing at centers for the developmentally disabled and mental health institutes; for repair or replacement of property damaged at the mental health institutes or at centers for the developmentally disabled; and for reimbursing the total cost of using, producing, and providing services, products, and care; and to transfer to the appropriation account under sub. (5) (kp) \$450,000 in fiscal year 2018-19 for funding peer-run respite centers for veterans. All moneys received as payments from medical assistance on and after August 1, 1978; as payments from all other sources including other payments under s. 46.10 and payments under s. 51.437 (4rm) (c) received on and after July 1, 1978; as medical assistance payments, other payments under s. 46.10, and payments under s. 51.42 (3) (as) 2. received on and after January 1, 1979; as payments for the rental of state-owned housing and other institutional facilities at centers for the developmentally disabled and mental health institutes; for the sale of electricity, steam, or chilled water; as payments in restitution of property damaged at the mental health institutes or at centers for the developmentally disabled; for the sale of surplus property, including vehicles, at the mental health institutes or at centers for the developmentally disabled; and for other services, products, and care shall be credited to this appropriation, except that any payment under s. 46.10 received for the care or treatment of patients admitted under s. 51.10, 51.15, or 51.20 for which the state is liable under s. 51.05 (3), of forensic patients committed under ch. 971 or 975, admitted under ch. 975, or transferred under s. 51.35 (3), or of patients transferred from a state prison under s. 51.37 (5), to the Mendota Mental Health Institute or the Winnebago Mental Health Institute shall be treated as general purpose revenue earned, as defined under s. 20.001 (4); and except that moneys received under s. 51.06 (6) may be expended only as provided in s. 13.101 (17).

SECTION 377h. 20.435 (2) (gz) of the statutes is cre- **Vetoed** ated to read:

20.435 (2) (gz) Costs of housing persons on supervised release. All moneys received under s. 980.08 (4) (dm) 4. for payment of costs associated with housing persons on supervised release.

SECTION 378. 20.435 (4) (a) of the statutes is amended to read:

20.435 (4) (a) General program operations. The amounts in the schedule for general program operations, including health care financing regulation, administration, field services, operation of the council on physical In Part

disabilities under s. 46.29, and medical assistance eligibility determinations under s. 49.45 (2) (a) 3.

SECTION 378k. 20.435 (4) (bf) of the statutes is created to read:

20.435 (4) (bf) *Graduate medical training support grants*. As a continuing appropriation, the amounts in the schedule to award grants to support graduate medical training programs under s. 146.64.

SECTION 379. 20.435 (4) (hs) of the statutes is renumbered 20.435 (1) (hs).

SECTION 379ar. 20.435 (4) (L) of the statutes is amended to read:

20.435 (4) (L) Fraud and error reduction. All moneys received as the state's share of the recovery of overpayments and incorrect payments under ss. 49.497 (1) and (1m), 49.793 (2) (a), and 49.847, all moneys received from counties and tribal governing bodies as a result of any error reduction activities under ss. 49.197 and 49.845, and all moneys credited to this appropriation account under ss. 49.497 (2) (b), 49.793 (2) (b), and 49.847 (3) (b), and 20 percent of moneys received as a result of the audit under 2017 Wisconsin Act (this act), section 9120 (6p), for any contracts under s. 49.197 (5), for any activities to reduce error and fraud under s. 49.845, to pay federal sanctions under the food stamp program, and for food stamp reinvestment activities under reinvestment agreements with the federal department of agriculture that are designed to improve the food stamp program.

SECTION 379as. 20.435 (4) (L) of the statutes, as affected by 2017 Wisconsin Act (this act), is amended to read:

20.435 (4) (L) Fraud and error reduction. All moneys received as the state's share of the recovery of overpayments and incorrect payments under ss. 49.497 (1) and (1m), 49.793 (2) (a), and 49.847, all moneys received from counties and tribal governing bodies as a result of any error reduction activities under ss. 49.197 and 49.845, and all moneys credited to this appropriation account under ss. 49.497 (2) (b), 49.793 (2) (b), and 49.847 (3) (b), and 20 percent of moneys received as a result of the audit under 2017 Wisconsin Act (this act), section 9120 (6p), for any contracts under s. 49.197 (5), for any activities to reduce error and fraud under s. 49.845, to pay federal sanctions under the food stamp program, and for food stamp reinvestment activities under reinvestment agreements with the federal department of agriculture that are designed to improve the food stamp program.

SECTION 379b. 20.435 (4) (x) of the statutes is amended to read:

20.435 (4) (x) Medical Assistance trust fund; children's services; Badger Care health care program. From the Medical Assistance trust fund, all moneys received under s. 49.45 (39) (bm) for reducing waiting lists for

children's long-term care services or other programs benefitting children and all moneys received for the Badger Care health care program under s. 49.665.

SECTION 379c. 20.435 (4) (xc) of the statutes is amended to read:

20.435 (4) (xc) Hospital assessment fund; hospital payments. From the hospital assessment fund, the amounts in the schedule all moneys received from the assessment under s. 50.38 (2) (a), except amounts transferred to the Medical Assistance trust fund under s. 50.38 (8), to reimburse eligible hospitals for services provided under the Medical Assistance Program under subch. IV of ch. 49, make payments to health maintenance organizations under s. 49.45 (59), provide supplemental funds to rural hospitals under s. 49.45 (5m) (am), make supplemental payments to Level I adult trauma centers under s. 49.45 (6y) (ap), make supplemental payments to hospitals based on performance under s. 49.45 (6y) (ar), and make refunds under s. 50.38 (6), and make the transfer under s. 50.38 (8).

SECTION 379d. 20.435 (4) (xe) of the statutes is amended to read:

20.435 (4) (xe) Critical access hospital assessment fund; hospital payments. From the critical access hospital assessment fund, all moneys received from the assessment under s. 50.38 (2) (b), except moneys appropriated under s. 20.285 (1) (qe) and (qj) and except amounts transferred to the Medical Assistance trust fund under s. 50.38 (10), to make payments to critical access hospitals required under s. 49.45 (3) (e) 12. for services provided under the Medical Assistance Program under subch. IV of ch. 49; and to make refunds under s. 50.38 (6m); and to make the transfer under s. 50.38 (10).

SECTION 379f. 20.435 (5) (bc) of the statutes is amended to read:

20.435 (5) (bc) Grants for community programs. The amounts in the schedule for grants for and contracts to establish community programs under s. 46.48, for pretrial intoxicated driver intervention grants under s. 51.49, to award a grant under 2017 Wisconsin Act (this act), section 9120 (6d), and for opioid treatment programs under s. 51.422. Notwithstanding ss. 20.001 (3) (a) and 20.002 (1), the department may transfer funds between fiscal years under this paragraph. Except for amounts authorized to be carried forward under s. 46.48 and as otherwise provided in this paragraph, all funds allocated but not encumbered by December 31 of each year lapse to the general fund on the next January 1 unless carried forward to the next calendar year by the joint committee on finance. Notwithstanding ss. 20.001 (3) (a) and 20.002 (1), the department shall transfer from this appropriation account to the appropriation account for the department of children and families under s. 20.437 (2) (dz) funds allocated by the department under s. 46.48 (30) but unexpended on June 30 of each year.

SECTION 379g. 20.435 (5) (bc) of the statutes, as affected by 2017 Wisconsin Act (this act), is amended to read:

20.435 (5) (bc) Grants for community programs. The amounts in the schedule for grants for and contracts to establish community programs under s. 46.48, for pretrial intoxicated driver intervention grants under s. 51.49, to award a grant under 2017 Wisconsin Act (this act), section 9120 (6d), and for opioid treatment programs under s. 51.422. Notwithstanding ss. 20.001 (3) (a) and 20.002 (1), the department may transfer funds between fiscal years under this paragraph. Except for amounts authorized to be carried forward under s. 46.48 and as otherwise provided in this paragraph, all funds allocated but not encumbered by December 31 of each year lapse to the general fund on the next January 1 unless carried forward to the next calendar year by the joint committee on finance. Notwithstanding ss. 20.001 (3) (a) and 20.002 (1), the department shall transfer from this appropriation account to the appropriation account for the department of children and families under s. 20.437 (2) (dz) funds allocated by the department under s. 46.48 (30) but unexpended on June 30 of each year.

Vetoed In Part

SECTION 379j. 20.435 (5) (kd) of the statutes is created to read:

20.435 (5) (kd) Youth crisis stabilization facilities. The amounts in the schedule for the purposes of providing grants to youth crisis stabilization facilities under s. 51.042. All moneys transferred by the joint committee on finance through the appropriation account under s. 20.865 (4) (g) shall be credited to this appropriation

SECTION 379k. 20.435 (5) (kd) of the statutes, as created by 2017 Wisconsin Act (this act), is repealed.

SECTION 379p. 20.435 (5) (kp) of the statutes is created to read:

20.435 (5) (kp) Veterans peer-run respite center. All

In Part

moneys transferred from the appropriation account under sub. (2) (gk) to make payments to an organization that establishes a peer-run respite center that provides services to veterans.

SECTION 379r. 20.435 (5) (kp) of the statutes, as created by 2017 Wisconsin Act (this act), is repealed.

SECTION 380. 20.435 (6) (jm) of the statutes is amended to read:

20.435 (6) (jm) Licensing and support services. The amounts in the schedule for the purposes specified in ss. 48.685 (2) (am) and (b) 1., (3) (a), (am), and (b), and (bm), and (5) (a), 48.686 (2) (am), (3) (am) and (bm), and (5) (a), 49.45 (47), 50.02 (2), 50.025, 50.065 (2) (am) and (b) 1., (3) (a) and (b), and (5), 50.13, 50.135, 50.36 (2), 50.49 (2) (b), 50.495, 50.52 (2) (a), 50.57, 50.981, and 146.40 (4r) (b) and (er), and subch. VI of ch. 50 and to conduct health facilities plan and rule development activities, for accrediting nursing homes, convalescent homes, and homes for the aged, to conduct capital construction

and remodeling plan reviews under ss. 50.02 (2) (b) and 50.36 (2), and for the costs of inspecting, licensing or certifying, and approving facilities, issuing permits, and providing technical assistance, that are not specified under any other paragraph in this subsection. All moneys received under ss. 48.685 (8), 48.686 (2) (ag), 49.45 (42) (c), 49.45 (47) (c), 50.02 (2), 50.025, 50.065 (8), 50.13, 50.36 (2), 50.49 (2) (b), 50.495, 50.52 (2) (a), 50.57, 50.93 (1) (c), and 50.981, all moneys received from fees for the costs of inspecting, licensing or certifying, and approving facilities, issuing permits, and providing technical assistance, that are not specified under any other paragraph in this subsection, and all moneys received under s. 50.135 (2) shall be credited to this appropriation account.

SECTION 381. 20.435 (7) (bg) of the statutes is renumbered 20.435 (1) (bg).

SECTION 382. 20.435 (7) (bm) of the statutes is renumbered 20.435 (1) (bm).

SECTION 383. 20.435 (7) (br) of the statutes is renumbered 20.435 (1) (br).

SECTION 384. 20.435 (7) (bt) of the statutes is amended to read:

20.435 (7) (bt) Early intervention services for infants and toddlers with disabilities. As a continuing appropriation, the amounts in the schedule for the early intervention services under s. 51.44, including services described under s. 49.45 (54) (c).

SECTION 385. 20.435 (7) (c) of the statutes is renumbered 20.435 (1) (cx).

SECTION 386. 20.435 (7) (cg) of the statutes is renumbered 20.435 (1) (cg).

SECTION 387. 20.435 (7) (d) of the statutes is renumbered 20.435 (1) (da).

SECTION 388. 20.435 (7) (dh) of the statutes is renumbered 20.435 (1) (dh).

SECTION 389. 20.435 (7) (kc) of the statutes is renumbered 20.435 (1) (kc).

SECTION 390. 20.435 (7) (kn) of the statutes is renumbered 20.435 (1) (kn).

SECTION 391. 20.435 (7) (ky) of the statutes is amended to read:

20.435 (7) (ky) Interagency and intra-agency aids. Except as provided in par. sub. (1) (kc), all moneys received from other state agencies and all moneys received by the department from the department for aids to individuals and organizations relating to long-term care services, for the purposes for which received.

SECTION 392. 20.435 (7) (kz) of the statutes is amended to read:

20.435 (7) (kz) Interagency and intra-agency local assistance. Except as provided in par. sub. (1) (kn), all moneys received from other state agencies and all moneys received by the department from the department for local assistance relating to long-term care services, for the purposes for which received.

Vetoed

Vetoed In Part Vetoed In Part Vetoed In Part **SECTION 392c.** 20.435 (8) (a) of the statutes is amended to read:

20.435 (8) (a) General program operations. The amounts in the schedule for executive, management and policy and budget services and activities, and for travel reimbursement for families with firsthand experience with children's mental health services who participate in meetings arranged by the office of children's mental health.

SECTION 392g. 20.437 (1) (e) of the statutes is amended to read:

20.437 (1) (e) Services for sex-trafficking victims. The Biennially, the amounts in the schedule for treatment and services for sex-trafficking victims under s. 48.48 (19).

SECTION 393L. 20.437 (1) (er) of the statutes is created to read:

20.437 (1) (er) *Grants for services for homeless and runaway youth.* The amounts in the schedule for grants to support programs that provide services for homeless and runaway youth under s. 49.1385.

SECTION 394. 20.437 (1) (jm) of the statutes is amended to read:

20.437 (1) (jm) *Licensing activities*. All moneys received from licensing activities under ss. 48.60, 48.62, 48.625, and 938.22 (7), and from fees under ss. 48.615, 48.625, 48.685 (8), and 938.22 (7) (b) and (c), and from fees under s. 48.685 (8) charged to entities other than child care centers or child care providers, for the costs of licensing child welfare agencies under s. 48.60, foster homes under s. 48.62, group homes under s. 48.625, and shelter care facilities under s. 938.22 (7) and for the purposes specified in s. 48.685 (2) (am) and (b) 1., (3) (a) and (b), and (5) (a) with respect to those entities.

SECTION 395. 20.437 (1) (kp) of the statutes is repealed.

SECTION 396. 20.437 (1) (kz) of the statutes is amended to read:

20.437 (1) (kz) Interagency and intra-agency aids; tribal placements and guardianships. The amounts in the schedule to be used for unexpected or unusually high-cost out-of-home care placements of Indian children by tribal courts, other than placements to which par. (kp) applies including placements of Indian juveniles who have been adjudicated delinquent, and for subsidized guardianship payments under s. 48.623 (1) or (6) for guardianships of Indian children ordered by tribal courts. All moneys transferred from the appropriation account under s. 20.505 (8) (hm) 21. shall be credited to this appropriation account. Notwithstanding s. 20.001 (3) (a), the unencumbered balance on June 30 of each year shall revert to the appropriation account under section 20.505 (8) (hm).

SECTION 397. 20.437 (2) (jn) of the statutes is amended to read:

20.437 (2) (jn) Child care licensing and certification activities. All moneys received from licensing activities under s. 48.65, from certifying activities under s. 48.651, and from fees under ss. 48.65 (3) and, 48.651 (2), and from fees under s. 48.685 (8) charged to child care centers and child care providers 48.686 (2) (ag) for the costs of licensing child care centers under s. 48.65 and of certifying child care providers under s. 48.651 and for the purposes specified in s. 48.685 48.686 (2) (am), (ar), and (b) 1. and 2., (3) (am) and (bm), and (5) (a) with respect to child care centers and child care providers.

SECTION 398. 20.445 (1) (b) of the statutes is amended to read:

20.445 (1) (b) Workforce training; programs, grants, and services. As a continuing appropriation, the amounts in the schedule for the apprenticeship completion award program under s. 106.05 (2), local youth apprenticeship grants under s. 106.13 (3m), youth summer jobs programs under s. 106.18, employment transit assistance grants under s. 106.26, workforce training grants and services under s. 106.27 (1), (1g), (1j), and (1r), and teacher development program grants under s. 106.272, career and technical education incentive grants under s. 106.273, technical education equipment grants under s. 106.275, apprentice programs under subch. I of ch. 106, and the grant described in 2017 Wisconsin Act (this act), section 9151 (5q).

SECTION 398b. 20.445 (1) (b) of the statutes, as affected by 2017 Wisconsin Act (this act), is amended to read:

20.445 (1) (b) Workforce training; programs, grants, and services. As a continuing appropriation, the amounts in the schedule for the local youth apprenticeship grants under s. 106.13 (3m), youth summer jobs programs under s. 106.18, employment transit assistance grants under s. 106.26, workforce training grants and services under s. 106.27 (1), (1g), (1j), and (1r), teacher development program grants under s. 106.272, career and technical education incentive grants under s. 106.273, technical education equipment grants under s. 106.275, and apprentice programs under subch. I of ch. 106, and the grant described in 2017 Wisconsin Act (this act), section 9151 (5q).

SECTION 399. 20.445 (1) (bm) of the statutes is amended to read:

20.445 (1) (bm) Workforce training; administration. Biennially, the amounts in the schedule for the administration of the apprenticeship completion award program under s. 106.05 (2), the local youth apprenticeship grant program under s. 106.13 (3m), the youth summer jobs program under s. 106.18, the employment transit assistance grant program under s. 106.26, the workforce training program under s. 106.27, and the teacher development program grants under s. 106.272, the career and technical education incentive grant program under s.

106.273, the technical education equipment grant program under s. 106.275, and the apprentice programs under subch. I of ch. 106.

SECTION 400. 20.445 (1) (d) of the statutes is created to read:

20.445 (1) (d) *Reimbursement for tuition payments*. The amounts in the schedule to reimburse school districts for payments under s. 118.55 (5) (e) 2.

SECTION 401. 20.445 (1) (g) of the statutes is amended to read:

20.445 (1) (g) Gifts and grants. All Except as provided in par. (gr), all moneys received as gifts or grants to carry out the purposes for which made.

SECTION 405. 20.445 (5) (n) of the statutes is amended to read:

20.445 (5) (n) Federal program aids and operations. All moneys received from the federal government, as authorized by the governor under s. 16.54, for the state administration of continuing programs and all federal moneys received for the purchase of goods and services under ch. 47 and for the purchase of vocational rehabilitation programs for individuals and organizations, to be expended for the purposes specified. The department shall, in each fiscal year, transfer \$600,000 of the moneys from the account under this paragraph to the appropriation account under s. 20.435 (7) (1) (kc).

SECTION 408c. 20.455 (2) (hd) of the statutes is amended to read:

20.455 (2) (hd) Internet crimes against children. All moneys transferred under 2015 Wisconsin Act 369, section 12m (1) and under 2017 Wisconsin Act (this act), section 9228 (1p) shall be credited to this appropriation account for criminal investigative operations and law enforcement relating to Internet crimes against children, prosecution of Internet crimes against children, and activities of state and local Internet crimes against children task forces.

SECTION 408e. 20.455 (2) (hm) of the statutes is renumbered 20.465 (3) (L) and amended to read:

20.465 (3) (L) Public safety interoperable communication system; general usage fees. The amounts in the schedule to operate a statewide public safety interoperable communication system. All moneys received from users as fees under s. 165.25 (17) (bm) 323.29 (3) (b) 2. shall be credited to this appropriation account.

SECTION 408f. 20.455 (2) (jb) of the statutes is amended to read:

20.455 (2) (jb) Crime laboratory equipment and supplies. The amounts in the schedule for the maintenance, repair, upgrading, and replacement costs of the laboratory equipment, for supplies used to maintain, repair, upgrade, and replace that equipment, and for operating costs, in the state and regional crime laboratories. All moneys transferred from par. (Lm) (Lp) for the purpose of this appropriation shall be credited to this appropriation. Notwithstanding s. 20.001 (3) (a), the unencum-

bered balance on June 30 of each year shall be transferred to the appropriation account under par. (Lm) (Lp).

SECTION 408j. 20.455 (2) (jc) of the statutes is created to read:

20.455 (2) (jc) Law enforcement overtime grants. The amounts in the schedule for grants under s. 165.986 (7). All moneys transferred under 2017 Wisconsin Act (this act), section 9228 (9p) shall be credited to this appropriation account.

SECTION 408m. 20.455 (2) (jd) of the statutes is created to read:

20.455 (2) (jd) Alternatives to incarceration grant program. The amounts in the schedule to provide grants under s. 165.95 (2) to counties that are not a recipient of a grant under the alternatives to incarceration grant program on the effective date of this paragraph [LRB inserts date]. All moneys transferred under 2017 Wisconsin Act (this act), section 9228 (15t) shall be credited to this appropriation account.

SECTION 408n. 20.455 (2) (ka) of the statutes is renumbered 20.465 (3) (ks) and amended to read:

20.465 (3) (ks) *Public safety interoperable communication system; state fees*. The amounts in the schedule to operate a statewide public safety interoperable communication system. All moneys received from public safety agencies that are state agencies as fees under s. 165.25 (17) (am) 323.29 (3) (b) 1. shall be credited to this appropriation account.

SECTION 408p. 20.455 (2) (kd) of the statutes is amended to read:

20.455 (2) (kd) *Drug law enforcement, crime laboratories, and genetic evidence activities*. The amounts in the schedule for activities relating to drug law enforcement, drug law violation prosecution assistance, criminal investigative operations, and activities of the state and regional crime laboratories. All moneys transferred to this appropriation account from the appropriation account under par. (Lm) (Lp) shall be credited to this appropriation account. Notwithstanding s. 20.001 (3) (a), the unencumbered balance on June 30 of each year shall revert to the appropriation account under par. (Lm) (Lp).

SECTION 408s. 20.455 (2) (ki) of the statutes is renumbered 20.465 (3) (km).

SECTION 408t. 20.455 (2) (Lm) of the statutes is amended to read:

20.455 (2) (Lm) Crime laboratories; deoxyribonucleic acid analysis. All moneys received from crime laboratories and drug law enforcement surcharges authorized under s. 165.755 and deoxyribonucleic acid analysis surcharges authorized under s. 973.046 (1r) to provide deoxyribonucleic acid analysis, to administer the appropriation account under par. (Lp) for administering s. 165.77, to pay and for paying for the costs of mailing and materials under s. 165.76 for the submission of biological specimens by the departments of corrections and

health services and by persons in charge of law enforcement and tribal law enforcement agencies, to transfer to the appropriation account under par. (jb) the amounts in the schedule under par. (jb), and to transfer to the appropriation accounts under par. (kd) and s. 20.475 (1) (km) the amounts in the schedule under par. (kd) and s. 20.475 (1) (km). All moneys transferred to this appropriation account from the appropriation account under par. (Lp) shall be credited to this appropriation account.

SECTION 408v. 20.455 (2) (Lp) of the statutes is created to read:

20.455 (2) (Lp) *Crime laboratories; deoxyribonucleic acid analysis surcharges*. All moneys received from the crime laboratories and drug law enforcement surcharges under s. 165.755 and deoxyribonucleic acid analysis surcharges under s. 973.046 (1r) to transfer to the appropriation account under par. (jb) the amounts in the schedule under par. (kd) the amounts in the schedule under par. (kd), to transfer to the appropriation account under s. 20.475 (1) (km) the amounts in the schedule under s. 20.475 (1) (km), and to transfer to the appropriation account under par. (Lm) the amount determined under s. 165.25 (18).

SECTION 409b. 20.455 (3) (k) of the statutes is repealed.

SECTION 409f. 20.455 (5) (kr) of the statutes is created to read:

20.455 (**5**) (kr) *Court appointed special advocates.* The amounts in the schedule to provide grants under s. 165.967. All moneys transferred under 2017 Wisconsin Act (this act), section 9228 (19p) shall be credited to this appropriation account.

SECTION 409g. 20.455 (5) (kr) of the statutes, as created by 2017 Wisconsin Act (this act), is repealed.

SECTION 410. 20.465 (1) (k) of the statutes is repealed.

SECTION 410c. 20.465 (3) (df) of the statutes is created to read:

20.465 (3) (df) Regional emergency response grants. As a continuing appropriation, the amounts in the schedule to pay grants under s. 323.70 (6m) for the replacement of equipment used in emergency responses to releases of hazardous substances.

SECTION 411. 20.465 (3) (dm) of the statutes is created to read:

20.465 (3) (dm) *Mobile field force grants*. As a continuing appropriation, the amounts in the schedule for grants awarded under s. 323.62 to local law enforcement agencies.

SECTION 411e. 20.465 (3) (ke) of the statutes is created to read:

20.465 (3) (ke) *Interagency and intra-agency assist-ance*. All moneys received from the department or any other state agency regarding emergency management

services and interoperability to carry out the purposes for which received.

SECTION 411m. 20.465 (3) (q) of the statutes is created to read:

20.465 (3) (q) *Interoperability council*. From the police and fire protection fund, the amounts in the schedule for general program operations of the interoperability council.

SECTION 411s. 20.465 (3) (qm) of the statutes is created to read:

20.465 (3) (qm) *Next Generation 911*. From the police and fire protection fund, the amounts in the schedule for the department to make and administer contracts under s. 256.35 (3s) (b) and for the 911 subcommittee to administer its duties under s. 256.35 (3s) (d).

SECTION 412h. 20.475 (1) (km) of the statutes is amended to read:

20.475 (1) (km) Deoxyribonucleic acid evidence activities. The amounts in the schedule for deoxyribonucleic acid evidence activities. All moneys transferred to this appropriation account from the appropriation account under s. 20.455 (2) (Lm) (Lp) shall be credited to this appropriation account. Notwithstanding s. 20.001 (3) (a), the unencumbered balance on June 30 of each year shall revert to the appropriation account under s. 20.455 (2) (Lm) (Lp).

SECTION 412k. 20.485 (1) (b) of the statutes is repealed.

SECTION 413. 20.485 (1) (gk) of the statutes, as affected by 2015 Wisconsin Act 55, is amended to read:

20.485 (1) (gk) *Institutional operations*. amounts in the schedule for the care of the members of the Wisconsin veterans homes under s. 45.50, for the payment of stipends under s. 45.50 (2m) (f), for the transfer of moneys to the appropriation account under s. 20.435 (4) (ky) for payment of the state share of the medical assistance costs related to the provision of stipends under s. 45.50 (2m) (f), for the payment of assistance to indigent veterans under s. 45.43 to allow them to reside at the Wisconsin Veterans Home at Union Grove, for the transfer of moneys to the appropriation account accounts under par. pars. (kc) and (kg), and for the payment of grants under s. 45.82. Not more than 1 percent of the moneys credited to this appropriation account may be used for the payment of assistance to indigent veterans under s. 45.43. All moneys received under par. (m) and s. 45.51 (7) (b) and (8) and all moneys received for the care of members under medical assistance, as defined in s. 49.43 (8), shall be credited to this appropriation account. Except for the moneys transferred under this paragraph to the appropriation account under par. (kc), no moneys may be expended from this appropriation for the purposes specified in par. (kc).

SECTION 413k. 20.485 (1) (hm) of the statutes is repealed.

SECTION 414. 20.485 (1) (kc) of the statutes is created to read:

20.485 (1) (kc) Electric energy derived from renewable resources. The amounts in the schedule for the premium cost incurred for the generation or purchase of electric energy derived from renewable resources under s. 16.75 (12). All moneys transferred from the appropriation account under par. (gk) shall be credited to this appropriation account.

SECTION 414b. 20.485 (1) (mj) of the statutes is repealed.

SECTION 414d. 20.485 (1) (u) of the statutes is repealed.

SECTION 414f. 20.485 (2) (a) of the statutes is repealed.

SECTION 414h. 20.485 (2) (b) of the statutes is repealed.

SECTION 414i. 20.485 (2) (d) of the statutes is repealed.

SECTION 414k. 20.485 (2) (e) of the statutes is repealed.

SECTION 414m. 20.485 (2) (eg) of the statutes is repealed.

SECTION 414n. 20.485 (2) (rm) of the statutes is amended to read:

20.485 (2) (rm) *Veterans assistance programs; fish and game vouchers*. Biennially, the amounts in the schedule for general program operations of the veterans assistance program under s. 45.43, and for grants under s. 45.03 (13) (j), and for reimbursements to the department of natural resources under s. 29.1945 (2).

SECTION 4140. 20.485 (2) (rn) of the statutes is created to read:

20.485 (2) (rn) *Fish and game vouchers*. Biennially, the amounts in the schedule for reimbursements to the department of natural resources under s. 29.1945 (2).

SECTION 414p. 20.485 (2) (tf) of the statutes is amended to read:

20.485 (2) (tf) Veterans tuition reimbursement program; grants to nonprofit organizations that serve veterans and their families. Biennially, the amounts in the schedule for the veterans tuition reimbursement program under s. 45.20 (2) and for grants to nonprofit organizations under s. 45.46. Notwithstanding s. 20.001 (3) (a), the department may encumber moneys under this appropriation for the biennium up to 60 days after the end of that biennium if an estimate is first submitted to the department of administration showing the amounts that will be encumbered during that 60-day period.

SECTION 414q. 20.485 (2) (th) of the statutes is created to read:

20.485 (2) (th) *Grants to nonprofit organizations.* Biennially, the amounts in the schedule for grants to nonprofit organizations under s. 45.46.

SECTION 414r. 20.485 (2) (vm) of the statutes is amended to read:

20.485 (2) (vm) Assistance to needy veterans and veteran start—up businesses. The amounts in the schedule for aid payments under s. 45.40 and for the grant to VETransfer, Inc., under s. 45.45.

SECTION 414s. 20.485 (2) (vs) of the statutes is created to read:

20.485 (2) (vs) *Grants to Camp American Legion*. The amounts in the schedule for grants to the Wisconsin department of the American Legion under s. 45.41 (5) to operate Camp American Legion.

SECTION 414w. 20.485 (2) (vu) of the statutes is created to read:

20.485 (2) (vu) *Grants to American Indian tribes and bands*. The amounts in the schedule for grants to American Indian tribes and bands under s. 45.82 (4).

SECTION 414y. 20.485 (2) (vw) of the statutes is amended to read:

20.485 (2) (vw) Payments to veterans organizations for claims service; grants for the operation of Camp American Legion; grants to American Indian tribes and bands. The amounts in the schedule for payments to veterans organizations for claims services under s. 45.41 (2) (am) and (3m), for grants to the Wisconsin department of the American Legion under s. 45.41 (5) to operate Camp American Legion, and for grants to American Indian tribes and bands under s. 45.82 (4).

SECTION 415. 20.485 (2) (vx) of the statutes is amended to read:

20.485 (2) (vx) *County grants*. The amounts in the schedule for the administration and payment of grants under s. 45.82.

SECTION 415g. 20.485 (2) (vy) of the statutes is repealed.

SECTION 415m. 20.485 (2) (yg) of the statutes is repealed.

SECTION 415s. 20.485 (3) (v) of the statutes is repealed.

SECTION 416. 20.505 (1) (cg) of the statutes is repealed.

SECTION 416m. 20.505 (1) (id) 2. of the statutes is amended to read:

20.505 (1) (id) 2. The amount transferred to s. 20.455 (2) (ki) 20.465 (3) (km) shall be the amount in the schedule under s. 20.455 (2) (ki) 20.465 (3) (km).

SECTION 418. 20.505 (1) (ir) of the statutes is renumbered 20.155 (1) (i).

SECTION 419. 20.505 (1) (ki) of the statutes is amended to read:

20.505 (1) (ki) Postage costs. As a continuing appropriation, the amounts in the schedule to pay state agency postage costs. All moneys received from state agencies for the payment of state agency postage costs shall be credited to this appropriation account to pay state agency postage costs.

SECTION 421. 20.505 (1) (kp) of the statutes is repealed.

SECTION 422. 20.505 (1) (kr) of the statutes is amended to read:

20.505 (1) (kr) Legal services; relocation assistance. The amounts in the schedule to provide legal services under s. 16.004 (15) and to perform the duties under ss. 32.19 to 32.27. All moneys received from assessments under s. ss. 16.004 (15) (b) and 32.25 (4) shall be credited to this appropriation account.

SECTION 424. 20.505 (1) (r) of the statutes is repealed.

SECTION 427. 20.505 (1) (tb) of the statutes is renumbered 20.144 (3) (tb) and amended to read:

20.144 (3) (tb) Payment of qualified higher education expenses and refunds; college tuition and expenses program. From the tuition trust fund, a sum sufficient for the payment of qualified higher education expenses and refunds under s. 16.64 224.48 (5) and (7).

SECTION 428. 20.505 (1) (td) of the statutes is renumbered 20.144 (3) (td) and amended to read:

20.144 (3) (td) Administrative expenses; college tuition and expenses program. From the tuition trust fund, the amounts in the schedule for the administrative expenses of the college tuition and expenses program under s. 16.64 224.48, including the expense of promoting the program.

SECTION 429. 20.505 (1) (tf) of the statutes is renumbered 20.144 (3) (tf) and amended to read:

20.144(3) (tf) Payment of qualified higher education expenses and refunds; college savings program trust fund. From the college savings program trust fund, a sum sufficient for the payment of qualified higher education expenses and refunds under s. 16.641 224.50 (2) and (3).

SECTION 430. 20.505 (1) (th) of the statutes is renumbered 20.144 (3) (th) and amended to read:

20.144 (3) (th) Administrative expenses; college savings program trust fund. From the college savings program trust fund, the amounts in the schedule for the administrative expenses of the college savings program under s. 16.641 224.50, including the expense of promoting the program.

SECTION 431. 20.505 (1) (tj) of the statutes is renumbered 20.144 (3) (tj) and amended to read:

20.144 (3) (tj) Payment of qualified higher education expenses and refunds; college savings program bank deposit trust fund. From the college savings program bank deposit trust fund, a sum sufficient for the payment of qualified higher education expenses and refunds under s. 16.641 224.50 (2) and (3).

SECTION 432. 20.505 (1) (tL) of the statutes is renumbered 20.144 (3) (tL) and amended to read:

20.144 (3) (tL) Administrative expenses; college savings program bank deposit trust fund. From the college savings program bank deposit trust fund, the amounts in the schedule for the administrative expenses of the college savings program under s. 16.641 224.50, including the expense of promoting the program.

SECTION 433. 20.505 (1) (tn) of the statutes is renumbered 20.144 (3) (tn) and amended to read:

20.144 (3) (tn) Payment of qualified higher education expenses and refunds; college savings program credit union deposit trust fund. From the college savings program credit union deposit trust fund, a sum sufficient for the payment of qualified higher education expenses and refunds under s. 16.641 224.50 (2) and (3).

SECTION 434. 20.505 (1) (tp) of the statutes is renumbered 20.144 (3) (tp) and amended to read:

20.144 (3) (tp) Administrative expenses; college savings program credit union deposit trust fund. From the college savings program credit union deposit trust fund, the amounts in the schedule for the administrative expenses of the college savings program under s. 16.641 224.50, including the expense of promoting the program.

SECTION 435. 20.505 (1) (ub) of the statutes is amended to read:

20.505 (1) (ub) Land information program, state operations: reviews of municipal incorporations and annexations: planning grants. From the land information fund, all moneys received by the department under s. 59.72 (5) (a), except moneys appropriated under par. (ud), the amounts in the schedule for the land information program under s. 16.967 and for reviews of proposed municipal incorporations and annexations by the department and for the purpose of providing aids under s. 16.965.

SECTION 436. 20.505 (1) (uc) of the statutes is created to read:

20.505 (1) (uc) Land information program; local aids. From the land information fund, all moneys received by the department under s. 59.72 (5) (a), except moneys appropriated under par. (ub), for aids to counties under s. 16.967 (7).

SECTION 438. 20.505 (4) (er) (title) of the statutes is amended to read:

20.505 **(4)** (er) (title) *Service award program; state matching awards.*

SECTION 439. 20.505 (4) (hc) of the statutes is repealed.

SECTION 440. 20.505 (4) (k) of the statutes is amended to read:

20.505 (4) (k) Waste facility siting board; general program operations. The amounts in the schedule for the general program operations of the waste facility siting board. All moneys transferred from the appropriation account under s. 20.370 (2) (4) (eg) shall be credited to this appropriation account.

SECTION 441. 20.505 (4) (s) of the statutes, as affected by 2015 Wisconsin Act 55, is amended to read:

20.505 (4) (s) School districts; telecommunications Telecommunications access for educational agencies. infrastructure grants, and teacher training grants. Biennially, from the universal service fund, the amounts in the schedule to make payments to telecommunications

providers under contracts under s. 16.971 (13), (14), and (15) to the extent that the amounts due are not paid from the appropriation under sub. (1) (is), to make payments to telecommunications providers under contracts under s. 16.971 (16) to the extent that the amounts due are not paid from the appropriation under sub. (1) (kL), to make grants to school district consortia under s. 16.997 (7), to make information technology infrastructure grants under s. 16.9945, and to make educational technology teacher training grants under s. 16.996. Notwithstanding s. 20.001 (3) (b), the unencumbered balance on June 30 of each odd–numbered year shall be transferred to the appropriation account under s. 20.155 (3) (r).

SECTION 442. 20.505 (4) (s) of the statutes, as affected by 2017 Wisconsin Act (this act), is amended to read:

20.505 (4) (s) Telecommunications access for educational agencies, infrastructure grants, and teacher training grants. Biennially, from the universal service fund, the amounts in the schedule to make payments to telecommunications providers under contracts under s. 16.971 (13), (14), and (15) to the extent that the amounts due are not paid from the appropriation under sub. (1) (is), to make payments to telecommunications providers under contracts under s. 16.971 (16) to the extent that the amounts due are not paid from the appropriation under sub. (1) (kL), to make grants to school district consortia under s. 16.997 (7), to make information technology infrastructure grants under s. 16.9945, and to make educational technology teacher training grants under s. 16.996. Notwithstanding s. 20.001 (3) (b), the unencumbered balance on June 30 of each odd-numbered year shall be transferred to the appropriation account under s. 20.155 (3) (r).

SECTION 443. 20.505 (4) (t) of the statutes is repealed.

SECTION 444. 20.505 (4) (tm) of the statutes is repealed.

SECTION 445. 20.505 (4) (tu) of the statutes is repealed.

SECTION 446. 20.505 (4) (tw) of the statutes is repealed.

SECTION 447. 20.505 (5) (ka) of the statutes is amended to read:

20.505 (5) (ka) Facility operations and maintenance; police and protection functions. The amounts in the schedule for the purpose of financing the costs of operation of state—owned or operated facilities that are not funded from other appropriations, including custodial and maintenance services; for minor projects; for utilities, fuel, heat, and air conditioning; for assessments levied by the department under s. 16.847 (3) for costs incurred and savings generated at departmental facilities; for facility design services provided to agencies under s. 16.849; and for costs incurred under ss. 16.858 and 16.895 by or on behalf of the department; and for police

and protection functions under s. 16.84 (2) and (3). All moneys received from state agencies for the operation of such facilities, from parking rental fees established under s. 16.843 (2) (bm) and miscellaneous other sources, from assessments under s. 16.895, from the performance of gaming protection functions under s. 16.84 (3), and from the fees assessed under s. 16.849, and all moneys transferred from the appropriation account under s. 20.865 (2) (e) for this purpose shall be credited to this appropriation account. No moneys may be expended from this appropriation for the purposes specified in par. (kg).

SECTION 448. 20.505 (5) (kg) of the statutes is created to read:

20.505 (5) (kg) Electric energy derived from renewable resources. The amounts in the schedule for the premium cost incurred for the generation or purchase of electric energy derived from renewable resources. All moneys received from agencies, as defined in s. 16.75 (12) (a) 1., for this purpose shall be credited to this appropriation account.

SECTION 449. 20.505 (5) (ks) of the statutes is created to read:

20.505 (5) (ks) Security services. The amounts in the schedule to provide security services at multitenant state buildings or multitenant state facilities. All moneys received from charges to state agencies for those services under s. 16.84 (2) shall be credited to this appropriation account.

SECTION 450. 20.505 (7) (fm) of the statutes is amended to read:

20.505 (7) (fm) Shelter for homeless and transitional housing grants. Biennially, the amounts in the schedule for transitional housing grants under s. 16.306 and for grants to agencies and shelter facilities for homeless individuals and families as provided under s. 16.308. Notwithstanding ss. 20.001 (3) (a) and 20.002 (1), the department may transfer funds between fiscal years under this paragraph.

SECTION 451. 20.505 (7) (fr) of the statutes is renumbered 20.435 (5) (fr) and amended to read:

20.435 (5) (fr) *Mental health for homeless individuals*. The amounts in the schedule for mental health services for homeless individuals under s. 16.311 51.047.

SECTION 452. 20.505 (7) (ft) of the statutes is created to read:

20.505 (7) (ft) *Employment grants*. The amounts in the schedule for grants to municipalities under s. 16.313.

SECTION 453. 20.505 (7) (kg) of the statutes is amended to read:

20.505 (7) (kg) *Housing program services*. All moneys received from other state agencies for housing program services, including all moneys required under s. 49.175 (1) (f) to be credited to this appropriation account, for the purpose of providing housing program services. Notwithstanding s. 20.001 (3) (c), at the end of each fiscal year, any unencumbered balance in this appropriation

account attributable to the moneys credited under s. 49.175 (1) (f) shall revert to one or more of the appropriation accounts specified in s. 49.175 (1) (intro.), as determined by the secretary of administration.

SECTION 454. 20.505 (8) (hm) (intro.) of the statutes is amended to read:

20.505 (8) (hm) *Indian gaming receipts.* (intro.) All moneys required to be credited to this appropriation under s. 569.06, all moneys transferred under 2001 Wisconsin Act 16, sections 9201 (5mk), 9205 (1mk), 9210 (3mk), 9223 (5mk), 9224 (1mk), 9225 (1mk), 9231 (1mk), 9237 (4mk), 9240 (1mk), 9251 (1mk), 9256 (1mk), 9257 (2mk), and 9258 (2mk), and all moneys that revert to this appropriation account from the appropriation accounts specified in subds. 1c. to 19., 22., and 23., less the amounts appropriated under par. (h) and s. 20.455 (2) (gc), for the purpose of 2017 Wisconsin Act (this act), section 9101 (10), and for the purpose of annually transferring the following amounts:

SECTION 455. 20.505 (8) (hm) (intro.) of the statutes, as affected by 2017 Wisconsin Act (this act), is amended to read:

20.505 (8) (hm) *Indian gaming receipts*. (intro.) All moneys required to be credited to this appropriation under s. 569.06, all moneys transferred under 2001 Wisconsin Act 16, sections 9201 (5mk), 9205 (1mk), 9210 (3mk), 9223 (5mk), 9224 (1mk), 9225 (1mk), 9231 (1mk), 9237 (4mk), 9240 (1mk), 9251 (1mk), 9256 (1mk), 9257 (2mk), and 9258 (2mk), and all moneys that revert to this appropriation account from the appropriation accounts specified in subds. 1c. to 19., 22., and 23., less the amounts appropriated under par. (h) and s. 20.455 (2) (gc), for the purpose of 2017 Wisconsin Act (this act), section 9101 (1), and for the purpose of annually transferring the following amounts:

SECTION 456. 20.505 (8) (hm) 8d. of the statutes is amended to read:

20.505 (8) (hm) 8d. The amount transferred to s. 20.370 (4) (1) (kk) shall be the amount in the schedule under s. 20.370 (4) (1) (kk).

SECTION 457. 20.505 (8) (hm) 17f. of the statutes is repealed.

SECTION 458. 20.505 (8) (hm) 18dm. of the statutes is amended to read:

20.505 (8) (hm) 18dm. The amount transferred to s. 20.435 (7) (1) (kn) shall be the amount in the schedule under s. 20.435 (7) (1) (kn).

SECTION 460. 20.505 (8) (hm) 21d. of the statutes is repealed.

Vetoed SECTION 460r. 20.548 of the statutes is created to In Part read:

20.548 Prosecutor board. There is appropriated to the prosecutor board for the following program:

(1) COORDINATION AND ADMINISTRATION OF PROSECUTOR FUNCTIONS. (a) *Program administration*. The

amounts in the schedule for program administration costs of the office of state prosecutors.

(g) *Gifts*, *grants*, *and proceeds*. All moneys received from gifts and grants and all proceeds from services, conferences, and sales of publications and promotional materials for the purposes for which made or received.

SECTION 461. 20.550 (1) (a) of the statutes is amended to read:

20.550 (1) (a) *Program administration operation*. The <u>Biennially, the</u> amounts in the schedule for program administration costs the operation of the office of the state public defender, excluding the costs under pars. <u>par.</u> (e) and (fb).

SECTION 462. 20.550 (1) (b) of the statutes is repealed.

SECTION 463. 20.550 (1) (c) of the statutes is repealed.

SECTION 464. 20.550 (1) (d) of the statutes is repealed.

SECTION 465. 20.550 (1) (e) of the statutes is repealed.

SECTION 466. 20.550 (1) (em) of the statutes is repealed.

SECTION 467. 20.550 (1) (f) of the statutes is repealed.

SECTION 468. 20.566 (1) (gn) of the statutes is repealed.

SECTION 469. 20.566 (7) (v) of the statutes is amended to read:

20.566 (7) (v) Investment and local impact fund. From the investment and local impact fund, all moneys received under s. 70.395 (1e) and (2) (dc) and (dg), less the moneys appropriated under s. 20.370 (2) (4) (gr), to be disbursed under ss. 70.395 (2) (d) to (g), 293.33 (4), 293.65 (5) (a), 295.443, and 295.61 (9) (a) and (c).

SECTION 469d. 20.566 (8) (b) of the statutes is created to read:

20.566 (8) (b) *Retailer compensation*. The amounts in the schedule to pay compensation to retailers under s. 565.10 (14) (b).

SECTION 476. 20.680 (2) (h) of the statutes is amended to read:

20.680 (2) (h) *Materials and services*. All moneys received from providing services and selling documents under s. 758.19 (2), except moneys received for those services and documents related to the consolidated court automation program credited to par. (j), to provide services and sell documents related to uniform forms, special reports, photocopies and pamphlets under s. 758.19 (2), except those services and documents related to the consolidated court automation program.

SECTION 477. 20.680 (2) (j) of the statutes is amended to read:

20.680 (2) (j) Court information systems. All moneys related to the consolidated court automation program

Vetoed In Part Vetoed In Part received from providing services and selling documents under s. 758.19 (2), all moneys received under s. 758.19 (4m), all moneys received under ss. 814.61, 814.62, and 814.63 that are required to be credited to this appropriation account under those sections, and \$6 of each \$21.50 received under s. 814.86 (1) for the operation of circuit court automated information systems under s. 758.19 (4) and to provide services and sell documents under s. 758.19 (2) related to uniform forms, special reports, photocopies, and pamphlets that are related to the consolidated court automation program.

SECTION 478. 20.680 (2) (kf) of the statutes is created to read:

20.680 (2) (kf) Interagency and intra-agency assistance. All moneys received from a court or transferred from any state agency for services provided to the court or state agency.

Vetoed In Part

SECTION 480b. 20.765 (4) (title) of the statutes is amended to read:

20.765 (4) (title) Capitol offices relocation: 100th ANNIVERSARY OF STATE CAPITOL.

SECTION 480c. 20.765 (4) (b) of the statutes is created

20.765 (4) (b) Celebration of 100th anniversary of state capitol; general purpose revenue. The amounts in the schedule to fund activities related to the celebration of the 100th anniversary of the state capitol.

SECTION 480cg. 20.765 (4) (h) of the statutes is created to read:

20.765 (4) (h) Celebration of 100th anniversary of state capitol; program revenue. All moneys received from revenues generated from activities related to the celebration of the 100th anniversary of the state capitol to lapse to the general fund the first \$50,000 credited to this appropriation account in each fiscal year and to transfer the remainder to the appropriation account under s. 20.855 (3) (k).

SECTION 480cp. 20.835 (1) (c) of the statutes is amended to read:

20.835 **(1) (c)** Expenditure restraint program account. A sum sufficient to make the payments under s. 79.05 (3), plus the amounts under s. 79.05 (7).

SECTION 480d. 20.835 (1) (f) of the statutes is created to read:

20.835 (1) (f) State aid; personal property tax exemption. A sum sufficient to make the state aid payments under s. 79.096.

SECTION 480m. 20.835 (1) (r) of the statutes is amended to read:

20.835 (1) (r) County and municipal aid account; police and fire protection fund. From the police and fire protection fund, after deducting the amounts appropriated from that fund under s. ss. 20.155 (3) (t) and 20.465 (3) (q) and (qm), all moneys received from the fees collected under s. 196.025 (6) to make the payments under s. 79.035.

SECTION 481m. 20.835 (2) (cp) of the statutes is created to read:

20.835 (2) (cp) Electronics and information technology manufacturing zone credit. A sum sufficient to make the payments under ss. 71.07 (3wm) (d) and 71.28 (3wm)

SECTION 481p. 20.835 (2) (d) of the statutes is created to read:

20.835 (2) (d) Research credit. A sum sufficient to make the payments under ss. 71.07 (4k) (e) 2. a., 71.28 (4) (k) 1., and 71.47 (4) (k) 1.

SECTION 482. 20.835 (3) (title) of the statutes is amended to read:

20.835 (3) (title) STATE PROPERTY TAX CREDITS RELIEF. SECTION 483. 20.835 (3) (ef) of the statutes is created

20.835 (3) (ef) Transfer to conservation fund; forestry. A sum sufficient in an amount equal to the amount to be paid into the conservation fund as determined under s. 70.58 (3) for the forestry purposes described under s. 70.58 (1). The amounts may be paid at such intervals during each fiscal year as the secretary of administration considers appropriate or necessary.

SECTION 483m. 20.855 (3) (k) of the statutes is cre-**Vetoed** ated to read:

In Part

20.855 (3) (k) Capitol restoration and relocation planning; program revenue. Biennially, the amounts in the schedule for purposes related to capitol restoration and relocation planning. All moneys transferred from the appropriation account under s. 20.765 (4) (h) shall be credited to this appropriation account.

SECTION 484. 20.855 (4) (h) of the statutes is created to read:

20.855 (4) (h) Volkswagen settlement funds. All moneys received from the trustee of the settlement funds, as defined in s. 16.047 (1) (a), for the replacement of vehicles in the state fleet under s. 16.047 (2) and for the grants under s. 16.047 (4m). No more than \$21,000,000 may be expended from this appropriation in fiscal year 2017–18. No moneys may be expended from this appropriation after June 30, 2027.

SECTION 485. 20.866 (1) (u) of the statutes is amended to read:

20.866 (1) (u) Principal repayment and interest. A sum sufficient from moneys appropriated under sub. (2) (zp) and ss. 20.115 (2) (d) and (7) (b) and (s), 20.190 (1) (c), (d), (i), and (j), 20.225 (1) (c) and (i), 20.245 (1) (e) and (j), 20.250 (1) (c) and (e), 20.255 (1) (d), 20.285 (1) (d), (je), and (gj), 20.320 (1) (c) and (t) and (2) (c), 20.370 (7) (aa), (ac), (ad), (ag), (aq), (ar), (at), (au), (bq), (br), (cb), (cc), (cd), (cg), (cq), (cr), (cs), (ct), (ea), (eq), and (er), 20.395 (6) (af), (aq), (ar), and (au), 20.410 (1) (e), (ec), and (ko) and (3) (e), 20.435 (2) (ee), 20.465 (1) (d), 20.485 (1) (f) and (go), (3) (t) and (4) (qm), 20.505 (4) (es), (et), (ha), and (hb) and (5) (c), (g), and (kc), 20.855 (8) (a), and 20.867 (1) (a) and (b) and (3) (a), (b), (bb), (bc), (bd), (be), (bf), (bg), (bh), (bj), (bL), (bm), (bn), (bq), (br), (bt), (bu), (bv), (bw), (bx), (cb), (cd), (cf), (ch), (cj), (cq), (cr), (cs), (g), (h), (i), (kd), and (q) for the payment of principal, interest, premium due, if any, and payment due, if any, under an agreement or ancillary arrangement entered into under s. 18.06 (8) (a) relating to any public debt contracted under subchs. I and IV of ch. 18.

SECTION 485p. 20.866 (2) (s) (intro.) of the statutes is amended to read:

20.866 (2) (s) *University of Wisconsin; academic facilities*. (intro.) From the capital improvement fund, a sum sufficient for the board of regents of the University of Wisconsin System to acquire, construct, develop, enlarge or improve university academic educational facilities and facilities to support such facilities. The state may contract public debt in an amount not to exceed \$2,341,609,100 \$2,552,521,100 for this purpose. Of this amount:

SECTION 485r. 20.866 (2) (t) of the statutes is amended to read:

20.866 (2) (t) *University of Wisconsin; self-amortizing facilities*. From the capital improvement fund, a sum sufficient for the board of regents of the University of Wisconsin System to acquire, construct, develop, enlarge, or improve university self-amortizing educational facilities and facilities to support such facilities. The state may contract public debt in an amount not to exceed \$2,709,353,100 \$2,740,855,400 for this purpose. Of this amount, \$4,500,000 is allocated only for the University of Wisconsin–Madison indoor practice facility for athletic programs and only at the time that ownership of the facility is transferred to the state.

SECTION 486. 20.866 (2) (tc) of the statutes is amended to read:

20.866 (2) (tc) Clean water fund program. From the capital improvement fund, a sum sufficient for the purposes of s. 281.57 (10m) and (10r) and to be transferred to the environmental improvement fund for the purposes of the clean water fund program under ss. 281.58 and 281.59. The state may contract public debt in an amount not to exceed \$686,743,200 \$646,283,200 for this purpose. Of this amount, the amount needed to meet the requirements for state deposits under 33 USC 1382 is allocated for those deposits. Of this amount, \$8,250,000 is allocated to fund the minority business development and training program under s. 200.49 (2) (b). Moneys from this appropriation account may be expended for the purposes of s. 281.57 (10m) and (10r) only in the amount by which the department of natural resources and the department of administration determine that moneys available under par. (tn) are insufficient for the purposes of s. 281.57 (10m) and (10r).

SECTION 487. 20.866 (2) (td) of the statutes is amended to read:

20.866 (2) (td) Safe drinking water loan program. From the capital improvement fund, a sum sufficient to be transferred to the environmental improvement fund for the safe drinking water loan program under s. 281.61. The state may contract public debt in an amount not to exceed \$65,600,000 \$71,400,000 for this purpose.

SECTION 488. 20.866 (2) (tf) of the statutes is amended to read:

20.866 (2) (tf) *Natural resources; nonpoint source.* From the capital improvement fund, a sum sufficient for the department of natural resources to fund nonpoint source water pollution abatement projects under s. 281.65 (4c) and (4e). The state may contract public debt in an amount not to exceed \$37,900,000 \$44,050,000 for this purpose.

SECTION 489. 20.866 (2) (th) of the statutes is amended to read:

20.866 (2) (th) *Natural resources; urban nonpoint source cost–sharing*. From the capital improvement fund, a sum sufficient for the department of natural resources to provide cost–sharing grants for urban nonpoint source water pollution abatement and storm water management projects under s. 281.66, to provide municipal flood control and riparian restoration cost–sharing grants under s. 281.665, and to make the grant under 2007 Wisconsin Act 20, section 9135 (1i). The state may contract public debt in an amount not to exceed \$49,900,000 \$53.600,000 for this purpose. Of this amount, \$500,000 is allocated in fiscal biennium 2001–03 for dam rehabilitation grants under s. 31.387.

SECTION 489m. 20.866 (2) (tu) of the statutes is amended to read:

20.866 (2) (tu) Natural resources; segregated revenue supported facilities. From the capital improvement fund, a sum sufficient for the department of natural resources to acquire, construct, develop, enlarge, or improve natural resource administrative office, laboratory, equipment storage, or maintenance facilities and to acquire, construct, develop, enlarge, or improve state recreation facilities and state fish hatcheries. The state may contract public debt in an amount not to exceed \$102,365,300 \$108,171,100 for this purpose.

SECTION 490. 20.866 (2) (tx) of the statutes is amended to read:

20.866 (2) (tx) *Natural resources; dam safety projects*. From the capital improvement fund, a sum sufficient for the department of natural resources to provide financial assistance to counties, cities, villages, towns, and public inland lake protection and rehabilitation districts for dam safety projects under s. 31.385. The state may contract public debt in an amount not to exceed \$21,500,000 \$25,500,000 for this purpose.

SECTION 492. 20.866 (2) (uv) of the statutes is amended to read:

20.866 (2) (uv) *Transportation, harbor improvements.* From the capital improvement fund, a sum sufficient for the department of transportation to provide grants for harbor improvements. The state may contract public debt in an amount not to exceed \$105,900,000 \$120,000,000 for this purpose.

SECTION 493. 20.866 (2) (uw) of the statutes is amended to read:

20.866 (2) (uw) Transportation; rail acquisitions and improvements. From the capital improvement fund, a sum sufficient for the department of transportation to acquire railroad property under ss. 85.08 (2) (L) and 85.09; and to provide grants and loans for rail property acquisitions and improvements under s. 85.08 (4m) (c) and (d). The state may contract public debt in an amount not to exceed \$238,300,000 \$250,300,000 for these purposes.

SECTION 493m. 20.866 (2) (ux) of the statutes is amended to read:

20.866 (2) (ux) Corrections; correctional facilities. From the capital improvement fund, a sum sufficient for the department of corrections to acquire, construct, develop, enlarge, or improve adult and juvenile correctional facilities. The state may contract public debt in an amount not to exceed \$882,346,900 \$926,679,900 for this purpose.

SECTION 493s. 20.866 (2) (v) of the statutes is amended to read:

20.866 (2) (v) Health services; mental health and secure treatment facilities. From the capital improvement fund, a sum sufficient for the department of health services to acquire, construct, develop, enlarge, or extend mental health and secure treatment facilities. The state may contract public debt in an amount not to exceed \$185,951,200 \$208,646,200 for this purpose.

SECTION 494. 20.866 (2) (we) of the statutes is amended to read:

20.866 (2) (we) *Agriculture; soil and water.* From the capital improvement fund, a sum sufficient for the department of agriculture, trade and consumer protection to provide for soil and water resource management under s. 92.14. The state may contract public debt in an amount not to exceed \$61,075,000 \$68,075,000 for this purpose.

SECTION 494m. 20.866 (2) (ws) of the statutes is amended to read:

20.866 (2) (ws) Administration; energy conservation projects; capital improvement fund. From the capital improvement fund, a sum sufficient for the department of administration to provide funding to agencies, as defined in s. 16.70 (1e), for energy conservation construction projects at state facilities under the jurisdiction of the agencies pursuant to s. 16.847 (2). The state may contract public debt in an amount not exceeding \$200,000,000 \$220,000,000 for this purpose.

SECTION 495. 20.866 (2) (xm) of the statutes is amended to read:

20.866 (2) (xm) Building commission; refunding tax-supported and self-amortizing general obligation debt. From the capital improvement fund, a sum sufficient to refund the whole or any part of any unpaid indebtedness used to finance tax-supported or self-amortizing facilities. In addition to the amount that may be contracted under par. (xe), the state may contract public debt in an amount not to exceed \$5,285,000,000 \$6,785,000,000 for this purpose. Such indebtedness shall be construed to include any premium and interest payable with respect thereto. Debt incurred by this paragraph shall be repaid under the appropriations providing for the retirement of public debt incurred for tax-supported and self-amortizing facilities in proportional amounts to the purposes for which the debt was refinanced. No moneys may be expended under this paragraph unless the true interest costs to the state can be reduced by the expenditure.

SECTION 495b. 20.866 (2) (y) of the statutes is amended to read:

20.866 (2) (y) Building commission; housing state departments and agencies. From the capital improvement fund, a sum sufficient to the building commission for the purpose of housing state departments and agencies. The state may contract public debt in an amount not to exceed \$820,767,100 \$917,767,100 for this purpose.

SECTION 495f. 20.866 (2) (z) (intro.) of the statutes is amended to read:

20.866 (2) (z) Building commission; other public purposes. (intro.) From the capital improvement fund, a sum sufficient to the building commission for relocation assistance and capital improvements for other public purposes authorized by law but not otherwise specified in this chapter. The state may contract public debt in an amount not to exceed \$2,491,765,400 \$2,677,933,400 for this purpose. Of this amount:

SECTION 495k. 20.866 (2) (zcq) of the statutes is created to read:

20.866 (2) (zcq) La Crosse Center. From the capital improvement fund, a sum sufficient for the building commission to provide a grant to the city of La Crosse for the remodeling and expansion of the La Crosse Center as specified in s. 13.48 (33e). The state may contract public debt in an amount not to exceed \$5,000,000 for this purpose.

SECTION 495p. 20.866 (2) (zcr) of the statutes is created to read:

20.866 (2) (zcr) St. Ann Center for Intergenerational Care, Inc.; Bucyrus Campus. From the capital improvement fund, a sum sufficient for the building commission to provide a grant to St. Ann Center for Intergenerational Care, Inc., for the completion of its Bucyrus Campus as specified in s. 13.48 (33m). The state may contract public debt in an amount not to exceed \$5,000,000 for this purpose.

SECTION 495s. 20.866 (2) (zcs) of the statutes is created to read:

20.866 (2) (zcs) Brown County innovation center. From the capital improvement fund, a sum sufficient for the building commission to provide a grant to Brown County for the construction of a science, technology, engineering, and mathematics innovation center as specified in s. 13.48 (33s). The state may contract public debt in an amount not to exceed \$5,000,000 for this purpose.

SECTION 495w. 20.866 (2) (zj) of the statutes is amended to read:

20.866 (2) (zj) Military affairs; armories and military facilities. From the capital improvement fund, a sum sufficient for the department of military affairs to acquire, construct, develop, enlarge, or improve armories and other military facilities. The state may contract public debt in an amount not to exceed \$46,272,700 \$56,490,800 for this purpose.

SECTION 495y. 20.866 (2) (zm) of the statutes is amended to read:

20.866 (2) (zm) Veterans affairs; veterans facilities. From the capital improvement fund, a sum sufficient for the department of veterans affairs to acquire, construct, develop, enlarge, or improve facilities at state veterans homes, veterans cemeteries, and the veterans museum. The state may contract public debt in an amount not to exceed \$10,686,100 \$15,018,700 for this purpose.

SECTION 496. 20.866 (2) (zn) of the statutes is amended to read:

20.866 (2) (zn) *Veterans affairs; self-amortizing mortgage loans*. From the capital improvement fund, a sum sufficient for the department of veterans affairs for loans to veterans under s. 45.37 (6) (a). The state may contract public debt in an amount not to exceed \$2,400,840,000 \$2,127,540,000 for this purpose.

SECTION 496d. 20.866 (2) (zp) of the statutes is amended to read:

20.866 (2) (zp) *Veterans affairs; self-amortizing facilities*. From the capital improvement fund, a sum sufficient for the department of veterans affairs to acquire, construct, develop, enlarge, or improve facilities at state veterans homes. The state may contract public debt in an amount not to exceed \$69,948,700 \$77,995,100 for this purpose.

SECTION 496h. 20.867 (3) (cq) of the statutes is created to read:

20.867 (3) (cq) Principal repayment, interest, and rebates; La Crosse Center. A sum sufficient to reimburse s. 20.866 (1) (u) for the payment of principal and interest costs incurred in financing the remodeling and expansion of the La Crosse Center as specified in s. 13.48 (33e), to make the payments determined by the building commission under s. 13.488 (1) (m) that are attributable to the proceeds of obligations incurred in financing the project, and to make payments under an agreement or ancillary arrangement entered into under s. 18.06 (8) (a).

SECTION 496p. 20.867 (3) (cr) of the statutes is created to read:

20.867 (3) (cr) Principal repayment, interest, and rebates; St. Ann Center for Intergenerational Care, Inc.; Bucyrus Campus. A sum sufficient to reimburse s. 20.866 (1) (u) for the payment of principal and interest costs incurred in financing the completion of the Bucyrus Campus of St. Ann Center for Intergenerational Care, Inc., as specified in s. 13.48 (33m), to make the payments determined by the building commission under s. 13.488 (1) (m) that are attributable to the proceeds of obligations incurred in financing the project, and to make payments under an agreement or ancillary arrangement entered into under s. 18.06 (8) (a).

SECTION 496t. 20.867 (3) (cs) of the statutes is created to read:

20.867 (3) (cs) Principal repayment, interest, and rebates; Brown County innovation center. A sum sufficient to reimburse s. 20.866 (1) (u) for the payment of principal and interest costs incurred in financing the construction of a science, technology, engineering, and mathematics innovation center as specified in s. 13.48 (33s), to make the payments determined by the building commission under s. 13.488 (1) (m) that are attributable to the proceeds of obligations incurred in financing the project, and to make payments under an agreement or ancillary arrangement entered into under s. 18.06 (8) (a).

SECTION 497. 20.903 (2) (bp) of the statutes is created to read:

20.903 (2) (bp) Notwithstanding sub. (1), liabilities may be created and moneys expended from the appropriations under s. 20.435 (2) (g), (gk), and (kx) in an additional amount not exceeding the value of the equipment and buildings for operations financed under s. 20.435 (2) (g), (gk), and (kx).

SECTION 498. 20.903 (2) (c) of the statutes is amended to read:

20.903 (2) (c) All expenditures authorized by this subsection are subject to the estimate approval procedure provided in s. 16.50 (2). Notwithstanding pars. (a), (b) and, (bn), and (bp), the maximum amounts that may be expended from a program revenue or program revenue — service appropriation which is limited to the amounts in the schedule are the amounts in the schedule, except as authorized by the department of administration under s. 16.515 or the joint committee on finance under s. 13.101.

SECTION 499. 20.905 (1) of the statutes is amended to read:

20.905 (1) Manner of payment. Payments to the state may be made in legal tender, postal money order, express money order, bank draft, or certified check. Payments to the state may also be made by personal check or individual check drawn in the ordinary course of business unless otherwise required by individual state agencies. Payments to the state made by a debit or credit card approved by the depository selection board secretary of

administration or his or her designee may be accepted by state agencies. Prior to authorizing the use of a card, the depository selection board secretary of administration or his or her designee shall determine how any charges associated with the use of the card shall be paid, unless the method of payment of such charges is specified by law. Unless otherwise specifically prohibited by law, payments to the state may be made by electronic funds trans-

SECTION 500. 20.905 (2) of the statutes is amended to read:

20.905 (2) PROTESTED PAYMENT. If a personal check tendered to make any payment to the state is not paid by the bank on which it is drawn, if an electronic funds transfer does not take place because of insufficient funds, or if a demand for payment under a debit or credit card transaction is not paid by the bank upon which demand is made, the person by whom the check has been tendered, the person whose funds were to be electronically transferred, or the person entering into the debit or credit card transaction shall remain liable for the payment of the amount for which the check was tendered, the amount that was to be electronically transferred, or the amount agreed to be paid by debit or credit card and for all legal penalties, additions and a charge set by the depository selection board secretary of administration or his or her designee which is comparable to charges for unpaid drafts made by establishments in the private sector. In addition, the officer to whom the check was tendered, to whom the electronic funds transfer was promised, or to whom the debit or credit card was presented may, if there is probable cause to believe that a crime has been committed, provide any information or evidence relating to the crime to the district attorney of the county having jurisdiction over the offense for prosecution as provided by law. If any license has been granted upon any such check, any such electronic funds transfer, or any such debit or credit card transaction, the license shall be subject to cancellation for the nonpayment of the check, the failure to make the electronic funds transfer, or failure of the bank to honor the demand for payment authorized by debit or credit card.

SECTION 501. 20.906 (6) of the statutes is amended to read:

20.906(6) DIRECT DEPOSITS. The governor or the secretary of administration may require state agencies making deposits under this section to make direct deposits to any depository designated by the depository selection board secretary of administration or his or her designee, if such a requirement is advantageous or beneficial to this state.

SECTION 502. 20.920 (2) (c) of the statutes is amended to read:

20.920 (2) (c) All moneys in a contingent fund, except petty cash accounts established under s. 16.52 (7), shall be deposited in a separate account in a public depos-

itory approved by the depository selection board secretary or his or her designee. The agency head of each state agency having a contingent fund is responsible for all disbursements from the fund, but the agency head may delegate the responsibility for administration of the fund to a custodian, who shall be an employee of the agency. State agency invoices which qualify for payment from a contingent fund may be paid by check, share draft or other draft drawn by the agency head or custodian against the account. No such invoice need be submitted for audit prior to disbursement. After making each disbursement, the agency head shall file with the secretary a claim for reimbursement of the contingent fund on a voucher which shall be accompanied by a copy of the invoice to be reimbursed. Upon audit and approval of the claim by the secretary, the department of administration shall reimburse the contingent fund with the total amount lawfully paid therefrom.

SECTION 506. 20.923 (4) (e) 2. of the statutes is amended to read:

20.923 (4) (e) 2. Employment relations commission: chairperson and members.

SECTION 507d. 20.923 (4) (e) 13. of the statutes is **Vetoed** created to read:

In Part

20.923 (4) (e) 13. Transportation projects commission: director.

SECTION 507g. 20.923 (4) (f) 7x. of the statutes is created to read:

In Part

20.923 (4) (f) 7x. State prosecutors office: executive

SECTION 508f. 20.923 (6) (hs) of the statutes is created to read:

20.923 (6) (hs) State prosecutors office: legislative

SECTION 510m. 23.09 (2) (d) (intro.) of the statutes is amended to read:

23.09 (2) (d) Lands, acquisition. (intro.) Acquire by purchase, lease or agreement, and receive by gifts or devise, lands or waters suitable for the purposes enumerated in this paragraph, and maintain such lands and waters for such purposes; and, except for the purpose specified under subd. 12., may condemn lands or waters suitable for such purposes after obtaining approval of the appropriate standing committees of each house of the legislature as determined by the presiding officer thereof:

SECTION 512m. 23.09 (15) of the statutes is created to read:

23.09 (15) FEES FOR HUNTING PHEASANTS ON DEPART-MENT LAND. If the department requires payment of a fee in order to hunt pheasants on land under its management and control, all of those fees shall be credited to the appropriation account under s. 20.370 (1) (hw).

SECTION 514c. 23.0917 (5g) (c) 1. of the statutes is amended to read:

23.0917 (5g) (c) 1. In this paragraph, "unobligated amount" means the amount by which the annual bonding authority for the subprograms under subs. (3), (4), and (4j) in fiscal years 2011–12, 2012–13, and 2013–14, 2014–15, and 2015–16 exceeded the amounts that the department obligated from the moneys appropriated under s. 20.866 (2) (ta) for those subprograms for those fiscal years, but not including the amount by which the annual bonding authority for the purpose under sub. (3) (br) in fiscal year years 2013–14, 2014–15, and 2015–16 exceeded the amount obligated for that purpose in that fiscal year.

SECTION 514d. 23.0917 (5g) (c) 2. b. of the statutes is amended to read:

23.0917 (**5g**) (c) 2. b. The amount necessary for the purpose under sub. (4) (cm) 5. but not more than \$5,000,000 \$12,500,000.

SECTION 514g. 23.0917 (5g) (d) of the statutes is created to read:

23.0917 (**5g**) (d) 1. In this paragraph, "unobligated amount" means the amount by which the annual bonding authority for the subprograms under subs. (3), (4), and (4j) in fiscal years 2014–15 and 2015–16 exceeded the amounts that the department obligated from the moneys appropriated under s. 20.866 (2) (ta) for those subprograms for those fiscal years, but not including the amount by which the annual bonding authority for the purpose under sub. (3) (br) in fiscal years 2014–15 and 2015–16 exceeded the amount obligated for that purpose in that fiscal year.

- 2. The department shall obligate the unobligated amount as follows:
- a. The amount necessary for a grant to Iron County to rebuild the Saxon Harbor campground and marina but not more than \$1,000,000.
- b. The amount necessary for the purpose under s. 23.0963 but not more than \$1,000,000.
- c. The amount necessary for a grant for no more than 50 percent of the cost of reconstructing Eagle Tower in Peninsula State Park but not more than \$750,000.

d. The amount necessary for a grant to the city of Horicon to enhance a shelter located near the Palmatory scenic overlook on the south side of the Horicon Marsh Wildlife Area but not more than \$500,000 if, by June 30, 2019, the department and the city submit to the joint committee on finance a written plan for using the grant funds and if, within 14 working days after receiving the plan, the committee cochairpersons do not inform the department of an objection to the plan or, if the cochairpersons inform the department of an objection to the plan, the committee approves the plan by a majority vote.

e. The amount necessary for a grant to the cities of Neenah and Menasha for no more than 50 percent of the cost of constructing 2 pedestrian bridges across the Fox River and pedestrian trails to connect the bridges to existing pedestrian trails but not more than \$415,300 and subject to the limitation that the total amount obligated under

this subd. 1. e. and s. 23.197 (16) may not exceed \$2,015,300.

SECTION 514n. 23.0917 (5g) (e) of the statutes is created to read:

23.0917 (**5g**) (e) 1. In this paragraph, "unobligated amount" means the amount by which the annual bonding authority for the subprograms under subs. (3), (4), and (4j) in fiscal years 2014–15, 2015–16, and 2016–17 exceeded the amounts that the department obligated from the moneys appropriated under s. 20.866 (2) (ta) for those subprograms for those fiscal years, but not including the amount by which the annual bonding authority for the purpose under sub. (3) (br) in fiscal years 2014–15, 2015–16, and 2016–17 exceeded the amount obligated for that purpose in that fiscal year.

2. Of the unobligated amount, the department shall obligate an amount necessary for the purpose under s. 281.665 (4) (c), but not more than \$14,600,000.

SECTION 514r. 23.0963 of the statutes is created to read:

23.0963 Acquisition of the Canadian Pacific Railway corridor. From the appropriation under s. 20.866 (2) (ta) and subject to s. 23.0917 (5g) (d) 2. b., the department shall use the amount necessary, but not more than \$1,000,000, to acquire the railway corridor known as the Canadian Pacific Railway corridor in Racine County, beginning at approximately Vandenboom Road in the town of Dover and extending northeast approximately 13 miles to CTH "H" in the village of Sturtevant.

SECTION 515. 23.16 (1) of the statutes is amended to read:

23.16 (1) PUBLICATION. The department may produce, issue, or reprint magazines or other periodicals on a periodic basis as it determines, pertaining to fish and game, forests, parks, environmental quality, and other similar subjects of general information. The department shall produce 4 printed issues of the Wisconsin Natural Resources Magazine annually, provide the content of those printed issues on its Internet site, and provide additional magazine content on its Internet site. The department may distribute its magazines and periodicals by subscription. The department shall charge a fee for any of its magazines or periodicals, except that no fee may be charged to a person who is provided a subscription to the Wisconsin Natural Resources Magazine under s. 29.235.

SECTION 516. 23.22 (2) (d) of the statutes is amended to read:

23.22 (2) (d) Under the program established under par. (a), the department shall set aside \$42,000 from the appropriation under s. 20.370 (4) (1) (ku) during fiscal year 2013–14 to be used for a project to improve the sea lamprey barrier on the Kewaunee River at the Besadny Anadromous Fish Facility. Upon either the receipt or commitment of funding in the amount of \$78,000 from

Vetoed In Part

Vetoed In Part

Vetoed In Part one or more governmental units, the department shall release the amount set aside for the project.

SECTION 517. 23.22 (2) (e) of the statutes is amended to read:

23.22 (2) (e) Under the program established under par. (a), the department shall set aside \$262,500 from the appropriation under s. 20.370 (4) (1) (ku) during fiscal year 2013–14 to be used for a project to construct a sea lamprey barrier on the Nemadji River. Upon either the receipt or commitment of funding in the amount of \$487,500 from one or more governmental units, the department shall release the amount set aside for the project.

SECTION 518. 23.22 (2) (f) of the statutes is amended to read:

23.22 (2) (f) Under the program established under par. (a) and from the appropriation under s. 20.370 (4) (1) (kc), the department may expend up to \$400,000 to carry out sea lamprey control projects and up to \$120,000 to conduct surveys of sea lamprey larvae on any inland lakes, tributaries of Lake Michigan or Lake Superior, or harbors of Lake Michigan or Lake Superior.

SECTION 519. 23.27 (5) of the statutes is amended to read:

23.27 (5) NATURAL AREAS LAND ACQUISITION; COM-MITMENT UNDER THE WISCONSIN NATURAL AREAS HER-ITAGE PROGRAM. It is the intent of the legislature to initiate additional natural areas land acquisition activities with moneys available from the appropriations under ss. 20.370 (1) (gr) (fu) and 20.866 (2) (ta), (tt), and (tz) under the Wisconsin natural areas heritage program. This commitment is separate from and in addition to the continuing commitment under sub. (4). Moneys available from the appropriations under ss. 20.370 (1) (gr) (fu) and 20.866 (2) (ta), (tt), and (tz) under the Wisconsin natural areas heritage program may not be used to acquire land through condemnation. The department may not acquire land under this subsection unless the land is suitable for dedication under the Wisconsin natural areas heritage program and upon purchase or as soon after purchase as practicable the department shall take all necessary action to dedicate the land under the Wisconsin natural areas heritage program. Except as provided in s. 23.0915 (2), the department may not expend from the appropriation under s. 20.866 (2) (tz) more than \$500,000 in each fiscal year for natural areas land acquisition activities under this subsection and for grants for this purpose under s. 23.096.

SECTION 520. 23.27 (6) of the statutes is amended to read:

23.27 (6) SALE; CREDIT. Moneys received by the state from the sale of any area on state—owned land under the department's management or control which is withdrawn from the state natural areas system shall be credited to the appropriation under s. 20.370 (1) (gr) (fu). An amount equal to the value of any area on state—owned land under

the department's management or control which is withdrawn from the state natural areas system but remains in state ownership shall be credited to the appropriation under s. 20.370 (1) (gr) (fu).

SECTION 521. 23.29 (2) of the statutes is amended to read:

23.29 (2) CONTRIBUTIONS; STATE MATCH. The department may accept contributions and gifts for the Wisconsin natural areas heritage program. The department shall convert donations of land which it determines, with the advice of the council, are not appropriate for the Wisconsin natural areas heritage program into cash. The department shall convert other noncash contributions into cash. These moneys shall be deposited in the conservation fund and credited to the appropriation under s. 20.370 (1) (gr) (fu). These moneys shall be matched by an equal amount released from the appropriation under s. 20.866 (2) (ta), (tt), or (tz) or from any combination of these appropriations to be used for natural areas land acquisition activities under s. 23.27 (5). The department shall determine how the moneys being released are to be allocated from these appropriations.

SECTION 521f. 23.33 (5m) (title) of the statutes is amended to read:

23.33 **(5m)** (title) SAFETY <u>ENHANCEMENT</u> PROGRAM. **SECTION 521g.** 23.33 (5m) (a) of the statutes is amended to read:

23.33 (5m) (a) The department shall establish a program to award grants provide funding to organizations that meet the eligibility requirements under par. (b).

SECTION 521h. 23.33 (5m) (b) (intro.) of the statutes is amended to read:

23.33 (5m) (b) (intro.) To be eligible for a grant funding under this subsection, an organization shall meet all of the following requirements:

SECTION 521i. 23.33 (5m) (b) 2. of the statutes is amended to read:

23.33 (5m) (b) 2. The organization promotes the operation of all-terrain vehicles and utility terrain vehicles in a manner that is safe and responsible and that does not harm the environment.

SECTION 521j. 23.33 (5m) (b) 3. of the statutes is amended to read:

23.33 (5m) (b) 3. The organization promotes the operation of all-terrain vehicles and utility terrain vehicles in a manner that does not conflict with the laws, rules, and departmental policies that relate to the operation of all-terrain vehicles or utility terrain vehicles.

SECTION 521k. 23.33 (5m) (b) 4. of the statutes is amended to read:

23.33 (5m) (b) 4. The interest of the organization is limited to the recreational operation of all-terrain vehicles and utility terrain vehicles on all-terrain vehicle trails and other interconnected areas that are off the highways.

SECTION 521L. 23.33 (5m) (b) 5. of the statutes is amended to read:

23.33 (5m) (b) 5. The organization has a board of directors that has a majority of members who are representatives of all-terrain vehicle or utility terrain vehicle clubs.

SECTION 521m. 23.33 (5m) (b) 6. of the statutes is amended to read:

23.33 (**5m**) (b) 6. The organization provides support to all–terrain vehicle <u>and utility terrain vehicle</u> clubs.

SECTION 521n. 23.33 (5m) (c) (intro.) of the statutes is amended to read:

23.33 (5m) (c) (intro.) An organization receiving -a grant funding under this subsection shall use the grant moneys to promote and provide support to the program established under sub. (5) by conducting activities that include all of the following:

SECTION 5210. 23.33 (5m) (c) 1. of the statutes is amended to read:

23.33 (5m) (c) 1. Collecting data on the recreational operation of all-terrain vehicles off the highways and utility terrain vehicles.

SECTION 521p. 23.33 (5m) (c) 3. of the statutes is amended to read:

23.33 (5m) (c) 3. Attempting to increase participation by current and future all-terrain vehicle and utility terrain vehicle operators and owners in the program established under sub. (5) (d).

SECTION 521q. 23.33 (5m) (c) 4. of the statutes is amended to read:

23.33 (**5m**) (c) 4. Assisting the department of natural resources and the department of tourism in creating an outreach program to inform local communities of appropriate all–terrain vehicle and utility terrain vehicle use in their communities and of the economic benefits that may be gained from promoting tourism to attract all–terrain vehicle and utility terrain vehicle operators.

SECTION 521r. 23.33 (5m) (c) 5. of the statutes is amended to read:

23.33 (5m) (c) 5. Attempting to improve and maintain its relationship with the department of natural resources, the department of tourism, all-terrain vehicle and utility terrain vehicle dealers, all-terrain vehicle and utility terrain vehicle manufacturers, off-highway motorcycle clubs, as defined in s. 23.335 (1) (r), off-highway motorcycle alliances, other organizations that promote the recreational operation of off-highway motorcycles, snowmobile clubs, as defined in s. 350.138 (1) (e), snowmobile alliances, as defined in s. 350.138 (1) (d), and other organizations that promote the recreational operation of snowmobiles.

SECTION 521s. 23.33 (5m) (c) 6. of the statutes is amended to read:

23.33 (5m) (c) 6. Recruiting, assisting in the training of, and providing support to a corps of volunteers that will assist in providing instruction on the safe and respon-

sible operation of all-terrain vehicles <u>and utility terrain</u> <u>vehicles</u> that is given in the field to all-terrain vehicle <u>and utility terrain vehicle</u> operators.

SECTION 521t. 23.33 (5m) (c) 7. of the statutes is amended to read:

23.33 (5m) (c) 7. Publishing a manual in cooperation Cooperating with the department that shall be used to train volunteers to recruit, train, and manage volunteer trail patrol ambassadors in monitoring the recreational operation of all-terrain vehicles and utility terrain vehicles for safety issues and other issues that relate to the responsible operation of all-terrain vehicles and utility terrain vehicles.

SECTION 521u. 23.33 (5m) (d) of the statutes is amended to read:

23.33 (5m) (d) The department shall pay the grants provide funding under this subsection from the appropriation under s. 20.370 (5) (cx).

SECTION 521v. 23.33 (5m) (e) of the statutes is created to read:

23.33 (5m) (e) The department shall annually determine the amount necessary to provide funding under this subsection. The amount shall be the greater of \$297,000 or the amount calculated by multiplying 80 cents by the number of all-terrain vehicles and utility terrain vehicles registered as of the last day of February of the previous fiscal year.

SECTION 522. 23.335 (15) (d) of the statutes is amended to read:

23.335 (**15**) (d) The department shall pay the grants from the appropriation under s. 20.370 (1) (9) (jb).

SECTION 523. 23.335 (20) (b) (intro.) of the statutes is amended to read:

23.335 (20) (b) Off-highway motorcycle projects. (intro.) The department may use funding from the appropriation under s. 20.370 (1) (9) (jb) for off-highway motorcycle projects that are undertaken by the state or by local governmental units. Any of the following types of off-highway motorcycle projects are eligible for funding:

SECTION 524. 23.40 (3) (e) of the statutes is amended to read:

23.40 (3) (e) The department shall credit any environmental impact statement fee for a project involving the generation of electricity to the appropriation under s. $20.370 \, (3) \, (9) \, (dh)$.

SECTION 525. 23.425 (2) (b) of the statutes is amended to read:

23.425 (2) (b) The fees collected by the department under par. (a) for the use of the MacKenzie environmental center shall be deposited in the general fund and credited to the appropriation under s. 20.370 (9) (1) (gb).

SECTION 526. 23.426 of the statutes is amended to read:

23.426 Programs at the Horicon Marsh education and visitor center. The department may establish and

charge fees for educational programs that the department provides at the Horicon Marsh education and visitor center. The fees collected under this section shall be deposited in the general fund and credited to the appropriation account under s. 20.370 (9) (1) (gh).

SECTION 526g. 24.40 (1) of the statutes is renumbered 24.40 (1r) and amended to read:

24.40 (1r) Every board, commission, department and agency of the state having real estate belonging to the state under its control may grant easements in said property for public utility or telecommunications service through, over, <u>under</u>, along or to said property, including without limitation by enumeration the necessary poles, wires, structures, lines, <u>fiber</u>, conduits, pipes or pipe lines for heat, light, water, gas, sewer, power, telecommunications <u>service</u>, telegraph and transmission of messages.

SECTION 526r. 24.40 (1g) of the statutes is created to read:

24.40 (**1g**) In this section, "telecommunications service" has the meaning given in s. 182.017 (1g) (cq).

SECTION 527. 24.40 (3) of the statutes is created to read:

24.40 (3) Notwithstanding s. 28.02 (5) or any contrary rule promulgated by the department, if the department grants an easement under sub. (1) for the construction of broadband infrastructure in underserved areas, as designated under s. 196.504 (2) (d), the department may not require any appraisal or the payment of any fee to grant the easement.

SECTION 527m. 25.17 (1) (yo) of the statutes is repealed.

SECTION 528. 25.17 (2) (f) of the statutes is amended to read:

25.17 (2) (f) Invest the moneys belonging to the college savings program trust fund, the college savings program bank deposit trust fund, and the college savings program credit union deposit trust fund in a manner consistent with the guidelines established under s. 16.641 224.50 (2) (c), unless the moneys are under the management and control of a vendor selected under s. 16.255 224.51. In making investments under this paragraph, the investment board shall accept any reasonable terms and conditions that the college savings program board specifies and is relieved of any obligations relevant to prudent investment of the fund, including the standard of responsibility under s. 25.15 (2).

SECTION 529. 25.19 (3) of the statutes is amended to read:

25.19 (3) The secretary of administration or his or her designee shall, at the direction of the depository selection board under s. 34.045 (1) (bm), allocate bank service costs to the funds incurring those costs.

SECTION 530. 25.29 (1) (a) of the statutes is amended to read:

25.29 (1) (a) Except as provided in ss. 25.293 and 25.295, all moneys accruing to the state for or in behalf

of the department under chs. 26, 27, 28, 29, 169, and 350, subchs. I and VI of ch. 77 and ss. 23.09 to 23.31, 23.325, 23.33, 23.335, except as provided in s. 25.40 (1) (bt), 23.35 to 23.42, 23.50 to 23.99, 30.50 to 30.55, 70.58, 71.10 (5), 71.30 (10), and 90.21, including grants received from the federal government or any of its agencies except as otherwise provided by law.

SECTION 531. 25.29 (7) (intro.) of the statutes is amended to read:

25.29 (7) (intro.) All of the proceeds of the tax which is levied moneys received under s. 70.58, and all moneys paid into the state treasury as the counties' share of compensation of emergency fire wardens under s. 26.14 shall be used for acquiring, preserving and developing the forests of the state, including the acquisition of lands owned by counties by virtue of any tax deed and of other lands suitable for state forests, and for the development of lands so acquired and the conduct of forestry thereon, including the growing and planting of trees; for forest and marsh fire prevention and control; for grants to forestry cooperatives under s. 36.56; for compensation of emergency fire wardens; for maintenance, permanent property and forestry improvements; for other forestry purposes authorized by law and for the payment of aid for forests as authorized in s. 28.11 and subchs. I and VI of ch. 77.

SECTION 532. 25.29 (7) (a) of the statutes is amended to read:

25.29 (7) (a) Eight percent of the tax levied moneys received under s. 70.58 or of the funds provided for in lieu of the levy shall be used to acquire and develop forests of the state for the purposes or capable of providing the benefits described under s. 28.04 (2) within areas approved by the department and the governor and located within the region composed of Manitowoc, Calumet, Winnebago, Sheboygan, Fond du Lac, Ozaukee, Washington, Dodge, Milwaukee, Waukesha, Jefferson, Racine, Kenosha, Walworth, Rock, and Outagamie counties.

SECTION 533. 25.29 (7) (b) of the statutes is amended to read:

25.29 (7) (b) An additional 4 percent of the tax levied moneys received under s. 70.58 or of the funds provided in lieu of the levy shall be used to purchase forests for the state for the purposes or capable of providing the benefits described under s. 28.04 (2) within areas approved by the department and the governor and located within the region specified under par. (a).

SECTION 533m. 25.36 (1) of the statutes is amended to read:

25.36 (1) Except as provided in sub. (2), all moneys appropriated or transferred by law shall constitute the veterans trust fund which shall be used for the lending of money to the mortgage loan repayment fund under s. 45.37 (5) (a) 12. and for the veterans programs under ss. 20.485 (2) (m), (tm), (u), (vy), and (z), and (5) (mn), (v), (vo), and (zm), 45.03 (19), 45.07, 45.20, 45.21, 45.40

(1m), 45.41, 45.42, 45.43, and 45.82 and administered by the department of veterans affairs, including all moneys received from the federal government for the benefit of veterans or their dependents, and for the veteran grant jobs pilot program under s. 38.31 administered by the technical college system board; all moneys paid as interest on and repayment of loans under the post-war rehabilitation fund; soldiers rehabilitation fund, veterans housing funds as they existed prior to July 1, 1961; all moneys paid as interest on and repayment of loans under this fund; all moneys paid as expenses for, interest on, and repayment of veterans trust fund stabilization loans under s. 45.356, 1995 stats.; all moneys paid as expenses for, interest on, and repayment of veterans personal loans; the net proceeds from the sale of mortgaged properties related to veterans personal loans; all mortgages issued with the proceeds of the 1981 veterans home loan revenue bond issuance purchased with moneys in the veterans trust fund; all moneys received from the state investment board under s. 45.42 (8) (b); all moneys received from the veterans mortgage loan repayment fund under s. 45.37 (7) (a) and (c); and all gifts of money received by the board of veterans affairs for the purposes of this fund.

SECTION 534. 25.40 (1) (a) 14. of the statutes is amended to read:

25.40 (1) (a) 14. Fees received under ss. $85.51_{\overline{5}}$ 348.105, and 348.26 (2) that are deposited in the general fund and credited to the appropriation account under s. 20.395 (5) (dg).

SECTION 535. 25.40 (1) (a) 22. of the statutes is amended to read:

25.40 (1) (a) 22. Moneys received under s. 341.14 (6r) (b) 10. that are deposited into the general fund and credited to the appropriation account under s. $\frac{20.435}{(1)}$ (20.395 (5) (gi).

SECTION 536. 25.40 (1) (a) 24. of the statutes is amended to read:

25.40 (1) (a) 24. Moneys received under ss. 341.14 (6r) (b) 11. and 343.21 (1) (o) that are deposited into the general fund and credited to the appropriation account under s. 20.435 (1) (g) 20.395 (5) (gj).

SECTION 537. 25.43 (3) of the statutes is amended to read:

25.43 (3) Except for the purpose of investment as provided in s. 25.17 (2) (d), the environmental improvement fund may be used only for the purposes authorized under ss. 20.320 (1) (r), (s), (sm), (t), and (x), and (2) (s) and (x) and (3) (q), 20.370 (4) (mt), (mx) and (nz), (8) (mr) and (9) (mt), (mx) and (ny), 20.505 (1) (v), (x) and (y), 281.58, 281.59, 281.60, 281.61 and, 281.62, and 283.31.

SECTION 538. 25.46 (4) of the statutes is amended to read:

25.46 (4) The moneys specified under s. 94.681 (7) (a) 1. and 2. for environmental management.

SECTION 539. 25.46 (4s) of the statutes is amended to read:

25.46 (**4s**) The fees imposed under s. 94.681 (3m) and (4) for environmental management.

SECTION 540. 25.465 (3) of the statutes is amended to read:

25.465 (3) The fees collected under s. 94.681 (2), (5) and (6) (a) 3. (bm), except as provided in s. 94.681 (7) (a).

SECTION 541. 25.465 (8) of the statutes is amended to read:

25.465 **(8)** The fees collected under s. 94.72 (5) (b) and (6) (a) 1. and 2. and (i).

SECTION 542. 25.468 of the statutes is amended to read:

25.468 Agricultural chemical cleanup fund. There is established a separate nonlapsible trust fund designated as the agricultural chemical cleanup fund, to consist of all revenues collected under ss. 94.64 (3r) (b) and (4) (a) 5., 94.681 (3) and (6) (a) 4. (c), 94.685 (3) (a) 2., 94.703 (3) (a) 2. and 3., 94.704 (3) (a) 2. and 94.73 (5) (e) and (8).

SECTION **543.** 25.77 (13) of the statutes is repealed. SECTION **543c.** 25.77 (14) of the statutes is repealed. SECTION **544.** 25.80 of the statutes is amended to read:

25.80 Tuition trust fund. There is established a separate nonlapsible trust fund designated as the tuition trust fund, consisting of all revenue from enrollment fees and the sale of tuition units under s. 16.64 224.48.

SECTION 545. 25.85 of the statutes is amended to read:

25.85 College savings program trust fund. There is established a separate nonlapsible trust fund designated as the college savings program trust fund, consisting of all revenue from enrollment fees for and contributions to college savings accounts under s. 16.641 224.50 and from distributions and fees paid by the vendor under s. 16.255 224.51 (3) other than revenue from those sources that is deposited in the college savings program bank deposit trust fund or the college savings program credit union deposit trust fund.

SECTION 546. 25.853 of the statutes is amended to read:

25.853 College savings program bank deposit trust fund. There is established a separate nonlapsible trust fund designated as the college savings program bank deposit trust fund, consisting of all revenue from enrollment fees for and contributions to college savings accounts under s. 46.641 224.50 in which the investment instrument is an account held by a state or national bank, a state or federal savings bank, a state or federal savings and loan association, or a savings and trust company that has its main office or home office or a branch office in this state and that is insured by the Federal Deposit Insurance Corporation, and all revenue from distributions and fees

paid by the vendors of those investment instruments under s. 16.255 224.51 (3).

SECTION 547. 25.855 of the statutes is amended to read:

25.855 College savings program credit union deposit trust fund. There is established a separate non-lapsible trust fund designated as the college savings program credit union deposit trust fund, consisting of all revenue from enrollment fees for and contributions to college savings accounts under s. 16.641 224.50 in which the investment instrument is an account held by a state or federal credit union, including a corporate central credit union organized under s. 186.32, that has its main office or home office or a branch office located in this state and that is insured by the National Credit Union Administration, and all revenue from distributions and fees paid by the vendors of those investment instruments under s. 16.255 224.51 (3).

SECTION **547m.** 25.98 of the statutes is repealed. **SECTION 548.** 26.11 (6) of the statutes is amended to read:

26.11 (6) The department, as the director of the effort, may suppress a forest fire on lands located outside the boundaries of intensive or extensive forest fire protection districts but not within the limits of any city or village if the town responsible for suppressing fires within its boundaries spends more than \$3,000, as determined by rates established by the department, on suppressing the forest fire and if the town chairperson makes a request to the department for assistance. Persons participating in the suppression efforts shall act at the direction of the department after the department begins suppression efforts under this subsection. Funds expended by the state under this subsection shall be expended from the appropriation under s. 20.370 (1) (2) (mv).

SECTION 549. 26.11 (7) (a) of the statutes is amended to read:

26.11 (7) (a) Notwithstanding s. 20.001 (3) (c), if the sum of the unencumbered balances in the appropriation accounts under s. 20.370 (1) (2) (cs) and (mz) exceeds \$1,000,000 on June 30 of any fiscal year, the amount in excess of \$1,000,000 shall lapse from the appropriation account under s. 20.370 (1) (2) (cs) to the conservation fund, except as provided in par. (b).

SECTION 550. 26.11 (7) (b) of the statutes is amended to read:

26.11 (7) (b) Notwithstanding s. 20.001 (3) (c), if the amount in the appropriation account under s. 20.370 (1) (2) (cs) is insufficient for the amount that must lapse under par. (a), the remainder that is necessary for the lapse shall lapse from the appropriation account under s. 20.370 (1) (2) (mz).

SECTION 551. 26.14 (3) of the statutes is renumbered 26.14 (3) (a) and amended to read:

26.14(3) (a) Emergency fire wardens, and all persons employed by them or by any other duly appointed fire

warden for the purpose of suppressing forest fires, shall receive such hourly pay as the department may determine, for the time actually employed. Equipment operators and other specialists shall be paid the prevailing wage rate for comparable skills in each locality. And in addition thereto the The department may also allow the cost of meals, transportation, and disbursements for emergency equipment. One—half of such expense shall be paid by the state and one—half by

(b) Of the expenses incurred under par. (a) the state shall pay one—half and the county where such the service was performed shall pay one—half.

SECTION 552. 26.14 (3) (c) of the statutes is created to read:

26.14 (3) (c) If the state receives any payment of damages under sub. (9) (b), the county's share of expenses under par. (b) is reduced by the amount by which the damages received exceed the state's share of expenses under par. (b). If, at the time the damages are paid, the county has already paid its share of expenses to the state, the state shall reimburse the county the amount by which the damages received exceed the state's share of expenses.

SECTION 553. 26.14 (9) (b) of the statutes is amended to read:

26.14 (9) (b) Any person who sets a fire on any land and allows such fire to escape and become a forest fire shall be liable for all expenses incurred in the suppression of the fire by the state or town in which the fire occurred. For purposes of this paragraph, the state is considered to incur all expenses described under sub. (3). An action under this paragraph shall be commenced within the time provided by s. 893.91 or be barred.

SECTION 554. 26.39 (2) of the statutes is amended to read:

26.39 (2) FORESTRY EDUCATION CURRICULUM; SCHOOLS. Using the moneys appropriated under s. 20.370 (1) (2) (cu), the department, in cooperation with the Center for Environmental Education in the College of Natural Resources at the University of Wisconsin–Stevens Point, shall develop a forestry education curriculum for grades kindergarten to 12.

SECTION 554g. 27.01 (2) (a) of the statutes is amended to read:

27.01 (2) (a) Acquire by purchase, lease or agreement lands or waters suitable for state park purposes and may acquire such lands and waters by condemnation after obtaining approval of the senate and assembly committees on natural resources. The power of condemnation may not be used for the purpose of establishing or extending a recreational trail; a bicycle way, as defined in s. 340.01 (5s); a bicycle lane, as defined in s. 340.01 (5e); or a pedestrian way, as defined in s. 346.02 (8) (a).

SECTION 557. 27.01 (7) (f) 2. of the statutes is amended to read:

27.01 (7) (f) 2. Except as provided in subds. 3. and 4. and par. (gm) 4., the <u>department shall charge a</u> fee for a daily vehicle admission receipt is <u>of not less than</u> \$7.85 <u>but not more than \$12.85</u>, as <u>determined by the secretary</u>, for any vehicle <u>which that</u> has Wisconsin registration plates.

SECTION 558. 27.01 (7) (f) 3. of the statutes is amended to read:

27.01 (7) (f) 3. Subject to par. (gm) 5., the <u>department shall charge a</u> fee for a daily vehicle admission receipt <u>of not less than \$10.85 but not more than \$15.85</u>, as <u>determined by the secretary</u>, for a motor bus that has Wisconsin registration plates is \$10.85.

SECTION 559. 27.01 (7) (f) 4. of the statutes is amended to read:

27.01 (7) (f) 4. Notwithstanding subd. 3. and subject to par. (gm) 5., the <u>department shall charge a</u> fee for a daily vehicle admission receipt <u>of not less than \$3.35 but not more than \$8.35</u>, as <u>determined by the secretary</u>, for a motor bus <u>which that has Wisconsin registration plates and primarily transports residents from nursing homes located in this state is \$3.35, for any motor bus which has Wisconsin registration plates.</u>

SECTION 561. 27.01 (7) (g) 2. of the statutes is amended to read:

27.01 (7) (g) 2. Except as provided in subds. 3. and 4., the <u>department shall charge a</u> fee for a daily vehicle admission receipt <u>of not less than \$10.85 but not more than \$15.85</u>, as determined by the secretary, for any vehicle that has a registration plate or plates from another state is \$10.85.

SECTION 562. 27.01 (7) (g) 3. of the statutes is amended to read:

27.01 (7) (g) 3. Subject to par. (gm) 5., the <u>department shall charge a</u> fee for a daily vehicle admission receipt <u>of not less than \$14.85 but not more than \$19.85</u> for a motor bus that has a registration plate or plates from another state is \$14.85.

SECTION 563. 27.01 (7) (g) 4. of the statutes is amended to read:

27.01 (7) (g) 4. Notwithstanding subd. 3. and subject to par. (gm) 5., the <u>department shall charge a</u> fee for a daily vehicle admission receipt <u>of not less than \$5.85 but not more than \$10.85</u> for a motor bus-which that has a registration plate from another state and primarily transports residents from nursing homes located in this state is \$5.85, for any motor bus which has a registration plate or plates from another state.

SECTION 564. 27.01 (10) (d) 1. of the statutes is amended to read:

27.01 (10) (d) 1. The department shall charge a camping fee of not less than \$15 but not more than \$20 <u>\$30</u>, as determined by the secretary, for each night at a campsite in a state campground for a resident camping party, except as provided under par. (fm).

SECTION 565. 27.01 (10) (d) 2. of the statutes is amended to read:

27.01 (10) (d) 2. The department shall charge a camping fee of not less than \$19 but not more than \$25 <u>\$35</u>, as determined by the secretary, for each night at a campsite in a state campground for a nonresident camping party, except as provided under par. (fm).

SECTION 565b. 27.01 (10) (g) 5. of the statutes is amended to read:

27.01 (10) (g) 5. An Except as provided in subd. 6., an additional camping fee of \$10 per night for a camping party that uses electricity supplied at a state campground if the campsite has an electric receptacle.

SECTION 565d. 27.01 (10) (g) 6. of the statutes is created to read:

27.01 (10) (g) 6. An additional camping fee of \$15 per night for a camping party that uses electricity supplied at a state campground located in Devil's Lake State Park, High Cliff State Park, Kohler–Andrae State Park, Peninsula State Park, or Willow River State Park, if the campsite has an electric receptacle.

SECTION 565e. 27.01 (13) of the statutes is amended to read:

27.01 (13) POLICE SUPERVISION. The department shall have police supervision over all state parks, and its duly appointed wardens or representatives in charge of any state park may arrest, with or without warrant, any person within such park area, committing an offense against the laws of the state or in violation of any rule or regulation of the department in force in such state park, and deliver such person to the proper court of the county wherein such offense has been committed and make and execute a complaint charging such person with the offense committed. The district attorney of the county wherein such offense has been committed shall appear and prosecute all actions arising under this subsection.

SECTION 565f. 27.01 (15) (b) 1. of the statutes is amended to read:

27.01 (**15**) (b) 1. No more than 30 35 percent of all state park campsites in the state have electric receptacles.

SECTION 565g. 27.019 (10) of the statutes is amended to read:

27.019 (10) ACQUISITION OF LAND. Any county in which there does not exist a county park commission acting through its rural planning committee may acquire by gift, grant, devise, donation, or purchase, condemnation or otherwise, with the consent of the county board, a sufficient tract or tracts of land for the reservation for public use of river fronts, lake shores, picnic groves, outlook points from hilltops, places of special historic interest, memorial grounds, parks, playgrounds, sites for public buildings, and reservations in and about and along and leading to any or all of the same, and to develop and maintain the same for public use. The power of condemnation may not be used for the purpose of establishing or extend-

ing a recreational trail; a bicycle way, as defined in s. 340.01 (5s); a bicycle lane, as defined in s. 340.01 (5e); or a pedestrian way, as defined in s. 346.02 (8) (a).

SECTION 565h. 27.05 (3) of the statutes is amended to read:

27.05 (3) Acquire, in the name of the county, by purchase, land contract, lease, condemnation, or otherwise, with the approval and consent of the county board, such tracts of land or public ways as it deems suitable for park purposes; including lands in any other county not more than three–fourths of a mile from the county line; but no land so acquired shall be disposed of by the county without the consent of said commission, and all moneys received for any such lands, or any materials, so disposed of, shall be paid into the county park fund hereinafter established. The power of condemnation may not be used for the purpose of establishing or extending a recreational trail; a bicycle way, as defined in s. 340.01 (5s); a bicycle lane, as defined in s. 340.01 (5e); or a pedestrian way, as defined in s. 346.02 (8) (a).

SECTION 565p. 27.065 (1) (a) of the statutes is amended to read:

27.065 (1) (a) The county board of any county which shall have adopted a county system of parks or a county system of streets and parkways, pursuant to s. 27.04, may acquire the lands necessary for carrying out all or part of such plan by gift, purchase, condemnation or otherwise; provided, however, that no lands shall be acquired by condemnation unless and until the common council of the city or the board of trustees of the village or the board of supervisors of the town wherein such land is situated shall consent thereto. The power of condemnation may not be used for the purpose of establishing or extending a recreational trail; a bicycle way, as defined in s. 340.01 (5s); a bicycle lane, as defined in s. 340.01 (5e); or a pedestrian way, as defined in s. 346.02 (8) (a). The cost of acquiring such lands by purchase or condemnation may be paid in whole or in part by the county or by the property to be benefited thereby, as the county board shall direct but in no case shall the amount assessed to any parcel of real estate exceed the benefits accruing thereto; provided, that no assessment for paying the cost of acquiring lands may be levied or collected against the property to be benefited until the governing body of the city, village or town where such lands are located has by resolution determined that the public welfare will be promoted thereby. Title to all lands acquired hereunder shall be an estate in fee simple.

SECTION 565t. 27.08 (2) (b) of the statutes is amended to read:

27.08 (2) (b) To acquire in the name of the city for park, parkway, boulevard or pleasure drive purposes by gift, devise, bequest or condemnation, either absolutely or in trust, money, real or personal property, or any incorporeal right or privilege; except that no lands may be acquired by condemnation for the purpose of establish-

ing or extending a recreational trail; a bicycle way, as defined in s. 340.01 (5s); a bicycle lane, as defined in s. 340.01 (5e); or a pedestrian way, as defined in s. 346.02 (8) (a). Gifts to any city of money or other property, real or personal, either absolutely or in trust, for park, parkway, boulevard or pleasure drive purposes shall be accepted only after they shall have been recommended by the board to the common council and approved by said council by resolution. Subject to the approval of the common council the board may execute every trust imposed upon the use of property or property rights by the deed, testament or other conveyance transferring the title of such property to the city for park, parkway, boulevard or pleasure drive purposes.

SECTION 565w. 27.08 (2) (c) of the statutes is amended to read:

27.08 (2) (c) Subject to the approval of the common council to buy or lease lands in the name of the city for park, parkway, boulevard or pleasure drive purposes within or without the city and, with the approval of the common council, to sell or exchange property no longer required for its purposes. Every city is authorized, upon recommendation of its officers, board or body having the control and management of its public parks, to acquire by condemnation in the name of the city such lands within or without its corporate boundaries as it may need for public parks, parkways, boulevards and pleasure drives. The power of condemnation may not be used for the purpose of establishing or extending a recreational trail; a bicycle way, as defined in s. 340.01 (5s); a bicycle lane, as defined in s. 340.01 (5e); or a pedestrian way, as defined in s. 346.02 (8) (a).

SECTION 567. 28.05 (3) (c) of the statutes is amended to read:

28.05 (3) (c) Of the amount received by the department from each timber sale for which the department used the services of a cooperating forester or a private contractor under this subsection, the department shall credit to the appropriation account under s. 20.370 (1) (2) (cy) an amount equal to the portion of the sale proceeds that the department is required to pay to the cooperating forester or private contractor.

SECTION 570. 28.15 (4) of the statutes is amended to read:

28.15 (4) The department shall pay the initial costs of administering and implementing a cooperative agreement and any contracts entered into under sub. (3) from the appropriation under s. 20.370 + (1) + (2) + (20) +

SECTION 571. 28.15 (5) of the statutes is amended to read:

28.15 (5) On June 30 of each fiscal year, 10 percent of the revenues received by the department in that fiscal year from the sale of timber from federal land under a cooperative agreement under this section shall lapse from the appropriation account under s. 20.370 (1) (2) (cz) to

the conservation fund. These amounts shall be lapsed until the total amount lapsed equals \$750,000.

SECTION 572b. 29.164 (2) (a) of the statutes is amended to read:

29.164 (2) (a) Department authority. The department may regulate and limit the hunting of wild turkeys by issuing licenses under this section. The department may not require the possession or validation of a wild turkey carcass tag and may not require that a carcass tag be attached to a lawfully killed wild turkey.

SECTION 572c. 29.164 (2) (c) 2. of the statutes is amended to read:

29.164 (2) (c) 2. If the department establishes a wild turkey hunting zone where or a season time period during which wild turkey hunting is permitted, no person may hunt wild turkeys in that wild turkey hunting zone or during that season time period unless the person has a wild turkey hunting license and a valid wild turkey hunting stamp as required under subd. 1. and unless the person has a wild turkey hunting tag authorization that is valid for that zone and that time period.

SECTION 572d. 29.164 (3) (a) 1. of the statutes is amended to read:

29.164 (3) (a) 1. If the department requires wild turkey hunting licenses under sub. (2) (a) and the number of applications for wild turkey hunting tags authorizations for a given wild turkey hunting zone or a given wild turkey hunting season time period exceeds the number of available wild turkey hunting tags authorizations allocated by the department for that zone or that season time period, the department shall issue wild turkey hunting licenses and tags authorizations for that zone or that season time period according to the cumulative preference system under this subsection.

SECTION 572e. 29.164 (3) (a) 2. of the statutes is amended to read:

29.164 (3) (a) 2. If the department requires wild turkey hunting licenses under sub. (2) (a) and the number of applications for wild turkey hunting tags authorizations for a given wild turkey hunting zone or a given wild turkey hunting season time period does not exceed the number of available wild turkey hunting tags authorizations allocated by the department for that zone or that season time period, the department shall issue a wild turkey hunting license and tag authorization to each applicant.

SECTION 572f. 29.164 (3) (e) of the statutes is amended to read:

29.164 (3) (e) *Notification; issuance; payment.* The department shall issue a notice of approval to those qualified applicants selected to receive a wild turkey hunting license and tag authorization under par. (a). A person who receives a notice of approval and who pays the license fee in the manner required by the department shall be issued a wild turkey hunting license, subject to ss. 29.024 and 54.25 (2) (c) 1. d., and a tag an authorization.

The department may not charge a fee for -a tag an authorization that is issued under this paragraph.

SECTION 572g. 29.164 (3m) (a) (intro.) of the statutes is amended to read:

29.164 (**3m**) (a) (intro.) Notwithstanding sub. (3), the department may issue wild turkey hunting licenses and tags <u>authorizations</u> without requiring that the licenses and tags <u>authorizations</u> be issued pursuant to the cumulative preference system under sub. (3) to members of the U.S. armed forces who are all of the following:

SECTION 572h. 29.164 (3m) (b) of the statutes is amended to read:

29.164 (**3m**) (b) The department may not impose any deadline or other restriction on the timing for applications or issuing licenses or tags <u>authorizations</u> under this subsection. If a license and tag <u>authorization</u> under this subsection are issued during the applicable hunting season, the license and tag <u>authorization</u> shall authorize hunting beginning on the date of issuance.

SECTION 572i. 29.164 (3m) (c) of the statutes is amended to read:

29.164 (**3m**) (c) An applicant for a wild turkey hunting license under this subsection shall be <u>is</u> eligible for the same number of wild turkey hunting tags <u>authorizations</u> for each applicable season time period as an applicant who is issued tags <u>authorizations</u> under the cumulative preference system under sub. (3).

SECTION 572j. 29.164 (4) (title) of the statutes is amended to read:

29.164 **(4)** (title) WILD TURKEY HUNTING STAMPS; ADDITIONAL TAGS AUTHORIZATIONS.

SECTION 572k. 29.164 (4) (b) of the statutes is amended to read:

29.164 (4) (b) Additional tags authorizations. The department may issue the wild turkey hunting tags authorizations that were allocated for a given wild turkey hunting zone or season time period under sub. (3) (a) 2. but that were not issued. The department shall charge the fee specified in s. 29.563 (2) (f) or (g) for each of these additional tags authorizations. The issuance of -a tag an authorization under this paragraph does not affect the priority that the person receiving the tag authorization may have under the cumulative preference system.

SECTION **572L.** 29.171 (3) of the statutes is repealed. SECTION **572m.** 29.172 (3) of the statutes is repealed. SECTION **572n.** 29.173 (3) of the statutes is repealed. SECTION **572ng.** 29.192 (1) (b) of the statutes is amended to read:

29.192 (1) (b) Requiring registration of each farm on which Canada goose hunting is allowed and registration of each goose killed at the farm.

SECTION 572nj. 29.192 (1m) of the statutes is created to read:

29.192 (1m) The department may not do any of the following:

- (a) Require a person to indicate on his or her hunting permit or otherwise record each Canada goose killed by the person.
- (b) Require a person to report to the department more than once annually each Canada goose killed by the person.

SECTION 5720. 29.211 (3) of the statutes is repealed. SECTION 572p. 29.216 (3) of the statutes is repealed. SECTION 572q. 29.217 (3) of the statutes is repealed. SECTION 573. 29.219 (3) (c) of the statutes is amended to read:

29.219 (3) (c) *Use of fees.* The department shall deposit receipts from the sale of resident 2–day sports fishing licenses under this subsection in the conservation fund. The department shall credit 50 percent of these receipts to the appropriation under s. 20.370 (4) (1) (ku).

SECTION 574. 29.219 (3m) (c) of the statutes is amended to read:

29.219 (**3m**) (c) *Use of fees.* The department shall deposit receipts from the sale of 2–day inland lake trout fishing licenses under this subsection in the conservation fund. The department shall credit 50 percent of these receipts to the appropriation account under s. 20.370 (4) (1) (ky).

SECTION 575. 29.228 (7) (c) of the statutes is amended to read:

29.228 (7) (c) *Use of fees.* The department shall deposit receipts from the sale of nonresident 2–day sports fishing licenses under this subsection in the conservation fund. The department shall credit 50 percent of these receipts to the appropriation under s. 20.370 (4) (1) (ku).

SECTION 576. 29.2285 (3) (e) of the statutes is amended to read:

29.2285 (3) (e) *Use of moneys from fees.* The department shall deposit the receipts from the sale of sturgeon hook and line tags issued under this subsection into the conservation fund and shall credit these receipts to the appropriation account under s. 20.370 (4) (1) (ky).

SECTION **576g.** 29.231 (4) of the statutes is repealed. SECTION **576r.** 29.235 (4) of the statutes is repealed. SECTION **578.** 29.237 (5) of the statutes is amended or read:

29.237 (5) The department shall deposit receipts from the sale of sturgeon spearing licenses under this subsection into the conservation fund and shall credit these receipts to the appropriation account under s. 20.370 (4) (1) (kw) for assessing and managing the lake sturgeon stock and fishery in the Lake Winnebago system, for improving and maintaining lake sturgeon habitat in the Lake Winnebago and upper Fox and Wolf rivers system, and for administering this section.

SECTION 578b. 29.324 (1) (c) of the statutes is repealed.

SECTION 578e. 29.324 (1) (d) of the statutes is repealed.

SECTION 578h. 29.324 (2) (intro.) and (a) of the statutes are consolidated, renumbered 29.324 (2) and amended to read:

29.324 (2) Except as provided in sub. (4), any Any member of a group deer hunting party may kill a deer for another member of the group deer hunting party if both of the following conditions exist: (a) At at the time and place of the kill, the person who kills the deer is in contact with the person for whom the deer is killed.

SECTION 578L. 29.324 (2) (b) of the statutes is repealed.

SECTION **578p.** 29.324 (3) of the statutes is repealed. SECTION **578r.** 29.324 (4) of the statutes is repealed. SECTION **578u.** 29.347 (2) of the statutes is amended to read:

29.347 (2) DEER OR ELK ELK CARCASS TAGS; DEER AND ELK REGISTRATION. Except as provided under ss. 29.324 (3) and 29.349 (2), any person who kills a deer shall immediately validate a deer carcass tag that is authorized for the type of deer killed. Any person who kills an elk shall immediately validate an elk carcass tag. Except as provided under sub. (2m) or s. 29.89 (6), no person may possess, control, store, transport, transfer, or dispose of a deer carcass unless a deer carcass tag has been validated in the manner required by the department. Except as provided under sub. (2m), no person may possess, control, store, transport, transfer, or dispose of an elk carcass unless an elk carcass tag has been validated in the manner required by the department. A person who kills a deer or an elk shall register the deer or elk in the manner required by the department. No person may possess, control, store, transport, transfer, or dispose of a deer or elk carcass unless in compliance with rules promulgated by the department under s. 29.063 (3).

SECTION 578v. 29.347 (2m) (b) of the statutes is amended to read:

29.347 (**2m**) (b) Subject to sub. (6), any person who retains a tag under par. (a) or who complies with s. 29.349 (2) (bm), if applicable, may give deer or elk meat to another person. The person who receives the <u>a</u> gift of deer or elk meat is not required to possess a tag.

SECTION 578y. 29.349 (2) (b) (intro.) of the statutes is amended to read:

29.349 (2) (b) (intro.) No person may take possession of the carcass of a bear or wild turkey killed in the manner specified in par. (a) and remove the carcass from the scene of the accident unless one of the following apply applies:

SECTION 578ym. 29.349 (4) of the statutes is renumbered 84.07 (7) and amended to read:

84.07 (7) DEER KILLED BY VEHICLES. The department shall establish a program contract with counties, municipalities, or private entities for the removal and disposal of deer killed by vehicles on state trunk highways.

Vetoed In Part

SECTION 579. 29.506 (7m) (a) of the statutes is amended to read:

29.506 (7m) (a) The department shall issue a taxidermy school permit to a person who applies for the permit; who, on August 15, 1991, holds a valid taxidermist permit issued under this section; and who, on August 15, 1991, operates a taxidermy school approved by the educational approval board under s. 38.50 38.51, 1989 stats.

SECTION 579e. 29.559 (1) (c) of the statutes is amended to read:

29.559 (1) (c) Any person, including the department, who issues a wild turkey hunting tag authorization under s. 29.164 (4) (b) or a sturgeon hook and line tag under s. 29.2285 (3) (b) shall collect, in addition to the statutory fee, an issuing fee for each authorization or tag that the person is issued. A person appointed under s. 29.024 (6) (a) 2., 3., or 4. may retain 15 cents for each issuing fee of each authorization or tag to compensate for services in issuing the authorization or tag.

SECTION 579m. 29.563 (2) (f) of the statutes is amended to read:

29.563 (2) (f) Resident tags authorizations. Each additional wild turkey hunting tag authorization issued to a resident under s. 29.164 (4) (b): \$9.75.

SECTION 579s. 29.563 (2) (g) of the statutes is amended to read:

29.563 (2) (g) Nonresident tags authorizations. Each additional wild turkey hunting tag authorization issued to a nonresident under s. 29.164 (4) (b): \$14.75.

SECTION 581g. 29.563 (12) (a) 2. of the statutes is amended to read:

29.563 (12) (a) 2. Archer, crossbow, sports, or conservation patron, except as provided in subd. 2m: \$14.25 if deer carcass tags are included; \$11.25 after open season and deer carcass tags are not included.

SECTION 581r. 29.563 (14) (c) 6. of the statutes is amended to read:

29.563 (14) (c) 6. Each wild turkey hunting tag authorization issued under s. 29.164 (4) (b) or sturgeon hook and line tag issued under s. 29.2285 (3) (b): 25 cents.

SECTION 582. 29.564 (2) of the statutes is amended to read:

29.564 (2) All moneys collected under sub. (1), less the amount retained as authorized under sub. (1m), shall be deposited into the account under s. 20.370 (3) (is) (9)

SECTION 582g. 29.888 (title) of the statutes is amended to read:

29.888 (title) Wolf depredation program; wolf damage claims.

SECTION 582h. 29.888 (5) of the statutes is created

29.888 (5) (a) From the sources described in par. (c), the department shall pay damage claims for damage caused by wolves during the period when the wolf is listed on the federal endangered list or on the state endangered list.

- (b) The department shall pay damage claims under par. (a) as soon as practicable after determining that the claim is eligible to be paid. The department may not Vetoed make the payments under par. (a) on a prorated basis.
- (c) The department shall make the payments under Vetoed par. (a) from available federal funds to the extent permit- In Part ted by federal law. If the department determines that the amount available from federal funds is insufficient in a given fiscal year to make all of these payments, the department shall make the remainder of the payments from the appropriation accounts under s. 20.370 (1) (fb), (fe), and (fs). If the department determines that the amount available under s. 20.370 (1) (fb), (fe), and (fs) is insufficient in a given fiscal year to pay the claims under par. (a) that remain after federal funds are used, the **Vetoed** department may request the joint committee on finance to take action under s. 13.101. The requirement of a finding of emergency under s. 13.101 (3) (a) 1. does not apply to such a request.

SECTION 582m. 29.889 (4) (bn) of the statutes is amended to read:

29.889 (4) (bn) Type of wildlife damage. In order to be eligible for wildlife damage abatement assistance, the type of wildlife damage to be abated shall be limited to damage to commercial seedings or crops growing on agricultural land, damage to crops that have been harvested for sale or further use but that have not been removed from the agricultural land, damage to orchard trees or nursery stock or, damage to apiaries or livestock, or damage to crops or grasses grown for use by a bird hunting preserve licensed under ch. 169.

SECTION 582p. 29.889 (5) (bs) of the statutes is created to read:

29.889 (5) (bs) Abatement authorized. Notwithstanding par. (bm), for damage caused by elk to crops or grasses grown for use by a bird hunting preserve licensed under ch. 169, a participating county may recommend fencing the affected property as a damage abatement measure.

SECTION 582r. 29.89 (6) of the statutes is repealed. **SECTION 583.** 29.984 (2) of the statutes is amended to read:

29.984 (2) Use of commercial fish protection sur-CHARGE FUNDS. All moneys collected from commercial fish protection surcharges shall be credited to the appropriation under s. 20.370 (4) (1) (kr).

SECTION 584. 29.9905 (2) of the statutes is amended

29.9905 (2) Use of Great Lakes resource sur-CHARGE FUNDS. All moneys collected from Great Lakes resource surcharges shall be credited to the appropriation under s. 20.370 (4) (1) (kr).

In Part

In Part

SECTION 584g. 30.01 (1c) of the statutes is renumbered 30.01 (1c) (a) and amended to read:

30.01 (1c) (a) "Boat shelter" means a structure in navigable waters with a roof but no walls and, except as provided in par. (b), no sides, designed and constructed for the purpose of providing cover for a berth place for watercraft, which has a roof but does not have walls or sides. Such a structure may include a device for lifting

SECTION 584j. 30.01 (1c) (b) of the statutes is created to read:

30.01 (1c) (b) "Boat shelter" includes a structure under par. (a) that has temporary sides made of flexible material with a minimum openness factor of 5 percent if all of the following conditions are met:

- 1. The sides are placed and maintained by the owner or easement holder of adjacent riparian land or his or her agent.
- 2. The boat is registered under subch. V or exempt from registration requirements under s. 30.51 and either has a wooden hull or is designated as a boat with significant historic or cultural value, as determined by the state historical society or a local or county historical society established under s. 44.03.
- 3. The sides are located entirely within the riparian owner's riparian zone.
- 4. There are no more than 2 boat shelters for the first 100 feet of the riparian owner's shoreline footage and no more than one additional boat shelter for each additional 50 feet of the riparian owner's shoreline footage. For purposes of this subdivision, shoreline footage is measured along a straight line connecting points where property lines meet the ordinary high-water mark.
- 5. The sides are placed no less than 36 inches above the water surface.
- 6. The structure is no more than 24 feet in length, unless the boat that will be sheltered is longer than 24 feet from bow to stern.

SECTION 584m. 30.01 (5r) of the statutes is created to read:

30.01 (5r) "Riparian zone" means the area that extends from riparian land waterward to the line of navigation as determined by a method that establishes riparian zone lines between adjacent riparian owners in a manner that equitably apportions access to the line of navigation.

SECTION 584r. 30.12 (1g) (jm) of the statutes is created to read:

- 30.12 (1g) (jm) Riprap in an amount not to exceed 200 linear feet that is placed in a river or inland lake, or in an amount not to exceed 300 linear feet that is placed in a Great Lakes water body, and to which all of the following apply:
- 1. The riprap is clean fieldstone or quarry stone with a diameter of no less than 6 inches and no greater than 48 inches.

- 2. The toe of the riprap does not extend more than 8 feet waterward of the ordinary high-water mark.
- 3. The final riprap slope is not steeper than one foot horizontal to 1.25 feet vertical.
- 4. The riprap does not reach an elevation higher than 36 inches above the ordinary high-water mark or above the storm-wave height, as calculated using a method established by the department by rule, whichever is
- 5. No fill material or soil is placed in a wetland and, aside from riprap and, under subd. 7, gravel, no fill material or soil is placed below the ordinary high-water mark of any navigable waterway.
- 6. The riprap follows the natural contour of the shoreline.
- 7. Filter fabric or clean-washed gravel is used as a filter layer under the riprap.

SECTION 585. 30.52 (3m) (b) of the statutes is amended to read:

30.52 (3m) (b) All moneys collected under par. (a), less the amount retained as authorized under par. (am), shall be deposited into the account under s. 20.370 (3) (is)

SECTION 585g. 32.015 of the statutes is created to read:

32.015 Limitations. Property may not be acquired by condemnation to establish or extend a recreational trail; a bicycle way, as defined in s. 340.01 (5s); a bicycle lane, as defined in s. 340.01 (5e); or a pedestrian way, as defined in s. 346.02 (8) (a).

SECTION 585h. 32.02 (1) of the statutes is amended **Vetoed**

In Part

32.02 (1) Any county, town, village, city, including villages and cities incorporated under general or special acts, school district, the department of health services, the department of corrections, the board of regents of the University of Wisconsin System, the building commission, a commission created by contract under s. 66.0301, with the approval of the municipality in which condemnation is proposed, a commission created by contract under s. 66.0301 or 66.0303 that is acting under s. 66.0304, if the condemnation occurs within the boundaries of a member of the commission, or any public board or commission, for any lawful purpose, but in the case of city and village boards or commissions approval of that action is required to be granted by the governing body. A mosquito control commission, created under s. 59.70 (12), and a local professional football stadium district board, created under subch. IV of ch. 229, may not acquire property by condemnation.

SECTION 585i. 32.02 (3) of the statutes is amended to **Vetoed** read:

In Part

32.02 (3) Any Subject to s. 32.03 (7), any railroad corporation, any grantee of a permit to construct a dam to develop hydroelectric energy for sale to the public, any Wisconsin plank or turnpike road corporation, any

Vetoed In Part drainage corporation, any interstate bridge corporation, or any corporation formed under chapter 288, laws of 1899, for any public purpose authorized by its articles of incorporation.

SECTION 585k. 32.03 (7) of the statutes is created to read:

32.03 (7) A railroad corporation may not acquire by condemnation any property or property interest that exceeds a width of 100 feet unless law is enacted that includes the legislative findings that the acquisition serves the public interest and that authorizes the acquisition.

SECTION 585m. 32.25 (4) of the statutes is created to read:

32.25 (4) The department of administration may assess condemnors required to file relocation payment plans and relocation assistance service plans under sub. (1). The department of administration shall prescribe a methodology to determine the amount of the assessments such that the amount of an assessment reflects the approximate costs incurred by the department in connection with reviewing and approving the plans filed by the condemnor. Assessments under this subsection shall be paid to the department of administration and credited to the appropriation account under s. 20.505 (1) (kr).

SECTION 585n. 32.28 (1) of the statutes is renumbered 32.28 (1) (intro.) and amended to read:

32.28 (1) (intro.) In this section, "litigation:

(b) "Litigation expenses" means the sum of the costs, disbursements and expenses, including reasonable attorney, appraisal and engineering fees necessary to prepare for or participate in actual or anticipated proceedings before the condemnation commissioners, board of assessment or any court under this chapter.

SECTION 585ng. 32.28 (1) (a) of the statutes is created to read:

32.28 (1) (a) "Consumer price index" means the average of the consumer price index over each 12-month period, all items, U.S. city average, as determined by the bureau of labor statistics of the U.S. department of labor.

SECTION 585nm. 32.28 (3) (d) of the statutes is amended to read:

32.28 (3) (d) The award of the condemnation commission under s. 32.05 (9) or 32.06 (8) exceeds the jurisdictional offer or the highest written offer prior to the jurisdictional offer by at least \$700 the amount specified in sub. (4) and at least 15 percent and neither party appeals the award to the circuit court;

SECTION 585nr. 32.28 (3) (e) of the statutes is amended to read:

32.28 (3) (e) The jury verdict as approved by the court under s. 32.05 (11) exceeds the jurisdictional offer or the highest written offer prior to the jurisdictional offer by at least \$700 the amount specified in sub. (4) and at least 15 percent;

SECTION 585pg. 32.28 (3) (f) of the statutes is amended to read:

32.28 (3) (f) The condemnee appeals an award of the condemnation commission which exceeds the jurisdictional offer or the highest written offer prior to the jurisdictional offer by at least \$700 the amount specified in sub. (4) and at least 15 percent, if the jury verdict as approved by the court under s. 32.05 (10) or 32.06 (10) exceeds the award of the condemnation commission by at least \$700 and at least 15 percent;

SECTION 585pm. 32.28 (3) (g) of the statutes is amended to read:

32.28 (3) (g) The condemnor appeals the award of the condemnation commission, if the jury verdict as approved by the court under s. 32.05 (10) or 32.06 (10) exceeds the jurisdictional offer or the highest written offer prior to the jurisdictional offer by at least \$700 the amount specified in sub. (4) and at least 15 percent;

SECTION 585pr. 32.28 (3) (h) of the statutes is amended to read:

32.28 (3) (h) The condemnee appeals an award of the condemnation commission which does not exceed the jurisdictional offer or the highest written offer prior to the jurisdictional offer by 15 percent, if the jury verdict as approved by the court under s. 32.05 (10) or 32.06 (10) exceeds the jurisdictional offer or the highest written offer prior to the jurisdictional offer by at least \$700 the amount specified in sub. (4) and at least 15 percent; or

SECTION 585q. 32.28 (3) (i) of the statutes is amended to read:

32.28 (3) (i) The condemnee appeals an assessment of damages and benefits under s. 32.61 (3), if the judgment is at least \$700 the amount specified in sub. (4) and at least 15 percent greater than the award made by the city.

SECTION 585qm. 32.28 (4) of the statutes is created to read:

32.28 (4) (a) The amount for the purposes of sub. (3) (d) to (i) shall be \$2,700, adjusted as specified in par. (b).

(b) Beginning on January 1, 2018, and annually on January 1 thereafter, the department of administration shall adjust the dollar amount specified in par. (a) by an amount equal to that dollar amount multiplied by the percentage change in the consumer price index for the prior year, rounded to the nearest dollar. The department shall publish the dollar amounts on its Internet site. Notwithstanding s. 227.10, the adjusted dollar amounts need not be promulgated as rules under ch. 227.

SECTION 585r. 32.51 (1) (intro.) of the statutes is amended to read:

32.51(1) PURPOSES. (intro.) In addition to the powers granted under subch. I <u>and subject to the limitations under s. 32.015</u>, any city may condemn or otherwise acquire property under this subchapter for:

SECTION 586. 34.045 (title) of the statutes is repealed and recreated to read:

34.045 (title) Secretary of administration.

SECTION 587. 34.045 (1) (intro.) of the statutes is amended to read:

34.045 (1) (intro.) The depository selection board secretary of administration or his or her designee shall:

SECTION 588. 34.045 (1) (bm) of the statutes is amended to read:

34.045 (1) (bm) Direct the secretary of administration to maintain Maintain compensating balances, or direct the investment board to pay bank service costs as allocated by the secretary of administration under s. 25.19 (3) directly from the income account of the state investment fund, or by a combination of such methods.

SECTION 589. 34.045 (2) of the statutes is amended to read:

34.045 (2) In the exercise of its authority, the depository selection board The secretary of administration or his or her designee shall require any state department or agency to submit to it him or her for prior review, elimination, consolidation, renegotiation, or confirmation any existing service contract or service proposed by the department or agency.

SECTION 590. 34.045 (3) of the statutes is amended to read:

34.045 (3) The board secretary of administration or his or her designee may, for cause, disapprove any contract submitted to it under sub. (2) if it he or she finds the proposed contract to be in violation of the guidelines established under sub. (1), or to have been improperly negotiated or to be otherwise illegal. If the board secretary of administration or his or her designee fails to disapprove a proposed contract within 60 days after it is submitted by the department or agency, the contract shall be deemed approved. The board secretary of administration or his or her designee shall provide written justification for disapproving a contract proposed by a state agency or department. A disapproval is subject to judicial review under ch. 227.

SECTION 591. 34.045 (4) of the statutes is amended to read:

34.045 (4) State agencies and departments shall provide the board secretary of administration or his or her designee with a written justification for any proposed contract award for service.

SECTION 594g. 36.09 (1) (e) of the statutes is amended to read:

36.09 (1) (e) The Subject to par. (em), the board shall appoint a president of the system; a chancellor for each institution; a dean for each college campus; the state geologist; the director of the laboratory of hygiene; the director of the psychiatric institute; the state cartographer; and the requisite number of officers, other than the vice presidents, associate vice presidents, and assistant vice presidents of the system; faculty; academic staff;

and other employees and fix the salaries, subject to the limitations under par. (j) and s. 230.12 (3) (e), the duties and the term of office for each. The board shall fix the salaries, subject to the limitations under par. (j) and s. 230.12 (3) (e), and the duties for each chancellor, vice president, associate vice president, and assistant vice president of the system. No sectarian or partisan tests or any tests based upon race, religion, national origin, or sex shall ever be allowed or exercised in the appointment of the employees of the system.

SECTION 594r. 36.09 (1) (em) of the statutes is created to read:

36.09 (1) (em) Neither the board nor any institution may adopt any policy or promulgate any rule that requires the board to consider for appointment as president of the system, chancellor, or vice chancellor only those individuals who are faculty members of the system or another institution of higher education, who have been granted tenure within the system or another institution of higher education, or who hold the highest level of academic degree in a field of study or profession.

SECTION 600m. 36.11 (6) (c) of the statutes is amended to read:

36.11 (6) (c) By April 10, 1998, and annually thereafter February 10 of each year, the board shall develop and submit to the higher educational aids board for its review under s. 39.285 (1) a proposed formula for the awarding of grants under s. 39.435, except for grants awarded under s. 39.435 (2) or (5), for the upcoming academic next fiscal year to students enrolled in the system.

SECTION 602m. 36.11 (56) of the statutes is amended to read:

36.11 (56) Travel policies. Effective July 1, 2013, the board shall establish travel policies for system employees and a schedule for the reimbursement of system employees for travel expenses. Beginning on the effective date of this subsection [LRB inserts date], except with respect to contracts in effect on that date, this schedule may not provide for reimbursement of system employees for lodging expenses incurred in this state at a rate that exceeds the maximum rate for lodging expenses incurred in the same location in this state under the approved uniform travel schedule incorporated under s. 20.916 (8) (b) into the current compensation plan under s. 230.12 (1).

SECTION 602p. 36.11 (57) of the statutes is created to read:

36.11 (57) CLASSIFICATION OF SEGREGATED FEES. (a) The board shall revise its policies regarding student segregated fees to ensure that the classification of those fees as allocable or nonallocable is consistent across institutions

(b) The board shall submit the revised policies under par. (a) to the joint committee on finance. If the cochairpersons of the committee do not notify the board that the committee has scheduled a meeting for the purpose of reviewing the revised policies within 14 working days after the date of the submittal, the revised policies are considered approved and may be implemented. If, within 14 working days after the date of the submittal, the cochairpersons of the committee notify the board that the committee has scheduled a meeting for the purpose of reviewing the revised policies, the revised policies may be implemented only upon approval of the committee.

(c) Notwithstanding s. 36.27 (1), the board may not approve an increase in the allocable portion of segregated fees at any institution until the joint committee on finance has approved the board's revised policies in the manner provided in par. (b).

SECTION 602r. 36.11 (58) of the statutes is created to read:

36.11 (58) Transfers to the University of Wis-CONSIN OSHKOSH FOUNDATION. The board may not transfer funds to the University of Wisconsin Oshkosh Foundation, Inc., unless the transfer is first approved by legislative enactment.

SECTION 603m. 36.112 of the statutes is created to read:

- 36.112 Performance funding; innovation fund. (1) DEFINITION. In this section, "institution" includes the extension.
- (2) GOALS; METRICS. (a) The legislature hereby establishes the following goals for the system:
 - 1. Growing and ensuring student access.
- 2. Improving and excelling at student progress and completion.
 - 3. Expanding contributions to the workforce.
- 4. Enhancing operational efficiency and effectiveness.
- (b) For each goal specified in par. (a), the Board of Regents shall identify at least 4 metrics to measure an institution's progress toward meeting the goal. As the Board of Regents determines is appropriate, the board may specify different metrics for the extension. For each goal, each institution shall select one of the metrics for improving its performance and one of the metrics for maintaining excellence.
- (3) OUTCOMES-BASED FUNDING FORMULA. (a) The Board of Regents shall develop a formula for distributing under par. (b) the amount allocated under sub. (4) among the institutions based on each institution's performance with respect to the metrics the institution selects under sub. (2) (b), except that no more than 30 percent of the amount allocated in a fiscal year may be distributed based on the metrics selected for maintaining excellence.
- (b) By no later than February 15, 2018, the Board of Regents shall submit the formula developed under par. (a) to the joint committee on finance. If the cochairpersons of the joint committee on finance do not notify the Board of Regents within 14 working days after the date of submittal that the committee has scheduled a meeting to review the formula, the Board of Regents shall use the

formula to distribute the amount allocated under sub. (4) **Vetoed** among the institutions. If, within 14 working days after In Part the date of submittal, the cochairpersons of the joint committee on finance notify the Board of Regents that the committee has scheduled a meeting to review the formula, the Board of Regents may use the formula to distribute the amount allocated under sub. (4) among the institutions only as modified or approved by the committee. The joint committee on finance shall consult with the appropriate standing committee in each house before modifying or approving the formula.

- (4) ALLOCATION. In each fiscal year beginning in fiscal year 2018-19, the Board of Regents shall allocate \$26,250,000 of the amount appropriated under s. 20.285 (1) (a) to distribute to institutions under the formula under sub. (3) (b).
- (5) REPORT. (a) Beginning in fiscal year 2018–19, the Board of Regents shall submit an annual report to the joint committee on finance that describes how the Board of Regents distributed in the fiscal year the amount allocated under (4) to the institutions under the formula under sub. (3) (b). The report shall describe all of the following:
- 1. The amount distributed to each institution under the formula in the fiscal year.
- 2. The performance of each institution with respect to all the metrics identified by the Board of Regents under sub. (2) (b).
- 3. The methodology used to make the distributions based on each institution's performance with respect to the metrics selected by the institution.
- 4. The performance of the system as a whole with **In Part** respect to all the metrics identified by the Board of Regents under sub. (2) (b).
- 5. Any other information used to administer the requirements of this section.
- (b) The Board of Regents shall make the report required under par. (a) available to the public, and each institution shall post the report on its Internet site.
- (6) Innovation fund. In fiscal year 2017–18, the Board of Regents shall allocate \$5,000,000 of the amount appropriated under s. 20.285 (1) (a) for the board to distribute to institutions to increase enrollments in highdemand degree programs identified under sub. (7) (b) . Vetoed The Board of Regents shall make the distribution through In Part a competitive process involving a request for proposals from the institutions.
- (7) OTHER DUTIES. The Board of Regents shall do all of the following:
- (a) Identify baseline data sets for the goals specified in sub. (2) (a).
- (b) Identify degree programs that qualify as high Vetoed demand for each institution.
- (c) Approve a peer group for each institution that includes institutions of higher education with comparable missions and service populations.

Vetoed

In Part

Vetoed In Part

Vetoed

In Part

In Part

SECTION 604. 36.115 (2) of the statutes is amended to read:

36.115 (2) The Except as provided in sub. (8), the board shall develop a personnel system that is separate and distinct from the personnel system under ch. 230 for all system employees except system employees assigned to the University of Wisconsin–Madison.

SECTION 605. 36.115 (3) of the statutes is amended to read:

36.115 (3) The Except as provided in sub. (8), the chancellor shall develop a personnel system that is separate and distinct from the personnel system under ch. 230 for all system employees assigned to the University of Wisconsin–Madison.

SECTION 606. 36.115 (5) (b) of the statutes is amended to read:

36.115 (5) (b) The board may not implement the personnel system developed under sub. (2) unless it has been approved by the joint committee on employment relations. This paragraph does not apply to revisions made under sub. (8) (b).

SECTION 607. 36.115 (5) (c) of the statutes is amended to read:

36.115 (5) (c) The chancellor may not implement the personnel system developed under sub. (3) unless it has been approved by the joint committee on employment relations. This paragraph does not apply to revisions made under sub. (8) (b).

SECTION 608. 36.115 (7) of the statutes is amended to read:

36.115 (7) The Except as provided in sub. (8), the board shall establish and maintain consistent employment relations policies and practices for all system employees except system employees assigned to the University of Wisconsin–Madison. The, and the chancellor shall establish and maintain consistent employment relations policies and practices for all system employees assigned to the University of Wisconsin–Madison.

SECTION 609. 36.115 (8) of the statutes is created to read:

- 36.115 (8) (a) No later than January 1, 2018, the board shall develop and implement a plan that includes all of the following for each institution within the system, including the University of Wisconsin–Madison:
- 1. Policies for monitoring teaching workloads of faculty and instructional academic staff, including requirements for individual faculty and instructional academic staff members to report the number of hours spent teaching to the system administration.
- 2. Policies for rewarding faculty and instructional academic staff who teach more than a standard academic load.
- (b) The board and the chancellor shall revise the personnel systems developed under subs. (2) and (3) and the employment relations policies and practices established under sub. (7) as necessary to ensure that the systems,

policies, and practices are consistent with the plan required under par. (a).

SECTION 610m. 36.25 (13i) of the statutes is created to read:

36.25 (13i) PRECISION MEDICINE MOLECULAR TUMOR BOARD. From the appropriation under s. 20.285 (1) (a), the Board of Regents shall allocate \$490,000 in each fiscal year to expand precision medicine at the University of Wisconsin Carbone Cancer Center through the Precision Medicine Molecular Tumor Board by increasing access to precision medicine for cancer patients in this state, providing genomics resources to patients in need, and developing a precision medicine statewide database.

SECTION 610p. 36.25 (54) of the statutes is created to read:

36.25 (54) SCHOOL OF ENGINEERING AT UNIVERSITY OF WISCONSIN-GREEN BAY. The board may create a school of engineering at the University of Wisconsin-Green Bay.

SECTION 612m. 36.27 (2) (br) of the statutes is created to read:

36.27 (2) (br) 1. Subject to subd. 3., a student who meets all of the following requirements is entitled to the exemption under par. (a):

- a. The student has resided in this state for 6 months immediately preceding the beginning of any semester or session in which the student enrolls at an institution.
- b. The student is a member of the Wisconsin national guard or a reserve unit of the U.S. armed forces when he or she enrolls in an institution and has been a member for the 6-month period under subd. 1. a.
- c. While enrolled in an institution, the student continues to be a member of the Wisconsin national guard or reserve unit of the U.S. armed forces, or is honorably discharged or released under honorable conditions from the national guard or a reserve unit of the U.S. armed forces.
- 2. If a student receiving the exemption under this paragraph withdraws from an institution during a semester or session because he or she is called into state active duty or into active service with the U.S. armed forces for at least 30 days, the student is entitled to the exemption under this paragraph if he or she reenrolls in an institution during the semester in which he or she is discharged, demobilized, or deactivated from active duty or in the next succeeding semester.
- 3. The exemption under this paragraph is limited to 128 credits or 8 semesters, whichever is longer.

SECTION 613. 36.27 (3n) (a) 1m. a. of the statutes is amended to read:

36.27 (3n) (a) 1m. a. A person who has served on active duty under honorable conditions in the U.S. armed forces, in forces incorporated as part of the U.S. armed forces, in the national guard, or in a reserve component of the U.S. armed forces; who was a resident of this state at the time of entry into that service or resided in this state for at least 5 consecutive years after the person attained

the age of 18; and who, while a resident of this state, died on active duty, died as the result of a service—connected disability, or died in the line of duty while on active or inactive duty for training purposes.

SECTION 614. 36.27 (3n) (a) 1m. b. of the statutes is amended to read:

36.27 (**3n**) (a) 1m. b. A person who was a resident of this state at the time of entry into service described in subd. 1m. a. or resided in this state for at least 5 consecutive years immediately preceding the beginning of any semester or session for which the person's spouse or child described in par. (b) 1., 2., or 3. registers at an institution, and who the U.S. department of veteran affairs has awarded at least a 30 percent service—connected disability rating under 38 USC 1114 or 1134.

SECTION 614d. 36.27 (3n) (b) (intro.) of the statutes is amended to read:

36.27 (**3n**) (b) (intro.) Except as provided in par. pars. (bd) and (bg), the board shall grant full remission of academic fees and segregated fees for 128 credits or 8 semesters, whichever is longer, less the number of credits or semesters for which the person received remission of fees under s. 38.24 (7) and less the amount of any academic fees or segregated fees paid under 38 USC 3319, to any resident student who maintains a cumulative grade point average of at least 2.0 and is also any of the following:

SECTION 614h. 36.27 (3n) (bd) of the statutes is created to read:

36.27 (**3n**) (bd) If an eligible veteran was not a resident of this state at the time of entry into service described in par. (a) 1m. a., the board may grant a remission of academic fees and segregated fees under this subsection only if the eligible veteran's spouse or child described in par. (b) 1., 2., or 3. has resided in this state for at least 5 consecutive years immediately preceding the spouse's or child's enrollment in an institution.

SECTION 622m. 36.31 (title) of the statutes is amended to read:

36.31 (title) Coordination with other educational agencies; credit for military education.

SECTION 623m. 36.31 (4) of the statutes is created to read:

- 36.31 (4) Upon receiving from the federal department of defense a student's official joint services transcript or Community College of the Air Force transcript, the institution or college campus in which the student is enrolled shall do the following:
- (a) Accept all American Council on Education credit recommendations included in the official joint services transcript and award academic credit to the student in accordance with these recommendations.
- (b) Accept all credits included in the Community College of the Air Force transcript and award academic credit to the student accordingly.

SECTION 623p. 36.64 (5) (a) of the statutes is renumbered 36.64 (5).

SECTION 623q. 36.64 (5) (b) of the statutes is repealed.

SECTION 625. 36.65 (2) (dm) of the statutes is created to read:

36.65 (2) (dm) *Teaching hours*. Aggregate data on teaching hours reported under s. 36.115 (8) (a) 1.

SECTION 626. 36.65 (6) of the statutes is created to read:

- 36.65 (6) ACCOUNTABILITY DASHBOARD. (a) In this subsection, "accountability dashboard" means the accountability dashboard that the board publishes on the system's Internet site.
- (b) The board shall publish aggregate data on teaching hours reported under s. 36.115 (8) (a) 1. on the accountability dashboard. The board shall make accessible via links on the accountability dashboard the teaching hours reported by individual faculty and academic staff members under s. 36.115 (8) (a) 1.

SECTION 626m. 36.68 of the statutes is created to read:

36.68 The Tommy G. Thompson Center on Public Leadership. (1) Definitions. In this section:

- (a) "Board" means the public leadership board.
- (b) "Center" means the Tommy G. Thompson Center on Public Leadership.
- (2) Mission. There is created at the University of Wisconsin–Madison the Tommy G. Thompson Center on Public Leadership that has as its mission the facilitation of research, teaching, outreach, and needed policy reforms regarding effective public leadership that improves the practice of American government. The center shall endeavor to carry out its mission throughout all the universities of the system.
- (3) DIRECTOR. Upon the joint recommendation of the chancellor of the University of Wisconsin–Madison and the dean of the College of Letters and Science at the University of Wisconsin–Madison, the board shall appoint a director of the center for a 3–year term.
 - (4) POWERS. The center may do any of the following:
- (a) Make grants to support faculty research that objectively studies public leadership in American political and legal institutions, policy making, and policy implementation.
- (b) Conduct an annual competitive grant contest for encouraging faculty research described in par. (a) that has direct and urgent relevance for the state of Wisconsin.
- (c) Publicize the findings of the research that is supported by the center.
- (d) Foster public debate over critical issues regarding effective public leadership and facilitate dialogue between academics and policy makers on those issues.

- (e) Conduct and promote programs and events that bring timely political and policy issues to the attention of larger audiences.
- (f) Take any other action that is consistent with the mission of the center.
- (5) SPEAKING ENGAGEMENTS. In each fiscal year, not less than \$500,000 of the amount appropriated under s. 20.285 (1) (b) shall be allocated by the board for speaking engagements sponsored by the center at campuses other than the University of Wisconsin–Madison campus.
- (6) EXPENDITURES. The director of the center shall propose an annual budget itemizing expenditures of the moneys appropriated under s. 20.285 (1) (b), including expenditures for grants to proposed recipients, which shall be subject to the approval of the board. Upon approval by the board, the director may make substantive changes to the annual budget only with the subsequent approval of the board.

SECTION 628m. 38.04 (7m) of the statutes is amended to read:

38.04 (7m) FINANCIAL AIDS. By April 10, 1998, and annually thereafter February 10 of each year, the board shall develop and submit to the higher educational aids board for its review under s. 39.285 (1) a proposed formula for the awarding of grants under s. 39.435, except for grants awarded under s. 39.435 (2) or (5), for the upcoming academic next fiscal year to students enrolled in the technical colleges.

SECTION 629. 38.04 (11) (a) 2. of the statutes is amended to read:

38.04 (11) (a) 2. In consultation with the state superintendent of public instruction, the board shall establish, by rule, a uniform format for district boards to use in reporting the number of pupils attending district schools under ss. 38.12 (14) and 118.15 (1) (b), (cm), and (d) and 118.55 (7r) and in reporting pupil participation in technical preparation programs under s. 118.34, including the number of courses taken for advanced standing in the district's associate degree program and for technical college credit.

SECTION 629m. 38.04 (14) (b) of the statutes is amended to read:

38.04 (14) (b) The Subject to s. 38.12 (7) (b), the board may direct the district boards to establish written policies relating to any matter not enumerated under par. (a).

SECTION 630. 38.04 (21) (a) of the statutes is amended to read:

38.04 (21) (a) The number of pupils who attended district schools under ss. $\underline{38.12}$ (14) and $\underline{118.15}$ (1) (b), (cm), and (d) and $\underline{118.55}$ (7r) in the previous school year.

SECTION 631. 38.04 (21) (c) of the statutes is amended to read:

38.04(21) (c) The number of persons who applied for admission to a technical college in the previous school year, who previously earned technical college credit

under s. 118.55 (7r) 38.12 (14) and who applied for admission within one year of graduating from high school.

SECTION 632f. 38.12 (7) of the statutes is renumbered 38.12 (7) (a).

SECTION 632h. 38.12 (7) (b) of the statutes is created to read:

- 38.12 (7) (b) Upon receiving from the federal department of defense a student's official joint services transcript or Community College of the Air Force transcript, the technical college in which the student is enrolled shall do the following:
- 1. Accept all American Council on Education credit recommendations included in the official joint services transcript and award academic credit to the student in accordance with these recommendations.
- 2. Accept all credits included in the Community College of the Air Force transcript and award academic credit to the student accordingly.

SECTION 632m. 38.14 (15) of the statutes is created to read:

38.14 (15) SCHOLARSHIPS. The district board may use funds received from participation in an auction of digital broadcast spectrum administered by the federal communications commission to provide scholarships that cover the fees established under s. 38.24 (1m) for students who meet eligibility criteria established by the district board.

SECTION 632p. 38.16 (3) (a) 2w. of the statutes is amended to read:

38.16 (3) (a) 2w. "Revenue" means the sum of the tax levy and, property tax relief aid under sub. (4), and payments received under s. 79.096.

SECTION 633. 38.22 (1) (intro.) of the statutes is amended to read:

38.22 (1) (intro.) Except as provided in subs. (1m) and (1s) and s. 118.55 (7r) 38.12 (14), every person who is at least the age specified in s. 118.15 (1) (b) is eligible to attend a technical college if the person is:

SECTION 636. 38.24 (7) (a) 1m. a. of the statutes is amended to read:

38.24 (7) (a) 1m. a. A person who has served on active duty under honorable conditions in the U.S. armed forces, in forces incorporated as part of the U.S. armed forces, in the national guard, or in a reserve component of the U.S. armed forces; who was a resident of this state at the time of entry into that service or resided in this state for at least 5 consecutive years after the person attained the age of 18; and who, while a resident of this state, died on active duty, died as the result of a service—connected disability, or died in the line of duty while on active or inactive duty for training purposes.

SECTION 637. 38.24 (7) (a) 1m. b. of the statutes is amended to read:

38.24 (7) (a) 1m. b. A person who was a resident of this state at the time of entry into service described in subd. 1m. a. or resided in this state for at least 5 consecu-

tive years immediately preceding the beginning of any semester or session for which the person's spouse or child described in par. (b) 1., 2., or 3. registers at a technical college, and who the U.S. department of veteran affairs has awarded at least a 30 percent service—connected disability rating under 38 USC 1114 or 1134.

SECTION 637d. 38.24 (7) (b) (intro.) of the statutes is amended to read:

38.24 (7) (b) (intro.) Except as provided in par. pars. (bd) and (bg), the district board shall grant full remission of fees for 128 credits or 8 semesters, whichever is longer, less the number of credits or semesters for which the person received remission of fees from any other district board under this subsection and from the Board of Regents under s. 36.27 (3n) (b) and less the amount of any fees paid under 38 USC 3319, to any resident student who maintains a cumulative grade point average of at least 2.0 and is also any of the following:

SECTION 637h. 38.24 (7) (bd) of the statutes is created to read:

38.24 (7) (bd) If an eligible veteran was not a resident of this state at the time of entry into service described in par. (a) 1m. a., the district board may grant a remission of academic fees and segregated fees under this subsection only if the eligible veteran's spouse or child described in par. (b) 1., 2., or 3. has resided in this state for at least 5 consecutive years immediately preceding the spouse's or child's enrollment in a technical college.

SECTION 639. 38.28 (1) (a) of the statutes is renumbered 38.28 (1).

SECTION 640. 38.28 (1) (b) and (c) of the statutes are repealed.

SECTION 650i. 38.28 (1m) (a) 1. of the statutes is amended to read:

38.28 (1m) (a) 1. "District aidable cost" means the annual cost of operating a technical college district, including debt service charges for district bonds and promissory notes for building programs or capital equipment, but excluding all expenditures relating to auxiliary enterprises and community service programs, all expenditures funded by or reimbursed with federal revenues, all receipts under ss. 38.12 (9) and (14), 38.14 (3) and (9), and 118.15 (2) (a), and 118.55 (7r), all receipts from grants awarded under ss. 38.04 (8), (28), and (31), 38.14 (11), 38.26, 38.27, 38.31, 38.33, 38.38, and 38.42, all fees collected under s. 38.24, and driver education and chauffeur training aids.

SECTION 650j. 38.28 (2) (b) 2. of the statutes is amended to read:

38.28 (2) (b) 2. The most current equalized values certified by the department of revenue shall be used in aid determinations. Equalized determinations, except equalized values shall include the full value of property that is exempt under s. 70.11 (39) and (39m) as determined under s. 79.095 (3) for 2016.

SECTION 650k. 38.28 (2) (bm) 2. d. of the statutes is amended to read:

38.28 (2) (bm) 2. d. In fiscal year 2016–17 and each fiscal year thereafter, the percentage is 30 percent.

SECTION 663m. 38.50 (title) of the statutes is renumbered 440.52 (title).

SECTION 664. 38.50 (1) (intro.) of the statutes is renumbered 440.52 (1) (intro.).

SECTION 665m. 38.50(1)(a) of the statutes is renumbered 440.52(1)(a) and amended to read:

440.52 (1) (a) Notwithstanding s. 38.01 (2), "board" "Board" means the educational approval board.

SECTION 666. 38.50 (1) (b) of the statutes is renumbered 440.52 (1) (b).

SECTION 667. 38.50 (1) (c) of the statutes is renumbered 440.52 (1) (c).

SECTION 668. 38.50 (1) (d) of the statutes is renumbered 440.52 (1) (d).

SECTION 669m. 38.50 (1) (e) of the statutes is renumbered 440.52 (1) (e).

SECTION 670. 38.50 (1) (f) of the statutes is renumbered 440.52 (1) (f).

SECTION 671m. 38.50(1)(g) of the statutes is renumbered 440.52(1)(g).

SECTION 672m. 38.50 (2) of the statutes is renumbered 440.52 (2).

SECTION 673m. 38.50 (3) of the statutes is renumbered 440.52 (3).

SECTION 674m. 38.50 (5) of the statutes is renumbered 440.52 (5) and amended to read:

440.52 (5) EMPLOYEES, QUARTERS. The board shall employ a person to perform the duties of an executive secretary and any other persons under the classified service that may be necessary to carry out the board's responsibilities. The person performing the duties of the executive secretary shall be in charge of the administrative functions of the board. The board shall, to the maximum extent practicable, keep its office with the technical college system board department.

SECTION 680m. 38.50 (7) of the statutes is renumbered 440.52 (7).

SECTION 685m. 38.50 (8) of the statutes is renumbered 440.52 (8).

SECTION 686. 38.50 (10) (title) of the statutes is renumbered 440.52 (10) (title).

SECTION 687. 38.50 (10) (a) of the statutes is renumbered 440.52 (10) (a) and amended to read:

440.52 (10) (a) *Authority*. All proprietary schools shall be examined and approved by the board before operating in this state. Approval shall be granted to schools meeting the criteria established by the board for a period not to exceed one year. No school may advertise in this state unless approved by the board. All approved schools shall submit quarterly reports, including information on enrollment, number of teachers and their qual-

ifications, course offerings, number of graduates, number of graduates successfully employed, and such other information as the board considers necessary. If a school closure results in losses to students, parents, or sponsors, the board may authorize the full or partial payment of those losses from the appropriation under s. 20.292 (2) (gm) 20.165 (1) (it).

SECTION 688m. 38.50 (10) (b) of the statutes is renumbered 440.52 (10) (b).

SECTION 689m. 38.50 (10) (c) of the statutes is renumbered 440.52 (10) (c).

SECTION 694m. 38.50 (10) (cm) of the statutes is renumbered 440.52 (10) (cm).

SECTION 695. 38.50 (10) (d) of the statutes is renumbered 440.52 (10) (d).

SECTION 696. 38.50 (10) (e) of the statutes is renumbered 440.52 (10) (e).

SECTION 697. 38.50 (10) (f) of the statutes is renumbered 440.52 (10) (f).

SECTION 698. 38.50 (11) (title) of the statutes is renumbered 440.52 (11) (title).

SECTION 699. 38.50 (11) (a) of the statutes is renumbered 440.52 (11) (a).

SECTION 700m. 38.50 (11) (b) of the statutes is renumbered 440.52 (11) (b).

SECTION 701m. 38.50 (11) (c) of the statutes is renumbered 440.52 (11) (c).

SECTION 702. 38.50 (11) (d) of the statutes is renumbered 440.52 (11) (d) and amended to read:

440.52 (11) (d) The board or association shall preserve a student record that comes into the possession of the board or association under par. (b) 1. or 2. and shall keep the student record confidential as provided under 20 USC 1232g and 34 CFR part 99. A student record in the possession of the board is not open to public inspection or copying under s. 19.35 (1). Upon request of the person who is the subject of a student record or an authorized representative of that person, the board or association shall provide a copy of the student record to the requester. The board or association may charge a fee for providing a copy of a student record. The fee shall be based on the administrative cost of taking possession of, preserving, and providing the copy of the student record. All fees collected by the board under this paragraph shall be credited to the appropriation account under s. 20.292 (2) (i) 20.165 (1) (jv).

SECTION 703m. 38.50 (12) of the statutes is renumbered 440.52 (12).

SECTION 704d. 38.50 (13) (title) of the statutes is renumbered 440.52 (13) (title).

SECTION 704f. 38.50 (13) (a) of the statutes is renumbered 440.52 (13) (a).

SECTION 704h. 38.50 (13) (b) of the statutes is renumbered 440.52 (13) (b).

SECTION 704j. 38.50 (13) (c) of the statutes is renumbered 440.52 (13) (c).

SECTION 704m. 38.50 (13) (d) of the statutes is renumbered 440.52 (13) (d) and amended to read:

440.52 (13) (d) The board may charge a fee for evaluating an educational institution under par. (a) 2. e. in an amount that is sufficient to cover all costs that the board incurs in evaluating the institution. All fees collected by the board under this paragraph shall be credited to the appropriation account under s. 20.292 (2) (g) 20.165 (1) (ir).

SECTION 704n. 39.285 (1) of the statutes is renumbered 39.285 (1) (a) and amended to read:

39.285 (1) (a) By May 1, 1998, and annually thereafter March 1 of each year, the board shall approve, modify, or disapprove any proposed formula for the awarding of grants for the upcoming academic next fiscal year submitted under sub. (2) or (3) or s. 36.11 (6) (c) or 38.04 (7m).

SECTION 704ng. 39.285 (1) (b) of the statutes is created to read:

39.285 (1) (b) If the board determines during a fiscal year that any formula approved under par. (a) during the prior fiscal year needs to be modified during the fiscal year in order to expend the entire amount appropriated for grants to students under s. 39.30 or 39.435, except s. 39.435 (2) or (5), in that fiscal year, the board shall submit the modified formula to the joint committee on finance. If the cochairpersons of the committee do not notify the board that the committee has scheduled a meeting for the purpose of reviewing the modified formula within 14 working days after the date of the submittal, the modified formula may be implemented as proposed by the board. If, within 14 working days after the date of the submittal, the cochairpersons of the committee notify the board that the committee has scheduled a meeting for the purpose of reviewing the modified formula, the modified formula may be implemented only upon approval of the commit-

SECTION 704nm. 39.285 (2) of the statutes is amended to read:

39.285 (2) By April 10, 1998, and annually thereafter February 10 of each year, the Wisconsin Association of Independent Colleges and Universities shall develop and submit to the board for its review under sub. (1) a proposed formula for the awarding of grants under s. 39.30 for the upcoming academic next fiscal year to students enrolled at private institutions of higher education.

SECTION 704nr. 39.285 (3) of the statutes is amended to read:

39.285 (3) By April 10, 1998, and annually thereafter February 10 of each year, each tribally controlled college in this state is requested to develop and submit to the board for its review under sub. (1) a proposed formula for the awarding of grants under s. 39.435, except for grants awarded under s. 39.435 (2) or (5), for the upcoming academic next fiscal year to students enrolled at that tribally controlled college.

SECTION 704pg. 39.385 (3) (b) of the statutes is amended to read:

39.385 (3) (b) He or she graduated from a graduate medical education training program in this state with an emphasis on primary care medicine or psychiatry.

SECTION 704pm. 39.385 (4) (b) of the statutes is amended to read:

39.385 (4) (b) A physician or psychiatrist may apply for participation in the grant program by submitting an application to the board in a form prescribed by the board. A physician or psychiatrist may submit that application while the physician or psychiatrist is participating in a graduate medical education training program described in sub. (3) (b), but must submit that application before accepting employment or any other affiliation as a primary care physician or psychiatrist in an underserved area in this state. The application shall include such information as the board may require to establish that the physician or psychiatrist meets the eligibility requirements specified in sub. (3) for participation in the grant program and any other information the board may require by rule promulgated under sub. (5).

SECTION 704q. 39.397 of the statutes is created to read:

- 39.397 School leadership loan program. (1) The board shall establish a loan program for students selected by the board in a competitive application process who meet all of the following requirements:
- (a) Are enrolled in a school leadership program at an institution of the University of Wisconsin System that permits students to earn degrees in less time than a traditional school leadership program and that includes a required mentorship or apprenticeship component.
- (b) Are nominated by a superintendent of a school
- (2) (a) Loans under sub. (1) shall be made from the appropriation under s. 20.235 (1) (cu). Except as provided in par. (b), loans under sub. (1) shall be repaid at an annual interest rate of 5 percent.
- (b) The board shall forgive 25 percent of the loan and 25 percent of the interest on the loan for each school year the recipient satisfies all of the following:
- 1. The recipient is employed in a school leadership position in an elementary or secondary school in this state.
- 2. The recipient receives a rating of proficient or distinguished on the educator effectiveness system or the equivalent in a school that does not use the educator effectiveness system.
- (3) The board shall deposit in the general fund as general purpose revenue-earned all repayments of loans made under sub. (1) and the interest on the loans.

SECTION 704r. 39.435 (7) of the statutes is repealed. **SECTION 705.** 40.02 (8) (b) 3. of the statutes is created to read:

40.02 (8) (b) 3. For the purpose of determining a beneficiary of a deferred compensation plan under par. (a) 2., a surviving domestic partner.

SECTION 706. 40.02 (21d) (intro.) of the statutes is amended to read:

40.02 (21d) (intro.) "Domestic partnership" means a relationship between 2 individuals, who submitted an affidavit of domestic partnership to the department before the effective date of this subsection [LRB inserts date], that satisfies all of the following:

SECTION 707f. 40.03 (6) (m) of the statutes is created **Vetoed**

In Part

40.03 (6) (m) 1. In consultation with the division of personnel management in the department of administration, shall annually, by April 1, submit to the joint committee on finance any changes it proposes to make to the group health insurance programs under subch. IV, other than programs under ss. 40.51 (7) and 40.55, for the following year. If the cochairpersons of the joint committee on finance do not notify the group insurance board that the committee has scheduled a meeting for the purpose of reviewing the proposed changes within 21 working days after the date of the group insurance board's submittal of the proposed changes, the group insurance board may implement the proposed changes. If, within 21 working days after the date of the group insurance board's submittal of the proposed changes, the cochairpersons of the committee notify the group insurance board that the committee has scheduled a meeting for the purpose of reviewing the proposed changes, the group insurance board may not implement the proposed changes without the approval of the committee.

2. In consultation with the division of personnel management in the department of administration, submit to the joint committee on finance any changes it proposes to make to the group health insurance programs under subch. IV, other than programs under ss. 40.51 (7) and 40.55, for the following year that were not submitted to the joint committee on finance under subd. 1. if the proposed changes would have a financial impact or would affect covered benefits. If the cochairpersons of the joint committee on finance do not notify the group insurance board that the committee has scheduled a meeting for the purpose of reviewing the proposed changes within 21 working days after the date of the group insurance board's submittal of the proposed changes, the group insurance board may implement the proposed changes. If, within 21 working days after the date of the group insurance board's submittal of the proposed changes, the cochairpersons of the committee notify the group insurance board that the committee has scheduled a meeting for the purpose of reviewing the proposed changes, the group insurance board may not implement the proposed changes without the approval of the committee.

SECTION 708. 40.51 (2m) (a) of the statutes is amended to read:

40.51 (2m) (a) In addition to the restriction under par. (b), neither a domestic partner of an eligible employee nor a stepchild of a current domestic partnership may not become be covered under a group health insurance plan under this subchapter unless the eligible employee submits an affidavit, designed by the group insurance board, attesting that the eligible employee and his or her domestic partner satisfy the requirements for a domestic partnership under s. 40.02 (21d). The eligible employee shall submit this affidavit to his or her employer at the time the eligible employee first enrolls in a group health insurance plan under this subchapter or at the time the eligible employee requests a change in dependent status while the eligible employee is enrolled in a group health insurance plan under this subchapter. Upon the dissolution of a domestic partnership, the eligible employee shall submit in a timely manner to his or her employer an affidavit, designed by the group insurance board, attesting to the dissolution of the domestic partnership.

SECTION 709. 40.51 (2m) (b) of the statutes is amended to read:

40.51 (2m) (b) If an eligible employee is divorced or was a domestic partner in a dissolved domestic partnership, the eligible employee may not enroll a new spouse or domestic partner in a group health insurance plan under this subchapter until 6 months have elapsed since the date of the divorce or dissolved domestic partnership.

SECTION 709g. 40.51 (6) of the statutes is amended to read:

Vetoed In Part

40.51 (6) This state shall offer to all of its employees at least 2 insured or uninsured health care coverage plans providing substantially equivalent hospital and medical benefits, including a health maintenance organization or a preferred provider plan, if those health care plans are determined by the group insurance board to be available in the area of the place of employment and are approved by the group insurance board. The group insurance board shall place each of the plans into one of -3-5 tiers established in accordance with standards adopted by the group insurance board. The tiers shall be separated according to the employee's share of premium costs.

SECTION 710. 40.513 (3) (b) of the statutes is amended to read:

40.513 (3) (b) The employee's spouse or domestic partner is receiving health care coverage under s. 40.51 (6).

SECTION 711. 40.52(2) of the statutes is amended to read:

40.52 (2) Health insurance benefits under this subchapter shall be integrated, with exceptions determined appropriate by the group insurance board, with benefits under federal plans for hospital and health care for the aged and disabled. Exclusions and limitations with respect to benefits and different rates may be established

for persons eligible under federal plans for hospital and health care for the aged and disabled in recognition of the utilization by persons within the age limits eligible under the federal program. The plan may include special provisions for spouses, domestic partners, and other dependents covered under a plan established under this subchapter where one spouse or domestic partner is eligible under federal plans for hospital and health care for the aged but the others are not eligible because of age or other reasons. As part of the integration, the department may, out of premiums collected under s. 40.05 (4), pay premiums for the federal health insurance.

SECTION 712. 40.55(1) of the statutes is amended to read:

40.55 (1) Except as provided in sub. (5), the state shall offer, through the group insurance board, to eligible employees under s. 40.02 (25) (bm) and to state annuitants long—term care insurance policies which have been filed with the office of the commissioner of insurance and which have been approved for offering under contracts established by the group insurance board. The state shall also allow an eligible employee or a state annuitant to purchase those policies for his or her spouse, domestic partner, or parent.

SECTION 719d. 40.65 (7) (am) 1. of the statutes is amended to read:

40.65 (7) (am) 1. To the surviving spouse or domestic partner until the surviving spouse remarries or the surviving domestic partner enters into a new domestic partnership or marries, if the spouse was married to the participant on the date that the participant was disabled under sub. (4) or the domestic partner was in a domestic partnership with the participant on the date that the participant was disabled under sub. (4), 50 percent of the participant's monthly salary at the time of death, but reduced by any amount payable under sub. (5) (b) 1. to 6.

SECTION 719e. 40.65 (7) (am) 1g. of the statutes is created to read:

40.65 (7) (am) 1g. To the surviving spouse until the surviving spouse remarries, if the spouse was in a domestic partnership with the participant on the date that the participant was disabled under sub. (4) and the disability under sub. (4) occurred before January 1, 2018, 50 percent of the participant's monthly salary at the time of death, but reduced by any amount payable under sub. (5) (b) 1. to 6.

SECTION 719f. 40.65 (7) (am) 1m. of the statutes is created to read:

40.65 (7) (am) 1m. To the surviving domestic partner until the surviving domestic partner marries, if the domestic partner was in a domestic partnership with the participant on the date that the participant was disabled under sub. (4) and the disability under sub. (4) occurred before January 1, 2018, 50 percent of the participant's monthly salary at the time of death, but reduced by any amount payable under sub. (5) (b) 1. to 6.

SECTION 719g. 40.65 (7) (am) 3. of the statutes is amended to read:

40.65 (7) (am) 3. The total monthly amount paid under subds. 1. 1g., 1m., and 2. may not exceed 70 percent of the participant's monthly salary at the time of death reduced by any amounts under sub. (5) (b) 1. to 6. that relate to the participant's work record.

SECTION 719h. 40.65 (7) (ar) 1. a. of the statutes is amended to read:

40.65 (7) (ar) 1. a. To the surviving spouse or domestic partner until the surviving spouse or domestic partner remarries or enters into a new domestic partnership, if the surviving spouse was married to the participant on the date that the participant was disabled under sub. (4) or the domestic partner was in a domestic partnership with the participant on the date that the participant was disabled under sub. (4), 70 percent of the participant's monthly salary at the time of death, but reduced by any amount payable under sub. (5) (b) 1. to 6.

SECTION 719i. 40.65(7) (ar) 1. ag. of the statutes is created to read:

40.65 (7) (ar) 1. ag. To the surviving spouse until the surviving spouse remarries, if the spouse was in a domestic partnership with the participant on the date that the participant was disabled under sub. (4) and the disability under sub. (4) occurred before January 1, 2018, 70 percent of the participant's monthly salary at the time of death, but reduced by any amount payable under sub. (5) (b) 1. to 6.

SECTION 719j. 40.65 (7) (ar) 1. am. of the statutes is created to read:

40.65 (7) (ar) 1. am. To the surviving domestic partner until the surviving domestic partner marries, if the domestic partner was in a domestic partnership with the participant on the date that the participant was disabled under sub. (4) and the disability under sub. (4) occurred before January 1, 2018, 70 percent of the participant's monthly salary at the time of death, but reduced by any amount payable under sub. (5) (b) 1. to 6.

SECTION 725. 40.80 (2r) (a) 2. of the statutes is amended to read:

40.80 (2r) (a) 2. Assigns all or part of a participant's accumulated assets held in a deferred compensation plan under this subchapter to a spouse, former spouse, domestic partner, former domestic partner, child, or other dependent to satisfy a family support or marital property obligation.

SECTION 726. 41.41 (10) (b) of the statutes is amended to read:

41.41 (10) (b) Each year, the department shall ascertain from the clerk of each taxation district in which the reserve or any land acquired by the board is located the aggregate gross general property tax rate for the taxation district, exclusive of the rate that applies under s. 70.58 and without respect to the school levy tax credit under s. 79.10.

SECTION 727. 41.41 (10) (c) 1. of the statutes is amended to read:

41.41 (10) (c) 1. Except as provided in par. (d), on or before each January 31, the department shall pay to the treasurer of each taxation district specified in par. (b), with respect to all land in the Kickapoo valley reserve and all land acquired by the board on or before January 1 of the preceding year, an amount determined by multiplying the estimated value of the land equated to the average level of assessment in the taxation district by the aggregate gross general property tax rate, exclusive of the rate that applies under s. 70.58 and without respect to the school levy tax credit under s. 79.10, that would apply to the land in that taxation district for that year if it were taxable.

SECTION **727p.** 43.24 (6) of the statutes is repealed. SECTION **728.** 44.16 (title) of the statutes is amended to read:

44.16 (title) **Circus World Museum Foundation. SECTION 729.** 44.16 (1) of the statutes is amended to read:

44.16 (1) The Except as provided in sub. (3), the historical society may enter into a lease agreement with the Circus World Museum Foundation, Inc., for the purpose of operating Circus World Museum, located in Baraboo, Wisconsin. The lease agreement shall not include any provision for the payment of a percentage of gross admissions income at Circus World Museum to the historical society.

SECTION 730. 44.16 (3) of the statutes is created to read:

44.16 (3) If a lease agreement under sub. (1) is in effect on the effective date of this subsection [LRB inserts date], the lease agreement shall terminate on January 1, 2018, or on the date that any termination occurs as provided under the lease agreement, whichever is earlier.

SECTION 731. 44.20 (1) of the statutes is renumbered 44.20 (1) (a).

SECTION 732. 44.20 (1) (b) of the statutes is created to read:

44.20 (1) (b) The historical society shall operate and maintain Circus World Museum. If a lease agreement under s. 44.16 (1) is in effect on the effective date of this paragraph [LRB inserts date], this paragraph does not apply until the termination date of the lease agreement that is specified in s. 44.16 (3).

SECTION 732m. 45.03 (13) (p) (intro.) of the statutes is amended to read:

45.03 **(13)** (p) (intro.) Before June 30 September 15 of each even–numbered year, submit to the joint committee on finance a report describing the condition of the veterans trust fund. The report shall include information regarding all of the following:

SECTION 732n. 45.03 (13) (p) 1. of the statutes is amended to read:

45.03 (13) (p) 1. The projected revenues and expenditures of the veterans trust fund beginning with <u>for</u> the <u>current</u> fiscal year that starts immediately after the submittal of the report and each fiscal year of the following fiscal biennium.

SECTION 732p. 45.03 (13) (p) 1m. of the statutes is created to read:

45.03 (13) (p) 1m. The actual revenues and expenditures of the veterans trust fund for the previous fiscal year.

SECTION 733. 45.20 (1) (d) of the statutes is amended to read:

45.20 (1) (d) "Tuition," when referring to the University of Wisconsin System, means academic fees and segregated fees; when referring to the technical colleges, means "program fees" and "additional fees" as described in s. 38.24 (1m) and (1s); and when referring to a high school, a school that is approved under s. 45.03 (11), or a proprietary school that is approved under s. 38.50 440.52, means the charge for the courses for which a person is enrolled.

SECTION 734. 45.20 (2) (a) 1. of the statutes is amended to read:

45.20 (2) (a) 1. The department shall administer a tuition reimbursement program for eligible veterans enrolling as undergraduates in any institution of higher education in this state, enrolling in a school that is approved under s. 45.03 (11), enrolling in a proprietary school that is approved under s. 38.50 440.52, enrolling in a public or private high school, enrolling in a tribal school, as defined in s. 115.001 (15m), in any grade from 9 to 12, or receiving a waiver of nonresident tuition under s. 39.47.

SECTION 735. 45.20 (2) (a) 2. (intro.) of the statutes is amended to read:

45.20 (2) (a) 2. (intro.) A veteran who is a resident of this state and otherwise qualified to receive benefits under this subsection may receive the benefits under this subsection upon the completion of any correspondence courses or part–time classroom study from an institution of higher education located outside this state, from a school that is approved under s. 45.03 (11), or from a proprietary school that is approved under s. 38.50 440.52, if any of the following applies:

SECTION 736. 45.20 (2) (c) 1. of the statutes is amended to read:

45.20 (2) (c) 1. A veteran who meets the eligibility requirements under par. (b) 1. may be reimbursed upon satisfactory completion of an undergraduate semester in any institution of higher education in this state, or upon satisfactory completion of a course at any school that is approved under s. 45.03 (11), any proprietary school that is approved under s. 38.50 440.52, any public or private high school, any tribal school, as defined in s. 115.001 (15m), that operates any grade from 9 to 12, or any institution from which the veteran receives a waiver of non-

resident tuition under s. 39.47. Except as provided in par. (e), the amount of reimbursement may not exceed the total cost of the veteran's tuition minus any grants or scholarships that the veteran receives specifically for the payment of the tuition, or, if the tuition is for an undergraduate semester in any institution of higher education, the standard cost of tuition for a state resident for an equivalent undergraduate semester at the University of Wisconsin–Madison, whichever is less.

SECTION 737. 45.20(2)(d) 1. (intro.) of the statutes is amended to read:

45.20 (2) (d) 1. (intro.) Subject to subd. 1m., a veteran's eligibility for reimbursement under this subsection at any institution of higher education in this state, at a school that is approved under s. 45.03 (11), at a proprietary school that is approved under s. 38.50 440.52, at a public or private high school, at a tribal school, as defined in s. 115.001 (15m), that operates any grade from 9 to 12, or at an institution where he or she is receiving a waiver of nonresident tuition under s. 39.47 is limited to the following:

SECTION 738h. 45.21 (2) (a) of the statutes is amended to read:

45.21 (2) (a) The veteran is enrolled in a training course in a technical college under ch. 38 or in a proprietary school in the state approved by the educational approval board under s. 38.50 440.52, other than a proprietary school offering a 4–year degree or 4–year program, or is engaged in a structured on–the–job training program that meets program requirements promulgated by the department by rule.

SECTION 738j. 45.21 (2) (a) of the statutes, as affected by 2017 Wisconsin Act (this act), is amended to read:

45.21 (2) (a) The veteran is enrolled in a training course in a technical college under ch. 38 or in a proprietary school in the state approved by the educational approval board department of safety and professional services under s. 440.52, other than a proprietary school offering a 4-year degree or 4-year program, or is engaged in a structured on—the—job training program that meets program requirements promulgated by the department by rule.

SECTION 739g. 45.41 (2) (intro.) of the statutes is renumbered 45.41 (2) (am) and amended to read:

45.41 (2) (am) Upon application the department may make a payment to any state veterans organization that establishes that it, or its national organization, or both, has maintained a full-time service office at the regional office for 5 consecutive years out of the 10-year period immediately preceding the application. Any payment shall be as follows, calculated based on the total amount of all salaries and travel expenses under sub. (3) paid during the previous fiscal year by the state veterans organization to employees engaged in veterans claims service and stationed at the regional office; The department shall

pay an amount equal to 50 percent of all salaries and travel expenses under sub. (3) or \$100,000, whichever is less, to a recipient under this paragraph.

SECTION 739h. 45.41 (2) (a) of the statutes is repealed.

SECTION 739i. 45.41 (2) (d) of the statutes is repealed.

SECTION 739j. 45.41 (2) (e) of the statutes is amended to read:

45.41 (2) (e) An organization that receives a payment under par. (a) or (d) (am) shall maintain records as required by the department concerning the organization's expenditure of the payment. That organization shall give the department access to those records upon request of the department, and the department may audit those records.

SECTION 739k. 45.41 (3m) of the statutes is amended to read:

45.41 (3m) If the total amount of payments committed to be paid under sub. (2) (a) and (d) (am) exceeds the amount available for the payments from the appropriation under s. 20.485 (2) (vw), the department shall prorate the reimbursement payments among the state veterans organizations receiving the payments.

SECTION 739m. 45.41 (4) (a) of the statutes is amended to read:

45.41 (4) (a) From the appropriation under s. 20.485 (2) (s), the department shall annually provide a payment of \$120,000 \$200,000 to the Wisconsin department of the Disabled American Veterans for the provision of transportation services to veterans.

SECTION 739n. 45.41 (5) of the statutes is amended to read:

45.41 (5) From the appropriation under s. 20.485 (2) (vw) (vs), the department may annually grant up to \$50,000 \$75,000 to the Wisconsin department of the American Legion for the operation of Camp American Legion.

SECTION 739pg. 45.45 of the statutes is repealed. **SECTION 739pm.** 45.46 of the statutes is amended to read:

45.46 Grants to nonprofit organizations that serve veterans and their families. From the appropriation under s. 20.485 (2) (tf) (th), the department may make grants of up to -a total of \$250,000 annually to nonprofit organizations, as defined in s. 108.02 (19), and no more than \$25,000 to each grant recipient, to provide financial assistance or other services to veterans and their dependents.

SECTION 739pv. 45.50 (2m) (d) of the statutes is

SECTION 739py. 45.50 (4) (b) of the statutes is amended to read:

45.50 (4) (b) The department may accept gifts, bequests, grants, or donations of money or of property from private sources to be administered by the department for the purposes of veterans homes. All moneys received shall be paid into the general fund and appropriated as provided in s. 20.485 (1) (h), except that gifts or grants received specifically for the purposes of the geriatric program at veterans homes are appropriated as provided in s. 20.485 (1) (hm). The department may not apply to the gifts and bequests fund interest on certificate of savings deposits for those members who do not receive maximum monthly retained income. The department shall establish for those persons upon their request individual accounts with savings and interest applied as the member requests.

SECTION 739qg. 45.57 (1) of the statutes is renum- **Vetoed** bered 45.57 (1) (a) and amended to read:

In Part

45.57 (1) (a) The Subject to par. (b), the department may transfer all or part of the unencumbered balance of any of the appropriations under s. 20.485 (1) (g), (gd), (gk), or (i) to the veterans trust fund or to the veterans mortgage loan repayment fund.

SECTION 739qm. 45.57 (1) (b) of the statutes is cre-

45.57 (1) (b) Before transferring all or part of an appropriation balance under par. (a), the department shall notify the joint committee on finance in writing of the proposed balance transfer. If the cochairpersons of the committee do not notify the department within 14 working days after the date of the department's notification that the committee has scheduled a meeting for the purpose of reviewing the proposed balance transfer, the balance transfer may be made as proposed by the department. If, within 14 working days after the date of the department's notification, the cochairpersons of the committee notify the department that the committee has scheduled a meeting for the purpose of reviewing the proposed balance transfer, the balance may be made only upon approval of the committee.

SECTION 739r. 45.58 of the statutes is amended to read:

45.58 Grants to local governments. From the appropriation under s. 20.485 (1) (kj), the department may make up to \$300,000 each fiscal biennium in grants to cities, villages, and towns that provide services to veterans homes and other facilities for veterans. A city, village, or town may not expend grant moneys it receives under this section for any purpose other than providing fire and emergency medical services to veterans homes and other facilities for veterans.

SECTION 739s. 45.70 (1m) of the statutes is repealed. SECTION 740. 45.82 (2) of the statutes is amended to

45.82 (2) The department of veterans affairs shall award a grant annually, on a reimbursable basis as specified in this subsection, to a county that meets the standards developed under this section if the county executive, administrator, or administrative coordinator certifies to the department that it employs a county veterans service officer who, if chosen after April 15, 2015, is chosen from a list of candidates who have taken a civil service examination for the position of county veterans service officer developed and administered by the bureau of merit recruitment and selection in the department of administration, or is appointed under a civil service competitive examination procedure under s. 59.52 (8) or ch. 63. The department of veterans affairs shall twice yearly reimburse grant recipients for documented expenses under sub. (5), subject to the following annual reimbursement limits: grant shall be \$8,500 for a county with a population of less than 20,000, \$10,000 for a county with a population of 20,000 to 45,499, \$11,500 for a county with a population of 45,500 to 74,999, and \$13,000 for a county with a population of 75,000 or more. The department of veterans affairs shall use the most recent Wisconsin official population estimates prepared by the demographic services center when making grants under this subsection.

SECTION 741. 45.82 (4) of the statutes is amended to read:

45.82 (4) The department shall provide grants on a reimbursable basis as specified in this subsection to the governing bodies of federally recognized American Indian tribes and bands from the appropriation under s. 20.485 (2) (km) or (vw) (vu) if that governing body enters into an agreement with the department regarding the creation, goals, and objectives of a tribal veterans service officer, appoints a veteran to act as a tribal veterans service officer, and gives that veteran duties similar to the duties described in s. 45.80 (5), except that the veteran shall report to the governing body of the tribe or band. The department shall twice yearly reimburse grant recipients for documented expenses under sub. (5), may make annual grants in an amount not to exceed \$15,000 per grant under this subsection and shall promulgate rules to implement this subsection.

SECTION 742m. 45.82 (5) of the statutes is repealed. **SECTION 744.** 45.82 (6) of the statutes is repealed.

SECTION 744av. 46.03 (1m) of the statutes is created to read:

Vetoed In Part

46.03 (1m) Institute appropriation surplus. After June 30 of each even—numbered fiscal year, determine the unencumbered amount remaining in the appropriation account under s. 20.435 (2) (gk) and provide this information to county and tribal human services departments. If the unencumbered amount in the appropriation account under s. 20.435 (2) (gk) on June 30 of an even—numbered fiscal year exceeds 17 percent over the amount of expenditures made during the even—numbered fiscal year, the department shall consult with county and tribal human services departments to develop a proposal for the use of that excess amount. The department shall submit the proposal for use of the excess amount, if an excess amount exists, in its next biennial budget request.

SECTION 744b. 46.036 (4) (c) of the statutes is amended to read:

46.036 (4) (c) Unless waived by the department, biennially, or annually if required under federal law, provide the purchaser with a certified financial and compliance audit report if the care and services purchased exceed \$25,000 \$100,000. The audit shall follow standards that the department prescribes.

SECTION 744c. 46.036 (5m) (b) 1. of the statutes is amended to read:

46.036 (5m) (b) 1. Subject to subd. 2. and pars. (e) and (em), if If revenue under a contract for the provision of a rate-based service exceeds allowable costs incurred in the contract period, the contract shall allow the provider may to retain from the surplus generated by that rate-based service up to 5 percent of the revenue received under the contract. A provider that retains a surplus under this subdivision shall use that retained surplus to cover a deficit between revenue and allowable costs incurred in any preceding or future contract period for the same rate-based service that generated the surplus or to address the programmatic needs of clients served by the same rate-based service that generated the surplus unless a uniform rate is established by rule under subd. 4., in which case the contract shall allow the provider to retain the uniform percentage rate established by the rule. The retained surplus is the property of the provider.

SECTION 744d. 46.036 (5m) (b) 2. of the statutes is repealed.

SECTION 744e. 46.036 (5m) (b) 3. of the statutes is created to read:

46.036 (5m) (b) 3. If on December 31 of any year the provider's accumulated surplus from all contract periods ending during that year for a rate-based service exceeds the allowable retention rate under subd. 1., the provider shall provide written notice of that excess to all purchasers of the rate-based service. Upon the written request of such a purchaser received no later than 6 months after the date of the notice, the provider shall refund the purchaser's proportional share of that excess. If the department determines based on an audit or fiscal review that the amount of the excess identified by the provider was incorrect, the department may seek to recover funds after the 6-month period has expired. The department shall commence any audit or fiscal review under this subdivision within 6 years after the end of the contract period.

SECTION 744f. 46.036 (5m) (b) 4. of the statutes is created to read:

46.036 (5m) (b) 4. The department, in consultation with the department of children and families and the department of corrections, shall promulgate rules to implement this subsection including all of the following:

a. Requiring that contracts for rate-based services under this subsection allow a provider to retain from any

surplus revenue up to 5 percent of the total revenue received under the contract, or a different percentage rate determined by the department. The percentage rate established under this subd. 4. a. shall apply uniformly to all rate-based service contracts under this subsection.

b. Establishing a procedure for reviewing rate-based service contracts to determine whether a contract complies with the provisions of this subsection.

SECTION 744g. 46.036 (5m) (e) of the statutes is amended to read:

46.036 (5m) (e) Notwithstanding par. (b) 1. and 2., the department or a county department under s. 46.215, 46.22, 46.23, 51.42, or 51.437 that purchases care and services from an inpatient alcohol and other drug abuse treatment program that is not affiliated with a hospital and that is licensed as a community-based residential facility, may allocate to the program an amount that is equal to the amount of revenues received by the program that are in excess of the allowable costs incurred in the period of a contract between the program and the department or the county department for purchase of care and services under this section. The department or the county department may make the allocation under this paragraph only if the funds so allocated do not reduce any amount of unencumbered state aid to the department or the county department that otherwise would lapse to the general fund.

SECTION 744h. 46.036 (5m) (em) of the statutes is amended to read:

46.036 (5m) (em) Notwithstanding pars. (b) 1. and 2. and (e), a county department under s. 46.215, 51.42, or 51.437 providing client services in a county having a population of 500,000 or more or a nonstock, nonprofit corporation providing client services in such a county may not retain a surplus under par. (b) 1., accumulate funds under par. (b) 2., or allocate an amount under par. (e) from revenues that are used to meet the maintenanceof-effort requirement under the federal temporary assistance for needy families program under 42 USC 601 to 619.

SECTION 745. 46.057 (2) of the statutes is amended to read:

46.057 (2) From the appropriation account under s. 20.410 (3) (ba), the department of corrections shall transfer to the appropriation account under s. 20.435 (2) (kx) \$1,365,500 in each fiscal year and, from the appropriation account under s. 20.410 (3) (hm), the department of corrections shall transfer to the appropriation account under s. 20.435 (2) (kx) \$2,929,200 \$2,869,200 in fiscal year 2015-16 2017-18 and \$2,997,600 <u>\$2,932,600</u> in fiscal year 2016-17 2018-19, for services for juveniles placed at the Mendota juvenile treatment center. The department of health services may charge the department of corrections not more than the actual cost of providing those services.

SECTION 746. 46.269 of the statutes is created to read:

46.269 Determining financial eligibility for longterm care programs. To the extent approved by the federal government, the department or its designee shall exclude any assets accumulated in a person's independence account, as defined in s. 49.472 (1) (c), and any income or assets from retirement benefits earned or accumulated from income or employer contributions while employed and receiving state-funded benefits under s. 46.27 or medical assistance under s. 49.472 in determining financial eligibility and cost-sharing requirements, if any, for a long-term care program under s. 46.27, 46.275, or 46.277, for the family care program that provides the benefit defined in s. 46.2805 (4), for the Family Care Partnership program, or for the self-directed services option, as defined in s. 46.2897 (1).

SECTION 747. 46.283 (5) of the statutes is amended to read:

46.283 (5) Funding. From the appropriation accounts under s. 20.435 (1) (n), (4) (b), (bd), (bm), (gm), (pa), and (w), and (7) (b) and (md), the department may contract with organizations that meet standards under sub. (3) for performance of the duties under sub. (4) and shall distribute funds for services provided by resource

SECTION 747w. 46.2899 (2), (3) and (4) of the **Vetoed** statutes are amended to read:

In Part

46.2899 (2) WAIVER PROGRAM. The department shall request a waiver, or a modification of a waiver, from the federal centers for medicare and medicaid services in order to receive the federal medical assistance percentage for home-based and community-based services provided to individuals who are developmentally disabled and who received post-secondary education on the grounds of health care institutions. If the waiver or modification of the waiver is approved, the department shall operate a waiver program to provide those services to no more than 100 individuals per month per year.

- (3) ELIGIBILITY. The department shall consider as eligible for the waiver program described under sub. (2) only individuals who are receiving post-secondary education in a setting that is distinguishable from the health care institution. The department shall set the financial eligibility requirements and functional eligibility requirements for the waiver program described under sub. (2) the same as the financial eligibility requirements and functional eligibility requirements for the selfdirected services option except for the requirement to be an individual who is developmentally disabled and who is receiving post-secondary education on the grounds of a <u>health care</u> institution.
- (4) SERVICES AND BENEFITS. The department shall provide the same services under the waiver program described in sub. (2) as it provides under the self-directed services option. The department shall determine the funding amount for a waiver program participant under

Vetoed In Part this section <u>based on what the individual would receive</u> <u>if enrolled in the self-directed services option</u>.

SECTION 748. 46.29 (1) (intro.) of the statutes is amended to read:

46.29 (1) (intro.) From the appropriation account under s. 20.435 (4) (1) (a), the department shall distribute at least \$16,100 in each fiscal year for operation of the council on physical disabilities. The council on physical disabilities shall do all of the following:

SECTION 749. 46.295 (1) of the statutes is amended to read:

46.295 (1) The department may, on the request of any hearing–impaired person, city, village, town, or county or private agency, provide funds from the appropriation accounts under s. 20.435 (4) (1) (da) and (hs) and (7) (d) to reimburse interpreters for hearing–impaired persons for the provision of interpreter services.

SECTION 750. 46.297 (1) of the statutes is amended to read:

46.297 (1) ASSISTANCE. From the appropriation under s. 20.435 (7) (d) (1) (da), the department shall, subject to the availability of funds, provide assistance to hearing–impaired persons to secure telecommunication devices capable of serving their needs. Except in extraordinary circumstances, the department shall purchase or provide funds for the purchase of telecommunication devices.

SECTION 751. 46.48 (1) of the statutes is amended to read:

46.48 (1) GENERAL. From the appropriation accounts under s. 20.435 (1) (b), (5) (bc), and (7) (bc), the department shall award grants for community programs as provided in this section.

SECTION 752. 46.48 (32) of the statutes is amended to read:

46.48 (32) PEER-RUN RESPITE CENTER CONTRACTS. The department shall contract with a peer-run organization to establish peer-run respite centers for individuals experiencing mental health conditions or substance abuse. Notwithstanding sub. (1), the department may make payments to an organization that establishes peer-run respite centers that provide services to veterans from the appropriation under s. 20.435 (5) (kp).

Vetoed In Part SECTION 752b. 46.48 (32) of the statutes, as affected by 2017 Wisconsin Act (this act), is amended to read: 46.48 (32) PEER-RUN RESPITE CENTER CONTRACTS. The department shall contract with a peer-run organization to establish peer-run respite centers for individuals experiencing mental health conditions or substance abuse. Notwithstanding sub. (1), the department may make payments to an organization that establishes peer-run respite centers that provide services to veterans from the appropriation under s. 20.435 (5) (kp).

SECTION 753. 46.80 (2m) (b) of the statutes is amended to read:

46.80 (**2m**) (b) May operate the foster grandparent project specified under 42 USC 5011 (a). If the department operates that project, the department shall distribute funds from the appropriation under s. 20.435 (7) (1) (dh) to supplement any federal foster grandparent project funds received under 42 USC 5011 (a).

SECTION 754. 46.80 (5) (a) of the statutes is amended to read:

46.80 (**5**) (a) From the appropriation under s. 20.435 (7) (1) (dh), the department shall provide a state supplement to the federal congregate nutrition projects under 42 USC 3030e, in effect on April 30, 1980, which will promote expansion of projects throughout the state and, from the appropriation under s. $20.435 \frac{(7)}{(1)}$ (kn), the department shall provide a state supplement to the federal congregate nutrition projects of \$450,000 for home-delivered meals and \$50,000 for congregate meals. Except as provided in par. (b), the department shall allocate these funds based on the formulas developed by the department under sub. (2m) (a) 2. A county that receives federal funds for congregate nutrition projects on or after July 1, 1977, may not receive under this paragraph an amount that is less than the 1976-77 allocation as a result of the program expansion. This paragraph does not require that federal limitations on the use of federal congregate nutrition funds for home delivered meals apply to the state supplement.

SECTION 755. 46.81 (2) of the statutes is amended to read:

46.81 (2) From the appropriation account under s. 20.435 (7) (1) (dh), the department shall allocate \$2,298,400 in each fiscal year to aging units to provide benefit specialist services for older individuals. The department shall ensure that each aging unit receives funds and shall take into account the proportion of the state's population of low–income older individuals who reside in a county.

SECTION 756. 46.81 (5) of the statutes is amended to read:

46.81 (5) From the appropriation under s. 20.435 (7) (1) (dh) the department shall allocate \$182,500 in each fiscal year to area agencies on aging. Each area agency on aging shall use the funds for training, supervision and legal back—up services for benefit specialists within its area.

SECTION 757. 46.85 (3m) (a) of the statutes is amended to read:

46.85 (**3m**) (a) From the appropriation under s. 20.435 (7) (1) (dh), the department shall provide a state supplement to federally funded Senior Companion Program and Retired Senior Volunteer Program units that were in operation on December 1, 1988, and administered by qualified public and non–profit private agencies.

SECTION 758. 46.85 (3m) (b) (intro.) of the statutes is amended to read:

46.85 (**3m**) (b) (intro.) From the appropriation under s. 20.435 (7) (1) (dh), the department shall allocate funds, based on the percentage of the state's population of low-income persons over age 60 who reside in each county or are members of an American Indian tribe, and distribute the funds to counties and federally recognized tribal governing bodies to supplement any of the following:

SECTION 759. 46.856 (2) (intro.) of the statutes is amended to read:

46.856 (2) (intro.) From the appropriation under s. 20.435 (7) (1) (bg), the department shall award a grant to at least one public agency or private nonprofit organization to do all of the following:

SECTION 760. 46.90 (5m) (a) of the statutes is amended to read:

46.90 (5m) (a) Upon responding to a report, the elder-adult-at-risk agency or the investigative agency shall determine whether the elder adult at risk or any other individual involved in the alleged abuse, financial exploitation, neglect, or self-neglect is in need of services under this chapter or ch. 47, 49, 51, 54, or 55. From the appropriation under s. 20.435 (7) (1) (dh), the department shall allocate to selected counties not less than \$25,000 in each fiscal year, and within the limits of these funds and of available state and federal funds and of county funds appropriated to match the state and federal funds, the elder-adult-at-risk agency shall provide the necessary direct services to the elder adult at risk or other individual or arrange for the provision of the direct services with other agencies or individuals. Those direct services provided shall be rendered under the least restrictive conditions necessary to achieve their objec-

SECTION 762. 46.96 (2) of the statutes is amended to read:

46.96 (2) The department shall make grants from the appropriations under s. 20.435 (7) (e), (1) (cx) and (kc), and (7) (na) to independent living centers for nonresidential services to severely disabled individuals.

SECTION 763. 46.96 (2d) of the statutes is amended to read:

46.96 (**2d**) The department shall make grants from the appropriations under s. 20.435 (1) (cx) and (7) (c) and (na) for the purposes for which the federal moneys are received, including for independent living services.

SECTION 764. 46.977 (2) (a) of the statutes is amended to read:

46.977 (2) (a) From the appropriation under s. 20.435 (7) (1) (cg), the department may under this section, based on the criteria under par. (c), award grants to applying organizations for the purpose of training and assisting guardians for individuals found incompetent under ch. 54. No grant may be paid unless the awardee provides matching funds equal to 10 percent of the amount of the award.

SECTION 765. 46.986 (2) (a) (intro.) of the statutes is amended to read:

46.986 **(2)** (a) (intro.) From the appropriation account under s. 20.435 (7) (1) (br), the department shall contract for the administration of life–span respite care projects with an organization to which all of the following apply:

SECTION 766. 46.995 (2g) of the statutes is created to read:

46.995 (2g) (a) The department may require a county to maintain a specified level of contribution for the disabled children's long—term support program. The department shall determine the amount of contribution that a county is required to maintain based on the historical county expenditures for the disabled children's long—term support program.

(b) Beginning in the 2017–19 fiscal biennium and thereafter, counties shall cooperate with the department to determine an equitable funding methodology and county contribution mechanism for contribution for the disabled children's long–term support program under par. (a) and to ensure that county contributions determined by the department are expended for the disabled children's long–term support program in the counties.

SECTION 767. 46.995 (2r) of the statutes is created to read:

46.995 (2r) The department may contract with a county or a group of counties to deliver disabled children's long-term support program services.

SECTION 768. 48.13 (3) of the statutes is amended to read:

48.13 (3) Who has been the victim of abuse, as defined in s. 48.02 (1) (a), (b), (c), (d), (e), (f), or (b) to (g), including injury that is self–inflicted or inflicted by another;

SECTION 769. 48.13 (3m) of the statutes is amended to read:

48.13 (3m) Who is at substantial risk of becoming the victim of abuse, as defined in s. 48.02 (1) (a), (b), (c), (d), (e), (f), or (b) to (g), including injury that is self–inflicted or inflicted by another, based on reliable and credible information that another child in the home has been the victim of such abuse;

SECTION 772. 48.563 (2) of the statutes is amended to read:

48.563 (2) COUNTY ALLOCATION. For children and family services under s. 48.569 (1) (d), the department shall distribute not more than \$68,264,800 \$70,211,100 in fiscal year 2015-16 2017-18 and \$68,327,900 \$74,308,000 in fiscal year 2016-17 2018-19.

SECTION 773. 48.57 (3m) (am) (intro.) of the statutes is amended to read:

48.57 (**3m**) (am) (intro.) From the appropriations under s. 20.437 (2) (dz), (md), (me), and (s), the department shall reimburse counties having populations of less

than 750,000 for payments made under this subsection and shall make payments under this subsection in a county having a population of 750,000 or more. Subject to par. (ap), a county department and, in a county having a population of 750,000 or more, the department shall make payments in the amount of \$226 \$238 per month beginning on January 1, 2014 2018, and \$232 \$244 per month beginning on January 1, 2015 2019, to a kinship care relative who is providing care and maintenance for a child if all of the following conditions are met:

SECTION 774. 48.57 (3n) (am) (intro.) of the statutes is amended to read:

48.57 (**3n**) (am) (intro.) From the appropriations under s. 20.437 (2) (dz), (md), (me), and (s), the department shall reimburse counties having populations of less than 750,000 for payments made under this subsection and shall make payments under this subsection in a county having a population of 750,000 or more. Subject to par. (ap), a county department and, in a county having a population of 750,000 or more, the department shall make monthly payments for each child in the amount of \$226 \$238 per month beginning on January 1, 2014 2018, and \$232 \$244 per month beginning on January 1, 2015 2019, to a long–term kinship care relative who is providing care and maintenance for that child if all of the following conditions are met:

SECTION 775. 48.62 (4) of the statutes is amended to read:

48.62 (4) Monthly payments in foster care shall be provided according to the rates specified in this subsection. Beginning on January 1, 2014 2018, the rates are \$226 \$238 for care and maintenance provided for a child of any age by a foster home that is certified to provide level one care, as defined in the rules promulgated under sub. (8) (a) and, for care and maintenance provided by a foster home that is certified to provide care at a level of care that is higher than level one care, \$375 \$394 for a child under 5 years of age; \$410 \$431 for a child 5 to 11 years of age; \$466 \$490 for a child 12 to 14 years of age; and \$487 \$511 for a child 15 years of age or over. Beginning on January 1, 2015 2019, the rates are \$232 \$244 for care and maintenance provided for a child of any age by a foster home that is certified to provide level one care, as defined in the rules promulgated under sub. (8) (a) and, for care and maintenance provided by a foster home that is certified to provide care at a level of care that is higher than level one care, \$384 \$404 for a child under 5 years of age; \$420 \$442 for a child 5 to 11 years of age; \$478 \$502 for a child 12 to 14 years of age; and \$499 \$524 for a child 15 years of age or over. In addition to these grants for basic maintenance, the department, county department, or licensed child welfare agency shall make supplemental payments for foster care to a foster home that is receiving an age-related rate under this subsection that are commensurate with the level of care that the foster home is certified to provide and the needs of the child

who is placed in the foster home according to the rules promulgated by the department under sub. (8) (c).

SECTION 776. 48.65 (1) of the statutes is amended to read:

48.65 (1) No person may for compensation provide care and supervision for 4 or more children under the age of 7 for less than 24 hours a day unless that person obtains a license to operate a child care center from the department. To obtain a license under this subsection to operate a child care center, a person must meet the minimum requirements for a license established by the department under s. 48.67, meet the requirements specified in s. 48.685 48.686, and pay the license fee under sub. (3). A license issued under this subsection is valid until revoked or suspended, but shall be reviewed every 2 years as provided in s. 48.66 (5).

SECTION 777. 48.651 (1) (intro.) of the statutes is amended to read:

48.651 (1) (intro.) Except as provided in s. 49.155 (4) (c), no person, other than a child care center licensed under s. 48.65 or established or contracted for under s. 120.13 (14), may receive payment for providing child care services for an individual who is determined eligible for a child care subsidy under s. 49.155 unless the person is certified, according to the standards adopted by the department under s. 49.155 sub. (1d), by the department in a county having a population of 750,000 or more, a county department, or an agency with which the department contracts under sub. (2). To be certified under this section, a person must meet the minimum requirements for certification established by the department under s. 49.155 sub. (1d), meet the requirements specified in s. 48.685 48.686, and pay the fee specified in sub. (2). The department in a county having a population of 750,000 or more, a county department, or an agency contracted with under sub. (2) shall certify the following categories of child care providers:

SECTION 778. 48.651 (1) (a) of the statutes is amended to read:

48.651 (1) (a) Level I certified family child care providers, as established by the department under s. 49.155 <u>sub.</u> (1d). No provider may be certified under this paragraph if the provider is a relative of all of the children for whom the provider provides care.

SECTION 779. 48.651 (1) (b) of the statutes is amended to read:

48.651 (1) (b) Level II certified family child care providers, as established by the department under s. 49.155 sub. (1d).

SECTION 780. 48.651 (1d) (b) of the statutes is created to read:

48.651 (1d) (b) 1. A level I certified family child care provider shall successfully complete department—approved preservice health and safety training in the topics specified in subd. 1. a. to j. by no later than the date of certification. A level II certified family child care provider

or an employee or volunteer of a level I or level II certified family child care provider who is not the primary provider of care and supervision for children shall successfully complete department—approved preservice health and safety training in the topics specified in subd. 1. a. to j. by no later than the end of the orientation period available under 42 USC 9858c (c) (2) (I) (i) (XI). The health and safety training required under this subdivision shall include training in all of the following topics:

- a. The prevention and control of infectious diseases, including by means of immunizations.
- b. The prevention of sudden infant death syndrome and use of safe sleeping practices.
- c. The administration of medication, consistent with parental consent.
- d. The prevention of and response to emergencies due to allergic reactions to food or other allergens.
- e. Building and physical premises safety, including identification of and protection from electrical hazards, bodies of water, vehicular traffic, and other hazards that can cause bodily injury.
- f. The prevention of shaken baby syndrome and abusive head trauma.
- g. Emergency preparedness and response planning for emergencies resulting from natural disaster or human-caused events.
- h. The handling and storage of hazardous materials and the appropriate disposal of biocontaminants.
- i. If applicable, appropriate precautions in transporting children.
 - j. First aid and cardiopulmonary resuscitation.
- 2. A child care provider or employee or volunteer of a child care provider shall also complete ongoing in—service training on an annual basis including training on the topics listed under subd. 1. a. to j.

SECTION 781. 48.651 (2) of the statutes is amended to read:

48.651 (2) The department in a county having a population of 750,000 or more or a county department shall certify child care providers under sub. (1) or the department may contract with a Wisconsin Works agency, as defined in s. 49.001 (9), child care resource and referral agency, Indian tribe, or other agency to certify child care providers under sub. (1) in a particular geographic area or for a particular Indian tribal unit. The department in a county having a population of 750,000 or more or a county department that certifies child care providers under sub. (1) may charge a fee to cover the costs of certifying those providers. An agency or Indian tribe contracted with under this subsection may charge a fee specified by the department to supplement the amount provided by the department under the contract for certifying child care providers.

SECTION 782. 48.651 (2m) of the statutes is repealed. SECTION 783. 48.651 (3) (a) of the statutes is amended to read:

48.651 (3) (a) If a child care provider certified under sub. (1) is convicted of a serious crime, as defined in s. 48.685 48.686 (1) (c) 3m., or if a caregiver specified in s. 48.685 48.686 (1) (ag) 1. a. or a nonclient resident, as defined in s. 48.685 48.686 (1) (bm), of the child care provider is convicted or adjudicated delinquent for committing a serious crime, as defined in s. 48.686 (1) (c), on or after his or her 12th 10th birthday, or if the department provides written notice of a decision under s. 48.686 (4p) that the child care provider, caregiver, or nonclient resident is ineligible for certification, employment, or residence at the child care provider, the department in a county having a population of 750,000 or more, a county department, or an agency contracted with under sub. (2) shall revoke the certification of the child care provider immediately upon providing written notice of revocation and the grounds for revocation and an explanation of the process for appealing the revocation.

SECTION 784. 48.651 (3) (b) of the statutes is amended to read:

48.651 (3) (b) If a child care provider certified under sub. (1) is the subject of a pending criminal charge alleging that the person has committed a serious crime, as defined in s. 48.685 48.686 (1) (c) 3m., or if a caregiver specified in s. 48.685 48.686 (1) (ag) 1. -a. or a nonclient resident, as defined in s. 48.685 48.686 (1) (bm), of the child care provider is the subject of a pending criminal charge or delinquency petition alleging that the person has committed a serious crime on or after his or her 12th 10th birthday, the department in a county having a population of 750,000 or more, a county department, or an agency contracted with under sub. (2) shall immediately suspend the certification of the child care provider until the department, county department, or agency obtains information regarding the final disposition of the charge or delinquency petition indicating that the person is not ineligible to be certified under sub. (1).

SECTION 785. 48.66(1)(a) of the statutes is amended to read:

48.66 (1) (a) Except as provided in s. 48.715 (6) and (7), the department shall license and supervise child welfare agencies, as required by s. 48.60, group homes, as required by s. 48.625, shelter care facilities, as required by s. 938.22, and child care centers, as required by s. 48.65. The department may license foster homes, as provided by s. 48.62, and may license and supervise county departments in accordance with the procedures specified in this section and in ss. 48.67 to 48.74. The department may supervise a child care program established or contracted for under s. 120.13 (14) that receives payment under s. 49.155 for the child care provided. In the discharge of this duty the department may inspect the records and visit the premises of all child welfare agencies, group homes, shelter care facilities, and child care centers and visit the premises of all foster homes in which children are placed. The department may also inspect the

records and visit the premises of all child care programs established or contracted for under s. 120.13 (14) that receive payment under s. 49.155 for the child care provided.

SECTION 786. 48.66 (5) of the statutes is amended to read:

48.66 (5) A child welfare agency, group home, child care center, or shelter care facility license, other than a probationary license, is valid until revoked or suspended, but shall be reviewed every 2 years after the date of issuance as provided in this subsection. At least 30 days prior to the continuation date of the license, the licensee shall submit to the department an application for continuance of the license in the form and containing the information that the department requires. If the minimum requirements for a license established under s. 48.67 are met, the application is approved, the applicable fees referred to in ss. 48.68 (1) and, 48.685 (8), and 48.686 (2) (ag) are paid, and any forfeiture under s. 48.715 (3) (a) or penalty under s. 48.76 that is due is paid, the department shall continue the license for an additional 2-year period, unless sooner suspended or revoked. If the application is not timely filed, the department shall issue a warning to the licensee. If the licensee fails to apply for continuance of the license within 30 days after receipt of the warning, the department may revoke the license as provided in s. 48.715 (4) and (4m) (b).

SECTION 787. 48.68 (1) of the statutes is amended to read:

48.68 (1) After receipt of an application for a license, the department shall investigate to determine if the applicant meets the minimum requirements for a license adopted by the department under s. 48.67 and meets the requirements specified in s. 48.685, if or 48.686, whichever is applicable. In determining whether to issue or continue a license, the department may consider any action by the applicant, or by an employee of the applicant, that constitutes a substantial failure by the applicant or employee to protect and promote the health, safety, and welfare of a child. Upon satisfactory completion of this investigation and payment of the fee required under s. 48.615 (1) (a) or (b), 48.625 (2) (a), 48.65 (3) (a), or 938.22 (7) (b), the department shall issue a license under s. 48.66 (1) (a) or, if applicable, a probationary license under s. 48.69 or, if applicable, shall continue a license under s. 48.66 (5). At the time of initial licensure and license renewal, the department shall provide a foster home licensee with written information relating to the monthly foster care rates and supplemental payments specified in s. 48.62 (4), including payment amounts, eligibility requirements for supplemental payments, and the procedures for applying for supplemental payments.

SECTION 788. 48.685 (1) (ag) 1. b. of the statutes is amended to read:

48.685 (1) (ag) 1. b. A person who has, or is seeking, a license, certification or contract to operate an entity,

who is receiving, or is seeking, payment under s. 48.623 (6) (am) for operating an entity, or who is seeking payment under s. 48.623 (6) (bm) for operating an entity.

SECTION 789. 48.685 (1) (am) of the statutes is renumbered 48.685 (1) (am) (intro.) and amended to read:

48.685 (1) (am) (intro.) "Client" means a person who receives direct care or treatment services from an entity or from a caregiver specified in par. (ag) 1. am. or from a child care program under s. 48.686 (1) (aj), including all of the following:

SECTION 790. 48.685 (1) (am) 1., 2. and 3. of the statutes are created to read:

48.685 (1) (am) 1. An adopted child for whom adoption assistance payments are being made under s. 48.975.

- 2. A child for whom subsidized guardianship payments are being made under s. 48.623.
- 3. A person who is 18 to 21 years old, is receiving independent living services under 42 USC 677 (a), is no longer placed in out–of–home care, and is residing in the foster home in which he or she was previously placed.

SECTION 791. 48.685 (1) (b) of the statutes is amended to read:

48.685 (1) (b) "Entity" means a child welfare agency that is licensed under s. 48.60 to provide care and maintenance for children, to place children for adoption, or to license foster homes: a foster home that is licensed under s. 48.62; an interim caretaker to whom subsidized guardianship payments are made under s. 48.623 (6); a person who is proposed to be named as a successor guardian in a successor subsidized guardianship agreement under s. 48.623 (2); a group home that is licensed under s. 48.625; a shelter care facility that is licensed under s. 938.22; a child care center that is licensed under s. 48.65 or established or contracted for under s. 120.13 (14); a child care provider that is certified under s. 48.651; an organization that facilitates delegations of the care and custody of children under s. 48.979; or a temporary employment agency that provides caregivers to another entity.

SECTION 792. 48.685 (1) (bm) of the statutes is amended to read:

48.685 (1) (bm) "Nonclient resident" means a person, including a person who is under 18 years of age, but not under 10 years of age, who resides, or is expected to reside, at an entity or with a caregiver specified in par. (ag) 1. am., who is not a client of the entity or caregiver, and who has, or is expected to have, regular, direct contact with clients of the entity or caregiver.

SECTION 793. 48.685 (1) (c) 3m. of the statutes is repealed.

SECTION 793g. 48.685 (1) (c) 3r. of the statutes is created to read:

48.685 (1) (c) 3r. For purposes of licensing a foster home for the placement of a child or of providing subsidized guardianship payments to an interim caretaker

under s. 48.623 (6) (am) or to a person seeking those payments as a successor guardian under s. 48.623 (6) (bm), or of permitting a person to be a caregiver or nonclient resident of a licensed foster home, any violation listed in subd. 1. to 3. or sub. (5) (bm) 1. to 4.

SECTION 794. 48.685 (1) (c) 4. of the statutes is amended to read:

48.685 (1) (c) 4. A violation of the law of any other state or United States jurisdiction that would be a violation listed in subd. 1., 2., 3., or 3m. 3r. if committed in this state

SECTION 795. 48.685 (2) (am) (intro.) of the statutes is amended to read:

48.685 (2) (am) (intro.) The department, a county department, an agency contracted with under s. 48.651 (2), or a child welfare agency, or a school board shall obtain all of the following with respect to a caregiver specified in sub. (1) (ag) 1. b., or a nonclient resident of an entity, and a person under 18 years of age, but not under 12 years of age, who is a caregiver of a child care center that is licensed under s. 48.65 or established or contracted for under s. 120.13 (14) or of a child care provider that is certified under s. 48.651:

SECTION 796. 48.685 (2) (am) 5. of the statutes is amended to read:

48.685 (2) (am) 5. Information maintained by the department of health services under this section and under ss. 48.623 (6) (am) 2. and (bm) 5., 48.651 (2m), 48.75 (1m), and 48.979 (1) (b), and 120.13 (14) regarding any denial to the person of a license, or continuation or renewal of a license, certification, or a contract to operate an entity, or of payments under s. 48.623 (6) for operating an entity, for a reason specified in sub. (4m) (a) 1. to 5. and regarding any denial to the person of employment at, a contract with, or permission to reside at an entity or of permission to reside with a caregiver specified in sub. (1) (ag) 1. am. for a reason specified in sub. (4m) (b) 1. to 5. If the information obtained under this subdivision indicates that the person has been denied a license, or continuation or renewal of a license, certification, a contract, payments, employment, a contract, or permission to reside as described in this subdivision, the department, a county department, an agency contracted with under s. 48.651 (2), or a child welfare agency, or a school board need not obtain the information specified in subds. 1. to

SECTION 797. 48.685 (2) (ar) of the statutes is repealed.

SECTION 798. 48.685 (2) (b) 1. (intro.), a., b., c. and d. of the statutes are renumbered 48.685 (2) (b) (intro.), 1m., 2m., 3m. and 4m.

SECTION 799. 48.685 (2) (b) 1. e. of the statutes is renumbered 48.685 (2) (b) 5m. and amended to read:

48.685 (2) (b) 5m. Information maintained by the department of health services under this section and under ss. 48.623 (6) (am) 2. and (bm) 5., 48.651 (2m),

48.75 (1m), and 48.979 (1) (b), and 120.13 (14) regarding any denial to the person of a license, or continuation or renewal of a license, certification, or a contract to operate an entity, or of payments under s. 48.623 (6) for operating an entity, for a reason specified in sub. (4m) (a) 1. to 5. and regarding any denial to the person of employment at, a contract with, or permission to reside at an entity or of permission to reside with a caregiver specified in sub. (1) (ag) 1. am. for a reason specified in sub. (4m) (b) 1. to 5. If the information obtained under this subd. 1. e. subdivision indicates that the person has been denied a license, continuation or renewal of a license, certification, a contract, payments, employment, or permission to reside as described in this subd. 1. e. subdivision, the entity need not obtain the information specified in subd. 1. a. to d. subds. 1m. to 4m.

SECTION 800. 48.685 (2) (b) 2. of the statutes is repealed.

SECTION 801. 48.685 (2) (b) 4. of the statutes is repealed.

SECTION 802. 48.685 (2) (bb) of the statutes is amended to read:

48.685 (2) (bb) If information obtained under par. (am) or (b) 1. indicates a charge of a serious crime, but does not completely and clearly indicate the final disposition of the charge, the department, county department, agency contracted with under s. 48.651 (2), child welfare agency, school board, or entity shall make every reasonable effort to contact the clerk of courts to determine the final disposition of the charge. If a background information form under sub. (6) (a) or (am) indicates a charge or a conviction of a serious crime, but information obtained under par. (am) or (b) 1. does not indicate such a charge or conviction, the department, county department, agency contracted with under s. 48.651 (2), child welfare agency, school board, or entity shall make every reasonable effort to contact the clerk of courts to obtain a copy of the criminal complaint and the final disposition of the complaint. If information obtained under par. (am) or (b) 1., a background information form under sub. (6) (a) or (am), or any other information indicates a conviction of a violation of s. 940.19 (1), 940.195, 940.20, 941.30, 942.08, 947.01 (1), or 947.013 obtained not more than 5 years before the date on which that information was obtained, the department, county department, agency contracted with under s. 48.651 (2), child welfare agency, school board, or entity shall make every reasonable effort to contact the clerk of courts to obtain a copy of the criminal complaint and judgment of conviction relating to that violation.

SECTION 803. 48.685 (2) (bd) of the statutes is amended to read:

48.685 (2) (bd) Notwithstanding pars. (am) and (b) 1., the department, a county department, an agency contracted with under s. 48.651 (2), or a child welfare agency, or a school board is not required to obtain the

information specified in par. (am) 1. to 5., and an entity is not required to obtain the information specified in par. (b) 1. a. to e. 1m. to 5m., with respect to a person under 18 years of age whose background information form under sub. (6) (am) indicates that the person is not ineligible to be employed at, contracted with, or permitted to reside at an entity or permitted to reside with a caregiver specified under sub. (1) (ag) 1. am. of the entity for a reason specified in sub. (4m) (b) 1. to 5. and with respect to whom the department, county department, contracted agency, child welfare agency, school board, or entity otherwise has no reason to believe that the person is ineligible to be employed, contracted with, or permitted to reside at an entity for any of those reasons. This paragraph does not preclude the department, a county department, an agency contracted with under s. 48.651 (2), or a child welfare agency, or a school board from obtaining, at its discretion, the information specified in par. (am) 1. to 5. with respect to a person described in this paragraph who is a nonclient resident or a prospective nonclient resident of an entity.

SECTION 804. 48.685 (2) (bg) of the statutes is amended to read:

48.685 (2) (bg) If an entity employs or contracts with a caregiver for whom, within the last year, the information required under par. (b) 1. a. to c. and e. 1m. to 3m. and 5m. has already been obtained by another entity, the entity may obtain that information from that other entity, which shall provide the information, if possible, to the requesting entity. If an entity cannot obtain the information required under par. (b) 1. a. to c. and e. 1m. to 3m. and 5m. from another entity or if an entity has reasonable grounds to believe that any information obtained from another entity is no longer accurate, the entity shall obtain that information from the sources specified in par. (b) 1. a. to c. and e. 1m. to 3m. and 5m.

SECTION 805. 48.685 (2) (bm) of the statutes is amended to read:

48.685 (2) (bm) If the person who is the subject of the search under par. (am), (ar), or (b) 1. is not a resident of this state, or if at any time within the $\frac{3}{5}$ years preceding the date of the search that person has not been a resident of this state, or if the department, county department, agency contracted with under s. 48.651 (2), child welfare agency, school board, or entity determines that the person's employment, licensing, or state court records provide a reasonable basis for further investigation, the department, county department, contracted agency, child welfare agency, school board, or entity shall make a good faith effort to obtain from any state or other United States jurisdiction in which the person is a resident or was a resident within the $\frac{3}{5}$ years preceding the date of the search information that is equivalent to the information specified in par. (am) 1., (ar), or (b) 1. a. 1m. The department, county department, contracted agency, child welfare agency, school board, or entity may require the person to

be fingerprinted on 2 fingerprint cards, each bearing a complete set of the person's fingerprints, or by other technologies approved by law enforcement agencies. The department of justice may provide for the submission of the fingerprint cards or fingerprints by other technologies to the federal bureau of investigation for the purposes of verifying the identity of the person fingerprinted and obtaining records of his or her criminal arrests and convictions.

SECTION 806. 48.685 (2) (br) of the statutes is repealed.

SECTION 807. 48.685 (3) (a) of the statutes is amended to read:

48.685 (3) (a) Subject to par. (am), every Every 4 years or at any time within that period that the department, a county department, or a child welfare agency considers appropriate, the department, county department, or child welfare agency shall request the information specified in sub. (2) (am) 1. to 5. for all caregivers specified in sub. (1) (ag) 1. b. who are licensed, certified, or contracted to operate an entity, or who are receiving payments under s. 48.623 (6) (am) for operating an entity, and for all persons who are nonclient residents of such a caregiver.

SECTION 808. 48.685 (3) (am) of the statutes is repealed.

SECTION 809. 48.685 (3) (b) of the statutes is amended to read:

48.685 (3) (b) Subject to par. (bm), every Every 4 years or at any time within that period that an entity considers appropriate, the entity shall request the information specified in sub. (2) (b) 1. a. to e. 1m. to 5m. for all persons who are caregivers specified in sub. (1) (ag) 1. a. or am. of the entity and for all nonclient residents of a caregiver specified in sub. (1) (ag) 1. am. of the entity.

SECTION 810. 48.685 (3) (bm) of the statutes is repealed.

SECTION 811. 48.685 (3m) of the statutes is amended to read:

48.685 (3m) Notwithstanding subs. (2) (b) 1. and (3) (b), if the department, a county department, an agency contracted with under s. 48.651 (2), or a child welfare agency, or a school board has obtained the information required under sub. (2) (am) or (3) (a) or (am) with respect to a person who is a caregiver specified in sub. (1) (ag) 1. b. and that person is also an employee, contractor, or nonclient resident of an entity, the entity is not required to obtain the information specified in sub. (2) (b) 1. or (3) (b) with respect to that person.

SECTION 812. 48.685 (4m) (a) (intro.) of the statutes is amended to read:

48.685 (**4m**) (a) (intro.) Notwithstanding s. 111.335, and except as provided in par. (ad) and sub. (5), the department may not license, or continue or renew the license of, a person to operate an entity, the department in a county having a population of 750,000 or more, a

county department, or an agency contracted with under s. 48.651 (2) may not certify a child care provider under s. 48.651, a county department or a child welfare agency may not license, or renew the license of, a foster home under s. 48.62, and the department in a county having a population of 750,000 or more or a county department may not provide subsidized guardianship payments to an interim caretaker under s. 48.623 (6) (am) or to a person seeking those payments as a successor guardian under s. 48.623 (6) (bm), and a school board may not contract with a person under s. 120.13 (14), if the department, county department, contracted agency, or school board knows or should have known any of the following:

SECTION 813. 48.685 (4m) (a) 1. of the statutes is amended to read:

48.685 (4m) (a) 1. That the person has been convicted of a serious crime or, if the person is an applicant for issuance or continuation of a license to operate a child care center or for initial certification under s. 48.651 or for renewal of that certification or if the person is proposing to contract with a school board under s. 120.13 (14) or to renew a contract under that subsection, that the person has been convicted of a serious crime or adjudicated delinquent on or after his or her 12th 10th-birthday for committing a serious crime or that the person is the subject of a pending criminal charge or delinquency petition alleging that the person has committed a serious crime on or after his or her 12th 10th birthday.

SECTION 814. 48.685 (4m) (ad) of the statutes is amended to read:

48.685 (4m) (ad) The department, a county department, or a child welfare agency may license a foster home under s. 48.62; the department may license a child care center under s. 48.65; the department in a county having a population of 750,000 or more, a county department, or an agency contracted with under s. 48.651 (2) may certify a child care provider under s. 48.651; or the department in a county having a population of 750,000 or more or a county department may provide subsidized guardianship payments to an interim caretaker under s. 48.623 (6) (am) or to a person seeking those payments as a successor guardian under s. 48.623 (6) (bm); and a school board may contract with a person under s. 120.13 (14), conditioned on the receipt of the information specified in sub. (2) (am) and (ar) indicating that the person is not ineligible to be so licensed, certified, or provided those payments, or contracted with for a reason specified in par. (a) 1. to 5.

SECTION 815. 48.685 (4m) (b) 1. of the statutes is amended to read:

48.685 (**4m**) (b) 1. That the person has been convicted of a serious crime or, if the person is a caregiver or nonclient resident of a child care center that is licensed under s. 48.65 or established or contracted for under s. 120.13 (14) or of a child care provider that is certified

under s. 48.651, that the person has been convicted of a serious crime or adjudicated delinquent on or after his or her 12th 10th birthday for committing a serious crime or that the person is the subject of a pending criminal charge or delinquency petition alleging that the person has committed a serious crime on or after his or her 12th 10th birthday.

SECTION 816. 48.685 (4m) (c) of the statutes is amended to read:

48.685 (4m) (c) If the background information form completed by a person under sub. (6) (am) indicates that the person is not ineligible to be employed or contracted with for a reason specified in par. (b) 1. to 5., an entity may employ or contract with the person for not more than 60 45 days pending the receipt of the information sought under sub. (2) (am) or (b) 4. If the background information form completed by a person under sub. (6) (am) indicates that the person is not ineligible to be permitted to reside at an entity or with a caregiver specified in sub. (1) (ag) 1. am. for a reason specified in par. (b) 1. to 5. and if an entity otherwise has no reason to believe that the person is ineligible to be permitted to reside at an entity or with that caregiver for any of those reasons, the entity may permit the person to reside at the entity or with the caregiver for not more than 60 45 days pending receipt of the information sought under sub. (2) (am) or (b) \pm . An entity shall provide supervision for a person who is employed, contracted with, or permitted to reside as permitted under this paragraph.

SECTION 817. 48.685 (4m) (d) of the statutes is created to read:

48.685 (4m) (d) If the department learns that a caregiver or nonclient resident is the subject of a pending investigation for a crime or offense that, under this subsection or sub. (5), could result in a bar to employment as a caregiver or residence at an entity, the department may notify the entity of the pending investigation.

SECTION 818. 48.685 (5) (a) of the statutes is amended to read:

48.685 (5) (a) Subject to pars. par. (bm) and (br), the department may license to operate an entity, the department in a county having a population of 750,000 or more, a county department, or an agency contracted with under s. 48.651 (2) may certify under s. 48.651, a county department or a child welfare agency may license to operate a foster home under s. 48.62, the department in a county having a population of 750,000 or more or a county department may provide subsidized guardianship payments under s. 48.623 (6), and a school board may contract with under s. 120.13 (14) to a person who otherwise may not be so licensed, certified, or contracted with or provided those payments for a reason specified in sub. (4m) (a) 1. to 5., and an entity may employ, contract with, or permit to reside at the entity or permit to reside with a caregiver specified in sub. (1) (ag) 1. am. of the entity a person who otherwise may not be so employed, provided payments, contracted with, or permitted to reside at the entity or with that caregiver for a reason specified in sub. (4m) (b) 1. to 5., if the person demonstrates to the department, the county department, the contracted agency, the or child welfare agency, or the school board or, in the case of an entity that is located within the boundaries of a reservation, to the person or body designated by the Indian tribe under sub. (5d) (a) 3., by clear and convincing evidence and in accordance with procedures established by the department by rule or by the tribe that he or she has been rehabilitated.

SECTION 819. 48.685 (5) (bm) (intro.) of the statutes is amended to read:

48.685 (5) (bm) (intro.) For purposes of licensing a foster home for the placement of a child on whose behalf foster care maintenance payments under s. 48.62 (4) will be provided or of providing subsidized guardianship payments to an interim caretaker under s. 48.623 (6) (am) or to a person seeking those payments as a successor guardian under s. 48.623 (6) (bm), no person, including a caregiver or nonclient resident under this section, who has been convicted of any of the following offenses may be permitted to demonstrate that he or she has been rehabilitated:

SECTION 820. 48.685 (5) (br) of the statutes is repealed.

SECTION 821. 48.685 (5c) (a) of the statutes is amended to read:

48.685 (**5c**) (a) Any person who is permitted but fails under sub. (5) (a) to demonstrate to the department, an agency contracted with under s. 48.651 (2), or a child welfare agency that he or she has been rehabilitated may appeal to the secretary or his or her designee. Any person who is adversely affected by a decision of the secretary or his or her designee under this paragraph has a right to a contested case hearing under ch. 227.

SECTION 822. 48.685 (5c) (c) of the statutes is repealed.

SECTION 823. 48.685 (5m) of the statutes is amended to read:

48.685 (5m) Notwithstanding s. 111.335, the department may refuse to license a person to operate an entity, a county department or a child welfare agency may refuse to license a foster home under s. 48.62, the department in a county having a population of 750,000 or more or a county department may refuse to provide subsidized guardianship payments to a person under s. 48.623 (6), and an entity may refuse to employ or contract with a caregiver or permit a nonclient resident to reside at the entity or with a caregiver specified in sub. (1) (ag) 1. am. of the entity if the person has been convicted of an offense that is not a serious crime, but that is, in the estimation of the department, county department, child welfare agency, or entity, substantially related to the care of a client. Notwithstanding s. 111.335, the department may refuse to license a person to operate a child care center,

the department in a county having a population of 750,000 or more, a county department, or an agency contracted with under s. 48.651 (2) may refuse to certify a child care provider under s. 48.651, a school board may refuse to contract with a person under s. 120.13 (14), and a child care center that is licensed under s. 48.65 or established or contracted for under s. 120.13 (14) or a child care provider that is certified under s. 48.651 may refuse to employ or contract with a caregiver or permit a nonclient resident to reside at the child care center or child care provider if the person has been convicted of or adjudicated delinquent on or after his or her 12th birthday for an offense that is not a serious crime, but that is, in the estimation of the department, county department, contracted agency, school board, child care center, or child care provider, substantially related to the care of a client.

SECTION 824. 48.685 (6) (a) of the statutes is amended to read:

48.685 (6) (a) Except as provided in this paragraph, the department shall require any person who applies for issuance, continuation, or renewal of a license to operate an entity, a county department or a child welfare agency shall require any person who applies for issuance or renewal of a license to operate a foster home under s. 48.62, and the department in a county having a population of 750,000 or more or a county department shall require any person who applies for subsidized guardianship payments under s. 48.623 (6) to complete a background information form that is provided by the department. The department shall require any person who applies for issuance, but not continuation, of a license to operate a child care center under s. 48.65, a school board shall require any person who proposes to contract, but not renew a contract, with the school board under s. 120.13 (14), and the department in a county having a population of 750,000 or more, a county department, or an agency contracted with under s. 48.651 (2) shall require any child care provider who applies for initial certification, but not renewal of that certification, under s. 48.651 to complete a background information form that is provided by the department.

SECTION 825. 48.685 (6) (am) of the statutes is amended to read:

48.685 (6) (am) Except as provided in this paragraph, every Every 4 years an entity shall require all of its caregivers and all nonclient residents of the entity or of a caregiver specified in sub. (1) (ag) 1. am. of the entity to complete a background information form that is provided to the entity by the department. -A child care center that is licensed under s. 48.65 or established or contracted for under s. 120.13 (14) or a child care provider that is certified under s. 48.651 is exempt from the 4—year requirement, but shall require any new caregiver or nonclient resident to complete a background information form that is provided to the child care center or child care provider by the department.

SECTION 826. 48.685 (6) (b) 1. of the statutes is amended to read:

48.685 (6) (b) 1. For caregivers who are licensed by the department, for persons under 18 years of age, but not under 12 years of age, who are caregivers of a child care center that is licensed under s. 48.65 or established or contracted for under s. 120.13 (14) or of a child care provider that is certified under s. 48.651, for persons who are nonclient residents of an entity that is licensed by the department, and for other persons specified by the department by rule, the entity shall send the background information form to the department.

SECTION 827. 48.685 (6) (b) 2. of the statutes is amended to read:

48.685 (6) (b) 2. For caregivers who are licensed or certified by a county department or an agency contracted with under s. 48.651 (2), for persons who are nonclient residents of an entity that is licensed or certified by a county department or an agency contracted with under s. 48.651 (2), and for other persons specified by the department by rule, the entity shall send the background information form to the county department or contracted agency.

SECTION 828. 48.685 (6) (b) 4. of the statutes is repealed.

SECTION 829. 48.685 (8) of the statutes is amended to read:

48.685 (8) The department, the department of health services, a county department, an agency contracted with under s. 48.651 (2), or a child welfare agency, or a school board may charge a fee for obtaining the information required under sub. (2) (am) or (ar) or (3) (a) or (am), for providing information to an entity to enable the entity to comply with sub. (2) (b) 1. or (3) (b), or for obtaining and submitting fingerprints under sub. (2) (bm) or (br). The fee may not exceed the reasonable cost of obtaining the information or of obtaining and submitting fingerprints. No fee may be charged to a nurse aide, as defined in s. 146.40 (1) (d), for obtaining or maintaining information or for obtaining and submitting fingerprints if to do so would be inconsistent with federal law.

SECTION 830. 48.685 (9) of the statutes is created to read:

48.685 (9) The department may promulgate any rules necessary for the administration of this section.

SECTION 831. 48.686 of the statutes is created to read: 48.686 Criminal history and child abuse record search; child care. (1) In this section:

- (ag) "Caregiver" means any of the following:
- 1. A person who is any of the following:
- a. An employee or independent contractor of a child care program.
- b. Involved in the care or supervision of clients of a child care program or has unsupervised access to clients of a child care program.

- 2. A person who has, or is seeking, a license, certification, or contract to operate a child care program.
- (aj) "Child care program" means a child care center that is licensed under s. 48.65 or established or contracted for under s. 120.13 (14), a child care provider that is certified under s. 48.651, or a temporary employment agency that provides caregivers to another child care program.
- (am) "Client" means a person who receives direct care from a child care program, from an entity under s. 48.685 (1) (b) or from a caregiver specified in s. 48.685 (1) (ag) 1. am., including all of the following:
- 1. An adopted child for whom adoption assistance payments are being made under s. 48.975.
- 2. A child for whom subsidized guardianship payments are being made under s. 48.623.
- 3. A person who is 18 to 21 years old, is receiving independent living services under 42 USC 677 (a) from an agency, is no longer placed in out–of–home care, and is residing in the foster home in which he or she was previously placed.
- (ar) "Contractor" means, with respect to a child care program, a person, or that person's agent, who provides services to the child care program under an express or implied contract or subcontract.
- (bm) "Nonclient resident" means a person who is age 10 or older, who resides, or is expected to reside, at a child care program, and who is not a client of the child care program or caregiver.
- (br) "Reservation" means land in this state within the boundaries of a reservation of a tribe or within the bureau of Indian affairs service area for the Ho-Chunk Nation.
 - (c) "Serious crime" means any of the following:
- 1. A violation of s. 940.12, 940.22 (2) or (3), 940.285 (2), 940.29, 940.295, or 942.09 (2).
- 2. A violation of s. 940.302 (2) if s. 940.302 (2) (a) 1. b. applies.
- 3. An offense under ch. 948 that is a felony, other than a violation of s. 948.22 (2).
- 4. A violation of s. 940.19 (3), 1999 stats., or of s. 940.19 (2), (4), (5), or (6) or 940.20 (1) or (1m), if the victim is the spouse of the person.
- 5. A violation of s. 940.01, 940.02, 940.03, 940.05, 940.06, 940.21, 940.225 (1), (2), or (3), 940.23, 940.305, 940.31, 941.20 (2) or (3), 941.21, 943.04, 943.10 (2), 943.32 (2), or 948.21 (1) (a).
- 6. Only for a caregiver, as defined in par. (ag) 2., a violation of s. 943.201, 943.203, or 943.38 (1) or (2); a violation of s. 943.34 (1), 943.395 (1), 943.41 (3) (e), (4) (a), (5), (6), or (6m), 943.45 (1), 943.455 (2), 943.46 (2), 943.47 (2), 943.50 (1m), or 943.70 (2) (a) or (am) or (3) (a) that is a felony; or an offense under subch. IV of ch. 943 that is a felony.
- 7. A violation of sub. (2) or s. 48.685 (2), (3), (4m) (b), or (6), 2015 stats., if the violation involves the provision of false information to or the intentional withholding

of information from, the department, a county department, an agency contracting under s. 48.651 (2), a school board, or a child care program.

- 8. An offense involving fraudulent activity as a participant in the Wisconsin Works program under ss. 49.141 to 49.161, including as a recipient of a child care subsidy under s. 49.155, or as a recipient of aid to families with dependent children under s. 49.19, medical assistance under subch. IV of ch. 49, food stamps benefits under the food stamp program under 7 USC 2011 to 2036, supplemental security income payments under s. 49.77, payments for the support of children of supplemental security income recipients under s. 49.775, or health care benefits under the Badger Care health care program under s. 49.665.
- 9. A violation of s. 125.075 (1), 125.085 (3) (a) 2., 125.105 (2) (b), 125.66 (3), 125.68 (12), 940.09, 940.19 (2), (4), (5), or (6), 940.20, 940.203, 940.205, 940.207, 940.25, or 943.23 (1g), a violation of s. 948.51 (2) that is a felony under s. 948.51 (3) (b) or (c), a violation of s. 346.63 (1), (2), (5), or (6) that is a felony under s. 346.65 (2) (am) 5., 6., or 7. or (f), (2j) (d), or (3m), or an offense under ch. 961 that is a felony, if the person completed his or her sentence, including any probation, parole, or extended supervision, or was discharged by the department of corrections, less than 5 years before the date of the investigation under sub. (2) (am).
- 10. A violation of s. 948.22 (2), if the person completed his or her sentence, including any probation, parole, or extended supervision, or was discharged by the department of corrections, less than 5 years before the date of the investigation under sub. (2) (am), unless the person has paid all arrearages due and is meeting his or her current support obligations.
- 11. A violation of the law of any other state or United States jurisdiction that would be a violation listed in subd. 1. to 10. if committed in this state.
- 12. A violation of the laws of another state or United States jurisdiction that if committed in this state would constitute felony battery under s. 940.19 (2), (4), (5), or (6) or 940.20, a felony offense of domestic abuse, as defined in s. 813.12 (1) (am), a sex offense or a violent crime under ch. 948, or a violation of 940.225 if the victim was a child.
- (2) (a) The department shall require any person who applies for issuance of an initial license to operate a child care center under s. 48.65, a school board shall require any person who proposes an initial contract with the school board under s. 120.13 (14), and the department in a county having a population of 750,000 or more, a county department, or an agency contracted with under s. 48.651 (2) shall require any child care provider who applies for initial certification under s. 48.651 to submit the information required for a background check request under par. (ag). A school board, county department, or

- contracted agency shall submit the completed background information request to the department.
- (ab) Each child care program shall submit a request to the department for a criminal background check for each potential caregiver and potential nonclient resident prior to the date on which an individual becomes a caregiver or nonclient resident, and at least once during every 5-year period for each existing caregiver or nonclient resident, except if all of the following apply:
- 1. The caregiver, potential caregiver, nonclient resident, or potential nonclient resident has received a background check as described in par. (am) while employed or seeking employment by another child care program within the state within the last 5 years.
- 2. The department provided to the child care program under subd. 1. a qualifying background check result for the caregiver, potential caregiver, nonclient resident, or potential nonclient resident.
- 3. The caregiver, potential caregiver, nonclient resident, or potential nonclient resident is employed by or resides at a child care program within the state or has been separated from employment or residence at a child care program within the state for a period of not more than 180 consecutive days.
- (ag) 1. A request for a background check to the department under par. (a) or (ab) shall be in the manner and on forms prescribed by the department, and shall include all of the following:
- a. Fingerprints of the subject that meet the standards of the department.
- b. Any additional information that the department deems necessary to perform the criminal background check.
- 2. A request for a criminal background check is considered submitted on the day that the department receives all of the information required under subd. 1.
- 3. The requester of a background check under this paragraph shall submit all fees required by the department pursuant to the instructions provided by the department, not to exceed the actual cost of conducting the criminal background check.
- (am) Upon receipt of a request submitted under par. (a) or (ab), the department shall obtain all of the following with respect to a caregiver or a nonclient resident who is not under 10 years of age:
- 1. A fingerprint-based criminal history search from the records maintained by the department of justice.
- 2. Information that is contained in the registry under s. 146.40 (4g) regarding any findings against the person.
- 3. Information maintained by the department of safety and professional services regarding the status of the person's credentials, if applicable.
- 4. Information maintained by the department regarding any final determination under s. 48.981 (3) (c) 5m. or, if a contested case hearing is held on such a determina-

tion, any final decision under s. 48.981 (3) (c) 5p. that the person has abused or neglected a child.

- 5. Information maintained by the department of health services under s. 48.685 regarding any denial to the person of a license, continuation or renewal of a license, certification, or a contract to operate an entity or a child care program, for a reason specified in s. 48.685 (4m) (a) 1. to 5. and regarding any denial to the person of employment at, a contract with, or permission to reside at an entity or a child care program for a reason specified in s. 48.685 (4m) (a) 1. to 5.
- 6. Information that is contained in the sex offender registry under s. 301.45 regarding whether the person has committed a sex offense that is a serious crime.
- 7. A fingerprint–based criminal history search using the federal bureau of investigation next generation identification.
- 8. A search of the national crime information center's national sex offender registry.
- 9. A search of the following registries, repositories, or databases in the state where the caregiver or nonclient resident resided for the period starting on the date 5 years prior to the department's receipt of the background check request and ending on the date the department received the background check request:
 - a. The state criminal registry or repository.
 - b. The state sex offender registry or repository.
- c. The state-based child abuse and neglect registry and database.
- 10. A search of the department's criminal background check records.
- (ar) After receiving a request under par. (a) or (ab), the department shall conduct the criminal background check as expeditiously as possible and shall make a good faith effort to complete all components of the criminal background check no later than 45 days after the date on which the request was submitted.
- (bb) If information obtained under par. (am) indicates a charge of a serious crime, but does not completely and clearly indicate the final disposition of the charge, the department shall make every reasonable effort to contact the clerk of courts to determine the final disposition of the charge. If information submitted to the department under par. (ag) indicates a charge or a conviction of a serious crime, but information obtained under par. (am) does not indicate such a charge or conviction, the department shall make every reasonable effort to contact the clerk of courts to obtain a copy of the criminal complaint and the final disposition of the complaint. If information obtained under par. (am), information submitted under par. (ag), or any other information indicates a conviction of a violation of s. 940.19 (1), 940.195, 940.20, 941.30, 942.08, 947.01 (1), or 947.013 obtained not more than 5 years before the date on which that information was obtained, the department shall make every reasonable effort to contact the clerk of courts to obtain a copy of the

criminal complaint and judgment of conviction relating to that violation.

- (bd) Notwithstanding par. (am), the department is not required to obtain the information specified in par. (am) 1. to 10., with respect to a person under 18 years of age whose background check request under par. (ag) indicates that the person is not ineligible to be permitted to reside at a child care program for a reason specified in sub. (4m) (a) 1. to 8. and with respect to whom the department otherwise has no reason to believe that the person is ineligible to be permitted to reside for any of those reasons. This paragraph does not preclude the department from obtaining, at its discretion, the information specified in par. (am) 1. to 10. with respect to a person described in this paragraph who is a nonclient resident or a potential nonclient resident of a child care program.
- (br) The department shall require the person who is the subject of a search under par. (am) to be fingerprinted on 2 fingerprint cards, each bearing a complete set of the person's fingerprints, or by other technologies approved by law enforcement agencies, unless the person has previously been fingerprinted under this paragraph.
- (3) (am) Every year or at any time that the department considers appropriate, the department may request the information specified in sub. (2) (am) 1. to 5. for all caregivers under sub. (1) (ag) 2., nonclient residents of such a caregiver, and caregivers under sub. (1) (ag) 1. who have direct contact with clients. For the purposes of this paragraph, "direct contact" means face—to—face physical proximity to a client that affords the opportunity to commit abuse or neglect of a client or to misappropriate the property of a client.
- (bm) Annually, by January 1, the department shall submit a report to the appropriate standing committees of the legislature under s. 13.172 (3) describing the report prepared under sub. (4p) (a) with respect to caregivers specified in sub. (1) (ag) 2., specifically any information indicating that the caregiver is ineligible under sub. (4m) (a) to be licensed, certified, or contracted to operate a child care program, and describing any action taken in response to the receipt of information under sub. (2) (am) indicating that such a caregiver is so ineligible.
- (4) (a) A child care program that violates sub. (2), (3), or (4m) (a) is subject to a forfeiture of not more than \$1,000 and to other sanctions specified by the department by rule.
- (b) A person who provides false information to the department under sub. (2) is subject to a forfeiture of not more than \$1,000 and to other sanctions specified by the department by rule.
- (4m) (a) Notwithstanding s. 111.335, and except as provided in par. (ad) and sub. (5), the department may not license, or continue or renew the license of, a person to operate a child care center under s. 48.65, the department in a county having a population of 750,000 or more, a county department, or an agency contracted with under

- s. 48.651 (2) may not certify a child care provider under s. 48.651, a school board may not contract with a person under s. 120.13 (14), and a child care program may not employ or contract with a caregiver specified in sub. (1) (ag) 1. if the department, county department, contracted agency, school board, or child care program knows or should have known any of the following:
- 1. That the person has been convicted of a serious crime or adjudicated delinquent on or after his or her 10th birthday for committing a serious crime or that the person is the subject of a pending criminal charge or delinquency petition alleging that the person has committed a serious crime on or after his or her 10th birthday.
- 3. That a unit of government or a state agency, as defined in s. 16.61 (2) (d), has made a finding that the person has abused or neglected any client or misappropriated the property of any client.
- 4. That a final determination has been made under s. 48.981 (3) (c) 5m. or, if a contested case hearing is held on such a determination, a final decision has been made under s. 48.981 (3) (c) 5p. that the person has abused or neglected a child.
- 5. That the department has determined the person ineligible to be licensed to operate a child care center under s. 48.65, to be certified to operate a child care provider under s. 48.651, to contract with a school board under s. 120.13 (14), to be employed as a caregiver at a child care program, or to be a nonclient resident at a child care program.
- 6. That the person has refused to provide information under sub. (2) (ag), or that the person refused to participate in, cooperate with, or submit required information for the criminal background check described in sub. (2) (am), including fingerprints.
- 7. That the person knowingly made a materially false statement in connection with the person's criminal background check described in sub. (2).
- 8. That the person knowingly omitted material information requested in connection with the person's criminal background check conducted under sub. (2).
- (ad) The department may license a child care center under s. 48.65; the department in a county having a population of 750,000 or more, a county department, or an agency contracted with under s. 48.651 (2) may certify a child care provider under s. 48.651; and a school board may contract with a person under s. 120.13 (14), conditioned on the receipt of the information specified in sub. (4p) (a) indicating that the person is not ineligible to be so licensed, certified, or contracted with for a reason specified in par. (a) 1. to 8.
- (c) A child care program may employ or contract with a potential caregiver or permit a potential nonclient resident to reside at the child care program for up to 45 days from the date a background check request is submitted to the department pending the completion of the department's report under sub. (4p) (a) if the department

- provides a preliminary report under sub. (4p) (c) to the child care program indicating that the potential caregiver or nonclient resident is not ineligible to work or reside at a child care program. At all times that children in care are present, an individual who received a qualifying result on a background check described in sub. (2) (am) within the past 5 years must supervise a potential employee or nonclient resident permitted to work or reside at the child care program under this paragraph.
- (4p) (a) The department shall provide the results of the criminal background check to the child care program in a written report that indicates only that the individual on whom the background check was conducted is eligible or ineligible for employment or to reside at the child care program, without revealing any disqualifying crime or other information regarding the individual.
- (b) The department shall provide the results of the criminal background check to the individual on whom the background check was conducted in a written report that indicates whether the individual is eligible or ineligible for employment or to reside at the child care program. If the individual is ineligible for employment or to reside at the child care program, the department's report shall include information on each disqualifying crime and information on the right to appeal.
- (c) Before the department completes its report under par. (a), a caregiver under sub. (1) (ag) 2. may submit a written request to the department for a preliminary report indicating whether a potential caregiver or nonclient resident is eligible to work or reside at a child care program under sub. (4m) (c). If the department receives such a request, it shall provide a written preliminary report to that caregiver indicating whether the individual is barred from employment as a caregiver or residence as a nonclient resident on the basis of a background check under sub. (2) (am) 1. or 7. If the individual is ineligible for employment or residence at a child care program based on the results of the preliminary report, the department shall also provide a preliminary report to the individual containing information related to each disqualifying crime
- (d) The results of a report under par. (c) may not be appealed by the individual until receipt of the department's report under par. (b) following completion of all components of the criminal background check.
- (4s) (a) An individual who is the subject of the department's report on the results of a criminal background check may appeal the department's decision. Only the person who is the subject of the department's report may appeal the department's decision. Neither the child care program nor any other person may appeal the department's decision.
- (b) An appeal request shall be submitted to the department at the address, e-mail address, or fax number identified in the statement of appeal rights no later than 60 days after the date of the department's decision, unless

the appellant requests, and the department grants, an extension for a specific amount of time prior to expiration of the 60 day appeal period. Extensions may be granted for good cause shown.

- (c) An appeal shall be submitted in the manner and on forms prescribed by the department, and must include all of the following information:
- 1. The information or issue disputed by the individual.
- 2. Any information known to the individual, or available to the individual through the exercise of reasonable diligence, that supports the individual's position.
- 3. The current or last known names, addresses, telephone numbers, and email addresses of any persons known or believed to have information relevant to determination of the appeal.
- 4. Copies of any documents or other materials in the possession of the individual, or reasonably available to the individual, that support the individual's position regarding the disputed information.
- (e) The department shall attempt to verify the accuracy of the information challenged by the appellant, including making reasonable good faith efforts to locate any missing information regarding the disqualifying crime that is relevant to the issue identified for appeal.
- (f) The department shall sustain the results of its criminal background check report if supported by a preponderance of the available evidence.
- (g) The department shall issue its appeal decision in writing. If the results of the original report are sustained upon review, the decision shall indicate the department's efforts to verify the accuracy of the information challenged by the individual. The decision shall also indicate any additional reconsideration and appeal rights available to the appellant.
- (h) An appellant under this subsection may seek reconsideration of the department's decision under par.(g) by the secretary or the secretary's designee.
- (i) A request for reconsideration detailing the basis for the request must be sent to the secretary at the address, e-mail address, or fax number identified in the department's decision no later than 30 days after the date of the department's decision.
- (j) The secretary or secretary's designee shall issue his or her reconsideration decision in writing and shall include information about any additional appeal rights available to the individual.
- (k) A denial of reconsideration under this subsection is a final decision of the department, and the appellant has a right to a contested case hearing under ch. 227.
- (L) The appeal and reconsideration process set forth in this subsection is the exclusive method for disputing a criminal history background report issued by the department. The department's decision may not be appealed in a ch. 68 or 227 proceeding challenging the denial of a license, certification, or contract to operate a child care

- program based on the department's criminal history background check report or challenging any other child care regulatory action taken in reliance upon that report.
- (m) Notwithstanding s. 19.35, the department may not publicly release or disclose the results of any criminal individual background report it issues, except that the department may release aggregated data by crime as listed in sub. (1) (c) from criminal background check results so long as the data does not contain personally identifiable information. The department may disclose and use information obtained in conducting criminal background checks as necessary during an appeal or reconsideration under this subsection.
- (5) (a) Subject to par. (br), the department may license to operate a child care program, the department in a county having a population of 750,000 or more, a county department, or an agency contracted with under s. 48.651 (2) may certify under s. 48.651, and a school board may contract with under s. 120.13 (14) a person who otherwise may not be licensed, certified, or contracted with for a reason specified in sub. (4m) (a) 1. to 8., and a child care program may employ, contract with, or permit to reside at the child care program a person who otherwise may not be so employed, contracted with, or permitted to reside for a reason specified in sub. (4m) (a) 1. to 8., if the person demonstrates to the department, the county department, the contracted agency, or the school board or, in the case of a child care program that is located within the boundaries of a reservation, to the person or body designated by the Indian tribe under sub. (5d) (a) 3., by clear and convincing evidence and in accordance with procedures established by the department by rule or by the tribe that he or she has been rehabilitated.
- (br) No person who has been convicted or adjudicated delinquent on or after his or her 10th birthday for committing any of the offenses identified in sub. (1) (c) 1. to 8. or 12. or for a violation of the law of any other state or United States jurisdiction that would be a violation listed in sub. (1) (c) 1. to 8. if committed in this state or who is the subject of a pending criminal charge or delinquency petition alleging that the person has committed any of those offenses on or after his or her 10th birthday may be permitted to demonstrate that he or she has been rehabilitated.
- (cm) Notwithstanding sub. (4m) (a) 1., if a person was convicted or adjudicated delinquent on or after his or her 10th birthday for committing any of the offenses listed in sub. (1) (c) 9. or 10. and the person completed his or her sentence, including any probation, parole, or extended supervision, or was discharged by the department of corrections, 5 or more years before the date of the investigation under sub. (2) (am), then the conviction or delinquency adjudication alone does not make the person ineligible to be licensed as a child care center under s. 48.65, certified as a child care provider under s. 48.651, contracted with under s. 120.13 (14), or employed by,

contracted with, or permitted to reside at a child care program and, with respect to that conviction or delinquency adjudication, the person need not demonstrate that he or she has been rehabilitated under par. (a) before being so licensed, certified, contracted with, employed, or permitted to reside.

- (5c) (a) Any person who is permitted but fails under sub. (5) (a) to demonstrate to the department that he or she has been rehabilitated may appeal to the secretary or his or her designee. Any person who is adversely affected by a decision of the secretary or his or her designee under this paragraph has a right to a contested case hearing under ch. 227.
- (b) Any person who is permitted but fails under sub. (5) (a) to demonstrate to a county department or an agency contracted with under s. 48.651 (2) that he or she has been rehabilitated may appeal to the director of the county department or his or her designee. Any person who is adversely affected by a decision of the director or his or her designee under this paragraph has a right to appeal the decision under ch. 68.
- (c) Any person who is permitted but fails under sub. (5) (a) to demonstrate to a school board that he or she has been rehabilitated may appeal to the state superintendent of public instruction or his or her designee. Any person who is adversely affected by a decision of the state superintendent or his or her designee under this paragraph has a right to a contested case hearing under ch. 227.
- (**5d**) (a) Any Indian tribe that chooses to conduct rehabilitation reviews under sub. (5) shall submit to the department a rehabilitation review plan that includes all of the following:
- 1. The criteria to be used to determine if a person has been rehabilitated.
- 2. The title of the person or body designated by the Indian tribe to whom a request for review must be made.
- 3. The title of the person or body designated by the Indian tribe to determine whether a person has been rehabilitated.
- 3m. The title of the person or body, designated by the Indian tribe, to whom a person may appeal an adverse decision made by the person specified under subd. 3. and whether the Indian tribe provides any further rights to appeal.
- 4. The manner in which the Indian tribe will submit information relating to a rehabilitation review to the department so that the department may include that information in its report to the legislature required under sub. (5g).
- 5. A copy of the form to be used to request a review and a copy of the form on which a written decision is to be made regarding whether a person has demonstrated rehabilitation.
- (b) If, within 90 days after receiving the plan, the department does not disapprove the plan, the plan shall be considered approved. If, within 90 days after receiv-

ing the plan, the department disapproves the plan, the department shall provide notice of that disapproval to the Indian tribe in writing, together with the reasons for the disapproval. The department may not disapprove a plan unless the department finds that the plan is not rationally related to the protection of clients. If the department disapproves the plan, the Indian tribe may, within 30 days after receiving notice of the disapproval, request that the secretary review the department's decision. A final decision under this paragraph is not subject to further review under ch. 227.

- (5g) On January 1 of each year, the department shall submit a report to the legislature under s. 13.172 (2) that specifies the number of persons in the previous year who have requested to demonstrate that they have been rehabilitated under sub. (5) (a), the number of persons who successfully demonstrated that they have been rehabilitated under sub. (5) (a), and the reasons for the success or failure of a person who has attempted to demonstrate that he or she has been rehabilitated.
- (5m) Notwithstanding s. 111.335, the department may refuse to license a person to operate a child care center, the department in a county having a population of 750,000 or more, a county department, or an agency contracted with under s. 48.651 (2) may refuse to certify a child care provider under s. 48.651, a school board may refuse to contract with a person under s. 120.13 (14), and a child care program may refuse to employ or contract with a caregiver or permit a nonclient resident to reside at the child care program if the person has been convicted of or adjudicated delinquent on or after his or her 10th birthday for an offense that is not a serious crime, but that is, in the estimation of the department, substantially related to the care of a client. The department shall notify the provider and the individual of the results of a substantially related determination pursuant to the process set forth in sub. (4p) for criminal background check determinations. The individual shall have the same appeal rights as set forth in sub. (4s), and the same appeal procedures apply.
- (7) The department shall conduct throughout the state periodic training sessions that cover procedures and uses of criminal background investigations; reporting and investigating misappropriation of property or abuse or neglect of a client; and any other material that will better enable entities to comply with the requirements of this section.
- (8) The department may promulgate any rules necessary for the administration of this section.

SECTION 832. 48.715 (4g) (a) of the statutes is amended to read:

48.715 (**4g**) (a) If a person who has been issued a license under s. 48.66 (1) (a) or a probationary license under s. 48.69 to operate a child care center is convicted of a serious crime, as defined in s. 48.685 <u>48.686</u> (1) (c) 3m., or, if a caregiver specified in s. 48.685 <u>48.686</u> (1)

(ag) 1. -a. or a nonclient resident, as defined in s. 48.685 48.686 (1) (bm), of the child care center is convicted or adjudicated delinquent for committing a serious crime on or after his or her 12th 10th birthday, or if the results of a criminal background check conducted under s. 48.686 indicate that the person, caregiver, or nonclient resident is not eligible to be licensed, certified, or employed or to reside at a child care program, the department shall revoke the license of the child care center immediately upon providing written notice of revocation and the grounds for revocation and an explanation of the process for appealing the revocation.

SECTION 833. 48.715 (4g) (b) of the statutes is amended to read:

48.715 (4g) (b) If a person who has been issued a license under s. 48.66 (1) (a) or a probationary license under s. 48.69 to operate a child care center is the subject of a pending criminal charge alleging that the person has committed a serious crime, as defined in s. 48.685 48.686 (1) (c) 3m., or if a caregiver specified in s. 48.685 48.686 (1) (ag) 1. -a. or a nonclient resident, as defined in s. 48.685 48.686 (1) (bm), of the child care center is the subject of a pending criminal charge or delinquency petition alleging that the person has committed a serious crime on or after his or her 12th 10th birthday, the department shall immediately suspend the license of the child care center until the department obtains information regarding the final disposition of the charge or delinquency petition indicating that the person is not ineligible to be licensed to operate a child care center.

SECTION 834. 48.73 of the statutes is amended to read:

48.73 Inspection of licensees and school district child care programs. The department may visit and inspect each child welfare agency, foster home, group home, and child care center licensed by the department, and for that purpose shall be given unrestricted access to the premises described in the license. The department may visit and inspect each child care program established or contracted for under s. 120.13 (14) that receives payment under s. 49.155 for the child care provided, and for that purpose shall be given unrestricted access to the premises used for the child care program.

SECTION 835a. 48.981 (3) (c) 5r. of the statutes is amended to read:

48.981 (3) (c) 5r. Within 15 days after a final determination is made under subd. 5m. that a specific person has abused or neglected a child or, if a contested case hearing is held on such a determination, within 15 days after a final decision is made under subd. 5p. determining that a specific person has abused or neglected a child, the county department or, in a county having a population of 750,000 or more, the department or a licensed child welfare agency under contract with the department shall provide the subunit of the department that administers s₋ ss. 48.685 and 48.686 with information about the person

who has been determined to have abused or neglected the child.

SECTION 835t. 48.981 (3) (cr) of the statutes is created to read:

48.981 (3) (cr) Contracts to perform child protective services. With the approval of the department, a county department may contract with one or more county departments or the department in a county having a population of 750,000 or more under s. 66.0301 to fulfill the county department's duties under this subsection and sub. (3m).

SECTION 836. 48.981 (7) (cp) of the statutes is amended to read:

48.981 (7) (cp) Notwithstanding par. (a), an agency may disclose a determination made before January 1, 2015, that a person has abused or neglected a child for purposes of a background check under s. 48.685, 48.686, or 50.065 only if that determination has not been reversed or modified on appeal and may disclose such a determination made on or after January 1, 2015, for those purposes only as provided in sub. (3) (c) 5r. Nothing in this paragraph prevents the disclosure of a report or record as otherwise permitted under this subsection.

SECTION 837. 49.133 (1m) (a) of the statutes is amended to read:

49.133 (1m) (a) If a child care provider is convicted of a serious crime, as defined in s. 48.685 48.686 (1) (c) 3m., or if a caregiver specified in s. 48.685 48.686 (1) (ag) 1. -a. or a nonclient resident, as defined in s. 48.685 48.686 (1) (bm), of the child care provider is convicted or adjudicated delinquent for committing a serious crime on or after his or her 12th 10th birthday or if the department provides written notice under s. 48.686 (4p) that the child care provider, caregiver, or nonclient resident is ineligible for certification, employment, or residence at the child care provider, the department or a county department under s. 46.215, 46.22, or 46.23 shall refuse to pay the child care provider for any child care provided under s. 49.132, 1995 stats., or any other program beginning on the date of the conviction or delinquency adjudication.

SECTION 838. 49.133 (1m) (b) of the statutes is amended to read:

49.133 (1m) (b) If a child care provider is the subject of a pending criminal charge alleging that the person has committed a serious crime, as defined in s. 48.685 48.686 (1) (c) 3m., or if a caregiver specified in s. 48.685 48.686 (1) (ag) 1. –a. or a nonclient resident, as defined in s. 48.685 48.686 (1) (bm), of the child care provider is the subject of a pending criminal charge or delinquency petition alleging that the person has committed a serious crime on or after his or her 12th 10th birthday, the department or county department under s. 46.215, 46.22, or 46.23 shall immediately suspend payment to the child care provider for any child care provided under s. 49.132, 1995 stats., or any other program until the department obtains information regarding the final disposition of the

charge or delinquency petition indicating that the person is not ineligible to receive such a payment.

SECTION 839. 49.133 (2m) (intro.) of the statutes is amended to read:

49.133 (**2m**) (intro.) The department or a county department under s. 46.215, 46.22, or 46.23 may refuse to pay a child care provider for child care provided under s. 49.132, 1995 stats., or any other program if any of the following applies to the child care provider or to a caregiver specified in s. 48.685 48.686 (1) (ag) 1. -a. or nonclient resident, as defined in s. 48.685 48.686 (1) (bm), of the child care provider:

SECTION 840. 49.133 (2m) (a) of the statutes is amended to read:

49.133 (**2m**) (a) The person has been convicted of or adjudicated delinquent on or after his or her 12th 10th birthday for an offense that is not a serious crime, as defined in s. 48.685 48.686 (1) (c) 3m., but the department, county department under s. 46.215, 46.22, or 46.23, agency contracted with under s. 48.651 (2), or school board determines under s. 48.685 48.686 (5m) that the offense substantially relates to the care of children or the department or county department determines that the offense substantially relates to the operation of a business.

SECTION 841. 49.133 (2m) (b) of the statutes is amended to read:

49.133 (2m) (b) The person is a caregiver specified in s. 48.685 48.686 (1) (ag) 1. -a. or a nonclient resident, as defined in s. 48.685 48.686 (1) (bm), and is the subject of a pending criminal charge that the department, county department under s. 46.215, 46.22, or 46.23, agency contracted with under s. 48.651 (2), or school board determines substantially relates to the care of children.

SECTION 842. 49.137 (2) (a) of the statutes is amended to read:

49.137 (2) (a) From the allocation under s. 49.155 (1g), the department may award grants to child care providers that meet the quality of care standards established under s. 49.155 (1d) (b) to improve the retention of skilled and experienced child care staff. In awarding grants under this subsection, the department shall consider the applying child care provider's total enrollment of children and average enrollment of children who receive or are eligible for publicly funded care from the child care provider.

SECTION 843. 49.137 (3) (a) of the statutes is amended to read:

49.137 (3) (a) From the allocation under s. 49.155 (1g), the department may award grants to child care providers for assistance in meeting the quality of care standards established under s. 49.155 (1d) (b).

SECTION 844n. 49.1385 of the statutes is created to read:

49.1385 Grants for services for homeless and runaway youth. The department may award not more than

\$100,000 in each fiscal year in grants to support programs that provide services for homeless and runaway youth.

SECTION 845. 49.155 (1) (am) of the statutes is repealed.

SECTION 846. 49.155 (1) (b) of the statutes is repealed.

SECTION 847. 49.155 (1) (bm) of the statutes is created to read:

49.155 (1) (bm) "Liquid assets" means an individual's financial resources that are cash or can be quickly converted to cash without incurring penalties, including cash on hand, as well as funds in checking, savings, money market, and credit union share accounts. "Liquid assets" does not include any financial resources designated by the department by rule as excluded for purposes of sub. (1m) (cm).

SECTION 848. 49.155 (1) (cm) of the statutes is created to read:

49.155 (1) (cm) "Temporary break" means an individual's time—limited absence from an authorized activity due to illness, leave to care for an individual's family member, a student or holiday break, an interruption in work for a seasonal worker who is not working between regular industry work seasons, or any other cessation of an authorized activity as long as the individual continues to be employed or enrolled in the authorized activity and the absence does not exceed 3 months.

SECTION 849. 49.155 (1d) (title) of the statutes is amended to read:

49.155 (**1d**) (title) CHILD CARE CERTIFICATION RULES QUALITY OF CARE STANDARDS.

SECTION 850. 49.155 (1d) (a) (intro.) of the statutes is renumbered 48.651 (1d) (a) and amended to read:

48.651 (1d) (a) The department shall promulgate rules establishing standards for the certification of child care providers under s. 48.651 sub. (1). The department shall consult with the child abuse and neglect prevention board before promulgating those rules. In establishing the requirements for certification under this paragraph for certification of a child care provider, the department shall include a requirement that all providers and all employees and volunteers of a provider who provide care and supervision for children receive, before the date on which the provider is certified or the employment or volunteer work commences, whichever is applicable, all of the following: the minimum health and safety training required under par. (b).

SECTION 851. 49.155 (1d) (a) 1. of the statutes is repealed.

SECTION 852. 49.155 (1d) (a) 2. of the statutes is repealed.

SECTION 853. 49.155 (1d) (am) of the statutes is repealed.

SECTION 854. 49.155 (1d) (b) of the statutes is renumbered 49.155 (1d).

SECTION 855. 49.155 (1m) (intro.) of the statutes is amended to read:

49.155 (1m) ELIGIBILITY. (intro.) Except as provided in s. 49.155 sub. (3g), the department shall determine, contract with a county department or agency to determine, or contract with a county department or agency to share determination of the eligibility of individuals residing in a particular geographic region or who are members of a particular Indian tribal unit for child care subsidies under this section. Under this section, and subject to sub. (2), an individual may receive a subsidy for child care for a child who has not attained the age of 13 or, if the child is disabled, who has not attained the age of 19, if the individual meets all of the following conditions:

SECTION 856. 49.155 (1m) (a) (intro.) of the statutes is amended to read:

49.155 (1m) (a) (intro.) The Subject to sub. (2), the individual is a parent of a child who meets the requirement under s. 49.145 (2) (c) and who is under the age of 13 or, if the child is disabled, is under the age of 19; or is a relative who, under s. 48.57 (3m) or (3n) or 48.62, is providing care and maintenance for a child who meets the requirement under s. 49.145 (2) (c) and who is under the age of 13 or, if the child is disabled, is under the age of 19; and child care services for that child are needed in order for the individual to do participate in an approved activity. An individual who is eligible to receive a child care subsidy under this subsection shall remain eligible for that subsidy for a period of 3 months after the individual permanently ceases participation in the approved activity or until the department or the county department or agency redetermines the individual's eligibility, whichever is earlier. In this paragraph, "approved activity" means any of the following:

SECTION 857. 49.155 (1m) (a) 1. of the statutes is amended to read:

49.155 (**1m**) (a) 1. Meet Meeting the school attendance requirement under s. 49.26 (1) (ge).

SECTION 858. 49.155 (1m) (a) 1m. (intro.) of the statutes is amended to read:

49.155 (1m) (a) 1m. (intro.) Obtain Obtaining a high school diploma or participate participating in a course of study meeting the standards established by the state superintendent of public instruction for the granting of a declaration of equivalency of high school graduation, if the individual is not subject to the school attendance requirement under s. 49.26 (1) (ge) and at least one of the following conditions is met:

SECTION 859. 49.155 (1m) (a) 2. of the statutes is amended to read:

49.155 (**1m**) (a) 2. Work Working in an unsubsidized job, including training provided by an employer during the regular hours of employment.

SECTION 860. 49.155 (1m) (a) 3. of the statutes is amended to read:

49.155 (**1m**) (a) 3. Work Working in a Wisconsin works employment position, including participation in job search, orientation, and training activities under s. 49.147 (2) (a) and in education or training activities under s. 49.147 (3) (am), (4) (am), or (5) (bm).

SECTION 861. 49.155 (1m) (a) 3m. of the statutes is amended to read:

49.155 (**1m**) (a) 3m. Participate Participating in a job search or work experience component of the food stamp employment and training program under s. 49.79 (9).

SECTION 862. 49.155 (1m) (a) 3r. of the statutes is amended to read:

49.155 (**1m**) (a) 3r. Participate Participating in the Transform Milwaukee Jobs program, or the Transitional Jobs program, under s. 49.163.

SECTION 863. 49.155 (1m) (a) 4. of the statutes is amended to read:

49.155 (1m) (a) 4. Participate Participating in basic education, including an English as a 2nd language course; literacy tutoring; or a course of study meeting the standards established by the state superintendent of public instruction under s. 115.29 (4) for the granting of a declaration of equivalency of high school graduation, if the department or the county department or agency determining eligibility determines that basic education would facilitate the individual's efforts to maintain employment. An individual may receive aid under this subdivision for up to 2 years.

SECTION 864. 49.155 (1m) (a) 5. of the statutes is amended to read:

49.155 (**1m**) (a) 5. Participate Participating in a course of study at a technical college, or participate participating in educational courses that provide an employment skill, as determined by the department, if the department or the county department or agency determining eligibility determines that the course or courses would facilitate the individual's efforts to maintain employment. An individual may receive aid under this subdivision for up to 2 years.

SECTION 865. 49.155 (1m) (a) 6. of the statutes is created to read:

49.155 (**1m**) (a) 6. Taking a temporary break from an authorized activity specified in subds. 1. to 5.

SECTION 866. 49.155 (1m) (br) of the statutes is created to read:

49.155 (**1m**) (br) The child is immunized as required under s. 252.04. Notwithstanding s. 252.04 (3), for purposes of this paragraph the immunization requirement may only be waived for reasons of health or religion.

SECTION 867. 49.155 (1m) (c) 1. (intro.) of the statutes is amended to read:

49.155 (**1m**) (c) 1. (intro.) Except as provided in subds. <u>1d.</u>, 1g., 1h., 1m., 2., and 3., the gross income of the individual's family is at or below 185 percent of the poverty line for a family the size of the individual's

family or, for an individual who is already receiving a child care subsidy under this section, the gross income of the individual's family is at or below 200 percent of the poverty line for a family the size of the individual's family. In calculating the gross income of the family, the department or county department or agency determining eligibility shall include court–ordered child or family support payments received by the individual, if those support payments exceed \$1,250 per month, and income described under s. 49.145 (3) (b) 1. and 3., except that, in calculating farm and self–employment income, the department or county department or agency determining eligibility shall include the sum of the following:

SECTION 868. 49.155 (1m) (c) 1d. a. of the statutes is created to read:

49.155 (1m) (c) 1d. a. Notwithstanding sub. (5) (b), if the individual is already receiving a child care subsidy under this section and the gross income of the individual's family exceeds 200 percent of the poverty line for a family the size of the individual's family, the individual's copayment amount under sub. (5) increases by \$1 for every \$3 by which the individual's family's gross income exceeds 200 percent of the poverty line for a family the size of the individual's family.

SECTION 868b. 49.155 (1m) (c) 1d. b. of the statutes is created to read:

49.155 (**1m**) (c) 1d. b. Notwithstanding subd. 1d. a., if the gross income of an individual's family exceeds 85 percent of the state median income for a family the size of the individual's family, the individual is not eligible to receive a child care subsidy under this section.

SECTION 869. 49.155 (1m) (cm) of the statutes is created to read:

49.155 (**1m**) (cm) The total liquid assets of the individual's family do not exceed \$25,000. This paragraph does not apply if the individual is any of the following:

- 1. A foster parent of the child.
- 2. A subsidized guardian or interim caretaker of the child under s. 48.623.
- 3. A relative of the child who is providing care for the child under a court order and receiving payments under s. 48.57 (3m) or (3n) on behalf of the child.

SECTION 870. 49.155 (2) of the statutes is created to read:

49.155 (2) ELIGIBILITY BASED ON THE CHILD'S AGE. Notwithstanding sub. (1m) (intro.) and (a) (intro.), an individual does not lose eligibility for a child care subsidy for a child who attains the age of 13 or, if the child is disabled, attains the age of 19 until the department or the county department or agency redetermines the individual's eligibility.

SECTION 871. 49.155 (4) (a) of the statutes is amended to read:

49.155 (4) (a) An eligible individual shall choose whether the child care will be provided by a child care center licensed under s. 48.65, a Level I certified family

child care provider <u>certified under s. 48.651 (1) (a)</u>, a Level II certified family child care provider <u>certified under s. 48.651 (1) (b)</u>, or a child care program provided or contracted for by a school board under s. 120.13 (14).

SECTION 872. 49.155 (6) (b) of the statutes is amended to read:

49.155 (6) (b) The department shall set maximum payment rates for Level I certified family child care providers certified under s. 48.651 (1) (a) for services provided to eligible individuals under this section. The maximum rates set under this paragraph may not exceed 75 percent of the rates established under par. (a).

SECTION 873. 49.155 (6) (d) of the statutes is amended to read:

49.155 (6) (d) The department may promulgate rules to establish a system of rates or a program of grants for child care providers that meet the higher quality of care standards established by rules promulgated under sub. (1d) (b). If a system of rates is established under this paragraph, the rates under that system shall be higher than the rates established under pars. (a) to (c).

SECTION 874. 49.155 (6g) (a) 5. of the statutes is created to read:

49.155 (**6g**) (a) 5. The department shall take into consideration child learning and development and shall promote continuity of care when authorizing hours of child care. The department is not required to limit authorized hours based on the individual's schedule of activities under sub. (1m) (a) or the number of hours the individual spends in those activities.

SECTION 875. 49.155 (6g) (b) 4. of the statutes is created to read:

49.155 (**6g**) (b) 4. Any reduction in hours due to a temporary break from an authorized activity.

SECTION 876. 49.155 (7) (a) 1. of the statutes is amended to read:

49.155 (7) (a) 1. If a child care provider is convicted of a serious crime, as defined in s. 48.685 48.686 (1) (c) 3m., or if a caregiver specified in s. 48.685 48.686 (1) (ag) 1. a. or a nonclient resident, as defined in s. 48.685 48.686 (1) (bm), of the child care provider is convicted or adjudicated delinquent for committing a serious crime on or after his or her 12th 10th birthday or if the department provides written notice under s. 48.686 (4p) that the child care provider, caregiver, or nonclient resident is ineligible for certification, employment, or residence at the child care provider, the department or the county department under s. 46.215, 46.22, or 46.23 shall refuse to allow payment to the child care provider for any child care provided under this section beginning on the date of the conviction or delinquency adjudication.

SECTION 877. 49.155 (7) (a) 2. of the statutes is amended to read:

49.155 (7) (a) 2. If a child care provider is the subject of a pending criminal charge alleging that the person has committed a serious crime, as defined in s. 48.685 48.686

(1) (c) 3m., or if a caregiver specified in s. 48.685 48.686 (1) (ag) 1. -a. or a nonclient resident, as defined in s. 48.685 48.686 (1) (bm), of the child care provider is the subject of a pending criminal charge or delinquency petition alleging that the person has committed a serious crime on or after his or her 12th 10th birthday, the department or the county department under s. 46.215, 46.22, or 46.23 shall immediately suspend payment to the child care provider for any child care provided under this section until the department obtains information regarding the final disposition of the charge or delinquency petition indicating that the person is not ineligible to receive such a payment.

SECTION 878. 49.155 (7) (b) (intro.) of the statutes is amended to read:

49.155 (7) (b) (intro.) The department or the county department under s. 46.215, 46.22, or 46.23 may refuse to allow payment to a child care provider for child care provided under this section if any of the following applies to the child care provider or to a caregiver specified in s. 48.685 48.686 (1) (ag) 1. a. or nonclient resident, as defined in s. 48.685 48.686 (1) (bm), of the child care provider:

SECTION 879. 49.155 (7) (b) 1. of the statutes is amended to read:

49.155 (7) (b) 1. The person has been convicted of or adjudicated delinquent on or after his or her 12th 10th birthday for committing an offense that is not a serious crime, as defined in s. 48.685 48.686 (1) (c) 3m., but the department, county department, agency contracted with under s. 48.651 (2), or school board determines under s. 48.685 48.686 (5m) that the offense substantially relates to the care of children or the department or county department determines that the offense substantially relates to the operation of a business.

SECTION 880. 49.155 (7) (b) 2. of the statutes is amended to read:

49.155 (7) (b) 2. The person is a caregiver specified in s. 48.685 48.686 (1) (ag) 1. -a. or a nonclient resident, as defined in s. 48.685 48.686 (1) (bm), and is the subject of a pending criminal charge that the department, county department, agency contracted with under s. 48.651 (2), or school board determines substantially relates to the care of children.

SECTION 881. 49.162 (1) (bg) of the statutes is created to read:

49.162 (1) (bg) "Controlled substance abuse screening" means a questionnaire, a criminal background check, or any other controlled substance abuse screening mechanism identified by the department by rule.

SECTION 882. 49.162 (1) (bm) of the statutes is created to read:

49.162 (1) (bm) "Dependent child" has the meaning given in s. 49.141 (1) (c).

SECTION 883. 49.162 (1) (br) of the statutes is created to read:

49.162 (1) (br) "Group member" means an adult member of an individual's Wisconsin Works group whose income or assets are included in determining the individual's eligibility for a program.

SECTION 884. 49.162 (1) (c) 4. of the statutes is created to read:

49.162 (1) (c) 4. A Wisconsin Works employment position.

SECTION 885. 49.162 (1) (e) of the statutes is created to read:

49.162 (1) (e) "Wisconsin Works" has the meaning given in s. 49.141 (1) (p).

SECTION 886. 49.162 (1) (f) of the statutes is created to read:

49.162 (1) (f) "Wisconsin Works employment position" has the meaning given in s. 49.141 (1) (r).

SECTION 887. 49.162 (1) (g) of the statutes is created to read:

49.162 (1) (g) "Wisconsin Works group" has the meaning given in s. 49.141 (1) (s).

SECTION 888. 49.162 (2) of the statutes is renumbered 49.162 (2) (a) and amended to read:

49.162 (2) (a) Beginning on the effective date of the rules promulgated under sub. (7), or on the effective date of the emergency rules promulgated under 2015 Wisconsin Act 55, section 9106 (2c), whichever is earlier, Except as provided in sub. (2m), in order to participate in a program, an individual who applies to participate in a program or who registers for a program under sub. (1) (c) 3.7and, with respect to an individual applying for a program under sub. (1) (c) 4., all of the individual's group members shall complete a controlled substance abuse screening questionnaire. If, on the basis of answers to the questionnaire the screening results, the administering agency determines that there is a reasonable suspicion that an individual who is otherwise eligible for a program or any of the individual's group members is abusing a controlled substance, the administering agency shall require the individual or group member to undergo a test for the use of a controlled substance. If

(b) Except as provided in sub. (4m), if the individual or group member refuses to submit to a test under par. (a), the individual is not eligible to participate in a program until the individual or group member complies with the requirement to undergo a test for the use of a controlled substance.

SECTION 889. 49.162 (2m) of the statutes is created to read:

49.162 (2m) (a) The screening and testing requirements under sub. (2) do not apply to an individual if the individual is any of the following:

- 1. A custodial parent of a child who is 8 weeks old or less.
- 2. A woman who is in a pregnancy that is medically verified and that is shown by medical documentation to be at risk.

- 3. A participant in a Wisconsin Works employment position who moves to an unsubsidized employment position and receives case management services under s. 49.1475.
 - 4. A dependent child.
- (b) The screening and testing requirements under sub. (2) do not apply to a group member if the group member is any of the following:
- 1. A custodial parent of a child who is 8 weeks old or less.
- 2. A woman who is in a pregnancy that is medically verified and that is shown by medical documentation to be at risk.
- 3. Specified as exempt from the screening and testing requirements by department rule.

SECTION 890. 49.162 (3) of the statutes is amended to read:

49.162 (3) If an individual <u>or group member</u> who undergoes a test under sub. (2) tests negative for the use of a controlled substance, or tests positive for the use of a controlled substance but presents evidence satisfactory to the administering agency that the individual <u>or group member</u> possesses a valid prescription for each controlled substance for which the individual <u>or group member</u> tests positive, the individual <u>or group member</u> will have satisfactorily completed the substance abuse testing requirements under this section.

SECTION 891. 49.162 (4) (a) of the statutes is amended to read:

49.162 (4) (a) If an individual <u>or group member</u> who undergoes a test under sub. (2) tests positive for the use of a controlled substance without presenting evidence of a valid prescription as described in sub. (3), the administering agency shall require the individual <u>or group member</u> to participate in substance abuse treatment to remain eligible to participate in a program. If the individual <u>or group member</u> refuses to participate in substance abuse treatment, the individual is not eligible to participate in a program until the individual <u>or group member</u> complies with the requirement to participate in substance abuse treatment.

SECTION 892. 49.162 (4) (b) of the statutes is amended to read:

49.162 (4) (b) During the time that an individual or group member is receiving substance abuse treatment under par. (a), the administering agency shall require the individual or group member to undergo random testing for the use of a controlled substance. For Except as provided in sub. (4m), for the individual to remain eligible for a program, the individual or his or her group member must cooperate with the testing and the results of the tests must be negative or, if any results are positive, the individual or group member must present evidence of a valid prescription as described in sub. (3). If the results of any test during treatment are positive for the use of a controlled substance and the individual or group member

does not present evidence of a valid prescription for the controlled substance, the individual or group member shall have the opportunity to begin the treatment again one time, as determined by the administering agency. If Except as provided in sub. (4m), if the individual or group member begins the substance abuse treatment again, he or she shall remain the individual remains eligible for a program as long as the results of all tests for the use of a controlled substance during the subsequent treatment are negative for the use of a controlled substance or, if any results are positive, the individual or group member presents evidence of a valid prescription for the controlled substance.

SECTION 893. 49.162 (4) (c) of the statutes is amended to read:

49.162 (4) (c) If an individual <u>or group member</u> receiving treatment under par. (b) completes treatment and, at the conclusion of the treatment, tests negative for the use of a controlled substance or presents evidence of a valid prescription for any controlled substance for which the individual <u>or group member</u> tests positive, the individual <u>or group member</u> will have satisfactorily completed the substance abuse testing requirements under this section.

SECTION 894. 49.162 (4m) of the statutes is created to read:

- 49.162 (4m) (a) If an individual applying for a community service job under s. 49.147 (4) or a transitional placement under s. 49.147 (5) or his or her group member undergoes a test under sub. (2), tests positive for the use of a controlled substance without presenting evidence of a valid prescription as described in sub. (3), and refuses to participate in substance abuse treatment under sub. (4) (a) or if the individual or his or her group member fails to cooperate with the testing or treatment requirements under sub. (4) (b), the individual remains eligible only for the monthly grant portion of the community service job or transitional placement under s. 49.148 (1) (b) or (c) and only to the extent described in par. (b).
- (b) 1. In determining the monthly grant for which an individual is eligible under par. (a), the department shall reduce the amount that would otherwise have been established under s. 49.148 (1) (b) or (c) by an amount that reflects the fact that the monthly grant is to be used exclusively for the benefit of the dependent children in the individual's Wisconsin Works group and not for the benefit of the individual.
- 2. If an individual is eligible for a monthly grant under the circumstances described in par. (a), the department shall pay the monthly grant through a protective payee structure, under which the monthly grant is paid to a protective payee who is not the individual and who holds the money and uses it exclusively for the benefit of the dependent children in the individual's Wisconsin Works group.

- 3. An individual's partial eligibility under par. (a) ends on the earlier of the following dates:
- a. The date on which the individual again becomes eligible for full participation in a Wisconsin Works employment position.
- b. Twelve months after the date on which the individual or his or her group member meets the circumstances described under par. (a), as determined by the department.

SECTION 895. 49.162 (7) of the statutes is amended to read:

49.162 (7) The department shall promulgate rules to implement the substance abuse screening, testing, and treatment requirements under this section <u>and the monthly grant eligibility and protective payee structure under sub. (4m).</u>

SECTION 895g. 49.1635 (4) of the statutes is amended to read:

49.1635 (4) Not more than 10 percent of the total funds received by the Wisconsin Trust Account Foundation <u>under sub. (1)</u> may be used for administration.

SECTION 895m. 49.1635 (5) of the statutes is created to read:

- 49.1635 (5) (a) From the allocation under s. 49.175 (1) (j), the department shall make a grant of \$500,000 in each fiscal year to Wisconsin Trust Account Foundation, Inc., for distribution of annual awards of not more than \$75,000 per year per program to programs that provide legal services to persons who are eligible under par. (b) 2. if all of the following apply:
- 1. Wisconsin Trust Account Foundation, Inc., submits a plan to the department detailing the proposed use of the grant; the proposed use of the grant conforms to the requirements under par. (b); and the secretary of the department, or his or her designee, approves the plan.
- 2. Wisconsin Trust Account Foundation, Inc., enters into an agreement with the department that specifies the conditions for the use of the grant proceeds, and the conditions conform to the requirements under par. (b) and include training, reporting, and auditing requirements.
- 3. Wisconsin Trust Account Foundation, Inc., agrees in writing to submit to the department the reports required under par. (c) by the times required under par. (c).
- (b) 1. Subject to subd. 3., the grant may be used only to provide legal services in civil matters related to domestic abuse, sexual abuse, or restraining orders or injunctions for individuals at risk under s. 813.123.
- 2. The recipients of the legal services under a grant under this subsection shall be individuals who are eligible for temporary assistance for needy families under 42 USC 601 et seq. and whose gross incomes are at or below 200 percent of the poverty line. For purposes of this subdivision, gross income shall be determined in the same way as gross income is determined for purposes of eligibility for a Wisconsin Works employment position, as

- defined in s. 49.141 (1) (r), including the exclusion of any payments or benefits made under any federal law that exempts those payments or benefits from consideration in determining eligibility for any federal means—tested program.
- 3. The legal services provided by a grant under this subsection shall be provided only in matters for which federal temporary assistance for needy families block grant funds under 42 USC 601 et seq. may be used.
- 4. The grant proceeds may not be used for legal services for litigation against the state.
- (c) For each fiscal year in which the department makes a grant under this subsection, Wisconsin Trust Account Foundation, Inc., shall submit to the department, within 3 months after spending the full amount of that grant, a report detailing how the grant proceeds were used. The department may not make a grant in a subsequent fiscal year unless Wisconsin Trust Account Foundation, Inc., submits the report under this paragraph within the time required and the department determines that the grant proceeds were used in accordance with the approved plan under par. (a) 1., the agreement under par. (a) 2., and the requirements under par. (b).

SECTION 896. 49.175 (1) (a) of the statutes is amended to read:

49.175 (1) (a) *Wisconsin Works benefits*. For Wisconsin Works benefits, \$83,000,000 \$42,500,000 in fiscal year 2015–16 2017–18 and \$83,000,000 \$44,625,000 in fiscal year 2016–17 2018–19.

SECTION 897. 49.175 (1) (b) of the statutes is amended to read:

49.175 (1) (b) Wisconsin Works agency contracts; job access loans. For contracts with Wisconsin Works agencies under s. 49.143 and for job access loans under s. 49.147 (6), \$58,336,500 in each fiscal year \$52,000,000 in fiscal year 2017–18 and \$54,600,000 in fiscal year 2018–19.

SECTION 898. 49.175 (1) (c) of the statutes is created to read:

49.175 (1) (c) Case management incentive payments. For supplement payments to individuals under s. 49.255, \$2,700,000 in fiscal year 2017–18 and \$2,700,000 in fiscal year 2018–19.

SECTION 899. 49.175 (1) (d) of the statutes is created to read:

49.175 (1) (d) Families and Schools Together. For the families and schools together program in 5 Milwaukee elementary schools to be chosen by the department, \$250,000 in each fiscal year.

SECTION 901. 49.175 (1) (f) of the statutes is created to read:

49.175 (1) (f) Homeless case management services grants. For grants to shelter facilities under s. 16.3085, \$500,000 in each fiscal year. All moneys allocated under this paragraph shall be credited to the appropriation account under s. 20.505 (7) (kg).

SECTION 902. 49.175 (1) (g) of the statutes is amended to read:

49.175 (1) (g) State administration of public assistance programs and overpayment collections. For state administration of public assistance programs and the collection of public assistance overpayments, \$15,080,200 \$15,987,000 in fiscal year 2015–16 2017–18 and \$15,295,800 \$15,902,900 in fiscal year 2016–17 2018–19.

SECTION 903. 49.175 (1) (i) of the statutes is amended to read:

49.175 (1) (i) *Emergency assistance*. For emergency assistance under s. 49.138 and for transfer to the department of administration for low–income energy or weatherization assistance programs, \$8,500,000 \$7,000,000 in each fiscal year 2015–16 and \$8,400,000 in fiscal year 2016–17.

SECTION 903m. 49.175 (1) (j) of the statutes is amended to read:

49.175 (1) (j) Grants for providing civil legal services. For the grants under 2015 Wisconsin Act 55, section 9106 (2q), s. 49.1635 (5) to Wisconsin Trust Account Foundation, Inc., for distribution to programs that provide civil legal services to low–income families, \$500,000 in each fiscal year.

SECTION 904. 49.175 (1) (k) of the statutes is amended to read:

49.175 (1) (k) *Transform Milwaukee and Transitional Jobs programs*. For contract costs under the Transform Milwaukee Jobs program and the Transitional Jobs program under s. 49.163, \$6,000,000 \$7,000,000 in fiscal year 2015–16 2017–18 and \$7,000,000 \$8,000,000 in fiscal year 2016–17 2018–19.

SECTION 905. 49.175 (1) (n) of the statutes is amended to read:

49.175 (1) (n) Fostering futures: connections count. For funding community connectors to interact with vulnerable families with young children and to connect families with formal and informal community support, \$360,300 in fiscal year 2016–17 2017–18 and \$560,300 in fiscal year 2018–19.

SECTION 906. 49.175 (1) (p) of the statutes is amended to read:

49.175 (1) (p) *Direct child care services*. For direct child care services under s. 49.155, \$262,064,700 \$289,215,200 in fiscal year 2015–16 2017–18 and \$280,719,700 \$310,369,200 in fiscal year 2016–17 2018–19.

SECTION 907. 49.175 (1) (q) of the statutes is amended to read:

49.175 (1) (q) *Child care state administration and licensing activities.* For state administration of child care programs under s. 49.155 and for child care licensing activities, \$35,244,600 \$36,189,400 in fiscal year 2015–16 2017–18 and \$33,248,300 \$36,030,000 in fiscal year 2016–17 2018–19.

SECTION 908. 49.175 (1) (qm) of the statutes is amended to read:

49.175 (1) (qm) *Quality care for quality kids*. For the child care quality improvement activities specified in s. 49.155 (1g), \$15,492,700 \$15,652,700 in each fiscal year.

SECTION 909. 49.175 (1) (r) of the statutes is amended to read:

49.175 (1) (r) *Children of recipients of supplemental security income*. For payments made under s. 49.775 for the support of the dependent children of recipients of supplemental security income, \$31,338,200 \$26,938,000 in each fiscal year.

SECTION 910. 49.175 (1) (s) of the statutes is amended to read:

49.175 (1) (s) *Kinship care and long–term kinship care assistance*. For kinship care and long–term kinship care payments under s. 48.57 (3m) (am) and (3n) (am), for assessments to determine eligibility for those payments, and for agreements under s. 48.57 (3t) with the governing bodies of Indian tribes for the administration of the kinship care and long–term kinship care programs within the boundaries of the reservations of those tribes, \$21,222,700 \$22,012,100 in fiscal year 2015–16 2017–18 and \$21,435,000 \$22,741,200 in fiscal year 2016–17 2018–19.

SECTION 911. 49.175 (1) (t) of the statutes is amended to read:

49.175 (1) (t) Safety and out-of-home placement services. For services provided to ensure the safety of children who the department or a county determines may remain at home if appropriate services are provided, and for services provided to families with children placed in out-of-home care, \$3,647,200 \$6,282,500 in fiscal year 2015-16 2017-18 and \$5,392,700 \$7,314,300 in fiscal year 2016-17 2018-19. To receive funding under this paragraph, a county shall match a percentage of the amount received that is equal to the percentage the county is required to match for a distribution under s. 48.563 (2) as specified by the schedule established by the department under s. 48.569 (1) (d).

SECTION 912. 49.175 (1) (u) of the statutes is amended to read:

49.175 (1) (u) *Prevention services*. For services to prevent child abuse or neglect in counties having a population of 750,000 or more, \$1,389,600, \$5,289,600 in each fiscal year.

SECTION 913. 49.175 (1) (v) of the statutes is amended to read:

49.175 (1) (v) General education development. For general education development testing and preparation for individuals who are eligible for temporary assistance for needy families under 42 USC 601 et seq., \$127,000 \$115,000 in each fiscal year 2015–16 and \$115,000 in fiscal year 2016–17.

SECTION 916. 49.175 (1) (y) of the statutes is created to read:

49.175 (1) (y) Offender reentry demonstration project. For the offender reentry demonstration project under s. 49.37 (1), \$187,500 in fiscal year 2017–18 and \$250,000 in fiscal year 2018–19.

SECTION 918. 49.175 (1) (z) of the statutes is amended to read:

49.175 (1) (z) Grants to the Boys and Girls Clubs of America. For grants to the Wisconsin Chapter of the Boys and Girls Clubs of America to fund programs that improve social, academic, and employment skills of youth who are eligible to receive temporary assistance for needy families under 42 USC 601 et seq., focusing on study habits, intensive tutoring in math and English, and exposure to career options and role models, \$1,175,000 \$1,275,000 in each fiscal year. Grants provided under this paragraph may not be used by the grant recipient to replace funding for programs that are being funded, when the grant proceeds are received, with moneys other than those from the appropriations specified in sub. (1) (intro.). The total amount of the grants includes funds for the Green Bay Boys and Girls Clubs for the BE GREAT: Graduate program in the amount of matching funds that the program provides, up to \$75,000 in each fiscal year, to be used only for activities for which federal Temporary Assistance for Needy Families block grant moneys may be used. The total amount of the grants also includes funds to be equally distributed among the Milwaukee, Oshkosh, and Appleton Boys and Girls Clubs for the BE GREAT: Graduate program in the amount of matching funds that the program provides, up to \$100,000 in each fiscal year, to be used only for activities for which federal Temporary Assistance for Needy Families block grant moneys may be used.

SECTION 919. 49.175 (1) (zh) of the statutes is amended to read:

49.175 (1) (zh) Earned income tax credit supplement. For the transfer of moneys from the appropriation account under s. 20.437 (2) (md) to the appropriation account under s. 20.835 (2) (kf) for the earned income tax credit, \$67,600,000 \$69,700,000 in each fiscal year 2015–16 and \$69,700,000 in fiscal year 2016–17.

SECTION 921. 49.255 of the statutes is created to read: **49.255 Case management incentive payments.** An individual who receives case management services under s. 49.1475 is eligible to receive from the department a supplement of \$50 per month over a period of 12 months if the individual meets the federal work participation requirements under 42 USC 607.

SECTION 922. 49.26 (1) (ge) of the statutes is renumbered 49.26 (1) (ge) 1. (intro.) and amended to read:

49.26 (1) (ge) 1. (intro.) An individual fails to meet the school attendance requirement if the individual <u>meets</u> at least one of the following conditions:

- a. The individual is either not enrolled in school or was not enrolled in the immediately preceding semester or is a habitual truant.
- <u>2.</u> The Wisconsin works Works agency or county department shall verify school enrollment and attendance.

SECTION 923. 49.26 (1) (ge) 1. b. of the statutes is created to read:

49.26 (1) (ge) 1. b. During the immediately preceding semester, the individual was either not enrolled in school or was a habitual truant.

SECTION 923b. 49.34 (4) (c) of the statutes is amended to read:

49.34 (4) (c) Unless waived by the department, biennially, or annually if required under federal law, provide the purchaser with a certified financial and compliance audit report if the care and services purchased exceed \$25,000 \$100,000. The audit shall follow standards that the department prescribes.

SECTION 923c. 49.34 (5m) (b) 1. of the statutes is amended to read:

49.34 (5m) (b) 1. Subject to subds. 2. and 3. and par. (em), if If revenue under a contract for the provision of a rate-based service exceeds allowable costs incurred in the contract period, the contract shall allow the provider may to retain from the surplus generated by that ratebased service up to 5 percent of the contract amount. A provider that retains a surplus under this subdivision shall use that retained surplus to cover a deficit between revenue and allowable costs incurred in any preceding or future contract period for the same rate-based service that generated the surplus or to address the programmatic needs of clients served by the same rate-based service that generated the surplus. This subdivision does not apply to a child welfare agency that is authorized under s. 48.61 (7) to license foster homes, a group home, as defined in s. 48.02 (7), or a residential care center for children and youth, as defined in s. 48.02 (15d) revenue received under the contract unless a uniform rate is established by rule under subd. 5., in which case the contract shall allow the provider to retain the uniform percentage rate established by the rule. The retained surplus is the property of the provider.

SECTION 923d. 49.34 (5m) (b) 2. of the statutes is repealed.

SECTION 923e. 49.34 (5m) (b) 3. of the statutes is repealed.

SECTION 923f. 49.34 (5m) (b) 4. of the statutes is created to read:

49.34 (5m) (b) 4. If on December 31 of any year the provider's accumulated surplus from all contract periods ending during that year for a rate-based service exceeds the allowable retention rate under subd. 1., the provider shall provide written notice of that excess to all purchasers of the rate-based service. Upon the written

request of such a purchaser received no later than 6 months after the date of the notice, the provider shall refund the purchaser's proportional share of that excess. If the department determines based on an audit or fiscal review that the amount of the excess identified by the provider was incorrect, the department may seek to recover funds after the 6-month period has expired. The department shall commence any audit or fiscal review under this subdivision within 6 years after the end of the contract period.

SECTION 923g. 49.34 (5m) (b) 5. of the statutes is created to read:

- 49.34 (5m) (b) 5. The department, in consultation with the department of health services and the department of corrections, shall promulgate rules to implement this subsection including all of the following:
- a. Requiring that contracts for rate—based services under this subsection allow a provider to retain from any surplus revenue up to 5 percent of the total revenue received under the contract, or a different percentage rate determined by the department. The percentage rate established under this subd. 5. a. shall apply uniformly to all rate—based service contracts under this subsection.
- b. Establishing a procedure for reviewing rate—based service contracts to determine whether a contract complies with the provisions of this subsection.

SECTION 923h. 49.34 (5m) (em) of the statutes is amended to read:

49.34 (5m) (em) Notwithstanding par. (b) 1. and 2., a county department under s. 46.215, 51.42, or 51.437 providing client services in a county having a population of 750,000 or more or a nonstock, nonprofit corporation providing client services in such a county may not retain a surplus generated by a rate–based service or accumulate funds from more than one contract period for a rate–based service from revenues that are used to meet the maintenance–of–effort requirement under the federal temporary assistance for needy families program under 42 USC 601 to 619.

SECTION 923i. 49.343 (5) (c) of the statutes is amended to read:

49.343 (5) (c) The identification of the measurements specified in sub. (6) (a) and the development of the payment levels specified in sub. (6) (a).

SECTION 923j. 49.343 (6) (a) (intro.) and 1. of the statutes are consolidated, renumbered 49.343 (6) (a) and amended to read:

49.343 (6) (a) For purposes of implementing a performance—based contracting system, the department, in cooperation with the advisory committee created under sub. (5), shall do all of the following: 1. Identify identify measurements by which to evaluate the performance of providers in meeting both the goals for the children placed in their care and the goals for the out—of—home care system in this state and adjust, as needed, those measurements.

SECTION 923k. 49.343 (6) (a) 2. of the statutes is repealed.

SECTION 923L. 49.343 (6) (b) of the statutes is repealed.

SECTION 923m. 49.343 (6) (c) and (d) of the statutes are amended to read:

- 49.343 (6) (c) Beginning on January 1, 2011, the department shall select a representative sample of providers and evaluate the performance of those providers in attaining the measurements identified under par. (a) 4. Based on that evaluation, the department, in consultation with the advisory committee created under sub. (5), shall adjust, as needed, those measurements by December 31, 2011.
- (d) Beginning on January 1, 2013, the department shall evaluate the performance of all providers in this state in attaining the measurements identified under par. (a) 4. Based on that evaluation, the department, in consultation with the advisory committee created under sub. (5), shall adjust, as needed, those measurements by December 31, 2013, and in subsequent years as determined necessary by the department.

SECTION 924. 49.37 of the statutes is created to read: 49.37 Offender reentry demonstration project.

- (1) Beginning in fiscal year 2017–18, the department of children and families shall establish a 5-year offender reentry demonstration project focused on noncustodial fathers in a 1st class city.
- (2) Upon completion of the demonstration project under sub. (1) and by June 30, 2023, the department of children and families shall conduct an evaluation of the demonstration project.

SECTION 924p. 49.45 (3m) (a) (intro.) and (b) 3. a. of the statutes are amended to read:

49.45 (**3m**) (a) (intro.) Subject to par. (c) and notwith-standing sub. (3) (e), from the appropriations under s. 20.435 (4) (b) and (o), in each fiscal year, the department shall pay to hospitals that serve a disproportionate share of low–income patients an amount equal to the sum of \$15,000,000 \$27,500,000, as the state share of payments, and the matching federal share of payments. The department may make a payment to a hospital under this subsection under the calculation method described in par. (b) if the hospital meets all of the following criteria:

(b) 3. a. No single hospital receives more than \$2,500,000 \$4,600,000.

SECTION 924r. 49.45 (3p) of the statutes is created to read:

49.45 (**3p**) RURAL CRITICAL CARE ACCESS SUPPLEMENT. (a) Subject to par. (c) and notwithstanding sub. (3) (e), from the appropriations under s. 20.435 (4) (b) and (o), in each fiscal year, the department shall pay to hospitals that would meet the criteria under sub. (3m) (a) except that the hospitals do not provide obstetric services an amount equal to the sum of \$250,000, as the state share of payments, and the matching federal share of payments.

The department may make a payment to a hospital under this subsection under a calculation method determined by the department that provides a fee-for-service supplemental payment that increases as the hospital's percentage of inpatient days for Medical Assistance recipients at the hospital increases.

- (b) The department shall ensure that the total amount of moneys available to pay hospitals described under this subsection is distributed in each fiscal year.
- (c) The department shall limit the maximum payment to hospitals under this subsection such that the amount of payment is in accordance with federal rules concerning any hospital specific limit.
- (d) The department shall seek any necessary approval from the federal department of health and human services to implement the hospital payment supplement described under par. (a). If approval is necessary and approval from the federal department of health and human services is received, the department shall implement the payment methodology described under par. (a). If approval is necessary and the federal department of health and human services does not approve, the department may not implement the hospital payment supplement under par. (a).

Vetoed In Part **SECTION 926p.** 49.45 (9r) of the statutes is created to read:

49.45 **(9r)** COMPLEX REHABILITATION TECHNOLOGY. (a) In this subsection:

- 1. "Complex needs patient" means an individual with a diagnosis or medical condition that results in significant physical impairment or functional limitation.
- 2. "Complex rehabilitation technology" means items classified within Medicare as durable medical equipment that are individually configured for individuals to meet their specific and unique medical, physical, and functional needs and capacities for basic activities of daily living and instrumental activities of daily living identified as medically necessary. "Complex rehabilitation technology" includes complex rehabilitation manual and power wheelchairs, adaptive seating and positioning items, and other specialized equipment such as standing frames and gait trainers, as well as options and accessories related to any of these items.
- 3. "Individually configured" means having a combination of sizes, features, adjustments, or modifications that a qualified complex rehabilitation technology supplier can customize to the specific individual by measuring, fitting, programming, adjusting, or adapting as appropriate so that the device operates in accordance with an assessment or evaluation of the individual by a qualified health care professional and is consistent with the individual's medical condition, physical and functional needs and capacities, body size, period of need, and intended use.

- 4. "Medicare" means coverage under Part A or Part Vetoed B of Title XVIII of the federal Social Security Act, 42 In Part USC 1395 et seq.
- "Qualified complex rehabilitation technology professional" means an individual who is certified as an assistive technology professional by the Rehabilitation Engineering and Assistive Technology Society of North America.
- 6. "Qualified complex rehabilitation technology supplier" means a company or entity that meets all of the following criteria:
- a. Is accredited by a recognized accrediting organization as a supplier of complex rehabilitation technology.
- b. Is an enrolled supplier for purposes of Medicare reimbursement that meets the supplier and quality standards established for durable medical equipment suppliers, including those for complex rehabilitation technology under Medicare.
- c. Is an employer of at least one qualified complex rehabilitation technology professional to analyze the needs and capacities of the complex needs patient in consultation with qualified health care professionals, to participate in the selection of appropriate complex rehabilitation technology for those needs and capacities of the complex needs patient, and to provide training in the proper use of the complex rehabilitation technology.
- d. Requires a qualified complex rehabilitation technology professional to be physically present for the evaluation and determination of appropriate complex rehabilitation technology for a complex needs patient.
- e. Has the capability to provide service and repair by qualified technicians for all complex rehabilitation technology it sells.
- f. Provides written information at the time of delivery of the complex rehabilitation technology to the complex needs patient stating how the complex needs patient may receive service and repair for the complex rehabilitation technology.
- 7. "Qualified health care professional" means any of the following:
- a. A physician or physician assistant licensed under subch. II of ch. 448.
- b. A physical therapist licensed under subch. III of ch. 448.
- c. An occupational therapist licensed under subch VII of ch. 448.
 - d. A chiropractor licensed under ch. 446.
- (b) The department shall promulgate rules and other policies for use of complex rehabilitation technology by recipients of Medical Assistance. The department shall include in the rules all of the following:
- 1. Designation of billing codes as complex rehabilitation technology including creation of new billing codes or modification of existing billing codes. The depart-

ment shall include provisions allowing quarterly updates to the designations under this subdivision.

- 2. Establishment of specific supplier standards for companies or entities that provide complex rehabilitation technology and limiting reimbursement only to suppliers that are qualified complex rehabilitation technology suppliers.
- 3. A requirement that Medical Assistance recipients who need a manual wheelchair, power wheelchair, or other seating component to be evaluated by all of the following:
- a. A qualified health care professional who does not have a financial relationship with a qualified complex rehabilitation technology supplier.
- b. A qualified complex rehabilitation technology professional.
- 4. Establishment and maintenance of payment rates for complex rehabilitation technology that are adequate to ensure complex needs patients have access to complex rehabilitation technology, taking into account the significant resources, infrastructure, and staff needed to appropriately provide complex rehabilitation technology to meet the unique needs of complex needs patients.
- 5. A requirement for contracts with the department that managed care plans providing services to Medical Assistance recipients comply with this subsection and the rules promulgated under this subsection.
- 6. Protection of access to complex rehabilitation technology for complex needs patients.
- (c) This subsection is not intended to affect coverage of speech generating devices, including healthcare common procedure coding system codes E2500, E2502, E2504, E2506, E2508, E2510, E2511, E2512, and E2599, under the Medical Assistance program.

SECTION 927. 49.45 (23) (g) 1. f. of the statutes is created to read:

49.45 (23) (g) 1. f. Provide employment and training services to childless adults receiving Medical Assistance under this subsection.

SECTION 928b. 49.45 (23) (g) 2. of the statutes is repealed.

SECTION 928d. 49.45 (23) (g) 3. and 4. of the statutes In Part are created to read:

> 49.45 (23) (g) 3. If the secretary of the federal department of health and human services approves any portion of the waiver amendment requested under subd. 1., the department shall, no later than the first day of the 4th month beginning after that approval, submit to the joint committee on finance a report that includes all of the following:

- a. A description of each component of the waiver amendment that is approved and any pertinent information on the department's plan for implementation.
- b. An estimate of the effect of implementation of the approved portions of the waiver amendment on enrollment in and the budget of the Medical Assistance pro-

gram in the fiscal biennium in which approval occurs and in future fiscal bienniums.

4. The department may not implement any approved portion of the waiver amendment requested under subd. 1. unless the joint committee on finance meets under s. 13.10 and approves the implementation of that portion of the waiver amendment. In a meeting under s. 13.10 to review the report submitted under subd. 3., the joint committee on finance may approve or disapprove of the waiver amendment portions that are approved by the federal department of health and human services or may modify the waiver amendment only by removing one or more components of the waiver amendment. The department may implement the waiver amendment only as approved by the joint committee on finance, including any modifications. The department shall, if necessary to implement the waiver amendment as modified by the joint committee on finance, submit a subsequent waiver amendment request to the federal department of health and human services that is consistent with the committee's actions.

SECTION 928f. 49.45 (24n) of the statutes is created to read:

49.45 (24n) REIMBURSEMENT FOR DENTAL SERVICES BY FACILITIES SERVING INDIVIDUALS WITH DISABILITIES. (a) Subject to approval of the federal department of health and human services under par. (b), the department shall distribute moneys in each fiscal year to increase the Medical Assistance reimbursement rates for all eligible dental services rendered by facilities that provide at least 90 percent of their dental services to individuals with cognitive and physical disabilities, as determined by the department. Under this subsection, the enhanced reimbursement rates for dental services would equal 200 percent of the Medical Assistance reimbursement rates that would otherwise be paid for these dental services.

(b) The department shall request any waiver from and submit any amendments to the state Medical Assistance plan to the federal department of health and human services necessary for the Medical Assistance reimbursement rate increase under par. (a). If any necessary waiver request or state plan amendment request is approved, the department shall implement par. (a) beginning on the effective date of the waiver or plan amendment.

SECTION 928g. 49.45 (26g) of the statutes is created **Vetoed** to read:

In Part

- 49.45 (26g) INTENSIVE CARE COORDINATION PROGRAM. (a) Subject to par. (h), the department shall create and implement a program to reimburse hospitals and health care systems for intensive care coordination services provided to recipients of Medical Assistance under this subchapter who are not enrolled in coverage under Medicare, 42 USC 1395 et seq.
- (b) The department shall select hospitals and health care systems to receive reimbursement under this subsection that submit to the department a description of their

Vetoed

In Part

Vetoed In Part intensive care coordination program that includes all of the following:

- 1. A statement that the hospital or health care system will use emergency department utilization data to identify recipients of Medical Assistance to receive intensive care coordination to reduce use of the emergency department by those Medical Assistance recipients.
- 2. The method the hospital or health care system uses to identify for intensive care coordination a Medical Assistance recipient who uses the emergency department frequently. The hospital or health care system shall specify how it defines frequent emergency department use and may use criteria such as whether a recipient of Medical Assistance visits the emergency room 3 or more times within 30 days, 6 or more times within 90 days, or 7 or more times within 12 months.
- 3. A description of the hospital's or health care system's intensive care coordination team consisting of health care providers other than solely physicians, such as nurses; social workers, case managers, or care coordinators; behavioral health specialists; and schedulers.
- 4. That the hospital or health care system provides to a Medical Assistance recipient enrolled in intensive care coordination through the hospital or health care system all of the following, as appropriate to his or her care:
- a. Discharge instructions and contacts for following up on care and treatment.
 - b. Referral information.
 - c. Appointment scheduling.
 - d. Medication instructions.
- e. Intensive care coordination by a social worker, case manager, or care coordinator to connect the Medical Assistance recipient to a primary care provider or to a managed care organization.
- Information about other health and social resources, such as transportation and housing.
- 5. The outcomes intended to result from intensive care coordination by the hospital or health care system. Outcomes for a Medical Assistance recipient during a 6-month or 12-month period may include successful connection to primary care or the managed care organization as evidenced by 2 or 3 primary care appointments, successful connection to behavioral health resources and alcohol and other drug abuse resources, as needed, or a decrease in use of the emergency room.
 - (c) The department shall do all of the following:
- 1. Respond to the hospital or health care system indicating if additional information is required to determine eligibility for the reimbursement program under this subsection.
- 2. If the hospital or health care system is eligible for the reimbursement program under this subsection, provide a description of the process for enrolling Medical Assistance recipients in intensive care coordination for reimbursement.

- (d) The department shall provide as reimbursement **Vetoed** for intensive care coordination to eligible hospitals and health care systems participating in the program under this subsection \$500 for each Medical Assistance recipient who is not enrolled in coverage under Medicare, 42 USC 1395 et seq., the hospital or health care system enrolls in intensive care coordination. The initial enrollment for each recipient lasts for 6 months, and the health care provider may enroll the Medical Assistance recipient in one additional 6-month period for an additional \$500 reimbursement payment. The department shall pay no more than \$1,500,000 cumulatively in each fiscal year from all funding sources for reimbursements under this paragraph.
- (e) Annually, each hospital and health care system that is eligible for the reimbursement program under this subsection shall submit a report to the department containing all of the following:
- 1. The number of Medical Assistance recipients served by intensive care coordination.
- 2. For each Medical Assistance recipient who is not enrolled in coverage under Medicare, 42 USC 1395 et seq., the number of emergency department visits for a period before enrollment of that recipient in intensive care coordination and the number of emergency department visits for the same recipient during the same period after enrollment in intensive care coordination.
- 3. Any demonstrated outcomes, such as those described in par. (b) 5., for Medical Assistance recipients.
- (f) For each hospital or health care system eligible for the reimbursement program under this subsection, the department shall calculate the costs saved to the Medical Assistance program by avoiding emergency department visits by subtracting the sum of reimbursements made under par. (d) to the hospital or health care system from the sum of costs of visits to the emergency department as reported under par. (e) 2. that were expected to occur without intensive care coordination. If the result of the calculation is positive, the department shall distribute half of the amount saved to the hospital or health care system subject to par. (h).
- (g) No later than 24 months after the date on which the first hospital or health care system is able to enroll individuals in the intensive care coordination program under this subsection, the department shall submit a report to the joint committee on finance summarizing the information reported under par. (e) including the costs saved by avoiding emergency department visits as calculated under par. (f).
- The department shall seek any necessary approval from the federal department of health and human services to implement the program under this subsection. If the federal department of health and human services disapproves the request for approval, the department may implement the reimbursement under par. (d),

In Part

the savings distribution under par. (f), or both or any part of the program under this subsection.

SECTION 928h. 49.45 (29y) of the statutes is created to read:

49.45 (29y) MENTAL HEALTH CONSULTATION REIM-BURSEMENT. (a) In this subsection, "clinical consultation" means, for a student up to age 21, communication from a mental health professional or a qualified treatment trainee working under the supervision of a mental health professional to another individual who is working with the client to inform, inquire, and instruct regarding all of the following and to direct and coordinate clinical service components:

- 1. The client's symptoms.
- 2. Strategies for effective engagement, care, and intervention for the client.
- 3. Treatment expectations for the client across service settings.
- (b) The department shall, subject to any approval necessary from the federal department of health and human services, reimburse clinical consultation from the Medical Assistance program under this subchapter.

Vetoed In Part

- (c) By March 31, 2019, the department shall submit a report to the joint committee on finance on the utilization of the clinical consultation services under this subsection.
- (d) The department may not provide the reimbursement for clinical consultation that occurs after June 30, 2019.

SECTION 928n. 49.45 (39) (bm) of the statutes is repealed.

SECTION 928r. 49.45 (47m) of the statutes is created to read:

- 49.45 (47m) Family Care funding. (a) In this subsection, "care management organization" means a care management organization under contract with the department of health services as described under s. 46.284.
- (b) The department shall collaborate with care management organizations and the federal centers for Medicare and Medicaid services to develop an allowable payment mechanism to increase the direct care and services portion of the capitation rates to address the direct caregiver workforce challenges in the state.

Vetoed In Part

Vetoed

In Part

- (c) By December 31, 2017, the department shall seek any federal approval necessary from the federal centers for Medicare and Medicaid services to implement the payment mechanism developed under par. (b).
- (d) The department may not implement the plan developed under this subsection unless the department receives federal approval under par. (c). The department may submit one or more requests to the joint committee on finance under s. 13.10 to supplement the appropriation under s. 20.435 (4) (b) from the appropriation under s. 20.865 (4) (a) for implementation of the payment mechanism under par. (b). The department may only use mon-

eys for the payment mechanism under par. (b) if the joint Vetoed committee on finance approves the request under this paragraph. Notwithstanding s. 13.101, the joint committee on finance is not required to find that an emergency exists before making a supplementation under this para- In Part

SECTION 928t. 49.45 (53m) of the statutes is created to read:

49.45 (53m) COVERAGE PROGRAM FOR INSTITUTIONS FOR MENTAL DISEASE. Subject to any necessary waiver approval of the federal department of health and human services, or as otherwise permitted under federal law, the department may, if federal funding participation is available, provide Medical Assistance coverage of services provided in an institution for mental disease to persons ages 21 to 64.

SECTION 929. 49.45 (54) (b) of the statutes is repealed.

SECTION 930. 49.45 (54) (c) of the statutes is created to read:

49.45 (54) (c) Special services. From the appropriations under s. 20.435 (4) (b) and (o) and (7) (bt), the department may pay the costs of services provided under the early intervention program under s. 51.44 that are included in program participant's individualized family service plan and that were not authorized for payment under the state Medicaid plan or a department policy before July 1, 2017, including any services under the early intervention program under s. 51.44 that are delivered by a type of provider that becomes certified to provide Medical Assistance service on July 1, 2017, or after.

SECTION 931. 49.46 (1) (em) of the statutes is created to read:

49.46 (1) (em) To the extent approved by the federal government, for the purposes of determining financial eligibility and any cost-sharing requirements of an individual under par. (a) 6m., 14., or 14m., (d) 2., or (e), the department or its designee shall exclude any assets accumulated in a person's independence account, as defined in s. 49.472 (1) (c), and any income or assets from retirement benefits earned or accumulated from income or employer contributions while employed and receiving state-funded benefits under s. 46.27 or medical assistance under s. 49.472.

SECTION 931n. 49.46 (2) (b) 6. dm. of the statutes is **Vetoed** created to read:

49.46 (2) (b) 6. dm. Subject to the requirements under s. 49.45 (9r), durable medical equipment that is considered complex rehabilitation technology, excluding speech generating devices.

SECTION 931p. 49.46 (2) (b) 6. e. of the statutes is amended to read:

49.46 (2) (b) 6. e. Subject to the limitation under s. 49.45 (30r), inpatient hospital, skilled nursing facility and intermediate care facility services for patients of any institution for mental diseases who are under 21 years of In Part

Vetoed

In Part

age, are under 22 years of age and who were receiving these services immediately prior to reaching age 21, or are 65 years of age or older, or are otherwise permitted under s. 49.45 (53m).

SECTION 932. 49.46 (2) (b) 17. of the statutes is amended to read:

49.46 (2) (b) 17. Services under s. 49.45 (54) (b) (c) for children participating in the early intervention program under s. 51.44, that are provided by a special educator

SECTION 932n. 49.46 (2) (dm) of the statutes is amended to read:

49.46 (2) (dm) Benefits Except as provided under s. 49.45 (53m), benefits under this section may not include payment for services to individuals aged 21 to 64 who are residents of an institution for mental diseases and who are otherwise eligible for medical assistance, except for individuals under 22 years of age who were receiving these services immediately prior to reaching age 21 and continuously thereafter and except for services to individuals who are on convalescent leave or are conditionally released from the institution for mental diseases. For purposes of this paragraph, the department shall define "convalescent leave" and "conditional release" by rule.

SECTION 933. 49.47 (4) (c) 1. of the statutes is amended to read:

49.47 (4) (c) 1. Except To the extent approved by the federal government and except as provided in par. (am), eligibility exists if income does not exceed 133 1/3 100 percent of the maximum aid to families with dependent children payment under s. 49.19 (11) poverty line for the applicant's family size or the combined benefit amount available under supplemental security income under 42 USC 1381 to 1383c and state supplemental aid under s. 49.77 whichever is lower. In this subdivision "income" includes earned or unearned income that would be included in determining eligibility for the individual or family under s. 49.19 or 49.77, or for the aged, blind or disabled under 42 USC 1381 to 1385. "Income" does not include earned or unearned income which would be excluded in determining eligibility for the individual or family under s. 49.19 or 49.77, or for the aged, blind or disabled individual under 42 USC 1381 to 1385.

SECTION 933n. 49.47 (6) (c) 4. of the statutes is amended to read:

49.47 (6) (c) 4. Services Except as provided under s. 49.45 (53m), services to individuals aged 21 to 64 who are residents of an institution for mental diseases and who are otherwise eligible for medical assistance, except for individuals under 22 years of age who were receiving these services immediately prior to reaching age 21 and continuously thereafter and except for services to individuals who are on convalescent leave or are conditionally released from the institution for mental diseases. For purposes of this subdivision, the department shall define "convalescent leave" and "conditional release" by rule.

SECTION 934. 49.472 (3) (a) of the statutes is amended to read:

49.472 (3) (a) The individual's family's net income is less than 250 percent of the poverty line for a family the size of the individual's family. In calculating the net income, the department shall apply all of the exclusions specified under 42 USC 1382a (b) and to the extent approved by the federal government shall exclude medical and remedial expenditures and long-term care costs in excess of \$500 per month that would be incurred by the individual in absence of coverage under the medical assistance purchase plan or a Medicaid long-term care program.

SECTION 935. 49.472 (3) (b) of the statutes is amended to read:

49.472 (3) (b) The individual's assets do not exceed \$15,000. In determining assets, the department may not include assets that are excluded from the resource calculation under 42 USC 1382b (a) of, assets accumulated in an independence account, and, to the extent approved by the federal government, assets from retirement benefits accumulated from income or employer contributions while employed and receiving medical assistance under this section or state—funded benefits under s. 46.27. The department may exclude, in whole or in part, the value of a vehicle used by the individual for transportation to paid employment.

SECTION 936. 49.472 (3) (f) of the statutes is amended to read:

49.472 (3) (f) The individual maintains premium payments under sub. (4) (am) and, if applicable and to the extent approved by the federal government, premium payments calculated by the department in accordance with sub. (4) (bm), unless the individual is exempted from premium payments under sub. (4) (b) (dm) or (5).

SECTION 937. 49.472 (3) (g) of the statutes is amended to read:

49.472 (3) (g) The individual is engaged in gainful employment or is participating in a program that is certified by the department to provide health and employment services that are aimed at helping the individual achieve employment goals. To the extent approved by the federal government, an individual shall prove gainful employment and earned income to the department by providing wage income or prove in–kind work income by federal tax filing documentation. To qualify as gainful income, the amount of in–kind income shall be equal to or greater than the minimum amount for which federal income tax reporting is required.

SECTION 938. 49.472 (4) (a) (intro.) of the statutes is renumbered 49.472 (4) (am) and amended to read:

49.472 (4) (am) Except To the extent approved by the federal government and except as provided in par. (b) pars. (dm) and (em) and sub. (5), an individual who is eligible for medical assistance under sub. (3) and receives medical assistance under this section shall pay a monthly

premium of \$25 to the department. The department shall establish the monthly premiums by rule in accordance with the following guidelines:

SECTION 939. 49.472 (4) (a) 1. of the statutes is repealed.

SECTION 940. 49.472 (4) (a) 2. of the statutes is repealed.

SECTION 941. 49.472 (4) (a) 2m. of the statutes is repealed.

SECTION 942. 49.472 (4) (a) 3. of the statutes is repealed.

SECTION 943. 49.472 (4) (b) of the statutes is repealed.

SECTION 944. 49.472 (4) (bm) of the statutes is created to read:

49.472 (4) (bm) To the extent approved by the federal government, in addition to the \$25 monthly premium under par. (am), an individual who receives medical assistance under this section and whose individual income exceeds 100 percent of the poverty line for a single–person household shall pay 3 percent of his or her adjusted earned and unearned monthly income under par. (cm) that is in excess of 100 percent of the poverty line.

SECTION 945. 49.472 (4) (cm) of the statutes is created to read:

49.472 (4) (cm) For the purposes of par. (bm), an individual's adjusted earned and unearned monthly income is calculated by subtracting from the individual's earned and unearned monthly income his or her actual out–of–pocket medical and remedial expenses, long–term care costs, and impairment–related work expenses.

SECTION 946. 49.472 (4) (dm) of the statutes is created to read:

49.472 (4) (dm) The department shall temporarily waive an individual's monthly premium under par. (am) and, if applicable, par. (bm) when the department determines that paying the premium would be an undue hardship on the individual.

SECTION 947. 49.472 (4) (em) of the statutes is created to read:

49.472 (4) (em) If the department determines that a state plan amendment or waiver of federal Medicaid law is necessary to implement the premium methodology under this subsection and changes to the income and asset eligibility under sub. (3) and s. 49.47 (4) (c) 1., the department shall submit a state plan amendment or waiver request to the federal department of health and human services requesting those changes. If a state plan amendment or waiver is not necessary or if the federal department of health and human services does not disapprove the state plan amendment or waiver request, the department may implement subs. (3) and (4) and s. 49.47 (4) (c) 1. with any adjustments from the federal department of health and human services. If the federal department of health and human services disapproves the state plan amendment or waiver request in whole or in part, the

department may implement the income and asset eligibility requirements and premium methodology under subs. (3) and (4), 2015 stats., and s. 49.47 (4) (c) 1., 2015 stats.

SECTION 948. 49.472 (5) of the statutes is amended to read:

49.472 (5) COMMUNITY OPTIONS PARTICIPANTS. From the appropriation under s. 20.435 (4) (bd), the department may pay all or a portion of the monthly premium calculated under sub. (4) (a) for an individual who is a participant in the community options program under s. 46.27 (11).

SECTION 949. 49.472 (6) (a) of the statutes is amended to read:

49.472 (6) (a) Notwithstanding sub. (4) (a) 3., from the appropriation accounts under s. 20.435 (4) (b), (gm), or (w), the department shall, on the part of an individual who is eligible for medical assistance under sub. (3), pay premiums for or purchase individual coverage offered by the individual's employer if the department determines that paying the premiums for or purchasing the coverage will not be more costly than providing medical assistance.

SECTION 950. 49.497 (1m) (a) of the statutes is amended to read:

49.497 (1m) (a) If, after notice that an incorrect payment was made, a recipient, or parent of a minor recipient, who is liable for repayment of an incorrect payment fails to repay the incorrect payment or enter into, or comply with, an agreement for repayment, the department may bring an action to enforce the liability or may issue an order to compel payment of the liability. The department shall issue the order to compel payment personally or by any type of mail service that requires a signature of acceptance from the recipient at the address of the person who is liable for repayment as it appears on the records of the department. The refusal or failure to accept or receive the order to compel payment by the person who is liable for repayment does not prevent the department from enforcing the order to compel repayment. Any person aggrieved by an order issued by the department under this paragraph may appeal the order as a contested case under ch. 227 by filing with the department a request for a hearing within 30 days after the date of the order. The only issue at the hearing shall be the determination by the department that the person has not repaid the incorrect payment or entered into, or complied with, an agreement for repayment.

SECTION 951. 49.497 (1m) (b) of the statutes is amended to read:

49.497 (1m) (b) If any recipient, or parent of a minor recipient, named in an order to compel payment issued under par. (a) fails to pay the department any amount due under the terms of the order and no contested case to review the order is pending and the time for filing for a contested case review has expired, the department may present a <u>certified</u> true and accurate copy of the order to

the circuit court for any county. The sworn statement of the secretary An affidavit from the collections unit of the department responsible for recoveries under this section shall be evidence of the incorrect payment. The circuit court shall, without notice, render judgment in accordance with the order. A judgment rendered under this paragraph shall have the same effect and shall be entered in the judgment and lien docket and may be enforced in the same manner as if the judgment had been rendered in an action tried and determined by the circuit court.

SECTION 952. 49.497 (1r) (a) of the statutes is amended to read:

49.497 (1r) (a) The department may recover any penalty assessment not paid under s. 49.471 (9) (c) from the employer against which the penalty was assessed. If, after notice that payment of a penalty is overdue, the employer who is liable fails to pay the penalty amount, or enter into or comply with an agreement for payment, the department may bring an action to enforce the liability or may issue an order to compel payment of the liability. The department shall issue the order to compel payment personally or by any type of mail service that requires a signature of acceptance from the recipient at the address of the employer who is liable for repayment as it appears on the records of the department. The refusal or failure to accept or receive the order to compel payment by the employer who is liable for repayment does not prevent the department from enforcing the order to compel repayment. Any person aggrieved by an order issued by the department under this paragraph may appeal the order as a contested case under ch. 227 by filing with the department a request for a hearing within 30 days after the date of the order. The only issue at the hearing shall be the determination by the department that the person has not paid the penalty or entered into, or complied with, an agreement for payment.

SECTION 953. 49.497 (1r) (b) of the statutes is amended to read:

49.497 (1r) (b) If any employer named in an order to compel payment issued under par. (a) fails to pay the department any amount due under the terms of the order and no contested case to review the order is pending and the time for filing for a contested case review has expired, the department may present a certified true and accurate copy of the order to the circuit court for any county. The sworn statement of the secretary An affidavit from the collections unit of the department responsible for recoveries under this section shall be evidence of the failure to pay the penalty. The circuit court shall, without notice, render judgment in accordance with the order. A judgment rendered under this paragraph shall have the same effect and shall be entered in the judgment and lien docket and may be enforced in the same manner as if the judgment had been rendered in an action tried and determined by the circuit court.

SECTION 954. 49.79 (1) (am) 4. of the statutes is renumbered 49.79 (1) (ap) 4.

SECTION 955. 49.79 (1) (am) 5. of the statutes is renumbered 49.79 (1) (ap) 5.

SECTION 957. 49.79 (1) (ap) (intro.) of the statutes is created to read:

49.79 (1) (ap) (intro.) "Able-bodied adult without dependents" means an able-bodied adult who is not any of the following:

SECTION 958. 49.79 (1) (bg) of the statutes is created to read:

49.79 (1) (bg) "Custodial parent" has the meaning given in s. 49.141 (1) (b).

SECTION 959. 49.79 (1) (em) of the statutes is created to read:

49.79 (1) (em) "Noncustodial parent" has the meaning given in s. 49.141 (1) (h).

SECTION 960. 49.79 (1r) of the statutes is created to read:

49.79 (**1r**) ELIGIBILITY; ASSET LIMIT. (a) In this subsection:

- 1. "Elderly, blind, or disabled individual" has the meaning given for "elderly or disabled member" in 7 USC 2012 (j).
- 2. "Household" has the meaning given in 7 USC 2012 (m).
- 3. "Liquid assets" means an individual's financial resources that are cash or can be converted to cash without incurring penalties, excluding the equity value of vehicles or of a home serving as the individual's primary residence. "Liquid assets" does not include any financial resources designated by the department by rule as excluded for the purposes of this subsection.
- (b) Subject to par. (c), an individual who is not an elderly, blind, or disabled individual is ineligible to participate in the food stamp program in a month in which the household of which the individual is a member has liquid assets of more than \$25,000.
- (c) If necessary, the department shall request a waiver from the U.S. department of agriculture to implement this subsection. If the U.S. department of agriculture disapproves the waiver request, the department may not implement this subsection.

SECTION 960c. 49.79 (1t) of the statutes is created to read:

49.79 (1t) FINANCIAL RECORD MATCHING PROGRAM. (a) *Definitions*. In this subsection:

- 1. "Account" means a demand deposit account, checking account, negotiable withdrawal order account, savings account, time deposit account, or money market mutual fund account.
- 2. "Financial institution" has the meaning given in s. 49.45 (4m) (a) 3.
- 3. "Other individual" means an individual whose resources are required by law to be disclosed to deter-

mine the eligibility of an applicant for or recipient of food stamp program benefits.

- (b) Matching program and agreements. 1. The department shall operate a financial record matching program under this subsection for the purpose of verifying the assets of applicants for and recipients of food stamp program benefits and other individuals.
- 2. The department shall enter into agreements with financial institutions doing business in this state to operate the financial record matching program under this subsection. An agreement shall require the financial institution to participate in the financial record matching program by electing either the financial institution matching option under par. (c) or the state matching option under par. (d). Any changes to the conditions of the agreement shall be submitted by the financial institution or the department at least 60 days before the effective date of the change. The department shall furnish the financial institution with a signed copy of the agreement.
- 3. The department shall reimburse a financial institution up to \$125 per calendar quarter for participating in the financial record matching program under this subsection, except that a financial institution that is also participating in the financial record matching program under s. 49.45 (4m) is eligible for reimbursement under only the program under s. 49.45 (4m).
- 4. To the extent feasible, the information to be exchanged under the matching program shall be provided by electronic data exchange as prescribed by the department in the agreement under subd. 2.
- (c) Financial institution matching option. If a financial institution with which the department has an agreement under par. (b) elects the financial institution matching option under this paragraph, all of the following apply:
- 1. At least once each calendar quarter, the department shall provide to the financial institution, in the manner specified in the agreement under par. (b) 2., information regarding applicants for and recipients of food stamp program benefits and other individuals. The information shall include names and social security or other taxpayer identification numbers.
- 2. Based on the information received under subd. 1., the financial institution shall take actions necessary to determine whether any applicant for or recipient of food stamp program benefits or other individual has an ownership interest in an account maintained at the financial institution. If the financial institution determines that an applicant, recipient, or other individual has an ownership interest in an account at the financial institution, the financial institution shall provide the department with a notice containing the applicant's, recipient's, or other individual's name, address of record, social security number or other taxpayer identification number, and account information. The account information shall include the account number, the account type, the nature

- of the ownership interest in the account, and the balance of the account at the time that the record match is made. The notice under this subdivision shall be provided in the manner specified in the agreement under par. (b) 2. and, to the extent feasible, by an electronic data exchange.
- (d) *State matching option*. If a financial institution with which the department has an agreement under par. (b) elects the state matching option under this paragraph, all of the following apply:
- 1. At least once each calendar quarter, the financial institution shall provide the department with information concerning all accounts maintained at the financial institution. For each account maintained at the financial institution, the financial institution shall notify the department of the name and social security number or other tax identification number of each person having an ownership interest in the account, together with a description of each person's interest. The information required under this subdivision shall be provided in the manner specified in the agreement under par. (b) 2. and, to the extent feasible, by an electronic data exchange.
- 2. The department shall take actions necessary to determine whether any applicant for or recipient of food stamp program benefits or other individual has an ownership interest in an account maintained at the financial institution providing information under subd. 1. Upon the request of the department, the financial institution shall provide to the department, for each applicant, recipient, or other individual who matches information provided by the financial institution under subd. 1., the address of record, the account number and account type, and the balance of the account.
- (e) Use of information by financial institution; penalty. A financial institution participating in the financial record matching program under this subsection, and the employees, agents, officers, and directors of the financial institution, may use information received from the department under par. (c) only for the purpose of matching records and may use information provided by the department in requesting additional information under par. (d) only for the purpose of providing the additional information. Neither the financial institution nor any employee, agent, officer, or director of the financial institution may disclose or retain information received from the department concerning applicants for or recipients of food stamp program benefits or other individuals. Any person who violates this paragraph may be fined not less than \$50 nor more than \$1,000 or imprisoned in the county jail for not less than 10 days or more than one year
- (f) Use of information by department. The department may use information provided by a financial institution under this subsection only for matching records under par. (d), for administering the financial record matching program under this subsection, and for determining eligibility or continued eligibility under this sub-

chapter. The department may not disclose or retain information received from a financial institution under this subsection concerning account holders who are not applicants for or recipients of food stamp program benefits or other individuals.

(g) Financial institution liability. A financial institution is not liable to any person for disclosing information to the department under this subsection or for any other action that the financial institution takes in good faith to comply with this subsection.

SECTION 961. 49.79 (6m) of the statutes is created to read:

- 49.79 (6m) ELIGIBILITY DENIAL; CHILD SUPPORT NON-COMPLIANCE. (a) In this subsection, what constitutes a refusal to cooperate is determined by the department in accordance with 7 USC 2015 (l) and (m) and any federal regulations promulgated under 7 USC 2015 (l) and (m).
- (b) An individual is ineligible to participate in the food stamp program in a month in which any of the following is true:
 - 1. The individual satisfies all of the following:
- a. The individual is a custodial parent of or lives with and exercises parental control over a child who is under the age of 18 and who has an absent parent.
- b. The individual refuses to cooperate fully, in good faith, with efforts directed at establishing or enforcing any support order or obtaining any other payments or property to which that individual or the child may have rights.
- c. The individual does not have good cause for refusing to cooperate, as determined by the department in accordance with 7 USC 2015 (l) (2) and any federal regulations promulgated under 7 USC 2015 (l) (2).
- 2. The individual is a noncustodial parent of a child under the age of 18 and the individual refuses to cooperate in providing or obtaining support for the child.

SECTION 962. 49.79 (6q) of the statutes is created to read:

- 49.79 (**6q**) ELIGIBILITY DENIAL; PATERNITY. (a) In this subsection, the department shall determine what constitutes a refusal to cooperate in accordance with 7 USC 2015 (l) and (m) and any federal regulations promulgated under 7 USC 2015 (l) and (m).
- (b) An individual is ineligible to participate in the food stamp program in a month in which any of the following is true:
 - 1. The individual satisfies all of the following:
- a. The individual is a custodial parent of or lives with and exercises parental control over a child who is under the age of 18 and who has an absent parent.
- b. The individual refuses to cooperate fully, in good faith, with applicable efforts directed at establishing the paternity of the child.
- c. The individual does not have good cause for refusing to cooperate, as determined by the department in

- accordance with 7 USC 2015 (l) (2) and any federal regulations promulgated under 7 USC 2015 (l) (2).
- 2. The individual is one of the following and refuses to cooperate fully, in good faith, with efforts directed at establishing the paternity of the child:
- a. Alleged to be the father under s. 767.80 of a child under the age of 18.
- b. A noncustodial parent of a child under the age of 18 for whom paternity has not been established.

SECTION 963. 49.79 (6t) of the statutes is created to read:

- 49.79 (6t) ELIGIBILITY DENIAL; DELINQUENT SUPPORT. An individual is ineligible to participate in the food stamp program in a month in which the individual is obligated by court order to provide support payments and is delinquent in making those court—ordered payments, unless any of the following is true:
- (a) The delinquency balance equals less than 3 months of the court–ordered support payment amount.
- (b) A court or a county child support agency under s. 59.53 (5) is allowing the individual to delay the child support payments.
- (c) The individual is complying with a payment plan approved by a county child support agency under s. 59.53 (5) to provide support for the child of the individual.
- (d) The individual is participating in an employment and training program, as determined by the department.

SECTION 963b. 49.79 (6u) of the statutes is created to read:

- 49.79 (**6u**) IMPLEMENTATION OF PATERNITY AND CHILD SUPPORT REQUIREMENTS. (a) The department may not implement sub. (6m), (6q), or (6t) unless all of the following are satisfied with respect to that subsection:
- 1. The department of children and families determines that the requirement of the subsection as it pertains to child support and paternity order establishment and compliance is able to be implemented in a way that is substantially state budget neutral in regard to child support fees.
- 2. The department of health services or the department of children and families has obtained any necessary approval from the federal government to implement the subsection in a budget—neutral manner in regard to child support fees.
- 3. The department of health services and the department of children and families have notified the governor and the joint committee on finance that the subsection may be implemented in a budget—neutral manner in regard to child support fees and that all necessary federal approval is obtained.
- (b) If the criteria under par. (a) for sub. (6m), (6q), or (6t) are satisfied to be implemented, the applicable subsection takes effect on the first day of the 6th month beginning after the date that the department of children

and families has made the notification to the governor and the joint committee on finance under par. (a) 3.

SECTION 964. 49.79 (9) (a) 1. of the statutes is amended to read:

49.79 (9) (a) 1. The department shall administer an employment and training program for recipients under the food stamp program and may contract with county departments under ss. 46.215, 46.22, and 46.23, multicounty consortia, local workforce development boards established under 29 USC 2832, tribal governing bodies, or other organizations to carry out the administrative functions. A county department, multicounty consortium, local workforce development board, tribal governing body, or other organization may subcontract with a Wisconsin Works agency or another provider to administer the employment and training program under this subsection. Except as provided in subds. 2. and 3., the department may require able individuals who are 18 to 60 years of age, or a subset of those individuals to the extent allowed by the federal government, who are not participants in a Wisconsin Works employment position to participate in the employment and training program under this subsection.

Vetoed In Part

SECTION 964d. 49.79 (9) (e) of the statutes is created to read:

49.79 (9) (e) The department shall ensure that all applicants for and recipients of the food stamp program are provided information about the employment and training program under this subsection at least once every 6 months and that all able-bodied adults without dependents are referred to the employment and training program under this subsection regardless of whether they are required to comply with a work requirement.

SECTION 965. 49.79 (10) (title) of the statutes is amended to read:

49.79 (10) (title) Eligibility and work require-MENTS FOR ABLE-BODIED ADULTS WITHOUT DEPENDENTS.

SECTION 966. 49.79 (10) (a) 1. of the statutes is amended to read:

49.79 (10) (a) 1. The department shall require an able-bodied adult without dependents who is participating in the food stamp program to fulfill the work requirement defined under 7 CFR 273.24 (a) (1).

SECTION 967. 49.79 (10) (a) 2. of the statutes is amended to read:

49.79 (10) (a) 2. If an able-bodied adult without dependents does not fulfill the work requirement, the department may limit the eligibility of the able-bodied adult's eligibility adult without dependents for food stamps to no more than 3 months during a 3-year period.

SECTION 968. 49.79 (10) (a) 3. of the statutes is amended to read:

49.79 (10) (a) 3. The department may exempt up to 15 percent of the able-bodied adults without dependents who are participating in the food stamp program from the time limit under subd. 2.

SECTION 968g. 49.79 (11) of the statutes is created to read:

49.79 (11) Treatment of inactive accounts; EXPUNGEMENT OF UNUSED BENEFITS. (a) If, for a period of 6 months or longer, an individual or household that is receiving benefits under this section through an electronic benefit transfer system uses no benefits that have been posted to the individual's or household's benefit account, the department shall remove all benefits from the account electronically and store them offline. The benefits being stored offline shall be made available to the individual or household again within 48 hours after a request by the individual or a member of the household to restore the benefits or upon reapplication by the individual or household for benefits under this section, whichever is applicable. The department shall attempt to notify the individual or household before benefits are removed from the account under this paragraph and shall describe the steps that the individual or household must take to get the benefits returned to the account.

- (b) The department shall expunge any benefits that have not been used after a period of one year, regardless of whether either of the following applies:
- 1. The benefits have been removed from an inactive benefit account under par. (a) and are being stored offline.
 - 2. The benefits are still posted to an active account.
- (c) The department shall seek any necessary approval from the U.S. department of agriculture to implement this subsection. If the U.S. department of agriculture disapproves, the department may not implement this subsection.

SECTION 969n. 50.14 (1) (am) of the statutes is cre- **Vetoed** ated to read:

In Part

50.14 (1) (am) "Institution for mental diseases" has the meaning given in s. 49.43 (6m).

SECTION 969p. 50.14 (2) (intro.) of the statutes is amended to read:

50.14 (2) (intro.) For Except as provided under sub. (2d), for the privilege of doing business in this state, there is imposed on all licensed beds of a facility an assessment in the following amount per calendar month per licensed bed of the facility:

SECTION 969r. 50.14 (2d) of the statutes is created to read:

50.14 (2d) (a) The department shall request approval from the secretary of the federal department of health and human services for the state to allow an exemption from the assessment described under sub. (2) for county government-owned institutions for mental diseases and facilities that are state licensed but not certified to participate in the Medicaid or Medicare programs.

(b) To the extent approved by the federal department of health and human services under par. (a), the requirements under this section do not apply to a county government-owned institution for mental diseases or a facility that is state licensed but not certified to participate in the

Medicaid or Medicare programs, effective on July 1, 2017, or the date on which the state receives federal approval, whichever is later.

SECTION 971p. 50.38 (3) of the statutes is amended to read:

50.38 (3) The department shall establish the percentage that is applicable under sub. (2) (a) and (b) so that the total amount of assessments collected under sub. (2) (a) in a state fiscal year is equal to the amount in the schedule under s. 20.005 (3) for the appropriation under s. 20.435 (4) (xc) for that fiscal year \$414,507,300.

SECTION 971r. 50.38 (6) (b) of the statutes is amended to read:

50.38 (6) (b) On June 30 of each state fiscal year, the department shall, from the appropriation account under s. 20.435 (4) (xc), refund to eligible hospitals, other than critical access hospitals, the difference between the amount in the schedule under s. 20.005 (3) for that appropriation and the amount any amounts not expended or encumbered from that appropriation in the fiscal year or transferred under sub. (8).

SECTION 971t. 50.38 (6m) (b) of the statutes is amended to read:

50.38 (**6m**) (b) On June 30 of each state fiscal year, the department shall, from the appropriation account under s. 20.435 (4) (xe), refund to critical access hospitals any unencumbered moneys in the critical access hospital assessment fund amounts not expended or encumbered from that appropriation in the fiscal year or transferred under sub. (10).

SECTION 971v. 50.38 (8) of the statutes is amended to read:

50.38 **(8)** In each state fiscal year, the secretary of administration shall transfer from the hospital assessment fund to the Medical Assistance trust fund an amount equal to the amount in the schedule under s. 20.005 (3) for the appropriation under s. 20.435 (4) (xc) collected under sub. (2) (a) for that fiscal year minus the state share of payments to hospitals required under s. 49.45 (3) (e) 11., and minus any refunds paid to hospitals from the hospital assessment fund under sub. (6) (a) in that fiscal year.

SECTION 973. 50.49 (6) (a) of the statutes is amended to read:

50.49 (6) (a) Except as provided in s. 50.498, the department shall issue a home health agency license if the applicant is fit and qualified, and if the home health agency meets the requirements established by this section. The Except as provided in par. (am), the department, or its designated representatives, shall make such inspections and investigations as are necessary to determine the conditions existing in each case and file written reports. Each licensee shall annually file a report with the department.

SECTION 974. 50.49 (6) (am) of the statutes is created to read:

50.49 (6) (am) In lieu of performing its own inspection or investigation under par. (a), the department may recognize as evidence for purposes of licensure accreditation of the home health agency by an organization that is approved by the federal centers for Medicare and Medicaid services and that meets any requirements established by the department. The home health agency shall provide the department with a copy of the report by the accreditation organization of each periodic review the organization conducts of the home health agency for the department's use in tracking compliance, investigating complaints, and conducting further surveys.

SECTION 976. 50.92 (4) (b) of the statutes is amended to read:

50.92 (4) (b) In lieu of inspecting or investigating a hospice under sub. (3) prior to issuance of a license, the department may accept evidence that a hospice applying for licensure under s. 50.93 has been inspected under and is currently in compliance with the hospice requirements of the joint commission for the accreditation of health organizations as a hospice from an organization that is approved by the federal centers for Medicare and Medicaid services and that meets any requirements established by the department. A hospice shall provide the department with a copy of the report by the joint commission for the accreditation of health organizations organization of each periodic review the association organization conducts of the hospice.

SECTION 977. 51.042 of the statutes is created to read: 51.042 Youth crisis stabilization facilities. (1) DEFINITIONS. In this section:

- (a) "Crisis" means a situation caused by an individual's apparent mental disorder that results in a high level of stress or anxiety for the individual, persons providing care for the individual, or the public and that is not resolved by the available coping methods of the individual or by the efforts of those providing ordinary care or support for the individual.
- (b) "Youth crisis stabilization facility" is a treatment facility with a maximum of 8 beds that admits a minor to prevent or de–escalate the minor's mental health crisis and avoid admission of the minor to a more restrictive setting.
- (2) CERTIFICATION REQUIRED; EXEMPTION. (a) No person may operate a youth crisis stabilization facility without a certification from the department. The department may limit the number of certifications it grants to operate a youth crisis stabilization facility.
- (b) A youth crisis stabilization facility that has a certification from the department under this section is not subject to facility regulation under ch. 48.
- (3) ADMISSION OF MINORS. A minor may be admitted to a youth crisis stabilization facility under this section by a court order under s. 51.20 (13) (a) 3. or through the procedure under s. 51.13. No person may transport a minor

to a youth crisis stabilization facility for detention under s. 51.15.

(4) RULES. The department may promulgate rules to implement this section.

SECTION 978f. 51.20 (5) of the statutes is renumbered 51.20 (5) (a) and amended to read:

51.20 (5) (a) The hearings which are required to be held under this chapter shall conform to the essentials of due process and fair treatment including the right to an open hearing, the right to request a closed hearing, the right to counsel, the right to present and cross-examine witnesses, the right to remain silent and the right to a jury trial if requested under sub. (11). The parent or guardian of a minor who is the subject of a hearing shall have the right to participate in the hearing and to be represented by counsel. All proceedings under this chapter shall be reported as provided in SCR 71.01.

(b) The court may determine to hold a hearing under this section at the institution at which the individual is detained, whether or not located in the same county as the court with which the petition was filed, unless the individual or his or her attorney objects.

SECTION 978k. 51.20 (5) (c) of the statutes is created to read:

51.20 (5) (c) 1. In a county in which the courthouse is 100 miles or more away from the facility at which the subject of the hearing is detained and has videoconferencing capabilities that meet the technical and operational standards under s. 885.54, the court may conduct the hearing under this section by videoconference unless both the corporation counsel and counsel representing the subject of the hearing object.

2. Subdivision 1. does not preclude a court from conducting a hearing by videoconference in circumstances other than described under subd. 1.

SECTION 979. 51.44 (3) (d) of the statutes is created to read:

51.44 (**3**) (d) From the appropriation under s. 20.435 (7) (bt), the department may pay the nonfederal share of Medical Assistance costs for services provided under s. 49.45 (54) (c).

SECTION 979k. 51.45 (13) (d) of the statutes is amended to read:

51.45 (13) (d) Whenever it is desired to involuntarily commit a person, a preliminary hearing shall be held under this paragraph. The purpose of the preliminary hearing shall be to determine if there is probable cause for believing that the allegations of the petition under par. (a) are true. The court shall assure that the person is represented by counsel at the preliminary hearing by referring the person to the state public defender, who shall appoint counsel for the person without a determination of indigency, as provided in s. 51.60. Counsel shall have access to all reports and records, psychiatric and otherwise, which have been made prior to the preliminary hearing. The person shall be present in person or by videoconference under s. 51.20 (5) (c) at the preliminary hearing and shall be afforded a meaningful opportunity to be heard. Upon failure to make a finding of probable cause under this paragraph, the court shall dismiss the petition and discharge the person from the custody of the county department.

SECTION 979p. 51.61 (1) (z) of the statutes is created **Vetoed** to read:

In Part

51.61 (1) (z) In the case of a patient committed under ch. 980, have the right to have a county department submit a report under s. 980.08 (4) (dm) within the time frame specified under that paragraph.

SECTION 980s. 59.17 (2) (b) 3. (intro.) of the statutes **Vetoed** is renumbered 59.17 (2) (b) 3. and amended to read:

In Part

59.17 (2) (b) 3. Exercise the authority under s. 59.52 (6) (a) that would otherwise be exercised by a county board, except that the county board may continue to exercise the authority under s. 59.52 (6) with regard to land that is zoned as a park on or after July 14, 2015, other than land zoned as a park in the city of Milwaukee that is located within the area west of Lincoln Memorial Drive, south of E. Michigan Street, east of N. Van Buren Street, and north of E. Clybourn Avenue. With regard to the sale, acquisition, or lease as landlord or tenant of property, other than certain park land as described in this subdivision, the county executive's action need not must be consistent with established county board policy and may take effect without submission to or approval by must be approved by the county board. The proceeds of the sale of property as authorized under this subdivision shall first be applied to any debt attached to the property. Before the county executive's sale of county land may take effect, a majority of the following must sign a document, a copy of which will be attached to the bill of sale and a copy of which will be retained by the county, certifying that they believe the sale is in the best interests of the county: to take effect. The county board may only approve or reject the contract as negotiated by the county executive.

SECTION 980se. 59.17 (2) (b) 3. a. to c. of the statutes are repealed.

SECTION 981. 59.20 (3) (a) of the statutes is amended to read:

59.20 (3) (a) Every sheriff, clerk of the circuit court, register of deeds, treasurer, comptroller, register of probate, clerk, and county surveyor shall keep his or her office at the county seat in the offices provided by the county or by special provision of law; or if there is none, then at such place as the board directs. The board may also require any elective or appointive county official to keep his or her office at the county seat in an office to be provided by the county. All such officers shall keep their offices open during the usual business hours of any day except Sunday, as the board directs. With proper care, the officers shall open to the examination of any person all books and papers required to be kept in his or her office and permit any person so examining to take notes and copies of such books, records, papers, or minutes therefrom except as authorized in par. (c) and ss. $19.36 (10) \leftrightarrow (12) \text{ and } (11)$ and 19.59 (3) (d) or under ch. 69.

Vetoed In Part **SECTION 981e.** 59.255 (2) (L) of the statutes is created to read:

59.255 (2) (L) The comptroller shall administer accounts payable, payroll, accounting, and financial information systems.

Vetoed In Part **SECTION 981h.** 59.52 (6) (intro.) of the statutes is amended to read:

59.52 (6) PROPERTY. (intro.) Except as provided in s. 59.17 (2) (b) 3., the The board may:

SECTION 981m. 59.52 (6) (a) of the statutes is amended to read:

Vetoed In Part

59.52 (6) (a) How acquired; purposes. Take Except as provided in s. 59.17 (2) (b) 3., take and hold land acquired under ch. 75 and acquire, lease or rent property, real and personal, for public uses or purposes of any nature, including without limitation acquisitions for county buildings, airports, parks, recreation, highways, dam sites in parks, parkways and playgrounds, flowages, sewage and waste disposal for county institutions, lime pits for operation under s. 59.70 (24), equipment for clearing and draining land and controlling weeds for operation under s. 59.70 (18), ambulances, acquisition and transfer of real property to the state for new collegiate institutions or research facilities, and for transfer to the state for state parks and for the uses and purposes specified in s. 23.09 (2) (d). The power of condemnation may not be used to acquire property for the purpose of establishing or extending a recreational trail; a bicycle way, as defined in s. 340.01 (5s); a bicycle lane, as defined in s. 340.01 (5e); or a pedestrian way, as defined in s. 346.02

SECTION 982b. 59.52 (7) of the statutes is amended to read:

59.52 (7) JOINT COOPERATION. The board may join with the state, other counties and municipalities in a cooperative arrangement as provided by s. 66.0301, including the acquisition, development, remodeling, construction, equipment, operation and maintenance of land, buildings and facilities for regional projects, whether or not such projects are located within the county. If a county is required to establish or maintain an agency, department, commission, or any other office or position to carry out a county responsibility, and the county joins with another county or municipality by entering into an intergovernmental cooperation contract under s. 66.0301 (2) to jointly carry out the responsibility. the jointly established or maintained agency, department, commission, or any other office or position to which the contract applies fulfills the county's obligation to establish or maintain such entities or positions until the contract entered into under s. 66.0301 (2) expires or is terminated by the parties. In addition, if 2 or more counties enter into an intergovernmental cooperation contract and

create a commission under s. 66.0301 (2) to jointly or regionally administer a function or project, the commission shall be considered to be a single entity that represents, and may act on behalf of, the joint interests of the signatories to the contract entered into under s. 66.0301 (2).

SECTION 982f. 59.52 (31) (e) of the statutes is Vetoed repealed. In Part

SECTION 982i. 59.69 (10) (ab) of the statutes is renumbered 59.69 (10) (ab) (intro.) and amended to read: 59.69 (10) (ab) (intro.) In this subsection "nonconforming use":

3. "Nonconforming use" means a use of land, a dwelling, or a building that existed lawfully before the current zoning ordinance was enacted or amended, but that does not conform with the use restrictions in the current ordinance.

SECTION 982ib. 59.69 (10) (ab) 1. of the statutes is created to read:

59.69 (10) (ab) 1. "Contiguous" means sharing a common boundary or being separated only by a waterway, section line, public road, private road, transportation right-of-way, or utility right-of-way.

SECTION 982ic. 59.69 (10) (ab) 2. of the statutes is created to read:

59.69 (10) (ab) 2. "Nonconforming quarry site" means land on which a quarry existed lawfully before the quarry became a nonconforming use, and includes any parcel of land that, as of the effective date of this subdivision [LRB inserts date], is contiguous to the land on which the quarry is located, is under the common ownership, leasehold, or control of the person who owns, leases, or controls the land on which the quarry is located, and is located in the same political subdivision.

SECTION 982id. 59.69 (10) (ab) 4. of the statutes is created to read:

59.69 (**10**) (ab) 4. "Quarry" has the meaning given in s. 66.0414 (2) (d).

SECTION 982ie. 59.69 (10) (ab) 5. of the statutes is created to read:

59.69 (**10**) (ab) 5. "Quarry operations" has the meaning given in s. 66.0414 (2) (e).

SECTION 982if. 59.69 (10) (ap) of the statutes is created to read:

59.69 (10) (ap) Notwithstanding par. (am), an ordinance enacted under this section may not prohibit the continued operation of a quarry at a nonconforming quarry site. For purposes of this paragraph, the continued operation of a quarry includes conducting quarry operations in an area of a nonconforming quarry site in which quarry operations have not previously been conducted.

SECTION 982ig. 59.69 (10s) of the statutes is created to read:

59.69 (10s) RENEWAL OF QUARRY PERMITS. (a) Except as provided in par. (b), a county shall, upon submission by a quarry operator of an application for renewal of a

Vetoed In Part Vetoed In Part

permit, as defined in s. 66.0414 (2) (a), renew the permit if the permit has a duration of less than 10 years.

(b) A county may deny the renewal of a permit, as defined in s. 66.0414 (2) (a), having a duration of less than 10 years if the holder of the permit fails to cure a material violation of a condition of the permit after reasonable notice from the county of the violation and a reasonable opportunity for the quarry operator to cure the violation.

SECTION 982m. 60.61 (5) (ab) of the statutes is renumbered 60.61 (5) (ab) (intro.) and amended to read: 60.61 (5) (ab) (intro.) In this subsection "nonconforming use":

2. "Nonconforming use" means a use of land, a dwelling, or a building that existed lawfully before the current zoning ordinance was enacted or amended, but that does not conform with the use restrictions in the current ordinance.

SECTION 982mb. 60.61 (5) (ab) 1. of the statutes is created to read:

60.61 (**5**) (ab) 1. "Nonconforming quarry site" has the meaning given in s. 59.69 (10) (ab) 2.

SECTION 982mc. 60.61 (5) (ab) 3. of the statutes is created to read:

60.61 (**5**) (ab) 3. "Quarry" has the meaning given in s. 66.0414 (2) (d).

SECTION 982md. 60.61 (5) (ab) 4. of the statutes is created to read:

60.61 (**5**) (ab) 4. "Quarry operations" has the meaning given in s. 66.0414 (2) (e).

SECTION 982me. 60.61 (5) (as) of the statutes is created to read:

60.61 (5) (as) Notwithstanding par. (am), an ordinance enacted under this section may not prohibit the continued operation of a quarry at a nonconforming quarry site. For purposes of this paragraph, the continued operation of a quarry includes conducting quarry operations in an area of a nonconforming quarry site in which quarry operations have not previously been conducted.

SECTION 982mf. 60.61 (5s) of the statutes is created to read:

60.61 (**5s**) RENEWAL OF QUARRY PERMITS. (a) Except as provided in par. (b), a town shall, upon submission by a quarry operator of an application for renewal of a permit, as defined in s. 66.0414 (2) (a), renew the permit if the permit has a duration of less than 10 years.

(b) A town may deny the renewal of a permit, as defined in s. 66.0414 (2) (a), having a duration of less than 10 years if the holder of the permit fails to cure a material violation of a condition of the permit after reasonable notice from the town of the violation and a reasonable opportunity for the quarry operator to cure the violation

SECTION 982p. 60.782 (2) (d) of the statutes is amended to read:

60.782 (2) (d) Lease or acquire, including by condemnation, any real property situated in this state that may be needed for the purposes of s. 23.09 (19), 23.094 (3g) or 30.275 (4). The power of condemnation may not used to acquire property for the purpose of establishing or extending a recreational trail; a bicycle way, as defined in s. 340.01 (5s); a bicycle lane, as defined in s. 340.01 (5e); or a pedestrian way, as defined in s. 346.02 (8) (a).

SECTION 982pg. 61.34 (3) of the statutes is renumbered 61.34 (3) (a) and amended to read:

61.34 (3) (a) The Except as provided in par. (b), the village board may acquire property, real or personal, within or outside the village, for parks, libraries, recreation, beautification, streets, water systems, sewage or waste disposal, harbors, improvement of watercourses, public grounds, vehicle parking areas, and for any other public purpose; may acquire real property within or contiguous to the village, by means other than condemnation, for industrial sites; may improve and beautify the same; may construct, own, lease and maintain buildings on such property for instruction, recreation, amusement and other public purposes; and may sell and convey such property. Condemnation shall be as provided by ch. 32.

SECTION 982pm. 61.34 (3) (b) of the statutes is created to read:

61.34 (3) (b) The village board may not use the power of condemnation to acquire property for the purpose of establishing or extending a recreational trail; a bicycle way, as defined in s. 340.01 (5s); a bicycle lane, as defined in s. 340.01 (5e); or a pedestrian way, as defined in s. 346.02 (8) (a).

SECTION 982pr. 62.22 (1) of the statutes is renumbered 62.22 (1) (a) and amended to read:

62.22 (1) (a) The Except as provided in par. (b), the governing body of any city may by gift, purchase or condemnation acquire property, real or personal, within or outside the city, for parks, recreation, water systems, sewage or waste disposal, airports or approaches thereto, cemeteries, vehicle parking areas, and for any other public purpose; may acquire real property within or contiguous to the city, by means other than condemnation, for industrial sites; may improve and beautify the same; may construct, own, lease and maintain buildings on such property for public purposes; and may sell and convey such property. The power of condemnation for any such purpose shall be as provided by ch. 32.

SECTION 982pt. 62.22 (1) (b) of the statutes is created to read:

62.22 (1) (b) The governing body of any city may not use the power of condemnation to acquire property for the purpose of establishing or extending a recreational trail; a bicycle way, as defined in s. 340.01 (5s); a bicycle lane, as defined in s. 340.01 (5e); or a pedestrian way, as defined in s. 346.02 (8) (a).

SECTION 982q. 62.23 (7) (ab) of the statutes is renumbered 62.23 (7) (ab) (intro.) and amended to read: 62.23 (7) (ab) Definition Definitions. (intro.) In this subsection "nonconforming use":

2. "Nonconforming use" means a use of land, a dwelling, or a building that existed lawfully before the current zoning ordinance was enacted or amended, but that does not conform with the use restrictions in the current ordinance.

SECTION 982qb. 62.23 (7) (ab) 1. of the statutes is created to read:

62.23 (7) (ab) 1. "Nonconforming quarry site" has the meaning given in s. 59.69 (10) (ab) 2.

SECTION 982qc. 62.23 (7) (ab) 3. of the statutes is created to read:

62.23 (7) (ab) 3. "Quarry" has the meaning given in s. 66.0414 (2) (d).

SECTION 982qd. 62.23 (7) (ab) 4. of the statutes is created to read:

62.23 (7) (ab) 4. "Quarry operations" has the meaning given in s. 66.0414 (2) (e).

SECTION 982ge. 62.23 (7) (hd) of the statutes is created to read:

62.23 (7) (hd) Nonconforming quarry sites. Notwithstanding par. (h), an ordinance enacted under this subsection may not prohibit the continued operation of a quarry at a nonconforming quarry site. For purposes of this paragraph, the continued operation of a quarry includes conducting quarry operations in an area of a nonconforming quarry site in which quarry operations have not previously been conducted.

SECTION 982qs. 62.23 (17) (a) (intro.) of the statutes is amended to read:

62.23 (17) (a) (intro.) Cities Except as provided in par. (am), cities may acquire by gift, lease, purchase or condemnation any lands (a) within its corporate limits for establishing, laying out, widening, enlarging, extending and maintaining memorial grounds, streets, squares, parkways, boulevards, parks, playgrounds, sites for public buildings, and reservations in and about and along and leading to any or all of the same; (b) any lands adjoining or near to such city for use, sublease or sale for any of the following purposes:

SECTION 982r. 62.23 (17) (am) of the statutes is created to read:

62.23 (17) (am) Cities may not use the power of condemnation to acquire property for the purpose of establishing or extending a recreational trail; a bicycle way, as defined in s. 340.01 (5s); a bicycle lane, as defined in s. 340.01 (5e); or a pedestrian way, as defined in s. 346.02 (8) (a).

Vetoed In Part

SECTION 982s. 62.23 (19) of the statutes is created to read:

62.23 (19) RENEWAL OF QUARRY PERMITS. (a) Except as provided in par. (b), a city shall, upon submission by a quarry operator of an application for renewal of a permit, as defined in s. 66.0414 (2) (a), renew the permit if the permit has a duration of less than 10 years.

(b) A city may deny the renewal of a permit, as defined in s. 66.0414 (2) (a), having a duration of less than 10 years if the holder of the permit fails to cure a material violation of a condition of the permit after reasonable notice from the city of the violation and a reasonable opportunity for the quarry operator to cure the viola-

SECTION 982t. 66.0102 of the statutes is created to **Vetoed** read:

In Part

- 66.0102 Conflicts with statutory provisions. (1) In this section, "political subdivision" means a city, village, town, or county.
- (2) A political subdivision may not enforce an ordinance if any of the following applies:
- (a) A statutory provision prohibits the political subdivision from enforcing the ordinance.
- (b) The ordinance logically conflicts with a statutory provision.
- (c) The ordinance defeats the purpose of a statutory provision.
- (d) The ordinance violates the spirit of a statutory

SECTION 983e. 66.0137 (5) (b) of the statutes is amended to read:

66.0137 (5) (b) The state or a local governmental unit may provide for the payment of premiums or cost sharing for hospital, surgical and other health and accident insurance and life insurance for employees and officers, their spouses, and dependent children, and their domestic partners under ch. 770 and dependent children. A local governmental unit may also provide for the payment of premiums or cost sharing for hospital and surgical care for its retired employees. In addition, a local governmental unit may, by ordinance or resolution, elect to offer to all of its employees a health care coverage plan through a program offered by the group insurance board under ch. 40. A local governmental unit that elects to participate under s. 40.51 (7) is subject to the applicable sections of ch. 40 instead of this subsection.

SECTION 984b. 66.0301 (2) of the statutes is amended to read:

66.0301 (2) Subject to s. 59.794 (2), and in addition to the provisions of any other statutes specifically authorizing cooperation between municipalities, unless those statutes specifically exclude action under this section, any municipality may contract with other municipalities and with federally recognized Indian tribes and bands in this state, for the receipt or furnishing of services or the joint exercise of any power or duty required or authorized by law. If municipal or tribal parties to a contract have varying powers or duties under the law, each may act under the contract to the extent of its lawful powers and duties. A contract under this subsection may bind the contracting parties for the length of time specified in the contract. This section shall be interpreted liberally in favor of cooperative action between municipalities and between municipalities and Indian tribes and bands in this state. If a municipality is required to establish or maintain an agency, department, commission, or any other office or position to carry out a municipal responsibility, and the municipality joins with another municipality by entering into an intergovernmental cooperation contract under this subsection to jointly carry out the responsibility, the jointly established or maintained agency, department, commission, or any other office or position to which the contract applies fulfills, subject to sub. (7), the municipality's obligation to establish or maintain such entities or positions until the contract entered into under this subsection expires or is terminated by the parties. In addition, if 2 or more municipalities enter into an intergovernmental cooperation contract and create a commission under this section to jointly or regionally administer a function or project, the commission shall be considered, subject to sub. (7), to be a single entity that represents, and may act on behalf of, the joint interests of the signatories to the contract entered into under this section.

SECTION 984c. 66.0301 (7) of the statutes is created to read:

66.0301 (7) With regard to a contract entered into under sub. (2) between 2 or more counties, which relates to the provision of services or facilities under a contract with an officer or agency of the state, the contract may not take effect unless it is approved in writing by the officer or chief of the agency that has authority over the contract for the provision of services or facilities. The contract must be approved or disapproved in writing by the officer or chief of the agency with regard to the matters within the scope of the contract for the provision of services or facilities within 90 days after receipt of the contract. Any disapproval shall detail the specific respects in which the proposed contract fails to demonstrate that the signatories intend to fulfill their contractual responsibilities or obligations. If the officer or chief of the agency fails to approve or disapprove of the contract entered into under sub. (2) within 90 days after receipt, the contract shall be considered approved by the officer or chief of the agency.

Vetoed In Part **SECTION 984g.** 66.0304 (1) (c) of the statutes is amended to read:

66.0304 (1) (c) "Commission" means an entity created by two or more political subdivisions, who contract with each other under s. 66.0301 (2) or 66.0303 (2), for the purpose of issuing bonds exercising the powers under this section.

SECTION 984gb. 66.0304 (1) (e) of the statutes is amended to read:

66.0304(1) (e) "Participant" means any public or private entity or unincorporated association, including a federally recognized Indian tribe or band, <u>and including</u> a business entity created under sub. (4e), that contracts

with a commission for the purpose of financing or refinancing a project that is owned, sponsored, or controlled by the public or private entity or unincorporated association. **Vetoed**In Part

SECTION 984gc. 66.0304 (1) (f) of the statutes is amended to read:

66.0304 (1) (f) "Political subdivision" means any city, village, town, or county in this state or any city, village, town, county, district, authority, agency, commission, or other similar governmental entity, or tribal government in another state or office, department, authority, or agency of any such other state or territory of the United States.

SECTION 984gd. 66.0304 (3) (a) of the statutes is amended to read:

66.0304 (3) (a) Two or more political subdivisions may create a commission for the purpose of issuing bonds exercising the powers granted under this section by entering into an agreement to do so under s. 66.0301 (2) or 66.0303 (2), except that upon its creation all of the initial members of -a- the commission shall be political subdivisions that are located in this state. A commission that is created as provided in this section is a unit of government, and a body corporate and politic, that is separate and distinct from, and independent of, the state and the political subdivisions which are parties to the agreement.

SECTION 984ge. 66.0304 (4) (a) of the statutes is amended to read:

66.0304 (4) (a) Adopt <u>and amend</u> bylaws, <u>policies</u>, <u>and procedures</u> for the regulation of its affairs and the conduct of its business.

SECTION 984gf. 66.0304 (4) (c) of the statutes is amended to read:

66.0304 (4) (c) Acquire, buy, own, operate, sell, lease as lessor or lessee, encumber, mortgage, hypothecate, pledge, assign, gift, or otherwise transfer any real, personal, tangible, or intangible property or interest in property that is located within or outside of this state.

SECTION 984gg. 66.0304 (4) (f) of the statutes is amended to read:

66.0304 (4) (f) Employ or appoint agents, employees, finance professionals, <u>counsel</u>, and special advisers as the commission finds necessary and fix their compensation.

SECTION 984gh. 66.0304 (4) (p) of the statutes is amended to read:

66.0304 **(4)** (p) Purchase bonds issued by or on behalf of, or held by, any participant, any state or a department, authority, or agency of the any state, or any political subdivision or subunit of a political subdivision, or the federal government or subunit of the federal government. Bonds purchased under this paragraph may be held by the commission or sold, in whole or in part, separately or together with other bonds issued by the commission.

SECTION 984gi. 66.0304 (4e) of the statutes is created to read:

66.0304 (4e) Creation of Business Entity. In this subsection, "business entity" means any nonprofit or forprofit corporation, limited liability company, partnership, or other business organization or entity. A commission may create one or more business entities of which the commission is the sole or controlling owner, member, manager, or partner, provided that the purpose of the business entity is to carry out or assist the commission in carrying out all or part of the commission's powers under sub. (4) with respect to projects located outside this state. Control may consist of the power to appoint a majority of, or veto any proposed appointment to, the governing body of a business entity created under this subsection. A business entity created under this subsection shall have such powers, consistent with the laws of the jurisdiction in which the business entity is organized, as are delegated to it by the commission and set forth in its organizational documents or in the resolution authorizing its creation. A business entity created under this subsection may be created or organized under the laws of this state or any state or territory of the United States. A business entity created under this subsection is entitled to the same exemptions and immunities that apply to a commission under this section. A business entity created under this subsection and a commission may make loans to, borrow money from, and acquire or assign or transfer property to or from, one another or any participant. A business entity created under this subsection is a legal entity that is separate and distinct from the commission, and its assets and liabilities may not be consolidated or commingled with those of a commission or any other business entity created under this subsection. A commission may not be held accountable for the actions, omissions, debts, or liabilities of any business entity created under this subsection. A business entity created under this subsection may not be held accountable for the actions, omissions, debts, or liabilities of the commission that creates it, or of any other business entity created under this subsection.

SECTION 984gj. 66.0304 (4s) of the statutes is created to read:

66.0304 (4s) Participation in Projects. In connection with a project located outside this state, the commission, directly or through a business entity created under sub. (4e), may participate in any new markets or other tax credit, subsidy, grant, loan, or credit enhancement program and may participate in any federal, state, or local government program established for the purpose of fostering economic development, including disaster relief, clean or renewable energy, housing assistance, water efficiency, transportation, or any other economic development in which the commission or a business entity created under sub. (4e) is eligible to participate, regardless of whether participation by the commission or a business entity involves the issuance of bonds by the commission or by any other issuer. In connection with the participation described in this subsection, the commission may exercise any or all of the powers under sub. (4) (c) and (g) Vetoed to (L), or it may delegate those powers to a business entity created under sub. (4e).

SECTION 984gk. 66.0304 (5) (a) 1. of the statutes is amended to read:

66.0304 (5) (a) 1. The face form of the bond shall include the date of issuance and the date of maturity.

SECTION 984gL. 66.0304 (5) (a) 2. of the statutes is amended to read:

66.0304 (5) (a) 2. The face form of the bond shall include the statements required under subs. (9) (c) and (11) (d).

SECTION 984gm. 66.0304 (5) (ae) of the statutes is created to read:

66.0304 (5) (ae) A bond resolution, trust agreement or indenture, or other agreement providing for issuance of the bonds may provide that the facsimile, electronic, or digital signature of any person authorized to execute documents, including bonds, on behalf of the commission shall be considered to be the legal equivalent of a manual signature on specified documents or all documents, and such signatures are valid and binding for all purposes.

SECTION 984gn. 66.0304 (5) (am) (intro.) of the statutes is amended to read:

66.0304 (5) (am) (intro.) Notwithstanding par. (a), as an alternative to specifying the matters required to be specified in the bond resolution under par. (a), the resolution may specify members of the board or officers or employees of the commission, by name or position, to whom the commission delegates authority to determine which of the matters under specified par. (a), and any other matters that the commission deems appropriate, for inclusion in the trust agreement, indenture, or other agreement providing for issuance of the bonds as finally executed. A resolution under this paragraph shall specify at least all of the following:

SECTION 984go. 66.0304 (5) (am) 3. of the statutes is amended to read:

66.0304 (5) (am) 3. The maximum interest rate to be borne by the bonds <u>expressed as a numerical percentage</u> and without regard to any penalty, default, or taxable rate that may be applicable to the bonds.

SECTION 984gp. 66.0304 (5) (b) 1. of the statutes is amended to read:

66.0304 **(5)** (b) 1. Early mandatory or optional redemption or purchase in lieu of redemption or tender, as provided in the resolution.

SECTION 984gq. 66.0304 (5) (b) 3. of the statutes is amended to read:

66.0304 (5) (b) 3. A trust agreement or indenture containing, or other agreement providing for issuance of the bonds, any of which contains any terms, conditions, and covenants that the commission determines to be necessary or appropriate, but such terms, conditions, and covenants may not be in conflict with the resolution.

In Part

Vetoed In Part

SECTION 984gqf. 66.0304 (5) (e) of the statutes is amended to read:

66.0304 (5) (e) The commission shall send notification to the department of administration and the department of revenue, on a form prescribed by the department of revenue, whenever a bond is issued under this section.

SECTION 984gr. 66.0304 (6) (e) of the statutes is repealed.

SECTION 984gs. 66.0304 (7) (a) of the statutes is amended to read:

66.0304 (7) (a) The commission may secure bonds by a trust agreement or, indenture by and between the commission and one or more corporate trustees, or other agreement providing for the issuance of the bonds. A bond resolution, trust agreement, or indenture, or other agreement providing for the issuance of the bonds may contain provisions for pledging the pledge or assignment by the commission of properties, revenues, and other tangible or intangible collateral, including contractual rights; holding and disbursing funds; protecting and enforcing the rights and remedies of bondholders; restricting individual rights of action by bondholders; and amendments, and any other provisions the commission determines to be reasonable and proper for the security of the bondholders or contracts entered into under this section in connection with the bonds.

SECTION 984gt. 66.0304 (8) of the statutes is amended to read:

66.0304 (8) NO PERSONAL LIABILITY. No board member director, officer, employee, or agent of the commission, of any member, or of a business entity created under sub. (4e) is liable personally on the bonds or any contract entered into by the commission or business entity or subject to any personal liability or accountability by reason of the contract or the issuance of the bonds, unless the personal liability or accountability is the result of the willful misconduct of such person.

SECTION 984gu. 66.0304 (9) (b) of the statutes is amended to read:

66.0304 (9) (b) The state and the political subdivisions who are parties to the agreement creating a commission under this section, the members, and the political subdivisions approving financing under sub. (11) (a) are not liable on bonds or any other contract entered into under this section, or for any other debt, obligation, or liability of the commission or a business entity created under sub. (4e), whether in tort, contract, or otherwise.

SECTION 984gv. 66.0304 (9) (c) of the statutes is amended to read:

66.0304 (9) (c) The bonds are not a debt of the state or the political subdivisions contracting to create a commission under this section, the members, or the political subdivisions approving financing under sub. (11) (a). A bond issue under this section does not obligate the state or a political subdivision to levy any tax or make any appropriation for payment of the bonds. All bonds issued

by a commission are payable solely from the funds Vetoed pledged for their payment in accordance with the bond resolution or, trust agreement or, indenture, or other agreement providing for their issuance. All bonds shall contain, on their face, a statement regarding the obligations of the state, the political subdivisions who are parties to the agreement creating the commission, the members, the political subdivisions approving financing under sub. (11) (a), and the commission as set forth in this paragraph.

SECTION 984gw. 66.0304 (9) (d) of the statutes is created to read:

66.0304 (9) (d) Projects not located in this state that are financed or refinanced by bonds of a commission, including any project owned, operated, leased from or to, or otherwise controlled by a participant or by the commission, are not considered public projects of this state, and are not subject to procurement, contracting, construction, tax, acquisition, construction, or improvements laws of this state that are applicable to public proj-

SECTION 984gx. 66.0304 (10) (b) of the statutes is amended to read:

66.0304 (10) (b) A commission shall maintain an accounting system in accordance with generally accepted accounting principles and shall have its financial statements and debt covenants audited annually by an independent certified public accountant, except that the commission by a unanimous vote may decide to have an audit performed under this paragraph every 2 years.

SECTION 984gy. 66.0304 (10) (d) of the statutes is created to read:

66.0304 (10) (d) Within 30 days of the close of each calendar quarter, a commission and any entity created under sub. (4e) shall file a report with the secretary of administration and the legislative audit bureau containing information showing the amount of bonds issued by the commission and any such entity in the previous quarter, the names of the borrowers and the project associated with the bonds, the types of bonds that were issued, the location of the project associated with the bonds, and a statement of the bond issuance fees that the commission and any such entity received in relation to each of the bond issues identified in the report.

SECTION 984h. 66.0304 (11) (a) of the statutes is renumbered 66.0304 (11) (a) 1. (intro.) and amended to read:

66.0304 (11) (a) 1. (intro.) -A Except as provided in subd. 2., a commission may not issue bonds to finance a capital improvement project in any state or territory of the United States unless a political subdivision within whose boundaries the project is to be located has approved the financing of the project. A commission may not issue bonds to finance a capital improvement project in this state unless all of the political subdivisions within whose boundaries the project is to be located has

have approved the financing of the project. An approval under this paragraph subdivision may be made by the one of the following:

a. The governing body of the political subdivision or, except or its designee.

b. Except for a 1st class city or a county in which a 1st class city is located, by the highest ranking executive or administrator of the political subdivision or his or her

SECTION 984hb. 66.0304 (11) (a) 1. c. of the statutes is created to read:

66.0304 (11) (a) 1. c. An applicable elected representative of the political subdivision, if any, as defined in section 147 (f) (2) (E) of the Internal Revenue Code, except that for a 1st class city, or a county in which a 1st class city is located, such approval may be given only by the governing body of the city or county.

SECTION 984hc. 66.0304 (11) (a) 2. of the statutes is created to read:

66.0304 (11) (a) 2. Except for financing a capital improvement project in this state, the commission may issue bonds to finance a capital improvement project without receiving the approval under subd. 1. if the financing is approved in accordance with section 147 (f) of the Internal Revenue Code.

SECTION 984hd. 66.0304 (11) (a) 3. of the statutes is created to read:

66.0304 (11) (a) 3. Bonds issued under this section are not considered issued for the purpose of financing a capital improvement project if the bond proceeds are used for any of the following purposes:

- a. To finance a facility if the facility was placed in service for federal tax purposes by the participant or a related person prior to the commission issuing the bonds and if no more than 10 percent of the bond proceeds are used to finance the construction, expansion, rehabilitation, renovation, or remodeling of capital improvements.
- b. To finance the acquisition of a facility, by a participant or by the commission, if no more than 10 percent of the bond proceeds are used to finance the construction, expansion, rehabilitation, renovation, or remodeling of the facility.
- c. To finance the commission's purchase either of bonds issued by a different issuer or of leases or contracts from a 3rd-party provider, and those bonds, leases, or contracts are used or were used to finance in whole or in part the construction, expansion, rehabilitation, renovation, or remodeling of real or tangible personal property.

SECTION 984he. 66.0304 (11) (bm) of the statutes is amended to read:

66.0304 (11) (bm) A project may be located outside of the United States or outside a territory of the United States if the any participant or borrower, including a coborrower, of proceeds of bonds issued to finance or refinance the project in whole or in part is incorporated organized under the laws of and has its principal place of business in any state or territory of the United States or Vetoed a territory of the United States. To the extent that this paragraph applies to a borrower, it also applies to a participant if the participant is a nongovernmental entity.

SECTION 984hf. 66.0304 (11) (c) of the statutes is amended to read:

66.0304 (11) (c) Any action brought to challenge the validity of the issuance of a bond under this section, or the enforceability of a contract entered into under this section, must be commenced in circuit court within 30 days of the commission adopting a resolution authorizing the issuance of the bond or the execution of the contract or be barred. Section 893.77 does not apply to bonds issued under this section.

SECTION 984hg. 66.0304 (12) of the statutes is amended to read:

66.0304 (12) STATE PLEDGE. The state pledges to and agrees with the bondholders, and persons that enter into contracts with a commission under this section, that the state will not limit, impair, or alter the rights and powers vested in a commission by this section, including the rights and powers under sub. (4), before the commission has met and discharged the bonds, and any interest due on the bonds, and has fully performed its contracts, unless adequate provision is made by law for the protection of the bondholders or those entering into contracts with a commission. The commission may include this pledge in a contract with bondholders. The pledge and agreement described in this subsection do not create any liability on any bonds or contracts of the commission on the part of the state, the members, or any other political subdivision of the state, or any political subdivision approving financing under sub. (11) (a), which liability shall be expressly limited as provided in sub. (9).

SECTION 984ht. 66.0307 (3) (d) 4m. of the statutes is created to read:

66.0307 (3) (d) 4m. Identify all highways within the territory covered by the plan of which each participating municipality has jurisdiction.

SECTION 984i. 66.0307 (6) of the statutes is amended to read:

66.0307 (6) BINDING ELEMENTS OF COOPERATIVE PLAN. If a cooperative plan is approved by the department under sub. (5) or an amended plan is approved under sub. (8), provisions in the plan to maintain existing boundaries, the boundary changes in the plan, the schedule for those changes, the plan for delivery of services, including road maintenance, and the schedule for those services are binding on the parties to the plan and have the force and effect of a contract.

Section 984ig. 66.0414 of the statutes is created to **Vetoed** read:

66.0414 Quarry operations. (1) Construction. (a) Nothing in this section may be construed to affect the authority of a political subdivision to regulate land use for a purpose other than quarry operations.

In Part

In Part

- (b) Nothing in this section may be construed to exempt a quarry from a regulation of general applicability placed by a political subdivision that applies to other property in the political subdivision that is not a quarry unless the regulation is inconsistent with this section.
- (c) Except for making unenforceable, under sub. (4) (b), (d), or (f), an ordinance or other limit on quarry operations, this section may not be interpreted to affect a legal claim that involves an ordinance or other limit on quarry operations that is in effect on January 1, 2017.
 - (2) DEFINITIONS. In this section:
- (a) "Permit" means a form of approval granted by a political subdivision for the operation of a quarry.
- (b) "Political subdivision" means a city, village, town, or county.
- (c) "Public works project" means a federal, state, county, or municipal project that involves the construction, maintenance, or repair of a public transportation facility or other public infrastructure and in which non-metallic minerals are used.
- (d) "Quarry" means the surface area from which nonmetallic minerals, including soil, clay, sand, gravel, and construction aggregate, that are used primarily for a public works project or a private construction or transportation project are extracted and processed.
- (e) "Quarry operations" means the extraction and processing of minerals at a quarry and all related activities, including blasting, vehicle and equipment access to the quarry, and loading and hauling of material to and from the quarry.
- (3) LIMITATIONS ON LOCAL REGULATION. (a) *Permits*.

 1. Consistent with the requirements and limitations in this subsection, except as provided in subd. 2. or 3., a political subdivision may require a quarry operator to obtain a zoning or non–zoning permit to conduct quarry operations.
- 2. a. Except as provided in subd. 2. b., a political subdivision may not require a quarry operator to obtain a zoning or non–zoning permit if the quarry operator conducts quarry operations at a quarry unless the political subdivision enacts an ordinance that requires the permit.
- b. A political subdivision may require a quarry operator to obtain a permit to conduct quarry operations at a nonconforming quarry site, as defined in s. 59.69 (10) (ab) 2., if quarry operations at the nonconforming quarry site have been previously discontinued for a period of 12 consecutive months after the political subdivision enacted the permit requirement.
- 3. A political subdivision may not require a quarry operator to obtain a zoning or non–zoning permit to conduct quarry operations unless the political subdivision enacts an ordinance that requires the permit.
- (b) Applicability of local limit. If a political subdivision enacts a non–zoning ordinance regulating the operation of a quarry that was not in effect when quarry opera-

- tions began at a quarry, the limit does not apply to that quarry or to land that, as of the effective date of this subdivision [LRB inserts date], is contiguous, as defined in s. 59.69 (10) (ab) 1., to the land on which the quarry is located, is under the common ownership, leasehold, or control of the person who owns, leases, or controls the land on which the quarry is located, and is located in the same political subdivision.
 - (c) *Blasting*. 1. In this paragraph:
- a. "Affected area" means an area within a certain radius of a blasting site that may be affected by a blasting operation, as determined using a formula established by the department of safety and professional services by rule that takes into account a scaled—distance factor and the weight of explosives to be used.
- b. "Airblast" means an airborne shock wave caused by a blast.
- c. "Flyrock" means rock that is propelled through the air as a result of a blast.
- d. "Ground vibration" means a shaking of the ground caused by the elastic wave emanating from a blast.
- 2. Except as provided under subds. 3. and 4., a political subdivision may not limit blasting at a quarry.
- 3. A political subdivision may require the operator of a quarry to do any of the following:
- a. Before beginning a blasting operation at the quarry, provide notice of the blasting operation to each political subdivision in which any part of the quarry is located and to owners of dwellings or other structures within the affected area.
- b. Before beginning a blasting operation at the quarry, cause a 3rd party to conduct a building survey of any dwellings or other structures within the affected area.
- c. Before beginning a blasting operation at the quarry, cause a 3rd party to conduct a survey of and test any wells within the affected area.
- d. Maintain records and prepare and submit reports related to blasting operations at the quarry.
- e. Comply with other properly adopted local blasting regulations that are not related to airblast, flyrock, or ground vibration.
- 4. A political subdivision may suspend a permit for a violation of the requirements under s. 101.15 relating to blasting and rules promulgated by the department of safety and professional services under s. 101.15 (2) (e) relating to blasting only if the department of safety and professional services determines that a violation of the requirements or rules has occurred and only for the duration of the violation as determined by the department of safety and professional services.
- (d) Water quality or quantity. 1. Except as provided under subds. 2. to 5., a political subdivision may not do any of the following with respect to the operation of a quarry:
 - a. Establish or enforce a water quality standard.

In Part

Vetoed In Part

- b. Issue permits, including permits for discharges to the waters of the state, or any other form of approval related to water quality or quantity.
- c. Impose any restriction related to water quality or quantity.
- d. Impose any requirements related to monitoring of water quality or quantity.
- 2. A political subdivision may require the operator of a quarry to conduct and provide water quality and quantity baseline testing and ongoing quality testing, to occur not more frequently than annually, of all wells within 1,000 feet of the perimeter of a quarry site when a new high capacity well is added to an existing quarry site or a new quarry site is established. A testing requirement under this subdivision may not impose any standard that is more stringent than the standards for groundwater quality required by rules promulgated by the department of natural resources. The political subdivision may request a report of well testing results within 30 days of the completion of testing and the quarry operator shall provide the results within that time. Any person offered the opportunity to have a well tested under this subdivision but who knowingly refuses testing waives any claim against a quarry operator related to the condition of the well if, within 90 days of the offer, the quarry operator records with the register of deeds for the county in which the well is located a written and sworn certification that the person refused the offer.
- 3. A political subdivision that imposes a requirement to conduct any ongoing water quality or quantity testing of wells adjacent to existing quarry sites prior to the effective date of this subdivision [LRB inserts date], may continue to do so.
- 4. A political subdivision may take actions related to water quality or quantity that are specifically required or authorized by state law.
- 5. A political subdivision may enforce properly adopted local water regulations but may suspend a permit for a violation of state law or rules promulgated by the department of natural resources relating to water quality or quantity only if the department of natural resources determines that a violation of state law or rules has occurred and only for the duration of the violation, as determined by the department of natural resources.
- (e) Air quality. 1. Notwithstanding s. 285.73, and except as provided under subds. 2. to 4., a political subdivision may not do any of the following with respect to the operation of a quarry:
- a. Establish or enforce an ambient air quality standard, standard of performance for new stationary sources, or other emission limitation related to air quality.
- b. Issue permits or any other form of approval related to air quality.
 - c. Impose any restriction related to air quality.
- d. Impose any requirement related to monitoring air quality.

- 2. A political subdivision may require the operator of **Vetoed** a quarry to use best management practices to limit offsite fugitive dust and may enforce properly adopted fugitive dust regulations.
- 3. A political subdivision may take actions related to air quality that are specifically required or authorized by state law.
- 4. A political subdivision may suspend a permit for a violation of state law or rules promulgated by the department of natural resources relating to air quality only if the department of natural resources determines that a violation of state law or rules has occurred and only for the duration of the violation, as determined by the department of natural resources.
- (f) Noise. A political subdivision may not limit the noise emitted from a quarry site, as measured off the property where the quarry is located without the use of a hearing protector, to be less than 76.5 percent of the decibel standards established under 30 CFR 62.100 to 62.190.
- (g) Quarry production. A political subdivision may not limit any of the following:
- 1. The quantity of minerals extracted from or processed by a quarry.
 - 2. The depth of mineral extraction at a quarry.
- 3. The number of truck loads that exits a quarry or the number of trucks that enters a quarry unless the purpose of the limit is to protect the structural condition of a roadway within the political subdivision.
 - 4. The times that any of the following may occur:
- a. Quarry operations if the materials produced by the quarry will be used in a public works project that requires construction work to be performed during the night or an emergency repair except that a political subdivision may limit the number of consecutive days that a quarry operator may conduct quarry operations during the hours of darkness to 5 consecutive days.
- b. The transportation of unloaded equipment within a quarry.
- c. Maintenance of vehicles, equipment, or buildings at a quarry.
 - d. Administrative activities at a quarry.
- e. Entry of unloaded trucks into a quarry at the times during which a quarry is permitted to operate unless the purpose of the limit is to protect the structural condition of a roadway within the political subdivision.
- 5. The hours of quarry operations conducted at a quarry to less than 72 hours per week, excluding hours on Sundays and holidays.
- (h) Setbacks. 1. A political subdivision may not establish a setback requirement for quarry operations that is more than 200 feet from the boundary of the property of a quarry.
- 2. Notwithstanding subd. 1., a political subdivision that enacts an ordinance imposing setback requirements shall allow a quarry operator to conduct quarry operations nearer to the boundary of the property of the quarry

In Part

Vetoed In Part than the distance of the setback requirement if all of the following apply:

- a. Each property owner of a lot that is located within 200 feet of the boundary of a quarry consents in writing to that conduct of quarry operations.
- b. The guarry operator provides the clerk of the political subdivision with a copy of the written agreement under subd. 2. a.
- c. The quarry operator records the written agreement under subd. 2. a. against the property described in subd. 2. a. in the office of the register of deeds for the county in which the land is located.
- (i) Quarry permit requirements. 1. A political subdivision may not add a condition to a permit during the duration of the permit unless the permit holder consents.
- 2. If a political subdivision requires a quarry to comply with another political subdivision's ordinance as a condition for obtaining a permit, the political subdivision that grants the permit may not require the quarry operator to comply with a provision of the other political subdivision's ordinance that is enacted after the permit is granted and while the permit is in effect.
- 3. a. A town may not require, as a condition for granting a permit to a quarry operator, that the quarry operator satisfy a condition that a county requires in order to grant a permit that is imposed by a county ordinance enacted after the county grants a permit to the quarry operator.
- b. A county may not require, as a condition for granting a permit to a quarry operator, that the quarry operator satisfy a condition that a town requires in order to grant a permit that is imposed by a town ordinance enacted after the town grants a permit to the quarry operator.
- 4. a. Except as provided in subd. 4. b., a political subdivision shall, upon submission of a permit renewal application by a quarry operator, renew the permit if the permit has a duration of less than 10 years. As a condition of renewing a permit, a political subdivision may require that a quarry operator satisfy a condition that the law authorizes the political subdivision to require.
- b. A political subdivision may deny the renewal of a permit having a duration of less than 10 years if the permit holder fails to cure a material violation of a condition of the permit after reasonable notice from the political subdivision of the violation and a reasonable opportunity for the operator to cure the violation.
- (k) Mining permit requirements. A political subdivision may not impose a condition on a permit for quarry operations that is inconsistent with the requirements of this section or s. 295.12.
- (4) Previous restrictions. (a) Except as provided in par. (b) or (d), and notwithstanding sub. (3), if a political subdivision has in effect on January 1, 2017, an ordinance that is more restrictive than this section, the political subdivision may maintain and enforce that ordinance.
- (b) If a political subdivision has in effect on January 1, 2017, an ordinance that contains a prohibition or

requirement that violates the prohibition or limitation Vetoed under sub. (3) (c) 2., (d) 1., (e) 1., or (g) 4. a., the prohibition or requirement does not apply and may not be enforced.

- (c) Except as provided in par. (d), and notwithstanding sub. (3), if a political subdivision has in effect on January 1, 2017, a requirement, not based on the political subdivision's authority under ch. 295, that a quarry operator obtain a non-zoning permit that is more restrictive than this section, the political subdivision may maintain and enforce that requirement if the political subdivision had authority to impose that requirement.
- (d) A requirement described under par. (c) that violates the prohibition or limitation under sub. (3) (c) 2., (d) 1., (e) 1., or (g) 4. a. does not apply and may not be enforced.
- (e) Notwithstanding sub. (3), a zoning or non–zoning permit that is held by a quarry operator and in effect on January 1, 2017, remains in effect for the duration of the permit.
- (f) A condition that a political subdivision requires to be satisfied in order to grant a zoning or non-zoning permit that is in effect on January 1, 2017, does not apply and may not be enforced if either of the following applies:
- 1. The political subdivision does not have authority to require that the condition be satisfied in order to grant the zoning or non-zoning permit.
- 2. The condition violates the prohibition or limitation under sub. (3) (c) 2., (d) 1., (e) 1., or (g) 4. a.

SECTION 984ij. 66.0414 (3) (f) of the statutes, as created by 2017 Wisconsin Act (this act), is amended to read:

66.0414 (3) (f) Noise. A political subdivision may not limit the noise emitted from a quarry, as measured off the property where the quarry is located without the use of a hearing protector, to be less than 76.5 percent of the decibel standards established under 30 CFR 62.100 to 62.190. A political subdivision may require trucks and other equipment that are owned or controlled by a quarry operator, when used in quarry operations during the hours of darkness, to use a white noise alarm instead of a beeping alarm for worker and vehicle safety.

SECTION 984j. 66.0433 (title) of the statutes is amended to read:

66.0433 (title) Licenses for nonintoxicating and soda water beverages.

SECTION 984jb. 66.0433 (1) (title) of the statutes is repealed.

SECTION 984jc. 66.0433 (1) of the statutes is renumbered 66.0433.

SECTION 984jd. 66.0433 (2) of the statutes is repealed.

SECTION 984p. 66.0510 of the statutes is created to

66.0510 Benefits to officers, employees, agents. (1) DEFINITIONS. In this section:

- (a) "Employee benefit plan" means a plan as defined in 29 USC 1002 (3).
- (b) "Local governmental unit" has the definition given in s. 66.0131 (1) (a).
- (2) BENEFITS. If a local governmental unit provides an employee benefit plan to its officers, agents, and employees, the plan may cover only such officers, agents, and employees and their spouses and dependent children.

SECTION 984pg. 66.0602 (2) of the statutes is renumbered 66.0602 (2) (a) and amended to read:

66.0602 (2) (a) Except as provided in subs. (3), (4), and (5), no political subdivision may increase its levy in any year by a percentage that exceeds the political subdivision's valuation factor. The Except as provided in par. (b), the base amount in any year, to which the limit under this section applies, shall be the actual levy for the immediately preceding year. In determining its levy in any year, a city, village, or town shall subtract any tax increment that is calculated under s. 59.57 (3) (a), 60.85 (1) (L), or 66.1105 (2) (i). The base amount in any year, to which the limit under this section applies, may not include any amount to which sub. (3) (e) 8. applies.

SECTION 984q. 66.0602 (2) (b) of the statutes is created to read:

66.0602 (2) (b) For purposes of par. (a), in 2018, and in each year thereafter, the base amount to which the limit under this section applies is the actual levy for the immediately preceding year, plus the amount of the payment under s. 79.096, and the levy limit is the base amount multiplied by the valuation factor, minus the amount of the payment under s. 79.096.

SECTION 985. 66.0602 (2m) (a) of the statutes is amended to read:

66.0602 (**2m**) (a) If a political subdivision's levy for the payment of any general obligation debt service, including debt service on debt issued or reissued to fund or refund outstanding obligations of the political subdivision and interest on outstanding obligations of the political subdivision, on debt originally issued before July 1, 2005, is less in the current year than it was in the previous year, the political subdivision shall reduce its levy limit in the current year by an amount equal to the amount that its levy was reduced as described in this subsection. This subsection does not apply to a political subdivision in any year in which the political subdivision does not increase its levy increase limit as allowed under sub. (3) (f) 1.

SECTION 985c. 66.0602 (2m) (b) 1. of the statutes is amended to read:

66.0602 (**2m**) (b) 1. In this paragraph, "covered service" means garbage collection, fire protection, snow plowing, street sweeping, or storm water management, except that garbage collection may not be a covered service for any political subdivision that owned and operated a landfill on January 1, 2013. With regard to fire protection, "covered service" does not include the

production, storage, transmission, sale and delivery, or furnishing of water for public fire protection purposes.

SECTION 985e. 66.0602 (4) (a) of the statutes is amended to read:

66.0602 (4) (a) A political subdivision may exceed the levy increase limit under sub. (2) if its governing body adopts a resolution to that effect and if the resolution is approved in a referendum. The resolution shall specify the proposed amount of increase in the levy beyond the amount that is allowed under sub. (2), the purpose for which the increase will be used, and shall specify whether the proposed amount of increase is for the next fiscal year only or if it will apply on an ongoing basis. With regard to a referendum relating to the 2005 levy, or any levy in an odd-numbered year thereafter, the political subdivision may call a special referendum for the purpose of submitting the resolution to the electors of the political subdivision for approval or rejection. With regard to a referendum relating to the 2006 levy, or any levy in an even-numbered year thereafter, the referendum shall be held at the next succeeding spring primary or election or partisan primary or general election.

SECTION 985g. 66.0602 (4) (c) of the statutes is amended to read:

66.0602 (4) (c) The referendum shall be held in accordance with chs. 5 to 12. The political subdivision shall provide the election officials with all necessary election supplies. The form of the ballot shall correspond substantially with the standard form for referendum ballots under ss. 5.64 (2) and 7.08 (1) (a). The question shall be submitted as follows: "Under state law, the increase in the levy of the (name of political subdivision) for the tax to be imposed for the next fiscal year, (year), is limited to%, which results in a levy of \$.... Shall the (name of political subdivision) be allowed to exceed this limit and increase the levy for the next fiscal year, (year), for (purpose for which the increase will be used), by a total of%, which results in a levy of \$....?". In preparing the ballot question for a referendum held at a partisan primary in 2014, as it relates to the allowable amount of levy rate increase and the total amount of a levy, a county with a population of at least 30,000, but no more than 40,000, that is adjacent to a county with a population exceeding 450,000, shall use the most recent data that it has and the most recent data that is available from the department of revenue.

SECTION 985h. 66.0603 (1m) (g) of the statutes is created to read:

66.0603 (1m) (g) A technical college district that receives funds from participation in an auction of digital broadcast spectrum administered by the federal communications commission may hold those funds in trust and may invest and reinvest those funds in the same manner authorized for investments under s. 881.01. Funds held in trust under this paragraph may only be distributed from

the trust in a manner consistent with ch. 38 and in accordance with the terms of the trust. Any trust formed pursuant to this paragraph shall be separate from any other trust created by, or under the control of, the technical college district.

SECTION 985k. 66.0603 (3) (d) of the statutes is created to read:

66.0603 (3) (d) 1. In addition to the authority granted under sub. (2), a technical college district may delegate the investment authority over the funds described under sub. (1m) (g) to an investment manager who meets the requirements and qualifications specified in the trust's investment policy and who is registered as an investment adviser under the Investment Advisers Act of 1940, 15 USC 80b-3.

2. If a technical college district has established a trust described in sub. (1m) (g), it shall annually publish a written report that states the amount in the trust, the investment return earned by the trust since the last report was published, the total disbursements made from the trust since the last report was published, and the name of the investment manager if investment authority has been delegated under subd. 1.

SECTION 985L. 66.0615 (1) (bs) of the statutes is created to read:

66.0615 (1) (bs) "Lodging marketplace" means an entity that provides a platform through which an unaffiliated 3rd party offers to rent a short–term rental to an occupant and collects the consideration for the rental from the occupant.

SECTION 985m. 66.0615 (1) (de) of the statutes is created to read:

66.0615 (1) (de) "Occupant" means a person who rents a short–term rental through a lodging marketplace.

SECTION 985n. 66.0615 (1) (df) of the statutes is created to read:

66.0615 (1) (df) "Owner" means the person who owns the residential dwelling that has been rented.

SECTION 9850. 66.0615 (1) (di) of the statutes is created to read:

66.0615 (1) (di) "Residential dwelling" means any building, structure, or part of the building or structure, that is used or intended to be used as a home, residence, or sleeping place by one person or by 2 or more persons maintaining a common household, to the exclusion of all others.

SECTION 985p. 66.0615 (1) (dk) of the statutes is created to read:

66.0615 (1) (dk) "Short-term rental" means a residential dwelling that is offered for rent for a fee and for fewer than 29 consecutive days.

SECTION 985q. 66.0615 (1m) (a) of the statutes is amended to read:

66.0615 (1m) (a) The governing body of a municipality may enact an ordinance, and a district, under par.

(e), may adopt a resolution, imposing a tax on the privilege of furnishing, at retail, except sales for resale, rooms or lodging to transients by hotelkeepers, motel operators. lodging marketplaces, owners of short-term rentals, and other persons furnishing accommodations that are available to the public, irrespective of whether membership is required for use of the accommodations. A tax imposed under this paragraph may be collected from the consumer or user, but may not be imposed on sales to the federal government and persons listed under s. 77.54 (9a). A tax imposed under this paragraph by a municipality shall be paid to the municipality and, with regard to any tax revenue that may not be retained by the municipality, shall be forwarded to a tourism entity or a commission if one is created under par. (c), as provided in par. (d). Except as provided in par. (am), a tax imposed under this paragraph by a municipality may not exceed 8 percent. Except as provided in par. (am), if a tax greater than 8 percent under this paragraph is in effect on May 13, 1994, the municipality imposing the tax shall reduce the tax to 8 percent, effective on June 1, 1994.

SECTION 985r. 66.0615 (5) of the statutes is created to read:

66.0615 (5) (a) A lodging marketplace shall register with the department of revenue, on forms prepared by the department, for a license to collect taxes imposed by the state related to a short–term rental and to collect room taxes imposed by a municipality. After a lodging marketplace applies for and receives such a license, it shall do all of the following:

- 1. If a short–term rental is rented through the lodging marketplace, collect sales and use taxes from the occupant and forward such amounts to the department of revenue.
- 2. If a short–term rental that is rented through the lodging marketplace is located in a municipality that imposes a room tax, collect the room tax from the occupant and forward it to the municipality.
- 3. Notify the owner of a short–term rental that the lodging marketplace has collected and forwarded the taxes described in subds. 1. and 2.
- (b) A municipality may not impose and collect a room tax from the owner of a short–term rental if the municipality collects the room tax on the residential dwelling under par. (a) 2.

SECTION 993. 66.0903 (1) (c) of the statutes is amended to read:

66.0903 (1) (c) "Hourly basic rate of pay" has the meaning given in s. 16.856 (1) (b), 2015 stats.

SECTION 994. 66.0903 (1) (f) of the statutes is amended to read:

66.0903 (1) (f) "Prevailing hours of labor" has the meaning given in s. 16.856 (1) (e), 2015 stats.

SECTION 995. 66.0903 (1) (g) of the statutes is amended to read:

66.0903 (1) (g) "Prevailing wage rate" includes the meanings given under s. 66.0903 (1) (g), 2013 stats., and s. 16.856 (1) (f), 2015 stats.

SECTION 996. 66.0903 (1) (j) of the statutes is amended to read:

66.0903 (1) (j) "Truck driver" has the meaning given in s. 16.856 (1) (j) includes an owner–operator of a truck.

SECTION 996g. 66.1014 of the statutes is created to read:

66.1014 Limits on residential dwelling rental prohibited. (1) In this section:

- (a) "Political subdivision" means any city, village, town, or county.
- (b) "Residential dwelling" means any building, structure, or part of the building or structure, that is used or intended to be used as a home, residence, or sleeping place by one person or by 2 or more persons maintaining a common household, to the exclusion of all others.
- (2) (a) Subject par. (d), a political subdivision may not enact or enforce an ordinance that prohibits the rental of a residential dwelling for 7 consecutive days or longer.
- (b) If a political subdivision has in effect on the effective date of this paragraph [LRB inserts date], an ordinance that is inconsistent with par. (a) or (d), the ordinance does not apply and may not be enforced.
- (c) Nothing in this subsection limits the authority of a political subdivision to enact an ordinance regulating the rental of a residential dwelling in a manner that is not inconsistent with the provisions of pars. (a) and (d).
- (d) 1. If a residential dwelling is rented for periods of more than 6 but fewer than 29 consecutive days, a political subdivision may limit the total number of days within any consecutive 365–day period that the dwelling may be rented to no fewer than 180 days. The political subdivision may not specify the period of time during which the residential dwelling may be rented, but the political subdivision may require that the maximum number of allowable rental days within a 365–day period must run consecutively. A person who rents the person's residential dwelling shall notify the clerk of the political subdivision in writing when the first rental within a 365–day period begins.
- 2. Any person who maintains, manages, or operates a short–term rental, as defined in s. 66.0615 (1) (dk), for more than 10 nights each year, shall do all of the following:
- a. Obtain from the department of agriculture, trade and consumer protection a license as a tourist rooming house, as defined in s. 97.01 (15k).
- b. Obtain from a political subdivision a license for conducting such activities, if a political subdivision enacts an ordinance requiring such a person to obtain a license

SECTION 996h. 66.1105 (2) (f) 2. e. of the statutes is created to read:

66.1105 (2) (f) 2. e. For a tax incremental district in the city of Milwaukee, direct or indirect expenses related to operating a rail fixed guideway transportation system, as defined in s. 85.066 (1), in the city of Milwaukee.

SECTION 996j. 66.1105 (5) (h) 5. of the statutes is created to read:

66.1105 (**5**) (h) 5. Notwithstanding the 2 consecutive year provision described in subd. 1., the village of Kimberly may adopt a resolution and proceed under this paragraph with regard to Tax Incremental District Number 6, which was created on September 12, 2016. To act under this subdivision, the village of Kimberly must adopt a resolution under subd. 1. not later than September 30, 2017, and shall provide the department of revenue with all required materials no later than October 31, 2017.

SECTION 996p. 66.1109 (5) (d) of the statutes is created to read:

66.1109 (5) (d) If real property that is specially assessed as authorized under this section is of mixed use such that part of the real property is exempted from general property taxes under s. 70.11 or is residential, or both, and part of the real property is taxable, the municipality may specially assess as authorized under this section only the percentage of the real property that is not tax—exempt or residential. This paragraph applies only to a 1st class city.

SECTION 996pm. 67.05 (6a) (a) 2. (intro.) of the statutes is amended to read:

67.05 (**6a**) (a) 2. (intro.) Except as provided under pars. (b) and (c) and subs. (7) and (15), and subject to the limit on the number of referendums that may be called in any calendar year under subd. 2. a., if the board of any school district, or the electors at a regularly called school district meeting, by a majority vote adopt an initial resolution to raise an amount of money by a bond issue, the school district clerk shall, within 10 days, publish notice of such adoption as a class 1 notice under ch. 985 or post the notice as provided under s. 10.05. The notice shall state the maximum amount proposed to be borrowed, the purpose of the borrowing, that the resolution was adopted under this subdivision and the place where and the hours during which the resolution may be inspected. The school board shall also do one of the following:

SECTION 996pr. 67.05 (6a) (a) 2. a. of the statutes is amended to read:

67.05 (**6a**) (a) 2. a. Direct the school district clerk to call a special election for the purpose of submitting submit the resolution to the electors for approval or rejection, or direct that the resolution be submitted at the next regularly scheduled spring primary or election or partisan primary or general election, provided such election is to be held not earlier than 70 days after the adoption of the resolution. The school board may direct the school district clerk to call a special election on the Tuesday after the first Monday in November in an odd—numbered year for

Vetoed In Part

the purpose of submitting the resolution to the electors for approval or rejection, provided the special election is held not earlier than 70 days after the adoption of the resolution. A school board may proceed under this subd. 2. a. and under s. 121.91 (3) (a) 1. no more than 2 times in any calendar year. The resolution shall not be effective unless adopted by a majority of the school district electors voting at the referendum.

SECTION 996q. 67.05 (6a) (a) 2. c. of the statutes is created to read:

67.05 (**6a**) (a) 2. c. For a school district that has experienced a natural disaster, including a fire, that causes the school district's costs to increase, direct the school district clerk to call a special referendum to be held within the 6-month period immediately following the natural disaster, provided the special referendum is to be held not sooner than 70 days after the adoption of the initial resolution. The resolution shall not be effective unless adopted by a majority of the school district electors voting at the referendum.

SECTION 996t. 67.05 (6a) (am) 1. of the statutes is amended to read:

67.05 (**6a**) (am) 1. If the public hearing under par. (a) 2. b. is for informational purposes only and, within 30 days after the public hearing, a petition is filed with the school district clerk for a referendum on the resolution signed by at least 7,500 electors of the school district or at least 20 percent of the school district electors, as determined under s. 115.01 (13), whichever is less, the resolution shall not be effective unless adopted by a majority of the school district electors voting at the referendum. Subject to the limit therein, the school board shall hold the referendum in accordance with par. (a) 2. a. The question submitted shall be whether the initial resolution shall or shall not be approved.

SECTION 996u. 67.12 (12) (a) of the statutes is amended to read:

67.12 (12) (a) Any municipality may issue promissory notes as evidence of indebtedness for any public purpose, as defined in s. 67.04 (1) (b), including but not limited to paying any general and current municipal expense, and refunding any municipal obligations, including interest on them. Each note, plus interest if any, shall be repaid within 10 years after the original date of the note, except that notes issued under this section for purposes of ss. 119.498, 145.245 (12m), 281.58, 281.59, 281.60, 281.61, and 292.72, issued to raise funds to pay a portion of the capital costs of a metropolitan sewerage district, or issued by a 1st class city or a county having a population of 500,000 or more, to pay unfunded prior service liability with respect to an employee retirement system, shall be repaid within 20 years after the original date of the note.

SECTION 996w. 67.12 (12) (h) of the statutes is amended to read:

67.12 (12) (h) Paragraph (e) 2. does not apply to borrowing by the school board of a school district created by a reorganization under s. 117.105, or by the school board from which territory is detached to create a school district under s. 117.105, for the purpose of financing any assets or liabilities apportioned to the school district or assets apportioned to another school district under s. 117.105 (1m) er. (2m), or (4m).

SECTION 997d. 70.11 (3) (a) of the statutes is renumbered 70.11 (3) (a) 1. and amended to read:

70.11 (3) (a) 1. Grounds Except as provided in subd. 2., grounds of any incorporated college or university, not exceeding 80 acres.

SECTION 997e. 70.11 (3) (a) 2. of the statutes is created to read:

70.11 (3) (a) 2. Grounds of any incorporated college or university, not exceeding 150 acres, if the college or university satisfies all of the following criteria:

- a. It is a nonprofit organization.
- b. It was founded before January 1, 1900.
- c. Its total annual undergraduate enrollment is at least 5,000 students, not including students receiving online instruction only.

SECTION 997em. 70.11 (4) (a) of the statutes is renumbered 70.11 (4) (a) 1.

SECTION 997f. 70.11 (4) (a) 2. of the statutes is created to read:

70.11 (4) (a) 2. For purposes of subd. 1., beginning with the property tax assessments as of January 1, 2018, property owned by a church or religious association necessary for location and convenience of buildings includes property necessary for the location and convenience of a building that the church or religious association intends to construct to replace a building destroyed by fire, natural disaster, or criminal act, regardless of whether preconstruction planning or construction has begun. This subdivision applies only for the first 25 years after the year in which the building is destroyed.

SECTION 997g. 70.11 (11) of the statutes is amended to read:

70.11 (11) BIBLE CAMPS. All real property not exceeding 30 40 acres and the personal property situated therein, of any Bible camp conducted by a religious non-profit corporation organized under the laws of this state, so long as the property is used for religious purposes and not for pecuniary profit of any individual.

SECTION 997h. 70.11 (39) of the statutes is amended to read:

70.11 (39) COMPUTERS. If the owner of the property fulfills the requirements under s. 70.35, mainframe Mainframe computers, minicomputers, personal computers, networked personal computers, servers, terminals, monitors, disk drives, electronic peripheral equipment, tape drives, printers, basic operational programs, systems software, and prewritten software. The exemp-

tion under this subsection does not apply to custom software, fax machines, copiers, equipment with embedded computerized components or telephone systems, including equipment that is used to provide telecommunications services, as defined in s. 76.80 (3). For the purposes of s. 79.095, the exemption under this subsection does not apply to property that is otherwise exempt under this chapter.

SECTION 997i. 70.11 (39m) of the statutes is amended to read:

70.11 (**39m**) If the owner of the property fulfills the requirements under s. 70.35, cash <u>Cash</u> registers and fax machines, excluding fax machines that are also copiers.

SECTION 997j. 70.111 (27) of the statutes is created to read:

- 70.111 (27) MACHINERY, TOOLS, AND PATTERNS. (a) In this subsection, "machinery" means a structure or assemblage of parts that transmits force, motion, or energy from one part to another in a predetermined way by electrical, mechanical, or chemical means. "Machinery" does not include a building.
- (b) Beginning with the property tax assessments as of January 1, 2018, machinery, tools, and patterns, not including such items used in manufacturing.
- (c) A taxing jurisdiction may include the most recent valuation of personal property described under par. (b) that is located in the taxing jurisdiction for purposes of complying with debt limitations applicable to the jurisdiction.

SECTION 997L. 70.35 (1) of the statutes is amended to read:

70.35 (1) To determine the amount and value of any personal property for which any person, firm, or corporation should be assessed, any assessor may examine such person or the managing agent or officer of any firm or corporation under oath as to all such items of personal property, the taxable value thereof as defined in s. 70.34 if the property is taxable and the fair market value if the property is exempt under s. 70.11 (39) or (39m). In the alternative the assessor may require such person, firm, or corporation to submit a return of such personal property and of the taxable value thereof. There shall be annexed to such return the declaration of such person or of the managing agent or officer of such firm or corporation that the statements therein contained are true.

SECTION 997m. 70.35 (2) of the statutes is amended to read:

70.35 (2) The return shall be made and all the information therein requested given by such person on a form prescribed by the assessor with the approval of the department of revenue which shall provide suitable schedules for such information bearing on value as the department deems necessary to enable the assessor to determine the true cash value of the taxable personal property, and of the personal property that is exempt under s. 70.11 (39) and (39m), that is owned or in the pos-

session of such person on January 1 as provided in s. 70.10. The return may contain methods of deriving assessable values from book values and for the conversion of book values to present values, and a statement as to the accounting method used. No person shall be required to take detailed physical inventory for the purpose of making the return required by this section.

SECTION 997n. 70.36 (1m) of the statutes is repealed. SECTION 998. 70.57 (4) (b) 1. of the statutes is amended to read:

70.57 (4) (b) 1. For the year in which the error occurred, apportion county, school district, technical college district, and metropolitan sewerage district property taxes, and state forestation taxes under s. 70.58, to the taxation district using the taxation district's erroneous valuation.

SECTION 999. 70.57 (4) (b) 2. of the statutes is amended to read:

70.57 (4) (b) 2. For the year in which the error occurred, apportion county, school district, technical college district, and metropolitan sewerage district property taxes, and state forestation taxes under s. 70.58, to the taxation district using the taxation district's correct valuation.

SECTION 1000. 70.58 (1) of the statutes is amended to read:

70.58 (1) Except as provided in sub. subs. (2) and (3), there is levied an annual tax of two-tenths of one mill for each dollar of the assessed valuation of the property of the state as determined by the department of revenue under s. 70.57, for the purpose of acquiring, preserving and developing the forests of the state and for the purpose of forest crop law and county forest law administration and aid payments, for grants to forestry cooperatives under s. 36.56, and for the acquisition, purchase and development of forests described under s. 25.29 (7) (a) and (b), the proceeds of the tax to be paid into the conservation fund. The tax shall not be levied in any year in which general funds are appropriated for the purposes specified in this section, equal to or in excess of the amount which the tax would produce and no tax shall be levied under this section beginning with the property tax assessments as of January 1, 2017.

SECTION 1001. 70.58 (2) of the statutes is amended to read:

70.58 (2) In each of 3 years beginning with the property tax assessments as of January 1, 2005, the department of revenue shall adjust the rate of the tax imposed under this section so that the percentage increase from the previous year in the total amount levied under this section does not exceed 2.6 percent. The rate determined by the department of revenue for the property tax assessment as of January 1, 2007, shall be the rate of the tax imposed under this section for all subsequent years, ending with the property tax assessments as of January 1, 2017.

SECTION 1002. 70.58 (3) of the statutes is created to read:

70.58 (3) In fiscal year 2017–18, and in each fiscal year thereafter, an amount equal to 0.1697 mills for each dollar of the assessed valuation of the property of the state as determined by the department of revenue under s. 70.57 shall be transferred from the general fund to the conservation fund for the purposes described under sub. (1)

SECTION 1002m. 71.01 (1as) of the statutes is created to read:

71.01 (1as) "Broadcaster" means a television or radio station licensed by the federal communications commission, a television or radio broadcast network, a cable television network, or a television distribution company. "Broadcaster" does not include a cable service provider, a direct broadcast satellite system, or an Internet content distributor.

SECTION 1003. 71.01 (6) (b) of the statutes is repealed.

SECTION 1004. 71.01 (6) (j) 1. of the statutes is amended to read:

71.01 (6) (j) 1. For taxable years beginning after December 31, 2013, and before January 1, 2017, for individuals and fiduciaries, except fiduciaries of nuclear decommissioning trust or reserve funds, "Internal Revenue Code" means the federal Internal Revenue Code as amended to December 31, 2013, except as provided in subds. 2. and 3. and subject to subd. 4.

SECTION 1005. 71.01 (6) (j) 3. i. of the statutes is created to read:

71.01 (6) (j) 3. i. Section 2004 of P.L. 114-41.

SECTION 1006. 71.01 (6) (j) 3. j. of the statutes is created to read:

71.01 (**6**) (j) 3. j. Sections 503 and 504 of P.L. 114–74. **SECTION 1007.** 71.01 (6) (j) 3. k. of the statutes is created to read:

71.01 (**6**) (j) 3. k. Sections 103, 104, 124, 168, 184, 185, 190, 204, 303, 306, 336, and 341 of division Q of P.L. 114–113.

SECTION 1008. 71.01 (6) (j) 3. L. of the statutes is created to read:

71.01 (**6**) (**j**) 3. L. P.L. 114–239.

SECTION 1009. 71.01 (6) (k) of the statutes is created to read:

71.01 (6) (k) 1. For taxable years beginning after December 31, 2016, for individuals and fiduciaries, except fiduciaries of nuclear decommissioning trust or reserve funds, "Internal Revenue Code" means the federal Internal Revenue Code as amended to December 31, 2016, except as provided in subds. 2. and 3. and s. 71.98 and subject to subd. 4.

2. For purposes of this paragraph, "Internal Revenue Code" does not include the following provisions of federal public laws for taxable years beginning after December 31, 2016: section 13113 of P.L 103–66; sections 1, 3,

4, and 5 of P.L. 106-519; sections 101, 102, and 422 of P.L 108-357; sections 1310 and 1351 of P.L. 109-58; section 11146 of P.L. 109-59; section 403 (q) of P.L. 109-135; section 513 of P.L. 109-222; sections 104 and 307 of P.L. 109-432; sections 8233 and 8235 of P.L. 110–28; section 11 (e) and (g) of P.L. 110–172; section 301 of P.L. 110-245; section 15351 of P.L. 110-246; section 302 of division A, section 401 of division B, and sections 312, 322, 502 (c), 707, and 801 of division C of P.L. 110-343; sections 1232, 1241, 1251, 1501, and 1502 of division B of P.L. 111-5; sections 211, 212, 213, 214, and 216 of P.L. 111-226; sections 2011 and 2122 of P.L. 111-240; sections 753, 754, and 760 of P.L. 111-312; section 1106 of P.L. 112-95; sections 104, 318, 322, 323, 324, 326, 327, and 411 of P.L. 112-240; P.L. 114-7; section 1101 of P.L. 114-74; section 305 of division P of P.L. 114–113; and sections 112, 123, 125 to 128, 143, 144, 151 to 153, 165 to 167, 169 to 171, 189, 191, 307, 326, and 411 of division Q of P.L. 114-113.

- 3. For purposes of this paragraph, "Internal Revenue Code" does not include amendments to the federal Internal Revenue Code enacted after December 31, 2016.
- 4. For purposes of this paragraph, the provisions of federal public laws that directly or indirectly affect the Internal Revenue Code, as defined in this paragraph, apply for Wisconsin purposes at the same time as for federal purposes, except that changes made by section 4007 (b) of P.L. 114–41, section 1102 of P.L. 114–74, sections 105, 111, 113 to 115, 133, 301, 302, 304, 305, 308, 311, 313 to 323, 325, 331, and 343 to 345 of division Q of P.L. 114–113 first apply for taxable years beginning after December 31, 2016.

SECTION 1012. 71.04 (7) (dh) 2. b. of the statutes is amended to read:

71.04(7) (dh) 2. b. The service relates to tangible personal property that is located in this state at the time that the service is received or tangible personal property that is delivered directly or indirectly to customers in this state

SECTION 1013. 71.04 (7) (dh) 2. c. of the statutes is amended to read:

71.04 (7) (dh) 2. c. The service is provided to <u>purchased by</u> an individual who is physically present in this state at the time that the service is received.

SECTION 1013d. 71.04 (7) (dh) 3. of the statutes is amended to read:

71.04 (7) (dh) 3. If Except as provided in subd. 4., if the purchaser of a service receives the benefit of a service in more than one state, the gross receipts from the performance of the service are included in the numerator of the sales factor according to the portion of the service received in this state.

SECTION 1013e. 71.04 (7) (dh) 4. of the statutes is created to read:

71.04 (7) (dh) 4. For taxable years beginning after December 31, 2018, a broadcaster's gross receipts from

advertising are in this state only if the advertiser's commercial domicile is in this state. With regard to a broadcaster who is a member of a combined group, as defined in s. 71.255 (1) (a), this subdivision does not apply to the gross receipts of the members who are not broadcasters.

SECTION 1013f. 71.04 (7) (dj) 1. (intro.) of the statutes is amended to read:

71.04 (7) (dj) 1. (intro.) Except as provided in <u>subd.</u> 2. and par. (df), gross royalties and other gross receipts received for the use or license of intangible property, including patents, copyrights, trademarks, trade names, service names, franchises, licenses, plans, specifications, blueprints, processes, techniques, formulas, designs, layouts, patterns, drawings, manuals, technical know-how, contracts, and customer lists, are sales in this state if any of the following applies:

SECTION 1013g. 71.04 (7) (dj) 1. a. of the statutes is amended to read:

71.04 (7) (dj) 1. a. The purchaser or licensee uses the intangible property in the operation of a trade or business at a location in this state. If Except as provided in subd. 2.. if the purchaser or licensee uses the intangible property in the operation of a trade or business in more than one state, the gross royalties and other gross receipts from the use of the intangible property shall be divided between those states having jurisdiction to impose an income tax on the taxpayer in proportion to the use of the intangible property in those states.

SECTION 1013h. 71.04 (7) (dj) 2. of the statutes is created to read:

71.04 (7) (dj) 2. For taxable years beginning after December 31, 2018, a broadcaster's gross royalties and other gross receipts received for the use or license of intangible property are sales in this state only if the commercial domicile of the purchaser or licensee is in this state and the purchaser or licensee has a direct connection or relationship with the broadcaster pursuant to a contract under which the royalties or receipts are derived. With regard to a broadcaster who is a member of a combined group, as defined in s. 71.255 (1) (a), this subdivision does not apply to the gross royalties and receipts of the members who are not broadcasters.

SECTION 1013i. 71.04 (7) (g) of the statutes is created to read:

71.04 (7) (g) 1. For taxable years beginning after December 31, 2018, the amount of a broadcaster's gross receipts from advertising and the use or license of intangible property, as determined under pars. (dh) 4. and (dj) 2., shall be adjusted as follows:

- a. Determine the amount of the numerator of the sales factor for a broadcaster as provided in this subsection.
- b. Multiply .01 by the total amount of the domestic gross receipts of the broadcaster from advertising and royalties and other gross receipts for the use or license of intangible property.

- c. Determine the numerator of the sales for a broadcaster by substituting the amount determined under subd. 1. b. for the total amount determined under subd. 1. a.
- d. Except as provided in subd. 1. e., if the amount of the numerator determined under subd. 1. c. is more than the amount determined under subd. 1. a., substitute the amount of total gross receipts determined under subd. 1. b. for the total amount of the gross receipts determined under subd. 1. a. For purposes of this subd. 1. d., the amount of the numerator for a broadcaster is the amount determined under subd. 1. c.
- e. If the amount of the numerator computed under subd. 1. c. is more than 140 percent of the amount determined under subd. 1. a., adjust the total amount of the gross receipts under subd. 1. a. so that the amount of the numerator for a broadcaster is 140 percent of the numerator otherwise determined under subd. 1. a.
- 2. The department may promulgate rules to administer this paragraph.

SECTION 1014. 71.05 (1) (c) 13. of the statutes is created to read:

71.05 **(1) (c)** 13. An entity described under, or an entity whose bonds are issued under, s. 66.1201, 66.1333, or 66.1335.

SECTION 1015. 71.05 (6) (a) 26. (intro.) of the statutes is amended to read:

71.05 (6) (a) 26. (intro.) For the taxable year in which a distribution is received, all of the following amounts distributed from a college savings account, as described in s. 16.641 224.50:

SECTION 1016. 71.05 (6) (a) 26. c. of the statutes is amended to read:

71.05 (6) (a) 26. c. To the extent that an amount is not otherwise added back under this subdivision, any amount withdrawn from a college savings account, as described in s. 16.641 224.50, for any purpose if the withdrawn amount was contributed to the account within 365 days of the day on which the amount was withdrawn from such an account and if the withdrawn amount was previously subtracted under par. (b) 32.

SECTION 1016v. 71.05 (6) (b) 22. of the statutes is amended to read:

71.05 (6) (b) 22. For taxable years beginning after December 31, 1995, an amount up to \$5,000 that is expended during the period that consists of the year to which the claim relates and the prior 2 taxable years, by a full—year resident of this state who is an adoptive parent, for adoption fees, court costs or legal fees relating to the adoption of a child, for whom a final order of adoption has been entered under s. 48.91 (3) or by an order of a court of any other state, or upon registration of a foreign adoption under s. 48.97 (2), during the taxable year.

SECTION 1017. 71.05 (6) (b) 23. of the statutes is amended to read:

71.05 (6) (b) 23. Any increase in value of a tuition unit that is purchased under a tuition contract under s. 16.64 224.48, except that the subtraction under this subdivision may not be claimed by any individual who received a refund under s. 16.64 224.48 (7) (a) 2., 3. or 4.

SECTION 1018. 71.05 (6) (b) 28. (intro.) of the statutes is amended to read:

71.05 **(6)** (b) 28. (intro.) An amount paid by a claimant for tuition expenses and mandatory student fees for a student who is the claimant or who is the claimant's child and the claimant's dependent who is claimed under section 151 (c) of the Internal Revenue Code, to attend any university, college, technical college or a school approved under s. 38.50 440.52, that is located in Wisconsin or to attend a public vocational school or public institution of higher education in Minnesota under the Minnesota–Wisconsin reciprocity agreement under s. 39.47, calculated as follows:

SECTION 1019. 71.05 (6) (b) 28. h. of the statutes is amended to read:

71.05 (6) (b) 28. h. No modification may be claimed under this subdivision for an amount paid for tuition expenses and mandatory student fees, as described under this subdivision, if the source of the payment is an amount withdrawn from a college savings account, as described in s. 16.641 224.50 or from a college tuition and expenses program, as described in s. 16.64 224.48, and if the owner of the account or a parent, grandparent, great—grandparent, aunt, or uncle of the beneficiary, who contributed to the account, has claimed a deduction under subd. 32. or 33. that relates to such an amount.

SECTION 1020. 71.05 (6) (b) 31. of the statutes is amended to read:

71.05 (6) (b) 31. Any increase in value of a college savings account, as described in s. 16.641 224.50, except that the subtraction under this subdivision may not be claimed by any individual who has made a nonqualified withdrawal, as described in s. 16.641 224.50 (2) (e).

SECTION 1021. 71.05 (6) (b) 32. (intro.) of the statutes is amended to read:

71.05 (6) (b) 32. (intro.) An amount paid into a college savings account, as described in s. 46.641 224.50, in the taxable year in which the contribution is made or on or before the 15th day of the 4th month beginning after the close of a taxpayer's taxable year to which this subtraction relates, by the owner of the account or by any other individual, for the benefit of any beneficiary of an account, calculated as follows, except that each amount that is subtracted under this subdivision may be subtracted only once:

SECTION 1022. 71.05 (6) (b) 32m. of the statutes is amended to read:

71.05 (6) (b) 32m. Consistent with the limitations specified in subd. 32., for rollovers occurring after April 15, 2015, any principal amount rolled over to a college savings account, as described in s. 16.641 224.50, from

another state's qualified tuition program, as described in 26 USC 529 (c) (3) (C) (i). Amounts eligible for the subtraction under this subdivision that are in excess of the annual limits specified under subd. 32. may be carried forward to future taxable years of the taxpayer without limitation, other than the limits specified in subd. 32. ae. and am.

SECTION 1023. 71.05 (6) (b) 33. (intro.) of the statutes is amended to read:

71.05 (6) (b) 33. (intro.) An amount paid into a college tuition and expenses program, as described in s. 16.64 224.48, by the owner of the account or by a parent, grandparent, great–grandparent, aunt, or uncle of the beneficiary, if the beneficiary of the account is one of the following: the claimant; the claimant's child; the claimant's grandchild; the claimant's great–grandchild; or the claimant's niece or nephew; calculated as follows:

SECTION 1023s. 71.05 (6) (b) 53. of the statutes is created to read:

71.05 (6) (b) 53. The value of any Olympic, Paralympic, or Special Olympics medal won by an individual in an Olympic, Paralympic, or Special Olympics competition, and the amount of any payment such an individual receives from the U.S. Olympic Committee or from the Special Olympics Board of Directors, but only to the extent that the committee made the payment because the individual won an Olympic, Paralympic, or Special Olympics medal.

SECTION 1024. 71.05 (8) (b) 1. of the statutes is amended to read:

71.05 (8) (b) 1. A Except as provided in s. 71.80 (25), a Wisconsin net operating loss may be carried back against Wisconsin taxable income of the previous 2 years and then carried forward against Wisconsin taxable incomes of the next 20 taxable years, if the taxpayer was subject to taxation under this chapter in the taxable year in which the loss was sustained incurred, to the extent not offset against other income of the year of loss and to the extent not offset against Wisconsin modified taxable income of the 2 years preceding the loss and of any year between the loss year and the taxable year for which the loss carry-forward is claimed. In this paragraph, "Wisconsin modified taxable income" means Wisconsin taxable income with the following exceptions: a net operating loss deduction or offset for the loss year or any taxable year before or thereafter is not allowed, the deduction for long-term capital gains under subs. (6) (b) 9. and 9m. and (25) is not allowed, the amount deductible for losses from sales or exchanges of capital assets may not exceed the amount includable in income for gains from sales or exchanges of capital assets and "Wisconsin modified taxable income" may not be less than zero.

SECTION 1035m. 71.07 (2dy) (b) of the statutes is amended to read:

71.07 (**2dy**) (b) *Filing claims*. Subject to the limitations under this subsection and ss. 238.301 to 238.306 or

ss. 560.701 to 560.706, 2009 stats., for taxable years beginning after December 31, 2008, a claimant may claim as a credit against the tax imposed under s. 71.02 or 71.08, up to the amount of the tax, the amount authorized for the claimant under s. 238.303 or s. 560.703, 2009 stats.

SECTION 1035n. 71.07 (2dy) (d) 2. of the statutes is amended to read:

71.07 (2dy) (d) 2. If a claimant's certification is revoked under s. 238.305 or s. 560.705, 2009 stats., or if a claimant becomes ineligible for tax benefits under s. 238.302 or s. 560.702, 2009 stats., the claimant may not claim credits under this subsection for the taxable year that includes the day on which the certification is revoked; the taxable year that includes the day on which the claimant becomes ineligible for tax benefits; or succeeding taxable years and the claimant may not carry over unused credits from previous years to offset the tax imposed under s. 71.02 or 71.08 for the taxable year that includes the day on which certification is revoked; the taxable year that includes the day on which the claimant becomes ineligible for tax benefits; or succeeding taxable years.

SECTION 1035p. 71.07 (3q) (b) (intro.) of the statutes is amended to read:

71.07 (3q) (b) Filing claims. (intro.) Subject to the limitations provided in this subsection and s. 238.16 or s. 560.2055, 2009 stats., for taxable years beginning after December 31, 2009, a claimant may claim as a credit against the taxes imposed under ss. s. 71.02 and 71.08 any of the following:

SECTION 1035x. 71.07 (3q) (c) 3. of the statutes is amended to read:

71.07 (3q) (c) 3. The maximum amount of credits that may be awarded under this subsection and ss. 71.28 (3q) and 71.47 (3q) for the period beginning on January 1, 2010, and ending on June 30, 2013, is \$14,500,000, not including the amount of any credits reallocated under s. 238.15 (3) (d), 2015 stats., or s. 560.205 (3) (d), 2009

SECTION 1036. 71.07 (3q) (d) 2. of the statutes is amended to read:

71.07 (3q) (d) 2. If the allowable amount of the claim under par. (b) exceeds the tax otherwise due under ss. s. 71.02 and 71.08, the amount of the claim not used to offset the tax due shall be certified by the department of revenue to the department of administration for payment by check, share draft, or other draft drawn from the appropriation account under s. 20.835 (2) (bb), except that the amounts certified under this subdivision for taxable years beginning after December 31, 2009, and before January 1, 2012, shall be paid in taxable years beginning after December 31, 2011. Notwithstanding s. 71.82, no interest shall be paid on amounts certified under this subdivision.

SECTION 1036e. 71.07 (3s) (b) of the statutes is amended to read:

71.07 (3s) (b) The tax imposed under s. 71.02 or 71.08 shall be reduced by an amount equal to the sales and use tax under ch. 77 paid by the person in such taxable year on fuel and electricity consumed in manufacturing tangible personal property in this state. Shareholders in a tax-option corporation and partners may claim the credit under this subsection, based on eligible sales and use taxes paid by the partnership or tax-option corporation, in proportion to the ownership interest of each partner or shareholder. The partnership or tax-option corporation shall calculate the amount of the credit which may be claimed by each partner or shareholder and shall provide that information to the partner or shareholder.

SECTION 1036f. 71.07 (3t) (b) of the statutes is amended to read:

71.07 (3t) (b) Credit. Subject to the limitations provided in this subsection and in s. 560.28, 2009 stats., for taxable years beginning after December 31, 2007, a claimant may claim as a credit, amortized over 15 taxable years starting with the taxable year beginning after December 31, 2007, against the tax imposed under s. 71.02 and 71.08, up to the amount of the tax, an amount equal to the claimant's unused credits under s. 71.07 (3s).

SECTION 1036h. 71.07 (3w) (b) (intro.) of the statutes **Vetoed** is amended to read:

71.07 (3w) (b) Filing claims; payroll. (intro.) Subject to the limitations provided in this subsection and s. 238.399 or s. 560.799, 2009 stats., a claimant may claim as a credit against the tax imposed under s. 71.02 or 71.08 an amount calculated as follows:

SECTION 1036Lm. 71.07 (3w) (bm) 1, of the statutes is amended to read:

71.07 (3w) (bm) 1. In addition to the credits under par. (b) and subds. 2., 3., and 4., and subject to the limitations provided in this subsection and s. 238.399 or s. 560.799, 2009 stats., a claimant may claim as a credit against the tax imposed under s. 71.02 or 71.08 an amount equal to a percentage, as determined under s. 238.399 or s. 560.799, 2009 stats., not to exceed 100 percent, of the amount the claimant paid in the taxable year to upgrade or improve the job-related skills of any of the claimant's full-time employees, to train any of the claimant's full-time employees on the use of job-related new technologies, or to provide job-related training to any full-time employee whose employment with the claimant represents the employee's first full-time job. This subdivision does not apply to employees who do not work in an enterprise zone.

SECTION 1037bc. 71.07 (3w) (bm) 2. of the statutes is amended to read:

71.07 (3w) (bm) 2. In addition to the credits under par. (b) and subds. 1., 3., and 4., and subject to the limitations provided in this subsection and s. 238.399 or s. In Part

560.799, 2009 stats., a claimant may claim as a credit against the tax imposed under s. 71.02 or 71.08 an amount equal to the percentage, as determined under s. 238.399 or s. 560.799, 2009 stats., not to exceed 7 percent, of the claimant's zone payroll paid in the taxable year to all of the claimant's full-time employees whose annual wages are greater than the amount determined by multiplying 2,080 by 150 percent of the federal minimum wage in a tier I county or municipality, not including the wages paid to the employees determined under par. (b) 1., or greater than \$30,000 in a tier II county or municipality, not including the wages paid to the employees determined under par. (b) 1., and who the claimant employed in the enterprise zone in the taxable year, if the total number of such employees is equal to or greater than the total number of such employees in the base year. A claimant may claim a credit under this subdivision for no more than 5 consecutive taxable years.

SECTION 1037bd. 71.07 (3w) (bm) 3. of the statutes is amended to read:

71.07 (3w) (bm) 3. In addition to the credits under par. (b) and subds. 1., 2., and 4., and subject to the limitations provided in this subsection and s. 238.399 or s. 560.799, 2009 stats., for taxable years beginning after December 31, 2008, a claimant may claim as a credit against the tax imposed under s. 71.02 or 71.08 up to 10 percent of the claimant's significant capital expenditures, as determined under s. 238.399 (5m) or s. 560.799 (5m), 2009 stats.

SECTION 1037be. 71.07 (3w) (bm) 4. of the statutes is amended to read:

71.07 (3w) (bm) 4. In addition to the credits under par. (b) and subds. 1., 2., and 3., and subject to the limitations provided in this subsection and s. 238.399 or s. 560.799, 2009 stats., for taxable years beginning after December 31, 2009, a claimant may claim as a credit against the tax imposed under s. 71.02 or 71.08, up to 1 percent of the amount that the claimant paid in the taxable year to purchase tangible personal property, items, property, or goods under s. 77.52 (1) (b), (c), or (d), or services from Wisconsin vendors, as determined under s. 238.399 (5) (e) or s. 560.799 (5) (e), 2009 stats., except that the claimant may not claim the credit under this subdivision and subd. 3. for the same expenditures.

SECTION 1037d. 71.07 (3w) (c) 1. of the statutes is renumbered 71.07 (3w) (c) 1. a. and amended to read:

Vetoed In Part

71.07 (3w) (c) 1. a. If For claims filed before January 1, 2018, if the allowable amount of the claim under this subsection exceeds the taxes otherwise due on the claimant's income under s. 71.02, the amount of the claim that is not used to offset those taxes shall be certified by the department of revenue to the department of administration for payment by check, share draft, or other draft drawn from the appropriation under s. 20.835 (2) (co).

Notwithstanding s. 71.82, no interest shall be paid on amounts certified under this subd. 1. a.

SECTION 1037e. 71.07 (3w) (c) 1. b. of the statutes is **Vetoed** created to read:

In Part

71.07 (3w) (c) 1. b. For claims filed after December 31, 2017, claims under this subsection shall be made to the Wisconsin Economic Development Corporation using policies and procedures established by the corporation board. The corporation shall certify valid claims to the department of administration for payment by check, share draft, or other draft drawn from the appropriation under s. 20.835 (2) (co). Notwithstanding s. 71.82, no interest shall be paid on amounts certified under this subd. 1. b.

SECTION 1037f. 71.07 (3w) (c) 2. of the statutes is amended to read:

71.07 (3w) (c) 2. Partnerships For claims filed before January 1, 2018, partnerships, limited liability companies, and tax-option corporations may not claim the credit under this subsection, but the eligibility for, and the amount of, the credit are based on their payment of amounts described under pars. (b) and (bm). A partnership, limited liability company, or tax-option corporation shall compute the amount of credit that each of its partners, members, or shareholders may claim and shall provide that information to each of them. Partners, members of limited liability companies, and shareholders of tax-option corporations may claim the credit in proportion to their ownership interests. For claims filed after December 31, 2017, partnerships, limited liability companies, and tax-option corporations may claim the credit under this subsection as provided under subd. 1. b.

SECTION 1037g. 71.07 (3w) (c) 3. of the statutes is amended to read:

71.07 (3w) (c) 3. No For claims filed before January 1, 2018, no credit may be allowed under this subsection unless the claimant includes with the claimant's return a copy of the claimant's certification for tax benefits under s. 238.399 (5) or (5m) or s. 560.799 (5) or (5m), 2009 stats.

SECTION 1037h. 71.07 (3w) (d) of the statutes is amended to read:

71.07 (**3w**) (d) Administration. Section 71.28 (4) (g) and (h), as it applies to the credit under s. 71.28 (4), applies to the credit under this subsection. Claimants shall include with their returns a copy of their certification for tax benefits, and a copy of the verification of their expenses, from the department of commerce or the Wisconsin Economic Development Corporation. This paragraph does not apply to claims filed after December 31, 2017.

SECTION 1037i. 71.07 (3wm) of the statutes is created to read:

71.07 (3wm) ELECTRONICS AND INFORMATION TECH-NOLOGY MANUFACTURING ZONE CREDIT. (a) Definitions. In this subsection:

- 1. "Claimant" means a person who is certified to claim tax benefits under s. 238.396 (3) and who files a claim under this subsection.
- 2. "Full-time employee" means an individual who is employed in a job for which the annual pay is at least \$30,000 and who is offered retirement, health, and other benefits that are equivalent to the retirement, health, and other benefits offered to an individual who is required to work at least 2,080 hours per year.
- 3. "State payroll" means the amount of payroll apportioned to this state, as determined under s. 71.25 (8).
- 4. "Wages" means wages under section 3306 (b) of the Internal Revenue Code, determined without regard to any dollar limitations.
- 5. "Zone" means a zone designated under s. 238.396 (1m).
- 6. "Zone payroll" means the amount of state payroll that is attributable to wages paid by the claimant to fulltime employees for services that are performed in the zone or that are performed outside the zone, but within the state, and for the benefit of the operations within the zone, as determined by the Wisconsin Economic Development Corporation. "Zone payroll" does not include the amount of wages paid to any full-time employees that exceeds \$100,000.
- (b) Filing claims; payroll. Subject to the limitations provided in this subsection and s. 238.396, a claimant may claim as a credit an amount calculated as follows:
- 1. Determine the zone payroll for the taxable year for full-time employees employed by the claimant.
- 2. Multiply the amount determined under subd. 1. by 17 percent.
- (bm) Filing supplemental claims. In addition to claiming the credit under par. (b), and subject to the limitations under this subsection and s. 238.396, a claimant may claim as a credit up to 15 percent of the claimant's significant capital expenditures in the zone in the taxable year, as determined under s. 238.396 (3m).

Vetoed In Part

(c) Limitations. Partnerships, limited liability companies, and tax-option corporations may claim the credit under this subsection as provided under par. (d). The Wisconsin Economic Development Corporation may recover credits claimed under this paragraph that are revoked or otherwise invalid from the partnership, limited liability company, or tax-option corporation or from the individual partner, member, or shareholder.

Vetoed In Part

(d) Administration. Claims under this subsection shall be made to the Wisconsin Economic Development Corporation using policies and procedures established by the corporation board. The corporation shall certify valid claims to the department of administration for payment by check, share draft, or other draft drawn from the appropriation under s. 20.835 (2) (cp). Notwithstanding s. 71.82, no interest shall be paid on amounts certified under this subdivision.

SECTION 1037t. 71.07 (3y) (b) (intro.) of the statutes **Vetoed** is amended to read:

71.07 (3y) (b) Filing claims. (intro.) Subject to the limitations provided in this subsection and s. 238.308, for taxable years beginning after December 31, 2015, a claimant may claim as a credit against the tax imposed under ss. 71.02 and 71.08 all of the following:

SECTION 1037u. 71.07 (3y) (c) 1. of the statutes is amended to read:

71.07 (3y) (c) 1. Partnerships For claims filed before January 1, 2018, partnerships, limited liability companies, and tax-option corporations may not claim the credit under this subsection, but the eligibility for, and the amount of, the credit are based on their payment of amounts under par. (b). A partnership, limited liability company, or tax-option corporation shall compute the amount of credit that each of its partners, members, or shareholders may claim and shall provide that information to each of them. Partners, members of limited liability companies, and shareholders of tax-option corporations may claim the credit in proportion to their ownership interests. For claims filed after December 31, 2017, partnerships, limited liability companies, and taxoption corporations may claim the credit under this subsection as provided under par. (d) 2. b.

SECTION 1037v. 71.07 (3y) (c) 2. of the statutes is amended to read:

71.07 (3y) (c) 2. No For claims filed before January 1, 2018, no credit may be allowed under this subsection unless the claimant includes with the claimant's return a copy of the claimant's certification for tax benefits under s. 238.308.

SECTION 1037w. 71.07 (3y) (d) 1. of the statutes is amended to read:

71.07 (3y) (d) 1. Section 71.28 (4) (e), (g), and (h), as it applies to the credit under s. 71.28 (4), applies to the credit under this subsection. This subdivision does not apply to claims filed after December 31, 2017.

SECTION 1038g. 71.07 (3y) (d) 2. of the statutes is renumbered 71.07 (3y) (d) 2. a. and amended to read:

71.07 (3y) (d) 2. a. If For claims filed before January Vetoed 2018, if the allowable amount of the claim under par. (b) exceeds the tax otherwise due under ss. 71.02 and 71.08, the amount of the claim not used to offset the tax due shall be certified by the department of revenue to the department of administration for payment by check, share draft, or other draft drawn from the appropriation account under s. 20.835 (2) (bg). Notwithstanding s. 71.82, no interest shall be paid on amounts certified under this subd. 2. a.

SECTION 1038h. 71.07 (3y) (d) 2. b. of the statutes is **Vetoed** created to read:

71.07 (3y) (d) 2. b. For claims filed after December 31, 2017, claims under this subsection shall be made to In Part

In Part

In Part

the Wisconsin Economic Development Corporation using policies and procedures established by the corporation board. The corporation shall certify valid claims to the department of administration for payment by check, share draft, or other draft drawn from the appropriation under s. 20.835 (2) (bg). Notwithstanding s. 71.82, no interest shall be paid on amounts certified under this subd. 2. b.

SECTION 1038m. 71.07 (4k) (b) 4. a. of the statutes is amended to read:

71.07 (4k) (b) 4. a. Except as provided in subds. 5. and 6., for taxable years beginning after December 31, 2014, an individual, a partner of a partnership, a shareholder of a tax-option corporation, or a member of a limited liability company may claim a credit against the tax imposed under s. 71.02 or 71.08, as allocated under par. (d), an amount equal to 5.75 percent of the amount by which the individual's, partnership's, tax-option corporation's, or limited liability company's qualified research expenses for the taxable year exceed 50 percent of the average qualified research expenses for the 3 taxable years immediately preceding the taxable year for which the claimant claims the credit. If the individual, partnership, tax-option corporation, or limited liability company had no qualified research expenses in any of the 3 taxable years immediately preceding the taxable year for which the claimant claims the credit, the claimant may claim an amount equal to 2.875 percent of the individual's, partnership's, tax-option corporation's, or limited liability company's qualified research expenses for the taxable year for which the claimant claims the credit.

SECTION 1038p. 71.07 (4k) (b) 5. a. of the statutes is amended to read:

71.07 (4k) (b) 5. a. For taxable years beginning after December 31, 2014, an individual, a partner of a partnership, a shareholder of a tax-option corporation, or a member of a limited liability company may claim a credit against the tax imposed under s. 71.02 or 71.08, as allocated under par. (d), an amount equal to 11.5 percent of the amount by which the individual's, partnership's, taxoption corporation's, or limited liability company's qualified research expenses for the taxable year exceed 50 percent of the average qualified research expenses for the 3 taxable years immediately preceding the taxable year for which the claimant claims the credit. If the individual, partnership, tax-option corporation, or limited liability company had no qualified research expenses in any of the 3 taxable years immediately preceding the taxable year for which the claimant claims the credit, the claimant may claim an amount equal to 5.75 percent of the individual's, partnership's, tax-option corporation's, or limited liability company's qualified research expenses for the taxable year for which the claimant claims the credit.

SECTION 1038r. 71.07 (4k) (b) 6. a. of the statutes is amended to read:

71.07 (4k) (b) 6. a. For taxable years beginning after December 31, 2014, an individual, a partner of a partnership, a shareholder of a tax-option corporation, or a member of a limited liability company may claim a credit against the tax imposed under s. 71.02 or 71.08, as allocated under par. (d), an amount equal to 11.5 percent of the amount by which the individual's, partnership's, taxoption corporation's, or limited liability company's qualified research expenses for the taxable year exceed 50 percent of the average qualified research expenses for the 3 taxable years immediately preceding the taxable year for which the claimant claims the credit. If the individual, partnership, tax-option corporation, or limited liability company had no qualified research expenses in any of the 3 taxable years immediately preceding the taxable year for which the claimant claims the credit, the claimant may claim an amount equal to 5.75 percent of the individual's, partnership's, tax-option corporation's, or limited liability company's qualified research expenses for the taxable year for which the claimant claims the credit.

SECTION 1038s. 71.07 (4k) (e) of the statutes is renumbered 71.07 (4k) (e) 1. and amended to read:

71.07 (**4k**) (e) 1. Section For taxable years beginning before January 1, 2018, s. 71.28 (4) (b) to (h), as it applies to the credit under s. 71.28 (4), applies to the credits under this subsection.

SECTION 1038t. 71.07 (4k) (e) 2. of the statutes is created to read:

71.07 (**4k**) (e) 2. For taxable years beginning after December 31, 2017, s. 71.28 (4) (b) to (e), (g), and (h), as it applies to the credit under s. 71.28 (4), applies to the credits under this subsection. For taxable years beginning after December 31, 2017, if the allowable amount of the claim under par. (b) 4., 5., or 6. exceeds the tax otherwise due under s. 71.02 or 71.08, all of the following apply:

a. The amount of the claim not used to offset the tax due, not to exceed 10 percent of the allowable amount of the claim under par. (b) 4., 5., or 6., shall be certified by the department of revenue to the department of administration for payment by check, share draft, or other draft drawn from the appropriation account under s. 20.835 (2) (d).

b. The amount of the claim not used to offset the tax due and not certified for payment under subd. 2. a. may be carried forward and credited against Wisconsin income taxes otherwise due for the following 15 taxable years to the extent not offset by these taxes otherwise due in all intervening years between the year in which the expense was incurred and the year in which the carryforward credit is claimed.

SECTION 1039. 71.07 (5) (b) of the statutes is amended to read:

71.07 (5) (b) Subtract the standard deduction under s. 71.05 (22), notwithstanding the limitation by such frac-

tion of that amount as Wisconsin adjusted gross income is of federal adjusted gross income described in s. 71.05 (22) (g) and (h), from the amount under par. (a).

SECTION 1040e. 71.07 (5b) (b) 1. of the statutes is amended to read:

71.07 (**5b**) (b) 1. For taxable years beginning after December 31, 2004, subject to the limitations provided under this subsection and s. 238.15 or s. 560.205, 2009 stats., and except as provided in subd. 2., a claimant may claim as a credit against the tax imposed under ss. s. 71.02 and 71.08, up to the amount of those taxes, 25 percent of the claimant's investment paid to a fund manager that the fund manager invests in a business certified under s. 238.15 (1) or s. 560.205 (1), 2009 stats.

SECTION 1040g. 71.07 (5b) (d) 3. of the statutes is amended to read:

71.07 (**5b**) (d) 3. Except as provided under s. 238.15 (3) (d) (intro.), for investments made after December 31, 2007, if an investment for which a claimant claims a credit under par. (b) is held by the claimant for less than 3 years, the claimant shall pay to the department, in the manner prescribed by the department, the amount of the credit that the claimant received related to the investment.

SECTION 1040h. 71.07 (5d) (b) (intro.) of the statutes is amended to read:

71.07 (**5d**) (b) *Filing claims*. (intro.) Subject to the limitations provided in this subsection and in s. 238.15 or s. 560.205, 2009 stats., a claimant may claim as a credit against the tax imposed under s. 71.02 or 71.08, up to the amount of those taxes, the following:

SECTION 1040p. 71.07 (5d) (d) 1. of the statutes is amended to read:

71.07 (**5d**) (d) 1. Except as provided under s. 238.15 (3) (d) (intro.), for investments made after December 31, 2007, if an investment for which a claimant claims a credit under par. (b) is held by the claimant for less than 3 years, the claimant shall pay to the department, in the manner prescribed by the department, the amount of the credit that the claimant received related to the investment.

Vetoed In Part **SECTION 1041e.** 71.07 (5m) (e) of the statutes is created to read:

71.07 (5m) (e) *Sunset*. No credit may be claimed under this subsection for taxable years beginning after December 31, 2016.

SECTION 1041s. 71.07 (5n) (b) (intro.) of the statutes is amended to read:

71.07 (**5n**) (b) *Filing claims*. (intro.) Subject to the limitations provided in this subsection, a claimant may claim as a credit against the tax imposed under ss. s. 71.02 and 71.08, up to the amount of the tax, an amount equal to one of the following percentages of the claimant's eligible qualified production activities income in the taxable year:

SECTION 1042. 71.07 (5n) (d) 3. of the statutes is created to read:

71.07 (**5n**) (d) 3. The amount of the eligible qualified production activities income that a claimant may claim in computing the credit under par. (b) shall be reduced by the amount of the qualified production activities income taxed by another state upon which the credit under sub. (7) is claimed.

SECTION 1043. 71.07 (5r) (a) 2. of the statutes is amended to read:

71.07 (**5r**) (a) 2. "Course of instruction" has the meaning given in s. 38.50 ± 440.52 (1) (c).

SECTION 1044. 71.07 (5r) (a) 6. b. of the statutes is amended to read:

71.07 (**5r**) (a) 6. b. A school approved under s. 38.50 440.52, if the delivery of education occurs in this state.

SECTION 1044p. 71.07 (7) (a) of the statutes is renumbered 71.07 (7) (a) (intro.) and amended to read:

71.07 (7) (a) (intro.) In this subsection, "state":

<u>2. "State"</u> includes the District of Columbia, but does not include the commonwealth of Puerto Rico or the several territories organized by Congress.

SECTION 1044r. 71.07 (7) (a) 1. of the statutes is created to read:

71.07 (7) (a) 1. "Net Wisconsin income tax" means the gross Wisconsin income tax less all nonrefundable credits that may be claimed by that taxpayer, except the credit for taxes paid to other states.

SECTION 1044t. 71.07 (7) (b) of the statutes is amended to read:

71.07 (7) (b) If Subject to conditions and limitations in pars. (c) and (d), if a resident individual, estate or trust pays a net income tax to another state, that resident individual, estate or trust may credit the net tax paid to that other state on that income against the net income tax otherwise payable to the state on income of the same year. The credit may not be allowed unless the income taxed by the other state is also considered income for Wisconsin tax purposes. The credit may not be allowed unless claimed within the time provided in s. 71.75 (2), but s. 71.75 (4) does not apply to those credits. For purposes of this paragraph, amounts declared and paid under the income tax law of another state are considered a net income tax paid to that other state only in the year in which the income tax return for that state was required to be filed. Income and franchise taxes paid to another state by a tax-option corporation, partnership, or limited liability company that is treated as a partnership may be claimed as a credit under this paragraph by that corporation's shareholders, that partnership's partners, or that limited liability company's members who are residents of this state and who otherwise qualify under this para-

SECTION 1044u. 71.07 (7) (c) and (d) of the statutes are created to read:

71.07 (7) (c) The credit may not exceed an amount determined by multiplying the taxpayer's net Wisconsin income tax by a ratio derived by dividing the income subject to tax in the other state that is also subject to tax in Wisconsin while the taxpayer is a resident of Wisconsin, by the taxpayer's Wisconsin adjusted gross income.

(d) The limitation in par. (c) does not apply to income that is taxed by one of the 4 states that border this state.

SECTION 1050p. 71.07 (9m) (a) 2m. of the statutes is amended to read:

71.07 (**9m**) (a) 2m. For taxable years beginning after December 31, 2013, any person may claim as a credit against taxes otherwise due under s. 71.02 or 71.08, up to the amount of those taxes, an amount equal to 20 percent of the costs of qualified rehabilitation expenditures, as defined in section 47 (c) (2) of the Internal Revenue Code, for certified historic structures on property located in this state, if the cost of the person's qualified rehabilitation expenditures is at least \$50,000 and the rehabilitated property is placed in service after December 31, 2013.

SECTION 1050r. 71.07 (9m) (a) 3. of the statutes is amended to read:

71.07 (9m) (a) 3. For taxable years beginning after December 31, 2013, any person may claim as a credit against taxes otherwise due under s. 71.02 or 71.08, up to the amount of those taxes, an amount equal to 20 percent of the costs of qualified rehabilitation expenditures, as defined in section 47 (c) (2) of the Internal Revenue Code, for qualified rehabilitated buildings, as defined in section 47 (c) (1) of the Internal Revenue Code, on property located in this state, if the cost of the person's qualified rehabilitation expenditures is at least \$50,000 and the rehabilitated property is placed in service after December 31, 2013, and regardless of whether the rehabilitated property is used for multiple or revenue-producing purposes. No credit may be claimed under this subdivision for property listed as a contributing building in the state register of historic places or in the national register of historic places and no credit may be claimed under this subdivision for nonhistoric, nonresidential property converted into housing if the property has been previously used for housing.

SECTION 1050u. 71.07 (9m) (h) of the statutes is amended to read:

71.07 (**9m**) (h) Any person, including a nonprofit entity described in section 501 (c) (3) of the Internal Revenue Code, may sell or otherwise transfer the credit under par. (a) 2m. or 3., in whole or in part, to another person who is subject to the taxes imposed under s. 71.02, 71.08, 71.23, or 71.43, if the person notifies the department of the transfer, and submits with the notification a copy of the transfer documents, and the department certifies ownership of the credit with each transfer.

SECTION 1051. 71.07 (9m) (i) of the statutes is created to read:

71.07 (**9m**) (i) If a person who claims a credit under this subsection and a credit under section 47 of the Internal Revenue Code for the same qualified rehabilitation expenditures is required to repay any amount of the credit claimed under section 47 of the Internal Revenue Code, the person shall repay to the department a proportionate amount of the credit claimed under this subsection.

SECTION 1051g. 71.07 (9r) (a) of the statutes is amended to read:

71.07 (**9r**) (a) For taxable years beginning on or after August 1, 1988, any natural person may credit against taxes otherwise due under s. 71.02 or 71.08 an amount equal to 25 percent of the costs of preservation or rehabilitation of historic property located in this state, including architectural fees and costs incurred in preparing nomination forms for listing in the national register of historic places in Wisconsin or the state register of historic places, if the nomination is made within 5 years prior to submission of a preservation or rehabilitation plan under par. (b) 3. b., and if the physical work of construction or destruction in preparation for construction begins after December 31, 1988, except that the credit may not exceed \$10,000, or \$5,000 for married persons filing separately, for any preservation or rehabilitation project.

SECTION 1052e. 71.08 (5) of the statutes is created to read:

71.08 (5) SUNSET. This section does not apply to taxable years beginning after December 31, 2016.

SECTION 1052k. 71.09 (1) (b) of the statutes is **In Part** amended to read:

71.09 (1) (b) "Tax shown on the return" and "tax for the taxable year" mean the net tax imposed under s. 71.02 after reduction for exemptions to, and credits against, that tax but before reduction by amounts withheld under subch. X and before reduction for amounts paid as estimated tax under this section for that tax plus the tax imposed under s. 71.08 before reduction for amounts paid as estimated tax under this section for that tax plus the surcharge imposed under s. 77.93 before reduction for amounts paid as estimated tax under this section for that surcharge.

SECTION 1052L. 71.09 (2) of the statutes is amended to read:

71.09 (2) Who shall pay. Every individual, estate and trust deriving income subject to taxation under this chapter, other than wages as defined in s. 71.63 (6) upon which taxes are withheld by the individual's employer under subch. X, shall pay estimated income tax₅ and the surcharge under s. 77.93 and alternative minimum tax. This section does not apply to any person on active duty with the U.S. armed forces while stationed outside the continental United States. This section does not apply to any taxable year ending before the date 2 years after the date of a decedent's death with respect to the estate of such decedent or any trust all of which is treated under

Vetoed In Part subpart E of part I of subchapter J of chapter 1 of the internal revenue code as owned by the decedent and to which the residue of the decedent's estate will pass under his or her will. This section does not apply to any trust that is subject to tax under this chapter on unrelated business taxable income as defined under section 512 of the internal revenue code. Those trusts are subject to estimated tax payments under s. 71.29.

SECTION 1052p. 71.09 (11) (b) of the statutes is amended to read:

71.09 (11) (b) The preceding taxable year was 12 months, the taxpayer had no liability under s. 71.02 or 71.08 for that year and the taxpayer was a resident of this state for all of that year.

SECTION 1052u. 71.10 (4) (f) of the statutes is amended to read:

71.10 (4) (f) Alternative For taxable years beginning before January 1, 2017, alternative minimum tax under s. 71.08, including any surtax on alternative minimum tax.

SECTION 1052w. 71.10 (4) (fr) of the statutes is amended to read:

71.10 (4) (fr) Research credit under s. 71.07 (4k), except as provided under par. (i).

SECTION 1052y. 71.10 (4) (i) of the statutes is amended to read:

71.10 (4) (i) The total of claim of right credit under s. 71.07 (1), farmland preservation credit under ss. 71.57 to 71.61, farmland preservation credit, 2010 and beyond under s. 71.613, homestead credit under subch. VIII, farmland tax relief credit under s. 71.07 (3m), dairy manufacturing facility investment credit under s. 71.07 (3p), jobs tax credit under s. 71.07 (3q), meat processing facility investment credit under s. 71.07 (3r), woody biomass harvesting and processing credit under s. 71.07 (3rm), food processing plant and food warehouse investment credit under s. 71.07 (3rn), business development credit under s. 71.07 (3y), research credit under s. 71.07 (4k) (e) 2. a., film production services credit under s. 71.07 (5f), film production company investment credit under s. 71.07 (5h), veterans and surviving spouses property tax credit under s. 71.07 (6e), enterprise zone jobs credit under s. 71.07 (3w), beginning farmer and farm asset owner tax credit under s. 71.07 (8r), earned income tax credit under s. 71.07 (9e), estimated tax payments under s. 71.09, and taxes withheld under subch. X.

SECTION 1057d. 71.22 (1e) of the statutes is created to read:

71.22 (1e) "Broadcaster" means a television or radio station licensed by the federal communications commission, a television or radio broadcast network, a cable television network, or a television distribution company. "Broadcaster" does not include a cable service provider, a direct broadcast satellite system, or an Internet content distributor.

SECTION 1058. 71.22 (4) (b) of the statutes is repealed.

SECTION 1059. 71.22 (4) (j) 1. of the statutes is amended to read:

71.22 (4) (j) 1. For taxable years beginning after December 31, 2013, and before January 1, 2017, "Internal Revenue Code" means the federal Internal Revenue Code as amended to December 31, 2013, except as provided in subds. 2. and 3. and subject to subd. 4., and except as provided in sub. (4m) and ss. 71.26 (2) (b) and (3), 71.34 (1g), and 71.42 (2).

SECTION 1060. 71.22 (4) (j) 3. i. of the statutes is created to read:

71.22 (4) (j) 3. i. Section 2004 of P.L. 114-41.

SECTION 1061. 71.22 (4) (j) 3. j. of the statutes is created to read:

71.22 (**4**) (j) 3. j. Sections 503 and 504 of P.L. 114–74. **SECTION 1062.** 71.22 (4) (j) 3. k. of the statutes is created to read:

71.22 (**4**) (j) 3. k. Sections 103, 104, 124, 168, 184, 185, 190, 204, 303, 306, 336, and 341 of division Q of P.L. 114–113.

SECTION 1063. 71.22 (4) (j) 3. L. of the statutes is created to read:

71.22 (4) (j) 3. L. P.L. 114-239.

SECTION 1064. 71.22 (4) (k) of the statutes is created to read:

71.22 (4) (k) 1. For taxable years beginning after December 31, 2016, "Internal Revenue Code" means the federal Internal Revenue Code as amended to December 31, 2016, except as provided in subds. 2. and 3. and subject to subd. 4., and except as provided in sub. (4m) and ss. 71.26 (2) (b) and (3), 71.34 (1g), 71.42 (2), and 71.98.

2. For purposes of this paragraph, "Internal Revenue Code" does not include the following provisions of federal public laws for taxable years beginning after December 31, 2016: section 13113 of P.L 103-66; sections 1, 3, 4, and 5 of P.L. 106-519; sections 101, 102, and 422 of P.L 108-357; sections 1310 and 1351 of P.L. 109-58; section 11146 of P.L. 109-59; section 403 (q) of P.L. 109-135; section 513 of P.L. 109-222; sections 104 and 307 of P.L. 109-432; sections 8233 and 8235 of P.L. 110-28; section 11 (e) and (g) of P.L. 110-172; section 301 of P.L. 110-245; section 15351 of P.L. 110-246; section 302 of division A, section 401 of division B, and sections 312, 322, 502 (c), 707, and 801 of division C of P.L. 110–343; sections 1232, 1241, 1251, 1501, and 1502 of division B of P.L. 111-5; sections 211, 212, 213, 214, and 216 of P.L. 111-226; sections 2011 and 2122 of P.L. 111-240; sections 753, 754, and 760 of P.L. 111-312; section 1106 of P.L. 112–95; sections 104, 318, 322, 323, 324, 326, 327, and 411 of P.L. 112-240; P.L. 114-7; section 1101 of P.L. 114-74; section 305 of division P of P.L. 114-113; and sections 112, 123, 125 to 128, 143, 144, 151 to 153, 165 to 167, 169 to 171, 189, 191, 307, 326, and 411 of division Q of P.L. 114–113.

- 3. For purposes of this paragraph, "Internal Revenue Code" does not include amendments to the federal Internal Revenue Code enacted after December 31, 2016.
- 4. For purposes of this paragraph, the provisions of federal public laws that directly or indirectly affect the Internal Revenue Code, as defined in this paragraph, apply for Wisconsin purposes at the same time as for federal purposes, except that changes made by section 4007 (b) of P.L. 114–41, section 1102 of P.L. 114–74, sections 105, 111, 113 to 115, 133, 301, 302, 304, 305, 308, 311, 313 to 323, 325, 331, and 343 to 345 of division Q of P.L. 114–113 first apply for taxable years beginning after December 31, 2016.

SECTION 1065. 71.22 (4m) (b) of the statutes is repealed.

SECTION 1066. 71.22 (4m) (j) 1. of the statutes is amended to read:

71.22 (4m) (j) 1. For taxable years beginning after December 31, 2013, and before January 1, 2017, "Internal Revenue Code", for corporations that are subject to a tax on unrelated business income under s. 71.26 (1) (a), means the federal Internal Revenue Code as amended to December 31, 2013, except as provided in subds. 2. and 3. and subject to subd. 4.

SECTION 1067. 71.22 (4m) (j) 3. i. of the statutes is created to read:

71.22 (**4m**) (j) 3. i. Section 2004 of P.L. 114–41.

SECTION 1068. 71.22 (4m) (j) 3. j. of the statutes is created to read:

71.22 (**4m**) (j) 3. j. Sections 503 and 504 of P.L. 114–74.

SECTION 1069. 71.22 (4m) (j) 3. k. of the statutes is created to read:

71.22 (**4m**) (j) 3. k. Sections 103, 104, 124, 168, 184, 185, 190, 204, 303, 306, 336, and 341 of division Q of P.L. 114–113.

SECTION 1070. 71.22 (4m) (j) 3. L. of the statutes is created to read:

71.22 (4m) (j) 3. L. P.L. 114-239.

SECTION 1071. 71.22 (4m) (k) of the statutes is created to read:

- 71.22 (4m) (k) 1. For taxable years beginning after December 31, 2016, "Internal Revenue Code", for corporations that are subject to a tax on unrelated business income under s. 71.26 (1) (a), means the federal Internal Revenue Code as amended to December 31, 2016, except as provided in subds. 2. and 3. and s. 71.98 and subject to subd. 4.
- 2. For purposes of this paragraph, "Internal Revenue Code" does not include the following provisions of federal public laws for taxable years beginning after December 31, 2016: section 13113 of P.L 103–66; sections 1, 3, 4, and 5 of P.L. 106–519; sections 101, 102, and 422 of P.L 108–357; sections 1310 and 1351 of P.L. 109–58;

- section 11146 of P.L. 109-59; section 403 (q) of P.L. 109-135; section 513 of P.L. 109-222; sections 104 and 307 of P.L. 109-432; sections 8233 and 8235 of P.L. 110-28; section 11 (e) and (g) of P.L. 110-172; section 301 of P.L. 110-245; section 15351 of P.L. 110-246; section 302 of division A, section 401 of division B, and sections 312, 322, 502 (c), 707, and 801 of division C of P.L. 110-343; sections 1232, 1241, 1251, 1501, and 1502 of division B of P.L. 111-5; sections 211, 212, 213, 214, and 216 of P.L. 111-226; sections 2011 and 2122 of P.L. 111-240; sections 753, 754, and 760 of P.L. 111-312; section 1106 of P.L. 112-95; sections 104, 318, 322, 323, 324, 326, 327, and 411 of P.L. 112-240; P.L. 114-7; section 1101 of P.L. 114-74; section 305 of division P of P.L. 114-113; and sections 112, 123, 125 to 128, 143, 144, 151 to 153, 165 to 167, 169 to 171, 189, 191, 307, 326, and 411 of division Q of P.L. 114–113.
- 3. For purposes of this paragraph, "Internal Revenue Code" does not include amendments to the federal Internal Revenue Code enacted after December 31, 2016.
- 4. For purposes of this paragraph, the provisions of federal public laws that directly or indirectly affect the Internal Revenue Code, as defined in this paragraph, apply for Wisconsin purposes at the same time as for federal purposes, except that changes made by section 4007 (b) of P.L. 114–41, section 1102 of P.L. 114–74, sections 105, 111, 113 to 115, 133, 301, 302, 304, 305, 308, 311, 313 to 323, 325, 331, and 343 to 345 of division Q of P.L. 114–113 first apply for taxable years beginning after December 31, 2016.

SECTION 1075. 71.25 (9) (dh) 2. b. of the statutes is amended to read:

71.25 (9) (dh) 2. b. The service relates to tangible personal property that is located in this state at the time that the service is received or tangible personal property that is delivered directly or indirectly to customers in this state.

SECTION 1076. 71.25 (9) (dh) 2. c. of the statutes is amended to read:

71.25 (9) (dh) 2. c. The service is provided to purchased by an individual who is physically present in this state at the time that the service is received.

SECTION 1076d. 71.25 (9) (dh) 3. of the statutes is amended to read:

71.25 (9) (dh) 3. If Except as provided in subd. 4. if the purchaser of a service receives the benefit of a service in more than one state, the gross receipts from the performance of the service are included in the numerator of the sales factor according to the portion of the service received in this state.

SECTION 1076e. 71.25 (9) (dh) 4. of the statutes is created to read:

71.25 (9) (dh) 4. For taxable years beginning after December 31, 2018, a broadcaster's gross receipts from advertising are in this state only if the advertiser's commercial domicile is in this state. With regard to a broad-

caster who is a member of a combined group, as defined in s. 71.255 (1) (a), this subdivision does not apply to the gross receipts of the members who are not broadcasters.

SECTION 1076f. 71.25 (9) (dj) (intro.) of the statutes is renumbered 71.25 (9) (dj) 1. (intro.) and amended to read:

71.25 (9) (dj) 1. (intro.) Except as provided in <u>subd.</u> 2m. and par. (df), gross royalties and other gross receipts received for the use or license of intangible property, including patents, copyrights, trademarks, trade names, service names, franchises, licenses, plans, specifications, blueprints, processes, techniques, formulas, designs, layouts, patterns, drawings, manuals, technical know-how, contracts, and customer lists, are sales in this state if any of the following applies:

SECTION 1076g. 71.25 (9) (dj) 1. of the statutes is renumbered 71.25 (9) (dj) 1. a. and amended to read:

71.25 (9) (dj) 1. a. The purchaser or licensee uses the intangible property in the operation of a trade or business at a location in this state. If Except as provided in subd. 2m., if the purchaser or licensee uses the intangible property in the operation of a trade or business in more than one state, the gross royalties and other gross receipts from the use of the intangible property shall be divided between those states having jurisdiction to impose an income tax on the taxpayer in proportion to the use of the intangible property in those states.

SECTION 1076h. 71.25 (9) (dj) 2. of the statutes is renumbered 71.25 (9) (dj) 1. b.

SECTION 1076i. 71.25 (9) (dj) 2m. of the statutes is created to read:

71.25 (9) (dj) 2m. For taxable years beginning after December 31, 2018, a broadcaster's gross royalties and other gross receipts received for the use or license of intangible property are sales in this state only if the commercial domicile of the purchaser or licensee is in this state and the purchaser or licensee has a direct connection or relationship with the broadcaster pursuant to a contract under which the royalties or receipts are derived. With regard to a broadcaster who is a member of a combined group, as defined in s. 71.255 (1) (a), this subdivision does not apply to the gross royalties and receipts of the members who are not broadcasters.

SECTION 1076j. 71.25 (9) (dj) 3. of the statutes is renumbered 71.25 (9) (dj) 1. c.

SECTION 1076k. 71.25 (9) (g) of the statutes is created to read:

71.25 (9) (g) 1. For taxable years beginning after December 31, 2018, the amount of a broadcaster's gross receipts from advertising and the use or license of intangible property, as determined under pars. (dh) 4. and (dj) 2m., shall be adjusted as follows:

- a. Determine the amount of the numerator of the sales factor for a broadcaster as provided in this subsection.
- b. Multiply .01 by the total amount of the domestic gross receipts of the broadcaster from advertising and

royalties and other gross receipts for the use or license of intangible property.

- c. Determine the numerator of the sales for a broadcaster by substituting the amount determined under subd. 1. b. for the total amount determined under subd. 1. a.
- d. Except as provided in subd. 1. e., if the amount of the numerator determined under subd. 1. c. is more than the amount determined under subd. 1. a., substitute the amount of total gross receipts determined under subd. 1. b. for the total amount of the gross receipts determined under subd. 1. a. For purposes of this subd. 1. d., the amount of the numerator for a broadcaster is the amount determined under subd. 1. c.
- e. If the amount of the numerator computed under subd. 1. c. is more than 140 percent of the amount determined under subd. 1. a., adjust the total amount of the gross receipts under subd. 1. a. so that the amount of the numerator for a broadcaster is 140 percent of the numerator otherwise determined under subd. 1. a.
- 2. The department may promulgate rules to administer this paragraph.

SECTION 1078m. 71.26 (2) (a) 4. of the statutes is amended to read:

71.26 (2) (a) 4. Plus the amount of the credit computed under s. 71.28 (1dm), (1dx), (1dy), (3g), (3h), (3n), (3p), (3q), (3r), (3rm), (3rn), (3t), (3w), (3y), (5e), (5f), (5g), (5h), (5i), (5j), (5k), (5r), (5rm), (6n), (8r), and (9s) and not passed through by a partnership, limited liability company, or tax–option corporation that has added that amount to the partnership's, limited liability company's, or tax–option corporation's income under s. 71.21 (4) or 71.34 (1k) (g).

SECTION 1079. 71.26 (2) (b) 2. of the statutes is repealed.

SECTION 1080. 71.26 (2) (b) 10. a. of the statutes is amended to read:

71.26 (2) (b) 10. a. For taxable years beginning after December 31, 2013, and before January 1, 2017, for a corporation, conduit, or common law trust which qualifies as a regulated investment company, real estate mortgage investment conduit, real estate investment trust, or financial asset securitization investment trust under the Internal Revenue Code, "net income" means the federal regulated investment company taxable income, federal real estate mortgage investment conduit taxable income, federal real estate investment trust or financial asset securitization investment trust taxable income of the corporation, conduit, or trust as determined under the Internal Revenue Code.

SECTION 1081. 71.26 (2) (b) 10. d. of the statutes is amended to read:

71.26 (2) (b) 10. d. For purposes of subd. 10. a., "Internal Revenue Code" does not include amendments to the federal Internal Revenue Code enacted after December 31, 2013, except that "Internal Revenue Code" includes the provisions of P.L. 113–97, P.L.

113–159, P.L. 113–168, section 302901 of P.L. 113–287, sections 171, 172, and 201 to 221 of P.L. 113–295, sections 102, 105, and 207 of division B of P.L. 113–295, P.L. 114–14, and P.L. 114–26, section 2004 of P.L. 114–41, sections 503 and 504 of P.L. 114–74, sections 103, 104, 124, 168, 184, 185, 190, 204, 303, 306, 336, and 341 of division Q of P.L. 114–113, and P.L. 114–239.

SECTION 1082. 71.26 (2) (b) 11. of the statutes is created to read:

71.26 (2) (b) 11. a. For taxable years beginning after December 31, 2016, for a corporation, conduit, or common law trust which qualifies as a regulated investment company, real estate mortgage investment conduit, real estate investment trust, or financial asset securitization investment trust under the Internal Revenue Code, "net income" means the federal regulated investment company taxable income, federal real estate mortgage investment conduit taxable income, federal real estate investment trust or financial asset securitization investment trust taxable income of the corporation, conduit, or trust as determined under the Internal Revenue Code.

- b. For purposes of subd. 11. a., "Internal Revenue Code" means the federal Internal Revenue Code as amended to December 31, 2016, except as provided in subd. 11. c. and d. and s. 71.98 and subject to subd. 11. e.
- c. For purposes of subd. 11. a., "Internal Revenue Code" does not include the following provisions of federal public laws for taxable years beginning after December 31, 2016: section 13113 of P.L 103–66; sections 1, 3, 4, and 5 of P.L. 106-519; sections 101, 102, and 422 of P.L 108-357; sections 1310 and 1351 of P.L. 109-58; section 11146 of P.L. 109-59; section 403 (q) of P.L. 109-135; section 513 of P.L. 109-222; sections 104 and 307 of P.L. 109-432; sections 8233 and 8235 of P.L. 110-28; section 11 (e) and (g) of P.L. 110-172; section 301 of P.L. 110-245; section 15351 of P.L. 110-246; section 302 of division A, section 401 of division B, and sections 312, 322, 502 (c), 707, and 801 of division C of P.L. 110-343; sections 1232, 1241, 1251, 1501, and 1502 of division B of P.L. 111-5; sections 211, 212, 213, 214, and 216 of P.L. 111-226; sections 2011 and 2122 of P.L. 111-240; sections 753, 754, and 760 of P.L. 111-312; section 1106 of P.L. 112-95; sections 104, 318, 322, 323, 324, 326, 327, and 411 of P.L. 112-240; P.L. 114-7; section 1101 of P.L. 114-74; section 305 of division P of P.L. 114–113; and sections 112, 123, 125 to 128, 143, 144, 151 to 153, 165 to 167, 169 to 171, 189, 191, 307, 326, and 411 of division Q of P.L. 114-113.
- d. For purposes of subd. 11. a., "Internal Revenue Code" does not include amendments to the federal Internal Revenue Code enacted after December 31, 2016.
- e. For purposes of subd. 11. a., the provisions of federal public laws that directly or indirectly affect the Internal Revenue Code, as defined in this subdivision, apply for Wisconsin purposes at the same time as for federal purposes, except that changes made by section 4007 (b)

of P.L. 114–41, section 1102 of P.L. 114–74, sections 105, 111, 113 to 115, 133, 301, 302, 304, 305, 308, 311, 313 to 323, 325, 331, and 343 to 345 of division Q of P.L. 114–113 first apply for taxable years beginning after December 31, 2016.

SECTION 1082m. 71.26 (3) (cf) of the statutes is created to read:

71.26 (3) (cf) For taxable years beginning after December 31, 2016, section 118 (a) (relating to nonshare-holder contributions to capital) is modified so that the amount of income and franchise tax credits under s. 71.28 (3q) (b), (3w) (b) and (bm) 1., 2., and 4., (3wm) (b) and (bm), and (3y) (b) that is not included in federal taxable income is added to federal taxable income.

SECTION 1083. 71.26 (4) (a) of the statutes is amended to read:

71.26 (4) (a) Except as provided in par. (b) and s. 71.80 (25), a corporation, except a tax-option corporation or an insurer to which s. 71.45 (4) applies, may offset against its Wisconsin net business income any Wisconsin net business loss sustained incurred in any of the next 20 immediately preceding taxable years, if the corporation was subject to taxation under this chapter in the taxable year in which the loss was sustained incurred, to the extent not offset by other items of Wisconsin income in the loss year and by Wisconsin net business income of any year between the loss year and the taxable year for which an offset is claimed. For purposes of this subsection, Wisconsin net business income or loss shall consist of all the income attributable to the operation of a trade or business in this state, less the business expenses allowed as deductions in computing net income. The Wisconsin net business income or loss of corporations engaged in business within and without the state shall be determined under s. 71.25 (6) and (10) to (12). Nonapportionable losses having a Wisconsin situs under s. 71.25 (5) (b) shall be included in Wisconsin net business loss; and nonapportionable income having a Wisconsin situs under s. 71.25 (5) (b), whether taxable or exempt, shall be included in other items of Wisconsin income and Wisconsin net business income for purposes of this sub-

SECTION 1083x. 71.28 (3q) (c) 3. of the statutes is amended to read:

71.28 (**3q**) (c) 3. The maximum amount of credits that may be awarded under this subsection and ss. 71.07 (3q) and 71.47 (3q) for the period beginning on January 1, 2010, and ending on June 30, 2013, is \$14,500,000, not including the amount of any credits reallocated under s. 238.15 (3) (d), 2015 stats., or s. 560.205 (3) (d), 2009 stats.

SECTION 1084. 71.28 (3q) (d) 2. of the statutes is amended to read:

71.28 (**3q**) (d) 2. If the allowable amount of the claim under par. (b) exceeds the tax otherwise due under s. 71.23, the amount of the claim not used to offset the tax

due shall be certified by the department of revenue to the department of administration for payment by check, share draft, or other draft drawn from the appropriation account under s. 20.835 (2) (bb), except that the amounts certified under this subdivision for taxable years beginning after December 31, 2009, and before January 1, 2012, shall be paid in taxable years beginning after December 31, 2011. Notwithstanding s. 71.82, no interest shall be paid on amounts certified under this subdivision.

Vetoed In Part

SECTION 1085ba. 71.28 (3w) (b) (intro.) of the statutes is amended to read:

71.28 (3w) (b) Filing claims; payroll. (intro.) Subject to the limitations provided in this subsection and s. 238.399 or s. 560.799, 2009 stats., a claimant may claim as a credit against the tax imposed under s. 71.23 an amount calculated as follows:

SECTION 1085bb. 71.28 (3w) (bm) 1. of the statutes is amended to read:

71.28 (3w) (bm) 1. In addition to the credits under par. (b) and subds. 2., 3., and 4., and subject to the limitations provided in this subsection and s. 238.399 or s. 560.799, 2009 stats., a claimant may claim as a credit against the tax imposed under s. 71.23 an amount equal to a percentage, as determined under s. 238.399 or s. 560.799, 2009 stats., not to exceed 100 percent, of the amount the claimant paid in the taxable year to upgrade or improve the job-related skills of any of the claimant's full-time employees, to train any of the claimant's fulltime employees on the use of job-related new technologies, or to provide job-related training to any full-time employee whose employment with the claimant represents the employee's first full-time job. This subdivision does not apply to employees who do not work in an enterprise zone.

SECTION 1085bc. 71.28 (3w) (bm) 2. of the statutes is amended to read:

71.28 (3w) (bm) 2. In addition to the credits under par. (b) and subds. 1., 3., and 4., and subject to the limitations provided in this subsection and s. 238.399 or s. 560.799, 2009 stats., a claimant may claim as a credit against the tax imposed under s. 71.23 an amount equal to the percentage, as determined under s. 238.399 or s. 560.799, 2009 stats., not to exceed 7 percent, of the claimant's zone payroll paid in the taxable year to all of the claimant's full-time employees whose annual wages are greater than the amount determined by multiplying 2,080 by 150 percent of the federal minimum wage in a tier I county or municipality, not including the wages paid to the employees determined under par. (b) 1., or greater than \$30,000 in a tier II county or municipality, not including the wages paid to the employees determined under par. (b) 1., and who the claimant employed in the enterprise zone in the taxable year, if the total number of such employees is equal to or greater than the total number of such employees in the base year. A claimant may claim a credit under this subdivision for no more than 5 consecutive taxable years.

SECTION 1085bd. 71.28 (3w) (bm) 3. of the statutes is amended to read:

71.28 (3w) (bm) 3. In addition to the credits under par. (b) and subds. 1., 2., and 4., and subject to the limitations provided in this subsection and s. 238.399 or s. 560.799, 2009 stats., for taxable years beginning after December 31, 2008, a claimant may claim as a credit against the tax imposed under s. 71.23 up to 10 percent of the claimant's significant capital expenditures, as determined under s. 238.399 (5m) or s. 560.799 (5m), 2009 stats.

SECTION 1085be. 71.28 (3w) (bm) 4. of the statutes is amended to read:

71.28 (3w) (bm) 4. In addition to the credits under par. (b) and subds. 1., 2., and 3., and subject to the limitations provided in this subsection and s. 238.399 or s. 560.799, 2009 stats., for taxable years beginning after December 31, 2009, a claimant may claim as a credit against the tax imposed under s. 71.23, up to 1 percent of the amount that the claimant paid in the taxable year to purchase tangible personal property, items, property, or goods under s. 77.52 (1) (b), (c), or (d), or services from Wisconsin vendors, as determined under s. 238.399 (5) (e) or s. 560.799 (5) (e), 2009 stats., except that the claimant may not claim the credit under this subdivision and subd. 3. for the same expenditures.

SECTION 1085d. 71.28 (3w) (c) 1. of the statutes is renumbered 71.28 (3w) (c) 1. a. and amended to read:

71.28 (3w) (c) 1. a. If For claims filed before January Vetoed 1, 2018, if the allowable amount of the claim under this subsection exceeds the taxes otherwise due on the claimant's income under s. 71.23, the amount of the claim that is not used to offset those taxes shall be certified by the department of revenue to the department of administration for payment by check, share draft, or other draft drawn from the appropriation under s. 20.835 (2) (co). Notwithstanding s. 71.82, no interest shall be paid on amounts certified under this subd. 1. a.

SECTION 1085e. 71.28 (3w) (c) 1. b. of the statutes is **Vetoed** created to read:

71.28 (3w) (c) 1. b. For claims filed after December 31, 2017, claims under this subsection shall be made to the Wisconsin Economic Development Corporation using policies and procedures established by the corporation board. The corporation shall certify valid claims to the department of administration for payment by check, share draft, or other draft drawn from the appropriation under s. 20.835 (2) (co). Notwithstanding s. 71.82, no interest shall be paid on amounts certified under this

SECTION 1085f. 71.28 (3w) (c) 2. of the statutes is amended to read:

subd. 1. b.

71.28 (3w) (c) 2. Partnerships For claims filed before January 1, 2018, partnerships, limited liability comVetoed In Part

In Part

In Part

panies, and tax-option corporations may not claim the credit under this subsection, but the eligibility for, and the amount of, the credit are based on their payment of amounts described under pars. (b) and (bm). A partnership, limited liability company, or tax-option corporation shall compute the amount of credit that each of its partners, members, or shareholders may claim and shall provide that information to each of them. Partners, members of limited liability companies, and shareholders of tax-option corporations may claim the credit in proportion to their ownership interests. For claims filed after December 31, 2017, partnerships, limited liability companies, and tax-option corporations may claim the credit under this subsection as provided under subd. 1. b.

SECTION 1085g. 71.28 (3w) (c) 3. of the statutes is amended to read:

71.28 (3w) (c) 3. No For claims filed before January 1, 2018, no credit may be allowed under this subsection unless the claimant includes with the claimant's return a copy of the claimant's certification for tax benefits under s. 238.399 (5) or (5m) or s. 560.799 (5) or (5m), 2009 stats.

SECTION 1085h. 71.28 (3w) (d) of the statutes is amended to read:

71.28 (3w) (d) Administration. Subsection (4) (g) and (h), as it applies to the credit under sub. (4), applies to the credit under this subsection. Claimants shall include with their returns a copy of their certification for tax benefits, and a copy of the verification of their expenses, from the department of commerce or the Wisconsin Economic Development Corporation. This paragraph does not apply to claims filed after December 31, 2017.

SECTION 1085i. 71.28 (3wm) of the statutes is created to read:

- 71.28 (3wm) ELECTRONICS AND INFORMATION TECH-NOLOGY MANUFACTURING ZONE CREDIT. (a) Definitions. In this subsection:
- 1. "Claimant" means a person who is certified to claim tax benefits under s. 238.396 (3) and who files a claim under this subsection.
- 2. "Full-time employee" means an individual who is employed in a job for which the annual pay is at least \$30,000 and who is offered retirement, health, and other benefits that are equivalent to the retirement, health, and other benefits offered to an individual who is required to work at least 2,080 hours per year.
- 3. "State payroll" means the amount of payroll apportioned to this state, as determined under s. 71.25 (8).
- 4. "Wages" means wages under section 3306 (b) of the Internal Revenue Code, determined without regard to any dollar limitations.
- 5. "Zone" means a zone designated under s. 238.396 (1m).
- 6. "Zone payroll" means the amount of state payroll that is attributable to wages paid by the claimant to fulltime employees for services that are performed in the

zone or that are performed outside the zone, but within the state, and for the benefit of the operations within the zone, as determined by the Wisconsin Economic Development Corporation. "Zone payroll" does not include the amount of wages paid to any full-time employees that exceeds \$100,000.

- (b) Filing claims; payroll. Subject to the limitations provided in this subsection and s. 238.396, a claimant may claim as a credit an amount calculated as follows:
- 1. Determine the zone payroll for the taxable year for full-time employees employed by the claimant.
- 2. Multiply the amount determined under subd. 1. by 17 percent.
- (bm) Filing supplemental claims. In addition to claiming the credit under par. (b), and subject to the limitations under this subsection and s. 238.396, a claimant may claim as a credit up to 15 percent of the claimant's significant capital expenditures in the zone in the taxable year, as determined under s. 238.396 (3m).
- (c) Limitations. Partnerships, limited liability companies, and tax-option corporations may claim the credit In Part under this subsection as provided under par. (d). The Wisconsin Economic Development Corporation may recover credits claimed under this paragraph that are revoked or otherwise invalid from the partnership, limited liability company, or tax-option corporation or from the individual partner, member, or shareholder.
- (d) Administration. Claims under this subsection Vetoed shall be made to the Wisconsin Economic Development Corporation using policies and procedures established by the corporation board. The corporation shall certify valid claims to the department of administration for payment by check, share draft, or other draft drawn from the appropriation under s. 20.835 (2) (cp). Notwithstanding s. 71.82, no interest shall be paid on amounts certified under this subdivision.

SECTION 1086b. 71.28 (3y) (b) (intro.) of the statutes is amended to read:

71.28 (3y) (b) Filing claims. (intro.) Subject to the limitations provided in this subsection and s. 238.308, for taxable years beginning after December 31, 2015, a claimant may claim as a credit against the tax imposed under s. 71.23 all of the following:

SECTION 1086d. 71.28 (3y) (c) 1. of the statutes is amended to read:

71.28 (3y) (c) 1. Partnerships For claims filed before January 1, 2018, partnerships, limited liability companies, and tax-option corporations may not claim the credit under this subsection, but the eligibility for, and the amount of, the credit are based on their payment of amounts under par. (b). A partnership, limited liability company, or tax-option corporation shall compute the amount of credit that each of its partners, members, or shareholders may claim and shall provide that information to each of them. Partners, members of limited liability companies, and shareholders of tax-option corporaVetoed

In Part

Vetoed In Part tions may claim the credit in proportion to their ownership interests. For claims filed after December 31, 2017, partnerships, limited liability companies, and taxoption corporations may claim the credit under this subsection as provided under par. (d) 2. b.

SECTION 1086e. 71.28 (3y) (c) 2. of the statutes is amended to read:

71.28 (**3y**) (c) 2. No For claims filed before January 1, 2018, no credit may be allowed under this subsection unless the claimant includes with the claimant's return a copy of the claimant's certification for tax benefits under s. 238.308.

SECTION 1086f. 71.28 (3y) (d) 1. of the statutes is amended to read:

71.28 (**3y**) (d) 1. Subsection (4) (e), (g), and (h), as it applies to the credit under sub. (4), applies to the credit under this subsection. This subdivision does not apply to claims filed after December 31, 2017.

SECTION 1086g. 71.28 (3y) (d) 2. of the statutes is renumbered 71.28 (3y) (d) 2. a. and amended to read:

71.28 (3y) (d) 2. a. If For claims filed before January 1, 2018, if the allowable amount of the claim under par. (b) exceeds the tax otherwise due under s. 71.23, the amount of the claim not used to offset the tax due shall be certified by the department of revenue to the department of administration for payment by check, share draft, or other draft drawn from the appropriation account under s. 20.835 (2) (bg). Notwithstanding s. 71.82, no interest shall be paid on amounts certified under this subd. 2. a.

Vetoed In Part

Vetoed In Part

SECTION 1086h. 71.28 (3y) (d) 2. b. of the statutes is created to read:

71.28 (**3y**) (d) 2. b. For claims filed after December 31, 2017, claims under this subsection shall be made to the Wisconsin Economic Development Corporation using policies and procedures established by the corporation board. The corporation shall certify valid claims to the department of administration for payment by check, share draft, or other draft drawn from the appropriation under s. 20.835 (2) (bg). Notwithstanding s. 71.82, no interest shall be paid on amounts certified under this subd. 2. b.

SECTION 1086i. 71.28(4)(k) of the statutes is created to read:

71.28 (4) (k) *Refunds*. Notwithstanding par. (f), for taxable years beginning after December 31, 2017, if the allowable amount of the claim under par. (ad) 4., 5., or 6. exceeds the tax otherwise due under s. 71.23, all of the following apply:

1. The amount of the claim not used to offset the tax due, not to exceed 10 percent of the allowable amount of the claim under par. (ad) 4., 5., or 6., shall be certified by the department of revenue to the department of administration for payment by check, share draft, or other draft drawn from the appropriation account under s. 20.835 (2) (d).

2. The amount of the claim not used to offset the tax due and not certified for payment under subd. 1. may be carried forward and credited against Wisconsin income or franchise taxes otherwise due for the following 15 taxable years to the extent not offset by these taxes otherwise due in all intervening years between the year in which the expense was incurred and the year in which the carryforward credit is claimed.

SECTION 1086m. 71.28 (5b) (d) 3. of the statutes is amended to read:

71.28 (**5b**) (d) 3. Except as provided under s. 238.15 (3) (d) (intro.), for investments made after December 31, 2007, if an investment for which a claimant claims a credit under par. (b) is held by the claimant for less than 3 years, the claimant shall pay to the department, in the manner prescribed by the department, the amount of the credit that the claimant received related to the investment.

SECTION 1087. 71.28 (5r) (a) 2. of the statutes is amended to read:

71.28 (5r) (a) 2. "Course of instruction" has the meaning given in s. 38.50 ± 440.52 (1) (c).

SECTION 1088. 71.28 (5r) (a) 6. b. of the statutes is amended to read:

71.28 (**5r**) (a) 6. b. A school approved under s. 38.50 440.52, if the delivery of education occurs in this state.

SECTION 1088u. 71.28 (6) (h) of the statutes is amended to read:

71.28 (6) (h) Any person, including a nonprofit entity described in section 501 (c) (3) of the Internal Revenue Code, may sell or otherwise transfer the credit under par. (a) 2m. or 3., in whole or in part, to another person who is subject to the taxes imposed under s. 71.02, 71.08, 71.23, or 71.43, if the person notifies the department of the transfer, and submits with the notification a copy of the transfer documents, and the department certifies ownership of the credit with each transfer.

SECTION 1089. 71.28 (6) (i) of the statutes is created to read:

71.28 (6) (i) If a person who claims a credit under this subsection and a credit under section 47 of the Internal Revenue Code for the same qualified rehabilitation expenditures is required to repay any amount of the credit claimed under section 47 of the Internal Revenue Code, the person shall repay to the department a proportionate amount of the credit claimed under this subsection.

SECTION 1090j. 71.30 (3) (c) of the statutes is amended to read:

71.30 (3) (c) Research credit under s. 71.28 (4), except as provided under par. (f).

SECTION 1090k. 71.30 (3) (f) of the statutes is amended to read:

71.30 (3) (f) The total of farmland preservation credit under subch. IX, farmland tax relief credit under s. 71.28 (2m), dairy manufacturing facility investment credit

under s. 71.28 (3p), jobs credit under s. 71.28 (3q), meat processing facility investment credit under s. 71.28 (3r), woody biomass harvesting and processing credit under s. 71.28 (3rm), food processing plant and food warehouse investment credit under s. 71.28 (3rn), enterprise zone jobs credit under s. 71.28 (3w), business development credit under s. 71.28 (3y), research credit under s. 71.28 (4) (k) 1., film production services credit under s. 71.28 (5f), film production company investment credit under s. 71.28 (5h), beginning farmer and farm asset owner tax credit under s. 71.28 (8r), and estimated tax payments under s. 71.29.

SECTION 1091. 71.34 (1g) (b) of the statutes is repealed.

SECTION 1092. 71.34 (1g) (j) 1. of the statutes is amended to read:

71.34 (**1g**) (j) 1. For taxable years beginning after December 31, 2013, and before January 1, 2017, for tax option corporations, "Internal Revenue Code" means the federal Internal Revenue Code as amended to December 31, 2013, except as provided in subds. 2., 3., and 5. and subject to subd. 4.

SECTION 1093. 71.34 (1g) (j) 3. i. of the statutes is created to read:

71.34 (**1g**) (j) 3. i. Section 2004 of P.L. 114–41.

SECTION 1094. 71.34 (1g) (j) 3. j. of the statutes is created to read:

71.34 (**1g**) (j) 3. j. Sections 503 and 504 of P.L. 114–74.

SECTION 1095. 71.34 (1g) (j) 3. k. of the statutes is created to read:

71.34 (**1g**) (j) 3. k. Sections 103, 104, 124, 168, 184, 185, 190, 204, 303, 306, 336, and 341 of division Q of PL, 114–113

SECTION 1096. 71.34 (1g) (j) 3. L. of the statutes is created to read:

71.34 (**1g**) (j) 3. L. P.L. 114–239.

SECTION 1097. 71.34 (1g) (k) of the statutes is created to read:

- 71.34 (**1g**) (k) 1. For taxable years beginning after December 31, 2016, for tax option corporations, "Internal Revenue Code" means the federal Internal Revenue Code as amended to December 31, 2016, except as provided in subds. 2., 3., and 5. and s. 71.98 and subject to subd. 4.
- 2. For purposes of this paragraph, "Internal Revenue Code" does not include the following provisions of federal public laws for taxable years beginning after December 31, 2016: section 13113 of P.L 103–66; sections 1, 3, 4, and 5 of P.L. 106–519; sections 101, 102, and 422 of P.L 108–357; sections 1310 and 1351 of P.L. 109–58; section 11146 of P.L. 109–59; section 403 (q) of P.L. 109–135; section 513 of P.L. 109–222; sections 104 and 307 of P.L. 109–432; sections 8233 and 8235 of P.L. 110–28; section 11 (e) and (g) of P.L. 110–172; section 301 of P.L. 110–245; section 15351 of P.L. 110–246; sec-

tion 302 of division A, section 401 of division B, and sections 312, 322, 502 (c), 707, and 801 of division C of P.L. 110–343; sections 1232, 1241, 1251, 1501, and 1502 of division B of P.L. 111–5; sections 211, 212, 213, 214, and 216 of P.L. 111–226; sections 2011 and 2122 of P.L. 111–240; sections 753, 754, and 760 of P.L. 111–312; section 1106 of P.L. 112–95; sections 104, 318, 322, 323, 324, 326, 327, and 411 of P.L. 112–240; P.L. 114–7; section 1101 of P.L. 114–74; section 305 of division P of P.L. 114–113; and sections 112, 123, 125 to 128, 143, 144, 151 to 153, 165 to 167, 169 to 171, 189, 191, 307, 326, and 411 of division Q of P.L. 114–113.

- 3. For purposes of this paragraph, "Internal Revenue Code" does not include amendments to the federal Internal Revenue Code enacted after December 31, 2016.
- 4. For purposes of this paragraph, the provisions of federal public laws that directly or indirectly affect the Internal Revenue Code, as defined in this paragraph, apply for Wisconsin purposes at the same time as for federal purposes, except that changes made by section 4007 (b) of P.L. 114–41, section 1102 of P.L. 114–74, sections 105, 111, 113 to 115, 133, 301, 302, 304, 305, 308, 311, 313 to 323, 325, 331, and 343 to 345 of division Q of P.L. 114–113 first apply for taxable years beginning after December 31, 2016.
- 5. For purposes of this paragraph, section 1366 (f) of the Internal Revenue Code (relating to pass—through of items to shareholders) is modified by substituting the tax under s. 71.35 for the taxes under sections 1374 and 1375 of the Internal Revenue Code.

SECTION 1098. 71.42 (2) (b) of the statutes is repealed.

SECTION 1099. 71.42 (2) (j) 1. of the statutes is amended to read:

71.42 (2) (j) 1. For taxable years beginning after December 31, 2013, and before January 1, 2017, "Internal Revenue Code" means the federal Internal Revenue Code as amended to December 31, 2013, except as provided in subds. 2. to 4. and subject to subd. 5.

SECTION 1100. 71.42 (2) (j) 3. i. of the statutes is created to read:

71.42 (2) (j) 3. i. Section 2004 of P.L. 114-41.

SECTION 1101. 71.42 (2) (j) 3. j. of the statutes is created to read:

71.42 (**2**) (j) 3. j. Sections 503 and 504 of P.L. 114–74. **SECTION 1102.** 71.42 (2) (j) 3. k. of the statutes is created to read:

71.42 (**2**) (j) 3. k. Sections 103, 104, 124, 168, 184, 185, 190, 204, 303, 306, 336, and 341 of division Q of P.L. 114–113.

SECTION 1103. 71.42 (2) (j) 3. L. of the statutes is created to read:

71.42 (2) (j) 3. L. P.L. 114-239.

SECTION 1104. 71.42 (2) (k) of the statutes is created to read:

71.42 (2) (k) 1. For taxable years beginning after December 31, 2016, "Internal Revenue Code" means the federal Internal Revenue Code as amended to December 31, 2016, except as provided in subds. 2. to 4. and s. 71.98 and subject to subd. 5.

- 2. For purposes of this paragraph, "Internal Revenue Code" does not include the following provisions of federal public laws for taxable years beginning after December 31, 2016: section 13113 of P.L 103–66; sections 1, 3, 4, and 5 of P.L. 106-519; sections 101, 102, and 422 of P.L 108-357; sections 1310 and 1351 of P.L. 109-58; section 11146 of P.L. 109-59; section 403 (q) of P.L. 109-135; section 513 of P.L. 109-222; sections 104 and 307 of P.L. 109-432; sections 8233 and 8235 of P.L. 110-28; section 11 (e) and (g) of P.L. 110-172; section 301 of P.L. 110-245; section 15351 of P.L. 110-246; section 302 of division A, section 401 of division B, and sections 312, 322, 502 (c), 707, and 801 of division C of P.L. 110-343; sections 1232, 1241, 1251, 1501, and 1502 of division B of P.L. 111-5; sections 211, 212, 213, 214, and 216 of P.L. 111-226; sections 2011 and 2122 of P.L. 111-240; sections 753, 754, and 760 of P.L. 111-312; section 1106 of P.L. 112-95; sections 104, 318, 322, 323, 324, 326, 327, and 411 of P.L. 112–240; P.L. 114–7; section 1101 of P.L. 114-74; section 305 of division P of P.L. 114–113; and sections 112, 123, 125 to 128, 143, 144, 151 to 153, 165 to 167, 169 to 171, 189, 191, 307, 326, and 411 of division Q of P.L. 114-113.
- 3. For purposes of this paragraph, "Internal Revenue Code" does not include amendments to the federal Internal Revenue Code enacted after December 31, 2016.
- 4. For purposes of this paragraph, "Internal Revenue Code" does not include section 847 of the federal Internal
- 5. For purposes of this paragraph, the provisions of federal public laws that directly or indirectly affect the Internal Revenue Code, as defined in this paragraph, apply for Wisconsin purposes at the same time as for federal purposes, except that changes made by section 4007 (b) of P.L. 114-41, section 1102 of P.L. 114-74, sections 105, 111, 113 to 115, 133, 301, 302, 304, 305, 308, 311, 313 to 323, 325, 331, and 343 to 345 of division Q of P.L. 114-113 first apply for taxable years beginning after December 31, 2016.

SECTION 1108. 71.45 (4) (a) of the statutes is amended to read:

71.45 (4) (a) Except as provided in par. (b) and s. 71.80 (25), insurers computing tax under this subchapter may subtract from Wisconsin net income any Wisconsin net business loss sustained incurred in any of the next 20 immediately preceding taxable years, if the insurer was subject to taxation under this chapter in the taxable year in which the loss was incurred, to the extent not offset by Wisconsin net business income of any year between the loss year and the taxable year for which an offset is claimed and computed without regard to sub. (2) (a) 8.

and 9. and this subsection and limited to the amount of net income, but no loss incurred for a taxable year before taxable year 1987 by a nonprofit service plan of sickness care under ch. 148, or dental care under s. 447.13 may be treated as a net business loss of the successor service insurer under ch. 613 operating by virtue of s. 148.03 or 447.13.

SECTION 1108x. 71.47 (3q) (c) 3. of the statutes is amended to read:

71.47 (3q) (c) 3. The maximum amount of credits that may be awarded under this subsection and ss. 71.07 (3q) and 71.28 (3q) for the period beginning on January 1, 2010, and ending on June 30, 2013, is \$14,500,000, not including the amount of any credits reallocated under s. 238.15 (3) (d), 2015 stats., or s. 560.205 (3) (d), 2009

SECTION 1109. 71.47 (3q) (d) 2. of the statutes is amended to read:

71.47 (3q) (d) 2. If the allowable amount of the claim under par. (b) exceeds the tax otherwise due under s. 71.43, the amount of the claim not used to offset the tax due shall be certified by the department of revenue to the department of administration for payment by check, share draft, or other draft drawn from the appropriation account under s. 20.835 (2) (bb), except that the amounts certified under this subdivision for taxable years beginning after December 31, 2009, and before January 1, 2012, shall be paid in taxable years beginning after December 31, 2011. Notwithstanding s. 71.82, no interest shall be paid on amounts certified under this subdivision.

Section 1110ba. 71.47 (3w) (b) (intro.) of the Vetoed statutes is amended to read:

71.47 (3w) (b) Filing claims; payroll. (intro.) Subject to the limitations provided in this subsection and s. 238.399 or s. 560.799, 2009 stats., a claimant may claim as a credit against the tax imposed under s. 71.43 an amount calculated as follows:

SECTION 1110bb. 71.47 (3w) (bm) 1. of the statutes is amended to read:

71.47 (3w) (bm) 1. In addition to the credits under par. (b) and subds. 2., 3., and 4., and subject to the limitations provided in this subsection and s. 238.399 or s. 560.799, 2009 stats., a claimant may claim as a credit against the tax imposed under s. 71.43 an amount equal to a percentage, as determined under s. 238.399 or s. 560.799, 2009 stats., not to exceed 100 percent, of the amount the claimant paid in the taxable year to upgrade or improve the job-related skills of any of the claimant's full-time employees, to train any of the claimant's fulltime employees on the use of job-related new technologies, or to provide job-related training to any full-time employee whose employment with the claimant represents the employee's first full-time job. This subdivision does not apply to employees who do not work in an enterprise zone.

In Part

Vetoed In Part **SECTION 1110bc.** 71.47 (3w) (bm) 2. of the statutes is amended to read:

71.47 (3w) (bm) 2. In addition to the credits under par. (b) and subds. 1., 3., and 4., and subject to the limitations provided in this subsection and s. 238.399 or s. 560.799, 2009 stats., a claimant may claim as a credit against the tax imposed under s. 71.43 an amount equal to the percentage, as determined under s. 238.399 or s. 560.799, 2009 stats., not to exceed 7 percent, of the claimant's zone payroll paid in the taxable year to all of the claimant's full-time employees whose annual wages are greater than the amount determined by multiplying 2,080 by 150 percent of the federal minimum wage in a tier I county or municipality, not including the wages paid to the employees determined under par. (b) 1., or greater than \$30,000 in a tier II county or municipality, not including the wages paid to the employees determined under par. (b) 1., and who the claimant employed in the enterprise zone in the taxable year, if the total number of such employees is equal to or greater than the total number of such employees in the base year. A claimant may claim a credit under this subdivision for no more than 5 consecutive taxable years.

SECTION 1110bd. 71.47 (3w) (bm) 3. of the statutes is amended to read:

71.47 (**3w**) (bm) 3. In addition to the credits under par. (b) and subds. 1., 2., and 4., and subject to the limitations provided in this subsection and s. 238.399 or s. 560.799, 2009 stats., for taxable years beginning after December 31, 2008, a claimant may claim as a credit against the tax imposed under s. 71.43 up to 10 percent of the claimant's significant capital expenditures, as determined under s. 238.399 (5m) or s. 560.799 (5m), 2009 stats.

SECTION 1110be. 71.47 (3w) (bm) 4. of the statutes is amended to read:

71.47 (**3w**) (bm) 4. In addition to the credits under par. (b) and subds. 1., 2., and 3., and subject to the limitations provided in this subsection and s. 238.399 or s. 560.799, 2009 stats., for taxable years beginning after December 31, 2009, a claimant may claim as a credit against the tax imposed under s. 71.43, up to 1 percent of the amount that the claimant paid in the taxable year to purchase tangible personal property, items, property, or goods under s. 77.52 (1) (b), (c), or (d), or services from Wisconsin vendors, as determined under s. 238.399 (5) (e) or s. 560.799 (5) (e), 2009 stats., except that the claimant may not claim the credit under this subdivision and subd. 3. for the same expenditures.

SECTION 1110d. 71.47 (3w) (c) 1. of the statutes is renumbered 71.47 (3w) (c) 1. a. and amended to read:

Vetoed In Part 71.47 (**3w**) (c) 1. a. **If** For claims filed before January 1, 2018, if the allowable amount of the claim under this subsection exceeds the taxes otherwise due on the claimant's income under s. 71.43, the amount of the claim that is not used to offset those taxes shall be certified by

the department of revenue to the department of administration for payment by check, share draft, or other draft drawn from the appropriation under s. 20.835 (2) (co). Notwithstanding s. 71.82, no interest shall be paid on amounts certified under this subd. 1. a.

SECTION 1110e. 71.47 (3w) (c) 1. b. of the statutes is created to read:

Vetoed In Part

71.47 (**3w**) (c) 1. b. For claims filed after December 31, 2017, claims under this subsection shall be made to the Wisconsin Economic Development Corporation using policies and procedures established by the corporation board. The corporation shall certify valid claims to the department of administration for payment by check, share draft, or other draft drawn from the appropriation under s. 20.835 (2) (co). Notwithstanding s. 71.82, no interest shall be paid on amounts certified under this subd. 1. b.

SECTION 1110f. 71.47 (3w) (c) 2. of the statutes is amended to read:

71.47 (3w) (c) 2. Partnerships For claims filed before January 1, 2018, partnerships, limited liability companies, and tax-option corporations may not claim the credit under this subsection, but the eligibility for, and the amount of, the credit are based on their payment of amounts described under pars. (b) and (bm). A partnership, limited liability company, or tax-option corporation shall compute the amount of credit that each of its partners, members, or shareholders may claim and shall provide that information to each of them. Partners, members of limited liability companies, and shareholders of tax-option corporations may claim the credit in proportion to their ownership interests. For claims filed after December 31, 2017, partnerships, limited liability companies, and tax-option corporations may claim the credit under this subsection as provided under subd. 1. b.

SECTION 1110g. 71.47 (3w) (c) 3. of the statutes is amended to read:

71.47 (**3w**) (c) 3. No For claims filed before January 1, 2018, no credit may be allowed under this subsection unless the claimant includes with the claimant's return a copy of the claimant's certification for tax benefits under s. 238.399 (5) or (5m) or s. 560.799 (5) or (5m), 2009 stats.

SECTION 1110h. 71.47 (3w) (d) of the statutes is amended to read:

71.47 (**3w**) (d) *Administration*. Section 71.28 (4) (g) and (h), as it applies to the credit under s. 71.28 (4), applies to the credit under this subsection. Claimants shall include with their returns a copy of their certification for tax benefits, and a copy of the verification of their expenses, from the department of commerce or the Wisconsin Economic Development Corporation. This paragraph does not apply to claims filed after December 31, 2017.

SECTION 1111b. 71.47 (3y) (b) (intro.) of the statutes is amended to read:

Vetoed In Part

71.47 (3y) (b) Filing claims. (intro.) Subject to the limitations provided in this subsection and s. 238.308, for taxable years beginning after December 31, 2015, a claimant may claim as a credit against the tax imposed under s. 71.43 all of the following:

SECTION 1111d. 71.47 (3y) (c) 1. of the statutes is amended to read:

71.47 (3y) (c) 1. Partnerships For claims filed before January 1, 2018, partnerships, limited liability companies, and tax-option corporations may not claim the credit under this subsection, but the eligibility for, and the amount of, the credit are based on their payment of amounts under par. (b). A partnership, limited liability company, or tax-option corporation shall compute the amount of credit that each of its partners, members, or shareholders may claim and shall provide that information to each of them. Partners, members of limited liability companies, and shareholders of tax-option corporations may claim the credit in proportion to their ownership interests. For claims filed after December 31, 2017, partnerships, limited liability companies, and taxoption corporations may claim the credit under this subsection as provided under par. (d) 2. b.

SECTION 1111e. 71.47 (3y) (c) 2. of the statutes is amended to read:

71.47 (3y) (c) 2. No For claims filed before January 1, 2018, no credit may be allowed under this subsection unless the claimant includes with the claimant's return a copy of the claimant's certification for tax benefits under s. 238.308.

SECTION 1111f. 71.47 (3y) (d) 1. of the statutes is amended to read:

71.47 (3y) (d) 1. Section 71.28 (4) (e), (g), and (h), as it applies to the credit under s. 71.28 (4), applies to the credit under this subsection. This subdivision does not apply to claims filed after December 31, 2017.

SECTION 1111g. 71.47 (3y) (d) 2. of the statutes is renumbered 71.47 (3y) (d) 2. a. and amended to read:

Vetoed In Part

71.47 (3y) (d) 2. a. If For claims filed before January 1, 2018, if the allowable amount of the claim under par. (b) exceeds the tax otherwise due under s. 71.43, the amount of the claim not used to offset the tax due shall be certified by the department of revenue to the department of administration for payment by check, share draft, or other draft drawn from the appropriation account under s. 20.835 (2) (bg). Notwithstanding s. 71.82, no interest shall be paid on amounts certified under this subd. 2. a.

Vetoed In Part

SECTION 1111h. 71.47 (3y) (d) 2. b. of the statutes is created to read:

71.47 (3y) (d) 2. b. For claims filed after December 31, 2017, claims under this subsection shall be made to the Wisconsin Economic Development Corporation using policies and procedures established by the corporation board. The corporation shall certify valid claims to the department of administration for payment by check, share draft, or other draft drawn from the appropriation under s. 20.835 (2) (bg). Notwithstanding s. 71.82, no Vetoed interest shall be paid on amounts certified under this In Part subd. 2. b.

SECTION 1111k. 71.47 (4) (k) of the statutes is created to read:

71.47 (4) (k) Refunds. Notwithstanding par. (f), for taxable years beginning after December 31, 2017, if the allowable amount of the claim under par. (ad) 4., 5., or 6. exceeds the tax otherwise due under s. 71.43, all of the following apply:

- 1. The amount of the claim not used to offset the tax due, not to exceed 10 percent of the allowable amount of the claim under par. (ad) 4., 5., or 6., shall be certified by the department of revenue to the department of administration for payment by check, share draft, or other draft drawn from the appropriation account under s. 20.835 (2)
- 2. The amount of the claim not used to offset the tax due and not certified for payment under subd. 1. may be carried forward and credited against Wisconsin income or franchise taxes otherwise due for the following 15 taxable years to the extent not offset by these taxes otherwise due in all intervening years between the year in which the expense was incurred and the year in which the carryforward credit is claimed.

SECTION 1111m. 71.47 (5b) (d) 3. of the statutes is amended to read:

71.47 (**5b**) (d) 3. Except as provided under s. 238.15 (3) (d) (intro.), for investments made after December 31, 2007, if an investment for which a claimant claims a credit under par. (b) is held by the claimant for less than 3 years, the claimant shall pay to the department, in the manner prescribed by the department, the amount of the credit that the claimant received related to the investment.

SECTION 1112. 71.47 (5r) (a) 2. of the statutes is amended to read:

71.47 (5r) (a) 2. "Course of instruction" has the meaning given in s. 38.50 440.52 (1) (c).

SECTION 1113. 71.47 (5r) (a) 6. b. of the statutes is amended to read:

71.47 (**5r**) (a) 6. b. A school approved under s. 38.50 440.52, if the delivery of education occurs in this state.

SECTION 1113u. 71.47 (6) (h) of the statutes is amended to read:

71.47 (6) (h) Any person, including a nonprofit entity described in section 501 (c) (3) of the Internal Revenue Code, may sell or otherwise transfer the credit under par. (a) 2m. or 3., in whole or in part, to another person who is subject to the taxes imposed under s. 71.02, 71.08, 71.23, or 71.43, if the person notifies the department of the transfer, and submits with the notification a copy of the transfer documents, and the department certifies ownership of the credit with each transfer.

SECTION 1114. 71.47 (6) (i) of the statutes is created to read:

71.47 (6) (i) If a person who claims a credit under this subsection and a credit under section 47 of the Internal Revenue Code for the same qualified rehabilitation expenditures is required to repay any amount of the credit claimed under section 47 of the Internal Revenue Code, the person shall repay to the department a proportionate amount of the credit claimed under this subsection.

SECTION 1114j. 71.49 (1) (c) of the statutes is amended to read:

71.49 (1) (c) Research credit under s. 71.47 (4). except as provided under par. (f).

SECTION 1114k. 71.49 (1) (f) of the statutes is amended to read:

71.49 (1) (f) The total of farmland preservation credit under subch. IX, farmland tax relief credit under s. 71.47 (2m), dairy manufacturing facility investment credit under s. 71.47 (3p), jobs credit under s. 71.47 (3q), meat processing facility investment credit under s. 71.47 (3r), woody biomass harvesting and processing credit under s. 71.47 (3rm), food processing plant and food warehouse investment credit under s. 71.47 (3m), enterprise zone jobs credit under s. 71.47 (3w), business development credit under s. 71.47 (3y), research credit under s. 71.47 (4) (k) 1., film production services credit under s. 71.47 (5f), film production company investment credit under s. 71.47 (5h), beginning farmer and farm asset owner tax credit under s. 71.47 (8r), and estimated tax payments under s. 71.48.

SECTION 1115. 71.52 (1d) of the statutes is created to read:

71.52 (**1d**) "Disabled" means an individual who is unable to engage in any substantial gainful employment by reason of a medically determinable physical or mental impairment which has lasted or is reasonably expected to last for a continuous period of not less than 12 months.

SECTION 1116. 71.52 (1e) of the statutes is created to read:

71.52 (**1e**) "Disqualified loss" means the sum of the following amounts, exclusive of net gains from the sale or exchange of capital or business assets and exclusive of net profits:

- (a) Net loss from sole proprietorships.
- (b) Net capital loss.
- (c) Net loss from sales of business property, excluding loss from involuntary conversions.
- (d) Net loss from rental real estate, royalties, partnerships, tax-option S corporations, trusts, estates, and real estate mortgage investment conduits.
 - (e) Net farm loss.

SECTION 1117. 71.52 (1m) of the statutes is created to read:

71.52 (1m) "Farmer," "farming," and "farm premises" have the meanings given in s. 102.04 (3).

SECTION 1118. 71.52 (6) of the statutes is amended to read:

71.52 (6) "Income" means the sum of Wisconsin adjusted gross income and the following amounts, to the extent not included in Wisconsin adjusted gross income: maintenance payments (except foster care maintenance and supplementary payments excludable under section 131 of the internal revenue code), support money, cash public assistance (not including credit granted under this subchapter and amounts under s. 46.27), cash benefits paid by counties under s. 59.53 (21), the gross amount of any pension or annuity (including railroad retirement benefits, all payments received under the federal social security act and veterans disability pensions), nontaxable interest received from the federal government or any of its instrumentalities, nontaxable interest received on state or municipal bonds, worker's compensation, unemployment insurance, the gross amount of "loss of time" insurance, compensation and other cash benefits received from the United States for past or present service in the armed forces, scholarship and fellowship gifts or income, capital gains, gain on the sale of a personal residence excluded under section 121 of the internal revenue code, dividends, income of a nonresident or part-year resident who is married to a full-year resident, housing allowances provided to members of the clergy, the amount by which a resident manager's rent is reduced, nontaxable income of an American Indian, nontaxable income from sources outside this state and nontaxable deferred compensation. Intangible drilling costs, depletion allowances and depreciation, including first-year depreciation allowances under section 179 of the internal revenue code, amortization, contributions to individual retirement accounts under section 219 of the internal revenue code, contributions to Keogh plans, net operating loss carry-backs and carry-forwards and, capital loss carry-forwards, and disqualified losses deducted in determining Wisconsin adjusted gross income shall be added to "income". "Income" does not include gifts from natural persons, cash reimbursement payments made under title XX of the federal social security act, surplus food or other relief in kind supplied by a governmental agency, the gain on the sale of a personal residence deferred under section 1034 of the internal revenue code or nonrecognized gain from involuntary conversions under section 1033 of the internal revenue code. Amounts not included in adjusted gross income but added to "income" under this subsection in a previous year and repaid may be subtracted from income for the year during which they are repaid. Scholarship and fellowship gifts or income that are included in Wisconsin adjusted gross income and that were added to household income for purposes of determining the credit under this subchapter in a previous year may be subtracted from income for the current year in determining the credit under this subchapter. A marital property agreement or unilateral statement under ch. 766 has no effect in computing "income" for a person whose homestead is not the same as the homestead of that person's spouse.

SECTION 1120. 71.54 (1) (g) 4. of the statutes is created to read:

71.54 (1) (g) 4. Except as provided in subds. 5. and 7., for claims filed in 2018 and thereafter and based on property taxes accrued or rent constituting property taxes accrued during the previous year, no credit may be allowed under this paragraph unless the claimant or the claimant's spouse is over the age of 61 at the close of the year to which the claim relates.

SECTION 1121. 71.54 (1) (g) 5., 6. and 7. of the statutes are created to read:

- 71.54(1) (g) 5. For claims filed in 2018 and thereafter and based on property taxes accrued or rent constituting property taxes accrued during the previous year, no credit may be allowed under this paragraph unless the claimant is disabled.
- 6. With regard to a claimant who is disabled, the claimant shall provide with his or her return proof that his or her disability is in effect for the taxable year to which the claim relates. Proof of disability may be demonstrated by any of the following:
- a. A statement from the Veteran's Administration certifying that the claimant is receiving a disability benefit due to 100 percent disability.
- b. A document, or copy of a document, from the Social Security Administration stating the date the disability began.
- c. A statement from a physician, as defined in s. 448.01 (5), stating the beginning date of the disability and whether the disability is permanent or temporary.
- 7. For claims filed in 2018 and thereafter and based on property taxes accrued or rent constituting property taxes accrued during the previous year, with regard to a claimant who is not disabled or who is under the age of 62 at the close of the year to which the claim relates, no credit may be allowed under this paragraph if the claimant had no earned income in the taxable year to which the claim relates.

SECTION 1125. 71.55 (10) of the statutes is created to read:

71.55 (10) FARMERS. Notwithstanding the provision in s. 71.52 (6) that requires the addition of certain disqualified losses to income, such an addition may not be made by a claimant who is a farmer whose primary income is from farming and whose farming generates less than \$250,000 in gross receipts from the operation of farm premises in the year to which the claim relates.

SECTION 1125s. 71.613 (2) (intro.) of the statutes is amended to read:

71.613 (2) FILING CLAIMS. (intro.) Subject to the limitations and conditions provided in sub. (3), a claimant may claim as a credit against the tax imposed under s. 71.02, 71.08, 71.23, or 71.43, an amount calculated by multiplying the claimant's qualifying acres by one of the

following amounts, and if the allowable amount of the claim exceeds the income taxes otherwise due on the claimant's income or if there are no Wisconsin income taxes due on the claimant's income, the amount of the claim not used as an offset against income taxes shall be certified by the department of revenue to the department of administration for payment to the claimant by check, share draft, or other draft from the appropriation under s. 20.835 (2) (do):

SECTION 1127. 71.65 (2) (b) of the statutes is amended to read:

71.65 (2) (b) Every resident of this state and every nonresident carrying on activities within this state, whether taxable or not under this chapter, who pays in any calendar year for services performed within this state by an individual remuneration which that is excluded from the definition of wages, in the amount of \$600 or more, shall, on or before February 28 January 31 of the year following the year in which the payments are made, furnish a statement, in such form as required by the department, disclosing the name of the payor, the name and address of the recipient of the payment, and the total amount paid in such the calendar year to such the recipient. The person who pays for the services shall, on or before January 31 of the year in which the statement is required to be furnished to the department that deadline, furnish the recipient of the payment with a copy of that the statement. In any case in which an individual receives wages and also remuneration for services which remuneration is excluded from such definition, both from the same payor, the wages and the excluded remuneration shall both be reported in the report required under this subsection in a manner satisfactory to the department, regardless of the amount of the excluded remuneration.

SECTION 1128. 71.65 (5) (a) (intro.), 1. and 2. of the statutes are consolidated, renumbered 71.65 (5) (a) and amended to read:

71.65 (5) (a) If an employer applies for an extension and shows good cause why an extension should be granted, the department may grant the following extensions for the following statements: 1. Thirty days a 30-day extension for filing a wage statement under sub. (1) or, an annual reconciliation report under sub. (3) (a) or (d). 2. Sixty days for filing, or a statement of nonwage payments under sub. (2) (b).

SECTION 1131. 71.70(1) of the statutes is amended to read:

71.70 (1) Persons other than corporations. Persons other than corporations deducting rent or royalties in determining taxable income shall file a report that shows the amounts and the name and address of all natural persons each individual who are residents is a resident of this state and to whom royalties of \$600 or more were are paid during the taxable year; and the amounts and the name and address of all natural persons each individual to whom rent of \$600 or more is paid during the taxable

year for property having a situs in this state. Such information shall be filed The person who deducts rent or royalties shall file the report on or before February 28 January 31 of the year following the year in which the payments were are made. The person who deducts rent or royalties shall, on or before January 31 of the year in which the report is required to be furnished that deadline, furnish the recipient of the payment with a copy of that the report.

SECTION 1132. 71.70(2) of the statutes is amended to read:

71.70 (2) CORPORATIONS. All corporations doing business in this state shall file, on or before March 15 January 31, any information relative to payments made within the preceding calendar year of rents and royalties to all natural persons individuals taxable thereon under this chapter. A The corporation that makes the payment shall, on or before January 31 of the year in which the statement is required to be furnished to the department that deadline, furnish the recipient of the payment with a copy of that the statement.

SECTION 1133. 71.71 (title) of the statutes is amended to read:

71.71 (title) Wages subject to withholding.

SECTION 1134. 71.715 of the statutes is created to read:

71.715 Wages not subject to withholding. (1) STATEMENT EMPLOYER MUST FURNISH TO EMPLOYEE. (a) Every employer, as defined in s. 71.63 (3), that pays in any calendar year wages, as defined in s. 71.63 (6), to an employee, as defined in s. 71.63 (2), from which the employer was not required to deduct and withhold from the employee under the general withholding provisions of subch. X., shall furnish to the employee, with respect to the wages paid by the employer to the employee during a calendar year, on or before January 31 of the year following the year in which the wages are paid, or, if the employee's employment is terminated before the close of a calendar year, on the day on which the last payment of wages is made, 2 legible copies of a written statement showing all of the following:

- 1. The name of the employer and the employer's Wisconsin income tax identification number, if any.
- 2. The name of the employee and the employee's social security number, if any.
- 3. The total amount of wages the employer paid in the calendar year to the employee.
- (b) An employee that receives a statement under par. (a) shall furnish the department one copy of the statement along with the employee's return for the year.
- (2) STATEMENT EMPLOYER MUST FILE. Every employer required to furnish a statement under sub. (1) (a) shall file, with respect to the wages paid by the employer to an employee as described in sub. (1) during the calendar year, on or before January 31 of the year fol-

lowing the year in which the wages are paid, one copy of the statement.

SECTION 1135. 71.72 of the statutes is amended to read:

71.72 Statement of nonwage payments. Every resident of this state and every nonresident carrying on activities within this state, whether taxable or not under this chapter, who pays in any calendar year for services performed within this state by an individual remuneration which that is excluded from the definition of wages in s. 71.63 (6), in the amount of \$600 or more, shall, on or before February 28 January 31 of the year following the year in which the payments were are made, file a statement disclosing the name of the payor, the name and address of the recipient of the payment, and the total amount paid in such the calendar year to such the recipient. The person who pays for the services shall, on or before January 31 of the year in which the statement is required to be furnished to the department that deadline, furnish the recipient of the payment with a copy of that the statement. In any case in which an individual receives wages, as defined in s. 71.63 (6), and also remuneration for services which remuneration is excluded from such definition, both from the same payor, the wages and the excluded remuneration shall both be reported in the statement required by s. 71.71 (2) in a manner satisfactory to the department, regardless of the amount of the excluded remuneration.

SECTION 1136. 71.73 (2) (intro.), (a), (b) and (c) of the statutes are consolidated, renumbered 71.73 (2) and amended to read:

71.73 (2) EXTENSIONS. If an employer a person applies for an extension and shows good cause why an extension should be granted, the department of revenue may grant the following extensions for the following statements: (a) Sixty days a 30-day extension for filing a rent and royalty statement under s. 71.70. (b) Thirty days for filing, a wage statement under s. 71.71. (c) Sixty days for filing, a wage statement under s. 71.715, or a statement of nonwage payments under s. 71.72.

SECTION 1139. 71.75 (7m) of the statutes is created to read:

71.75 (7m) The department shall not issue a refund to an employed individual before March 1 unless both the individual and the individual's employer have filed all required returns and forms with the department for the taxable year for which the individual claims a refund.

SECTION 1140. 71.77 (2n) of the statutes is created to read:

71.77 (**2n**) Notwithstanding sub. (2), the department may make an assessment within one year of receiving notice of revocation from the Wisconsin Economic Development Corporation to recover all or a part of any tax credit claimed by a taxpayer, but revoked by the corporation.

SECTION 1151. 71.80 (20) of the statutes is amended to read:

71.80 (20) ELECTRONIC FILING. If a person is required to file $50 \underline{10}$ or more wage statements or $50 \underline{10}$ or more of any one type of information return with the department, the person shall file the statements or the returns electronically, by means prescribed by the department.

SECTION 1152. 71.80 (25) of the statutes is created to read:

71.80 (25) NET OPERATING AND BUSINESS LOSS CARRY-FORWARD AND CARRY-BACK. (a) No offset of Wisconsin income may be made under s. 71.05 (8) (b) 1., 71.26 (4) (a), or 71.45 (4) (a) unless the incurred loss was computed on a return that was filed within 4 years of the unextended due date for filing the original return for the taxable year in which the loss was incurred.

(b) No carry–back of a loss may be allowed under s. 71.05 (8) (b) 1. unless claimed within 4 years of the unextended due date for filing the original return for the taxable year to which the loss is carried back.

SECTION 1153. 71.83 (1) (cf) of the statutes is created to read:

71.83 (1) (cf) *Inconsistent estate basis reporting*. If any portion of an underpayment of tax required to be shown on a Wisconsin return is the result of an inconsistent estate basis reporting, there shall be added to the tax an amount equal to 20 percent of that portion of the underpayment. For purposes of this paragraph, an inconsistent estate basis reporting occurs if the property basis claimed on a Wisconsin return exceeds the property basis determined under section 1014 (f) of the Internal Revenue Code. The department shall assess, levy, and collect the penalty under this paragraph in the same manner as it assesses, levies, and collects taxes under this chapter.

SECTION 1160d. 71.98 (8) of the statutes is created to read:

71.98 (8) CHARITABLE DISTRIBUTIONS FROM AN INDIVIDUAL RETIREMENT ACCOUNT. For taxable years beginning after December 31, 2017, section 408 (d) (8) of the Internal Revenue Code, relating to a tax—free qualified charitable distribution from an individual retirement account directly to a charitable organization.

SECTION 1162. 73.03 (69) (b) 1. of the statutes is amended to read:

73.03 (69) (b) 1. The business has at least 2 full—time employees and the amount of payroll compensation paid by the business in this state is equal to at least 50 percent of the amount of all payroll compensation paid by the business. An employee of a professional employer organization, as defined in s. 202.21 (5), or a professional employer group, as defined in s. 202.21 (4), who is performing services for a client is considered an employee solely of the client for purposes of this subdivision.

SECTION 1162m. 73.03 (71) (b) of the statutes is amended to read:

73.03 (71) (b) After the department makes the determination under par. (a), the department shall determine how much the individual income tax rates under s. 71.06 may be reduced in the following taxable year in order to eliminate the alternative minimum tax under s. 71.08 and decrease individual income tax revenue by the amount determined under par. (a). For purposes of this paragraph, the department shall calculate the tax rate reductions in proportion to the share of gross tax attributable to each of the tax brackets under s. 71.06 in effect during the most recently completed taxable year.

SECTION 1162n. 73.03 (71) (c) of the statutes is amended to read:

73.03 (71) (c) The department shall certify the determinations made under pars. (a) and (b) to the secretary of the department of administration, to the governor, and to the legislature and specify with that certification that the elimination of the alternative minimum tax and the new tax rates take effect in the taxable year following the taxable year in which the department makes the certification under this paragraph.

SECTION 1165d. 73.06 (3) of the statutes is amended to read:

73.06 (3) The department of revenue, through its supervisors of equalization, shall examine and test the work of assessors during the progress of their assessments and ascertain whether any of them is assessing property at other than full value or is omitting property subject to taxation from the roll. The department and such supervisors shall have the rights and powers of a local assessor for the examination of persons and property and for the discovery of property subject to taxation. If any property has been omitted or not assessed according to law, they shall bring the same to the attention of the local assessor of the proper district and if such local assessor shall neglect or refuse to correct the assessment they shall report the fact to the board of review. If it discovers errors in identifying or valuing property that is exempt under s. 70.11 (39) or (39m), the department shall change the specification of the property as taxable or exempt and shall change the value of the property. All disputes between the department, municipalities, and property owners about the taxability or value of property that is reported under s. 79.095 (2) (a) or of the property under s. 70.995 (12r) shall be resolved by using the procedures under s. 70.995 (8).

SECTION 1169. 77.14 of the statutes is amended to read:

77.14 Forest croplands information, protection, appropriation. The department of natural resources shall publish and distribute information regarding the method of taxation of forest croplands under this subchapter, and may employ a fire warden in charge of fire prevention in forest croplands. All actual and necessary expenses incurred by the department of natural resources

or by the department of revenue in the performance of their duties under this subchapter shall be paid from the appropriation made in s. 20.370 (1) (2) (mv) upon certification by the department incurring such expenses.

SECTION 1170. 77.22 (2) (c) of the statutes is repealed.

SECTION 1171. 77.22 (2) (d) of the statutes is repealed.

SECTION 1171r. 77.51 (1bm) of the statutes is created to read:

77.51 (**1bm**) "Beekeeping" means the business of moving, raising, producing, and other management of bees or bee products, regardless of the number of hives of bees managed.

SECTION 1172. 77.51 (2) of the statutes is amended to read:

77.51 (2) "Contractors" and "subcontractors" are the consumers of tangible personal property or items or goods under s. 77.52 (1) (b) or (d) used by them in real property construction activities, and the sales and use tax applies to the sale of tangible personal property or items or goods under s. 77.52 (1) (b) or (d) to them. A contractor engaged primarily in real property construction activities may use resale certificates only with respect to purchases of tangible personal property or items or goods under s. 77.52 (1) (b) or (d) which that the contractor has sound reason to believe the contractor will sell to customers for whom the contractor will not perform real property construction activities involving the use of such tangible personal property or items or goods under s. 77.52 (1) (b) or (d). In this subsection, "real property construction activities" means activities that occur at a site where tangible personal property or items or goods under s. 77.52 (1) (b) or (d) that are applied or adapted to the use or purpose to which real property is devoted are affixed to that real property, if the intent of the person who affixes that property is to make a permanent accession to the real property. In this subsection, "real property construction activities" does not include affixing property subject to tax under s. 77.52 (1) (c) to real property or affixing to real property tangible personal property that remains tangible personal property after it is affixed.

SECTION 1173. 77.51 (9) (a) of the statutes is renumbered 77.51 (9) (a) 1.

SECTION 1174. 77.51 (9) (a) 2. of the statutes is created to read:

77.51 (9) (a) 2. For purposes of subd. 1., it is presumed that a seller is not pursuing a vocation, occupation, or business or a partial vocation or occupation or part—time business as a vendor of tangible personal property, or items, property, or goods under s. 77.52 (1) (b), (c), or (d), or taxable services if the seller's total taxable sales price from sales of tangible personal property, items, property, and goods under s. 77.52 (1) (b), (c), and (d), and taxable services is less than \$2,000 during a calendar year.

SECTION 1176. 77.51 (12t) of the statutes is created to read:

77.51 (12t) "Real property construction activities" means activities that occur at a site where tangible personal property or items or goods under s. 77.52 (1) (b) or (d) that are applied or adapted to the use or purpose to which real property is devoted are affixed to that real property, if the intent of the person who affixes that property is to make a permanent accession to the real property. "Real property construction activities" does not include affixing property subject to tax under s. 77.52 (1) (c) to real property or affixing to real property tangible personal property that remains tangible personal property affer it is affixed.

SECTION 1177. 77.51 (13) (am) of the statutes is amended to read:

77.51 (13) (am) Any person making any retail sale of a motor vehicle, aircraft, snowmobile, recreational vehicle, as defined in s. 340.01 (48r), trailer, semitrailer, all-terrain vehicle, utility terrain vehicle, off-highway motorcycle, as defined in s. 23.335 (1) (q), or boat registered or titled, or required to be registered or titled, under the laws of this state or of the United States.

SECTION 1177d. 77.51 (13) (o) of the statutes is amended to read:

77.51 (13) (o) A person selling drugs for animals <u>or bees</u> to a veterinarian. As used in this paragraph, "animal" includes livestock, pets, and poultry.

SECTION 1177t. 77.52 (2) (a) 5. a. of the statutes is repealed.

SECTION 1178. 77.52 (2) (a) 10. of the statutes is amended to read:

77.52 (2) (a) 10. Except for the repair, service, alteration, fitting, cleaning, painting, coating, towing, inspection, and maintenance of any aircraft or aircraft parts; except for services provided by veterinarians; and except for installing or applying tangible personal property, or items or goods under sub. (1) (b) or (d), that, subject to par. (ag), when installed or applied, will constitute an addition or capital improvement of real property; the repair, service, alteration, fitting, cleaning, painting, coating, towing, inspection, and maintenance of all items of tangible personal property or items, property, or goods under sub. (1) (b), (c), or (d), unless, at the time of that repair, service, alteration, fitting, cleaning, painting, coating, towing, inspection, or maintenance, a sale in this state of the type of property, item, or good repaired, serviced, altered, fitted, cleaned, painted, coated, towed, inspected, or maintained would have been exempt to the customer from sales taxation under this subchapter, other than the exempt sale of a motor vehicle or truck body to a nonresident under s. 77.54 (5) (a) and other than nontaxable sales under s. 77.522 or unless the repair, service, alteration, fitting, cleaning, painting, coating, towing, inspection, or maintenance is provided under a contract that is subject to tax under subd. 13m. The tax imposed under this subsection applies to the repair, service, alteration, fitting, cleaning, painting, coating, towing, inspection, or maintenance of items listed in par. (ag), regardless of whether the installation or application of tangible personal property or items, property, or goods under sub. (1) (b), (c), or (d) related to the items is an addition to or a capital improvement of real property, except that the tax imposed under this subsection does not apply to the original installation or the complete replacement of an item listed in par. (ag), if that installation or replacement is a real property construction activity under s. 77.51 (2).

SECTION 1179. 77.52 (2m) (b) of the statutes is amended to read:

77.52 (**2m**) (b) With respect to the services subject to tax under sub. (2) (a) 7., 10., 11., and 20. and except as provided in s. 77.54 (60) (b) and (bm) 2., all tangible personal property or items, property, or goods under s. 77.52 (1) (b), (c), or (d) physically transferred, or transferred electronically, to the customer in conjunction with the selling, performing, or furnishing of the service is a sale of tangible personal property or items, property, or goods under s. 77.52 (1) (b), (c), or (d) separate from the selling, performing, or furnishing of the service.

SECTION 1181. 77.52 (13) of the statutes is amended to read:

77.52 (13) For the purpose of the proper administration of this section and to prevent evasion of the sales tax it shall be presumed that all receipts are subject to the tax until the contrary is established. The burden of proving that a sale of tangible personal property, or items, property, or goods under sub. (1) (b), (c), or (d), or services is not a taxable sale at retail is upon the person who makes the sale unless that person takes from the purchaser an electronic or a paper certificate, in a manner prescribed by the department, to the effect that the property, item, good, or service is purchased for resale or is otherwise exempt, except that no certificate is required for the sale of tangible personal property, or items, property, or goods under sub. (1) (b), (c), or (d), or services that are exempt under s. 77.54 (5) (a) 3., (7), (7m), (8), (10), (11), (14), (15), (17), (20n), (21), (22b), (31), (32), (35), (36), (37), (42), (44), (45), (46), (51), and (52), and (66).

SECTION 1182n. 77.522 (4) (h) of the statutes is repealed.

SECTION 1184. 77.53 (17) of the statutes is amended to read:

77.53 (17) This section does not apply to tangible personal property or items, property, or goods under s. 77.52 (1) (b), (c), or (d) purchased outside this state, as determined under s. 77.522, other than motor vehicles, boats, snowmobiles, recreational vehicles, as defined in s. 340.01 (48r), trailers, semitrailers, all-terrain vehicles, utility terrain vehicles, off-highway motorcycles, as defined in s. 23.335 (1) (q), and airplanes registered or titled or required to be registered or titled in this state, which is brought into this state by a nondomiciliary for

the person's own storage, use or other consumption while temporarily within this state when such property, item, or good is not stored, used or otherwise consumed in this state in the conduct of a trade, occupation, business or profession or in the performance of personal services for wages or fees.

SECTION 1185. 77.53 (18) of the statutes is amended to read:

77.53 (18) This section does not apply to the storage, use or other consumption in this state of household goods or items, property, or goods under s. 77.52 (1) (b), (c), or (d) for personal use or to aircraft, motor vehicles, boats, snowmobiles, mobile homes, manufactured homes, as defined in s. 101.91 (2), recreational vehicles, as defined in s. 340.01 (48r), trailers, semitrailers, all-terrain vehicles, and utility terrain vehicles, and off-highway motorcycles, as defined in s. 23.335 (1) (q), for personal use, purchased by a nondomiciliary of this state outside this state, as determined under s. 77.522, 90 days or more before bringing the goods, items, or property into this state in connection with a change of domicile to this state.

SECTION 1185p. 77.54 (3) (a) of the statutes is amended to read:

77.54(3) (a) The sales price from the sales sale of and the storage, use, or other consumption of tractors and machines, including accessories, attachments, and parts, lubricants, nonpowered equipment, and other tangible personal property, or items or property under s. 77.52 (1) (b) or (c), that are used exclusively and directly, or are consumed or lose their identities, in the business of farming, including dairy farming, agriculture, horticulture, floriculture, silviculture, beekeeping, and custom farming services, but excluding automobiles, trucks, and other motor vehicles for highway use; excluding personal property that is attached to, fastened to, connected to, or built into real property or that becomes an addition to, component of, or capital improvement of real property; and excluding tangible personal property, or items or property under s. 77.52 (1) (b) or (c), used or consumed in the erection of buildings or in the alteration, repair, or improvement of real property, regardless of any contribution that that personal property, or item or property under s. 77.52 (1) (b) or (c), makes to the production process in that building or real property and regardless of the extent to which that personal property, or item or property under s. 77.52 (1) (b) or (c), functions as a machine, except as provided in par. (c).

SECTION 1185t. 77.54 (3m) (intro.) of the statutes is amended to read:

77.54 (3m) (intro.) The sales price from the sale of and the storage, use, or other consumption of the following items if they are used exclusively by the purchaser or user in the business of farming; including dairy farming, agriculture, horticulture, floriculture, silviculture, beekeeping, and custom farming services:

SECTION 1185v. 77.54 (3m) (hm) of the statutes is created to read:

77.54 (3m) (hm) Bees, beehives, and bee combs.

SECTION 1185w. 77.54 (3m) (L) of the statutes is amended to read:

77.54 (**3m**) (L) Containers for fruits, vegetables, <u>bee products</u>, grain, hay, silage, and animal wastes.

SECTION 1186. 77.54 (7) (b) (intro.) of the statutes is amended to read:

77.54 (7) (b) (intro.) If the item transferred is a motor vehicle, snowmobile, recreational vehicle, as defined in s. 340.01 (48r), trailer, semitrailer, all-terrain vehicle, utility terrain vehicle, off-highway motorcycle, as defined in s. 23.335 (1) (q), or aircraft and the item is registered or titled, or required to be registered or titled, in this state or if the item is a boat that is registered or titled, or required to be registered or titled, or required to be registered or titled, in this state or under the laws of the United States, the exemption under par. (a) applies only if all of the following conditions are fulfilled:

SECTION 1186d. 77.54 (9m) of the statutes is amended to read:

77.54 (9m) The sales price from the sale of and the storage, use, or other consumption of tangible personal property, or items or property under s. 77.52 (1) (b) or (c), sold to a construction contractor who, in fulfillment of a real property construction activity, transfers the tangible personal property, or items or property under s. 77.52 (1) (b) or (c), to an entity described under sub. (9a) (b), (c), (d), (em), or (f), a technical college district, the Board of Regents of the University of Wisconsin System, an institution, as defined in s. 36.05 (9), a college campus, as defined in 36.05 (6m), or the University of Wisconsin-Extension, if such tangible personal property, or items or property, becomes a component of a facility in this state that is owned by the entity. In this subsection, "facility" means any building, shelter, parking lot, parking garage, athletic field, athletic park, storm sewer, water supply system, or sewerage and waste water treatment facility, but does not include a highway, street, or road.

SECTION 1187. 77.54 (20n) (d) of the statutes is created to read:

77.54 (20n) (d) The sales price from the sale of and the storage, use, or other consumption of prepared food that is sold by a retailer and that meets all of the following conditions:

- 1. The prepared food is not candy, soft drinks, or dietary supplements.
- 2. The retailer manufactures the prepared food in a building assessed as manufacturing property under s. 70.995, or that would be assessed as manufacturing property under s. 70.995 if the building was located in this state.
- 3. The retailer makes no retail sales of prepared food at the building described in subd. 2.
 - 4. Any of the following applies:

- a. The retailer freezes the prepared food prior to its sale and sells the prepared food at retail in a frozen state, and the prepared food is not sold with eating utensils that are provided by the retailer, as described in s. 77.51 (10m) (a) 3.
- b. The prepared food consists of more than 50 percent yogurt.

SECTION 1187d. 77.54 (23n) of the statutes is renumbered 77.54 (23n) (b) (intro.) and amended to read:

77.54 (23n) (b) (intro.) The sales price from the sales of tangible personal property and property under s. 77.52 (1) (c) to, and the storage, use, or other consumption of tangible personal property and property under s. 77.52 (1) (c) by, a person who is licensed to operate a commercial radio or television station in this state, if the tangible personal property or property under s. 77.52 (1) (c) is used any of the following:

1. Used exclusively and directly in, or is fuel or electricity consumed in, the origination or integration of various sources of program material for commercial radio or television transmissions that are generally available to the public free of charge without a subscription or service agreement. This subsection applies to vehicles licensed for highway use and equipment used to transmit or receive signals from a satellite.

SECTION 1187e. 77.54 (23n) (a) of the statutes is created to read:

77.54 (23n) (a) In this subsection, "program material" means material transmitted by a commercial radio or television station that is generally available to the public free of charge without a subscription or service agreement. "Program material" includes material used in origination.

SECTION 1187f. 77.54 (23n) (b) 2. to 5. of the statutes are created to read:

77.54 (23n) (b) 2. Used primarily for transmitting or receiving commercial radio or television program material, including a broadcast transmitter, a satellite dish, and a communications tower and the material used to construct the tower.

- 3. Leased space on a communications tower if the space is used exclusively for transmitting or receiving commercial radio or television program material.
- 4. A motor vehicle licensed for highway use and used exclusively in the origination of commercial radio or television program material.
- 5. A part, an accessory, or a supply, including fuel or electricity, that is used for any of the property that is exempt under in subds. 1. to 4.

SECTION 1187j. 77.54 (30) (a) 3. of the statutes is amended to read:

77.54 (**30**) (a) 3. Electricity sold for use in farming, including but not limited to agriculture, dairy farming, floriculture, silviculture, and horticulture, and beekeeping.

Vetoed In Part **SECTION 1187k.** 77.54 (30) (a) 5. of the statutes is amended to read:

77.54 (**30**) (a) 5. Fuel sold for use in farming, including but not limited to agriculture, dairy farming, floriculture, silviculture, and horticulture, and beekeeping.

SECTION 1187n. 77.54 (33) of the statutes is amended to read:

77.54 (33) The sales price from sales the sale of and the storage, use, or other consumption of drugs used on farm livestock, not including workstock, or on bees.

SECTION 1187p. 77.54 (37) of the statutes is amended to read:

77.54 (37) The sales price from revenues collected under s. 256.35 (3) and the surcharge established by rule by the public service commission under s. 256.35 (3m) (f) for customers of wireless providers, as defined in s. 256.35 (3m) (a) 6.

SECTION 1188. 77.54 (60) (a) of the statutes is renumbered 77.54 (60) (d) (intro.) and amended to read:

77.54 **(60)** (d) (intro.) In this subsection, "lump sum:

1. "Construction contract" means a contract to perform real property construction activities and to provide tangible personal property, items or property under s. 77.52 (1) (b) or (c), or taxable services and for which the contractor quotes the charge for labor, services of subcontractors, tangible personal property, items and property under s. 77.52 (1) (b) and (c), and taxable services as one price, including a contract for which the contractor itemizes the charges for labor, services of subcontractors, tangible personal property, items and property under s. 77.52 (1) (b) and (c), and taxable services as part of a schedule of values or similar document products.

SECTION 1189. 77.54 (60) (b) of the statutes is amended to read:

77.54 (60) (b) The sales price from the sale of and the storage, use, or other consumption of tangible personal property, items and property under s. 77.52 (1) (b) and (c), and taxable services products that are sold by a prime contractor as part of a lump sum construction contract, if the total sales price of all such taxable products is less than 10 percent of the total amount of the lump sum construction contract. Except as provided in par. (c), the prime contractor is the consumer of such taxable the products and shall pay the tax imposed under this subchapter on the taxable products.

SECTION 1190. 77.54 (60) (bm) of the statutes is created to read:

77.54 (60) (bm) 1. The sales price from the sale of and the storage, use, or other consumption of products that are sold by a subcontractor to a prime contractor, or to another subcontractor for eventual sale to the prime contractor, as part of a construction contract, if any of the following applies:

a. The total sales price of all products is less than 10 percent of the total amount of the construction contract.

- b. The products will be sold by the prime contractor as part of a construction contract, and that sale is exempt under par. (b).
- 2. Except as provided in par. (c), the subcontractor is the consumer of the products exempted under this paragraph and shall pay the tax imposed under this subchapter on the products.

SECTION 1191. 77.54 (60) (c) of the statutes is renumbered 77.54 (60) (c) (intro.) and amended to read:

77.54 (**60**) (c) (intro.) If the <u>lump sum construction</u> contract <u>under par.</u> (b) is <u>entered into with between a prime contractor and</u> an entity that is exempt from taxation under sub. (9a), the <u>all of the following apply:</u>

1. The prime contractor is the consumer of all taxable products used by the <u>prime</u> contractor in real property construction activities, but the <u>prime</u> contractor may purchase without tax, for resale, tangible personal property, items and property under s. 77.52 (1) (b) and (c), and taxable services <u>products</u> that are sold by the <u>prime</u> contractor to the entity as part of the <u>lump sum construction</u> contract with the entity and that are not consumed by the <u>prime</u> contractor in real property construction activities.

SECTION 1192. 77.54 (60) (c) 2. of the statutes is created to read:

77.54 (60) (c) 2. A subcontractor of the prime contractor is the consumer of all products used by the subcontractor in real property construction activities, but the subcontractor may purchase without tax, for resale, products that are sold by the subcontractor to the prime contractor or another subcontractor, as part of the subcontractor's construction contract under par. (bm), for resale to the entity and that are not consumed by the subcontractor in real property construction activities.

SECTION 1193. 77.54 (60) (d) 2. and 3. of the statutes are created to read:

77.54 (**60**) (d) 2. "Prime contractor" means a contractor who enters into a construction contract with an owner or lessee of real property, except for leased property under s. 77.52 (1) (c), to perform real property construction activities on the real property.

3. "Subcontractor" means a contractor who enters into a construction contract with a prime contractor or another subcontractor.

SECTION 1194b. 77.54 (65) of the statutes is created to read:

77.54 (65) (a) The sales price from the sale of and the storage, use, or other consumption of the following:

1. A video or electronic game sold in a tangible form to a person in the business of providing a taxable service through an amusement device if the video or electronic game is used exclusively for the amusement device. For purposes of this subdivision, a video or electronic game sold in a tangible form is a separate sale from the amusement device through which the video or electronic game is played if the sales price of the video or electronic game

is separately indicated from the sales price of the amusement device on the invoice, bill of sale, or similar document that the seller gives to the purchaser.

- 2. Tangible personal property sold to a person in the business of providing a taxable service through an amusement device if the tangible personal property is used exclusively as a prize awarded or transferred through the use of the amusement device.
- 3. Tournament or league entrance fees advertised and set aside as prize money.
- (b) For purposes of this subsection, "amusement device" means a single or multiplayer device, machine, or game played for amusement, the outcome of which depends at least in part on the skill, precision, dexterity, or knowledge of the person playing, but not predominantly on the element of chance. "Amusement device" includes a pinball machine, console machine, crane machine, claw machine, redemption game, stacker, arcade game, foosball or soccer table game, miniature racetrack or football machine, target or shooting gallery machine, basketball machine, shuffleboard table, kiddie ride game, Skee-Ball machine, air hockey machine, dart board, pool table, billiard table, or any other similar device, machine, or game. "Amusement device" does not include any device, machine, or game that is illegal to operate within this state.

SECTION 1194c. 77.54 (66) of the statutes is created to read:

77.54 (66) The sales price from the sale of and the storage, use, or other consumption of farm—raised fish, as defined in s. 95.001 (1) (ah), sold to a fish farm, as defined in s. 95.001 (1) (aj), that is registered with the department of agriculture, trade and consumer protection under s. 95.60 (3m), or to a person who holds a valid permit under s. 29.736 for the stocking of fish.

SECTION 1194d. 77.59 (2) of the statutes is amended to read:

77.59 (2) The Except as provided in sub. (2g), the department may, by field audit, determine the tax required to be paid to the state or the refund due to any person under this subchapter. The determination may be made upon the basis of the facts contained in the return being audited or upon any other information in the department's possession. The determination may be made on the basis of sampling, whether or not the person being audited has complete records of transactions and whether or not the person being audited consents. The department may examine and inspect the books, records, memoranda and property of any person in order to verify the tax liability of that person or of another person. The department may subpoena any person to give testimony under oath before it and to produce whatever books, records or memoranda are necessary in order to enable the department to verify the tax liability of that person or of another person. The determination shall be presumed to be correct and the burden of proving it to be incorrect shall be upon the person challenging its correctness. A determination by the department in a field audit becomes final at the expiration of the appeal periods provided in sub. (6), and the tax liability of the taxpayer for the period audited may not be subsequently adjusted except as provided in sub. (4) (b), (8) or (8m). If the taxpayer files or is required to file more than one return for the taxpayer's fiscal year or for a calendar year, the determination made by field audit for that fiscal or calendar year shall be based on the receipts, purchases, deductions and exemptions for the entire fiscal or calendar year.

SECTION 1194e. 77.59 (2g) of the statutes is created to read:

77.59 (2g) The department shall promulgate rules to establish criteria applicable to field audits conducted under this subchapter for which an auditor uses a statistical sampling method whereby the auditor randomly selects a sample of transactions and uses probability theory to evaluate the sample results. The department shall establish criteria under this subsection to provide that any person with less than \$10,000,000 in annual sales during any year at issue in a field audit may choose to have the audit conducted using statistical sampling as described in this subsection. In addition, the department shall establish criteria under this subsection that specifies the number of transactions necessary to qualify for statistical sampling and the maximum sample size.

SECTION 1198. 77.78 of the statutes is amended to read:

77.78 Registration. No motor vehicle, boat, snowmobile, recreational vehicle, as defined in s. 340.01 (48r), trailer, semitrailer, all-terrain vehicle, utility terrain vehicle, off-highway motorcycle, as defined in s. 23.335 (1) (q), or aircraft that is required to be registered by this state may be registered or titled by this state unless the registrant files a sales and use tax report and pays the county tax and special district tax at the time of registering or titling to the state agency that registers or titles the property. That state agency shall transmit those tax revenues to the department of revenue.

SECTION 1199. 77.82 (2m) (d) of the statutes is amended to read:

77.82 (2m) (d) All of the application recording fees collected under par. (a) shall be credited to the appropriation under s. $20.370 \frac{(1)}{(2)} (cr)$.

SECTION 1200. 77.82 (2m) (dm) 1. of the statutes is amended to read:

77.82 (**2m**) (dm) 1. Of each management plan fee, \$300 or the entire fee, whichever is less, that is collected under par. (ag) shall be credited to the appropriation under s. 20.370 + (1) + (2) +

SECTION 1201. 77.82 (2m) (dm) 2. of the statutes is amended to read:

77.82 (2m) (dm) 2. Any amount not credited to the appropriation under s. 20.370 (1) (2) (cx), as calculated

in subd. 1., shall be deposited into the conservation fund for forestry purposes.

SECTION 1202. 77.82 (4) of the statutes is amended to read:

77.82 (4) Additions to managed forest land. An owner of land that is designated as managed forest land may file an application with the department to designate as managed forest land an additional parcel of land if the additional parcel is at least 3 acres in size and is contiguous to any of that designated land. The application shall be accompanied by a nonrefundable \$20 application recording fee unless a different amount for the fee is established by the department by rule at an amount equal to the average expense to the department of recording an order issued under this subchapter. The fee shall be deposited in the conservation fund and credited to the appropriation under s. 20.370 (1) (2) (cr). The application shall be filed on a department form and shall contain any additional information required by the department. The tax rate applicable to an addition under this subsection shall be the tax rate currently applicable to the parcel to which the land is being added.

SECTION 1203. 77.88 (2) (ac) 1. of the statutes is amended to read:

77.88 (2) (ac) 1. If the land transferred under par. (a) meets the eligibility requirements under s. 77.82 (1) (a) and (b), the land shall continue to be designated as managed forest land if the transferee, within 30 days after a transfer of ownership, files a form provided by the department signed by the transferee. By signing the form, the transferee certifies to the department an intent to comply with the existing management plan for the land and any amendments to the plan. The transferee shall provide proof that each person holding any encumbrance on the land agrees to the designation. The transferee may designate an area of the transferred land closed to public access as provided under s. 77.83. The department shall issue an order continuing the designation of the land as managed forest land under the new ownership. The transferee shall pay a \$100 fee that will accompany the report. The fee shall be deposited in the conservation fund. Twenty dollars of the fee or a different amount of the fee as may be established under subd. 2. shall be credited to the appropriation under s. 20.370 (1) (2) (cr). The department shall immediately notify each person entitled to notice under s. 77.82 (8).

SECTION 1204. 77.88 (2) (ac) 2. of the statutes is amended to read:

77.88 (2) (ac) 2. The department may establish by rule a different amount of each fee under subd. 1. that will be credited to the appropriation under s. 20.370 (1) (2) (cr). The amount shall be equal to the average expense to the department of recording an order issued under this subchapter.

SECTION 1205. 77.89 (1) (b) of the statutes is amended to read:

77.89 (1) (b) The department shall distribute from the appropriation under s. 20.370 (1) (2) (mv) of the statutes \$1,000,000 in fiscal year 2015–16 and \$1,000,000 in fiscal year 2016–17 among treasurers of each municipality in which is located land subject to a managed forest land order that is designated as closed to public access under s. 77.83 (1). The department shall distribute to each municipal treasurer an amount in proportion to the number of acres of closed land in that municipality. The department shall make the payments for fiscal year 2015–16 before July 1, 2016. The department shall make the payments for fiscal year 2016–17 before July 1, 2017.

SECTION 1205m. 77.89 (1) (c) of the statutes is repealed.

SECTION 12050. 77.89 (2) (a) of the statutes is amended to read:

77.89 (2) (a) Each municipal treasurer shall pay 20 percent of each payment received under sub. (1) (a), and (b), and (c) and under ss. 77.84 (2) (a), (am), and (bp), 77.85, and 77.876 to the county treasurer and shall deposit the remainder in the municipal treasury. The payment to the county treasurer for money received before November 1 of any year shall be made on or before the November 15 after its receipt. For money received on or after November 1 of any year, the payment to the county treasurer shall be made on or before November 15 of the following year.

SECTION 1205p. 77.89 (2) (b) of the statutes is amended to read:

77.89 (2) (b) The municipal treasurer shall pay all 20 percent of the amounts received under s. 77.84 (2) (b) and (bm) to the county treasurer, as provided under ss. 74.25 and 74.30. The county treasurer shall, by June 30 of each year, pay all amounts received under this paragraph to the department. All amounts received by the department shall be credited to the conservation fund and shall be reserved for land acquisition, resource management activities, and grants under s. 77.895, and shall deposit the remainder in the municipal treasury.

SECTION 1206. 77.91 (4) of the statutes is amended to read:

77.91 (4) EXPENSES. Except as provided in sub. (5), the department's expenses for the administration of this subchapter shall be paid from the appropriation under s. 20.370 (1) (2) (mv).

SECTION 1207. 77.91 (5) of the statutes is amended to read:

77.91 (5) RECORDING. Each register of deeds who receives notice of an order under this subchapter shall record the action as provided under s. 59.43 (1c). The department shall pay the register of deeds the fee specified under s. 59.43 (2) (ag) 1. from the appropriation under s. 20.370 (1) (2) (cr). If the amount in the appropriation under s. 20.370 (1) (2) (cr) in any fiscal year is insufficient to pay the full amount required under this subsection in that fiscal year, the department shall pay the

balance from the appropriation under s. 20.370 (1) (2) (mv).

Vetoed In Part **SECTION 1208m.** 78.01 (2m) (f) of the statutes is amended to read:

78.01 (**2m**) (f) It is dyed diesel fuel and is sold for off-highway use other than use in a snowmobile, in a limited use off-highway motorcycle that is not registered for private use under s. 23.335 (3) (a), or in an all-terrain vehicle or utility terrain vehicle that is not registered for private use under s. 23.33 (2) (d) or (2g), or in a recreational motorboat or if no claim for a refund for the tax on the diesel fuel may be made under s. 78.75 (1m) (a) 3.

SECTION 1210. 79.035 (7) of the statutes is created to read:

79.035 (7) (a) The department of administration shall reduce the payment under this section to each county and municipality that receives a grant under s. 16.047 (4m) for replacement of public transit vehicles in an urban mass transit system by an amount determined as follows:

- 1. For an urban mass transit system that is eligible to receive state aid under s. 85.20 (4m) (a) 6. cm. or d. and serving a population exceeding 200,000, 75 percent of the total amount of grants received under s. 16.047 (4m).
- 2. For an urban mass transit system that is eligible to receive state aid under s. 85.20 (4m) (a) 7. and serving a population of at least 50,000, 20 percent of the total amount of grants received under s. 16.047 (4m).
- 3. For an urban mass transit system that is eligible to receive state aid under s. 85.20 (4m) (a) 8. and serving a population of less than 50,000, 10 percent of the total amount of grants received under s. 16.047 (4m).
- (b) Beginning with the first payment due under this section after the county or municipality receives a grant under s. 16.047 (4m), the department of administration shall apply the reduction determined under par. (a) for each county and municipality by reducing 10 consecutive annual payments under this section to the county or municipality by equal amounts. If in any year the reduction under this paragraph for a county or municipality exceeds the payment under this section for the county or municipality, the department of administration shall apply the excess amount of the reduction to the payment to the county or municipality under s. 79.04.

SECTION 1210b. 79.04 (1) (a) of the statutes is amended to read:

79.04 (1) (a) An amount from the public utility account determined by multiplying by 3 mills in the case of a town, and 6 mills in the case of a city or village, the first \$125,000,000 of the amount shown in the account, plus leased property, of each public utility except qualified wholesale electric companies, as defined in s. 76.28 (1) (gm), on December 31 of the preceding year for "production plant, exclusive of land," "general structures," and "substations," in the case of light, heat and power companies, electric cooperatives or municipal electric companies, for all property within a municipality in

accordance with the system of accounts established by the public service commission or rural electrification administration, less depreciation thereon as determined by the department of revenue and less the value of treatment plant and pollution abatement equipment, as defined under s. 70.11 (21), as determined by the department of revenue plus an amount from the public utility account determined by multiplying by 3 mills in the case of a town, and 6 mills in the case of a city or village, of the first \$125,000,000 of the total original cost of production plant, general structures, and substations less depreciation, land and approved waste treatment facilities of each qualified wholesale electric company, as defined in s. 76.28 (1) (gm), as reported to the department of revenue of all property within the municipality. The total of amounts, as depreciated, from the accounts of all public utilities for the same production plant is also limited to not more than \$125,000,000. The amount distributable to a municipality under this subsection and sub. (6) in any year shall not exceed \$300 times the population of the municipality, except that, beginning with payments in 2009, the amount distributable to a municipality under this subsection and sub. (6) in any year shall not exceed \$425 times the population of the municipality, except as provided under par. (am).

SECTION 1210c. 79.04 (1) (am) of the statutes is created to read:

79.04 (1) (am) The payment limitation under par. (a) does not apply to the amounts distributable to a municipality under this subsection and sub. (6) if the first distribution to the municipality that meets or exceeds the limitation occurs after 2010. This paragraph does not apply to distributions after 2022.

SECTION 1210d. 79.05 (7) of the statutes is created to read:

79.05 (7) (a) Beginning with the distributions in 2018 and ending with the distributions in 2022, the annual payment under this section to the village of Maine shall be the amount otherwise determined for the village under this section, plus \$583,000.

(b) Beginning with the distributions in 2018 and ending with the distributions in 2022, the annual payment under this section to the city of Janesville shall be the amount otherwise determined for the city under this section, plus \$583,000.

SECTION 1210e. 79.095 (1) (bh) of the statutes is created to read:

79.095 (1) (bh) "Inflation factor" means a percentage equal to the average annual percentage change in the U.S. consumer price index for all urban consumers, U.S. city average, as determined by the U.S. department of labor, for the 12 months ending on September 30 of the year before the payment under sub. (4) (b), except that the percentage under this paragraph shall not be less than zero.

SECTION 1210f. 79.095 (4) of the statutes is renumbered 79.095 (4) (a) and amended to read:

79.095 (4) (a) The Except as provided in par. (b), the department shall calculate the payments due each taxing jurisdiction under this section by multiplying the full value as of the January 1 of the preceding year of the property that is exempt under s. 70.11 (39) and (39m) and that is located in the jurisdiction by the full-value gross tax rate of the jurisdiction for the preceding year.

(c) The department shall certify the amount of the payment due each taxing jurisdiction to the department of administration, which shall make the payments on or before the 4th Monday in July. For purposes of ch. 121, school districts shall treat the payments made in July under this subsection as if they had been received in the previous school year.

SECTION 1210g. 79.095 (4) (b) of the statutes is created to read:

79.095 (4) (b) In 2018, each taxing jurisdiction shall receive a payment under this section equal to the payment it received in 2017, multiplied by 1.0147. In 2019, each taxing jurisdiction shall receive a payment under this section equal to the payment it received in the previous year, multiplied by one plus the inflation factor. In 2020, and in each year thereafter, each taxing jurisdiction shall receive a payment under this section equal to the payment it received in the previous year.

SECTION 1210h. 79.095 (5) of the statutes is created to read:

79.095 (5) SUNSET. Subsections (2), (3), and (4) (a) do not apply with regard to the payments made under this section after July 2017.

SECTION 1210p. 79.096 of the statutes is created to read:

79.096 State aid; personal property. (1) Beginning in 2019, and in each year thereafter, the department of administration shall pay to each taxing jurisdiction, as defined in s. 79.095 (1) (c), an amount equal to the property taxes levied on the items of personal property described under s. 70.111 (27) (b) for the property tax assessments as of January 1, 2017.

- (2) (a) Each municipality shall report to the department of revenue, in the time and manner determined by the department, the amount of the property taxes levied on the items of personal property described under s. 70.111 (27) (b) for the property tax assessments as of January 1, 2017, on behalf of the municipality and on behalf of other taxing jurisdictions.
- (b) Each taxing jurisdiction shall report to the department of revenue, in the time and manner determined by the department, any information the department considers necessary to administer this section.
- (3) Each taxing jurisdiction shall attribute to each tax incremental district within the taxing jurisdiction the district's proportionate share of the amount the taxing jurisdiction receives under sub. (1), except that this subsection does not apply after the district closes.

(4) The department of revenue shall certify the amount of the payment due each taxing jurisdiction under sub. (1) to the department of administration, and the department of administration shall make the payment on or before the first Monday in May.

SECTION 1211. 79.10 (7m) (cm) 1. a. of the statutes is amended to read:

79.10 (7m) (cm) 1. a. If, in any year, the total of the amounts determined under subs. (4), (5), and (5m) for any municipality is \$3,000,000 or more, the municipality, with the approval of the majority of the members of the municipality's governing body, may notify the department of administration to distribute the amounts directly to the municipality and the department of administration shall distribute the amounts at the time and in the manner provided under pars. (a) 1., (b) 1., and (c) 1. Beginning in 2018, if the municipality approves the distribution under this subd. 1. a. by enacting an ordinance and provides a copy of the ordinance to the department of administration and the department of revenue, the department of administration shall distribute the amounts determined under subs. (4), (5), and (5m) to the municipality as provided under this subd. 1. a. for the year in which the municipality enacts the ordinance and in all subsequent years until the municipality notifies the department of administration and the department of revenue that the municipality has repealed the ordinance or until the total amounts under subs. (4), (5), and (5m) to be distributed to the municipality in a year is less than \$3,000,000.

SECTION 1212. 79.14 of the statutes is amended to read:

79.14 School levy tax credit. The appropriation under s. 20.835 (3) (b), for the payments under s. 79.10 (4), is \$319,305,000 in 1994, 1995, and 1996; \$469,305,000 beginning in 1997 and ending in 2006; \$593,050,000 in 2007; \$672,400,000 in 2008; \$747,400,000 in 2009; \$732,550,000 in 2010, 2011, and 2012; \$747,400,000 in 2013, 2014, and 2015; and \$853,000,000 in 2016 and 2017; and \$940,000,000 in 2018 and in each year thereafter.

SECTION 1212m. 84.013 (3) (cb) of the statutes is cre- Vetoed ated to read:

In Part

84.013 (3) (cb) I 94 extending approximately 7.5 miles from US 12 to 130th Street near STH 65 in St. Croix

SECTION 1213. 84.013 (3) (km) of the statutes is repealed.

SECTION 1214. 84.013 (3) (rb) of the statutes is repealed.

SECTION 1215. 84.013 (3) (rj) of the statutes is repealed.

SECTION 1216. 84.013 (3) (tc) of the statutes is repealed.

Vetoed In Part

SECTION 1216bg. 84.013 (5) of the statutes is amended to read:

84.013 (5) Commencing with the 1985–87 biennial budget bill and biennially thereafter, the department shall request adjustments to the list of major highway projects under sub. (3) as listed projects are completed, projects are approved under sub. (6) and new projects are ready for construction. The department shall submit the proposed biennial adjustments for major highway projects to the transportation projects commission for review and recommendation as provided under s. 13.489. Submission of proposed adjustments to the transportation projects commission is not required after June 30, 2021.

SECTION 1216bi. 84.013 (6) of the statutes is amended to read:

84.013 (6) If following the enactment of the biennial budget bill the department determines that a highway project which was initially planned or designed as a reconditioning, reconstruction or resurfacing project is a major highway project and is ready for construction, the department shall submit the proposal for the specific project to the transportation projects commission for review and recommendation as provided under s. 13.489. After the transportation projects commission has submitted its report on the project, the department may request approval of the specific project as a major highway project from the joint committee on finance. If the joint committee on finance approves the project, the committee shall make such transfer of funds among the highway appropriations as deemed necessary and the department may proceed with construction. This subsection does not apply after June 30, 2021.

SECTION 1216c. 84.02 (8) (a) of the statutes is amended to read:

84.02 (8) (a) The department may make additions or deletions to the state trunk highway system by entering into a jurisdictional transfer agreement with any local unit of government a municipality or county. Addition to or deletion of any part of the state trunk highway system under this subsection may be made without regard to any mileage limitation or procedural requirement imposed under this section or chapter 518, laws of 1947.

SECTION 1216d. 84.02 (8) (b) of the statutes is amended to read:

84.02 (8) (b) The transfer of a highway under a jurisdictional transfer agreement must be may not take effect until the agreement is approved by the department and by resolution of the governing body of any the municipality or county board involved before the transfer of any highway becomes effective that is a party to the agreement.

SECTION 1216e. 84.02 (8) (c) of the statutes is amended to read:

84.02 (8) (c) A jurisdictional transfer agreement shall specify the length of time for which it is in effect and may contain any other terms and conditions that the department and the local unit of government may deem necessary regarding the maintenance or rehabilitation of any highway transferred under the agreement.

SECTION 1216f. 84.02 (8) (d) of the statutes is created to read:

84.02 (8) (d) The department and the municipality or county that is party to a jurisdictional transfer agreement shall maintain a record of the agreement.

SECTION 1216g. 84.02 (8m) of the statutes is created

84.02 (8m) REVERSION OF JURISDICTION. A highway that is under the jurisdiction of a municipality or county and that satisfies all of the following requirements shall be transferred to the jurisdiction of the department:

- (a) Prior to the effective date of this paragraph [LRB inserts date], jurisdiction of the highway was transferred under sub. (8) by the department to a municipality or county under a jurisdictional transfer agreement to which more than one municipality or county was party.
- (b) The municipality or county to which jurisdiction of the highway was transferred under par. (a) subsequently transferred under an agreement under s. 66.0307 territory in which the highway is located to another municipality or county and the agreement under s. 66.0307 did not specifically address jurisdiction of the highway.
- (c) Not later than 6 months after the effective date of this paragraph [LRB inserts date], the governing body of the transferor and transferee municipalities or counties under par. (b) adopt resolutions stating the intent of the municipality or county to revert jurisdiction of the highway to the department and provide a copy of the resolution to the department.

SECTION 1221m. 84.06 (14) of the statutes is created **Vetoed** to read:

84.06 (14)REPLACE-IN-KIND ALTERNATIVES REQUIRED. (a) In this subsection, "replace-in-kind alternative" means a project plan that does not include bicycle lanes, added lanes of travel, or significant design modifications that would include geometric or safety modifications, changes to highway alignment, or access points.

(b) The department shall conduct a study of and provide a cost estimate for a replace-in-kind alternative for each highway improvement project.

SECTION 1222. 84.062 of the statutes is repealed. **SECTION 1222m.** 84.07 (1) of the statutes is amended **Vetoed** to read:

84.07 (1) ROUTINE MAINTENANCE. Subject to sub. (1r), the state trunk highway system shall be maintained by the state at state expense. The department shall prescribe by rule specifications for such maintenance and may contract with any county highway committee or municipality to have all or certain parts of the work of maintaining the state trunk highways within or beyond the limits of the county or municipality, including interstate bridges, performed by the county or municipality, and any county or municipality may enter into such con-

In Part

In Part

tract. Maintenance activities include the application of protective coatings, the removal and control of snow, the removal, treatment and sanding of ice, interim repair of highway surfaces and adjacent structures, and all other operations, activities and processes required on a regular, continuing basis for the preservation of the highways on the state trunk system, and including the care and protection of trees and other roadside vegetation and suitable planting to prevent soil erosion or to beautify highways pursuant to s. 66.1037, and all routine measures deemed necessary to provide adequate traffic service. Maintenance activities also include the installation, replacement, rehabilitation, or maintenance of highway signs, highway lighting, and pavement markings, and the maintenance of traffic control signals and intelligent transportation systems. Maintenance activities do not include the removal and disposal of deer killed by vehicles on state trunk highways. The department may contract with a private entity for services or materials or both associated with the installation, replacement, rehabilitation, or maintenance of highway signs, highway lighting, and pavement markings and the maintenance of traffic control signals and intelligent transportation systems.

SECTION 1223. 84.41 (3) of the statutes is repealed. SECTION 1224. 84.59 (6) of the statutes is amended to read:

84.59 (6) The building commission may contract revenue obligations when it reasonably appears to the building commission that all obligations incurred under this section can be fully paid from moneys received or anticipated and pledged to be received on a timely basis. Except as provided in this subsection, the principal amount of revenue obligations issued under this section may not exceed \$3,931,472,900 \$4,055,372,900, excluding any obligations that have been defeased under a cash optimization program administered by the building commission, to be used for transportation facilities under s. 84.01 (28) and major highway projects for the purposes under ss. 84.06 and 84.09. In addition to the foregoing limit on principal amount, the building commission may contract revenue obligations under this section as the building commission determines is desirable to refund outstanding revenue obligations contracted under this section, to make payments under agreements or ancillary arrangements entered into under s. 18.55 (6) with respect to revenue obligations issued under this section, and to pay expenses associated with revenue obligations contracted under this section.

SECTION 1224m. 85.062 (3) (b) of the statutes is repealed.

SECTION 1224o. 85.066 (3) of the statutes is renumbered 85.066 (3) (intro.) and amended to read:

85.066 (3) COUNTIES CONTAINING EXPENDITURES RELATED TO CERTAIN TRANSPORTATION SYSTEMS IN A FIRST CLASS CITY. (intro.) A county containing a 1st class city The following may not incur any direct or indirect expensions.

ses, including the forfeiture of any revenue, relating to the operation or construction of a rail fixed guideway transportation system in the a 1st class city unless the expense incurred or revenue forfeited will be fully reimbursed by the 1st class city.

SECTION 1224p. 85.066 (3) (a) to (c) of the statutes are created to read:

85.066 (3) (a) Except as required to comply with the requirements under 49 USC 5329, the state.

- (b) An agency, as defined in s. 16.52 (7).
- (c) A county in which the 1st class city is located.

SECTION 1224r. 85.066 (4) of the statutes is created to read:

85.066 (4) REIMBURSEMENT. If a person restricted from incurring expenses under sub. (3), with the approval of the 1st class city, incurs a direct or indirect expense, including the forfeiture of any revenue, relating to the operation or construction of a rail fixed guideway transportation system in a 1st class city, the 1st class city shall fully reimburse the person for the expense.

SECTION 1224t. 85.09 (2) (a) of the statutes is amended to read:

85.09 (2) (a) The department of transportation shall have the first right to acquire, for present or future transportational or recreational purposes, any property used in operating a railroad or railway, including land and rails, ties, switches, trestles, bridges, and the like located on that property, that has been abandoned. The department of transportation may, in connection with abandoned rail property, assign this right to a state agency, the board of regents of the University of Wisconsin System, any county or municipality, or any transit commission. Acquisition by the department of transportation may be by gift, purchase, or condemnation in accordance with the procedure under s. 32.05, except that the power of condemnation may not be used to acquire property for the purpose of establishing or extending a recreational trail; a bicycle way, as defined in s. 340.01 (5s); a bicycle lane, as defined in s. 340.01 (5e); or a pedestrian way, as defined in s. 346.02 (8) (a). In addition to its property management authority under s. 85.15, the department of transportation may, subject to any prior action under s. 13.48 (14) (am) or 16.848 (1), lease and collect rents and fees for any use of rail property pending discharge of the department's duty to convey property that is not necessary for a public purpose. No person owning abandoned rail property, including any person to whom ownership reverts upon abandonment, may convey or dispose of any abandoned rail property without first obtaining a written release from the department of transportation indicating that the first right of acquisition under this subsection will not be exercised or assigned. No railroad or railway may convey any rail property prior to abandonment if the rail property is part of a rail line shown on the railroad's system map as in the process of abandonment, expected to be abandoned, or under study for possible abandonment

unless the conveyance or disposal is for the purpose of providing continued rail service under another company or agency. Any conveyance made without obtaining such release is void. The first right of acquisition of the department of transportation under this subsection does not apply to any rail property declared by the department to be abandoned before January 1, 1977. The department of transportation may acquire any abandoned rail property under this section regardless of the date of its abandonment.

SECTION 1225g. 85.20 (9) of the statutes is created to read:

85.20 (9) PROHIBITED EXPENDITURES. An eligible applicant may not use aids provided under this section for any purpose related to the operation of a rail fixed guideway transportation system, as defined in s. 85.066 (1), in a 1st class city.

SECTION 1226. 86.07 (2) (a) of the statutes is amended to read:

86.07 (2) (a) Subject to par. (b) and s. 86.16 (1m) (a) 2. and (c) and (6), no person shall make any excavation or fill or install any culvert or make any other alteration in any highway or in any manner disturb any highway or bridge without a permit therefor from the highway authority maintaining the highway. Such permit shall contain the statement and be subject to the condition that the work shall be constructed subject to such rules and regulations as may be prescribed by said authority and be performed and completed to its satisfaction, and in the case of temporary alterations that the highway or bridge shall be restored to its former condition, and that the permittee shall be liable to the town or county or state, as the case may be, for all damages which occur during the progress of said work or as a result thereof. Nothing herein shall abridge the right of the department, the county board or its highway committee, or any other local authority to make such additional rules, regulations and conditions not inconsistent herewith as may be deemed necessary and proper for the preservation of highways, or for the safety of the public, and to make the granting of any such permit conditional thereon. If any culvert is installed or any excavation or fill or any other alteration is made in violation of the provisions of this paragraph, the highway or bridge may be restored to its former condition by the highway authority in charge of the maintenance thereof at the expense of the violator; and any person who violates this paragraph shall be punished by a fine of not less than \$50 nor more than \$500, or by imprisonment not exceeding 6 months, or both.

SECTION 1226m. 86.16(1) of the statutes is amended to read:

86.16 (1) Any person, firm, or corporation, including any foreign corporation authorized to transact business in this state, may, subject to ss. 30.44 (3m), 30.45 and 196.491 (3) (d) 3m., with the written consent of the department with respect to state trunk highways, and

with the written consent of local authorities with respect to highways under their jurisdiction, including connecting highways, construct and operate lines, wires, or fiber for telecommunications service, as defined in s. 182.017 (1g) (cq), telegraph, telephone, or electric lines, or pipes or pipelines, for the purpose of transmitting voice, video, data, messages, water, liquid manure, heat, light, or power along, across, under, or within the limits of the highway.

SECTION 1227. 86.16 (6) of the statutes is created to read:

86.16 (6) If the department consents under sub. (1) to the construction of broadband infrastructure in underserved areas, as designated under s. 196.504 (2) (d), the department may not charge any fee for the initial issuance of any permit necessary to construct broadband infrastructure along, across, or within the limits of a highway.

SECTION 1227d. 86.19 (1g) of the statutes is renumbered 86.19 (1g) (am) and amended to read:

86.19 (1g) (am) 1. Subject to par. (b) subd. 2., the department shall erect and maintain 2 directional signs along eastbound and westbound I 94 and 4 directional signs along the exit ramps that correspond to the signs along the main roadway in Milwaukee County for the Basilica of St. Josaphat.

2. Upon receipt of sufficient contributions from interested parties, including any county, city, village, or town, to cover the costs of fabricating, erecting, and maintaining the signs specified in par. (a) subd. 1., the department shall erect and maintain the signs. No state funds, other than from the receipt of contributions under this paragraph subdivision, may be expended for the fabrication, erection, or maintenance of the signs.

SECTION 1227dm. 86.19 (1g) (d) of the statutes is created to read:

86.19 (**1g**) (d) 1. Subject to subd. 2., the department shall erect and maintain 2 directional signs along east-bound and westbound STH 54 for the Bergstrom Waterfowl Complex in the town of Bovina in Outagamie County.

2. Upon receipt of sufficient contributions from interested parties, including any county, city, village, or town, to cover the costs of fabricating, erecting, and maintaining the signs specified in subd. 1., the department shall erect and maintain the signs. No state funds, other than from the receipt of contributions under this subdivision, may be expended for the fabrication, erection, or maintenance of the signs.

SECTION 1227e. 86.19 (1g) (e) of the statutes is created to read:

86.19 (**1g**) (e) 1. Subject to subd. 2., the department shall erect and maintain all of the following directional signs for Soldiers Walk Memorial Park:

a. Two signs along eastbound and westbound I 94 at the STH 95 interchange and 2 signs along the exit ramps

that correspond to the signs along the main roadway in Jackson County.

- b. One sign along northbound STH 93 at the intersection with STH 95 in Trempealeau County.
- c. One sign along westbound STH 95 at the intersection with STH 93 in Trempealeau County.
- 2. No later than 6 months following receipt of sufficient contributions from interested parties, including any county, city, village, or town, to cover the costs of fabricating, erecting, and maintaining the signs specified in subd. 1., the department shall erect and maintain the signs. No state funds, other than from the receipt of contributions under this subdivision, may be expended for the fabrication, erection, or maintenance of the signs.

SECTION 1227em. 86.19 (1g) (f) of the statutes is created to read:

86.19 (**1g**) (f) The department shall erect and maintain 2 directional signs along eastbound and westbound I 94 at the Somers Road interchange in Kenosha County for Shoreland Lutheran High School.

SECTION 1227er. 86.19 (1g) (g) of the statutes is created to read:

- 86.19 (**1g**) (g) 1. Subject to subd. 2., the department shall erect and maintain 2 signs displaying the words "Town of Lawrence" along northbound and southbound I 41 at the north and south jurisdictional boundaries of the town of Lawrence in Brown County.
- 2. Upon receipt of sufficient contributions from interested parties, including any county, city, village, or town, to cover the costs of fabricating, erecting, and maintaining the signs specified in subd. 1., the department shall erect and maintain the signs. No state funds, other than from the receipt of contributions under this subdivision, may be expended for the fabrication, erection, or maintenance of the signs.

SECTION 1227p. 86.19 (1h) of the statutes is renumbered 86.19 (1g) (bm) and amended to read:

- 86.19 (**1g**) (bm) 1. Subject to par. (b) <u>subd. 2.</u>, the department shall erect and maintain 2 directional signs along eastbound and westbound I 90/94 and 2 directional signs along the exit ramps that correspond to the signs along the main roadway for the Wisconsin Basketball Coaches Association Hall of Fame in Columbia County.
- 2. Upon receipt of sufficient contributions from interested parties, including any county, city, village, or town, to cover the costs of fabricating, erecting, and maintaining the signs specified in par. (a) subd. 1., the department shall erect and maintain the signs. No state funds, other than from the receipt of contributions under this paragraph subdivision, may be expended for the fabrication, erection, or maintenance of the signs.

SECTION 1227t. 86.19 (1i) of the statutes is renumbered 86.19 (1g) (c), and 86.19 (1g) (c) 1. (intro.) and 2., as renumbered, are amended to read:

- 86.19 (**1g**) (c) 1. (intro.) Subject to par. (b) subd. 2., the department shall erect and maintain all of the following directional signs for the Iola Car Show:
- 2. Upon receipt of sufficient contributions from interested parties, including any county, city, village, or town, to cover the costs of fabricating, erecting, and maintaining the signs specified in par. (a) subd. 1., the department shall erect and maintain the signs. No state funds, other than from the receipt of contributions under this paragraph subdivision, may be expended for the fabrication, erection, or maintenance of the signs.

SECTION 1228. 86.30 (2) (a) 3. of the statutes is amended to read:

86.30 (2) (a) 3. For each mile of road or street under the jurisdiction of a municipality as determined under s. 86.302, the mileage aid payment shall be \$2,117 \$2,202 in calendar years 2013 and 2014 year 2017 and \$2,202 \$2,389 in calendar year 2015 2018 and thereafter.

SECTION 1229. 86.30 (9) (b) of the statutes is amended to read:

86.30 (9) (b) For the purpose of calculating and distributing aids under sub. (2), the amounts for aids to counties are \$94,615,600 \$98,400,200 in calendar years 2013 and 2014 year 2017 and \$98,400,200 \$111,093,800 in calendar year 2015 2018 and thereafter. These amounts, to the extent practicable, shall be used to determine the statewide county average cost—sharing percentage in the particular calendar year.

SECTION 1230. 86.30 (9) (c) of the statutes is amended to read:

86.30 (9) (c) For the purpose of calculating and distributing aids under sub. (2), the amounts for aids to municipalities are \$308,904,300 \$321,260,500 in calendar years 2013 and 2014 year 2017 and \$321,260,500 \$348,639,300 in calendar year 2015 2018 and thereafter. These amounts, to the extent practicable, shall be used to determine the statewide municipal average cost—sharing percentage in the particular calendar year.

SECTION 1231. 86.31 (3g) of the statutes is amended to read:

86.31 (**3g**) COUNTY TRUNK HIGHWAY IMPROVEMENTS — DISCRETIONARY GRANTS. From the appropriation under s. 20.395 (2) (ft), the department shall allocate \$5,127,000 in fiscal years 2011–12 and 2012–13, \$4,727,000 in fiscal year 2013–14, and \$5,127,000 in fiscal year years 2014–15 to 2016–17 and \$5,393,400 in fiscal year 2017–2018 and each fiscal year thereafter, to fund county trunk highway improvements with eligible costs totaling more than \$250,000. The funding of improvements under this subsection is in addition to the allocation of funds for entitlements under sub. (3).

SECTION 1232. 86.31 (3m) of the statutes is amended to read:

86.31 (3m) TOWN ROAD IMPROVEMENTS — DISCRETIONARY GRANTS. From the appropriation under s. 20.395 (2) (ft), the department shall allocate \$732,500 in fiscal year 2009—10 and in fiscal year 2010—11, and \$5,732,500 in fiscal year years 2011—12 to 2016—17 and \$5,923,600 in fiscal year 2017—18 and each fiscal year thereafter, to fund town road improvements with eligible costs totaling \$100,000 or more. The funding of improvements under this subsection is in addition to the allocation of funds for entitlements under sub. (3).

SECTION 1233. 86.31 (3r) of the statutes is amended to read:

86.31 (**3r**) MUNICIPAL STREET IMPROVEMENTS — DISCRETIONARY GRANTS. From the appropriation under s. 20.395 (2) (ft), the department shall allocate \$1,020,000 in fiscal year 2007–08, \$1,040,400 in fiscal year 2008–09, and \$976,500 in fiscal year years 2009–10 to 2016–17 and \$3,850,400 in fiscal year 2017–18 and each fiscal year thereafter, to fund municipal street improvement projects having total estimated costs of \$250,000 or more. The funding of improvements under this subsection is in addition to the allocation of funds for entitlements under sub. (3).

SECTION 1235. 89.03 (3) of the statutes is created to read:

89.03 (3) The examining board shall promulgate rules specifying a procedure for addressing allegations that a person licensed or certified by the veterinary examining board under this chapter has practiced as a veterinarian or veterinary technician while impaired by alcohol or other drugs or that his or her ability to practice is impaired by alcohol or other drugs, and for assisting a person licensed by the veterinary examining board under this chapter who requests to participate in the procedure or who requests assistance in obtaining mental health services. In promulgating rules under this subsection, the examining board shall seek to facilitate early identification of chemically dependent veterinarians or veterinary technicians and encourage their rehabilitation. The rules promulgated under this subsection may be used in conjunction with the formal disciplinary process under this chapter. The examining board may contract with another entity to administer the procedure specified under the rules promulgated under this subsection.

SECTION 1236. 92.14 (6) (c) of the statutes is created to read:

92.14 (6) (c) When preparing an annual grant allocation plan under par. (b), the department and the department of natural resources shall consider the existence and location of impaired water bodies that the department of natural resources has identified to the federal environmental protection agency under 33 USC 1313 (d) (1) (A) and agricultural enterprise areas designated under s. 91.84, and shall give priority to providing cost–sharing for nutrient management planning projects that are in or near, or that affect, those areas.

SECTION 1237m. 93.29 of the statutes is repealed. **SECTION 1238.** 93.33 of the statutes is renumbered 106.40, and 106.40 (3) (a) and (4s) (c), as renumbered, are amended to read:

106.40 (3) (a) The council shall create an executive committee that includes the secretary of agriculture, trade and consumer protection or his or her designee and, the state superintendent of public instruction or his or her designee, and the secretary of workforce development or his or her designee. The council shall select members of the executive committee so that fewer than half of the members of the executive committee are state employees. The executive committee shall provide guidance to the council and to staff that support the functions of the council. The executive committee shall meet between meetings of the council.

(4s) (c) Each of the individuals specified in s. 15.137 (2) 15.227 (15) (a) 8. and the chancellor of the University of Wisconsin–Extension, jointly or individually, shall annually prepare a review of agricultural education programs in the University of Wisconsin System, with input from or review by the University of Wisconsin System administration.

SECTION 1241d. 93.51 (3) (d) of the statutes is amended to read:

93.51 (3) (d) Advisors and the department shall keep confidential all information obtained in the process of providing advice or counseling, except for aggregate data compiled by the department at not less than a countywide level that does not contain any reference to the identity of any individual or individual farm. Any such confidential information contained in a record is not subject to the right of public inspection and copying under s. 19.35 (1). This paragraph does not apply to information relating to possible criminal misconduct.

SECTION 1241m. 93.59 (5) of the statutes is amended to read:

93.59 (5) In any fiscal year, the department may not provide more than \$20,000 \$40,000 to any single producer-led group or legal entity on behalf of the group.

SECTION 1242. 94.64 (3) (a) 1. of the statutes is amended to read:

94.64 (3) (a) 1. Except as provided in subd. 2., no person may manufacture or distribute fertilizer in this state without an annual license from the department. A separate license is required for each business location and each mobile unit at which the person manufactures fertilizer. A license shall expire expires on August 14 September 30 annually and is not transferable between persons or locations.

SECTION 1243. 94.64 (3) (c) of the statutes is created to read:

94.64 (3) (c) A person who has been issued a license under this subsection shall annually, on or before the date the person's license expires, notify the department that

the person intends to maintain, amend, or discontinue the license.

SECTION 1244. 94.64 (3r) (b) (intro.) of the statutes is amended to read:

94.64 (**3r**) (b) (intro.) Beginning with the license year that begins on August 15, 2013, a- A person applying for a license under sub. (3) shall pay the following agricultural chemical cleanup surcharges, unless the department establishes different surcharges under s. 94.73 (15):

SECTION 1245. 94.64 (3r) (b) 1. of the statutes is amended to read:

94.64 (**3r**) (b) 1. For each business location and each mobile unit that the applicant uses to manufacture fertilizer in this state, other than a business location or mobile unit that is also licensed under s. 94.685 or 94.703, \$11.20 \$20, except as provided in s. 94.73 (15).

SECTION 1246. 94.64 (3r) (b) 2. of the statutes is amended to read:

94.64 (**3r**) (b) 2. If the applicant distributes, but does not manufacture, fertilizer in this state, \$11.20 \$20, except as provided in s. 94.73 (15).

SECTION 1247. 94.64 (3r) (b) 3. of the statutes is created to read:

94.64 (**3r**) (b) 3. For each business location and each mobile unit that the applicant uses to distribute bulk fertilizer in this state, \$25 in addition to the surcharge under subd. 2., except as provided in s. 94.73 (15).

SECTION 1248. 94.64 (4) (a) 5. of the statutes is amended to read:

94.64 (4) (a) 5. An agricultural chemical cleanup surcharge of 35 10 cents per ton on all fertilizer that the person sells or distributes in this state after June 30, 2014, unless the department establishes a different surcharge under, except as provided in s. 94.73 (15).

SECTION 1249. 94.64 (5) (a) (intro.) of the statutes is amended to read:

94.64 (5) (a) *Requirement*. (intro.) A person who is required to pay fees or surcharges under sub. (4) shall do all of the following by August 14 annually, on or before the date the license expires:

SECTION 1250. 94.64 (5) (b) of the statutes is amended to read:

94.64 (5) (b) Extended deadline. The department may extend the filing deadline under par. (a) for up to 30 days for cause, in response to a request filed before August 14 the filing deadline.

SECTION 1251. 94.65 (2) (a) of the statutes is amended to read:

94.65 (2) (a) Except as provided under par. (b), no person may manufacture or distribute a soil or plant additive in this state unless the person first obtains an annual license from the department. Application for a license or for renewal of a license shall be made on forms provided by the department and shall be accompanied by an annual license fee of \$25. All licenses expire on March 31. A license expires on September 30 annually.

SECTION 1252. 94.65 (6) (a) (intro.) of the statutes is amended to read:

94.65 (6) (a) (intro.) Each person holding a permit for the distribution of a soil or plant additive under sub. (3) shall do all of the following:

SECTION 1253. 94.65 (6) (a) 1. of the statutes is amended to read:

94.65 (6) (a) 1. Annually by March 31, on or before the date the person's permit expires, file with the department a tonnage report setting forth the number of tons of each soil or plant additive distributed during the preceding year the 12 months ending on the preceding June 30 by that person, or by any other person authorized under sub. (3) (a) 2. to distribute under the name of that person and pay to the department a fee of 25 cents per ton so distributed. The minimum total fee is \$25.

SECTION 1254. 94.65 (6) (a) 3. of the statutes is amended to read:

94.65 (6) (a) 3. Annually by March 31, on or before the date the permit expires, pay to the department a research fee of 10 cents for each ton of soil or plant additive distributed as described in the tonnage report filed under subd. 1. The minimum research fee is \$1 for 10 tons or less. The department shall credit this fee to the appropriation account under s. 20.115 (7) (h).

SECTION 1255. 94.65 (6) (a) 4. of the statutes is amended to read:

94.65 (6) (a) 4. Annually by March 31, on or before the date the permit expires, pay to the department a groundwater fee of 10 cents for each ton of soil or plant additive distributed, as described in the tonnage report filed under subd. 1. The minimum groundwater fee is \$1 for 10 tons or less. All groundwater fees shall be credited to the environmental fund for environmental management.

SECTION 1256. 94.65 (6) (a) 5. of the statutes is created to read:

94.65 (6) (a) 5. Annually, on or before the date the permit expires, notify the department that the person intends to maintain, amend, or discontinue the permit.

SECTION 1257. 94.65 (6) (b) of the statutes is amended to read:

94.65 (6) (b) If by March 31 the date the permit expires a person holding a permit under sub. (3) has failed to file a tonnage report or to pay the inspection fee required under par. (a), the department may summarily suspend or revoke the permit or license issued under this section. A penalty of 10 percent of the inspection fee due shall be assessed against the permit holder for all inspection fees not paid when due. The minimum total penalty is \$10. An unpaid inspection fee or penalty shall constitute a debt owed the department by the permit holder until paid. The department may not issue or renew a license or issue a permit or amended permit to a person owing an unpaid inspection fee or penalty.

SECTION 1258. 94.68 (2) (a) (intro.) of the statutes is renumbered 94.68 (2) (intro.).

SECTION 1259. 94.68 (2) (a) 1. of the statutes is renumbered 94.68 (2) (am).

SECTION 1260. 94.68 (2) (a) 2. of the statutes is renumbered 94.68 (2) (bm) and amended to read:

94.68 (2) (bm) A report identifying each pesticide that the applicant sells or distributes for use in this state and the gross revenue that the applicant derived from the sale or distribution of each pesticide during the preceding year, as defined in s. 94.681 (1) (d).

SECTION 1261. 94.68 (2) (b) of the statutes is repealed.

SECTION 1262. 94.68 (3) of the statutes is amended to read:

94.68 (3) At least 15 days before a person holding a license under this section begins to sell or distribute for use in this state a pesticide product that was not identified in the person's most recent annual license application, the person shall file a supplementary report with the information required under sub. (2) (a) 2. (bm) and any fees and surcharges required under s. 94.681. The department may not disclose sales revenue information obtained under this subsection submitted under s. 94.68 (2) (a) 2., 2015 stats.

SECTION 1263. 94.681 (1) (cm) of the statutes is repealed.

SECTION 1264. 94.681 (1) (d) of the statutes is repealed.

SECTION 1265. 94.681 (2) (intro.) of the statutes is renumbered 94.681 (2) and amended to read:

94.681 (2) ANNUAL LICENSE FEE. An applicant for a license under s. 94.68 shall pay an annual license fee for each pesticide product that the applicant sells or distributes for use in this state. The during the license year. Except as provided in sub. (5) or (6), the amount of the fee is based on sales of pesticide products during the payment period. An applicant shall pay an estimated fee before the start of each license year as provided in sub. (3s) (a) and shall make a fee adjustment payment before the end of the license year if required under sub. (3s) (b). Except as provided in sub. (5) or (6), the fee for each pesticide product is as follows: \$500.

SECTION 1266. 94.681 (2) (a) of the statutes is repealed.

SECTION 1267. 94.681 (2) (b) of the statutes is repealed.

SECTION 1268. 94.681 (2) (c) of the statutes is repealed.

SECTION 1269. 94.681 (3) (intro.) of the statutes is renumbered 94.681 (3) and amended to read:

94.681 (3) Nonhousehold pesticides; Cleanup surcharge. An applicant for a license under s. 94.68 shall pay an agricultural chemical cleanup surcharge for each nonhousehold pesticide product that the applicant sells or distributes for use in this state. The amount of the sur-

charge is based on sales of nonhousehold pesticide products during the payment period. An applicant shall pay an estimated surcharge before the start of each license year as provided in sub. (3s) (a) and shall make a surcharge adjustment payment before the end of the license year if required by sub. (3s) (b) during the license year. Except as provided in sub. (6) or under s. 94.73 (15), the amount of the surcharge is as follows: \$30.

SECTION 1270. 94.681 (3) (a) of the statutes is repealed.

SECTION 1271. 94.681 (3) (b) of the statutes is repealed.

SECTION 1272. 94.681 (3) (c) of the statutes is repealed.

SECTION 1273. 94.681 (3m) of the statutes is repealed.

SECTION 1274. 94.681 (3s) (a) of the statutes is renumbered 94.681 (3s) and amended to read:

94.681 (3s) Payment of fees and surcharges. Before the start of a license year, and at least 15 days before beginning to sell a new pesticide product in this state, an applicant or licensee shall estimate the gross revenues that the applicant will receive from sales of each pesticide product during the payment period that ends during the year for which a license is sought under s. 94.68 and shall pay the amounts under subs. (2), (3), and (3m) based on that estimate. At least 15 days before beginning to sell a new pesticide product in this state, a licensee shall estimate the gross revenues that the applicant will receive from sales of that pesticide product during the payment period in which the licensee begins to sell the pesticide product and shall pay the amounts under subs. (2), (3), and (3m) based on that estimate pay the amounts due under subs. (2) and (3).

SECTION 1275. 94.681 (3s) (b) of the statutes is repealed.

SECTION 1276. 94.681 (3s) (c) of the statutes is repealed.

SECTION 1277. 94.681 (5) of the statutes is amended to read:

94.681 (5) UNREPORTED PESTICIDE; INCREASED LICENSE FEE. If a person applying for or holding a license under s. 94.68 sells or distributes a pesticide product for use in this state without having filed a report for the product under s. 94.68 (2) (a) 2. (bm) or (3), the license fee for that product is twice the amount determined under sub. (2), except that, if the pesticide product is exempt from federal registration under 40 CFR 152.25, the license fee for that product is \$250.

SECTION 1278. 94.681 (6) (a) (intro.) of the statutes is renumbered 94.681 (6) (intro.).

SECTION 1279. 94.681 (6) (a) 1. of the statutes is renumbered 94.681 (6) (am).

SECTION 1280. 94.681 (6) (a) 2. of the statutes is repealed.

SECTION 1281. 94.681 (6) (a) 3. of the statutes is renumbered 94.681 (6) (bm) and amended to read:

94.681 (6) (bm) By March December 31 of the year following the year in which the person stopped selling or distributing the pesticide product for use in this state, pay a final license fee of \$500 for the pesticide product, calculated under sub. (2) based on the sales of the pesticide product during the period specified in subd. 2.

SECTION 1282. 94.681 (6) (a) 4. of the statutes is renumbered 94.681 (6) (c) and amended to read:

94.681 (6) (c) If the product is a nonhousehold pesticide, pay a final agricultural chemical cleanup surcharge calculated under sub. (3) based on sales of the product during the period specified in subd. 2. of \$30.

SECTION 1283. 94.681 (6) (a) 5. of the statutes is repealed.

SECTION 1284. 94.681 (6) (b) of the statutes is repealed.

SECTION 1285. 94.681 (7) (a) (intro.) and 1. of the statutes are consolidated, renumbered 94.681 (7) (a) and amended to read:

94.681 (7) (a) *License fees*. The department shall deposit all license fees collected under subs. (2), (5) and (6) (a) 3. (bm) in the agrichemical management fund, except as follows: 1. The that the department shall deposit an amount equal to \$94 \$108 for each pesticide product for which an applicant pays a license fee in the environmental fund for environmental management.

SECTION 1286. 94.681 (7) (a) 2. of the statutes is repealed.

SECTION 1287. 94.681 (7) (b) of the statutes is amended to read:

94.681 (7) (b) *Nonhousehold pesticides; cleanup surcharge*. The department shall deposit the surcharges collected under subs. (3) and (6) (a) 4. (c) in the agricultural chemical cleanup fund.

SECTION 1288. 94.681 (7) (bm) of the statutes is repealed.

SECTION 1289. 94.685 (3) (a) 2. of the statutes is amended to read:

94.685 (3) (a) 2. An agricultural chemical cleanup surcharge of \$22.40, unless the department establishes a different surcharge under \$20, except as provided in s. 94.73 (15).

SECTION 1290. 94.703 (3) (a) 2. of the statutes is amended to read:

94.703 (3) (a) 2. An agricultural chemical cleanup surcharge of \$30.40, unless the department establishes a different surcharge under \$20, except as provided in subd. 3. or s. 94.73 (15).

SECTION 1291. 94.703 (3) (a) 3. of the statutes is created to read:

94.703 (3) (a) 3. If the applicant manufactures or distributes bulk pesticides in this state, an additional agricultural chemical cleanup surcharge of \$25, except as provided in s. 94.73 (15).

SECTION 1292. 94.704 (3) (a) 2. of the statutes is amended to read:

94.704 (3) (a) 2. An agricultural chemical cleanup surcharge of \$11.20, unless the department establishes a different surcharge under \$10, except as provided in s. 94.73 (15).

SECTION 1293. 94.72 (6) (a) 1. of the statutes is repealed.

SECTION 1294. 94.72 (6) (a) 2. of the statutes is amended to read:

94.72 (6) (a) 2. For commercial feeds distributed in this state on or after January 1, 2002, a A feed inspection fee of 23 cents per ton, except that if the person distributes less than 200 tons of commercial feed in a year, the feed inspection fee is \$46.

SECTION 1295. 94.72 (6) (a) 3. of the statutes is amended to read:

94.72 (6) (a) 3. Beginning on October 29, 1999, for commercial feeds distributed in this state a A weights and measures inspection fee of 2 cents per ton, except that if the person distributes less than 200 tons of commercial feed in a year, the weights and measures inspection fee is \$4.

SECTION 1296. 94.72 (6) (b) of the statutes is amended to read:

94.72 (6) (b) Responsibility. Except as provided in par. (d), if more than one manufacturer or distributor is involved in the chain of distribution, the one who first sells or distributes commercial feed in this state or to a person in this state for further sale is responsible for the payment of inspection fees for the feed. No inspection fees are required for commercial feeds sold under the name and label of another licensee if the inspection fees have been or will be paid by a previous manufacturer or distributor in the chain of distribution as evidenced by an invoice or sales receipt. No inspection fees have been or will be paid by a previous manufacturer or distributor in the chain of distribution as evidenced by an invoice or sales receipt.

SECTION 1297. 94.72 (6) (c) of the statutes is repealed.

SECTION 1298. 94.72 (6) (f) of the statutes is repealed.

SECTION 1299. 94.72 (6) (g) of the statutes is repealed.

SECTION 1300. 94.72 (6) (h) of the statutes is repealed.

SECTION 1301. 94.73 (3m) (w) of the statutes is repealed.

SECTION 1302. 94.73 (6) (b) of the statutes is amended to read:

94.73 (6) (b) Except as provided in pars. (c) and (e), the department shall reimburse a responsible person an amount equal to 75 percent of the corrective action costs incurred for each discharge site that are greater than

In Part

\$3,000 and less than \$400,000 for costs incurred before July 1, 2017, or that are greater than \$3,000 and less than \$650,000 for costs incurred on or after July 1, 2017.

SECTION 1303. 94.73 (6) (c) (intro.) of the statutes is amended to read:

94.73 (6) (c) (intro.) Except as provided in par. (e), the department shall reimburse a responsible person an amount equal to 75 percent of the corrective action costs incurred for each discharge site that are greater than \$7,500 and less than \$400,000 for costs incurred before July 1, 2017, or that are greater than \$7,500 and less than \$650,000 for costs incurred on or after July 1, 2017, if any of the following applies:

SECTION 1304. 94.73 (15) of the statutes is repealed and recreated to read:

94.73 (15) SURCHARGE ADJUSTMENTS. (a) On May 1 annually, the department shall determine the amount available in the agricultural chemical cleanup fund.

- (b) If the amount determined under par. (a) is more than \$1,500,000, the surcharges for the subsequent year shall be as follows:
 - 1. Under s. 94.64 (3r) (b) 1. and 2., \$0.
 - 2. Under s. 94.64 (3r) (b) 3., \$0.
 - 3. Under s. 94.64 (4) (a) 5., \$0.
 - 4. Under s. 94.681 (3), \$0.
 - 5. Under s. 94.685 (3) (a) 2., \$0.
 - 6. Under s. 94.703 (3) (a) 2., \$0.
 - 7. Under s. 94.703 (3) (a) 3., \$0.
 - 8. Under s. 94.704 (3) (a) 2., \$0.
- (c) If the amount determined under par. (a) is \$1,500,000 or less, but more than \$750,000, the surcharges for the subsequent license year shall be as follows:
 - 1. Under s. 94.64 (3r) (b) 1. and 2., \$10.
 - 2. Under s. 94.64 (3r) (b) 3., \$12.50.
 - 3. Under s. 94.64 (4) (a) 5., 5 cents per ton.
 - 4. Under s. 94.681 (3), \$15.
 - 5. Under s. 94.685 (3) (a) 2., \$10.
 - 6. Under s. 94.703 (3) (a) 2., \$10.
 - 7. Under s. 94.703 (3) (a) 3., \$12.50.
 - 8. Under s. 94.704 (3) (a) 2., \$5.

SECTION 1305. 94.74 of the statutes is repealed.

SECTION 1305g. 101.01 (11) (d) of the statutes is created to read:

101.01 (11) (d) A not-for-profit facility with the primary purpose of housing or rehabilitating abandoned, injured, or sick wildlife.

SECTION 1305m. 101.01 (12) (d) of the statutes is created to read:

101.01 (12) (d) A not-for-profit facility with the primary purpose of housing or rehabilitating abandoned, injured, or sick wildlife.

SECTION 1305p. 101.02 (7w) of the statutes is created to read:

101.02 (7w) Notwithstanding sub. (7) (a), and except as provided in this subsection and s. 66.0414 (3) (c), no city, village, town, or county may make or enforce a local Vetoed order that limits blasting at a quarry, as defined in s. 66.0414 (2) (d). A city, village, town, or county may petition the department for an order granting the city, village, town, or county the authority to impose additional restrictions and requirements related to blasting on the operator of a quarry, and the department may not charge a fee for the petition. If the department issues the order, the order may grant the city, village, town, or county the authority to impose restrictions and requirements related to blasting at the quarry that are more restrictive than the requirements under s. 101.15 related to blasting and rules promulgated by the department under s. 101.15 (2) (e) related to blasting. If a city, village, town, or county submits a petition under this subsection because of concerns regarding the potential impact of blasting on a qualified historic building, as defined in s. 101.121 (2) (c), the department may require the operator of the quarry to pay the costs of an impact study related to the qualified historic building.

SECTION 1306. 101.02 (20) (a) of the statutes is amended to read:

101.02 (20) (a) For purposes of this subsection, "license" means a license, permit, or certificate of certification or registration issued by the department for an occupation or profession under s. 101.122 (2) (c), 101.15 (2) (e), 101.16 (3g), 101.17, 101.178 (2) or (3) (a), 101.63 (2) or (2m), 101.653, 101.73 (5) or (6), 101.82 (1m), (1v), and (2), 101.935, 101.95, 101.951, 101.952, 101.96 (2), 101.985 (1) to (3), 145.02 (4), 145.035, 145.045, 145.15, 145.16, 145.165, 145.17, 145.175, 145.18, or 167.10 (6m) or under rules promulgated under ch. 101 or 145.

SECTION 1307. 101.02 (21) (a) of the statutes is amended to read:

101.02 (21) (a) In this subsection, "license" means a license, permit, or certificate of certification or registration issued by the department for an occupation or profession under s. 101.122 (2) (e), 101.15 (2) (e), 101.16 (3g), 101.17, 101.178 (2) or (3) (a), 101.63 (2) or (2m), 101.653, 101.73 (5) or (6), 101.82 (1m), (1v), and (2), 101.935, 101.95, 101.951, 101.952, 101.96 (2), 101.985 (1) to (3), 145.02 (4), 145.035, 145.045, 145.15, 145.16, 145.165, 145.17, 145.175, 145.18, or 167.10 (6m) or under rules promulgated under ch. 101 or 145.

SECTION 1308. 101.02 (24) (a) 2. of the statutes is amended to read:

101.02 (24) (a) 2. "License" means a license, permit, or certificate of certification or registration issued by the department for an occupation or profession under s. 101.122 (2) (c), 101.15 (2) (e), 101.16 (3g), 101.17, 101.178 (2) or (3) (a), 101.63 (2) or (2m), 101.653, 101.654, 101.73 (5) or (6), 101.82 (1m), (1v), and (2), 101.935, 101.95, 101.951, 101.952, 101.985 (1) to (3), 145.02 (4), 145.035, 145.045, 145.15, 145.16, 145.165, 145.17, 145.175, 145.18, or 167.10 (6m) or under rules promulgated under ch. 101 or 145.

Vetoed In Part

SECTION 1310. 101.122 of the statutes is repealed. SECTION 1311. 101.19 (1g) (i) of the statutes is repealed.

SECTION 1312. 101.19 (1r) of the statutes is amended to read:

101.19 (**1r**) Notwithstanding subs. (1g) and (1m), the department shall waive any fee imposed on an individual who is eligible for the veterans fee waiver program under s. 45.44 for a license, permit, or certificate of certification or registration issued by the department under s. 101.122 (2) (c), 101.15 (2) (e), 101.16 (3g), 101.17, 101.178 (2) or (3) (a), 101.63 (2) or (2m), 101.653, 101.73 (5) or (6), 101.82 (1m), (1v), and (2), 101.935, 101.95, 101.951, 101.952, 101.985 (1) to (3), 145.02 (4), 145.035, 145.045, 145.07 (12), 145.15, 145.16, 145.165, 145.17, 145.175, 145.18, or 167.10 (6m).

SECTION 1319g. 101.61 (1) of the statutes is amended to read:

101.61 (1) "Dwelling" means any building that contains one or 2 dwelling units. "Dwelling unit" means a structure or that part of a structure which is used or intended to be used as a home, residence or sleeping place by one person or by 2 or more persons maintaining a common household, to the exclusion of all others. "Dwelling" and "dwelling unit" do not include a primitive rural hunting cabin.

SECTION 1319r. 101.61 (3) of the statutes is created to read:

- 101.61 (3) "Primitive rural hunting cabin" means a structure that satisfies all of the following:
 - (a) The structure is not used as a home or residence.
- (b) The structure is used principally for recreational hunting activity.
 - (c) The structure does not exceed 2 stories in height.
 - (d) The structure satisfies any of the following:
- 1. The structure was constructed before December 31, 1997.
- 2. The structure results from alterations made to a structure described in subd. 1.
- 3. The structure replaces a structure described in subd. 1.

SECTION 1324m. 101.63 (11) of the statutes is created to read:

101.63 (11) Develop and maintain computer software available to the public that provides the information, tools, and calculations required for a person to determine whether plans for the construction of, addition to, or alteration of a dwelling comply with the energy efficiency requirements of the uniform dwelling code promulgated under sub. (1).

SECTION 1325g. 101.65 (1g) of the statutes is created to read:

101.65 (**1g**) May not exercise jurisdiction over the construction or inspection of primitive rural hunting cabins.

SECTION 1325r. 101.82 (1) of the statutes is amended to read:

101.82 (1) Promulgate by rule a state electrical wiring code that establishes standards for installing, repairing, and maintaining electrical wiring. The rules shall include separate standards, established in consultation with the <u>uniform</u> dwelling code council, that apply only to electrical wiring in camping units that are set in a fixed location in a campground for which a permit is issued under s. 254.47 97.67, that contain a sleeping place, and that are used for seasonal overnight camping. The rules do not apply to electrical wiring in primitive rural hunting cabins, as defined in s. 101.61 (3). Where feasible, the rules shall reflect nationally recognized standards.

SECTION 1332. 102.07 (12m) (a) 1. of the statutes is amended to read:

102.07 (12m) (a) 1. "Institution of higher education" means an institution within the University of Wisconsin System, a technical college, a tribally controlled college controlled by an Indian tribe that has elected under s. 102.05 (2) to become subject to this chapter, a school approved under s. 38.50 440.52, or a private, nonprofit institution of higher education located in this state.

SECTION 1384. 103.503 (1) (a) of the statutes is amended to read:

103.503 (1) (a) "Accident" means an incident caused, contributed to, or otherwise involving an employee that resulted or could have resulted in death, personal injury, or property damage and that occurred while the employee was performing the work described in s. 66.0903 (4), 2013 stats., or s. 16.856 (2m), 2015 stats., on a project of public works or while the employee was performing work on a public utility project.

SECTION 1385. 103.503 (1) (c) of the statutes is amended to read:

103.503 (1) (c) "Contracting agency" means a local governmental unit, as defined in s. 66.0903 (1) (d), or a state agency, as defined in s. 16.856 (1) (h), that has contracted for the performance of work on a project of public works or a public utility that has contracted for the performance of work on a public utility project.

SECTION 1386. 103.503 (1) (e) of the statutes is amended to read:

103.503 (1) (e) "Employee" means a laborer, worker, mechanic, or truck driver who performs the work described in s. 66.0903 (4), 2013 stats., or s. 16.856 (2m), 2015 stats., on a project of public works or on a public utility project.

SECTION 1387. 103.503 (1) (fm) of the statutes is created to read:

103.503 (1) (fm) "Local governmental unit" means a political subdivision of this state, a special purpose district in this state, an instrumentality or corporation of such a political subdivision or special purpose district, a

combination or subunit of any of the foregoing, or an instrumentality of the state and any of the foregoing.

SECTION 1388. 103.503 (1) (g) of the statutes is amended to read:

103.503 (1) (g) "Project of public works" means a project of public works that is subject to s. 16.856 or that would be subject to s. 66.0903, 2013 stats., if the project were erected, constructed, repaired, remodeled, or demolished prior to January 1, 2017, or that would be subject to s. 16.856, 2015 stats., if the project were erected, constructed, repaired, remodeled, or demolished prior to the effective date of this paragraph [LRB inserts date].

SECTION 1389. 103.503 (1) (j) of the statutes is created to read:

103.503 (1) (j) "State agency" means any office, department, independent agency, institution of higher education, association, society, or other body in state government created or authorized to be created by the constitution or any law, including the legislature and the courts. "State agency" also includes the University of Wisconsin Hospitals and Clinics Authority, the Fox River Navigational System Authority, and the Wisconsin Aerospace Authority.

SECTION 1390. 103.503 (2) of the statutes is amended to read:

103.503 (2) SUBSTANCE ABUSE PROHIBITED. No employee may use, possess, attempt to possess, distribute, deliver, or be under the influence of a drug, or use or be under the influence of alcohol, while performing the work described in s. 66.0903 (4), 2013 stats., or s. 16.856 (2m), 2015 stats., on a project of public works or while performing work on a public utility project. An employee is considered to be under the influence of alcohol for purposes of this subsection if he or she has an alcohol concentration that is equal to or greater than the amount specified in s. 885.235 (1g) (d).

SECTION 1391. 103.503 (3) (a) 2. of the statutes is amended to read:

103.503 (3) (a) 2. A requirement that employees performing the work described in s. 66.0903 (4), 2013 stats., or s. 16.856 (2m), 2015 stats., on a project of public works or performing work on a public utility project submit to random, reasonable suspicion, and post–accident drug and alcohol testing and to drug and alcohol testing before commencing work on the project, except that testing of an employee before commencing work on a project is not required if the employee has been participating in a random testing program during the 90 days preceding the date on which the employee commenced work on the project.

SECTION 1395. 106.04 of the statutes is repealed.

SECTION 1396. 106.125 of the statutes is created to

106.125 Early college credit program. On behalf of the school board of a school district and on behalf of

the governing body of a participating private school, as defined in s. 118.55 (1) (c), the department of workforce development shall pay to the department of public instruction the costs of tuition for a pupil who attends an institution of higher education under the program under s. 118.55 as provided under s. 118.55 (5) (e) 2. and 3.

SECTION 1397. 106.27 (1) of the statutes is renumbered 106.27 (1) (intro.) and amended to read:

106.27 (1) Workforce training grants. (intro.) From the appropriation under s. 20.445 (1) (b), the department shall award grants to public and private organizations for the development and implementation of workforce training programs. An organization that is awarded a grant under this subsection may use the grant for the training of unemployed and underemployed workers and incumbent employees of businesses in this state. As a condition of receiving a grant under this subsection, the department may require a public or private organization to provide matching funds at a percentage to be determined by the department. Grants awarded under this subsection may include any of the following:

SECTION 1398. 106.27 (1) (a) of the statutes is created to read:

106.27 (1) (a) Grants for collaborative projects among school districts, technical colleges, and businesses to provide high school students with industry–recognized certifications in high–demand fields, as determined by the department.

SECTION 1399. 106.27 (1) (b) of the statutes is created to read:

106.27 (1) (b) 1. Grants for programs that train teachers and that train individuals to become teachers, including teachers in dual enrollment programs.

- 2. In this paragraph:
- a. "Dual enrollment program" means a program or course of study designed to provide high school students the opportunity to gain credits in both technical college and high school, including transcripted credit programs or other educational services provided by contract between a school district and a technical college.
- b. "Teacher" includes an instructor at a technical college under ch. 38.

SECTION 1400. 106.27 (1) (c) of the statutes is created to read:

106.27 (1) (c) Grants for the development of public-private partnerships designed to improve workforce retention through employee support and training.

SECTION 1401. 106.27 (1) (d) of the statutes is created to read:

106.27 (1) (d) Grants to nonprofit organizations, institutions of higher education, as defined in 20 USC 1001 (a) and (b), and employers to increase the number of students who are placed with employers for internships.

SECTION 1402. 106.27 (1) (e) of the statutes is created to read:

106.27 (1) (e) Grants to community—based organizations for public—private partnerships to create and implement a nursing training program for middle school and high school students.

SECTION 1402c. 106.27 (1) (f) of the statutes is created to read:

106.27 (1) (f) Grants to school districts to fund building modifications needed to support school districts' technical education programs.

SECTION 1403. 106.27 (1j) of the statutes is created to read:

106.27 (1j) WORKFORCE TRAINING PROGRAM; MOBILE CLASSROOMS. (a) Of the amounts appropriated under s. 20.445 (1) (b), the department shall allocate up to \$1,000,000 for grants to the department of corrections to fund the creation and operation of mobile classrooms.

- (am) Of the amounts appropriated under s. 20.445 (1) (b), the department may allocate up to \$50,000 in each fiscal year for grants to fund the upkeep and maintenance of the mobile classrooms described under par. (a).
- (b) The mobile classrooms described under par. (a) shall be used to provide job skills training to individuals in underserved areas of this state, including inmates at correctional facilities who are preparing for reentry into the workforce.
- (c) The department of corrections may use the grant money awarded under par. (a) to purchase capital equipment, such as a mobile or modular unit, that will be used as a mobile classroom, including costs to modify the equipment to make it suitable for classroom instruction, and to purchase and install any furniture, equipment, and supplies necessary or desirable for outfitting the mobile classroom for the job skills training that will be provided in the mobile classroom.

SECTION 1404. 106.27 (1r) of the statutes is amended to read:

106.27 **(1r)** STUDENT INTERNSHIP COORDINATION. From the appropriation under s. 20.445 (1) (b), the department shall provide coordination between <u>non-profit organizations and</u> institutions of higher education, as defined in 20 USC 1001 (a) and (b), and employers to increase the number of students who are placed with employers for internships.

SECTION 1405. 106.27 (2g) (a) 2. of the statutes is amended to read:

106.27 (**2g**) (a) 2. Receive and review applications for grants under subs. (1) and, (1g), and (1j) (am) and prescribe the form, nature, and extent of the information that must be contained in an application for a grant under sub. (1) or, (1g), or (1j) (am).

SECTION 1406. 106.27 (3) of the statutes is amended to read:

106.27 (3) ANNUAL REPORT. Annually, by December 31, the department shall submit a report to the governor and the cochairpersons of the joint committee on finance providing an account of the department's activities and

expenditures under this section during the preceding fiscal year and detailing the amounts allocated to and expended for each of the programs, grants, and services specified in s. 20.445 (1) (b) and (bm) for that fiscal year. The report shall include information on the number of unemployed and underemployed workers and incumbent employees who participate in training programs under sub. (1) or (1j); the number of unemployed workers who obtain gainful employment, underemployed workers who obtain new employment, and incumbent employees who receive increased compensation after participating in such a training program; and the wages earned by those workers and employees both before and after participating in such a training program. The report shall also include information on the extent to which waiting lists for enrollment in courses and programs provided by technical colleges in high-demand fields are reduced as a result of grants under sub. (1g) (a), on the number of high school pupils students who participate in certification or training programs under sub. (1) (a) or (e) or (1g) (b), on the building modifications funded under sub. (1) (f) and the effect of those building modifications on the school districts' technical education programs, and on the number of persons with disabilities who participate in employment enhancement activities under sub. (1g) (c). In addition, the report shall provide information on the number of student interns who are placed with employers as a result of the coordination activities conducted under sub. (1r) or the grants awarded under sub. (1) (d).

SECTION 1407. 106.272 of the statutes is created to read:

106.272 Teacher development program grants.

- (1) From the appropriation under s. 20.445 (1) (b), the department shall award grants to the school board of a school district or to the governing body of a private school, as defined under s. 115.001 (3d), or to a charter management organization that has partnered with an educator preparation program approved by the department of public instruction and headquartered in this state to design and implement a teacher development program.
- (2) In awarding a grant under this section, the department shall do all of the following:
- (a) Consult with the department of public instruction to confirm that the teacher development program satisfies the requirements under s. 118.196 (2).
- (b) Consider the methods by which the school board, governing body, or charter management organization and the educator preparation program under sub. (1) will make the teacher development program affordable to participating employees.
- (c) Consider whether the school board, governing body, or charter management organization has agreed to contribute matching funds towards the teacher development program.

SECTION 1407g. 106.273 (3) (a) (intro.) of the statutes is amended to read:

106.273 (3) (a) (intro.) From the appropriation under s. 20.445 (1) (b), the department shall allocate not less than \$3,000,000 \$3,500,000 in each fiscal year for incentive grants to school districts under this section. From that allocation, the department shall, subject to par. (am), annually award to a school district \$1,000 per pupil for each pupil in the school district who, in the prior school year, obtained a high school diploma or a technical education high school diploma from a school in the school district, if all of the following apply:

SECTION 1407i. 106.273 (3) (am) of the statutes is created to read:

106.273 (3) (am) The department may not make a per pupil award of \$1,000 to a school district under par. (a) if the industry-recognized certification program completed by the pupil as a condition of the award under par. (a) 2. is an information technology instructional program developed under s. 115.455.

SECTION 1407k. 106.275 of the statutes is created to read:

106.275 Technical education equipment grants.

- (1) AWARDING OF GRANTS. (a) From the appropriation under s. 20.445 (1) (b), the department may allocate up to \$500,000 in each fiscal year for technical education equipment grants to school districts under this section. From that allocation, the department may award grants in the amount of not more than \$50,000 to school districts whose grant applications are approved under sub. (2) (b).
- (b) A school district that is awarded a grant under this section shall use the grant moneys awarded for the acquisition of equipment that is used in advanced manufacturing fields in the workplace, together with any software necessary for the operation of that equipment and any instructional material necessary to train pupils in the operation of that equipment.
- (c) As a condition of receiving a grant under this section, a school district shall provide matching funds equal to 200 percent of the grant amount awarded. The match may be in the form of money, or the monetary value of equipment, contributed from private sources, the school district, or both.
- (2) Grant application process. (a) A school district that wishes to receive a grant under this section shall apply for the grant in accordance with procedures and requirements established by the department under rules promulgated under sub. (4) (b) 1. A grant application shall describe the purpose and need for the grant, the projected outcomes that the school district is seeking to achieve as a result of receiving the grant, the amount and source of the matching funds required under sub. (1) (c), and any other information that the department may require under rules promulgated under sub. (4) (b) 1.

Vetoed In Part

(b) The department, in consultation with the advisory committee created under sub. (4) (a), shall review and evaluate a grant application submitted under par. (a) in accordance with procedures and criteria established by

- the department under rules promulgated under sub. (4) (b) 2. After completing that review and evaluation, the department shall notify the school district of the department's decision on the grant application.
- (3) REPORTING REQUIREMENTS. Each school district that receives a grant under this section shall file a report with the department by September 1 of each of the first 3 fiscal years following the fiscal year in which the grant was received. The report shall describe how the grant moneys were expended, describe the outcomes achieved as a result of receiving the grant, share the best practices employed by the school district regarding the training of pupils in the use of the equipment acquired with the grant moneys, include a plan for sustainability of that training. and provide such other information as the department may require under rules promulgated under sub. (4) (b)
- (4) IMPLEMENTATION OF GRANT PROGRAM. (a) The Vetoed secretary of workforce development shall create an advisory committee under s. 15.04 (1) (c) to assist the department in reviewing and evaluating grant applications under sub. (2) (b). The committee shall consist of 5 individuals appointed by the secretary each of whom represents a different industrial sector of the economy and a different geographic region of the state.
- (b) The department shall promulgate rules to implement this section. Those rules shall include all of the following:
- 1. Rules establishing the procedures and requirements for applying for a grant under sub. (2) (a), including the information that must be submitted with a grant application.
- 2. Rules establishing the procedures and criteria for awarding a grant under sub. (2) (b).
- 3. Rules governing the reporting requirements under sub. (3), including the information that must be provided in a report submitted under sub. (3).

SECTION 1407m. 106.277 of the statutes is created to read:

- 106.277 Grants for teacher training and recruitment. (1) From the appropriation under s. 20.445 (1) (bt), the department shall award a grant to a nonprofit organization that applies on a form prepared by the department to receive a grant under this section if all of the following apply:
- (a) The organization is described under section 501 (c) (3) or (4) of the Internal Revenue Code and exempt from taxation under section 501 (a) of the Internal Revenue Code.
- (b) The organization operates a program to recruit and prepare individuals to teach in public or private schools located in low-income or urban school districts in this state.
- (c) The organization submits an application no later than January 15 in the year for which the organization seeks a grant under this section and includes with its

In Part

application a description of the organization and its program and the manner in which grant funding will benefit the organization or has benefited the organization in the past.

- (2) The department shall establish a process for evaluating and assigning a score to each organization eligible to receive a grant under sub. (1). If the amount appropriated under s. 20.445 (1) (bt) is insufficient to make the payments required under sub. (1), the department shall give preference in evaluating grants under this section to a nonprofit organization for each of the following:
- (a) The program trains future teachers who are enrolled in an accredited college or university in this state concurrent with the training.
- (b) The program focuses on future teachers who plan to teach in public or private schools in this state as a profession.
- (c) The program provides continuing education and professional development.
- (d) The program attempts to place a majority of its total participants in public or private schools located in low–income or urban school districts in this state.
- (3) When awarding a grant under this section, the department may not consider the religious affiliation, if any, of the nonprofit organization or whether the organization has received funding from the state in the past.
- (4) By February 15 of each year, the department shall make its determination regarding successful applicants and shall notify those applicants that will receive a grant under this section.

SECTION 1444. 109.09 (1) of the statutes is amended to read:

109.09 (1) The department shall investigate and attempt equitably to adjust controversies between employers and employees as to alleged wage claims. The department may receive and investigate any wage claim that is filed with the department, or received by the department under s. 109.10 (4), no later than 2 years after the date the wages are due. The department may, after receiving a wage claim, investigate any wages due from the employer against whom the claim is filed to any employee during the period commencing 2 years before the date the claim is filed. The department shall enforce this chapter and s. 66.0903, 2013 stats., s. 103.49, 2013 stats., and s. 229.8275, 2013 stats., and s. 16.856, 2015 stats., and ss. 16.856, 103.02, 103.82, and 104.12. In pursuance of this duty, the department may sue the employer on behalf of the employee to collect any wage claim or wage deficiency and ss. 109.03 (6) and 109.11 (2) and (3) shall apply to such actions. Except for actions under s. 109.10, the department may refer such an action to the district attorney of the county in which the violation occurs for prosecution and collection and the district attorney shall commence an action in the circuit court having appropriate jurisdiction. Any number of wage claims or wage deficiencies against the same employer

may be joined in a single proceeding, but the court may order separate trials or hearings. In actions that are referred to a district attorney under this subsection, any taxable costs recovered by the district attorney shall be paid into the general fund of the county in which the violation occurs and used by that county to meet its financial responsibility under s. 978.13 (2) (b) for the operation of the office of the district attorney who prosecuted the action.

SECTION 1446. 111.07 (5) of the statutes is amended to read:

111.07 (5) The commission may make findings and orders or may authorize a commissioner or an examiner to make findings and orders. Any party in interest who is dissatisfied with the findings or order of a commissioner or examiner may file a written petition with the commission as a body to review the findings or order. If no petition is filed within 20 days from the date that a copy of the findings or order of the commissioner or examiner was mailed to the last-known address of the parties in interest, such findings or order shall be considered the findings or order of the commission as a body unless set aside, reversed, or modified by such commissioner the commission or examiner within such time. If the findings or order are set aside by the commissioner commission or examiner the status shall be the same as prior to the findings or order set aside. If the findings or order are reversed or modified by the commissioner or examiner, the time for filing petition with the commission shall run from the time that notice of such reversal or modification is mailed to the last-known address of the parties in interest. Within 45 days after the filing of such petition with the commission, the commission shall either affirm, reverse, set aside, or modify such findings or order, in whole or in part, or direct the taking of additional testimony. Such action shall be based on a review of the evidence submitted. If the commission is satisfied that a party in interest has been prejudiced because of exceptional delay in the receipt of a copy of any findings or order, it may extend the time another 20 days for filing a petition with the commission.

SECTION 1447. 111.07 (6) of the statutes is amended to read:

111.07 (6) The commission shall have the power to remove or transfer the proceedings pending before —a commissioner or an examiner. It may also, on its own motion, set aside, modify, or change any order, findings, or award, whether made by an individual commissioner, an examiner, or by the commission as a body, at any time within 20 days from the date thereof if it shall discover any mistake therein, or upon the grounds of newly discovered evidence.

SECTION 1448. 111.14 of the statutes is amended to read:

111.14 Penalty. Any person who shall willfully assault, resist, prevent, impede, or interfere with any

member of the commission or any of its agents or agencies in the performance of duties pursuant to this subchapter shall be punished by a fine of not more than \$500 or by imprisonment in the county jail for not more than one year, or both.

SECTION 1452. 111.322 (2m) (c) of the statutes is repealed.

SECTION 1453. 111.322 (2m) (d) of the statutes is amended to read:

111.322 (2m) (d) The individual's employer believes that the individual engaged or may engage in any activity described in pars. (a) to (e) (bm).

SECTION 1454. 111.335 (1) (cx) of the statutes is amended to read:

111.335 (1) (cx) Notwithstanding s. 111.322, it is not employment discrimination because of conviction record to refuse to employ or license, or to bar or terminate from employment or licensure, any individual who has been convicted of any offense under s. 38.50 440.52 (13) (c).

Vetoed In Part **SECTION 1460m.** 114.136 (2) (b) 3. of the statutes is created to read:

114.136 (2) (b) 3. An ordinance under this section may not prohibit the use of a physical barrier in lieu of compliance with a 48–hour drainage requirement for a storm detention pond that is located in a residential subdivision underlain by natural clay soil.

SECTION 1461p. 115.28 (7) (a) of the statutes is amended to read:

115.28 (7) (a) License all teachers for the public schools of the state; make rules establishing standards of attainment and procedures for the examination and licensing of teachers within the limits prescribed in ss. 118.19 (2) and (3), 118.191, 118.1915, 118.192, 118.193, 118.194, and 118.195, and 118.197; prescribe by rule standards, requirements, and procedures for the approval of teacher preparatory programs leading to licensure, including a requirement that, beginning on July 1, 2012, and annually thereafter, each teacher preparatory program located in this state shall submit to the department a list of individuals who have completed the program and who have been recommended by the program for licensure under this subsection, together with each individual's date of program completion, from each term or semester of the program's most recently completed academic year; file in the state superintendent's office all papers relating to state teachers' licenses; and register each such license.

SECTION 1462. 115.28 (7) (g) of the statutes is amended to read:

115.28 (7) (g) Notwithstanding s. 118.19 (3), (4m), (6) to (9), and (12) to (14), grant a charter school teaching license to any person who has a bachelor's degree and demonstrates, based upon criteria established by the department, that the person is proficient in the subject or subjects that he or she intends to teach. The license

authorizes the person to teach that subject or those subjects in a charter school. The license is valid for 3 years and is renewable for 3—year periods.

SECTION 1463. 115.28 (7) (gm) of the statutes is amended to read:

115.28 (7) (gm) Notwithstanding s. 118.19 (8), (9), and (11), grant an initial charter school principal license to any person who is licensed, or otherwise credentialed, to be a school principal in another state if the person holds the license or other credential in good standing, has completed at least 3 years of full–time classroom teaching, and is eligible for licensure under s. 118.19 (4) and (10). The license authorizes the person to be a principal of a charter school. The license is valid for 5 years and is renewable for 5—year periods.

SECTION 1463f. 115.28 (10o) (c) of the statutes is created to read:

115.28 (**10o**) (c) Beginning in November of the 2017–18 school year, annually determine whether any unified school district qualifies as an eligible unified school district, as defined in s. 115.999 (2m) (a).

SECTION 1463h. 115.28 (10o) (d) of the statutes is created to read:

115.28 (100) (d) By November 30, 2017, and each November 30 thereafter, notify the clerk of each village located in an eligible unified school district, as defined in s. 115.999 (2m) (a), and the school district administrator of the eligible unified school district, as defined in s. 115.999 (2m) (a), that the unified school district qualifies as an eligible unified school district, as defined in s. 115.999 (2m) (a).

SECTION 1464. 115.28 (45) of the statutes is created to read:

115.28 **(45)** Grants for Bullying Prevention. From the appropriation under s. 20.255 (3) (eb), award grants to a nonprofit organization, as defined in s. 108.02 (19), to provide training and an online bullying prevention curriculum for pupils in grades kindergarten to 8.

SECTION 1465. 115.28 (54m) of the statutes is amended to read:

115.28 **(54m)** NOTICE OF EDUCATIONAL OPTIONS. Include on the home page of the department's Internet site a link to information about all of the educational options available to children in the state who are at least 3 years old but not yet 18 years old, including public schools, private schools participating in a parental choice program, charter schools, virtual schools, full–time or part–time open enrollment, youth options, course options in a nonresident school district, the early college credit program, and options for pupils enrolled in a home–based private educational program.

SECTION 1467b. 115.28 (60) of the statutes is repealed.

SECTION 1467d. 115.28 (63) of the statutes, as created by 2017 Wisconsin Act 31, is renumbered 115.28 (63) (intro.) and amended to read:

115.28 **(63)** MENTAL HEALTH TRAINING PROGRAM. (intro.) Establish a mental health training support program under which the department provides training on the screening, brief intervention, and referral to treatment program, an all of the following evidence—based strategy strategies related to addressing mental health issues in schools, to school district staff and instructional staff of charter schools under s. 118.40 (2r) or (2x).

SECTION 1467h. 115.28 (63) (a) of the statutes is created to read:

115.28 (63) (a) The screening, brief intervention, and referral to treatment program.

SECTION 1467p. 115.28 (63) (b) of the statutes is created to read:

115.28 (63) (b) Trauma sensitive schools.

SECTION 1467t. 115.28 (63) (c) of the statutes is created to read:

115.28 (63) (c) Youth mental health first aid.

SECTION 1468g. 115.28 (64) of the statutes is created to read:

115.28 **(64)** *College Possible*. Annually, from the appropriation under s. 20.255 (3) (fc), award a grant to College Possible, Inc., to work with pupils and college students in Milwaukee. The grant amount shall be determined on a matching basis, under which College Possible, Inc., provides matching funds in an amount equal to 20 percent of the grant amount.

SECTION 1468m. 115.28 (65) of the statutes is created to read:

115.28 **(65)** WISCONSIN READING CORPS. In the 2017–18 and 2018–19 school years, distribute the amounts appropriated under s. 20.255 (3) (fr) to Wisconsin Reading Corps to provide one–on–one tutoring if Wisconsin Reading Corps provides matching funds of \$250,000 in each school year.

SECTION 1470. 115.364 of the statutes is created to read:

115.364 Aid for school mental health programs. (1) In this section:

- (a) "Eligible independent charter school" is a school under contract with one of the entities under s. 118.40 (2r) (b) 1. or with the director under s. 118.40 (2x) that increased the amount it expended in the preceding school year to employ, hire, or retain social workers over the amount it expended in the school year immediately preceding the preceding school year to employ, hire, or retain social workers.
- (am) "Eligible private school" means a private school participating in a parental choice program under s. 118.60 or 119.23 that increased the amount it expended in the preceding school year to employ, hire, or retain social workers over the amount it expended in the school year immediately preceding the preceding school year to employ, hire, or retain social workers.
- (b) "Eligible school district" is a school district that increased the amount it expended in the preceding school

- year to employ, hire, or retain social workers over the amount it expended in the school year immediately preceding the preceding school year to employ, hire, or retain social workers.
- (2) (a) Beginning in the 2018–19 school year and annually thereafter, the state superintendent shall do all of the following:
- 1. Subject to par. (b), from the appropriation under s. 20.255 (2) (da), pay to an eligible school district an amount equal to 50 percent of the amount by which the school district increased its expenditures in the preceding school year to employ, hire, or retain social workers over the amount it expended in the school year immediately preceding the preceding school year to employ, hire, or retain social workers.
- 2. Subject to par. (b), from the appropriation under s. 20.255 (2) (da), pay to an eligible independent charter school an amount equal to 50 percent of the amount by which the independent charter school increased its expenditures in the preceding school year to employ, hire, or retain social workers over the amount it expended in the school year immediately preceding the preceding school year to employ, hire, or retain social workers.
- 3. Subject to par. (b), from the appropriation under s. 20.255 (2) (da), pay to an eligible private school an amount equal to 50 percent of the amount by which the private school increased it expenditures in the preceding school year to employ, hire, or retain social workers over the amount it expended in the school year immediately preceding the preceding school year to employ, hire, or retain social workers.
- (b) 1. If the appropriation under s. 20.255 (2) (da) in any fiscal year is insufficient to pay the full amount of aid under par. (a), the state superintendent shall prorate state aid payments among the school districts, private schools, and independent charter schools eligible for the aid.
- 2. a. Subject to subd. 2. b., if, after making the payments required under par. (a), moneys remain in the appropriation account under s. 20.255 (2) (da), the state superintendent shall reimburse eligible school districts, private schools, and independent charter schools for an amount equal to expenditures made by the school district, private school, or independent charter school in the preceding school year to employ, hire, or retain social workers less the amount of increased expenditures for which the school district, private school, or independent charter school was reimbursed under par. (a).
- b. If the appropriation under s. 20.255 (2) (da) in any fiscal year is insufficient to pay the full amount of aid under subd. 2. a., the state superintendent shall prorate state aid payments among the school districts, private schools, and independent charter schools eligible for the aid
- (3) The department shall promulgate rules to implement and administer this section.

SECTION 1470g. 115.367 of the statutes is created to read:

- 115.367 School-based mental health services grants. (1) Grant Program. The department shall establish and administer a competitive program to award grants to school boards and operators of charter schools under s. 118.40 (2r) or (2x) for the purpose of collaborating with community mental health agencies to provide mental health services to pupils. School boards and operators of charter schools under s. 118.40 (2r) and (2x) may apply for a grant under this section individually or as a consortium of school boards, charter schools, or both. For purposes of this subsection, a "consortium of school boards" includes a cooperative educational service agency.
- (2) ELIGIBILITY CRITERIA. The department shall establish by rule the criteria the department will use to award grants under this section. The department shall include all of the following in the criteria:
- (a) That the applicant require providers and contractors who participate in its school–based mental health services program to bill the Medical Assistance program under subch. IV of ch. 49 and health insurance, as applicable, for any goods and services provided under the program.
- (b) That the applicant has sought or will seek out community funding or foundation grants to cover at least some of the expenses of the program that are not paid by the Medical Assistance program under subch. IV of ch. 49 or health insurance.
- (c) Additional application criteria, which may include that the proposed school mental health services program includes collaboration with counties, providers, or community groups; considers the needs of pupils and families; and includes a referral or intake process, a continuum of therapeutic services, consultation with school staff, and access to services regardless of income.
- (3) AWARDS. The department shall award grants under this section beginning in the 2018–19 school year. From the appropriation under s. 20.255 (2) (dt), the department shall award at least \$3,250,000 in grants under this section each school year.
- (4) RULES. The department shall promulgate rules to implement and administer this section.

SECTION 1471. 115.38 (1) (d) of the statutes is amended to read:

115.38 (1) (d) The number and percentage of resident pupils attending a course at an educational institution in a nonresident school district under s. 118.52, the number of nonresident pupils attending a course in the school district under s. 118.52, and the courses taken by those pupils.

SECTION 1472. 115.385 (1) (d) of the statutes is created to read:

- 115.385 (1) (d) For a school district and for each high school in that school district, all of the following information:
- 1. The number and percentage of pupils participating in the program under s. 118.55.
- 2. The number and percentage of pupils participating in a youth apprenticeship under s. 106.13.
- 3. The number of community service hours provided by pupils.
- 4. The number of advanced placement courses offered to and the number of advanced placement credits earned by pupils.
- 5. The number of pupils earning industry–recognized credentials through a technical education program established by a school board as described in s. 118.33 (1) (g) 1. c.

SECTION 1473. 115.385 (4) of the statutes is amended to read:

115.385 (4) Annually, each public school, including a charter school, and each private school participating in a parental choice program under s. 118.60 or 119.23 shall provide a copy of the school's accountability report to the parent or guardian of each pupil enrolled in or attending the school. Each school shall simultaneously provide to the parent or guardian of each pupil enrolled in the school a list of the educational options available to children who reside in the pupil's resident school district, including public schools, private schools participating in a parental choice program, charter schools, virtual schools, full—time or part—time open enrollment, youth options, course options in a nonresident school district, the early college credit program, and options for pupils enrolled in a home—based private educational program.

SECTION 1473k. 115.387 of the statutes is created to read:

115.387 School performance improvement grants. (1) In this section:

- (a) "Accountability report" means the school and school district accountability report published under s. 115.385.
- (b) "Eligible school" means any of the following that is located within the geographical boundaries of an eligible school district:
- 1. A public school that is under the control of a school board, including a charter school under contract with the school board.
- 2. A charter school established under s. 118.40 (2r) or (2x).
- 3. A private school participating in the program under s. 118.60 or 119.23.
- (c) "Eligible school district" means any of the following:

Vetoed In Part Vetoed In Part

- 1. A school district that was placed in the lowest performance category on the accountability report in the previous school year.
 - 2. A 1st class city school district.
- (d) 1. For purposes of a public school that is under the control of a school board, "number of pupils enrolled" has the meaning given in s. 115.437 (1).
- 2. For purposes of a charter school established under s. 118.40 (2r) or (2x), "number of pupils enrolled" means the number of pupils attending the charter school.
- 3. For purposes of a private school participating in the program under s. 118.60 or 119.23, "number of pupils enrolled" means the number of pupils attending the private school under the program under s. 118.60 or 119.23.
- (2) Beginning in the 2018–19 school year and in each school year thereafter, subject to sub. (4), the department shall award a grant in an amount determined under sub. (3) to each eligible school that satisfies all of the follow-
- (a) The eligible school develops a written school improvement plan to improve pupil performance in math and reading.
- (b) If the eligible school received a grant under this section in the previous school year, the numerical score that was the basis for the eligible school's performance category on the accountability report in the previous school year was greater than the numerical score that was the basis for the eligible school's performance category on the accountability report in the school year preceding the previous school year.
- (3) The department shall calculate the amount of an eligible school's grant under sub. (2) as follows:
- (a) Determine the total number of pupils enrolled in all eligible schools that are entitled to a grant under this section in the current school year.
- (b) Divide the amount appropriated under s. 20.255 (2) (dg) for the current school year by the number of pupils determined under par. (a).
- (c) Multiply the quotient determined in par. (b) by the number of pupils enrolled in the eligible school in the current school year.
- (4) In each school year, the department may not award a grant under sub. (2) before the department of administration approves the per pupil amount calculated under sub. (3) (b) for that school year.
- (5) A school board shall distribute funds it receives under this section to the school administrator of the eligible school that earned the award under this section.

SECTION 1474. 115.42 (3) of the statutes is repealed. **SECTION 1475.** 115.42 (4) (c) of the statutes is repealed.

SECTION 1475m. 115.423 of the statutes is created to read:

115.423 Rural school teacher talent pilot program. From the appropriation under s. 20.255 (1) (eg), the department shall award grants to cooperative educational service agencies to coordinate with universities and colleges to place undergraduate college students in rural school districts as student teachers and in practicums and internships. Grant moneys may be used to expand an existing student teacher, practicum, or internship program or to create a new program, but may not be used to maintain an existing program. Grant moneys may be used to cover the cooperative educational service agencies' costs to coordinate the program and to promileage reimbursement and stipends to participating undergraduates.

SECTION 1475p. 115.434 of the statutes is created to **Vetoed** read:

In Part

- 115.434 Shared services pilot program. (1) (a) Two or more school boards may enter into an agreement to apply for aid under this section to share the services of one or more administrative personnel.
- (b) To qualify for aid under this section, each applicant school board shall pass a resolution approving participation in the shared services program under this sec-
- (c) The school boards that have entered into an agreement to apply for aid under this section shall jointly submit a shared services plan to the department by July 1, 2018. The participating school boards shall include all of the following in the plan:
- 1. The position or positions the districts intend to share.
- 2. The position or positions that will be eliminated in each district.
- 3. The salary and fringe benefit costs of the positions described under subds. 1. and 2.
- 4. Information demonstrating that the shared services plan will result in a net reduction in filled administrative positions between the participating school districts.
- (d) A school board may enter into an agreement with a unit of government other than a school district to share administrative personnel under a shared services plan submitted under par. (c), but the unit of government other than the school district is not eligible for aid under this
- (e) There is no limit on the number of positions that participating school boards or a participating school board and a participating unit of government may propose to share under a shared services plan.
- (2) The department shall review and approve applications submitted under sub. (1) in the order in which the applications are received and shall approve applications until all moneys appropriated under s. 20.255 (2) (bt) have been encumbered.
- (3) (a) From the appropriation under s. 20.255 (2) (bt), the department shall, subject to sub. (4), make the following payment to each school district that jointly submitted an application under sub. (1) and whose shared services plan was approved by the department under sub. (2):

In Part

Vetoed In Part

- 1. In the first 3 school years of a shared services plan approved under sub. (2):
 - a. For a district administrator, \$40,000.
- b. For a human resources director, information technology coordinator, or business manager, \$22,500.
- c. For any non-faculty administrative position other than a position identified in subd. 1. a. or b. and other than a principal or assistant principal, \$17,500.
- 2. In the 4th school year of a shared services plan approved under sub. (2), subject to subd. 4., 50 percent of the amount received under subd. 1.
- 3. In the 5th school year of a shared services plan approved under sub. (2), subject to subd. 4., no payment.
- 4. If, before the beginning of the 4th school year of a shared services plan, each school district that is participating in the shared services plan enters into a whole grade sharing agreement under s. 118.50, for the 4th and 5th school years of the shared services plan, 100 percent of the amount under subd. 1.
- (b) The department shall make its first payments under this subsection by January 1, 2019.
- (4) (a) If one of the school boards that jointly submitted a shared services plan approved under sub. (2) hires an additional individual to staff a position covered under the shared services plan without eliminating the individual who is serving in that same position under the shared services plan, the department shall withdraw all school districts that were party to the shared services plan from the program under this section. A school district that is withdrawn under this paragraph may not receive any additional aid under sub. (3).
- (b) If a school district employee holds more than one position in each district and each position is covered under a shared services plan approved under sub. (2), each school district may receive aid under sub. (3) for only one of the positions covered under the shared services plan. In the event the school districts whose shared employee holds more than one position under the shared services plan would be eligible for more than one category of aid payment under sub. (3) (a) 1. a. to c., the department shall pay the higher aid amount to each school district for that shared school district employee.
- (5) No later than February 1, 2019, the department shall submit to the joint committee on finance a report containing all of the following information about the program under this section:
- (a) The number of school boards that jointly submitted an application and shared services plan to participate in the program.
- (b) The number of shared services plans approved by the department and the name of each school district participating in each such plan.
- (c) The number of administrative personnel positions to be shared under a shared services plan under this section.

- (d) The amount of funding encumbered under this Vetoed section to date.
- (e) The total anticipated reduction in salary and fringe benefit costs by each school district participating in a shared services plan and by all school districts participating in a shared services plan.

SECTION 1477d. 115.436 (3) (a) of the statutes is amended to read:

115.436 (3) (a) Beginning in the 2009-10 school year, from the appropriation under s. 20.255 (2) (ae) and subject to pars. par. (b) and (c), the department shall pay to each school district eligible for sparsity aid \$300 multiplied by the membership in the previous school year.

SECTION 1480p. 115.436 (3) (ap) of the statutes is created to read:

115.436 (3) (ap) In the school year in which a school district consolidation takes effect under s. 117.08 or 117.09 and in each of the subsequent 4 school years, notwithstanding sub. (2), the department shall pay the consolidated school district sparsity aid in an amount that is not less than 50 percent of the aggregate amount of sparsity aid received by the consolidating school districts in the school year prior to the school year in which the consolidation takes effect. This paragraph does not apply to a school district created by a consolidation under s. 117.08 or 117.09 that takes effect before July 1, 2019.

SECTION 1480r. 115.436 (3) (b) of the statutes is amended to read:

115.436 (3) (b) If the appropriation under s. 20.255 (2) (ae) in any fiscal year is insufficient to pay the full amount under par. pars. (a), (am), and (ap), the department shall prorate the payments among the eligible school districts entitled to aid under this subsection.

SECTION 1481g. 115.436 (3) (c) 1. of the statutes is renumbered 115.436 (3) (am) and amended to read:

115.436 (3) (am) Beginning in the 2015–16 2017–18 school year, in any fiscal year in which the department has paid the full amount due to eligible school districts under par. (a) and an unencumbered balance remains in from the appropriation under s. 20.255 (2) (ae), the department shall, subject to subd. 2. par. (b), pay to each school district that received aid under this section in the previous school year but does not satisfy the requirement under sub. (2) (a) in the current school year \$300 multiplied by the membership used to determine the payment 50 percent of the amount received by the school district under par. (a) in the previous school year.

SECTION 1481i. 115.436 (3) (c) 2. of the statutes is repealed.

SECTION 1481m. 115.437 (2) (a) of the statutes is amended to read:

115.437 (2) (a) Except as provided in par. (b), annually on the 4th Monday of March, the department shall pay to each school district an amount equal to the average of the number of pupils enrolled in the school district in the current and 2 preceding school years multiplied by \$75 in the 2013-14 school year, by \$150 in the 2014-15 and 2015-16 school years, and by \$250 in the 2016-17 school year, by \$450 in the 2017-18 school year, by \$654 in the 2018-19 school year, and by \$630 in each school year thereafter. The department shall make the payments from the appropriation under s. 20.255 (2) (aq).

SECTION 1482f. 115.438 of the statutes is created to read:

115.438 Personal electronic computing devices; grant program. (1) In this section:

- (a) "Membership" has the meaning given in s. 121.004 (5).
- (b) "Personal electronic computing device" means an electronic computing device that satisfies all of the following criteria:
- 1. The electronic computing device is a mobile device.
- 2. The electronic computing device is assignable to an individual pupil to be used solely by that pupil.
- 3. The electronic computing device may be used to access the Internet.
- (2) Any of the following may apply for a grant under this section:
 - (a) A school board.
- (b) An operator of a charter school under s. 118.40 (2r) or (2x).
 - (c) The governing body of a private school.
 - (d) A tribal school.
- (3) A recipient of a grant under this section may use the grant only for the following purposes:
- To purchase personal electronic computing devices.
- (b) To purchase software for personal electronic computing devices.
- (c) To purchase curriculum, including any related educational content or materials, a portion or all of which includes content that may be accessed on a personal electronic computing device.
- (d) To train professional staff on how to effectively incorporate personal electronic computing devices into a classroom and into the high school curriculum.
- (4) (a) Beginning in the 2018–19 school year and ending in the 2022-23 school year, from the appropriation under s. 20.255 (2) (aw) and subject to pars. (b) and (c), the department shall pay to each school district, operator of a charter school under s. 118.40 (2r) or (2x), governing body of a private school, and tribal school that applies for a grant under this section an amount calculated by multiplying \$125 by one of the following:
- 1. For a school district, the number of 9th grade pupils included in the school district's membership in the previous school year.
- 2. For an operator of a charter school under s. 118.40 (2r) or (2x), the number of 9th grade pupils attending the charter school in the current school year.

- 3. For a governing body of a private school, the number of 9th grade pupils attending the private school in the current school year.
- 4. For a tribal school, the number of 9th grade pupils attending the tribal school in the current school year.
- (b) As a condition of receiving a grant under par. (a), an applicant shall demonstrate to the satisfaction of the department that the applicant will provide matching funds in an amount equal to the grant award.
- (c) If the appropriation under s. 20.255 (2) (aw) in any fiscal year is insufficient to pay the full amount under par. (a), the department shall prorate the payments among the eligible applicants.

SECTION 1482j. 115.447 of the statutes is created to read:

115.447 Summer school programs; grants. (1) In this section, "eligible school district" means any of the following:

(a) A school district that was placed in the lowest per- Vetoed formance category on the accountability report published In Part under s. 115.385 in the previous school year.

(b) A 1st class city school district.

- (2) Beginning in the 2018–19 school year and in each In Part year thereafter, from the appropriation under s. 20.255 (2) (dj), the department shall award grants, on a competitive basis, to eligible school districts to do any of the fol- In Part lowing to increase pupil attendance, improve low-performing schools, improve academic achievement, or In Part expose pupils to innovative learning activities:
 - (a) Develop a summer school program.
 - (b) Redesign a summer school program.
 - (c) Implement a summer school program.
- (3) The department shall promulgate rules to implement this section.

SECTION 1482m. 115.45 (2) (a) of the statutes is amended to read:

115.45 (2) (a) No later than April 1, 2016 Annually, the department shall notify school boards, operators of charter schools under s. 118.40 (2r) and (2x), governing bodies of private schools, and administrators of homebased private educational programs that applications for grants under this section will be accepted from eligible teams through a date set forth in the notice. As a condition of receiving a grant under this section, an applicant eligible team shall demonstrate to the satisfaction of the department that the applicant eligible team will provide matching funds in an amount equal to the amount awarded under this section.

SECTION 1482p. 115.455 of the statutes is created to read:

115.455 Grant for information technology education. (1) (a) The department shall develop a competitive request-for-proposal process for the award of a grant to an entity to provide information technology education opportunities to public school pupils in grades 6 to 12, Vetoed In Part

Vetoed

Vetoed Vetoed technical college district students, and patrons of public libraries.

- (b) The department shall accept applications from entities responding to the request–for–proposal under par. (a) and shall, in the 2017–18 and 2018–19 school years, from the appropriation under s. 20.255 (2) (eb), award a grant to an entity that, subject to sub. (3), satisfies the requirements under sub. (2).
- (2) To be eligible for a grant under this section, the entity shall demonstrate that it has successfully offered an information technology instructional program in schools in this state and shall develop an instructional program that includes all of the following components, and shall ensure that the instructional program will be operated in 225 sites, including 16 public libraries:
- (a) A research-based curriculum emphasizing the development of information technology skills.
 - (b) Online access to the curriculum.
- (c) Instructional software for use in the classroom and at a pupil's home.
- (d) A curriculum that aligns with the coding and other techniques included on the computer science Advanced Placement examination. In developing the curriculum required under this paragraph, the entity shall ensure that a pupil who successfully completes the curriculum and passes the Advanced Placement examination is eligible for certification.
- (e) Certifications of skills and competencies in a broad range of information technology-related skill areas, including applications used most often in businesses.
- (f) Professional development and co-teaching for teachers and administrators, including teachers providing instruction in the information technology program.
- (g) Deployment and program support, including integration of the information technology instructional program with curriculum standards.
- (h) Opportunities for pupils completing the information technology program to earn college credit.
- (3) In awarding the grant under sub. (1), the department shall give preference to an entity that demonstrates that it has successfully provided high-quality information technology instructional programming and educational opportunities to pupils enrolled in or attending schools in this state.

SECTION 1483. 115.745 (1) of the statutes is amended to read:

115.745 (1) A school board or, a cooperative educational service agency, or an agency determined by the state superintendent to be eligible for designation under 42 USC 9836 as a head start agency, in conjunction with a tribal education authority, may apply to the department for a grant for the purpose of supporting innovative, effective instruction in one or more American Indian languages.

SECTION 1483m. 115.77 (1) of the statutes is amended to read:

115.77 (1) In sub. (1m) (a) to (d), except as provided in s. 118.51 (12) (b), if a child with a disability is attending a public school in a nonresident school district under s. 118.50, 118.51, or 121.84 (1) (a) or (4), "local educational agency" means the school district that the child is attending.

SECTION 1485m. 115.79 (1) (b) of the statutes is amended to read:

115.79 (1) (b) An educational placement is provided to implement a child's individualized education program. Except as provided in s. 118.51 (12) (b), if a child with a disability is attending a public school in a nonresident school district under s. 118.50, 118.51, or 121.84 (1) (a) or (4), the school board of the school district that the child is attending shall provide an educational placement for the child and shall pay tuition charges instead of the school district in which the child resides if required by the placement.

SECTION 1486ap. 115.7915 (1) (a) of the statutes is created to read:

115.7915 (1) (a) "Eligible school" means a private school located in this state.

SECTION 1486b. 115.7915 (2) (intro.) of the statutes is amended to read:

115.7915 (2) SCHOLARSHIP REQUIREMENTS. (intro.) Beginning in the 2016–17 school year, the department shall provide to a child with a disability a scholarship under sub. (4m) (a) to attend -a private an eligible school if all of the following apply:

SECTION 1486c. 115.7915 (2) (a) of the statutes is repealed.

SECTION 1486d. 115.7915 (2) (b) of the statutes is amended to read:

115.7915 (2) (b) The governing body of the private eligible school notified the department of its intent to participate in the program under this section.

SECTION 1486e. 115.7915 (2) (c) of the statutes, as affected by 2017 Wisconsin Act 36, is amended to read:

115.7915 (2) (c) The <u>eligible</u> school has been approved as a private school by the state superintendent under s. 118.165 (2) or is accredited by AdvancED, Wisconsin Religious and Independent Schools Accreditation, the Independent Schools Association of the Central States, Wisconsin Evangelical Lutheran Synod School Accreditation, Wisconsin Association of Christian Schools, National Lutheran School Accreditation, Christian Schools International, Association of Christian Schools International, the diocese or archdiocese within which the <u>private eligible</u> school is located, or any other organization recognized by the National Council for Private School Accreditation, as of the August 1 preceding the school term for which the scholarship is awarded.

SECTION 1486em. 115.7915 (2) (e) of the statutes is repealed.

SECTION 1487g. 115.7915 (2) (f) of the statutes, as affected by 2017 Wisconsin Act 36, is amended to read:

115.7915 (2) (f) The child's parent or guardian on behalf of the child, or, for a child with a disability who has reached the age of 18 and has not been adjudicated incompetent, the child, submitted an application for a scholarship under this section on a form prepared by the department that includes the document developed by the department under sub. (4) to the eligible school that the child will attend. A child's parent or guardian or a child with a disability who has reached the age of 18 may apply for a scholarship at any time during a school year and, subject to sub. (3) (b), a child may begin attending a private an eligible school under this section at any time during the school year.

SECTION 1487m. 115.7915 (2) (g) of the statutes is amended to read:

115.7915 (2) (g) The private eligible school has accepted the child's application to attend the private eligible school under a scholarship awarded under this section.

SECTION 1488g. 115.7915 (2) (h) of the statutes, as affected by 2017 Wisconsin Act 36, is amended to read:

115.7915 (2) (h) The child's parent or guardian consents to make the child available for a reevaluation within 60 days following a request for a reevaluation under this paragraph. The individualized education program team appointed for the child by the resident school district shall conduct the reevaluation, except that, if the child is attending -a private an eligible school under this section in a nonresident school district and the parent or guardian of the child provides written consent, an individualized education program team appointed for the child by that nonresident school district may conduct the reevaluation. Upon the request of the school board of the child's resident school district, the individualized education program team shall conduct the reevaluation required under this paragraph in the manner described under s. 115.782 (4) (a) 2. no more frequently than once every 3 years, determined from the date of the most recent evaluation or reevaluation conducted for the child under s. 115.782 or, for a child whose most recent evaluation or reevaluation was conducted more than 3 years before the child began attending -a private an eligible school under this section, the date the child began attending a private an eligible school under this section.

SECTION 1488m. 115.7915 (3) (a) of the statutes is amended to read:

115.7915 (3) (a) The governing body of <u>a private an eligible</u> school that intends to participate in the program under this section shall notify the department of its intent. The governing body of the <u>private eligible</u> school shall include in the notice under this paragraph the number of

spaces the <u>private eligible</u> school has available for children receiving a scholarship under this section.

SECTION 1488r. 115.7915 (3) (bm) of the statutes, as affected by 2017 Wisconsin Act 36, is amended to read:

115.7915 (3) (bm) Upon receipt of an application for a scholarship under sub. (2) (f), the governing body of the private eligible school shall determine whether the application satisfies the requirements under sub. (2), other than the requirement under sub. (2) (d), and shall request verification from the local education agency that developed the child's individualized education program or services plan that the child has an individualized education program or services plan in place that meets the requirement in sub. (2) (d). The governing body of the private eligible school shall also notify the child's resident school board that, pending verification that the requirements of sub. (2) have been satisfied, the child will be awarded a scholarship under this section. The local education agency shall, within 5 business days of receiving a request under this paragraph, provide the governing body of the private eligible school with a copy of the child's individualized education program or services plan.

SECTION 1492b. 115.7915 (4c) of the statutes is created to read:

115.7915 (**4c**) Private school; Statement of ACTUAL COSTS. Beginning in the 2018–19 school year, at the end of a school year in which a private school receives a scholarship under sub. (4m) on behalf of a child with a disability, the private school may submit to the department a financial statement and supporting documentation that shows the actual costs that the private school incurred to comply with sub. (6) (h) 1. for that child during that school year. The department shall provide the resident school board a copy of any financial statement and documentation it receives under this subsection.

SECTION 1492d. 115.7915 (4m) (a) 2. of the statutes is renumbered 15.7915 (4m) (a) 2. a. and amended to read:

15.7915 (**4m**) (a) 2. a. Beginning in In the 2017–18 school year, the sum of the scholarship amount under this paragraph for the previous school year; the amount of the per pupil revenue limit adjustment under s. 121.91 (2m) for the current school year, if positive; and the change in the amount of statewide categorical aid per pupil between the previous school year and the current school year, as determined under s. 118.40 (2r) (e) 2p., if positive.

SECTION 1492e. 115.7915 (4m) (a) 2. b. of the statutes is created to read:

115.7915 (4m) (a) 2. b. Beginning in the 2018–19 school year and subject to subd. 3., the sum of the scholarship amount under this subdivision for the previous school year; the amount of the per pupil revenue limit adjustment under s. 121.91 (2m) for the current school year, if positive; and the change in the amount of state-

wide categorical aid per pupil between the previous school year and the current school year, as determined under s. 118.40 (2r) (e) 2p., if positive.

SECTION 1492f. 115.7915 (4m) (a) 3. of the statutes is created to read:

115.7915 (**4m**) (a) 3. Beginning in the 2019–20 school year, if a private school submitted a financial statement for a child with a disability under sub. (4c) in the previous school year, the amount shown on the financial statement for that child for the previous school year, up to 150 percent of the amount calculated under subd. 2. b. for the current school year.

SECTION 1492g. 115.7915 (4m) (b) of the statutes is amended to read:

115.7915 (4m) (b) The department shall pay 25 percent of the total amount under par. (a) in September, 25 percent in November, 25 percent in February, and 25 percent in May. Each installment may consist of a single check for all children attending the private school who are receiving a scholarship under par. (a). The department shall include the entire amount under sub. (4p) in the November installment, but the payment shall be made in a separate check from the payment under this subsection.

SECTION 1492h. 115.7915 (4m) (c) of the statutes is amended to read:

115.7915 (**4m**) (c) The department may not make a scholarship payment under par. (a) or sub. (4p) to a private school on behalf of a child's parent or guardian unless the child's parent or guardian has acknowledged receiving the private school's profile under sub. (6) (g) in the manner provided by the department.

SECTION 1492i. 115.7915 (4m) (cm) of the statutes is created to read:

115.7915 (4m) (cm) If a private school receives a payment under par. (a) 3. and the amount shown on the financial statement submitted for the child with a disability under sub. (4c) in the previous school year is greater than 150 percent of the amount calculated under par. (a) 2. b. for the current school year, the department shall pay to the private school, from the appropriation under s. 20.255 (2) (az), the amount determined as follows:

- 1. Multiply the amount calculated under par. (a) 2. b. for the current school year by 1.5.
- 2. Subtract the product under subd. 1. from the amount shown on the financial statement.
- 3. Multiply the difference determined under subd. 2. by 0.90.

SECTION 1492j. 115.7915 (4m) (f) 1. a. of the statutes is amended to read:

115.7915 **(4m)** (f) 1. a. Determine the number of pupils sum of the amount paid for each child residing in the school district for whom a payment is made under par. (a) in that school year.

SECTION 1492k. 115.7915 (4m) (f) 1. b. of the statutes is repealed.

SECTION 1492L. 115.7915 (4m) (f) 1. df. of the statutes is created to read:

115.7915 (**4m**) (f) 1. df. Identify the children residing in the school district for whom a payment is made under sub. (**4p**) in that school year.

SECTION 1492m. 115.7915 (4m) (f) 1. dh. of the statutes is created to read:

115.7915 (**4m**) (f) 1. dh. Sum the payments made under sub. (4p) (a) for all of the children identified under subd. 1. df. for that school year.

SECTION 1492n. 115.7915 (4m) (f) 1. e. of the statutes is amended to read:

115.7915 (**4m**) (f) 1. e. Sum the amounts calculated under subd. 1. b. and a., d., and dh.

SECTION 1492p. 115.7915 (4p) of the statutes is created to read:

115.7915 (**4p**) SCHOLARSHIP PAYMENT; SUMMER SCHOOL. (a) In addition to the scholarship amount under sub. (4m), the department shall, subject to par. (b) and sub. (4m) (c) and in the manner described in sub. (4m) (b), pay to a private school participating in the scholarship program under this section, on behalf of the parent or guardian of a child who attended a private school under this section during the immediately preceding school term and who attends summer school in the private school under this section during a summer, an amount determined as follows:

- 1. Determine the scholarship amount under sub. (4m) that applied to the child in the immediately preceding school term.
- 2. If the child attended summer school for at least 15 days of summer instruction at the private school during that summer, multiply the amount under subd. 1. by 0.05.
- 3. If the child attended summer school for less than 15 days of summer instruction at the private school during that summer, multiply the amount under subd. 1. by 0.05, and multiply that product by the quotient determined by dividing the number of days of summer instruction the child attended during that summer by 15.
- (b) A participating private school may receive payment under par. (a) only if all of the following are satisfied:
- 1. The private school offers no fewer than 19 summer days of instruction during that summer.
- 2. Each summer day of instruction offered by the private school under subd. 1. is comprised of no fewer than 270 minutes of instruction.

SECTION 1492v. 115.7915 (6) (k) of the statutes is created to read:

115.7915 (6) (k) Annually, on or before September 15, file with the department a report stating its summer daily attendance for each day of summer school for the purpose of sub. (4p).

SECTION 1493r. 115.7915 (8) (c) of the statutes is amended to read:

115.7915 (8) (c) The state superintendent may withhold payment under sub. (4m) or (4p) from a private school participating in the program under this section if the private school violates this section.

SECTION 1494. 115.88 (1) of the statutes is renumbered 115.88 (1) (a) and amended to read:

115.88 (1) (a) A school board, board of control of a cooperative educational service agency, operator of a charter school established under s. 118.40 (2r) or (2x), operator of a school under a contract with a school board under s. 118.40 that is not an instrumentality of a school district, or, upon authorization of the county board, a county children with disabilities education board may employ, for a special education program, either full—time or part—time licensed teachers, licensed coordinators of special education, school nurses, licensed school social workers, licensed school psychologists, licensed school counselors, paraprofessionals, licensed consulting teachers to work with any teacher of regular education programs who has a child with a disability in a class, and any other personnel approved by the department. The

(b) A school board, board of control of a cooperative educational service agency, operator of a charter school established under s. 118.40 (2r) or (2x), operator of a school under a contract with a school board under s. 118.40 that is not an instrumentality of a school district, or, upon authorization of the county board, a county children with disabilities education board may contract with private or public agencies for substitute teaching and paraprofessional staffing services, physical or occupational therapy services, orientation and mobility training services, educational interpreter services, educational audiology, speech and language therapy, pupil transition services for eligible pupils who are 18 to 21 years old, or any service approved by the state superintendent, on the basis of demonstrated need. A school board may contract with a charter school to provide special education services to pupils attending the charter school if the charter school is under contract with the school board under s. 118.40 (2m) and the charter school is not an instrumentality of the school district.

SECTION 1495. 115.88 (1m) (a) of the statutes is amended to read:

115.88 (1m) (a) Subject to par. (b), upon receipt of the plan under s. 115.77 (4), if the state superintendent is satisfied that the special education program has been maintained during the preceding school year in accordance with law, the state superintendent shall certify to the department of administration in favor of each county, cooperative educational service agency, charter school established under s. 118.40 (2r) or (2x), and school district maintaining such special education program a sum equal to the amount expended by the county, agency, charter school, and school district during the preceding year for salaries of personnel and services enumerated in sub. (1); the salary portion of any authorized contract for

services under sub. (1); the salary portion of any contract to provide special education services to pupils attending a charter school, as authorized under sub. (1); (a) and (b) and other expenses approved by the state superintendent, as costs eligible for reimbursement from the appropriation under s. 20.255 (2) (b).

SECTION 1496. 115.88 (1m) (ag) of the statutes is created to read:

115.88 (1m) (ag) Subject to par. (b), if the state superintendent is satisfied that the special education program of an operator of a school under a contract with a school board under s. 118.40 that is not an instrumentality of a school district has been maintained during the previous school year in accordance with law, the state superintendent shall certify to the department of administration in favor of the school board a sum equal to the amount expended by the operator during the previous school year for salaries of personnel and services enumerated in sub. (1) (a) and (b) and other expenses approved by the state superintendent as costs eligible for reimbursement from the appropriation under s. 20.255 (2) (b). Within 30 days of its receipt, the school board shall pay to the operator of the school under a contract with the school board the aid received under this paragraph.

SECTION 1497. 115.88 (1m) (am) of the statutes is repealed.

SECTION 1498. 115.88 (1m) (b) of the statutes is amended to read:

115.88 (**1m**) (b) The department shall promulgate rules establishing the percentage of the salaries of school nurses, licensed school social workers, licensed school psychologists, and licensed school counselors that may be certified under pars. par. (a) and (am) as costs eligible for reimbursement. For each category of personnel, the department shall base the percentage on the average percentage of work time that the category spends providing services to children with disabilities, including conducting evaluations under s. 115.782.

SECTION 1499. 115.88 (2) of the statutes is repealed. **SECTION 1500.** 115.88 (2m) (title) of the statutes is amended to read:

115.88 **(2m)** (title) OTHER SPECIAL OR ADDITIONAL TRANSPORTATION; AID.

SECTION 1501. 115.88 (2m) of the statutes is renumbered 115.88 (2m) (b) and amended to read:

115.88 (2m) (b) If the state superintendent is satisfied that a school board, board of control, operator of a charter school established under s. 118.40 (2r) or (2x), or established as a noninstrumentality charter school under s. 118.40 (2m) transports children with disabilities and the state superintendent is satisfied that the operator of the charter school is complying with 20 USC 1400 to 14910 county children with disabilities education board has provided special or additional transportation during the previous school year as described under par. (a), the state superintendent shall certify to the department of adminis-

tration in favor of the school board, board of control, operator of the charter school, or county children with disabilities education board providing the transportation a sum equal to the amount that expended by the school board, board of control, operator of the charter school expended, or county children with disabilities education board during the previous school year for transportation under this subsection as costs eligible for reimbursement from the appropriations under s. 20.255 (2) (b). The state superintendent may audit costs under this subsection and adjust reimbursement to cover only actual, eligible costs.

SECTION 1502. 115.88 (2m) (a) of the statutes is created to read:

115.88 (2m) (a) A school board, board of control of a cooperative educational service agency, operator of a charter school established under s. 118.40 (2r) or (2x), or, upon authorization of the county board, a county children with disabilities education board shall provide special or additional transportation as required in the individualized education program developed for the child with a disability under s. 115.787 (2) or as required under s. 121.54 (3). The operator of a school under a contract with a school board under s. 118.40 that is not an instrumentality of the school district shall provide special or additional transportation under this paragraph if the contract between the operator and the school board requires the operator to provide the special or additional transportation.

SECTION 1503. 115.88 (2m) (c) of the statutes is created to read:

115.88 (2m) (c) If the state superintendent is satisfied that the operator of a school under a contract with a school board under s. 118.40 that is not an instrumentality of the school district has provided special or additional transportation during the previous school year as described under par. (a), the state superintendent shall certify to the department of administration in favor of the school board a sum equal to the amount expended by the operator during the previous school year for providing the transportation as costs eligible for reimbursement from the appropriation under s. 20.255 (2) (b). Within 30 days of its receipt, the school board shall pay to the operator of the school under a contract with the school board the aid received under this paragraph.

SECTION 1504. 115.88 (10) of the statutes is created to read:

115.88 (10) AUDIT OF ELIGIBLE COSTS. The state superintendent may audit costs under this section and adjust the amounts eligible for reimbursement to cover only actual, eligible costs.

SECTION 1504c. 115.881 (2) of the statutes is amended to read:

115.881 (2) For each child whose costs exceeded \$30,000 under sub. (1), the department shall, from the appropriation under s. 20.255 (2) (bd), pay an eligible applicant in the current school year an amount equal to

0.70 0.90 multiplied by that portion of the cost under sub. (1) that exceeded \$30,000.

SECTION 1504d. 115.884 (1) (intro.) of the statutes is amended to read:

115.884 (1) (intro.) In the 2016–17 school year and each school year thereafter, from the appropriation under s. 20.255 (2) (bf), the department shall award an incentive grant in the amount of \$1,000 per individual to a school district, or to an operator of a charter school established under s. 118.40 (2r) or (2x), that applies for a grant under this section and that demonstrates to the satisfaction of the department that the individual satisfies all of the following criteria:

SECTION 1504e. 115.884 (1) (a) (intro.) of the statutes is created to read:

115.884 (1) (a) (intro.) During the school year prior to the school year preceding the school year in which a school district or operator of the charter school applies to receive a grant under this section, all of the following criteria apply to the individual:

SECTION 1504f. 115.884 (1) (a) of the statutes is renumbered 115.884 (1) (a) 1. and amended to read:

115.884 (1) (a) 1. The individual was enrolled in a <u>high</u> school in the school district or in <u>a high school grade</u> in the charter school in the 2014–15 or 2015–16 school year and, at the time of his or her enrollment, an.

3. An individualized education program was in effect for the individual.

SECTION 1504g. 115.884 (1) (a) 2. of the statutes is created to read:

115.884 (1) (a) 2. The individual exited the high school in the school district or exited the charter school.

SECTION 1504h. 115.884 (1) (b) of the statutes is amended to read:

115.884 (1) (b) At the time the school district or the operator of the charter school applies to receive an incentive a grant under this section, one of the following criteria applies to the individual described in par. (a):

- 1. The individual had enrolled in a higher education program within one year of leaving exiting high school. In this subdivision, "higher education program" means a 4–year program at a college or university, a 2–year program at a college or community college, or a 2–year program at a technical college.
- 2. The individual had enrolled in other postsecondary education or training within one year of leaving exiting high school. In this subdivision, "other postsecondary education or training" includes a high school completion or equivalency program, a vocational school, an apprenticeship or short–term training program, an on–the–job training program, an adult education program, and a program, other than a 2–year program, at a vocational or technical school.
- 3. The individual had been, or remains, competitively employed within one year of <u>leaving exiting</u> high school. In this subdivision, "competitively employed"

means 90 days or more of cumulative or consecutive work paying minimum wage or greater for an average of at least 20 hours per week in a setting with others who are not disabled.

SECTION 1504i. 115.884 (2) of the statutes is amended to read:

115.884 (2) If the appropriation under s. 20.255 (2) (bf) in the 2016–17 school any fiscal year is insufficient to pay the full amount under sub. (1), the department shall prorate the amount of its payments among school districts and operators of charter schools established under s. 118.40 (2r) or (2x) that are eligible for an incentive grant under this section.

SECTION 1504j. 115.885 of the statutes is created to read:

- 115.885 Special education transition readiness grant program. (1) Beginning in the 2018–19 school year, the department shall awards grants to school districts and charter schools under s. 118.40 (2r) and (2x) to fund special education workforce transition support services, including pupil transportation, professional development for school personnel, and employing adequate school personnel.
- (2) The department shall award grants under sub. (1) from the appropriation under s. 20.255 (2) (bg). The department may not award a grant under sub. (1) in an amount that is less than \$25,000 nor more than \$100,000.
- (3) The department shall promulgate rules to implement and administer this section.

SECTION 1504k. 115.999 (2) (a) of the statutes is amended to read:

115.999 (2) (a) Within 120 days after receiving notice under s. 115.28 (10o) (b), the governor, the mayor, and the county executive shall compile a list of candidates for commissioner. The Subject to sub. (2m), the county executive shall select a commissioner using the procedure under s. 119.9001 (2) (a).

SECTION 1504L. 115.999 (2m) of the statutes is created to read:

- 115.999 (2m) SPECIAL PROVISIONS; CERTAIN UNIFIED SCHOOL DISTRICTS. (a) In this subsection, an "eligible unified school district" mean a unified school district that satisfies the following criteria:
- 1. The unified school district is an eligible school district.
- 2. The unified school district contains a city that has a population of more than 75,000.
- 3. The unified school district contains at least 2 villages.
- (b) 1. Within 120 days after receiving notice under s. 115.28 (10o) (d), an eligible unified school district may demonstrate to the department of administration that the school board of the eligible unified school district is not, directly or indirectly, delegating its authority to make decisions about providing benefits to its employees. If the department of administration certifies that the school

board of the eligible unified school district is not, directly or indirectly, delegating its authority to make decisions about providing benefits to its employees, the county executive may not select a commissioner under sub. (2) (a) unless the eligible unified school district satisfies all of the following criteria:

- a. The unified school district was assigned to the lowest performance category on the accountability reports published for the district under s. 115.385 (1) in the 3 most recent school years.
- b. The school district received intradistrict transfer aid under s. 121.85 (6) (a) in the 3 school years described under subd. 1. a.
- 2. If the department of administration does not certify that the school board of an eligible unified school district is not, directly or indirectly, delegating its authority to make decisions about providing benefits to its employees, the county executive shall select a commissioner under sub. (2) (a).
- (c) Within 60 days after receiving notice under s. 115.28 (100) (d), the village board of each village located within an eligible unified school district may consider a resolution to affirm or deny the village board's intent to create a new school district under s. 117.105 (4m). If a village board adopts a resolution affirming the village's intent to create a new school district, the village board shall include all of the following information in the resolution:
- 1. The territory of the new school district. The territory of the new school district shall correspond with village boundaries but may incorporate more than one village.
 - 2. The name of the new school district.
- 3. The type of the new school district and the grades to be taught by the new school district as described under s. 115.01 (2) and (3).
- 4. Whether the proposed effective date of the school district creation is July 1 of the following school year or July 1 of the second following year.
- (d) If a county executive may select a commissioner under par. (b) 1. and in the previous school year a village board did not adopt a resolution to affirm its intent to create a new school district, within 60 days of receiving notice under s. 115.28 (10o) (d) for the 3rd consecutive school year, the village board shall by resolution provide for a referendum on the question of whether to create a new school district. The village board shall include in the resolution all of the information described in par. (c) 1. to 4.

SECTION 1504m. 117.05 (1m) of the statutes is amended to read:

117.05 (**1m**) BOARD AND APPEAL PANEL MEETINGS. The state superintendent shall set the time and place for meetings of the board under ss. 117.10, 117.105 (2m) and (4m), 117.12 (5) and 117.132 and for meetings of appeal panels under ss. 117.12 (4) and 117.13.

SECTION 1504n. 117.05 (2) (a) of the statutes is amended to read:

117.05 (2) (a) *Board*. The state superintendent shall appoint 7 members of the board to perform any review under ss. 117.10, 117.105 (2m) and (4m), 117.12 (5), and 117.132. The 7 members shall include the state superintendent or his or her designee on the board, 2 board members from school districts with small enrollments, 2 board members from school districts with medium enrollments, and 2 board members from school districts with large enrollments. Any action of the board under this chapter requires the affirmative vote of at least 4 of the 7 members appointed under this paragraph.

SECTION 1504p. 117.05 (4) (a) (intro.) of the statutes is amended to read:

117.05 (4) (a) *Pending proceedings*. (intro.) A reorganization proceeding is pending from the date that a petition is filed under s. 117.105 (1) (a), 117.11 (2) or 117.12 (2) or a resolution is adopted under s. 117.08 (1), 117.09 (1), 117.10 (1), 117.105 (1) (b) or (4m), 117.13 (2) or 117.132 (2) until the date on which the latest of any of the following occurs:

SECTION 1504q. 117.05 (4) (d) 1. of the statutes is amended to read:

117.05 (4) (d) 1. Except as provided in subd. 2., no petition may be filed or resolution adopted for the creation of a new school district under s. 117.105 (1) (a) or (b) before the 5th July 1 following the filing of a petition under s. 117.105 (1) (a) or the adoption of a resolution under s. 117.105 (1) (b) or the date of an order issued under s. 117.105 (4m) (c) for any reorganization that includes any of the same territory.

SECTION 1504r. 117.05 (9) (a) 1m. of the statutes is created to read:

117.05 (9) (a) 1m. The village boards of villages and the affected school district under s. 117.105 (4m).

SECTION 1504u. 117.08 (6) of the statutes is created to read:

- 117.08 **(6)** STATE AID. From the appropriation under s. 20.255 (2) (br), the department shall pay to a school district created by a consolidation under this section that takes effect on or after July 1, 2019, the following amounts:
- (a) In the school year in which the consolidation takes effect and in each of the subsequent 4 school years, \$150 multiplied by the number of pupils enrolled, as defined in s. 115.437 (1), in the school district in that school year.
- (b) In the 5th school year following the school year in which the consolidation takes effect, 50 percent of the amount the school district received under par. (a) in the 4th year following the school year in which the consolidation takes effect.
- (c) In the 6th school year following the school year in which the consolidation takes effect, 25 percent of the amount the school district received under par. (a) in the

4th year following the school year in which the consolidation takes effect.

SECTION 1504v. 117.09 (6) of the statutes is created to read:

- 117.09 (6) STATE AID. From the appropriation under s. 20.255 (2) (br), the department shall pay to a school district created by a consolidation under this section that takes effect on or after July 1, 2019, the following amounts:
- (a) In the school year in which the consolidation takes effect and in each of the subsequent 4 school years, \$150 multiplied by the number of pupils enrolled, as defined in s. 115.437 (1), in the school district in that school year.
- (b) In the 5th school year following the school year in which the consolidation takes effect, 50 percent of the amount the school district received under par. (a) in the 4th year following the school year in which the consolidation takes effect.
- (c) In the 6th school year following the school year in which the consolidation takes effect, 25 percent of the amount the school district received under par. (a) in the 4th year following the school year in which the consolidation takes effect.

SECTION 1504w. 117.105 (4m) of the statutes is created to read:

- 117.105 **(4m)** PROCEDURES FOR CREATION OF A SCHOOL DISTRICT BASED ON AN OPPORTUNITY SCHOOLS AND PARTNERSHIP PROGRAM; CERTAIN VILLAGES. (a) In this subsection:
- 1. "Eligible unified school district" has the meaning given in s. 115.999 (2m) (a).
- 2. "Village board" means the village board of a village located in an eligible unified school district.
- (b) If a village board adopts a resolution under s. 115.999 (2m) (c) affirming the village board's intent to create a new school district or a resolution to provide a referendum under s. 115.999 (2m) (d), the village board shall submit the resolution to the board.
- (c) 1. Upon receipt of a resolution under par. (b), the board shall hold a public hearing on the proposed reorganization. After the hearing and after consulting with the school board of the eligible unified school district and the village boards of the villages located within the proposed school district, the board shall determine the apportionment of assets and liabilities between the eligible unified school district and the proposed school district according to the criteria under s. 66.0235 (2c).
- 2. By no later than 180 days after the village board receives notice under s. 115.28 (10o) (d), the board shall issue an order that includes the territory of the new school district, as described in the resolution adopted by the village board, and the apportionment of assets and liabilities as determined under subd. 1.
- (d) A binding referendum on a proposed reorganization ordered by the board under par. (c) 2. shall be held

in the territory of the school district proposed to be created by the reorganization.

- (e) 1. If a majority of the votes cast in the referendum held under par. (d) is in favor of the creation of the proposed school district, no later than 60 days after the referendum is held, the village boards of the villages included in the proposed school district shall adopt a resolution to designate all of the following for the proposed school district:
- a. The number of school board members under s. 120.01 or 120.41.
- b. The terms of initial members of the school board under s. 120.02 (3) (a).
- c. The method of election of school board members under s. 120.06 or 120.42.
- 2. A village that adopts a resolution under subd. 1. shall submit the resolution to the eligible unified school district.
- 3. If a majority of the votes cast in the referendum held under par. (d) is in favor of the creation of the proposed school district, after receiving a resolution under subd. 2., the school board of the eligible unified school district shall make and file an order of school district reorganization under s. 117.17 (2). In the order, the school board shall designate that the first election of school board members of the newly created school district shall occur at the regularly scheduled spring election immediately following the date on which the order is filed with the board. The reorganization shall take effect on the July 1 described in the resolution adopted by the village board under s. 115.999 (2m) (c).
- (f) Subsections (1) to (4) and ss. 117.14 and 117.15 do not apply to a reorganization under this subsection.

SECTION 1504x. 117.20 (1) (a) of the statutes is amended to read:

117.20 (1) (a) Except as provided in par. (b), if a referendum is required under ss. 117.08 to 117.11, it shall be held on the Tuesday after the first Monday in November following receipt of the petition or adoption of the resolution under s. 117.08 (3) (a), 117.09 (3) (a), 117.10 (3) (a) or 117.11 (4) (a). If a referendum is required under s. 117.105 (3), it shall be held on the Tuesday after the first Monday in the 2nd November following receipt of the petition or adoption of the resolution under s. 117.105 (1). If a referendum is required under s. 117.105 (4m), it shall be held on the Tuesday after the first Monday in November following the date an order is issued by the board under s. 117.105 (4m) (c).

SECTION 1504y. 117.22 (2) (bm) of the statutes is amended to read:

117.22 (2) (bm) If an order of reorganization is issued under s. 117.105, the first election of school board members shall be held at the spring election following the referendum under s. 117.105 (3) or (4m).

SECTION 1505f. 118.076 (3) (intro.) of the statutes is amended to read:

118.076 (3) (intro.) Beginning in the 2017–18 school year and subject to sub. (4), each school board operating any grade from 7 to 12, the operator of each charter school established under s. 118.40 (2r) or (2x) that operates any grade from 7 to 12, and the governing body of each private school that operates any grade from 7 to 12 shall do all of the following:

SECTION 1505h. 118.076 (4) of the statutes is created to read:

118.076 (4) (a) In this subsection, a "virtual school" is a school in which all or a portion of the instruction is provided through means of the Internet, and the pupils enrolled in and instructional staff employed by the school are geographically remote from each other. "Virtual school" includes a virtual charter school.

(b) A virtual school need not provide any instruction required under sub. (3) in a manner that requires the pupils receiving the instruction and instructional staff providing the instruction to be together in the same geographical location. A virtual school may provide all of the instruction required under sub. (3) through the means of the Internet.

SECTION 1506m. 118.16 (2) (c) of the statutes is amended to read:

118.16 (2) (c) Except as provided under pars. (cg) and (cr), shall notify the parent or guardian of a child who has been truant of the child's truancy and direct the parent or guardian to return the child to school no later than the next day on which school is in session or to provide an excuse under s. 118.15. The notice under this paragraph shall be given before the end of the 2nd school day after receiving a report of an unexcused absence. The notice may be made by electronic communication, personal contact, 1st class mail, or telephone call of which a written record is kept, except that. The school attendance officer shall attempt to give notice by personal contact or. telephone call shall be attempted, or, unless the parent or guardian has refused to receive electronic communication, electronic communication before notice by 1st class mail may be given.

SECTION 1506n. 118.19 (title) of the statutes is amended to read:

118.19 (title) Teacher certificates and licenses; administrator and pupil services professional licenses.

SECTION 1507. 118.19 (1) of the statutes is amended to read:

118.19 (1) Except as provided in <u>subs. (1b)</u> and (1c) and s. 118.40 (8) (b) 1. and 2., any person seeking to teach in a public school, including a charter school, or in a school or institution operated by a county or the state shall first procure a license or permit from the department.

SECTION 1507m. 118.19 (1b) of the statutes is created to read:

118.19 (**1b**) An individual may teach an online course in a subject and level in a public school, including a charter school, without a license or permit from the department if the individual holds a valid license or permit to teach the subject and level in the state from which the online course is provided.

SECTION 1508. 118.19 (1c) of the statutes is created to read:

- 118.19 (1c) (a) In this subsection, "institution of higher education" means an institution or college campus within the University of Wisconsin System, a technical college under ch. 38, or any private, nonprofit postsecondary institution that is a member of the Wisconsin Association of Independent Colleges and Universities.
- (b) A faculty member of an institution of higher education may teach in a public high school, including a charter school that operates only high school grades, without a license or permit from the department if the faculty member satisfies all of the following:
- 1. The faculty member is in good standing with the institution of higher education at which he or she is a member of the faculty.
- 2. The faculty member possesses a bachelor's degree.
- 3. The department conducts a background investigation of the faculty member and the results of the background investigation would not make the faculty member ineligible for a teaching license under sub. (4) or (10).

SECTION 1509. 118.19 (1m) of the statutes is amended to read:

118.19 (1m) (a) The department of public instruction may not issue or renew a license or permit or revalidate a license that has no expiration date unless the applicant provides the department of public instruction with his or her social security number. The department of public instruction may not disclose the social security number except to the department of revenue for the sole purpose of requesting certifications under s. 73.0301 and to the department of workforce development for the sole purpose of requesting certifications under s. 108.227.

(b) The department of public instruction may not issue or renew a license or permit or revalidate a license that has no expiration date if the department of revenue certifies under s. 73.0301 that the applicant, licensee, or permit holder is liable for delinquent taxes or if the department of workforce development certifies under s. 108.227 that the applicant, licensee, or permit holder is liable for delinquent unemployment insurance contributions

SECTION 1510. 118.19 (1r) of the statutes is amended to read:

118.19 (**1r**) (a) As provided in the memorandum of understanding under s. 49.857, the department of public instruction may not issue or renew a license or permit or revalidate a license that has no expiration date unless the applicant provides the department of public instruction

with his or her social security number. The department of public instruction may not disclose the social security number except to the department of children and families for the sole purpose of administering s. 49.22.

(b) As provided in the memorandum of understanding under s. 49.857, the department may not issue or renew a license or permit or revalidate a license that has no expiration date if the applicant, licensee or permit holder is delinquent in making court—ordered payments of child or family support, maintenance, birth expenses, medical expenses or other expenses related to the support of a child or former spouse or if the applicant, licensee or permit holder fails to comply, after appropriate notice, with a subpoena or warrant issued by the department of children and families or a county child support agency under s. 59.53 (5) and related to paternity or child support proceedings.

SECTION 1511. 118.19 (1s) (a) of the statutes is amended to read:

118.19 (1s) (a) Notwithstanding subs. (1m) and (1r), if an applicant does not have a social security number, the applicant, as a condition of applying for, or applying to renew or revalidate, a license under this section shall submit a statement made or subscribed under oath or affirmation to the department that the applicant does not have a social security number.

SECTION 1511p. 118.19 (3) (a) of the statutes is amended to read:

118.19(3) (a) No license to teach in any public school may be issued unless the applicant possesses a bachelor's degree including such professional training as the department by rule requires, except as permitted under par. (b) and ss. 115.28 (17) (a), 118.191, 118.1915, 118.192, 118.193, and 118.194, and 118.197. Notwithstanding s. 36.11 (16), no teacher preparatory program in this state may be approved by the state superintendent under s. 115.28 (7) (a), unless each student in the program is required to complete student teaching consisting of full days for a full semester following the daily schedule and semester calendar of the cooperating school. No license to teach in any public school may be granted to an applicant who completed a professional training program outside this state unless the applicant completed student teaching consisting of full days for a full semester following the daily schedule and semester calendar of the cooperating school or the equivalent, as determined by the state superintendent. The state superintendent may grant exceptions to the student teaching requirements under this paragraph when the midyear calendars of the institution offering the teacher preparatory program and the cooperating school differ from each other and would prevent students from attending classes at the institution in accordance with the institution's calendar. The state superintendent shall promulgate rules to implement this subsection. If for the purpose of granting a license to teach or for approving a teacher preparatory program the state superintendent requires that an institution of higher education be accredited, the state superintendent shall accept accreditation by a regional or national institutional accrediting agency recognized by the U.S. department of education or by a programmatic accrediting organization.

SECTION 1512. 118.19 (4m) of the statutes is amended to read:

118.19 (4m) The state superintendent may not issue or renew a license to teach the visually impaired unless the applicant demonstrates, based on criteria established by the state superintendent by rule, that he or she is proficient in reading and writing braille and in teaching braille. In promulgating rules under this subsection, the state superintendent shall take into consideration the standard used by the librarian of congress for certifying braille transcribers.

SECTION 1512c. 118.19 (7m) of the statutes is created to read:

118.19 (7m) The state superintendent shall grant a substitute teacher permit to an individual who is eligible for licensure under subs. (4) and (10), has an associate degree, and has successfully completed substitute teacher training.

SECTION 1512d. 118.19 (10) (a) of the statutes is renumbered 118.19 (10) (a) (intro.) and amended to read:

118.19 (10) (a) (intro.) In this subsection, "educational:

1. "Educational agency" has the meaning given in s. 115.31 (1) (b).

SECTION 1512h. 118.19 (10) (a) 2. of the statutes is created to read:

118.19 (**10**) (a) 2. "Pupil services professional" has the meaning given in s. 118.257 (1) (c).

SECTION 1512p. 118.19 (10) (b) 1. of the statutes is amended to read:

118.19 (10) (b) 1. Conduct a background investigation of each applicant for issuance or renewal of a license or permit, including a license or permit issued to a pupil services professional, and for a faculty member seeking to teach in a public high school without a license or permit.

SECTION 1512t. 118.19 (10) (b) 2. of the statutes is renumbered 118.19 (10) (b) 2. (intro.) and amended to read:

118.19 (**10**) (b) 2. (intro.) Over a 5-year period At least once every 5 years, conduct a background investigation of each person who satisfies all of the following:

- <u>a.</u> The person holds a license, issued by the state superintendent, <u>including a license issued to a pupil services professional</u>, that has no expiration date and who.
- <u>b. The person</u> is employed by an educational agency <u>or by a charter school established under s. 118.40 (2r) or (2x).</u>

SECTION 1516. 118.19 (10) (e) of the statutes is amended to read:

118.19 (10) (e) The state superintendent may issue or renew a license or permit conditioned upon the receipt of a satisfactory background investigation.

SECTION 1517. 118.19 (12) of the statutes is amended to read:

118.19 (12) Beginning on July 1, 1998, the department may not issue or renew a license that authorizes the holder to teach reading or language arts to pupils in any prekindergarten class or in any of the grades from kindergarten to 6 unless the applicant has successfully completed instruction preparing the applicant to teach reading and language arts using appropriate instructional methods, including phonics. The phonics instruction need not be provided as a separate course. In this subsection, "phonics" means a method of teaching beginners to read and pronounce words by learning the phonetic value of letters, letter groups and syllables.

SECTION 1518. 118.19 (17) of the statutes is repealed. SECTION 1519. 118.19 (18) of the statutes is created to read:

118.19 (18) (a) Beginning on the effective date of this paragraph [LRB inserts date], and subject to ss. 115.31 and 115.315, an individual who is applying for an initial teaching license, an initial administrator license, or an initial pupil services professional license under this section shall be subject to the requirement of this paragraph. The department may issue a provisional license for a term of 3 years to an individual who applies for an initial license under this paragraph.

- (b) 1. Except as provided in par. (c), and subject to ss. 115.31 and 115.315, a professional or master teaching license, administrator license, or pupil services license that is valid and current on the effective date of this subdivision [LRB inserts date], is a lifetime license and has no expiration date.
- 2. The department shall, subject to ss. 115.31 and 115.315, issue a provisional license to an individual who holds a valid and current initial teaching, administrator, or pupil services license on the effective date of this subdivision [LRB inserts date].
- 3. a. The department may issue a lifetime license under this subdivision to an individual who obtained a provisional license under subd. 2. or par. (a) if the individual has successfully completed 6 semesters of teaching, administrating, or pupil services experience and if the school board of each school district in which the individual served as a teacher, administrator, or pupil services professional certifies to the department that the semesters completed by that individual in that school district were completed successfully.
- b. An individual who does not complete 6 semesters of teaching, administrating, or pupil services experience within the 3-year term of a provisional license issued under par. (a) may apply to renew the provisional license. There is no limit to the number of times an individual may renew a provisional license under par. (a).

(c) If an individual who holds a lifetime license under par. (b) is not actively employed by a school district for 5 or more consecutive years, the department shall invalidate the lifetime license. An individual whose lifetime license has been invalidated under this paragraph may not revalidate the lifetime license until the individual applies for and obtains a provisional license under par. (a) and obtains the certification required under par. (b) 3. a.

SECTION 1519m. 118.1915 of the statutes is created to read:

118.1915 Licensure for Junior Reserve Officer Training Corps instructors. (1) Notwithstanding s. 118.19 (4m), (6) to (9), and (12) to (14), the department shall grant a license to an individual to provide instruction to pupils enrolled in a Junior Reserve Officer Training Corps program offered in the high school grades if the individual is eligible for licensure under s. 118.19 (4) and (10) and the individual satisfies all of the following:

- (a) Possesses a bachelor's degree.
- (b) Successfully completed a Junior Reserve Officer Training Corps Instructor Certification program.
- (2) A license under sub. (1) authorizes an individual to teach the courses for which the individual has successfully completed the Junior Reserve Officer Training Corps Instructor Certification process.

SECTION 1520. 118.193 (2) (c) of the statutes is repealed.

SECTION 1521. 118.193 (3) (c) of the statutes is repealed.

SECTION 1522. 118.193 (4) (a) of the statutes is repealed.

SECTION 1523. 118.193 (4) (b) of the statutes is renumbered 118.193 (4).

SECTION 1524. 118.196 of the statutes is created to read:

- 118.196 Teacher development program. (1) A school board, governing body of a private school, or a charter management organization may apply to the department of workforce development for a grant under s. 106.272 to design and implement a teacher development program that satisfies the requirements under sub. (2) with an educator preparation program approved by the department and headquartered in this state.
- (2) (a) The school board, governing body, or charter management organization and the educator preparation program under sub. (1) shall design the teacher development program to prepare employees of the school district, private school, or charter management organization who work closely with students to successfully complete the requirements for obtaining a permit under s. 118.192 or an initial teaching license under s. 118.19, including any standardized examination prescribed by the state superintendent as a condition for permitting or licensure.
- (b) To implement the teacher development program designed under par. (a), the school board, governing body, and charter management organization shall allow

- employees who are enrolled in the program to satisfy student teaching requirements in a school in the school district, in the private school, or in the charter management organization, and the partnering entity under sub. (1) shall prepare and provide intensive coursework for participating employees.
- (c) The school board, governing body, and charter management organization shall permit an individual who does not hold a bachelor's degree to enroll in a teacher development program developed and implemented under this section.
- (3) Notwithstanding s. 118.19 (3) (a), the department may issue an initial teaching license to an individual who completes a teacher development program designed and implemented under this section.

SECTION 1524g. 118.197 of the statutes is created to read:

118.197 Initial license to teach; alternative preparation programs. (1) Notwithstanding s. 118.19 (4m), (6) to (9), and (12) to (14), the department shall grant an initial license to teach to an individual who is eligible for licensure under s. 118.19 (4) and (10) and who satisfies all of the following:

- (a) The individual possesses a bachelor's degree.
- (b) The individual successfully completed an alternative teacher certification program operated by a non-stock, nonprofit organization described under section 501 (c) (3) of the Internal Revenue Code that satisfies all of the following criteria:
 - 1. The organization operates in at least 5 states.
- 2. The organization has been operating an alternative teacher certification program for at least 10 years.
- 3. The organization requires candidates to pass a subject area exam and the pedagogy exam known as the Professional Teaching Knowledge exam to receive a certificate under the alternative teacher certification.
- (2) A license under sub. (1) authorizes an individual to teach the subject and educational levels for which the individual has successfully completed an alternative teacher certification program. The department shall treat an initial license to teach granted under sub. (1) in the same manner the state superintendent treats an initial license to teach granted in accordance with s. 118.19.

SECTION 1525. 118.21 (2) of the statutes is amended to read:

118.21 (2) Any person who contracts to teach in any public school shall file in the office of the school district administrator, within 10 days after entering into such contract, a statement showing the date of expiration, if any, and the grade and character of certificate or license held. In any school district not having a school district administrator, the statement shall be filed with the school district clerk. Teachers employed by a cooperative educational service agency shall file the statement in the office of the agency coordinator. No order or warrant may be issued by the school district clerk in payment of

the salary of any teacher, unless the teacher has complied with this subsection.

SECTION 1528m. 118.33 (1m) (a) 1. of the statutes is amended to read:

118.33 (1m) (a) 1. Beginning in the 2016–17 school year, no school board, operator of a charter school under s. 118.40 (2r) or (2x), or governing body of a private school participating in a program under s. 118.60 or 119.23 may, except as provided in subd. 2. and subject to the policies under sub. (2) (m), grant a high school diploma to any pupil unless the pupil takes, during the high school grades, a civics test comprised of 100 questions that are identical to the 100 questions that may be asked of an individual during the process of applying for U.S. citizenship by the United States Citizenship and Immigration Services and the pupil correctly answers at least 60 65 of those questions.

SECTION 1529. 118.35 (4) of the statutes is amended to read:

118.35 (4) From the appropriation under s. 20.255 (2) (fy), the department shall award grants to nonprofit organizations, cooperative educational service agencies, institutions within the University of Wisconsin System, and the school district operating under ch. 119 districts for the purpose of providing to gifted and talented pupils those services and activities not ordinarily provided in a regular school program that allow such pupils to fully develop their capabilities.

SECTION 1531g. 118.40 (2r) (b) 1. c. of the statutes is amended to read:

118.40 (**2r**) (b) 1. c. On a pilot basis, the <u>The</u> chancellor of the University of Wisconsin–Parkside.

SECTION 1531h. 118.40 (2r) (b) 1. e. of the statutes is amended to read:

118.40 (**2r**) (b) 1. e. The Gateway Each technical college district board other than the Milwaukee area technical college district board.

SECTION 1531j. 118.40 (2r) (b) 1. eg. of the statutes is created to read:

118.40 (**2r**) (b) 1. eg. The chancellor of any institution in the University of Wisconsin System other than the University of Wisconsin–Milwaukee and the University of Wisconsin–Parkside.

SECTION 1531k. 118.40 (2r) (bm) 1. of the statutes is repealed.

SECTION 1531L. 118.40 (2r) (bm) 2. of the statutes is renumbered 118.40 (2r) (bm).

SECTION 1531m. 118.40 (2r) (c) of the statutes is repealed.

SECTION 1531n. 118.40 (2r) (cp) of the statutes is repealed.

SECTION 1532. 118.40 (2r) (e) 2p. a. of the statutes, as affected by 2015 Wisconsin Act 55, is amended to read:

118.40 (**2r**) (e) 2p. a. Add the amounts appropriated in the current fiscal year under s. 20.255 (2), except s.

20.255 (2) (ac), (aw), (az), (bb), (dg), (dj), (fm), (fp), (fq), (fr), (fu), (k), and (m); and s. 20.505 (4) (es); and the amount, as determined by the secretary of administration, of the appropriation under s. 20.505 (4) (s) allocated for payments to telecommunications providers under contracts with school districts and cooperative educational service agencies under s. 16.971 (13), for grants to school district consortia under s. 16.997 (7), and to make educational technology teacher training grants under s. 16.996.

SECTION 1532d. 118.40 (2r) (e) 3m. of the statutes is amended to read:

118.40 (2r) (e) 3m. The department shall pay 25 percent of the total amount in September, 25 percent in December, 25 percent in February, and 25 percent in June. The department shall send the check to the operator of the charter school. The department shall include the entire amount under par. (fm) 1. in the December installment, but the payment shall be made in a separate check from the payment under this paragraph.

SECTION 1532e. 118.40 (2r) (f) 2. of the statutes is amended to read:

118.40 (**2r**) (f) 2. The department shall pay 25 percent of the total amount due to an operator of a charter school in September, 25 percent in December, 25 percent in February, and 25 percent in June. The department shall send the check to the operator of the charter school. The department shall include the entire amount under par. (fm) 2. in the December installment, but the payment shall be made in a separate check from the payment under this paragraph.

SECTION 1532f. 118.40 (2r) (fh) of the statutes is created to read:

118.40 (**2r**) (fh) Annually, on or before September 15, an operator of a charter school authorized under this subsection shall file with the department a report stating its summer daily attendance for each day of summer school for the purpose of par. (fm).

SECTION 1532g. 118.40 (2r) (fm) of the statutes is created to read:

118.40 (2r) (fm) 1. Beginning in the 2018–19 school year, in addition to the payment under par. (e) and subject to subd. 3., for a pupil attending summer school at a charter school established by or under a contract with an entity under par. (b) 1. a. to f., the department shall pay to the operator of the charter school, in the manner described in par. (e) 3m., an amount determined as follows:

a. Determine the per pupil amount under par. (e) for attending the charter school in the immediately preceding school term.

b. If the pupil attended summer school for at least 15 days of summer instruction at the charter school during that summer, multiply the amount under subd. 1. a. by 0.05.

- c. If the pupil attended summer school for less than 15 days of summer instruction at the charter school during that summer, multiply the amount under subd. 1. a. by 0.05 by the quotient determined by dividing the number of days of summer instruction the pupil attended during that summer by 15.
- 2. Beginning in the 2018–19 school year, in addition to the payment under par. (f) and subject to subd. 3., for a pupil attending summer school at a charter school established by or under a contract with an entity under par. (b) 1. g. or h., the state superintendent shall pay to the operator of the charter school, in the manner described in sub. (f) 2., an amount determined as follows:
- a. Determine the per pupil amount under par. (f) for attending the charter school in the immediately preceding school term.
- b. If the pupil attended summer school for at least 15 days of summer instruction at the charter school during that summer, multiply the amount under subd. 2. a. by 0.05.
- c. If the pupil attended summer school for less than 15 days of summer instruction at the charter school during that summer, multiply the amount under subd. 2. a. by 0.05 by the quotient determined by dividing the number of days of summer instruction the pupil attended during that summer by 15.
- 3. An operator of a charter school may receive a per pupil payment under this paragraph if all of the following are satisfied:
- a. The charter school offers no fewer than 19 summer days of instruction during that summer.
- b. Each summer day of instruction offered by the charter school under subd. 3. a. is comprised of no fewer than 270 minutes of instruction.

SECTION 1532h. 118.40 (2r) (g) 1. a. of the statutes is amended to read:

118.40 (**2r**) (g) 1. a. Determine the number of pupils residing in the school district for whom a payment is made under par. (e) to an operator of a charter school established under contract with an entity under par. (b) 1. e., eg., or f. in that school year.

SECTION 1532j. 118.40 (2r) (g) 1. bf. of the statutes is created to read:

118.40 (**2r**) (g) 1. bf. Identify the pupils residing in the school district for whom a payment is made under par. (fm) to an operator of a charter school established under contract with an entity under par. (b) 1. e. or f. in that school year.

SECTION 1532k. 118.40 (2r) (g) 1. bn. of the statutes is created to read:

118.40 (**2r**) (g) 1. bn. Sum the payments made under par. (fm) 1. for all of the pupils identified under subd. 1. bf. that school year.

SECTION 1532m. 118.40 (2r) (g) 1. df. of the statutes is created to read:

118.40 (**2r**) (g) 1. df. Identify the pupils residing in the school district for whom a payment is made under par. (fm) to an operator of a charter school established under contract with an entity under par. (b) 1. g. or h. in that school year.

SECTION 1532n. 118.40 (2r) (g) 1. dn. of the statutes is created to read:

118.40 (**2r**) (g) 1. dn. Sum the payments made under par. (fm) 2. for all of the pupils identified under subd. 1. df. that school year.

SECTION 1532p. 118.40 (2r) (g) 1. e. of the statutes is amended to read:

118.40 (**2r**) (g) 1. e. Sum the amounts determined under subd. 1. b. and, bn., d., and dn.

SECTION 1532q. 118.40 (2x) (c) of the statutes is repealed.

SECTION 1532qm. 118.40 (2x) (cm) (intro.) of the statutes, as created by 2017 Wisconsin Act 30, is amended to read:

118.40 (2x) (cm) (intro.) Notwithstanding par. (e) (b) 1.., the director may enter into a contract to establish, as a pilot project, only one recovery charter school, to be located in this state and that operates only high school grades, if the term of the contract is limited to 4 consecutive school years and the contract requires the charter school operator to do all of the following:

SECTION 1532r. 118.40 (2x) (e) 2. of the statutes, as affected by 2017 Wisconsin Act 30, is amended to read:

118.40 (2x) (e) 2. The department shall pay 25 percent of the total amount required to be paid under this paragraph in September, 25 percent in December, 25 percent in February, and 25 percent in June. The department shall send the check to the operator of the charter school. The state superintendent shall include the entire amount under par. (em) 1. in the December installment, but the payment shall be made in a separate check from the payment under this paragraph.

SECTION 1532s. 118.40 (2x) (eh) of the statutes is created to read:

118.40 (2x) (eh) Annually, on or before September 15, an operator of a charter school authorized under this subsection shall file with the department a report stating its summer daily attendance for each day of summer school for the purpose of par. (em).

SECTION 1532t. 118.40 (2x) (em) of the statutes is created to read:

118.40 (2x) (em) 1. Beginning in the 2018–19 school year, in addition to the payment under par. (e) and subject to subd. 2., for a pupil attending summer school at a charter school established under this subsection, the state superintendent shall pay to the operator of the charter school, in the manner described in par. (e) 2., the amount determined under sub. (2r) (fm) 1. for the pupil.

- 2. An operator of a charter school may receive a per pupil payment under this paragraph if all of the following are satisfied:
- a. The charter school offers no fewer than 19 summer days of instruction during that summer.
- b. Each summer day of instruction offered by the charter school under subd. 2. a. is comprised of no fewer than 270 minutes of instruction.

SECTION 1532u. 118.40 (2x) (f) 1. c., d. and e. of the statutes are created to read:

118.40 (2x) (f) 1. c. Identify the pupils residing in the school district for whom a payment is made under par. (em) 1. in that school year.

- d. Sum the payments made under par. (em) 1. for all of the pupils identified under subd. 1. c. that school year.
- e. Sum the amounts determined under subd. 1. b. and d.

SECTION 1532v. 118.40 (3) (g) 1. of the statutes is amended to read:

118.40 (3) (g) 1. Except as provided in subds. 2. and. 3., and 4. and sub. (4) (ar) 1., a contract with a school board, an entity under sub. (2r) (b), or the director under sub. (2x) shall require that if the capacity of the charter school is insufficient to accept all pupils who apply, the charter school shall accept pupils at random.

SECTION 1532vg. 118.40 (3) (g) 4. of the statutes is created to read:

118.40 (3) (g) 4. A charter school established under a contract with a union high school district under sub. (2) or (2m) may give preference in enrollment to pupils who were enrolled during the previous school year in a charter school operating under a cooperative agreement with the charter school established under a contract with a union high school district.

SECTION 1532vr. 118.40 (3m) (b) of the statutes is amended to read:

118.40 (3m) (b) When contracting for the establishment of a charter school under this section, adhere to consider the principles and standards for quality charter schools established by the National Association of Charter School Authorizers.

SECTION 1532w. 118.40 (7) (am) 2. of the statutes is amended to read:

118.40 (7) (am) 2. A charter school established under sub. (2r) or (2x) or a private school located in the school district operating under ch. 119 that is converted to a charter school is not an instrumentality of any school district and no school board may employ any personnel for the charter school. If the chancellor of an institution in the University of Wisconsin-Parkside Wisconsin System other than the University of Wisconsin Milwaukee contracts for the establishment of a charter school under sub. (2r), the board of regents of the University of Wisconsin System may employ instructional staff for the charter school. If the Gateway a technical college district board other than the Milwaukee area technical college district board contracts for the establishment of a charter school under sub. (2r), Gateway the technical college may employ instructional staff for the charter school.

SECTION 1532x. 118.40 (8) (b) 1. of the statutes is amended to read:

118.40 (8) (b) 1. The governing body of a virtual charter school shall assign an appropriately licensed teacher for each online course offered by the virtual charter school. No In this subsection, an individual who holds a license or permit to teach a subject and level in the state from which the online course is provided is appropriately licensed to teach the subject and level in this state.

1r. The governing body of a virtual charter school may not permit a person holding only a permit to teach exclusively in a charter school may to teach in a virtual charter school, and no.

1g. The governing body of a virtual charter school may not permit a person holding both a license to teach exclusively in a charter school and a license to teach in other public schools may to teach, in a the virtual charter school, a subject or at-a level that is not authorized by the latter license.

SECTION 1534f. 118.50 (2) (a) of the statutes is amended to read:

118.50 (2) (a) A school board may not enter into, extend, or renew a whole grade sharing agreement after January 10 February 15 of the school year preceding the school year in which the agreement, extension, or renewal takes effect.

SECTION 1534g. 118.50 (2) (b) of the statutes is amended to read:

118.50 (2) (b) At least 150 60 days before entering into, extending, or renewing a whole grade sharing agreement, the school board shall adopt a resolution stating its intention to do so. Within 10 days after adoption of the resolution, the school district clerk shall publish notice of the adoption of the resolution as a class 1 notice under ch. 985 in a newspaper published in the school district or post a notice of the adoption of the resolution as provided in s. 10.05.

SECTION 1534p. 118.50 (5m) of the statutes is cre- **Vetoed** ated to read:

In Part

- 118.50 (5m) STATE AID. (a) Beginning in the 2018-19 school year and subject to par. (c), the department shall pay to a school board that enters into a whole grade sharing agreement the following amounts:
- 1. In each of the first 4 school years of the whole grade sharing agreement, \$150 multiplied by the number of pupils who, during the first school year of the whole grade sharing agreement, are enrolled in the school district in a grade level that is subject to the whole grade sharing agreement.
- 2. Subject to par. (b), in the 5th school year of the whole grade sharing agreement, 50 percent of the amount calculated under subd. 1.

Vetoed In Part

- (b) If, before the 5th school year of a whole grade sharing agreement, 2 or more school boards participating in the whole grade sharing agreement adopt resolutions ordering that the school districts be consolidated under s. 117.08 or 117.09 and the school boards are following the consolidation procedures under s. 117.08 or 117.09, the department shall, during the 5th and 6th school years of the whole grade sharing agreement, pay each school board that passed a resolution to consolidate the amount calculated under par. (a) 1. for that school board.
- (c) 1. If the appropriation under s. 20.255 (2) (bp) is insufficient to pay the full amount under this subsection, the funds shall be prorated among the entitled school boards.
- 2. Paragraph (a) applies to an original whole grade sharing agreement. If a whole grade sharing agreement is extended or renewed under this section, the additional school years are considered to be part of the original whole grade sharing agreement. The department shall consider a whole grade sharing agreement entered into between school boards that contains substantially similar terms to an expired whole grade sharing agreement, including that the same grades are subject to both agreements, to be an extension of the expired whole grade sharing agreement.

SECTION 1534s. 118.51 (1) (aj) of the statutes is created to read:

118.51 (1) (aj) "Free appropriate public education" has the meaning given in s. 115.76 (7).

SECTION 1535a. 118.51 (9) of the statutes is amended to read:

118.51 (9) APPEAL OF REJECTION. If the nonresident school board rejects an application under sub. (3) (a) or (7), the resident school board prohibits a pupil from attending public school in a nonresident school district under sub. (3m) (d), (6), or (7), or the nonresident school board prohibits a pupil from attending public school in the nonresident school district under sub. (11), the pupil's parent may appeal the decision to the department within 30 days after the decision. If the nonresident school board provides notice that the special education or related service is not available under sub. (12) (b), the pupil's parent may appeal the required transfer to the department within 30 days after receipt of the notice. The department shall affirm the school board's decision unless the department finds that the decision was arbitrary or unreasonable.

SECTION 1535c. 118.51 (12) (title) of the statutes is amended to read:

118.51 (12) (title) SPECIAL NONRESIDENT SCHOOL DISTRICT STATEMENT OF EDUCATIONAL COSTS; SPECIAL EDUCATION OR RELATED SERVICES.

SECTION 1535d. 118.51 (12) of the statutes is renumbered 118.51 (12) (b).

SECTION 1535e. 118.51 (12) (a) of the statutes is created to read:

118.51 (12) (a) Beginning in the 2018–19 school year, at the end of a school year in which a child with a disability who attends a nonresident school district under this section and receives special education or related services under subch. V of ch. 115 in the nonresident school district, the nonresident school board may submit to the department a financial statement that shows the actual costs the nonresident school board incurred to provide a free appropriate public education to the child during that school year. The department shall provide the resident school board with a copy of any financial statement it receives under this paragraph.

SECTION 1535g. 118.51 (16) (a) 1. of the statutes is amended to read:

118.51 (16) (a) 1. For each school district, the number of nonresident pupils attending public school in the school district under this section, other than pupils for whom a payment is made under sub. (17) (a) or, (c), or (cm).

SECTION 1535h. 118.51 (16) (a) 2. of the statutes is amended to read:

118.51 (16) (a) 2. For each school district, the number of resident pupils attending public school in a nonresident school district under this section, other than pupils for whom a payment is made under sub. (17) (a) ΘF_{\bullet} (c) or (cm).

SECTION 1535ig. 118.51 (16) (a) 3. b. of the statutes is amended to read:

118.51 (16) (a) 3. b. Beginning with the amount in the 2015–16 school year and, except as provided in subd. 3. c., in each school year thereafter, the sum of the amount determined under this subdivision for the previous school year; the amount of the per pupil revenue limit adjustment under s. 121.91 (2m) for the current school year, if positive; and the change in the amount of statewide categorical aid per pupil between the previous school year and the current school year, as determined under s. 118.40 (2r) (e) 2p., if positive.

SECTION 1535im. 118.51 (16) (a) 3. c. of the statutes is created to read:

118.51 (**16**) (a) 3. c. For the amount in the 2017–18 to 2020–21 school years, the amount determined under subd. 3. b. plus \$100.

SECTION 1535j. 118.51 (16) (c) of the statutes is amended to read:

118.51 (16) (c) If a pupil attends public school in a nonresident school district under this section for less than a full school term, the department shall prorate the state aid adjustments under this subsection and sub. (17) (c) and (cm) based on the number of days that school is in session and the pupil attends public school in the nonresident school district.

SECTION 1535k. 118.51 (16) (d) of the statutes is amended to read:

118.51 (16) (d) The department shall ensure that the aid adjustments under par. (b) and sub. (17) (c) and (cm)

do not affect the amount determined to be received by a school district as state aid under s. 121.08 for any other purpose.

SECTION 1535m. 118.51 (17) (title) of the statutes is amended to read:

118.51 (17) (title) STATE AID ADJUSTMENTS AND TUITION PUPIL TRANSFER AMOUNT AND PAYMENTS TO A NONRESIDENT SCHOOL BOARD; CHILDREN WITH DISABILITIES

SECTION 15350. 118.51 (17) (b) 2. b. of the statutes is amended to read:

118.51 (17) (b) 2. b. Beginning in In the 2017–18 school year, the per pupil transfer amount is the sum of the per pupil transfer amount for the previous school year; the amount of the per pupil revenue limit adjustment under s. 121.91 (2m) for the current school year, if positive; and the change in the amount of statewide categorical aid per pupil between the previous school year and the current school year, as determined under s. 118.40 (2r) (e) 2p., if positive.

SECTION 1535p. 118.51 (17) (b) 2. c. of the statutes is created to read:

118.51 (17) (b) 2. c. Beginning in the 2018–19 school year, and subject to subd. 3., the per pupil transfer amount is the sum of the per pupil transfer amount for the previous school year; the amount of the per pupil revenue limit adjustment under s. 121.91 (2m) for the current school year, if positive; and the change in the amount of statewide categorical aid per pupil between the previous school year and the current school year, as determined under s. 118.40 (2r) (e) 2p., if positive.

SECTION 1535q. 118.51 (17) (b) 3. of the statutes is created to read:

118.51 (17) (b) 3. Beginning in the 2019–20 school year, if a nonresident school board submitted a financial statement for a child with a disability under sub. (12) (a) in the previous school year, the per pupil transfer amount for that child is the amount shown on the financial statement for that child for the previous school year, up to \$30,000.

SECTION 1535r. 118.51 (17) (bm) of the statutes is created to read:

118.51 (17) (bm) 1. Beginning in the 2019–20 school year, the department shall determine all of the following for each school district:

- a. The number of nonresident pupils attending public school in the school district under this section who are receiving special education or related services under subch. V of ch. 115 and for whom no financial statement was submitted under sub. (12) (a) in that school year.
- b. The amount shown on each financial statement submitted in that school year under sub. (12) (a) by the school board of that school district for a nonresident pupil who is attending public school in the school district under this section and receiving special education or related services under subch. V of ch. 115. If the amount shown

- on any financial statement described in this subd. 1. b. exceeds \$30,000, for the purpose of subd. 2. b., the department shall identify the amount shown as \$30,000.
- c. The number of resident pupils attending public school in a nonresident school district under this section who are receiving special education or related services under subch. V of ch. 115 and for whom the nonresident school board did not submit a financial statement under sub. (12) (a) in that school year.
- d. The amount shown on each financial statement submitted in that school year under sub. (12) (a) for a pupil who is a resident of the school district and who is attending public school in a nonresident school district under this section and receiving special education or related services under subch. V of ch. 115. If the amount shown on any financial statement described in this subd. 1. d. exceeds \$30,000, for the purpose of subd. 2. e., the department shall identify the amount shown as \$30,000.
- 2. Beginning in the 2019–20 school year, for each school district, the department shall do all of the following:
- a. Multiply the number under subd. 1. a. by the per pupil transfer amount under par. (b) 2. c.
- b. Determine the sum of the amounts shown on the financial statements submitted as described under subd. 1. b.
- c. Determine the sum of the product under subd. 2. a. and the sum under subd. 2. b.
- d. Multiply the number under subd. 1. c. by the per pupil transfer amount under par. (b) 2. c.
- e. Determine the sum of the amounts shown on the financial statements submitted as described under subd. 1. d.
- f. Determine the sum of the product under subd. 2. d. and the sum under subd. 2. e.

SECTION 1535t. 118.51 (17) (c) of the statutes is amended to read:

118.51 (17) (c) 1. If the number determined in par. (b) 1. a. is greater than the number determined in par. (b) 1. b. for a school district, beginning in the 2016–17, 2017–18, and 2018–19 school year years, the department shall increase that school district's state aid payment under s. 121.08 by an amount equal to the difference multiplied by the an amount under par. (b) 2. a., b., or c.

2. If the number determined in par. (b) 1. a. is less than the number determined in par. (b) 1. b. for a school district, beginning in the 2016–17, 2017–18, and 2018–19 school year years, the department shall decrease that school district's state aid payment under s. 121.08 by an amount equal to the difference multiplied by the an amount under par. (b) 2. a., b., or c. If the state aid payment under s. 121.08 is insufficient to cover the reduction, the department shall decrease other state aid payments made by the department to the school district by the remaining amount. If the state aid payment under s. 121.08 and other state aid payments made by the depart-

ment to the school district are insufficient to cover the reduction, the department shall use the moneys appropriated under s. 20.255 (2) (cg) to pay the balance to school districts under subd. 1.

SECTION 1535v. 118.51 (17) (cm) of the statutes is created to read:

118.51 (17) (cm) 1. If the amount determined under par. (bm) 2. c. is greater than the amount determined under par. (bm) 2. f. for a school district, beginning in the 2019–20 school year, the department shall increase that school district's state aid payment under s. 121.08 by an amount equal to the difference.

2. If the amount determined under par. (bm) 2. c. is less than the amount determined under par. (bm) 2. f. for a school district, beginning in the 2019–20 school year, the department shall decrease that school district's state aid payment under s. 121.08 by an amount equal to the difference. If the state aid payment under s. 121.08 is insufficient to cover the reduction, the department shall decrease other state aid payments made by the department to the school district by the remaining amount. If the state aid payment under s. 121.08 and other state aid payments made by the department to the school district are insufficient to cover the reduction, the department shall use the moneys appropriated under s. 20.255 (2) (cg) to pay the balance to school districts under subd. 1.

SECTION 1536. 118.52 (title) of the statutes is repealed and recreated to read:

118.52 (title) Part-time open enrollment.

SECTION 1537. 118.52 (1) (am) of the statutes is repealed.

SECTION 1538. 118.52 (2) of the statutes is amended to read:

118.52 (2) APPLICABILITY. —A—Beginning in the 2018—19 school year, a pupil enrolled in a public school in the high school grades may attend an educational institution public school in a nonresident school district under this section for the purpose of taking a course offered by the educational institution nonresident school district. A pupil may attend no more than 2 courses at any time at educational institutions in nonresident school districts under this section.

SECTION 1539. 118.52 (3) (a) of the statutes is amended to read:

118.52 (3) (a) The parent of a pupil who wishes to attend an educational institution public school in a non-resident school district for the purpose of taking a course under this section shall submit an application, on a form provided by the department, to the educational institution at school board of the nonresident school district in which the pupil wishes to attend a course not later than 6 weeks prior to the date on which the course is scheduled to commence. The application shall specify the course that the pupil wishes to attend and may specify the school or schools at which the pupil wishes to attend the course. The educational institution nonresident school board

shall send a copy of the application to the pupil's resident school board, except that if the pupil is attending a school in a school district other than the pupil's resident school district pursuant to a whole grade sharing agreement under s. 118.50, the educational institution nonresident school district to which the pupil applies under this section shall send a copy of the application to the school board of the district in which the pupil is attending school pursuant to the whole grade sharing agreement.

SECTION 1540. 118.52 (3) (b) of the statutes is amended to read:

118.52 (3) (b) If an educational institution a nonresident school board receives more applications for a particular course than there are spaces available in the course, the educational institution nonresident school board shall determine which pupils to accept on a random basis.

SECTION 1541. 118.52 (3) (c) of the statutes is amended to read:

118.52(3) (c) No later than one week prior to the date on which the course is scheduled to commence, the educational institution nonresident school board shall notify the applicant and the resident school board, in writing, whether the application has been accepted and, if the application is accepted, the school at which the pupil may attend the course. If the applicant pupil is attending a school in a school district other than the pupil's resident school district pursuant to a whole grade sharing agreement under s. 118.50, the educational institution school board of the district to which the pupil applies under this section shall provide the notice required under this paragraph to the school board of the district in which the pupil is attending school pursuant to the whole grade sharing agreement. The acceptance applies only for the following semester, school year, or other session in which the course is offered. If the educational institution school board of the district to which the pupil applies under this section rejects an application, it shall include in the notice the reason for the rejection.

SECTION 1542. 118.52 (3) (d) 1. of the statutes is amended to read:

118.52 (3) (d) 1. If it denies an application to attend an educational institution public school in a nonresident school district under sub. (6), notify the applicant and the educational institution nonresident school board, in writing, that the application has been denied and include in the notice the reason for the rejection.

SECTION 1543. 118.52 (3) (e) of the statutes is amended to read:

118.52 (3) (e) Following receipt of a notice of acceptance but prior to the date on which the course is scheduled to commence, the pupil's parent shall notify the resident school board, or, if the pupil is attending school in a school district other than the pupil's resident school district pursuant to a whole grade sharing agreement under s. 118.50, the school board of the district in which the pupil is attending school, and the educational institution

school board of the district to which the pupil applies under this section of the pupil's intent to attend the course at in the educational institution school district to which the pupil applies under this section.

SECTION 1544. 118.52 (6) (a) of the statutes is amended to read:

118.52 (6) (a) *Individualized education program requirements*. The school board of a pupil's resident school district, or, if the pupil is attending school in a school district other than the pupil's resident school district pursuant to a whole grade sharing agreement under s. 118.50, the school board of the district in which the pupil is attending school, shall reject a pupil's application to attend a course at an educational institution in a public school in a nonresident school district under this section if the resident school board or the school board of the district in which the pupil is attending school, respectively, determines that the course conflicts with the individualized education program for the pupil under s. 115.787 (2).

SECTION 1545. 118.52 (6) (b) of the statutes is created to read:

118.52 (6) (b) *Undue financial burden*. The school board of a pupil's resident school district may reject an application to attend a course in a public school in a non-resident school district if the cost of the course would impose upon the resident school district an undue financial burden in light of the resident school district's total economic circumstances, including its revenue limit under subch. VII of ch. 121, its ability to pay tuition costs for the pupil, and the per pupil cost for children continuing to be served by the resident school district.

SECTION 1546. 118.52 (6) (c) of the statutes is repealed.

SECTION 1547. 118.52 (8) of the statutes is amended to read:

118.52 (8) APPEAL OF REJECTION. If an application is rejected under sub. (3) (c) or a pupil is prohibited from attending a course at an educational institution in a public school in a nonresident school district under sub. (6), the pupil's parent may appeal the decision to the department within 30 days after the decision. The department shall affirm the decision unless the department finds that the decision was arbitrary or unreasonable. The department's decision is final and is not subject to judicial review under subch. III of ch. 227.

SECTION 1548. 118.52 (9) of the statutes is amended to read:

118.52 (9) RIGHTS AND PRIVILEGES OF NONRESIDENT PUPILS. A pupil attending a course at an educational institution in a public school in a nonresident school district under this section has all of the rights and privileges of other pupils attending the educational institution residing in that school district and is subject to the same rules and regulations as those pupils residing in that school district.

SECTION 1549. 118.52 (10) of the statutes is amended to read:

118.52 (10) DISCIPLINARY RECORDS. Notwithstanding s. 118.125, the resident school board shall provide to the educational institution nonresident school board to which a pupil has applied under this section, upon request by that educational institution school board, a copy of any expulsion findings and orders, a copy of records of any pending disciplinary proceeding involving the pupil, a written explanation of the reasons for the expulsion or pending disciplinary proceeding, and the length of the term of the expulsion or the possible outcomes of the pending disciplinary proceeding.

SECTION 1550. 118.52 (11) (a) of the statutes is amended to read:

118.52 (11) (a) *Responsibility*. The parent of a pupil attending a course at an educational institution in a public school in a nonresident school district under this section is responsible for transporting the pupil to and from the course that the pupil is attending.

SECTION 1551. 118.52 (11) (b) of the statutes is amended to read:

118.52 (11) (b) Low-income assistance. The parent of a pupil who is attending a course at an educational institution in a public school in a nonresident school district under this section may apply to the department for reimbursement of the costs incurred by the parent for the transportation of the pupil to and from the pupil's residence or school in which the pupil is enrolled and the educational institution that school at which the pupil is attending for the course if the pupil and parent are unable to pay the cost of such transportation. The department shall determine the reimbursement amount and shall pay the amount from the appropriation under s. 20.255 (2) (cy). The department shall give preference under this paragraph to those pupils who satisfy the income eligibility criteria for a free or reduced-price lunch under 42 USC 1758 (b) (1).

SECTION 1552. 118.52 (12) (a) of the statutes is renumbered 118.52 (12) and amended to read:

118.52 (12) The resident school board shall pay to the educational institution nonresident school board, for each resident pupil attending a course at the educational institution in a public school in the nonresident school district under this section, an amount equal to the cost of providing the course to the pupil, calculated in a manner determined by the department. Except as provided in par. (b), the educational institution may not charge to or receive from the pupil or the pupil's resident school board any additional payment for a pupil attending a course at the educational institution under this section.

SECTION 1553. 118.52 (12) (b) of the statutes is repealed.

SECTION 1554. 118.55 (title) of the statutes is amended to read:

 $118.55 \ (title)$ Youth options Early college credit program.

SECTION 1554d. 118.55 (1) of the statutes is renumbered 118.55 (1) (intro.) and amended to read:

118.55 (1) Definition Definitions. (intro.) In this section, "institution:

(a) "Institution of higher education" means an all of the following:

1. An institution within the University of Wisconsin System, or a tribally controlled college or a.

<u>2. A</u> private, nonprofit institution of higher education located in this state.

SECTION 1554h. 118.55 (1) (b) of the statutes is created to read:

118.55 (1) (b) "Governing body of a private school" means a board elected or appointed to govern a private school or, if no board is appointed or elected to govern the school, any other person having direct charge of the private school.

SECTION 1554p. 118.55 (1) (c) of the statutes is created to read:

118.55 (1) (c) "Participating private school" means a private school attended by a pupil who has applied to take or is taking a course under this section.

SECTION 1556. 118.55 (2) (a) of the statutes is amended to read:

118.55 (2) (a) Subject to par. (b) and sub. (7t) (c), any public high school pupil enrolled in the 11th or 12th grade who is not attending a technical college under sub. (7r) or s. 38.12 (14) or 118.15 (1) (b) and any high school pupil attending a private school may enroll in an institution of higher education for the purpose of taking one or more nonsectarian courses at the institution of higher education, including during a summer semester or session. The pupil shall submit an application to the institution of higher education in the previous school semester. The pupil shall indicate on the application whether he or she will be taking the course or courses for high school credit or postsecondary credit or both, if applicable. The pupil shall also specify on the application that, if he or she is admitted, the institution of higher education may disclose the pupil's grades, the courses that he or she is taking, and his or her attendance record to the public or private school in which the pupil is enrolled.

SECTION 1556m. 118.55 (3) of the statutes is amended to read:

118.55 (3) NOTIFICATION OF SCHOOL BOARD; DETERMINATION OF HIGH SCHOOL CREDIT; NOTIFICATION OF POSTSECONDARY CREDIT. (a) A public school pupil who intends to enroll in an institution of higher education under this section shall notify the school board of the school district in which he or she is enrolled and a pupil attending a private school who intends to enroll in an institution of higher education under this section shall notify the governing body of the private school he or she attends of that intention no later than March 1 if the pupil intends to enroll in the fall semester, and no later than October 1 if the pupil intends to enroll in the spring semester. The

notice shall include the titles of the courses in which the pupil intends to enroll and the number of credits of each course, and shall specify whether the pupil will be taking the courses for high school or postsecondary credit.

(b) If the <u>public school</u> pupil specifies in the notice under par. (a) that he or she intends to take a course at an institution of higher education for high school credit, the school board shall determine whether the course is comparable to a course offered in the school district, and whether the course satisfies any of the high school graduation requirements under s. 118.33 and the number of high school credits to award the pupil for the course, if any. If the pupil attending a private school specifies in the notice under par. (a) that he or she intends to take a course at an institution of higher education for high school credit, the governing body of the participating private school shall determine whether the course is comparable to a course offered at the private school, whether the course satisfies any requirements necessary for high school graduation, and the number of high school credits to award the pupil for the course, if any. In cooperation with institutions of higher education, the state superintendent shall develop guidelines to assist school districts and participating private schools in making the determinations. The school board or governing body shall notify the pupil of its determinations, in writing, before the beginning of the semester in which the pupil will be enrolled. If the public school pupil disagrees with the school board's decision regarding comparability of courses, satisfaction of high school graduation requirements or the number of high school credits to be awarded, the pupil may appeal the school board's decision to the state superintendent within 30 days after the decision. The state superintendent's decision shall be final and is not subject to review under subch. III of ch. 227. If the pupil attending a participating private school disagrees with any decision of a governing body under this paragraph, the pupil may appeal the decision to the governing body within 30 days after the decision.

SECTION 1558. 118.55 (3) (c) of the statutes is created to read:

118.55 (3) (c) If the pupil specifies in the notice under par. (a) that he or she intends to take a course for postsecondary credit at an institution of higher education that is within the University of Wisconsin System, the board of regents of the University of Wisconsin System shall notify the pupil whether credits earned for the course are transferable between and within institutions within the system.

SECTION 1559. 118.55 (4) (a) of the statutes is renumbered 118.55 (4) (a) (intro.) and amended to read:

118.55 (4) (a) (intro.) An institution of higher education may shall admit a pupil to attend a course under this section only if it has all of the following apply:

2. There is space available in the course.

SECTION 1560. 118.55 (4) (a) 1. of the statutes is created to read:

118.55 (4) (a) 1. The pupil meets the requirements and prerequisites of the course.

SECTION 1560g. 118.55 (4) (b) of the statutes is amended to read:

118.55 (4) (b) If an institution of higher education admits a pupil, it shall notify the school board of the school district in which the pupil is enrolled or the governing body of the pupil's participating private school, in writing, within 30 days after the beginning of classes at the institution of higher education. The notification shall include the course or courses in which the pupil is enrolled.

SECTION 1560r. 118.55 (4) (c) of the statutes is amended to read:

118.55 (4) (c) If a pupil is not admitted to attend the course that he or she specified in the notice under sub. (3) (a) but is admitted to attend a different course, the pupil shall immediately notify the school board of the school district in which he or she is enrolled or the governing body of the pupil's participating private school and the school board or governing body shall inform the pupil of its determinations under sub. (3) (b) regarding the course to which the pupil was admitted as soon as practicable.

SECTION 1561. 118.55 (5) (intro.) of the statutes is amended to read:

118.55 (5) PAYMENT RESPONSIBILITY FOR AND DETERMINATION OF COSTS: PAYMENT AND REIMBURSEMENT FOR CERTAIN COSTS. (intro.) Subject to sub. (7t), within 30 days after the end of the semester, the school board of the school district in which a pupil attending an institution of higher education under this section is enrolled and the governing body of the participating private school attended by a pupil who is attending an institution of higher education under this section shall pay the institution of higher education, on behalf of the pupil, be responsible for the following amount for any course that is taken for high school credit and that is not comparable to a course offered in the school district:

SECTION 1561m. 118.55 (5) (a) of the statutes is amended to read:

118.55 (5) (a) If the pupil is attending an institution within the University of Wisconsin System, taking a course for high school credit, regardless of whether the course is also taken for postsecondary credit, and if the course is not comparable to a course offered in the school district, 75 percent of the actual cost of tuition, fees, books and other necessary materials directly related to for the course, as determined under par. (d). If a pupil attending a private school is taking a course for high school credit, regardless of whether the course is also taken for postsecondary credit, and if the course is not comparable to a course offered by the participating private school, 75 percent of the actual cost of tuition for the course, as determined under par. (d). If the pupil takes a

course described under this paragraph at a high school in a school district or at a participating private school, the school board of the school district or the governing body of the participating private school shall be responsible for the costs of books and other necessary materials for the course.

SECTION 1563. 118.55 (5) (b) of the statutes is created to read:

118.55 (5) (b) If the pupil is taking a course for post-secondary credit and if the course is not comparable to a course offered in the school district, 25 percent of the actual cost of tuition for the course, as determined under par. (d).

SECTION 1564. 118.55 (5) (c) of the statutes is repealed.

SECTION 1564m. 118.55 (5) (d) 1m. of the statutes is created to read:

118.55 (5) (d) 1m. For an institution of higher education under sub. (1) (a) that is a University of Wisconsin college campus, as defined in s. 36.05 (6m), one–half of the amount that would be charged for each credit assigned to the course to an individual who is a resident of this state and who is enrolled in the college campus as an undergraduate student. Subject to sub. (7t), neither the college campus nor the school board may charge any additional costs or fees to a pupil to attend a course under this section.

SECTION 1565. 118.55 (5) (d) 2. of the statutes is created to read:

118.55 (5) (d) 2. For an institution of higher education under sub. (1) (a) 2., one—third of the amount that would be charged for each credit assigned to a similar course offered by the University of Wisconsin—Madison to an individual who is a resident of this state and who is enrolled at the University of Wisconsin—Madison as an undergraduate student. Subject to sub. (7t), neither the institution of higher education nor the school board may charge any additional costs or fees to a pupil to attend a course under this section.

SECTION 1566. 118.55 (5) (e) of the statutes is created to read:

118.55 (5) (e) 1. Subject to sub. (7t), within 30 days after the end of the semester, the school board of the school district in which a pupil who attended an institution of higher education under this section was enrolled and the governing body of a participating private school attended by a pupil who attended the institution of higher education under this section shall pay the institution, on behalf of the pupil, the amount determined under par. (d) and shall submit an itemized report to the department of the amounts paid under this subdivision.

2. Subject to subd. 3., from the appropriation under s. 20.445 (1) (d), the secretary of the department of workforce development shall, on behalf of the school board of a school district in which a pupil who attended an institution of higher education under this section was enrolled

and on behalf of the governing body of a participating private school and a pupil who attended the private school and who attended an institution of higher education under this section, pay to the department of public instruction the following amount:

- a. For a pupil who took a course for high school credit, as described in par. (a), 25 percent of the actual cost of tuition for the course, as determined under par. (d). The department of public instruction shall reimburse the school board of the school district or the governing body of the private school the amount received from the department of workforce development under this subd. 2. a.
- b. For a pupil who took a course for postsecondary credit, as described in par. (b), 50 percent of the actual cost of tuition for the course, as determined under par. (d). The department of public instruction shall reimburse the school board of the school district or the governing body of the private school the amount received from the department of workforce development under this subd. 2. b.
- 3. If the appropriation under s. 20.445 (1) (d) in any fiscal year is insufficient to reimburse all school districts and all governing bodies eligible for the full amount of reimbursable tuition costs under subd. 2., the secretary of the department of workforce development shall notify the state superintendent, who shall prorate the amount of the payments under subd. 2. among eligible school districts and governing bodies.

SECTION 1566g. 118.55 (6) (a) of the statutes is amended to read:

118.55 (6) (a) Subject to sub. (7t), a pupil taking a course at an institution of higher education for high school credit under this section is not responsible for any portion of the tuition and fees for the course if the school board, or the state superintendent on appeal under sub. (3) (b), the governing body of the participating private school, or the governing body on appeal under sub. (3) (b) has determined that the course is not comparable to a course offered in the school district or at the participating private school, whichever is applicable.

SECTION 1566r. 118.55 (6) (b) of the statutes is amended to read:

118.55 (6) (b) A pupil taking a course at an institution of higher education for high school credit under this section is responsible for the tuition and fees for the course if the school board or the governing body of the participating private school has determined that the course is comparable to a course offered in the school district or at the participating private school, unless the state superintendent or the governing body reverses the decision of the school board's decision board or governing body, respectively, on appeal under sub. (3) (b).

SECTION 1567. 118.55 (6) (c) of the statutes is renumbered 118.55 (6) (c) 1. and amended to read:

118.55 (6) (c) 1. A Except as provided in subd. 2., a pupil taking a course under this section at an institution of higher education only for postsecondary credit under this section is responsible for 25 percent of the actual cost of tuition and fees for the course, as determined under sub. (5) (d). The school board of the school district in which the pupil attending an institution under this section is enrolled and the governing body of a participating private school attended by a pupil attending an institution of higher education under this section shall establish a written policy governing the timing and method for recovering from the pupil or the pupil's parent or guardian the pupil's share of tuition as specified in this subdivision.

SECTION 1568. 118.55 (6) (c) 2. of the statutes is created to read:

118.55 (6) (c) 2. The school board or the governing body of the participating private school shall waive the pupil's responsibility for costs under subd. 1. if the department determines that the cost of the course would pose an undue financial burden on the pupil's family.

SECTION 1569. 118.55 (7g) of the statutes is amended to read:

118.55 (7g) Transportation. The parent or guardian of a pupil who is attending an institution of higher education or technical college under this section and is taking a course for high school credit may apply to the state superintendent for reimbursement of the cost of transporting the pupil between the high school or participating private school in which the pupil is enrolled and the institution of higher education or technical college that the pupil is attending if the pupil and the pupil's parent or guardian are unable to pay the cost of such transportation. The state superintendent shall determine the reimbursement amount and shall pay the amount from the appropriation under s. 20.255 (2) (cw) (cy). The state superintendent shall give preference under this subsection to those pupils who satisfy the income eligibility criteria for a free or reduced-price lunch under 42 USC 1758 (b) (1).

SECTION 1570b. 118.55 (7r) (title) of the statutes is renumbered 38.12 (14) (title).

SECTION 1570c. 118.55 (7r) (a) (intro.) of the statutes is renumbered 38.12 (14) (a) (intro.).

SECTION 1570e. 118.55 (7r) (a) 1. of the statutes is renumbered 38.12 (14) (a) 1.

SECTION 1570g. 118.55 (7r) (a) 2. of the statutes is renumbered 38.12 (14) (a) 2.

SECTION 1570i. 118.55 (7r) (a) 3. of the statutes is renumbered 38.12 (14) (a) 3. and amended to read:

38.12 (14) (a) 3. The pupil notifies the school board of the school district in which the pupil resides of his or her intent to attend a technical college under this subsection by March 1 if the pupil intends to enroll in the fall semester and by October 1 if the pupil intends to enroll in the spring semester.

SECTION 1570j. 118.55 (7r) (a) 4. of the statutes is renumbered 38.12 (14) (a) 4.

SECTION 1570n. 118.55 (7r) (a) 5. of the statutes is renumbered 38.12 (14) (a) 5. and amended to read:

38.12 (14) (a) 5. The pupil is not ineligible under subsection. State of the program under this section.

SECTION 1570p. 118.55 (7r) (am) of the statutes is renumbered 38.12 (14) (am).

SECTION 1570r. 118.55 (7r) (b) of the statutes is renumbered 38.12 (14) (b).

SECTION 1570t. 118.55 (7r) (c) of the statutes is renumbered 38.12 (14) (c).

SECTION 1570v. 118.55 (7r) (d) of the statutes is renumbered 38.12 (14) (d) and amended to read:

38.12 (14) (d) Subject to sub. s. 118.55 (7t), for each pupil attending a technical college under this subsection, the school board shall pay to the technical college district board, in 2 installments payable upon initial enrollment and at the end of the semester, for those courses taken for high school credit, an amount equal to the cost of tuition, course fees, and books for the that a pupil at who is attending the technical college and who is a resident of this state would be charged, except that the school board is not responsible for payment for any courses that are comparable to courses offered in the school district.

SECTION 1570w. 118.55 (7r) (dm) of the statutes is renumbered 38.12 (14) (dm).

SECTION 1570x. 118.55 (7r) (e) of the statutes is renumbered 38.12 (14) (e) and amended to read:

38.12 (14) (e) The school board of the school district in which the pupil resides is not responsible for transporting a pupil attending a technical college under this subsection to or from the technical college that the pupil is attending.

SECTION 1570y. 118.55 (7r) (f) of the statutes is renumbered 38.12 (14) (f).

SECTION 1571. 118.55 (7t) (a) of the statutes is amended to read:

118.55 (7t) (a) A school board <u>or the governing body</u> <u>of a participating private school</u> may establish a written policy limiting the number of credits for which the school board <u>or governing body</u> will pay under <u>subs. sub.</u> (5) and (7r) <u>s. 38.12 (14)</u> (d) to the equivalent of 18 postsecondary semester credits per pupil.

SECTION 1572. 118.55 (7t) (b) of the statutes is renumbered 118.55 (5) (d) (intro.) and amended to read:

118.55 (5) (d) (intro.) If a school board <u>or the governing body of a participating private school</u> is required to pay tuition and fees on behalf of a pupil under sub. (5) (a) or (c) 1. or (7r) (d) this subsection, the tuition and fees charged for each credit assigned to the course may not exceed the following:

1. For an institution of higher education under sub. (1) (a) 1., other than a University of Wisconsin college campus, as defined in s. 36.05 (6m), one-third of the

amount that would be charged <u>-a pupil for each credit</u> assigned to the course to an individual who is a resident of this state and who is enrolled in the educational institution as an undergraduate student. Subject to sub. (7t), neither the institution of higher education nor the school board nor the governing body may charge any additional costs or fees to a pupil to attend a course under this section.

SECTION 1572m. 118.55 (7t) (c) of the statutes is amended to read:

118.55 (7t) (c) If a pupil receives a failing grade in a course, or fails to complete a course, at an institution of higher education or technical college for which the school board or the governing body of a participating private school has made payment, the pupil's parent or guardian, or the pupil if he or she is an adult, shall reimburse the school board or the governing body the amount paid on the pupil's behalf upon the request of the school board's request board or governing body. If a school board or governing body that requests reimbursement of a payment made under this section is not reimbursed as requested, the pupil on whose behalf the payment was made is ineligible for any further participation in the program under this section. For the purposes of this paragraph, a grade that constitutes a failing grade for a course offered in the school district or at the participating private school constitutes a failing grade for a course taken at an institution of higher education or technical college under this section.

SECTION 1573. 118.55 (8) (title) of the statutes is amended to read:

118.55 **(8)** (title) Program information: <u>Agree-Ment.</u>

SECTION 1574. 118.55 (8) of the statutes is renumbered 118.55 (8) (a) and amended to read:

118.55 (8) (a) Annually by October 1, each school board shall provide information about the program under this section to all pupils enrolled in the school district in the 8th, 9th, 10th, and 11th grades.

SECTION 1575. 118.55 (8) (b) of the statutes is created to read:

118.55 (8) (b) A school board or the governing body of a participating private school may enter into an agreement with an institution of higher education to facilitate the early college credit program under this section.

SECTION 1576. 118.57 (1) of the statutes is amended to read:

118.57 (1) Annually, by January 31, each school board shall publish as a class 1 notice, under ch. 985, and post on its Internet site a description of the educational options available to the child children in the school district, including public schools, private schools participating in a parental choice program, charter schools, virtual schools, full–time or part–time open enrollment, youth options, and course options in a nonresident school district, and the early college credit program.

SECTION 1577. 118.60 (1) (am) (intro.) of the statutes is renumbered 118.60 (1) (am) and amended to read:

118.60 (1) (am) "Eligible school district" means a the school district that, subject to sub. (1m), satisfies all of the following: was identified as an eligible school district under 2011 Wisconsin Act 32, section 9137 (3u).

SECTION 1578. 118.60 (1) (am) 1. to 4. of the statutes are repealed.

SECTION 1579. 118.60 (1m) of the statutes is repealed.

SECTION 1580g. 118.60 (2) (a) 1. c. of the statutes is amended to read:

118.60 (2) (a) 1. c. The family income of a pupil does not need to be verified under subd. 1. b. for a pupil who resided in a school district other than an eligible school district or a 1st class city school district in a school year, attended a participating private school in a school district other than an eligible school district or a 1st class city school district under the program under this section or s. 119.23 in that a school year, and applies to attend a participating private school in any other school district under this section in the school year immediately following that school year.

SECTION 1580k. 118.60 (2) (a) 2. (intro.) of the statutes, as affected by 2017 Wisconsin Act 36, is amended to read:

118.60 (2) (a) 2. (intro.) The Except as provided in par. (d), the pupil satisfies one or more of the following:

SECTION 1580p. 118.60 (2) (a) 2. g. of the statutes is created to read:

118.60 (2) (a) 2. g. If the pupil resides in a school district, other than an eligible school district or a 1st class city school district, the pupil was on a waiting list under sub. (3) (ar) 4. in any previous school year.

SECTION 1582. 118.60 (2) (a) 6. c. of the statutes is amended to read:

118.60 (2) (a) 6. c. Any teacher employed by the private school on July 1 of the first school year that begins after a school district is identified as an eligible school district under sub. (1m) or 2011 Wisconsin Act 32, section 9137 (3u), who has been teaching for at least the 5 consecutive years immediately preceding that July 1, and who does not satisfy the requirements under subd. 6. a. on that July 1, applies to the department on a form prepared by the department for a temporary, nonrenewable waiver from the requirements under subd. 6. a. The department shall promulgate rules to implement this subd. 6. c., including the form of the application and the process by which the waiver application will be reviewed. The application form shall require the applicant to submit a plan for satisfying the requirements under subd. 6. a., including the name of the accredited institution of higher education at which the teacher is pursuing or will pursue the bachelor's degree and the anticipated date on which the teacher expects to complete the bachelor's degree. No waiver granted under this subd. 6. c. is valid after July

31 of the 5th school year that begins after a school district is <u>was</u> identified as an eligible school district under sub. (1m) or 2011 Wisconsin Act 32, section 9137 (3u).

SECTION 1598at. 118.60 (2) (bm) of the statutes is amended to read:

118.60 (2) (bm) No pupil who resides in a school district, other than an eligible school district or a 1st class city school district, may attend a participating private school under this section unless the pupil is a member of a family that has a total family income that does not exceed an amount equal to 1.85 2.2 times the poverty level, determined in accordance with criteria established by the director of the federal office of management and budget. In this paragraph and sub. (3m), family income includes income of the pupil's parents or legal guardians. The Except as provided in par. (a) 1. c., the family income of the pupil shall be verified as provided in par. (a) 1. b. A pupil attending a private school under this section whose family income increases may continue to attend a private school under this section.

SECTION 1598b. 118.60 (2) (d) of the statutes is created to read:

118.60 (2) (d) 1. In this paragraph:

- a. "Agreement year" means the first school year during which a combined private school and a private school participating in the program under this section are governed by the same governing body under a governing body agreement.
- b. "Combined private school" means a private school that enters into a governing body agreement and did not participate in the program under this section in the school year preceding the agreement year.
- c. "Governing body agreement" means an agreement to be governed by the same governing body that is entered into by a private school participating in the program under this section and a private school that is not participating in the program under this section.
- 2. If a combined private school participates in the program under this section during the agreement year or the school year following the agreement year, the requirements under par. (a) 2. do not apply to a pupil who applies to attend the combined private school under this section during the agreement year or the school year following the agreement year.

SECTION 1598c. 118.60 (4m) (a) 2. of the statutes is repealed and recreated to read:

118.60 (4m) (a) 2. If the pupil attended summer school for at least 15 days of summer instruction at the private school during that summer, multiply the amount under subd. 1. by 0.05.

SECTION 1598d. 118.60 (4m) (a) 3. of the statutes is created to read:

118.60 (4m) (a) 3. If the pupil attended summer school for less than 15 days of summer instruction at the private school during that summer, multiply the amount under subd. 1. by 0.05 by the quotient determined by

dividing the number of days of summer instruction the pupil attended during that summer by 15.

SECTION 1598e. 118.60 (4m) (b) 3. of the statutes is repealed.

SECTION 1599. 119.04 (1) of the statutes is amended to read:

119.04 (1) Subchapters IV, V and VII of ch. 115, ch. 121 and ss. 66.0235 (3) (c), 66.0603 (1m) to (3), 115.01 (1) and (2), 115.28, 115.31, 115.33, 115.34, 115.343, 115.345, 115.363, <u>115.364</u>, 115.365 (3), <u>115.367</u>, 115.38 (2), 115.415, 115.445, 118.001 to 118.04, 118.045, 118.06, 118.07, 118.075, 118.076, 118.10, 118.12, 118.125 to 118.14, 118.145 (4), 118.15, 118.153, 118.16, 118.162, 118.163, 118.164, 118.18, 118.19, 118.196, 118.20, 118.223, 118.225, 118.24 (1), (2) (c) to (f), (6), (8), and (10), 118.245, 118.255, 118.258, 118.291, 118.292, 118.293, 118.30 to 118.43, 118.46, 118.50, 118.51, 118.52, 118.53, 118.55, 118.56, 120.12 (2m), (4m), (5), and (15) to (27), 120.125, 120.13 (1), (2) (b) to (g), (3), (14), (17) to (19), (26), (34), (35), (37), (37m), and (38), and (39), 120.137, 120.14, 120.20, 120.21 (3), and 120.25 are applicable to a 1st class city school district and board but not, unless explicitly provided in this chapter or in the terms of a contract, to the commissioner or to any school transferred to an opportunity schools and partnership program.

SECTION 1602d. 119.23 (2) (a) 1. d. of the statutes is amended to read:

119.23 (2) (a) 1. d. In this subd. 1. d., "eligible school district" has the meaning given in s. 118.60 (1) (am). The family income of a pupil does not need to be verified under subd. 1. b. for a pupil who resided in a school district other than the school district operating under this chapter in a school year, attended a participating private school under the program under this section or s. 118.60, in a school year and applies to attend a participating private school in the program under this section in the school year immediately following that school year.

SECTION 1619m. 119.23 (4m) (a) 2. of the statutes is repealed and recreated to read:

119.23 (4m) (a) 2. If the pupil attended summer school for at least 15 days of summer instruction at the private school during that summer, multiply the amount under subd. 1. by 0.05.

SECTION 1619n. 119.23 (4m) (a) 3. of the statutes is created to read:

119.23 (4m) (a) 3. If the pupil attended summer school for less than 15 days of summer instruction at the private school during that summer, multiply the amount under subd. 1. by 0.05 by the quotient determined by dividing the number of days of summer instruction the pupil attended during that summer by 15.

SECTION 1619s. 119.23 (4m) (b) 3. of the statutes is repealed.

SECTION 1623g. 120.12 (24) of the statutes is renumbered 120.12 (24) (a).

SECTION 1623r. 120.12 (24) (b) of the statutes is created to read:

120.12 (24) (b) Annually submit to the department of administration a report containing all of the following information regarding health care for school district employees:

- 1. Health care plan design.
- 2. Premium contributions.
- 3. Self-insurance contributions.
- 4. Deductibles, copayments, coinsurance, and other methods by which employees contribute to health care costs.

SECTION 1625. 120.13 (14) (a) of the statutes is amended to read:

120.13 (14) (a) Establish and provide or contract for the provision of child care programs for children. The school board may receive federal or state funds for this purpose. The school board may charge a fee for all or part of the cost of the service for participation in a child care program established under this subsection. Costs associated with a child care program under this subsection may not be included in shared costs under s. 121.07 (6). Child care programs established under this subsection shall meet the standards for licensed child care centers established by the department of children and families. Pursuant to ss. 48.66 (1) (a), 48.73, and 48.74, the department of children and families may visit and inspect the premises of, inspect the records of, and investigate and prosecute any alleged violations occurring at any child care program established or contracted for under this subsection that receives payment under s. 49.155 for the child care provided. If a school board proposes to contract for the provision of a child care program under this subsection or if on July 1, 1996, a school board is a party to a contract for the provision of a child care program under this subsection, the school board shall refer the proposed contractor to the department of children and families for the criminal history and child abuse record search required under s. 48.685. Each school board shall provide the department of health services with information about each person who is denied a contract for a reason specified in s. 48.685 (4m) (a) 1. to 5.

SECTION 1626. 120.13 (14) (a) of the statutes, as affected by 2017 Wisconsin Act (this act), is amended to read:

120.13 (14) (a) Establish and provide or contract for the provision of child care programs for children. The school board may receive federal or state funds for this purpose. The school board may charge a fee for all or part of the cost of the service for participation in a child care program established under this subsection. Costs associated with a child care program under this subsection may not be included in shared costs under s. 121.07 (6). Child care programs established under this subsection shall meet the standards for licensed child care centers established by the department of children and families. Pur-

suant to ss. 48.66 (1) (a), 48.73, and 48.74, the department of children and families may visit and inspect the premises of, inspect the records of, and investigate and prosecute any alleged violations occurring at any child care program established or contracted for under this subsection that receives payment under s. 49.155 for the child care provided. If a school board proposes to contract for the provision of a child care program under this subsection or if on July 1, 1996, a school board is a party to a contract for the provision of a child care program under this subsection, the school board shall refer the proposed contractor to the department of children and families for the criminal history and child abuse record search required under s. 48.685. Each school board shall provide the department of health services with information about each person who is denied a contract for a reason specified in s. 48.685 (4m) (a) 1. to 5 48.686.

SECTION 1627. 120.13 (14) (b) 1. of the statutes is amended to read:

120.13 (14) (b) 1. If a person who has contracted under par. (a) to provide a child care program is convicted of a serious crime, as defined in s. 48.685 48.686 (1) (c) 3m., or if a caregiver specified in s. 48.685 48.686 (1) (ag) 1. -a. or a nonclient resident, as defined in s. 48.685 48.686 (1) (bm), of the child care program is convicted or adjudicated delinquent for committing a serious crime on or after his or her 12th 10th birthday, the school board shall rescind the contract of the contractor immediately upon providing written notice of the rescission and the grounds for the rescission and an explanation of the process for appealing the rescission.

SECTION 1628. 120.13 (14) (b) 2. of the statutes is amended to read:

120.13 (14) (b) 2. If a person who has contracted under par. (a) to provide a child care program is the subject of a pending criminal charge alleging that the person has committed a serious crime, as defined in s. 48.685 48.686 (1) (c) 3m., or if a caregiver specified in s. 48.685 48.686 (1) (ag) 1. a. or a nonclient resident, as defined in s. 48.685 48.686 (1) (bm), of the child care program is the subject of a pending criminal charge or delinquency petition alleging that the person has committed a serious crime on or after his or her 12th 10th birthday, the school board shall immediately suspend the contract of the contractor until the school board obtains information regarding the final disposition of the charge or delinquency petition indicating that the person is not ineligible to provide a child care program under this subsection.

SECTION 1630d. 121.004 (6) of the statutes is amended to read:

121.004(6) NET COST. The "net cost" of a fund means the gross cost of that fund minus all nonduplicative revenues and other financing sources of that fund except property taxes, general aid, and aid received under s. ss. 79.095 (4) and 79.096. In this subsection, "nonduplicative revenues" includes federal financial assistance under

20 USC 236 to 245, to the extent permitted under federal law and regulations.

SECTION 1633g. 121.02 (1) (a) 2. of the statutes is amended to read:

121.02 (1) (a) 2. Subject to s. 118.40 (8) (b) 1... 2., and 3., ensure that all instructional staff of charter schools located in the school district hold a license or permit to teach issued by the department. For purposes of this subdivision, a virtual charter school is located in the school district specified in s. 118.40 (8) (a) and a charter school established under s. 118.40 (3) (c) 1. c. is located in the school district specified in s. 118.40 (3) (c) 1. c. The state superintendent shall promulgate rules defining "instructional staff" for purposes of this subdivision.

SECTION 1633i. 121.02 (1) (a) 3. of the statutes is created to read:

121.02 (1) (a) 3. Ensure that an individual who provides instruction in a subject and level through an online course offered from another state to pupils enrolled in a school in the school district holds a valid license or permit to teach the subject and level in the state from which the online course is provided.

SECTION 1635. 121.05 (1) (a) 5. of the statutes is amended to read:

121.05 (1) (a) 5. Pupils attending a technical college under s. 118.15 (1) (b) and pupils attending an institution of higher education or a technical college under s. 118.55.

SECTION 1635d. 121.06 (4) of the statutes is amended to read:

121.06 (4) For purposes of computing state aid under s. 121.08, equalized valuations calculated under sub. (1) and certified under sub. (2) shall include the full value of property that is exempt under s. 70.11 (39) and (39m) as determined under s. 79.095 (3) for 2016.

SECTION 1635e. 121.07 (2) (d) of the statutes is amended to read:

121.07 (2) (d) The number of pupils residing in the school district in the previous school year for whom a payment was made under s. 118.40 (2r) (e) to an operator of a charter school established under contract with an entity under s. 118.40 (2r) (b) 1. e., eg., or f. in the previous school year.

SECTION 1635em. 121.07 (6) (e) 1. of the statutes is amended to read:

121.07 (6) (e) 1. For a school district created by a consolidation under s. 117.08 or 117.09 that takes effect before July 1, 2019, in the school year in which the consolidation takes effect and in each of the subsequent 4 school years, the amounts under pars. (b) and (d) shall be multiplied by 1.15 and rounded to the next lowest dollar.

SECTION 1635f. 121.07 (7) (e) 1. of the statutes is amended to read:

121.07 (7) (e) 1. For a school district created by a consolidation under s. 117.08 or 117.09 that takes effect before July 1, 2019, in the school year in which the consolidation takes effect and in each of the subsequent 4

school years, the amounts under pars. (a) to (bm) shall be multiplied by 1.15 and rounded to the next lower dollar.

SECTION 1635h. 121.15 (4) (a) of the statutes is amended to read:

121.15 (4) (a) In this subsection, "state aid" has the meaning given in s. 121.90 (2) except that it excludes aid paid to school districts under s. ss. 79.095 (4) and 79.096.

SECTION 1638. 121.58 (2) (a) 4. of the statutes is amended to read:

121.58 (2) (a) 4. For each pupil so transported whose residence is more than 12 miles from the school attended, \$275 \$300 per school year in the 2014-15 2016-17 school year and \$300 \$365 per school year thereafter.

SECTION 1639. 121.58 (2) (am) of the statutes is amended to read:

121.58 (2) (am) State aid under par. (a) shall be reduced proportionately in the case of a pupil transported for less than a full school year because of nonenrollment. State aid for transportation shall not exceed the actual cost thereof. No state aid of any kind may be paid to a school district which charges the pupil transported or his or her parent or guardian any part of the cost of transportation provided under ss. 121.54 (1) to (3), (5), (6) and (10) and 121.57 or which willfully or negligently fails to transport all pupils for whom transportation is required under s. 121.54.

SECTION 1640. 121.58 (4) of the statutes is amended to read:

121.58 (4) State aid for summer class trans-PORTATION. Annually on or before October 1 of the year in which transportation is provided under s. 118.50 (3) (b) or 121.54 (4), or under s. 121.54 (10) if the transportation is provided by the nonresident school district that a pupil attends under s. 118.51 or 121.84 (4), the school district clerk shall file with the department a report, containing such information as the department requires, on transportation provided by the school board to and from summer classes. Upon receipt of such report and if the summer classes meet the requirements of s. 121.14 (1) (a) 1. or 2., state aid shall be paid for such transportation. A school district which provides such transportation shall be paid state aid for such transportation at the rate of \$4 \$10 per pupil transported to and from public school whose residence is at least 2 miles and not more than 5 miles by the nearest traveled route from the public school attended, and \$6 \$20 per pupil transported to and from public school whose residence is more than 5 miles by the nearest traveled route from the public school attended, if the pupil is transported 30 days or more. The state aid shall be reduced proportionately if the pupil is transported less than 30 days.

SECTION 1640b. 121.59 (2) (a) of the statutes is amended to read:

121.59 (2) (a) Divide the statewide school district transportation costs in the previous school year by the

statewide membership in the previous school year and multiply the quotient by $1.5 \underline{1.45}$.

SECTION 1640c. 121.59 (2m) of the statutes is created to read:

121.59 (2m) (a) Beginning in the 2017–18 school year and in any school year thereafter, if a school district was eligible to receive aid under sub. (2) in the immediately preceding school year but is ineligible to receive aid in the current school year because the number under sub. (2) (d) is not a positive number, the state superintendent shall, subject to par. (b), pay to that school district the amount determined as follows:

- 1. Determine the amount paid to the school district under sub. (2) (f) in the immediately preceding school year.
 - 2. Multiply the amount under subd. 1. by 0.5.
- (b) The sum of all payments under par. (a) may not exceed \$200,000 in any fiscal year. If in any school year the amount to which school districts are entitled under par. (a) exceeds \$200,000, the state superintendent shall prorate the payments among the eligible school districts.

SECTION 1640cm. 121.84 (4) (b) of the statutes is amended to read:

121.84(4) (b) If a pupil attends school in a school district outside the pupil's school district of residence under par. (a), s. 118.51 (12) (b), (14), (16), and (17) apply to the pupil as if the pupil were attending school in a nonresident school district under s. 118.51. If the pupil is rejected as a result of s. 118.51 (12) (b), s. 118.51 (9) applies.

SECTION 1640d. 121.90 (2) (am) 2. of the statutes is amended to read:

121.90 (2) (am) 2. Amounts under s. ss. 79.095 (4) and 79.096 for the current school year.

SECTION 1640g. 121.905 (1) of the statutes is **Vetoed** amended to read:

In Part

121.905 (1) In this section, "revenue ceiling" means \$9,000 in the 2011-12 school year and in the 2012-13 school year and \$9,100 in the 2013-14 \$9,300 in the 2017-18 school year, \$9,400 in the 2018-19 school year, \$9,500 in the 2019-20 school year, \$9,600 in the 2020-21 school year, \$9,700 in the 2021-22 school year, and \$9,800 in the 2022-23 school year and in any subsequent school year.

SECTION 1640i. 121.91 (3) (a) of the statutes is renumbered 121.91 (3) (a) 1. and amended to read:

121.91 (3) (a) 1. If a school board wishes to exceed the limit under sub. (2m) otherwise applicable to the school district in any school year, it shall promptly adopt a resolution supporting inclusion in the final school district budget of an amount equal to the proposed excess revenue. The resolution shall specify whether the proposed excess revenue is for a recurring or nonrecurring purpose, or, if the proposed excess revenue is for both recurring and nonrecurring purposes, the amount of the proposed excess revenue for each purpose. The resolution shall be filed as provided in s. 8.37. Within 10 days after adopting the resolution, the school board shall notify the department of the scheduled date of the that it will schedule a referendum for the purpose of submitting the resolution to the electors of the school district for approval or rejection and shall submit a copy of the resolution to the department. The Except as provided in subds. 2. and 3., the school board shall call a special referendum for the purpose of submitting the resolution to the electors of the school district for approval or rejection. In lieu of a special referendum, the school board may specify that schedule the referendum to be held at the next succeeding regularly scheduled spring primary or election or partisan primary or general election, if provided such election is to be held not sooner than 70 days after the filing of the resolution of the school board. A school board may proceed under this subdivision and under s. 67.05 (6a) 2. a. no more than 2 times in any calendar year. The school district clerk shall certify the results of the referendum to the department within 10 days after the referendum is held.

SECTION 1640m. 121.91 (3) (a) 2. of the statutes is created to read:

121.91 (3) (a) 2. The school board of a school district that experiences a natural disaster, including a fire, that causes the school district's costs to increase may call a special referendum to be held within the 6-month period immediately following the natural disaster, provided the special referendum is to be held not sooner than 70 days after the filing of the resolution of the school board under subd. 1.

Vetoed In Part

Vetoed

In Part

SECTION 1640p. 121.91 (3) (a) 3. of the statutes is created to read:

121.91 (3) (a) 3. The school board of a school district may call a special referendum to be held on the Tuesday after the first Monday in November in an odd–numbered year, provided the special referendum is to be held not sooner than 70 days after the filing of the resolution of the school board under subd. 1.

SECTION 1640r. 121.91 (3) (c) of the statutes is amended to read:

121.91 (3) (c) The A referendum under this subsection shall be held in accordance with chs. 5 to 12. The school district clerk shall provide the election officials with all necessary election supplies. The form of the ballot shall correspond substantially with the standard form for referendum ballots prescribed by the elections commission under ss. 5.64 (2) and 7.08 (1) (a). The question submitted shall be whether the limit under sub. (2m) may be exceeded by a specified amount. If the resolution provides that any of the excess revenue will be used for a nonrecurring purpose, the ballot in the election shall so state and shall specify the amount that will be used for a nonrecurring purpose. The limit otherwise applicable to the school district under sub. (2m) is increased by the

amount approved by a majority of those voting on the question.

SECTION 1640t. 121.91 (4) (L) 1. of the statutes is repealed.

SECTION 1640v. 121.91 (4) (L) 2. of the statutes is renumbered 121.91 (4) (L).

SECTION 1641m. 121.91 (4) (o) 4. of the statutes is created to read:

121.91 (4) (o) 4. Unless the resolution is adopted before January 1, 2018, subd. 1. applies only to a resolution adopted after December 31, 2018.

SECTION 1641p. 121.91 (4) (p) 1. of the statutes is **In Part** amended to read:

121.91 (4) (p) 1. The limit otherwise applicable to a school district under sub. (2m) in any school year is increased by the amount of any reduction to that school district's state aid payment made under s. 118.51 (16) (b) 2. and (c) or (17) (c) 2. or (cm) 2. in the previous school year for a pupil who was not included in the calculation of the number of pupils enrolled in that school district in the previous school year.

SECTION 1642. 125.04 (5) (a) 5. of the statutes is amended to read:

125.04 (5) (a) 5. Have successfully completed within the 2 years prior to the date of application a responsible beverage server training course at any location that is offered by a technical college district and that conforms to curriculum guidelines specified by the technical college system board or a comparable training course that is approved by the department or the educational approval board department of safety and professional services. This subdivision does not apply to an applicant who held, or who was an agent appointed and approved under sub. (6) of a corporation or limited liability company that held, within the past 2 years, a Class "A", "Class A" or "Class C" license or a Class "B" or "Class B" license or permit or a manager's or operator's license.

SECTION 1643. 125.17 (6) (a) (intro.) of the statutes is amended to read:

125.17 (6) (a) (intro.) Except as provided in par. (b), no municipal governing body may issue an operator's license unless the applicant has successfully completed a responsible beverage server training course at any location that is offered by a technical college district and that conforms to curriculum guidelines specified by the technical college system board or a comparable training course, which may include computer—based training and testing, that is approved by the department or the educational approval board department of safety and professional services, or unless the applicant fulfills one of the following requirements:

SECTION 1643d. 125.26 (1) of the statutes is amended to read:

125.26 (1) Every municipal governing body may issue Class "B" licenses for the sale of fermented malt beverages from premises within the municipality and

Vetoed In Part may authorize an official or body of the municipality to issue temporary Class "B" licenses under sub. (6). A Class "B" license authorizes retail sales of fermented malt beverages to be consumed either on the premises where sold or off the premises. A license may be issued after July 1. That license shall expire on the following June 30. Persons holding a Class "B" license may sell beverages containing less than 0.5 percent of alcohol by volume without obtaining a license under s. 66.0433 (4).

SECTION 1643h. 125.27 (1) (a) of the statutes is amended to read:

125.27 (1) (a) The department shall issue Class "B" permits to clubs holding a valid certificate issued under s. 73.03 (50) that are operated solely for the playing of golf or tennis and are commonly known as country clubs and to clubs that are operated solely for curling, ski jumping or yachting, if the club is not open to the general public and if no Class "B" licenses are issued by the governing body of the municipality in which the club is located. A Class "B" permit authorizes retail sales of fermented malt beverages to be consumed on the premises where sold. Persons holding a Class "B" permit may sell beverages containing less than 0.5 percent of alcohol by volume without obtaining a license under s. 66.0433 (1).

SECTION 1643p. 125.27 (2) (b) of the statutes is amended to read:

125.27 (2) (b) Persons holding a permit under par. (a) may sell beverages containing less than 0.5 percent of alcohol by volume without obtaining a license under s. 66.0433 (1).

SECTION 1643t. 125.27 (3) (c) of the statutes is amended to read:

125.27 (3) (c) A tribe holding a permit under par. (a) may sell beverages containing less than 0.5 percent of alcohol by volume without obtaining a license under s. 66.0433 (1).

SECTION 1644. 134.66 (2m) (b) of the statutes is amended to read:

134.66 (2m) (b) Paragraph (a) does not apply to an agent, employee, or independent contractor who has received the training described in par. (a) as part of a responsible beverage server training course or a comparable training course, as described in s. 125.04 (5) (a) 5., that was successfully completed by the agent, employee, or independent contractor. The department of health services shall make the training program developed or approved by that department under par. (a) available to the technical college system board, and that board shall include that training program or a comparable training program approved by that department in the curriculum guidelines specified by that board under s. 125.04 (5) (a) 5. The department of health services shall also make the training program developed or approved by that department under par. (a) available to any provider of a comparable training course, as described in s. 125.04 (5) (a) 5., on request, and the department of revenue or the educational approval board department of safety and professional services may approve a comparable training course under s. 125.04 (5) (a) 5. only if that training course includes the training program developed or approved by the department of health services under par. (a) or a comparable training program approved by that department.

SECTION 1644r. 139.32 (5) of the statutes is amended to read:

139.32 (5) Manufacturers, bonded direct marketers, and distributors who are authorized by the department to purchase tax stamps shall receive a discount of 0.7 0.8 percent of the tax paid on stamp purchases.

SECTION 1646t. 145.01 (4m) of the statutes is renumbered 145.01 (4m) (intro.) and amended to read:

145.01 (4m) FAILING PRIVATE ON-SITE WASTEWATER TREATMENT SYSTEM. (intro.) "Failing private on-site wastewater treatment system" has the meaning specified under s. 145.245 (4). means a private on-site wastewater treatment system that meets the criteria established by the department for determining if a private on-site wastewater treatment system is failing. A failing private on-site wastewater treatment system is one that causes or results in any of the following conditions:

SECTION 1646u. 145.01 (4m) (a) of the statutes is created to read:

145.01 **(4m)** (a) The discharge of sewage into surface water or groundwater.

SECTION 1646v. 145.01 (4m) (b) of the statutes is created to read:

145.01 (**4m**) (b) The introduction of sewage into zones of saturation that adversely affects the operation of a private on–site wastewater treatment system.

SECTION 1646w. 145.01 (4m) (c) of the statutes is created to read:

145.01 (4m) (c) The discharge of sewage to a drain tile or into zones of bedrock.

SECTION 1646x. 145.01 (4m) (d) of the statutes is created to read:

145.01 (4m) (d) The discharge of sewage to the surface of the ground.

SECTION 1646y. 145.01 (4m) (e) of the statutes is created to read:

145.01 (4m) (e) The failure to accept sewage discharges and back up of sewage into the structure served by the private on–site wastewater treatment system.

SECTION 1646ym. 145.02 (2) (a) of the statutes is amended to read:

145.02 (2) (a) The department shall have general supervision of all such plumbing and shall after public hearing prescribe and publish and enforce reasonable standards therefor which shall be uniform and of statewide concern so far as practicable. Any employee designated by the department may act for the department in holding the public hearing required under this subsection. To the extent that the historic building code applies

to the subject matter of these standards, the standards do not apply to a qualified historic building if the owner elects to be subject to s. 101.121. The standards do not apply to a primitive rural hunting cabin, as defined in s. 101.61 (3).

SECTION 1652g. 145.09 of the statutes is renumbered 145.09 (1m).

SECTION 1652h. 145.09 (2m) of the statutes is created to read:

145.09 (**2m**) The department shall accept for a journeyman plumber's examination a person to whom all of the following apply:

- (a) The person completed a plumbing apprenticeship under s. 106.025 or under the laws of any other state.
- (b) The person passed a journeyman plumber's examination in any other state.
- (c) The person has practiced for at least 5 years under a journeyman plumber's license or equivalent license issued by another state having licensure provisions governing plumbers that the department determines are substantially similar to the requirements under this chapter and the rules promulgated under this chapter, and the person has not been the subject of any disciplinary actions related to that license or any other equivalent license.

SECTION 1655g. 145.20 (5) (a) of the statutes is amended to read:

145.20 (5) (a) The department shall establish a maintenance program to be administered by governmental units responsible for the regulation of private on-site wastewater treatment systems. The department shall determine the private on-site wastewater treatment systems to which the maintenance program applies. At a minimum the maintenance program is applicable to all new or replacement private on-site wastewater treatment systems constructed in a governmental unit after the date on which the governmental unit adopts this program. The department may apply the maintenance program by rule to private on-site wastewater treatment systems constructed in a governmental unit responsible for the regulation of private on-site wastewater treatment systems on or before the date on which the governmental unit adopts the program. The department shall determine the private on-site wastewater treatment systems to which the maintenance program applies in governmental units that do not meet the conditions for eligibility under s. 145.245

SECTION 1655h. 145.20 (5) (am) of the statutes is amended to read:

145.20 (5) (am) Each governmental unit responsible for the regulation of private on–site wastewater treatment systems shall adopt and begin the administration of the program established under par. (a) before October 1, 2019. As part of adopting and administering the program, the governmental unit shall conduct and maintain an inventory of all the private on–site wastewater treat-

ment systems located in the governmental unit and shall complete the initial inventory before October 1, 2017. In order to be eligible for grant funding under s. 145.245, a governmental unit must comply with these deadlines.

SECTION 1655j. 145.245 of the statutes, as affected by 2017 Wisconsin Act (this act), is repealed.

SECTION 1655k. 145.245 (12m) of the statutes is repealed.

SECTION 1657b. 146.615 of the statutes is created to read:

146.615 Advanced practice clinician training grants. (1) In this section:

- (a) "Advanced practice clinician" means a physician assistant or an advanced practice nurse, including a nurse practitioner, certified nurse–midwife, clinical nurse specialist, or certified registered nurse anesthesiologist.
- (b) "Clinic" has the meaning given in s. 146.903 (1) (b).
 - (c) "Hospital" has the meaning given in s. 50.33 (2).
- (d) "Rural clinic" means a clinic that is located in a city, town, or village in this state that has a population of less than 20,000.
- (e) "Rural hospital" means a hospital that is located in a city, town, or village in this state that has a population of less than 20,000.
- (2) Beginning in fiscal year 2018–19, from the appropriation under s. 20.435 (1) (fk), subject to sub. (3), the department shall distribute grants to hospitals and clinics that provide new training opportunities for advanced practice clinicians. The department shall distribute the grants under this section to hospitals and clinics that apply, in the form and manner determined by the department, to receive grants and that satisfy the criteria under sub. (3).
- (3) (a) The department may distribute up to \$50,000 per fiscal year per hospital or clinic.
- (b) If the department distributes a grant to a hospital or clinic that has not previously received a grant under this section, the hospital or clinic receiving the grant may use the grant to create the education and infrastructure for training advanced practice clinicians or for activities authorized under par. (c). In distributing grants under this section, the department shall give preference to advanced practice clinician clinical training programs that include rural hospitals and rural clinics as clinical training locations.
- (c) If the department distributes a grant to a hospital or clinic that has previously received a grant under this section, the department shall require the hospital or clinic to use the grant to pay for the costs of operating a clinical training program for advanced practice clinicians, which may include any of the following:
 - 1. Required books and materials.
 - 2. Tuition and fees.
 - 3. Stipends for reasonable living expenses.

- 4. Preceptor costs, including preceptor compensation attributable to training, certification requirements, travel, and advanced practice clinician training.
- (d) A recipient awarded a grant under this section shall match through its own funding sources the amount of the grant distributed by the department for the purposes of operating an advanced practice clinician rotation.
- (4) A hospital or clinic sponsoring a training program for advanced practice clinicians supported by a grant under this section may determine what, if any, posteducation requirements must be fulfilled by participants in the training program for advanced practice clinicians.

SECTION 1657d. 146.616 of the statutes is created to read:

146.616 Allied health professional education and training grants. (1) In this section:

- (a) "Allied health professional" means any individual who is a health care provider other than a physician, registered nurse, dentist, pharmacist, chiropractor, or podiatrist and who provides diagnostic, technical, therapeutic, or direct patient care and support services to the patient.
- (b) "Clinic" has the meaning given in s. 146.903 (1) (b).
 - (c) "Hospital" has the meaning given in s. 50.33 (2).
- (d) "Rural clinic" means a clinic that is located in a city, town, or village in this state that has a population of less than 20,000.
- (e) "Rural hospital" means a hospital that is located in a city, town, or village in this state that has a population of less than 20,000.
- (2) Beginning in fiscal year 2018–19, from the appropriation under s. 20.435 (1) (fi), subject to subs. (3) to (5), the department shall distribute grants to hospitals, health systems, and educational entities that form health care education and training consortia for allied health professionals. The department shall distribute the grants under this section to hospitals, health systems, and educational entities that apply, in the form and manner determined by the department, to receive a grant and that satisfy the requirements established by the department under sub. (4)
- (3) (a) The department may distribute up to \$125,000 per fiscal year per consortium to be used for any of the following:
 - 1. Curriculum and faculty development.
 - 2. Tuition reimbursement.
 - 3. Clinical site or simulation expenses.
- (b) A recipient awarded a grant under this section shall match through its own funding sources the amount of the grant distributed by the department for the purposes of operating an allied health professional training consortium.

- (4) The department shall determine the requirements for the formation of health care education and training consortia for allied health professionals.
- (5) In distributing grants under this section, the department shall give preference to rural hospitals, health systems with a rural hospital or rural clinic, and rural educational entities.

SECTION 1662. 146.82 (2) (a) 16. of the statutes is amended to read:

146.82 (2) (a) 16. To a designated representative of the long–term care ombudsman under s. 16.009 (4), for the purpose of protecting and advocating the rights of an individual 60 years of age or older who resides in a long–term care facility, as specified in s. 16.009 (4) (b), or an individual 60 years of age or older who is an enrollee of the family care program, the Family Care Partnership Program, the program of all–inclusive care for the elderly, or the self–directed services option.

SECTION 1664. 146.98 of the statutes is repealed. SECTION 1672. 165.055 (2) of the statutes is amended to read:

165.055 (2) The deputy attorney general shall give a bond to the state in the sum of \$5,000, with good and sufficient sureties, to be approved by the governor, conditioned for the faithful performance of the deputy attorney general's duties and the <u>The</u> attorney general shall be responsible for all acts of the deputy attorney general.

SECTION 1672g. 165.10 of the statutes is created to read:

165.10 Limits on expenditure of discretionary settlement funds. Notwithstanding s. 20.455 (3), before the attorney general may expend settlement funds under s. 20.455 (3) (g) that are not committed under the terms of the settlement, the attorney general shall submit to the joint committee on finance a proposed plan for the expenditure of the funds. If the cochairpersons of the committee do not notify the attorney general within 14 working days after the submittal that the committee has scheduled a meeting for the purpose of reviewing the proposed plan, the attorney general may expend the funds to implement the proposed plan. If, within 14 working days after the submittal, the cochairpersons of the committee notify the attorney general that the committee has scheduled a meeting for the purpose of reviewing the proposed plan, the attorney general may expend the funds only to implement the plan as approved by the committee.

SECTION 1673d. 165.25 (17) (title) of the statutes is repealed.

SECTION 1673h. 165.25 (17) (intro.) of the statutes is renumbered 323.29 (3) (a) 1. and amended to read:

323.29 (3) (a) 1. Provide staff support for the interoperability council under s. 16.9645 and oversight of the development and operation of a statewide public safety interoperable communication system.

(b) The department may do any of the following: **SECTION 1673p.** 165.25 (17) (am) of the statutes is renumbered 323.29 (3) (b) 1. and amended to read:

323.29 (3) (b) 1. Charge a public safety agency, as defined in s. 256.35 (1) (g), that is a state agency a fee for use of the statewide public safety interoperable communication system under this subsection section.

SECTION 1673t. 165.25 (17) (bm) of the statutes is renumbered 323.29 (3) (b) 2. and amended to read:

323.29 (3) (b) 2. Charge a person that is not a state agency a fee for use of the statewide public safety interoperable communication system under this subsection section.

SECTION 1673v. 165.25 (18) and (19) of the statutes are created to read:

165.25 (18) CRIME LABORATORIES; DEOXYRIBONU-CLEIC ACID ANALYSIS. Determine the amount required to fund the appropriation account under s. 20.455 (2) (Lm).

(19) CRIME LABORATORIES; DEOXYRIBONUCLEIC ACID ANALYSIS SURCHARGES. If the appropriation account under s. 20.455 (2) (Lp) is anticipated to go into deficit, promptly notify the joint committee on finance in writing of the anticipated deficit.

SECTION 1673x. 165.95 (2) of the statutes is amended to read:

165.95 (2) The department of justice shall make grants to counties to enable them to establish and operate programs, including suspended and deferred prosecution programs and programs based on principles of restorative justice, that provide alternatives to prosecution and incarceration for criminal offenders who abuse alcohol or other drugs. The department of justice shall make the grants from the appropriations under s. 20.455 (2) (em), (jd), (kn), and (kv). The department of justice shall collaborate with the departments of corrections and health and family services in establishing this grant program.

SECTION 1674. 165.986 (1) of the statutes is amended to read:

165.986 (1) The department of justice shall provide grants from the appropriation under s. 20.455 (2) (kb) to cities to employ additional uniformed law enforcement officers whose primary duty is beat patrolling. A city is eligible for a grant under this section subsection in fiscal year 1994–95 if the city has a population of 25,000 or more. A city may receive a grant for a calendar year if the city applies for a grant before September 1 of the preceding calendar year. Grants shall be awarded to the 10 eligible cities submitting an application for a grant that have the highest rates of violent crime index offenses in the most recent full calendar year for which data is available under the uniform crime reporting system of the federal bureau of investigation.

SECTION 1675. 165.986 (2) of the statutes is amended to read:

165.986 (2) A city applying to the department of justice for a grant under this section sub. (1) shall include a

proposed plan of expenditure of the grant moneys. The grant moneys that a city receives under this section sub. (1) may be used for salary and fringe benefits only. Except as provided in sub. (3), the positions for which funding is sought must be created on or after April 21, 1994, and result in a net increase in the number of uniformed law enforcement officers assigned to beat patrol duties

SECTION 1676. 165.986 (3) (intro.) of the statutes is amended to read:

165.986 (3) (intro.) During the first 6 months of the first year of a grant <u>under sub. (1)</u>, a city may, with the approval of the department, use part of the grant for the payment of salary and fringe benefits for overtime provided by uniformed law enforcement officers whose primary duty is beat patrolling. A city may submit a request to the department for a 3-month extension of the use of the grant for the payment of overtime costs. To be eligible to use part of the first year's grant for overtime costs, the city shall provide the department with all of the following:

SECTION 1677. 165.986 (4) of the statutes is amended to read:

165.986 (4) The department shall develop criteria which, notwithstanding s. 227.10 (1), need not be promulgated as rules under ch. 227, for use in determining the amount to grant to cities under this section sub. (1). The department may not award an annual grant under sub. (1) in excess of \$150,000 to any city. The department shall review any application and plan submitted under sub. (2) to determine if that application and plan meet the requirements of this section. The grant that a city receives under this section sub. (1) may not supplant existing local resources.

SECTION 1678. 165.986 (5) of the statutes is amended to read:

165.986 (5) A city may receive a grant <u>under sub. (1)</u> for 3 consecutive years without submitting a new application each year. For each year that a city receives a grant <u>under sub. (1)</u>, the city shall provide matching funds of at least 25 percent of the amount of the grant.

SECTION 1679. 165.986 (6) of the statutes is amended to read:

165.986 (6) The department may make grants <u>under sub. (1)</u> to additional cities with a population of 25,000 or more after fiscal year 1994–95. Eligibility for <u>the grants under this subsection</u> shall be determined and allocations made as provided in this section.

SECTION 1680. 165.986 (7) of the statutes is created to read:

165.986 (7) From the appropriation under s. 20.455 (2) (jc), the department shall make grants in amounts determined by the department to cities with a population of 25,000 or more to reimburse overtime costs for uniformed law enforcement officers whose primary duty is beat patrolling, except that the department may award no

more \$400,000 to a city for a calendar year. The grants may be used for salary and fringe benefits only. The grants may be awarded only to the 10 eligible cities submitting an application for a grant that have the highest rates of violent crime index offenses in the most recent full calendar year for which data is available under the uniform crime reporting system of the federal bureau of investigation. A city may receive a grant for a calendar year if the city applies before September 1 of the preceding calendar year and provides the department all of the following:

- (a) The reasons why uniformed law enforcement officers assigned to beat patrol duties need to work over-time
- (b) The status of the hiring and training of new uniformed law enforcement officers who will have beat patrol duties.
- (c) A proposed plan of expenditure of the grant moneys.

SECTION 1680b. 165.989 of the statutes is created to read:

- 165.989 Community institution security cost reimbursement grants. (1) In this section, "community institution" means a building used by members of a community to engage in social gatherings, educational activities, or other community-building activities that is owned by a corporation, organization, or association described in section 501 (c) (3) of the Internal Revenue Code that is exempt from taxation under section 501 (a) of the Internal Revenue Code.
- (2) The department of justice shall establish policies and procedures for the distribution of grants from the appropriation under s. 20.455 (3) (g) to reimburse community institutions that have expanded security measures or installed additional security infrastructure in response to continuous or ongoing security threats that the institution has received. Grants may be awarded to pay reasonable and necessary security costs that shall be determined by the department in consultation with the community institution and local law enforcement agencies. Grant funds may not be awarded to pay for overtime costs of the community institution's employees or for the hiring of private security personnel in response to a security threat. Notwithstanding s. 227.10 (1), the department need not promulgate the required policies and procedures as rules under ch. 227.
- (3) Any community institution may apply to the department of justice for a grant under this section and shall include in the application detailed documentation of the security threats received, the corresponding expansion of security measures or installation of additional security infrastructure, and proof of the associated expenses incurred for which the community institution seeks a reimbursement grant. The department shall review each application and may award a grant to an eligible community institution for up to 50 percent of the

actual security expenses incurred by the community institution. Grants awarded under this section may not exceed \$200,000 per fiscal biennium.

SECTION 1680c. 165.989 of the statutes, as created by 2017 Wisconsin Act (this act), is repealed.

SECTION 1680d. 167.10 (3) (b) 8. of the statutes is created to read:

167.10 (3) (b) 8. The possession of fireworks by a person who is not a resident of this state if the person does not use the fireworks in this state.

SECTION 1680e. 167.10 (3) (bm) of the statutes is amended to read:

167.10 (3) (bm) Paragraph (a) applies to a person transporting fireworks under par. (b) 7. if, in the course of transporting the fireworks through a city, town, or village, the person remains in that city, town, or village for a period of at least 42 72 hours.

SECTION 1680f. 167.10 (3) (c) 6. of the statutes is amended to read:

167.10 (3) (c) 6. Any individual or group of individuals. A permit issued to a group of individuals confers the privileges under the permit to each member of the group.

SECTION 1680g. 167.10 (3) (f) 4. of the statutes is amended to read:

167.10(3) (f) 4. The date <u>or dates</u> and location of permitted use.

SECTION 1680h. 167.10 (6m) (d) of the statutes is amended to read:

167.10 (6m) (d) The department of safety and professional services shall issue a 4-year license to manufacture fireworks or devices listed under sub. (1) (e), (f), or (i) to (n) to a person who complies with the rules of the department promulgated under par. (e). Notwithstanding s. 101.19 (1g) (j), the license fee is \$100. The department may not issue a license to a person who does not comply with the rules promulgated under par. (e). The department may revoke a license under this subsection for the refusal to permit an inspection at reasonable times by the department or for a continuing violation of the rules promulgated under par. (e).

SECTION 1680m. 168.04 (4) of the statutes is created to read:

- 168.04 (4) (a) In this subsection, "gasoline-ethanol fuel blend" includes such a fuel blend for both automotive and nonautomotive uses.
- (b) Except as provided under par. (c), compliance with the requirements, established by the department by rule under sub. (1), of ASTM D4814–17 or the most current version of testing methods adopted by the department may be demonstrated by testing a gasoline–ethanol fuel blend or testing the gasoline base stock from which the gasoline–ethanol fuel blend is produced.
- (c) The department may promulgate rules that require that a gasoline-ethanol fuel blend and the gasoline base stock from which the gasoline-ethanol fuel blend is produced meet the requirements of ASTM

Vetoed In Part D4814–17, or the most current version of testing methods adopted by the department. A rule promulgated under this paragraph may not take effect sooner than July 1, 2019.

SECTION 1682. 178.0120 (2) (b) of the statutes is amended to read:

178.0120 (2) (b) The department may collect an expedited service fee, established by rule, for processing in an expeditious manner a record required or permitted to be filed with the department under this chapter, except that the fee to expedite processing to within one hour of filing shall be \$500 and the fee to expedite processing to within 4 hours of filing shall be \$250. Notwithstanding s. 178.0110, the \$500 or \$250 expedited processing fee, if applicable, applies to a partnership regardless of the date the partnership was formed.

SECTION 1690. 182.01 (4) (d) of the statutes is amended to read:

182.01 (4) (d) Processing, in an expeditious manner, a document required or permitted to be filed with the department, except that the fee to expedite processing to within one hour of filing shall be \$500 and the fee to expedite processing to within 4 hours of filing shall be \$250.

SECTION 1691. 182.028 of the statutes is amended to read:

182.028 School corporations. Any corporation formed for the establishment and maintenance of schools, academies, seminaries, colleges or universities or for the cultivation and practice of music shall have power to enact bylaws for the protection of its property, and provide fines as liquidated damages upon its members and patrons for violating the bylaws, and may collect the same in tort actions, and to prescribe and regulate the courses of instruction therein, and to confer such degrees and grant such diplomas as are usually conferred by similar institutions or as shall be appropriate to the courses of instruction prescribed, except that no corporation shall operate or advertise a school that is subject to s. 38.50 440.52 (10) without complying with the requirements of s. 38.50 440.52. Any stockholder may transfer his or her stock to the corporation for its use; and if the written transfer so provides the stock shall be perpetually held by the board of directors with all the rights of a stockholder, including the right to vote.

SECTION 1691c. 196.01 (5) (b) 7. of the statutes is created to read:

196.01 (5) (b) 7. A state agency, as defined in s. 20.001 (1), that may own, operate, manage, or control all or any part of a plant or equipment for the production, transmission, delivery, or furnishing of water either directly or indirectly for the public.

SECTION 1691d. 196.218 (3) (a) 2e. of the statutes is created to read:

196.218 (3) (a) 2e. No later than 30 days after the close of a fiscal year:

- a. The commission shall estimate the amount of unencumbered balances under s. 20.155 (1) (q) and (3) (rm) for that fiscal year that will transfer to the appropriation account under s. 20.155 (3) (r).
- b. The department of public instruction shall provide the commission with the department's estimate of the total amount of unencumbered balances under s. 20.255 (1) (q) and (3) (q), (qm), and (r) for that fiscal year that will transfer to the appropriation account under s. 20.155 (3) (r).
- c. The Board of Regents of the University of Wisconsin System shall provide the commission with the board's estimate of the amount of unencumbered balance under s. 20.285 (1) (q) for that fiscal year that will transfer to the appropriation account under s. 20.155 (3) (r).

SECTION 1691h. 196.218 (3) (a) 2m. of the statutes is created to read:

196.218 (3) (a) 2m. No later than 30 days after the close of a fiscal biennium, the department of administration shall provide the commission with the department's estimate of the amount of unencumbered balance under s. 20.505 (4) (s) for that fiscal biennium that will transfer to the appropriation account under s. 20.155 (3) (r).

SECTION 1691p. 196.218 (3) (a) 2s. of the statutes is created to read:

196.218 (3) (a) 2s. Thirty days after the close of a fiscal year or as soon as practicable thereafter, the commission shall determine the sum of the estimates specified in subd. 2e. a., b., and c. If the close of a fiscal year is also the close of a fiscal biennium, the sum shall include the estimate specified in subd. 2m. In the subsequent fiscal year, the commission shall transfer from the universal service fund to the appropriation account under s. 20.155 (3) (rm) an amount equal to \$2,000,000 less the sum determined under this subdivision.

SECTION 1691r. 196.218 (3) (a) 3. (intro.) of the statutes is amended to read:

196.218 (3) (a) 3. (intro.) The commission shall designate the method by which the contributions under this paragraph shall be calculated and collected. The method shall ensure that the contributions are sufficient to generate and, to the extent practicable, do not exceed the following amounts:

SECTION 1691t. 196.218 (3) (a) 3. am. of the statutes is created to read:

196.218 (3) (a) 3. am. The amount appropriated under s. 20.155 (3) (rm).

SECTION 1692. 196.218 (3) (a) 3. b. of the statutes is amended to read:

196.218 (**3**) (a) 3. b. The amounts appropriated under ss. 20.255 (1) (q) and (3) (q), (qm), and (r), 20.285 (1) (q), and 20.505 (4) (s), (t), (tm), (tu), and (tw).

SECTION 1693. 196.218 (5) (a) 10. of the statutes is amended to read:

196.218 (5) (a) 10. To make broadband expansion grants and administer the program under s. 196.504.

SECTION 1695. 196.504 (1) (a) of the statutes is renumbered 196.504 (1) (ac).

SECTION 1695e. 196.504 (1) (ab) of the statutes is created to read:

196.504 (1) (ab) "Economic development" has the meaning given in s. 196.796 (1) (c).

SECTION 1695m. 196.504 (1) (ad) of the statutes is created to read:

196.504 (1) (ad) "Fixed wireless service" has the meaning given in s. 77.51 (3rn), except that it does not include mobile wireless service, as defined in s. 77.51 (7k), or telecommunications services, as defined in s. 77.51 (21n), transmitted through the use of satellite.

SECTION 1695s. 196.504 (1) (c) of the statutes is created to read:

196.504(1)(c) "Unserved area" means an area of this state that is not served by an Internet service provider offering Internet service that is all of the following:

- 1. Fixed wireless service or wired service.
- 2. Provided at actual speeds of at least 20 percent of the upload and download speeds for advanced telecommunications capability as designated by the federal communications commission in its inquiries regarding advanced telecommunications capability under 47 USC 1302 (b).

SECTION 1698. 196.504 (2) (a) of the statutes is amended to read:

196.504 (2) (a) To make broadband expansion grants to eligible applicants for the purpose of constructing broadband infrastructure in underserved areas designated under par. (d). Grants awarded under this section shall be paid from the appropriation appropriations under s. 20.155 (3) (r). In each fiscal year, the total amount of the grants may not exceed \$1,500,000 and (rm).

SECTION 1699g. 196.504 (2) (c) of the statutes is amended to read:

196.504 (2) (c) To establish criteria for evaluating applications and awarding grants under this section. The criteria shall prohibit grants that have the effect of subsidizing the expenses of a provider of telecommunications service, as defined in s. 182.017 (1g) (cq), or the monthly bills of customers of those providers. The criteria shall give priority to projects that include matching funds, that involve public-private partnerships, that affect areas with no broadband service providers unserved areas, that are scalable, that promote economic development, that will not result in delaying the provision of broadband service to areas neighboring areas to be served by the proposed project, or that affect a large geographic area or a large number of underserved individuals or communities. When evaluating grant applications under this section, the commission shall consider the degree to which the proposed projects would duplicate existing broadband infrastructure, information about the presence of which is provided to the commission by the applicant or another person within a time period designated by the commission; the impacts of the proposed projects on the ability of individuals to access health care services from home and the cost of those services; and the impacts of the proposed projects on the ability of students to access educational opportunities from home.

SECTION 1699r. 196.504 (2) (e) of the statutes is created to read:

196.504 (2) (e) To designate areas of the state as unserved areas.

SECTION 1701. 196.858 (1) of the statutes is amended to read:

196.858 (1) The commission shall annually assess against local exchange and interexchange telecommunications utilities the total, not to exceed \$5,000,000, of the amounts appropriated under s. 20.505 (1) (ir) 20.155 (1) (i).

SECTION 1702. 196.858 (2) of the statutes is amended to read:

196.858 (2) The commission shall assess a sum equal to the annual total amount under sub. (1) to local exchange and interexchange telecommunications utilities in proportion to their gross operating revenues during the last calendar year. If total expenditures for telephone telecommunications relay service exceeded the payment made under this section in the prior year, the commission shall charge the remainder to assessed telecommunications utilities in proportion to their gross operating revenues during the last calendar year. A telecommunications utility shall pay the assessment within 30 days after the bill has been mailed to the assessed telecommunication utility. The bill constitutes notice of the assessment and demand of payment. Payments shall be credited to the appropriation account under s. 20.505 (1) (ir) 20.155 (1) (i).

SECTION 1703. 202.051 (3) (c) of the statutes is amended to read:

202.051 (3) (c) Pays the charge for an unpaid draft established by the depository selection board under s. 20.905 (2).

SECTION 1703m. 218.14 of the statutes is created to read:

218.14 Service fees. (1) A dealer may not assess a purchaser of a recreational vehicle an additional service fee or charge for performing a vehicle inspection or completing a form that is related to the sale of the recreational vehicle and required by law unless the dealer discloses the fee or charge to the purchaser and provides a statement on the purchase or lease contract in substantially the following form: "A service fee or charge is not required by law but may be charged to recreational vehicle purchasers or lessees for services related to compliance with state and federal laws, verifications, and public safety and must be reasonable." The amount of a service fee or

charge assessed by a dealer under this subsection may not be higher than the amount initially disclosed.

- (2) Upon request from a purchaser of a recreational vehicle, a dealer shall provide a written list of the services for which a service fee or charge under sub. (1) is assessed.
- (3) The department may audit a dealer to determine whether fees or charges assessed by the dealer under sub. (1) are reasonable.

SECTION 1703p. 224.42 (title) of the statutes is amended to read:

224.42 (title) Disclosure of financial records for Medical Assistance and food stamp program eligibil-

SECTION 1703r. 224.42 (2) of the statutes is amended to read:

224.42 (2) Financial record matching agree-MENTS. A financial institution is required to enter into an agreement with the department of health services under s. ss. 49.45 (4m) and 49.79 (1t).

SECTION 1703t. 224.42 (3) (a) and (b) of the statutes are amended to read:

224.42 (3) (a) Disclosing a financial record of an individual or other information to the department of health services in accordance with an agreement, and its participation in the program, under s. 49.45 (4m) or 49.79 (1t).

(b) Any other action taken in good faith to comply with s. 49.45 (4m) or 49.79 (1t).

SECTION 1704. 224.48 (1) (am) of the statutes is created to read:

224.48 (1) (am) "Department" means the department of financial institutions.

SECTION 1705. 224.50 (1) (c) of the statutes is created to read:

224.50 (1) (c) "Department" means the department of financial institutions.

SECTION 1706. 224.51 (1g) of the statutes is created to read:

224.51 (1g) In this section, "department" means the department of financial institutions.

SECTION 1712h. 227.118 of the statutes is created to read:

Vetoed In Part

227.118 Review of rules affecting state prosecutors office. (1) REPORT ON RULES AFFECTING STATE PROSE-CUTORS OFFICE. If a proposed rule directly affects the state prosecutors office, the agency proposing the rule shall, prior to submitting the proposed rule to the legislative council staff under s. 227.15, submit the proposed rule to the state prosecutors office. The state prosecutors office shall prepare a report on the proposed rule before it is submitted to the legislative council staff under s. 227.15. The state prosecutors office may request any information from other state agencies, local governments, individuals, or organizations that is reasonably necessary for the office to prepare the report. The state prosecutors office shall prepare the report within 30 days after the rule is **Vetoed** submitted to the office.

- (2) FINDINGS OF THE OFFICE TO BE CONTAINED IN THE REPORT. The report of the state prosecutors office shall contain information about the effect of the proposed rule on the state prosecutors office.
- (3) APPLICABILITY. This section does not apply to emergency rules promulgated under s. 227.24.

SECTION 1740g. 227.19 (3) (em) of the statutes is created to read:

227.19 (3) (em) The report of the state prosecutors office, if the proposed rule directly affects the state prosecutors office.

SECTION 1754. 227.55 of the statutes is renumbered 227.55 (1) and amended to read:

227.55 (1) Within 30 days after service of the petition for review upon the agency, or within such further time as the court may allow allows, the agency in possession of the record for the decision under review shall transmit to the reviewing court the original or a certified copy of the entire record of the proceedings in which the decision under review was made, including all pleadings, notices, testimony, exhibits, findings, decisions, orders, and exceptions, therein; but except that by stipulation of all parties to the review proceedings the record may be shortened by eliminating any portion thereof of the record. Any party, other than the agency that is a party, refusing to stipulate to limit the record may be taxed by the court for the additional costs. The Except as provided in sub. (2), the record may be typewritten or printed. The exhibits may be typewritten, photostated photocopied, or otherwise reproduced, or, upon motion of any party, or by order of the court, the original exhibits shall accompany the record. The court may require or permit subsequent corrections or additions to the record when deemed desirable.

SECTION 1755. 227.55 (2) of the statutes is created to read:

227.55 (2) In the case of a record under sub. (1) that is in the possession of the division of hearings and appeals, if any portion of the record is in the form of an audio or video recording, the division may transmit to the reviewing court a copy of that recording in lieu of preparing a transcript, unless the court requests a transcript.

SECTION 1757e. 230.08 (2) (e) 4. of the statutes is repealed.

SECTION 1757m. 230.08 (2) (fq) of the statutes is created to read:

230.08 (2) (fq) The director of the transportation projects commission.

SECTION 1758g. 230.08 (2) (qp) of the statutes is cre-

230.08(2) (qp) The executive director and legislative liaison in the office of state prosecutors.

SECTION 1761. 230.08 (4) (d) of the statutes is repealed.

In Part

Vetoed In Part

Vetoed In Part Vetoed In Part **SECTION 1761p.** 230.12 (1) (cm) of the statutes is created to read:

230.12 (1) (cm) Supplementary compensation; longevity awards for correctional officers and youth counselors. 1. In this paragraph:

- a. "Correctional officer" means an individual classified as a correctional officer or a correctional sergeant who is employed by the state and whose principal duty is the supervision of inmates at a prison, as defined in s. 302.01, or the supervision of persons committed under s. 980.06 at the secure mental health facility established under s. 46.055 or the Wisconsin resource center established under s. 46.056.
- b. "Youth counselor" means an individual classified as a youth counselor or a youth counselor—advanced who is employed by the state and whose principal duty is the supervision of juveniles held in a juvenile correctional facility, as defined in s. 938.02 (10p).
- 2. The administrator shall include in the compensation plan the following length of service awards for correctional officers and youth counselors:
- a. On the employee's 10th anniversary of service, \$250.
- b. On the employee's 15th anniversary of service, \$500.
- c. On the employee's 20th anniversary of service, \$750.
- d. On the employee's 25th anniversary of service, and each 5 year anniversary of service thereafter, \$1,000.

SECTION 1762. 230.13 (1) (intro.) of the statutes is amended to read:

230.13 (1) (intro.) Except as provided in sub. (3) and ss. 19.36 (10) to (12) and (11) and 103.13, the director and the administrator may keep records of the following personnel matters closed to the public:

Vetoed In Part **SECTION 1762s.** 230.33 (1) of the statutes is amended to read:

230.33 (1) A person appointed to an unclassified position by the governor, elected officer, judicial body, or prosecutor board, or by a legislative body or committee shall be granted a leave of absence without pay for the duration of the appointment and for 3 months thereafter, during which time the person has restoration rights to the former position or equivalent position in the department in which last employed in a classified position without loss of seniority. The person shall also have reinstatement privileges for 5 years following appointment to the unclassified service or for one year after termination of the unclassified appointment whichever is longer. Restoration rights and reinstatement privileges shall be forfeited if the reason for termination of the unclassified appointment would also be reason for discharge from the former position in the classified service.

SECTION 1765. 230.44 (4) (bm) of the statutes is amended to read:

230.44 (4) (bm) Upon request of an employee who files an appeal of the decision of the administrator made under s. 230.09 (2) (a) or (d), the appeal shall be heard by -a the commissioner or an attorney employed by the commission serving as arbitrator under rules promulgated for this purpose by the commission. In such an arbitration, the arbitrator shall orally render a decision at the conclusion of the hearing affirming, modifying, or rejecting the decision of the administrator. The decision of the arbitrator is final and is not subject to review by the commission. An arbitrator's decision may not be cited as precedent in any other proceeding before the commission or before any court. The arbitrator shall promptly file his or her decision with the commission. The decision of the arbitrator shall stand as the decision of the commission. The decision of the commission is subject to review under ss. 227.53 to 227.57 only on the ground that the decision was procured by corruption, fraud, or undue means or that the arbitrator or the commission exceeded the arbitrator's or the commission's power. The record of a proceeding under this paragraph shall be transcribed as provided in s. 227.44 (8).

SECTION 1766. 230.45 (1) (am) of the statutes is amended to read:

230.45 (1) (am) Designate a commissioner or Serve as an arbitrator, or designate an attorney employed by the commission to serve as an arbitrator, in arbitrations under s. 230.44 (4) (bm).

SECTION 1767. 230.85 (3) (b) of the statutes is amended to read:

230.85 (3) (b) If, after hearing, the division of equal rights finds that the respondent did not engage in or threaten a retaliatory action it shall order the complaint dismissed. The division of equal rights shall order the employee's appointing authority to insert a copy of the findings and orders into the employee's personnel file and, if the respondent is a natural person, order the respondent's appointing authority to insert such a copy into the respondent's personnel file. If the division of equal rights finds by unanimous vote that the employee filed a frivolous complaint it may order payment of the respondent's reasonable actual attorney fees and actual costs. Payment may be assessed against either the employee or the employee's attorney, or assessed so that the employee and the employee's attorney each pay a portion. To find a complaint frivolous the division of equal rights must find that s. 802.05 (2) or 895.044 has been violated.

SECTION 1769. 233.13 (intro.) of the statutes is amended to read:

233.13 Closed records. (intro.) Except as provided in ss. 19.36 (10) to (12) and (11) and 103.13, the authority may keep records of the following personnel matters closed to the public:

Vetoed In Part

SECTION 1769v. 238.115 (4) of the statutes is created to read:

238.115 (4) EXCEPTION. After March 31, 2018, this section does not apply to the tax credits under ss. 238.308, 238.396, and 238.399.

SECTION 1770. 238.123 of the statutes is repealed. **SECTION 1771.** 238.124 of the statutes is created to read:

- 238.124 Loan limitations. (1) The corporation may not originate any loan that is forgivable in whole or in part upon the loan recipient's achievement of one or more conditions or goals.
- (2) Each new lending program the corporation implements or administers shall adhere as closely as practicable to commonly accepted commercial lending practices. The corporation shall adopt policies and procedures implementing this subsection.

SECTION 1771s. 238.145 of the statutes is repealed. **SECTION 1774e.** 238.15 (3) (d) (intro.) of the statutes is renumbered 238.15 (3) (d) and amended to read:

238.15 (3) (d) Administration. The corporation, in consultation with the department of revenue, shall establish policies and procedures to administer this section and shall further define "bona fide angel investment" for purposes of s. 71.07 (5d) (a) 1. The aggregate amount of tax credits under s. 71.07 (5d) that may be claimed for investments in businesses certified under sub. (1) and of tax credits under ss. 71.07 (5b), 71.28 (5b), 71.47 (5b), and 76.638 that may be claimed for investments paid to fund managers certified under sub. (2) is \$30,000,000 per calendar year. The policies and procedures shall provide that a person who receives a credit under s. 71.07 (5b) or (5d), 71.28 (5b), 71.47 (5b), or 76.638 must keep the investment in a certified business, or with a certified fund manager, for no less than 3 years, unless the person's investment becomes worthless, as determined by the corporation, during the 3-year period or the person has kept the investment for no less than 12 months and a bona fide liquidity event, as determined by the corporation, occurs during the 3-year period. The policies and procedures shall permit the corporation to reallocate credits under this section in any calendar year that are unused in that calendar year to a person eligible for tax benefits, as defined under s. 238.30 (7) (e), if all of the following apply:

SECTION 1774f. 238.15 (3) (d) 1. of the statutes is repealed.

SECTION 1774g. 238.15 (3) (d) 2. of the statutes is repealed.

SECTION 1774k. 238.16 (4) (c) of the statutes is amended to read:

238.16 (4) (c) Subject to a reallocation by the corporation pursuant to policies and procedures adopted under s. 238.15 (3) (d), the The corporation may allocate up to \$10,000,000 in tax benefits under this section in any calendar year.

SECTION 1775. 238.17 of the statutes is renumbered 238.17 (1) and amended to read:

238.17 (1) For taxable years beginning after December 31, 2013, the corporation may certify a person to claim a tax credit under s. 71.07 (9m), 71.28 (6), or 71.47 (6), if the corporation determines that the person is conducting an eligible activity under s. 71.07 (9m), 71.28 (6), or 71.47 (6). No person may claim a tax credit under s. 71.07 (9m), 71.28 (6), or 71.47 (6) without first being certified under this section subsection.

SECTION 1775g. 238.17 (2) of the statutes is created to read:

238.17 (2) Beginning July 1, 2018, the corporation may not certify persons to claim more than a total of \$5,000,000 in tax credits for all projects undertaken on the same parcel.

Vetoed In Part

SECTION 1775m. 238.17 (3) of the statutes is created to read:

238.17 (3) (a) Except as provided in par. (b), the corporation may not certify a person for a tax credit under sub. (1) if the person is not subject to the taxes imposed under s. 71.02, 71.08, 71.23, or 71.43, except that the corporation may certify a nonprofit entity described under section 501 (c) (3) of the Internal Revenue Code for a tax credit under sub. (1) if the entity intends to sell or otherwise transfer the credit, as provided under s. 71.07 (9m), 71.28 (6) (h), or 71.47 (6) (h).

(b) The corporation may certify a nonprofit entity not described under section 501 (c) (3) of the Internal Revenue Code for a tax credit under sub. (1) by submitting the proposal for the certification to the joint committee on finance for approval. If the cochairpersons of the joint committee on finance do not notify the corporation within 14 working days after the date of the corporation's submittal that the committee has scheduled a meeting for the purpose of reviewing the proposal, the proposal may be implemented as proposed by the corporation. If, within 14 working days after the corporation's submittal, the cochairpersons of the joint committee on finance notify the corporation that the committee has scheduled a meeting for the purpose of reviewing the proposal, the proposal may be implemented only upon approval of the committee.

SECTION 1779L. 238.28 of the statutes is created to read:

238.28 Refundable tax credits. (1) POLICIES AND Vetoed PROCEDURES. The corporation shall adopt policies and In Part procedures implementing ss. 71.07 (3w) (c) 1. b., (3wm) (d), and (3y) (d) 2. b., 71.28 (3w) (c) 1. b., (3wm) (d), and (3y) (d) 2. b., and 71.47 (3w) (c) 1. b. and (3y) (d) 2. b.

(2) Use of Credits. It is the intent of the legislature Vetoed that all credits awarded under ss. 238.16, 238.308, 238.396, and 238.399 become a permanent part of the working capital structure of businesses claiming the credits.

SECTION 1779m. 238.308 (4) (b) of the statutes is amended to read:

238.308 (4) (b) Subject to a reallocation by the corporation under s. 238.15 (3) (d), the The corporation may allocate up to \$17,000,000 in tax benefits under this section in 2016 and up to \$22,000,000 per in tax benefits under this section each year thereafter. Any unused allocation may be carried forward.

SECTION 1779n. 238.308 (4) (c) of the statutes is created to read:

238.308 (4) (c) In any year, the corporation may exceed the annual limit on tax benefits specified in par. (b) by up to \$10,000,000 if all of the following apply:

- 1. The corporation notifies the joint committee on finance in writing of its proposal to exceed the annual limit on tax benefits specified in par. (b).
- 2. The corporation submits with its notification under subd. 1. evidence that shows the corporation's proposal is necessary to accomplish the corporation's statewide economic development objectives.
 - 3. Any of the following is true:
- a. The cochairpersons of the joint committee on finance fail to notify the corporation, within 14 working days after the date of the corporation's notification under subd. 1., that the committee has scheduled a meeting for the purpose of reviewing the corporation's proposal.
- b. The cochairpersons of the joint committee on finance notify the corporation that the committee has approved the corporation's proposal.

Vetoed In Part **SECTION 1783L.** 238.399 (3) (a) of the statutes is amended to read:

238.399 (3) (a) The corporation may designate not more than 30 areas in this state as enterprise zones.

SECTION 17830. 238.399 (5s) of the statutes is created to read:

238.399 (**5s**) CAP. (a) Except as provided in par. (b), the corporation may not authorize payments under ss. 71.07 (3w), 71.28 (3w), and 71.47 (3w) in any fiscal biennium that total more than \$80,600,000 in the aggregate.

(b) The corporation may submit a plan to exceed the aggregate amount specified under par. (a) to the cochairpersons of the joint committee on finance for review by the committee. If the cochairpersons of the committee do not notify the corporation that the committee has scheduled a meeting for the purpose of reviewing the proposed plan within 14 working days after the date of the corporation's submittal, the corporation may exceed the aggregate amount in accordance with its proposed plan. If, within 14 working days after the date of the corporation's submittal, the cochairpersons of the committee notify the corporation that the committee has scheduled a meeting for the purpose of reviewing the proposed plan, the corporation may not exceed the aggregate amount unless the committee approves the proposed plan. If the committee modifies and approves the proposed plan, the corporation may exceed the aggregate amount in accordance with the plan as modified by the committee.

Vetoed In Part

SECTION 1783q. 238.399 (6) (e) of the statutes is Vetoed repealed. In Part

SECTION 1784. 250.16 of the statutes is renumbered 341.14 (8v) and amended to read:

341.14 **(8v)** Payments to the Wisconsin Women's Health Foundation. (a) From the appropriation account under s. 20.435 (1) 20.395 (5) (gi), the department shall make payments to the Wisconsin Women's Health Foundation, Inc., to provide women's health outreach and education programs and support for women's health research that improves the quality of life for women and families in this state.

- (b) The agreement under this section shall require that the The Wisconsin Women's Health Foundation, Inc., shall provide, without fee and as a condition of receiving payments specified under this section subsection, any license or other approval required for use of any logo, trademark, trade name, word, or symbol to be used on or in association with special group registration plates under s. 341.14 sub. (6r) (f) 57.
- (c) The agreement under this section shall require that As a condition of receiving payments specified under this subsection, the Wisconsin Women's Health Foundation, Inc., shall annually submit to the attorney general and the presiding officer of each house of the legislature an audited financial statement of its use of the payments under this section subsection, prepared in accordance with generally accepted accounting principles.
- (d) Payments to the Wisconsin Women's Health Foundation, Inc., under this section subsection shall be discontinued by the department if the Wisconsin Women's Health Foundation, Inc., dissolves or is no longer exempt from taxation under section 501 (a) of the Internal Revenue Code.

SECTION 1785. 250.17 of the statutes is renumbered 341.14 (8w) and amended to read:

341.14 (8w) Organ and tissue donation. (a) From the appropriation account under s. 20.435 (1) (g) 20.395 (5) (gj), the department shall make payments to Donate Life Wisconsin, or an organization designated under par. (d) if Donate Life Wisconsin ceases to exist, to encourage organ and tissue donation by providing educational programs, promoting or advancing research and patient services, and, at the discretion of Donate Life Wisconsin, distributing portions of these payments to any other organ and tissue procurement and donation organization in this state that is exempt from taxation under section 501 (a) of the Internal Revenue Code, to be used for these same purposes.

(b) The agreement under this section shall require that Donate Life Wisconsin <u>shall</u> provide, without fee and as a condition of receiving payments specified under this <u>section</u> subsection, any license or other approval

Vetoed In Part Vetoed In Part required for use of any logo, trademark, trade name, word, or symbol to be used on or in association with special group registration plates under s. 341.14 sub. (6r) (f) 58.

- (c) The agreement under this section shall require that As a condition of receiving payments specified under this subsection. Donate Life Wisconsin shall annually submit to the attorney general and the presiding officer of each house of the legislature an audited financial statement of its use of the payments under this section subsection, prepared in accordance with generally accepted accounting principles. The agreement under this section shall also require that As a condition of receiving payments specified under this subsection. Donate Life Wisconsin shall enter into a contract with any organ and tissue donor organization to which it distributes funds under sub. (1) par. (a) requiring that organization to prepare and submit audited financial statements of that organization's use of funds received under sub. (1) par. (a).
- (d) The department shall discontinue payments to Donate Life Wisconsin under this section subsection if Donate Life Wisconsin dissolves or is no longer exempt from taxation under section 501 (a) of the Internal Revenue Code and the department, in consultation with the department of health services, shall designate a new recipient for payments under this section subsection. The new recipient must be a nonprofit organization that promotes organ and tissue donation and must comply with any requirement specified in this section subsection for Donate Life Wisconsin. Notwithstanding any other provision of this section subsection, the department shall not make any payments under this section subsection until Donate Life Wisconsin is properly formed and operational and is exempt from taxation under section 501 (a) of the Internal Revenue Code.

SECTION 1785r. 252.12 (2) (a) 8. (intro.) of the statutes is amended to read:

252.12 (2) (a) 8. 'Mike Johnson life care and early intervention services grants.' (intro.) The department shall award not more than \$3,677,000 \$4,000,000 in each fiscal year in grants to applying organizations for the provision of needs assessments; assistance in procuring financial, medical, legal, social and pastoral services; counseling and therapy; homecare services and supplies; advocacy; and case management services. These services shall include early intervention services. The department shall also award not more than \$74,000 in each year from the appropriation account under s. 20.435 (5) (md) for the services under this subdivision. The state share of payment for case management services that are provided under s. 49.45 (25) (be) to recipients of medical assistance shall be paid from the appropriation account under s. 20.435 (1) (am). All of the following apply to grants awarded under this subdivision:

SECTION 1791c. 252.185 of the statutes is created to read:

252.185 Communicable disease control and prevention. (1) From the appropriation under s. 20.435 (1) (cf), the department shall distribute moneys to local health departments to use for disease surveillance, contact tracing, staff development and training, improving communication among health care professionals, public education and outreach, and other infection control measures as required under this chapter. The department shall consider the following factors to establish an equitable allocation formula for the distribution of moneys under this section:

- (a) Base allocation, including at least some base amount for each local health department.
 - (b) General population.
 - (c) Target populations.
 - (d) Risk factors.
- (e) Geographic area, including consideration of the size of the service area or the density of population, or both.
- (2) By January 1, 2019, and biennially thereafter, each local health department shall submit to the division of the department that addresses public health issues a financial statement of its use of funds under this section.

SECTION 1791e. 253.12 (2) (a) (intro.) of the statutes is amended to read:

253.12 (2) (a) (intro.) Except as provided in par. (b), all of the following shall report in the manner prescribed by the department under sub. (3) (a) 3. a birth defect in an infant or child that is specified under sub. (3) (a) 2. or (d):

SECTION 1791g. 253.12 (2) (d) of the statutes is amended to read:

253.12 (2) (d) The department may not require a person specified under par. (a) 1. or 2. to report the name of an infant or child for whom a report is made under par. (a) if the parent or guardian of the infant or child refuses to consent states in writing that he or she refuses to the release of the name or address of the infant or child.

SECTION 1791j. 253.12 (3) (a) 2. of the statutes is amended to read:

253.12 (3) (a) 2. Specify by rule the <u>any</u> birth defects the <u>department determines</u> the existence of which requires a report under sub. (2) to be submitted to the department <u>and that the council under sub. (4) does not unanimously decide should be reported.</u>

SECTION 1791n. 253.12 (3) (a) 3m. of the statutes is created to read:

253.12 (3) (a) 3m. Require persons specified under sub. (2) (a) that are required to report to notify a parent or guardian of the infant or child who is diagnosed with a birth defect of the option to refuse to release the name and address of the infant or child to the registry.

SECTION 1791p. 253.12 (3) (c) of the statutes is repealed.

SECTION 1791r. 253.12 (3) (d) of the statutes is created to read:

253.12 (3) (d) The secretary, after reviewing recommendations of the council under sub. (4), shall maintain a list of specific birth defects the existence of which requires a report under sub. (2) to be submitted to the department and that the council unanimously decides are required to be reported.

SECTION 1791s. 253.12 (4) (a) of the statutes is amended to read:

253.12 (4) (a) Make recommendations to the department regarding the establishment of a registry that documents the diagnosis in the state of an infant or child who has a birth defect, as required under sub. (3) (a) 1. and regarding, the specific birth defects for which a report is required under sub. (2) on which the council unanimously decides, the rules that the department is required to promulgate under sub. (3) (a) 2. and 3. on the birth defects to be reported under sub. (2), and on the general content and format of the report under sub. (2) and procedures for submitting the report. The council shall also make recommendations regarding the content of a report that, because of the application of sub. (2) (d), does not contain the name of the subject of the report.

SECTION 1791u. 253.12 (5) (a) 5. of the statutes is created to read:

253.12 (5) (a) 5. The state registrar, the vital records system, and other data systems maintained by the department or another state or federal agency for purposes including determining whether multiple reports are made for an infant or child, matching reported information on infants or children with vital records and other registries, and making referrals to intervention and treatment.

SECTION 1792. 254.11 (9) of the statutes is amended to read:

254.11 (9) "Lead poisoning or lead exposure" means a level of lead in the blood of $40 \frac{5}{2}$ or more micrograms per 100 milliliters of blood.

SECTION 1799c. 256.35 (1) (em) of the statutes is created to read:

256.35 (1) (em) "Emergency number system" means any basic system, sophisticated system, or Next Generation 911, as defined in sub. (3s) (a) 3., regardless of technology platform.

SECTION 1799g. 256.35 (3m) of the statutes is repealed.

SECTION 1799n. 256.35 (3s) of the statutes is created to read:

256.35 (**3s**) NEXT GENERATION 911. (a) *Definitions*. In this subsection:

- 1. "Department" means the department of military affairs.
- 2. "Emergency services IP network" means a managed Internet protocol network that is used for emergency services and can be shared by all public safety answering points.

- 3. "Next Generation 911" means a statewide emergency number system regardless of technology platform that does all of the following:
- a. Provides standardized interfaces from requests for emergency assistance.
- b. Processes all types of requests for emergency assistance, including calls and nonvoice and multimedia messages.
- Acquires and integrates data useful to the delivery or routing and handling of requests for emergency assistance.
- d. Delivers requests for emergency assistance and data to appropriate public safety answering points and emergency responders.
- e. Supports data and communications needs for coordinated incident response and management.
- f. Provides a secure environment for emergency communications.
- 4. "Operational date," with respect to a county, means the date determined by the department on which Next Generation 911 begins to be fully operational in the county.
- 5. "Service supplier" has the meaning given in sub. (3) (a) 3.
- 6. "Service user" has the meaning given in sub. (3) (a) 4.
- (b) Emergency services IP network contracts. The department shall invite bids to be submitted under s. 16.75 and, from the appropriation under s. 20.465 (3) (qm), contract for the creation, operation, and maintenance of an emergency services IP network that to the greatest extent feasible relies on industry standards and existing infrastructure to provide all public safety answering points with the network necessary to implement Next Generation 911.
- (c) Existing contracts and charges. 1. The department shall determine the operational date for each county. If a contract under sub. (3) (b) 3. between a service supplier and a county is in effect immediately before the operational date determined for the county, the contract shall expire on the operational date and, except as provided in subd. 2., beginning on the operational date, the service supplier may not bill any service user for a charge levied by the county under sub. (3) (b) or impose a surcharge approved under sub. (3) (f). At least 30 days before a contract expires under this subdivision, the department shall provide written notice of the expiration to the county and service supplier.
- 2. If a contract terminates under subd. 1. before a service supplier has been fully compensated for nonrecurring services described in sub. (3) (b) 3. a., the service supplier may continue to bill service users for the charge levied by the county under sub. (3) (b) or impose a surcharge approved under sub. (3) (f) until the service sup-

plier is fully compensated for those nonrecurring services.

- (d) *911 subcommittee duties*. The 911 subcommittee shall do all of the following:
- 1. Advise the department on the contracts required under par. (b).
- 2. Advise the department on the statewide efforts, leveraging of existing infrastructure, and industry standards that are necessary to transition to Next Generation 911
- 3. Make recommendations to the department regarding federal sources of funding and the sustainable funding streams that are required to enable public safety answering points to purchase and maintain equipment necessary for Next Generation 911.
- 4. If funding is made available for the department or another state agency to make grants to public safety answering points for training or upgrading facilities or services or for implementing Next Generation 911, advise the department or other state agency on making the grants, including advising on eligibility criteria for the grants. The criteria shall include basic training and service standards that grant applicants must satisfy.
- Conduct a statewide 911 telecommunications system assessment.
- 6. Develop recommendations for service standards for public safety answering points.
- 7. Promote, facilitate, and coordinate interoperability across all public safety answering points with respect to telecommunications services and data systems, including geographic information systems.
- 8. Promote, facilitate, and coordinate consolidation of public safety answering point functions where consolidation would provide improved service, increased efficiency, or cost savings.
- 9. Undertake all of its duties in a manner that is competitively and technologically neutral.

SECTION 1799r. 256.35 (7) of the statutes is renumbered 256.35 (7) (intro.) and amended to read:

256.35 (7) LIABILITY EXEMPTION. (intro.) A telecommunications utility, wireless provider, as defined in sub. (3m) (a) 6., or local government, as defined in sub. (3m) (a) 4., All of the following shall not be liable to any person who uses an emergency number system created under this section or makes an emergency telephone call initially routed to a wireless public safety answering point, as defined in sub. (3m) (a) 7., 2015 stats.:

SECTION 1799w. 256.35 (7) (a) to (d) of the statutes are created to read:

256.35 (7) (a) A telecommunications utility.

- (b) A wireless provider, as defined in s. 256.35 (3m) (a) 6., 2015 stats.
- (c) A local government, as defined in s. 256.35 (3m) (a) 4., 2015 stats.
- (d) A person that supplies any service, product, equipment, or database, including any related emergency

notification service or process, that is used for or in conjunction with the installation, implementation, operation, or maintenance of the emergency number system and that is used by a public safety answering point.

SECTION 1799y. 281.145 of the statutes is created to read:

281.145 River and stream monitoring and study. (1) In this section:

- (a) "Nonpoint source" has the meaning given in s. 281.16 (1) (e).
- (b) "Point source" has the meaning given in s. 283.01 (12).
- (2) The department shall conduct a program to monitor and study the introduction of nutrients from point sources and nonpoint sources into the East and West Twin Rivers, the Manitowoc River, the Sheboygan River, and the streams that outlet to Lake Michigan and that lie in and between the Ahnapee River watershed and the Sauk Creek watershed. The department shall seek to do all of the following under this subsection:
- (a) Identify the amounts of nutrients being introduced into these waters.
- (b) Characterize and quantify the nutrients, in particular nitrogen and phosphorus, introduced into these waters from nonpoint sources relative to climate, land use, soil type, elevation, and drainage.
- (c) Collect water quality information from locations on these waters and from major tributaries and major impoundments to use in evaluating the biological, physical, and chemical properties of the water and to use as data in watershed and river models.
- (d) Use watershed and river models and the information collected under this subsection and from other sources to forecast the effect on water quality of different methods of reducing the amounts of nutrients introduced into these waters.
- (e) Develop tools to use in selecting and implementing methods of reducing the amounts of nutrients introduced into these waters.

SECTION 1805. 281.36 (11) (a) of the statutes is amended to read:

281.36 (11) (a) The department shall set a surcharge fee to be charged for each application to proceed under a wetland general permit that is issued under sub. (3g) (a) 4., 5., or 6. The surcharge fee shall be set on an annual basis by the department and may not exceed more than 50 percent of the market price, as determined by the department, for the equivalent purchase of credits from a mitigation bank. These fees shall be credited to the appropriation account under s. 20.370 (4) (9) (bm) for the restoration and creation of wetlands. The department may enter into agreements with other entities for the restoration and creation of such wetlands.

SECTION 1806. 281.36 (11) (b) of the statutes is amended to read:

281.36 (11) (b) Any wetland that is restored or created using funding from the appropriation under s. 20.370 (4) (9) (bm) shall be open to the public for hunting, fishing, trapping, cross—country skiing, or hiking or any combination thereof, but the department may establish reasonable restrictions on the use of the land by the public in order to protect public safety or to protect a unique plant or animal community.

SECTION 1806f. 281.57 (7) (c) 1. of the statutes is amended to read:

281.57 (7) (c) 1. Metropolitan sewerage districts that serve 1st class cities are limited in each fiscal year to receiving total grant awards not to exceed 33 percent of the sum of the amounts in the schedule for that fiscal year for the appropriation under s. 20.165 (2) (ke) and the amount authorized under sub. (10) for that fiscal year plus the unencumbered balance at the end of the preceding fiscal year for the amount authorized under sub. (10). This subdivision is not applicable to grant awards provided during fiscal years 1985–86, 1986–87, 1988–89 and 1989–90.

SECTION 1808. 281.58 (6) (b) 6. of the statutes is repealed.

SECTION 1813. 281.58 (12) (a) 1. of the statutes is renumbered 281.58 (12) (a) 1. (intro.) and amended to read:

281.58 (12) (a) 1. (intro.) Except as modified under par. (f) and except as restricted by sub. (8) (b), (c), (f) or (h), the interest rate for projects specified in sub. (7) (b) 1. to 5. is one of the following:

c. For a municipality that does not meet the requirements specified in subd. 1. a. or b., 75 percent of market interest rate for projects for which the subsidy was allocated from the amount under s. 281.59 (3e) (b), 2013 stats., for a biennium before the 2015–17 biennium and 70 percent of market interest rate for projects for which the financial assistance is allocated under this section for the 2015–17 biennium or later, and 55 percent of market interest rate for projects for which the financial assistance is allocated under this section for the 2017–19 biennium or later.

SECTION 1814. 281.58 (12) (a) 1. a. of the statutes is created to read:

281.58 (12) (a) 1. a. For a municipality that has a population of less than 1,000, and in which the median household income is 65 percent or less of the median household income in this state, zero percent of market interest rate.

SECTION 1815. 281.58 (12) (a) 1. b. of the statutes is created to read:

281.58 (12) (a) 1. b. For a municipality that has a population of less than 10,000, and in which the median household income is 80 percent or less of the median household income in this state, 33 percent of market interest rate.

SECTION 1816m. 281.58 (13) (b) (intro.) of the statutes is amended to read:

281.58 (13) (b) (intro.) A municipality with an application that is approved under sub. (9m) is eligible for state financial hardship assistance for the project costs that are eligible under the clean water fund program, except for costs to which sub. (8) (b), (c), (f) or (h) applies, if the initial application was submitted on or before June 30, 2017, the application, including the facility plan and the design plans and specifications, was completed on or before June 30, 2018, and the municipality meets all of the following criteria:

SECTION 1816t. 281.59 (1m) (c) of the statutes is repealed.

SECTION 1817m. 281.59 (9) (a) of the statutes is amended to read:

281.59 (9) (a) A loan approved under the elean water fund program, the safe drinking water loan program or the land recycling loan program shall be for no longer than 20 years, as determined by the department of administration, be fully amortized not later than 20 years after the original date of the financial assistance agreement, and require the repayment of principal and interest, if any, to begin not later than 12 months after the expected date of completion of the project that it funds, as determined by the department of administration.

SECTION 1817mc. 281.59 (9) (ag) of the statutes is created to read:

281.59 (9) (ag) A loan approved under the clean water fund program shall be for no longer than 30 years or the useful life of the project, whichever is less, as determined by the department of administration. The loan shall be fully amortized not later than 30 years after the original date of the financial assistance agreement or the end of the useful life of the project, whichever is less, as determined by the department of administration. Repayment of principal and interest, if any, shall begin not later than 12 months after the expected date of completion of the project that the loan funds, as determined by the department of administration.

SECTION 1818. 281.65 (4g) of the statutes is amended to read:

281.65 (4g) The department may contract with any person from the appropriation account under s. 20.370 (4) (9) (at) for services to administer or implement this section, including information and education and training services. The department shall allocate \$500,000 in each fiscal year from the appropriation account under s. 20.370 (4) (at) for contracts for educational and technical assistance related to the program under this section provided by the University of Wisconsin–Extension.

SECTION 1818g. 281.665 (4) (c) of the statutes is created to read:

281.665 (4) (c) 1. Notwithstanding pars. (a) and (b) and subject to subd. 2., the department shall provide a cost–sharing grant under this section for a project described under sub. (5) (d) in an amount sufficient to

accomplish the flood-control goals of the project as proposed in the application, but not to exceed \$14,600,000.

2. The department may not provide a cost-sharing grant under subd. 1. unless the department first notifies the cochairpersons of the joint committee on finance, in writing, that it intends to award the grant. The notice shall contain a description of the purposes proposed for expenditure of the moneys received as a part of the grant. If the cochairpersons of the committee do not notify the department that the committee has scheduled a meeting for the purpose of reviewing the proposed grant within 14 working days after the date of the department's notification, the moneys may be awarded as proposed by the department. If, within 14 working days after the date of the department's notification, the cochairpersons of the committee notify the department that the committee has scheduled a meeting for the purpose of reviewing the proposed grant, no moneys may be awarded without the approval of the committee.

SECTION 1818r. 281.665 (5) (d) of the statutes is created to read:

281.665 (5) (d) Notwithstanding pars. (a) to (c), during the 2017–19 fiscal biennium, the department shall consider an applicant to be eligible for a cost–sharing grant for a project under this section if the project is funded or executed in whole or in part by the U.S. army corps of engineers under 33 USC 701s.

SECTION 1819. 281.73 of the statutes is repealed. SECTION 1819m. 283.01 (12) (a) of the statutes is amended to read:

283.01 (12) (a) A discernible, confined, and discrete conveyance, including but not limited to any pipe, ditch, channel, tunnel, conduit, well, discrete fissure, container, rolling stock, concentrated animal feeding operation, or vessel or other floating craft from which pollutants may be discharged either into the waters of the state or into a publicly owned treatment works except for a conveyance that conveys only storm water. This term does not include agricultural storm water discharges and return flows from irrigated agriculture.

SECTION 1819n. 283.01 (12) (b) of the statutes is amended to read:

283.01 (12) (b) A discernible, confined, and discrete conveyance of storm water for which a permit is required under s. 283.33 (1). This term does not include agricultural storm water discharges and return flows from irrigated agriculture.

SECTION 1820. 283.33 (9) (c) of the statutes is amended to read:

283.33 (9) (c) All moneys collected under par. (a) shall be credited to the appropriation under s. 20.370 (4) (9) (bj).

SECTION 1821. 283.87 (4) of the statutes is amended to read:

283.87 (4) AIDS TO MUNICIPALITIES; ENVIRONMENTAL DAMAGE COMPENSATION. The department may make

grants to any county, city, village, or town for the acquisition or development of recreational lands and facilities from moneys appropriated under s. 20.370 (2) (4) (dv). Use and administration of the grant shall be consistent with any court order issued under sub. (3). A county, city, village, or town which receives a grant under this section is not required to share in the cost of a project under this section.

SECTION 1822. 285.69 (2) (c) (intro.) of the statutes is amended to read:

285.69 (2) (c) (intro.) The fees collected under pars. (a) and (e) shall be credited to the appropriations under s. 20.370 (2) (bg), (3) (bg), (4) (co), (8) (mg) and (9) (mh) for the following:

SECTION 1823. 285.69 (2e) (c) of the statutes is amended to read:

285.69 (**2e**) (c) The fees collected under this subsection shall be credited to the appropriation accounts under s. 20.370 (2) (bg), (3) (bg), (4) (co), (8) (mg), and (9) (mh) for the purposes in sub. (2) (c) 1. and 2.

SECTION 1824. 285.69 (2m) (bm) (intro.) of the statutes is amended to read:

285.69 (**2m**) (bm) (intro.) The fees collected under this subsection shall be credited to the appropriation account under s. 20.370 (2) (bh) (4) (cm) for the following purposes as they relate to stationary sources for which an operation permit is required under s. 285.60 but not under the federal clean air act:

SECTION 1825. 285.69 (3) (a) of the statutes is amended to read:

285.69 (3) (a) The department may promulgate rules for the payment and collection of fees for inspecting nonresidential asbestos demolition and renovation projects regulated by the department. The fees under this subsection for an inspection plus the fee under sub. (1) (c) may not exceed \$700 if the combined square and linear footage of friable asbestos-containing material involved in the project is less than 5,000. The fees under this subsection for an inspection plus the fee under sub. (1) (c) may not exceed \$1,325 if the combined square and linear footage of friable asbestos-containing material involved in the project is 5,000 or more. The fees collected under this subsection shall be credited to the appropriation under s. 20.370 (2) (bi) (4) (cn) for the direct and indirect costs of conducting inspections of nonresidential asbestos demolition and renovation projects regulated by the department and for inspecting property proposed to be used for a community fire safety training project.

SECTION 1826. 285.69 (7) of the statutes is amended to read:

285.69 (7) EMISSION REDUCTION CREDIT FEES. The department may promulgate rules for the payment of fees by persons who hold emission reduction credits that may be used to satisfy the offset requirements in s. 285.63 (2) (a) and that have been certified by the department. The rules may waive the payment of fees under this subsec-

tion for categories of emission reduction credits. The fees collected under this subsection shall be credited to the appropriation under s. 20.370 (2) (bg) (4) (co).

SECTION 1827L. 285.72 (title) of the statutes is amended to read:

285.72 (title) Air quality monitoring station stations.

SECTION 1827m. 285.72 of the statutes is renumbered 285.72 (1) and amended to read:

285.72 (1) From the appropriation under s. 20.370 (2) (ee) (4) (cv), the department shall fund the construction, operation, and maintenance of an air quality monitoring station in a county identified in its entirety as a nonattainment area for the 2008 8—hour national ambient air quality standard for ozone under 40 CFR 50.15 for the purpose of assessing ozone concentrations. The department may designate the monitoring station as a special purpose monitor under 40 CFR 58.20.

SECTION 1827n. 285.72 (2) of the statutes is created to read:

285.72 (2) From the appropriation under s. 20.370 (4) (cv), the department shall fund the operation and maintenance of an air quality monitoring station in a county where a sulfur dioxide monitor has been in place for 3 years as a result of sulfur dioxide monitoring requirements under 40 CFR part 51 and the data requirement rule for the 2010 one—hour sulfur dioxide primary national ambient air quality standard published in the federal register on August 21, 2015. The department may designate the monitoring station as a special purpose monitor under 40 CFR 58.20.

SECTION 1828. 287.91 (4) of the statutes is amended to read:

287.91 (4) The department of natural resources shall reimburse the department of justice for the expenses incurred in enforcing this chapter from the appropriation under s. $20.370 \frac{(2)}{(4)}$ (4) (ma).

SECTION 1829. 289.31 (7) (f) of the statutes is amended to read:

289.31 (7) (f) If the owner or operator of a site or facility subject to an order under par. (d) is a municipality, the municipality is responsible for conducting any monitoring ordered under par. (d). The department shall, from the environmental fund appropriation under s. 20.370 (2) (4) (dv), reimburse the municipality for the costs of monitoring that exceed an amount equal to \$3 per person residing in the municipality for each site or facility subject to an order under par. (d), except that the maximum reimbursement is \$100,000 for each site or facility. The department shall exclude any monitoring costs paid under the municipality's liability insurance coverage in calculating the municipal cost of monitoring a site or facility.

SECTION 1830. 289.43 (7) (e) 3. of the statutes is amended to read:

289.43 (7) (e) 3. All fees collected under this paragraph shall be credited to the appropriation under s. $20.370 \frac{(2)}{(4)} \frac{(4)}{(4g)}$.

SECTION 1831. 289.64 (6) of the statutes is amended to read:

289.64 **(6)** Use of solid waste facility siting BOARD FEES. The fees collected under sub. (2) shall be credited to the appropriation under s. $20.370 \frac{(2)}{(4)} \frac{(4)}{(8)}$ for transfer to the appropriation under s. $20.505 \frac{(4)}{(8)}$.

SECTION 1832. 289.68 (1) of the statutes is amended to read:

289.68 (1) PAYMENTS FROM THE WASTE MANAGEMENT FUND. The department may expend moneys in the waste management fund only for the purposes specified under subs. (3) to (6) and 1991 Wisconsin Act 39, section 9142 (2w). The department may expend moneys appropriated under s. 20.370 (2) (4) (dq) for the purposes specified under subs. (3) and (5) and 1991 Wisconsin Act 39, section 9142 (2w). The department may expend moneys appropriated under s. 20.370 (2) (4) (dt) for the purposes specified under sub. (4). The department may expend moneys appropriated under s. 20.370 (2) (4) (dy) and (dz) for the purposes specified under sub. (6).

SECTION 1833. 289.68 (3) of the statutes is amended to read:

289.68 (3) PAYMENTS FOR LONG-TERM CARE AFTER TERMINATION OF PROOF OF FINANCIAL RESPONSIBILITY. The department may spend moneys appropriated under s. 20.370 (2) (4) (dq) for the costs of long-term care of an approved facility for which the plan of operation was approved under s. 289.30 (6) before August 9, 1989, that accrue after the requirement to provide proof of financial responsibility expires under s. 289.41 (1m) (b) or (f) as authorized under s. 289.41 (11) (b) 2.

SECTION 1834. 289.68 (4) of the statutes is amended to read:

289.68 **(4)** Payment of Closure and Long-term Care Costs; Forfeited Bonds and Similar Moneys. The department may utilize moneys appropriated under s. 20.370 (2) <u>(4)</u> (dt) for the payment of costs associated with compliance with closure and long-term care requirements under s. 289.41 (11) (b) 1.

SECTION 1835. 289.68 (5) of the statutes is amended to read:

289.68 **(5)** PREVENTION OF IMMINENT HAZARD. The department may utilize moneys appropriated under s. 20.370 (2) (4) (dq) for the payment of costs associated with imminent hazards as authorized under s. 289.41 (11) (c) and (cm).

SECTION 1836. 289.68 (6) of the statutes is amended to read:

289.68 **(6)** Payment of corrective action, forfeited bonds and recovered moneys. The department may utilize moneys appropriated under s. $20.370 \frac{(2)}{(4)}$ (dy) and (dz) for the payment of costs of corrective action under s. 289.41 (11) (bm).

SECTION 1837. 292.11 (6) (a) of the statutes is amended to read:

292.11 (6) (a) Contingency plan; activities resulting from discharges. The department may utilize moneys appropriated under s. 20.370 (2) (4) (dv) and (my) (ms) in implementing and carrying out the contingency plan developed under sub. (5) and to provide for the procurement, maintenance, and storage of necessary equipment and supplies, personnel training, and expenses incurred in identifying, locating, monitoring, containing, removing, and disposing of discharged substances.

SECTION 1838. 292.11 (6) (b) of the statutes is amended to read:

292.11 **(6)** (b) *Limitation on equipment expenses.* No more than 25 percent of the moneys available under the appropriation under s. 20.370 (2) (4) (dv) or (my) <u>(ms)</u> during any fiscal year may be used for the procurement and maintenance of necessary equipment during that fiscal year.

SECTION 1839. 292.11 (6) (c) 2. of the statutes is amended to read:

292.11 (6) (c) 2. Reimbursements to the department under section 311, federal water pollution control act amendments of 1972, P.L. 92–500, shall be credited to the appropriation under s. 20.370 (2) (my) (4) (ms).

SECTION 1840. 292.31 (4) of the statutes is amended to read:

292.31 **(4)** Monitoring costs at nonapproved Facilities owned or operated by municipalities. Notwithstanding the environmental response rules under sub. (2) or the environmental repair authority, remedial action sequence, and emergency response requirements under sub. (3), the department shall pay that portion of the cost of any monitoring requirement which is to be paid under s. 289.31 (7) (f) from the appropriation under s. 20.370 (2) (4) (dv) prior to making other payments from that appropriation.

SECTION 1841. 292.31 (7) (am) 2. of the statutes is amended to read:

292.31 (7) (am) 2. The department may acquire an interest in property from any person as part of a remedial action conducted in cooperation with the federal environmental protection agency if the acquisition is necessary to implement the remedy. Under this subdivision, the department may acquire an interest in property that is necessary to ensure that restrictions on the use of land or groundwater are enforceable. The department may expend moneys from the appropriations under ss. 20.370 (2) (4) (dv) and 20.866 (2) (tg) if necessary to compensate a person for an interest in property acquired by the department under this subdivision.

SECTION 1842. 292.31 (7) (b) of the statutes is amended to read:

292.31 (7) (b) The department may expend moneys from the appropriations under ss. 20.370 (2) (4) (dv) and 20.866 (2) (tg) as required under 42 USC 9601, et seq. The department shall promulgate by rule criteria for the expenditure of moneys from the appropriations under ss. 20.370 (2) (4) (dv) and 20.866 (2) (tg). The criteria shall include consideration of the amount of moneys available in the appropriations under ss. 20.370 (2) (4) (dv) and 20.866 (2) (tg), the moneys available from other sources for the required sharing of costs, the differences between public and private sites or facilities, the potential for cost recovery from responsible parties, and any other appropriate factors.

SECTION 1843. 292.41 (6) (a) of the statutes is amended to read:

292.41 **(6)** (a) The department may utilize moneys appropriated under s. 20.370 (2) (4) (dv) and (my) (ms) in taking action under sub. (4). The department shall utilize these moneys to provide for the procurement, maintenance, and storage of necessary equipment and supplies, personnel training, and expenses incurred in locating, identifying, removing, and disposing of abandoned containers.

SECTION 1844. 292.41 (6) (b) of the statutes is amended to read:

292.41 **(6)** (b) No more than 25 percent of the total of all moneys available under the appropriation under s. 20.370 (2) (4) (dv) and (my) (ms) may be used annually for the procurement and maintenance of necessary equipment during that fiscal year.

SECTION 1845. 292.55 (2) of the statutes is amended to read:

292.55 (2) The department may assess and collect fees from a person to offset the costs of providing assistance under sub. (1). The department shall promulgate rules for the assessment and collection of fees under this subsection. Fees collected under this subsection shall be credited to the appropriation account under s. 20.370 (2) (4) (dh).

SECTION 1846. 292.57 (2) (b) of the statutes is amended to read:

292.57 (2) (b) Any moneys collected under this subsection shall be credited to the appropriation account under s. $20.370 \frac{(2)}{(4)} \frac{(4)}{(4h)}$.

SECTION 1847. 292.70 (7) of the statutes is amended to read:

292.70 (7) REVIEW AND PAYMENT. If a claim is filed under an agreement under sub. (2) or (3), the department shall review the claim to determine whether it is valid. A valid claim shall be paid from the appropriation under s. 20.370 (2) (4) (fq).

SECTION 1848. 292.94 of the statutes is amended to read:

292.94 Fees related to enforcement actions. The department may assess and collect fees from a person

who is subject to an order or other enforcement action for a violation of s. 292.11 or 292.31 to cover the costs incurred by the department to review the planning and implementation of any environmental investigation or environmental cleanup that the person is required to conduct. The department shall promulgate rules for the assessment and collection of fees under this section. Fees collected under this section shall be credited to the appropriation account under s. $20.370 \frac{(2)}{(4)} \frac{(4)}{(4)}$.

Vetoed In Part **SECTION 1849m.** 301.03 (21) of the statutes is created to read:

301.03 (21) By December 31, 2017, and every 2 years thereafter, submit a Wisconsin Results First Initiative report to the appropriate standing committees of the legislature under s. 13.172 (3).

SECTION 1850e. 301.08 (2) (e) of the statutes is amended to read:

301.08 (2) (e) The Except as provided in par. (em), the purchaser shall recover from provider agencies money paid in excess of the conditions of the contract from subsequent payments made to the provider.

SECTION 1850f. 301.08 (2) (em) of the statutes is created to read:

301.08 (2) (em) 1. In this paragraph:

- a. "Provider" means a nonstock corporation organized under ch. 181 that is a nonprofit corporation, as defined in s. 181.0103 (17), and that contracts under this section to provide client services on the basis of a unit rate per client service or a county department under s. 46.215, 46.22, 46.23, 51.42, or 51.437 that contracts under this section to provide client services on the basis of a unit rate per client service.
- b. "Rate-based service" means a service or a group of services, as determined by the department, that is reimbursed through a prospectively set rate and that is distinguishable from other services or groups of services by the purpose for which funds are provided for that service or group of services and by the source of funding for that service or group of services.
- 2. If revenue under a contract for the provision of a rate—based service exceeds allowable costs incurred in the contract period, the contract shall allow the provider to retain from the surplus up to 5 percent of the revenue received under the contract unless a uniform rate is established by rule under subd. 6., in which case the contract shall allow the provider to retain the uniform percentage rate established by the rule. The retained surplus is the property of the provider.
- 3. If on December 31 of any year the provider's accumulated surplus from all contract periods ending during that year for a rate—based service exceeds the allowable retention rate under subd. 2., the provider shall provide written notice of that excess to all purchasers of the rate—based service. Upon the written request of such a purchaser received no later than 6 months after the date of the notice, the provider shall refund the purchaser's propor-

tional share of that excess. If the department determines based on an audit or fiscal review that the amount of the excess identified by the provider was incorrect, the department may seek to recover funds after the 6-month period has expired. The department shall commence any audit or fiscal review under this subdivision within 6 years after the end of the contract period.

- 4. Notwithstanding subd. 2., a county department under s. 46.215 providing client services in a county having a population of 750,000 or more or a nonstock, nonprofit corporation providing client services in such a county may not retain a surplus generated by a rate—based service or accumulate funds from more than one contract period for a rate—based service from revenues that are used to meet the maintenance—of—effort requirement under the federal temporary assistance for needy families program under 42 USC 601 to 619.
- 5. All providers that are subject to this paragraph shall comply with any financial reporting and auditing requirements that the department may prescribe. Those requirements shall include a requirement that a provider provide to any purchaser and the department any information that the department needs to claim federal reimbursement for the cost of any services purchased from the provider and a requirement that a provider provide audit reports to any purchaser and the department according to standards specified in the provider's contract and any other standards that the department may prescribe.
- 6. The department, in consultation with the department of health services and the department of children and families, shall promulgate rules to implement this paragraph including all of the following:
- a. Requiring that contracts for rate-based services under this subsection allow a provider to retain from any surplus revenue up to 5 percent of the total revenue received under the contract, or a different percentage rate determined by the department. The percentage rate established under this subd. 6. a. shall apply uniformly to all rate-based service contracts under this paragraph.
- b. Establishing a procedure for reviewing rate—based service contracts to determine whether a contract complies with the provisions of this paragraph.

SECTION 1851. 301.16 (1r) of the statutes is amended to read:

301.16 (**1r**) In addition to the institutions under sub. (1), the department shall establish a medium security correctional institution for persons 15 years of age or over, but not more than 24 years of age, who have been placed in a state prison under s. 302.01. The medium security correctional institution under this subsection shall be known as the Racine Youthful Offender Correctional Facility and shall be located at the intersection of Albert Street and North Memorial Drive in the city of Racine. The department shall limit the number of prisoners who may be placed at the Racine Youthful Offender Correctional Facility to no more than 450 500 at any one time.

SECTION 1851g. 301.16 (1ww) of the statutes is created to read:

301.16 (1ww) In addition to the institutions under sub. (1), the department shall establish a geriatric correctional institution, as enumerated in 2017 Wisconsin Act (this act), section 9104 (1) (c) 1. d.

SECTION 1851r. 301.18 (1) (h) of the statutes is created to read:

301.18 (1) (h) Provide the facilities necessary for the geriatric correctional institution established under s. 301.16 (1ww).

SECTION 1854. 301.26 (4) (d) 2. of the statutes, as affected by 2015 Wisconsin Act 55, section 4270, is repealed and recreated to read:

301.26 (4) (d) 2. Beginning on July 1, 2017, and ending on June 30, 2018, the per person daily cost assessment to counties shall be \$390 for care in a Type 1 juvenile correctional facility, as defined in s. 938.02 (19), and \$390 for care for juveniles transferred from a juvenile correctional institution under s. 51.35 (3).

SECTION 1855. 301.26 (4) (d) 3. of the statutes, as affected by 2015 Wisconsin Act 55, section 4272, is repealed and recreated to read:

301.26 (4) (d) 3. Beginning on July 1, 2018, and ending on June 30, 2019, the per person daily cost assessment to counties shall be \$397 for care in a Type 1 juvenile correctional facility, as defined in s. 938.02 (19), and \$397 for care for juveniles transferred from a juvenile correctional institution under s. 51.35 (3).

SECTION 1855m. 302.01 (1) (k) of the statutes is created to read:

302.01 (1) (k) The geriatric correctional institution authorized under s. 301.16 (1ww).

Vetoed In Part

SECTION 1856c. 302.05 (title) of the statutes is amended to read:

302.05 (title) Wisconsin substance abuse rehabilitation program.

SECTION 1856e. 302.05 (1) (am) (intro.) and (b) of the statutes are consolidated, renumbered 302.05 (1) and amended to read:

302.05 (1) The department of corrections and the department of health services may designate a section of a mental health institute as a correctional treatment facility for the treatment of substance abuse of inmates transferred from Wisconsin state prisons. This section shall be administered by the department of corrections and shall be known as the Wisconsin substance abuse program. The department of corrections and the department of health services shall ensure that the residents at the institution and the residents in the substance abuse program: (b) The department of corrections and the department of health services shall, at any correctional facility the departments determine department determines is appropriate, provide a substance abuse treatment rehabilitation program for inmates for the purposes of the program described in sub. (3).

SECTION 1856f. 302.05 (1) (am) 1. and 2. of the **Vetoed** statutes are repealed.

SECTION 1856g. 302.05 (2) of the statutes is amended to read:

302.05 (2) Transfer to a correctional treatment facility for the treatment of substance abuse participation in a rehabilitation program described in sub. (1) shall be considered a transfer under s. 302.18.

SECTION 1857b. 302.05 (3) (b) of the statutes is amended to read:

302.05 (3) (b) Except as provided in par. (d), if the department determines that an eligible inmate serving a sentence other than one imposed under s. 973.01 has successfully completed a treatment rehabilitation program described in sub. (1), the parole commission shall parole the inmate for that sentence under s. 304.06, regardless of the time the inmate has served. If the parole commission grants parole under this paragraph, it shall require the parolee to participate in an intensive supervision program for drug abusers appropriate to the parolee's rehabilitation needs as a condition of parole.

SECTION 1857c. 302.05 (3) (c) 1. of the statutes is amended to read:

302.05 (3) (c) 1. Except as provided in par. (d), if the department determines that an eligible inmate serving the term of confinement in prison portion of a bifurcated sentence imposed under s. 973.01 has successfully completed a treatment rehabilitation program described in sub. (1), the department shall inform the court that sentenced the inmate.

SECTION 1857e. 302.05 (3) (c) 2. (intro.) of the statutes is amended to read:

302.05 (3) (c) 2. (intro.) Upon being informed by the department under subd. 1. that an inmate whom the court sentenced under s. 973.01 has successfully completed a treatment rehabilitation program described in sub. (1), the court shall modify the inmate's bifurcated sentence as follows:

SECTION 1857f. 302.05 (3) (d) of the statutes is amended to read:

302.05 (3) (d) The department may place intensive sanctions program participants in a treatment rehabilitation program described in sub. (1), but pars. (b) and (c) do not apply to those participants.

SECTION 1857h. 302.05 (3) (e) of the statutes is amended to read:

302.05 (3) (e) If an inmate is serving the term of confinement portion of a bifurcated sentence imposed under s. 973.01, the sentence was imposed before July 26, 2003 the effective date of this paragraph [LRB inserts date], and the inmate satisfies the criteria under par. (a) 1., the inmate may, with the department's approval, petition the sentencing court to determine whether he or she is eligible or ineligible to participate in the earned release program under this subsection during the term of confinement. The inmate shall serve a copy of the petition on the

Vetoed In Part district attorney who prosecuted him or her, and the district attorney may file a written response. The court shall exercise its discretion in granting or denying the inmate's petition but must do so no later than 90 days after the inmate files the petition. If the court determines under this paragraph that the inmate is eligible to participate in the earned release program, the court shall inform the inmate of the provisions of par. (c).

SECTION 1892c. 323.29 (title) of the statutes is created to read:

323.29 (title) Statewide public safety interoperable communication system.

SECTION 1892g. 323.29 (1) (title) of the statutes is created to read:

323.29 (1) (title) DEFINITIONS.

SECTION 1892L. 323.29 (1) (am) of the statutes is created to read:

323.29 (1) (am) "Department" means the department of military affairs.

SECTION 1892p. 323.29 (3) (title) of the statutes is created to read:

323.29 (3) (title) Department duties and powers.

SECTION 1892s. 323.29 (3) (a) (intro.) of the statutes is created to read:

323.29 (3) (a) (intro.) The department shall do all of the following:

SECTION 1892t. 323.29 (3) (a) 2. of the statutes is created to read:

323.29 (3) (a) 2. During the 2017–19 fiscal biennium, conduct a request for proposals regarding a statewide public safety interoperable communication system. The department shall require the submitted proposals to include all costs associated with their fulfillment, including costs to the state and local governments.

SECTION 1892x. 323.29 (4) of the statutes is created to read:

323.29 (4) DIRECTOR OF EMERGENCY COMMUNICATIONS. The adjutant general shall appoint a director of emergency communications within the division to serve at the pleasure of the adjutant general outside the classified service. The position shall be funded from the appropriation under s. 20.465 (3) (q).

SECTION 1893. 323.62 of the statutes is created to read:

323.62 Mobile field force grants. From the appropriation under s. 20.465 (3) (dm), the division may award grants to Wisconsin law enforcement agencies, as defined in s. 165.77 (1) (c), to fund crowd—control training and equipment used for crowd control.

SECTION 1893s. 323.70 (6m) of the statutes is created to read:

323.70 (6m) From the appropriation under s. 20.465 (3) (df), the division may award grants to local agencies with which the division contracts under sub. (2). A grant awarded under this subsection shall be used to fund the

replacement of equipment used in emergency responses to releases of hazardous substances under this section.

SECTION 1894. 341.14 (6r) (b) 10. of the statutes is amended to read:

341.14 (**6r**) (b) 10. An additional fee of \$25 that is in addition to the fee under subd. 2. shall be charged for the issuance or renewal of a plate issued on an annual basis for the special group specified under par. (f) 57. An additional fee of \$50 that is in addition to the fee under subd. 2. shall be charged for the issuance or renewal of a plate issued on the biennial basis for the special group specified under par. (f) 57. if the plate is issued or renewed during the first year of the biennial registration period or \$25 for the issuance or renewal if the plate is issued or renewed during the 2nd year of the biennial registration period. All moneys received under this subdivision, in excess of \$27,600 for the initial costs of production of the special group plate under par. (f) 57., shall be credited to the appropriation account under s. 20.435 (1) 20.395 (5) (gi). To the extent permitted under ch. 71, the fee under this subdivision is deductible as a charitable contribution for purposes of the taxes under ch. 71.

SECTION 1895. 341.14 (6r) (b) 11. of the statutes is amended to read:

341.14 (**6r**) (b) 11. An additional fee of \$25 that is in addition to the fee under subd. 2. shall be charged for the issuance or renewal of a plate issued on an annual basis for the special group specified under par. (f) 58. An additional fee of \$50 that is in addition to the fee under subd. 2. shall be charged for the issuance or renewal of a plate issued on the biennial basis for the special group specified under par. (f) 58. if the plate is issued or renewed during the first year of the biennial registration period or \$25 for the issuance or renewal if the plate is issued or renewed during the 2nd year of the biennial registration period. To the extent permitted under ch. 71, the fee under this subdivision is deductible as a charitable contribution for purposes of the taxes under ch. 71. All moneys received under this subdivision, in excess of \$43,200 for the initial costs of production of the special group plate under par. (f) 58., shall be credited to the appropriation account under s. 20.435 (1) (g) 20.395 (5) (gi).

SECTION 1895m. 341.25(1)(L) of the statutes is created to read:

341.25 **(1)** (L) 1. In this paragraph:

a. "Alternative fuel" means a gas, liquid, or other fuel that, with or without adjustment or manipulation such as adjustment or manipulation of pressure or temperature, is capable of being used for the generation of power to propel a motor vehicle, including, but not limited to, natural gas, compressed natural gas, liquefied natural gas, liquefied petroleum gas, hydrogen, hydrogen compressed natural gas, or hythane. Alternative fuel does not include motor fuel, electricity, leaded racing fuel, or an excluded liquid, as defined in 26 CFR 48.4081–1.

- b. "Hybrid electric vehicle" means a vehicle that is capable of using gasoline, diesel fuel, or alternative fuel to propel the vehicle but that is propelled to a significant extent by an electric motor that draws electricity from a battery that has a capacity of not less than 4 kilowatt hours and may be capable of being recharged from an external source of electricity.
- c. "Nonhybrid electric vehicle" means a vehicle that is propelled solely by electrical energy and that is not capable of using gasoline, diesel fuel, or alternative fuel to propel the vehicle.
- 2. If a motor truck or automobile is a hybrid electric vehicle, in addition to the fee under par. (a) or (c), a surcharge of \$75 shall be added to and collected with the fee for each automobile and for each motor truck registered under par. (c) at a gross weight of not more than 8,000 pounds.
- 3. If a motor truck or automobile is a nonhybrid electric vehicle, in addition to the fee under par. (a) or (c), a surcharge of \$100 shall be added to and collected with the fee for each automobile and for each motor truck registered under par. (c) at a gross weight of not more than 8,000 pounds.

SECTION 1895p. 342.06 (1) (eg) of the statutes is amended to read:

342.06 (1) (eg) Except as provided in par. (eh), if the applicant is an individual, the social security number of the applicant. The department of transportation may not disclose a social security number obtained under this paragraph to any person except to the department of children and families for the sole purpose of administering s. 49.22, to the department of workforce development for the sole purpose of enforcing or administering s. 108.22, and to the department of revenue for the purposes of administering state taxes and collecting debt, and to the elections commission for the sole purpose of allowing the chief election officer to comply with the terms of the agreement under s. 6.36 (1) (ae).

SECTION 1896. 343.14 (2) (gh) of the statutes is amended to read:

343.14 (2) (gh) A question as to whether the applicant wishes to designate an additional \$2 to support the efforts of Donate Life Wisconsin for the purposes described under s. 250.17 (1) 341.14 (8w).

SECTION 1896m. 343.14 (2j) of the statutes is amended to read:

343.14 (2j) Except as otherwise required to administer and enforce this chapter, the department of transportation may not disclose a social security number obtained from an applicant for a license under sub. (2) (bm) to any person except to the department of children and families for the sole purpose of administering s. 49.22, to the department of workforce development for the sole purpose of enforcing or administering s. 108.22, to the department of revenue for the purposes of administering state taxes and collecting debt, or to the driver licensing

agency of another jurisdiction, or to the elections commission for the sole purpose of allowing the chief election officer to comply with the terms of the agreement under s. 6.36 (1) (ae).

SECTION 1897. 343.21 (1) (o) of the statutes is amended to read:

343.21 (1) (o) In addition to any other fee under this subsection, \$2 for any person making a designation of an additional \$2 to support the efforts of Donate Life Wisconsin under s. 343.14 (2) (gh) or 343.50 (4). From the moneys received under this paragraph, 90 percent shall be deposited into the general fund and credited to the appropriation account under s. 20.435 (1) (g) 20.395 (5) (gj).

SECTION 1897m. 343.50 (8) (c) 3. of the statutes is created to read:

343.50 (8) (c) 3. Notwithstanding par. (b) and s. 343.14 (2j), the department may, upon request, provide to the elections commission for the sole purpose of allowing the chief election officer to comply with the terms of the agreement under s. 6.36 (1) (ae) any applicant information or identification card holder information maintained by the department of transportation and identified in s. 343.14 (2).

SECTION 1900. 348.105 (4) of the statutes is amended to read:

348.105 (4) An application for a permit shall be made to the department using an electronic application process established by the department. The department shall charge a fee of \$1,800 for a permit. All moneys received from fees imposed by the department under this subsection shall be deposited in the general transportation fund and credited to the appropriation account under s. 20.395 (5) (dg).

SECTION 1900j. 348.15 (3) (b) of the statutes is amended to read:

348.15 (3) (b) The gross weight imposed on the highway by the wheels of any one axle may not exceed 20,000 pounds or, if the vehicle or combination of vehicles is an implement of husbandry or agricultural commercial motor vehicle, or is a 2-vehicle combination transporting by trailer or semitrailer an implement of husbandry or agricultural commercial motor vehicle to or from a farm-related destination, and is operated on or before January 1, 2020, 23,000 pounds. In addition, the gross weight imposed on the highway by the wheels of the steering axle of a truck tractor may not exceed 13,000 pounds unless the manufacturer's rated capacity of the axle and the tires is sufficient to carry the weight, but not to exceed 20,000 pounds.

SECTION 1900k. 348.15 (3) (g) (intro.) of the statutes is amended to read:

348.15 (3) (g) (intro.) Notwithstanding par. (c), if the vehicle or combination of vehicles is an implement of husbandry or agricultural commercial motor vehicle, or is a 2-vehicle combination transporting by trailer or

semitrailer an implement of husbandry or agricultural commercial motor vehicle to or from a farm-related destination, and is operated on or before January 1, 2020, the gross weight imposed on the highway by any group of 2 or more consecutive axles of the vehicle or vehicle combination may not exceed the maximum gross weights in the following table for each of the respective distances between axles and the respective numbers of axles of a group:

SECTION 1900L. 348.15 (9) (g) of the statutes is repealed.

SECTION 1900Lm. 348.15 (11) of the statutes is created to read:

348.15 (11) The maximum gross weight for a a vehicle or combination of vehicles having 11 axles and transporting exclusively forest products or lumber operated on a highway without a permit may not exceed 164,000 pounds. Notwithstanding sub. (3) (a), (b), (br), and (c), there is no weight limitation per wheel, axle, or group of axles for a vehicle or combination of vehicles having 11 axles and transporting exclusively forest products or lumber. This subsection applies only on the following:

- (a) STH 13, from the junction of USH 2 and STH 13 in the city of Ashland to the intersection of STH 13 and Old Airport Road in Ashland County.
- (b) STH 70, from the junction of STH 70 and USH 45 in the city of Eagle River to the junction of STH 70 and USH 51 in Vilas County.

SECTION 1900m. 348.16 (2) of the statutes is amended to read:

348.16 (2) Except as provided in sub. (3) and s. 348.175 and subject to any modifications made by a city of the first class pursuant to s. 349.15 (3), no person, without a permit therefor, shall operate on a class "B" highway any vehicle or combination of vehicles imposing wheel, axle, group of axles, or gross weight on the highway exceeding 60 percent of the weights authorized in s. 348.15 (3). This subsection does not apply, from April 24, 2014, to January 1, 2020, to an implement of husbandry or agricultural commercial motor vehicle being operated or transported, or to a 2-vehicle combination transporting an implement of husbandry or agricultural commercial motor vehicle, as described in s. 348.15 (9) (c), (cm), (e), or (f) 1. or 1m.

SECTION 1900n. 348.17 (6) (a) 3. of the statutes is repealed.

SECTION 1900o. 348.21 (3t) (c) of the statutes is repealed.

SECTION 1900og. 348.27 (12) of the statutes is renumbered 348.27 (12) (a) and amended to read:

348.27 (12) (a) The department may issue an annual or consecutive month permit for the transportation of garbage, as defined in s. 289.01 (9), or refuse, in a self-compactor equipped vehicle which exceeds statutory weight and length limitations and for the return of the vehicle when empty. A permit under this subsection may

be issued for use on any highway within this state. In this subsection, "refuse" means combustible and noncombustible rubbish, including paper, wood, metal, glass, cloth and products thereof, litter and street rubbish, ashes, and lumber, concrete, and other debris resulting from the construction or demolition of structures.

SECTION 1900om. 348.27 (12) (b) of the statutes is created to read:

348.27 (12) (b) The department may issue an annual or consecutive month permit for the transportation of garbage, as defined in s. 289.01 (9), or refuse, in any of the following vehicles that exceed statutory weight and length limitations and for the return of the vehicle when empty:

- 1. A self-compactor equipped vehicle.
- 2. A roll-off equipped truck or roll-off trailer that uses all axles while transporting garbage or refuse.

SECTION 1900or. 348.27 (12) (c) of the statutes is created to read:

348.27 (12) (c) A permit under par. (b) may be issued for use on any highway within this state.

SECTION 1900p. 348.27 (19) (f) of the statutes is repealed.

SECTION 1900q. 348.28 (1) (a) of the statutes is amended to read:

348.28 (1) (a) Permits issued under ss. 348.25, 348.26 and 348.27, other than a permit described in s. 348.27 (19) (b) 5. b., shall be carried on the vehicle during operations so permitted. This paragraph does not apply after January 1, 2020, with respect to a permit issued under s. 348.27 (19).

SECTION 1900r. 348.28 (1) (b) of the statutes is amended to read:

348.28 (1) (b) Permits issued under s. 348.27 (19) that are required to be carried on the vehicle under par. (a) may be carried and produced in either printed or electronic format, including by display of electronic images on a cellular telephone or other electronic device. If the permit is displayed in electronic format on any cellular telephone or other electronic device, a traffic officer or inspector under s. 110.07 (3) may not view, and producing the permit in electronic format is not considered consent for the traffic officer or inspector to view, any content on the telephone or other device except the permit. This paragraph does not apply after January 1, 2020.

SECTION 1900sg. 350.12 (3j) (b) of the statutes is amended to read:

350.12 (**3j**) (b) Except as provided in par. (bg), the The fee for a trail use sticker issued for a snowmobile that is exempt from registration under sub. (2) (b) or (bn) is \$34.25 \$49.25. A trail use sticker issued for such a snowmobile may be issued only by the department and persons appointed by the department and expires on June 30 of each year.

SECTION 1900sm. 350.12 (3j) (bg) 2. of the statutes is repealed.

SECTION 1900sp. 350.12 (3j) (bg) 3. of the statutes is amended to read:

350.12 (**3j**) (bg) 3. The fee for a trail use sticker issued in fiscal years 2015–16 to 2018–19 for a snowmobile that is registered under sub. (3) and that is owned by a person who is both a member of a snowmobile club and a member of the Association of Wisconsin Snowmobile Clubs is \$9.25.

SECTION 1900sv. 350.12 (3j) (bg) 4. of the statutes is amended to read:

350.12 (**3j**) (bg) 4. The fee for a trail use sticker issued in fiscal years 2015–16 to 2018–19 for a snowmobile that is registered under sub. (3) and that is owned by a person who does not meet the requirements under subd. 3. is \$29.25.

SECTION 1900t. 350.12 (3j) (e) 2. of the statutes is amended to read:

350.12 (3j) (e) 2. Any person, including the department, who issues a trail use sticker or a temporary trail use receipt shall collect in addition to the fee under par. (b) or (bg) $\frac{2}{3}$, or 4. an issuing fee of 75 cents. The agent may retain 50 cents of the issuing fee to compensate the agent for the agent's services in issuing the temporary trail use receipt.

SECTION 1900v. 350.12 (4) (b) 1. of the statutes is amended to read:

350.12(4) (b) 1. State aids and funds for maintenance costs shall be 100 percent of the actual cost of maintaining the trail per year up to a \$250 \$300 per mile per year maximum, except as provided in pars. (bg) to (br). Qualifying trails are trails approved by the board as snowmobile trails. State aid for development may equal 100 percent of development expenses. Aids for major reconstruction or rehabilitation projects to improve bridges may equal 100 percent of eligible costs. Aids for trail rehabilitation projects may equal 100 percent of eligible costs. Development shall begin the same year the land is acquired. Moneys available for development shall be distributed on a 100 percent grant basis, 75 percent at the time of approval but no later than January 1 and 25 percent upon completion of the project. A county application may include a request for development, rehabilitation or maintenance of trails, or any combination thereof. Trail routes, sizes and specifications shall be prescribed only by the board.

SECTION 1900vm. 350.12 (4) (bg) 2. a. of the statutes is repealed.

SECTION 1900w. 350.12 (4) (bg) 2. b. of the statutes is renumbered 350.12 (4) (bg) 2. and amended to read:

350.12 **(4)** (bg) 2. For fiscal years 2016–17 to 2019–20, the <u>The</u> department shall calculate an amount equal to the number of trail use stickers issued under sub. (3j) in the previous fiscal year to owners of snowmobiles that are exempt from registration under sub. (2) multiplied by \$47 and shall credit this amount to the appropriation account under s. 20.370 (5) (cw). From the appropri

ation account under s. 20.370 (5) (cw), the department shall make payments to the department or a county for the purposes specified in par. (b). The department shall make payments under par. (bm) for trail maintenance costs that were incurred in the previous fiscal year and that exceed the maximum specified under par. (b) 1. before making payments for any of the other purposes specified in par. (b).

SECTION 1900wm. 350.12 (4) (bm) 1. of the statutes is amended to read:

350.12 (4) (bm) 1. The actual cost incurred by the department or the county in maintaining its trails that are qualified under par. (b) 1. or 4. in the previous fiscal year exceeds the maximum of \$250 \$300 per mile per year under par. (b) 1.

SECTION 1900x. 350.12 (4) (bm) 2. of the statutes is amended to read:

350.12 (4) (bm) 2. Of the actual cost incurred by the department or the county in maintaining its trails that are qualified under par. (b) 1. or 4. for the fiscal year applicable under subd. 1., the actual cost incurred in grooming the trails exceeds a maximum of \$150 \$200 per mile per year.

SECTION 1903. 440.03 (1c) of the statutes is created to read:

440.03 (1c) The department shall promulgate rules specifying a procedure for addressing allegations that a credential holder has practiced while impaired by alcohol or other drugs or that his or her ability to practice is impaired by alcohol or other drugs, and for assisting a credential holder who requests to participate in the procedure. In promulgating rules under this subsection, the department shall seek to facilitate early identification of chemically dependent credential holders and encourage their rehabilitation. The rules promulgated under this subsection may be used by the department, the real estate appraisers board, and all examining boards and affiliated credentialing boards attached to the department or an examining board in conjunction with the procedures established under sub. (1). The department may contract with another entity to administer the procedure specified under the rules promulgated under this subsection.

SECTION 1904. 440.03 (4m) of the statutes is created to read:

440.03 (4m) Except as otherwise permitted in chs. 440 to 480, the department may require a credential holder to submit proof of the continuing education programs or courses that he or she has completed only if a complaint is made against the credential holder.

SECTION 1929. 440.035 (2) of the statutes is created to read:

440.035 (2) Except as otherwise permitted in chs. 440 to 480, an examining board or affiliated credentialing board attached to the department or an examining board may require a credential holder to submit proof of the continuing education programs or courses that he or she

has completed only if a complaint is made against the credential holder.

SECTION 1931. 440.23 (2) (c) of the statutes is amended to read:

440.23 (2) (c) Pays the charge for an unpaid draft established by the depository selection board under s. 20.905 (2).

SECTION 1935. Subchapter V (title) of chapter 440 [precedes 440.51] of the statutes is amended to read:

CHAPTER 440

SUBCHAPTER V

PEDDLERS; PRIVATE SCHOOLS

SECTION 1936am. 440.52 (title) of the statutes, as affected by 2017 Wisconsin Act (this act), is repealed and recreated to read:

440.52 (title) Private trade, correspondence, business, and technical schools.

SECTION 1936b. 440.52 (1) (a) of the statutes, as affected by 2017 Wisconsin Act (this act), is repealed.

SECTION 1936c. 440.52 (1) (e) 8. of the statutes, as affected by 2017 Wisconsin Act (this act), is amended to read:

440.52 (1) (e) 8. Schools accredited by accrediting agencies recognized by the board department.

SECTION 1936d. 440.52 (1) (g) of the statutes, as affected by 2017 Wisconsin Act (this act), is amended to read:

440.52 (1) (g) "Teaching location" means the area and facilities designated for use by a school required to be approved by the board department under this section.

SECTION 1936e. 440.52 (2) of the statutes, as affected by 2017 Wisconsin Act (this act), is amended to read:

440.52 (2) RESPONSIBILITIES. The board department shall protect the general public by inspecting and approving private trade, correspondence, business, and technical schools doing business within this state, whether located within or outside this state, changes of ownership or control of the schools, teaching locations used by the schools, and courses of instruction offered by the schools and regulate the soliciting of students for correspondence or classroom courses and courses of instruction offered by the schools.

SECTION 1936f. 440.52 (3) of the statutes, as affected by 2017 Wisconsin Act (this act), is amended to read:

440.52 (3) RULE-MAKING POWER. The board department shall promulgate rules and establish standards necessary to administer this section.

SECTION 1936g. 440.52 (5) of the statutes, as affected by 2017 Wisconsin Act (this act), is repealed.

SECTION 1936h. 440.52 (7) (intro.) of the statutes, as affected by 2017 Wisconsin Act (this act), is amended to read:

440.52 (7) APPROVAL OF SCHOOLS GENERALLY. (intro.) To protect students, prevent fraud and misrepresentation in the sale and advertising of courses and courses of instruction, and encourage schools to maintain courses

and courses of instruction consistent in quality, content, and length with generally accepted educational standards, the board department shall do all of the following:

SECTION 1936i. 440.52 (7) (g) of the statutes, as affected by 2017 Wisconsin Act (this act), is amended to read:

440.52 (7) (g) Approve courses of instruction, schools, changes of ownership or control of schools, and teaching locations meeting the requirements and standards established by the board department and complying with rules promulgated by the board department; publish a list of the schools and courses of instruction approved and a list of the schools that are authorized to use the term "college," "university," "state," or "Wisconsin" in their names; and make those lists of the schools available on the board's department's Internet site.

SECTION 1936j. 440.52 (7) (h) of the statutes, as affected by 2017 Wisconsin Act (this act), is amended to read:

440.52 (7) (h) Issue permits to solicitors when all board department requirements have been met.

SECTION 1936k. 440.52 (7) (i) of the statutes, as affected by 2017 Wisconsin Act (this act), is amended to read:

440.52 (7) (i) Require schools to furnish a surety bond in an amount as provided by rule of the board department.

SECTION 1936L. 440.52 (8) (a) of the statutes, as affected by 2017 Wisconsin Act (this act), is amended to read:

440.52 **(8)** (a) *In general*. No solicitor representing any school offering any course or course of instruction shall sell any course or course of instruction or solicit students for a course or course of instruction in this state for a consideration or remuneration, except upon the actual business premises of the school, unless the solicitor first secures a solicitor's permit from the board department. If the solicitor represents more than one school, a separate permit shall be obtained for each school the solicitor represents.

SECTION 1936m. 440.52 (8) (b) of the statutes, as affected by 2017 Wisconsin Act (this act), is amended to read:

440.52 (8) (b) *Solicitor's permit*. The application for a solicitor's permit shall be made on a form furnished by the board department and shall be accompanied by a fee and a surety bond acceptable to the board department in the sum of \$2,000. The board department shall, by rule, specify the amount of the fee for a solicitor's permit. The bond may be continuous and shall be conditioned to provide indemnification to any student suffering loss as the result of any fraud or misrepresentation used in procuring his or her enrollment or as a result of the failure of the school to perform faithfully the agreement the solicitor made with the student, and may be supplied by the solicitor or by the school itself either as a blanket bond cover-

ing each of its solicitors in the amount of \$2,000 or the surety bond under sub. (7) (i). Upon approval of a permit, the board department shall issue an identification card to the solicitor giving his or her name and address, the name and address of the employing school, and certifying that the person whose name appears on the card is authorized to solicit students for the school. A permit shall be valid for one year from the date issued. Liability under this paragraph of the surety on the bond for each solicitor covered by the bond shall not exceed the sum of \$2,000 as an aggregate for any and all students for all breaches of the conditions of the bond. The surety of a bond may cancel the bond upon giving 30 days' notice in writing to the board department and shall be relieved of liability under this paragraph upon giving the notice for any breach of condition occurring after the effective date of the cancellation. An application for renewal shall be accompanied by a fee, a surety bond acceptable to the board department in the sum of \$2,000 if a continuous bond has not been furnished, and such information as the board department requests of the applicant. The board department shall, by rule, specify the amount of the fee for renewal of a solicitor's permit.

SECTION 1936n. 440.52 (8) (c) (intro.), 1., 2., 4. and 5. of the statutes, as affected by 2017 Wisconsin Act (this act), are amended to read:

440.52 **(8)** (c) Refusal or revocation of permit. (intro.) The board department may refuse to issue or renew, or may revoke, any solicitor's permit upon one or any combination of the following grounds:

- 1. Willful violation of this subsection or any rule promulgated by the board department under this section.
- 2. Furnishing false, misleading, or incomplete information to the board department.
- 4. Refusal by the school to be represented to allow reasonable inspection or to supply information after written request therefor by the board department.
- 5. Failure of the school which the solicitor represents to meet requirements and standards established by and to comply with rules promulgated by the board department under sub. (7).

SECTION 19360. 440.52 (8) (d) of the statutes, as affected by 2017 Wisconsin Act (this act), is amended to read:

440.52 (8) (d) *Notice of refusal to issue or renew permit.* Notice of refusal to issue or renew a permit or of the revocation of a permit shall be sent by registered mail to the last address of the applicant or permit holder shown in the records of the board department. Revocation of a permit shall be effective 10 days after the notice of revocation has been mailed to the permit holder.

SECTION 1936p. 440.52 (8) (e) of the statutes, as affected by 2017 Wisconsin Act (this act), is amended to read:

440.52 (8) (e) Request for appearance. Within 20 days of the receipt of notice of the board's department's refusal to issue or renew a permit or of the revocation of a permit, the applicant or holder of the permit may request permission to appear before the board department in person, with or without counsel, to present reasons why the permit should be issued, renewed, or reinstated. Upon receipt of a request, the board department shall grant a hearing to the applicant or holder of the permit within 30 days giving that person at least 10 days' notice of the date, time, and place.

SECTION 1936q. 440.52 (10) (a) of the statutes, as affected by 2017 Wisconsin Act (this act), is amended to read:

440.52 (10) (a) Authority. All proprietary schools shall be examined and approved by the board department before operating in this state. Approval shall be granted to schools meeting the criteria established by the board department for a period not to exceed one year. No school may advertise in this state unless approved by the board department. All approved schools shall submit quarterly reports, including information on enrollment, number of teachers and their qualifications, course offerings, number of graduates, number of graduates successfully employed, and such other information as the board department considers necessary. If a school closure results in losses to students, parents, or sponsors, the board department may authorize the full or partial payment of those losses from the appropriation under s. 20.165 (1) (it).

SECTION 1936r. 440.52 (10) (b) of the statutes, as affected by 2017 Wisconsin Act (this act), is amended to read:

440.52 (10) (b) *Application*. Application for initial approval of a school or a course of instruction, approval of a teaching location, change of ownership, or control of a school, renewal of approval of a school or reinstatement of approval of a school or course of instruction that has been revoked shall be made on a form furnished by the board department and shall be accompanied by a fee set by the board department under par. (c) and any other information as the board department considers necessary to evaluate the school in carrying out the purpose of this section.

SECTION 1936s. 440.52 (10) (c) (intro.) and 1. of the statutes, as affected by 2017 Wisconsin Act (this act), are amended to read:

440.52 (10) (c) Fees; rule making. (intro.) The board department shall promulgate rules to establish the fees paid to the board department under this subsection. In promulgating rules to establish the fees, the board department shall do all of the following:

1. Require that the amount of fees collected under this paragraph be sufficient to cover all costs that the board department incurs in examining and approving proprietary schools under this subsection.

SECTION 1936t. 440.52 (10) (cm) of the statutes, as affected by 2017 Wisconsin Act (this act), is amended to read:

440.52 (10) (cm) *Limit on student protection fee.* The board department shall discontinue collecting annual student protection fees under par. (c) 4. during the period that the balance in the fund created by those fees exceeds \$1,000,000.

SECTION 1936u. 440.52 (11) (b) 1. of the statutes, as affected by 2017 Wisconsin Act (this act), is amended to read:

440.52 (11) (b) 1. If a school operating in this state discontinues its operations, proposes to discontinue its operations, or is in imminent danger of discontinuing its operations as determined by the board department, if the student records of the school are not taken into possession under subd. 2., and if the board department determines that the student records of the school are in danger of being destroyed, secreted, mislaid, or otherwise made unavailable to the persons who are the subjects of those student records or the authorized representatives of those persons, the board department may take possession of those student records.

SECTION 1936v. 440.52 (11) (c) of the statutes, as affected by 2017 Wisconsin Act (this act), is amended to read:

440.52 (11) (c) If necessary to protect student records from being destroyed, secreted, mislaid, or otherwise made unavailable to the persons who are the subjects of those student records or the authorized representatives of those persons, the board department or association may seek a court order authorizing the board department or association to take possession of those student records.

SECTION 1936w. 440.52 (11) (d) of the statutes, as affected by 2017 Wisconsin Act (this act), is amended to read:

440.52 (11) (d) The board department or association shall preserve a student record that comes into the possession of the board department or association under par. (b) 1. or 2. and shall keep the student record confidential as provided under 20 USC 1232g and 34 CFR part 99. A student record in the possession of the board department is not open to public inspection or copying under s. 19.35 (1). Upon request of the person who is the subject of a student record or an authorized representative of that person, the board department or association shall provide a copy of the student record to the requester. The board department or association may charge a fee for providing a copy of a student record. The fee shall be based on the administrative cost of taking possession of, preserving, and providing the copy of the student record. All fees collected by the board department under this paragraph shall be credited to the appropriation account under s. 20.165 (1) (jv).

SECTION 1936x. 440.52 (12) (a) (intro.) and 1. of the statutes, as affected by 2017 Wisconsin Act (this act), are amended to read:

440.52 (12) (a) (intro.) No person that holds itself out to the public in any way as a legitimate institution of higher education may use the term "college" or "university" in the person's name unless the person provides an educational program for which the person awards an associate or higher degree and the person has accreditation recognized by the U.S. secretary of education, has the foreign equivalent of that accreditation, as determined by the board department, or has accreditation recognized by the Council for Higher Education Accreditation. This paragraph does not apply to any of the following:

1. A school that was doing business in this state with the approval of the <u>educational approval</u> board <u>under s.</u> 38.50, 2007 stats., prior to May 27, 2010.

SECTION 1936y. 440.52 (12) (b) of the statutes, as affected by 2017 Wisconsin Act (this act), is amended to read:

440.52 (12) (b) No school, including a school described in sub. (1) (e) 1. to 8., may use the term "state" or "Wisconsin" in its name if the use of that term operates to mislead the public into believing that the school is affiliated with the University of Wisconsin System or the technical college system, unless the school actually is so affiliated. This paragraph does not apply to a school described in sub. (1) (e) 1. that has accreditation recognized by the U.S. secretary of education, has the foreign equivalent of that accreditation, as determined by the board department, or has accreditation recognized by the Council for Higher Education Accreditation.

SECTION 1936z. 440.52 (13) (a) 2. a., b. and e. of the statutes, as affected by 2017 Wisconsin Act (this act), are amended to read:

440.52 (13) (a) 2. a. Has accreditation recognized by the U.S. secretary of education; has the foreign equivalent of that accreditation, as determined by the board department; or has accreditation recognized by the Council for Higher Education Accreditation.

b. Is approved by the board <u>department</u> to operate in this state.

e. Has been found by the <u>board department</u> to meet standards of academic quality comparable to those of an educational institution located in the United States that has accreditation recognized by the U.S. secretary of education or by the Council for Higher Education Accreditation to offer credentials of the type and level claimed.

SECTION 1936zm. 440.52 (13) (d) of the statutes, as affected by 2017 Wisconsin Act (this act), is amended to read:

440.52 (13) (d) The board department may charge a fee for evaluating an educational institution under par. (a) 2. e. in an amount that is sufficient to cover all costs that the board department incurs in evaluating the institution.

All fees collected by the board <u>department</u> under this paragraph shall be credited to the appropriation account under s. 20.165 (1) (jr).

SECTION 2149m. 460.05 (1) (e) 1. of the statutes is amended to read:

460.05 (1) (e) 1. Graduated from a school of massage therapy or bodywork therapy approved by the educational approval board under s. 38.50 440.52 that meets the requirements under s. 460.095 or completed a training program approved by the affiliated credentialing board under the rules promulgated under s. 460.04 (2) (b).

SECTION 2149p. 460.05 (1) (e) 1. of the statutes, as affected by 2017 Wisconsin Act (this act), is amended to read:

460.05 (1) (e) 1. Graduated from a school of massage therapy or bodywork therapy approved by the educational approval board department under s. 440.52 that meets the requirements under s. 460.095 or completed a training program approved by the affiliated credentialing board under the rules promulgated under s. 460.04 (2) (b).

SECTION 2212. 563.055 (2) (b) of the statutes is amended to read:

563.055 **(2)** (b) Pays the charge for an unpaid draft established by the depository selection board under s. 20.905 (2).

SECTION 2212g. 601.31 (1) (n) of the statutes is amended to read:

601.31 (1) (n) For appointing, or renewing an appointment of, an agent under s. 628.11, \$16 annually for resident agents or \$40 \$30 annually for nonresident agents, unless the commissioner sets a higher fee by rule, to be paid at times and under procedures set by the commissioner.

SECTION 2212m. 601.415 (13) of the statutes is created to read:

601.415 (13) Membership in the National Conference of Insurance Legislators. Annually, from the appropriation account under s. 20.145 (1) (g), the commissioner shall credit to the appropriation account under s. 20.765 (3) (g) an amount sufficient for the payment of annual dues by the legislature for membership in the National Conference of Insurance Legislators.

SECTION 2213. 605.03 (1) (a) of the statutes is amended to read:

605.03 (1) (a) *Mandatory coverage*. The Subject to par. (f), the property fund shall provide protection against fire and extended coverage perils. The coverage shall be at least as favorable as that customarily provided by policies filed with the commissioner for the use of private insurers in insuring comparable property.

SECTION 2214. 605.03 (1) (d) of the statutes is amended to read:

605.03 (1) (d) *Term of policy*. The <u>Subject to par. (f)</u> 1., the manager may prescribe the time periods for which coverage is to be provided.

SECTION 2215. 605.03 (1) (f) of the statutes is created to read:

605.03 (1) (f) Limits on issuance, renewal, and filing claims; final distribution. 1. No coverage under the property fund may be issued on or after July 1, 2017. No coverage may be renewed after December 31, 2017. No coverage may terminate later than December 31, 2018.

- 2. All claims must be filed with the property fund by no later than July 1, 2019. No claim filed after July 1, 2019, will be covered by the fund.
- 3. Upon the cessation of all operations of the property fund, the manager shall distribute any moneys remaining in the fund among the local governmental units that were insured under the fund on July 1, 2017.

SECTION 2216. 605.21 (1) of the statutes is amended to read:

605.21 (1) PLACING INSURANCE. The Subject to s. 605.03 (1) (f), the property fund shall insure property described in s. 605.02 after receipt from the clerk of the local governmental unit of a certified copy of the resolution authorizing insurance in the property fund. The clerk shall report to the manager each policy then in force upon such property, stating the property covered by the policy and the dates of issue and of expiration, the amounts and rates of insurance and the premiums. Property already insured shall become insured by the property fund as existing policies expire or are canceled. Thereafter the insurance on all property described in s. 605.02 shall be provided. Premiums shall be certified by the manager to the clerk of the appropriate unit.

SECTION 2217. 605.23 (1) of the statutes is amended to read:

605.23 (1) PAYMENT FOR LOSSES. The Subject to s. 605.03 (1) (f) 2., the manager shall determine within a reasonable time any loss on insured property owned by a local governmental unit or for which the unit is liable and promptly certify the amount to the department of administration, which shall issue a warrant on the property fund payable to the treasurer of the local governmental unit for the amount of the loss less any applicable amounts under s. 605.03 (2) or (3).

SECTION 2218. 605.35 of the statutes is repealed. **SECTION 2221.** 706.05 (12) of the statutes is amended to read:

706.05 (12) Every conveyance of any interest in real property offered for recordation shall be accompanied by the form under s. 77.22 (2). If the property is subject to certification under s. 101.122 (4) (a), waiver under s. 101.122 (4) (b) or stipulation under s. 101.122 (4) (c), the documents of conveyance offered for recordation shall have appended the certificate required under s. 101.122

(4) (a), a waiver under s. 101.122 (4) (b) or a stipulation under s. 101.122 (4) (c).

SECTION 2225d. 770.001 of the statutes is repealed. SECTION 2225f. 770.05 (intro.) of the statutes is amended to read:

770.05 Criteria for forming a domestic partnership. (intro.) Two Subject to s. 770.07 (1) (a), 2 individuals may form a domestic partnership if they satisfy all of the following criteria:

SECTION 2225p. 770.07 (1) (a) of the statutes is amended to read:

770.07 (1) (a) Individuals who wish to To form a domestic partnership shall, individuals must apply on or after the 31st day beginning after July 1, 2009, but no later than the effective date of this paragraph [LRB inserts datel, for a declaration of domestic partnership to the county clerk of the county in which at least one of the individuals has resided for at least 30 days immediately before applying.

SECTION 2225r. 770.07 (3) of the statutes is created

770.07 (3) No county clerk may issue a declaration of domestic partnership to individuals who apply after the effective date of this subsection [LRB inserts date].

SECTION 2226. 778.25 (1) (a) 7. of the statutes is repealed.

SECTION 2227. 778.25 (1) (b) of the statutes is amended to read:

778.25 (1) (b) The citation form provided by this section may serve as the initial pleading for the action and, except as provided in par. (c), is adequate process to give a court jurisdiction over the person if the citation is filed with the court.

SECTION 2228. 778.25 (1) (c) of the statutes is repealed.

SECTION 2229j. 779.14 (1e) (b) of the statutes is amended to read:

779.14 (1e) (b) All contracts that are in excess of \$30,000, as indexed under sub. (1s), and that are for performing, furnishing, or procuring labor, services, materials, plans, or specifications for a public improvement or public work shall contain a provision under which the prime contractor agrees, to the extent practicable, to maintain a list of all subcontractors, suppliers, and service providers performing, furnishing, or procuring labor, services, materials, plans, or specifications under the contract.

SECTION 2229k. 779.14 (1m) (c) 1. (intro.) of the statutes is amended to read:

779.14 (1m) (c) 1. (intro.) In the case of a contract with a contract price exceeding \$10,000, as indexed under sub. (1s), \$16,000 but not exceeding \$100,000, as indexed under sub. (1s) \$148,000:

SECTION 2229m. 779.14 (1m) (c) 2. (intro.) of the statutes is amended to read:

779.14 (1m) (c) 2. (intro.) In the case of a contract with a contract price exceeding \$100,000, as indexed under sub. (1s), \$148,000 but not exceeding \$250,000, as indexed under sub. (1s) \$369,000:

SECTION 2229n. 779.14 (1m) (c) 3. of the statutes is amended to read:

779.14(1m) (c) 3. In the case of a contract with a contract price exceeding \$250,000, as indexed under sub. (1s), \$369,000 the contract shall require the prime contractor to obtain a payment and performance bond meeting the requirements under par. (e).

SECTION 2229p. 779.14 (1m) (d) 1. (intro.) of the statutes is amended to read:

779.14 (**1m**) (d) 1. (intro.) In the case of a contract with a contract price exceeding \$10,000, as indexed under sub. (1s), \$16,000 but not exceeding \$50,000, as indexed under sub. (1s) \$74,000:

SECTION 2229r. 779.14 (1m) (d) 2. (intro.) of the statutes is amended to read:

779.14 (1m) (d) 2. (intro.) In the case of a contract with a contract price exceeding \$50,000, as indexed under sub. (1s), \$74,000 but not exceeding \$100,000, as indexed under sub. (1s) \$148,000:

SECTION 2229s. 779.14 (1m) (d) 3. of the statutes is amended to read:

779.14 (**1m**) (d) 3. Except as provided in sub. (4), in the case of a contract with a contract price exceeding \$100,000, as indexed under sub. (1s), \$148,000 the contract shall require the prime contractor to obtain a payment and performance bond meeting the requirements under par. (e).

SECTION 2229t. 779.14 (1s) of the statutes is repealed.

SECTION 2230s. 809.30 (2) (d) of the statutes is Vetoed amended to read:

809.30 (2) (d) Indigency redetermination. Except as provided in this paragraph, whenever a person whose trial counsel is appointed by the state public defender files a notice under par. (b) requesting public defender representation for purposes of postconviction or postdisposition relief, the prosecutor may, within 5 days after the notice is served and filed, file in the circuit court and serve upon the state public defender a request that the person's indigency be redetermined before counsel is appointed or transcripts are requested. This paragraph does not apply to a person who is entitled to be represented by counsel under s. 48.23, 51.60 (1), 55.105, or

SECTION 2233. 815.18 (3) (o) of the statutes is amended to read:

938.23, or 980.03 (2) (a).

815.18 (3) (o) Tuition units. Tuition units purchased under s. 16.64 224.48.

SECTION 2234. 815.18 (3) (p) of the statutes is amended to read:

815.18 (3) (p) College savings accounts. An interest in a college savings account under s. 16.641 224.50.

SECTION 2235. 846.167 (2) (a) of the statutes is amended to read:

846.167 (2) (a) If the purchaser is not the judgment creditor, before the court may confirm the sale, the purchaser shall provide the judgment creditor with any information required for the judgment creditor to complete the real estate transfer return under s. 77.22 and, if applicable, any information required for a certificate, waiver, or stipulation required under s. 101.122.

SECTION 2236. 846.167 (2) (b) 2. b. of the statutes is amended to read:

846.167 (2) (b) 2. b. Any other document required for the register of deeds to record the deed, including any certificate, waiver, or stipulation required under s. 101.122.

SECTION 2237. 846.167 (2) (c) of the statutes is amended to read:

846.167 (2) (c) No later than 10 days after the court confirms the sale, the judgment creditor shall provide to the court the receipt for submitting a transfer return under s. 77.22 and any certificate, waiver, or stipulation required under s. 101.122.

SECTION 2238. 846.167 (3) of the statutes is amended

846.167 (3) Upon the court confirming the sale of mortgaged premises located in a county and upon compliance by the purchaser with the terms of the sale and the payment of any balance of the sale price to be paid, unless otherwise ordered by the court, the clerk of the court shall transmit the deed to the mortgaged premises received under s. 846.16, the receipt for submitting a transfer return under s. 77.22, any certificate, waiver, or stipulation required under s. 101.122, the amount due under s. 59.43 (2) to record the deed and any other document required to record the deed, and the transfer fee, if any, to the register of deeds of the county.

SECTION 2245. 938.396 (2g) (o) of the statutes is amended to read:

938.396 (2g) (o) Criminal history record search. If a juvenile is adjudged delinquent for committing a serious crime, as defined in s. 48.685 (1) (c) or 48.686 (1) (c), the court clerk shall notify the department of justice of that fact. No other information from the juvenile's court records may be disclosed to the department of justice except by order of the court. The department of justice may disclose any information provided under this subsection only as part of a criminal history record search under s. 48.685 (2) (am) 1. or (b) 1. a. 1m. or s. 48.686 (2)

SECTION 2246. 938.485 (4) of the statutes is amended to read:

938.485 (4) REIMBURSEMENT OF TRIBES AND COUN-TIES FOR TRIBAL DELINQUENCY PLACEMENTS. Reimburse Indian tribes and county departments, from the appropriation under s. 20.437 (1) (kp) (kz), for unexpected or unusually high-cost out-of-home care placements of Indian juveniles who have been adjudicated delinquent

by tribal courts. In this subsection, "unusually high-cost out-of-home care placements" means the amount by which the cost to an Indian tribe or to a county department of out-of-home care placements of Indian juveniles who have been adjudicated delinquent by tribal courts exceeds \$50,000 in a fiscal year.

SECTION 2248m. 944.21 (8) (b) 3. a. of the statutes is amended to read:

944.21 (8) (b) 3. a. Is a technical college, is a school approved by the educational approval board under s. 38.50 440.52, or is a school described in s. 38.50 440.52 (1) (e) 6., 7. or 8.; and

SECTION 2248p. 944.21 (8) (b) 3. a. of the statutes, as affected by 2017 Wisconsin Act (this act), is amended to read:

944.21 (8) (b) 3. a. Is a technical college, is a school approved by the educational approval board department of safety and professional services under s. 440.52, or is a school described in s. 440.52 (1) (e) 6., 7. or 8.; and

SECTION 2249. 946.15 of the statutes is repealed.

SECTION 2249e. 946.91 (3) (c) 3. of the statutes is cre- **Vetoed** ated to read:

946.91 (3) (c) 3. Any payment made for sharing of cost savings under s. 49.45 (26g).

SECTION 2249g. 946.93 (5) (c) 3. of the statutes is created to read:

946.93 (5) (c) 3. Any payment made for sharing of cost savings under s. 49.45 (26g).

SECTION 2250m. 948.11 (4) (b) 3. a. of the statutes is amended to read:

948.11 (4) (b) 3. a. Is a technical college, is a school approved by the educational approval board under s. 38.50 440.52, or is a school described in s. 38.50 440.52 (1) (e) 6., 7. or 8.; and

SECTION 2250p. 948.11 (4) (b) 3. a. of the statutes, as affected by 2017 Wisconsin Act (this act), is amended to read:

948.11 (4) (b) 3. a. Is a technical college, is a school approved by the educational approval board department of safety and professional services under s. 440.52, or is a school described in s. 440.52 (1) (e) 6., 7. or 8.; and

SECTION 2251c. 961.385 (2) (cs) 1. of the statutes, as created by 2015 Wisconsin Act 266, is amended to read:

961.385 (2) (cs) 1. Require -a practitioner to review that a patient's records under the program be reviewed before the practitioner issues a prescription order for the patient. The review required under this subdivision may be performed by the practitioner or by the practitioner's agent in accordance with applicable standards of practice. This subdivision does not apply after April 1, 2020.

SECTION 2251g. 961.385 (2) (cs) 2. (intro.) of the statutes, as created by 2015 Wisconsin Act 266, is amended to read:

961.385 (2) (cs) 2. (intro.) The requirement under subd. 1. that a practitioner review a patient's records under the program be reviewed before the practitioner

issues a prescription order for the patient does not apply if any of the following is true:

SECTION 2251L. 961.385 (2) (cs) 2. d. of the statutes, as created by 2015 Wisconsin Act 266, is amended to

961.385 (2) (cs) 2. d. Due to emergency, it is not possible for the practitioner to review the patient's records under the program before the practitioner issues a prescription order for the patient.

SECTION 2251p. 961.385 (2) (cs) 2. e. of the statutes, as created by 2015 Wisconsin Act 266, is amended to read:

961.385 (2) (cs) 2. e. The practitioner is unable It is not possible to review the patient's records under the program because the digital platform for the program is not operational or due to other technological failure if the practitioner reports that failure is reported to the board.

SECTION 2251t. 961.385 (2) (f) of the statutes is amended to read:

961.385 (2) (f) Permit the board to refer to the appropriate licensing or regulatory board for discipline, or the appropriate law enforcement agency for investigation and possible prosecution, a pharmacist, pharmacy, or practitioner that fails to comply with rules promulgated under this subsection, including by failure to generate a record that is required by the program.

SECTION 2251x. 961.385 (2) (fm) of the statutes is created to read:

961.385 (2) (fm) Permit the board to refer a pharmacist, pharmacy, or practitioner to the appropriate law enforcement agency for investigation and possible prosecution when the board has determined that a criminal violation may have occurred.

SECTION 2251xg. 961.443 (2) of the statutes, as affected by 2017 Wisconsin Act 33, sections 1em, 1gm and 1im, and 2017 Wisconsin Act (this act), is repealed and recreated to read:

961.443 (2) IMMUNITY FROM CRIMINAL PROSECUTION. An aider is immune from prosecution under s. 961.573 for the possession of drug paraphernalia, under s. 961.41 (3g) for the possession of a controlled substance or a controlled substance analog, and under s. 961.69 (2) for possession of a masking agent under the circumstances surrounding or leading to his or her commission of an act described in sub. (1).

SECTION 2251xm. 961.443 (2) (a) of the statutes, as affected by 2017 Wisconsin Act 33, section 1g, is amended to read:

961.443 (2) (a) No aider may have his or her parole, probation, or extended supervision revoked, and an aider is immune from prosecution under s. 946.49 for bail jumping, under s. 961.573 for the possession of drug paraphernalia, under s. 961.41 (3g) for the possession of a controlled substance or a controlled substance analog, and under s. 961.69 (2) for possession of a masking agent, under the circumstances surrounding or leading to his or her commission of an act described in sub. (1) that occurs on or after July 19, 2017, if the aider's attempt to obtain assistance occurs immediately after the aider believes the other person is suffering from the overdose or other adverse reaction.

SECTION 2251xp. 961.443 (2) (b) 1. of the statutes, as created by 2017 Wisconsin Act 33, is amended to read:

961.443 (2) (b) 1. No aided person may have his or her parole, probation, or extended supervision revoked under the circumstances surrounding or leading to an aider's commission of an act described in sub. (1) that occurs on or after July 19, 2017, if the aided person completes a treatment program as a condition of his or her parole, probation, or extended supervision or, if a treatment program is unavailable or would be prohibitive financially, agrees to be imprisoned in the county jail for not less than 15 days.

SECTION 2251xs. 961.443 (2) (b) 2. of the statutes, as created by 2017 Wisconsin Act 33, is amended to read:

961.443 (2) (b) 2. If an aided person is subject to prosecution under s. 946.49 for bail jumping, under s. 961.573 for the possession of drug paraphernalia, under s. 961.41 (3g) for the possession of a controlled substance or a controlled substance analog, or under s. 961.69 (2) for possession of a masking agent under the circumstances surrounding or leading to an aider's commission of an act described in sub. (1) that occurs on or after July 19, 2017, the district attorney shall offer the aided person a deferred prosecution agreement that includes the completion of a treatment program. This subdivision does not apply to an aided person who is on parole, probation, or extended supervision and fails to meet a condition under subd. 1.

SECTION 2251y. 967.06 (2) (b) of the statutes is Vetoed amended to read:

967.06 (2) (b) If the person indicating that he or she wants to be represented by a lawyer is detained under ch. 48, 51, 55, or 938, or 980, the person shall be referred for appointment of counsel as provided under s. 48.23 (4), 51.60, 55.105, or 938.23 (4), or 980.03 (2) (a), whichever is applicable.

SECTION 2252. 967.06 (3) of the statutes is amended to read:

967.06 (3) In any case in which the state public defender provides representation to an indigent person, the public defender may request that the applicable court reporter or clerk of circuit court prepare and transmit any transcript or court record. The request shall be complied with. The state public defender shall, from the appropriation under s. 20.550 (1) (f) (a), compensate the court reporter or clerk of circuit court for the cost of preparing, handling, duplicating, and mailing the documents.

SECTION 2253. 970.05 (2) (b) of the statutes is amended to read:

970.05 (2) (b) When a transcript is requested under sub. (1) by the state public defender or by a private attor-

ney appointed under s. 977.08, the state public defender shall pay the cost of the original from the appropriation under s. 20.550(1) (f) (a) and any additional copies shall be paid for at the statutory rate by the party requesting the copies.

SECTION 2254. 971.23 (10) of the statutes is amended to read:

971.23 (10) PAYMENT OF COPYING COSTS IN CASES INVOLVING INDIGENT DEFENDANTS. When the state public defender or a private attorney appointed under s. 977.08 requests copies, in any format, of any item that is discoverable under this section, the state public defender shall pay any fee charged for the copies from the appropriation account under s. 20.550 (1) (f) (a). If the person providing copies under this section charges the state public defender a fee for the copies, the fee may not exceed the applicable maximum fee for copies of discoverable materials that is established by rule under s. 977.02 (9).

SECTION 2255. 973.013 (3m) of the statutes is amended to read:

973.013 (3m) If a person who has not attained the age of 16 18 years is sentenced to the Wisconsin state prisons, the department shall place the person at a juvenile correctional facility or a secured residential care center for children and youth, unless the department determines that placement in an institution under s. 302.01 is appropriate based on the person's prior record of adjustment in a correctional setting, if any; the person's present and potential vocational and educational needs, interests and abilities; the adequacy and suitability of available facilities; the services and procedures available for treatment of the person within the various institutions; the protection of the public; and any other considerations promulgated by the department by rule. The department may not place any person under the age of 18 years in the correctional institution authorized in s. 301.16 (1n). This subsection does not preclude the department from designating an adult correctional institution, other than the correctional institution authorized in s. 301.16 (1n), as a reception center for the person and subsequently transferring the person to a juvenile correctional facility or a secured residential care center for children and youth. Section 302.11 and ch. 304 apply to all persons placed in a juvenile correctional facility or a secured residential care center for children and youth under this subsection.

SECTION 2255p. 973.046 (3) of the statutes is amended to read:

973.046 (3) All moneys collected from deoxyribonucleic acid analysis surcharges shall be deposited by the secretary of administration as specified in s. 20.455 (2) (Lm) (Lp) and utilized under s. 165.77.

Vetoed In Part

SECTION 2257e. 977.02 (2m) of the statutes is amended to read:

977.02 (2m) Promulgate rules regarding eligibility for legal services under this chapter, including legal services for persons who are entitled to be represented by

counsel without a determination of indigency, as provided in s. 48.23 (4), 51.60, 55.105, or 938.23 (4), or 1 In Part 980.03 (2) (a).

SECTION 2257f. 977.02 (3) (intro.) of the statutes is amended to read:

977.02 (3) (intro.) Promulgate rules regarding the determination of indigency of persons entitled to be represented by counsel, other than persons who are entitled to be represented by counsel under s. 48.23, 51.60, 55.105, or 938.23, or 980.03 (2) (a) including the time period in which the determination must be made and the criteria to be used to determine indigency and partial indigency. The rules shall specify that, in determining indigency, the representative of the state public defender shall do all of the following:

SECTION 2257g. 977.05 (4) (gm) of the statutes is amended to read:

977.05 (4) (gm) In accordance with the standards under pars. (h) and (i), accept referrals from judges and courts for the provision of legal services without a determination of indigency of persons who are entitled to be represented by counsel under s. 48.23, 51.60, 55.105, or 938.23, or 980.03 (2) (a), appoint counsel in accordance with contracts and policies of the board, and inform the referring judge or court of the name and address of the specific attorney who has been assigned to the case.

SECTION 2257h. 977.05 (4) (h) of the statutes is amended to read:

977.05 (4) (h) Accept requests for legal services from persons who are entitled to be represented by counsel under s. 48.23, 51.60, 55.105, or 938.23, or 980.03 (2) (a) and from indigent persons who are entitled to be represented by counsel under s. 967.06 or who are otherwise so entitled under the constitution or laws of the United States or this state and provide such persons with legal services when, in the discretion of the state public defender, such provision of legal services is appropriate.

SECTION 2257i. 977.05 (4) (i) 9. of the statutes is created to read:

977.05 (4) (i) 9. Cases involving persons who are subject to petitions under ch. 980.

SECTION 2257j. 977.06 (2) (a) of the statutes is amended to read:

977.06 (2) (a) A person seeking to have counsel assigned for him or her under s. 977.08, other than a person who is entitled to be represented by counsel under s. 48.23, 51.60, 55.105, or 938.23, or 980.03 (2) (a), shall sign a statement declaring that he or she has not disposed of any assets for the purpose of qualifying for that assignment of counsel. If the representative or authority making the indigency determination finds that any asset was disposed of for less than its fair market value for the purpose of obtaining that assignment of counsel, the asset shall be counted under rules promulgated under s. 977.02 (3) at its fair market value at the time it was disposed of, minus the amount of compensation received for the asset.

Vetoed In Part

SECTION 2257k. 977.06 (2) (am) of the statutes is amended to read:

977.06 (2) (am) A person seeking to have counsel assigned for him or her under s. 977.08, other than a person who is entitled to be represented by counsel under s. 48.23, 51.60, 55.105, or 938.23, or 980.03 (2) (a), shall sign a statement declaring that the information that he or she has given to determine eligibility for assignment of counsel he or she believes to be true and that he or she is informed that he or she is subject to the penalty under par.

SECTION 2257L. 977.07 (1) (a) of the statutes is amended to read:

977.07 (1) (a) Determination of indigency for persons entitled to counsel shall be made as soon as possible and shall be in accordance with the rules promulgated by the board under s. 977.02 (3) and the system established under s. 977.06. No determination of indigency is required for a person who is entitled to be represented by counsel under s. 48.23, 51.60, 55.105, or 938.23, or 980.03 (2) (a).

SECTION 2257m. 977.07 (1) (c) of the statutes is amended to read:

977.07 (1) (c) For all referrals made under ss. 809.107, 809.30, 974.06 (3) (b) and 974.07 (11), except a referral of a person who is entitled to be represented by counsel under s. 48.23, 51.60, 55.105, or 938.23, or 980.03 (2) (a), a representative of the state public defender shall determine indigency. For referrals made under ss. 809.107, 809.30 and 974.06 (3) (b), except a referral of a person who is entitled to be represented by counsel under s. 48.23, 51.60, 55.105, or 938.23, or 980.03 (2) (a), the representative of the state public defender may, unless a request for redetermination has been filed under s. 809.30 (2) (d) or the person's request for representation states that his or her financial circumstances have materially improved, rely upon a determination of indigency made for purposes of trial representation under this section.

SECTION 2257n. 977.075 (4) of the statutes is amended to read:

977.075 (4) The board shall establish by rule a fee schedule that sets the maximum amount that a parent subject to s. 48.275 (2) (b) or 938.275 (2) (b) shall pay as reimbursement for legal services and sets the maximum amount that a person subject to s. 51.605 or, 55.107, or 980.0305 shall pay as reimbursement for legal services. The maximum amounts under this subsection shall be based on the average cost, as determined by the board, for each applicable type of case.

SECTION 22570. 977.08 (1) of the statutes is amended

977.08 (1) If the representative or the authority for indigency determinations specified under s. 977.07 (1) refers a case to or within the office of the state public defender or if a case is referred under s. 48.23 (4), 51.60,

55.105, or 938.23 (4), or 980.03 (2) (a), the state public **Vetoed** defender shall assign counsel according to subs. (3) and (4). If a defendant makes a request for change of attorney assignment, the change of attorney must be approved by the circuit court.

SECTION 2257p. 977.08 (2) (intro.) of the statutes is amended to read:

977.08 (2) (intro.) All attorneys in a county shall be notified in writing by the state public defender that a set of lists is being prepared of attorneys willing to represent persons referred under s. 48.23 (4), 51.60, 55.105, or 938.23 (4), or 980.03 (2) (a) and indigent clients in the following:

SECTION 2258. 977.085 (1) (a) of the statutes is amended to read:

977.085 (1) (a) Private bar and staff case loads at the trial and appellate levels and expenditures of moneys under s. 20.550 (1) (b) to (d) (a) for the current fiscal year.

SECTION 2259. 977.085 (1) (b) of the statutes is amended to read:

977.085 (1) (b) Projections for the private bar and staff case loads at the trial and appellate levels and for expenditures of moneys under s. 20.550 (1) (b) to (d) (a) for the remainder of the current fiscal year and for the next fiscal year.

SECTION 2260. 977.085 (1m) of the statutes is amended to read:

977.085 (1m) The projections under sub. (1) (b) shall include the number of cases projected to be assigned to the private bar and the number of cases for which reimbursement will be made under s. 20.550 (1) (d) (a).

SECTION 2261. 977.085 (2) (intro.) of the statutes is amended to read:

977.085 (2) (intro.) If the projections under sub. (1) (b) indicate that moneys are being expended under s. 20.550 (1) (d) (a) at a rate which will deplete the appropriation prior to the end of the current fiscal year, the board shall include in the report a plan to address the problem. The plan shall include proposals for one or more of the following:

SECTION 2261d. 977.085 (3) of the statutes is amended to read:

Vetoed In Part

977.085 (3) The board shall provide quarterly reports to the joint committee on finance on the status of reimbursement for or recoupment of payments under ss. 48.275, 51.605, 55.107, 757.66, 938.275, 977.06, 977.075 and, 977.076, and 980.0305, including the amount of revenue generated by reimbursement and recoupment. The quarterly reports shall include any alternative means suggested by the board to improve reimbursement and recoupment procedures and to increase the amount of revenue generated. The department of justice, district attorneys, circuit courts and applicable county agencies shall cooperate by providing any necessary information to the state public defender.

Vetoed

In Part

Vetoed In Part **SECTION 2261g.** 978.001 (1b), (1d) and (1n) of the statutes are created to read:

978.001 (1b) "Board" means the prosecutor board.

- (1d) "Executive director" means the executive director appointed under s. 978.003 (3).
 - (1n) "Office" means the state prosecutors office.

SECTION 2261h. 978.001 (1p) of the statutes is repealed.

SECTION 2261j. 978.003 of the statutes is created to read:

978.003 Board; duties. The board shall do all of the following:

- (1) Submit the budget in accordance with s. 16.42 after the executive director submits the budget to the board and the board approves it.
- (2) At least annually submit to the joint committee on finance recommendations on the allocation of prosecutor resources.
- (3) Appoint an attorney with experience in criminal prosecution as the executive director of the office.
- (4) Oversee, and set policy initiatives for, the executive director.
- (5) Review existing law or proposed legislation and make recommendations to the legislature.

SECTION 2261L. 978.004 of the statutes is created to read:

978.004 State prosecutors office executive director. (1) The executive director shall do all of the following:

- (a) Manage and direct the office subject to the policy initiatives set under s. 978.003 (4).
- (b) Prepare and submit to the board for its approval a budget and any personnel and employment policies that the board requires.
- (c) Prepare and submit to the board and other appropriate persons an annual report of the activities of the office in the form that the board directs.
- (d) Represent the board before the governor, the legislature, bar associations, courts, and other appropriate entities.
- (e) Appoint in the classified service an executive assistant and all other employees of the office. Before making an appointment under this paragraph, the executive director shall notify the board of any prospective appointment. If the board does not object to the prospective appointment within 7 working days after notification, the executive director may make the appointment. If the board objects to a prospective appointment, the executive director may not make the appointment until the board approves it.
- (f) Prepare fiscal estimates on bills affecting prosecutors or the office, including bills modifying or creating crimes or sentencing practices. To prepare a fiscal estimate, the executive director shall consult with and obtain data from district attorneys. The executive director shall transmit a draft fiscal estimate to the board. If the board

does not object to the draft fiscal estimate within 7 working days after receiving it, the executive director may submit the fiscal estimate. If the board objects to a draft fiscal estimate, the executive director may not submit the fiscal estimate until the board approves it.

(2) The executive director may identify methods and practices for district attorneys that promote professional competence, ethical practices, and evidence—based practices

SECTION 2261m. 978.005 of the statutes is created to read:

978.005 Limits on board and executive director. Neither the board nor the executive director may make any decision regarding the handling of any case nor interfere with any district attorney in carrying out professional duties. Neither the board nor the office may interfere

with or infringe upon the autonomy of a district attorney or upon the authority of a district attorney to manage his or her own prosecutorial unit.

SECTION 22610. 978.03 of the statutes is amended to read:

978.03 Deputies and assistants in certain prosecutorial units. (1) The district attorney of any prosecutorial unit having a population of 500,000 or more may appoint 7 deputy district attorneys and such assistant district attorneys as may be requested by the department of administration, or by the board, and authorized in accordance with s. 16.505. The district attorney shall rank the deputy district attorneys for purposes of carrying out duties under this section. The deputies, according to rank, may perform any duty of the district attorney, under the district attorney's direction. In the absence or disability of the district attorney, the deputies, according to rank, may perform any act required by law to be performed by the district attorney. Any such deputy must have practiced law in this state for at least 2 years prior to appointment under this section.

- (1m) The district attorney of any prosecutorial unit having a population of 200,000 or more but not more than 499,999 may appoint 3 deputy district attorneys and such assistant district attorneys as may be requested by the department of administration, or by the board, and authorized in accordance with s. 16.505. The district attorney shall rank the deputy district attorneys for purposes of carrying out duties under this section. The deputies, according to rank, may perform any duty of the district attorney, under the district attorney's direction. In the absence or disability of the district attorney, the deputies, according to rank, may perform any act required by law to be performed by the district attorney. Any such deputy must have practiced law in this state for at least 2 years prior to appointment under this section.
- (2) The district attorney of any prosecutorial unit having a population of 100,000 or more but not more than 199,999 may appoint one deputy district attorney and such assistant district attorneys as may be requested by

Vetoed In Part the department of administration, or by the board, and authorized in accordance with s. 16.505. The deputy may perform any duty of the district attorney, under the district attorney's direction. In the absence or disability of the district attorney, the deputy may perform any act required by law to be performed by the district attorney. The deputy must have practiced law in this state for at least 2 years prior to appointment under this section.

(3) Any assistant district attorney under sub. (1), (1m), or (2) must be an attorney admitted to practice law in this state and, except as provided in s. 978.043 (1), may perform any duty required by law to be performed by the district attorney. The district attorney of the prosecutorial unit under sub. (1), (1m), or (2) may appoint such temporary counsel as may be authorized by the department of administration board.

SECTION 2261q. 978.045 (1g) of the statutes is amended to read:

978.045 (1g) A court on its own motion may appoint a special prosecutor under sub. (1r) or a district attorney may request a court to appoint a special prosecutor under that subsection. Before a court appoints a special prosecutor on its own motion or at the request of a district attorney for an appointment that exceeds 6 hours per case, the court or district attorney shall request assistance from a district attorney, deputy district attorney or assistant district attorney from other prosecutorial units or an assistant attorney general. A district attorney requesting the appointment of a special prosecutor, or a court if the court is appointing a special prosecutor on its own motion, shall notify the department of administration, on a form provided by that department, of office that the district attorney's attorney or the court's inability court, whichever is appropriate, is unable to obtain assistance from another prosecutorial unit or from an assistant attorney general.

SECTION 2261qm. 978.045 (1r) (bm) (intro.) of the statutes is amended to read:

978.045 (1r) (bm) (intro.) The judge may appoint an attorney as a special prosecutor at the request of a district attorney to assist the district attorney in the prosecution of persons charged with a crime, in grand jury proceedings, in proceedings under ch. 980, or in investigations. The Except as provided under par. (bp), the judge may appoint an attorney as a special prosecutor only if the judge or the requesting district attorney submits an affidavit to the department of administration attesting that any of the following conditions exists:

Vetoed In Part

SECTION 2261r. 978.045 (1r) (bm) (intro.) of the statutes, as affected by 2017 Wisconsin Act (this act), is amended to read:

978.045 (1r) (bm) (intro.) The judge may appoint an attorney as a special prosecutor at the request of a district attorney to assist the district attorney in the prosecution of persons charged with a crime, in grand jury proceedings, in proceedings under ch. 980, or in investigations.

Except as provided under par. (bp), the judge may appoint an attorney as a special prosecutor only if the judge or the requesting district attorney submits an affidavit to the department of administration office attesting that any of the following conditions exists:

SECTION 2261rm. 978.045 (1r) (bp) of the statutes is created to read:

978.045 (1r) (bp) The judge may appoint an attorney as a special prosecutor to assist the district attorney in counties with a population of less than 45,000 if the department of administration certifies that the county has a significant case backlog and if a petition for such an appointment is approved by the affected county board. This paragraph does not apply after December 31, 2019.

SECTION 2261rt. 978.045 (1r) (cm) of the statutes is amended to read:

978.045 (1r) (cm) The judge may not appoint an attorney as a special prosecutor to assist the district attorney in John Doe proceedings under s. 968.26 unless a condition under par. (bm) 1. to 8. exists, par. (bp) applies, or unless the judge determines that a complaint received under s. 968.26 (2) (am) relates to the conduct of the district attorney to whom the judge otherwise would refer the complaint. This paragraph does not prohibit assistance authorized by s. 978.05 (8).

SECTION 2261s. 978.045 (2) of the statutes is Vetoed amended to read:

In Part

In Part

978.045 (2) If the department of administration office approves the appointment of a special prosecutor under sub. (1r), the court shall fix the amount of compensation for the attorney appointed according to the rates specified in s. 977.08 (4m) (b). The department of administration shall pay the compensation ordered by the court from the appropriation under s. 20.475 (1) (d). The court, district attorney, and the special prosecutor shall provide any information regarding a payment of compensation that the department requests. Any payment under this subsection earns interest on the balance due from the 121st day after receipt of a properly completed invoice or receipt and acceptance of the property or service under the order or contract, whichever is later, at the rate specified in s. 71.82 (1) (a) compounded monthly.

SECTION 2262. 978.05 (6) (a) of the statutes is amended to read:

978.05 (6) (a) Institute, commence or appear in all civil actions or special proceedings under and perform the duties set forth for the district attorney under ch. 980 and ss. 17.14, 30.03 (2), 48.09 (5), 59.55 (1), 59.64 (1), 70.36, 84.062 (8), 89.08, 103.92 (4), 109.09, 343.305 (9) (a), 806.05, 938.09, 938.18, 938.355 (6) (b) and (6g) (a), 946.86, 946.87, 961.55 (5), 971.14 and 973.075 to 973.077, perform any duties in connection with court proceedings in a court assigned to exercise jurisdiction under chs. 48 and 938 as the judge may request and perform all appropriate duties and appear if the district attorney is designated in specific statutes, including matters

within chs. 782, 976 and 979 and ss. 51.81 to 51.85. Nothing in this paragraph limits the authority of the county board to designate, under s. 48.09 (5), that the corporation counsel provide representation as specified in s. 48.09 (5) or to designate, under s. 48.09 (6) or 938.09 (6), the district attorney as an appropriate person to represent the interests of the public under s. 48.14 or 938.14.

Vetoed In Part

SECTION 2262c. 978.05 (9) of the statutes is amended to read:

978.05 (9) BUDGET. Prepare a biennial budget request for submission to the department executive director under s. 978.11 978.004 (1) (b) by September 1 of each even-numbered year.

SECTION 2262e. 978.11 of the statutes is repealed. **SECTION 2262g.** 978.12 (1) (c) of the statutes is amended to read:

978.12 (1) (c) Assistant district attorneys. Assistant district attorneys shall be employed outside the classified service. For purposes of salary administration, the administrator of the division of personnel management in the department of administration, in consultation with the office, shall establish one or more classifications for assistant district attorneys in accordance with the classification or classifications allocated to assistant attorneys general. Except as provided in ss. 111.93 (3) (b) and 230.12 (10), the salaries of assistant district attorneys shall be established and adjusted in accordance with the state compensation plan for assistant attorneys general whose positions are allocated to the classification or classifications established by the administrator of the division of personnel management in the department of administration under this paragraph.

Vetoed In Part

SECTION 2262L. 980.03 (2) (a) of the statutes is amended to read:

980.03 (2) (a) Counsel. If In any situation under this chapter in which the person claims or appears to be indigent has a right to be represented by counsel, the court shall refer the person to the authority for indigency determinations under s. 977.07 (1) and, if applicable, the appointment of as soon as practicable to the state public defender, who shall appoint counsel for the person under s. 977.08 without a determination of indigency.

SECTION 2262m. 980.0305 of the statutes is created to read:

980.0305 Reimbursement for counsel provided by the state. (1) INQUIRY. At or after the conclusion of a proceeding under this chapter in which the state public defender has provided counsel for a person, the court may inquire as to the person's ability to reimburse the state for the costs of representation. If the court determines that the person is able to make reimbursement for all or part of the costs of representation, the court may order the person to reimburse the state an amount not to exceed the maximum amount established by the public defender board under s. 977.075 (4). Upon the court's request, the state public defender shall conduct a determination of

indigency under s. 977.07 and report the results of the Vetoed determination to the court.

- (2) PAYMENT. Reimbursement ordered under this section shall be made to the clerk of courts of the county where the proceedings took place. The clerk of courts shall transmit payments under this section to the county treasurer, who shall deposit 25 percent of the payment amount in the county treasury and transmit the remainder to the secretary of administration. Payments transmitted to the secretary of administration shall be deposited in the general fund and credited to the appropriation account under s. 20.550 (1) (L).
- (3) REPORT. By January 31st of each year, the clerk of courts for each county shall report to the state public defender the total amount of reimbursements ordered under sub. (1) in the previous calendar year and the total amount of reimbursements paid to the clerk under sub. (2) in the previous year.

SECTION 2262n. 980.08 (4) (cm) and (e) of the statutes are consolidated, renumbered 980.08 (4) (dm) 1. (intro.) and amended to read:

980.08 (4) (dm) 1. (intro.) If the court finds that all of the criteria in par. (cg) are met, the court shall select a county to prepare a report under par. (e). Unless the court has good cause to select another county, the court shall select order the county of the person's county of residence, as determined by the department of health services under s. 980.105. An actual or alleged lack of available housing for the person within a county because of an ordinance or resolution in effect or proposed by the county or by a city, town, or village within the county may not constitute good cause to select another county under this paragraph. The court may not select a county where there is a facility in which persons committed to institutional care under this chapter are placed unless that county is also that person's county of residence. (e) The court shall order the county department under s. 51.42 in the county of intended placement to prepare a report, either independently or with the department of health services, identifying prospective residential options for community placement. In identifying prospective residential options, the county department shall consider the proximity of any potential placement to the residence of other persons on supervised release and to the residence of persons who are in the custody of the department of corrections and regarding whom a sex offender notification bulletin has been issued to law enforcement agencies under s. 301.46 (2m) (a) or (am). The, to prepare a report. The county shall create a temporary committee to prepare the report for the county. The committee shall consist of the county department under s. 51.42, a representative of the department of health services, a local probation or parole officer, the county corporation counsel or his or her designee. and a representative of the department of the county that is responsible for land use and planning or the department of the county that is responsible for land information. In

Vetoed In Part the report, the county shall identify an appropriate residential option in that county while the person is on supervised release and shall demonstrate that the county has contacted the landlord for that residential option and that the landlord has committed to enter into a lease. The county shall consider the following factors when identifying an appropriate residential option:

Vetoed In Part

- 2. When preparing the report, the county department shall consult with a local law enforcement agency having jurisdiction over the residential option. The law enforcement agency may submit a written report that provides information relating to the residential option, and, if the law enforcement agency submits a report, the county department shall include the agency's report when the county department submits its report to the department of health services.
- 4. The county shall submit its report to the department of health services within 60 120 days following the court order. A county that does not submit its report within 120 days violates the person's rights under s. 51.61, and each day that the county does not submit the report after the 120 days have expired constitutes a separate violation under s. 51.61. Notwithstanding s. 51.61 (7), any damages beyond costs and reasonable actual attorney fees recovered by the person for a violation shall be deposited into the appropriation account under s. 20.435 (2) (gz).

SECTION 22620. 980.08 (4) (d) of the statutes is repealed.

SECTION 2262p. 980.08 (4) (dm) 3. of the statutes is created to read:

980.08 (4) (dm) 3. To assist the county in identifying appropriate residential options for the report, within 30 days after the court orders the county to prepare the report, the department of health services shall determine the identity and location of known and registered victims of the person's acts by searching its victim database and consulting with the office of victim services in the department of corrections, the department of justice, and the county coordinator of victims and witnesses services in the county of intended placement, the county where the person was convicted, and the county of commitment. The county may consult with the department of health services on other matters while preparing the report and the department of health services shall respond as soon as practically possible.

SECTION 2262q. 980.08 (4) (em) of the statutes is repealed.

SECTION 2262r. 980.08 (4) (f) (intro.) of the statutes is renumbered 980.08 (4) (f) and amended to read:

980.08 (4) (f) The court shall direct the department to use any submissions under par. (d), the report submitted under par. (e), any report submitted under par. (em), and other residential options identified by the department (dm) to prepare a supervised release plan for the person-The department shall search its victim database, and con-

sult with the office of victim services in the department Vetoed of corrections, the department of justice, and the county coordinator of victims and witnesses services in the county of intended placement, the county where the person was convicted, and the county of commitment to determine the identity and location of known and registered victims of the person's acts. The department shall prepare a supervised release plan that identifies the proposed residence residential option the county identified in its report. The plan shall also address the person's need, if any, for supervision, counseling, medication, community support services, residential services, vocational services, and alcohol or other drug abuse treatment. The supervised release plan shall be submitted to the court within 90 30 days of the finding under par. (cg) after the county submitted its report under par. (dm). The court may grant-extensions one extension of up to 30 days of this time period for good cause. The plan shall do all of the following:

SECTION 2262s. 980.08 (4) (f) 1. of the statutes is repealed.

SECTION 2262t. 980.08 (4) (f) 2., 3. and 4. of the statutes are renumbered 980.08 (4) (dm) 1. a., b. and c. and amended to read:

980.08 (4) (dm) 1. a. Ensure that The distance between the person's placement is into a residence that is not less than 1,500 feet from and any school premises, child care facility, public park, place of worship, or youth center. A person is not in violation of a condition or rule of supervised release under sub. (7) (a) if any school premises, child care facility, public park, place of worship, or youth center is established within 1,500 feet from near the person's residence after he or she is placed in the residence under this section.

- b. If the person committed a sexually violent offense against an adult at risk, as defined in s. 55.01 (1e), or an elder adult at risk, as defined in s. 46.90 (1) (br), ensure that the distance between the person's placement is into a residence that is not less than 1,500 feet from and a nursing home or an assisted living facility. A person is not in violation of a condition or rule of supervised release under sub. (7) (a) if a nursing home or an assisted living facility is established within 1,500 feet from near the person's residence after he or she is placed in the residence under this section.
- c. If the person is a serious child sex offender, ensure that the distance between the person's placement is into a residence that is not on a property adjacent to and a property where a child's primary residence exists. For the purpose of this subdivision, adjacent properties are properties that share a property line without regard to a public or private road if the living quarters on each property are not more than 1,500 feet apart. A person is not in violation of a condition or rule of supervised release under sub. (7) (a) if a child establishes primary residence

in a property adjacent to near the person's residence after the person is placed in the residence under this section.

SECTION 2262u. 980.08 (4) (g) of the statutes is amended to read:

980.08 (4) (g) The court shall review the plan submitted by the department under par. (cm) (f). If the details of the plan adequately meet the treatment needs of the individual and the safety needs of the community, then the court shall approve the plan and determine that supervised release is appropriate. If the details of the plan do not adequately meet the treatment needs of the individual or the safety needs of the community, then the court shall determine that supervised release is not appropriate or direct the preparation of another supervised release plan to be considered by the court under this paragraph. If the plan is inadequate under this paragraph due to the residential option, the court shall order the county to identify and arrange to lease another residential option and to prepare a new report under par. (dm). If the plan is inadequate under this paragraph due to the treatment options, the court shall order the department to prepare another plan under par. (f).

SECTION 2262v. 980.08 (5m) of the statutes is repealed.

SECTION 2262w. 980.105 (2) of the statutes is created to read:

980.105 (2) If sub. (1m) is insufficient to determine the county of residence, the department shall find that the county of residence is the county in which, on the date that the person committed the sexually violent offense that resulted in the sentence, placement, or commitment that was in effect when the petition was filed under s. 980.02, the person would have been a resident for the purposes of social security disability insurance eligibility.

SECTION 2262x. 980.105 (2m) of the statutes is repealed.

SECTION 2262y. 990.01 (2) of the statutes is amended to read:

990.01 (2) ACQUIRE. "Acquire," when used in connection with a grant of power to any person, includes the acquisition by purchase, grant, gift or bequest. It includes the power to condemn only in the cases specified in s. 32.02 and subject to the limitations under s. 32.015.

SECTION 2264. 995.55 (1) (b) of the statutes is amended to read:

995.55 (1) (b) "Educational institution" means an institution of higher education, as defined in s. 108.02 (18); a technical college established under s. 38.02; a school, as defined in s. 38.50 440.52 (11) (a) 2.; a public school, as described in s. 115.01 (1); a charter school, as defined in s. 115.001 (1); a private school, as defined in s. 115.001 (3r); or a private educational testing service or administrator.

SECTION 2265. 2013 Wisconsin Act 229, section 6 (1), as last affected by 2015 Wisconsin Act 55, is amended to read:

[2013 Wisconsin Act 229] Section 6 (1) This act takes effect on July 1, 2017 2018, and first applies to bad Vetoed debts resulting from sales completed beginning on July 1, 2017 2018.

In Part Vetoed

SECTION 2265g. 2015 Wisconsin Act 55, section In Part 768kb is repealed.

SECTION 2265h. 2015 Wisconsin Act 55, section 768pb is repealed.

Section 2265m. 2015 Wisconsin Act 55, section 1458rb is repealed.

SECTION 2265p. 2015 Wisconsin Act 55, section 9426 (1q) is amended to read:

[2015 Wisconsin Act 55] Section 9426 (1q) WISCON-SIN COURT APPOINTED SPECIAL ADVOCATE ASSOCIATION GRANTS. The repeal of sections 20.455 (5) (es) and 165.967 of the statutes takes effect on July 1, 2017 2019.

SECTION 2265q. 2015 Wisconsin Act 55, section 9449 (1q) is repealed.

SECTION 2265r. 2017 Wisconsin Act 33, section 1y is amended to read:

[2017 Wisconsin Act 33] Section 1y This act takes effect on the day after publication July 19, 2017, except

(1) SUNSET FOR EVALUATION OF LEGISLATION. The treatment of section 961.443 (2) (title) (by Section 1em), (a) (by Section 1gm), and (b) (by Section 1im) of the statutes takes effect on the first day of the 37th month beginning after publication August 1, 2020.

SECTION 2265t. 2017 Wisconsin Act 26, section 1 (1) is amended to read:

[2017 Wisconsin Act 26] Section 1 (1) EXPANDING GRADUATE MEDICAL TRAINING IN AN ADDICTION SPECIALTY. From the appropriation under section 20.435 (4) (b) (bf) of the statutes and notwithstanding the funding limitations in section 146.64 (2) (c) 1. of the statutes, the department of health services may award grants to hospitals under section 146.64 of the statutes to increase the number of physicians trained in an addiction specialty. To receive a grant under this subsection, the hospital shall expand fellowship positions in addiction medicine or addiction psychiatry for physicians practicing family medicine, general internal medicine, general surgery, pediatrics, or psychiatry.

SECTION 2266r. DHS 116.04 (2) (d) of the administrative code is amended to read:

DHS 116.04 (2) (d) The department may not require a reporter under par. (a), (b) or (c) to provide the name of a child to the department if the child's parent or guardian does not consent states in writing that he or she refuses to the release of the name or address of the child to the department.

SECTION 2266s. DHS 116.05 (2) (a) (intro.) of the administrative code is amended to read:

DHS 116.05 (2) (a) (intro.) The department may release child-identifiable data only to persons specified in s. 253.12, Stats., and to the following persons:

SECTION 2266t. DHS 116.05 (4) of the administrative code is repealed.

SECTION 2266u. DOC 309.10 (1) of the administrative code is amended to read:

DOC 309.10 (1) Public officials, elected tribal officials, tribal judges, and members of private and public organizations who provide services to inmates may visit institutions with the approval of the warden. These visitors shall make arrangements for all such visits in advance with the warden to minimize interference with normal operations and activities. The warden may limit the duration of such visits for security reasons. A person who has not attained the age of 18 may not participate in any group visit except with the approval of the warden, unless the person is a family member on the inmate's approved visitor list.

SECTION 2266v. Trans 142.04 (7) of the administrative code is repealed.

SECTION 2266w. Trans 269.05 (2) of the administrative code is amended to read:

Trans 269.05 (2) An issuing authority may issue a permit only for the transportation of garbage or refuse, in a self–compactor equipped vehicle or a roll–off equipped truck or roll–off trailer that uses all axles while transporting garbage or refuse, or for the transportation of recyclable scrap. This includes the transportation of tools and equipment necessary for the safe and efficient pick–up and discharge of the garbage or refuse, or recyclable scrap, and for the return of the vehicle when empty.

SECTION 9101. Nonstatutory provisions; Administration.

- (1) ELIMINATION OF DEPOSITORY SELECTION BOARD.
- (a) Assets and liabilities. On the effective date of this paragraph, the assets and liabilities of the depository selection board become the assets and liabilities of the department of administration.
- (b) *Tangible personal property.* On the effective date of this paragraph, all tangible personal property, including records, of the depository selection board is transferred to the department of administration.
- (c) Contracts. All contracts entered into by the depository selection board in effect on the effective date of this paragraph remain in effect and are transferred to the department of administration. The department of administration shall carry out all obligations under such a contract unless modified or rescinded by the department of administration to the extent allowed under the contract.
- (d) *Pending matters*. Each matter pending with the depository selection board on the effective date of this paragraph is transferred to the department of administration, and all materials submitted to or actions taken by the depository selection board with respect to the pending matter are considered as having been submitted to or taken by the department of administration.

- (e) Rules and orders. All rules promulgated by the depository selection board that are in effect on the effective date of this paragraph remain in effect until their specified expiration dates or until amended or repealed by the department of administration. All orders issued by the depository selection board that are in effect on the effective date of this paragraph remain in effect until their specified expiration dates or until modified or rescinded by the department of administration.
- (2) TRANSFER OF COLLEGE SAVINGS PROGRAMS DUTIES TO THE DEPARTMENT OF FINANCIAL INSTITUTIONS.
- (a) Assets and liabilities. On the effective date of this paragraph, the assets and liabilities of the department of administration that are primarily related to the department's performance of duties under sections 16.64 and 16.641, 2015 stats., as determined by the secretary of administration, become the assets and liabilities of the department of financial institutions.
- (b) *Positions and employees*. On the effective date of this paragraph, 2.0 FTE SEG positions, and the incumbent employees holding those positions, in the department of administration responsible for the performance of duties under sections 16.64 and 16.641, 2015 stats., as determined by the secretary of administration, are transferred to the department of financial institutions.
- (c) Employee status. Employees transferred under paragraph (b) have all the rights and the same status under chapter 230 of the statutes in the department of financial institutions that they enjoyed in the department of administration immediately before the transfer. Notwithstanding section 230.28 (4) of the statutes, no employee transferred under paragraph (b) who has attained permanent status in class is required to serve a probationary period.
- (d) Tangible personal property. On the effective date of this paragraph, all tangible personal property, including records, of the department of administration that is primarily related to the department's performance of duties under sections 16.64, 16.641, and 16.642 (2), 2015 stats., as determined by the secretary of administration, is transferred to the department of financial institutions.
- (e) Contracts. All contracts entered into by the department of administration in effect on the effective date of this paragraph that are primarily related to the department's performance of duties under sections 16.64 and 16.641, 2015 stats., as determined by the secretary of administration, remain in effect and are transferred to the department of financial institutions. The department of financial institutions shall carry out any obligations under those contracts unless modified or rescinded by the department of financial institutions to the extent allowed under the contract.
- (f) Rules and orders. All rules promulgated by the department of administration in effect on the effective date of this paragraph that are primarily related to the department's performance of duties under sections 16.64

- and 16.641, 2015 stats., as determined by the secretary of administration, remain in effect until their specified expiration dates or until amended or repealed by the department of financial institutions. All orders issued by the department of administration in effect on the effective date of this paragraph that are primarily related to the department's performance of duties under sections 16.64 and 16.641, 2015 stats., as determined by the secretary of administration, remain in effect until their specified expiration dates or until modified or rescinded by the department of financial institutions.
- (g) *Pending matters*. Any matter pending with the department of administration that is primarily related to the department's performance of duties under sections 16.64 and 16.641, 2015 stats., as determined by the secretary of administration, is transferred to the department of financial institutions. All materials submitted to or actions taken by the department of administration with respect to the pending matter are considered as having been submitted to or taken by the department of financial institutions.
 - (3) Transfer of mental health services.
- (a) Assets and liabilities. On the effective date of this paragraph, the assets and liabilities of the department of administration that are primarily related to mental health services, as determined by the secretary of administration, become the assets and liabilities of the department of health services.
- (b) Tangible personal property. On the effective date of this paragraph, all tangible personal property, including records, of the department of administration that is primarily related to mental health services, as determined by the secretary of administration, is transferred to the department of health services.
- (c) Contracts. All contracts entered into by the department of administration in effect of the effective date of this paragraph that are primarily related to mental health services, as determined by the secretary of administration, remain in effect and are transferred to the department of health services. The department of health services shall carry out any obligations under those contracts unless modified or rescinded by the department of health services to the extent allowed under the contract.
- (d) Rules and orders. All rules promulgated by the department of administration in effect on the effective date of this paragraph that are primarily related to mental health services, as determined by the secretary of administration, remain in effect until their specified expiration dates or until amended or repealed by the department of health services. All orders issued by the department of administration in effect on the effective date of this paragraph that are primarily related to mental health services, as determined by the secretary of administration, remain in effect until their specified expiration dates or until modified or rescinded by the department of health services.

- (e) *Pending matters*. Any matter pending with the department of administration on the effective date of this paragraph that is primarily related to mental health services, as determined by the secretary of administration, is transferred to the department of health services. All materials submitted to or actions taken by the department of administration with respect to the pending matter are considered as having been submitted to or taken by the department of health services.
 - (4) TELECOMMUNICATIONS RELAY SERVICE.
- (a) *Position transfer.* On the effective date of this paragraph, 1.0 FTE PR position, and the incumbent employee holding that position, in the department of administration responsible for administering telecommunications relay service, as determined by the secretary of administration, is transferred to the public service commission.
- (b) Employee status. The employee transferred under paragraph (a) has all the rights and the same status under chapter 230 of the statutes in the public service commission that the employee enjoyed in the department of administration immediately before the transfer. Notwithstanding section 230.28 (4) of the statutes, if the employee transferred under paragraph (a) attained permanent status in class before the transfer, the employee is not required to serve a probationary period.
- (c) Contracts. All contracts entered into by the department of administration in effect on the effective date of this paragraph that are primarily related to telecommunications relay service, as determined by the secretary of administration, remain in effect and are transferred to the public service commission. The public service commission shall carry out any obligations under those contracts unless modified or rescinded by the commission to the extent allowed under the contracts.
- (d) Assets and liabilities. On the effective date of this paragraph, the assets and liabilities of the department of administration primarily related to telecommunications relay service, as determined by the secretary of administration, become the assets and liabilities of the public service commission.
- (e) Tangible personal property. On the effective date of this paragraph, all tangible personal property, including records, of the department of administration that is primarily related to telecommunications relay service, as determined by the secretary of administration, is transferred to the public service commission.
 - (7p) Prosecutor board.

(a) *Initial terms for prosecutor board members*. Notwithstanding section 15.77 of the statutes, of the members of the prosecutor board who are appointed as initial members, one member representing each district under section 752.11 (1) (b) and (d) of the statutes and one member under section 15.77 (3) of the statutes shall serve for a one–year term and one member representing the district under section 752.11 (1) (c) of the statutes, one mem-

Vetoed In Part

ber under section 15.77 (2) of the statutes, and one member under section 15.77 (3) of the statutes shall serve for a 2-year term.

- (b) Transfer of state prosecutors office.
- 1. 'Assets and liabilities.' On the effective date of this subdivision, the assets and liabilities of the department of administration that are primarily related to the state prosecutors office, as determined by the secretary of administration, become the assets and liabilities of the prosecutor board.
- 2. 'Tangible personal property.' On the effective date of this subdivision, all tangible personal property, including records, of the department of administration that is primarily related to the state prosecutors office, as determined by the secretary of administration, is transferred to the prosecutor board.
- 3. 'Contracts.' All contracts entered into by the department of administration that are primarily related to the state prosecutors office, as determined by the secretary of administration, in effect on the effective date of this subdivision, remain in effect and are transferred to the prosecutor board. The prosecutor board shall carry out any such contractual obligations unless modified or rescinded by the prosecutor board to the extent allowed under the contract.
- 4. 'Pending matters.' Any matter pending with the department of administration that is primarily related to the state prosecutors office, as determined by the secretary of administration, on the effective date of this subdivision, is transferred to the prosecutor board, and all materials submitted to or actions taken by the department of administration, with respect to the pending matter are considered as having been submitted to or taken by the prosecutor board.
- 5. 'Rules and orders.' All rules promulgated for the department of administration that are primarily related to the state prosecutors office, as determined by the secretary of administration, that are in effect on the effective date of this subdivision remain in effect until their specified expiration dates or until amended or repealed by the prosecutor board.
 - (c) Plan for office space for prosecutors office.
- 1. The prosecutor board, in consultation with the department of administration, shall, no later than March 1, 2018, submit to the joint committee on finance a plan to house the prosecutors office in the space that, on the effective date of this subdivision, is occupied by the director of the state prosecutors office.
- 2. The plan submitted under subdivision 1. shall include provisions for the acquisition or release, as appropriate, of space; the relocation, if necessary, of staff and tangible personal property; and any other provisions necessary for the transition. The plan shall provide office space for a legislative liaison and a space to accommodate meetings of the prosecutor board.

- 3. If the cochairpersons of the joint committee on Vetoed finance do not notify the prosecutor board within 14 working days after the date the plan is submitted under subdivision 1. that the committee has scheduled a meeting to take place for the purpose of reviewing the plan, the prosecutor board shall implement the plan. If, within 14 working days after the date the plan is submitted under subdivision 1., the cochairpersons of the joint committee on finance notify the prosecutor board that the committee has scheduled a meeting for the purpose of reviewing the plan, the prosecutor board shall incorporate into the plan all changes made by the committee and implement the plan.
 - (8f) JUDICIAL COMPENSATION; COMPENSATION PLAN.
- (a) The administrator of the division of personnel management in the department of administration shall, in establishing the state employee compensation plan under section 230.12 of the statutes for the 2017–19 biennium. consult with the chief justice of the Wisconsin Supreme Court regarding establishing salary adjustments for judges and justices for the 2017-19 biennium that exceed the equivalent of 2, 2 percent general wage adjustments.
- (b) If, on the effective date of this paragraph, the compensation plan under section 230.12 of the statutes has been adopted for the 2017-19 biennium, by no later than 30 days after the effective date of this paragraph, the administrator of the division of personnel management in the department of administration shall consult with the chief justice of the Wisconsin Supreme Court to discuss the possibility of proposing an amendment under section 230.12 (3) (c) of the statutes to include the judicial salary adjustments that exceed the equivalent of 2, 2 percent general wage adjustments for judges and justices during the 2017-19 biennium.
 - (9) Human resources services.
- (a) Definition. In this subsection, "shared services agency" has the meaning given in section 16.004 (20) (a) of the statutes.
- (b) Assets and liabilities. On the effective date of this paragraph, the assets and liabilities of a shared services agency that relate to human resources services and payroll and benefit services, as determined by the secretary of administration, become the assets and liabilities of the department of administration.
 - (c) Positions and employees.
- 1. On the effective date of this subdivision, all FTE positions in a shared services agency relating to human resources services and payroll and benefit services, as determined by the secretary of administration, and the incumbent employees holding those positions, are transferred to the department of administration.
- 2. Employees transferred under subdivision 1. have all the rights and the same status under chapter 230 of the statutes in the department of administration that they enjoyed in the shared services agency immediately

before the transfer. Notwithstanding section 230.28 (4) of the statutes, no employee transferred under subdivision 1. who has attained permanent status in class is required to serve a probationary period.

- (d) Personal property. On the effective date of this paragraph, all tangible personal property, including records, of a shared services agency that relate to human resources services and payroll and benefit services, as determined by the secretary of administration, are transferred to the department of administration.
- (e) Contracts. All contracts entered into by a shared services agency in effect on the effective date of this paragraph that are primarily related to human resources services and payroll and benefit services, as determined by the secretary of administration, remain in effect and are transferred to the department of administration.
- (10) YOUTH WELLNESS CENTER; TRIBAL PAYMENT. From the appropriation account under section 20.505 (8) (hm) of the statutes, the department of administration shall pay \$200,000 in fiscal year 2017–18 to American Indian tribes for performing a feasibility study for the creation of a youth wellness center and for developing a business plan for the creation of the youth wellness cen-
- (10t) COMMUNITY BLOCK GRANT PRIORITY. In the 2017–19 fiscal biennium, the department of administration shall, for purposes of awarding federal community development block grant funding, give priority to the extent allowed under federal law to a project that satisfies all of the following:
- (a) The project plans for or establishes public or private facilities for the provision of new water and sewer services primarily to residential users.
- (b) The new water service replaces service provided on the effective date of this paragraph by an entity other than a community water system, as defined in section 281.62 (1) (a) of the statutes, a cooperative association organized under chapter 185 of the statutes, or a private groundwater well.
- (c) The new sewer service replaces service provided on the effective date of this paragraph by an entity other than a public utility, as defined in section 196.01 (5) of the statutes, a private on-site wastewater treatment system, as defined in section 145.01 (12) of the statutes, or any other on-site form of sewage disposal.

Vetoed In Part

(11c) Information technology study. In consultation with the office of the commissioner of insurance, the department of administration shall prepare a report on information technology services provided during the 2017-19 fiscal biennium by the division of enterprise technology to the office of the commissioner of insurance. The report shall identify efficiencies associated with the office of the commissioner of insurance receiving information technology services from the division of enterprise technology rather than providing those services itself. The department of administration shall submit the report with its 2019-21 biennial budget request.

- (11i) WORKER'S COMPENSATION HEARINGS STUDY. The Vetoed division of hearings and appeals shall conduct a study of In Part the audio and video needs for worker's compensation hearings and the feasibility of using audio and video technology alternatives for those hearings. The division shall submit its findings to the worker's compensation advisory council no later than June 30, 2018. Based on the findings of the study, the council may submit a recommendation to the division regarding audio and video recording equipment sufficient to replace a court reporter for inclusion in the department of administration's 2019–21 biennial budget request.
- (11q) REPORT CONCERNING CERTAIN INFORMATION Vetoed TECHNOLOGY AND PROCUREMENT SERVICES POSITIONS. No In Part later than August 31, 2018, the department of administration shall submit a report to the joint committee on finance concerning the activities performed in the 2017–18 fiscal year by the 2.0 PR positions providing information technology services to state agencies and the 2.0 PR positions providing procurement services, created in budget determinations for this act for an information technology procurement initiative. The report shall include all of the following:
- (a) Accomplishments of the new positions, including system or process improvements and major information technology procurements that were done efficiently or effectively.
- (b) All additional savings or efficiencies that the department of administration estimates resulted from the activities of the new positions.
- (c) The department of administration's plans for additional improvements, projects, or work products for the new positions for the 2018–19 fiscal year.
- (11s) REPORT CONCERNING CERTAIN INFORMATION Vetoed TECHNOLOGY POSITIONS CONVERTED FROM CONTRACTOR In Part STATUS. No later than August 31, 2018, the department of administration shall submit a report to the joint committee on finance concerning the activities performed in the 2017-18 fiscal year by the permanent information technology positions converted from contractor staff in budget determinations for this act. The report shall include all of the following:
- (a) Accomplishments of the converted positions, including system or process improvements, progress or completion of projects, and finished work products.
- (b) All additional savings or efficiencies that the department of administration estimates resulted from the activities of the converted positions.
- (c) The department of administration's plans for additional improvements, projects, or work products for the converted positions for the 2018–19 fiscal year.
 - (11u) Position elimination report.

Vetoed In Part

- (a) Not later than January 1, 2018, the department of administration shall report to the cochairpersons of the joint committee on finance the funding source for, and the appropriation to be decreased with regard to, 10.0 vacant SEG FTE positions relating to forestry or parks to be eliminated in the department of natural resources.
- (b) The positions identified in paragraph (a) shall be eliminated and the appropriations shall be decreased in the final 2017–19 appropriation schedule in chapter 20 of the statutes.

Vetoed In Part (11w) Length of Service awards for Correctional Officers and Youth Counselors; Compensation Plan. If, on the effective date of this subsection, the compensation plan under section 230.12 of the statutes has been adopted for the 2017–19 biennium and the compensation plan does not include the supplemental compensation required under section 230.12 (1) (cm) of the statutes, by no later than 30 days after the effective date of this subsection, the administrator of the division of personnel management in the department of administration shall propose an amendment under section 230.12 (3) (c) of the statutes to include the supplemental compensation required under section 230.12 (1) (cm) of the statutes in the compensation plan for the 2017–19 biennium.

SECTION 9102. Nonstatutory provisions; Agriculture, Trade and Consumer Protection.

(2) Professional assistance procedures; emergency rules. The veterinary examining board may use the procedure under section 227.24 of the statutes to promulgate rules under section 89.03 (3) of the statutes for the period before the effective date of the permanent rule promulgated under section 89.03 (3) of the statutes but not to exceed the period authorized under section 227.24 (1) (c) of the statutes, subject to one extension of 60 days under section 227.24 (2) of the statutes. If the board uses this procedure to promulgate these rules, the board shall promulgate the rules no later than the 60th day after the

effective date of this subsection. Notwithstanding section 227.24 (1) (a), (2) (b), and (3) of the statutes, the board is not required to provide evidence that promulgating a rule under this subsection as an emergency rule is necessary for the preservation of the public peace, health, safety, or welfare and is not required to provide a finding of emergency for a rule promulgated under this subsection.

- (3) Transition Period.
- (a) Notwithstanding the annual period for a license specified in section 94.64 (3) of the statutes, a license issued on or after August 15, 2017, and before October 1, 2017, shall expire on September 30, 2018.
- (b) Notwithstanding the annual period for a license specified in section 94.65 (2) of the statutes, a license issued on or after April 1, 2017, and before October 1, 2017, shall expire on September 30, 2018.
- (c) Notwithstanding the annual reporting period for a permit specified in section 94.65 (6) (a) of the statutes, the reporting period for a permit issued on or after April 1, 2017, and before October 1, 2017, shall be January 1, 2017, to June 30, 2018.

SECTION 9103. Nonstatutory provisions; Arts Board.

(1p) State AID FOR THE ARTS. From the appropriation under section 20.380 (3) (b) of the statutes, the arts board shall make a grant of \$100,000 in fiscal year 2017–18 for the purpose of making improvements to or expanding an existing arts center to a county that borders the state of Illinois, has a population between 35,000 and 40,000 persons as of the 2010 U.S. census, and has an existing arts center. The arts board may not make the grant under this subsection unless the county receiving the grant provides at least an equal amount of funding from public or private sources.

SECTION 9104. Nonstatutory provisions; Building Commission.

Vetoed In Part

(1) 2017–19 AUTHORIZED STATE BUILDING PROGRAM. For the fiscal years beginning on July 1, 2017, and ending on June 30, 2019, the Authorized State Building Program is as follows:

(a) Building Commission	
1. Projects financed by general fund supported borrowing:	
a. State Capitol basement renovations	\$ 1,000,000
2. Agency totals:	
General fund supported borrowing	<u>1,000,000</u>
Total — All sources of funds	\$ 1,000,000
(b) Department of Administration	
1. Projects financed by program revenue supported borrowing:	
a. Southeast Wisconsin Law Enforcement	
Facility — Milwaukee	\$ 75,000,000
2. Projects financed with existing program revenue supported borrowing:	
a. State Office Building Replacement — land	
only — Milwaukee	4,000,000

2017 Assembly Bill 64	- 361 -	2017 Wisc	onsin Act 59
3. Agency totals:			
Program revenue suppor	rted borrowing		75,000,000
Existing program revent	ue supported borrowing		4,000,000
Total — All sources of f	funds	\$	79,000,000
(c) Department of Correction	TS		
1. Projects financed by genero	al fund supported borrowing:		
 a. Fox Lake Correction drinking water system Fox Lake 		\$	3,000,000
b. Waupun Correction behavior housing uni improvements — Wa	t life safety		6,981,000
c. Wisconsin Secure F inmate programs buil			8,870,000
d. Geriatric Correction and renovation of a factoric correctional institution			7,000,000
2. Agency totals:			
General fund supported	borrowing		25,851,000
Total — All sources of t	funds	\$	25,851,000
(d) DEPARTMENT OF HEALTH SER	VICES		
1. Projects financed by genera	al fund supported borrowing:		
 a. Mendota Mental He Lorenz Hall West sec 	ealth Institute — ure treatment units — Madison	\$	16,972,000
(Total project all fu	nding sources \$17,972,000)		
b. Mendota Mental Ho Boiler #1 replacemen			5,723,000
2. Projects financed by progre	am revenue:		
 a. Mendota Mental He Lorenz Hall West sec Madison 			1,000,000
	nding sources \$17,972,000)		1,000,000
3. Agency totals:	g 304120 \$17,572,000)		
General fund supported	borrowing		22,695,000
Program revenue	C		1,000,000
Total — All sources of f	funds	\$	23,695,000
(e) Department of Military A	FFAIRS		
1. Projects financed by genero	al fund supported borrowing:		
 a. National Guard Rea addition and renovati 		\$	6,972,200
(Total project all fu	nding sources \$24,170,900)		
b. National Guard Rearenovation phase II —			3,245,900
(Total project all fu	nding sources \$6,491,800)		
2. Projects financed by federa	al funds:		

2017 Wisconsin Act 59	- 362 -	2017 Ass	embly Bill 64
a. National Guard Readin			
addition and renovation -			17,198,700
	ng sources \$24,170,900)		
b. National Guard Readin			2 245 000
renovation phase II — M			3,245,900
(Total project all fundir	ig sources \$6,491,800)		
3. Agency totals:			10.210.100
General fund supported bor	rowing		10,218,100
Federal funds	1.	ф	<u>20,444,600</u>
Total — All sources of fund		\$	30,662,700
(f) DEPARTMENT OF NATURAL RESOU			
stewardship property devel	eneral fund supported borrowing — opment and local assistance funds:		
a. High Cliff State Park —	– family campground		0.44 = 0.0
expansion — Harrison		\$	841,700
am. Peninsula State Park — reconstruction — Gibralt	_		750,000
			750,000
(Total project all fundir			
b. Peninsula State Park — campground toilet showe	•		
replacement — Gibraltar	_		839,300
1m. Projects financed by program			
a. Pattison State Park — c Pattison Ranger Station –	construction of communications tower at — Superior		1,241,700
2. Projects financed by segregated	_		, ,
a. Science Operations Cer			
and chronic wasting disea	ase processing		
center addition — Monor	na		4,805,800
3. Projects financed by gift funds.	•		
a. Peninsula State Park —	_		
reconstruction — Gibralt	ar		1,772,100
4. Agency totals:			
Existing general fund suppo development and local as	orted borrowing — stewardship property		2,431,000
Segregated fund supported l			4,805,800
Program revenue	borrowing		1,241,700
Gift funds			1,241,700 1,772,100
Total — All sources of fund	c	\$	10,250,600
(g) State Fair Park		Ψ	10,230,000
1. Projects financed by gifts, gran	nts and other receipts.		
a. Cream Puff Pavilion —	_	\$	6,000,000
b. Dairy Education Center		Ψ	1,920,000
2. Agency totals:	West Allis		1,920,000
Gifts, grants, and other rece	ints		7,920,000
(h) State Historical Society	ipω		7,720,000
1. Projects financed by gifts, gran	nts and other receipts:		
1. 1 rojects jumiced by gijts, gran	as, and other receipts.		

2017 Assembly Bill 64	- 363 -	2017 Wisconsin Act 59
a. Old World Wiscon biergarten — Eagle	nsin — old brewery and	\$ 1,650,000
2. Agency totals:		Ψ 1,030,000
Gifts, grants, and other	r receints	1,650,000
(i) Department of Veterans	_	1,020,000
* *	eral fund supported borrowing:	
• • •	ns Home at King —	
electrical substation	_	\$ 1,048,600
(Total project all f	funding sources \$2,996,000)	
b. Wisconsin Veterar	ns Home at King —	
food service system	upgrades	2,450,300
	Funding sources \$7,001,000)	
	ns Home at King —	022.700
water improvement		833,700
	funding sources \$2,382,000)	
	ram revenue supported borrowing:	
a. Wisconsin Veteral food service system	ns Home at King —	4,550,700
	Funding sources \$7,001,000)	4,330,700
	ing program revenue supported borrowing:	
	ns Home at King —	
electrical substation	=	1,947,400
(Total project all f	Funding sources \$2,996,000)	
	ns Home at King —	
water improvement	S	1,548,300
(Total project all f	funding sources \$2,382,000)	
4. Projects financed by feder	ral funds:	
	Veterans Memorial	
	– crypts, irrigation, and	1 701 200
flag plaza	\$	1,701,300
	funding sources \$1,833,500)	
b. Southern Wiscons Cemetery at Union		
alignment	Grove headstone	3,444,500
_	Funding sources \$3,731,000)	
5. Projects financed by prog	_	
a. Central Wisconsin	Neterans Memorial	
	- crypts, irrigation, and	
flag plaza		132,200
	Funding sources \$1,833,500)	
	in Veterans Memorial	
Cemetery at Union realignment	Grove — neadstone	286,500
	Funding sources \$3,731,000)	200,300
6. Agency totals:	unding sources \$3,731,000)	
General fund supporte	d borrowing	4,332,600
Program revenue supp	_	4,550,700
1 Togram Tevenue supp	orted corrowing	7,550,700

2017 Wisconsin Act 59	- 364 -	2017 Assembly Bill 64
Existing program revenue supp Federal funds Program revenue	ported borrowing	3,495,700 5,145,800 418,700
Total — All sources of funds (j) University of Wisconsin System		\$ 17,943,500
1. Projects financed by general fund	**	
(Total project all funding s		\$ 23,839,000
b. Milwaukee — Northwest (Total project all funding s	sources \$52,180,000)	46,800,000
c. Parkside — Wyllie Hall Ro (Total project all funding s	sources \$35,886,000)	35,201,000
mechanical and industrial er		
(Total project all funding se. Platteville — construction	sources \$55,189,000) and renovation of Boebel Hall	54,602,000 23,772,000
f. Whitewater — utility corriction chiller plant upgrade		16,698,000
(Total project all funding s g. System — classroom reno	vations/	10 000 000
instructional technology imp 2. Projects financed by program reve		10,000,000
	Hall addition and renovation	\$ 19,307,000
Parking Lot 62	of a new parking ramp on the site of	20,647,000
(Total project all funding s c. Madison — Lathrop Drive	sources \$23,647,000) b/Bascom Hill utility repairs — phase I	
(Total project all funding s d. Parkside — Wyllie Hall re	sources \$32,656,000)	8,817,000
(Total project all funding s	_	685,000
mechanical and industrial er		587,000
(Total project all funding s f. River Falls — construction May Hall	sources \$55,189,000) In of an addition to and renovation of	4,955,000
•	ram revenue supported borrowing:	4,933,000
 a. Extension — Lowell Hall (Total project all funding s 		3,005,000
b. Milwaukee — Northwest (Total project all funding s		3,200,000
 c. Milwaukee — Sandburg F phase I (Total project all funding s 		31,000,000
d. Whitewater — utility corrictions of the chiller plant upgrade		11,902,000

(Total project all funding sources \$28,600,000)	
4. Projects financed by program revenue:	
a. Extension — Lowell Hall floors 2–4 renovation	1,000,000
(Total project all funding sources \$4,005,000)	
b. Madison — construction of a new parking ramp on the site of	
Parking Lot 62	
(Total project all funding sources \$23,647,000)	3,000,000
c. Milwaukee — Sandburg Hall renovation —	
phase I	2,500,000
(Total project all funding sources \$33,500,000)	
5. Projects financed by gifts, grants, and other receipts:	
a. Milwaukee — Northwest Quadrant renovation	2,180,000
(Total project all funding sources \$52,180,000)	
6. Agency totals:	
General fund supported borrowing	210,912,000
Program revenue supported borrowing	54,998,000
Existing program revenue supported borrowing	49,107,000
Program revenue	6,500,000
Gifts, grants, and other receipts	<u>2,180,000</u>
Total — All sources of funds	\$ 323,697,000
(k) Brown County STEM Innovation Center — Brown County	
1. Projects financed by general fund supported borrowing:	
a. Brown County STEM Innovation Center	\$ 5,000,000
(Total project all funding sources \$15,000,000)	
2. Projects financed by gifts, grants, and other receipts:	
a. Brown County STEM Innovation Center	10,000,000
(Total project all funding sources \$15,000,000)	
3. Agency totals:	
General fund supported borrowing	5,000,000
Gifts, grants, and other receipts	10,000,000
Total — All sources of funds	\$ 15,000,000
(L) La Crosse Center — La Crosse	
1. Projects financed by general fund supported borrowing:	
a. La Crosse Center	\$ 5,000,000
(Total project all funding sources \$47,000,000)	
2. Projects financed by gifts, grants, and other receipts:	
a. La Crosse Center	42,000,000
(Total project all funding sources \$47,000,000)	
3. Agency totals:	
General fund supported borrowing	5,000,000
Gifts, grants, and other receipts	42,000,000
Total — All sources of funds	\$ 47,000,000
(m) St. Ann Center for Intergenerational Care, Inc. — Milwaukee	
1. Projects financed by general fund supported borrowing:	
a. St. Ann Center for Intergenerational Care, Inc.	\$ 5,000,000

2017 Wisconsin Ac	t 59 –	366 –	2017 Assemb	bly Bill 64
(Tot	al project all funding sources \$2	25,268,200)		
2. Projects fin	anced by gifts, grants, and other	r receipts:		
a. St. A	Ann Center for Intergenerational	l Care, Inc.		20,268,200
(Tot	al project all funding sources \$2	25,268,200)		
3. Agency tota	els:			
General	fund supported borrowing			5,000,000
Gifts, gr	ants, and other receipts			20,268,200
Total —	All sources of funds		\$	25,268,200
(n) All agency	PROJECT FUNDING			
1. Projects fin	anced by general fund supporte	d borrowing:		
a. Faci	ility maintenance and repair		\$	97,868,000
(Tot	al program all funding sources	\$178,167,000)		
	ity repair and renovation			60,000,000
(Tot	al program all funding sources	\$113,903,300)		
	lth, safety, and environmental p			20,000,000
	al program all funding sources	\$33,016,300)		
	ventive maintenance			800,000
	al program all funding sources	\$900,000)		
	ital equipment acquisition			3,000,000
	al program all funding sources			
	anced by existing general fund s y — stewardship property devel			
a. Faci	ility maintenance and repair			3,708,400
(Tot	al program all funding sources	\$178,167,000)		
b. Util	ity repair and renovation			2,062,700
(Tot	al program all funding sources	\$113,903,300)		
c. Hea	lth, safety, and environmental p	rotection		47,900
(Tot	al program all funding sources	\$33,016,300)		
3. Projects fin	anced by program revenue supp	ported borrowing:		
a. Faci	ility maintenance and repair			12,500,000
(Tot	al program all funding sources	\$178,167,000)		
b. Util	ities repair and renovation			5,500,000
(Tot	al program all funding sources	\$113,903,300)		
c. Hea	lth, safety, and environmental p	rotection		4,000,000
(Tot	al program all funding sources	\$33,016,300)		
	anced by existing program rever	nue supported borrowing:		
a. Faci	ility maintenance and repair			15,000,000
(Tot	al program all funding sources	\$178,167,000)		
	ity repair and renovation			18,500,000
	al program all funding sources			
	grammatic remodeling and reno			4,000,000
	al program all funding sources	\$12,129,000)		
	rgy conservation			20,000,000
5. Projects fin	anced by segregated fund suppo	orted borrowing:		

2017 Assembly Bill 64	<i>−</i> 367 <i>−</i>	2017 Wisconsin Act 59
a. Facility mainten	ance and repair	1,000,000
•	all funding sources \$178,167,000)	
, 10	gregated fund supported revenue borrowing:	
a. Facility mainten		2,445,000
(Total program a	all funding sources \$178,167,000)	
b. Utility repair and	d renovation	2,241,000
(Total program a	all funding sources \$113,903,300)	
c. Health, safety, a	nd environmental protection	3,350,000
(Total program a	all funding sources \$33,016,300)	
d. Preventive main	tenance	100,000
(Total program a	all funding sources \$900,000)	
e. Programmatic re	emodeling and renovation	3,569,000
(Total program a	all funding sources \$12,129,000)	
f. Capital equipme	ent acquisition	175,000
(Total program a	all funding sources \$3,175,000)	
7. Projects financed by pro	ogram revenue:	
a. Facility mainten	ance and repair	29,874,400
(Total program a	all funding sources \$178,167,000)	
b. Utility repair and	_	19,582,400
(Total program a	all funding sources \$113,903,300)	
	nd environmental protection	800,000
· · · · · · · · · · · · · · · · · · ·	all funding sources \$33,016,300)	
	emodeling and renovation	3,361,000
	all funding sources \$12,129,000)	
8. Projects financed by fea	_	
a. Facility mainten		10,466,300
(Total program a	all funding sources \$178,167,000)	
b. Utility repair and	_	2,289,500
(Total program a	all funding sources \$113,903,300)	
	nd environmental protection	4,627,000
	all funding sources \$33,016,300)	
	emodeling and renovation	699,000
_	all funding sources \$12,129,000)	
9. Gifts, grants, and other	_	
a. Facility mainten	_	5,304,900
(Total program a	all funding sources \$178,167,000)	
b. Utility repair and	d renovation	3,727,700
(Total program a	all funding sources \$113,903,300)	
	nd environmental protection	191,400
	all funding sources \$33,016,300)	
	emodeling and renovation	500,000
	all funding sources \$12,129,000)	•
10. All agency totals:		
General fund suppor	ted borrowing	181,668,000

(o)

Total existing general fund supported borrowing authority —	
stewardship property development and local assistance funds	5,819,000
Program revenue supported borrowing	22,000,000
Existing program revenue supported borrowing	57,500,000
Segregated fund supported borrowing	1,000,000
Segregated fund supported revenue borrowing	11,880,000
Program revenue	53,617,800
Federal funds	18,081,800
Gifts, grants, and other receipts	9,724,000
Total — All sources of funds	\$ 361,290,600
SUMMARY	
Total general fund supported borrowing	\$ 471,676,700
Total existing general fund supported borrowing — stewardship	
property development and local assistance funds	8,250,000
Total program revenue supported borrowing	156,548,700
Total existing program revenue supported borrowing	114,102,700
Total segregated fund supported borrowing	5,805,800
Total segregated fund supported revenue borrowing	11,880,000
Total program revenue	62,778,200
Total gifts, grants, and other receipts	95,514,300
Total federal funds	43,672,200
Total — All sources of funds	\$ 970,228,600

- (2) PROGRAMS PREVIOUSLY AUTHORIZED. In addition to the projects and financing authority enumerated in subsection (1), the building and financing authority enumerated in the previous state building program is continued in the 2017–19 fiscal biennium.
- (3) Loans. During the 2017–19 fiscal biennium, the building commission may make loans from general fund supported borrowing or the building trust fund to state agencies, as defined in section 20.001 (1) of the statutes, for projects that are to be utilized for programs not funded by general purpose revenue and that are authorized in subsection (1).
- (4) 2013–15 AUTHORIZED STATE BUILDING PROGRAM CHANGES. In 2013 Wisconsin Act 20, section 9104 (1) (b) 1. d., under projects financed by general fund supported borrowing, the amount authorized for the project identified as "Green Bay Correctional Institution north and south cell halls improvements" is increased from \$3,750,000 to \$22,232,000 and the appropriate totals are adjusted accordingly.
- (5) 2015-17 Authorized State Building Program changes.
- (a) In 2015 Wisconsin Act 55, section 9104 (1) (e) 1. a., under projects financed by existing general fund supported borrowing, the amount authorized for the project identified as "Willow River State Park Little Falls Dam repair or replacement Hudson" is increased from

- \$3,041,700 to \$6,541,700 and the appropriate totals are adjusted accordingly.
- (b) In 2015 Wisconsin Act 55, section 9104 (1) (e) 2. a., under projects financed by existing stewardship borrowing, the amount authorized for the project identified as "Willow River State Park Little Falls Dam repair or replacement Hudson" is increased from \$5,000,000 to \$12,500,000 and the appropriate totals are adjusted accordingly.
- (6) 2013-15 Authorized State Building Program deletions.
- (a) In 2013 Wisconsin Act 20, section 9104 (1) (g) 3. k., under projects financed by program revenue supported borrowing, the 2013–15 Authorized State Building Program project identified as "Platteville residence hall and dining facility" is deleted and the appropriate totals are decreased accordingly.
- (b) In 2013 Wisconsin Act 20, section 9104 (1) (g) 3. o., under projects financed by program revenue supported borrowing, the 2013–15 Authorized State Building Program project identified as "Whitewater indoor tennis building" is deleted and the appropriate totals are decreased accordingly.
- (7) LA CROSSE CENTER. Notwithstanding section 13.48 (33e) (b) of the statutes, the building commission may not make a grant to the city of La Crosse for the remodeling and expansion of the La Crosse Center, as

enumerated in subsection (1) (L), under section 13.48 (33e) of the statutes, unless the department of administration has reviewed and approved plans for the project. Notwithstanding sections 16.85 (1) and 16.855 (1m) of the statutes, the department of administration may not supervise any services or work or let any contract for the project. Section 16.87 of the statutes does not apply to the project.

- (8) St. Ann Center for Intergenerational Care, INC.; BUCYRUS CAMPUS. Notwithstanding section 13.48 (33m) (b) of the statutes, the building commission may not make a grant to the St. Ann Center for Intergenerational Care, Inc., for the completion of its Bucyrus Campus, as enumerated in subsection (1) (m), under section 13.48 (33m) of the statutes, unless the department of administration has reviewed and approved plans for the project. Notwithstanding sections 16.85 (1) and 16.855 (1m) of the statutes, the department of administration may not supervise any services or work or let any contract for the project. Section 16.87 of the statutes does not apply to the project.
- (9) Brown County innovation center. Notwithstanding section 13.48 (33s) (b) of the statutes, the building commission shall not make a grant to Brown County for the construction of a science, technology, engineering, and mathematics innovation center, as enumerated in subsection (1) (k), under section 13.48 (33s) of the statutes, unless the department of administration has reviewed and approved plans for the project. Notwithstanding sections 16.85 (1) and 16.855 (1m) of the statutes, the department of administration shall not supervise any services or work or let any contract for the project. Section 16.87 of the statutes does not apply to the project.
- (10) CONSTRUCTION OF NATIONAL GUARD READINESS CENTERS. From the appropriation under section 20.867 (2) (r) of the statutes, the building commission shall allocate \$627,800, matched by \$1,883,900 of federal funds, to develop preliminary plans and specifications for the construction of national guard readiness centers located in the cities of Black River Falls, Viroqua, Wausau, and Wisconsin Rapids.
- (11) PLAN CONCERNING DEPARTMENT OF CORRECTIONS FACILITIES.
- (a) There is created a corrections facilities planning committee consisting of 3 members appointed by the governor, one of whom the governor shall designate as chair of the committee, and 6 members of the legislature, jointly appointed by the speaker of the assembly and the senate majority leader.
- (b) The corrections facilities planning committee shall develop a comprehensive, long-range master plan concerning department of corrections facilities and, no later than September 15, 2018, shall submit the plan to the governor and the appropriate standing committees of the legislature under section 13.172 (3) of the statutes.

- (c) The building commission shall allocate \$600,000 from the appropriation under section 20.867 (2) (r) of the statutes for the corrections facilities planning committee to develop the plan under paragraph (b) and the department of administration shall assist the committee in the performance of its functions.
- (d) The corrections facilities planning committee terminates upon submission of the plan under paragraph (b).
 - (12) Geriatric Correctional Institution.
- (a) No bonds may be issued for the geriatric correc- Vetoed tional institution enumerated under subsection (1) (c) 1. In Part d. without the approval of the joint committee on finance under paragraph (b).
- (b) The department of corrections may request the approval of the joint committee on finance for the bond issuance enumerated under subsection (1) (c) 1. d. and for In Part the release of funds from the appropriation under section 20.865 (4) (a) of the statutes for operating costs of that institution once the department of corrections has identified the location of the institution and determined the staffing and other operating costs of the institution.

SECTION 9105. Nonstatutory provisions; Child Abuse and Neglect Prevention Board.

Section 9106. Nonstatutory provisions; Children and Families.

- (1) EMERGENCY RULES. Using the procedure under section 227.24 of the statutes, the department of children and families may promulgate the rules required under sections 48.685 (9) and 48.686 (8) of the statutes as emergency rules. Notwithstanding section 227.24 (1) (a), (2) (b), and (3) of the statutes, the department of children and families is not required to provide evidence that promulgating a rule under this subsection as an emergency rule is necessary for the preservation of the public peace, health, safety, or welfare and is not required to provide a finding of emergency for a rule promulgated under this subsection.
- (2) WISCONSIN WORKS CONTROLLED SUBSTANCE SCREENING AND TESTING.
- (a) Using the procedure under section 227.24 of the statutes, the department of children and families may promulgate the rules required under section 49.162 (7) of the statutes for the period before the effective date of the permanent rule promulgated under section 49.162 (7) of the statutes but not to exceed the period authorized under section 227.24 (1) (c) of the statutes, subject to extension under section 227.24 (2) of the statutes. Notwithstanding section 227.24 (1) (a), (2) (b), and (3) of the statutes, the department is not required to provide evidence that promulgating a rule under this paragraph as an emergency rule is necessary for the preservation of the public peace, health, safety, or welfare and is not required to provide a finding of emergency for a rule promulgated under this paragraph.
- (b) The department of children and families shall present the statement of scope of any emergency rules

Vetoed

Vetoed In Part

promulgated under paragraph (a) to the governor for approval under section 227.135 (2) of the statutes no later than the 120th day after the effective date of this paragraph.

- (3t) RATE-BASED SERVICE CONTRACTS. If on the effective date of this subsection, the amount accumulated by a provider, as defined in section 49.34 (5m) (a) 1. of the statutes, from all contract periods ending before that date for all rate-based services, as defined in section 49.34 (5m) (a) 2. of the statutes, provided by the provider exceeds 10 percent of the provider's total contract amount for all rate-based services in the year before the effective date of this subsection, the provider shall provide written notice of that excess to all purchasers of that rate-based service and, upon the written request of such a purchaser received no later than 6 months after the date of the notice, shall return to the purchaser the purchaser's proportional share of that excess. If the department of children and families under section 49.34 (5m) of the statutes determines based on an audit or fiscal review that the amount of the excess identified by the provider was incorrect, the department of children and families may seek to recover funds after the 6-month period has expired. The department of children and families shall commence any audit or fiscal review under this subsection within 6 years after the end of the contract period.
 - (3w) Work Participation Rate.

Vetoed In Part (a) The department of children and families shall submit reports to the joint committee on finance that detail performance on work participation rate targets in the temporary assistance for needy families program, progress made on any compliance programs with the federal department of health and human services, and the appeals process for any penalties applied to the state under the temporary assistance for needy families program that are related to work participation rate requirements. The department of children and families shall submit the reports no later than September 15, 2017, March 16, 2018, September 14, 2018, and March 15, 2019

Vetoed In Part

Vetoed In Part

(b) On or before October 1, 2018, the department of children and families shall present to the joint committee on finance for its approval a plan to improve work participation rates in the temporary assistance for needy families program. The department may incorporate into the plan a request for a waiver under Section 1115 of the Social Security Act. If the cochairpersons of the joint committee on finance do not notify the department that the committee has scheduled a meeting for the purpose of reviewing the plan within 14 working days after the date the plan was submitted, the department shall implement the plan. If, within 14 working days after the date the plan was submitted, the cochairpersons of the committee notify the department that the committee has scheduled a meeting for the purpose of reviewing the plan, the department may not implement the plan unless the committee approves or modifies the plan. If the committee modifies the plan, the department may implement the plan only as modified by the committee.

SECTION 9107. Nonstatutory provisions; Circuit Courts.

SECTION 9108. Nonstatutory provisions; Corrections.

- (2t) RATE-BASED SERVICE CONTRACTS. If on the effective date of this subsection, the amount accumulated by a provider, as defined in section 301.08 (2) (em) 1. a. of the statutes, from all contract periods ending before that date for all rate-based services, as defined in section 301.08 (2) (em) 1. b. of the statutes, provided by the provider exceeds 10 percent of the provider's total contract amount for all rate-based services in the year before the effective date of this subsection, the provider shall provide written notice of that excess to all purchasers of that rate-based service and, upon the written request of such a purchaser received no later than 6 months after the date of the notice, shall return to the purchaser the purchaser's proportional share of that excess. If the department of corrections under section 301.08 (2) (em) of the statutes determines based on an audit or fiscal review that the amount of the excess identified by the provider was incorrect, the department of corrections may seek to recover funds after the 6-month period has expired. The department of corrections shall commence any audit or fiscal review under this subsection within 6 years after the end of the contract period.
- (2w) REPORT ON BODY-WORN CAMERAS BY CORRECTIONAL OFFICERS. By July 1, 2018, the department of corrections shall submit a report to the appropriate standing committees of the legislature under section 13.172 (3) of the statutes that does all of the following:
- (a) Identifies each institution at which body-worn cameras are being used.
- (b) Indicates how many body—worn cameras are being used at each institution under paragraph (a).
- (c) Compares the number of staff and inmate assaults reported in restrictive housing since body-worn cameras started being used to the number of staff and inmate assaults reported in restricted housing before body-worn cameras started being used.
 - (8w) ALCOHOL ABUSE TREATMENT PROGRAM.
- (a) The department of corrections shall design an alcohol abuse treatment program to provide intensive treatment in conjunction with a work release model that allows inmates to work in individual job placements. The department shall develop community job placements that are appropriately matched to each inmate's employment and educational skills and shall provide or arrange for appropriate transportation to and from job sites.
- (b) The department of corrections shall submit as part of its 2019–21 agency budget request a request for staffing and funding for the program under paragraph (a)

In Part

and any statutory changes that may be necessary to provide sentencing modifications to coordinate the program.

(c) Five years after the program under paragraph (a) begins operation, the department of corrections shall submit to the governor and the appropriate standing committees of the legislature under section 13.172 (3) of the statutes an evidence-based evaluation of the program's impact on inmates' long-term recovery from alcohol abuse problems and recidivism into the criminal justice system.

Vetoed In Part

- (22t) Report on serious mental illness among DEPARTMENT OF CORRECTIONS INMATES. By July 1, 2018, the department of corrections shall submit to the appropriate standing committees of the legislature under section 13.172 (3) of the statutes a report on serious mental illness among department of corrections inmates, which shall include all of the following:
- (a) The average number of inmates with a serious mental illness in each department of corrections institu-
- (b) The average number of inmates with a serious mental illness in each department of corrections institution restrictive housing unit.
- (c) The department of corrections' compliance status or alternative policies related to each of the U.S. department of justice's recommendations related to the use of restrictive housing for inmates with a serious mental illness.
- (d) An estimate of what additional resources, if any, are necessary to address serious mental illness within the department of corrections inmate population.

Vetoed In Part

- (31t) DEPARTMENT OF CORRECTIONS INMATE WORK OPPORTUNITIES.
- (a) By December 31, 2017, the department of corrections shall submit to the appropriate standing committees of the legislature under section 13.172 (3) of the statutes a report on department of corrections inmate work opportunities, which shall include all of the following:
- 1. A survey of existing work release programs at each department of corrections institution and the estimated number of inmates who participate in those programs at each department of corrections institution.
- 2. The estimated number of department of corrections inmates who continue to work after release from incarceration at a job at which he or she began working as an inmate in a work release program.
- 3. The costs assessed by the department of corrections on each department of corrections work release participant.
- (b) By December 31, 2017, the department of corrections shall submit to the appropriate standing committees of the legislature under section 13.172 (3) of the statutes a plan to increase employment opportunity incentives for department of corrections inmates.

SECTION 9109. Nonstatutory provisions; Court of Appeals.

SECTION 9110. Nonstatutory provisions; District Attorneys.

SECTION 9111. Nonstatutory provisions; Educational Approval Board.

- TEMPORARY ATTACHMENT OF EDUCATIONAL Vetoed APPROVAL BOARD TO DEPARTMENT OF SAFETY AND PROFES- In Part SIONAL SERVICES
- (a) Assets and liabilities. On the effective date of this paragraph, the assets and liabilities of the technical college system board primarily related to the functions of the educational approval board, as determined by the secretary of administration, become the assets and liabilities of the department of safety and professional services.
 - (b) Positions and employees.
- 1. On the effective date of this subdivision, all FTE positions, and the incumbent employees holding those positions, in the technical college system board performing duties primarily related to the functions of the educational approval board, as determined by the secretary of administration, are transferred to the department of safety and professional services.
- 2. Employees transferred under subdivision 1. have all the rights and the same status under chapter 230 of the statutes in the department of safety and professional services that they enjoyed in the technical college system board immediately before the transfer. Notwithstanding section 230.28 (4) of the statutes, no employee transferred under subdivision 1. who has attained permanent status in class is required to serve a probationary period.
- (c) Tangible personal property. On the effective date of this paragraph, all tangible personal property, including records, of the technical college system board that is primarily related to the functions of the educational approval board, as determined by the secretary of administration, is transferred to the department of safety and professional services.
- (d) Contracts. All contracts entered into by the technical college system board in effect on the effective date of this paragraph that are primarily related to the functions of the educational approval board, as determined by the secretary of administration, remain in effect and are transferred to the department of safety and professional services. The department of safety and professional services shall carry out any obligations under such a contract until the contract is modified or rescinded by the department of safety and professional services to the extent allowed under the contract.
- (e) Pending matters. Any matter pending with the technical college system board that is primarily related to the functions of the educational approval board, as determined by the secretary of administration, is transferred to the department of safety and professional services. All materials submitted to or actions taken by the technical college system board with respect to the pending matter are considered as having been submitted to or taken by the department of safety and professional services.

- (1q) Elimination of educational approval board AND TRANSFER OF FUNCTIONS.
- (a) Definition. In this subsection, "board" means the educational approval board.
- (b) Assets and liabilities. On the effective date of this paragraph, the assets and liabilities of the board become the assets and liabilities of the department of safety and professional services.

(bm) Positions and employees.

Vetoed In Part

Vetoed

In Part

1. On the effective date of this subdivision, all FTE positions, and the incumbent employees holding those positions, in the board are transferred to the department of safety and professional services.

- 2. Employees transferred under subdivision 1. have all the rights and the same status under chapter 230 of the statutes in the department of safety and professional services that they enjoyed in the board immediately before the transfer. Notwithstanding section 230.28 (4) of the statutes, no employee transferred under subdivision 1. who has attained permanent status in class is required to serve a probationary period.
- (c) Tangible personal property. On the effective date of this paragraph, all tangible personal property, including records, of the board is transferred to the department of safety and professional services.
- (d) Contracts. All contracts entered into by the board in effect on the effective date of this paragraph remain in effect and are transferred to the department of safety and professional services. The department of safety and professional services shall carry out any obligations under those contracts unless modified or rescinded by the department to the extent allowed under the contract.
 - (e) Rules and orders.
- 1. All rules promulgated by the board in effect on the effective date of this subdivision remain in effect until their specified expiration dates or until amended or repealed by the department of safety and professional services.
- 2. All orders issued by the board in effect on the effective date of this subdivision remain in effect until their specified expiration dates or until modified or rescinded by the department of safety and professional services.
- (f) Pending matters. Any matter pending with the board on the effective date of this paragraph is transferred to the department of safety and professional services. All materials submitted to or actions taken by the board are considered as having been submitted to or taken by the department of safety and professional services.

SECTION 9112. Nonstatutory provisions; Educational Communications Board.

SECTION 9113. Nonstatutory provisions; Elections Commission.

(1) DEPLETION OF FEDERAL HELP AMERICA VOTE ACT FUNDING FOR ELIGIBLE ELECTION ADMINISTRATION COSTS. During fiscal year 2018-19, the elections commission shall spend all available funds in the appropriation

account under section 20.510 (1) (x) of the statutes prior to spending any funds appropriated to the elections commission under section 20.510 (1) (a) of the statutes for the purpose of replacing election administration funding received from the federal government under the federal Help America Vote Act, Public Law 107-252. The elections commission may spend moneys appropriated under section 20.510 (1) (x) of the statutes only on election administration costs permissible under the federal Help America Vote Act, Public Law 107-252.

SECTION 9114. Nonstatutory provisions; **Employee Trust Funds.**

(1c) Consumer-driven health plan educational Vetoed

In Part

- (a) The department of employee trust funds shall develop a plan to conduct a consumer-driven health plan educational campaign before and during the annual enrollment period under the state health insurance plan for the 2019 calendar year. The educational campaign shall provide all of the following information:
- 1. The advantages of high-deductible health plans and health savings accounts.
- 2. Examples of individuals or families that may benefit from high-deductible health plans and health savings accounts.
- 3. Any consumer–driven health plan design changes or initiatives approved by the group insurance board for implementation by the department of employee trust funds.
- (b) No later than January 1, 2018, the department of employee trust funds shall submit the plan developed under paragraph (a), along with a request for any funding needed to conduct the educational campaign described under paragraph (a), to the joint committee on finance under section 13.10 of the statutes. The department of employee trust funds may not conduct the educational campaign unless the committee approves the plan.
- (1t) Group insurance board plan for state pro-GRAM RESERVES.
- (a) No later than March 1, 2018, the group insurance board shall submit to the joint committee on finance for review a plan that includes all of the following:
- 1. The amount of state program reserves as of December 31, 2017.
- 2. The amount of state program reserves that will be used during calendar year 2018 to reduce state program costs.
- 3. A projection of 2018 year-end state program reserves prepared by the group insurance board's consulting actuary.
- 4. The group insurance board's planned utilization of state program reserves in calendar year 2019.
- (b) If, within 21 working days after the date on which the group insurance board submitted the plan described under paragraph (a), the cochairpersons of the joint committee on finance do not notify the group insurance board

In Part

that the joint committee on finance has scheduled a meeting for the purpose of reviewing the plan, the group insurance board may implement the plan. If, within 21 working days after the date on which the group insurance board submitted the plan, the cochairpersons of the joint committee on finance notify the group insurance board that the joint committee on finance has scheduled a meeting for the purpose of reviewing the plan, the group insurance board may implement the plan only upon approval of the joint committee on finance.

- (2p) Group insurance board; group health pro-GRAM RESERVES.
- (a) During the 2017–19 fiscal biennium, the group insurance board shall use \$68,800,000 of the state group health program reserves established under section 40.03 (6) of the statutes to reduce state group health program
- (b) During the 2017–19 fiscal biennium, the group insurance board shall review its policies related to maintaining reserves for fully insured health plans. In conducting this review, the group insurance board shall review at least all of the following:
- 1. The history of changes in the participation of fully insured health plans in the group health insurance program.
- 2. The number of members affected by the discontinuation of fully insured health plans from year to year.
- 3. The dollar amount of claims or premiums associated with members that are affected by the discontinuation of fully insured health plans from year to year.
- (2w) STATE EMPLOYEE GROUP HEALTH PROGRAM SAV-INGS. The group insurance board shall attempt to ensure that state employee group health program costs, paid from general purpose revenues, are reduced by \$63,900,000 during the 2017-19 fiscal biennium. The reductions shall be achieved through a combination of the following:
- (a) Savings resulting from negotiations with insurers who provide health care coverage to state employees.
- Utilization of state group health program reserves.
- (c) Increased use of tiers under section 40.51 (6) of the statutes for state employee health insurance premium
- (d) Additional utilization of state group health program reserves during 2018 and 2019 if the group insurance board revises its reserve policy.
- (e) Health care plan design changes, with a focus on consumer-driven health care, provided that the changes do not increase total employee premium costs under the lowest tier plans under section 40.51 (6) of the statutes by more than 10 percent during 2018 and 2019. The costs include health insurance premiums, co-pays, deductibles, coinsurance, and out-of-pocket expenditures.

(f) Any other state employee health program or Vetoed health care plan changes, provided that they do not increase total employee health insurance premium costs under the lowest tier plans under section 40.51 (6) of the statutes by more than 10 percent during 2018 and 2019. The costs include health insurance premiums, co-pays, deductibles, coinsurance, and out-of-pocket expendi-

SECTION 9115. Nonstatutory provisions; Employment Relations Commission.

(1) ELIMINATION OF OFFICES OF COMMISSIONER. On the effective date of this subsection, the 3 offices of commissioner at the Employment Relations Commission are eliminated.

SECTION 9116. Nonstatutory provisions; Ethics Commission.

Section 9117. Nonstatutory provisions; Financial Institutions.

SECTION 9118. Nonstatutory provisions; Governor.

SECTION 9119. Nonstatutory provisions; Health and Educational Facilities Authority.

SECTION 9120. Nonstatutory provisions; Health Services.

- (1) EMERGENCY RULES ON YOUTH CRISIS STABILIZA-TION FACILITIES. The department of health services may promulgate emergency rules under section 227.24 of the statutes implementing certification of youth crisis stabilization facilities under section 51.042 of the statutes. Notwithstanding section 227.24 (1) (a) and (3) of the statutes, the department of health services is not required to provide evidence that promulgating a rule under this subsection as an emergency rule is necessary for the preservation of the public peace, health, safety, or welfare and is not required to provide a finding of emergency for a rule promulgated under this subsection. Notwithstanding section 227.24 (1) (c) and (2) of the statutes, emergency rules promulgated under this subsection remain in effect until July 1, 2019, or the date on which permanent rules take effect, whichever is sooner.
- (1b) SUPPLEMENT FOR YOUTH CRISIS STABILIZATION FACILITIES. During the 2017-19 fiscal biennium, the In Part department of health services may submit one or more requests to the joint committee on finance under section 13.10 of the statutes to supplement the appropriation under section 20.435 (5) (kd) of the statutes in a total of no more than \$1,245,500 from the appropriation account under section 20.435 (2) (gk) of the statutes for the purpose of providing one or more grants to a youth crisis stabilization facility under section 51.042 of the statutes. In a submission under this subsection, the department of health services shall describe its plan for distributing grant moneys, including the conditions the department would specify for the expenditure of grant moneys and the criteria the department proposes to use for selecting

Vetoed

In Part

grantees. The department of health services may not issue a request for proposals to award grants to a youth crisis stabilization facility until the joint committee on finance approves or modifies and approves the department's plan under this subsection.

- (1c) YOUTH CRISIS STABILIZATION FACILITY FUNDING PROPOSAL. The department of health services shall include in its 2019–21 biennial budget request a proposal for funding grants to youth crisis stabilization facilities under section 51.042 of the statutes with general fund moneys.
- (1g) PEER-RUN RESPITE CENTER FOR VETERANS. The department of health services shall include in its 2019–21 biennial budget request a proposal to provide ongoing general purpose revenue funding for a peer–run respite center that provides services to veterans.

Vetoed In Part

- (1t) Grace Period for County Reports. Notwithstanding sections 51.61 (1) (z) and 980.08 (4) (dm) 4. of the statutes, beginning on the effective date of this subsection and ending on the first day of the 13th month beginning after the effective date of this subsection, the county shall submit a report required under section 980.08 (4) (dm) of the statutes to the department of health services within 180 days, rather than 120 days, following the court order or be subject to action as provided in sections 51.61 (1) (z) and 980.08 (4) (dm) 4. of the statutes.
- (2) FOODSHARE EMPLOYMENT AND TRAINING PROGRAM REQUIREMENT PILOT PROGRAM.
- (a) The department of health services may implement a requirement for able–bodied adults to participate in the food stamp program's employment and training program under section 49.79 (9) of the statutes in no more than 2 vendor regions of the food stamp program's employment and training program beginning in April 2019 . The department may not impose the pilot program requirement under this paragraph after June 30, 2020.

(b) The department of health services shall evaluate the pilot program under paragraph (a) and, depending on the department's findings, submit a proposal for statewide expansion of the requirement to participate in the food stamp program's employment and training program in its 2021–23 biennial budget.

Vetoed In Part

Vetoed

In Part

Vetoed In Part

Vetoed

In Part

(c) During the 2017–19 fiscal biennium, the department of health services shall submit a detailed implementation plan for the pilot program under paragraph (a) and may submit one or more requests to the joint committee on finance under section 13.10 of the statutes to supplement the appropriations under section 20.435 (4) (a), (bm), (bn), and (bp) of the statutes from the appropriation under section 20.865 (4) (a) of the statutes for the purpose of implementing the pilot program under paragraph (a). The department of health services may only use moneys for the pilot program under paragraph (a) of the statutes if the joint committee on finance approves the request under this paragraph. Notwithstanding section 13.101 (3) of the statutes, the joint committee on finance is not

required to find that an emergency exists before making a supplementation under this paragraph.

- (2p) SUPPLEMENT FOR FOODSHARE CHILD SUPPORT AND PATERNITY COMPLIANCE. During the 2017–19 fiscal biennium, the department of health services or the department of children and families may submit one or more requests to the joint committee on finance to supplement the appropriations under section 20.435 (4) (a), (bm), and (bn) or 20.437 (2) (a) of the statutes from the appropriation under section 20.865 (4) (a) of the statutes for the purpose of implementing child support and paternity compliance for the food stamp program under section 49.79 (6m), (6q), or (6t) of the statutes, subject to section 49.79 (6u) of the statutes. If, within 14 days after the date of a department's submittal, the cochairpersons of the committee do not notify the department that the committee has scheduled a meeting to review the request, the supplement is considered approved. If the cochairpersons notify the department that the committee has scheduled a meeting to review the request, the supplement may be made only upon the approval of the committee. Notwithstanding section 13.101 (3) of the statutes, the joint committee on finance is not required to find that an emergency exists before making a supplementation under this subsection.
- (2s) FOODSHARE EMPLOYMENT AND TRAINING PROGRAM OUTCOMES REPORT. By February 1, 2018, the department of health services shall provide to the joint committee on finance an outcome report on the food stamp program's employment and training program under section 49.79 (9) of the statutes. The report shall include any proposed program improvements and contract modifications necessary based on the reported outcomes.
- (3t) RATE-BASED SERVICE CONTRACTS. If on the effective date of this subsection, the amount accumulated by a provider, as defined in section 46.036 (5m) (a) 1. of the statutes, from all contract periods ending before that date for all rate-based services, as defined in section 46.036 (5m) (a) 2. of the statutes, provided by the provider exceeds 10 percent of the provider's total contract amount for all rate-based services in the year before the effective date of this subsection, the provider shall provide written notice of that excess to all purchasers of that rate-based service and, upon the written request of such a purchaser received no later than 6 months after the date of the notice, shall return to the purchaser the purchaser's proportional share of that excess. If the department of health services under section 46.036 (5m) of the statutes determines based on an audit or fiscal review that the amount of the excess identified by the provider was incorrect, the department of health services may seek to recover funds after the 6-month period has expired. The department of health services shall commence any audit or fiscal review under this subsection within 6 years after the end of the contract period.

- (4g) Physical medicine pilot program.
- (a) Definitions. In this subsection:
- "Department" means the department of health services.
- 2. "Medical Assistance program" means the program under subchapter IV of chapter 49 of the statutes.
- 3. "Physical medicine" means rehabilitation techniques that aim to enhance and restore functional ability and quality of life to persons with physical impairments, injuries, or disabilities.
- (b) *Study*. The department shall study best practices for physical medicine and the impact the use of physical medicine has on the use and frequency of use of prescription and over—the—counter drugs and shall develop a proposal for a physical medicine pilot program to minimize prescription of addictive drugs for individuals who receive benefits under the Medical Assistance program by using chiropractic and physical and occupational therapy services that are reimbursed under the Medical Assistance program. In completing the study and developing the proposal under this paragraph, the department shall solicit input from persons that are interested in physical medicine, including those interested in chiropractic care and physical therapy.
- (c) *Report.* No later than April 1, 2018, the department shall submit a report of the study and the proposal for the pilot program under paragraph (b) to the legislature under section 13.172 (2) of the statutes. The department may not implement the pilot program under paragraph (b) unless the legislature directs or explicitly authorizes the department to implement the pilot program.
- (4j) Ambulatory surgical center reimbursement RATES UNDER MEDICAL ASSISTANCE PROGRAM. department of health services shall consult with ambulatory surgical centers to develop a plan to increase reimbursement rates for ambulatory surgical centers for services provided to recipients of Medical Assistance under subchapter IV of chapter 49 of the statutes. During the 2017-19 fiscal biennium, the department of health services may submit one or more requests to the joint committee on finance to supplement the appropriation under section 20.435 (4) (b) of the statutes in a total of no more than \$2,500,000 from the appropriation under section 20.865 (4) (a) of the statutes for the purpose of increasing reimbursement rates under the Medical Assistance program for ambulatory surgical centers in accordance with the plan established under this subsection. If, within 14 days after the date of the department's submittal, the cochairpersons of the committee do not notify the department of health services that the committee has scheduled a meeting to review the request, the supplement is considered approved. If the cochairpersons notify the department of health services that the committee has scheduled a meeting to review the request, the supplement may be made only upon the approval of the committee. Notwith-

standing section 13.101 (3) of the statutes, the joint committee on finance is not required to find that an emergency exists before making a supplementation under this subsection.

(4k) Family Care Partnership program. By Vetoed December 31, 2017, the department of health services In Part shall submit a request for a waiver of federal Medicaid law to the federal department of health and human services to expand the Family Care Partnership program, as described in section 49.496 (1) (bk) 3. of the statutes, statewide. If the federal department of health and human services approves the request, the department of health services shall, within 60 days of receiving notice of the approval, submit a plan for expansion of the Family Care Partnership program following the guidelines in the waiver to the joint committee on finance for approval. The department of health services may expand the Family Care Partnership program only as approved by the joint committee on finance. If the federal department of health and human services disapproves the request, the department of health services shall submit a report to the joint committee on finance describing the reasons the request was disapproved.

(5b) Nursing home bed licenses.

Vetoed In Part

- (a) In this subsection, "nursing home" has the meaning given in section 50.01 (3) of the statutes.
- (b) Notwithstanding sections 150.33, 150.35, and 150.39 of the statutes, from the nursing home beds that are available under section 150.31 of the statutes, the department of health services shall, following submission of the application under paragraph (c), redistribute 18 beds to a nursing home that satisfies all of the following:
- 1. On the effective date of this subdivision, it has a licensed bed capacity of no more than 30.
- 2. On the effective date of this subdivision, it is located in a county that has a population of at least 27,000, with the population of the county seat of no more than 9,200, and that is adjacent to a county with a population of at least 20,000.
- 3. It has requested the increase in the number of its licensed beds through a notice to the department of health services that includes its per diem operating and capital rates.
- (c) The department of health services shall approve an application from a nursing home that meets the qualifications under paragraph (b) within 30 days after the department of health services receives the application.
- (d) The department of health services shall develop a policy that specifies procedures for applying for, and receiving approval of, the transfer of available, licensed nursing home beds. The department of health services shall submit a report on the resulting policy to the joint committee on finance no later than July 1, 2018.
- (5f) EMERGENCY PHYSICIAN SERVICES AND REIMBURSEMENT WORKGROUP.

- (a) Under section 15.04 (1) (c) of the statutes, the department of health services shall establish a committee to examine medical services provided in hospital emergency departments to Medical Assistance recipients and make recommendations regarding potential savings in these services and increases to Medical Assistance reimbursement for emergency physician services. To the extent the committee determines appropriate, the committee may examine aspects of the healthcare system involving emergency care, including patient care practices, medication use and prescribing practices, billing and coding administration, organization of health care delivery systems, care coordination, patient financial incentives, and other aspects.
- (b) The committee under paragraph (a) shall consist of all of the following members appointed by the secretary of health services:
- 1. Two physicians practicing in Wisconsin representing a statewide physician—member organization of emergency physicians.
- 2. Two representatives of the division of the department of health services that addresses Medical Assistance services, with experience in emergency physician services, codes, and payment.
- 3. One representative who is a hospital emergency department administrator employed by a Wisconsin hospital or hospital—based health system.
- 4. One coding and billing specialist from an organization with expertise on and in the business of emergency medicine that contracts emergency physicians practicing in Wisconsin.
- (c) The committee may solicit input from others as it determines is necessary and appropriate.
- (d) The committee under this subsection must first convene no later than 60 days after the effective date of this paragraph and meet at least every 45 days until arriving at a set of recommendations.
- (e) The committee shall report its findings and recommendations to the joint committee on finance no later than September 1, 2018, and each recommendation must be made on the basis of a consensus of the committee.
 - (5h) Complex rehabilitation technology.

Vetoed In Part

- (a) The department of health services shall submit in proposed form the rules required under section 49.45 (9r) of the statutes, including the rules described under paragraph (b), to the legislative council staff under section 227.15 (1) of the statutes no later than the first day of the 13th month beginning after the effective date of this paragraph.
- (b) The department of health services shall include in the proposed rules submitted under paragraph (a) rules that designate the healthcare common procedure coding system codes that are used in the federal Medicare program for complex rehabilitation technology for the Medical Assistance program and are in accordance with section 49.45 (9r) of the statutes.

- (c) The department of health services shall in the proposed rules exempt the codes designated from any bidding or selective contracting requirements.
- (6d) CRITICAL ACCESS HOSPITAL GRANT. The department of health services shall award a grant of \$250,000 in the 2017–19 fiscal biennium to a critical access hospital to support the cost of a behavioral health crisis management system. The department of health services shall award the grant under this subsection to a critical access hospital that meets all of the following criteria:
- (a) The hospital presents a proposal that does all of the following:
- 1. Provides in–person triage, assessment, and brief intervention services to persons presenting in the hospital emergency department for reasons related to a behavioral health crisis.
- 2. Provides the services specified under subdivision 1. through video telemedicine consultation to persons presenting in hospital emergency departments, other than the hospital's emergency department, in the hospital's region.
- 3. Coordinates the transfer of persons who require care for a behavioral health condition to another facility, as appropriate.
- (b) The hospital is located in the northwestern part of this state in a county that borders Minnesota and that has immediate access to the interstate highway known as I 94.
- (c) The hospital provides alcohol and drug abuse assessment and treatment services and inpatient psychiatric services.
- (6p) MEDICAL ASSISTANCE AUDIT OF FAMILY PLANNING PROVIDERS.
- (a) In this subsection, "covered entity" has the meaning given in 42 USC 256b (a) (4) (C) and (K).
- (b) The office of the inspector general of the department of health services shall conduct an audit of all family planning service reimbursements paid to covered entities under the Medical Assistance program under subchapter IV of chapter 49 of the statutes for the period January 1, 2013, to December 31, 2016. The office of the inspector general shall conclude the audit under this subsection no later than June 30, 2019.
- (6t) COMMUNITY PROGRAM GRANTS. Notwithstanding the limitations on the amount of grants under section 46.48 of the statutes, in the 2019–21 fiscal biennium, from the appropriation account under section 20.435 (7) (bc) of the statutes, the department of health services may increase the amount of each grant or the number of grants awarded under section 46.48 of the statutes to expend the moneys described in Section 9220 (5r) of this act.

SECTION 9121. Nonstatutory provisions; Higher Educational Aids Board.

(1f) MINORITY TEACHER LOAN PROGRAM. Notwithstanding section 16.42 (1) (e) of the statutes, in submitting information under section 16.42 of the statutes for

purposes of the 2019 biennial budget act, the higher educational aids board shall submit information concerning the appropriation under section 20.235 (1) (cr) of the statutes as though the total amount appropriated under section 20.235 (1) (cr) of the statutes for the 2018–19 fiscal year was \$125,000 more than the total amount that was actually appropriated under section 20.235 (1) (cr) of the statutes for the 2018–19 fiscal year.

(2f) TEACHER LOAN PROGRAM. Notwithstanding section 16.42 (1) (e) of the statutes, in submitting information under section 16.42 of the statutes for purposes of the 2019 biennial budget act, the higher educational aids board shall submit information concerning the appropriation under section 20.235 (1) (ct) of the statutes as though the total amount appropriated under section 20.235 (1) (ct) of the statutes for the 2018–19 fiscal year was \$125,000 more than the total amount that was actually appropriated under section 20.235 (1) (ct) of the statutes for the 2018–19 fiscal year.

SECTION 9122. Nonstatutory provisions; Historical Society

- (1) CIRCUS WORLD MUSEUM.
- (a) Definitions. In this subsection:
- 1. "Foundation" means the Circus World Museum Foundation. Inc.
- 2. "Lease termination date" means the termination date of the lease agreement that is specified in section 44.16 (3) of the statutes.
- (b) *Employees*. If a lease agreement under section 44.16 (1) of the statutes is in effect on the effective date of this paragraph, the historical society shall offer employment to each individual who is employed by the foundation on the lease termination date, but only if vacant authorized or limited term positions are available for the individuals to fill and the historical society has funding for the positions.
- (c) Contracts. If a lease agreement under section 44.16 (1) of the statutes is in effect on the effective date of this paragraph, all contracts entered into by the foundation that are in effect on the lease termination date remain in effect and are transferred to the historical society on the lease termination date. The historical society shall carry out any contractual obligations under such a contract until the contract is modified or rescinded by the historical society to the extent allowed under the contract.

SECTION 9123. Nonstatutory provisions; Housing and Economic Development Authority.

SECTION 9124. Nonstatutory provisions; Insurance.

- (1) Transfer of information technology functions.
- (a) Assets and liabilities. On the effective date of this paragraph, the assets and liabilities of the office of the commissioner of insurance that are primarily related to information technology functions, as determined by the secretary of administration, become the assets and liabili-

ties of the division of enterprise technology in the department of administration.

- (b) Positions and employees. On the effective date of this paragraph, 12.5 FTE positions and the incumbent employees holding those positions in the office of the commissioner of insurance who are performing duties that are primarily related to information technology functions, as determined by the secretary of administration, are transferred to the division of enterprise technology in the department of administration.
- (c) *Employee status*. Employees transferred under paragraph (b) have all the rights and the same status under chapter 230 of the statutes in the division of enterprise technology in the department of administration that they enjoyed in the office of the commissioner of insurance immediately before the transfer. Notwithstanding section 230.28 (4) of the statutes, no employee transferred under paragraph (b) who has attained permanent status in class is required to serve a probationary period.
- (d) Tangible personal property. On the effective date of this paragraph, all tangible personal property, including records, of the office of the commissioner of insurance that is primarily related to information technology functions, as determined by the secretary of administration, is transferred to the division of enterprise technology in the department of administration.
- (e) Contracts. All contracts entered into by the office of the commissioner of insurance in effect on the effective date of this paragraph that are primarily related to information technology functions, as determined by the secretary of administration, remain in effect and are transferred to the division of enterprise technology in the department of administration. The division of enterprise technology in the department of administration shall carry out any obligations under those contracts unless modified or rescinded by the division of enterprise technology in the department of administration to the extent allowed under the contract.
- (f) Rules and orders. All rules promulgated by the office of the commissioner of insurance in effect on the effective date of this paragraph that are primarily related to information technology functions remain in effect until their specified expiration dates or until amended or repealed by the division of enterprise technology in the department of administration. All orders issued by the office of the commissioner of insurance in effect on the effective date of this paragraph that are primarily related to information technology functions remain in effect until their specified expiration dates or until modified or rescinded by the division of enterprise technology in the department of administration.
- (g) *Pending matters*. Any matter pending with the office of the commissioner of insurance on the effective date of this paragraph that is primarily related to information technology functions, as determined by the secretary of administration, is transferred to the division of enter-

prise technology in the department of administration. All materials submitted to or actions taken by the office of the commissioner of insurance with respect to the pending matter are considered as having been submitted to or taken by the division of enterprise technology in the department of administration.

SECTION 9125. Nonstatutory provisions; Investment Board.

SECTION 9126. Nonstatutory provisions; Joint Committee on Finance.

SECTION 9127. Nonstatutory provisions; Judicial Commission.

SECTION 9128. Nonstatutory provisions; Justice.

- (1q) Drug Law enforcement activities. Of the moneys in the appropriation account under section 20.455 (3) (g) of the statutes that have been received from a settlement and that are not committed to a specific purpose under the terms of the settlement, the department of justice shall allocate \$500,000 in the 2017–19 fiscal biennium to the division of criminal investigation to support overtime, fleet costs, training, and other supplies and services related to drug law enforcement.
 - (1w) Interoperable communications transfer.
- (a) Assets and liabilities. On the effective date of this paragraph, the assets and liabilities of the department of justice that are primarily related to interoperable communications, as determined by the secretary of administration, shall become the assets and liabilities of the department of military affairs.
- (b) Positions and employees. On the effective date of this paragraph, all positions and all incumbent employees holding those positions in the department of justice performing duties that are primarily related to interoperable communications, as determined by the secretary of administration, are transferred to the department of military affairs.
- (c) *Employee status*. Employees transferred under paragraph (b) have all the rights and the same status under chapter 230 of the statutes in the department of military affairs that they enjoyed in the department of justice immediately before the transfer. Notwithstanding section 230.28 (4) of the statutes, no employee so transferred who has attained permanent status in class is required to serve a probationary period.
- (d) *Tangible personal property*. On the effective date of this paragraph, all tangible personal property, including records, of the department of justice that is primarily related to interoperable communications, as determined by the secretary of administration, is transferred to the department military affairs.
- (e) *Pending matters*. Any matter pending with the department of justice on the effective date of this paragraph that is primarily related to interoperable communications, as determined by the secretary of administration, is transferred to the department of military affairs. All materials submitted to or actions taken by the department

- of justice with respect to the pending matter are considered as having been submitted to or taken by the department of military affairs.
- (f) Contracts. All contracts entered into by the department of justice in effect on the effective date of this paragraph that are primarily related to interoperable communications, as determined by the secretary of administration, remain in effect and are transferred to the department of military affairs. The department of military affairs shall carry out any obligations under those contracts unless modified or rescinded by the department of military affairs to the extent allowed under the contract.
- (g) Rules and orders. All rules promulgated by the department of justice in effect on the effective date of this paragraph that are primarily related to interoperable communications, as determined by the secretary of administration, remain in effect until their specified expiration dates or until amended or repealed by the department of military affairs. All orders issued by the department of justice in effect on the effective date of this paragraph that are primarily related to interoperable communications, as determined by the secretary of administration, remain in effect until their specified expiration dates or until modified or rescinded by the department of military affairs.
- (2p) STUDY OF THE SALE OR TRANSFER OF CRIME LABORATORY SUPPLIES. By January 5, 2018, the department of justice shall submit to the joint committee on finance a study of the implications of a sale or transfer of ammunition in the possession of the state crime laboratories to state and local law enforcement agencies for training purposes. The study shall address the manner in which other states dispose of ammunition in their crime laboratories and suggest ways in which Wisconsin may dispose of ammunition in the state crime laboratories.
- (3s) Funding for division of Criminal investigation overtime.
- (a) In the schedule under section 20.005 (3) of the statutes for the appropriation to the department of justice under section 20.455 (2) (a) of the statutes, the dollar amount for fiscal year 2017–18 is increased by \$558,900 for the purpose of funding division of criminal investigation overtime. In the schedule under section 20.005 (3) of the statutes for the appropriation to the department of justice under section 20.455 (2) (a) of the statutes, the dollar amount for fiscal year 2018–19 is increased by \$558,900 for the purpose of funding division of criminal investigation overtime.
- (b) Notwithstanding section 16.42 (1) (e) of the statutes, in submitting information under section 16.42 of the statutes for purposes of the 2019–21 biennial budget bill, the department of justice shall submit information concerning the appropriation under section 20.455 (2) (a) of the statutes as though the increase in that appropriation under paragraph (a) had not been made.

SECTION 9129. Nonstatutory provisions; Legislature.

- (2i) WISCONSINEYE EQUIPMENT.
- (a) Subject to paragraph (b), during the 2017–19 fiscal biennium, all of the following shall occur:
- 1. The assembly committee on organization shall authorize the expenditure of \$206,300 from the appropriation account under section 20.765 (1) (a) of the statutes and the senate committee on organization shall authorize the expenditure of \$206,300 from the appropriation account under section 20.765 (1) (b) of the statutes to assist in paying the costs of 21 WisconsinEye cameras in the state capitol building.
- 2. The chief justice of the supreme court shall authorize the expenditure of \$37,800 from the appropriation account under section 20.680 (1) (a) of the statutes to assist in paying the costs of 2 WisconsinEye cameras in the supreme court.
- (b) No moneys may be expended under paragraph (a) until all of the following have occurred:
- 1. The joint committee on legislative organization determines that WisconsinEye has secured \$226,200 in matching funds.
- 2. The joint committee on legislative organization has selected the location for the 21 WisconsinEye cameras in the state capitol building, and the chief justice of the supreme court has selected the location for the 2 WisconsinEye cameras in the supreme court.
- 3. The state capitol and executive residence board has approved the location and installation of all of the WisconsinEye cameras.

Vetoed In Part

- (2w) AUDIT OF STATE GROUP HEALTH INSURANCE PROGRAMS. The joint legislative audit committee is requested to direct the legislative audit bureau to perform a financial and performance evaluation audit of the state group health insurance programs, including a review of the group insurance board's compliance with its reserves policy, a review of the appropriateness of the group insurance board's policy regarding fully insured program reserves, and the circumstances that have created ongoing, frequent accumulation and use of reserves. If the joint legislative audit committee directs the legislative audit bureau to perform an audit, the legislative audit bureau shall file its report as described under section 13.94 (1) (b) of the statutes.
- (3s) SUPPLEMENTAL FUNDING FOR THE DEPARTMENT OF JUSTICE. In the schedule under section 20.005 (3) of the statutes for the appropriation to the joint committee on finance under section 20.865 (4) (a) of the statutes, the dollar amount for fiscal year 2017–18 is decreased by \$558,900 for the purpose of funding division of criminal investigation overtime. In the schedule under section 20.005 (3) of the statutes for the appropriation to the joint committee on finance under section 20.865 (4) (a) of the statutes, the dollar amount for fiscal year 2018–19 is

decreased by \$558,900 for the purpose of funding division of criminal investigation overtime.

SECTION 9130. Nonstatutory provisions; Lieutenant Governor.

SECTION 9131. Nonstatutory provisions; Local Government.

SECTION 9132. Nonstatutory provisions; Military Affairs.

- (1) Mobile field force grants. Notwithstanding section 16.42 (1) (e) of the statutes, in submitting information under section 16.42 of the statutes for purposes of the 2019–21 biennial budget bill, the department of military affairs shall submit information concerning the appropriation under section 20.465 (3) (dm) of the statutes as though that appropriation had not been made.
- (1w) Initial terms of 911 Subcommittee members. Notwithstanding section 15.315 (2) (a) of the statutes, as created by this act:
- (a) Of the initial members appointed to the 911 sub-committee by the governor, the terms of 6 of the members expire on July 1, 2019, the terms of 6 of the members expire on July 1, 2020, and the terms of 6 of the members expire on July 1, 2021.
- (b) The term of the initial member appointed to the 911 subcommittee by the adjutant general expires on July 1, 2021.
- (1x) INTEROPERABILITY REPORT. No later than January 1, 2019, the department of military affairs, in consultation with the interoperability council and the 911 subcommittee, shall submit in the manner provided under section 13.172 (2) of the statutes a report to the legislature and governor that includes all of the following:
- (a) Recommendations for changing the statutory authority of the interoperability council.
- (b) A description of the progress made toward creating a statewide public safety interoperable communication system.
- (c) A description of the obstacles that hinder progress toward interoperability.
- (d) Recommendations for legislative or executive action to promote interoperability.
- (1y) Consultation. In fiscal year 2017–18, the department of military affairs shall do all of the following:
- (a) Consult with the department of justice on how to effectively transfer from the department of justice to the department of military affairs resources and responsibilities relating to the interoperability council and the statewide public safety interoperable communication system.
- (b) Consult with the department of transportation and other state agencies on the effective use of staff at the department of transportation and other state agencies for the ongoing maintenance of a statewide public safety interoperable communication system.

(1yy) Transfer request. In fiscal year 2017–18, the department of military affairs may request the joint committee on finance to take action under section 13.10 of the statutes to transfer funding and positions from the department of transportation to the department of military affairs for the ongoing maintenance of a statewide public safety interoperable communication system.

SECTION 9133. Nonstatutory provisions; Natural Resources.

- (1) ENVIRONMENTAL FUND. Of the moneys in the environmental fund that are received for the purpose of environmental management, \$3,152,500 shall, in each fiscal year of the 2017-19 fiscal biennium, be considered to have been received for the purposes under section 20.370 (7) (cr) and (cs) of the statutes.
- (2) RELOCATION OF CHIEF STATE FORESTER AND DIVI-SION OF FORESTRY EMPLOYEES.
- (a) The department of natural resources shall relocate the headquarters for the chief state forester to an existing department facility north of STH 29 no later than January
- (b) The department of natural resources may allow individuals who are employed by the division of forestry on the effective date of this paragraph and located in the department office at 101 South Webster Street in the city of Madison to relocate to existing state-owned or stateleased facilities north of STH 29 designated by the chief state forester.
- (c) For the 2017–18 and 2018–19 fiscal years, for each employee who relocates under paragraph (b), the department of natural resources shall authorize payment of the moving expenses described under section 20.917 (1) (a) of the statutes.
- (d) By February 1, 2019, the department of natural resources shall submit a report to the governor and the cochairpersons of the joint committee on finance on the number of employees who have relocated under paragraph (b).
- (5) WISCONSIN NATURAL RESOURCES MAGAZINE EDI-TOR. The director of communications of the department of natural resources designated as a division administrator under section 230.08 (2) (e) 8. of the statutes shall serve as the editor of the Wisconsin Natural Resources Magazine.
- (6c) ALL-TERRAIN VEHICLES. If the department of natural resources maintains an administrative account that allocates moneys appropriated from the conservation fund to programs relating to all-terrain vehicles, the department of natural resources shall treat the appropriation under section 20.370 (5) (cx) of the statutes as an allocation from the administrative all-terrain vehicle
- (6f) SNOWMOBILE TRAIL STUDY. The snowmobile recreational council shall study options for ensuring adequate funding for development and maintenance of snowmobile trails in the state, especially currently

unfunded trails, and shall report its findings, conclusions, and recommendations to the joint committee on finance by January 1, 2019.

- (6p) Payments to counties and municipalities. The department of natural resources shall determine whether the amount payable to each county and municipality in 2018 under section 77.84 (2) of the statutes is less than the amount received by the respective counties and municipalities in 2015 under sections 77.07 and 77.87, 2013 stats. For each county and municipality for which the department of natural resources determines that the amount payable in 2018 is at least \$2,000 less than the amount received in 2015, the department of natural resources shall pay to each of those counties and municipalities an amount equal to the difference between the amount payable in 2018 and the amount received in 2015. The department of natural resources shall pay the amounts required under this subsection not later than March 1, 2018.
- (6q) Audit of moneys received for forestry ACTIVITIES. The joint legislative audit committee is requested to direct the legislative audit bureau to perform an audit of the moneys received by the department of natural resources for forestry activities and how those moneys are spent. If the committee directs the legislative audit bureau to perform an audit, the bureau shall file its report as described under section 13.94 (1) (b) of the statutes on or before June 30, 2019.
- (6r) COUNCIL ON FORESTRY REPORT. The governor's Vetoed council on forestry shall determine the relative priority of In Part expenditures from the department of natural resources administrative account that allocates moneys appropriated from the conservation fund to programs relating to forestry. The governor's council on forestry shall report its determinations and recommendations for the 2019-21 biennial budget to the governor, the department of natural resources, and the senate and assembly standing committees having jurisdiction over forestry matters no later than July 1, 2018.
- (6u) RECREATIONAL PASSPORT. The department of natural resources and the department of transportation shall jointly develop a plan to authorize the purchase of a recreational passport when an individual initiates or renews his or her annual vehicle registration. In the plan, the departments shall provide in detail the costs of implementing the plan, a timeline for implementing the plan, and the estimated revenue to be collected when the plan is fully implemented. One option that the departments shall evaluate in the plan is the creation of the recreational passport as a special license plate registration sticker that shows both the vehicle registration expiration year and an indicator that the sticker is also a recreational passport. The departments shall complete the plan in time for the plan to be included in the department of natural resources' 2019-21 biennial budget request.

Vetoed In Part Vetoed In Part

(7p) TAINTER LAKE BIOMANIPULATION PILOT. In the 2017–18 fiscal year, from the appropriation under section 20.370 (4) (mr) of the statutes, the department of natural resources shall expend not less than \$65,000 to conduct a pilot project to improve the water quality and fish habitat of Tainter Lake in Dunn County. The project shall include a comprehensive fish study, the removal of zooplanktivorous and benthivorous fish, and the introduction of piscivorous game fish.

- (7w) VILLAGE OF PLOVER GRANT. During the 2017-19 fiscal biennium, the department of natural resources shall award a grant of \$100,000 from the appropriation under section 20.370 (6) (gs) of the statutes to the Village of Plover for the purpose of employing an engineering firm and other experts qualified in wetland restoration and watershed enhancement to design solutions to increase the flow of the Little Plover River.
- (8t) Burlington storm water study. 2017–18 fiscal year, from the appropriation under section 20.370 (6) (dq) of the statutes, the department of natural resources shall provide a grant of \$50,000 to the Southeastern Wisconsin Regional Planning Commission to conduct a storm water management study for the city of Burlington. The study shall identify the location and destination of current storm water flows, identify city and private property owners that have storm water capacity issues, and provide recommendations for eliminating flooding at the city's major businesses.

SECTION 9134. Nonstatutory provisions; Public Defender Board.

SECTION 9135. Nonstatutory provisions; Public Instruction.

- (1p) IMPLEMENTATION OF EARLY COLLEGE CREDIT PRO-GRAM FOR THE FALL SEMESTER OF THE 2018-19 SCHOOL YEAR.
 - (a) In this subsection:
- 1. "Early college credit program" means the program under section 118.55 of the statutes, as affected by this
- 2. "Institution of higher education" means all of the following:
- a. An institution within the University of Wisconsin System or a tribally controlled college.
- b. A private, nonprofit institution of higher education located in this state.
- 3. "Participating private school" means a private school attended by a pupil who is enrolled in any of the grades 8 to 12.
- (b) Beginning on the effective date of this paragraph, the department of public instruction shall take whatever steps are necessary to implement the early college credit program under this act, including providing information to a school board, a participating private school, an institution of higher education, and a high school pupil who may be interested in enrolling in an institution of higher education in the fall semester of the 2018-19 school year

for the purpose of taking one or more nonsectarian courses at the institution of higher education under the early college credit program.

- (1t) VIRTUAL CHARTER SCHOOL FUNDING STUDY. The Vetoed department of public instruction shall, no later than Janu- In Part ary 1, 2019, prepare a report that compares the amount spent by the state for each pupil attending a virtual charter school under the program under section 118.51 of the statutes to the actual cost incurred by the virtual charter school to provide instruction to each such pupil. The department shall submit the report required under this subsection to the joint committee on finance and to the appropriate standing committees of the legislature under section 13.172 (3) of the statutes.
- (4f) SCHOOL-BASED MENTAL HEALTH SERVICES GRANT PROGRAM; ADVISORY COMMITTEE.
- (a) The state superintendent of public instruction shall establish an advisory committee under sections 15.04 (1) (c) and 227.13 of the statutes to make recommendations to the department of public instruction about the criteria the department is required to establish by rule under section 115.367 (2) of the statutes. The state superintendent of public instruction shall include on the committee established under this paragraph at least all of the following:
 - 1. A current or retired school administrator.
- 2. An individual who holds a license to teach issued by the department of public instruction or a license in a pupil services category under section PI 34.31 of the Wisconsin Administrative Code issued by the department of public instruction.
- 3. A provider of mental health services or a representative of an association that represents mental health service providers.
- 4. A family member of a pupil who is receiving or who may receive mental health services.
- 5. A representative of a school board or a charter school established under under section 118.40 (2r) or (2x) of the statutes.
- (b) The advisory committee established under paragraph (a) terminates upon the publication of the permanent rules required to be promulgated by the department under section 115.367 of the statutes, unless the state superintendent of public instruction elects to maintain the committee established under paragraph (a) after the permanent rules are published.
- (c) The department of public instruction may promulgate emergency rules under section 227.24 of the statutes to implement section 115.367 of the statutes, including establishing criteria under section 115.367 (2) of the statutes. Notwithstanding section 227.24 (1) (e) 1d. of the statutes, the department of public instruction is not required to prepare a statement of the scope if emergency rules are promulgated under this paragraph. Notwithstanding section 227.24 (1) (c) and (2) of the statutes, emergency rules promulgated under this paragraph

Vetoed In Part Vetoed In Part remain in effect until July 1, 2019, or the date on which permanent rules take effect, whichever is sooner. Not-withstanding section 227.24 (1) (a) and (3) of the statutes, the department is not required to provide evidence that promulgating a rule under this paragraph as an emergency rule is necessary for the preservation of the public peace, health, safety, or welfare and is not required to provide a finding of emergency for a rule promulgated under this paragraph.

Vetoed In Part

- (4p) Whole grade sharing agreement; aid. By February 1, 2019, the department of public instruction shall submit a report to the joint committee on finance that includes all of the following:
- (a) The number of school boards that applied for aid under section 118.50 (5m) of the statutes for the 2018–19 school year.
- (b) The number of school boards approved to receive aid under section 118.50 (5m) of the statutes for the 2018–19 school year.
- (c) For each school board approved to receive aid under section 118.50 (5m) of the statutes for the 2018–19 school year, all of the following:
 - 1. The name of the school board.
- 2. The number of grade levels that are subject to the whole grade sharing agreement.
- 3. The specific grade levels that are subject to the whole grade sharing agreement.
- 4. As of January 1, 2019, how much of the aid the school board is entitled to receive under section 118.50 (5m) of the statutes during the 2018–19 school year has been encumbered and how much has been expended.
- (4w) Reorganization of a low performing school district; study.
- (a) *Definition*. In this subsection, "eligible school district" means a school district that satisfied all of the following in the 2015–16 and 2016–17 school years:
- 1. The school district was assigned to the lowest performance category on the accountability reports published under section 115.385 (1) of the statutes.
- 2. The school district received intradistrict transfer aid under section 121.85 (6) (a) of the statutes.
- (b) *Contract*. The department of public instruction shall contract with an entity to study the effect of reorganizing an eligible school district to create one or more new school districts that consists of one or more villages located within the eligible school district and to report the results of the study to the department of public instruction by no later than 120 days after the contract is awarded to the entity. The contract shall require the entity to evaluate at least all of the following:
- 1. The estimated general and categorical school aid that the eligible school district and new school district or districts would be eligible to receive following the reorganization.

- 2. The impact the reorganization would have on the amount of property taxes paid by residents of the eligible school district and the new school district or districts.
- 3. An inventory of school buildings located in the eligible school district.
- 4. The assets and liabilities of the eligible school district.
- (c) Request for proposal. By no later than 30 days after the effective date of this paragraph, the department of public instruction shall issue a request for proposals for the study and report under paragraph (b).
- (d) Distribution of the report. The department of public instruction shall distribute the report submitted to the department under paragraph (b) to the village board of each village located in an eligible school district and to the school board of an eligible school district.
 - (5p) TEACHER LICENSURE; RULES.
- (a) The department of public instruction shall promulgate rules to revise chapter PI 34 of the administrative code. In promulgating rules under this paragraph, the department of public instruction shall simplify the teacher licensure system, to the extent practicable, by doing at least all of the following:
- 1. Simplifying the grade levels that a licensee is authorized to teach under his or her license.
 - 2. Creating broad field subject licenses.
- 3. Allowing school boards to increase the number of teachers in a school district by offering internships and residency opportunities.
- 4. Creating a permit that authorizes an individual who is enrolled in a teacher preparatory program to teach in public schools as part of an internship, residency program, or other equivalent training program.
- 5. Simplifying licensure reciprocity for individuals who hold a license in another state.
- 6. Expanding pathways for individuals who hold a license issued by the department to obtain additional licenses to fill positions in geographic areas and subject areas that are in need of educational personnel.
- (b) The department of public instruction may not promulgate rules required under paragraph (a) in a manner that decreases the quality standards for obtaining a license to teach from the department.
- (c) The department of public instruction shall submit in proposed form the rules required under paragraph (a) to the legislative council staff under section 227.15 (1) of the statutes no later than January 1, 2018.

SECTION 9136. Nonstatutory provisions; Public Lands, Board of Commissioners of.

SECTION 9137. Nonstatutory provisions; Public Service Commission.

SECTION 9138. Nonstatutory provisions; Revenue.

(1) Forestation state tax. For the property tax assessments as of January 1, 2017, the department of rev-

enue shall prescribe a form for the property tax bills prepared under section 74.09 of the statutes that indicates that the state no longer imposes the forestation state tax. The form shall also indicate the amount of the forestation state tax that the taxpayer paid in the previous year.

(1f) SALES AND USE TAX AUDITS; STATISTICAL SAM-PLING. Using the procedure under section 227.24 of the statutes, the department of revenue shall promulgate the rules required under section 77.59 (2g) of the statutes for the period before the effective date of the permanent rule promulgated under section 77.59 (2g) of the statutes but not to exceed the period authorized under section 227.24 (1) (c) of the statutes, subject to extension under section 227.24 (2) of the statutes. Notwithstanding section 227.24 (1) (a), (2) (b), and (3) of the statutes, the department is not required to provide evidence that promulgating a rule under this subsection as an emergency rule is necessary for the preservation of the public peace, health, safety, or welfare and is not required to provide a finding of emergency for a rule promulgated under this subsec-

SECTION 9139. Nonstatutory provisions; Safety and Professional Services.

(12) Professional assistance procedures; emer-GENCY RULES. The department may use the procedure under section 227.24 of the statutes to promulgate rules under section 440.03 (1c) of the statutes for the period before the effective date of the permanent rule promulgated under section 440.03 (1c) of the statutes but not to exceed the period authorized under section 227.24 (1) (c) of the statutes, subject to one extension of 60 days under section 227.24 (2) of the statutes. If the department uses this procedure to promulgate these rules, the department shall promulgate the rules no later than the 60th day after the effective date of this subsection. Notwithstanding section 227.24 (1) (a), (2) (b), and (3) of the statutes, the department is not required to provide evidence that promulgating a rule under this subsection as an emergency rule is necessary for the preservation of the public peace, health, safety, or welfare and is not required to provide a finding of emergency for a rule promulgated under this subsection.

- (17w) OCCUPATIONAL LICENSE STUDY.
- (a) Definitions. In this subsection:
- 1. "Department" means the department of safety and professional services.
- 2. "Occupational license" means any of the following:
- a. A license, permit, certification, registration, or other approval granted under section 167.10 (6m) or chapters 101, 145, or 440 to 480 of the statutes.
- b. A license, permit, certification, registration, or other approval not included under subdivision 2. a. if granted to a person by this state in order that the person may engage in a profession, occupation, or trade in this

state or in order that the person may use one or more titles in association with his or her profession, occupation, or

- (b) Report. No later than December 31, 2018, the department shall submit a report to the governor and the chief clerk of each house of the legislature for distribution to the legislature under section 13.172 (2) of the statutes. The report shall include the department's recommendations for the elimination of occupational licenses based on all of the following:
- 1. The department's evaluation of whether the unregulated practice of the profession, occupation, or trade can clearly harm or endanger the health, safety, or welfare of the public, and whether the potential for the harm is recognizable and not remote or speculative.
- 2. The department's evaluation of whether the public reasonably benefits from the occupational license requirement.
- 3. The department's evaluation of whether the public can be effectively protected by any means other than requiring an occupational license.
- 4. The department's analysis of whether licensure requirements for the regulated profession, occupation, or trade exist in other states.
- 5. The department's estimate of the number of individuals or entities that are affected by the occupational license requirement.
- 6. The department's estimate of the total financial burden imposed on individuals or entities as a result of the occupational licensure requirement, including education or training costs, examination fees, private credential fees, occupational license fees imposed by the state, and other costs individuals or entities incur in order to obtain the required occupational license.
- 7. Any statement or analysis provided by the agency or board administering the occupational license.
- 8. The department's evaluation of the tangible or intangible barriers people may face in obtaining an occupational license.

SECTION 9140. Nonstatutory provisions; Secretary of State.

SECTION 9141. Nonstatutory provisions; State Fair Park Board.

SECTION 9142. Nonstatutory provisions; Supreme Court

- (5f) DECISIONS OF LABOR AND INDUSTRY REVIEW COM- Vetoed MISSION. The chief justice of the supreme court is In Part requested to do all of the following:
- (a) Conduct a survey of decisions and orders of the labor and industry review commission under chapters 102 and 108 and sections 106.52 (4), 106.56 (4), and 111.39 of the statutes, citing the statutes interpreted by the commission and whether the decisions and orders were the subjects of actions for judicial review filed in circuit court.

(b) Submit a report of the survey's findings to the governor and to the joint committee on finance by July 1,

SECTION 9143. Nonstatutory provisions; Technical College System.

SECTION 9144. Nonstatutory provisions; Tourism.

- (1) FINANCIAL MANAGEMENT POSITION TRANSFER.
- (a) Employee transfer. On the effective date of this paragraph, 1.0 FTE position and the incumbent employee holding the position in the department of tourism who performs duties relating to financial management, as determined by the secretary of administration, is transferred to the department of administration.
- (b) Employee status. The employee transferred under paragraph (a) has all the rights and the same status under chapter 230 of the statutes in the department of administration that he or she enjoyed in the department of tourism immediately before the transfer. Notwithstanding section 230.28 (4) of the statutes, no employee so transferred who has attained permanent status in class is required to serve a probationary period.
- (2t) Marketing Campaign for the city of Pitts-VILLE. From the appropriation under section 20.380 (1) (b) of the statutes, in fiscal year 2017–18, the department of tourism shall award a grant of \$7,500 to the city of Pittsville for, as determined by the city, signage and a landmark to market the city of Pittsville as the geographical center of Wisconsin.
- (3t) IMPROVEMENT OF PROPERTY USED FOR SKI JUMPING COMPETITIONS.
- (a) From the appropriation under section 20.380 (1) (b) of the statutes, in fiscal year 2017–18, the department of tourism shall award a grant of \$75,000 to Vernon County for the purpose of improving structures and other property in Vernon County that are used to facilitate a national or international ski jumping competition.
- (b) Vernon County shall distribute all of the grant moneys under paragraph (a) within 2 years after its receipt of the grant moneys.
- (c) Each person receiving a distribution of grant moneys from Vernon County under paragraph (b) shall contribute to the improvements specified under paragraph (a) an amount from nonstate revenue sources that is equal to at least 33 percent of the amount of the distribution the person receives.

SECTION 9145. Nonstatutory provisions; Transportation.

In Part

Vetoed

(1f) Transportation engineering and construc-

Vetoed TION STUDY. The department of transportation in consultation with the transportation projects commission shall enter into an agreement with an independent engineering firm that has not previously conducted business with the state for the preparation, and delivery to the department and commission, of a report by no later than January 1, In Part 2019, that does all of the following:

- (a) Reviews the standards of all other states related to transportation engineering and highway construction and recommends any best practices.
- (b) Analyzes the process the department of transportation uses for determining project priority and assesses whether the process uses reasonable financing and completion time assumptions.
- (c) Evaluates the allocation of funds to the state highway rehabilitation, major highway development, and southeast Wisconsin freeway megaprojects programs.
 - (1t) Elimination of Positions.
- (a) The department of transportation shall eliminate 100 SEG positions in fiscal year 2017-18 and 100 SEG positions in fiscal year 2018–19.
- (b) The department of transportation shall eliminate private management consultant positions in the local assistance program as necessary to accomplish the lapses required under SECTION 9245 (2t) of this act.
- (c) No later than January 1, 2019, the department of transportation shall submit a report to the joint committee on finance identifying the positions eliminated under paragraphs (a) and (b) and the appropriation accounts to be reduced.
- (2f) Transportation projects commission funding. Vetoed Not later than March 1, 2018, the transportation projects In Part commission shall submit a request to the joint committee on finance for not more than an additional 4.0 GPRfunded positions. If the cochairpersons of the committee do not notify the commission within 14 working days after the submittal that the committee has scheduled a meeting for the purpose of reviewing the request, the commission may expend the funds. If, within 14 working days after the submittal, the cochairpersons of the committee notify the commission that the committee has scheduled a meeting for the purpose of reviewing the request, the commission may expend the funds only as approved by the committee.
- (2i) CONSTRUCTION WORK ON STH 23. If the cost of a construction contract related to a project enumerated under section 84.013 (3) (ra) of the statutes is lower than the cost projected by the department of transportation for that project, the department shall reserve the first \$19,400,000 of all such contract savings from all sources, including state and federal moneys and the proceeds of bonds, for construction work on STH 23 in the 2017-19 biennium. If the department of transportation does not expend these moneys by January 1, 2019, the department may expend the moneys on other major highway projects.
- (2j) REHABILITATION WORK ON STH 23. The department of transportation shall conduct rehabilitation work on STH 23 in Sheboygan County and Fond du Lac County in the 2017-19 biennium.
 - (3i) AIRPORT IMPROVEMENT PROJECT FUNDING.

- (a) Notwithstanding section 114.34 of the statutes, in fiscal year 2017–18, from the appropriation under section 20.395 (2) (dq) of the statutes, the department of transportation shall award a grant of \$4,000,000 to the city of Wisconsin Rapids for improvements to the Alexander Field airport.
- (b) Notwithstanding section 114.34 of the statutes, in fiscal year 2017–18, from the appropriation under section 20.395 (2) (dq) of the statutes, the department of transportation shall award a grant of \$1,700,000 to the Appleton International Airport for design services and construction related to making improvements necessary for the airport to be an airport rescue and firefighting facility.
- (4c) PESHTIGO FIRE MUSEUM DIRECTIONAL SIGNS. Notwithstanding any eligibility criteria or other criteria or specification under section 86.196 of the statutes, in fiscal year 2017–18, the department of transportation shall erect 2 tourist–oriented directional signs, one for each direction of travel, along USH 41 in Marinette County for the Peshtigo Fire Museum in the town of Peshtigo. The department may not charge a fee exceeding \$1,000 related to signs erected under this subsection, which fee shall be paid by the historical society from the appropriation account under section 20.245 (1) (a) of the statutes.
- (4d) FINCANTIERI BAY SHIPBUILDING HARBOR ASSIST-ANCE GRANT. In the 2017–18 fiscal year, from the appropriations under sections 20.395 (2) (cq) and 20.866 (2) (uv) of the statutes, notwithstanding the eligibility criteria under section 85.095 of the statutes, the department of transportation shall award a grant under section 85.095 (2) (a) of the statutes to Fincantieri Bay Shipbuilding for dredging and dockwall construction. The amount of the grant awarded under this subsection shall be \$3,200,000 or the total cost of the project, whichever is less.

Vetoed

In Part

Vetoed

In Part

Vetoed

In Part Vetoed

In Part

- $\begin{tabular}{ll} (4w) & {\hbox{\bf STUDY OF CONSOLIDATION OF SEGREGATED FUNDS}\\ \hbox{\bf IN LOCAL PROGRAM} \ . \end{tabular}$
- (a) The department of transportation shall study the effects of consolidating state moneys in the surface transportation program and replacing these funds with federal moneys from the state highway program and shall report its findings to the joint committee on finance no later than May 1, 2018.

(b) The department of transportation may submit a request to make transfers of state and federal moneys between the surface transportation program and state highway program to the joint committee on finance under section 13.10 of the statutes. A request made under this paragraph shall include an estimate of the potential savings or costs to local governments and the state that could be associated with the request.

(5f) DISASTER DAMAGE AID PAYMENT. Notwithstanding section 86.34 (1m) and (2) of the statutes, in the 2017–18 fiscal year, from the appropriation under section 20.395 (1) (fs) of the statutes, the department of trans-

portation shall make an aid payment of \$64,000 to the town of Lafayette in Chippewa County.

- (6b) TOLLING IMPLEMENTATION STUDY.
- (a) The department of transportation shall enter into a contract under which the department of transportation may expend not more than \$2,500,000 from the appropriation under section 20.395 (4) (aq) of the statutes for the purpose of the contractor conducting a tolling implementation study that includes all of the following:
- 1. An analysis to support the completion of the federal tolling application process.
- 2. A tolling concepts of operation plan that outlines the policies, procedures, and operations needed to govern roadway tolling.
- 3. A traffic and revenue analysis including the revenue needed to support toll revenue–supported debt.
- 4. An evaluation or reevaluation of federal environmental requirements, including required documentation.
- (b) No later than January 1, 2019, the contractor conducting the study under paragraph (a) shall report its findings to the department of transportation and the legislature under section 13.172 (2) of the statutes.
- (7m) REIMBURSEMENT TO VILLAGE OF RIB LAKE. In the 2017–18 fiscal year, from the appropriation under section 20.395 (2) (js) of the statutes, the department of transportation shall provide \$20,000 to the village of Rib Lake in Taylor County to reimburse the village for costs incurred in the design of a safe routes to school project.
- (8m) Traffic control signals in city of Green-FIELD. The department of transportation shall study the potential installation of traffic control signals at the intersection of Layton Avenue and 124th Street near the entrance and exit to I 43 in the city of Greenfield in Milwaukee County. The department of transportation shall report its findings to the senate and assembly standing committees having jurisdiction over transportation matters no later than June 30, 2018.

(10c) STATE HIGHWAY 154 REHABILITATION. In the 2017–19 fiscal biennium, the department of transportation shall complete the rehabilitation project on STH 154 in Sauk County between the village of Loganville and the Richland County–Sauk County border. The project shall include milling, overlay, and safety improvements to existing facilities.

(10d) Prohibiting funding of North Leg of Zoo Interchange Project. Notwithstanding section 84.0145 (3) (b) of the statutes, in the 2017–19 fiscal biennium, the department of transportation may not provide funding from any source, including state and federal moneys, the proceeds of bonds, or savings realized from project costs that are lower than the costs projected by the department of transportation, for the north leg of the Zoo interchange project between Swan Boulevard and Burleigh Street in Milwaukee County.

SECTION 9146. Nonstatutory provisions; Treasurer.

Vetoed

In Part

Vetoed

SECTION 9147. Nonstatutory provisions; University of Wisconsin Hospitals and Clinics Authority; Medical College of Wisconsin.

SECTION 9148. Nonstatutory provisions; University of Wisconsin System.

- (2) UNIVERSITY OF WISCONSIN FLEXIBLE OPTION PROGRAMS. The Board of Regents of the University of Wisconsin System shall ensure that, no later than December 1, 2019, the total number of accredited competency—based degree and certificate programs offered under the University of Wisconsin Flexible Option platform is increased by at least 25 percent over the total number of such programs that are offered on the effective date of this subsection.
- (2p) TEACHER RESIDENCY PROGRAM. No later than December 1, 2019, the Board of Regents of the University of Wisconsin System shall ensure that at least one school of education within the University of Wisconsin System has partnered with a school district to develop a teacher residency program.
- (2q) Annual financial audit of the University of Wisconsin System.
 - (a) Definitions. In this subsection:
- 1. "Board" has the meaning given in section 36.05 (2) of the statutes.
- 2. "System" has the meaning given in section 36.05 (12) of the statutes.

Vetoed (b) *No financial audit by legislative audit bureau.*In Part Notwithstanding section 13.94 (1) (t) of the statutes, the Vetoed legislative audit bureau shall not conduct a financial audit In Part of the system for the 2017–18 and 2018–19 fiscal years.

- (c) Contract for financial audit. The board shall contract with an independent accounting firm licensed under chapter 442 of the statutes for purposes of conducting an annual financial audit of the system for fiscal year 2017–18 and fiscal year 2018–19. This accounting firm shall report to the board and shall provide all of the following to the board, the governor, the joint legislative audit committee, and the joint committee on finance:
 - 1. The audited financial statements.
 - 2. Performance improvement observations.
- 3. A management letter complete with internal control deficiencies and audit differences.
- (d) Legislative audit bureau assistance. The accounting firm with which the board contracts under paragraph (c) may use the legislative audit bureau to assist in conducting the audit to the extent the work relied upon does not modify the audit opinion with the exception of accepting the prior year's unqualified opinion.
- (3t) RESIDENT UNDERGRADUATE TUITION. Notwithstanding section 36.27 (1) (a) of the statutes, the Board of Regents of the University of Wisconsin System may not charge resident undergraduates enrolled in an institution or college campus in the 2017–18 or 2018–19 academic year more in academic fees than it charged resident

undergraduates enrolled in that institution or college campus in the 2016–17 academic year.

SECTION 9149. Nonstatutory provisions; Veterans Affairs.

- (1e) Crisis intervention services demonstration PROGRAM. From the appropriation under section 20.485 (2) (u) of the statutes, the department of veterans affairs shall allocate \$60,000 each fiscal year of the 2017–19 fiscal biennium for a crisis intervention services demonstration program administered by the department.
- (1f) VETERANS TRUST FUND REPORT. No later than January 1, 2018, the department of veterans affairs shall submit a report to the joint committee on finance that includes all of the following:
- (a) A description and analysis of the department's administrative costs supported by the veterans trust fund.
- (b) A description and analysis of the department's administrative costs supported by revenue generated from Wisconsin veterans homes.
- (c) The department's proposals for changes to the department's programs, administrative structure, or position levels and salaries to increase efficiency or lower administrative costs.
 - (d) The following proposals:
- 1. A proposed long-term plan to maintain the solvency of the veterans trust fund that includes the use of transfers from appropriations for Wisconsin veterans homes.
- 2. A proposed long—term plan to maintain the solvency of the veterans trust fund that does not include the use of transfers from appropriations for Wisconsin veterans homes.
- (1g) RESPONSE TO AUDIT. The department of veterans affairs shall do all of the following to implement the recommendations contained in the legislative audit bureau's Report 17–8 relating to the Wisconsin Veterans Home at King:
- (a) Promulgate rules amending chapter VA 6 of the Wisconsin Administrative Code to establish a formula for calculating private pay rates for nursing home and assisted living care at Wisconsin veterans homes and that clearly define rate-setting terms, including "costs of care" under section VA 6.01 (16) of the Wisconsin Administrative Code. The department shall present the statement of scope of the rules required under this paragraph to the governor for approval under section 227.135 (2) of the statutes no later than July 1, 2018.
- (b) No later than July 1, 2018, submit a report to the joint committee on finance and the joint legislative audit committee that includes all of the following:
- 1. The cash balance the department believes is appropriate to maintain in the appropriation account under section 20.485 (1) (gk) of the statutes.
- 2. A description of the department's effort to develop and routinely update a detailed plan for the management

and proposed use of the cash balance in the Wisconsin veterans home PR appropriation accounts.

- (c) No later than July 1, 2018, submit a report to the joint committee on finance and the joint legislative audit committee that includes all of the following:
- 1. A description of the department's efforts to establish a systematic process for comprehensively identifying and assessing the capital—related project needs of all Wisconsin veterans homes.
- 2. A description of the department's efforts to use the information gathered under the process described in sub-division 1. to complete a 10-year facilities plan for the Wisconsin veterans homes and to help develop the department's required 6-year facilities plans in the future.
- (2f) Study concerning women veterans. From the appropriation under section 20.485 (2) (u) of the statutes, the department of veterans affairs shall allocate \$20,000 in the 2017–19 fiscal biennium for the purpose of conducting a study concerning the needs of women veterans in Wisconsin.

SECTION 9150. Nonstatutory provisions; Wisconsin Economic Development Corporation.

- (1) GPR EXPENDITURE LIMITATION. Notwithstanding the cap on expenditures under section 20.192 (1) (a) of the statutes, no more than \$1,519,500 may be expended from that appropriation in fiscal year 2017–18.
- (3f) FABRICATION LABORATORY GRANT PROGRAM. The Wisconsin Economic Development Corporation shall develop and implement an economic development program that is similar to the fabrication laboratory grant program under section 238.145, 2015 stats. From the appropriation under section 20.192 (1) (a) or (r) of the statutes, the Wisconsin Economic Development Corporation shall allocate at least \$500,000 in each fiscal year of the 2017–19 fiscal biennium for the purpose of awarding grants under that economic development program.
 - (3i) Fabrication Laboratory Grants.
 - (a) Definitions. In this subsection:
- 1. "Fabrication laboratory" means a high-technology workshop equipped with computer-controlled additive and subtractive manufacturing components.
- 2. "Nonprofit organization" means an organization that is described in section 501 (c) (3) or (6) of the Internal Revenue Code and that is exempt from federal income tax.
- (b) *Grants*. From the appropriation under section 20.192 (1) (a) or (r) of the statutes, the Wisconsin Economic Development Corporation shall allocate \$100,000 in fiscal year 2017–18 and \$300,000 in fiscal year 2018–19 for the purpose of awarding grants to one or more nonprofit organizations for the provision of services to school districts in this state in which the school board has adopted a resolution to initiate a fabrication laboratory. The services provided under a grant awarded under this paragraph shall include the development of

curricula for fabrication laboratories, in–school coaching and ongoing professional development for district personnel, and other assistance related to fabrication laboratories, as determined by the Wisconsin Economic Development Corporation.

(3t) RECONCILIATION.

Vetoed In Part

- (a) If August 2017 Special Session Assembly Bill 1, as shown by Senate Substitute Amendment 1, or August 2017 Special Session Senate Bill 1, as shown by Senate Substitute Amendment 1, is enacted substantially without change, then the treatment of sections 20.835 (2) (cp), 71.07 (3wm), and 71.28 (3wm) of the statutes in this act supersedes the treatment in those bills as shown by those substitute amendments.
- (b) If August 2017 Special Session Assembly Bill 1, as shown by Senate Substitute Amendment 1, or August 2017 Special Session Senate Bill 1, as shown by Senate Substitute Amendment 1, is not enacted substantially without change, then the treatment of sections 20.835 (2) (cp), 71.07 (3wm), and 71.28 (3wm) of the statutes in this act and all cross–references to section 238.396 of the statutes in this act are void.
- (3w) Marathon County Development Corporation.
- (a) *Definition*. In this subsection, "minority-owned business" means a sole proprietorship, partnership, limited liability company, joint venture, or corporation that is at least 30 percent owned by a minority group member or members, as defined in section 16.287 (1) (f) of the statutes, who are U.S. citizens or persons lawfully admitted to the United States for permanent residence, as defined under 8 USC 1101 (a) (20).
- (b) Revolving loan fund grant. From the appropriation under section 20.192 (1) (a) or (r) of the statutes, the Wisconsin Economic Development Corporation shall grant \$100,000 to MCDEVCO, Inc., doing business as the Marathon County Development Corporation, in the 2017–18 fiscal year for a revolving loan program for minority—owned businesses in Marathon County.

SECTION 9151. Nonstatutory provisions; Workforce Development.

- (1) FAST FORWARD GRANTS FOR TECHNICAL COLLEGES. Of the amounts appropriated to the department of workforce development under section 20.445 (1) (b) of the statutes, the department shall allocate not less than \$5,000,000 in fiscal year 2017–18 for grants to technical colleges for workforce training programs under section 106.27 (1) of the statutes.
- (2) FAST FORWARD GRANTS FOR NURSING TRAINING PROGRAMS. Of the amounts appropriated to the department of workforce development under section 20.445 (1) (b) of the statutes, the department shall allocate not less than \$1,500,000 in the 2017–19 fiscal biennium for grants for nursing training programs under section 106.27 (1) (e) of the statutes.
 - (3) Worker's compensation position transfer.

- (a) *Employee transfer*. On the effective date of this paragraph, 4.5 FTE positions and the incumbent employees holding those positions in the department of workforce development who perform duties relating to worker's compensation hearings, as determined by the secretary of administration, are transferred to the department of administration.
- (b) *Employee status*. The employees transferred under paragraph (a) have all the rights and the same status under chapter 230 of the statutes in the department of administration that the employees enjoyed in the department of workforce development immediately before the transfer. Notwithstanding section 230.28 (4) of the statutes, no employee transferred under paragraph (a) who has attained permanent status in class is required to serve a probationary period.
- (5q) Grant to support the Building Occupational Skills for Success program. From the appropriation under section 20.445 (1) (b) of the statutes, the department of workforce development shall provide a grant of \$200,000 in fiscal year 2017–18 to the Milwaukee Development Corporation for the purpose of supporting the Building Occupational Skills for Success program that is designed to provide students with the skills and tools needed to become future business owners if all of the following are satisfied:
- (a) The Milwaukee Development Corporation or any other organization provides equal matching funds to support the Building Occupational Skills for Success program.
- (b) The Milwaukee Development Corporation agrees to make the Building Occupational Skills for Success program curriculum available, upon request, to any school board, operator of a charter school authorized under section 118.40 (2r) or (2x) of the statutes, private school, tribal school, or any nonprofit organization after the program has been implemented in Milwaukee Public Schools for one year.
- (c) The Milwaukee Development Corporation agrees to submit a report to the joint committee on finance that summarizes the results of the Building Occupational Skills for Success program by no later than 90 days after the program has been implemented in Milwaukee Public Schools for one year.
- (5y) TECHNICAL EDUCATION EQUIPMENT GRANTS; EMERGENCY RULES. Using the procedure under section 227.24 of the statues, the department of workforce development may promulgate the rules required under section 106.275 of the statutes for the period before the effective date of the permanent rules promulgated under section 106.275 of the statues but not to exceed the period authorized under section 227.24 (1) (c) of the statues, subject to extension under section 227.24 (2) of the statutes. Notwithstanding section 227.24 (1) (a), (2) (b), and (3) of the statutes, the department is not required to provide evidence that promulgating a rule under this subsection as an

emergency rule is necessary for the preservation of the public peace, health, safety, or welfare and is not required to provide a finding of emergency for a rule promulgated under this subsection.

SECTION 9152. Nonstatutory provisions; Other.

- (1) Study on public benefits and chronic absenteeism. The departments of children and families, public instruction, health services, and workforce development, together with any other relevant programs or agencies the departments identify as appropriate, shall collaborate to prepare a report on the population overlap of families that receive public benefits and children who are absent from school for 10 percent or more of the school year. The agencies shall submit the report on or before December 30, 2018, to the governor and appropriate standing committees of the legislature under section 13.172 (3) of the statutes.
- (1i) REQUIRED GENERAL FUND STRUCTURAL BALANCE. Section 20.003 (4m) of the statutes shall not apply to the action of the legislature in adopting any legislation during the 2017–18 legislative session.

SECTION 9201. Fiscal changes; Administration.

(1) DIVISION OF PERSONNEL MANAGEMENT LAPSE. Notwithstanding section 20.001 (3) (a) of the statutes, from the appropriation account to the department of administration under section 20.505 (1) (kz) of the statutes, there is lapsed to the general fund \$2,800,000 in fiscal year 2018–19.

SECTION 9202. Fiscal changes; Agriculture, Trade and Consumer Protection.

SECTION 9203. Fiscal changes; Arts Board.

SECTION 9204. Fiscal changes; Building Commission.

SECTION 9205. Fiscal changes; Child Abuse and Neglect Prevention Board.

SECTION 9206. Fiscal changes; Children and Families.

SECTION 9207. Fiscal changes; Circuit Courts.

SECTION 9208. Fiscal changes; Corrections.

SECTION 9209. Fiscal changes; Court of Appeals.

SECTION 9210. Fiscal changes; District Attorneys. SECTION 9211. Fiscal changes; Educational

SECTION 9211. Fiscal changes; Educationa Approval Board.

SECTION 9212. Fiscal changes; Educational Communications Board.

SECTION 9213. Fiscal changes; Elections Commission.

SECTION 9214. Fiscal changes; Employee Trust Funds.

SECTION 9215. Fiscal changes; Employment Relations Commission.

(1) UNSPENT PROGRAM REVENUE. Notwithstanding section 20.001 (3) (a) of the statutes, at the end of each fiscal year in the 2017–19 fiscal biennium, there is lapsed to the general fund any unencumbered balance exceeding 10 percent of that fiscal year's expenditures from the

appropriation account under section 20.425 (1) (i) of the statutes.

SECTION 9216. Fiscal changes; Ethics Commission.

SECTION 9217. Fiscal changes; Financial Institutions.

SECTION 9218. Fiscal changes; Governor. SECTION 9219. Fiscal changes; Health and Educational Facilities Authority.

SECTION 9220. Fiscal changes; Health Services.

- (1m) Graduate medical training support grants.
- (a) In the schedule under section 20.005 (3) of the statutes for the appropriation to the department of health services under section 20.435 (4) (b) of the statutes, the dollar amount for fiscal year 2017–18 is decreased by \$63,000 for grants to support graduate medical training under section 146.64 of the statutes. In the schedule under section 20.005 (3) of the statutes for the appropriation to the department of health services under section 20.435 (4) (b) of the statutes, the dollar amount for fiscal year 2018–19 is decreased by \$63,000 for grants to support graduate medical training under section 146.64 of the statutes.
- (b) In the schedule under section 20.005 (3) of the statutes for the appropriation to the department of health services under section 20.435 (4) (bf) of the statutes, the dollar amount for fiscal year 2017–18 is increased by \$63,000 for grants to support graduate medical training under section 146.64 of the statutes. In the schedule under section 20.005 (3) of the statutes for the appropriation to the department of health services under section 20.435 (4) (bf) of the statutes, the dollar amount for fiscal year 2018–19 is increased by \$63,000 for grants to support graduate medical training under section 146.64 of the statutes.
- (5p) GENERAL AIDS AND LOCAL ASSISTANCE. In the 2019–21 fiscal biennium, the department of health services may expend from the appropriation under section 20.435 (1) (b) of the statutes in addition to the amount in the schedule under section 20.005 (3) of the statutes, 20 percent of the moneys received from overpayments and incorrect payments as a result of the audit under SECTION 9120 (6p) for the purposes for which the appropriation under section 20.435 (1) (b) of the statutes is made.
- (5q) COMMUNITY AIDS AND MEDICAL ASSISTANCE PAYMENTS. In the 2019–21 fiscal biennium, the department of health services may expend from the appropriation under section 20.435 (7) (b) of the statutes in addition to the amount in the schedule under section 20.005 (3) of the statutes, 20 percent of the moneys received from overpayments and incorrect payments as a result of the audit under Section 9120 (6p) for the purposes for which the appropriation under section 20.435 (7) (b) of the statutes is made.
- (5r) Grants for community programs. In the 2019–21 fiscal biennium, the department of health ser-

vices may expend from the appropriation under section 20.435 (7) (bc) of the statutes in addition to the amount in the schedule under section 20.005 (3) of the statutes, 20 percent of the moneys received from overpayments and incorrect payments as a result of the audit under Section 9120 (6p) of this act for the purposes for which the appropriation under section 20.435 (7) (bc) of the statutes is made.

(5s) Services for Sex-trafficking victims. In the 2019–21 fiscal biennium, the department of children and families may expend from the appropriation under section 20.437 (1) (e) of the statutes in addition to the amount in the schedule under section 20.005 (3) of the statutes, 20 percent of the moneys received from overpayments and incorrect payments as a result of the audit under Section 9120 (6p) of this act for the purposes for which the appropriation under section 20.437 (1) (e) of the statutes is made.

SECTION 9221. Fiscal changes; Higher Educational Aids Board.

SECTION 9222. Fiscal changes; Historical Society. SECTION 9223. Fiscal changes; Housing and Economic Development Authority.

SECTION 9224. Fiscal changes; Insurance.

SECTION 9225. Fiscal changes; Investment Board. SECTION 9226. Fiscal changes; Joint Committee on Finance.

SECTION 9227. Fiscal changes; Judicial Commission.

SECTION 9228. Fiscal changes; Justice.

- (1p) Internet crimes against children. There is transferred from the appropriation account under section 20.455 (2) (Lp) of the statutes to the appropriation account under section 20.455 (2) (hd) of the statutes \$750,000 in each fiscal year in the 2017–19 fiscal biennium.
- (9p) Law enforcement overtime grants. Of the moneys in the appropriation account under section 20.455 (3) (g) of the statutes that have been received from a settlement and that are not committed to a specific purpose under the terms of the settlement, there is transferred \$2,000,000 in the 2017–18 fiscal year to the appropriation account under section 20.455 (2) (jc) of the statutes to provide grants under the law enforcement overtime grant program.
- (15t) ALTERNATIVES TO INCARCERATION GRANT PROGRAM. Of the moneys in the appropriation account under section 20.455 (3) (g) of the statutes that have been received from a settlement and that are not committed to a specific purpose under the terms of the settlement, there is transferred \$500,000 in the 2017–18 fiscal year to the appropriation account under section 20.455 (2) (jd) of the statutes to provide grants under the alternatives to incarceration grant program.
- (19p) WISCONSIN COURT APPOINTED SPECIAL ADVOCATE ASSOCIATION GRANTS. Of the moneys in the appro-

Board.

priation account under section 20.455 (3) (g) of the statutes that have been received from a settlement and that are not committed to a specific purpose under the terms of the settlement, there is transferred \$160,000 in the 2017–18 fiscal year to the appropriation account under section 20.455 (5) (kr) of the statutes to provide grants under section 165.967 of the statutes.

SECTION 9229. Fiscal changes; Legislature. SECTION 9230. Fiscal changes; Lieutenant Governor.

SECTION 9231. Fiscal changes; Local Government. SECTION 9232. Fiscal changes; Military Affairs. SECTION 9233. Fiscal changes; Natural Resources. SECTION 9234. Fiscal changes; Public Defender

SECTION 9235. Fiscal changes; Public Instruction. SECTION 9236. Fiscal changes; Public Lands, **Board of Commissioners of.**

SECTION 9237. Fiscal changes; Public Service Commission.

- (1) UNIVERSAL SERVICE FUND TRANSFER. There is transferred from the universal service fund to the appropriation account under section 20.155 (3) (r) of the statutes \$6,000,000 in the 2017-18 fiscal year.
 - (2) Federal E-rate transfer.
- (a) There is transferred from the appropriation account under section 20.505 (4) (mp) of the statutes to the appropriation account under section 20.155 (3) (r) of the statutes \$5,000,000 in the 2017-18 fiscal year.
- (b) There is transferred from the appropriation account under section 20.505 (4) (mp) of the statutes to the universal service fund \$7,500,000 in fiscal year 2017 - 18.

SECTION 9238. Fiscal changes; Revenue.

(1b) GENERAL PROGRAM OPERATIONS. Notwithstanding section 20.001 (3) (a) of the statutes, from the appropriation account to the department of revenue under section 20.566 (3) (a) of the statutes, there is lapsed to the general fund \$134,500 in fiscal year 2017-18 and \$80,600 in fiscal year 2018-19.

SECTION 9239. Fiscal changes; Safety and Professional Services.

SECTION 9240. Fiscal changes; Secretary of State. SECTION 9241. Fiscal changes; State Fair Park Board.

SECTION 9242. Fiscal changes; Supreme Court. SECTION 9243. Fiscal changes; Technical College System.

SECTION 9244. Fiscal changes; Tourism. SECTION 9245. Fiscal changes; Transportation.

(1) Transfer from Petroleum Inspection fund to TRANSPORTATION FUND. There is transferred from the petroleum inspection fund to the transportation fund \$24,000,000 in each fiscal year of the 2017-19 fiscal biennium.

- (2t) LAPSE OF MONEYS. Notwithstanding section 20.001 (3) (a) to (c) of the statutes, there is lapsed to the transportation fund from the unencumbered balances of appropriation accounts affected by changes to overtime management and consultant funding on local programs under Section 9145 (1t) of this act and from which the positions eliminated under Section 9145 (1t) of this act were funded \$13,000,000 in fiscal year 2017-18 and \$13,000,000 in fiscal year 2018-19.
- (3i) Lapse of moneys related to land sales. Notwithstanding section 20.001 (3) (c) of the statutes, from the appropriation account to the department of transportation under section 20.395 (4) (ew) of the statutes, there is lapsed to the transportation fund \$3,300,000 in fiscal year 2017–18 and \$700,000 in fiscal year 2018–19.

SECTION 9246. Fiscal changes; Treasurer.

SECTION 9247. Fiscal changes; University of Wisconsin Hospitals and Clinics Authority; Medical College of Wisconsin.

SECTION 9248. Fiscal changes; University of Wisconsin System.

Section 9249. Fiscal changes; Veterans Affairs. SECTION 9250. Fiscal changes; Wisconsin Economic Development Corporation.

SECTION 9251. Fiscal changes; Workforce Devel-

SECTION 9252. Fiscal changes; Other. SECTION 9301. Initial applicability; Administration.

- (2) DIVISION OF HEARINGS AND APPEALS; TRANSCRIPTS. The renumbering and amendment of section 227.55 of the statutes and the creation of section 227.55 (2) of the statutes first apply to petitions for review submitted under section 227.53 of the statutes on the effective date of this subsection.
- (2f) Cost-benefit analysis of lease and purchase Vetoed OPTIONS. The renumbering and amendment of section In Part 16.84 (5) of the statutes and the creation of section 16.84 (5) (b) of the statutes first apply to leases entered into, renewed, or extended on the effective date of this subsection.
- (6s) SERVICE AWARD PROGRAM PAYMENTS. The treatment of section 16.25 (3) (d) 1. of the statutes first applies to a payment from the appropriation under section 20.505 (4) (er) of the statutes that relates to a municipal contribution paid in 2017.
- (6w) CONDEMNATION LITIGATION EXPENSES. The treatment of section 32.28 (3) (d), (e), (f), (g), (h), and (i) and (4) of the statutes, the renumbering and amendment of section 32.28 (1) of the statutes, and the creation of section 32.28 (1) (a) of the statutes first apply to written offers and jurisdictional offers made on the effective date of this subsection.

SECTION 9302. Initial applicability; Agriculture, Trade and Consumer Protection.

- (1) PESTICIDE, FERTILIZER, AND COMMERCIAL FEED LICENSING.
- (a) The treatment of section 94.64 (4) (a) 5. of the statutes first applies to fertilizer sold or distributed on July 1, 2018.
- (b) The treatment of section 94.72 (6) (a) 2. and 3. of the statutes first applies to commercial feeds distributed on January 1, 2018.
- (c) The treatment of section 94.64 (3) (a) 1. of the statutes first applies to licenses for which the license period begins on the August 15 following the effective date of this subsection.
- (d) The treatment of sections 94.64 (3r) (b) (intro.), 1., 2., and 3. and 94.73 (15) (b) 1., 2., and 3. and (c) 1., 2., and 3. of the statutes first applies to licenses for which the license period begins on the October 1 following the effective date of this subsection.
- (e) The treatment of sections 94.68 (3), 94.681 (1) (cm) and (d), (3m), (6) (b), and (7) (bm), 94.685 (3) (a) 2., 94.703 (3) (a) 2. and 3., 94.704 (3) (a) 2., and 94.73 (15) (b) 4., 5., 6., 7., and 8. and (c) 4., 5., 6., 7., and 8. of the statutes, the repeal of sections 94.68 (2) (b) and 94.681 (2) (a), (b), and (c), (3) (a), (b), and (c), (3s) (b) and (c), and (6) (a) 2. and 5. of the statutes, the renumbering of sections 94.68 (2) (a) (intro.) and 1. and 94.681 (6) (a) (intro.) and 1. of the statutes, and the renumbering and amendment of sections 94.68 (2) (a) 2. and 94.681 (2) (intro.), (3) (intro.), (3s) (a), and (6) (a) 3. and 4. of the statutes first apply to licenses for which the license period begins on January 1, 2018.
- (f) The treatment of section 94.65 (2) (a) and (6) (a) 1. of the statutes first applies to licenses and permits for which the license or permit period begins on April 1, 2018
- (g) The treatment of section 94.681 (7) (a) 2. of the statutes and the consolidation, renumbering, and amendment of section 94.681 (7) (a) (intro.) and 1. of the statutes first apply to fees collected for licenses for which the license period begins on January 1, 2018.
- (h) The treatment of section 94.72 (6) (c), (f), (g), and (h) of the statutes first applies to manufacturers and distributors with a license for which the license period begins on March 1, 2018.

SECTION 9303. Initial applicability; Arts Board. SECTION 9304. Initial applicability; Building Commission.

Section 9305. Initial applicability; Child Abuse and Neglect Prevention Board.

SECTION 9306. Initial applicability; Children and Families.

- (1) ELIGIBILITY FOR WISCONSIN SHARES CHILD CARE SUBSIDY.
- (a) The treatment of section 49.155 (1m) (a) (intro.) (as it relates to continued eligibility after ceasing participation in an approved activity) of the statutes first applies to an individual who is eligible to receive a child care sub-

- sidy under section 49.155 (1m) of the statutes and who permanently ceases participation in an approved activity, as defined in section 49.155 (1m) (a) of the statutes, on the effective date of this subsection.
- (b) The treatment of section 49.155 (1) (cm), (1m) (a) 6., and (6g) (b) 4. of the statutes first applies to an individual who is eligible to receive a child care subsidy under section 49.155 (1m) of the statutes and who takes a temporary break, as defined in section 49.155 (1) (cm) of the statutes, from an approved activity, as defined in section 49.155 (1m) (a) of the statutes, on the effective date of this subsection.
- (2) WISCONSIN WORKS CONTROLLED SUBSTANCE SCREENING AND TESTING. The treatment of sections 49.162 (1) (bg), (3), (4) (a), (b), and (c), (4m), and (7) of the statutes first applies to an individual who applies to participate in a program or who registers for a program under section 49.162 (1) (c) 3. of the statutes or, with respect to an individual who applies to participate in a program under section 49.162 (1) (c) 4. of the statutes, to the individual's group members on the effective date of the rules promulgated under section 49.162 (1) (7) of the statutes to implement the changes made by this act or on the effective date of the emergency rules promulgated under Section 9106 (2) (a) of this act, whichever is earlier.
- (3) WISCONSIN SHARES ASSET LIMIT. The treatment of section 49.155 (1) (bm) and (1m) (cm) of the statutes first applies to individuals whose eligibility for Wisconsin Shares is determined or redetermined on the effective date of this subsection.
- (4t) RATE-BASED SERVICE CONTRACTS. The treatment of sections 49.34 (4) (c) and (5m) (b) 1., 2., 3., 4., and 5. and (em) and 49.343 (5) (c) and (6) (a) (intro.), 1., and 2., (b), (c), and (d) of the statutes first apply to a contract under which a provider commences performance on the effective date of this subsection.

 $\begin{array}{lll} \textbf{SECTION} & \textbf{9307. Initial} & \textbf{applicability;} & \textbf{Circuit} \\ \textbf{Courts.} \end{array}$

SECTION 9308. Initial applicability; Corrections.

(1t) RATE-BASED SERVICE CONTRACTS. The treatment of section 301.08 (2) (e) and (em) of the statutes first applies to a contract under which a provider commences performance on the effective date of this subsection.

SECTION 9309. Initial applicability; Court of Appeals.

SECTION 9310. Initial applicability; District Attorneys.

SECTION 9311. Initial applicability; Educational Approval Board.

SECTION 9312. Initial applicability; Educational Communications Board.

SECTION 9313. Initial applicability; Elections Commission.

SECTION 9314. Initial applicability; Employee Trust Funds.

- (1) HEALTH CARE BENEFITS FOR DOMESTIC PARTNERS. The treatment of sections 40.51 (2m) (a) and (b) and 40.52 (2) of the statutes first applies to coverage under group insurance plans offered by the group insurance board on January 1, 2018.
- (2) DEFERRED COMPENSATION PLANS; DOMESTIC PARTNERS. The treatment of section 40.02 (8) (b) 3. of the statutes first applies to benefits paid to a survivor of a participant who dies on January 1, 2018.

- (3c) Health Care Coverage Plan Tiers. The treatment of section 40.51 (6) of the statutes first applies to health care coverage plans offered for calendar year 2018.
- (3p) Submission of Proposed Changes to Group Health Insurance Programs. The treatment of section 40.03 (6) (m) 2. of the statutes first applies to changes the group insurance board proposes to make to the group health insurance program under subchapter IV of chapter 40 of the statutes, other than programs under sections 40.51 (7) and 40.55 of the statutes, for the 2018 program year.
- (4p) APPOINTMENT OF MEMBERS TO THE GROUP INSURANCE BOARD. The treatment of section 15.07 (1) (b) 24. of the statutes first applies to members of the group insurance board who are appointed on the effective date of this subsection.

SECTION 9315. Initial applicability; Employment Relations Commission.

SECTION 9316. Initial applicability; Ethics Commission.

SECTION 9317. Initial applicability; Financial Institutions.

SECTION 9318. Initial applicability; Governor.

SECTION 9319. Initial applicability; Health and Educational Facilities Authority.

SECTION 9320. Initial applicability; Health Services.

(1) MEDICAL ASSISTANCE INCOME; MEDICAL ASSISTANCE PURCHASE PLAN. The treatment of sections 46.269, 49.46 (1) (em), 49.47 (4) (c) 1., and 49.472 (3) (a), (b), (f), and (g), (4) (a) (intro.), 1., 2., 2m., and 3., (b), (bm), (cm), (dm), and (em), (5), and (6) (a) of the statutes first applies to determinations of initial eligibility and cost—sharing and reviews for continued eligibility and cost—sharing on the effective date of this subsection or on the first day of the 4th month beginning after the date of federal approval of the state plan amendment or waiver request, whichever is later.

Vetoed In Part

(1t) SUPERVISED RELEASE. The treatment of sections 20.435 (2) (gz), 51.61 (1) (z), 980.08 (4) (cm), (d), (dm) 3., (e), (em), (f) (intro.), 1., 2., 3., and 4., and (g) and (5m), and 980.105 (2) and (2m) of the statutes and Section 9120 (1t) of this act first apply to petitions pending under section 980.08 of the statutes on the effective date of this subsection.

- (2p) FOODSHARE ASSET REQUIREMENT. The treatment of section 49.79 (1r) of the statutes first applies to initial determinations of applications and redeterminations of eligibility for the food stamp program on the effective date of this subsection or on the date the waiver under section 49.79 (1r) (c) of the statutes, if a waiver is necessary, is in effect, whichever is later.
- (2t) RATE-BASED SERVICE CONTRACTS. The treatment of section 46.036 (4) (c) and (5m) (b) 1., 2., 3., and 4., (e), and (em) of the statutes first applies to a contract under which a provider commences performance on the effective date of this subsection.

SECTION 9321. Initial applicability; Higher Educational Aids Board.

Section 9322. Initial applicability; Historical Society.

Section 9323. Initial applicability; Housing and Economic Development Authority.

SECTION 9324. Initial applicability; Insurance.

SECTION 9325. Initial applicability; Investment Board.

SECTION 9326. Initial applicability; Joint Committee on Finance.

SECTION 9327. Initial applicability; Judicial Commission.

SECTION 9328. Initial applicability; Justice.

SECTION 9329. Initial applicability; Legislature.

SECTION 9330. Initial applicability; Lieutenant Governor.

SECTION 9331. Initial applicability; Local government.

- (1) LEVY LIMIT NEGATIVE ADJUSTMENT FOR DEBT SERVICE. The treatment of section 66.0602 (2m) (a) of the statutes first applies to a levy that is imposed in December 2017.
- (1w) Levy Limit Negative adjustment fire Protection. The treatment of section 66.0602 (2m) (b) 1. of the statutes first applies to a levy that is imposed in December 2017.
- (3b) COOPERATIVE PLANS. The treatment of section 66.0307 (3) (d) 4m. and (6) of the statutes first applies to a cooperative plan submitted to the department of administration on the effective date of this subsection.
- (3f) Provision of Benefits. The treatment of sections 66.0137 (5) (b) and 66.0510 of the statutes first applies to a local governmental unit's officers, employees, and agents, and their spouses, dependent children, and domestic partners who are covered by a collective bargaining agreement under subchapter IV of chapter 111 of the statutes, or another contract, that contains provisions inconsistent with that treatment on the day on which the agreement or contract expires or is terminated, extended, modified, or renewed, whichever occurs first.
- (3t) The treatment of section 66.1105 (2) (f) 2. e. of the statutes first applies to a tax incremental district that

is in existence or created on the effective date of this subsection.

- (3w) Levy Limit Referenda.
- (a) The treatment of section 66.0602 (4) (a) of the statutes first applies to a resolution to exceed a levy increase limit adopted on the effective date of this paragraph.
- (b) The treatment of section 66.0602 (4) (c) of the statutes first applies to a referendum to exceed a levy increase limit authorized by a resolution adopted on the effective date of this paragraph.
- (4f) Business improvement districts. The treatment of section 66.1109 (5) (d) of the statutes first applies to a special assessment that is imposed on the effective date of this subsection.
- (5t) ROOM TAX. The treatment of section 66.0615 (1m) (a) and (5) of the statutes first applies to a lodging marketplace that registers with the department of revenue on the effective date of this subsection.
- (6p) The treatment of sections 48.981 (3) (cr), 59.52 (7), and 66.0301 (2) and (7) of the statutes first applies to contracts entered into on the effective date of this subsec-

Vetoed In Part

(7t) LAND TRANSFER AUTHORITY, MILWAUKEE COUNTY EXECUTIVE AND BOARD. The treatment of sections 59.17 (2) (b) 3. (intro.) and a. to c. and 59.52 (6) (intro.) and (a) (as it relates to land transactions in Milwaukee County) and (31) (e) of the statutes first applies to a land transaction for which a contract has been entered into after September 1, 2018.

SECTION 9332. Initial applicability; Military Affairs.

Section 9333. Initial applicability; Natural Resources.

- (1) FEE WAIVERS. The treatment of sections 24.40 (3) and 86.16 (6) of the statutes first applies to easements granted or construction permits issued on the effective date of this subsection.
- (2) Camping fees. The treatment of section 27.01 (10) (d) 1. and 2. of the statutes first applies to campsite reservations made on January 1, 2018.

applicability; SECTION 9334. Initial **Public** Defender Board.

applicability; SECTION **9335.** Initial Public Instruction.

- (1) PUPIL TRANSPORTATION AID. The treatment of section 121.58 (2) (am) and (4) of the statutes first applies to state aid for transportation provided in the 2017-18 school year.
- (1f) SCHOOL DISTRICT BORROWING. The treatment of section 67.05 (6a) (a) 2. a. and c. and (am) 1. of the statutes first applies to a resolution to issue a bond adopted by the school board of a school district on the effective date of this subsection.
- (1g) SCHEDULING OF SCHOOL DISTRICT REFERENDUMS. The treatment of section 121.91 (3) (c) of the statutes, the

renumbering and amendment of section 121.91 (3) (a) of the statutes, and the creation of section 121.91 (3) (a) 2. and 3. of the statutes first apply to a resolution to exceed the revenue limit under section 121.91 (2m) of the In Part statutes adopted by the school board of a school district on the effective date of this subsection.

- (2c) ADDITIONAL SPECIAL EDUCATION AID. The treatment of section 115.881 (2) of the statutes first applies to aid paid in the 2017-18 school year.
- INDEPENDENT CHARTER SCHOOLS; SUMMER SCHOOL PAYMENTS. The treatment of section 118.40 (2r) (fm) and (2x) (em) 1. of the statutes first applies to pupils who attend summer school during the summer of 2018.
- (3p) PARENTAL CHOICE PROGRAMS; SUMMER SCHOOL PAYMENTS. The treatment of sections 118.60 (4m) (a) 2. and 3. and (b) 3. and 119.23 (4m) (a) 2. and 3. and (b) 3. of the statutes first applies to payments for pupils who attend summer school during the summer of 2018.
- (3q) SPECIAL NEEDS SCHOLARSHIP PROGRAM; SUM-MER SCHOOL PAYMENTS. The treatment of section 115.7915 (4p) of the statutes first applies to payments for pupils who attend summer school during the summer of 2018.
- (4d) PARENTAL CHOICE PROGRAMS; INCOME ELIGIBIL-ITY LIMIT FOR STATEWIDE PROGRAM AND RECIPROCITY BETWEEN PROGRAMS. The treatment of sections 118.60 (2) (a) 1. c., and (bm) and 119.23 (2) (a) 1. d. of the statutes first applies to a pupil who applies to attend a participating private school in a program under section 118.60 or 119.23 of the statutes in the 2018-19 school vear.
- (5d) SPECIAL NEEDS SCHOLARSHIP PROGRAM; ELIGI-BILITY. The treatment of section 115.7915 (2) (a) and (e) of the statutes first applies to applications to participate in the Special Needs Scholarship Program in the 2018-19 school year.

SECTION 9336. Initial applicability; Public Lands, **Board of Commissioners of.**

Section 9337. Initial applicability; Public Service Commission.

SECTION 9338. Initial applicability; Revenue.

- (1) INTERNAL REVENUE CODE UPDATE. The treatment of section 71.83 (1) (cf) of the statutes first applies to property for which a federal estate tax return is filed after July 31, 2015.
- (1f) BIBLE CAMPS. The treatment of section 70.11 (11) of the statutes first applies to the property tax assessments as of January 1, 2018.
- (2) SALES AND USE TAX EXEMPTION FOR OCCASIONAL SALES. The renumbering of section 77.51 (9) (a) of the statutes and the creation of section 77.51 (9) (a) 2. of the statutes first apply to sales beginning on January 1, 2018.
- (2e) Grounds of a college or university. The renumbering and amendment of section 70.11 (3) (a) of the statutes and the creation of section 70.11 (3) (a) 2. of

the statutes first apply to the property tax assessments as of January 1, 2018.

- (3t) OTHER STATE TAX CREDIT. The treatment of section 71.07 (7) (b), (c), and (d) of the statutes, the renumbering and amendment of section 71.07 (7) (a) of the statutes, and the creation of section 71.07 (7) (a) 1. of the statutes first apply to taxable years beginning on January 1, 2017.
- (4) Bonds issued by WHEFA. The treatment of section 71.05 (1) (c) 13. of the statutes first applies to taxable years beginning on January 1 of the year in which this subsection takes effect, except that if this subsection takes effect after July 31 the treatment of section 71.05 (1) (c) 13. of the statutes first applies to taxable years beginning on January 1 of the year following the year in which this subsection takes effect.
- (7) SOURCING SERVICES. The treatment of sections 71.04 (7) (dh) 2. b. and c. and 71.25 (9) (dh) 2. b. and c. of the statutes first applies to taxable years beginning on January 1, 2017.
- (8) OTHER STATE TAX CREDIT. The treatment of section 71.07 (5n) (d) 3. of the statutes first applies to taxable years beginning on January 1, 2017.
- (10) NET BUSINESS LOSS CARRY–FORWARD AND CARRY–BACK. The treatment of sections 71.05 (8) (b) 1., 71.26 (4) (a), 71.45 (4) (a), and 71.80 (25) of the statutes first applies to a loss claimed on the effective date of this subsection regardless of the year in which the loss was incurred.
- (11) INTEREST ON TAX REFUNDS. The treatment of sections 71.07 (3q) (d) 2., (3w) (c) 1., and (3y) (d) 2., 71.28 (3q) (d) 2., (3w) (c) 1., and (3y) (d) 2., and 71.47 (3q) (d) 2., (3w) (c) 1., and (3y) (d) 2. of the statutes first applies to refunds paid on the effective date of this subsection.
- (12) Itemized deductions credit, nonresidents. The treatment of section 71.07 (5) (b) of the statutes first applies to taxable years beginning on January 1, 2017.
- (13) FILING OF CERTAIN INFORMATION RELATED TO INCOME AND FRANCHISE TAXES. The treatment of sections 71.65 (2) (b), 71.70 (1) and (2), 71.715, and 71.72 of the statutes first applies to payments made on January 1, 2017
- (13i) SUBTRACT MODIFICATION, CERTAIN ADOPTION FEES. The treatment of section 71.05 (6) (b) 22. of the statutes first applies to taxable years beginning after December 31, 2016.
- (13p) ELIMINATE THE ALTERNATIVE MINIMUM TAX. The treatment of sections 71.07 (2dy) (b) and (d) 2.; 71.07 (3q) (b) (intro.); 71.07 (3q) (d) 2. (as it relates to the elimination of the alternative minimum tax); 71.07 (3s) (b); 71.07 (3t) (b); 71.07 (4k) (b) 4. a., 5. a., and 6. a.; 71.07 (5b) (b) 1.; 71.07 (5d) (b) (intro.); 71.07 (5n) (b) (intro.); 71.07 (9m) (a) 2m. and 3. and (h); 71.07 (9r) (a); 71.08 (5); 71.09 (1) (b), (2), and (11) (b); 71.10 (4) (f); 71.28 (6) (h); 71.47 (6) (h); 71.613 (2) (intro.); and 73.03 (71) (b)

and (c) of the statutes first applies to taxable years beginning after December 31, 2018.

- (14) EXTENSION OF TIME TO FILE CERTAIN INFORMATION. The treatment of sections 71.65 (5) (a) (intro.), 1., and 2. and 71.73 (2) (intro.), (a), (b), and (c) of the statutes first applies to an extension applied for on the effective date of this subsection.
- (15) ELECTRONIC FILING. The treatment of section 71.80 (20) of the statutes first applies to a statement or return required to be filed in 2018.
- (16) HOMESTEAD CREDIT; INVESTMENT LOSS LIMITS. The treatment of sections 71.52 (1e), (1m), and (6) and 71.55 (10) of the statutes first applies to taxable years beginning after December 31, 2017.
- (16c) OLYMPIC, PARALYMPIC, AND SPECIAL OLYMPICS MEDALS. The treatment of section 71.05 (6) (b) 53. of the statutes first applies, retroactively, to taxable years beginning after December 31, 2015.
- (18) LUMP SUM CONTRACT SALES TAX EXEMPTION. The treatment of sections 77.52 (2m) (b) and 77.54 (60) (b) and (bm) of the statutes, the renumbering and amendment of section 77.54 (60) (a) and (c) of the statutes, and the creation of section 77.54 (60) (c) 2. and (d) 2. and 3. of the statutes first apply to a contract that is entered into or extended, modified, or renewed on the effective date of this subsection.
- (18d) Building Materials sales Tax exemption. The treatment of section 77.54 (9m) of the statutes first applies to contracts entered into on July 1, 2018.

SECTION 9339. Initial applicability; Safety and Professional Services.

- (1) Information included on Real estate transfer RETURN FORM. The treatment of section 77.22 (2) (c) and (d) of the statutes first applies to a real estate transfer return form submitted for recording on the effective date of this subsection.
- (2) CERTIFICATION OF RENTAL UNIT ENERGY EFFICIENCY. The treatment of section 101.122 (4) of the statutes first applies to a transfer of a rental unit on the effective date of this subsection.
- (3) FORMAL REQUISITES FOR RECORDING CONVEYANCE. The treatment of sections 101.122 (6) and 706.05 (12) of the statutes first applies to a deed, conveyance, or other document of transfer submitted for recording on the effective date of this subsection.
- (4) CITATION PROCEDURE FOR CERTAIN LIMITED VIOLATIONS. The treatment of section 778.25 (1) (a) 7., (b), and (c) of the statutes first applies to an action to recover a forfeiture commenced on the effective date of this subsection.
- (5) CONFIRMATION OF SALE AND TRANSMITTAL OF DEED IN POPULOUS COUNTIES. The treatment of section 846.167 (2) (a), (b) 2. b., and (c) and (3) of the statutes first applies to a sale made by a sheriff or referee that is confirmed on the effective date of this subsection.

(7f) FIREWORKS MANUFACTURER LICENSING. The treatment of section 167.10 (6m) (d) of the statutes first applies to an application for a license or license renewal under that section received by the department of safety and professional services on the effective date of this subsection.

SECTION 9340. Initial applicability; Secretary of State.

SECTION 9341. Initial applicability; State Fair Park Board.

SECTION 9342. Initial applicability; Supreme Court.

SECTION 9343. Initial applicability; Technical College System.

SECTION 9344. Initial applicability; Tourism. SECTION 9345. Initial applicability; Transportation.

- (4b) JURISDICTIONAL TRANSFER AGREEMENTS. The treatment of section 84.02 (8) (a), (b), (c), and (d) of the statutes first applies to a jurisdictional transfer agreement entered into on the effective date of this subsection.
- (4p) REGISTRATION FEE FOR HYBRID ELECTRIC AND NONHYBRID ELECTRIC VEHICLES. The treatment of section 341.25 (1) (L) of the statutes first applies to applications for registration received by the department of transportation on the effective date of this subsection.

Vetoed In Part

- (4t) REPLACE-IN-KIND ALTERNATIVES. The treatment of section 84.06 (14) of the statutes first applies to preparations for a highway improvement project commenced on the effective date of this subsection.
- (4w) Transportation of raw forest products and LUMBER. The treatment of section 348.15 (11) of the statutes first applies to vehicles operated on the effective date of this subsection.

SECTION 9346. Initial applicability; Treasurer.

SECTION 9347. Initial applicability; University of Wisconsin Hospitals and Clinics Authority; Medical College of Wisconsin.

SECTION 9348. Initial applicability; University of Wisconsin System.

- (3c) Nonresident Tuition Exemption for National GUARD MEMBERS. The treatment of section 36.27 (2) (br) of the statutes first applies to a person who enrolls in an institution in the University of Wisconsin System in the academic year beginning after the effective date of this subsection.
- (3p) EMPLOYEE REIMBURSEMENT FOR LODGING EXPEN-SES. The treatment of section 36.11 (56) of the statutes first applies to claims for reimbursement made on the effective date of this subsection.

SECTION 9349. Initial applicability; Veterans Affairs.

(1) Grants. The treatment of section 45.41 (2) (intro.) of the statutes first applies to an application for payment the department of veterans affairs receives from

a state veterans organization under section 45.41 (2) of the statutes on April 1, 2018.

SECTION 9350. Initial applicability; Wisconsin **Economic Development Corporation.**

(1) New Loans. The treatment of section 238.124 (1) of the statutes first applies to a loan the Wisconsin Economic Development Corporation originates on the effective date of this subsection.

SECTION 9351. Initial applicability; Workforce Development.

SECTION 9352. Initial applicability; Other.

- (3) ELIMINATION OF PREVAILING WAGE LAW. The treatment of sections 16.856, 19.36 (3) and (12), 59.20 (3) (a), 84.062, 84.41 (3), 106.04, 109.09 (1), 111.322 (2m) (c) and (d), 230.13 (1) (intro.), 233.13 (intro.), 946.15, and 978.05 (6) (a) of the statutes first applies, with respect to a project of public works that is subject to bidding, to a project for which the request for bids is issued on the effective date of this subsection and, with respect to a project of public works that is not subject to bidding, to a project the contract for which is entered into on the effective date of this subsection.
- (5t) CONDEMNATION AUTHORITY FOR RECREATIONAL TRAILS. The treatment of sections 23.09 (2) (d) (intro.), 27.01 (2) (a), 27.019 (10), 27.05 (3), 27.065 (1) (a), 27.08 (2) (b) and (c), 32.015, 32.51 (1) (intro.), 59.52 (6) (a) (as it relates to condemnation for recreational trails, bicycle ways, bicycle lanes, and pedestrian ways), 60.782 (2) (d), 61.34 (3) (b), 62.22 (1) (b), 62.23 (17) (a) (intro.) and (am), 85.09 (2) (a), and 990.01 (2) of the statutes and the renumbering and amendment of sections 61.34 (3) and 62.22 (1) of the statutes first apply to condemnation proceedings in which title to the subject property has not vested in the condemnor on the effective date of this subsection.

SECTION 9400. Effective dates; general. Except as otherwise provided in Sections 9401 to 9452 of this act, this act takes effect on July 1, 2017, or on the day after publication, whichever is later.

SECTION 9401. Effective dates; Administration.

- (1p) PROSECUTOR BOARD. The treatment of sections Vetoed 13.093 (2) (a), 13.0967, 15.105 (7), 15.77, 16.971 (9) (by **In Part** SECTION 171b) and (10), 20.548, 20.923 (4) (f) 7x. and (6) (hs), 227.118, 227.19 (3) (em), 230.08 (2) (qp), 230.33 (1), 978.001 (1b), (1d), (1n), and (1p), 978.003, 978.004, 978.005, 978.03, 978.045 (1g), (1r) (bm) (intro.) (by SEC-TION 2261r), and (2), 978.05 (9), 978.11, and 978.12 (1) (c) of the statutes and Section 9101 (7p) of this act take effect on February 1, 2018.
- INFORMATION TECHNOLOGY INFRASTRUCTURE FUNDING; SUNSET. The treatment of section 20.505 (4) (s) (by Section 442) of the statutes takes effect on July 1, 2019.

sion.

- (4) HUMAN RESOURCES SERVICES. The treatment of section 16.004 (20) of the statutes and Section 9101 (9) of this act takes effect on July 1, 2018.
- (5) YOUTH WELLNESS CENTER; TRIBAL PAYMENT. The treatment of section 20.505 (8) (hm) (by SECTION 455) of the statutes takes effect on July 1, 2019.

SECTION 9402. Effective dates; Agriculture, Trade and Consumer Protection.

SECTION 9403. Effective dates; Arts Board. SECTION 9404. Effective dates; Building Commis-

Section 9405. Effective dates; Child Abuse and Neglect Prevention Board.

SECTION 9406. Effective dates; Children and Families.

- (1) FOSTER CARE AND KINSHIP CARE RATES. The treatment of sections 48.57 (3m) (am) (intro.) and (3n) (am) (intro.) and 48.62 (4) of the statutes takes effect on January 1, 2018, or on the day after publication, whichever is later.
- (2) CHILD CARE BACKGROUND CHECKS. The treatment of sections 20.435 (6) (jm), 20.437 (1) (jm) and (2) (jn), 48.65 (1), 48.651 (1) (intro.), (a), and (b), (1d) (b), (2), (2m), and (3) (a) and (b), 48.66 (5), 48.68 (1), 48.685 (1) (ag) 1. b., (b), (bm), and (c) 3m. and 4., (2) (am) (intro.) and 5., (ar), (b) 1. (intro), a., b., c., d., and e., 2., and 4., (bb), (bd), (bg), (bm), and (br), (3) (a), (am), (b), and (bm), (3m), (4m) (a) (intro.) and 1., (ad), (b) 1., (c), and (d), (5) (a), (bm) (intro.), and (br), (5c) (a) and (c), (5m), (6) (a), (am), and (b) 1., 2., and 4., (8), and (9), 48.686, 48.715 (4g) (a) and (b), 48.981 (7) (cp), 49.133 (1m) (a) and (b) and (2m) (intro.), (a), and (b), 49.137 (2) (a) and (3) (a), 49.155 (1) (am) and (b), (1d) (title), (a) (intro.), 1., and 2., (am), and (b), (4) (a), (6) (b) and (d), (7) (a) 1. and 2. and (b) (intro.), 1., and 2., 120.13 (14) (a) (by SEC-TION 1626) and (b) 1. and 2., and 938.396 (2g) (o) of the statutes, the renumbering and amendment of section 48.685 (1) (am) of the statutes, and the creation of section 48.685 (1) (am) 1., 2., and 3. of the statutes take effect on September 30, 2018.
- (3) CHILD CARE SUBSIDY COPAYMENTS. The treatment of section 49.155 (1m) (c) 1. (intro.) and 1d. of the statutes takes effect on July 1, 2018, or on the first day of the 12th month beginning after publication, whichever is later.
- (4t) RATE-BASED SERVICE CONTRACTS. The treatment of sections 49.34 (4) (c) and (5m) (b) 1., 2., 3., 4., and 5. and (em) and 49.343 (5) (c) and (6) (a) (intro.), 1., and 2., (b), (c), and (d) of the statutes and Section 9306 (4t) of this act take effect on the January 1 after publication.

SECTION 9407. Effective dates; Circuit Courts. SECTION 9408. Effective dates; Corrections.

(1) JUVENILE CORRECTIONAL SERVICES DAILY RATES. The treatment of section 301.26 (4) (d) 2. and 3. of the statutes takes effect on July 1, 2017, or on the 2nd day after publication, whichever is later.

(3t) RATE-BASED SERVICE CONTRACTS. The treatment of section 301.08 (2) (e) and (em) of the statutes and SECTION 9308 (1t) of this act take effect on the January 1 after publication.

SECTION 9409. Effective dates; Court of Appeals. SECTION 9410. Effective dates; District Attorneys. SECTION 9411. Effective dates; Educational Approval Board.

(1p) Temporary attachment of educational Vetoed APPROVAL BOARD TO DEPARTMENT OF SAFETY AND PROFES- In Part SIONAL SERVICES. The repeal of sections 15.945 (title) and 20.292 (2) (title) of the statutes, the renumbering of section 38.50 (title), (1) (intro.), (b), (c), (d), (e), (f), and (g), (2), (3), (7), (8), (10) (title), (b), (c), (cm), (d), (e), and (f), (11) (title), (a), (b), and (c), (12), and (13) (title), (a), (b), and (c) of the statutes, the renumbering and amendment of sections 15.945 (1), 20.292 (2) (g), (gm), and (i), and 38.50 (1) (a), (5), (10) (a), (11) (d), and (13) (d) of the statutes, the amendment of sections 15.406 (6) (a) 1. (by SECTION 52m), 45.20 (1) (d) and (2) (a) 1. and 2. (intro.), (c) 1., and (d) 1. (intro.), 45.21 (2) (a) (by Section 738h), 71.05 (6) (b) 28. (intro.), 71.07 (5r) (a) 2. and 6. b., 71.28 (5r) (a) 2. and 6. b., 71.47 (5r) (a) 2. and 6. b., 102.07 (12m) (a) 1., 111.335 (1) (cx), 182.028, subchapter V (title) of chapter 440, 460.05 (1) (e) 1. (by Section 2149m), 944.21 (8) (b) 3. a. (by Section 2248m), 948.11 (4) (b) 3. a. (by Section 2250m), and 995.55 (1) (b) of the statutes, and Section 9111 (1p) of this act take effect on January 1, 2018, or on the day after publication, whichever is later.

(1q) Elimination of Educational approval board AND TRANSFER OF FUNCTIONS. The repeal of sections 15.07 (5) (i), 15.405 (18), and 440.52 (1) (a) and (5) of the statutes, the amendment of sections 15.406 (6) (a) 1. (by SECTION 520), 29.506 (7m) (a), 45.21 (2) (a) (by SECTION 738j), 125.04 (5) (a) 5., 125.17 (6) (a) (intro.), 134.66 (2m) (b), 440.52 (1) (e) 8. and (g), (2), (3), (7) (intro.), (g), (h), and (i), (8) (a), (b), (c) (intro.), 1., 2., 4., and 5., (d), and (e), (10) (a), (b), (c) (intro.) and 1., and (cm), (11) (b) 1., (c), and (d), (12) (a) (intro.) and 1. and (b), and (13) (a) 2. a., b., and e. and (d), 460.05 (1) (e) 1. (by Section 2149p), 944.21 (8) (b) 3. a. (by Section 2248p), and 948.11 (4) (b) 3. a. (by SECTION 2250p) of the statutes, the repeal and recreation of sections 15.675 (1) (d) and 440.52 (title) of the statutes, and Section 9111 (1q) of this act take effect on July 1, 2018, or on the day after publication, whichever is later.

Vetoed In Part

SECTION 9412. Effective dates; Educational Communications Board.

SECTION 9413. Effective dates; Elections Commission.

SECTION 9414. Effective dates; Employee Trust Funds.

(1) DUTY DISABILITY SURVIVORSHIP BENEFITS AND DEFERRED COMPENSATION PLANS; DOMESTIC PARTNERS. The treatment of sections 40.02 (8) (b) 3., 40.65 (7) (am)

1., 1g., 1m., and 3. and (ar) 1. a., ag., and am., and 40.80 (2r) (a) 2. of the statutes takes effect on January 1, 2018.

SECTION 9415. Effective dates; Employment Relations Commission.

SECTION 9416. Effective dates; Ethics Commis-

Section 9417. Effective dates; Financial Institutions.

SECTION 9418. Effective dates; Governor. SECTION 9419. Effective dates; Health and Educational Facilities Authority.

SECTION 9420. Effective dates; Health Services.

- (1) MEDICAL ASSISTANCE INCOME; MEDICAL ASSIST-ANCE PURCHASE PLAN. The treatment of sections 46.269, 49.46 (1) (em), 49.47 (4) (c) 1., and 49.472 (3) (a), (b), (f), and (g), (4) (a) (intro.), 1., 2., 2m., and 3., (b), (bm), (cm), (dm), and (em), (5), and (6) (a) of the statutes and SEC-TION 9320 (1) of this act take effect on July 1, 2018.
- (2) FOODSHARE ASSET REQUIREMENT. The treatment of section 49.79 (1r) of the statutes and Section 9320 (2p) of this act take effect on July 1, 2018.
- (3f) CHILDREN'S LONG-TERM SUPPORT WAIVER PRO-GRAM. The treatment of sections 20.435 (4) (x), 25.77 (14), and 49.45 (39) (bm) of the statutes takes effect on December 31, 2017.
- (3k) Graduate medical training support grants. If January 2017 Special Session Assembly Bill 7 is enacted into law, the treatment of 2017 Wisconsin Act (January 2017 Special Session Assembly Bill 7), SEC-TION 1 (1) (by Section 2265t) and Section 9220 (1m) of this act take effect on the 2nd day after publication of this act or of 2017 Wisconsin Act (January 2017 Special Session Assembly Bill 7), whichever is later. If January 2017 Special Session Assembly Bill 7 is not enacted into law in the 2017-18 legislative session, the treatment of 2017 Wisconsin Act (January 2017 Special Session Assembly Bill 7), Section 1 (1) (by Section 2265t) and SECTION 9220 (1m) of this act are void.

Vetoed In Part

- (3t) YOUTH CRISIS STABILIZATION FACILITY. The repeal of section 20.435 (5) (kd) of the statutes takes effect on July 1, 2019.
- (3u) RATE-BASED SERVICE CONTRACTS. The treatment of section 46.036 (4) (c) and (5m) (b) 1., 2., 3., and 4., (e), and (em) of the statutes and Section 9320 (2t) of this act take effect on the January 1 after publication.
- (4b) MEDICAL ASSISTANCE AUDIT OF FAMILY PLANNING SERVICES.
- (a) The treatment of section 20.435 (4) (L) (by SEC-TION 379ar) of the statutes takes effect on July 1, 2019.
- (b) The treatment of section 20.435 (4) (L) (by SEC-TION 379as) of the statutes takes effect on July 1, 2021.

Vetoed In Part

(4f) PEER-RUN RESPITE CENTER FOR VETERANS. The treatment of sections 20.435 (2) (gk) (by Section 377b) and 46.48 (32) (by Section 752b) of the statutes and the repeal of section 20.435 (5) (kp) of the statutes take effect on July 1, 2019.

(6d) CRITICAL ACCESS HOSPITAL GRANT. The treatment of section 20.435 (5) (bc) (by Section 379g) of the statutes takes effect on July 1, 2019.

SECTION 9421. Effective dates; Higher Educational Aids Board.

SECTION 9422. Effective dates; Historical Society. SECTION 9423. Effective dates; Housing and Economic Development Authority.

SECTION 9424. Effective dates; Insurance.

- (1) Transfer of information technology func-TIONS. SECTION 9124 (1) of this act takes effect on October 1, 2017.
- (3f) Nonresident agent appointment fees. The treatment of section 601.31 (1) (n) of the statutes takes effect on January 1, 2019.

SECTION 9425. Effective dates; Investment Board. SECTION 9426. Effective dates; Joint Committee on Finance.

SECTION 9427. Effective dates; Judicial Commission.

SECTION 9428. Effective dates; Justice.

- (1q) Community institution security cost reim-BURSEMENT GRANTS. The repeal of section 165.989 of the statutes takes effect on July 1, 2019.
- (1r) Court appointed special advocate associa-TION GRANTS.
- (a) The repeal of section 20.455 (5) (kr) of the statutes takes effect on July 1, 2019.
- (b) Section 2265p of this act takes effect on June 29, 2017.

Section 9429. Effective dates; Legislature. SECTION 9430. Effective dates; Lieutenant Gover-

SECTION 9431. Effective dates; Local Government.

- (1i) QUARRY REGULATION; ORDINANCE IN CONFLICT Vetoed WITH STATUTE. The treatment of sections 59.69 (10) (ap) In Part and (10s), 60.61 (5) (as) and (5s), 62.23 (7) (hd) and (19), and 101.02 (7w) of the statutes, the renumbering and amendment of sections 59.69 (10) (ab), 60.61 (5) (ab), and 62.23 (7) (ab) of the statutes, and the creation of sections 59.69 (10) (ab) 1., 2., 4., and 5., 60.61 (5) (ab) 1., 3., and 4., 62.23 (7) (ab) 1., 3., and 4., and 66.0414 of the statutes take effect on April 1, 2018.
- (2i) REGULATION OF QUARRY NOISE. The amendment of section 66.0414 (3) (f) of the statutes takes effect on April 1, 2019.

SECTION 9432. Effective dates; Military Affairs. SECTION 9433. Effective dates; Natural Resources.

- (1) VEHICLE ADMISSION RECEIPTS. The treatment of section 27.01 (7) (f) 2., 3., and 4. and (g) 2., 3., and 4. of the statutes takes effect on January 1, 2018.
- (2) Camping fees. The treatment of section 27.01 (10) (d) 1. and 2. and (g) 5. and 6. of the statutes takes effect on January 1, 2018.

(4w) VILLAGE OF PLOVER GRANT. The repeal of section 20.370 (6) (gs) of the statutes takes effect on July 1, 2019.

SECTION 9434. Effective dates; Public Defender Board.

SECTION 9435. Effective dates: Public Instruction.

- (1f) CONSOLIDATION AID. The treatment of section 20.255 (2) (br) of the statutes takes effect on July 1, 2019.
- (1p) PART TIME OPEN ENROLLMENT. The repeal of sections 20.255 (2) (cw) and 118.52 (1) (am), (6) (c), and (12) (b) of the statutes, the renumbering and amendment of section 118.52 (12) (a) of the statutes, the amendment of sections 20.255 (2) (cy), 115.28 (54m), 115.38 (1) (d), 115.385 (4), 118.52 (2), (3) (a), (b), (c), (d) 1., and (e), (6) (a), (8), (9), (10), and (11) (a) and (b), 118.57 (1), and 121.05 (1) (a) 5. of the statutes, the repeal and recreation of section 118.52 (title) of the statutes, and the creation of section 118.52 (6) (b) of the statutes take effect on July
- (1q) Early college credit program. The repeal of section 118.55 (5) (c) of the statutes, the renumbering and amendment of section 118.55 (1), (4) (a), (6) (c), (7t) (b), and (8) of the statutes, the amendment of section 118.55 (title), (2) (a), (3), (4) (b) and (c), (5) (intro.) and (a), (6) (a) and (b), (7g), (7t) (a) and (c), and (8) (title) of the statutes, and the creation of sections 20.445 (1) (d), 106.125, and 118.55 (1) (b), (c), and (d), (3) (c), (4) (a) 1., (5) (b), (d) 1m. and 2., and (e), (6) (c) 2., and (8) (b) of the statutes take effect on July 1, 2018.
- (1w) SCHEDULING SCHOOL DISTRICT REFERENDUMS. The treatment of sections 7.52 (8), 8.06, 67.05 (6a) (a) 2. (intro.), a., and c. and (am) 1., and 121.91 (3) (c) of the statutes, the renumbering and amendment of section 121.91 (3) (a) of the statutes, the creation of section 121.91 (3) (a) 2. and 3. of the statutes, and Section 9335 (1f) and (1g) of this act take effect on January 1, 2018.
- (1x) REORGANIZATION OF A LOW PERFORMING SCHOOL DISTRICT; STUDY FUNDING. The repeal of section 20.255 (1) (fp) of the statutes takes effect on July 1, 2019.

SECTION 9436. Effective dates; Public Lands, **Board of Commissioners of.**

SECTION 9437. Effective dates; Public Service Commission.

Vetoed In Part

Vetoed

In Part

(1t) Public utility definition. The treatment of section 196.01 (5) (b) 7. of the statutes takes effect on the first day of the 13th month beginning after publication.

SECTION 9438. Effective dates; Revenue.

- (2) BAD DEBT ADJUSTMENTS. The treatment of 2013 Wisconsin Act 229, section 6 (1) takes effect retroactively to June 30, 2017.
- Vetoed In Part
- (2i) SALES TAX ON BROADCAST EQUIPMENT. The renumbering and amendment of section 77.54 (23n) of the statutes and the creation of section 77.54 (23n) (a) and (b) 2. to 5. of the statutes take effect on July 1, 2019.
- (2p) Amusement devices and tournament or ENTRANCE FEES. The treatment of section 77.54 (65) of

- the statutes takes effect on the first day of the 3rd month beginning after publication.
- (3) LUMP SUM CONTRACT SALES TAX EXEMPTION. The treatment of sections 77.52 (2m) (b) and 77.54 (60) (b) and (bm) of the statutes, the renumbering and amendment of section 77.54 (60) (a) and (c) of the statutes, the creation of section 77.54 (60) (c) 2. and (d) 2. and 3. of the statutes, and Section 9338 (18) take effect on the first day of the 3rd month beginning after publication.
- (3d) BUILDING MATERIALS SALES TAX EXEMPTION. The treatment of section 77.54 (9m) of the statutes takes effect on July 1, 2018.
- (3f) SALES AND USE TAX AUDITS; STATISTICAL SAM-PLING. The treatment of section 77.59 (2) and (2g) of the statutes takes effect on the first day of the 6th month beginning after publication.
- (3m) RECREATIONAL MOTORBOATS. The treatment of **Vetoed** section 78.01 (2m) (f) of the statutes takes effect retroac- In Part tively on July 1, 2013.
- (4f) Tax stamp discount for manufacturers, BONDED DIRECT MARKETERS, AND DISTRIBUTORS. The treatment of section 139.32 (5) of the statutes takes effect on the first day of the 3rd month beginning after publication.
- (4i) SALES AND USE TAX EXEMPTIONS FOR BEEKEEPING. The treatment of sections 77.51 (1bm) and (13) (o) and 77.54 (3) (a), (3m) (intro.), (hm), and (L), (30) (a) 3. and 5., and (33) of the statutes takes effect on the first day of the 3rd month beginning after publication.
- (4w) SALES TAX TREATMENT OF INTERNET ACCESS SER-VICES. The treatment of sections 77.52 (2) (a) 5. a. and 77.522 (4) (h) of the statutes takes effect on July 1, 2020.

SECTION 9439. Effective dates; Safety and Professional Services.

- (1) RENTAL UNIT ENERGY EFFICIENCY. The treatment of sections 77.22 (2) (c) and (d), 101.02 (20) (a), (21) (a), and (24) (a) 2., 101.122, 101.19 (1g) (i) and (1r), 706.05 (12), 778.25 (1) (a) 7., (b), and (c), and 846.167 (2) (a), (b) 2. b., and (c) and (3) of the statutes and Section 9339 (1), (2), (3), (4), and (5) of this act take effect on the first day of the 4th month beginning after publication.
- (4t) PRIVATE ON-SITE WASTEWATER TREATMENT SYS-TEM REPLACEMENT OR REHABILITATION GRANT PROGRAM; SUNSET. The treatment of sections 20.165 (2) (j) and (ke), 145.20 (5) (a) and (am), 145.245 (by Section 1655j), and 281.57 (7) (c) 1. of the statutes, the renumbering and amendment of section 145.01 (4m) of the statutes, and the creation of section 145.01 (4m) (a), (b), (c), (d), and (e) of the statutes take effect on June 30, 2021.

SECTION 9440. Effective dates; Secretary of State. SECTION 9441. Effective dates; State Fair Park

SECTION 9442. Effective dates; Supreme Court. **SECTION 9443. Effective dates; Technical College** System.

SECTION 9444. Effective dates; Tourism.

(1t) IMPROVEMENT OF PROPERTY USED FOR SKI JUMPING COMPETITIONS. The treatment of section 20.380 (1) (b) (by Section 359n) of the statutes takes effect on July 1, 2018.

SECTION 9445. Effective dates; Transportation.

Vetoed In Part

- (1f) Transportation projects commission mem-BERSHIP. The treatment of section 13.489 (1g) of the statutes takes effect on January 1, 2018.
- (1i) RECREATIONAL VEHICLE DEALER SERVICE FEES. Notwithstanding section 227.265 of the statutes, the repeal of section Trans 142.04 (7), Wisconsin Administrative Code, takes effect on the day after publication.
- (1p) REGISTRATION FEE FOR HYBRID ELECTRIC AND NONHYBRID ELECTRIC VEHICLES. The treatment of section 341.25 (1) (L) of the statutes and Section 9345 (4p) of this act take effect on January 1, 2018.

SECTION 9446. Effective dates: Treasurer.

SECTION 9447. Effective dates; University of Wisconsin Hospitals and Clinics Authority; Medical College of Wisconsin.

SECTION 9448. Effective dates; University of Wisconsin System.

(3p) EMPLOYEE REIMBURSEMENT FOR LODGING EXPEN-SES. The treatment of section 36.11 (56) of the statutes and Section 9348 (3p) of this act take effect on the first day of the 4th month beginning after publication.

SECTION 9449. Effective dates; Veterans Affairs.

(1f) Grants to local governments providing ser-VICES TO VETERANS HOMES. The treatment of 2015 Wisconsin Act 55, sections 768kb, 768pb, 1458rb, and 9449 (1q) takes effect retroactively to June 30, 2017.

SECTION 9450. Effective dates; Wisconsin Economic Development Corporation.

SECTION 9451. Effective dates; Workforce Development.

(2q) Grant to support the Building Occupational SKILLS FOR SUCCESS PROGRAM. The treatment of section 20.445 (1) (b) (by Section 398b) of the statutes takes effect on July 1, 2018.

SECTION 9452. Effective dates; Other.

- (2f) DOMESTIC PARTNERSHIP. The treatment of sections 66.0137 (5) (b), 66.0510, 770.001, 770.05 (intro.), and 770.07 (1) (a) and (3) of the statutes and SECTION 9331 (3f) of this act take effect on the first day of the 7th month beginning after publication.
- (2w) Elimination of prevailing wage law. The Vetoed treatment of sections 16.856, 19.36 (3) and (12), 59.20 (3) (a), 84.062, 84.41 (3), 106.04, 109.09 (1), 111.322 (2m) (c) and (d), 230.13 (1) (intro.), 233.13 (intro.), 946.15, and 978.05 (6) (a) of the statutes and SECTION 9352 (3) of this act take effect on September 1, 2018.
- (2x) SUNSET FOR EVALUATION OF LEGISLATION. The repeal and recreation of section 961.443 (2) of the statutes takes effect on August 1, 2020.

In Part